### **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

### Form S-4 REGISTRATION STATEMENT

**UNDER** THE SECURITIES ACT OF 1933

# ProFrac Holding Corp. (Exact Name of Registrant as Specified in Its Charter)

Delaware	1389	87-2424964
(State or Other Jurisdiction of	(Primary Standard	(I.R.S. Employer
Incorporation or Organization)	Industrial Classification Code Number)	Identification No.)
	333 Shops Boulevard, Suite 301 Willow Park, Texas 76087 (254) 776-3722	
(Address, Including Zip Code,	and Telephone Number, Including Area Code, of Registrant's 1	Principal Executive Offices)
	I T	
	Lance Turner Chief Financial Officer 333 Shops Boulevard, Suite 301 Willow Park, Texas 76087 (254) 776-3722	
(Name, Address, Includ	ling Zip Code, and Telephone Number, Including Area Code, of	f Agent For Service)
	Copies to:	
Samuel P. Williams, Esq. James E. Bedar, Esq. Brown Rudnick LLP One Financial Center Boston, MA 02111 (617) 856-8200	Joshua Shapiro Chief Financial Officer U.S. Well Services, Inc. 1360 Post Oak Boulevard, Suite 1800 Houston, TX 77056 (832) 562-3730	Corey C. Brown, Esq. Adam K. Nalley, Esq. Kevin J. Poli, Esq. Porter Hedges LLP 1000 Main Street, 36th Floor Houston, TX 77002 (713) 226-6644
Approximate date of commencement of proposed the merger described in the enclosed proxy statement/inform	I sale to the public: As soon as practicable after the effective date on nation statement/prospectus.	of this Registration Statement and upon completion of
If the securities being registered on this Form are be check the following box: $\hfill\Box$	ing offered in connection with the formation of a holding company	and there is compliance with General Instruction G,
If this form is filed to register additional securities for registration statement number of the earlier effective registration.	or an offering pursuant to Rule 462(b) under the Securities Act, checation statement for the same offering. $\Box$	ck the following box and list the Securities Act
If this form is a post-effective amendment filed purs number of the earlier effective registration statement for the	cuant to Rule 462(d) under the Securities Act, check the following became offering. $\ \square$	ox and list the Securities Act registration statement
	rge accelerated filer, an accelerated filer, a non-accelerated filer, a succelerated filer," "smaller reporting company," and "emerging grov	
Large accelerated filer $\Box$		Accelerated filer
Non-accelerated filer ⊠		Smaller reporting company
		Emerging growth company
If an emerging growth company, indicate by check accounting standards provided pursuant to Section 7(a)(2)(1)	mark if the registrant has elected not to use the extended transition p B) of the Securities Act. $\Box$	eriod for complying with any new or revised financial
If applicable, place an X in the box to designate the	appropriate rule provision relied upon in conducting this transaction	1:
Exchange Act Rule 13e-4(i) (Cross-Border Issuer T	ender Offer)	
Exchange Act Rule 14d-l(d) (Cross-Border Third-Pa	arty Tender Offer)	
	nt on such date or dates as may be necessary to delay its effective on statement shall thereafter become effective in accordance with	

registration statement shall become effective on such date as the SEC, acting pursuant to said Section 8(a), may determine.

The information in this proxy statement/information statement/prospectus is not complete and may be changed. ProFrac Holding Corp. may not issue the securities offered by this proxy statement/information statement/prospectus until the registration statement containing this proxy statement/information statement/prospectus has been declared effective by the Securities and Exchange Commission. This proxy statement/information statement/prospectus does not constitute an offer to sell or a solicitation of an offer to buy these securities, nor shall there be any sale of these securities, in any jurisdiction in which such offer, solicitation or sale is not permitted or would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction.

### PROXY STATEMENT/INFORMATION STATEMENT/PROSPECTUS

SUBJECT TO COMPLETION, DATED AUGUST30, 2022



# NOTICE OF ACTION BY WRITTEN CONSENT AND INFORMATION STATEMENT WE ARE NOT ASKING YOU FOR A PROXY AND YOU ARE REQUESTED NOT TO SEND US A PROXY.

To the Stockholders of ProFrac Holding Corp.:

This notice of action by written consent and the accompanying proxy statement/information statement/prospectus are being furnished to holders of the Class A Common Stock, \$0.01 per share par value ("ProFrac Class A Common Stock"), and Class B Common Stock, \$0.01 per share par value ("ProFrac Class B Common Stock"), of ProFrac Holding Corp., a Delaware corporation ("ProFrac").

On June 21, 2022, ProFrac entered into an Agreement and Plan of Merger (the 'Merger Agreement') by and among ProFrac, U.S. Well Services, Inc., a Delaware corporation ("USWS"), and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of ProFrac ('Merger Sub''). The Merger Agreement provides for, among other things, the merger of Merger Sub with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of ProFrac (the "Merger"). A copy of the Merger Agreement is included as Annex A to the accompanying proxy statement/information statement/prospectus.

Pursuant to the Merger Agreement, subject to the terms and conditions set forth therein:

- At the effective time of the Merger (the 'Effective Time'), each share of Class A Common Stock of USWS, par value \$0.0001 per share, (the "USWS Common Stock") will be converted automatically into the right to receive 0.3366 (the 'Exchange Ratio') shares of ProFrac Class A Common Stock (the "Merger Consideration"), as adjusted from the agreed upon Merger Agreement Exchange Ratio of 0.0561 to account for the 1-for-6 reverse stock split implemented by USWS on August 4, 2022.
- USWS will take all requisite action so that, effective as of immediately prior to the Effective Time, all shares of Series A Redeemable Convertible Preferred Stock of USWS, par value \$0.0001 per share (the "USWS Series A Preferred Stock"), and all USWS Equity Linked Convertible Notes (as defined in the accompanying proxy statement/information statement/prospectus) issued and outstanding at such time will be converted into a certain number of shares USWS Common Stock. For more information about the conversion of USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes into USWS Common Stock, see the sections titled "The Merger Agreement Treatment of USWS Series A Preferred Stock and Treatment of USWS Equity Linked Convertible Notes."
- Subject to the consummation of the Warrant Sale (as defined in the accompanying proxy statement/information statement/prospectus), at the Effective Time, the USWS Term C Loan Warrants (as defined in the accompanying proxy statement/information statement/prospectus) issued and outstanding immediately prior to the Effective Time (which shall be held by ProFrac following the consummation of the Warrant Sale) will be automatically canceled and will cease to exist and no consideration will be delivered in exchange therefor. For more information about the Warrant Sale, see the section titled "The Merger Agreement Related Agreements."

• Prior to the Effective Time, ProFrac will execute an amendment to each of the warrant agreements that govern the USWS Rollover Warrants (as defined in the accompanying proxy statement/information statement/prospectus) as necessary to assume each USWS Rollover Warrant as of the Effective Time in accordance with the terms thereof and the Merger Agreement. Each USWS Rollover Warrant shall remain outstanding following the Merger through its prescribed term but shall represent the right to receive upon valid exercise thereof shares of ProFrac Class A Common Stock equal to the product of (A) the number of shares of USWS Common Stock subject to such USWS Rollover Warrant immediately prior to the Effective Time and (B) the Exchange Ratio. Additionally, each USWS Rollover Warrant will be amended such that the exercise price of such USWS Rollover Warrant will equal the current exercise price of such USWS Rollover Warrant divided by the Exchange Ratio. For more information about the USWS Rollover Warrants, see the section titled "The Merger Agreement — Treatment of USWS Warrants — Rollover Warrants."

Further, pursuant to the Merger Agreement, USWS agreed that it will take all requisite action so that:

- at the Effective Time, each share of USWS Common Stock subject to vesting, repurchase, or other lapse of restrictions (a 'USWS Restricted Share') that is outstanding and unvested under the USWS Amended and Restated 2018 Stock Incentive Plan immediately prior to the Effective Time will be canceled in exchange for the right to receive the Merger Consideration and, in lieu of any fractional shares, cash;
- immediately prior to the Effective Time, each then-outstanding deferred stock unit or restricted stock unit of USWS (each, a \*USWS DSU") will be canceled and converted into the right to receive the Merger Consideration and, in lieu of any fractional shares, cash; and
- immediately prior to the Effective Time, all outstanding USWS Pool A Performance Awards and USWS Pool B Performance Awards (each as defined in the accompanying proxy statement/information statement/prospectus) will be canceled and converted into the right to receive a certain amount of Merger Consideration and such amount will be dependent upon the holder and the type of the performance award. For more information about the exchange of the USWS Pool A Performance Awards and USWS Pool B Performance Awards for Merger Consideration, see the section titled "The Merger Agreement Treatment of USWS Equity Awards"

Based on the number of shares of USWS Common Stock (after giving effect to the conversion of the USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes), USWS Restricted Shares, USWS DSUs, USWS Pool A Performance Awards and USWS Pool B Performance Awards outstanding as of the date of the accompanying proxy statement/information statement/prospectus, an aggregate of approximately 12.8 million shares of ProFrac Class A Common Stock will be issued in exchange therefor which, based on the ProFrac Class A Common Stock June 21, 2022 closing price, would be approximately \$275 million. The issuance by ProFrac of the shares of ProFrac Class A Common Stock comprising the Merger Consideration is referred to herein as the "ProFrac Stock Issuance."

Immediately following the completion of the Merger, based on the aggregate amount of capital stock and shares of ProFrac Class A Common Stock outstanding as of August 8, 2022, it is anticipated that persons who were stockholders of ProFrac and USWS immediately prior to the Merger will own approximately 91.76% and 8.24%, respectively, of the then outstanding ProFrac capital stock and will own approximately 76.34% and 23.66%, respectively, of the then outstanding ProFrac Class A Common Stock.

Dan Wilks and Farris Wilks, together with certain of their affiliates, including THRC Holdings, LP, a Texas limited partnership (**THRC Holdings**"), and THRC Management, LLC, a Texas limited liability company (**THRC Management**" and, together with THRC Holdings and Dan Wilks and Farris Wilks, collectively, the "**Wilks Parties**"), collectively hold a controlling interest in ProFrac. Certain Wilks Parties also own certain securities of USWS as further described in the section titled "*The Merger — Interests of ProFrac Executive Officers and Directors in the Merger*." Upon the consummation of the Merger, the Wilks Parties will receive an aggregate of 4,138,596 shares of ProFrac Class A Common Stock as Merger Consideration (assuming, solely for the purpose of this calculation, a hypothetical closing date of October 31, 2022), which, based on the ProFrac Class A Common Stock June 21, 2022 closing price, would be approximately \$88.9 million.

The Board of Directors of ProFrac (the "ProFrac Board") formed a special committee of the ProFrac Board comprised solely of independent and disinterested directors of the ProFrac Board (the "ProFrac Special Committee") because of the actual or potential conflicts of interest of certain officers of ProFrac and members of the ProFrac Board as more particularly described in the section titled "The Merger — Interests of ProFrac Executive Officers and Directors in the Merger."

On June 20, 2022, the ProFrac Special Committee unanimously (i) determined that the Merger Agreement and the transactions contemplated thereby were fair to, and in the best interests of, ProFrac and the holders of ProFrac Class A Common Stock, excluding THRC Holdings and its affiliates (''ProFrac Unaffiliated Stockholders''), and (ii) approved and declared advisable the Merger Agreement and the transactions contemplated thereby. The ProFrac Special Committee further unanimously (i) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) determine that the Merger Agreement and the transactions contemplated thereby were fair to, and in the best interests of, ProFrac and the ProFrac Unaffiliated Stockholders and (y) approve and declare advisable the Merger Agreement and the transactions contemplated thereby, and (ii) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) direct that the ProFrac Stock Issuance be submitted to ProFrac's stockholders for their approval and (y) recommend that ProFrac's stockholders approve the ProFrac Stock Issuance.

On June 21, 2022, the ProFrac Board, acting on the unanimous recommendation and approval of the ProFrac Special Committee, unanimously (i) approved and declared advisable the Merger Agreement and the transactions contemplated thereby, (ii) approved and declared advisable the execution and delivery by ProFrac of the Merger Agreement, the performance by ProFrac of its covenants and agreements contained therein and the consummation of the transactions contemplated by the Merger Agreement, (iii) determined that the Merger Agreement and the transactions contemplated thereby are fair to, and in the best interests of, ProFrac and the ProFrac Unaffiliated Stockholders, and (iv) resolved to recommend that ProFrac's stockholders approve the ProFrac Stock Issuance.

On June 21, 2022, and in accordance with and concurrently with the execution and delivery of the Merger Agreement, THRC Holdings and Farris Wilks (collectively, the "ProFrac Controlling Stockholders") approved the ProFrac Stock Issuance by delivery of a written consent (the "ProFrac Stockholder Written Consent"). As of June 21, 2022, the ProFrac Controlling Stockholders held the majority of the voting power represented by the outstanding ProFrac Class A Common Stock. Therefore, the delivery of the ProFrac Stockholder Written Consent satisfied the requirement in the Merger Agreement and under the Nasdaq rules that the affirmative vote of the holders of at least a majority of the voting power of the ProFrac Class A Common Stock approve the ProFrac Stock Issuance. No additional ProFrac stockholder approval is required for the Merger, the Merger Agreement and the other transactions contemplated thereby, including the ProFrac Stock Issuance. As a result, ProFrac has not solicited and will not be soliciting your vote for the approval of the ProFrac Stock Issuance, the Merger and the Merger Agreement and the other transactions contemplated thereby and does not intend to call a meeting of stockholders for purposes of voting on the adoption of the ProFrac Stock Issuance, the Merger Agreement and the other transactions contemplated thereby.

The completion of the Merger is subject to approval by the Stockholders of USWS at a Special Meeting thereof to be held on [\_\_], 2022. Concurrently with the execution and delivery of the Merger Agreement, certain Stockholders of USWS, including THRC Holdings, (each, a "USWS Supporting Stockholder" and, collectively, the "USWS Supporting Stockholders") entered into a Voting Agreement with ProFrac (the "Voting Agreement"). Pursuant to the Voting Agreement, each USWS Supporting Stockholder agreed to, among other things, (i) support and vote in favor of the Merger Agreement, the Merger and the other transactions contemplated by the Merger Agreement, (ii) take (and refrain from taking) certain other actions in connection with the transactions contemplated by the Merger Agreement, and (iii) to the extent applicable, deliver a Conversion Notice (as defined in the Merger Agreement) to convert each USWS Supporting Stockholder's shares of USWS Series A Preferred Stock at the Merger Conversion Ratio (as defined in the Merger Agreement). The USWS Supporting Stockholders together beneficially own, in the aggregate, approximately 44% of the currently outstanding shares of USWS Common Stock, which includes 9,568 shares of USWS Common Stock, or approximately 0.07%, held by THRC Holdings.

This notice of action by written consent and the proxy statement/information statement/prospectus shall constitute notice to you from ProFrac that the ProFrac Stock Issuance has been approved by the holders of a majority of the voting power of the ProFrac Class A Common Stock pursuant to the ProFrac Stockholder Written Consent in lieu of a meeting in accordance with the Merger Agreement and Section 228 of the General Corporation Law of the State of Delaware.

The proxy statement/information statement/prospectus accompanying this letter provides you with more specific information concerning the ProFrac Stock Issuance, the Merger and the Merger Agreement and the transactions contemplated thereby. We encourage you to carefully read the proxy statement/information statement/prospectus and the copy of the Merger Agreement.

Sincerely,

Matthew D. Wilks *Executive Chairman and Director* [ •], 2022

Neither the Securities and Exchange Commission nor any state securities regulatory agency has approved or disapproved of the transactions described in the accompanying proxy statement/information statement/prospectus, passed upon the merits or fairness of the Merger or related transactions or passed upon the adequacy or accuracy of the disclosure in the accompanying proxy statement/information statement/prospectus. Any presentation to the contrary constitutes a criminal offense.

The Merger described herein involves certain risks and uncertainties. See the section titled "Risk Factors" beginning on page 26.



## NOTICE OF SPECIAL MEETING OF STOCKHOLDERS To Be Held On [ ], 2022

Dear Stockholders of U.S. Well Services, Inc.:

NOTICE IS HEREBY GIVEN that a special meeting of stockholders of U.S. Well Services, Inc. ('USWS'') will be held on [ ], 2022 at [\_\_\_\_] a.m., local time, solely by means of remote communication in an audio-only virtual meeting format, at www.proxydocs.com/USWS (the "USWS Special Meeting").

The purpose of the USWS Special Meeting is to consider and to vote on the following proposals:

- to adopt the Agreement and Plan of Merger, dated as of June 21, 2022 (the 'Merger Agreement'), among USWS, ProFrac Holding Corp.
   ("ProFrac") and Thunderclap Merger Sub I, Inc., an indirect subsidiary of ProFrac ('Merger Sub"), a copy of which is attached as <u>Annex A</u>
   to the accompanying proxy statement/information statement/prospectus, and the transactions contemplated thereby, including the merger,
   pursuant to which Merger Sub will merge with and into USWS, with USWS surviving as an indirect subsidiary of ProFrac (the "Merger");
- 2. to approve, for purposes of complying with the Nasdaq listing rules, the issuance of shares of USWS Class A common stock, par value \$0.0001 per share (the "USWS Common Stock"), to be issued by USWS upon the conversion of the USWS Series A Redeemable Convertible Preferred Stock (the "USWS Series A Preferred Stock") and the Convertible Senior Secured (Third Lien) PIK Notes (the "Equity Linked Convertible Notes"), in an amount equal to 20% or more of USWS Common Stock outstanding;
- to approve an amendment to the USWS Certificate of Designations of Series A Preferred Stock to modify certain terms relating to the conversion rights of USWS Series A Preferred Stock;
- 4. to approve an amendment to our Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan (the 'A&R LTIP') to increase the maximum number of shares of USWS Common Stock that may be issued under the A&R LTIP by 2,000,000 shares; and
- 5. if submitted to a vote of USWS Stockholders, to approve an adjournment of this USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of the foregoing proposals if there are not sufficient votes for those proposals.

The board of directors of USWS (the "USWS Board") formed a special committee comprised solely of independent and disinterested directors of the USWS Board (the "USWS Special Committee") to review, negotiate and approve the Merger, the Merger Agreement, and the other transactions contemplated thereby because of the actual or potential conflict of interest of certain members of the USWS Board, as more particularly described in the section titled "The Merger — Interests of USWS Executive Officers and Directors in the Merger." The USWS Special Committee considered these actual or potential conflicts of interest, among other matters, in approving the Merger and in recommending to the USWS Board the adoption of the Merger Agreement and the transactions contemplated thereby.

Acting on the unanimous recommendation and approval of the USWS Special Committee, the USWS Board has unanimously determined that the Merger, the Merger Agreement, and the other transactions contemplated thereby are advisable and in the best interests of USWS and its stockholders. The USWS Board recommends that USWS stockholders vote "FOR" the adoption of the Merger Agreement and the transactions contemplated thereby, including the Merger, "FOR" the issuance of shares of USWS Common Stock to be issued by USWS

upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, "FOR" the approval of the amendment to the USWS Certificate of Designations of Series A Preferred Stock, "FOR" the approval of the amendment to the A&R LTIP, and "FOR" the adjournment of the USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of any of the proposals to be acted upon at the USWS Special Meeting.

USWS and ProFrac cannot complete the Merger unless:

- the proposal to adopt the Merger Agreement is approved by the affirmative vote of a majority of the shares of USWS Common Stock outstanding on the record date of the USWS Special Meeting;
- the proposal to approve the issuance of shares of USWS Common Stock to be issued by USWS upon conversion of the USWS Series A
  Preferred Stock and the USWS Equity Linked Convertible Notes is approved by the affirmative vote of the holders of a majority of the total
  votes cast at the USWS Special Meeting; and
- the proposal to approve the amendment to the USWS Certificate of Designations of Series A Preferred Stock is approved by a majority of the shares of USWS Common Stock outstanding on the record date of the USWS Special Meeting.

Your failure to vote will have the same effect as a vote against the adoption of the Merger Agreement, the issuance of the USWS Common

Stock, and the approval of the amendment to the Certificate of Designations of Series A Preferred Stock.				
The close of business on [	], 2022 has been fixed as the record date, which is referred	to as the USWS record date, for the determination		
of USWS stockholders who are entitled	d to notice of, and to vote at, the USWS Special Meeting or any a	djournments or postponements of the USWS		
Special Meeting. Only holders of USW	VS Common Stock of record at the close of business on [	], 2022 are entitled to notice of, and to vote at,		
the USWS Special Meeting or any adjo-	ournments or postponements of the USWS Special Meeting. A co	mplete list of the holders of USWS Common Stock		
entitled to vote at the USWS Special M	Meeting will be available for examination by any USWS stockhold	ler, for any purpose germane to the USWS Special		

Meeting, at the USWS principal executive offices at 1360 Post Oak Blvd., Suite 1800, Houston, TX 77056, for a period of ten days before the USWS Special Meeting, between the hours of 9:00 a.m. and 4:00 p.m., local time, and at the USWS Special Meeting during the entire time of the meeting.

USWS directs your attention to the accompanying proxy statement/information statement/prospectus accompanying this notice for a more complete statement regarding the matters proposed to be acted upon at the USWS Special Meeting. USWS encourages you to read the entire proxy statement/information statement/prospectus carefully, including the Merger Agreement, which is attached as Annex A to the proxy statement/information statement/prospectus, and the section entitled "Risk Factors" beginning on page 26.

YOUR VOTE IS VERY IMPORTANT. WHETHER OR NOT YOU PLAN TO ATTEND THE USWS SPECIAL MEETING, PLEASE VOTE AS SOON AS POSSIBLE BY MAIL, BY TELEPHONE OR THROUGH THE INTERNET. INSTRUCTIONS ON THESE DIFFERENT WAYS TO VOTE YOUR PROXY ARE FOUND ON THE ENCLOSED PROXY CARD. YOU MAY REVOKE YOUR PROXY AT ANY TIME BEFORE IT IS VOTED AT THE USWS SPECIAL MEETING BY FOLLOWING THE PROCEDURES SET FORTH IN THE ACCOMPANYING PROXY STATEMENT/INFORMATION STATEMENT/PROSPECTUS. REMEMBER, YOUR VOTE IS IMPORTANT, SO PLEASE VOTE PROMPTLY!

By Order of the Board of Directors,	
T 1D 1	
Joel Broussard Chairman of the Board	

### ADDITIONAL INFORMATION

### **ProFrac**

ProFrac is subject to the informational filing requirements of the Securities Exchange Act of 1934, as amended (the 'Exchange Act'), and, in accordance therewith, is obligated to file reports, proxy statements and other information with the Securities and Exchange Commission (the "SEC") relating to its business, financial condition and other matters. At the present time, ProFrac is not eligible to incorporate by reference the information filed by ProFrac with the SEC into this proxy statement/information statement/prospectus and accordingly has included the required business and financial information of ProFrac in this proxy statement/information statement/prospectus. Documents filed with the SEC by ProFrac are available free of charge on the investor relations portion of the ProFrac website at www.pfholdingscorp.com. You will also be able to obtain free copies of these documents and other documents containing important information about ProFrac, once such documents are filed with the SEC, through the website maintained by the SEC at www.sec.gov.

You may also request copies of this proxy statement/information statement/prospectus or other information about ProFrac, without charge, upon written or oral request to ProFrac's principal executive offices at the following address and telephone number:

Attention: Investor Relations ProFrac Holding Corp. 333 Shops Boulevard, Suite 301 Willow Park, TX 76087 (254) 776-3722

### USWS

This proxy statement/information statement/prospectus incorporates by reference important business and financial information about USWS from other documents filed by USWS with the SEC, that are not included in or delivered with this proxy statement/information statement/prospectus. This information is available to you without charge upon your written or oral request. For a list of the USWS documents incorporated by reference into this proxy statement/information statement/prospectus, see "Where You Can Find More Information" beginning on page 276. You can obtain the documents that are incorporated by reference into this proxy statement/information statement/prospectus, without charge, from the Investor Relations section of USWS' website at www.uswellservices.com or by requesting them in writing or by telephone as set forth below:

U.S. Well Services, Inc. 1360 Post Oak Boulevard Suite 1800 Houston, TX 77056 (832) 562-3730

If you would like to request any documents, for a timely delivery of the documents, please do so by  $[\bullet]$ , 2022, which is five (5) business days prior to the date of the USWS Special Meeting.

The contents of the websites of the SEC, ProFrac, USWS or any other entity are not incorporated in the accompanying proxy statement/information statement/prospectus. The information about how you can obtain certain information and/or documents that are incorporated by reference in the accompanying proxy statement/information statement/prospectus at these websites is being provided only for your convenience.

### ABOUT THIS PROXY STATEMENT/INFORMATION STATEMENT/PROSPECTUS

This proxy statement/information statement/prospectus, which forms a part of the registration statement on FormS-4 filed with the SEC by ProFrac, constitutes a prospectus of ProFrac under the Securities Act of 1933, as amended (the "Securities Act"), with respect to the shares of ProFrac Class A Common Stock issuable to the holders of USWS Common Stock and USWS Equity Awards (as defined in this proxy statement/information statement/prospectus) in connection with the Merger. This proxy statement/information statement/prospectus also registers (i) the publicly traded USWS SPAC Warrants and (ii) the shares of ProFrac Class A Common Stock that will be issuable upon exercise of the currently outstanding USWS SPAC Warrants, which warrants shall remain outstanding after the consummation of the Merger but represent the right to purchase, upon exercise thereof, shares of ProFrac Class A Common Stock (rather than shares of USWS Common Stock) based on the Exchange Ratio and correspondingly adjusted exercise price. This proxy statement/information statement/prospectus also constitutes an information statement for ProFrac and a proxy statement for USWS under the Exchange Act. This proxy statement/information statement/prospectus also constitutes a notice of action taken by the ProFrac Controlling Stockholders with respect to the approval of the ProFrac Stock Issuance and a notice of meeting with respect to the USWS Special Meeting.

You should rely only on the information contained in or incorporated by reference into this proxy statement/information statement/prospectus. No one has been authorized to provide you with information that is different from that contained in, or incorporated by reference into, this proxy statement/information statement/prospectus is dated as of the date set forth on the cover hereof. You should assume that the information contained in this proxy statement/information statement/prospectus is accurate only as of such date. You should also assume that the information incorporated by reference into this proxy statement/information statement/prospectus is only accurate as of the date of such incorporated document. Neither the mailing of this proxy statement/information statement/prospectus to ProFrac stockholders or USWS stockholders nor the issuance by ProFrac of shares of ProFrac Class A Common Stock pursuant to the Merger Agreement will create any implication to the contrary.

The sections titled "Questions and Answers About the Merger" and "Summary" below highlight select information from this proxy statement/information statement/prospectus, but they do not include all of the information that may be important to you. To better understand the Merger and the Merger Agreement, and for a more complete description of the legal terms thereof, you should carefully read this entire proxy statement/information statement/prospectus, including the section entitled "Risk Factors" and the Merger Agreement, a copy of which is attached as Annex A hereto, as well as the documents that are incorporated by reference into this proxy statement/information statement/prospectus. See "Where You Can Find More Information."

This proxy statement/information statement/prospectus does not constitute an offer to sell, or a solicitation of an offer to buy, any securities, or the solicitation of a proxy in any jurisdiction to or from any person to whom it is unlawful to make any such offer or solicitation in such jurisdiction. Information contained in this proxy statement/information statement/prospectus regarding ProFrac and its business, operations, management and other matters has been provided by ProFrac, and information contained in this proxy statement/information statement/prospectus regarding USWS and its business, operations, management and other matters has been provided by USWS.

### TABLE OF CONTENTS

EDECHIENTI VIIGED TERMO	Page :
FREQUENTLY USED TERMS	1
QUESTIONS AND ANSWERS ABOUT THE MERGER	vii
SUMMARY	1
The Companies	1
The Merger	2
Selected Historical and Pro Forma Consolidated Financial Data of ProFrac	13
Selected Historical Consolidated Financial Data of USWS	18
Unaudited Comparative Per Share Data	21
COMPARATIVE PER SHARE MARKET PRICE AND DIVIDEND INFORMATION	24
RISK FACTORS	26
Risks Related to the Merger	26
Risks Related to the Combined Company Following the Merger	30
Risks Related to ProFrac's Business	32
Risks Related to ProFrac's Class A Common Stock	54
Risks Related to USWS' Business and Industry	66
CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS	79
THE MERGER	81
Structure of the Merger	81
Merger Consideration	81
Background of the Merger	81
ProFrac Board and ProFrac Special Committee Reasons for the Merger	93
Opinion of the Financial Advisor to the ProFrac Special Committee	98
Interests of ProFrac Executive Officers and Directors in the Merger	105
USWS Board and USWS Special Committee Reasons for the Merger	108
Recommendations of the USWS Board and the USWS Special Committee	112
Opinions of the Financial Advisor to the USWS Special Committee	115
Interests of USWS Executive Officers and Directors in the Merger	122
Board of Directors and Management of USWS Following Completion of the Merger	125
Board of Directors and Management of ProFrac Following the Completion of the Merger	125
Accounting Treatment	125
Federal Securities Laws Consequences; Stock Transfer Restriction Agreements	125
Appraisal Rights	126

	1 age
MATERIAL UNITED STATES FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER  REGULATORY AND OTHER APPROVALS REQUIRED FOR THE MERGER  THE COMPANIES  THE MERGER AGREEMENT  Explanatory Note Regarding Summary of Merger Agreement and Representations and Warranties in the Merger Agreement  Structure of the Merger  Merger Consideration  Exchange of Shares  Fractional Shares  Treatment of USWS Series A Preferred Stock  Treatment of USWS Equity Linked Convertible Notes  Treatment of USWS Warrants  Treatment of USWS Equity Awards  Representations and Warranties  Covenants  Conditions to Completion of the Merger  Termination of the Merger Agreement  Termination Fees  Other Expenses  Amendments; Waivers  Related Agreements  THE USWS SPECIAL MEETING  Date, Time and Place  Purpose of the USWS Special Meeting	127
REGULATORY AND OTHER APPROVALS REQUIRED FOR THE MERGER	131
THE COMPANIES	133
THE MERGER AGREEMENT	134
Explanatory Note Regarding Summary of Merger Agreement and Representations and Warranties in the Merger Agreement	134
Structure of the Merger	134
Merger Consideration	134
Exchange of Shares	135
<u>Fractional Shares</u>	135
Treatment of USWS Series A Preferred Stock	135
Treatment of USWS Equity Linked Convertible Notes	135
Treatment of USWS Warrants	136
Treatment of USWS Equity Awards	136
Representations and Warranties	137
<u>Covenants</u>	138
Conditions to Completion of the Merger	144
Termination of the Merger Agreement	145
<u>Termination Fees</u>	147
Other Expenses	147
Amendments; Waivers	147
Related Agreements	148
THE USWS SPECIAL MEETING	150
Date, Time and Place	150
Purpose of the USWS Special Meeting	150
Board Recommendations	150
USWS Record Date; Shares Entitled to Vote	150
Stock Ownership of Directors and Executive Officers	151
Quorum Requirement	151
Votes Required to Approve USWS Proposals	151
<u>Voting of Proxies</u>	152
Revocation of Proxies	153
Solicitation of Proxies	153
Householding	153
MERGER PROPOSAL (PROPOSAL 1)	154

	Page
ISSUANCE OF ADDITIONAL SHARES PROPOSAL (PROPOSAL 2)	155
SERIES A CERTIFICATE OF DESIGNATIONS AMENDMENT PROPOSAL (PROPOSAL 3)	157
AMENDMENT TO A&R LTIP (PROPOSAL 4)	158
USWS ADJOURNMENT PROPOSAL (PROPOSAL 5)	160
INFORMATION ABOUT PROFRAC	161
PROFRAC'S MANAGEMENT AND BOARD OF DIRECTORS	185
MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF PROFRAC	211
INFORMATION ABOUT USWS	235
MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF USWS	246
DESCRIPTION OF PROFRAC CAPITAL STOCK	254
COMPARISON OF STOCKHOLDER RIGHTS AND CORPORATE GOVERNANCE MATTERS	264
<u>LEGAL MATTERS</u>	273
<u>EXPERTS</u>	273
TRANSFER AGENT AND REGISTRAR	273
FUTURE STOCKHOLDER PROPOSALS	274
HOUSEHOLDING OF PROXY MATERIALS	275
WHERE YOU CAN FIND MORE INFORMATION	276
INDEX TO FINANCIAL STATEMENTS	F-i
LIST OF ANNEXES	
ANNEX A AGREEMENT AND PLAN OF MERGER	
ANNEX B PROFRAC STOCKHOLDER WRITTEN CONSENT	
ANNEX C OPINION OF JEFFERIES LLC	
ANNEX D OPINION OF PIPER SANDLER & CO.	

### FREQUENTLY USED TERMS

- "1933 Act" or "Securities Act" means the Securities Act of 1933, as amended.
- "1934 Act" or "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Alpine" means Alpine Silica, LLC, a Texas limited liability company and indirect wholly-owned subsidiary of ProFrac.
- "Award Amendment" means the amendment to the USWS Pool A Performance Awards and USWS Pool B Performance Awards, substantially in the form of <a href="Exhibit C">Exhibit C</a> to the Merger Agreement.
- "Best Flow" means Best Pump & Flow LP, a Texas limited partnership and indirect wholly-owned subsidiary of ProFrac.
- "Bylaws" means, with respect to ProFrac, the Amended and Restated Bylaws of ProFrac, dated as of May 17, 2022, and, with respect to USWS, the Amended and Restated Bylaws of USWS, dated as of February 15, 2017, in each case as amended.
- "Certificate of Designations" means that certain Certificate of Designations of the USWS Series A Preferred Stock, dated May 24, 2019, as it may be amended, restated or otherwise modified from time to time following the date of this proxy statement/information statement/prospectus, including pursuant to the COD Amendment.
- "Certificate of incorporation" means, with respect to ProFrac, the Amended and Restated Certificate of Incorporation of ProFrac, dated as of May 17, 2022, and, with respect to USWS, the Second Amended and Restated Certificate of Incorporation of USWS, dated as of November 9, 2018, in each case as amended.
- "Closing" means the closing of the Merger.
- "Closing Date" means the date of the Closing.
- "Code" means the Internal Revenue Code of 1986, as amended.
- "COD Amendment" means the amendment to the Certificate of Designations, substantially in the form of Exhibit B to the Merger Agreement.
- "Crestview Partners" means certain funds or other investment vehicles managed by or otherwise affiliated with Crestview Advisors, L.L.C.
- "Delaware Secretary of State" means the Secretary of State of the State of Delaware.
- "DGCL" means the General Corporation Law of the State of Delaware, as amended from time to time.
- "E&P" means exploration and production.
- "Effective Time" means the time in which the Merger becomes effective.
- "ESG" means environmental, social and governance.
- "Exchange Ratio" means the ratio of 0.3366 shares of ProFrac Class A Common Stock per issued and outstanding share of USWS Common Stock that will be issued to holders of shares of USWS Common Stock in connection with the Merger.

- "February 2022 Warrant Agreement" means certain Warrant Agreement, dated as of February 28, 2022, by and between USWS and Continental Stock Transfer & Trust Company.
- "FTSI" means FTS International, Inc., a Delaware corporation and wholly owned subsidiary of ProFrac LLC.
- "FTSI Acquisition" means the March 4, 2022 acquisition by ProFrac LLC of all of the outstanding stock of FTSI, pursuant to the FTSI Merger Agreement, for an aggregate purchase price of approximately \$407.5 million.
- "FTSI Merger Agreement" means the Agreement and Plan of Merger, dated as of October 21, 2021, by and among FTSI, ProFrac LLC and ProFrac Acquisitions, Inc.
- "GAAP" means generally accepted accounting principles in the United States.
- "HHP" means hydraulic horsepower.
- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.
- "Jefferies" means Jefferies LLC, independent financial advisor to the ProFrac Special Committee.
- "March 2022 Warrant Agreement" means certain Warrant Agreement, dated as of March 1, 2022, by and between USWS and Continental Stock Transfer & Trust Company.
- "Merger" means the merger of Merger Sub with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of ProFrac.
- "Merger Agreement" means the Agreement and Plan of Merger, dated as of June 21, 2022, as it may further be amended or supplemented from time to time, by and among ProFrac, Merger Sub, and USWS. A copy of the Merger Agreement is attached to this proxy statement/information statement/prospectus as <u>Annex A</u>.
- "Merger Consideration" means the shares of ProFrac Class A Common Stock issuable to holders of USWS Common Stock and USWS Equity Awards in connection with the Merger pursuant to the Merger Agreement.
- "Merger Sub" means Thunderclap Merger Sub I, Inc., a Delaware corporation, and an indirect subsidiary of ProFrac.
- "Nasdaq" means Nasdaq Global Select Market or Nasdaq Capital Market, as applicable.
- "OPEC+" means members of the Organization of Petroleum Exporting Countries, Russia and other oil-producing countries, collectively.
- "Piper Sandler" means Piper Sandler & Co., independent financial advisor to the USWS Special Committee.
- "ProFrac" means ProFrac Holding Corp., a Delaware corporation.
- "ProFrac Board" means the board of directors of ProFrac.
- "ProFrac Class A Common Stock" means shares of Class A common stock, par value \$0.01 per share, of ProFrac.

- "ProFrac Class B Common Stock" means shares of Class B common stock, par value \$0.01 per share, of ProFrac.
- "ProFrac Common Stock" means the ProFrac Class A Common Stock and ProFrac Class B Common Stock, collectively.
- "ProFrac Controlling Stockholders" means Dan Wilks and Farris Wilks and their affiliates, including THRC Holdings, which collectively own the majority of the voting power of the ProFrac Class A Common Stock and ProFrac Class B Common Stock.
- "ProFrac IPO" means the initial public offering ProFrac Class A Common Stock consummated on May 17, 2022.
- "ProFrac LLC" means ProFrac Holdings, LLC, a Delaware limited liability company and subsidiary of ProFrac.
- "ProFrac II LLC" means ProFrac Holdings II, LLC, a Delaware limited liability company and indirect wholly-owned subsidiary of ProFrac.
- "ProFrac LLC Unit Holders" means the holders of common units of ProFrac LLC.
- "ProFrac Predecessor" means ProFrac LLC and its subsidiaries on a consolidated basis prior to the ProFrac IPO.
- "ProFrac Prospectus" means that certain prospectus, dated May 12, 2022, filed with the SEC by ProFrac pursuant to Rule 424(b)(4) under the 1933 Act on May 16, 2022, in connection with the ProFrac IPO.
- "ProFrac Special Committee" refers to the special committee of the ProFrac Board comprised solely of independent and disinterested directors of ProFrac.
- "ProFrac Stockholders" refers to holders of capital stock of ProFrac as of the time immediately before the Effective Time.
- "ProFrac Stockholders' Agreement' means the Stockholders' Agreement, dated as of May 17, 2022, by and among ProFrac, THRC Holdings, Farris C. Wilks, FARJO Holdings, LP and the Farris and Jo Ann Wilks 2022 Family Trust.
- "ProFrac Stock Issuance" means the issuance of shares of ProFrac Class A Common Stock to the holders of USWS Common Stock and USWS Equity Awards in connection with the Merger.
- "ProFrac Stockholder Approval" means the affirmative vote of the holders of a majority of the shares of ProFrac Class A Common Stock entitled to vote with respect to the approval of the ProFrac Stock Issuance.
- "ProFrac Trading Price" means an amount equal to the average of the volume weighted average price per share of ProFrac Class A Common Stock on Nasdaq (as reported by Bloomberg L.P. or, if not reported therein, in another authoritative source mutually selected by ProFrac and USWS) on each of the ten (10) consecutive trading days ending with the complete trading day immediately prior to the Effective Time.
- "ProFrac Unaffiliated Stockholders" means the holders of ProFrac Class A Common Stock, excluding the ProFrac Controlling Stockholders.
- "ProFrac Stockholder Written Consent" means the written consent executed on June 21, 2022 by the ProFrac Controlling Stockholders approving the ProFrac Stock Issuance.
- "Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002, as amended.

- "SEC" means the Securities and Exchange Commission.
- "SPS Acquisition" means the acquisition by ProFrac II LLC of the SP Companies pursuant to the SP Purchase Agreement.
- "SP Companies" means SP Monahans and SP Sales, collectively.
- "SP Monahans" means SP Silica of Monahans, LLC, a Delaware limited liability company.
- "SP Purchase Agreement' means the Membership Interest Purchase Agreement, dated as of June 16, 2022, by and among ProFrac II LLC, FoxRock Ranch Holding Company, LLC and the SP Companies, pursuant to which ProFrac II LLC has agreed to purchase 100% of the issued and outstanding membership interests of each of the SP Companies for an aggregate purchase price of approximately \$90 million in cash.
- "SP Sales" means SP Silica Sales, LLC, a Delaware limited liability company.
- "Tax Receivable Agreement" means the Tax Receivable Agreement, dated as of May 17, 2022, by and among ProFrac, the TRA Holders and the agents named therein.
- "Term C Loan Side Letter" means the letter agreement, dated as of February 28, 2022, as amended, by and among the Term C Lenders, the Term C Loan Warrant Holders, USWS, USWS Holdings, USWS LLC and the other subsidiaries of USWS party thereto, entered into in connection with the establishment of a new last-out senior secured term loan credit facility under the Senior Secured Term Loan Agreement.
- "THRC Holdings" means THRC Holdings, LP, a Texas limited partnership.
- "TRA Holders" means those certain ProFrac LLC Unit Holders that entered into the Tax Receivable Agreement.
- "Treasury Regulations" means the regulations promulgated under the Code.
- "Trigger Date" means the first date on which the ProFrac Controlling Stockholders no longer individually or collectively beneficially own (or otherwise have the right to vote or direct the vote of) more than 50% of the outstanding shares of ProFrac Common Stock.
- "USWS" means U.S. Well Services, Inc., a Delaware corporation.
- "USWS Board" means the board of directors of USWS.
- "USWS Borrower" means U.S. Well Services, LLC, a Delaware limited liability company and subsidiary of USWS in its capacity as borrower under the Senior Secured Term Loan Agreement.
- "USWS Common Stock" means the Class A common stock, par value \$0.0001 per share, of USWS.
- "USWS DSU" means the deferred stock unit or restricted stock unit of USWS, in each case representing a right to receive one share of USWS Common Stock granted under the USWS LTIP.
- "USWS Equity Awards" means, collectively, the USWS DSUs, USWS Restricted Shares, the USWS Pool A Performance Awards and USWS Pool B Performance Awards.
- "USWS Equity Linked Convertible Notes" means those outstanding convertible promissory notes convertible into USWS Common Stock and issued in accordance with that certain Note Purchase Agreement dated as of June 24, 2021, as amended, by and among USWS, ProFrac Predecessor, THRC Holdings, Crestview III USWS TE, LLC, Crestview III USWS TE, LLC, and Wilmington Savings Fund Society, FSB, as collateral agent for the

purchaser parties thereto, as each of such promissory notes may be amended, restated or otherwise modified from time to time following the date hereof.

"USWS February Term C Loan Warrants" means those outstanding warrants of USWS issued in accordance with that certain February 2022 Warrant Agreement, as each, some or all may be amended, restated or otherwise modified from time to time.

"USWS LTIP" or "A&R LTIP" means the Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan. USWS LTIP and A&R LTIP are used interchangeably throughout this document.

"USWS March Term C Loan Warrants" means those outstanding warrants of USWS issued in accordance with that certain March 2022 Warrant Agreement, as each, some or all may be amended, restated or otherwise modified from time to time.

"USWS Merger Proposal" means the USWS proposal to approve the adoption of the Merger Agreement and the transactions contemplated thereby, including the Merger.

"USWS Pool A Performance Awards" means the Performance Awards (Pool A) as granted under the USWS LTIP, as evidenced by that certain U.S. Well Services, Inc. Terms and Conditions Performance Award (Pool A).

"USWS Pool B Performance Awards" means the Performance Awards (Pool B) as granted under the USWS LTIP, as evidenced by that certain U.S. Well Services, Inc. Terms and Conditions Performance Award (Pool B).

"USWS Private Placement Agent Warrants" means those outstanding placement agent common stock warrants of USWS issued on March 11, 2022 to H.C. Wainwright & Co., LLC as compensation for its service as placement agent to the Company for the private placement of such warrants, as each may be amended, restated or otherwise modified from time to time.

"USWS Private Placement Investor Warrants" means those outstanding common stock warrants of USWS issued to investors in a private placement on March 11, 2022, as each may be amended, restated or otherwise modified from time to time.

"USWS Private Placement Warrants" means, collectively, the USWS Private Placement Agent Warrants and the USWS Private Placement Investor Warrants.

"USWS RDO Warrants" means those outstanding warrants of USWS issued in a registered direct offering on March 11, 2022.

"USWS Restricted Shares" means those shares of USWS Common Stock subject to vesting, repurchase or other lapse of restrictions granted under the USWS LTIP.

"USWS Rollover Warrants" means, collectively, the USWS SPAC Warrants, USWS Series A Warrants, USWS Private Placement Warrants and USWS RDO Warrants.

"USWS Series A Preferred Stock" means the Series A Redeemable Convertible Preferred Stock, par value \$0.0001 per share, of USWS.

"USWS Series A Warrants" means those outstanding warrants of USWS issued in accordance with that certain Warrant Agreement, dated as of May 24, 2019, by and between USWS and Continental Stock Transfer & Trust Company, as each may be amended, restated or otherwise modified from time to time.

"USWS SPAC Warrants" means those outstanding warrants of USWS issued in accordance with that certain Warrant Agreement, dated as of March 9, 2017, by and between USWS and Continental Stock Transfer & Trust Company, as each may be amended, restated or otherwise modified from time to time.

"USWS Special Committee" refers to the special committee of the USWS Board consisting solely of independent and disinterested directors of USWS, which the USWS Board established for the purpose of reviewing, evaluating and deliberating with respect to, and if deemed appropriate by the USWS Special Committee, negotiating the Merger Agreement and the transactions contemplated thereby, including the Merger.

"USWS Special Meeting" means the special meeting of the USWS Stockholders, to be held by means of remote communication in an audio-only virtual meeting format, at www.proxydocs.com/USWS at [a.m.][p.m] Central Time on [ ], 2022, and any adjournments thereof, to consider the proposals described in this proxy statement/information statement/prospectus.

"USWS Stockholders" refers to holders of capital stock of USWS as of the time immediately before the Effective Time.

"USWS Stockholder Approval" means the vote of the USWS Stockholders required to approve the Merger Agreement and all the transactions contemplated thereby, including the Merger, in accordance with USWS organizational documents.

"USWS Term C Loan Warrants" means, collectively, the USWS March Term C Loan Warrants and the USWS February Term C Loan Warrants.

"USWS Warrants" means, collectively, the USWS SPAC Warrants, the USWS Series A Warrants, the USWS Term C Loan Warrants, the USWS Private Placement Warrants and the USWS RDO Warrants.

"Warrant Purchase Agreement' means the Warrant Purchase Agreement, dated as of June 21, 2022, by and among ProFrac and the holders of USWS Term C Loan Warrants.

"Warrant Sale" means the purchase by ProFrac of the USWS Term C Loan Warrants pursuant to the Warrant Purchase Agreement for an aggregate purchase price of \$2,639,999.82.

### QUESTIONS AND ANSWERS ABOUT THE MERGER

The following are some questions that you may have regarding the Merger, the issuance of the shares of ProFrac Class A Common Stock to the holders of USWS Common Stock and USWS Equity Awards in connection with the Merger, and other matters being considered at the USWS Special Meeting and the answers to those questions. ProFrac and USWS urge you to carefully read the entirety of this proxy statement/information statement/prospectus, including the annexes hereto and the documents referred to herein, because the information in this section does not provide all the information that might be important to you with respect to the Merger, the issuance of shares of ProFrac Class A Common Stock in connection with the Merger, and the other matters being considered at the USWS Special Meeting.

### Questions and Answers Regarding the Merger

### Q: Why am I receiving this proxy statement/information statement/prospectus?

A: You are receiving this proxy statement/information statement/prospectus because ProFrac and USWS have entered into the Merger Agreement, pursuant to the terms of which, among other things, and subject to the conditions included in the Merger Agreement, ProFrac has agreed to acquire USWS by means of the merger of Merger Sub with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of ProFrac. A copy of the Merger Agreement is attached to this proxy statement/information statement/prospectus as Annex A.

### ProFrac Stockholders:

The Merger Agreement requires that the affirmative vote of the holders of a majority of the shares of ProFrac Class A Common Stock approve the ProFrac Stock Issuance. This approval was obtained through the execution and delivery by the ProFrac Controlling Stockholders of the ProFrac Stockholder Written Consent on June 21, 2022. In addition, all other conditions to the Merger Agreement must be satisfied or waived, please see section "The Merger Agreement – Conditions to Completion of the Merger." For ProFrac Stockholders, this proxy statement/information statement/prospectus contains important information about the ProFrac Stock Issuance approval, the Merger, the other transactions in relation thereto, and other actions taken in connection with the ProFrac Stockholder Written Consent. ProFrac is not asking ProFrac Stockholders for a proxy and ProFrac Stockholders are not requested to send ProFrac a proxy.

### USWS Stockholders:

Also, in order to complete the Merger, and in accordance with the DGCL, USWS Stockholders must approve and adopt the Merger Agreement and the transactions contemplated thereby, among other proposals as described in this proxy statement/information statement/prospectus.

The proposal to adopt the Merger Agreement requires the approval by the affirmative vote of a majority of the shares of USWS Common Stock outstanding on the record date. The proposal to approve the issuance of shares of USWS Common Stock to be issued by USWS upon conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes requires the approval by the affirmative vote of the holders of a majority of the total votes cast at the USWS Special Meeting. The proposal to approve the amendment to the USWS Certificate of Designations of Series A Preferred Stock requires the approval by a majority of the shares of USWS Common Stock outstanding on the record date. The proposal to amend the A&R LTIP to increase the number of shares issuable thereunder requires the affirmative vote of a majority of the shares of USWS Common Stock present in person or represented by proxy at the meeting and entitled to vote on this matter.

For USWS Stockholders, this proxy statement/information statement/prospectus, which you should read carefully, contains important information about the Merger, the issuance of USWS Common Stock in connection

with the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes and other matters being considered at the USWS Special Meeting. USWS is asking you for a proxy.

### Q: What will I receive in the Merger?

A: *ProFrac Stockholders*. At the Effective Time, other than the ProFrac Stockholders that own securities in USWS and are entitled to receive Merger Consideration pursuant to the Merger Agreement, each ProFrac Stockholder will hold the same number of shares of ProFrac Common Stock that such stockholder held immediately prior to consummation of the Merger.

USWS Stockholders. At the Effective Time, each share of USWS Common Stock will be converted automatically into the right to receive 0.3366 shares of ProFrac Class A Common Stock. Pursuant to the terms of the Merger Agreement, the agreed-upon Exchange Ratio resulted in the right to receive 0.0561 shares of ProFrac Class A Common Stock, which Exchange Ratio has subsequently been adjusted to 0.3366 to account for the 1-for-6 reverse stock split implemented by USWS on August 4, 2022. No fractional share interests, as a result of conversion of shares of USWS Common Stock into shares of ProFrac Class A Common Stock, shall entitle the owner thereof to any shares of ProFrac or to vote or to any other rights of a holder of ProFrac Class A Common Stock. All fractional shares to which a holder of USWS Common Stock would be entitled to receive shall be aggregated and rounded to three decimal places. In lieu of any such fractional shares, holders of USWS Common Stock who would otherwise be entitled to such fractional shares shall be entitled to an amount in cash, without interest, rounded to the nearest cent. The amount of cash for fractional shares will be calculated by multiplying the fraction of a share of ProFrac Class A Common Stock to which the USWS Stockholder would be entitled to receive in the Merger by the ProFrac Trading Price. For additional information on the exchange of USWS Common Stock for the Merger Consideration, please see "The Merger Agreement — Exchange of Shares."

USWS Equity Awards: At the Effective Time, the holders of the USWS Equity Awards will be entitled to receive Merger Consideration. For additional information on the cancellation and conversion of USWS Equity Awards for Merger Consideration, please see the sections titled "The Merger Agreement — Merger Consideration" and "The Merger Agreement — Exchange of Shares"

Upon the consummation of the Merger, the Wilks Parties will receive an aggregate of 4,138,596 shares of ProFrac Class A Common Stock as Merger Consideration (assuming, solely for the purpose of this calculation, a hypothetical Closing Date of October 31, 2022), which, based on the ProFrac Class A Common Stock June 21, 2022 closing price, would be approximately \$88.9 million. For more information regarding the Wilks Parties' financial and investment interest in the Merger, please see "The Merger — Interests of ProFrac Executive Officers and Directors in the Merger."

### Q: What happens to USWS' outstanding Series A Preferred Stock?

A: Pursuant to the Merger Agreement, USWS agreed to take all requisite action so that, effective as of immediately prior to the Effective Time, (i) each holder of USWS Series A Preferred Stock issued and outstanding at such time may convert such stock into shares of USWS Common Stock at the Merger Conversion Ratio (as defined below), and (ii) any shares of USWS Series A Preferred Stock issued and outstanding immediately prior to the Effective Time not so converted by the holder at the Merger Conversion Ratio will automatically convert into shares of USWS Common Stock at the then-effective conversion rate, as calculated pursuant to the Certificate of Designations.

The "Merger Conversion Ratio" means, for each share of USWS Series A Preferred Stock, the quotient of (i) its liquidation preference as of the date of the conversion and (ii) Merger Conversion Price. The "Merger Conversion Price" will be \$7.32, subject to adjustment.

### Q: What happens to USWS' outstanding Equity Linked Convertible Notes?

A: Pursuant to the Merger Agreement, USWS agreed to take all requisite action so that, effective as of immediately prior to the Effective Time, each USWS Equity Linked Convertible Note issued and outstanding at such time shall automatically convert into a number of shares of USWS Common Stock equal to the quotient obtained by dividing (i) the amount of outstanding aggregate principal amount, plus accrued and unpaid interest, owing under such USWS Equity Linked Convertible Note through July 9, 2022, by (ii) \$7.32.

### Q: What happens to USWS' outstanding warrants?

A: USWS Term C Loan Warrants: Pursuant to the Warrant Purchase Agreement dated as of June 21, 2022, immediately prior to the Effective Time and conditioned upon the consummation of the Merger, the holders of the USWS Term C Loan Warrants will sell all of their respective USWS Term C Loan Warrants to ProFrac or its subsidiaries in exchange for an aggregate purchase price of \$2,639,999.82. Upon the consummation of this Warrant Sale, at the Effective Time, the USWS Term C Loan Warrants will be automatically canceled and will cease to exist and no consideration will be delivered in exchange therefor.

### USWS Rollover Warrants:

Prior to the Effective Time, ProFrac will execute an amendment to each of the warrant agreements that govern the USWS Rollover Warrants as necessary to assume each warrant as of the Effective Time in accordance with the terms thereof and the Merger Agreement.

Each USWS Rollover Warrant, as amended, shall remain outstanding following the consummation of the Merger through its prescribed term, but shall become the right to receive, upon valid exercise thereof, shares of ProFrac Class A Common Stock equal to the product of (A) the number of shares of USWS Common Stock subject to such USWS Rollover Warrant immediately prior to the Effective Time and (B) the Exchange Ratio. Additionally, each USWS Rollover Warrant will be amended such that the exercise price of such USWS Rollover Warrant will equal the current exercise price of such USWS Rollover Warrant divided by the Exchange Ratio.

It is anticipated that the publicly traded USWS SPAC Warrants will be registered and traded on Nasdaq under the symbol "PFHCW".

### Q: Who will own ProFrac immediately following the Merger?

A: ProFrac and USWS estimate that, immediately following the completion of the Merger, based on the aggregate amount of capital stock and shares of ProFrac Class A Common Stock outstanding as of August 8, 2022, persons who were stockholders of ProFrac and USWS immediately prior to the Merger will own approximately 91.76% and 8.24%, respectively, of the then outstanding ProFrac capital stock and will own approximately 76.34% and 23.66%, respectively, of the then outstanding ProFrac Class A Common Stock.

At the Effective Time, the Wilks Parties will receive an aggregate of 4,138,596 shares of ProFrac Class A Common Stock as Merger Consideration (assuming, solely for the purpose of this calculation, a hypothetical closing date of October 31, 2022), which, based on the ProFrac Class A Common Stock June 21, 2022 closing price, would be approximately \$88.9 million. For more information regarding the Wilks Parties financial and investment interest in the Merger, please see "The Merger — Interests of ProFrac Executive Officers and Directors in the Merger."

### Q: Will the shares of ProFrac Class A Common Stock that I acquire in connection with the Merger receive a dividend?

A: After the closing of the Merger, as a holder of ProFrac Class A Common Stock, you will receive the same dividends on shares of ProFrac Class A Common Stock, if and when declared, that all other holders of ProFrac Class A Common Stock will receive for any dividend with a record date that occurs after the Effective Time. ProFrac has not paid a regular dividend on the ProFrac Class A Common Stock historically and has no present intention of instituting a regular dividend.

### Q: Will the shares of ProFrac Class A Common Stock received at the consummation of the Merger be traded on an exchange?

A: Yes. It is a condition to the consummation of the Merger that the shares of ProFrac Class A Common Stock issuable to the holders of USWS Common Stock and USWS Equity Awards in connection with the Merger be approved for listing on the Nasdaq, upon official notice of issuance. ProFrac Class A Common Stock is currently traded on Nasdaq under the symbol "PFHC."

### Q: How will ProFrac Stockholders be affected by the Merger?

A: At the Effective Time, other than the ProFrac Stockholders that own securities in USWS and are entitled to receive Merger Consideration pursuant to the Merger Agreement, each ProFrac Stockholder will hold the same number of shares of ProFrac securities that such stockholder held immediately prior to consummation of the Merger. As a result of the Merger, ProFrac Stockholders will own shares in a larger company with more assets. However, because ProFrac will be issuing additional shares of ProFrac Class A Common in connection with the ProFrac Stock Issuance, each share of ProFrac Class A Common Stock issued and outstanding immediately prior to the Merger will represent a smaller percentage of the aggregate number of shares of ProFrac Class A Common Stock issued and outstanding after the Merger.

### Q: Is the merger expected to be taxable to stockholders?

A: The Merger is expected to be treated as a taxable transaction for U.S. federal income tax purposes. Accordingly, a U.S. holder (as defined below in the section entitled "Material United States Federal Income Tax Consequences of the Merger") of USWS Common Stock generally will recognize gain or loss, if any, in an amount equal to the difference between (i) the sum of the fair market value as of the effective time of the Merger of any ProFrac Class A Common Stock received by such USWS Stockholder in the Merger and any cash received by such U.S. holder in the Merger in lieu of fractional shares of ProFrac Class A Common Stock and (ii) such USWS Stockholder's adjusted tax basis in the USWS Common Stock exchanged therefor.

Except in certain circumstances described in the section entitled "Material United States Federal Income Tax Consequences of the Merger") of USWS Common Stock generally will not be subject to U.S. federal income or withholding tax on any gain recognized on the exchange of USWS Common Stock for ProFrac Class A Common Stock and cash in the Merger. The summary above is qualified in its entirety by the more detailed discussion provided in the section entitled "Material United States Federal Income Tax Consequences of the Merger" beginning on page 127. Tax matters can be complicated and the tax consequences of the Merger to you will depend on your particular tax situation. We recommend that you consult your tax advisor to determine the tax consequences of the Merger to you.

### Questions and Answers Regarding the USWS Special Meeting and Voting

### Q: When and where is the USWS Special Meeting?

A: The USWS Special Meeting will be held on [ ], 2022 at [ : ] a.m./p.m. solely by means of remote communication in an audio-only virtual meeting format at www.proxydocs.com/USWS.

### Q: What are USWS Stockholders being asked to vote on?

A: USWS Stockholders are being asked to approve the following proposals at the USWS Special Meeting:

- To adopt the Merger Agreement and the transactions contemplated thereby, including the Merger;
- To approve, for purposes of complying with the Nasdaq listing rules, the issuance of shares of USWS Common Stock to be issued by USWS
  upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, in an amount equal to 20% or
  more of USWS Common Stock outstanding;
- To approve an amendment to the Certificate of Designations;
- To approve an amendment to the A&R LTIP to increase the maximum number of shares of USWS Common Stock that may be issued under the A&R LTIP by 2,000,000 shares; and
- If submitted to a vote of USWS Stockholders, to approve an adjournment of this USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of the foregoing proposals if there are not sufficient votes for those proposals.

### Q: What vote by USWS Stockholders is required to approve the proposals at the USWS Special Meeting?

A: The proposal to adopt the Merger Agreement requires the approval by the affirmative vote of a majority of the shares of USWS Common Stock outstanding on the record date. The proposal to approve the issuance of shares of USWS Common Stock to be issued by USWS upon conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes requires the approval by the affirmative vote of the holders of a majority of the total votes cast at the USWS Special Meeting. The proposal to approve the amendment to the Certificate of Designations requires the approval by a majority of the shares of USWS Common Stock outstanding on the record date. The proposal to amend the A&R LTIP to increase the number of shares issuable thereunder requires the affirmative vote of a majority of the shares of USWS Common Stock present in person or represented by proxy at the meeting and entitled to vote on this matter.

### Q: Who can vote at, and what are the record dates of, the USWS Special Meeting?

A: All holders of record of USWS Common Stock as of the close of business on [ ], 2022 are entitled to vote at the USWS Special Meeting and any adjournment or postponement thereof for which a new record date has not yet been established.

### Q: Why is my vote important?

A: Your vote is important, regardless of the number of shares of USWS Common Stock you own, as the Closing of the Merger is condition upon, among other things, the approval of the Merger and certain related proposals by the USWS Stockholders.

### Q: How does USWS Board recommend that I vote?

A: Acting on the unanimous recommendation and approval of the USWS Special Committee, the USWS Board has unanimously determined that the Merger Agreement, the Merger and the other transactions contemplated therein are advisable and in the best interests of USWS and its stockholders and therefore recommends USWS Stockholders vote "FOR" the USWS Merger Proposal, "FOR" the issuance of shares of USWS Common Stock to be issued by USWS upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, "FOR" the approval of the amendment to the Certificate of Designations, "FOR" the approval of the amendment to the A&R LTIP, and "FOR" the adjournment of the USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of any of the proposals to be acted upon at the USWS Special Meeting. For additional information regarding how the USWS Board recommends that USWS Stockholders vote, see the section titled "The Merger—USWS Board and USWS Special Committee Reasons for the Merger—Recommendations of the USWS Board and the USWS Special Committee."

### Q: How do I participate in the USWS Special Meeting?

A: To participate in the USWS Special Meeting, you must have your control number that is shown on your proxy card or the instructions that accompanied your proxy materials. You may access the Special Meeting by visiting www.proxydocs.com/USWS. You will be able to submit questions during the meeting via the website.

#### Q: How many votes may I cast?

A: Each issued and outstanding share of USWS Common Stock entitles its holder of record to vote on each matter to be considered at the USWS Special Meeting. The USWS Stockholders of record on the record date are the only USWS Stockholders that are entitled to receive notice of, and to vote at, the USWS Special Meeting or any adjournments or postponements thereof.

### Q: How many shares are entitled to vote at the USWS Special Meeting?

A: Each share of USWS Common Stock outstanding on the record date is entitled to one vote on the proposals to be voted on at the USWS Special Meeting. On the record date, there were [ ] shares of USWS Common Stock outstanding.

### Q: What constitutes a quorum at the USWS Special Meeting?

A: In accordance with the bylaws of USWS, a quorum exists if the holders of the outstanding shares of USWS Common Stock representing a majority of the voting power of all outstanding shares of USWS Common Stock entitled to vote as of the record date are present in person or by proxy at the USWS Special Meeting, including proxies on which abstentions (withholding authority to vote) are indicate. Abstentions and broker non-votes will be counter for purposes of establishing a quorum at the USWS Special Meeting. If less than a quorum is presented at the USWS Special Meeting, the presiding officer at the meeting or a majority of the shares so represented may move to adjourn the meeting. In the event a quorum is present at the USWS Special Meeting, but sufficient votes to approve any of the items proposed by the USWS Board have not been received, the presiding officer of the meeting or the persons named as proxies may propose one or more adjournments of the USWS Special Meeting to permit further solicitation of proxies. A stockholder vote may be taken on one or more of the proposals in this proxy statement/information statement/prospectus prior to such adjournment if sufficient proxies have been received and it is otherwise appropriate.

### O: Who will count the votes?

A: The votes will be counted by the inspector of elections appointed for the USWS Special Meeting.

### Q: What happens if I sell my shares before the Special Meeting?

A: If you sell or otherwise transfer your shares after the record date, but before the Special Meeting, unless special arrangements (such as provisions of a proxy) are made between you and the person(s) to whom you transfer your shares and each of you notifies the Secretary of USWS in writing of such a special arrangement, you will retain your right to vote such shares at the Special Meeting, but will otherwise have transferred ownership of your shares.

### Q: What is the difference between holding shares of record and holdings shares as a beneficial owner of shares of USWS Common Stock?

A: If your shares of USWS Common Stock are registered directly in your name with the registrar and transfer agent of USWS, Continental Stock Transfer & Trust Company, you are considered the "stockholder of record" of those shares in your account and you have the right to vote those shares. If your shares of USWS Common Stock

are held in an account with a broker or other nominees, you are deemed to be a "beneficial stockholder" of those shares, which are held in "street name." The broker or other nominee is deemed to be the "stockholder of record" for those shares. As a beneficial owner, you have the right to instruct the broker or other nominee on how to vote those shares in accordance with the instructions provided by your broker or other nominee.

Stockholders of USWS who receive a proxy card must sign and return each applicable proxy card in accordance with their respective instructions or submit a proxy by telephone or via the Internet at www.proxypush.com/USWS with respect to shares of USWS Common Stock in order to ensure the voting of the shares owned. The shares represented by your proxy will be voted at the USWS Special Meeting as directed by your proxy or voting instructions.

### Q: If my shares are held in "street name" by my broker, bank, trustee or other nominee, will my broker, bank, trustee or other nominee vote my shares for me?

A: If you are a beneficial owner whose USWS Common Stock is held of record by your broker, bank, trustee or other nominee, you must instruct the broker, bank, trustee or other nominee how to vote your shares of USWS Common Stock. Brokers, banks, trustees and other nominees who hold shares of USWS Common Stock in "street name" typically have the authority to vote in their discretion on "routine" proposals when they have not received instructions on how to vote from the beneficial owner. However, brokers, banks, trustees and other nominees typically are not allowed to exercise their voting discretion on matters that are "non-routine" without specific instructions on how to vote from the beneficial owner. Under the current rules of Nasdaq, all of the proposals being considered at the USWS Special Meeting are non-routine. Therefore, brokers, banks, trustees and other nominees do not have discretionary authority to vote on any of the proposals.

A "broker non-vote" with respect to shares of USWS Common Stock occurs when (i) a share of USWS Common Stock held by a broker, bank, trustee or other nominee is present or represented at a meeting of holders of USWS Common Stock, (ii) the beneficial owner of such shares of USWS Common Stock has not instructed his, her or its broker, bank, trustee or other nominee does not have discretionary voting power on such proposal. Brokers, banks, trustees and other nominees do not have discretionary voting authority with respect to any of the proposals being considered at the USWS Special Meeting; therefore, if a beneficial owner of shares of USWS Common Stock held in "street name" does not give voting instructions to the broker, bank, trustee or other nominee, then those shares of USWS Common Stock will not be present at the USWS Special Meeting or represented by proxy at the Special Meeting because the broker non-votes will not count toward the quorum requirements. As a result, it is expected that there will not be any broker non-votes at the USWS Special Meeting.

Please follow the voting instructions provided by your broker, bank, trustee or other holder of record. Please note that you may not vote shares held in "street name" by returning a proxy card directly to USWS or by voting at the USWS Special Meeting unless you provide a "legal proxy," which you must obtain from your broker, bank, trustee or other holder of record. Further, brokers, banks, trustees or other holders of record who hold shares of USWS Common Stock on your behalf may not give a proxy to vote those shares without specific voting instructions from you.

Your vote is important, and you are urged to provide voting instructions for your shares by following the instructions provided on the enclosed proxy card(s) or voting instruction card(s). Please vote promptly.

### Q: What happens if I fail to vote, abstain from voting or fail to instruct my broker, bank, trustee or other nominee to vote my shares?

A: You are strongly encouraged to vote. Your abstention from voting will have the effect of a vote "AGAINST" the USWS Merger Proposal, the proposal to approve the amendment to the A&R LTIP; and your failure to vote or the failure to instruct your

broker, bank, trustee or other nominee to vote will have the effect of a vote "AGAINST" the USWS Merger Proposal, the proposal to approve the amendment to the Certificate of Designations and the proposal to approve the amendment to the A&R LTIP.

### Q: What should I do if I receive more than one set of voting materials for the USWS Special Meeting?

A: This means you own shares of USWS Common Stock that are registered under different names or are in more than one account. For example, you may own some shares directly as a USWS Stockholder of record and other shares through a broker, or you may own shares through more than one broker. In these situations, you will receive multiple sets of proxy materials. You must vote, sign and return all of the proxy cards or follow the instructions for any alternative voting procedure on each of the proxy cards that you receive in order to vote all of the shares you own.

### Q: What will happen if I return my proxy card without indicating how to vote?

A: If you sign and send in your proxy and do not indicate how you want to vote on any particular proposal, the USWS Common Stock represented by your proxy will be voted "FOR" that proposal.

### Q: What do I do if I want to change my vote?

A: You can change your vote at any time before your proxy is voted at the USWS Special Meeting. You can do this in one of four ways:

- · you can send a signed notice of revocation;
- you can grant a new, valid proxy by executing a new proxy card bearing a later date;
- you can vote at a later time by telephone or through the Internet; or
- if you are a holder of record, you can attend the applicable special meeting and vote in person, but your attendance alone will not revoke any proxy that you have previously given.

If you choose either of the first two methods, your notice of revocation or your new proxy must be received by the USWS Corporate Secretary at the address provided under "The Companies" beginning on page 1 no later than the beginning of the USWS Special Meeting. If you are a USWS Stockholder, you can find further details on how to revoke your proxy in "The USWS Special Meeting — Revocation of Proxies" beginning on page 153.

If you hold your shares in "street name," please refer to your proxy card or the information forwarded by your bank, broker or other holder of record to see which options are available to you.

### Q: Can I dissent and require appraisal of my shares?

A: ProFrac Stockholders. Under Delaware law, ProFrac Stockholders have no right to an appraisal of the "fair value" of their shares in connection with the Merger.

USWS Stockholders. Under Delaware law, USWS Stockholders have no right to an appraisal of the "fair value" of their shares in connection with the Merger.

### Q: Do any of the officers or directors of USWS have interests in the Merger that may differ from or be in addition to my interests as a USWS Stockholder?

A: Yes, certain officers and directors of USWS have interests in the Merger that may differ from or be in addition to the interests of USWS Stockholders. For more information on these interests, see "The Merger — Interests of USWS Executive Officers and Directors in the Merger" beginning on page 122.

### Q: What do I do now?

A: Carefully read and consider the information contained in and referred herein into this proxy statement/information statement/prospectus, including its annexes.

For USWS Stockholders, in order for your shares to be represented at the USWS Special Meeting:

- you can vote by telephone or through the Internet by following the instructions included on your proxy card;
- you can indicate on the enclosed proxy card how you would like to vote and sign and return the proxy card in the accompanying
  pre-addressed postage paid envelope; or
- you can attend the USWS Special Meeting and vote virtually at www.proxydocs.com/USWS.

If you hold your shares in "street name," please refer to your proxy card or the information forwarded by your bank, broker or other holder of record to see which options are available to you.

### Q: Should stockholders send in their stock certificates with their proxies?

A: No. USWS Stockholders should not send in their stock certificates at this time. If the Merger is consummated, ProFrac's exchange agent will send former USWS Stockholders a letter of transmittal and written instructions explaining what they must do to exchange their USWS stock certificates or transfer uncertificated shares for the Merger Consideration payable to them.

ProFrac Stockholders will retain their current stock certificates after the Merger and should not send in their stock certificates.

### Q: When do you expect to complete the Merger?

A: ProFrac and USWS are working to complete the Merger as quickly as practicable. However, ProFrac and USWS cannot assure you when or if the Merger will be completed. Completion of the Merger is subject to satisfaction or waiver of the conditions specified in the Merger Agreement, including, among other matters, the receipt of the USWS Stockholder Approval. It is possible that factors outside the control of both companies could result in the Merger being completed later than expected. Although the exact timing of completion of the Merger cannot be predicted with certainty, ProFrac and USWS anticipate completing the Merger in the fourth quarter of 2022. See "The Merger Agreement — Conditions to Completion of the Merger" beginning on page 144.

### Q: What happens if the Merger is not completed?

A: If the USWS Merger Proposal is not approved by the USWS Stockholders or the Merger is not completed for any other reason, USWS Stockholders will not receive any payment for shares of USWS Common Stock they own. Instead, USWS will remain an independent public company, USWS Common Stock and its public warrants will continue to be listed and traded on Nasdaq and registered under the 1934 Act and USWS will continue to file periodic reports with the SEC.

Under specified circumstances in the Merger Agreement, USWS may be required to pay ProFrac a termination fee of \$8,000,000 or reimburse ProFrac for certain expenses in an amount up to \$3,000,000, in each case, depending on the termination event. See "The Merger Agreement — Termination Fees" beginning on page 147.

### Q: Whom should I call with questions?

A: If you have any questions about the USWS Special Meeting, the Merger, the ProFrac Stock Issuance, the USWS Proposals or how to submit your proxy, or if you need additional copies of this proxy statement/

information statement/prospectus or documents referenced herein, the applicable enclosed proxy card or voting instructions, you should contact:

For ProFrac Stockholders:

ProFrac Holding Corp. 333 Shops Boulevard, Suite 301 Willow Park, TX 76087 Attention: Investor Relations Email: investors@profrac.com

### For USWS Stockholders

U.S. Well Services, Inc. 1360 Post Oak Boulevard, Suite 1800 Houston, Texas 77056 Attention: Corporate Secretary Email: esimonson@uswellservices.com

### Q: How can I submit questions to be answered during the Special Meeting?

A: If you wish to submit a question to be answered during the USWS Special Meeting, you may do so. You may submit a question prior to the USWS Special Meeting at any time before 11:59 p.m. ET on [ ], 2022, by logging into www.proxypush.com/USWS and entering your 16-digit control number.

Questions pertinent to meeting matters, if any, will be answered during the USWS Special Meeting, subject to time constraints.

### Q: How can I inspect the list of Stockholders of Record?

A: A complete list of the holders of USWS Common Stock entitled to vote at the USWS Special Meeting will be available for examination by any USWS Stockholder, for any purpose germane to the USWS Special Meeting, at the USWS principal executive offices at 1360 Post Oak Boulevard, Suite 1800, Houston, Texas 77056, for a period of ten days before the USWS Special Meeting, between the hours of 9:00 a.m. and 4:00 p.m., local time, and at the USWS Special Meeting during the entire time of the meeting.

### Q: Where can I find the voting results of the USWS Special Meeting?

A: USWS plans to announce preliminary voting results at the USWS Special Meeting, and will publish results on a Current Report on Form8-K filed with the SEC following the Special Meeting.

### **SUMMARY**

This summary highlights information contained elsewhere in this proxy statement/information statement/prospectus. It does not contain all of the information that may be important to you. You are urged to read carefully this entire proxy statement/information statement/prospectus, including the attached annexes, and the other documents to which this proxy statement/information statement/prospectus refers you in order for you to fully understand the proposed Merger. See "Where You Can Find More Information" beginning on page 276. Each item in this summary refers to the page of this proxy statement/information statement/prospectus on which that subject is discussed in more detail.

### The Companies

### ProFrac (see page 133)

ProFrac Holding Corp. 333 Shops Boulevard, Suite 301 Willow Park, TX 76087 (254) 776-3722 www.pfholdingscorp.com

ProFrac is a growth-oriented, vertically integrated and innovation-driven energy services company providing hydraulic fracturing, completion services and other complementary products and services to leading upstream oil and gas companies engaged in the E&P of North American unconventional oil and natural gas resources. Founded in 2016, ProFrac was built to be the go-to service provider for E&P companies' most demanding hydraulic fracturing needs. ProFrac is focused on employing new technologies to significantly reduce "greenhouse gas" ("GHG") emissions and increase efficiency in what has historically been an emissions-intensive component of the unconventional E&P development process.

The information contained on ProFrac's website is expressly not incorporated by reference into this proxy statement/information statement/prospectus and you should not consider information contained on ProFrac's website to be part of this proxy statement/information statement/prospectus.

### Merger Sub (see page 133)

Thunderclap Merger Sub I, Inc. 333 Shops Boulevard, Suite 301 Willow Park, TX 76087 (254) 776-3722

Merger Sub is an indirect subsidiary of ProFrac that was incorporated under the laws of the State of Delaware in June 2022. Merger Sub does not engage in any operations and exists solely to facilitate the Merger.

### USWS (see page 133)

U.S. Well Services, Inc. 1360 Post Oak Boulevard, Suite 1800 Houston, Texas 77056 (832) 562-3730 www.uswellservices.com

USWS is a technology-focused oilfield service company focused on electric pressure pumping services for oil and natural gas E&P companies in the United States. USWS' patented electric pressure pumping technology

provides one of the first fully electric, mobile well stimulation systems powered by locally supplied natural gas including field gas sourced directly from the wellhead. USWS' electric pressure pumping technology decreases emissions, sound pollution and truck traffic while generating operating efficiencies including significant customer fuel cost savings versus conventional diesel-powered fleets. The USWS fleet consists mostly of all-electric, mobile pressure pumping equipment and other auxiliary heavy equipment that performs stimulation services. USWS' Clean Fleet® electric fleets replace the traditional engines, transmissions, and radiators used in conventional diesel fleets with electric motors powered by electricity.

The information contained on USWS' website is expressly not incorporated by reference into this proxy statement/information statement/prospectus, and you should not consider information contained on USWS' website to be part of this proxy statement/information statement/prospectus.

### The Merger

The Merger Agreement is attached as <u>Annex A</u> to this proxy statement/information statement/prospectus. ProFrac and USWS encourage you to carefully read the Merger Agreement in its entirety because it is the principal legal agreement that governs the Merger.

### Structure of the Merger (see page 134)

Subject to the terms and conditions of the Merger Agreement, at the Effective Time, among other matters, Merger Sub will merge with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of ProFrac.

### Merger Consideration (see page 134)

As a result of the Merger, each share of USWS Common Stock will be converted automatically into the right to receive 0.3366 shares of ProFrac Class A Common Stock. ProFrac will not issue any fractional shares of ProFrac Class A Common Stock in connection with the Merger. Instead, USWS Stockholders will receive cash in lieu of any fractional shares of ProFrac Class A Common Stock that they would have otherwise received in the Merger. The amount of cash for fractional shares will be calculated by multiplying the fraction of a share of ProFrac Class A Common Stock to which the USWS Stockholder would be entitled to receive in the Merger by the ProFrac Trading Price.

ProFrac Stockholders will continue to own their existing shares of ProFrac Class A Common Stock after the Merger.

### **ProFrac Stock Issuance**

On June 21, 2022, in accordance with the Merger Agreement, the ProFrac Controlling Stockholders approved the ProFrac Stock Issuance by delivery of the ProFrac Stockholder Written Consent. Since the ProFrac Stock Issuance will result in the issuance of shares of ProFrac Class A Common Stock constituting 20% or more of the outstanding ProFrac Class A Common Stock, Nasdaq rules require such issuance to be approved by the holders of the majority of outstanding ProFrac Class A Common Stock. The ProFrac Stockholder Written Consent constitutes such approval. As of June 21, 2022, the ProFrac Controlling Stockholders held the majority of the voting power of the outstanding ProFrac Class A Common Stock. Therefore, the delivery of the ProFrac Stockholder Written Consent satisfied the requirement in the Merger Agreement and under the Nasdaq rules that the affirmative vote of the holders of at least a majority of the voting power of the ProFrac Class A Common Stock approve the ProFrac Stock Issuance. No additional approval of the stockholders of ProFrac is required to approve the ProFrac Stock Issuance, Merger Agreement, the Merger and the other transactions. As a result, ProFrac has not solicited and will not be soliciting your vote for the approval of the ProFrac Stock Issuance, the

Merger Agreement, the Merger or the other transactions and does not intend to call a meeting of stockholders for purposes of voting on the adoption of the ProFrac Stock Issuance, the Merger Agreement, the Merger or the other transactions. A copy of the ProFrac Stockholder Written Consent is attached to this proxy statement/information statement/prospectus as Annex B. For information on the recommendation of the ProFrac Board, please see 'The Merger — ProFrac Board and ProFrac Special Committee Reasons for the Merger.'

### Ownership of the Combined Company After the Merger

ProFrac will issue approximately 12.8 million shares of ProFrac Class A Common Stock to USWS Stockholders (after giving effect to the conversions of the USWS Series A Preferred Stock, USWS Equity Linked Convertible Notes and USWS Equity Awards) in the Merger based on the number of shares of USWS Common Stock and ProFrac Class A Common Stock outstanding as of August 29, 2022. At the completion of the Merger, it is expected that there will be outstanding approximately 54 million shares of ProFrac Class A Common Stock.

Immediately following the completion of the Merger, based on the aggregate amount of capital stock and shares of ProFrac Class A Common Stock outstanding as of August 8, 2022, it is anticipated that persons who were stockholders of ProFrac and USWS immediately prior to the Merger will own approximately 91.76 % and 8.24%, respectively, of the then outstanding ProFrac capital stock and will own approximately 76.34 % and 23.66%, respectively, of the then outstanding ProFrac Class A Common Stock. At the Effective Time, other than the ProFrac Stockholders that own securities in USWS and are entitled to receive Merger Consideration pursuant to the Merger Agreement, each ProFrac Stockholder will hold the same number of shares of ProFrac Class A Common Stock unstanding immediately prior to consummation of the Merger. However, upon the ProFrac Stock Issuance, each share of ProFrac Class A Common Stock outstanding immediately prior to the consummation of the Merger will represent a smaller percentage of the aggregate number of shares of ProFrac Class A Common Stock outstanding after the consummation of the Merger. Following the consummation of the Merger, each share of ProFrac Class A Common Stock will represent an equity interest in a combined company with more assets.

### Recommendations of the ProFrac and USWS Boards

Recommendations to ProFrac Stockholders. The ProFrac Board, acting on the unanimous recommendation and approval of the ProFrac Special Committee, unanimously determined that the Merger Agreement and the transactions contemplated thereby were in the best interests of ProFrac and the ProFrac Unaffiliated Stockholders, approved, adopted and declared advisable the Merger Agreement and the transactions contemplated thereby, approved and declared advisable certain ancillary agreements contemplated by the Merger Agreement or otherwise entered into in connection with the Merger, directed that the ProFrac Stock Issuance proposal be submitted to ProFrac Stockholders for approval as required by Nasdaq rules, and resolved to recommend that ProFrac Stockholders approve the ProFrac Stock Issuance. The ProFrac Board unanimously recommended that ProFrac Stockholders vote "FOR" the ProFrac Stock Issuance proposal.

Recommendations to USWS Stockholders. The USWS Board, acting on the unanimous recommendation and approval of the USWS Special Committee, has unanimously determined that the Merger Agreement and the transactions contemplated thereby, including the Merger, are advisable and fair to and in the best interests of USWS and its stockholders. The USWS Board recommends that USWS Stockholders vote: "FOR" the USWS Merger Proposal, "FOR" the issuance of shares of USWS Common Stock to be issued by USWS upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, "FOR" the approval of the amendment to the Certificate of Designations, "FOR" the approval of the amendment to the A&R LTIP, and "FOR" the adjournment of the USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of any of the proposals to be acted upon at the USWS Special Meeting. For additional information see "The USWS Special Meeting — Board Recommendations" beginning on page 150.

### **Opinions of Financial Advisors**

Opinion of the Financial Advisor to the ProFrac Special Committee (see page 98 and Annex C)

The ProFrac Special Committee has retained Jefferies as its independent financial advisor in connection with the Merger. As part of this engagement, Jefferies delivered a written opinion, dated June 21, 2022, to the ProFrac Special Committee as to the fairness, from a financial point of view and as of such date, to ProFrac of the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement.

The full text of Jefferies' opinion, which describes the various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken by Jefferies, is attached as Annex C to this proxy statement/information statement/prospectus and is incorporated here by reference. ProFrac encourages you to read the opinion carefully and in its entirety. Jefferies' opinion was provided for the use and benefit of the ProFrac Special Committee (in its capacity as such) in its evaluation of the Exchange Ratio from a financial point of view and did not address any other aspect of the Merger or any other matter. Jefferies' opinion did not address the relative merits of the Merger or other transactions contemplated by the Merger Agreement as compared to any alternative transaction or opportunity that might be available to ProFrac, nor did it address the underlying business decision by ProFrac to engage in the Merger or any term, aspect or implication of any other agreement entered into in connection with, or contemplated by or resulting from, the Merger or otherwise. Jefferies' opinion did not constitute a recommendation as to how the ProFrac Special Committee or the ProFrac Board, and does not constitute a recommendation as to how any securityholder, should vote or act with respect to the Merger or any other matter. The summary of Jefferies' opinion is qualified in its entirety by reference to the full text of Jefferies' opinion.

Opinion of the Financial Advisor to the USWS Special Committee (see page 115 and Annex D)

USWS retained Piper Sandler to act as an independent financial advisor to the USWS Special Committee in connection with the proposed Merger and participated in certain of the negotiations leading to the execution of the Merger Agreement. On June 21, 2022, prior to the execution of the Merger Agreement, Piper Sandler separately delivered its written opinion to the USWS Special Committee and the USWS Board, to the effect that, as of such date, the exchange ratio was fair, from a financial point of view, to the holders of USWS Common Stock issued and outstanding immediately prior to the effective time of the Merger, in their capacity as holders of USWS Common Stock.

The full text of Piper Sandler's opinion is attached as <u>Annex D</u> to this proxy statement/information statement/prospectus. The opinion outlines the procedures followed, assumptions made, matters considered and qualifications and limitations on the review undertaken by Piper Sandler in rendering its opinion. The description of the opinion set forth below is qualified in its entirety by reference to the full text of the opinion. Holders of USWS Common Stock are urged to read the entire opinion carefully in connection with their consideration of the proposed Merger.

### Treatment of USWS Series A Preferred Stock (see page 135)

Pursuant to the Merger Agreement, USWS agreed to take all requisite action so that, effective as of immediately prior to the Effective Time, (i) each holder of USWS Series A Preferred Stock issued and outstanding at such time may convert such stock into shares of USWS Common Stock at the Merger Conversion Ratio (as defined below), and (ii) any shares of USWS Series A Preferred Stock issued and outstanding immediately prior to the Effective Time not so converted at the Merger Conversion Ratio will automatically convert into shares of USWS Common Stock at the then-effective conversion rate as calculated pursuant to USWS' Certificate of Designations.

The "Merger Conversion Ratio" means, for each share of USWS Series A Preferred Stock, the quotient of (i) its liquidation preference as of the date of the conversion and (ii) Merger Conversion Price. The "Merger Conversion Price" will be \$7.32, subject to adjustment.

For more information about the amendment to the Certificate of Designation, see "The Merger Agreement — Related Agreements."

### Treatment of USWS Equity Linked Convertible Notes (see page 135)

Pursuant to the Merger Agreement, USWS agreed to take all requisite action so that, effective as of immediately prior to the Effective Time, each USWS Equity Linked Convertible Note issued and outstanding at such time shall automatically convert into a number of shares of USWS Company Stock equal to the quotient obtained by dividing (i) the amount of outstanding aggregate principal amount, plus accrued and unpaid interest, owing under such USWS Equity Linked Convertible Note through July 9, 2022, by (ii) \$7.32.

For more information about the amendments to the USWS Equity Linked Convertible Notes, see "The Merger Agreement — Related Agreements."

### Treatment of USWS Warrants (see page 136)

USWS Term C Loan Warrants

Subject to the consummation of the Warrant Sale, at the Effective Time, the USWS Term C Loan Warrants issued and outstanding immediately prior to the Effective Time (which shall be held by ProFrac following the consummation of the Warrant Sale) will be automatically canceled and will cease to exist and no consideration will be delivered in exchange therefor.

For more information about the Warrant Sale, please see "The Merger Agreement-Related Agreements" on page 148.

#### Rollover Warrants

Prior to the Effective Time, ProFrac will execute an amendment to each of the warrant agreements that govern the USWS Rollover Warrants as necessary to assume each warrant as of the Effective Time in accordance with the terms thereof and the Merger Agreement.

Each USWS Rollover Warrant shall remain outstanding following the Merger through its prescribed term but shall represent the right to receive upon valid exercise thereof shares of ProFrac Class A Common Stock equal to the product of (A) the number of shares of USWS Common Stock subject to such USWS Rollover Warrant immediately prior to the Effective Time and (B) the Exchange Ratio. Additionally, each USWS Rollover Warrant will be amended such that the exercise price of such USWS Rollover Warrant will equal the current exercise price of such USWS Rollover Warrant divided by the Exchange Ratio.

Upon consummation of the Merger and effectiveness of this Registration Statement on Form S-4, the publicly tradable warrants of USWS, currently listed on the Nasdaq Capital Market under the symbol "USWSW," shall become publicly tradable warrants of ProFrac and shall be listed on the Nasdaq Global Select Market under the symbol "PFHCW." Such warrants shall remain outstanding following the consummation of the Merger and shall be amended and restated upon the effectiveness and consummation of the Merger such that they represent the right to purchase, upon exercise thereof, shares of ProFrac Class A Common Stock (rather than shares of USWS Class A Common Stock) based on the Exchange Ratio and correspondingly adjusted exercise price.

For more information about the USWS Rollover Warrants, see the section titled "The Merger Agreement — Treatment of USWS Warrants – Rollover Warrants."

### Treatment of USWS Equity Awards (see page 136)

Pursuant to the Merger Agreement, USWS will take, or cause to be taken, all actions necessary so that:

- at the Effective Time, the USWS Restricted Shares that are outstanding and unvested under the USWS LTIP immediately prior to the
  Effective Time, will be canceled in exchange for the right to receive the Merger Consideration and, in lieu of any fractional shares,
  cash:
- immediately prior to the Effective Time, each USWS DSU will be canceled and converted into the right to receive the Merger Consideration and, in lieu of any fractional shares, cash (treating such USWS DSU in the same manner as if it were an outstanding share of USWS Common Stock for such purposes); and
- immediately prior to the Effective Time, all outstanding USWS Pool A Performance Awards and USWS Pool B Performance Awards will be canceled and converted into the right to receive a certain amount of Merger Consideration and such amount will be dependent upon the holder and the type of such performance award, as further described in the accompanying proxy statement/information statement/prospectus.

### Interests of ProFrac's and USWS' Executive Officers and Directors in the Merger (see pages 105 and 122)

ProFrac

The ProFrac Stockholders should note that some directors and executive officers of ProFrac have interests in the Merger that are different from, or are in addition to, the interests of ProFrac's stockholders generally. These interests may present them with actual or potential conflicts of interest.

Specifically, Ladd Wilks, the Chief Executive Officer of ProFrac, is the son of Farris Wilks and the nephew of Dan Wilks and Matthew D. Wilks, the Executive Chairman of the ProFrac Board, is the son of Dan Wilks and the nephew of Farris Wilks. Additionally, Matthew D. Wilks is the Vice President of Investments of THRC Holdings, an affiliate of Dan Wilks and Farris Wilks.

The Wilks Parties collectively hold a controlling interest in ProFrac and beneficially own certain securities of USWS, as described in this proxy statement/information statement/prospectus.

Upon the consummation of the Merger, the Wilks Parties will receive an aggregate of 4,138,596 shares of ProFrac Class A Common Stock as Merger Consideration (assuming, solely for the purpose of this calculation, a hypothetical Closing Date of October 31, 2022), which, based on the ProFrac Class A Common Stock June 21, 2022 closing price, would be approximately \$88.9 million.

These interests are described in more detail in the section titled "The Merger — Interests of ProFrac Executive Officers and Directors in the Merger" beginning on page 105. The ProFrac Special Committee and ProFrac Board were aware of these interests and considered them, among other matters, in approving the Merger, the Merger Agreement and the transactions contemplated thereby.

### USWS

In considering the USWS Board's recommendation that USWS Stockholders vote in favor of the proposals described in this proxy statement/information statement/prospectus, USWS Stockholders should be aware that some USWS executive officers and directors may have interests that may be different from, or in addition to, USWS Stockholders' interests, including but not limited to, their receipt of change of control payments under existing USWS Equity Awards in connection with the Merger, entry into new change of control agreements and retention and severance agreements with ProFrac, new employment arrangements with ProFrac, accelerated vesting of USWS Equity Awards at the Effective Time, and the right of USWS' executive officers and directors to continued indemnification and insurance coverage for acts or omissions occurring before the Merger.

In addition, some of USWS' directors hold certain convertible securities and may have interests that are different from, or in addition to, USWS Stockholders' interest. For example, David Matlin, a member of the USWS Board, is a lender to USWS and holds USWS Term C Loan Warrants and will receive \$122,791 in the Warrant Sale. Adam Klein, a member of the USWS Board, is a partner at Crestview Partners. Crestview Partners is a lender to USWS and holds certain convertible securities in USWS, including the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, both of which are convertible into additional shares of USWS Common Stock.

These interests are described in more detail in the section titled "The Merger — Interests of USWS Executive Officers and Directors in the Merger" beginning on page 122. The USWS Special Committee and USWS Board were aware of these interests and considered them, among other matters, in recommending the adoption of the Merger Agreement and approval of the Merger, the Merger Agreement and the transactions contemplated thereby.

### Appraisal Rights (see page 126)

Under the DGCL, neither the ProFrac Stockholders nor the USWS Stockholders have a right to an appraisal of the "fair value" of their shares in connection with the Merger.

### Material United States Federal Income Tax Consequences of the Merger (see page 127)

The Merger is expected to be treated as a taxable transaction for U.S. federal income tax purposes. Accordingly, a U.S. holder (as defined below in the section entitled "Material United States Federal Income Tax Consequences of the Merger") of USWS Common Stock generally will recognize gain or loss, if any, in an amount equal to the difference between (i) the sum of the fair market value as of the effective time of the Merger of any ProFrac Class A Common Stock received by such USWS Stockholder in the Merger and any cash received by such U.S. holder in the Merger in lieu of fractional shares of ProFrac Class A Common Stock and (ii) such USWS Stockholder's adjusted tax basis in the USWS Common Stock exchanged therefor.

Except in certain circumstances described in the section entitled "Material United States Federal Income Tax Consequences of the Merger – Tax Consequences to Non-U.S. Holders", a Non-U.S. holder (as defined below in the section entitled "Material United States Federal Income Tax Consequences of the Merger") of USWS Common Stock generally will not be subject to U.S. federal income or withholding tax on any gain recognized on the exchange of USWS Common Stock for ProFrac Class A Common Stock and cash in the Merger.

The summary above is qualified in its entirety by the more detailed discussion provided in the section entitled 'Material United States Federal Income Tax Consequences of the Merger" beginning on page 127. Tax matters can be complicated and the tax consequences of the Merger to you will depend on your particular tax situation. We recommend that you consult your tax advisor to determine the tax consequences of the Merger to you.

### Accounting Treatment (see page 125)

ProFrac prepares its financial statements in accordance with GAAP. The Merger will be accounted for using the purchase method of accounting. As discussed under "The Merger — Accounting Treatment" on page 125, based on the facts that ProFrac board members and senior management will control and represent a majority of the board of directors and senior management of the combined company, as well as the terms of the exchange, pursuant to which USWS Stockholders will receive a premium of approximately 68% (as of the date preceding the Merger announcement) over the fair market value of their shares on such date, ProFrac is considered to be the acquirer of USWS for accounting purposes. ProFrac will allocate the purchase price to the fair value of USWS' assets and liabilities at the Effective Time, with the excess purchase price being recorded as goodwill.

### Regulatory and Other Approvals Required for the Merger (see page 131)

ProFrac and USWS have each agreed to use their reasonable best efforts in order to obtain all regulatory approvals required in order to consummate and make effective the Merger and the other transactions contemplated by the Merger Agreement. These include approvals by the U.S. Federal Trade Commission, the U.S. Department of Justice and state antitrust authorities. ProFrac and USWS each submitted their antitrust filings with the U.S. Department of Justice and the U.S. Federal Trade Commission on July 6, 2022, which commenced the initial waiting period of 30 days. On August 6, 2022, the waiting period under the HSR Act expired without a request for additional information. Based on the review and analyses performed ProFrac and USWS do not believe that any other antitrust related filings are required in any foreign jurisdictions.

### Listing of ProFrac Class A Common Stock (see page 132)

ProFrac has agreed to use its best efforts to cause the shares of ProFrac Class A Common Stock to be issued as the Merger Consideration to be approved for listing on Nasdaq, subject to official notice of issuance, prior to the Effective Time. ProFrac Class A Common Stock is currently traded on Nasdaq under the symbol "PFHC."

#### Representations, Warranties and Covenants (see page 137)

The parties to the Merger Agreement have made representations, warranties and covenants that are customary for transactions of this nature. The representations and warranties of the respective parties to the Merger Agreement will not survive the closing of the Merger.

The Merger Agreement includes customary covenants of the parties, including, among others, covenants (a) for the parties to conduct their respective businesses in the ordinary course during the interim period between the date of the execution of the Merger Agreement and the consummation of the Merger, (b) that the parties not engage in certain kinds of transactions during the interim period between the date of the execution of the Merger Agreement and the consummation of the Merger, (c) that ProFrac use commercially reasonable efforts to obtain and consummate a financing to fund the repayment in connection with the Closing of certain indebtedness of USWS and its subsidiaries as promptly as reasonably possible following the date of the Merger Agreement, (d) that USWS reasonably cooperate with such financing efforts and (e) providing for ProFrac and USWS to cooperate in the preparation, and obtaining the effectiveness of, the Registration Statement on Form S-4, which includes this proxy statement/information statement/prospectus.

USWS further agreed (a) to amend the terms of the USWS Equity Linked Convertible Notes and the Certificate of Designations regarding the USWS Series A Preferred Stock, and to use commercially reasonable efforts to amend the USWS Pool A Performance Awards and USWS Pool B Performance Awards not already amended pursuant to an executed Award Amendment, (b) not to solicit proposals relating to alterative business combination transactions or, subject to certain exceptions, enter into discussions or an agreement concerning or provide confidential information in connection with any proposals for alternative business combination transactions and (c) as soon as practicable following the date of the Merger Agreement convene and hold the USWS Special Meeting for the purposes of obtaining the USWS Stockholder Approval.

For more information about the Amendments to the Certificate of Designations and USWS Equity Linked Convertible Notes, see "The Merger Agreement — Related Agreements."

### Conditions to Completion of the Merger (see page 144)

The obligations of each of ProFrac and USWS to complete the Merger are subject to the satisfaction at or before the Effective Time of a number of customary conditions, including, among others: (a) receipt of the USWS Stockholder Approval, (b) the absence of any applicable law or order prohibiting the consummation of the Merger or the ProFrac Stock Issuance, (c) the applicable waiting period under the HSR Act relating to the

transactions contemplated by the Merger Agreement shall have expired or terminated, (d) ProFrac's Registration Statement on FormS-4, which includes this proxy statement/information statement/prospectus, being effective and not subject to any stop order that is in effect or pending proceeding seeking a stock order, and (e) approval for the listing on Nasdaq of the shares of ProFrac Class A Common Stock to be issued in the Merger, subject to official notice of issuance.

In addition, the obligations of each of ProFrac and USWS to complete the Merger are subject to the satisfaction or waiver at or before the Effective Time of the following additional conditions, among others: (a) the representations and warranties of the other party shall be accurate to the extent specified in the Merger Agreement; (b) the other party shall have performed in all material respects all of its obligations required to be performed by it at or prior to the Closing Date and any such failure to perform shall have been cured on or prior to the Closing Date; and (c) no Parent Material Adverse Effect or Company Material Adverse Effect, as such terms are defined in the Merger Agreement shall have occurred and be continuing with respect to the other party since June 21, 2022.

### Termination of the Merger Agreement (see page 145)

ProFrac and USWS may mutually agree in writing to terminate the Merger Agreement at any time prior to the Effective Time. In addition, either ProFrac or USWS may also terminate the Merger Agreement, under specified circumstances, if the Merger is not completed by March 21, 2023, or under other circumstances described in this proxy statement/information statement/prospectus.

#### Termination Fees and Other Expenses (see page 147)

Generally, all costs and expenses incurred in connection with the Merger Agreement and the transactions contemplated thereby will be paid by the party incurring those costs or expenses, subject to the specific exceptions discussed in this proxy statement/information statement/prospectus. The Merger Agreement provides that, upon termination of the Merger Agreement under specified circumstances, USWS may be required to pay ProFrac a termination fee of \$8,000,000 or reimburse ProFrac for certain expenses in an amount up to \$3,000,000, in each case, depending on the termination event.

#### Related Agreements (see page 148)

Voting Agreement

Concurrently with the execution and delivery of the Merger Agreement, certain USWS Stockholders, including THRC Holdings (each, a 'USWS Supporting Stockholder" and, collectively, the "USWS Supporting Stockholders"), entered into the Voting Agreement with ProFrac. Pursuant to the Voting Agreement, each USWS Supporting Stockholder agreed to, among other things, support and vote in favor of the Merger Agreement, the Merger and the other transactions contemplated by the Merger Agreement. At the close of business on [ ], 2022, the record date for the USWS Special Meeting, the USWS Supporting Stockholders beneficially owned [ ] shares of USWS Common Stock, or approximately [ ]% of the shares of USWS Common Stock outstanding on that date.

Warrant Purchase Agreement

Concurrently with the execution and delivery of the Merger Agreement, the holders of the Term C Loan Warrants (collectively, the 'Term C Loan Warrant Holders'), including Crestview Partners and David Matlin, entered into a Warrant Purchase Agreement. Pursuant to the Warrant Purchase Agreement, immediately prior to the Effective Time and conditioned upon the consummation of the Merger, the Term C Loan Warrant Holders will sell all of their respective Term C Loan Warrants to ProFrac or its subsidiaries in exchange for an aggregate purchase price of \$2,639,999.82.

#### Amendment to Certificate of Designations

Under the terms of the Merger Agreement, USWS is obligated to take all requisite actions so that, effective as of immediately prior to the Effective Time, each holder of shares of USWS Series A Preferred Stock issued and outstanding at such time would have the option to convert such shares into shares of USWS Common Stock. USWS will take all requisite action to cause the Certificate of Designations to be amended prior to the Closing by adoption of an amendment to the Certificate of Designations.

### Amendments to USWS Equity Linked Convertible Notes

Under the terms of the Merger Agreement, USWS also agreed to cause the USWS Equity Linked Convertible Notes to be amended by execution of amendments thereto (the "Convertible Note Amendments"). On June 21, 2022, each of the Convertible Note Amendments was executed by USWS and the respective payees of the USWS Equity Linked Convertible Notes and became effective, including Convertible Note Amendments with respect to the USWS Equity Linked Convertible Notes held by Crestview Partners and by THRC Holdings, one of the Wilks Parties. The Convertible Note Amendments amended the terms of the Equity Linked Convertible Notes in order to provide for, among other things, the conversion by USWS, pursuant to the terms of the Merger Agreement and as of immediately prior to the Effective Time (any such conversion, a "Note Merger Conversion"), of all of the outstanding principal and interest then owing under each USWS Equity Linked Convertible Note into a number of shares of USWS Common Stock equal to the quotient obtained by dividing (i) the amount of such outstanding principal and interest owing through July 9, 2022, by (ii) \$7.32.

#### Amendments to Rollover Warrants

Prior to the Effective Time, ProFrac will execute an amendment to each of the warrant agreements that govern the USWS Rollover Warrants, as necessary to assume each such warrant as of the Effective Time in accordance with the terms thereof and the Merger Agreement. Each USWS Rollover Warrant shall remain outstanding following the consummation of the Merger through its prescribed term, but shall become the right to receive upon valid exercise thereof shares of ProFrac Class A Common Stock (rather than shares of USWS Common Stock). For more information about the USWS Rollover Warrants, see the section titled "The Merger Agreement — Treatment of USWS Warrants — Rollover Warrants"

#### Board of Directors and Management of USWS Following Completion of the Merger (see page 125)

Following completion of the Merger, USWS will be an indirect subsidiary of ProFrac. The officers and directors of Merger Sub immediately prior to the Effective Time shall be the officers and directors of USWS following the consummation of the Merger. The business and affairs of USWS will be managed by or under the direction of its board of directors. It is expected that the sole director and executive officers of USWS following completion of the Merger will be as follows:

Name	Age	Title
Ladd Wilks	36	Chief Executive Officer
Lance Turner	42	Chief Financial Officer
Coy Randle	60	Chief Operations Officer
Robert Willette	46	Chief Legal Officer
Matthew D. Wilks	39	Director

### Board of Directors and Executive Officers of ProFrac Following Completion of the Merger (see page 125)

Following completion of the Merger, the current directors and executive officers of ProFrac are expected to continue in their current positions, other than for changes that may be publicly announced by ProFrac in the future in the normal course.

#### Special Meeting of USWS Stockholders

#### USWS Special Meeting (see page 150)

Meeting. The USWS Special Meeting will be held on [ ], 2022, at [ ] a.m., local time, at www.proxydocs.com/USWS. At the USWS Special Meeting, USWS Stockholders will be asked:

- 1. to approve the USWS Merger Proposal;
- to approve, for purposes of complying with the Nasdaq listing rules, the issuance of shares of USWS Common Stock to be issued by USWS
  upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, in an amount equal to 20% or
  more of USWS Common Stock outstanding;
- to approve an amendment to the Certificate of Designations to modify certain terms relating to the conversion rights of the USWS Series A
  Preferred Stock:
- to approve an amendment to the A&R LTIP, to increase the maximum number of shares of USWS Common Stock that may be issued under the A&R LTIP by 2,000,000 shares; and
- 5. if submitted to a vote of USWS Stockholders, to approve an adjournment of the USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of the adoption of the Merger Agreement if there are not sufficient votes for such proposal.

Record Date; Votes. USWS has fixed the close of business on [ ], 2022 as the record date, which is referred to as the USWS record date, for determining the USWS Stockholders entitled to receive notice of and to vote at the USWS Special Meeting. Only holders of record of USWS Common Stock on the USWS record date are entitled to receive notice of and vote at the USWS Special Meeting, and any adjournment or postponement thereof.

Each share of USWS Common Stock is entitled to one vote. On the USWS record date, there were [ ] shares of USWS Common Stock entitled to vote at the USWS Special Meeting.

Required Vote. The proposals require different percentages of votes in order to approve them:

- the proposal for the approval of the USWS Merger Proposal, requires the affirmative vote of a majority of the outstanding shares of USWS Common Stock entitled to vote at the USWS Special Meeting;
- the proposal to approve the issuance of shares of USWS Common Stock to be issued by USWS upon the conversion of the USWS Series A
  Preferred Stock and the USWS Equity Linked Convertible Notes requires the affirmative vote of the holders of a majority of the total votes
  cast at the USWS Special Meeting;
- the proposal to approve the amendment to the Certificate of Designations requires the affirmative vote of a majority of the outstanding shares of USWS Common Stock entitled to vote at the USWS Special Meeting;
- the proposal to approve the amendment to the A&R LTIP requires the affirmative vote of a majority of the shares of voting stock present in person or represented by proxy at the meeting and entitled to vote; and
- the proposal to adjourn the USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of the proposal to adopt the
  Merger Agreement, requires the affirmative vote of a majority of the votes cast by stockholders present in person or represented by proxy,
  whether or not a quorum is present, at the USWS Special Meeting.

Adoption of the Merger Agreement and the transactions contemplated thereby, including the Merger, by USWS Stockholders is a condition to the completion of the Merger.

Stock Ownership of Directors and Executive Officers. As of the close of business on the USWS record date, directors and executive officers of USWS and their respective affiliates owned and were entitled to vote [ ] shares of USWS Common Stock, or approximately [ ]% of the shares of USWS Common Stock outstanding on that date. To the knowledge of USWS, the directors and executive officers of USWS and their respective affiliates intend to vote all of their shares of USWS Common Stock they are entitled to vote in favor of all USWS proposals at the USWS Special Meeting. Two members of the USWS Board, Joel Broussard and David Matlin, are parties to the Voting Agreement with ProFrac.

#### Selected Historical and Pro Forma Consolidated Financial Data of ProFrac

Unless the context otherwise requires, all references in this section to "ProFrac," "we," "us," or "our" refer to ProFrac Holding Corp. together with its consolidated subsidiaries.

The following table presents selected historical consolidated financial data of ProFrac Predecessor and the unaudited pro forma financial data of ProFrac as of the dates and for the periods indicated. The selected historical consolidated financial data as of and for the years ended December 31, 2021 and 2020 was derived from the audited financial statements appearing elsewhere in this proxy statement/information statement/prospectus. The summary historical consolidated financial data as of and for the six months ended June 30, 2022 and 2021 was derived from the unaudited historical financial statements included elsewhere in this proxy statement/information statement/prospectus. The unaudited historical financial statements were prepared on a basis consistent with that used in preparing our audited consolidated financial statements and include all adjustments, consisting of normal and recurring items, that we consider necessary for a fair presentation of our financial position and results of operations for the unaudited periods. The unaudited pro forma financial data was derived from the unaudited pro forma financial statements included elsewhere in this proxy statement/information statement/prospectus. Historical results are not necessarily indicative of future results.

The summary unaudited pro forma balance sheet data as of June 30, 2022 has been prepared to give pro forma effect to the anticipated acquisition of USWS, as if it had occurred on June 30, 2022. The summary unaudited pro forma statement data of operations for the six months ended June 30, 2022 and year ended December 31, 2021 have been prepared to give pro forma effect to (i) the expansion of ProFrac LLC's term loan credit facility (the "Old Term Loan Credit Facility") and related purchase of all the series A-1 and series B-1 preferred units in Basin Production and Completion LLC ("BPC" and, such repurchases collectively, the "BPC Acquisition"), (ii) the entry into the New Term Loan Credit Facility and the application of borrowings thereunder to fund a portion of the purchase price in the FTSI Acquisition and associated expenses and to repay in full the Old Term Loan Credit Facility, (iii) the issuance of subordinated debt to THRC Holdings and Equify Financial, LLC ("Equify Financial"), the proceeds of which were used to fund a portion of the purchase price in the FTSI Acquisition, (iv) the completion of the FTSI Acquisition, (v) the Corporate Reorganization described under "Information About ProFrac", (vi) the IPO and the application of the net proceeds therefrom, and (vii) the anticipated acquisition of USWS as if they had occurred on January 1, 2021.

This information is subject to and gives effect to the assumptions and adjustments described in the notes accompanying the unaudited pro forma financial statements included elsewhere in this proxy statement/information statement/prospectus. The unaudited pro forma financial data is presented for informational purposes only, should not be considered indicative of actual results of operations that would have been achieved had such transactions been consummated on the dates indicated and does not purport to be indicative of statements of financial position or results of operations as of any future date or for any future period.

The summary historical consolidated and combined and unaudited pro forma financial data presented below should be read in conjunction with "Risk Factors," "Management's Discussion and Analysis of Financial Condition and Results of Operations of ProFrac" and the consolidated financial statements of ProFrac Predecessor and the related notes and the pro forma financial statements of ProFrac and the related notes and other financial data included elsewhere in this proxy statement/information statement/prospectus. Among other things, the historical and pro forma financial statements include more detailed information regarding the basis of presentation for the information in the following tables.

	_	ProFrac Historical		
		Six months ended June 30,		Six months ended June 30,
Statement of Operations Data:	_	2022	2021	2022
Total revenues	\$	934,824	\$324,405	\$ 1,121,373
Total cost of revenues, exclusive of depreciation, depletion and amortization	Ψ	573,199	245,014	728,810
Depreciation, depletion and amortization		108,280	70,365	144,017
Loss on disposal of assets, net		1,989	4,075	4,964
Selling, general and administrative		121.675	27,872	164,832
Interest expense, net		22,723	12,222	32,561
Loss on extinguishment of debt		17,095	12,222	18,746
Other income		(9,220)	(240)	(11,674)
Reorganization items, net		(9,220)	(240)	(129)
Income tax provision (benefit)		4,864	(308)	4,864
Net income (loss)		94,219	(34,595)	34,382
Less: net (income) loss attributable to ProFrac Predecessor		(79,867)	34,476	34,362
Less: net (income) loss attributable to Prorrac Fredecessor  Less: net loss attributable to noncontrolling interests		8,288	119	8,288
Less: net ross autroutable to noncontrolling interests  Less: net income attributable to redeemable noncontrolling interests		(16,082)	119	(27,833)
Ç				
Net income attributable to ProFrac Holding Corp.	\$	6,558	\$ —	\$ 14,837
Pro Forma Per Share Information:				
Net loss per Class A share:				
Basic	\$			\$ 0.27
Diluted		0.16		0.27
Weighted average Class A shares outstanding:				
Basic		40,123		54,016
Diluted		40,145		54,038
Balance Sheet Data (as of end of period):				
Cash and equivalents	\$	,		\$ 69,548
Property, plant and equipment, net		664,245		904,387
Total assets		1,663,625		2,195,538
Total long-term debt		427,961		599,289
Redeemable noncontrolling interests		2,024,687		2,024,687
Total stockholders' deficit	(	(1,316,530)		(1,066,373)
Cash Flow Statement Data:				
Net cash provided by operating activities	\$	,	\$ 26,780	
Net cash used in investing activities		(388,330)	(38,451)	
Net cash provided by financing activities		374,244	26,101	
Other Data:				
Adjusted EBITDA(1)	\$	,	\$ 51,759	\$ 313,956
Capital expenditures		116,069	53,607	178,806

		ProFrac Historical	
		ended	Year ended
	2021	ber 31, 2020	December 31, 2021
Statement of Operations Data:			
Total revenues	\$768,353	\$ 547,679	\$ 1,424,066
Total cost of revenues, exclusive of depreciation, depletion, and amortization	570,122	432,570	1,108,233
Depreciation, depletion, and amortization	140,687	150,662	273,447
Loss (gain) on disposal of assets, net	9,777	8,447	(9,924)
Selling, general and administrative	65,592	51,014	166,393
Litigation settlement	_	_	35,000
Interest expense, net	25,788	23,276	58,734
Loss on extinguishment of debt, net	515	_	29,182
Reorganization items, net	_	_	894
Other expense (income)	(404)	(324)	1,233
Income tax (benefit) provision	(186)	582	(143)
Net loss	(43,538)	(118,548)	(238,983)
Less: net loss attributable to ProFrac Predecessor	42,420	117,405	(== 0,5 00)
Less: net loss attributable to noncontrolling interests	1,118	1,143	1,162
Less: net loss attributable to redeemable noncontrolling interests	_		155, 125
Net loss attributable to ProFrac Holding Corp.	<u> </u>	<u> </u>	\$ (82,696)
Pro Forma Per Share Information:	Ψ	Ψ	\$ (02,070)
Net loss per Class A share:			
Basic			\$ (1.53)
Diluted			(1.53)
Weighted average Class A shares outstanding:			(1.55)
Basic			54.016
Diluted			54,016
Balance Sheet Data (as of end of period):			5 1,010
Cash and equivalents	\$ 5,376	\$ 2,952	
Property, plant and equipment, net	363,687	429,684	
Total assets	664,570	577,277	
Total long-term debt	269,773	260,229	
Total equity	148,110	176,812	
Cash Flow Statement Data:	- 10,111		
Net cash provided by operating activities	\$ 43,942	\$ 45,054	
Net cash used in investing activities	(78,383)	(44,617)	
Net cash provided by (used in) financing activities	36,865	(15,322)	
Other Data:	2 3,000	(,-==)	
Adjusted EBITDA(1)	\$134,688	\$ 72,797	\$ 187,753
Capital expenditures	87,400	48,037	166,524
1	,	.,,	, . = .

<sup>(1)</sup> Adjusted EBITDA is a Non-GAAP Financial Measure. See the tables entitled "Reconciliation and Calculation of Non-GAAP Financial and Operational Measures" below. For definitions of the non-GAAP financial measures of Adjusted EBITDA, see "Note regarding Non-GAAP Financial Measures" in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations of ProFrac."

# Reconciliation and Calculation of Non-GAAP Financial and Operational Measures

Reconciliation of net income (loss) to Adjusted EBITDA and Adjusted EBITDA per fleet for the six months ended June 30, 2022 and 2021:

		ProFrac Historical	
	Six month	hs ended	Pro Forma Six months ended June 30,
	2022	2021	2022
Net income (loss)	\$ 94,219	\$(34,595)	\$ 34,382
Interest expense, net	22,723	12,222	32,561
Income tax (benefit) provision	4,864	(308)	4,864
Depreciation, depletion, and amortization	108,280	70,365	144,017
Loss on disposal of assets, net	1,989	4,075	4,964
Loss on extinguishment of debt	17,095	_	18,746
Bad debt expense, net of recoveries	5	_	5
Severance charges		_	458
Reorganization cost	55	_	(74)
Gain on foreign currency transactions	(46)	_	(46)
Litigation accrual	4,000	_	4,000
Stock compensation expense	1,455		9,563
Stock compensation expense related to deemed contribution	38,849	_	38,849
Change in fair value of warrant liabilities	_	_	(831)
Fleet laydown and reactivation costs	_	_	2,265
Replacement of damaged equipment	_	_	422
Unrealized gain on investments, net	(8,526)	_	(8,526)
Transaction and related costs	17,082	_	28,337
Adjusted EBITDA	\$302,044	\$ 51,759	\$ 313,956
Active fleets	26	14	36
Adjusted EBITDA per fleet	\$ 11,485	\$ 3,620	\$ 8,625
Net income (loss) per fleet	3,582	(2,419)	945

Reconciliation of net loss to Adjusted EBITDA and Adjusted EBITDA per fleet for the year ended December 31, 2021 and 2020:

	ProFrac Historical Year ended		Historical I		ProFrac Pro Forma Year ended
	Decem	ber 31,	December 31,		
	2021	2020	2021		
Net loss	\$ (43,538)	\$(118,548)	\$ (238,983)		
Interest expense, net	25,788	23,276	58,734		
Income tax (benefit) provision	(186)	582	(143)		
Depreciation, depletion, and amortization.	140,687	150,662	273,447		
Loss (gain) on disposal of assets, net	9,777	8,447	(9,924)		
Loss on extinguishment of debt	515	_	29,182		
Bad debt expense, net of recoveries	(1,164)	2,778	(1,012)		
Severance charges	500	_	2,326		
Reorganization cost	2,060	_	2,954		
Supply commitment charges	_	5,600	_		
Loss (gain) on foreign currency transactions	249	_	249		
Litigation settlement	_	_	35,000		
Stock compensation expense	_	_	12,817		
Change in fair value of warrant liabilities	_	_	2,152		
Fleet laydown and reactivation costs	_	_	6,578		
Sales and use tax audit expense	_	_	206		
Transaction and related costs			14,170		
Adjusted EBITDA	<u>\$134,688</u>	\$ 72,797	\$ 187,753		
Active fleets	14	11	35		
Adjusted EBITDA per fleet	\$ 9,621	\$ 6,618	\$ 5,380		
Net loss per fleet	(3,110)	(10,777)	(6,848)		

#### Selected Historical Consolidated Financial Data of USWS

The following table presents selected historical consolidated financial data of USWS as of the dates and for the periods indicated. The selected historical consolidated financial data as of and for the years ended December 31, 2021 and 2020 was derived from the audited financial statements appearing elsewhere in this proxy statement/information statement/prospectus. The summary historical consolidated financial data as of and for the six months ended June 30, 2022 and 2021 was derived from the unaudited historical financial statements included elsewhere in this proxy statement/information statement/prospectus. The unaudited historical financial statements were prepared on a basis consistent with that used in preparing our audited consolidated financial statements and include all adjustments, consisting of normal and recurring items, that we consider necessary for a fair presentation of our financial position and results of operations for the unaudited periods. Historical results are not necessarily indicative of future results.

The selected historical consolidated financial data presented below should be read in conjunction with 'Risk Factors,' "Management's Discussion and Analysis of Financial Condition and Results of Operations of USWS" and the consolidated financial statements of USWS and the related notes and other financial data included elsewhere in this proxy statement/information statement/prospectus. Among other things, the historical financial statements include more information regarding the basis of presentation for the information in the following table.

### U.S. WELL SERVICES, INC. Selected Historical Consolidated Financial Data

(in thousands, except for per share amounts) (unaudited)

	Six Montl June		Year Ended December 31,		
	2022	2021	2021	2020	
Statement of Operations Data:					
Revenue	\$ 109,914	\$ 155,057	\$ 250,463	\$ 244,007	
Costs and expenses:					
Cost of services (excluding depreciation and amortization)	95,932	121,883	221,364	187,803	
Depreciation and amortization	11,577	20,942	35,444	80,353	
Selling, general and administrative expenses	17,778	14,604	32,578	43,632	
Impairment of long-lived assets	_	_	_	147,543	
Litigation settlement	_	35,000	35,000	_	
Loss (gain) on disposal of assets	2,980	1,891	(21,896)	7,112	
Interest expense, net	(17,530)	(13,516)	(33,370)	(25,226)	
Change in fair value of warrant liabilities	831	(7,287)	(2,152)	6,342	
Patent license sales	_	22,500	22,500	_	
Loss on extinguishment of debt, net	(1,651)	(839)	(6,142)	_	
Other income	1,623	52	515	108	
Income tax benefit		(27)	(27)	(824)	
Net loss	(35,080)	(38,326)	(70,649)	(240,388)	
Net loss attributable to noncontrolling interest		(44)	(44)	(11,048)	
Net loss attributable to U.S. Well Services, Inc.	\$ (35,080)	\$ (38,282)	\$ (70,605)	\$ (229,340)	

		Six Months Ended June 30,		Ended ber 31,
	2022	2021	2021	2020
Net loss attributable to U.S. Well Services, Inc. stockholders pe	r common share:			
Basic and diluted (1)	\$ (3.27)	\$ (10.12)	\$ (14.64)	\$ (79.94)
Weighted average common shares outstanding:				
Basic and diluted (1)	11,347	3,991	5,399	3,085
Balance Sheet Data (as of end of period):				
Cash and cash equivalents	\$ 17,268	\$ 57,544	\$ 6,384	\$ 3,693
Property and equipment, net	\$ 194,943	\$ 213,301	\$ 162,664	\$ 235,332
Total assets	\$ 298,665	\$ 379,630	\$ 243,557	\$ 323,223
Convertible senior notes	\$ 116,183	\$ 85,677	\$ 105,769	\$ —
Total long-term debt	\$ 173,938	\$ 304,180	\$ 181,047	\$ 297,421
Total mezzanine equity	\$ 26,092	\$ 44,961	\$ 23,866	\$ 73,661
Total equity (deficit)	\$ (116,047)	\$ (123,345)	\$ (129,103)	\$ (105,212)
Cash Flow Statement Data:				
Net cash provided by (used in) operating activities	\$ 3,559	\$ (28,371)	\$ (19,277)	\$ 8,616
Net cash provided by (used in) investing activities	\$ (30,999)	\$ (16,288)	\$ 56,156	\$ (34,999)
Net cash provided by (used in) financing activities	\$ 36,324	\$ 97,460	\$ (33,021)	\$ (9,759)
Other Financial and Operational Data:				
Capital Expenditures	\$ 48,337	\$ 24,841	\$ 57,724	\$ 55,943
Adjusted EBITDA (2)	\$ 3,950	\$ 48,377	\$ 39,989	\$ 31,146

Periods presented have been adjusted to reflect the 1-for-3.5 reverse stock split on September 30, 2021 and the 1-for-6 reverse stock split on August 4, 2022.
 Adjusted EBITDA is a Non-GAAP Financial Measure. See the tables entitled "Reconciliation and Calculation of Non-GAAP Financial and

<sup>(2)</sup> Adjusted EBITDA is a Non-GAAP Financial Measure. See the tables entitled "Reconciliation and Calculation of Non-GAAP Financial and Operational Measures" below.

# U.S. WELL SERVICES, INC.

# RECONCILIATION OF NET LOSS (GAAP) TO EBITDA AND ADJUSTED EBITDA (NON-GAAP)

(in thousands)

(unaudited)

		Six Months Ended June 30,		Ended iber 31,
	2022	2021	2021	2020
Net loss	\$ (35,080)	\$ (38,326)	\$ (70,649)	\$ (240,388)
Interest expense, net	17,530	13,516	33,370	25,226
Income tax benefit	_	(27)	(27)	(824)
Depreciation and amortization	11,577	20,942	35,444	80,353
EBITDA	(5,973)	(3,895)	(1,862)	(135,633)
Impairment loss	` <u></u>		<u> </u>	147,543
Litigation settlement	_	35,000	35,000	_
Loss (gain) on disposal of assets	2,980	1,891	(21,896)	7,112
Change in fair value of warrant liabilities	(831)	7,287	2,152	(6,342)
Loss on extinguishment of debt, net	1,651	839	6,142	_
Share-based compensation	1,613	3,661	11,694	10,056
Fleet start-up, laydown and reactivation costs	2,265	2,301	6,185	3,033
Severance, business restructuring, and market-driven costs	458	1,144	1,826	5,377
Transaction related costs	1,365	149	149	_
Replacement of damaged equipment	243	_	_	_
Non-recurring labor and mobilization costs	_	_	393	_
Other	179	_	206	_
Adjusted EBITDA	3,950	48,377	39,989	31,146
Patent license sales		(22,500)	(22,500)	
Adjusted EBITDA, excluding patent license sales	\$ 3,950	\$ 25,877	\$ 17,489	\$ 31,146

#### **Unaudited Comparative Per Share Data**

The following table presents ProFrac's and USWS' historical and pro forma per share data as of and for the six months ended June 30, 2022, and as of and for the year ended December 31, 2021. The pro forma information is provided as if the Merger had been completed on January 1, 2021. The information in the table below has been derived from and should be read in conjunction with the historical consolidated financial statements of ProFrac and USWS included elsewhere in this proxy statement/information statement/prospectus.

Unaudited pro forma combined per share data as of and for the six months ended June 30, 2022, and as of and for the year ended December 31, 2021, were derived and should be read in conjunction with the unaudited pro forma combined financial data included in "Unaudited Pro Forma Condensed Combined Financial Information" and the related notes thereto in this proxy statement/information statement/prospectus. The pro forma information is presented for illustrative purposes only and is not necessarily indicative of the operating results that would have occurred if the Merger had been completed as of the beginning of the period or the combined financial position of the companies as of June 30, 2022.

	Six Months  Ended June 30,  2022			ar Ended ember 31, 2021
		(unaudit	ed)	
ProFrac Historical Data				
Basic and diluted net income (loss) per share	\$	0.16		(1)
Book value per share	\$	4.32(2)		(1)
USWS Historical Data				
Basic and diluted net income (loss) per share	\$	(3.27)	\$	(14.64)
Book value per share	\$	(9.05)	\$	(14.57)
ProFrac Pro Forma Combined Data				
Basic and diluted net income (loss) per share	\$	0.27	\$	(1.53)
Book value per share	\$	5.58(2)		(3)

- (1) No net income or book value attributable to ProFrac Holding Corp. for year ended December 31, 2021 because the period precedes the issuance of shares of Class A Common Stock as a result of the ProFrac IPO on May 17, 2022.
- (2) The book value per share is calculated on an as converted basis assuming each share of Class B Common Stock is exchanged for one share of Class A Common Stock.
- (3) The proforma balance sheet has been prepared as of the most recently completed interim period filed pursuant to Regulation S-X, Rule 3-01. As such, no proforma balance sheet has been prepared as of December 31, 2021.

#### **Summary of Risk Factors**

There are a number of risks that USWS Stockholders should consider in connection with the Merger Agreement and the transactions contemplated thereunder. These risks are discussed more fully in the section titled "Risk Factors." Any of these risks could materially adversely affect the business, financial condition and results of operation of ProFrac, USWS or the combined company, and the actual outcome of matters as to which forward-looking statements are made in this proxy statement/information statement/prospectus. You should carefully consider all of the risk factors together with all of the other information included in, or incorporated by reference into, this proxy statement/information statement/prospectus. These risks include, but are not limited to, the following:

- Because the market price of ProFrac Class A Common Stock may fluctuate, and the Closing Date of the Merger is not yet ascertainable, USWS Stockholders cannot be certain of the precise value of the Merger Consideration they will receive in the Merger.
- · The combined company will incur significant transaction and Merger-related costs in connection with the Merger.

- Whether or not the Merger is completed, the announcement and pendency of the Merger could impact or cause disruptions in USWS' and ProFrac's businesses, which could have an adverse effect on their results of operations and financial condition.
- Certain directors and executive officers of ProFrac and USWS have interests that may be different from, or in addition to, interests of ProFrac and USWS Stockholders generally.
- The Merger could have a dilutive effect on ProFrac's earnings per share calculated in accordance with GAAP, which may adversely affect the market price of ProFrac Class A Common Stock following the Merger.
- ProFrac may be required to incur significant additional indebtedness in order to finance the Merger, which will limit the combined company's operating flexibility, and could adversely affect the combined company's operations and financial results and prevent the combined company from fulfilling its obligations.
- The combined company may not be able to generate sufficient cash flow to service all of its obligations, including its obligations under its new credit facilities.
- ProFrac's business and financial performance depends on the oil and natural gas industry and particularly on the level of capital spending and E&P activity within the United States and in the basins in which ProFrac operates, and a decline in prices for oil and natural gas may have an adverse effect on ProFrac's revenue, cash flows, profitability and growth.
- ProFrac faces significant competition that may cause a loss of market share.
- Reliance upon a few large customers may adversely affect ProFrac's revenue and operating results.
- A negative shift in investor sentiment of the oil and gas industry has had and could in the future have adverse effects on ProFrac's customers' operations and ability to raise debt and equity capital.
- ProFrac's operations require substantial capital and ProFrac may be unable to obtain needed capital or financing on satisfactory terms or at all, which could limit ProFrac's ability to grow.
- The growth of ProFrac's business through recently completed acquisitions and potential future acquisitions may expose ProFrac to various
  risks, including those relating to difficulties in identifying suitable, accretive acquisition opportunities and integrating businesses, assets and
  personnel, as well as difficulties in obtaining financing for targeted acquisitions and the potential for increased leverage or debt service
  requirements.
- ProFrac may not be able to generate sufficient cash flow to service all of its obligations, including obligations under its recently amended credit and other financing facilities.
- ProFrac's indebtedness and liquidity needs could restrict ProFrac's operations and make ProFrac more vulnerable to adverse economic conditions.
- Conflicts of interest could arise in the future between ProFrac, on the one hand, and Dan Wilks and Farris Wilks and entities owned by or
  affiliated with them, on the other hand, concerning among other things, business transactions, potential competitive business activities or
  business opportunities.
- An active, liquid and orderly trading market for ProFrac Class A common stock may not develop or be maintained, and ProFrac's stock price
  may be volatile.
- The unaudited pro forma financial data included in this proxy statement/information statement/ prospectus may not be representative of ProFrac's actual financial condition and results of operations in the future.
- The ProFrac Controlling Stockholders have the ability to direct the voting of a majority of ProFrac's voting stock, and their interests may conflict with those of ProFrac's other stockholders.

- USWS' transition from the diesel pressure pumping market has negatively impacted and may continue to negatively impact its liquidity and its ability to generate revenues and service its outstanding indebtedness for a period of time.
- USWS' business depends on the level of capital spending and E&P activity by the onshore oil and natural gas industry in the United States, and the level of such activity is affected by industry conditions that are beyond USWS' control.
- USWS' level of current and future indebtedness could adversely affect USWS' financial condition.
- USWS' debt financing agreements subject it to financial and other restrictive covenants. These restrictions may limit USWS' operational or financial flexibility and could subject USWS to potential defaults under its credit facilities.
- · If USWS is unable to fully protect its intellectual property rights, USWS may suffer a loss in its competitive advantage or market share.
- USWS' ability to expand its operations relies in part on USWS' ability to market its Clean Fleet® technology, and advancements in well service technologies could have a material adverse effect on USWS' business, financial condition and results of operations.

#### COMPARATIVE PER SHARE MARKET PRICE AND DIVIDEND INFORMATION

### **Market Price**

Shares of ProFrac Class A Common Stock are listed on Nasdaq under the symbol "PFHC," and shares of USWS Common Stock are listed on Nasdaq under the symbol "USWS." The following table presents the closing prices of ProFrac Class A Common Stock and USWS Common Stock on (i) June 21, 2022, the last trading day before the public announcement of the Merger, and (ii) August 29, 2022, the latest practicable trading day before the date of this proxy statement/information statement/prospectus. The table also shows the estimated implied value of the per share Merger Consideration for each share of USWS Common Stock for each of the same two days, which per share value is calculated as the product of (i) the applicable ProFrac per share value and (ii) the exchange ratio of 0.3366 of a share of ProFrac Class A Common Stock per each share of USWS Common Stock, rounded to the nearest cent.

	ProFrac C	ass A				lied per Value of	
	Common	Common Stock US		USWS Common		Merger	
Date	Closing P	Closing Price		osing Price	Consideration		
June 21, 2022	\$	21.49	\$	4.32(1)	\$	7.26(1)	
August 29, 2022	\$	21.55	\$	6.96	\$	7.25	

(1) The USWS Common Stock Closing Price and the Implied per Share Value of Merger Consideration for June 21, 2022 has been adjusted to account for the 1-for-6 reverse stock split implemented by USWS on August 4, 2022.

The above table shows only a historical comparison. This comparison may not provide meaningful information to USWS Stockholders in connection with making decisions with respect to the Merger. In addition, because the exchange ratio will not be adjusted for changes in the market price of either ProFrac Class A Common Stock or USWS Common Stock, the market value of ProFrac Class A Common Stock that USWS Stockholders will have the right to receive on the date the Merger is completed may vary significantly from the market value of the ProFrac Class A Common Stock that USWS Stockholders would receive if the Merger was completed on the date of this proxy statement/information statement/prospectus. USWS Stockholders are urged to obtain current market information for shares of ProFrac Class A Common Stock and USWS Common Stock and to review carefully the other information contained in this proxy statement/information statement/prospectus or incorporated herein by reference in making any decisions with respect to the Merger. See "Where You Can Find More Information" for instructions on how to obtain the information that has been incorporated by reference. Historical performance is not necessarily indicative of any performance to be expected in the future. See also "Risk Factors" and "Cautionary Statement Regarding Forward-Looking Statements."

#### Dividend and Share Data - ProFrac

Holders of shares of ProFrac Class A Common Stock are entitled to ratably receive dividends when and if declared by the ProFrac Board out of funds legally available for that purpose, subject to any statutory or contractual restrictions on the payment of dividends and to any prior rights and preferences that may be applicable to any outstanding preferred stock. ProFrac has not paid a regular dividend on the ProFrac Class A Common Stock historically and has no present intention of instituting a regular dividend.

As of August 8, 2022, there were 41,239,957 shares of ProFrac Class A Common Stock outstanding, and there were 3 holders of record of ProFrac Class A Common Stock.

## $Dividend\ and\ Share\ Data-USWS$

Holders of shares of USWS Common Stock are entitled to receive dividends from USWS when, as and if declared by the USWS Board, subject to the consent of the holders of shares of USWS Series A Preferred Stock.

USWS has not paid any dividends since its inception and it does not intend to pay regular cash dividends in the foreseeable future.

As of August 15, 2022, there were approximately 12,851,347 shares of USWS Common Stock outstanding, and there were 59 holders of record of USWS Common Stock.

#### RISK FACTORS

In addition to the other information contained in or incorporated by reference into this proxy statement/information statement/prospectus, including the matters addressed under the caption "Cautionary Statement Regarding Forward-Looking Statements," USWS Stockholders should carefully consider the following risk factors in deciding whether to approve the Merger. You should read this proxy statement/information statement/prospectus and its Annexes and the other documents incorporated by reference into this proxy statement/information statement/prospectus carefully and in their entirety. See "Where You Can Find More Information."

#### Risks Related to the Merger

Because the market price of ProFrac Class A Common Stock may fluctuate, and the Closing Date of the Merger is not yet ascertainable, USWS Stockholders cannot be certain of the precise value of the Merger Consideration they will receive in the Merger.

Upon completion of the Merger, each share of USWS Common Stock issued and outstanding immediately prior to the Merger will be cancelled and converted into the right to receive Merger Consideration equal to 0.3366 of a share of ProFrac Class A Common Stock. The Merger Consideration is fixed and will not be adjusted due to any increases or decreases in the price of ProFrac Class A Common Stock or USWS Common Stock. However, the value of ProFrac Class A Common Stock to be received by USWS Stockholders in the Merger will depend upon the market price of a share of ProFrac Class A Common Stock upon the completion of the Merger. Therefore, if the price of ProFrac Class A Common Stock declines, USWS Stockholders will receive less value for their shares upon the completion of the Merger than the implied value of the Merger Consideration calculated on the date the Merger Agreement was signed or on the date of the USWS Special Meeting.

The Merger may not be completed until a period of time after the USWS Special Meeting, during which time the market value of ProFrac Class A Common Stock and USWS Common Stock will likely fluctuate. Therefore, at the time of the USWS Special Meeting, USWS Stockholders will not know the exact market value of ProFrac Class A Common Stock that will be issued in connection with the Merger. The market price of a share of ProFrac Class A Common Stock at the time the Merger is completed is likely to be different, and may be lower, than it was at the time the Merger Agreement was signed and at the time of the USWS Special Meeting. The closing price of ProFrac Class A Common Stock on the Nasdaq Global Select Market on June 21, 2022, the last trading day prior to announcement of the Merger Agreement, was \$21.49 per share. From June 21, 2022 through the date of this proxy statement/information statement/prospectus, the closing trade price of ProFrac Class A Common Stock ranged from a high of \$21.54 per share to a low of \$14.25 per share.

Stock price changes may result from a variety of factors, including, but not limited to:

- changes in the business, operations or prospects of ProFrac or USWS;
- · catastrophic events, both natural and man-made;
- · governmental, litigation and regulatory developments or considerations;
- quarterly fluctuations in either of the companies' actual or anticipated operating results;
- · announcements of technological innovations;
- new products or product enhancements by either of the companies or their respective competitors;
- · developments in patents or other intellectual property rights and litigation;
- · developments in relationships with either of the companies' customers and suppliers;
- interest rates;
- general market and economic conditions;

- market assessments as to whether and when the Merger will be completed and market assessments of the condition, results or prospects of either company's business;
- epidemics, pandemics, or other major public health issues, such as the COVID-19 pandemic;
- wars, such as the conflict between Russia and Ukraine, and their potential impacts on global markets; and
- changes in reimbursement rates.

USWS Stockholders are urged to obtain current market prices for ProFrac Class A Common Stock and USWS Common Stock when they consider whether to approve the proposals required to complete the Merger at the USWS Special Meeting.

#### Failure to complete the Merger could negatively impact the stock prices and the future business and financial results of ProFrac and USWS.

If the Merger is not completed, the ongoing businesses of ProFrac or USWS may be adversely affected and ProFrac and USWS will be subject to several risks, including the following:

- USWS being required, upon termination of the Merger Agreement under specified circumstances, to pay to ProFrac a termination fee of \$8,000,000 or to reimburse ProFrac for certain expenses in an amount up to \$3,000,000, in each case, depending on the termination event;
- · having to pay significant costs relating to the Merger, such as legal, accounting, financial advisor and printing fees;
- the attention of management of ProFrac and USWS will have been diverted to the Merger instead of on such company's own operations and pursuit of other opportunities that could have been beneficial to such company; and
- customer perception may be negatively impacted which could affect the ability of ProFrac and USWS to compete for, or to win, new and renewal business in the marketplace.

Consummation of the Merger is contingent upon the satisfaction of a number of conditions, including approval by USWS Stockholders, which may not be satisfied or completed on a timely basis, if at all. Failure to complete this transaction could negatively impact ProFrac's and USWS' stock price, business, financial condition, results of operations or prospects.

Consummation of the Merger is subject to conditions to Closing that are not wholly within the control of ProFrac and USWS, including, among other things, adoption of the Merger Agreement by the holders of a majority of the outstanding shares of USWS Common Stock entitled to vote on the adoption of the Merger Agreement. There can be no assurance that each of the conditions with respect to the Merger will be satisfied or waived in a timely manner, if at all, and the Merger may be delayed or not consummated. If the conditions are not satisfied or waived in a timely manner and the Merger is not consummated, ProFrac and USWS may lose some or all of the intended or perceived benefits of the Merger, which could cause ProFrac's and USWS' stock price to decline and harm their respective businesses.

The combined company may be unable to successfully integrate ProFrac's and USWS' operations or to realize the anticipated cost savings, revenues and other benefits of the Merger. As a result, the value of the combined company's common stock may be adversely affected.

ProFrac and USWS entered into the Merger Agreement because each company believes that the Merger will be beneficial to each of ProFrac, USWS and their respective stockholders. Currently, each company operates as an independent public company. Achieving the anticipated benefits of the Merger will depend in part upon

whether the two companies integrate their businesses in an efficient and effective manner. The companies may not be able to accomplish this integration process smoothly or successfully. The necessity of coordinating geographically separated organizations, systems and facilities and addressing possible differences in business backgrounds, corporate cultures and management philosophies may increase the difficulties of integration. The companies operate numerous systems, including those involving management information, purchasing, accounting and finance, sales, billing, employee benefits, payroll and regulatory compliance. The integration of certain operations following the Merger will require the dedication of significant management resources, which may temporarily distract management's attention from the day-to-day business of the combined company. Employee uncertainty and lack of focus during the integration process may also disrupt the business of the combined company and result in undesired employee attrition. Any inability of management to successfully integrate the operations of the two companies could have a material adverse effect on the business and results of operations of the combined company. The companies may not be able to achieve the anticipated operating and cost synergies or long-term strategic benefits of the Merger. An inability to realize the full extent of, or any of, the anticipated benefits of the Merger, as well as any delays encountered in the integration process, could have an adverse effect on the business and results of operations of the combined company, which may affect the value of the shares of the combined company's common stock after the completion of the Merger.

Due to legal restrictions, USWS and ProFrac have been able to conduct only limited planning regarding the integration of the two companies following the Merger and have not yet determined the exact nature of how the businesses and operations of the two companies will be combined after the Merger. The actual integration may result in additional and unforeseen expenses, and the anticipated benefits of the integration plan may not be realized.

#### The combined company will incur significant transaction and Merger-related costs in connection with the Merger.

ProFrac and USWS expect to incur significant costs associated with combining the operations of the two companies. The substantial majority of the expenses resulting from the Merger will be comprised of transaction costs related to the Merger, systems consolidation costs, and business integration and employment—related costs. ProFrac and USWS will also incur transaction fees and costs related to formulating integration plans. Additional unanticipated costs may be incurred in the integration of the two companies' businesses. Due to legal restrictions, USWS and ProFrac have been able to conduct only limited planning regarding the integration of the two companies and have not yet been able to formulate detailed integration plans to deliver anticipated synergies. Although ProFrac and USWS expect that the elimination of duplicative costs, as well as the realization of other efficiencies related to the integration of the businesses, should allow them to offset incremental transaction and Merger-related costs over time, this net benefit may not be achieved in the near term, or at all.

Whether or not the Merger is completed, the announcement and pendency of the Merger could impact or cause disruptions in USWS' and ProFrac's businesses, which could have an adverse effect on their results of operations and financial condition.

Whether or not the Merger is consummated, the announcement and pendency of the Merger could impact or cause disruptions in the businesses of ProFrac and USWS, which could have an adverse effect on their results of operations and financial condition. Specifically:

- current and prospective customers and suppliers of ProFrac and USWS may experience uncertainty associated with the Merger, including
  with respect to current or future business relationships with ProFrac, USWS or the combined company and may attempt to negotiate changes
  in existing business relationships or consider entering into business relationships with parties other than ProFrac, USWS or the combined
  company, either before or after completion of the Merger;
- USWS and ProFrac employees may experience uncertainty about their future roles with the combined company, which might adversely
  affect USWS' and ProFrac's ability to retain and hire key employees;

- if the Merger is completed, the accelerated equity-based awards and payment of "change in control" benefits to some members of USWS' and ProFrac's management on completion of the Merger could result in increased difficulty or cost in retaining USWS' and ProFrac's officers and employees; and
- the attention of management of each of USWS and ProFrac may be directed toward the completion of the Merger and transaction-related considerations and may be diverted from the day-to-day business operations of their respective companies.

#### The ProFrac Controlling Stockholders have material interests in USWS.

The ProFrac Controlling Stockholders have substantial investments and financial interests in USWS and, therefore, an acquisition of USWS by ProFrac could give rise to certain actual or potential conflicts of interest. For more information about such conflicts of interest, please see "The Merger" Interests of ProFrac Executive Officers and Directors in the Merger." As a result, the ProFrac Board formed the ProFrac Special Committee and provided the ProFrac Special Committee with the power and authority to, among other things, evaluate, review, consider and negotiate a potential transaction involving the acquisition of USWS by ProFrac. The ProFrac Board also resolved that the Board would not approve a potential transaction involving the acquisition of USWS by ProFrac without a prior favorable recommendation of such transaction by the ProFrac Special Committee. Following multiple meetings with its independent financial and legal advisors, the ProFrac Special Committee (i) determined that the Merger Agreement and the transactions contemplated thereby are fair to, and in the best interests of, ProFrac and the ProFrac Unaffiliated Stockholders, (ii) approved and declared advisable the Merger Agreement and the transactions contemplated thereby, (iii) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) determine that the Merger Agreement and the transactions contemplated thereby are fair to, and in the best interests of, ProFrac and the ProFrac Unaffiliated Stockholders and (y) approve and declare advisable the Merger Agreement and the transactions contemplated thereby, and (iv) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) direct that the ProFrac Stock Issuance be submitted to the ProFrac stockholders for their approval and (y) recommend that the ProFrac stockholders approve the ProFrac Stock Issuance. Notwithstanding the foregoing, there can be no definitive assurance that the efforts of the ProFrac Special Committee in connection with the Merger were sufficient in neutralizing or mitigating these actual or potential conflicts of interest. The determinations and recommendations of the ProFrac Special Committee were based on then-available information as of the date of such determinations and recommendations and does not reflect any subsequent events.

# Certain directors and executive officers of ProFrac and USWS have interests that may be different from, or in addition to, interests of ProFrac and USWS Stockholders generally.

Some directors and executive officers of ProFrac and USWS may have interests in the Merger that differ from yours. For example, some directors and executive officers of USWS have employment, indemnification and severance arrangements, rights to acceleration of stock options and other equity-based awards, and other benefits they may receive on a change in control of which provide them with interests in the Merger that may be different from yours. You should be aware of these interests when you consider your board of directors' recommendation that you vote in favor of the proposals submitted at your stockholder meeting. For more information on these interests, please see the sections titled "The Merger — Interests of ProFrac Executive Officers and Directors in the Merger" and "The Merger — Interests of USWS Executive Officers and Directors in the Merger."

# The deal-protection provisions of the Merger Agreement may deter alternative business combinations and could negatively impact the stock prices of ProFrac and USWS if the Merger Agreement is terminated.

As a result of certain provisions of the Merger Agreement, it is possible that a third party who might be interested in pursuing a business combination proposal with ProFrac or USWS would be discouraged from doing so. Any such proposal might be advantageous to the stockholders of ProFrac and USWS when compared to the terms and conditions of the transaction described in this proxy statement/information statement/prospectus. In

particular, the termination fee provision of the Merger Agreement may deter third parties from proposing alternative business combinations that might result in greater value to ProFrac and USWS Stockholders than the Merger. In addition, in the event the Merger Agreement is terminated in circumstances that obligate USWS to pay a termination fee to ProFrac or to reimburse ProFrac for certain expenses, USWS' stock price may decline as a result of making any such payment.

The Merger could have a dilutive effect on ProFrac's earnings per share calculated in accordance with GAAP, which may adversely affect the market price of ProFrac Class A Common Stock following the Merger.

The Merger could have a dilutive effect on earnings per share of ProFrac calculated in accordance with GAAP primarily due to the amortization of the intangible assets in connection with the Merger. These expectations are based on preliminary estimates which may materially change after the completion of the Merger. The combined company could also encounter additional transaction and integration-related costs or other factors such as the failure to realize all of the benefits anticipated in the Merger. All of these factors could cause further dilution to the earnings per share of the combined company or cause a decrease in the price of common stock of the combined company.

Charges to earnings resulting from the application of the purchase method of accounting may adversely affect the market value of the combined company's common stock following the Merger.

In accordance with GAAP, ProFrac will be considered the acquirer of USWS for accounting purposes. ProFrac will account for the Merger using the purchase method of accounting, which could result in additional depreciation and amortization expenses to the combined company's earnings that could adversely affect the market value of ProFrac's common stock following the completion of the Merger. Under the purchase method of accounting, ProFrac will allocate the total purchase price to the assets acquired and liabilities assumed from USWS based on their fair values as of the date of the Effective Time of the Merger, and record any excess of the purchase price over those fair values as goodwill. For certain tangible and intangible assets, recording their fair values as of the completion date of the Merger could result in the combined company incurring significant additional depreciation and/or amortization expense that exceed the combined amounts recorded by ProFrac and USWS prior to the Merger. This increased expense will be recorded by the combined company over the useful lives of the underlying assets. In addition, to the extent the value of goodwill or intangible assets were to become impaired, the combined company would be required to recognize impairment losses relating to those assets.

### Risks Related to the Combined Company Following the Merger

ProFrac may be required to incur significant additional indebtedness in order to finance the Merger, which will limit the combined company's operating flexibility, and could adversely affect the combined company's operations and financial results and prevent the combined company from fulfilling its obligations.

ProFrac may be required to incur significant additional indebtedness in order to fund its various obligations in connection with the Merger, including the repayment of approximately \$170 million of USWS debt currently projected to be outstanding at Closing and the payment of expenses incurred in connection with the Merger. ProFrac also expects to assume approximately \$55 million of USWS' outstanding indebtedness in connection with the Merger.

Although ProFrac expects to funds its obligations in connection with the Merger through its New Term Loan Credit Facility (as defined below) and/or its New ABL Credit Facility (as defined below), available cash, additional loans and/or notes and cash flows generated from operations, ProFrac cannot assure that these sources will be sufficient, or that it will be able to refinance any of its outstanding indebtedness, and/or obtain additional financing on favorable terms, or at all, or that its cash flows generated from operations will be sufficient, in order to satisfy its obligations in connection with the Merger, or otherwise.

ProFrac's ability to refinance any of its outstanding indebtedness, and/or obtain additional financing in order to satisfy its obligations in connection with the Merger will depend on, among other things, ProFrac's financial

condition at the time, restrictions in agreements governing its indebtedness, and other factors, including the financial condition of USWS, restrictions in agreements governing the indebtedness that ProFrac is assuming, and the condition of the financial markets and the markets in which ProFrac competes. As a result, ProFrac cannot assure that it will be able to refinance any of its outstanding indebtedness, and/or obtain additional financing on favorable terms, or at all, or that cash flows generated from operations will be sufficient, in order to satisfy ProFrac's obligations in connection with the acquisition transactions, or otherwise.

To the extent ProFrac is able to refinance any of its outstanding indebtedness, and/or obtain additional financing, such level of indebtedness may:

- · make it more difficult for the combined company to satisfy its obligations with respect to its outstanding indebtedness;
- · increase the combined company's vulnerability to general adverse economic and industry conditions, including increases in interest rates;
- require the combined company to dedicate a substantial portion of its cash flow from operations to interest and principal payments on its
  indebtedness, which would reduce the availability of its cash flow to fund working capital, capital expenditures, expansion efforts and other
  general corporate purposes;
- · limit the combined company's flexibility in planning for, or reacting to, changes in its business and the industry in which it operates;
- · place the combined company at a competitive disadvantage compared to its competitors that have less debt; and
- limit the combined company's ability to borrow additional funds for working capital, capital expenditures, general corporate purposes and acquisitions.

In addition, the terms of any financing obligations to be incurred in connection with any refinancing of any of ProFrac's outstanding indebtedness, and/or additional financing are likely to contain covenants that restrict the combined company's ability to engage in certain transactions and may impair the combined company's ability to respond to changing business and economic conditions, including, among other things, limitations on the combined company's ability to:

- · incur additional indebtedness;
- · pay dividends and make distributions;
- repurchase stock;
- · make certain investments;
- create liens:
- · engage in transactions with affiliates;
- · merge with or acquire another company; and
- transfer and sell assets.

The combined company's ability to comply with these provisions may be affected by general economic conditions, political decisions, industry conditions and other events beyond its control. The combined company's failure to comply with the covenants contained in any amended or new financing facilities could result in an event of default, which could materially and adversely affect the combined company's business, results of operation and financial condition.

If there were an event of default under one of the combined company's debt instruments or a change of control of the combined company, the holders of the defaulted debt could cause all amounts outstanding with

respect to that debt to be due and payable immediately and may be cross-defaulted to other debt. The combined company's assets or cash flow may not be sufficient to fully repay borrowings under its outstanding debt instruments if accelerated upon an event of default, and there is no guarantee that the combined company would be able to repay, refinance or restructure the payments on those debt securities.

The combined company may not be able to generate sufficient cash flow to service all of its obligations, including its obligations under its new credit facilities.

The combined company's ability to make payments on and to refinance its indebtedness, including the indebtedness incurred in connection with the Merger, and to fund planned capital expenditures, strategic transactions and expansion efforts will depend on the combined company's ability to generate cash in the future. This, to a certain extent, is subject to general economic, financial, competitive, legislative, regulatory and other factors that are beyond the control of the combined company.

The combined company's business may not be able to generate sufficient cash flow from operations, and the combined company cannot assure that future borrowings will be available to it in amounts sufficient to enable it to pay its indebtedness as such indebtedness matures and to fund its other liquidity needs. If this is the case, the combined company will need to refinance all or a portion of its indebtedness on or before maturity, and there can be no assurance that it will be able to refinance any of its indebtedness, including its new credit facilities, on commercially reasonable terms, or at all. The combined company may need to adopt one or more alternatives, such as reducing or delaying planned expenses and capital expenditures, selling assets, restructuring debt, or obtaining additional equity or debt financing. These financing strategies may not be affected on satisfactory terms, if at all. The combined company's ability to refinance its indebtedness or obtain additional financing, or to do so on commercially reasonable terms, will depend on, among other things, its financial condition at the time, restrictions in agreements governing its indebtedness, and other factors, including the condition of the financial markets and the markets in which the combined company will compete.

If the combined company does not generate sufficient cash flow from operations, and additional borrowings, refinancings or proceeds of asset sales are not available to it, the combined company may not have sufficient cash to enable it to meet all of its obligations.

The aggregate ownership and voting interest of the current USWS Stockholders in ProFrac after the Merger will be lower than what USWS Stockholders currently have in USWS and they will exercise less influence over management of ProFrac than they currently exercise over management of USWS.

#### Risks Related to ProFrac's Business

ProFrac's business and financial performance depends on the oil and natural gas industry and particularly on the level of capital spending and E&P activity within the United States and in the basins in which ProFrac operates, and a decline in prices for oil and natural gas may have an adverse effect on ProFrac's revenue, cash flows, profitability and growth.

Demand for most of ProFrac's services depends substantially on the level of capital expenditures in the United States by companies in the oil and natural gas industry. As a result, ProFrac's operations are dependent on the levels of capital spending and activity in oil and gas exploration, development and production. A prolonged reduction in oil and gas prices would generally depress the level of oil and natural gas exploration, development, production, and well completion activity and would result in a corresponding decline in the demand for the hydraulic fracturing services that ProFrac provides. The significant decline in oil and natural gas prices that occurred in 2020 caused a reduction in ProFrac's customers' spending and associated drilling and completion activities, which had an adverse effect on ProFrac's revenue. While oil and natural gas prices have since increased, should prices again decline, similar declines in ProFrac's customers' spending would have an adverse effect on ProFrac's revenue. In addition, a worsening of these conditions may result in a material adverse impact

on certain of ProFrac's customers' liquidity and financial position resulting in further spending reductions, delays in the collection of amounts owing to ProFrac and similar impacts.

Many factors over which ProFrac has no control affect the supply of and demand for, and ProFrac's customers' willingness to explore, develop and produce oil and natural gas, and therefore, influence prices for ProFrac's services, including:

- the U.S. and non-U.S. supply of, and demand for, oil and natural gas;
- · the level of prices, and expectations about future prices, of oil and natural gas;
- the level of global oil and natural gas E&P;
- the cost of exploring for, developing, producing and delivering oil and natural gas;
- the supply of and demand for drilling and hydraulic fracturing equipment;
- global or national health concerns, including health epidemics such as the ongoing COVID-19 pandemic;
- the expected decline rates of current production;
- inability to acquire or maintain necessary permits or mining or water rights;
- the price and quantity of foreign imports;
- political and economic conditions in oil and natural gas producing countries and regions, including the United States, the Middle East, Africa, South America and Russia;
- actions by the members of OPEC+ and other oil-producing countries with respect to oil production levels and announcements of potential changes in such levels;
- speculative trading in crude oil and natural gas derivative contracts;
- the level of consumer product demand;
- the discovery rates of new oil and natural gas reserves;
- · the availability of water resources, suitable proppant and chemical additives in sufficient quantities for use in hydraulic fracturing fluids;
- contractions in the credit market;
- the strength or weakness of the U.S. dollar;
- available pipeline and other transportation capacity;
- · the levels of oil and natural gas storage;
- adverse weather conditions and other natural disasters;
- U.S. and non-U.S. tax policy;
- U.S. and non-U.S. governmental approvals and regulatory requirements and conditions;
- · the continued threat of terrorism and the impact of military and other action, including military action in the Middle East;
- · technical advances affecting energy consumption;
- the proximity and capacity of oil and natural gas pipelines and other transportation facilities;
- · the price and availability of alternative fuels and energy sources;
- · uncertainty in capital commodities markets and the ability of oil and natural gas producers to raise equity capital and debt financing;

- merger and divestiture activity among oil and natural gas producers;
- cyclical/seasonal business and dependence upon spending of ProFrac's customers;
- competition among oilfield service and equipment providers;
- changes in transportation regulations that result in increased costs or administrative burdens; and
- overall domestic and global economic conditions.

These factors and the volatility of the energy markets make it extremely difficult to predict future oil and natural gas price movements with any certainty. Such a decline would have a material adverse effect on ProFrac's business, results of operation and financial condition.

# The COVID-19 pandemic reduced demand for ProFrac's services and could, in the future, have a material adverse effect on ProFrac's operations, business and financial results.

ProFrac faces risks related to public health crises, including the COVID-19 pandemic. The effects of the COVID-19 pandemic, including travel bans, prohibitions on group events and gatherings, shutdowns of certain businesses, curfews, shelter-in-place orders and recommendations to practice social distancing in addition to other actions taken by both businesses and governments, resulted in a significant and swift reduction in international and U.S. economic activity. The collapse in the demand for oil caused by this unprecedented global health and economic crisis contributed to the significant decrease in crude oil prices in 2020 and adversely impacted the demand for ProFrac's services and could, in the future, have a material adverse effect on ProFrac's operations, business and financial results.

Since the beginning of 2021, the distribution of COVID-19 vaccines progressed and many government-imposed restrictions were relaxed or rescinded. However, ProFrac continue to monitor the effects of the pandemic on ProFrac's customers, operations, and employees. These effects have included, and may continue to include, adverse revenue and net income effects, financial health of ProFrac's customers and therefore their ability to drill and complete wells or pay for services provided, financial health of ProFrac's suppliers and therefore their ability to deliver necessary goods and services, disruptions to ProFrac's operations, and ultimately the financial health and results of ProFrac.

The extent to which ProFrac's operating and financial results are affected by COVID-19 will depend on various factors and consequences beyond ProFrac's control, such as the emergence of more contagious and harmful variants of the COVID-19 virus, the duration and scope of the pandemic, additional actions by businesses and governments in response to the pandemic, and the speed and effectiveness of responses to combat the virus. COVID-19, and the volatile regional and global economic conditions stemming from the pandemic, could also aggravate the other risk factors that ProFrac identifies herein. While the effects of the COVID-19 pandemic have lessened recently in the United States, ProFrac cannot predict the duration or future effects of the pandemic, or more contagious and harmful variants of the COVID-19 virus, and such effects may materially adversely affect ProFrac's operating and financial results in a manner that is not currently known to ProFrac or that ProFrac does not currently consider to present significant risks to its operations.

# The cyclical nature of the oil and natural gas industry may cause ProFrac's operating results to fluctuate.

ProFrac derives its revenues from companies in the oil and natural gas E&P industry, a historically cyclical industry with levels of activity that are significantly affected by the levels and volatility of oil and natural gas prices. ProFrac has experienced, and may in the future experience, significant fluctuations in operating results as a result of the reactions of ProFrac's customers to changes in oil and natural gas prices. For example, prolonged low commodity prices experienced by the oil and natural gas industry during 2015, 2016 and recently in 2020, combined with adverse changes in the capital and credit markets, caused many E&P companies to reduce their capital budgets and drilling activity. This resulted in a significant decline in demand for oilfield services and

adversely impacted the prices oilfield services companies could charge for their services. In addition, a majority of the service revenue ProFrac earns is based upon a charge for a relatively short period of time (for example, a day, a week or a month) for the actual period of time the service is provided to ProFrac's customers. By contracting services on a short-term basis, ProFrac is exposed to the risks of a rapid reduction in market prices and utilization and resulting volatility in ProFrac's revenues.

#### The ongoing military action between Russia and Ukraine could adversely affect ProFrac's business, financial condition and results of operations.

On February 24, 2022, Russian military forces commenced a military operation in Ukraine. Although the length, impact and outcome of the ongoing military conflict in Ukraine is highly unpredictable, this conflict could lead to significant market and other disruptions, including significant volatility in commodity prices and supply of energy resources, instability in financial markets, supply chain interruptions, political and social instability, changes in consumer or purchaser preferences as well as increase in cyberattacks and espionage.

Russia's recognition of two separatist republics in the Donetsk and Luhansk regions of Ukraine and subsequent military action against Ukraine have led to an unprecedented expansion of sanction programs imposed by the United States, the European Union, the United Kingdom, Canada, Switzerland, Japan and other countries against Russia, Belarus, the Crimea Region of Ukraine, the so-called Donetsk People's Republic and the so-called Luhansk People's Republic, including, among others:

- blocking sanctions against some of the largest state-owned and private Russian financial institutions (and their subsequent removal from the Society for Worldwide Interbank Financial Telecommunication ("SWIFT") payment system) and certain Russian businesses, some of which have significant financial and trade ties to the European Union;
- blocking sanctions against Russian and Belarusian individuals, including the Russian President, other politicians and those with government connections or involvement in Russian military activities; and
- blocking of Russia's foreign currency reserves as well as expansion of sectoral sanctions and export and trade restrictions, limitations on investments and access to capital markets and bans on various Russian imports.

The situation is rapidly evolving as a result of the conflict in Ukraine, and the United States, the European Union, the United Kingdom and other countries may implement additional sanctions, export controls or other measures against Russia, Belarus and other countries, regions, officials, individuals or industries in the respective territories. Such sanctions and other measures, as well as the existing and potential further responses from Russia or other countries to such sanctions, tensions and military actions, could adversely affect the global economy and financial markets and could adversely affect ProFrac's business, financial condition and results of operations. ProFrac's manufacturing operations would be potentially vulnerable to interruptions in the supply of certain materials and metals, such as nickel, which are incorporated into raw materials ProFrac obtains from suppliers and uses in its manufacturing processes.

# ProFrac's business may be affected by sanctions, export controls and similar measures targeting Russia, as well as other responses to Russia's military action in Ukraine.

As a result of Russia's military action in Ukraine, governmental authorities in the United States, the European Union and the United Kingdom, among others, launched an expansion of coordinated sanctions and export control measures, including:

- · blocking sanctions on some of the largest state-owned and private Russian financial institutions (and their subsequent removal from SWIFT);
- blocking sanctions against Russian and Belarusian individuals, including the Russian President, other politicians and those with government connections or involved in Russian military activities;

- blocking sanctions against certain Russian businessmen and their businesses, some of which have significant financial and trade ties to the European Union;
- blocking of Russia's foreign currency reserves and prohibition on secondary trading in Russian sovereign debt and certain transactions with the Russian Central Bank, National Wealth Fund and the Ministry of Finance of the Russian Federation;
- · expansion of sectoral sanctions in various sectors of the Russian and Belarusian economies and the defense sector;
- · United Kingdom sanctions introducing restrictions on providing loans to, and dealing in securities issued by, persons connected with Russia;
- restrictions on access to the financial and capital markets in the European Union, as well as prohibitions on aircraft leasing operations;
- sanctions prohibiting most commercial activities of U.S. and EU persons in Crimea and Sevastopol;
- enhanced export controls and trade sanctions targeting Russia's imports of technological goods as a whole, including tighter controls on
  exports and reexports of dual-use items, stricter licensing policy with respect to issuing export licenses, and/or increased use of end-use"
  controls to block or impose licensing requirements on exports, as well as higher import tariffs and a prohibition on exporting luxury goods to
  Russia and Belarus;
- · closure of airspace to Russian aircraft; and
- ban on imports of Russian oil, liquefied natural gas and coal to the United States.

As the conflict in Ukraine continues, there can be no certainty regarding whether the governmental authorities in the United States, the European Union, the United Kingdom or other countries will impose additional sanctions, export controls or other measures targeting Russia, Belarus and other territories.

ProFrac's business must be conducted in compliance with applicable economic and trade sanctions laws and regulations, including those administered and enforced by the U.S. Department of Treasury's Office of Foreign Assets Control, the U.S. Department of State, the U.S. Department of Commerce, the United Nations Security Council and other relevant governmental authorities. ProFrac must be ready to comply with the existing and any other potential additional measures imposed in connection with the conflict in Ukraine.

ProFrac does not currently have contracts directly with the entities or businesses on the sanctions list and ProFrac does not currently have operations, nor does ProFrac directly source materials from, Russia, Belarus, the Crimea Region of Ukraine, the so-called Donetsk People's Republic or the so-called Luhansk People's Republic. ProFrac continuously reviews and monitors its contractual relationships with suppliers and customers to establish whether any are the target of the applicable sanctions. In the unlikely event that ProFrac identifies a party with which ProFrac has a business relationship that is the target of applicable sanctions, ProFrac would immediately activate a legal analysis of what gives rise to the business relationship, including any contract, to estimate the most appropriate course of action to comply with the sanction regulations, together with the impact of a contractual termination according to the applicable law, and then proceed as required by the regulatory authorities. However, given the range of possible outcomes, the full costs, burdens, and limitations on ProFrac's and its customers' and business partners' businesses are currently unknown and may become significant.

Furthermore, even if an entity is not formally subject to sanctions, customers and business partners of such entity may decide to reevaluate or cancel projects with such entity for reputational or other reasons. As result of the ongoing conflict in Ukraine, many U.S. and other multi-national businesses across a variety of industries, including consumer goods and retail, food, energy, finance, media and entertainment, tech, travel and logistics, manufacturing and others, have indefinitely suspended their operations and paused all commercial activities in Russia and Belarus. Depending on the extent and breadth of sanctions, export controls and other measures that

may be imposed in connection with the conflict in Ukraine, it is possible that ProFrac's business, financial condition and results of operations could be materially and adversely affected.

### ProFrac faces significant competition that may cause a loss of market share.

The oilfield services industry is highly competitive and has relatively few barriers to entry. The principal competitive factors impacting sales of ProFrac's services are price, reputation and technical expertise, equipment and service quality and health and safety standards. The market is also fragmented and includes numerous small companies capable of competing effectively in ProFrac's markets on a local basis, as well as several large companies that possess substantially greater financial and other resources than ProFrac does. ProFrac's larger competitors' greater resources could allow those competitors to compete more effectively than ProFrac can. For instance, ProFrac's larger competitors may offer services at below-market prices or bundle ancillary services at no additional cost to customers. ProFrac competes with large national and multi-national companies that have longer operating histories, greater financial, technical and other resources and greater name recognition than ProFrac does. Several of ProFrac's competitors provide a broader array of services and have a stronger presence in more geographic markets.

Some jobs are awarded on a bid basis, which further increases competition based on price. Pricing is often the primary factor in determining which qualified contractor is awarded a job. The competitive environment may be further intensified by mergers and acquisitions among oil and natural gas companies or other events that have the effect of reducing the number of available customers. As a result of a combination of continued pressure from increased competition which began during the second half of 2018 and 2019 and decreased demand for ProFrac's services in 2020 due to the COVID-19 pandemic, ProFrac had to lower the prices for its services, which adversely affected ProFrac's results of operations. If competition remains the same or increases as a result of a continued industry downturn or future industry downturns, ProFrac may be required to lower its prices, which would adversely affect its results of operations. In the future, ProFrac may lose market share or be unable to maintain or increase prices for ProFrac's present services or to acquire additional business opportunities, which could have a material adverse effect on ProFrac's business, financial condition, results of operations and cash flows.

ProFrac's competitors may be able to respond more quickly to new or emerging technologies and services and changes in customer requirements. The amount of equipment available may exceed demand, which could result in active price competition. In addition, depressed commodity prices lower demand for hydraulic fracturing equipment, which results in excess equipment and lower utilization rates. In addition, some E&P companies have commenced completing their wells using their own hydraulic fracturing equipment and personnel. Any increase in the development and utilization of in-house fracturing capabilities by ProFrac's customers could decrease the demand for ProFrac's services and have a material adverse impact on ProFrac's business.

In addition, competition among oilfield service and equipment providers is affected by each provider's reputation for safety and quality. ProFrac cannot assure that it will be able to maintain its competitive position.

ProFrac's business depends upon ProFrac's ability to obtain specialized equipment, parts and key raw materials from third-party suppliers, and ProFrac may be vulnerable to delayed deliveries and future price increases.

While ProFrac operates a vertically integrated business, ProFrac purchases certain specialized equipment, parts and raw materials from third party suppliers and affiliates. At times during the commodity price cycle, there is a high demand for hydraulic fracturing and other oilfield services and extended lead times to obtain equipment and raw materials needed to provide these services. Should ProFrac's current suppliers be unable or unwilling to provide the necessary equipment, parts or raw materials or otherwise fail to deliver the products timely and in the quantities required, any resulting delays in the provision of ProFrac's services could have a material adverse effect on ProFrac's business, financial condition, results of operations and cash flows. In addition, future price

increases for this type of equipment, parts and raw materials could negatively impact ProFrac's ability to purchase new equipment, to update or expand ProFrac's existing fleet, to timely repair equipment in ProFrac's existing fleet or meet the current demands of ProFrac's customers.

ProFrac currently relies on a limited number of suppliers for major equipment to build new electric-powered hydraulic fracturing fleets utilizing Clean Fleet® technology, and ProFrac's reliance on these vendors exposes ProFrac to risks including price and timing of delivery.

ProFrac currently relies on a limited number of suppliers for major equipment to build new electric-powered hydraulic fracturing fleets utilizing Clean Fleet® technology, including USWS. During periods in which fracturing services are in high demand, ProFrac may experience delays in obtaining certain parts that are used in fabricating and assembling ProFrac's fleets. If demand for hydraulic fracturing fleets or the components necessary to build such fleets increases or these vendors face financial distress or bankruptcy, these vendors may not be able to provide the components necessary to construct ProFrac's electric-powered hydraulic fracturing fleets on schedule or at the current price. If this were to occur, ProFrac could be required to seek other suppliers for major equipment to build ProFrac's electric-powered hydraulic fracturing fleets, which may adversely affect ProFrac's revenues or increase ProFrac's costs.

#### Reliance upon a few large customers may adversely affect ProFrac's revenue and operating results.

The majority of ProFrac's revenue is generated from hydraulic fracturing services. Due to the large percentage of ProFrac's revenue historically derived from hydraulic fracturing services with recurring customers and the limited availability of ProFrac's fracturing units, ProFrac has had some degree of customer concentration. ProFrac's top ten customers represented approximately 50.8% and 52.8% of ProFrac's consolidated revenue for the years ended December 31, 2021 and 2020. It is likely that ProFrac will depend on a relatively small number of customers for a significant portion of ProFrac's revenue in the future. If a major customer fails to pay us, cash flow from operations would be impacted and ProFrac's operating results and financial condition could be harmed. Additionally, if ProFrac were to lose any material customer, ProFrac may not be able to redeploy equipment at similar utilization or pricing levels and such loss could have an adverse effect on ProFrac's business until the equipment is redeployed at similar utilization or pricing levels.

ProFrac is exposed to counterparty credit risk. Nonpayment and nonperformance by ProFrac's customers, suppliers or vendors could adversely impact ProFrac's operations, cash flows and financial condition.

Weak economic conditions and widespread financial distress, including the significantly reduced global and national economic activity caused by the COVID-19 pandemic, could reduce the liquidity of ProFrac's customers, suppliers or vendors, making it more difficult for them to meet their obligations to ProFrac. ProFrac is therefore subject to heightened risks of loss resulting from nonpayment or nonperformance by ProFrac's customers, suppliers and vendors. Severe financial problems encountered by ProFrac's customers, suppliers and vendors could limit ProFrac's ability to collect amounts owed to ProFrac, or to enforce the performance of obligations owed to ProFrac under contractual arrangements. In the event that any of ProFrac's customers was to enter into bankruptcy, ProFrac could lose all or a portion of the amounts owed to ProFrac by such customer, and ProFrac may be forced to cancel all or a portion of ProFrac's service contracts with such customer at significant expense to ProFrac.

In addition, nonperformance by suppliers or vendors, who have committed to provide ProFrac with critical products or services, could raise ProFrac's costs or interfere with ProFrac's ability to successfully conduct ProFrac's business. All of the above may be exacerbated in the future as the COVID-19 outbreak and the governmental responses to the outbreak continue. These factors, combined with volatile prices of oil and natural gas, may precipitate a continued economic slowdown and/or a recession.

Oil and natural gas companies' operations using hydraulic fracturing are substantially dependent on the availability of water. Restrictions on the ability to obtain water for E&P activities and the disposal of flowback and produced water may impact their operations and have a corresponding adverse effect on ProFrac's business, results of operations and financial condition.

Water is an essential component of shale oil and natural gas production during both the drilling and hydraulic fracturing processes. ProFrac's oil and natural gas producing customers' access to water to be used in these processes may be adversely affected due to reasons such as periods of extended drought, privatization, third party competition for water in localized areas or the implementation of local or state governmental programs to monitor or restrict the beneficial use of water subject to their jurisdiction for hydraulic fracturing to assure adequate local water supplies. The occurrence of these or similar developments may result in limitations being placed on allocations of water due to needs by third party businesses with more senior contractual or permitting rights to the water. ProFrac's customers' inability to locate or contractually acquire and sustain the receipt of sufficient amounts of water could adversely impact their E&P operations and have a corresponding adverse effect on ProFrac's business, results of operations and financial condition.

Moreover, the imposition of new environmental regulations and other regulatory initiatives could include increased restrictions on ProFrac's producing customers' ability to dispose of flowback and produced water generated by hydraulic fracturing or other fluids resulting from E&P activities. Applicable laws impose restrictions and strict controls regarding the discharge of pollutants into waters of the United States and require that permits or other approvals be obtained to discharge pollutants to such waters. Additionally, regulations implemented under both federal and state laws prohibit the discharge of produced water and sand, drilling fluids, drill cuttings and certain other substances related to the natural gas and oil industry into coastal waters. These laws provide for civil, criminal and administrative penalties for any unauthorized discharges of pollutants and unauthorized discharges of reportable quantities of oil and hazardous substances. Compliance with current and future environmental regulations and permit requirements governing the withdrawal, storage and use of surface water or groundwater necessary for hydraulic fracturing of wells and any inability to secure transportation and access to disposal wells with sufficient capacity to accept all of ProFrac's flowback and produced water on economic terms may increase ProFrac's customers' operating costs and could result in restrictions, delays, or cancellations of ProFrac's customers' operations, the extent of which cannot be predicted.

#### ProFrac relies on a few key employees whose absence or loss could adversely affect ProFrac's business.

Many key responsibilities within ProFrac's business have been assigned to a small number of employees. The loss of their services could adversely affect ProFrac's business. In particular, the loss of the services of one or more members of ProFrac's executive team, including ProFrac's Chief Executive Officer, Executive Chairman, Chief Operating Officer, Chief Legal Officer and Chief Financial Officer, could disrupt ProFrac's operations. ProFrac does not maintain "key person" life insurance policies on any of ProFrac's employees. As a result, ProFrac is not insured against any losses resulting from the death of ProFrac's key employees.

# If ProFrac is unable to employ a sufficient number of skilled and qualified workers, ProFrac's capacity and profitability could be diminished and ProFrac's growth potential could be impaired.

The delivery of ProFrac's services requires skilled and qualified workers with specialized skills and experience who can perform physically demanding work. As a result of the volatility of the oilfield services industry and the demanding nature of the work, workers may choose to pursue employment in fields that offer a more desirable work environment at wage rates that are competitive. ProFrac's ability to be productive and profitable will depend upon ProFrac's ability to employ and retain skilled workers. In addition, ProFrac's ability to expand ProFrac's operations depends in part on ProFrac's ability to increase the size of ProFrac's skilled labor force. The demand for skilled workers is high, and the supply is limited. As a result, competition for experienced oilfield service personnel is intense, and ProFrac faces significant challenges in competing for crews and management with large and well-established competitors. A significant increase in the wages paid by competing

employers could result in a reduction of ProFrac's skilled labor force, increases in the wage rates that ProFrac must pay, or both. If either of these events were to occur, ProFrac's capacity and profitability could be diminished and ProFrac's growth potential could be impaired.

A negative shift in investor sentiment of the oil and gas industry has had and could in the future have adverse effects on ProFrac's customers' operations and ability to raise debt and equity capital.

Certain segments of the investor community have developed negative sentiment towards investing in ProFrac's industry. Recent equity returns in the sector versus other industry sectors have led to lower oil and gas and related services representation in certain key equity market indices. In addition, some investors, including investment advisors and certain sovereign wealth funds, pension funds, university endowments and family foundations, have stated policies to disinvest in the oil and gas sector based on their social and environmental considerations. Certain other stakeholders have also pressured commercial and investment banks and other lenders and investors to stop financing oil and gas production and related infrastructure projects, which adversely affects ProFrac's customers. Such developments, including environmental activism and initiatives aimed at limiting climate change and reducing air pollution, could result in downward pressure on the stock prices of oilfield service companies, including ProFrac's. This may also potentially result in a reduction of available capital funding for potential transactions, impacting ProFrac's future financial results.

Additionally, negative public perception regarding ProFrac's industry may lead to increased regulatory scrutiny, which may, in turn, lead to new state and federal safety and environmental laws, regulations, guidelines or enforcement interpretations. Additionally, environmental groups, landowners, local groups and other advocates may oppose ProFrac's customers' operations through organized protests, attempts to block or sabotage ProFrac's customers' operations, intervene in regulatory or administrative proceedings involving ProFrac's customers' assets, or file lawsuits or other actions designed to prevent, disrupt or delay the development or operation of ProFrac's customers' assets. These actions may cause operational delays or restrictions, increased operating costs, additional regulatory burdens and increased risk of litigation for ProFrac's customers, which could reduce ProFrac's customers' production levels over time and, as a result, may reduce demand for ProFrac's services. Moreover, governmental authorities exercise considerable discretion in the timing and scope of permit issuance and the public may engage in the permitting process, including through intervention in the courts. Negative public perception could cause the permits that ProFrac's customers require to conduct their operations to be withheld, delayed or burdened by requirements that restrict ProFrac's customers' ability to profitably conduct their businesses, which would also reduce demand for ProFrac's services. Ultimately, this could make it more difficult to secure funding for ProFrac's operations.

In addition, organizations that provide information to investors on corporate governance and related matters have developed ratings processes for evaluating companies on their approach to ESG matters. Such ratings are used by some investors to inform their investment and voting decisions. Unfavorable ESG ratings and recent activism directed at shifting funding away from companies with fossil fuel-related assets could lead to increased negative investor sentiment toward ProFrac and ProFrac's industry and to the diversion of investment to other industries, which could have a negative impact on the price of ProFrac Common Stock and ProFrac's or ProFrac's customers' access to and cost of capital. Also, institutional lenders may decide not to provide funding for fossil fuel energy companies based on climate change-related concerns, which could affect ProFrac's or ProFrac's customers' access to capital for potential growth projects.

ProFrac's operations require substantial capital and ProFrac may be unable to obtain needed capital or financing on satisfactory terms or at all, which could limit ProFrac's ability to grow.

The oilfield services industry is capital intensive. In conducting ProFrac's business and operations, ProFrac has made, and expects to continue to make, substantial capital expenditures. ProFrac's total capital expenditures were approximately \$48.0 million for the year ended December 31, 2020 and \$87.4 million for the year ended December 31, 2021. ProFrac has historically financed capital expenditures primarily with cash generated by

operations, equipment and vendor financing, borrowings under ProFrac's Old ABL Credit Facility (as defined herein) and other debt financing. Following the completion of this offering, ProFrac intends to finance ProFrac's capital expenditures primarily with cash on hand, cash flow from operations and borrowings under ProFrac's New ABL Credit Facility. Further, any disruptions or continuing volatility in the global financial markets may lead to an increase in interest rates or a contraction in credit availability impacting ProFrac's ability to finance ProFrac's operations. This could put ProFrac at a competitive disadvantage or interfere with ProFrac's growth plans. Further, ProFrac's actual capital expenditures for 2022 or future years could exceed ProFrac's capital expenditure budget. In the event ProFrac's capital expenditure requirements at any time are greater than the amount ProFrac has available, ProFrac could be required to seek additional sources of capital, which may include debt financing, joint venture partnerships, sales of assets, offerings of debt or equity securities or other means. ProFrac may not be able to obtain any such alternative source of capital. ProFrac may be required to curtail or eliminate contemplated activities. If ProFrac can obtain alternative sources of capital, the terms of such alternative may not be favorable to us. In particular, the terms of any debt financing may include covenants that significantly restrict ProFrac's operations. ProFrac's inability to grow as planned may reduce ProFrac's chances of maintaining and improving profitability.

The growth of ProFrac's business through recently completed acquisitions and potential future acquisitions may expose ProFrac to various risks, including those relating to difficulties in identifying suitable, accretive acquisition opportunities and integrating businesses, assets and personnel, as well as difficulties in obtaining financing for targeted acquisitions and the potential for increased leverage or debt service requirements.

ProFrac has pursued and intends to continue to pursue selected, accretive acquisitions of complementary assets and businesses. Acquisitions involve numerous risks, including:

- unanticipated costs and exposure to liabilities assumed in connection with the acquired business or assets, including but not limited to
  environmental liabilities and title issues;
- · difficulties in integrating the operations and assets of the acquired business and the acquired personnel;
- complexities associated with managing a larger, more complex, integrated business;
- · limitations on ProFrac's ability to properly assess and maintain an effective internal control environment over an acquired business;
- potential losses of key employees, customers and business partners of the acquired business;
- performance shortfalls at one or both of the companies as a result of the diversion of management's attention from theirday-to-day responsibilities caused by completing an acquisition and integrating an acquired business into the combined company;
- · risks of entering markets in which ProFrac has limited prior experience; and
- · increases in ProFrac's expenses and working capital requirements.

The process of integrating an acquired business, including in connection with ProFrac's recently completed FTSI Acquisition and the expected future completion of each of the SPS Acquisition and the Merger, may involve unforeseen costs and delays or other operational, technical and financial difficulties and may require a significant amount of time and resources. For example, ProFrac may experience difficulties in integrating FTSI's, the SP Companies' or USWS' operations into ProFrac's business and in realizing expected benefits and synergies from such acquisitions. The integration process may involve unforeseen difficulties and may require a disproportionate amount of ProFrac's managerial and financial resources. If ProFrac is unable to successfully integrate the operations of FTSI, the SP Companies or USWS with ProFrac's business, ProFrac may be unable to achieve consolidation savings and may incur unanticipated costs and liabilities. Likewise, ProFrac does not have a formal estimate of mineral reserves for West Munger, and the productivity of that site could be less than ProFrac is anticipating. ProFrac could also encounter difficulties in the development of the mining and sand processing capacity at ProFrac's West Munger Facility and may not realize the expected benefits from ProFrac's

investments in Flotek Industries, Inc. ('Flotek'') and FHE USA LLC ('FHE''). ProFrac's failure to incorporate the acquired business and assets into ProFrac's existing operations successfully or to minimize any unforeseen operational difficulties could have a material adverse effect on ProFrac's business, liquidity position, financial condition, prospects and results of operations. Furthermore, there is intense competition for acquisition opportunities in ProFrac's industry. Competition for acquisitions may increase the cost of, or cause ProFrac to refrain from, completing acquisitions.

In addition, ProFrac may not have sufficient capital resources to complete any additional acquisitions, such as the non-binding letter of intent ProFrac signed on August 18, 2022, see "Information About ProFrac — Recent Events." Historically, ProFrac has financed ProFrac's acquisitions primarily with funding from ProFrac's equity investors, commercial borrowings and cash generated by operations. ProFrac may incur substantial indebtedness to finance future acquisitions and also may issue equity, debt or convertible securities in connection with such acquisitions. Debt service requirements could represent a significant burden on ProFrac's results of operations and financial condition, and the issuance of additional equity or convertible securities could be dilutive to ProFrac's existing shareholders. Furthermore, ProFrac may not be able to obtain additional financing as needed or on satisfactory terms.

ProFrac's ability to continue to grow through acquisitions and manage growth will require ProFrac to continue to invest in operational, financial and management information systems and to attract, retain, motivate and effectively manage ProFrac's employees. The inability to effectively manage the integration of acquisitions, including in connection with ProFrac's Corporate Reorganization (as defined below), could reduce ProFrac's focus on current operations, which, in turn, could negatively impact ProFrac's earnings and growth. ProFrac's financial position and results of operations may fluctuate significantly from period to period, based on whether or not significant acquisitions are completed in particular periods.

Consummation of the Merger is contingent upon the satisfaction of a number of conditions, including approval by USWS' Stockholders, which may not be satisfied or completed on a timely basis, if at all. Failure to complete the Merger could negatively impact ProFrac's stock price, business, financial condition, results of operations or prospects.

Consummation of the Merger is subject to conditions to closing that are not wholly within ProFrac's control, including, among other things, adoption of the Merger Agreement by the holders of a majority of the outstanding shares of USWS Common Stock entitled to vote on the adoption of the Merger Agreement. ProFrac cannot assure that each of the conditions with respect to the Merger will be satisfied or waived in a timely manner, if at all, and the Merger may be delayed or not consummated. If the conditions are not satisfied or waived in a timely manner and the Merger is delayed or not consummated, ProFrac may lose some or all of the intended or perceived benefits of the Merger, which could cause ProFrac's stock price to decline and harm its business.

ProFrac is also subject to additional risks in connection with the Merger, including, without limitation: (1) the parties' ability to meet expectations regarding the timing and completion of the Merger; (2) the occurrence of any event, change or other circumstance that could give rise to the termination of the Merger Agreement; (3) the effect of the announcement or pendency of the Merger on ProFrac's business relationships, operating results, and business generally; (4) risks that the Merger disrupts ProFrac's current plans and operations; (5) the amount of the costs, fees, expenses and other charges related to the Merger; (6) the outcome of any legal proceedings that may be instituted against ProFrac or any of its affiliates relating to the transactions contemplated by the Merger; (7) the restrictions imposed on ProFrac's business and operations pursuant to any affirmative or negative covenants set forth in the documents relating to the Merger, and the potential impact of such covenants on ProFrac's business; (8) the risk that the Merger will divert management's attention resulting in a potential disruption of ProFrac's current business plan; (9) potential difficulties in employee retention arising from the Merger; (10) the ability to obtain certain governmental and regulatory approvals, including the

expiration or termination of the HSR Act waiting period (which has been satisfied); and (11) the ability to integrate acquired assets and personnel into ProFrac's existing business model and realize the expected value of resulting operational synergies from the Merger.

# ProFrac may not be able to generate sufficient cash flow to service all of its obligations, including obligations under its recently amended credit and other financing facilities.

ProFrac's ability to fund its various obligations in connection with the Merger, make payments on or refinance any of its outstanding indebtedness, and/or obtain additional financing, including any indebtedness incurred to finance the Merger, and to fund planned capital expenditures, other strategic transactions and expansion efforts, will depend on ProFrac's ability to generate cash in the future. This, to a certain extent, is subject to general economic, financial, competitive, legislative, regulatory and other factors that are beyond ProFrac's control.

ProFrac's business may not be able to generate sufficient cash flows from operations, and ProFrac cannot assure that future borrowings will be available to ProFrac in amounts sufficient to enable ProFrac to pay its indebtedness or any future indebtedness as such indebtedness matures and to fund its other liquidity needs. If this is the case, ProFrac will need to refinance all or a portion of its indebtedness on or before maturity, and ProFrac cannot assure that it will be able to refinance any of its indebtedness on commercially reasonable terms, or at all. ProFrac may need to implement one or more alternatives, such as reducing or delaying planned expenses and capital expenditures, selling assets, restructuring debt, or obtaining additional equity or debt financing. These financing strategies may not be executed on satisfactory terms, if at all. ProFrac's ability to refinance its indebtedness or obtain additional financing, and to do so on commercially reasonable terms, will depend on, among other things, ProFrac's financial condition at the time, restrictions in agreements governing ProFrac's indebtedness, and other factors, including the condition of the financial markets and the markets in which ProFrac competes.

If ProFrac does not generate sufficient cash flows from operations, and additional borrowings, refinancings or proceeds of asset sales are not available to ProFrac, ProFrac may not have sufficient cash to enable it to meet all of its obligations, including its various obligations in connection with the Merger.

# Concerns over general economic, business or industry conditions may have a material adverse effect on ProFrac's results of operations, liquidity and financial condition.

Concerns over global economic conditions, geopolitical issues, including the recent conflict between Russia and Ukraine, supply chain disruptions, interest rates, inflation, the availability and cost of credit and the United States and foreign financial markets have contributed to increased economic uncertainty and diminished expectations for the global economy. These factors, combined with volatility in commodity prices, business and consumer confidence and unemployment rates, have precipitated an economic slowdown. Concerns about global economic growth have had a significant adverse impact on global financial markets and commodity prices. If the economic climate in the United States or abroad deteriorates, worldwide demand for petroleum products could diminish further, which could impact the price at which oil, natural gas and natural gas liquids can be sold, which could affect the ability of ProFrac's customers to continue operations and ultimately adversely impact ProFrac's results of operations, liquidity and financial condition.

### Inflation may adversely affect ProFrac's operating results.

Inflationary factors such as increases in the labor costs, material costs and overhead costs may adversely affect ProFrac's operating results. ProFrac does not believe that inflation has had a material impact on ProFrac's financial position or results of operations to date; however, a high rate of inflation, including a continuation of inflation at the current rate, may have an adverse effect on ProFrac's reputation, business, financial condition, cash flows and results of operations.

#### ProFrac's indebtedness and liquidity needs could restrict ProFrac's operations and make ProFrac more vulnerable to adverse economic conditions.

ProFrac's existing and future indebtedness, whether incurred in connection with acquisitions, operations or otherwise, and limited access to liquidity may adversely affect ProFrac's operations and limit ProFrac's growth, and ProFrac may have difficulty making debt service payments on such indebtedness as payments become due. ProFrac's level of indebtedness may affect ProFrac's operations in several ways, including the following:

- · increasing ProFrac's vulnerability to general adverse economic and industry conditions;
- the covenants that are contained in the agreements governing ProFrac's indebtedness could limit ProFrac's ability to borrow funds, dispose of
  assets, pay dividends and make certain investments;
- ProFrac's debt covenants could also affect ProFrac's flexibility in planning for, and reacting to, changes in the economy and in ProFrac's industry;
- any failure to comply with the financial or other debt covenants, including covenants that impose requirements to maintain certain financial ratios, could result in an event of default, which could result in some or all of ProFrac's indebtedness becoming immediately due and payable;
- ProFrac's level of debt could impair ProFrac's ability to obtain additional financing, or obtain additional financing on favorable terms, in the
  future for working capital, capital expenditures, acquisitions or other general corporate purposes; and
- ProFrac's business may not generate sufficient cash flow from operations to enable ProFrac to meet ProFrac's obligations under ProFrac's indebtedness.

# Restrictions in ProFrac's debt agreements and any future financing agreements may limit ProFrac's ability to finance future operations, meet capital needs or capitalize on potential acquisitions and other business opportunities.

The operating and financial restrictions and covenants in existing and future debt agreements could restrict ProFrac's ability to finance future operations, meet capital needs or to expand or pursue ProFrac's business activities. For example, ProFrac's debt agreements will restrict or limit ProFrac's ability to:

- · grant liens;
- · incur additional indebtedness;
- · engage in a merger, consolidation or dissolution;
- enter into transactions with affiliates;
- sell or otherwise dispose of assets, businesses and operations;
- materially alter the character of ProFrac's business as conducted at the closing of this offering; and
- make acquisitions, investments and capital expenditures and pay dividends.

Furthermore, ProFrac's debt agreements contain certain other operating and financial covenants. ProFrac's ability to comply with the covenants and restrictions contained in ProFrac's debt agreements may be affected by events beyond ProFrac's control, including prevailing economic, financial and industry conditions. If market or other economic conditions deteriorate, ProFrac's ability to comply with these covenants may be impaired. If ProFrac violates any of the restrictions, covenants, ratios or tests in ProFrac's debt agreements, a significant portion of ProFrac's indebtedness may become immediately due and payable, and ProFrac's lenders' commitment to make further loans to ProFrac may terminate. ProFrac might not have, or be able to obtain, sufficient funds to make these accelerated payments. Any subsequent replacement of ProFrac's debt agreements or any new indebtedness could have similar or greater restrictions.

An increase in interest rates would increase the cost of servicing ProFrac's indebtedness and could reduce ProFrac's profitability, decrease ProFrac's liquidity and impact ProFrac's solvency.

A number of ProFrac's existing debt agreements provide for, and ProFrac's future debt agreements may provide for, debt incurred thereunder to bear interest at variable rates. As a result, increases in interest rates could increase the cost of servicing such indebtedness and materially reduce ProFrac's profitability and cash flows.

ProFrac's operations are subject to unforeseen interruptions and hazards inherent in the oil and natural gas industry, for which ProFrac may not be adequately insured and which could cause ProFrac to lose customers and substantial revenue.

ProFrac's operations are exposed to the risks inherent to ProFrac's industry, such as equipment defects, vehicle accidents, fires, explosions, blowouts, surface cratering, uncontrollable flows of gas or well fluids, pipe or pipeline failures, abnormally pressured formations and various environmental hazards, such as oil spills and releases of, and exposure to, hazardous substances. For example, ProFrac's operations are subject to risks associated with hydraulic fracturing, including any mishandling, surface spillage or potential underground migration of fracturing fluids, including chemical additives. In addition, ProFrac's operations are exposed to potential natural disasters, including blizzards, tornadoes, storms, floods, other adverse weather conditions and earthquakes. The occurrence of any of these events could result in substantial losses to ProFrac due to injury or loss of life, severe damage to or destruction of property, natural resources and equipment, pollution or other environmental damage, clean-up responsibilities, regulatory investigations and penalties or other damage resulting in curtailment or suspension of ProFrac's operations. The cost of managing such risks may be significant. The frequency and severity of such incidents will affect operating costs, insurability and relationships with customers, employees and regulators. In particular, ProFrac's customers may elect not to purchase ProFrac's services if they view ProFrac's environmental or safety record as unacceptable, which could cause ProFrac to lose customers and substantial revenues.

ProFrac's insurance may not be adequate to cover all losses or liabilities ProFrac may suffer. Furthermore, ProFrac may be unable to maintain or obtain insurance of the type and amount ProFrac desires at reasonable rates. As a result of market conditions, premiums and deductibles for certain of ProFrac's insurance policies have increased and could escalate further. In addition, sub-limits have been imposed for certain risks. In some instances, certain insurance could become unavailable or available only for reduced amounts of coverage. If ProFrac were to incur a significant liability for which ProFrac is not fully insured, it could have a material adverse effect on ProFrac's business, results of operations and financial condition. In addition, ProFrac may not be able to secure additional insurance or bonding that might be required by new governmental regulations. This may cause ProFrac to restrict ProFrac's operations, which might severely impact ProFrac's financial position.

Since hydraulic fracturing activities are part of ProFrac's operations, they are covered by ProFrac's insurance against claims made for bodily injury, property damage and clean-up costs stemming from a sudden and accidental pollution event. However, ProFrac may not have coverage if ProFrac is unaware of the pollution event and unable to report the "occurrence" to ProFrac's insurance company within the time frame required under ProFrac's insurance policy. In addition, these policies do not provide coverage for all liabilities, and the insurance coverage may not be adequate to cover claims that may arise, or ProFrac may not be able to maintain adequate insurance at rates ProFrac considers reasonable. A loss not fully covered by insurance could have a material adverse effect on ProFrac's financial position, results of operations and cash flows.

Inaccuracies in ProFrac's estimates of mineral reserves and resource deposits, or deficiencies in ProFrac's title to those deposits, could result in ProFrac's inability to mine the deposits or require ProFrac to pay higher than expected costs.

ProFrac bases its mineral reserve and resource estimates on engineering, economic and geological data assembled and analyzed by ProFrac's mining engineers, which are reviewed periodically by outside firms.

However, commercial silica reserve estimates are necessarily imprecise and depend to some extent on statistical inferences drawn from available drilling data, which may prove unreliable. There are numerous uncertainties inherent in estimating quantities and qualities of commercial silica reserves and non-reserve commercial silica deposits and costs to mine recoverable reserves, many of which are beyond ProFrac's control and any of which could cause actual results to differ materially from ProFrac's expectations. These uncertainties include:

- geological and mining conditions and/or effects from prior mining that may not be fully identified by available data or that may differ from experience;
- assumptions regarding the effectiveness of ProFrac's mining, quality control and training programs;
- assumptions concerning future prices of commercial silica products, operating costs, mining technology improvements, development costs and reclamation costs; and
- assumptions concerning future effects of regulation, including the issuance of required permits and taxes by governmental agencies.

In addition, title to, and the area of, mineral properties and water rights may also be disputed. Mineral properties sometimes contain claims or transfer histories that examiners cannot verify. A successful claim that ProFrac does not have title to one or more of ProFrac's properties or lack appropriate water rights could cause ProFrac to lose any rights to explore, develop and extract any minerals on that property, without compensation for ProFrac's prior expenditures relating to such property. Any inaccuracy in ProFrac's estimates related to ProFrac's mineral reserves and non-reserve mineral deposits, or ProFrac's title to such deposits, could result in ProFrac's inability to mine the deposits or require ProFrac to pay higher than expected costs.

Additionally, a portion of ProFrac's Alpine reserves are located on approximately 630 acres that ProFrac leases pursuant to a lease that terminates in 2052 and requires that ProFrac commence production from the leased premises by January 1, 2032. If ProFrac does not commence mining activities by January 1, 2032, ProFrac's lease of this property would terminate and ProFrac would lose its interest in these reserves.

### A terrorist attack or armed conflict could harm ProFrac's business.

Terrorist activities, anti-terrorist efforts and other armed conflicts involving the United States could adversely affect the U.S. and global economies and could prevent ProFrac from meeting financial and other obligations. ProFrac could experience loss of business, delays or defaults in payments from payors or disruptions of fuel supplies and markets if pipelines, production facilities, processing plants, refineries or transportation facilities are direct targets or indirect casualties of an act of terror or war. Such activities could reduce the overall demand for oil and natural gas, which, in turn, could also reduce the demand for ProFrac's services. Terrorist activities and the threat of potential terrorist activities and any resulting economic downturn could adversely affect ProFrac's results of operations, impair ProFrac's ability to raise capital or otherwise adversely impact ProFrac's ability to realize certain business strategies.

## Increasing trucking regulations may increase ProFrac's costs and negatively impact ProFrac's results of operations.

In connection with ProFrac's business operations, including the transportation and relocation of ProFrac's hydraulic fracking equipment and shipment of frac sand, ProFrac operates trucks and other heavy equipment. As such, ProFrac operates as a motor carrier in providing certain of ProFrac's services and therefore are subject to regulation by the United States Department of Transportation ("DOT") and by various state agencies. These regulatory authorities exercise broad powers, governing activities such as the authorization to engage in motor carrier operations, driver licensing, insurance requirements, financial reporting and review of certain mergers, consolidations and acquisitions, and transportation of hazardous materials. ProFrac's trucking operations are subject to possible regulatory and legislative changes that may increase ProFrac's costs. Some of these possible changes include increasingly stringent environmental regulations, changes in the hours of service regulations

which govern the amount of time a driver may drive or work in any specific period, onboard black box recorder device requirements or limits on vehicle weight and size.

Interstate motor carrier operations are subject to safety requirements prescribed by the DOT. To a large degree, intrastate motor carrier operations are subject to state safety regulations that mirror federal regulations. Matters such as the weight and dimensions of equipment are also subject to federal and state regulations. From time to time, various legislative proposals are introduced, including proposals to increase federal, state, or local taxes, including taxes on motor fuels, which may increase ProFrac's costs or adversely impact the recruitment of drivers. ProFrac cannot predict whether, or in what form, any increase in such taxes applicable to ProFrac will be enacted.

Certain motor vehicle operators require registration with the DOT. This registration requires an acceptable operating record. The DOT periodically conducts compliance reviews and may revoke registration privileges based on certain safety performance criteria that could result in a suspension of operations.

## ProFrac may be subject to claims for personal injury and property damage, which could materially adversely affect ProFrac's financial condition and results of operations.

ProFrac operates with most of its customers under master service agreements ("MSAs"). ProFrac endeavors to allocate potential liabilities and risks between the parties in the MSAs. Generally, under ProFrac's MSAs, including those relating to ProFrac's hydraulic fracturing services, ProFrac assumes responsibility for, including control and removal of, pollution or contamination which originates above surface and originates from ProFrac's equipment or services. ProFrac's customer assumes responsibility for, including control and removal of, all other pollution or contamination, which may occur during operations, including that which may result from seepage or any other uncontrolled flow of drilling fluids. ProFrac may have liability in such cases if ProFrac is negligent or commits willful acts. Generally, ProFrac's customers also agree to indemnify ProFrac against claims arising from their employees' personal injury or death to the extent that, in the case of ProFrac's hydraulic fracturing operations, their employees are injured or their properties are damaged by such operations, unless resulting from ProFrac's gross negligence or willful misconduct. Similarly, ProFrac generally agrees to indemnify ProFrac's customers for liabilities arising from personal injury to or death of any of ProFrac's employees, unless resulting from gross negligence or willful misconduct of the customer. In addition, ProFrac's customers generally agree to indemnify ProFrac for loss or destruction of customer-owned property or equipment and in turn, ProFrac agrees to indemnify its customers for loss or destruction of property or equipment ProFrac owns. Losses due to catastrophic events, such as blowouts, are generally the responsibility of the customer. However, despite this general allocation of risk, ProFrac might not succeed in enforcing such contractual allocation, might incur an unforeseen liability falling outside the scope of such allocation or may be required to enter into an MSA with terms that vary from the above allocations of risk. Litigation arising from a catastrophic occurrence at a location where ProFrac's equipment and services are being used may result in ProFrac being named as a defendant in lawsuits asserting large claims. As a result, ProFrac may incur substantial losses which could materially and adversely affect ProFrac's financial condition and results of operation.

# ProFrac is subject to cyber security risks. A cyber incident could occur and result in information theft, data corruption, operational disruption and/or financial loss.

The oil and natural gas industry has become increasingly dependent on digital technologies to conduct certain processing activities. For example, ProFrac depends on digital technologies to perform many of ProFrac's services and process and record operational and accounting data. At the same time, cyber incidents, including deliberate attacks or unintentional events, have increased. The U.S. government has issued public warnings that indicate that energy assets might be specific targets of cyber security threats. ProFrac's technologies, systems and networks, and those of ProFrac's vendors, suppliers and other business partners, may become the target of cyberattacks or information security breaches that could result in the unauthorized release, gathering, monitoring, misuse, loss or destruction of proprietary and other information, or other disruption of ProFrac's business

operations. In addition, certain cyber incidents, such as surveillance, may remain undetected for an extended period. ProFrac's systems and insurance coverage for protecting against cyber security risks may not be sufficient. As cyber incidents continue to evolve, ProFrac may be required to expend additional resources to continue to modify or enhance ProFrac's protective measures or to investigate and remediate any vulnerability to cyber incidents. ProFrac's insurance coverage for cyberattacks may not be sufficient to cover all the losses ProFrac may experience as a result of such cyberattacks.

## If ProFrac is unable to fully protect ProFrac's intellectual property rights, ProFrac may suffer a loss in ProFrac's competitive advantage or market share.

While ProFrac has acquired licenses from USWS to construct electric-powered hydraulic fracturing fleets utilizing Clean Flee® technology, ProFrac does not have patents or patent applications relating to many of ProFrac's key processes and technology. If ProFrac is not able to maintain the confidentiality of ProFrac's trade secrets, or if ProFrac's competitors are able to replicate ProFrac's technology or services, ProFrac's competitive advantage would be diminished. ProFrac also cannot ensure that any patents ProFrac may obtain in the future would provide ProFrac with any significant commercial benefit or would allow ProFrac to prevent ProFrac's competitors from employing comparable technologies or processes.

### ProFrac may be adversely affected by disputes regarding intellectual property rights of third parties.

Third parties from time to time may initiate litigation against ProFrac by asserting that the conduct of ProFrac's business infringes, misappropriates or otherwise violates intellectual property rights. ProFrac may not prevail in any such legal proceedings related to such claims, and ProFrac's products and services may be found to infringe, impair, misappropriate, dilute or otherwise violate the intellectual property rights of others. If ProFrac is sued for infringement and loses, ProFrac could be required to pay substantial damages and/or be enjoined from using or selling the infringing products or technology. Any legal proceeding concerning intellectual property could be protracted and costly regardless of the merits of any claim and is inherently unpredictable and could have a material adverse effect on ProFrac's financial condition, regardless of its outcome.

If ProFrac were to discover that its technologies or products infringe valid intellectual property rights of third parties, ProFrac may need to obtain licenses from these parties or substantially re-engineer its products in order to avoid infringement. ProFrac may not be able to obtain the necessary licenses on acceptable terms, or at all, or be able to re-engineer its products successfully. If ProFrac's inability to obtain required licenses for its technologies or products prevents ProFrac from selling its products, that could adversely impact ProFrac's financial condition and results of operations.

Additionally, ProFrac currently licenses certain third-party intellectual property in connection with its business, and the loss of any such license could adversely impact ProFrac's financial condition and results of operations.

# Seasonal weather conditions, natural disasters, public health crises, and other catastrophic events outside of ProFrac's control could severely disrupt normal operations and harm ProFrac's business.

ProFrac's operations are located in different regions of the United States. Some of these areas are adversely affected by seasonal weather conditions, primarily in the winter and spring. However, as evidenced by the severe winter weather experienced in the southern United States and Canada during February 2021, weather-related hazards can exist in almost all the areas where ProFrac operates. During periods of heavy snow, ice or rain, ProFrac may be unable to move its equipment between locations or obtain adequate supplies of raw material or fuel, thereby reducing ProFrac's ability to provide services and generate revenues. The exploration activities of ProFrac's customers may also be affected during such periods of adverse weather conditions. Additionally, extended drought conditions in ProFrac's operating regions could impact ProFrac's ability or ProFrac's customers' ability to source sufficient water or increase the cost for such water. As a result, a natural disaster or

inclement weather conditions could severely disrupt the normal operation of ProFrac's business and adversely impact ProFrac's financial condition and results of operations. Climate change may exacerbate the likelihood or intensity of such natural disasters or inclement weather conditions. Furthermore, if an area in which ProFrac operates or the market demand for oil and natural gas is affected by a public health crisis, such as COVID-19, or other similar catastrophic event outside of ProFrac's control, ProFrac's business and results of operations could suffer.

ProFrac's fleet includes substantial legacy capacity that may require increased levels of maintenance and capital expenditures to be maintained in good operating condition, is less efficient, and may be subject to a higher likelihood of mechanical failure, an inability to economically return to service or requirement to be scrapped. If ProFrac is unable to manage retiring some portion of ProFrac's fleet efficiently, or if ProFrac is unable meet the changing needs of ProFrac's customers, ProFrac's results will deteriorate and ProFrac's financial position and cash flows could be materially adversely affected.

As of June 30, 2022, approximately 40% of ProFrac's pumps have Tier IV engines and approximately 50% have dual fuel capabilities. Many of the fleets ProFrac acquired in the FTSI Acquisition are substantially older. This legacy portion of ProFrac's fleet is generally less technologically advanced than ProFrac's upgraded fleets, may require additional maintenance and capital expenditures to be kept in good operating condition and as a consequence may be subject to longer or more frequent periods of unavailability. Prolonged periods of unavailability of one or more of ProFrac's older fleets could have a material adverse effect on ProFrac's financial position, results of operations and cash flows. In addition, ProFrac expects that the fleets acquired in the FTSI Acquisition may be less attractive and less fuel efficient than ProFrac's competitors' newer fleets, putting ProFrac at a competitive disadvantage. While ProFrac is retiring 650,000 HHP of FTSI's older, emissions-intensive fleets, ProFrac may be unable to successfully replace those retired fleets with comparable production that meets ProFrac's lower-emissions profile or ProFrac's desired rate of return for ProFrac's fleets.

ProFrac's financial results may be materially adversely affected by the inclusion of Flotek's financial statements in ProFrac's consolidated financial statements, and ProFrac does not have the benefit of Flotek's cash or liquidity.

Due to ProFrac's determination that Flotek is a variable interest entity of which ProFrac is the primary beneficiary, Flotek's financial statements have been included in ProFrac's consolidated financial statements from May 17, 2022. Consequently, ProFrac's financial results may be materially adversely affected if Flotek reports poor or worsened financial results. In addition, ProFrac does not have the ability to access or deploy Flotek's cash or liquidity in its operations, which may limit ProFrac's ability to mitigate the impact of the inclusion of Flotek's financial statements in its consolidated financial statements.

## Risks Related to Environmental and Regulatory Matters

ProFrac's operations and the operations of ProFrac's customers are subject to environmental, health and safety laws and regulations, and future compliance, claims, and liabilities relating to such matters may have a material adverse effect on ProFrac's results of operations, financial position or cash flows.

The nature of ProFrac's operations, and those of ProFrac's customers, including the handling, transporting and disposing of a variety of fluids and substances, including hydraulic fracturing fluids and other regulated substances, air emissions, and wastewater discharges exposes ProFrac and ProFrac's customers to some risk of environmental liability, including the release of pollutants from oil and natural gas wells and associated equipment to the environment. ProFrac is also subject to laws and regulations associated with sand mining and equipment manufacturing operations, including the processing, and the related storage, handling, transportation and disposal of raw materials, products and wastes. The cost of compliance with these laws can be significant. Failure to properly handle, transport or dispose of these materials or otherwise conduct ProFrac's operations in accordance with these and other environmental, health and safety laws could expose ProFrac to substantial

liability for administrative, civil and criminal penalties, cleanup and site restoration costs and liability associated with releases of such materials, damages to natural resources and other damages, as well as potentially impair ProFrac's ability to conduct ProFrac's operations. Such liability is commonly on a strict, joint and several liability basis, without regard to fault. Liability may be imposed as a result of ProFrac's conduct that was lawful at the time it occurred or the conduct of, or conditions caused by, prior operators or other third parties. Neighboring landowners and other third parties may file claims against ProFrac for personal injury or property damage allegedly caused by the release of pollutants into the environment. Environmental, health and safety laws and regulations have changed in the past, and they may change in the future and become more stringent. Current and future claims and liabilities may have a material adverse effect on ProFrac because of potential adverse outcomes, defense costs, diversion of management resources, unavailability of insurance coverage and other factors. The ultimate costs of these liabilities are difficult to determine and may exceed any reserves ProFrac may have established. If existing environmental, health and safety requirements or enforcement policies change, ProFrac may be required to make significant unanticipated capital and operating expenditures. For more information, see "Information About ProFrac — Environmental and Occupational Health and Safety Regulations."

### ProFrac's operations, and those of ProFrac's customers, are subject to a series of risks arising from climate change.

Climate change continues to attract considerable public and scientific attention. As a result, numerous proposals have been made and are likely to continue to be made at the international, national, regional and state levels of government to monitor and limit emissions of carbon dioxide, methane and other GHGs. These efforts have included consideration of cap-and-trade programs, carbon taxes, GHG reporting and tracking programs and regulations that directly limit GHG emissions from certain sources.

In the United States, no comprehensive climate change legislation has been implemented at the federal level. However, President Biden has established addressing climate change as a priority of his administration and has issued several executive orders addressing climate change. Moreover, following the U.S. Supreme Court finding that GHG emissions constitute a pollutant under the federal Clean Air Act ("CAA"), the EPA has adopted regulations that, among other things, establish construction and operating permit reviews for GHG emissions from certain large stationary sources, require the monitoring and annual reporting of GHG emissions from certain petroleum and natural gas system sources in the United States, and together with the DOT, set GHG emissions and fuel economy standards for vehicles in the United States. The regulation of methane from oil and gas facilities has been subject to uncertainty in recent years. The EPA previously had promulgated new source performance standards ("NSPS") imposing limitations on methane emissions from sources in the oil and gas sector. Subsequently, in September 2020, the Trump Administration rescinded those methane standards and removed the transmission and storage segments from the oil and gas source category under the CAA's NSPS. However, in June 2021, President Biden signed a resolution passed by the U.S. Congress under the Congressional Review Act nullifying the September 2020 rule, effectively reinstating the prior standards. In November 2021, as required by President Biden's executive order, the EPA proposed new regulations to expand NSPS requirements for oil and gas sector sources and establish comprehensive standards of performance and emission guidelines for methane and volatile organic compound emissions from existing operations in the oil and gas sector, including the exploration and production, transmission, processing, and storage segments. The EPA has announced that the agency hopes to finalize these rulemakings by the end of 2022. Once finalized, the regulations are likely to be subject to legal challenge and will also need to be incorporated into the states' implementation plans, which will need to be approved by the EPA in individual rulemakings that could also be subject to legal challenge. The reinstatement of direct regulation of methane emission for new sources and the promulgation of requirements for existing oil and gas customers could result in increased costs for ProFrac's customers and consequently adversely affect demand for ProFrac's services.

Separately, various states and groups of states have adopted or are considering adopting legislation, regulation or other regulatory initiatives that are focused on such areas as GHG cap and trade programs, carbon

taxes, reporting and tracking programs, and restriction of emissions. For example, several states, including Pennsylvania and New Mexico, have proposed or adopted regulations restricting the emission of methane from E&P activities. At the international level, the United Nations-sponsored "Paris Agreement" requires member states to submit non-binding, individually-determined reduction goals known as Nationally Determined Contributions every five years after 2020. President Biden has recommitted the United States to the Paris Agreement and, in April 2021, announced a goal of reducing the United States' emissions by 50-52% below 2005 levels by 2030. In November 2021, President Biden released "The Long-Term Strategy of the United States: Pathways to Net-Zero Greenhouse Gas Emissions by 2050," which, among other things, explains that the U.S. and EUare co-leading the "Global Methane Pledge" that aims to cut global methane pollution at least 30% by 2030 relative to 2020 levels. The impacts of these orders, pledges, agreements, and any legislation or regulation promulgated to fulfill the United States' commitments under the Paris Agreement, cannot be predicted at this time.

Governmental, scientific, and public concern over the threat of climate change arising from GHG emissions has resulted in increasing political risks in the United States, including climate change related pledges made by certain candidates now in public office. On January 27, 2021, President Biden issued an executive order that called for substantial action on climate change, including, among other things, the increased use of zero-emission vehicles by the federal government, the elimination of subsidies provided to the fossil fuel industry, and increased emphasis on climate-related risks across government agencies and economic sectors. The Biden Administration has also issued orders temporarily suspending the issuance of authorizations, and suspending the issuance of new leases pending a study, for oil and gas development on federal lands. For more information, see ProFrac's regulatory disclosure titled "Information About ProFrac — Regulation of Hydraulic Fracturing and Related Activities." As a result, ProFrac cannot predict the full impact of these developments or whether the Biden Administration may pursue further restrictions. Other actions that could be pursued by the Biden Administration may include the imposition of more restrictive requirements for the establishment of pipeline infrastructure or the permitting of LNG export facilities, as well as more restrictive GHG emission limitations for oil and gas facilities. Litigation risks are also increasing as a number of entities have sought to bring suit against various oil and natural gas companies in state or federal court, alleging among other things, that such companies created public nuisances by producing fuels that contributed to climate change or alleging that the companies have been aware of the adverse effects of climate change for some time but defrauded their investors or customers by failing to adequately disclose those impacts.

There are also increasing financial risks for fossil fuel producers as shareholders currently invested in fossil-fuel energy companies may elect in the future to shift some or all of their investments into non-fossil fuel related sectors. Institutional lenders who provide financing to fossil fuel energy companies also have become more attentive to sustainable lending practices and some of them may elect not to provide funding for fossil fuel energy companies. There is also a risk that financial institutions will be required to adopt policies that have the effect of reducing the funding provided to the fossil fuel sector. Recently, President Biden signed an executive order calling for the development of a "climate finance plan" and, separately, the Federal Reserve announced that is has joined the Network for Greening the Financial System, a consortium of financial regulators focused on addressing climate-related risks in the financial sector. Limitation of investments in and financing for fossil fuel energy companies could result in the restriction, delay or cancellation of drilling programs or development or production activities. Additionally, the SEC recently proposed new rules relating to the disclosure of a range of climate-related risks. ProFrac is currently assessing this rule but at this time ProFrac cannot predict the costs of implementation or any potential adverse impacts resulting from the rule. To the extent this rule is finalized as proposed, ProFrac or ProFrac's customers could incur increased costs related to the assessment and disclosure of climate-related risks. In addition, enhanced climate disclosure requirements could accelerate the trend of certain stakeholders and lenders restricting or seeking more stringent conditions with respect to their investments in certain carbon intensive sectors.

The adoption and implementation of new or more stringent international, federal or state legislation, regulations or other regulatory initiatives that impose more stringent standards for GHG emissions from the oil

and natural gas sector or otherwise restrict the areas in which this sector may produce oil and natural gas or generate the GHG emissions could result in increased costs of compliance or costs of consuming, and thereby reduce demand for oil and natural gas, which could reduce demand for ProFrac's services. Additionally, political, litigation and financial risks may result in ProFrac's customers restricting or cancelling production activities, incurring liability for infrastructure damages as a result of climatic changes, or impairing their ability to continue to operate in an economic manner, which also could reduce the demand for ProFrac's services. One or more of these developments could have a material adverse effect on ProFrac's business, financial condition and results of operation. Finally, many scientists have concluded that increasing concentrations of GHG in the atmosphere may produce climate changes that have significant physical effects, such as increased frequency and severity of storms, droughts, and floods and other climate events that could have an adverse effect on ProFrac's customers' operations.

Federal, state and local legislative and regulatory initiatives relating to hydraulic fracturing as well as governmental reviews and investment practices for such activities may serve to limit future oil and natural gas E&P activities and could have a material adverse effect on ProFrac's results of operations and business.

Various federal, state and local legislative and regulatory initiatives have been, or could be undertaken which could result in additional requirements or restrictions being imposed on hydraulic fracturing operations. Currently, hydraulic fracturing is generally exempt from federal regulation under the Safe Drinking Water Act Underground Injection Control (the "SDWA UIC") program and is typically regulated by state oil and gas commissions or similar agencies. However, certain federal agencies have increased scrutiny and regulation. For example, in late 2016, the EPA released a final report on the potential impacts of hydraulic fracturing on drinking water resources, concluding that "water cycle" activities associated with hydraulic fracturing may impact drinking water resources under certain limited circumstances. Additionally, the EPA has asserted regulatory authority pursuant to the SDWA UIC program over hydraulic fracturing activities involving the use of diesel fuel in the fracturing fluid and issued guidance of such activities. Furthermore, the U.S. Bureau of Land Management (the "BLM") published a final rule in 2015 that established stringent standards relating to hydraulic fracturing on federal and Native American lands. The rule was rescinded, but the rescission is currently on appeal to the U.S. Court of Appeals for the Ninth Circuit. Similarly, the EPA has adopted rules on the capture of methane and other emissions released during hydraulic fracturing. In addition to federal regulatory actions, legislation has been introduced, but not enacted, in U.S. Congress to provide for federal regulation of hydraulic fracturing and to require disclosure of the chemicals used in the hydraulic fracturing process.

Separately, the Biden Administration has taken action to restrict E&P activities, including hydraulic fracturing, on public lands. For more information, see "Information About ProFrac — Environmental and Occupational Health and Safety Regulations—Regulation of Hydraulic Fracturing and Related Activities."

Many states and local governments have also adopted regulations that impose more stringent permitting, disclosure, disposal and well-construction requirements on hydraulic fracturing operations, including states where ProFrac or ProFrac's customers operate, such as Texas, Colorado and North Dakota. States could also elect to place prohibitions on hydraulic fracturing, as several states have already done. In addition, some states have adopted broader sets of requirements related to oil and gas development more generally that could impact hydraulic fracturing activities. Separately, state and federal regulatory agencies have at times focused on a possible connection between hydraulic fracturing related activities, including the underground injection of wastewater into disposal wells, and the increased occurrence of seismic activity. Regulators in some states have imposed, or are considering imposing, additional requirements in the permitting of produced water disposal wells or otherwise to assess any relationship between seismicity and the use of such wells. To the extent any new regulations are adopted to restrict hydraulic fracturing activities or the disposal of fluids associated with such activities, it may adversely affect ProFrac's customers and, as a result, demand for ProFrac's services. For more information see "Information About ProFrac — Environmental and Occupational Health and Safety Regulations—Regulation of Hydraulic Fracturing and Related Activities.

Increased regulation and attention given to the hydraulic fracturing process could lead to greater opposition to, and litigation concerning, oil and natural gas production activities using hydraulic fracturing techniques.

Additional legislation or regulation could also lead to operational delays for ProFrac's customers or increased operating costs in the production of oil and natural gas, including from the developing shale plays, or could make it more difficult for ProFrac and ProFrac's customers to perform hydraulic fracturing. The adoption of any additional laws or regulations regarding hydraulic fracturing or further restrictions on the availability of capital for hydraulic fracturing could potentially cause a decrease in the completion of new oil and natural gas wells and an associated decrease in demand for ProFrac's services and increased compliance costs and time. Such a decrease could have a material adverse effect on ProFrac's liquidity, consolidated results of operations, and consolidated financial condition. Moreover, the increased competitiveness of alternative energy sources (such as wind, solar, geothermal, tidal and biofuels) or increased focus on reducing the use of combustion engines in transportation (such as governmental mandates that ban the sale of new gasoline-powered automobiles) could reduce demand for hydrocarbons and therefore for ProFrac's services, which would lead to a reduction in ProFrac's revenues.

### Conservation measures, commercial development and technological advances could reduce demand for oil and natural gas and ProFrac's services.

Fuel conservation measures, alternative fuel requirements, increasing consumer demand for alternatives to oil and natural gas, technological advances in fuel economy and energy generation devices could reduce demand for oil and natural gas, resulting in reduced demand for oilfield services. The impact of the changing demand for oil and natural gas services and products may have a material adverse effect on ProFrac's business, financial condition, results of operations and cash flows.

The commercial development of economically-viable alternative energy sources and related products (such as electric vehicles, wind, solar, geothermal, tidal, fuel cells and biofuels) could have a similar effect. In addition, certain U.S. federal income tax deductions currently available with respect to oil and natural gas exploration and development, including the allowance of percentage depletion for oil and natural gas properties, may be eliminated as a result of proposed legislation. Any future decreases in the rate at which oil and natural gas reserves are discovered or developed, whether due to the passage of legislation, increased governmental regulation leading to limitations, or prohibitions on exploration and drilling activity, including hydraulic fracturing, or other factors, could have a material adverse effect on ProFrac's business and financial condition, even in a stronger oil and natural gas price environment.

## Additional restrictions on drilling activities intended to protect certain species of wildlife may adversely affect ProFrac's ability to conduct completion activities.

In the United States, the Endangered Species Act (the 'ESA") restricts activities that may affect endangered or threatened species or their habitats. Similar protections are offered to migratory birds under the Migratory Bird Treaty Act (the "MBTA"). To the extent species that are listed under the ESA or similar state laws, or are protected under the MBTA, inhabit the areas where ProFrac or ProFrac's customers operate, ProFrac's operations and the operations of ProFrac's customers could be adversely impacted. Moreover, drilling activities may be delayed, restricted or precluded in protected habitat areas or during certain seasons, such as breeding and nesting seasons. The listing of new species under the ESA in the areas where ProFrac's customers operate similarly has the potential to adversely impact ProFrac's operations and demand for ProFrac's services as a result of restrictions on oil and gas activities. For example, recently there have been renewed calls to review protections currently in place for the dunes sagebrush lizard, whose habitat includes parts of the Permian Basin, and to reconsider listing the species under the ESA, and, separately, a lawsuit has been filed to list the eastern hellbender salamander, whose habitat includes parts of the Appalachian Basin. Additionally, on June 1, 2021, U.S. Fish & Wildlife Service (the "FWS") proposed to list two distinct population segments of the lesser prairie-chicken under the ESA. Various stakeholders have, in consultation with the FWS, developed a voluntary conservation plan to protect the dunes sagebrush lizard habitat and limit disturbance of the dunes sagebrush

lizard by participants' activities. The voluntary conservation plan is known as a Candidate Conservation Agreement with Assurances (\*CCAA\*). ProFrac has joined the CCAA in an effort to mitigate potential impacts on ProFrac's business of a listing of the dunes sagebrush lizard by the FWS.

In addition, as a result of one or more settlements approved by the FWS, the agency was required to make a determination on the listing of numerous other species as endangered or threatened under the ESA by the end of the FWS' 2017 fiscal year. The FWS did not meet that deadline, but continues to evaluate whether to take action with respect to those species. Separately, on March 23, 2022, the FWS proposed a rule to redesignate the northern long-eared bat from a threatened species to an endangered species under the ESA. The designation of previously unidentified endangered or threatened species could cause ProFrac's operations to become subject to operating restrictions or bans, and limit future development activity in affected areas. The FWS and similar state agencies may designate critical or suitable habitat areas that they believe are necessary for the survival of threatened or endangered species. In October 2021, the Biden administration published two rules that reversed changes made by the Trump administration, namely to the definition of "habitat" and a policy that made it easier to exclude territory from critical habitat. It is possible these rules could increase the portion of ProFrac's customers' operating areas that could be designated as critical habitat. Such a designation could materially restrict use of or access to federal, state and private lands.

### Risks Related to ProFrac's Class A Common Stock

ProFrac Holding Corp. is a holding company. ProFrac Holding Corp.'s only material asset is its equity interest in ProFrac LLC, and ProFrac Holding Corp. is accordingly dependent upon distributions from ProFrac LLC to pay taxes, make payments under the Tax Receivable Agreement and cover its corporate and other overhead expenses.

ProFrac Holding Corp. is a holding company and has no material assets other than its equity interest in ProFrac LLC. ProFrac Holding Corp. has no independent means of generating revenue. To the extent ProFrac LLC has available cash, ProFrac intends to cause ProFrac LLC to make (i) generally pro rata distributions to the holders of ProFrac LLC Units, including ProFrac Holding Corp., in an amount at least sufficient to allow ProFrac Holding Corp. to pay its taxes (and those of its wholly owned subsidiaries) and to make payments under the Tax Receivable Agreement it will enter into with the TRA Holders and any subsequent tax receivable agreement that it may enter into in connection with future acquisitions and (ii) non-pro rata payments to ProFrac Holding Corp. to reimburse it for its corporate and other overhead expenses. To the extent that ProFrac Holding Corp. needs funds and ProFrac LLC or its subsidiaries are restricted from making such distributions or payments under applicable law or regulation or under the terms of any current or future financing arrangements, or are otherwise unable to provide such funds, ProFrac's liquidity and financial condition could be materially adversely affected.

Moreover, because ProFrac Holding Corp. will have no independent means of generating revenue, ProFrac Holding Corp.'s ability to make tax payments under the Tax Receivable Agreement will be dependent on the ability of ProFrac LLC to make distributions to ProFrac Holding Corp. in an amount sufficient to cover ProFrac Holding Corp.'s tax obligations (and those of its wholly owned subsidiaries) and obligations under the Tax Receivable Agreement. This ability, in turn, may depend on the ability of ProFrac LLC's subsidiaries to make distributions to it. ProFrac intends that such distributions from ProFrac LLC and its subsidiaries be funded with cash from operations or from future borrowings. The ability of ProFrac LLC, its subsidiaries and other entities in which it directly or indirectly holds an equity interest to make such distributions will be subject to, among other things, (i) the applicable provisions of Texas law (or other applicable jurisdiction) that may limit the amount of funds available for distribution and (ii) restrictions in relevant debt instruments issued by ProFrac LLC or its subsidiaries and other entities in which it directly or indirectly holds an equity interest. To the extent that ProFrac Holding Corp. is unable to make payments under the Tax Receivable Agreement for any reason, such payments will be deferred and will accrue interest until paid.

Conflicts of interest could arise in the future between ProFrac, on the one hand, and Dan Wilks and Farris Wilks and entities owned by or affiliated with them, on the other hand, concerning among other things, business transactions, potential competitive business activities or business opportunities.

Conflicts of interest could arise in the future between ProFrac, on the one hand, and Dan Wilks, Farris Wilks and entities owned by or affiliated with them, on the other hand, concerning among other things, business transactions, potential competitive business activities or business opportunities. Dan Wilks, Farris Wilks and other businesses owned by or affiliated with them operating in the energy and oilfield services industries. In the normal course of business, ProFrac has engaged in transactions with some of these companies. For more information, please see "Information About ProFrac — Certain Relationships and Related Party Transactions." Furthermore, Dan Wilks, Farris Wilks and other businesses owned by or affiliated with them may now, or in the future, directly or indirectly, compete with ProFrac for investment or business opportunities.

Dan Wilks, Farris Wilks and their affiliates are not restricted from owning assets or engaging in businesses that compete directly or indirectly with ProFrac and do not have any duty to refrain from engaging, directly or indirectly, in the same or similar business activities or lines of business as ProFrac, including those business activities or lines of business deemed to be competing with ProFrac, or doing business with any of ProFrac's clients, customers or vendors

Dan Wilks, Farris Wilks or their affiliates may become aware, from time to time, of certain business opportunities (such as acquisition opportunities) and may direct such opportunities to other businesses in which they have invested, in which case ProFrac may not become aware of or otherwise have the ability to pursue such opportunities. In addition, Dan Wilks, Farris Wilks and their affiliates may dispose of their interests in energy or other oilfield services companies or other assets in the future, without any obligation to offer ProFrac the opportunity to purchase any of those interests or assets.

In any of these matters, the interests of Dan Wilks, Farris Wilks and their affiliates and other business owned by or affiliated with them may differ or conflict with the interests of ProFrac's other shareholders. Any actual or perceived conflicts of interest with respect to the foregoing could have an adverse impact on the trading price of ProFrac Class A Common Stock.

The requirements of being a public company, including compliance with the reporting requirements of the Exchange Act and the requirements of Sarbanes-Oxley, may strain ProFrac's resources, increase ProFrac's costs and distract management, and ProFrac may be unable to comply with these requirements in a timely or cost-effective manner.

As a newly public company, ProFrac is required to comply with new laws, regulations and requirements, certain corporate governance provisions of Sarbanes-Oxley, related regulations of the SEC and the requirements of Nasdaq. Complying with these statutes, regulations and requirements occupies a significant amount of time of ProFrac's Board and management and causes ProFrac to incur significant costs and expenses. As a public company, ProFrac is required to:

- · maintain a comprehensive compliance function;
- comply with rules promulgated by Nasdaq;
- · continue to prepare and distribute periodic public reports in compliance with ProFrac's obligations under the federal securities laws;
- · establish, maintain and update various internal policies, such as those relating to insider trading; and
- involve and retain to a greater degree outside counsel and accountants in the above activities.

In addition, being a public company subject to these rules and regulations may make it more difficult and more expensive for ProFrac to obtain director and officer liability insurance and ProFrac may be required to accept reduced policy limits and coverage or incur substantially higher costs to obtain the same or similar coverage. As

a result, it may be more difficult for ProFrac to attract and retain qualified individuals to serve on ProFrac's Board or as executive officers. As a newly public company, ProFrac is continuing to evaluate the impact of these rules, and ProFrac cannot predict or estimate the amount of costs ProFrac may incur or the timing of such costs.

ProFrac will be required to comply with certain provisions of Section 404 of Sarbanes-Oxley as early as ProFrac's fiscal year ending December 31, 2022. Section 404 of Sarbanes-Oxley requires that ProFrac document and test ProFrac's internal control over financial reporting and issue management's assessment of ProFrac's internal control over financial reporting. Section 404 of Sarbanes-Oxley also requires that ProFrac's independent registered public accounting firm opine on those internal controls upon becoming a large accelerated filer, as defined in the SEC rules, or otherwise ceasing to qualify as an emerging growth company under the JOBS Act. ProFrac is evaluating its existing controls against the standards adopted by the Committee of Sponsoring Organizations of the Treadway Commission. During the course of ProFrac's ongoing evaluation and integration of the internal control over financial reporting, ProFrac may identify areas requiring improvement, and ProFrac may have to design enhanced processes and controls to address issues identified through this review. For example, ProFrac anticipates the need to hire additional administrative and accounting personnel to conduct ProFrac's financial reporting.

ProFrac cannot be certain at this time that it will be able to successfully complete the procedures, certification and attestation requirements of Section 404 of Sarbanes-Oxley or that ProFrac or ProFrac's independent registered public accounting firm will not identify material weaknesses in ProFrac's internal control over financial reporting. If ProFrac fails to comply with the requirements of Section 404 of Sarbanes-Oxley or if ProFrac or ProFrac's independent registered public accounting firm identify and report such material weaknesses, the accuracy and timeliness of the filing of ProFrac's annual and quarterly reports may be materially adversely affected and could cause investors to lose confidence in ProFrac's reported financial information, which could have a negative effect on the price of ProFrac Class A Common Stock. In addition, a material weakness in the effectiveness of ProFrac's internal control over financial reporting could result in an increased chance of fraud and the loss of customers, reduce ProFrac's ability to obtain financing and require additional expenditures to comply with these requirements, each of which could have a material adverse effect on ProFrac's business, results of operations and financial condition.

## An active, liquid and orderly trading market for ProFrac Class A common stock may not develop or be maintained, and ProFrac's stock price may be volatile.

Prior to the ProFrac IPO, ProFrac Class A Common Stock was not traded on any market. An active, liquid and orderly trading market for ProFrac Class A Common Stock may not develop or be maintained. Active, liquid and orderly trading markets usually result in less price volatility and more efficiency in carrying out investors' purchase and sale orders. The market price of ProFrac Class A Common Stock could vary significantly as a result of a number of factors, some of which are beyond ProFrac's control. In the event of a drop in the market price of ProFrac Class A Common Stock, you could lose a substantial part or all of your investment in ProFrac Class A Common Stock.

The following is a non-exhaustive list of factors that could affect ProFrac's stock price:

- ProFrac's operating and financial performance;
- quarterly variations in ProFrac's financial and operating results;
- the public reaction to ProFrac's press releases, ProFrac's other public announcements and ProFrac's filings with the SEC;
- strategic actions by ProFrac's competitors;
- ProFrac's failure to meet revenue or earnings estimates by research analysts or other investors;
- · changes in revenue or earnings estimates, or changes in recommendations or withdrawal of research coverage, by equity research analysts;

- speculation in the press or investment community;
- the failure of research analysts to cover ProFrac Class A Common Stock;
- sales of ProFrac Class A Common Stock by ProFrac or other shareholders, or the perception that such sales may occur;
- changes in accounting principles, policies, guidance, interpretations or standards;
- additions or departures of key management personnel;
- actions by ProFrac's stockholders;
- · general market conditions, including fluctuations in commodity prices;
- · domestic and international economic, legal and regulatory factors unrelated to ProFrac's performance; and
- the realization of any risks described under this "Risk Factors" section.

The stock markets in general have experienced extreme volatility that has often been unrelated to the operating performance of particular companies. These broad market fluctuations may adversely affect the trading price of ProFrac Class A Common Stock. Securities class action litigation has often been instituted against companies following periods of volatility in the overall market and in the market price of a company's securities. Such litigation, if instituted against ProFrac, could result in very substantial costs, divert ProFrac's management's attention and resources and materially harm ProFrac's business, operating results and financial condition.

## The unaudited pro forma financial data included in this proxy statement/information statement/prospectus may not be representative of ProFrac's actual financial condition and results of operations in the future.

The unaudited pro forma financial data included in this proxy statement/information statement/prospectus is presented for illustrative purposes only and is not necessarily indicative of what ProFrac's actual financial position or results of operations would have been as of the dates indicated, nor is it indicative of ProFrac's future operating results or financial position. The preparation of the pro forma financial information is based upon available information and certain assumptions and estimates that ProFrac currently believes are reasonable. There may be differences between preliminary estimates in the pro forma financial information and the final accounting presentation, which could result in material differences from the pro forma information presented in this prospectus with respect to ProFrac's estimated financial position and results of operations. Accordingly, ProFrac's business, assets, cash flows, results of operations and financial condition may differ significantly from those indicated by the unaudited pro forma financial data included in this proxy statement/information statement/prospectus. In addition, the assumptions used in preparing the unaudited pro forma financial data may not prove to be accurate and other factors may affect ProFrac's financial condition or results of operations. Any potential decline in ProFrac's financial condition or results of operations may cause significant variations in ProFrac's stock price.

# The ProFrac Controlling Stockholders have the ability to direct the voting of a majority of ProFrac's voting stock, and their interests may conflict with those of ProFrac's other stockholders.

As of August 29, 2022, the ProFrac Controlling Stockholders owned approximately 87% of ProFrac's voting stock. As a result, the ProFrac Controlling Stockholders are able to control matters requiring stockholder approval, including the election of directors, changes to ProFrac's organizational documents and significant corporate transactions. This concentration of ownership makes it unlikely that any other holder or group of holders of ProFrac Class A Common Stock will be able to affect the way ProFrac is managed or the direction of ProFrac's business. The interests of the ProFrac Controlling Stockholders with respect to matters potentially or actually involving or affecting ProFrac, such as future acquisitions, financings and other corporate opportunities and attempts to acquire ProFrac, may conflict with the interests of ProFrac's other stockholders.

For example, the ProFrac Controlling Stockholders may have different tax and other positions from ProFrac, especially in light of the Tax Receivable Agreement, that could influence their decisions regarding whether and when to support the disposition of assets, the incurrence or refinancing of new or existing indebtedness, or the termination of the Tax Receivable Agreement and acceleration of ProFrac's obligations thereunder. In addition, the determination of future tax reporting positions, the structuring of future transactions and the handling of any challenge by any taxing authority to ProFrac's tax reporting positions may take into consideration tax or other considerations of the ProFrac Controlling Stockholders which may differ from the considerations of ProFrac or ProFrac's other stockholders. Please read "Information About ProFrac — Certain Relationships and Related Party Transactions—Tax Receivable Agreement."

Furthermore, in connection with the ProFrac IPO, ProFrac entered into the ProFrac Stockholders' Agreement with the ProFrac Controlling Stockholders that addresses the right to designate nominees for election to the ProFrac Board. See "Information About ProFrac — Certain Relationships and Related Party Transactions—Stockholders' Agreement." The existence of significant stockholders may have the effect of deterring hostile takeovers, delaying or preventing changes in control or changes in management, or limiting the ability of ProFrac's other stockholders to approve transactions that they may deem to be in the best interests of ProFrac. Moreover, the ProFrac Controlling Stockholders' concentration of stock ownership may adversely affect the trading price of ProFrac Class A Common Stock to the extent investors perceive a disadvantage in owning stock of a company with significant stockholders.

### A significant reduction by the ProFrac Controlling Stockholders of their ownership interests in ProFrac could adversely affect ProFrac.

ProFrac believes that the ProFrac Controlling Stockholders' substantial ownership interest in ProFrac provides them with an economic incentive to assist ProFrac to be successful. Upon the expiration or earlier waiver of the lock-up restrictions on transfers or sales of ProFrac's securities agreed to in connection with the ProFrac IPO, the ProFrac Controlling Stockholders will not be subject to any obligation to maintain their ownership interest in ProFrac and may elect at any time thereafter to sell all or a substantial portion of or otherwise reduce their ownership interest in ProFrac. If the ProFrac Controlling Stockholders sell all or a substantial portion of their ownership interests in ProFrac, they may have less incentive to assist in ProFrac's success. Such actions could adversely affect ProFrac's ability to successfully implement ProFrac's business strategies which could adversely affect ProFrac's cash flows or results of operations.

ProFrac's certificate of incorporation and restated bylaws, as well as Delaware law, contain provisions that could discourage acquisition bids or merger proposals, which may adversely affect the market price of ProFrac Class A Common Stock and could deprive ProFrac's investors of the opportunity to receive a premium for their shares.

ProFrac's certificate of incorporation authorizes the ProFrac Board to issue preferred stock without stockholder approval in one or more series, designate the number of shares constituting any series, and fix the rights, preferences, privileges and restrictions thereof, including dividend rights, voting rights, rights and terms of redemption, redemption price or prices and liquidation preferences of such series. If the ProFrac Board elects to issue preferred stock, it could be more difficult for a third party to acquire ProFrac. In addition, some provisions of ProFrac's certificate of incorporation and bylaws could make it more difficult for a third party to acquire control of ProFrac, even if the change of control would be beneficial to ProFrac's stockholders. These provisions include the following:

- until ProFrac ceases to be a controlled company, the members of the ProFrac Board designated by the parties to the ProFrac Stockholders' Agreement will have a majority of the voting power of ProFrac's Board;
- after ProFrac ceases to be a controlled company, dividing the ProFrac Board into three classes of directors, with each class serving staggered three-year terms;

- after ProFrac ceases to be a controlled company, and subject to the terms of the ProFrac Stockholders' Agreement, providing that all
  vacancies, including newly created directorships, may, except as otherwise required by law or, if applicable, the rights of holders of a series
  of preferred stock, only be filled by the affirmative vote of a majority of directors then in office, even if less than a quorum (prior to such
  time, vacancies may also be filled by stockholders holding a majority of the outstanding shares);
- after ProFrac ceases to be a controlled company, permitting any action by stockholders to be taken only at an annual meeting or special meeting rather than by a written consent of the stockholders, subject to the rights of any series of preferred stock with respect to such rights;
- after ProFrac ceases to be a controlled company, permitting special meetings of ProFrac's stockholders to be called only by ProFrac's Chief Executive Officer, the Executive Chairman of the ProFrac Board and the ProFrac Board pursuant to a resolution adopted by the affirmative vote of a majority of the total number of authorized directors whether or not there exist any vacancies in previously authorized directorships;
- after ProFrac ceases to be a controlled company, and subject to the rights of the holders of shares of any series of ProFrac's preferred stock and the terms of the ProFrac Stockholders' Agreement, requiring the affirmative vote of the holders of at least 66 2/3% in voting power of all then outstanding ProFrac Common Stock entitled to vote generally in the election of directors, voting together as a single class, to remove any or all of the directors from office at any time, and directors will be removable only for "cause";
- prohibiting cumulative voting in the election of directors;
- establishing advance notice provisions for stockholder proposals and nominations for elections to the ProFrac Board to be acted upon at meetings of stockholders; and
- · providing that the ProFrac Board is expressly authorized to adopt, or to alter or repeal ProFrac's bylaws.

In addition, certain change of control events will have the effect of accelerating the payments due under the Tax Receivable Agreement, which could be substantial and accordingly serve as a disincentive to a potential acquirer of the Company. Please see "Risk Factors — Risks Related to ProFrac Class A Common Stock—In certain cases, payments under the Tax Receivable Agreement may be accelerated and/or significantly exceed the actual benefits, if any, ProFrac realizes in respect of the tax attributes subject to the Tax Receivable Agreement."

ProFrac's certificate of incorporation designates the Court of Chancery of the State of Delaware as the sole and exclusive forum for certain types of actions and proceedings that may be initiated by ProFrac's Stockholders, which could limit the ability of ProFrac's Stockholders to obtain a favorable judicial forum for disputes with ProFrac or ProFrac's directors, officers, employees or agents.

ProFrac's certificate of incorporation provides that, unless ProFrac consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware, or, if such court does not have subject matter jurisdiction thereof, the federal district court of the State of Delaware, will, to the fullest extent permitted by applicable law, be the sole and exclusive forum for (i) any derivative action, suit or proceeding brought on ProFrac's behalf, (ii) any action, suit or proceeding asserting a claim of breach of a fiduciary duty owed by any of ProFrac's current or former directors, officers, employees or stockholders to ProFrac or ProFrac's Stockholders, (iii) any action, suit or proceeding asserting a claim arising pursuant to any provision of the DGCL, ProFrac's certificate of incorporation or ProFrac's bylaws (as either may be amended or restated) or as to which the DGCL confers jurisdiction on the Court of Chancery of the State of Delaware, or (iv) any action, suit or proceeding asserting a claim governed by the internal affairs doctrine, in each such case subject to such Court of

Chancery having personal jurisdiction over the indispensable parties named as defendants therein. Any person or entity purchasing or otherwise acquiring any interest in shares of ProFrac's capital stock will be deemed to have

notice of, and consented to, the provisions of ProFrac's certificate of incorporation described in the preceding sentences. This choice of forum provision may limit a stockholder's ability to bring a claim in a judicial forum that it finds favorable for disputes with ProFrac or ProFrac's directors, officers, employees or stockholders, which may discourage such lawsuits against ProFrac and such persons. However, stockholders will not be deemed to have waived ProFrac's compliance with the federal securities laws and the rules and regulations thereunder and the exclusive forum provision would not apply to suits brought to enforce a duty or liability created by the Securities Act or the Exchange Act.

ProFrac's certificate of incorporation also provides that the federal district courts of the United States will be the exclusive forum for any complaint asserting a cause of action under the Securities Act. Section 22 of the Securities Act creates concurrent jurisdiction for federal and state courts over all suits brought to enforce any duty or liability created by the Securities Act or the rules and regulations thereunder. Accordingly, there is uncertainty as to whether a court would enforce this forum provision providing for exclusive jurisdiction of federal district courts with respect to suits brought to enforce any duty or liability created by the Securities Act. If a court were to find these provisions of ProFrac's certificate of incorporation inapplicable to, or unenforceable in respect of, one or more of the specified types of actions or proceedings, ProFrac may incur additional costs associated with resolving such matters in other jurisdictions, which could adversely affect ProFrac's business, financial condition or results of operations.

ProFrac does not presently anticipate paying cash dividends on ProFrac Class A Common Stock and ProFrac's existing debt agreements place certain restrictions on ProFrac's ability to do so. Consequently, your only opportunity to achieve a return on your shares of ProFrac Class A Common Stock is if the price of ProFrac's Class A Common Stock appreciates.

While ProFrac looks forward to the opportunity to pay dividends in the future, ProFrac does not presently anticipate paying any cash dividends on ProFrac Class A Common Stock in the foreseeable future. In addition, ProFrac's existing debt agreements place, and ProFrac expects ProFrac's future debt agreements will place, certain restrictions on ProFrac's ability to pay cash dividends. Consequently, unless ProFrac revises its dividend policy, your only opportunity to achieve a return on your investment in ProFrac will be if you sell your ProFrac Class A Common Stock at a price greater than the value received for it in connection with the Merger. There is no guarantee that the price of ProFrac Class A Common Stock that will prevail in the market will ever exceed the value received for it in connection with the Merger.

Future sales of ProFrac Class A Common Stock in the public market, or the perception that such sales may occur, could reduce ProFrac's stock price, and any additional capital raised by ProFrac through the sale of equity or convertible securities may dilute your ownership in us.

ProFrac may issue or sell additional shares of ProFrac Class A Common Stock or securities that are convertible or exchangeable therefor. After the completion of the Merger, it is expected that ProFrac will have 54,018,996 outstanding shares of ProFrac Class A Common Stock. Following the completion of the Merger, the ProFrac Controlling Stockholders will own 30,805,623 shares of ProFrac Class A Common Stock and 97,447,865 shares of ProFrac Class B Common Stock, or approximately 82.7% of ProFrac's total outstanding shares. In addition, certain ProFrac LLC Unit Holders are party to a registration rights agreement, which requires ProFrac to effect the registration of the shares of ProFrac Class A Common Stock that they receive in exchange for their ProFrac LLC Units in certain circumstances no earlier than the expiration of the lock-up period contained in the underwriting agreement entered into in connection with the ProFrac IPO.

In connection with the ProFrac IPO, ProFrac filed a registration statement with the SEC onForm S-8 providing for the registration of 2,951,566 shares of ProFrac Class A Common Stock issued or reserved for issuance under ProFrac's long term incentive plan. Subject to the satisfaction of vesting conditions, the expiration of lock-up agreements and the requirements of Rule 144, shares registered under the registration statement onForm S-8 may be made available for resale immediately in the public market without restriction.

In addition, ProFrac is party to a registration rights agreement with the prior owners of West Munger (such agreement, the 'West Munger Registration Rights Agreement' and such persons, the "West Munger Sellers"). Under the West Munger Registration Rights Agreement, ProFrac has agreed to file a resale shelf registration statement on Form S-1 that registers under the Securities Act the offer and resale of the shares of ProFrac Class A Common Stock owned by the West Munger Sellers or their permitted transferees. ProFrac filed the resale registration statement on Form S-1 on August 2, 2022, which was declared effective by the SEC on August 5, 2022. On August 8, 2022, ProFrac filed with the SEC a prospectus pursuant to Rule 424(b) under the Securities Act in connection with the resale of the shares of ProFrac Class A Common Stock by the West Munger Sellers named therein.

ProFrac cannot predict the size of future issuances of ProFrac Class A Common Stock or securities convertible into ProFrac Class A Common Stock or the effect, if any, that future issuances and sales of shares of ProFrac Class A Common Stock will have on the market price of ProFrac Class A Common Stock. Sales of substantial amounts of ProFrac Class A Common Stock (including shares issued in connection with the Merger or another acquisition), or the perception that such sales could occur, may adversely affect prevailing market prices of ProFrac Class A Common Stock.

# ProFrac is required to make payments under the Tax Receivable Agreement for certain tax benefits that it may claim, and the amounts of such payments could be significant.

In connection with the closing of the ProFrac IPO, ProFrac entered into the Tax Receivable Agreement with the TRA Holders. This agreement generally provides for the payment by ProFrac to the TRA Holders of 85% of the net cash savings, if any, in U.S. federal, state and local income tax and franchise tax (computed using simplifying assumptions to address the impact of state and local taxes) that ProFrac actually realizes (or is deemed to realize in certain circumstances) in periods after the ProFrac IPO as a result of certain increases in tax basis available to ProFrac as a result of acquisitions of ProFrac LLC Units in connection with the ProFrac IPO or pursuant to the exercise of the Redemption Right or the Call Right (as such terms are defined in the ProFrac LLC Agreement and further described under the section "Information About ProFrac — Certain Relationships and Related Party Transactions — Redemption Rights") and certain benefits attributable to imputed interest. ProFrac will retain the benefit of the remaining 15% of any actual net cash tax savings.

The term of the Tax Receivable Agreement will continue until all tax benefits that are subject to the Tax Receivable Agreement have been utilized or expired, unless ProFrac experiences a change of control (as defined in the Tax Receivable Agreement, which includes certain mergers, asset sales, or other forms of business combinations) or the Tax Receivable Agreement otherwise terminates early (at ProFrac's election or as a result of ProFrac's breach or the commencement of bankruptcy or similar proceedings by or against ProFrac) and ProFrac makes the termination payments specified in the Tax Receivable Agreement in connection with such change of control or other early termination. In the event that the Tax Receivable Agreement is not terminated, the payments under the Tax Receivable Agreement are anticipated to commence in 2023 and to continue for 15 years after the date of the last redemption of the ProFrac LLC Units.

The payment obligations under the Tax Receivable Agreement are ProFrac's obligations and not obligations of ProFrac LLC, and ProFrac expects that the payments required to be made under the Tax Receivable Agreement will be substantial. Estimating the amount and timing of payments that may become due under the Tax Receivable Agreement is by its nature imprecise. For purposes of the Tax Receivable Agreement, net cash tax savings generally are calculated by comparing ProFrac's actual tax liability (determined by using the actual applicable U.S. federal income tax rate and an assumed combined state and local income and franchise tax rate) to the amount ProFrac would have been required to pay had it not been able to utilize any of the tax benefits subject to the Tax Receivable Agreement. The actual increases in tax basis covered by the Tax Receivable Agreement, as well as the amount and timing of any payments under the Tax Receivable Agreement, will vary depending on a number of factors, including the timing of any redemption of ProFrac LLC Units, the price of ProFrac Class A Common Stock at the time of each redemption, the extent to which such redemptions are taxable

transactions, the amount of the redeeming ProFrac LLC Unit Holder's tax basis in its ProFrac LLC Units at the time of the relevant redemption, the depreciation and amortization periods that apply to the increase in tax basis, the amount and timing of taxable income ProFrac generates in the future, the U.S. federal income tax rates then applicable, and the portion of ProFrac's payments under the Tax Receivable Agreement that constitute imputed interest or give rise to depreciable or amortizable tax basis. Any distributions made by ProFrac LLC to ProFrac in order to enable ProFrac to make payments under the Tax Receivable Agreement, as well as any corresponding pro rata distributions made to the ProFrac LLC Unit Holders, could have an adverse impact on ProFrac's liquidity.

The payments under the Tax Receivable Agreement will not be conditioned upon a TRA Holder having a continued ownership interest in ProFrac or ProFrac LLC. For additional information regarding the Tax Receivable Agreement, see "Information About ProFrac—Certain Relationships and Related Party Transactions—Tax Receivable Agreement."

In certain cases, payments under the Tax Receivable Agreement may be accelerated and/or significantly exceed the actual benefits, if any, ProFrac realizes in respect of the tax attributes subject to the Tax Receivable Agreement.

If ProFrac experiences a change of control (as defined under the Tax Receivable Agreement, which includes certain mergers, asset sales and other forms of business combinations) or the Tax Receivable Agreement otherwise terminates early (at ProFrac's election or as a result of ProFrac's breach or the commencement of bankruptcy or similar proceedings by or against ProFrac), ProFrac's obligations under the Tax Receivable Agreement would accelerate and ProFrac would be required to make an immediate payment equal to the present value of the anticipated future payments to be made by it under the Tax Receivable Agreement (determined by applying a discount rate equal to (i) the greater of (A) 0.25% and (B) the 180-Day Average Secured Overnight Financing Rate ("SOFR"), plus (ii) 150 basis points) and such payment is expected to be substantial. The calculation of anticipated future payments will be based upon certain assumptions and deemed events set forth in the Tax Receivable Agreement, including (i) that ProFrac has sufficient taxable income to fully utilize the tax benefits covered by the Tax Receivable Agreement, and (ii) that any ProFrac LLC Units (other than those held by ProFrac) outstanding on the termination date are deemed to be redeemed on the termination date. Any early termination payment may be made significantly in advance of, and may materially exceed, the actual realization, if any, of the future tax benefits to which the termination payment relates.

If ProFrac experiences a change of control (as defined under the Tax Receivable Agreement) or the Tax Receivable Agreement otherwise terminates early, ProFrac's obligations under the Tax Receivable Agreement could have a substantial negative impact on ProFrac's liquidity and could have the effect of delaying, deferring or preventing certain mergers, asset sales, or other forms of business combinations or changes of control. For example, if ProFrac had experienced a change of control or the Tax Receivable Agreement had otherwise been terminated immediately after the ProFrac IPO, the estimated termination payments would, in the aggregate, have been approximately \$600 million (calculated using a discount rate equal to (i) the greater of (A) 0.25% and (B) SOFR, plus (ii) 150 basis points, applied against an undiscounted liability of \$700 million calculated at the 21% U.S. federal corporate income tax rate and estimated applicable state and local income tax rates). The foregoing amount is merely an estimate and the actual payment could differ materially. There can be no assurance that ProFrac will be able to satisfy its obligations under the Tax Receivable Agreement.

Please read "Information About ProFrac — Certain Relationships and Related Party Transactions—Tax Receivable Agreement."

In the event that payment obligations under the Tax Receivable Agreement are accelerated in connection with certain mergers, other forms of business combinations or other changes of control, the consideration payable to holders of ProFrac Class A Common Stock could be substantially reduced.

If ProFrac experiences a change of control (as defined under the Tax Receivable Agreement, which includes certain mergers, asset sales and other forms of business combinations), ProFrac would be obligated to make a substantial immediate lump-sum payment, and such payment may be significantly in advance of, and may materially exceed, the actual realization, if any, of the future tax benefits to which the payment relates. As a result of this payment obligation, holders of ProFrac Class A Common Stock could receive substantially less consideration in connection with a change of control transaction than they would receive in the absence of such obligation. Further, any payment obligations under the Tax Receivable Agreement will not be conditioned upon the TRA Holders' having a continued interest in ProFrac or ProFrac LLC. Accordingly, the TRA Holders' interests may conflict with those of the holders of ProFrac Class A Common Stock. Please read "Risk Factors—Risks Related to ProFrac Class A Common Stock—In certain cases, payments under the Tax Receivable Agreement may be accelerated and/or significantly exceed the actual benefits, if any, ProFrac Holding Corp. realizes in respect of the tax attributes subject to the Tax Receivable Agreement" and "Information About ProFrac—Certain Relationships and Related Party Transactions—Tax Receivable Agreement."

## ProFrac will not be reimbursed for any payments made under the Tax Receivable Agreement in the event that any tax benefits are subsequently disallowed.

Payments under the Tax Receivable Agreement will be based on the tax reporting positions that ProFrac will determine. The U.S. Internal Revenue Service ("IRS") or another taxing authority may challenge all or part of the tax basis increases covered by the Tax Receivable Agreement, as well as other related tax positions ProFrac take, and a court could sustain such challenge. The TRA Holders will not reimburse ProFrac for any payments previously made under the Tax Receivable Agreement if any tax benefits that have given rise to payments under the Tax Receivable Agreement are subsequently disallowed, except that excess payments made to any TRA Holder will be netted against future payments that would otherwise be made to such TRA Holder, if any, after ProFrac's determination of such excess (which determination may be made a number of years following the initial payment and after future payments have been made). As a result, in such circumstances, ProFrac could make payments that are greater than ProFrac's actual cash tax savings, if any, and ProFrac may not be able to recoup those payments, which could materially adversely affect ProFrac's liquidity.

If ProFrac LLC were to become a publicly traded partnership taxable as a corporation for U.S. federal income tax purposes, ProFrac and ProFrac LLC might be subject to potentially significant tax inefficiencies, and

ProFrac would not be able to recover payments previously made by ProFrac under the Tax Receivable Agreement even if the corresponding tax benefits were subsequently determined to have been unavailable due to such status.

ProFrac intends to continue to operate such that ProFrac LLC does not become a publicly traded partnership taxable as a corporation for U.S. federal income tax purposes. A "publicly traded partnership" is a partnership the interests of which are traded on an established securities market or are readily tradable on a secondary market or the substantial equivalent thereof. Under certain circumstances, redemptions of ProFrac LLC Units pursuant to the Redemption Right (or acquisitions of ProFrac LLC Units pursuant to the Call Right) or other transfers of ProFrac LLC Units could cause ProFrac LLC to be treated as a publicly traded partnership. Applicable U.S. Treasury regulations provide for certain safe harbors from treatment as a publicly traded partnership, and ProFrac intends to continue to operate such that redemptions or other transfers of ProFrac LLC Units qualify for one or more such safe harbors. For example, ProFrac intends to continue to limit the number of unitholders of ProFrac LLC, and the ProFrac LLC Agreement entered into in connection with the closing of the ProFrac IPO, provides for limitations on the ability of unitholders of ProFrac LLC to transfer their ProFrac LLC Units and will provide ProFrac, as the managing member of ProFrac LLC, with the right to impose restrictions (in addition to those already in place) on the ability of unitholders of ProFrac LLC to redeem their ProFrac LLC Units pursuant to the Redemption Right to the extent ProFrac Holding Corp. believes it is necessary to ensure that ProFrac LLC will continue to be treated as a partnership for U.S. federal income tax purposes.

If ProFrac LLC were to become a publicly traded partnership taxable as a corporation for U.S. federal income tax purposes, significant tax inefficiencies might result for ProFrac and for ProFrac LLC, including as a result of ProFrac's inability to file a consolidated U.S. federal income tax return with ProFrac LLC. In addition, ProFrac might not be able to realize tax benefits covered under the Tax Receivable Agreement, and ProFrac would not be able to recover any payments previously made by ProFrac under the Tax Receivable Agreement, even if the corresponding tax benefits (including any claimed increase in the tax basis of ProFrac LLC's assets) were subsequently determined to have been unavailable.

Changes in effective tax rates, or adverse outcomes resulting from other tax increases or an examination of ProFrac's income or other tax returns, could adversely affect ProFrac's results of operations and financial condition.

Any changes in ProFrac's effective tax rates or tax liabilities could adversely affect ProFrac's results of operations and financial condition. ProFrac's future effective tax rates could be subject to volatility or adversely affected by a number of factors, including:

- · changes in the valuation of ProFrac's deferred tax assets and liabilities;
- expected timing and amount of the release of any tax valuation allowances;
- expansion into or future activities in new jurisdictions;
- the availability of tax deductions, credits, exemptions, refunds and other benefits to reduce tax liabilities;
- · tax effects of share-based compensation; and
- · changes in tax laws, tax regulations, accounting principles, or interpretations or applications thereof.

In addition, an adverse outcome arising from an examination of ProFrac's income or other tax returns could result in higher tax exposure, penalties, interest or other liabilities that could have an adverse effect on ProFrac's operating results and financial condition.

### ProFrac may issue preferred stock whose terms could adversely affect the voting power or value of ProFrac Class A Common Stock.

ProFrac's certificate of incorporation authorizes ProFrac to issue, without the approval of ProFrac's stockholders, one or more classes or series of preferred stock having such designations, preferences, limitations and relative rights, including preferences over ProFrac Class A Common Stock respecting dividends and distributions, as ProFrac's Board may determine. The terms of one or more classes or series of preferred stock could adversely impact the voting power or value of ProFrac Class A Common Stock. For example, ProFrac might grant holders of preferred stock the right to elect some number of ProFrac's directors in all events or on the happening of specified events or the right to veto specified transactions. Similarly, the repurchase or redemption rights or liquidation preferences ProFrac might assign to holders of preferred stock could affect the residual value of ProFrac Class A Common Stock

If ProFrac were deemed to be an investment company under the Investment Company Act of 1940, as amended (the "1940 Act"), applicable restrictions could make it impractical for ProFrac to continue ProFrac's business as contemplated and could have a material adverse effect on ProFrac's business, financial condition and results of operations.

Under Sections 3(a)(1)(A) and (C) of the 1940 Act, a company generally will be deemed to be an "investment company" for purposes of the 1940 Act if (i) it is, or holds itself out as being, engaged primarily, or proposes to engage primarily, in the business of investing, reinvesting or trading in securities or (ii) it engages, or proposes to engage, in the business of investing, reinvesting, owning, holding or trading in securities and it owns or proposes to acquire investment securities having a value exceeding 40% of the value of its total assets (exclusive of U.S. government securities and cash items) on an unconsolidated basis. ProFrac does not believe that ProFrac is an "investment company," as such term is defined in either of those sections of the 1940 Act.

As the sole managing member of ProFrac LLC, ProFrac controls and operates ProFrac LLC. On that basis, ProFrac believes that ProFrac's interest in ProFrac LLC is not an "investment security" as that term is used in the 1940 Act. However, if ProFrac were to cease participation in the management of ProFrac LLC, ProFrac's interest in ProFrac LLC could be deemed to be an "investment security" for purposes of the 1940 Act.

Although ProFrac and ProFrac LLC intend to continue to conduct ProFrac's operations so that ProFrac will not be deemed an investment company, if ProFrac were to be deemed an investment company, restrictions imposed by the 1940 Act, including limitations on ProFrac's capital structure and ProFrac's ability to transact with affiliates, could make it impractical for ProFrac to continue ProFrac's business as contemplated and could have a material adverse effect on ProFrac's business, financial condition and results of operations.

ProFrac expects to be a "controlled company" within the meaning of the Nasdaq rules following the completion of the Merger and, as a result, will qualify for and intend to rely on exemptions from certain corporate governance requirements.

Because the ProFrac Controlling Stockholders will own 30,805,623 shares of ProFrac Class A Common Stock and 97,447,865 ProFrac LLC Units (and an equal number of shares of ProFrac Class B Common Stock), representing approximately 82.7% of the voting power of ProFrac following the completion of the Merger, ProFrac expects to be a controlled company as of the completion of the Merger under Sarbanes-Oxley and rules of Nasdaq. Additionally, the ProFrac Controlling Stockholders are currently, and ProFrac expects that they will continue to be following the completion of the Merger, deemed a group for purposes of certain rules and regulations of the SEC as a result of the ProFrac Stockholders' Agreement. Under the Nasdaq rules, a company of which more than 50% of the voting power is held by another person or group of persons acting together is a controlled company and may elect not to comply with certain Nasdaq corporate governance requirements, including the requirements that:

- a majority of the board of directors consists of independent directors, as defined under the rules of Nasdaq;
- the nominating and governance committee be composed entirely of independent directors with a written charger addressing the committee's purpose and responsibilities; and
- the compensation committee be composed entirely of independent directors with a written charter addressing the committee's purpose and responsibilities.

These requirements will not apply to ProFrac as long as ProFrac remains a controlled company. ProFrac currently intends to continue to utilize some or all of these exemptions. Accordingly, you may not have the same protections afforded to stockholders of companies that are subject to all of the corporate governance requirements of Nasdaq. See section titled "Information about ProFrac — ProFrac's Management and Board of Directors."

For as long as ProFrac is an emerging growth company, ProFrac will not be required to comply with certain reporting requirements, including those relating to accounting standards and disclosure about ProFrac's executive compensation, that apply to other public companies.

ProFrac is classified as an "emerging growth company" under the JOBS Act. For as long as ProFrac is an emerging growth company, which may be up to five full fiscal years following the ProFrac IPO, unlike other public companies, ProFrac will not be required to, among other things: (i) provide an auditor's attestation report on management's assessment of the effectiveness of ProFrac's system of internal control over financial reporting pursuant to Section 404(b) of Sarbanes-Oxley; (ii) comply with any new requirements adopted by the Public Company Accounting Oversight Board requiring mandatory audit firm rotation or a supplement to the auditor's report in which the auditor would be required to provide additional information about the audit and the financial statements of the issuer; (iii) provide certain disclosures regarding executive compensation required of larger

public companies; or (iv) hold nonbinding advisory votes on executive compensation. Additionally, as an emerging growth company, ProFrac is required to have only two years of audited financial statements and only two years of related Management's Discussion and Analysis of Financial Condition and Results of Operations disclosure. ProFrac will remain an emerging growth company for up to five years from the completion of the ProFrac IPO, although ProFrac will lose that status sooner if ProFrac has more than \$1.07 billion of revenues in a fiscal year, have more than \$700.0 million in market value of ProFrac Class A Common Stock held by non-affiliates, or issue more than \$1.0 billion of non-convertible debt over a three-year period.

To the extent that ProFrac relies on any of the exemptions available to emerging growth companies, you will receive less information about ProFrac's executive compensation and internal control over financial reporting than issuers that are not emerging growth companies. Additionally, ProFrac intends to continue to take advantage of the extended transition periods for the adoption of new or revised financial accounting standards under the JOBS Act until ProFrac is no longer an emerging growth company. ProFrac's election to use the transition periods permitted by this election may make it difficult to compare ProFrac's financial statements to those of non-emerging growth companies and other emerging growth companies that have opted out of the extended transition periods permitted under the JOBS Act and who will comply with new or revised financial accounting standards.

If some investors find ProFrac Class A Common Stock to be less attractive as a result, there may be a less active trading market for ProFrac's Class A Common Stock and ProFrac's stock price may be more volatile.

If securities or industry analysts cease publishing research reports or publish unfavorable research about ProFrac's business, the price and trading volume of its Class A Common Stock could decline.

The trading market for ProFrac Class A Common Stock will depend in part on the research reports that securities or industry analysts publish about ProFrac or ProFrac's business. If one or more of the analysts who covers ProFrac downgrades ProFrac's securities, the price of ProFrac's securities would likely decline. Moreover, if one or more of these analysts ceases to cover ProFrac or fails to publish regular reports on ProFrac, interest in the purchase of ProFrac's securities could decrease, which could cause the price of ProFrac Class A Common Stock and other securities and their trading volume to decline.

### Risks Related to USWS' Business and Industry

USWS' transition from the diesel pressure pumping market has negatively impacted and may continue to negatively impact its liquidity and its ability to generate revenues and service its outstanding indebtedness for a period of time.

Since USWS' announcement in May 2021 of its commitment to becoming anall-electric pressure pumping services provider, USWS has sold most of its legacy, diesel-powered pressure pumping equipment, which has resulted in a reduction in the number of fleets USWS has available to provide pressure pumping services. Until USWS can complete the build out of additional all-electric pressure pumping equipment, USWS expects to generate less revenue, which has adversely impacted and may continue to adversely impact its ability to service its outstanding indebtedness. Additionally, the decrease in revenue has resulted in and may continue to result in a reduction in the borrowing base available under USWS' ABL Credit Facility, which may adversely impact USWS' liquidity. Furthermore, USWS also expanded its operations into new geographical areas as its fleet availability has been reduced, which has resulted in a reduction in USWS' economies of scale which has and may continue to have an adverse impact on the profitability of its operations.

USWS' business depends on the level of capital spending and E&P activity by the onshore oil and natural gas industry in the United States, and the level of such activity is affected by industry conditions that are beyond USWS' control.

USWS' business is directly affected by the willingness of its customers to make expenditures to explore for, develop and produce oil and natural gas from onshore resources in the United States. The willingness of USWS'

customers to undertake these activities depends largely upon prevailing industry conditions that are influenced by numerous factors over which USWS has no control, including:

- · prices, and expectations about future prices, for oil and natural gas;
- domestic and foreign supply of, and demand for, oil and natural gas and related products;
- · the supply of and demand for hydraulic fracturing and other oilfield services and equipment in the United States;
- the cost of exploring for, developing, producing and delivering oil and natural gas;
- · available pipeline, storage and other transportation capacity;
- lead times associated with acquiring equipment and products and availability of qualified personnel;
- · the discovery rates of new oil and natural gas reserves;
- federal, state and local regulation of hydraulic fracturing and other oilfield service activities, as well as exploration and production activities, including public pressure on governmental bodies and regulatory agencies to regulate USWS' industry;
- · the availability of water resources, suitable proppant and chemicals in sufficient quantities for use in hydraulic fracturing fluids;
- · geopolitical developments and political instability in oil and natural gas producing countries;
- · actions of OPEC+, its members and other state-controlled oil companies relating to oil price and production controls;
- advances in exploration, development and production technologies or in technologies affecting energy consumption;
- · the price and availability of alternative fuels and energy sources;
- weather conditions, natural disasters and other catastrophic events such as an epidemic or pandemic disease outbreak;
- uncertainty in capital and commodities markets and the ability of oil and natural gas producers to raise equity capital and debt financing;
- U.S. federal, state and local and non-U.S. governmental regulations and taxes; and
- epidemics, pandemics, or other major public health issues, such as the COVID-19 pandemic.

The oil and natural gas industry is volatile. A prolonged economic slowdown or recession in the United States, adverse events relating to the energy industry or regional, national, and global economic conditions and factors could negatively impact exploration and production activity and the level of drilling and completion activity by some of USWS' customers. This volatility may result in a decline in the demand for, or adversely affect the price of, USWS' services. In addition, material declines in oil and natural gas prices, the development of oil and natural gas reserves in USWS' market areas or drilling or completion activity in the U.S. oil and natural gas shale regions, could have a material adverse effect on USWS' business, financial condition, prospects, results of operations and cash flows.

## The volatility of oil and natural gas prices may adversely affect the demand for USWS' services and negatively impact USWS' results of operations.

The demand for USWS' services is substantially influenced by current and anticipated crude oil and natural gas commodity prices and the related levels of capital spending and drilling activity in the areas in which USWS has operations. Volatility or weakness in crude oil and natural gas commodity prices (or the perception that crude oil and natural gas commodity prices will decrease) affects the spending patterns of USWS' customers, and the products and services USWS provides are, to a substantial extent, deferrable in the event oil and natural gas companies reduce capital expenditures. As a result, USWS may experience lower utilization of, and may be forced to lower its rates for, its equipment and services.

Historical prices for crude oil and natural gas have been extremely volatile and are expected to continue to be volatile. The market prices for crude oil and natural gas depend on factors beyond USWS' control, including worldwide and domestic supplies of crude oil and natural gas and actions taken by foreign oil and gas producing nations. Prolonged periods of price instability in the oil and natural gas industry will adversely affect the demand for USWS' products and services, its financial condition, prospects and results of operations and its ability to service its debt or fund capital expenditures.

For example, the price of oil fell significantly in 2020, due to the COVID-19 pandemic and its impact on the worldwide economy and global demand for oil. As a result, E&P companies moved to significantly cut costs, both by decreasing drilling and completion activity and by demanding price concessions from their service providers, including providers of pressure pumping services. In turn, service providers, including pressure pumping service providers, were forced to lower their operating costs and capital expenditures, while continuing to operate their businesses in an extremely competitive environment.

Oil and natural gas prices continued to fluctuate in fiscal year 2021, with the ongoingCOVID-19 pandemic contributing to volatility and uncertainty. Towards the end of the year, prices increased significantly and with the increase in commodity prices demand for USWS' services improved. However, persistent commodity price volatility, supply chain disruptions and deteriorating economic conditions could impact USWS' near-term business prospects and ability to forecast future performance. USWS expects its customers to react to commodity price volatility by adjusting their level of capital spending accordingly.

Additionally, fuel conservation measures, alternative fuel requirements and increasing consumer demand for alternatives to oil and natural gas could reduce the demand for oil and natural gas products, creating downward pressure on commodity prices and the prices USWS is able to charge for its services.

### USWS' level of current and future indebtedness could adversely affect USWS' financial condition.

As of June 30, 2022, USWS had \$7 million of borrowings outstanding, with available capacity of \$17.9 million, under its ABL Credit Facility. USWS' ABL Credit Facility is scheduled to mature on April 1, 2025. As of June 30, 2022, USWS had \$102.8 million of borrowings outstanding under its Term A Loan and Term B Loan (collectively the "Senior Secured Term Loan") and an additional \$22.5 million in loans outstanding under its last-out senior secured term loan credit facility ("Term C Loans") pursuant to its Senior Secured Term Loan Agreement. USWS' Senior Secured Term Loan and Term C Loans are scheduled to mature on December 5, 2025. On USWS' Senior Secured Term Loan, USWS is required to make quarterly principal payments of \$1.25 million until March 31, 2023 and \$5.0 million from June 30, 2023 through September 30, 2025, with final payment due at maturity. USWS' Term C Loans have a PIK interest rate of 14.0% and contains provisions with up to a 100% premium payable upon any repayment, prepayment or acceleration. USWS' obligations under its ABL Credit Facility, Senior Secured Term Loan and Term C Loans are secured by substantially all of USWS' assets

As of June 30, 2022, USWS had \$25.0 million outstanding under its USDA Loan. USWS' USDA Loan is scheduled to mature on November 12, 2030, subject to equal monthly principal payments beginning on December 12, 2023. USWS' USDA Loan is secured by certain of its pressure pumping equipment.

As of June 30, 2022, the aggregate outstanding balance under USWS' equipment financing notes was \$6.8 million, of which \$3.4 million is due within one year. USWS' equipment financing notes are secured by certain of its pressure pumping equipment.

Certain of USWS' debt instruments include provisions, such as PIK interest, repayment and other premiums, and fees which will result in the amount of outstanding debt to increase significantly over time. Upon maturity of USWS' indebtedness, it will be required to repay, extend or refinance its indebtedness. USWS may not be able to extend, replace or refinance any one or all of its existing debt financing agreements on terms reasonably acceptable to USWS, or at all. If USWS is unable to meet its debt service obligations, its lenders under the ABL Credit Facility, Senior Secured Term Loan, Term C Loans, USDA Loan, or equipment financing notes can seek to foreclose on USWS' assets.

As of June 30, 2022, USWS had \$133.8 million of principal, including PIK interest, outstanding of the Convertible Senior Notes, which are convertible into shares of USWS' Common Stock at the option of the holders. The Convertible Senior Notes, subject to earlier conversion, are due and payable on June 5, 2026 in shares of Common Stock equal to the entire outstanding and unpaid principal balance, plus any PIK interest, subject to certain limitations on the number of shares of USWS Common Stock that may be issued and which would require USWS to settle the conversion in payment partially in cash.

USWS' ability to meet its debt service obligations will be dependent upon future performance, which in turn will be subject to general economic conditions, industry cycles and financial, business, and other factors affecting its operations, many of which are beyond its control. USWS' business may not continue to generate sufficient cash flow from operations to pay its debt service obligations when due. Moreover, USWS may incur additional indebtedness, which would increase the amount of cash flow USWS needs to service debt obligations. If USWS is unable to generate sufficient cash flow from operations, it may be required to sell assets, to restructure or refinance all or a portion of such indebtedness or to obtain additional financing. USWS cannot be sure, however, that it would be able to sell assets, restructure, or refinance all or a portion of its indebtedness or obtain additional financing on commercially reasonable terms or at all. Moreover, any failure to make scheduled payments of interest and principal on USWS' outstanding indebtedness would likely result in a reduction of its credit profile, which could harm USWS' ability to incur additional indebtedness on acceptable terms. To the extent inadequate liquidity or other considerations require USWS to seek to restructure or refinance its indebtedness, USWS' ability to do so will depend on rumerous factors, including many beyond its control, such as the condition of the capital markets and USWS' financial condition at such time. Any new or refinancing or restructuring of USWS' indebtedness could be at higher interest rates or other unfavorable terms and may require USWS to comply with more onerous covenants, which could further restrict its business operations. Furthermore, its current and future indebtedness may discourage acquisition bids or merger proposals, which may adversely affect the market price of USWS' Common Stock.

USWS' debt financing agreements subject it to financial and other restrictive covenants. These restrictions may limit USWS' operational or financial flexibility and could subject USWS to potential defaults under its credit facilities.

USWS' debt financing agreements subject it to restrictive covenants, including, among other things, limitations (each of which is subject to certain exceptions) on USWS' ability to incur debt, grant liens, enter into transactions resulting in fundamental changes (such as mergers or sales of all or substantially all of its assets) and asset sales or other types of dispositions, restrict subsidiary dividends or other subsidiary distributions, enter into transactions with affiliates and swap counterparties, make investments and restricted payments, permit subsidiaries to provide guarantees to other material debt, and enter into leases and sale and lease back arrangements.

Additionally, USWS' ABL Credit Facility is subject to a springing fixed charge coverage covenant. For a description of the covenants under its ABL Credit Facility, please see "Note 11 – Debt" in the notes to the USWS financial statements beginning on page F-134. If USWS is unable to remain in compliance with the covenants of its ABL Credit Facility, then amounts outstanding thereunder may be accelerated and become due immediately. USWS might not have, or be able to obtain, sufficient funds to make these accelerated payments, and any such acceleration could have a material adverse effect on its financial condition and results of operations.

Moreover, subject to the limits contained in its debt financing agreements, USWS may incur substantial additional debt from time to time. Any borrowings USWS may incur in the future would have several important consequences for its future operations, including that:

covenants contained in the documents governing such indebtedness may require USWS to meet or maintain certain financial tests, which
may affect USWS' flexibility in planning for, and reacting to, changes in its industry, such as being able to take advantage of acquisition
opportunities when they arise;

- USWS' ability to obtain additional financing for working capital, capital expenditures, acquisitions, general corporate and other purposes may be limited;
- · USWS may be competitively disadvantaged to its competitors that are less leveraged or have greater access to capital resources; and
- USWS may be more vulnerable to adverse economic and industry conditions.

### The phase-out of the London Interbank Offered Rate, or LIBOR, may adversely affect a portion of USWS' outstanding debt.

In July 2017, the United Kingdom's Financial Conduct Authority, which regulates LIBOR, announced that it intends to phase out LIBOR by the end of 2021. The ICE Benchmark Administration, the administrator of LIBOR, has ceased to publish USD LIBOR for the one week and two month USD LIBOR tenors. Further, on March 5, 2021, the ICE Benchmark Administration announced its intention to cease the publication of the remaining USD LIBOR tenors after June 30, 2023. While this announcement extends the transition period to June 2023, the United States Federal Reserve concurrently issued a statement advising banks to stop new USD LIBOR issuances by the end of 2021. In light of these recent announcements, the future of LIBOR at this time is uncertain and any changes in the methods by which LIBOR is determined or regulatory activity related to LIBOR's phase-out could cause LIBOR to perform differently than in the past or cease to exist. Changes in the method of determining LIBOR, or the replacement of LIBOR with an alternative floating borrowing rate, may adversely affect USWS' borrowing costs. While USWS' ABL Credit Facility and Senior Secured Term Loan are scheduled to mature in April 2025 and December 2025, respectively, potential changes, or uncertainty related to such potential changes in interest rate benchmarks may adversely affect USWS' ability to refinance its indebtedness. USWS cannot predict the effect of the potential changes to LIBOR or the establishment and use of alternative floating borrowing rates on the portion of its outstanding debt that is LIBOR based. Challenges in changing to a different borrowing interest rate may result in less favorable pricing on certain of USWS' debt instruments and could have an adverse effect on its financial results and cash flow.

# USWS' operations are subject to unforeseen interruptions and hazards inherent in the oil and natural gas industry, for which USWS may not be adequately insured.

USWS' operations are exposed to the risks inherent to its industry, such as equipment defects, vehicle accidents, fires, explosions, blowouts, surface cratering, uncontrollable flows of gas or well fluids, pipe or pipeline failures, abnormally pressured formations and various environmental hazards, such as oil spills and releases of, and exposure to, hazardous substances. For example, USWS' operations are subject to risks associated with pressure pumping, including any mishandling, surface spillage or potential underground migration of fracturing fluids, including chemical additives. In addition, USWS' operations are exposed to potential natural disasters, including blizzards, tornadoes, storms, floods, other adverse weather conditions and earthquakes. The occurrence of any of these events could result in substantial losses to USWS' business due to personal injury or loss of life, severe damage to or destruction of property, natural resources and equipment, pollution or other environmental damage or other damage resulting in curtailment or suspension of USWS' operations. Litigation arising from operations where USWS' services are provided, may cause USWS to be named as a defendant in lawsuits asserting potentially large claims including claims for exemplary damages. The cost of managing such risks may be significant, and the frequency and severity of such incidents may affect operating costs, insurability and relationships with customers, employees, and regulators. USWS' customers may elect not to purchase USWS' services if they view USWS' environmental or safety record as unacceptable, which could cause USWS to lose customers and substantial revenues.

USWS' insurance may not be adequate to cover all losses or liabilities it may suffer, and the insurance coverage may not be adequate to cover claims that may arise. USWS is not fully insured against all risks, either because insurance is not available or coverage is excluded from USWS' policy, or because of the high premium costs relative to perceived risk. Furthermore, USWS may be unable to maintain or obtain insurance of the type

and amount it desires at reasonable rates. Insurance rates in the past have been subject to wide fluctuation and changes in coverage could result in less coverage, increases in cost or higher deductibles and retentions or the imposition of sub-limits for certain risks. In addition, USWS may not be able to secure additional insurance or bonding that might be required by new governmental regulations. If USWS was to incur a significant liability for which USWS is not fully insured, it could have a material adverse effect on USWS' business, results of operations and financial condition.

USWS' long-term contracts are subject to certain risks, including counterparty payment risks, inability to renew or replace at favorable economic terms, and changing market conditions that result in higher costs without offsetting revenue escalations.

USWS generally has long-term written contractual arrangements with its customers on most of its equipment. The counterparties to USWS' contractual arrangements are subject to various market risks that impact their businesses and, as a result, they may be unable to make payments to USWS pursuant to the payment terms set forth in such contractual arrangements. Additionally, as contracts with USWS customers come up for replacement or renewal, changing market conditions may prevent USWS from replacing or renewing the contracts on comparable terms. USWS' ability to achieve favorable terms under these expiring contracts could be affected by many factors, including prolonged reduced commodity prices, decrease in demand for USWS services or increased competition in the markets USWS serves. If USWS is unable to replace or renew the expiring agreements on comparable terms, it could materially adversely affect USWS' business, financial condition, results of operations and cash flows, including its ability to make cash distributions to its shareholders.

With no long-term contract in place, such customers could cease buying USWS' services at any time, for any reason, with little or no recourse. If multiple customers or a material customer with whom USWS did not have a long-term contract in place elected not to purchase its services, USWS' business prospects, financial condition, and results of operations could be adversely affected.

USWS is dependent on a few customers in a single industry. The loss of one or more significant customers could adversely affect USWS' financial condition, prospects, and results of operations.

USWS' customers are engaged in the oil and natural gas E&P business in the United States. Historically, USWS has been dependent upon a few customers for a significant portion of its revenue. For the year ended December 31, 2021, three customers each accounted for greater than 10% of USWS' total consolidated revenues and collectively represented 50.4% of its consolidated revenues. It is likely that USWS will continue to derive a significant portion of its revenue from a relatively small number of customers in the future. Additionally, the oil and natural gas industry is characterized by frequent consolidation activity and, recently, by frequent financial distress and bankruptcy filings. Changes in ownership of USWS' customers or bankruptcy filings by its customers may result in the loss of, or reduction in, business from those customers. If USWS was to lose any material customer, or if a major customer fails to pay or delays in paying for USWS' services, USWS may not be able to redeploy its equipment at similar utilization or pricing levels or within a short period of time and such loss could have a material adverse effect on USWS' business, financial condition, prospects and results of operations.

Rising and volatile material costs and supply chain constraints or disruptions have adversely affected and could continue to adversely affect USWS' results of operations.

USWS uses certain products supplied by customers in its operations, including water and sand. As a result of supply chain constraints, USWS' customers experienced shortages of water and sand, which led to significant downtime during the fourth quarter of 2021. This downtime, and any continued downtime resulting from continued supply chain constraints and related supply shortages, have had and may continue to have a negative impact on USWS' ability to generate revenue. In addition, continued supply chain constraints may result in rising costs that USWS pays for supplies, which would decrease its ability to perform its services at a profitable level.

It is also possible that supply chain constraints or disruptions will result in significant delivery delays with respect to the equipment that USWS needs to build its new Nyx Clean Fleets<sup>®</sup>. If USWS is unable to build out its new Nyx Clean Fleets<sup>®</sup> as a result of any delivery delays, USWS may not be in position to enter into contracts with customers, which would negatively impact its results of operations.

USWS currently relies on a limited number of suppliers for major equipment to build new and upgrade existing electric fleets to its current custom Clean Fleet® design, and USWS' reliance on these vendors exposes it to risks including price and timing of delivery. USWS' failure to meet current commitments to these vendors could result in its inability to build fleets in the future and may impact its ability to repair and replace certain equipment.

USWS currently relies on a limited number of suppliers for major equipment to build its new fleets and upgrade any existing electric fleets as needed to its current custom Clean Fleet® design. During periods in which pressure pumping services are in high demand, USWS has experienced delays in obtaining certain parts that are used in fabricating and assembling its fleets. If demand for fleets or the components necessary to build such fleets increases or these vendors face financial distress or bankruptcy, these vendors may not be able to provide the new or upgraded fleets on schedule or at the current price. If this were to occur, USWS could be required to seek other suppliers for major equipment to build or upgrade its fleets, which may adversely affect its revenues or increase its costs. USWS has made certain commitments to purchase new fleets from these vendors. USWS' failure to meet these commitments could impact its ability to build new fleets in the future and impair its ability to repair or replace certain equipment on existing fleets.

USWS relies on a limited number of suppliers to provide generators to power USWS' custom Clean Fleet design, which exposes it to risks relating to operating performance and availability of supply. An inability to secure power generation assets could materially impact USWS' operations.

USWS' current custom Clean Fleet® requires the use of power generators in order to operate. USWS currently relies on a limited number of suppliers to provide it with the power generators required to operate its Clean Fleet® and, as a result, USWS is exposed to certain supply chain and operational risks with respect to such power generators. As the availability of power generators remains limited as a result of current supply chain disruptions, USWS may not be able to obtain new or replacement power generators as needed to operate its fleets, which would impact USWS' ability to provide pressure pumping services to its customers and would have a materially adverse impact on its results of operations.

USWS relies on a limited number of third parties for proppant and chemicals and delays in deliveries and increases in the cost of such materials could harm its business, results of operations and financial condition.

USWS has established relationships with a limited number of suppliers of its raw materials (such as proppant and chemicals). Should any of USWS' current suppliers be unable to provide the necessary materials or otherwise fail to deliver the materials in a timely manner and in the quantities required, any resulting delays in the provision of services could have a material adverse effect on USWS' business, results of operations and financial condition. Additionally, increasing and volatile costs of such materials may negatively impact demand for USWS' services or the profitability of its business operations. In the past, USWS' industry faced sporadic proppant shortages associated with pressure pumping operations requiring work stoppages, which adversely impacted the operating results of several competitors. USWS may not be able to mitigate any future shortages of materials, including proppant.

USWS' assets require significant amounts of capital for maintenance, upgrades and refurbishment and may require significant capital expenditures for new equipment.

USWS' fleets and other completion service-related equipment require significant capital investment in maintenance, upgrades and refurbishment to maintain their competitiveness. The costs of components and labor

may increase in the future which will require USWS to incur additional costs to maintain, upgrade and/or refurbish its fleets. USWS' fleets and other equipment typically do not generate revenue while they are undergoing maintenance, upgrades, or refurbishment. Any maintenance, upgrade or refurbishment project for USWS' assets could increase its indebtedness or reduce cash available for other opportunities. Furthermore, such projects may require proportionally greater capital investments as a percentage of total asset value, which may make such projects difficult to finance on acceptable terms. To the extent USWS is unable to fund such projects, USWS may have less equipment available for service or its equipment may not be attractive to potential or current customers. Additionally, competition or advances in technology within USWS' industry may require USWS to update or replace existing fleets or build or acquire new fleets. Such demands on USWS' capital or reductions in demand for USWS' fleets and the increase in cost of labor necessary for such maintenance and improvement, in each case, could have a material adverse effect on USWS' business, liquidity position, financial condition, prospects and results of operations and may increase its costs.

An inability to access capital on favorable terms to fund USWS' capital commitments may have an adverse impact on its business, liquidity position, financial condition, prospects and results of operations and may require USWS to raise capital on unfavorable terms that could result in dilution to its shareholders.

USWS has entered into certain agreements for the build out of its new next-generationall-electric fleets, some of which will require significant capital commitments. If USWS is not able to fund such capital commitments from its generated cash flows, USWS may be required to access the capital or credit markets. USWS' current or future capital structure, operating performance or financial condition may not permit USWS to access the capital or credit markets or to obtain other financing at the times, in the amounts and on the terms necessary or advisable for USWS to successfully carry out its business strategy or to fund its capital commitments, which would have an adverse impact on USWS' business, liquidity position, financial condition, prospects and results of operations and could result in dilution to its shareholders.

USWS is subject to federal, state, and local laws and regulations regarding issues of health, safety, and protection of the environment, including with respect to its pressure pumping operations. Under these laws and regulations, USWS may become liable for penalties, damages or costs of remediation or other corrective measures. Any changes in laws or government regulations could increase USWS' costs of doing business.

USWS' operations are subject to stringent federal, state, and local laws and regulations relating to, among other things, protection of natural resources, clean air and drinking water, endangered species, GHGs, nonattainment areas, the environment, health and safety, chemical use and storage, waste management, waste disposal, corrective action and transportation of waste and other hazardous and nonhazardous materials. USWS' operations involve risks of environmental liability, including leakage from an operator's casing during its operations or accidental spills onto or into surface or subsurface soils, surface water or groundwater. Some environmental laws and regulations may impose strict liability, joint and several liability, or both. In some situations, USWS could be exposed to liability as a result of its conduct that was lawful at the time it occurred or the conduct of, or conditions caused by, third parties without regard to whether USWS caused or contributed to the conditions. Additionally, environmental concerns, including clean air, drinking water contamination and seismic activity, have prompted investigations that could lead to the enactment of regulations, limitations, restrictions or moratoria that could potentially result in the shutdown of USWS' operations, fines and penalties (administrative, civil or criminal), revocations of permits to conduct business, expenditures for remediation or other corrective measures and/or claims for liability for property damage, exposure to hazardous materials, exposure to hazardous waste, nuisance or personal injuries. Sanctions for noncompliance with applicable environmental laws, regulations, permits and other requirements may also include the assessment of administrative, civil, or criminal penalties, revocation of permits and temporary or permanent cessation of operations in a particular location and issuance of corrective action orders. Such claims or sanctions and related costs could cause USWS to incur substantial costs or losses and could have a material adverse effect on its business, financial condition, prospects, and results of operations. Additionally, an increase in regulatory requirements, limitations, restrictions or moratoria on oil and natural gas exploration and completion activities at a federal, state or local level or changes in the way these requirements are interpreted or enforced could significantly delay or interrupt USWS' operations, limit the amount of work USWS can perform, increase

USWS' costs of compliance, or increase the cost of its services, thereby possibly having a material adverse impact on its financial condition. For more information about regulations and laws regarding issues of health, safety, and protection of the environment in USWS' industry, please see "Information About USWS – Business—Environmental and Occupational Health and Safety Regulations."

In December 2016, the EPA issued a study of the potential impacts of hydraulic fracturing on drinking water and groundwater. The EPA report states that there is scientific evidence that hydraulic fracturing activities can impact drinking resources under some circumstances and identifies certain conditions in which the EPA believes the impact of such activities on drinking water and groundwater can be more frequent or severe. The EPA study could spur further initiatives to regulate hydraulic fracturing. Additionally, state legislatures, state regulatory agencies and local municipalities may consider legislation, regulations, or ordinances, respectively that could affect all aspects of the oil and natural gas industry and occasionally take action to restrict or further regulate hydraulic fracturing operations. Some states, counties and municipalities have enacted or are considering moratoria on hydraulic fracturing or zoning ordinances, which could impose a de facto ban on drilling and hydraulic fracturing operations. Currently, it is not possible to estimate the potential impact on USWS' business of these state and municipal actions or the enactment of additional federal or state legislation or regulations affecting hydraulic fracturing. Compliance, stricter regulations, or the consequences of any failure to comply by USWS could have a material adverse effect on its business, financial condition, prospects and results of operations. For more information about regulations relating to hydraulic fracturing, please see "Information About USWS – Business—Environmental and Occupational Health and Safety Regulations."

Furthermore, many states in which USWS operates require the disclosure of some or all the chemicals used in USWS' pressure pumping operations. Certain aspects of one or more of these chemicals may be considered proprietary by USWS or its chemical suppliers. Disclosure of USWS' proprietary chemical information to third parties or to the public, even if inadvertent, could diminish the value of USWS' trade secrets or those of USWS' chemical suppliers and could result in competitive harm to USWS, which could have an adverse impact on USWS' business, financial condition, prospects and results of operations. Additionally, USWS' business could be affected by a moratorium or increased regulation of companies in its supply chain, such as sand mining by USWS' proppant suppliers, which could limit USWS' access to supplies and increase the costs of its raw materials. At this time, it is not possible to estimate how these various restrictions could affect USWS' ongoing operations.

Many regulatory and legislative bodies routinely evaluate the adequacy and effectiveness of laws and regulations affecting the oil and gas industry. Reducing emissions of volatile organic compounds and hazardous air pollutants is one of the sectors designated for increased enforcement by the EPA, which, in addition to state and local governing bodies, will continue to regulate USWS' industry in the years to come. Laws and regulations protecting the environment, especially those related to GHGs and climate change, generally have become more stringent over time, and USWS expects them to continue to do so. This could lead to material increases in USWS' costs and liability exposure for future environmental compliance and remediation and may negatively impact demand for USWS' services. For example, oil and natural gas exploration and production may decline because of environmental requirements, including land use policies responsive to environmental concerns. Additionally, if USWS expands the size or scope of its operations, USWS could be subject to regulations that are more stringent than the requirements under which USWS is currently allowed to operate or require additional authorizations to continue operations. Compliance with this additional regulatory burden could increase USWS' operating or other costs.

Additionally, failure to comply with government, industry or USWS' own environmental, health and safety laws and regulations, or failure to comply with USWS' compliance or reporting requirements, could tarnish USWS' reputation for safety and quality and have a material adverse effect on USWS' competitive position. In addition, customers maintain their own compliance and reporting requirements, and if USWS does not perform in accordance with their requirements, USWS could lose business from its customers, many of whom have an increased focus on environmental and safety issues.

Climate change legislation, regulations restricting emissions of greenhouse gases or other action taken by public or private entities related to climate change could result in increased operating costs and reduced demand for the crude oil and natural gas produced by USWS' customers.

In response to findings that emissions of carbon dioxide, methane and other GHGs present an endangerment to public health and the environment, the EPA has issued regulations to restrict emissions of GHGs under existing provisions of the Clean Air Act. From time to time, Congress has considered legislation to reduce emissions of GHGs but no such legislation has yet been adopted by Congress. Some states have individually or in regional cooperation, imposed restrictions on GHG emissions under various policies and approaches, including establishing a cap on emissions, requiring efficiency measures, or providing incentives for pollution reduction, use of renewable energy sources, or use of replacement fuels with lower carbon content. In the future, the United States has also chosen to adhere to international agreements targeting GHGs reductions (e.g., the Paris Agreement). The adoption of legislation or regulatory programs to reduce emissions of GHGs could require USWS to incur additional operating costs, such as costs to purchase and operate emissions control systems, to acquire emissions allowances or to comply with new regulatory or reporting requirements. Any such legislation or regulatory programs could also increase the cost of consuming, and thereby reduce demand for, the oil and natural gas USWS' customers produce. Consequently, legislation and regulatory programs to reduce emissions of GHGs could have an adverse effect on USWS' business, financial condition, and results of operations. For more information about climate change legislation, please see "Information about USWS – Business—Environmental and Occupational Health and Safety Regulations."

Furthermore, increasing attention to the risks of climate change has resulted in an increased possibility of lawsuits or investigations brought by public and private entities against oil and natural gas companies in connection with their GHG emissions. Should USWS be targeted by any such litigation or investigations, USWS may incur liability, which, to the extent that societal pressures or political or other factors are involved, could be imposed without regard to the causation of, or contribution to, the asserted damage, or to other mitigating factors. The ultimate impact of GHGs emissions-related agreements, legislation and measures on USWS' financial performance is highly uncertain because USWS is unable to predict with certainty, for a multitude of individual jurisdictions, the outcome of political decision-making processes and the variables and tradeoffs that inevitably occur in connection with such processes.

### If USWS is unable to fully protect its intellectual property rights, USWS may suffer a loss in its competitive advantage or market share.

USWS has been granted or have received notice of allowance for 81 patents and have an additional 249 patents pending. If USWS is not able to maintain the confidentiality of its trade secrets or fails to adequately protect its intellectual property rights USWS has now or acquires in the future, its competitive advantage would be diminished. Additionally, competitors may be able to replicate USWS' technology or services protected by its intellectual property rights. USWS cannot be sure that any patents USWS has been granted or may obtain in the future would provide USWS with any significant commercial benefit or would allow USWS to prevent its competitors from employing comparable technologies or processes.

### USWS may be adversely affected by disputes regarding intellectual property rights of third parties.

Third parties from time to time may initiate litigation against USWS by asserting that the conduct of USWS' business infringes, misappropriates, or otherwise violates intellectual property rights, or USWS may initiate legal proceedings in order to enforce its intellectual property rights. If USWS initiates such proceedings or if any third parties bring a claim of intellectual property infringement against USWS, USWS may be subject to costly and time-consuming litigation, diverting the attention of management and its employees. USWS may not prevail in any such legal proceedings related to such claims, and its products and services may be found to infringe, impair, misappropriate, dilute, or otherwise violate the intellectual property rights of others. If USWS is unsuccessful in defending such claims, it could be required to pay substantial damages and/or be enjoined from

using or selling the infringing products or technology. Any legal proceeding concerning intellectual property could be protracted and costly regardless of the merits of any claim and is inherently unpredictable and could have a material adverse effect on USWS' financial condition, regardless of its outcome.

If USWS was to discover that its technologies or products infringe valid intellectual property rights of third parties, USWS may need to obtain licenses from these parties or substantially re-engineer its products in order to avoid infringement. USWS may not be able to obtain the necessary licenses on acceptable terms, or at all, or be able to re-engineer its products successfully. If USWS' inability to obtain required licenses for its technologies or products prevents USWS from selling its products, it could adversely impact USWS' financial condition and results of operations.

### Changes in transportation regulations may increase USWS' costs and negatively impact its results of operations.

USWS is subject to various transportation regulations, including as a motor carrier by the U.S. Department of Transportation and by various federal and state agencies, whose regulations include certain permit requirements of highway and safety authorities. These regulatory authorities exercise broad powers over USWS' trucking operations, generally governing such matters as the authorization to engage in motor carrier operations, safety, equipment testing, driver requirements and specifications and insurance requirements. The trucking industry is subject to possible regulatory and legislative changes that may impact USWS' operations, such as changes in fuel emissions limits, hours of service regulations that govern the amount of time a driver may drive or work in any specific period and limits on vehicle weight and size. As the federal government continues to develop and propose regulations relating to fuel quality, engine efficiency and GHG emissions, USWS may experience an increase in costs related to truck purchases and maintenance, impairment of equipment productivity, a decrease in the residual value of vehicles, unpredictable fluctuations in fuel prices and an increase in operating expenses. Increased truck traffic may contribute to deteriorating road conditions in some areas where USWS' operations are performed. USWS' operations, including routing and weight restrictions, could be affected by road construction, road repairs, detours and state and local regulations and ordinances restricting access to certain routes or times on specific roadways. Proposals to increase federal, state, or local taxes, including taxes on motor fuels, are also made from time to time, and any such increase would increase USWS' operations will be enacted and to what extent any such legislation or regulations could increase its costs or otherwise adversely affect its business or operations.

Oil and natural gas companies are substantially dependent on the availability of water during pressure pumping operations. Restrictions on the ability to obtain water for exploration and production activities and the disposal of flowback and produced water may impact their operations and have a corresponding adverse effect on USWS' business, results of operations and financial condition.

Water is an essential component of shale oil and natural gas production during pressure pumping operations. USWS' oil and natural gas producing customers' access to water to be used in this process may be adversely affected due to reasons such as periods of extended drought, private, third-party competition for water in localized areas or the implementation of local or state governmental programs to monitor or restrict the beneficial use of water subject to their jurisdiction for hydraulic fracturing to assure adequate local water supplies. The occurrence of these or similar developments may result in limitations being placed on allocations of water due to needs by third-party businesses with more senior contractual or permitting rights to the water. USWS' customers' inability to locate or contractually acquire and sustain the receipt of sufficient amounts of water could adversely impact their E&P operations and have a corresponding adverse effect on USWS' business, results of operations and financial condition.

Moreover, the imposition of new environmental regulations and other regulatory initiatives could include increased restrictions on USWS' producing customers' ability to dispose of flowback and produced water

generated in hydraulic fracturing or other fluids resulting from E&P activities. For more information about water-related regulations, please see "Information About USWS – Business – Environmental and Occupational Health and Safety Regulations." Compliance with current and future environmental regulations and permit requirements governing the withdrawal, storage and use of surface water or groundwater necessary for hydraulic fracturing of wells and any inability to secure transportation and access to disposal wells with sufficient capacity to accept all of USWS' flowback and produced water on economic terms may increase its customers' operating costs and cause delays, interruptions or termination of its customers' operations, the extent of which cannot be predicted. In addition, the legal requirements related to the disposal of produced water into a non-producing geologic formation by means of underground injection wells are subject to change based on concerns of the public or governmental authorities regarding such disposal activities. One such concern arises from seismic events near underground disposal wells that are used for the disposal by injection of produced water resulting from oil, natural gas, and natural gas liquids activities. In response to concerns regarding induced seismicity, regulators in some states have imposed, or are considering imposing, additional requirements in the permitting of produced water disposal wells to assess any relationship between seismicity and the use of such wells. Among other things, these rules require companies seeking permits for disposal wells to provide seismic activity data in permit applications, provide for more frequent monitoring and reporting for certain wells and allow the state to modify, suspend or terminate permits on grounds that a disposal well is likely to be, or determined to be, causing seismic activity. States may issue orders to temporarily shut down or to curtail the injection depth of existing wells in the vicinity of seismic events.

Another consequence of seismic events may be lawsuits alleging that disposal well operations have caused damage to neighboring properties or otherwise violated state and federal rules regulating waste disposal. These developments could result in additional regulation and restrictions on the use of injection wells by USWS. Increased regulation and attention given to induced seismicity could also lead to greater opposition, including litigation to limit or prohibit oil, natural gas and natural gas liquids activities utilizing injection wells for produced water disposal.

Any one or more of these developments may result in USWS or its vendors having to limit disposal well volumes, disposal rates and pressures or locations, or require USWS or its vendors to shut down or curtail the injection into disposal wells, which events could have a material adverse effect on USWS' business, financial condition and results of operations.

USWS' ability to expand its operations relies in part on USWS' ability to market its Clean Flee® technology, and advancements in well service technologies could have a material adverse effect on USWS' business, financial condition and results of operations.

The pressure pumping industry is characterized by rapid and significant technological advancements and introductions of new products and services using new technologies, some of which may be subject to patent or other intellectual property protections. For example, USWS uses its patented Clean Fleet® technology as a competitive advantage in the markets it serves. As competitors and others use or develop new or comparable technologies in the future, USWS may lose market share or be placed at a competitive disadvantage. Further, USWS may face competitive pressure to develop, implement or acquire certain new technologies at a substantial cost. Some of USWS' competitors may have greater financial, technical and personnel resources than USWS does, which may allow them to gain technological advantages or implement new technologies before USWS can. Additionally, USWS may be unable to implement new technologies or services at all, on a timely basis or at an acceptable cost. New technology could also make it easier for USWS' customers to vertically integrate their operations, thereby reducing or eliminating the need for USWS' services. Limits on USWS' ability to effectively use or implement new technologies may have a material adverse effect on USWS' business, financial condition, and results of operations.

#### USWS may record losses or impairment charges related to idle assets or assets that it sells.

Prolonged periods of low utilization, changes in technology or the sale of assets below their carrying value may cause USWS to experience losses. These events could result in the recognition of impairment charges that negatively impact USWS' financial results. Significant impairment charges because of a decline in market conditions or otherwise could have a material adverse effect on USWS' results of operations in future periods.

### USWS' ability to use its net operating loss carryforwards (NOLs) to offset future income may be limited.

The ability to utilize USWS' NOL carryforwards to reduce taxable income in future years could become subject to significant limitations under Section 382 of the Code if USWS undergoes an ownership change. In general, an "ownership change" under Section 382 occurs if the percentage of stock owned by an entity's 5% shareholder (as defined for tax purposes) increases by more than 50 percentage points over a rolling three-year period. In the event of an ownership change, Section 382 of the Code imposes an annual limitation on the amount of taxable income a corporation may offset with NOL carryforwards. The annual limitation is generally equal to the value of the stock of the corporation immediately before the ownership change, multiplied by the long-term tax-exempt rate, a rate published monthly by the Internal Revenue Service. Any unused annual limitation may generally be carried over to later years until the NOL carryforwards expire.

Due to historical performance of cumulative losses, the NOL carryforwards are fully reserved with a valuation allowance. USWS intends to maintain a full valuation allowance on these deferred tax assets until there is sufficient evidence to support a full or partial reversal.

You should read and consider the risk factors specific to USWS' business that will also affect the combined company after the Merger. These risks are described in Part I, Item 1A of USWS' Annual Report on Form 10-K for the fiscal year ended December 31, 2021, and in other documents that are incorporated by reference into this proxy statement/information statement/prospectus. See "Where You Can Find More Information" for the location of information incorporated by reference in this proxy statement/information statement/prospectus.

### CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

Certain statements in this proxy statement/information statement/prospectus may be considered "forward-looking statements" within the meaning of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995. In some cases, the reader can identify forward-looking statements by words such as "may," "should," "expect," "intend," "will," "estimate," "anticipate," "predict," or similar words. Forward-looking statements relate to future events or ProFrac's, USWS' or the combined company's future financial or operating performance. These forward-looking statements include, among other things, statements regarding: the expected benefits of the Merger, including any resulting synergies and positive impact on earnings, competitive advantages, expanded active fleet and electric fleet portfolio, increased value, improved efficiency, cost savings including fuel cost savings, access to and rights in acquired intellectual property, emissions minimization and other expected advantages of the Merger; the anticipated timing of the Merger; the likelihood and ability of the parties to successfully consummate the Merger; the services to be offered by the combined company; the markets in which ProFrac and USWS operate; the financial forecasts and projections of ProFrac's and USWS' respective management teams; business strategies, debt levels, industry environment and growth opportunities; the projected value of operational synergies, including value expected to result from license fee savings; and expectations regarding ProFrac's ability to finance USWS' debt. Such forward-looking statements are based upon assumptions made by ProFrac and USWS as of the date hereof and are subject to risks, uncertainties, and other factors that could cause actual results to differ materially from those expressed or implied by such forward-looking statements. Factors that may cause actual results to differ materially from current expectations include, but are not limited to: the risk that the Merger may not be completed in a timely manner or at all, which may adversely affect the price of ProFrac's and USWS' securities; the failure to satisfy the conditions to the consummation of the Merger, including the approval of the Merger by the stockholders of USWS, and the receipt of certain governmental and regulatory approvals; the failure by ProFrac to obtain any financing it may need to complete the Merger on favorable terms or at all; the effect of the announcement or pendency of the Merger on ProFrac's and USWS' business relationships, performance, and businesses generally; risks that the Merger disrupts current plans of ProFrac or USWS and may cause potential difficulties in employee retention as a result of the Merger; the outcome of any legal proceedings that may be instituted against ProFrac or USWS or any of their respective affiliates related to the Merger Agreement and the Merger; changes to the proposed structure of the Merger that may be required or appropriate as a result of applicable laws or regulations or as a condition to obtaining regulatory approval of the Merger; the impact on the price of ProFrac's and USWS' securities, including volatility resulting from changes in the competitive and highly regulated industries in which ProFrac and USWS operate, variations in performance across competitors, changes in laws and regulations affecting ProFrac's and USWS' business and changes in the combined company's capital structure; the ability to implement business plans, forecasts, and other expectations after the completion of the Merger, and identify and realize additional opportunities; the ability to integrate acquired assets and personnel into ProFrac's existing business model and realize the expected value of resulting operational synergies; the ability to successfully and sustainably execute on current business strategies and plans for growth; risks relating to acquisitions generally; and other risks and uncertainties detailed in ProFrac's final prospectus relating to its initial public offering (File No. 333-261255) declared effective by the SEC on May 12, 2022, as well as in ProFrac's prospectus relating to the resale of shares of ProFrac Class A Common Stock by the West Munger Sellers (File No. 333-266465) declared effective by the SEC on August 5, 2022, and in its subsequent reports on Form 10-Q, including in the sections thereof captioned "Risk Factors" and "Cautionary Statement Regarding Forward-Looking Statements," as well as in its subsequent reports on Form8-K, all of which are filed with the SEC and available on the SEC's website at www.sec.gov and on ProFrac's website at www.pfholdingscorp.com, and in USWS' Annual Report on Form 10-K for the fiscal year ended December 31, 2021 and in its subsequent reports on Form 10-Q, including in the sections thereof captioned "Risk Factors," "Cautionary Note Regarding Forward Looking Statements" and "Disclosure Regarding Forward-Looking Statements," as well as in its subsequent reports on Form8-K, all of which are filed with the SEC and available on the SEC's website at www.sec.gov and on USWS' website at www.uswellservices.com. The foregoing list of factors is not exhaustive. There may be additional risks that neither ProFrac nor USWS presently know or that ProFrac or USWS currently believe are immaterial that could also cause actual results to differ from those contained in the forward-looking statements.

You should carefully consider the foregoing factors and the other risks and uncertainties described under "Risk Factors" in this proxy statement/information statement/prospectus, and in other documents filed by ProFrac and USWS from time to time with the SEC, which are available on the SEC's website at www.sec.gov. These filings identify and address other important risks and uncertainties that could cause actual events and results to differ materially from those contained in the forward-looking statements.

Nothing in this proxy statement/information statement/prospectus should be regarded as a representation by any person that the forward-looking statements set forth herein will be achieved or that any of the contemplated results of such forward looking statements will be achieved, including without limitation any expectations about ProFrac's, USWS' or the combined company's operational and financial performance or achievements. Readers are cautioned not to put undue reliance on forward-looking statements. ProFrac and USWS assume no obligation and, except as required by law, do not intend to update or revise these forward-looking statements, whether as a result of new information, future events, or otherwise. ProFrac and USWS give no assurance that it will achieve its expectations, including that the Merger will be consummated.

#### THE MERGER

The following discussion contains important information relating to the Merger. You are urged to read this discussion together with the Merger Agreement and related documents attached as Annexes to this proxy statement/information statement/prospectus for a more complete understanding of the proposed Merger and the transactions related thereto. In addition, important business and financial information about each of ProFrac and USWS is included in this proxy statement/information statement/prospectus. Please see "Where You Can Find More Information."

# Structure of the Merger

The Merger Agreement provides for the merger of Merger Sub with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of ProFrac. The Merger will become effective when the parties to the Merger Agreement file a certificate of merger with the Secretary of State of the State of Delaware (or at such later time as ProFrac and USWS mutually agree and specify in the certificate of merger).

### Merger Consideration

At the Effective Time, each issued and outstanding share of USWS Common Stock will be converted automatically into the right to receive 0.3366 shares of ProFrac Class A Common Stock. USWS Stockholders will receive cash in lieu of any fractional shares of ProFrac Class A Common Stock that would have otherwise been received in the Merger. See "The Merger Agreement — Fractional Shares" beginning on page 134.

Immediately following the completion of the Merger, based on the aggregate amount of capital stock and shares of ProFrac Class A Common Stock outstanding as of August 8, 2022, it is anticipated that persons who were stockholders of ProFrac and USWS immediately prior to the Merger will own approximately 91.76% and 8.24%, respectively, of the then outstanding ProFrac capital stock and will own approximately 76.34% and 23.66%, respectively, of the then outstanding ProFrac Class A Common Stock.

## **Background of the Merger**

On several occasions, each of ProFrac's and USWS' strategic review and evaluation of potential business combinations have included an assessment of the other's business, operations and intellectual property. Also, given the size of the industry and similar business focus of ProFrac and USWS, the companies have been familiar with each other's businesses for a number of years and have, from time to time, discussed potential strategic transactions. Starting in June of 2019, members of the management teams of ProFrac and USWS from time to time have met or otherwise communicated with each other and their respective advisors and with representatives of other companies regarding industry trends and potential strategic alternatives, including acquisitions, business combinations and other strategic transactions. In addition, members of the Wilks family and ProFrac management have each, from time to time, discussed various matters regarding USWS with other USWS securityholders, including potential transactions in USWS securities.

Throughout this timeframe, until its IPO on May 12, 2022, ProFrac was a private company, wholly-owned by the Wilks family.

On June 27, 2019, Ladd Wilks (ProFrac's Chief Executive Officer), Matthew D. Wilks (the Executive Chairman of the ProFrac Board, and the then President and Chief Financial Officer of ProFrac), Brian Uhlmer (the then Senior Vice President of Strategy and Finance for ProFrac), and Morgan Neff (an investment professional from the Wilks' family office), met with Joel Broussard (the current Chairman of the USWS Board and the then President and Chief Executive Officer of USWS), Kyle O'Neill (the current President and Chief Executive Officer of USWS and the then Chief Financial Officer of USWS), Matt Bernard (the then Chief Administrative Officer of USWS) and Josh Shapiro (the current Chief Financial Officer of USWS and the then Vice President of Finance for USWS) to discuss the potential benefits of consolidation in the pressure pumping sector. Based upon the June 27, 2019 meeting, ProFrac and USWS management teams determined to enter into a confidentiality agreement in order to engage in discussions and due diligence regarding a potential business combination. Following that, the discussions regarding a business combination stalled as ProFrac and USWS remained focused on their respective businesses.

On November 13, 2019, Mr. Uhlmer met with Messrs. Broussard and Shapiro while at an investor conference and expressed interest in speaking with Adam Klein, a director of USWS and partner at Crestview Partners, USWS' largest investor. Mr. Uhlmer mentioned that it could make sense for ProFrac and USWS to resume discussions but only if Crestview Partners would be supportive. Mr. Uhlmer asked for and received the contact information of Mr. Klein.

On November 19, 2019, Mr. Matthew D. Wilks and Mr. Uhlmer met with Mr. Klein, in his capacity as a representative of USWS' largest investor, to discuss Crestview Partners' investment thesis in USWS and to gauge whether he would support a potential business combination between ProFrac and USWS. After a high level conversation, Mr. Klein indicated that such a business combination could make strategic sense in the future, but the parties did not discuss a potential combination in the near term.

On December 12, 2019, Mr. Uhlmer met with Messrs. Broussard and O'Neill to discuss preliminary due diligence requirements in connection with a potential business combination.

On December 26, 2019, Mr. Uhlmer expressed to Mr. Shapiro ProFrac's openness to continuing to conduct due diligence on USWS' business despite the decision of ProFrac's owners to focus on its business operations rather than pursue a potential business combination. Despite Mr. Uhlmer's outreach, the parties did not continue their conversations because, among other things, the business combination under discussion at that time would have resulted in ProFrac being a part of a public company, which was not then an attractive option to ProFrac.

On February 19, 2021, Mr. Matthew D. Wilks contacted Messrs. Broussard and O'Neill to discuss the idea of the possibility of a mutually beneficial transaction. The conversation was high level and no details, terms, specifics or confidential information were discussed.

On March 4, 2021, ProFrac and USWS entered into an updated and extended confidentiality agreement. That same day, Messrs. Matthew D. Wilks and Shapiro spoke regarding valuation and structure for a potential business combination. Specifically, they discussed a reverse merger structure in which ProFrac would merge into USWS, valuing the combined companies between \$1.8 to \$2 billion, with ProFrac Stockholders owning 70% of the combined company, and a concurrent equity raise of \$100 to \$200 million.

During March 2021, ProFrac and USWS continued to conduct mutual due diligence, discussed matters related to operating performance, capital structure and valuation and analyzed issues relating to various potential transaction structures.

On April 1, 2021, ProFrac delivered a preliminary,non-binding term sheet to the USWS Board that proposed a combination of the two companies valuing USWS' total enterprise value at \$500 million with the ProFrac Stockholders owning 70% of the combined entity. The term sheet included a requirement that the combined entity raise \$100 million in equity at closing but provided that \$40 million of this would be backstopped by the Wilks family office. On April 1, 2021, the closing price for USWS Common Stock was \$1.13 per share (or \$23.73 on a split-adjusted basis), reflecting an implied enterprise value of \$508 million.

On April 2, 2021, the USWS Board, together with the USWS management team, held a special meeting to discuss the proposed business combination with ProFrac. The USWS Board determined to engage legal and financial advisors to assist it in evaluating the proposed business combination with ProFrac.

On April 5, 2021, USWS contacted the law firm of Porter Hedges LLP ('Porter Hedges''), which had historically served as outside counsel to USWS, to inform Porter Hedges of the potential business combination with ProFrac. Also, on April 5, 2021, the USWS Board held a meeting at which the USWS management team and its financial advisor, Piper Sandler, updated the USWS Board concerning its discussions with ProFrac that had occurred to date as well as its views on the proposed business combination with ProFrac. Piper Sandler additionally informed the USWS Board that ProFrac had engaged Piper Sandler to assist with diligence on a separate and unrelated matter. At that meeting, in consultation with Piper Sandler, and the USWS management

team, the USWS Board developed and approved a counterproposal, and authorized Piper Sandler to convey that counterproposal to ProFrac. The USWS Board determined to work with Piper Sandler, despite its ongoing work with ProFrac, because of Piper Sandler's extensive knowledge of USWS' business, intellectual property and capital structure, resulting from its prior work as USWS' financial advisor. As time was of the essence, Piper Sandler's ability to quickly understand USWS' capital needs was imperative and bringing in another advisor would have resulted in potentially critical delays in the process.

On April 8, 2021, Piper Sandler delivered the counterproposal on behalf of USWS to ProFrac, reflecting, among other things, an increase in USWS Stockholders' ownership of the combined company from 30% to 38.5%, an increase in the backstop by the Wilks family office of the equity raise from \$40 million and provisions regarding the composition of the board of directors of the combined company. ProFrac's proposal was silent on board composition. On April 8, 2022, the closing price of USWS Common Stock was \$1.03 (or \$21.63 on a split-adjusted basis), reflecting an implied total enterprise value of \$499 million in USWS' counterproposal. There was no initial response from ProFrac to this counterproposal and, after that, conversations regarding the business combination stalled.

On April 14, 2021, ProFrac's internal counsel spoke with USWS' outside litigation attorneys regarding USWS' intellectual property defense strategy. The parties decided to move forward with this aspect of ProFrac's due diligence in the event that conversations regarding the business combination resumed.

In late April 2021, ProFrac informed USWS that it intended to focus on its business operations rather than continue to pursue the proposed business combination.

On June 1, 2021, the Superior Court of the State of Delaware ruled against USWS in a litigation matter with Smart Sand related to a claim against USWS for a breach of a contract in which Smart Sand claimed damages against USWS of approximately \$54 million, which was not a matter related to the intellectual property matters ProFrac was reviewing in its due diligence. USWS appealed this decision, but in consideration of a potential need to settle the litigation matter for a meaningful amount of cash, USWS engaged Piper Sandler to explore sources of liquidity alternatives, including, among other things, restructuring and raising new capital. In its outreach, Piper Sandler contacted 43 groups, with 15 groups receiving materials. As part of its investor outreach efforts, Piper Sandler contacted Mr. Matthew D. Wilks who acted as a representative for THRC Holdings regarding a potential strategic investment in USWS. During the course of conversations with Piper Sandler, Mr. Matthew D. Wilks expressed an interest in potentially licensing USWS' intellectual property related to the USWS Clean Fleet®, electric hydraulic fracturing technology and equipment. Piper Sandler also contacted Crestview Partners and certain individuals affiliated with Mr. Matlin.

In connection with its review of various liquidity alternatives, the USWS Board formed a special committee consisting of Messrs. Treadwell, Burnett, Carroll and Habachy (the "USWS 2021 Special Committee"). The USWS 2021 Special Committee met on June 13, 2021, June 14, 2021 and June 15, 2021 to review and discuss various transaction structures and alternatives. The full USWS Board met on June 2, 2021, June 3, 2021 and June 7, 2021 to discuss the same

On June 24, 2021, the USWS 2021 Special Committee and the USWS Board, each acting by written consent, approved and USWS entered into a Note Purchase Agreement, as amended, pursuant to which it issued \$84.0 million of Convertible Senior Secured PIK Notes ("Notes") convertible into shares of USWS Common Stock (the "Equity Linked Notes") and \$22.5 million of Notes convertible into licenses to build three hydraulic fracturing fleets using USWS' Clean Fleet\* technology (the "License Linked Notes"). In connection with the transaction, THRC Holdings, one of the Wilks Parties, purchased \$25.0 million of the Equity Linked Notes and ProFrac purchased \$22.5 million of the License Linked Notes. On June 25, 2021, USWS issued an additional \$19.0 million of Equity Linked Notes, but no additional Notes were issued to ProFrac or any of the Wilks Parties. Crestview Partners also purchased \$20.0 million of the Equity Linked Notes for cash and \$20.0 million in Equity Linked Notes in exchange for 15,588 shares of USWS Series A Preferred Stock in the transaction, which was approved by the USWS 2021 Special Committee and the full USWS Board.

USWS agreed to settle the litigation with Smart Sand for \$35.0 million in cash and the entry into an agreement to provide Smart Sand certain rights of first refusal related to the supply of frac sand.

On June 29, 2021, ProFrac exercised its right to convert its License Linked Notes in full and USWS entered into a license agreement with ProFrac, which provided a five-year option to purchase up to 20 licenses to build and operate electric fleets using USWS' Clean Fleet® technology. Upon entry into the license agreement, ProFrac purchased three licenses to build and operate three electric fleets, each valued at \$7.5 million.

On July 7, 2021, the USWS 2021 Special Committee and the full USWS Board, each acting by written consent, approved an additional issuance of Notes.

On July 12, 2021, USWS issued an additional \$6.0 million of Equity Linked Notes, of which \$5.5 million was purchased by THRC Holdings and Farris Wilks, in the aggregate.

Between July 2021 and January 2022, representatives of USWS and ProFrac engaged in certain technical conversations regarding electric fleet technology in connection with the license agreement between the parties, but during that time there were no conversations regarding a potential business combination or any other transactions.

On January 11, 2022, the USWS Board held a special meeting at which the USWS management team and Piper Sandler discussed potential strategic alternatives to address liquidity and for funding growth capital expenditures related to newbuild electric fleets on order. Based on this meeting, the USWS Board engaged Piper Sandler on January 21, 2022, to (i) assist USWS with a debt capital markets process to address liquidity, including consideration of a last-out term loan financing, equipment financings and amendments to USWS's existing Senior Secured Term Loan Facility and (ii) review strategic alternatives and conduct outreach to counterparties regarding a potential business combination.

Between January 21, 2022 and early February 2022, Piper Sandler contacted ProFrac and four other strategic counterparties to assess interest levels in either a business combination with USWS or license sales of USWS's electric frac technology.

On January 26, 2022, the USWS Board formed a special committee of its independent and disinterested directors, consisting of David Treadwell, Richard Burnett, Ryan Carroll and Steve Habachy the ("USWS January Special Committee") to review and evaluate the terms of potential debt financing in light of Crestview Partners' potential participation in the transaction and its status as a large debt and equity holder in USWS and its representation on the USWS Board.

On February 9, 2022, USWS entered into confidentiality agreements with Company A and Company B. USWS answered preliminary due diligence questions from Company A and invited Company B to the facility where the USWS Clean Fleets® were being built. The process with both potential counterparties did not move quickly and as USWS was able to raise additional capital in the interim period, USWS was no longer interested in pursuing these potential transactions and these discussions did not result in further engagement.

On February 17, 2022, ProFrac delivered, by email to Piper Sandler, anon-binding indication of interest offering to acquire USWS at a total enterprise value between \$280 million and \$370 million. On February 17, 2022, the closing price of USWS Common Stock was \$0.94 (or \$5.64 on a split-adjusted basis), reflecting an implied total enterprise value of \$388 million for ProFrac's offer. This offer was considered and rejected by the USWS Board.

Simultaneously with the strategic alternatives review, Piper Sandler contacted both existing and potential new lenders to USWS to address liquidity. A total of 47 potential debt financing sources were contacted, but other than THRC Holdings, there was no third-party interest in additional lending to USWS. Throughout this process, USWS faced a growing working capital deficit, as well as milestone payments for its new build fleets that it needed to meet in order to comply with customer expectations. Given these liquidity needs, USWS decided to move forward with a potential financing transaction despite the limited third-party interest resulting from Piper Sandler's outreach.

On February 28, 2022, the USWS January Special Committee approved, and recommended that the USWS Board also authorize and approve, (i) creating an additional tranche of last-out term loans ("Term C Loans"), (ii) issuing \$10.0 million of Term C Loans and USWS Term C Loan Warrants to certain Crestview Partners related entities and (iii) authorizing USWS to issue up to \$25.0 million of additional Term C Loans. On the same day the USWS Board unanimously approved these same transactions and the issuances to certain Crestview Partners related entities were consummated.

On March 1, 2022, USWS (i) issued an additional \$10.0 million of Term C Loans to THRC Holdings and \$1.5 million of Term C Loans to David Matlin and (ii) issued USWS March Term C Loan Warrants to TRHC Holdings and David Matlin.

On May 23, 2022, ProFrac's Board held a meeting with the management of ProFrac to discuss a potential business combination with USWS, review the relative valuation and potential synergies from a potential business combination, and the potential terms and conditions for a potential business combination. The ProFrac Board authorized ProFrac's management to explore the possibility of a potential business combination with USWS.

On May 25, 2022, ProFrac delivered anon-binding indication of interest (**\*IOI\***) to the USWS Board through Piper Sandler. Pursuant to the IOI, ProFrac indicated its interest in potentially acquiring (a) all of the capital stock of USWS, on a debt-free basis, at a total enterprise value of \$380 million in exchange for a mix, to be determined, of cash and ProFrac Class A Common Stock, or (b) all of the USWS Class A Common Stock at \$0.70 per share, paid in shares of ProFrac Class A Common Stock, together with the assumption by ProFrac of certain debt and the payment of certain obligations of USWS, subject to a minimum cash on hand threshold of \$40 million. On May 25, 2022, the closing price of USWS Common Stock was \$0.61 per share (or \$3.66 on a split-adjusted basis), reflecting an implied total enterprise value of \$324 million for ProFrac's proposal.

On May 26, 2022, the USWS Board held a special meeting at which the USWS management team and Piper Sandler discussed the IOI. During the meeting, the USWS Board and the USWS management team agreed that the offer warranted further consideration and requested that Piper Sandler and Porter Hedges perform preliminary financial and legal analysis of the IOI.

On May 29, 2022, Mr. Matthew D. Wilks, Mr. Lance Turner (the Chief Financial Officer of ProFrac), and Mr. Rob Willette (the Chief Legal Officer of ProFrac), conducted a virtual due diligence meeting with Messrs. O'Neill and Shapiro, along with Piper Sandler, to discuss USWS' operations, the duration of its customer contracts and USWS' commercial strategy. During the meeting, Piper Sandler, on behalf of USWS, requested that ProFrac submit a revised IOI to acquire USWS reflecting any valuation and capital structure implications from diligence findings.

Between late May and early June 2022, representatives of ProFrac held discussions with Piper Sandler and USWS regarding the price per share set forth in the IOI and a go-shop provision with fiduciary out if and when the parties were to negotiate definitive documentation.

In late May 2022, ProFrac Board members discussed establishing a special committee of the ProFrac Board's independent and disinterested members to evaluate the potential transaction in light of the Wilks Parties' significant beneficial ownership of USWS securities, as well as the possibility that ProFrac might enter into additional transactions with other significant beneficial holders of USWS securities, which could benefit the Wilks Parties, in order to consummate a business combination with USWS. In advance of the formal appointment of the ProFrac Special Committee, Terry Glebocki, Gerald Haddock and Stacy Nieuwoudt, who are independent, disinterested members of the ProFrac Board and comprise ProFrac's Audit Committee, began working together informally to oversee ProFrac's efforts regarding the potential business combination. Ms. Glebocki, the Chairwoman of ProFrac's Audit Committee, initiated efforts to retain independent legal and financial advisors to advise the ProFrac Special Committee in its evaluation of the potential business combination. In connection with its preliminary activities, the ProFrac Special Committee initially engaged Latham & Watkins LLP ("Latham") to serve as its independent legal counsel. Latham served in that capacity from May 31 through June 8, 2022. Following further interviews of potential independent legal advisors, the ProFrac Special Committee engaged Kirkland & Ellis LLP ("Kirkland") to serve as its independent legal advisor on June 12, 2022.

On May 26, 2022, Piper Sandler informed members of USWS management of the transactions for which Piper Sandler was engaged as an advisor by ProFrac.

On May 31, 2022, the USWS Board held a special meeting, at which the USWS management team, Porter Hedges and Piper Sandler were present, to discuss a potential business combination with ProFrac and review the relative valuation and potential synergies from a potential business combination. In light of the compelling rationale for a combination, the USWS Board determined to continue conversations and mutual due diligence with ProFrac.

On June 2, 2022, at the request of ProFrac's management, Brown Rudnick LLP, ProFrac's outside counsel ('Brown Rudnick"), shared new potential deal terms with Latham, for the ProFrac Special Committee's consideration.

Later, on June 2, 2022, Mr. Matthew D. Wilks presented ProFrac management's business rationale for the potential business combination with USWS to the independent members of the ProFrac Board which would comprise the ProFrac Special Committee, via a telephonic meeting, along with additional details regarding the Wilks Parties' financial interests related to the potential acquisition. Also present at this telephonic presentation were Mr. Willette, Latham and Brown Rudnick.

On June 3, 2022, the ProFrac Board formed the ProFrac Special Committee to evaluate, review and consider the proposed business combination, and, as appropriate, negotiate the proposed business combination and the terms thereof. Following a determination that each (i) were not members of the ProFrac management, (ii) were not affiliated with and were independent of any persons with a potential interest in the proposed business combination (other than an affiliation with ProFrac by virtue of serving as a director of ProFrac), and (iii) did not have a direct or indirect interest in the proposed business combination that was in addition to, different from, or otherwise adverse to the interests of ProFrac and its stockholders, generally, Terry Glebocki, Gerald Haddock and Stacy Nieuwoudt were appointed as members of the ProFrac Special Committee.

On June 3, 2022, the ProFrac Special Committee and ProFrac management, together with Latham and Brown Rudnick, held a meeting to review and revise a non-binding revised indication of interest ("Revised IOI"). Also, on June 3, 2022, the ProFrac Special Committee, by unanimous written consent, approved ProFrac sending the Revised IOI to USWS, and a representative of ProFrac sent the Revised IOI to Piper Sandler, for USWS' consideration. Under the proposed terms of the Revised IOI, ProFrac offered to acquire all of the USWS Class A Common Stock at \$1.06 per share, paid in shares of ProFrac Class A Common Stock, together with the assumption by ProFrac of certain debt and payment of certain obligations of USWS, subject to the minimum cash on hand threshold of \$40 million. The closing price of USWS Common Stock on June 3, 2022 was \$0.85 per share (or \$5.10 on a splitadjusted basis), reflecting an implied total enterprise value of \$343 million.

Members of USWS Management reviewed the various conflicts resulting from Piper Sandler's work with ProFrac and discussed the same with Messrs. Treadwell and Broussard and, on June 5, 2022, informed Piper Sandler that they could proceed with providing a fairness opinion.

On June 6, 2022, the USWS Board held a special meeting with members of management and its financial and legal advisors to further discuss a potential business combination with ProFrac and the Revised IOI. Representatives of Piper Sandler updated the USWS Board on the implied valuation of the proposed business combination based on the Revised IOI assuming the then current trading prices and analyzed the potential value and risk associated with remaining a standalone company versus combining with ProFrac. Based on the analysis, as well as the strong rationale for combining with ProFrac and review of other strategic alternatives available to USWS, the USWS Board agreed that Piper Sandler should propose an exchange ratio of 0.0561 shares of ProFrac Class A Common Stock (at the time valued at \$1.10 per share) for each share of USWS Common Stock, and directed Piper Sandler to contact ProFrac and convey the counterproposal. Piper Sandler delivered this feedback later that day. Additionally, on behalf of the USWS Board, Piper Sandler requested a 45 day "go-shop" period and a termination fee of \$3.0 million during the go-shop period and \$5.0 million after the go-shop period concluded.

On the morning of June 8, 2022, Mr. Matthew D. Wilks held a meeting with members of the ProFrac Board to discuss the terms and conditions of the Revised IOI and the counterproposal received from Piper Sandler.

Also, on the morning on June 8, 2022, a representative of the Wilks Parties indicated to ProFrac management flexibility as to the type of consideration they would accept as USWS securityholders in any transaction, if it was to be determined that such an approach might mitigate any potential or actual conflict of interest concerns.

Later, on June 8, 2022, Mr. Willette held discussions with Mr. O'Neill and Mr. Shapiro, along with representatives from Brown Rudnick and Porter Hedges, concerning USWS' outstanding debt and equity securities, as well as its long-term incentive plan. Also on June 8, 2022, USWS granted ProFrac and its legal advisors access to a "virtual data room," allowing ProFrac to perform its review of previously requested due diligence documentation.

In addition, on June 8, 2022, the USWS Board formed a special committee of its independent and disinterested directors, consisting of David Treadwell, Ryan Carroll and Steve Habachy (the "USWS Special Committee"), to evaluate the proposed business combination with ProFrac as well as any alternative transactions, in light of Crestview Partners' large equity stake and debt holdings in USWS and its representation on the USWS Board, as well as the expectation that in any merger transaction Crestview Partners would be requested to enter into an agreement to vote in favor of the transaction and agree to the treatment of its convertible equity securities in USWS. Paul Hastings LLP ("Paul Hastings") was engaged as legal advisor to the USWS Special Committee.

Between June 8, 2022 and June 21, 2022, representatives of ProFrac had discussions with Crestview Partners and other significant holders of convertible securities in USWS regarding the treatment of their securities in the potential business combination, and their willingness to support such a transaction.

On June 9, 2022, Messrs. Turner and Shapiro conducted a virtual meeting with representatives of Piper Sandler to discuss potential synergies that may result from the potential business combination of ProFrac and USWS.

On June 10, 2022, Brown Rudnick discussed with representatives of ProFrac management the terms of a preliminary,non-binding indication of interest in connection with the potential purchase by ProFrac of all of the securities of USWS held by Crestview Partners and its affiliates (the "Crestview IOI"). Brown Rudnick sent the Crestview IOI to the ProFrac Special Committee for its review. Following the ProFrac Special Committee's approval, a representative of ProFrac sent the Crestview IOI to Crestview Partners. Pursuant to the Crestview IOI, ProFrac indicated its interest in: (i) adjusting the timing and terms of the conversion into USWS Common Stock of the shares of USWS Series A Preferred Stock from \$6.67 to \$1.34 and require no later than July 19, 2022 Crestview Partners and its affiliates to elect to either (A) convert all of its shares of USWS Series A Preferred Stock into USWS Common Stock or (B) be paid in cash by USWS at the adjusted conversion price of \$1.34; (ii) terminating and cancelling the USWS February Term C Loan Warrants in exchange for cash at \$1.276 per share, less the strike price of such USWS February Term C Loan Warrants; (iii) terminating and cancelling the USWS March Term C Loan Warrants; (iv) amending the terms of the USWS Equity Linked Convertible Notes such that the conversion price for each USWS Equity Linked Convertible Notes holders would elect to convert all of their respective outstanding principal and interest amounts under the USWS Equity Linked Convertible Notes holders would elect to convert all of their respective outstanding principal and interest amounts under the USWS Equity Linked Convertible Notes holders would elect to convert all of their respective outstanding principal and interest amounts under the USWS Equity Linked Convertible Notes into USWS Common Stock no later than July 19, 2022; and (v) tolling interest accruing on the Term C Loans owned by Crestview Partners and its affiliates commencing on June 30, 2022. All similarly situated securityholders in each class of

On June 10, 2022, the ProFrac management team, the USWS management team, Piper Sandler, and members of both the ProFrac and USWS Boards (attended by Mr. David Treadwell and Mr. Adam Klein of the USWS Board) met virtually to conduct a mutual business and operational due diligence session. The companies discussed key commercial, operational, strategic and financial due diligence related to ProFrac and USWS.

On June 11, 2022, Mr. Klein contacted Mr. Matthew D. Wilks to respond to the ProFrac Crestview Proposal on a number of points. First, Mr. Klein advised that the proposed conversion price of \$1.34 for both the Series A Preferred Stock and the PIK Notes was insufficient and noted that it was Crestview Partners' understanding that such a conversion would require USWS shareholder approval. Second, Crestview Partners would not want to convert its Series A Preferred Stock or PIK Notes into common unless there was certainty that the merger would close. Third, Crestview Partners would want its outside advisor expenses reimbursed by USWS. Following that discussion, Mr. Wilks agreed, subject to the approval of the ProFrac Special Committee, that the conversion prices on the Series A Preferred and the PIK Notes would be \$1.22 and that Crestview Partners advisor expenses would be reimbursed at closing. Mr. Wilks agreed to have ProFrac's legal counsel connect with Vinson & Elkins LLP, counsel to Crestview Partners ("Vinson & Elkins") on whether a pre-closing conversion would require a USWS shareholder vote.

Later that same day (June 11, 2022), Mr. Willette sent Porter Hedges an initial draft of the Merger Agreement, which had been prepared by Brown Rudnick. The draft Merger Agreement reflected, among other things, a taxable two-step transaction structure with a mixture of cash and stock consideration, a 45-day go-shop period, certain closing conditions, including conditions relating to the receipt by ProFrac and USWS of the necessary consents and/or waivers under their respective credit facilities and other debt agreements, as well as customary representations, warranties and interim operating covenants.

On June 12, 2022, the ProFrac Special Committee held a telephonic meeting attended by Mr. Willette and representatives of Kirkland to discuss ProFrac management's financial projections and the analyses to be performed by representatives of Jefferies, which the ProFrac Special Committee, in consultation with Kirkland, had chosen to retain as its financial advisor and to provide, if requested by the ProFrac Special Committee, an opinion regarding a potential business combination with USWS. Additionally, on June 12, 2022, Mr. Willette sent Porter Hedges a second draft of the Merger Agreement. In this revised draft, the go-shop period was shortened from 45 days to 30 days.

Also, on June 12, 2022, Messrs. Turner and Shapiro, along with representatives of Piper Sandler, held further discussions regarding potential synergies created by the proposed business combination.

Beginning on June 13, 2022, ProFrac, USWS, Piper Sandler and the legal advisors to ProFrac, USWS and their respective special committees held daily update calls to discuss the status of transaction documentation, key outstanding diligence or documentation items, and other items. Also, on June 13, 2022, the USWS Board and the USWS management held a special meeting with Piper Sandler and Porter Hedges to review the then current status of the transaction and key proposed terms.

On June 13, 2022, the USWS Special Committee and management held a special meeting with representatives from Piper Sandler, Porter Hedges and Paul Hastings to review the then current status of the transaction and key proposed terms. At the meeting, each member of the USWS Special Committee was asked to disclose any conflicts of interest, either directly or indirectly that may pertain to the proposed transaction with ProFrac. Mr. Carroll disclosed that TCW Asset Management Company, LLC ("TCW") held an interest in the debt of ProFrac. The members of the USWS Special Committee determined that TCW's interest in ProFrac was immaterial and would not be an impediment to Mr. Carroll serving on the committee.

On June 13, 2022, the ProFrac Special Committee held a telephonic meeting attended by representatives of Kirkland and Jefferies. Representatives of Kirkland provided an overview of the ProFrac Special Committee's fiduciary duties and key issues in the draft Merger Agreement prepared by Brown Rudnick and sent to USWS on June 12, 2022.

On June 13, 2022, Porter Hedges sent Kirkland and Brown Rudnick an updated draft of the Merger Agreement reflecting comments from USWS, the USWS Special Committee and Crestview Partners, that contemplated, among other items, (a) removal of a financing cooperation covenant, (b) removal of the financing condition and conditions relating to the receipt by ProFrac and USWS of the necessary consents and/or waivers under their respective credit facilities and other debt agreements, (c) a termination fee to be paid by USWS to

ProFrac equal to 3% of the equity value of the transaction for a termination following the Go-Shop Period End Date and 1% of the equity value of the transaction for a termination prior to the Go-Shop Period End Date, (d) a 45-day go-shop period and (e) the insertion of a specific performance provision.

After consultation with the ProFrac Special Committee, on June 14, 2022, representatives from Kirkland sent a draft Support Agreement to Porter Hedges, Paul Hastings, Brown Rudnick and Vinson & Elkins LLP. The draft Support Agreement provided, among other things, that USWS insiders would vote their shares of USWS in favor of the Merger Agreement and against any alternative acquisition agreement.

On June 14, 2022, the ProFrac Special Committee held a telephonic meeting attended by representatives of Kirkland. At this meeting, the ProFrac Special Committee approved entering into an engagement letter with Jefferies to act as the ProFrac Special Committee's financial advisor in connection with the proposed transaction, which engagement letter was subsequently entered into on June 15, 2022. Kirkland then led a discussion regarding key issues set forth in Porter Hedges' revised draft of the Merger Agreement, including, among other items, the proposal on the termination fee to be paid by USWS to ProFrac, closing conditions, and the go-shop, non-solicitation and change in recommendation provisions of the Merger Agreement. In addition, representatives of Kirkland conveyed USWS' desire for an all-stock transaction structured as a one-step merger instead of a transaction with a mix of cash and stock consideration structured as a two-step tender offer as contemplated by Brown Rudnick's June 12<sup>th</sup> draft of the Merger Agreement.

On June 15, 2022, the ProFrac Special Committee held a telephonic meeting attended by representatives of Kirkland, Jefferies and Potter Anderson & Corroon LLP ("Potter Anderson"), the ProFrac Special Committee's Delaware counsel in connection with the potential transaction. During the meeting, the participants reviewed the intellectual property of USWS and its impact on the valuation of USWS. Additionally, representatives of Jefferies provided an update on its review of the proposed transaction. Kirkland and Potter Anderson again reviewed the ProFrac Special Committee's fiduciary duties and discussed generally the purpose and impact of obtaining a fairness opinion in a conflicted transaction.

In addition, on June 15, 2022, the ProFrac Special Committee held another telephonic meeting attended by representatives of Kirkland, Brown Rudnick, Potter Anderson and Mr. Matthew D. Wilks. During the meeting, the ProFrac Special Committee discussed with Mr. Matthew D. Wilks the timing of the proposed business combination between ProFrac and USWS, as well as the deal protection provisions contemplated by Porter Hedges' June 13th draft of the Merger Agreement.

Later, on June 15, 2022, the ProFrac Special Committee held an additional telephonic meeting attended by representatives of Kirkland. During the meeting, the ProFrac Special Committee and representatives of Kirkland further discussed the timing of the proposed transaction and of the delivery of the fairness opinion and determined to request a presentation from members of ProFrac management regarding the potential business combination.

On June 15, 2022, Vinson & Elkins sent Kirkland and Brown Rudnick, Porter Hedges and Paul Hastings (a) initial drafts of the amendment to USWS Equity Linked Convertible Notes and COD Amendment reflecting, among other things, a conversion price for the USWS Equity Linked Convertible Notes and Series A Redeemable Convertible Preferred Stock of \$1.22 and, due to structural considerations, a deadline for conversion of USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes into USWS Common Stock of immediately before closing of the Merger rather than July 19, 2022 (as initially set forth in the Crestview IOI) and (b) a revised draft of the Support Agreement that, among other things, provided that the obligation for certain USWS insiders to vote their shares in favor of the transaction would fall away in the event there was an Adverse Recommendation Change (as defined in the Merger Agreement).

On June 15, 2022, Kirkland sent Porter Hedges an updated draft of the Merger Agreement that contemplated, among other items, (a) an all stock-transaction structured as an one-step merger as previously requested by USWS, (b) reinsertion of the financing cooperation covenant, and (c) a counterproposal that the termination fee to be paid by USWS to ProFrac be based on the enterprise value of the transaction given the size of USWS' equity value relative to its outstanding debt and the potential repayment by ProFrac of such debt at the closing of the proposed business combination.

In the evening of June 15, 2022, the USWS Special Committee held a telephonic meeting with representatives of Paul Hastings and Piper Sandler to discuss, among other things, the status of the negotiations, possible responses to key issues in the Merger Agreement and the timing of the Jefferies opinion.

In the evening of June 15, 2022, the ProFrac Special Committee had an additional meeting by videoconference attended by representatives of Kirkland, Potter Anderson and ProFrac's management. Representatives of ProFrac's management provided the ProFrac Special Committee with a presentation relating to the businesses of ProFrac and USWS. In addition, ProFrac management discussed with the ProFrac Special Committee certain financial aspects of the proposed business combination and its strategic rationale, including the expansion of ProFrac's fleet through the addition of USWS' electric fleets, savings on fuel costs and costs of repair and maintenance and license fees otherwise payable to USWS, the expansion of ProFrac's portfolio of intellectual property, expected synergies resulting from the potential business combination and ProFrac management's valuation of the net present value of earnings that would result from the potential business combination. ProFrac management also provided the ProFrac Special Committee with an overview of the due diligence conducted by ProFrac and Brown Rudnick to date.

On June 16, 2022, representatives of Kirkland verbally conveyed to Porter Hedges the importance of removing the "go shop" provision and "force the vote" provision from the Merger Agreement in order to increase certainty of closing to ProFrac.

Also on June 16, 2022, the USWS Special Committee held a telephonic meeting with representatives of USWS' management team, Piper Sandler and Paul Hastings. At the meeting, a representative of Paul Hastings discussed with the USWS Special Committee certain legal and governance considerations and protocols relating to the USWS Special Committee's consideration of a potential business combination between USWS and ProFrac. Representatives from Paul Hastings also discussed certain issues raised by ProFrac's counsel during negotiations relating to the termination fee, "go-shop" and "force the vote" provisions.

On June 16, 2022, Porter Hedges sent Kirkland and Brown Rudnick an updated draft of the Merger Agreement that contemplated, among other items, (a) a proposed end date of six months from the execution of the Merger Agreement and (b) a counterproposal to set the termination fee to be paid by USWS to ProFrac at \$5 million with a termination fee of \$1.667 million in the event USWS accepts a Superior Proposal (as defined elsewhere in this proxy statement/information statement/prospectus) prior to the Go-Shop Period End Date.

On June 16, 2022, Vinson & Elkins sent Kirkland, Brown Rudnick, Porter Hedges and Paul Hastings an initial draft of the Warrant Purchase Agreement and the PIK Interest Letter Agreement.

On June 18, 2022, Mr. Turner and Mr. Shapiro, with assistance from representatives of Jefferies and Piper Sandler, conducted a virtual due diligence session during which USWS' capital structure was discussed.

Also, on June 18, 2022, representatives from Kirkland sent revised drafts to Porter Hedges of the Merger Agreement and Support Agreement along with the amendment to USWS Equity Linked Convertible Notes, the COD Amendment and the Warrant Purchase Agreement. Due to structural considerations, the timing of conversion of USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes into USWS Common Stock had changed from the Crestview IOI from July 19 to immediately before closing of the Merger. Also reflecting negotiations of ProFrac and Crestview Partners, the conversion prices for the USWS Equity Linked Convertible Notes and USWS Series A Preferred Stock changed from \$1.34 in the Crestview IOI to \$1.22 and the holders of the USWS Series A Preferred Stock no longer had the right to elect to receive cash. The revised draft of the Merger Agreement contemplated a single-step merger and included a "force the vote" provision requiring USWS to submit the transaction to its stockholders regardless of a Change of Recommendation by the USWS Board. In addition, as discussed with the ProFrac Special Committee, this revised draft of the Merger Agreement eliminated the go-shop period and increased the termination fee to \$15 million. The revised draft of the Support Agreement provided for, among other things, the obligation of certain USWS insiders to vote their shares in favor of the transaction even in the event there was an Adverse Recommendation Change.

Between June 18, 2022 and June 19, 2022, Kirkland sent Porter Hedges updated drafts of the Merger Agreement that contemplated, among other items, (a) removal of the "go-shop" provision in the Merger Agreement paired with a "force the vote" provision, (b) a counterproposal to set the end date at nine months following the execution of the Merger Agreement, and (c) a counterproposal to set the termination fee to be paid by USWS to ProFrac at \$15 million.

On June 19, 2022, the USWS management team, the USWS Special Committee, and representatives from Crestview Partners, Porter Hedges, Vinson & Elkins, and Piper Sandler held a call to discuss material issues outstanding related to the Merger Agreement. Based on the consensus views reached on this call, a representative of Paul Hastings sent a list of key open issues to representatives of Brown Rudnick and Kirkland, which list highlighted, among other issues, maintaining a fiduciary termination right with respect to superior proposals and a \$5 million termination fee, while accepting the removal of the go-shop provision and accepting that the voting agreements would not terminate upon a mere change of board recommendation (but the aggregate shares subject to the voting commitment would be cut back to 35% of USWS' outstanding shares in such event (the "Voting Reduction").

Also on June 19, 2022, Vinson & Elkins sent a revised draft of the amendment to USWS Equity Linked Convertible Notes and the Support Agreement to Kirkland and Brown Rudnick. The revised draft of the USWS Equity Linked Convertible Notes provided that, among other things, in certain circumstances including in the event that the Merger is consummated, interest on USWS Equity Linked Convertible Notes would cease accruing as of July 9, 2022. The revised draft of the Support Agreement provided for, among other things, the Voting Reduction.

On the morning of June 20, 2022, Porter Hedges sent Kirkland and Brown Rudnick an updated draft of the Merger Agreement that contemplated, among other items, (a) revisions to the tax treatment of the potential transaction, (b) a counterproposal to set the termination fee to be paid by USWS to ProFrac at \$5 million, and (c) removal of the "force the vote" provision.

Later on June 20, 2022, representatives of ProFrac communicated to representatives of Piper Sandler that ProFrac needed additional time to determine whether ProFrac was willing to sign transaction documents contemplating a tax-free reorganization. Calls among Paul Hastings, Mr. Treadwell and other members of the USWS Special Committee were held to discuss timing and other considerations. The USWS Special Committee determined that it was in the best interests of the USWS stockholders to concede the taxable deal structure in light of (i) the ease with which USWS Stockholders could sell a relatively small percentage of ProFrac's outstanding common stock being received in the Merger into a highly liquid market to satisfy tax obligations and (ii) the risk to the transaction of a potential one-week or longer delay to signing. The USWS Special Committee further determined that it was in the best interests of USWS Stockholders to continue to push for the fiduciary termination right.

Later in the morning on June 20, 2022, representatives of Kirkland, Brown Rudnick, Paul Hastings and Porter Hedges held a call to discuss the key open items. Following discussions, the parties agreed on a taxable deal structure and to restore USWS' fiduciary termination right with respect to an unsolicited superior proposal.

On June 20, 2022, the USWS Special Committee met with representatives from Paul Hastings and Piper Sandler to discuss the potential business combination. Representatives from Paul Hastings reviewed the duties of the USWS Board in the context of the proposed business combination as well as a detailed summary of the terms of the Merger Agreement, the Support Agreement, the amendment to the USWS Equity Linked Convertible Notes, the COD Amendment, the Warrant Purchase Agreement and other ancillary documents. Representatives from Piper Sandler presented their valuation analysis of both USWS and ProFrac, and answered questions from the members of the USWS Special Committee concerning their analyses. Piper Sandler indicated to the USWS Special Committee that they would be ready to deliver its fairness opinion. In light of the various reasons considered and discussions between the USWS Special Committee and its legal and financial advisors, subject to (i) the transaction documents being finalized in a manner satisfactory to USWS' management team and Paul Hastings and reflecting discussions with the USWS Special Committee, (ii) agreement on a termination fee of no

more than \$8 million and (iii) the delivery of the Piper opinion, members of the USWS Special Committee unanimously (a) determined that the Merger Agreement and related transactions were fair to, and in the best interest of, USWS and the holders of shares of USWS Common Stock, and recommended submission of the Merger Agreement and related transactions to the USWS Board, and (b) recommended that the USWS Board approve the Merger Agreement and the related transactions, and determine that the Merger Agreement and the related transactions are advisable, fair to, and in the best interest of, USWS and the holders of USWS Common Stock and that the transaction be submitted to the USWS Stockholders for their approval.

Later on June 20, 2022, following the USWS Special Committee meeting, the USWS Board met with representatives of Porter Hedges, Piper Sandler, Crestview Partners and Paul Hastings, to discuss the potential business combination. Representatives of Piper Sandler began the meeting by indicating that ProFrac had just announced its deals with each of Signal Peak and with Flotek, after which Piper Sandler presented orally its financial analysis of the transaction. Representatives of Paul Hastings then led a detailed discussion on the negotiations of the Merger Agreement and the terms approved by the USWS Special Committee. The USWS Special Committee then informed the USWS Board that it had unanimously approved the proposed transaction, subject to (i) the finalization of transaction documents, (ii) a termination fee of no more than \$8 million and (iii) the delivery of the fairness opinion of Piper Sandler.

On June 20, 2022, the ProFrac Special Committee held a meeting attended by representatives of Kirkland and Jefferies. During the initial portion of the meeting, representatives of Jefferies provided an update regarding its review of the proposed transaction, and noted that representatives of Jefferies would re-join the meeting at a later time to review with the ProFrac Special Committee Jefferies' financial analysis of the proposed transaction. Kirkland then provided a summary of the terms of Merger Agreement, the Support Agreement and the other ancillary agreements, including the COD Amendment, the Warrant Purchase Agreement, the amendment to the Equity Linked Convertible Notes and the PIK Interest Letter Agreement, and discussed with the ProFrac Special Committee antitrust and intellectual property matters. Following this summary, representatives of ProFrac management and Brown Rudnick joined the meeting. The ProFrac Special Committee discussed with management the terms of the Merger Agreement, the Support Agreement and the other ancillary agreements, including the issues relating to the removal of the "force the vote" provision as well as discussion regarding antitrust matters. Representatives of Brown Rudnick also updated the ProFrac Special Committee on the due diligence conducted in connection with the proposed business combination, and following such update representatives of ProFrac management and Brown Rudnick left the meeting. Representatives of Jefferies then re-joined the meeting and reviewed its financial analysis of the Exchange Ratio with the ProFrac Special Committee and rendered an oral opinion, confirmed by delivery of a written opinion dated June 21, 2022, to the ProFrac Special Committee to the effect that, as of that date and based on and subject to various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken as described in the opinion, the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement was fair, from a financial point of view, to ProFrac. After discussion and deliberation, the ProFrac Special Committee adopted resolutions, in which the ProFrac Special Committee unanimously (i) determined that the Merger Agreement, and the transactions contemplated thereby, were advisable, fair to and in the best interest of ProFrac and unaffiliated stockholders of ProFrac and (ii) approved and declared advisable the Merger Agreement and the transactions contemplated thereby. The ProFrac Special Committee further unanimously (i) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) determine that the Merger Agreement and the transactions contemplated thereby were fair to and in the best interests of ProFrac and the unaffiliated stockholders of ProFrac and (y) approve and declare advisable the Merger Agreement and the transactions contemplated thereby, and (ii) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) direct the ProFrac Stock Issuance be submitted to ProFrac's stockholders for their approval and (y) recommend that ProFrac's stockholders approve of the ProFrac Stock Issuance.

Later that day, Kirkland sent to Porter Hedges an updated draft of the Merger Agreement that contemplated, among other items, (a) revisions to the structure of the potential transaction, as agreed by the parties, (b) acceptance of Porter Hedges' removal of the "force the vote" provision, and (c) a counterproposal to set the

termination fee to be paid by USWS to ProFrac at \$12 million. Several discussions between representatives of Piper Sandler and Paul Hastings and representatives of ProFrac were held to reduce the termination fee to \$8 million. Final discussions continued through the evening and into the morning of June 21, 2022.

Following the distribution by Kirkland of the revised Merger Agreement, ProFrac, USWS and their respective legal advisors further revised and finalized the Merger Agreement and disclosure schedules thereto, the Support Agreement and the other ancillary agreements. Each member of the ProFrac Special Committee subsequently approved the final version of the Merger Agreement, which contemplated an \$8 million termination fee to be paid by USWS to ProFrac.

On June 21, 2022, Piper Sandler delivered its written opinion, confirming that as of such date and based upon and subject to the assumptions, procedures, factors, qualifications and limitations set forth therein, the Exchange Ratio provided for in the business combination was fair, from a financial point of view, to USWS. Following delivery of Piper Sandler's written opinion, the USWS Special Committee and USWS Board determined that conditions to their recommendation and approval were satisfied.

Following approval by the ProFrac Board on June 21, 2022 upon the recommendation of the ProFrac Special Committee, and approval by the USWS Board on June 21, 2022 upon the recommendation of the USWS Special Committee, the parties entered into the Merger Agreement on June 21, 2022.

On June 22, 2022, ProFrac and USWS each filed a Current Report on Form8-K with the SEC, announcing the entry into the Merger Agreement.

### ProFrac Board and ProFrac Special Committee Reasons for the Merger

On June 20, 2022, the ProFrac Special Committee unanimously (i) determined that the Merger Agreement and the transactions contemplated thereby were fair to, and in the best interest of, ProFrac and the ProFrac Unaffiliated Stockholders and (ii) approved and declared advisable the Merger Agreement and the transactions contemplated thereby. The ProFrac Special Committee further unanimously (i) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) determine that the Merger Agreement and the transactions contemplated thereby were fair to, and in the best interest of, ProFrac and the ProFrac Unaffiliated Stockholders and (y) approve and declare advisable the Merger Agreement and the transactions contemplated thereby and (ii) resolved, subject to the terms and conditions set forth in the Merger Agreement, to recommend that the ProFrac Board (x) direct that the ProFrac Stock Issuance be submitted to the ProFrac Stockholders for their approval and (y) recommend that the ProFrac Stockholders approve the ProFrac Stock Issuance.

In making the above determinations, the ProFrac Special Committee consulted with ProFrac's management, other members of the ProFrac Board, and independent financial and legal advisors to the ProFrac Special Committee.

On June 21, 2022, the ProFrac Board, acting on the unanimous recommendation and approval of the ProFrac Special Committee, unanimously (i) determined that the Merger Agreement and the transactions contemplated thereby were fair to, and in the best interests of, ProFrac and the ProFrac Unaffiliated Stockholders, (ii) authorized, approved, and declared advisable the Merger Agreement and the transactions contemplated thereby, and (iii) directed that the ProFrac Stock Issuance be submitted to ProFrac's Stockholders for their approval, and recommended that ProFrac's Stockholders approve the ProFrac Stock Issuance. The ProFrac Board unanimously recommended that ProFrac Stockholders vote "FOR" the ProFrac Stock Issuance proposal.

In reaching its decision to approve the Merger Agreement and to recommend that ProFrac Stockholders approve the ProFrac Stock Issuance, the ProFrac Board consulted with ProFrac's management, the ProFrac Special Committee, and its legal advisors.

The ProFrac Special Committee and the ProFrac Board considered a number of factors when evaluating the Merger, many of which support their determination that the Merger and the other related transactions contemplated by the Merger Agreement were fair to, and in the best interests of, ProFrac and the ProFrac Unaffiliated Stockholders. The ProFrac Special Committee and the ProFrac Board considered these factors as a whole and without assigning relative weights to each such factor, and overall considered the relevant factors to be favorable to, and in support of, their determinations and recommendations.

*Strategic Considerations*. The ProFrac Board and the ProFrac Special Committee considered a number of factors pertaining to the strategic rationale for the Merger as supporting its decision to approve the Merger, including the following:

- the expansion of ProFrac's presence in the electric fleet offering by adding 9 electric fleets currently active or under construction;
- the expansion of ProFrac's portfolio to 44 active fleets, including 28 NextGen, low emissions fleets;
- the elimination of the patent license agreement between ProFrac and USWS that requires ProFrac to pay a license fee for each new electric fleet that ProFrac constructs. For more information about the patent license agreement, see section titled "The Merger Interests of ProFrac Executive Officers and Directors in the Merger—Commercial Arrangements with USWS";
- the cost savings opportunity that could be realized if future ProFrac electric fleets are constructed using ProFrac's in-house manufacturing capabilities;
- the belief that the Merger will improve ProFrac's efficiency for its operations, lower operation and maintenance costs, lower diesel costs and provide an unrivaled cost savings to its customers;
- the fact that the Merger will provide ProFrac access to USWS' expansive intellectual property portfolio, including over 80 and 240 granted and pending patents, respectively;
- · the belief that the Merger will be accretive to ProFrac's 2023 financial results, and that it will maintain ProFrac's conservative balance sheet;
- · the belief that the Merger will create value going forward and will ensure that ProFrac has access to USWS' Clean Fleet technology; and
- the belief that the combination of vertical integration, fuel savings from natural gas-powered fleets and technological innovation enable ProFrac to create value and generate best-in-class customer offering.

Other Factors Considered by the ProFrac Board and the ProFrac Special Committee. In addition to considering the factors described above, the ProFrac Board and the ProFrac Special Committee considered the following additional factors, all of which they viewed as supporting the decision to approve the Merger:

- the business, operations, financial condition, earnings and prospects of each of ProFrac, USWS and the combined company;
- current and historical prices and trading information with respect to each of ProFrac's and USWS' common stock, which assisted the ProFrac Special Committee and the ProFrac Board in their respective conclusion that the Exchange Ratio was fairly priced;
- the fact that the Merger Consideration represented a premium to USWS Stockholders of approximately 68% based on the closing prices of each company's stock on Nasdaq on June 21, 2022, the last trading day before the Merger was publicly announced, and that the Exchange Ratio is fixed, which the ProFrac Special Committee and the ProFrac Board believed was consistent with market practice for mergers of this type and with the strategic purpose of the Merger;
- the expected treatment of the Merger as a taxable transaction, as described in the section entitled "Material United States Federal Income Tax Consequences of the Merger";

- the current and prospective competitive climate in the industries in which ProFrac and USWS operate, including the potential for consolidation, and the alternatives reasonably available to ProFrac if it did not pursue the Merger;
- the expected benefits to customers and suppliers of ProFrac and USWS, and the opportunities for the employees of the combined company;
- the belief that the terms and conditions of the Merger Agreement are reasonable, including:
  - the fact that they were determined througharm's-length negotiations conducted at the direction of the ProFrac Special Committee, USWS, USWS Special Committee, and their respective representatives and advisors;
  - · the fact that the representations and warranties and covenants are generally reciprocal;
  - the fact that the conditions to Closing are limited;
  - the ability of ProFrac, under certain circumstances, to terminate the Merger Agreement without any termination fees payable to USWS:
  - the fact that upon termination of the Merger Agreement under specified circumstances, USWS may be required to pay ProFrac a
    termination fee of \$8,000,000 or reimburse ProFrac for certain expenses in an amount up to \$3,000,000, in each case, depending on the
    termination event:
  - · the fact that the termination fee provisions are the product of negotiations; and
  - · the fact that the size of the termination fees are reasonable in light of the size and benefits of the Merger.
- · the anticipated market capitalization, earnings and adjusted earnings per share and capital structure of the combined company; and
- · in addition, the ProFrac Board considered the factors described in the section titled "Factors Considered by the ProFrac Special Committee"

The ProFrac Board and the ProFrac Special Committee weighed these advantages and opportunities against a number of other factors identified in its deliberations that weighed negatively against the Merger, including:

- the possibility that the Merger might not be completed or that completion might be delayed or subject to conditions that may be imposed by governmental authorities;
- the challenges inherent in the combination of two businesses of the size and scope of ProFrac and USWS and the possible diversion of
  management's attention for an extended period of time;
- the risk that the combined company might not retain key employees despite its best efforts;
- the risk of not capturing all the anticipated cost savings and operational synergies between ProFrac and USWS and the risk that other anticipated benefits might not be realized;
- risks associated with the incurrence of significant indebtedness to complete the Merger, such as the increased leverage of the combined company and restrictions on operations of the combined company following the completion of the Merger, as well as risks relating to ProFrac's ability to fund its obligations in connection with the Merger;
- regulatory and litigation risks associated with the transaction or with the combination of the two companies;
- the fees and expenses associated with completing the Merger and retaining key personnel;
- the fact that some directors and executive officers of ProFrac have interests in the Merger that may be different from, or are in addition to, the interests of ProFrac's stockholders generally;
- the fact that ProFrac's Controlling Stockholders have a substantial interest in USWS;

- the fact that the Controlling Stockholders of ProFrac immediately prior to the Merger will own approximately 82.7% of the combined company immediately following the Merger (assuming conversion of all of USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes):
- the fact that the Merger could have a dilutive effect on earnings per share of ProFrac calculated in accordance with GAAP;
- · the fact that ProFrac expects to incur additional charges related to write-offs in connection within-process research and development; and
- the other risks of the type and nature described under "Risk Factors", and the matters described under "Cautionary Statement Regarding Forward-Looking Statements."

### Factors Considered by the ProFrac Special Committee

As a result of the actual or potential conflicts of interests listed under the section 'Interests of ProFrac Executive Officers and Directors in the Merger," the ProFrac Board formed the ProFrac Special Committee and provided the ProFrac Special Committee with the power and authority to, among other things, evaluate, review, consider and negotiate a potential transaction involving the acquisition of USWS by ProFrac. The ProFrac Special Committee consists of Gerald Haddock, Stacy Nieuwoudt and Terry Glebocki.

The ProFrac Special Committee considered the following procedural safeguards and process matters to be generally positive or favorable in making its determinations and approvals and related recommendations to the ProFrac Board:

- the ProFrac Special Committee consists of three independent directors who (i) are not members of ProFrac's management, (ii) are not affiliated with and are independent of any persons with a potential interest in a potential transaction involving the acquisition of USWS by ProFrac (other than an affiliation with ProFrac by virtue of serving as a director of ProFrac), and (iii) did not have a direct or indirect interest in a potential transaction involving the acquisition of USWS by ProFrac that was in addition to, different from, or otherwise adverse to the interests of ProFrac and its stockholders, generally;
- the ProFrac Board empowered the ProFrac Special Committee to evaluate, review, consider and negotiate a potential transaction involving the acquisition of USWS by ProFrac;
- the ProFrac Special Committee had the power to "say no" with respect to a potential transaction involving the acquisition of USWS by ProFrac, in that the ProFrac Board resolved to not approve a potential transaction involving the acquisition of USWS by ProFrac without a prior favorable recommendation of such transaction by the ProFrac Special Committee;
- · the ProFrac Special Committee retained and was advised by, experienced and qualified outside legal and financial advisors;
- the ProFrac Special Committee held more than 20 meetings as part of its evaluation, review, consideration and negotiation of a potential transaction involving the acquisition of USWS by ProFrac;
- the ProFrac Special Committee received the opinion, dated June 21, 2022, of Jefferies to the ProFrac Special Committee as to the fairness, from a financial point of view and as of such date, to ProFrac of the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement, which opinion was based on and subject to various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken as further described in the heading "Opinion of the Financial Advisor to the ProFrac Special Committee";
- the terms and conditions of the Merger Agreement were determined througharm's-length negotiations conducted at the direction of the ProFrac Special Committee and USWS Special Committee and their respective representatives and advisors; and

the compensation of the members of the ProFrac Special Committee was in no way contingent on their approval of any transaction.

This discussion of the information and factors considered by the ProFrac Special Committee and the ProFrac Board in reaching their respective conclusion and recommendations includes all of the material factors considered by the ProFrac Special Committee and the ProFrac Board but is not intended to be exhaustive and is not provided in any specific order or ranking. In view of the wide variety of factors considered by the ProFrac Special Committee and the ProFrac Board in evaluating the Merger Agreement and the related transactions contemplated thereby, and the complexity of these matters, neither the ProFrac Special Committee nor the ProFrac Board found it practicable to, and did not attempt to, quantify, rank or otherwise assign relative weight to those factors. In addition, different members of the ProFrac Special Committee and the ProFrac Board may have given different weight to different factors. The ProFrac Special Committee and the ProFrac Board did not reach any specific conclusion with respect to any of the factors considered and instead conducted an overall analysis of such factors and determined that, in the aggregate, the potential benefits considered outweighed the potential risks or possible negative consequences of approving the Merger Agreement and the issuance of ProFrac Class A Common Stock pursuant to the Merger Agreement.

It should be noted that this explanation of the reasoning of the ProFrac Special Committee and the ProFrac Board and all other information presented in this section is forward-looking in nature and, therefore, should be read in light of the factors discussed in the section titled "Cautionary Statement Regarding Forward-Looking Statements."

# **ProFrac Prospective Financial Information**

ProFrac prefers not to make public projections as to future performance, earnings or other results due to the unpredictability of the underlying assumptions and estimates. However, in connection with its evaluation of the Merger and related transactions, ProFrac's management prepared and provided to USWS and to the ProFrac Board certain non-public, internal financial projections concerning ProFrac's future performance as a standalone business for the fiscal years ending December 31, 2022 through 2025. These projections do not give effect to the Merger.

ProFrac has included below a summary of these financial forecasts, which it refers to as the "projections." In compiling the projections, ProFrac's management considered historical performance, combined with estimates regarding revenues, gross profit, working capital, adjusted EBITDA, as defined below, capital expenditures and total free cash flows. ProFrac's management also provided the projections to Jefferies and approved the projections for Jefferies' use in connection with Jefferies' financial analysis and opinion. The summary of the projections is not being included in this proxy statement/information statement/prospectus to influence a stockholder's decision whether to vote for or against the proposal to adopt the Merger Agreement but is being included because the projections were provided to USWS, ProFrac's Board, and Jefferies.

The projections included in this proxy statement/information statement/prospectus have been prepared by, and are the responsibility of, ProFrac's management. The projections were not prepared with a view toward public disclosure or toward complying with GAAP, the published guidelines of the SEC regarding projections or the guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information. Grant Thornton, ProFrac's independent registered public accounting firm, has not examined, compiled or performed any procedures with respect to the projections and, accordingly, does not express an opinion or any other form of assurance with respect thereto.

The projections were based on numerous variables and assumptions that are inherently uncertain and many of which are beyond the control of ProFrac's management. ProFrac's management has limited visibility into future activity levels and pricing for ProFrac's services in future periods because they depend on several macroeconomic, global commodity price levels, and industry specific variables outside of ProFrac's control. As a result, the assumptions upon which the projections were based necessarily involve judgments with respect to, among other things, industry performance, general business, economic, regulatory, market and financial conditions, as well as matters specific to ProFrac's business, all of which are difficult or impossible to predict accurately and many of which are beyond ProFrac's control.

In light of the foregoing factors and the uncertainties inherent in the projections, stockholders are cautioned not to place undue, if any, reliance on the projections.

The following table and other information below is a summary of the projections (dollars in millions, except as noted).

		Management Projections			
	F	Fiscal Year Ending December 31			
	2022E	2023E	2024E	2025E	
Revenue	\$2,218	\$2,703	\$3,178	\$3,507	
Adjusted EBITDA (1)	\$ 721	\$ 992	\$1,151	\$1,184	
Operating Cash Flow (2)	\$ 389	\$ 741	\$ 904	\$ 958	
Capital Expenditures	\$ (268)	\$ (270)	\$ (304)	\$ (300)	
Acquisitions	\$ (90)	\$ —	\$ —	\$ —	
Free Cash Flow to Debt Service (3)	\$ 31	\$ 471	\$ 600	\$ 658	

- (1) Adjusted EBITDA means net income before (i) interest expense, net, (ii) income tax provision, (iii) depreciation, depletion and amortization, (iv) loss on disposal of assets, (v) stock-based compensation and (vi) other non-recurring charges.
- (2) Operating Cash Flow means net income adjusted fornon-cash items and changes in working capital.
- (3) Free Cash Flow to Debt Services means Operating Cash Flow less Capital Expenditures and cash paid for Acquisitions.

# Opinion of the Financial Advisor to the ProFrac Special Committee

The ProFrac Special Committee retained Jefferies as its independent financial advisor in connection with the Merger. In connection with this engagement, the ProFrac Special Committee requested that Jefferies evaluate the fairness, from a financial point of view, to ProFrac of the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement. At a meeting of the ProFrac Special Committee held on June 20, 2022, Jefferies rendered an oral opinion, confirmed by delivery of a written opinion dated June 21, 2022, to the ProFrac Special Committee to the effect that, as of that date and based on and subject to various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken as described in its opinion, the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement was fair, from a financial point of view, to ProFrac.

The full text of Jefferies' opinion, which describes the various assumptions made, procedures followed, matters considered and limitations and qualifications on the review undertaken by Jefferies, is attached as Annex C to this information statement and is incorporated herein by reference. ProFrac encourages you to read the opinion carefully and in its entirety. Jefferies' opinion was provided for the use and benefit of the ProFrac Special Committee (in its capacity as such) in its evaluation of the Exchange Ratio from a financial point of view and did not address any other aspect of the Merger or any other matter. Jefferies' opinion did not address the relative merits of the Merger or other transactions contemplated by the Merger Agreement as compared to any alternative transaction or opportunity that might be available to ProFrac, nor did it address the underlying business decision by ProFrac to engage in the Merger or any term, aspect or implication of any other agreement entered into in connection with, or contemplated by or resulting from, the Merger or otherwise. Jefferies' opinion did not constitute a recommendation as to how the ProFrac Special Committee or the ProFrac Board, and does not constitute a recommendation as to how any securityholder, should vote or act with respect to the Merger or any other matter. The following summary is qualified in its entirety by reference to the full text of Jefferies' opinion.

In arriving at its opinion, Jefferies, among other things:

reviewed a draft, dated June 21, 2022, of the Merger Agreement;

- reviewed certain publicly available financial and other information about USWS and ProFrac;
- reviewed certain information furnished to Jefferies and approved for its use by ProFrac, including financial forecasts and analyses, relating to
  the business, operations and prospects of USWS prepared by management of USWS (the "USWS Forecasts");
- reviewed certain information furnished to Jefferies and approved for its use by ProFrac, including financial forecasts and analyses, relating to the business, operations and prospects of ProFrac prepared by management of ProFrac (the "ProFrac Forecasts");
- reviewed certain estimates furnished to Jefferies and approved for its use by ProFrac as to potential cost savings and revenue synergies (including the amount and timing thereof) expected by the management of ProFrac to result from the Merger (the "Synergies");
- · held discussions with members of senior management of ProFrac concerning the matters described in the second through fifth clauses above;
- reviewed the share trading price history and valuation multiples for USWS Common Stock and ProFrac Class A Common Stock and compared them with those of certain publicly traded companies that Jefferies deemed relevant;
- considered certain potential pro forma financial effects of the Merger on ProFrac utilizing the financial forecasts and estimates relating to ProFrac and USWS and the potential Synergies referred to above; and
- · conducted such other financial studies, analyses and investigations as Jefferies deemed appropriate.

In Jefferies' review and analysis and in rendering its opinion, Jefferies assumed and relied upon, but did not assume any responsibility to independently investigate or verify, the accuracy and completeness of all financial and other information that was supplied or otherwise made available to Jefferies by ProFrac or that was publicly available to Jefferies (including, without limitation, the information described above), or that was otherwise reviewed by Jefferies. Jefferies relied on assurances of the management of ProFrac that it was not aware of any facts or circumstances that would make any of the foregoing information incomplete, inaccurate or misleading. In Jefferies' review, Jefferies did not obtain any independent evaluation or appraisal of any of the assets or liabilities (contingent, accrued, derivative, off-balance sheet or otherwise), nor did Jefferies conduct a physical inspection of any of the properties or facilities of, USWS or ProFrac, and Jefferies was not furnished with and assumed no responsibility to obtain, any such evaluations, appraisals or physical inspections. Jefferies did not evaluate the solvency or fair value of USWS, ProFrac or any other entity under any laws relating to bankruptcy, insolvency or similar matters.

With respect to the financial forecasts provided to and reviewed by Jefferies, Jefferies noted that projecting future results of any company is inherently subject to uncertainty. The ProFrac Special Committee informed Jefferies, however, and Jefferies assumed, at the direction of the ProFrac Special Committee, that the USWS Forecasts were reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of USWS as to the future financial performance of USWS and the other matters covered thereby. Jefferies also assumed, at the direction of the ProFrac Special Committee, that the ProFrac Forecasts were reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of ProFrac as to the future financial performance of ProFrac and the other matters covered thereby. With respect to the future financial performance of ProFrac giving pro forma effect to the Merger, at the direction of the ProFrac Special Committee and with its consent, Jefferies assumed that the Synergies would be realized in the amounts and at the times projected. Jefferies expressed no opinion as to the USWS Forecasts or the ProFrac Forecasts or the assumptions on which they were based, including with respect to the Synergies.

Jefferies relied upon the assessments of the management of ProFrac as to, among other things, (i) the potential impact on USWS and ProFrac of market, competitive, seasonal, macroeconomic, geopolitical and other conditions, trends and developments in and prospects for, and governmental, regulatory and legislative matters relating to or affecting, the energy industry and the geographic regions and local communities in which USWS and ProFrac operate, and (ii) the ability of ProFrac to integrate the businesses of USWS and ProFrac and to realize the potential Synergies. Jefferies assumed that there would not be any developments with respect to any such matters that would have an adverse effect on USWS, ProFrac or the Merger (including the contemplated benefits thereof) or that otherwise would be meaningful in any respect to Jefferies' analyses or opinion.

Jefferies' opinion was based on economic, monetary, regulatory, market and other conditions existing and which could be evaluated as of the date thereof. Jefferies expressly disclaimed any undertaking or obligation to advise any person of any change in any fact or matter affecting its opinion of which Jefferies becomes aware after the date thereof. Jefferies further noted that the current volatility and disruption in the energy, credit and financial markets relating to, among other things, the COVID-19 pandemic and armed conflict in or relating to Ukraine, may or may not have an effect on USWS or ProFrac, and Jefferies did not express an opinion as to the effects of such volatility or such disruption on the Merger or any party to the Merger.

Jefferies made no independent investigation of, and Jefferies expressed no view or opinion as to, any legal, regulatory, accounting or tax matters affecting or relating to USWS or ProFrac, and Jefferies assumed the correctness in all respects material to its analyses and opinion of all legal, regulatory, accounting and tax advice given to USWS, ProFrac and the ProFrac Special Committee, including, without limitation, with respect to changes in, or the impact of, accounting standards or tax and other laws, regulations and governmental and legislative policies affecting USWS or ProFrac or the Merger and legal, regulatory, accounting and tax consequences of the terms of, and transactions contemplated by, the Merger Agreement and related documents to USWS and ProFrac and their respective stockholders. In addition, in preparing its opinion, Jefferies did not take into account any tax consequences of the transaction to any holder of USWS Common Stock or ProFrac Common Stock. Jefferies assumed that the Merger would be consummated in accordance with the terms of the Merger Agreement without waiver, modification or amendment of any term, condition or agreement and in compliance with all applicable laws, documents and other requirements and that the final form of the Merger Agreement would be substantially similar to the last draft reviewed by Jefferies also assumed that in the course of obtaining the necessary governmental, regulatory or third-party approvals, consents, waivers and releases for the Merger or otherwise, including with respect to any divestitures or other requirements, no delay, limitation, restriction or condition would be imposed or occur that would have an adverse effect on USWS, ProFrac or the contemplated benefits of the Merger or that otherwise would be material in any respect to Jefferies' analyses or opinion.

In addition, Jefferies was not requested to and did not provide advice concerning the structure, the specific amount of the Exchange Ratio, or any other aspects of the Merger, or to provide services other than the delivery of Jefferies' opinion. Jefferies was not authorized to and did not solicit any expressions of interest from any other parties with respect to the purchase or sale of all or any part of USWS or ProFrac or any other alternative transaction. Jefferies did not participate in negotiations with respect to the terms of the Merger and related transactions. Consequently, Jefferies assumed that such terms were the most beneficial terms from ProFrac's perspective that could under the circumstances be negotiated among the parties to such transactions, and no opinion was expressed whether any alternative transaction might result in consideration more favorable to ProFrac than that contemplated by the Merger Agreement.

Jefferies' opinion did not address the relative merits of the transactions contemplated by the Merger Agreement as compared to any alternative transaction or opportunity that might be available to USWS or ProFrac, nor did it address the underlying business decision by USWS or ProFrac to engage in the Merger or the terms of the Merger Agreement or the documents referred to therein, including the form or structure of the Merger or any term, aspect or implication of any other agreements, arrangements or understandings entered into in connection with, or contemplated by or resulting from the Merger or otherwise. Jefferies was not asked to address, and its opinion did not address, the fairness to, or any other consideration of, the holders of any class of securities, creditors or other constituencies of USWS or ProFrac or any other party, other than the fairness of the

Exchange Ratio provided for in the Merger pursuant to the Merger Agreement. Jefferies expressed no view or opinion as to the actual value of ProFrac Class A Common Stock when issued in connection with the Merger or the price at which shares of USWS Common Stock or ProFrac Class A Common Stock or any other securities may trade or otherwise be transferrable at any time. Furthermore, Jefferies did not express any view or opinion as to the fairness, financial or otherwise, of the amount or nature of any compensation or other consideration payable to or to be received by any of USWS' or ProFrac's officers, directors or employees, or any class of such persons, in connection with the Merger relative to the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement or otherwise. Jefferies' opinion was authorized by the Fairness Committee of Jefferies LLC.

In connection with rendering its opinion to the ProFrac Special Committee, Jefferies performed certain financial and comparative analyses, including those described below. The following summary is not a complete description of all analyses performed and factors considered by Jefferies in connection with its opinion. The preparation of a financial opinion is a complex process involving subjective judgments and is not necessarily susceptible to partial analysis or summary description. With respect to the selected public companies analyses summarized below, no company used as a comparison was identical or directly comparable to ProFrac, USWS or the Merger nor were individual multiples observed for each of the selected companies, in isolation, independently determinative of the results of Jefferies' selected public companies analyses. These analyses necessarily involved complex considerations and judgments concerning financial and operating characteristics and other factors that could affect the public trading, acquisition or other values of the companies or transactions concerned.

Jefferies believes that its analyses and the summary below must be considered as a whole and in context and that selecting portions of its analyses and factors or focusing on information presented in tabular format, without considering all analyses and factors or the narrative description of the analyses, could create a misleading or incomplete view of the processes underlying Jefferies' analyses and opinion. Jefferies did not draw, in isolation, conclusions from or with regard to any one factor or method of analysis for purposes of its opinion, but rather arrived at its ultimate opinion based on the results of all analyses undertaken by it and assessed as a whole.

The estimates of the future performance of ProFrac and USWS in or underlying Jefferies' analyses are not necessarily indicative of future results or values, which may be significantly more or less favorable than those estimates. In performing its analyses, Jefferies considered industry performance, general business and economic conditions and other matters, many of which are beyond the control of ProFrac and USWS. Estimates of the financial value of companies or businesses do not purport to be appraisals or necessarily reflect the prices at which companies, businesses or securities actually may be sold or acquired. Accordingly, the estimates used in,

and the implied reference ranges resulting from, any particular analysis described below are inherently subject to substantial uncertainty and should not be taken as Jefferies' view of the actual value of ProFrac, USWS or their respective businesses or securities.

The Exchange Ratio was determined through negotiations between ProFrac and USWS, and the decision by ProFrac to enter into the Merger Agreement was solely that of the ProFrac Board based on the recommendation of the ProFrac Special Committee. Jefferies' opinion and financial analyses were only one of many factors considered by the ProFrac Special Committee in its evaluation of the Exchange Ratio and should not be viewed as determinative of the views of the ProFrac Special Committee, the ProFrac Board or ProFrac management with respect to the Merger or the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement.

# Financial Analyses

The summary of the financial analyses described in this section entitled 'Financial Analyses' is a summary of the material financial analyses reviewed with the ProFrac Special Committee and performed by Jefferies in connection with its analyses and opinion. The financial analyses summarized below include information presented in tabular format. In order to fully understand Jefferies' financial analyses, the tables must be read together with the text of each summary. The tables alone do not constitute a complete description of the financial analyses. Considering the data below without considering the full narrative description of the financial analyses, including the methodologies and assumptions underlying the analyses, could create a

misleading or incomplete view of Jefferies' financial analyses. The order in which the financial analyses summarized below appear does not necessarily reflect the relative importance or weight given to such analyses. The following quantitative information, to the extent that it is based on market data, is based on market data as it existed on or before June 17, 2022 and is not necessarily indicative of current or future market conditions.

For purposes of the financial analyses described below, the term "Exchange Ratio" means the exchange ratio as defined in the Merger Agreement as of June 21, 2022 of 0.0561 of a share of ProFrac Class A Common Stock for each outstanding share of USWS Common Stock. The term "Implied Merger Consideration" means \$1.19 per share, calculated based on the Exchange Ratio of 0.0561 and the closing price of \$21.26 as of June 17, 2022 for the ProFrac Class A Common Stock. The term "fully diluted shares" with respect to ProFrac includes the impact from outstanding ProFrac dilutive securities, and the term "fully diluted shares" with respect to USWS includes the effect of the conversion of USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes and the impact of USWS dilutive securities.

On August 4, 2022, USWS effected a 1-for-6 reverse split of its Common Stock. After giving effect to USWS' reverse split, the Exchange Ratio as defined in the Merger Agreement has been adjusted from 0.0561 to 0.3366.

### ProFrac Financial Analyses

Selected Public Companies Analysis. Jefferies reviewed publicly available financial, stock market and operating information of ProFrac and the following selected publicly traded companies in the oilfield services industry that Jefferies considered generally relevant for purposes of analysis, based on Jefferies' professional judgment and taking into account, among other things, business and financial profiles and market capitalizations of such companies, collectively referred to as the "ProFrac selected companies":

- Liberty Energy Inc.
- · NexTier Oilfield Solutions Inc.
- ProPetro Holding Corp.
- · RPC, Inc.

Jefferies reviewed, among other information and to the extent publicly available, enterprise values of the ProFrac selected companies, calculated as fully diluted equity values based on closing stock prices on June 17, 2022, plus total debt, preferred equity and non-controlling interests, minus cash and cash equivalents (as applicable), as a multiple of such company's estimated earnings before interest, taxes, depreciation and amortization, referred to as EBITDA, for the calendar years 2022, 2023 and 2024, which we refer to as 2022E EBITDA, 2023E EBITDA and 2024E EBITDA, respectively. Financial data of the ProFrac selected companies were based on publicly available research analysts' estimates, public filings and other publicly available information.

The results of this analysis were as follows:

## Selected Public Companies Analysis - ProFrac

Financial Metric	Low	High	Mean	Median
2022E EBITDA	3.6x	6.9x	5.3x	5.3x
2023E EBITDA	2.6x	5.3x	3.9x	3.8x
2024E EBITDA	2.4x	5.0x	3.5x	3.3x

Jefferies applied a selected range of enterprise value to estimated EBITDA multiples of 4.25x to 6.25x, 3.0x to 5.0x, and 2.5x to 4.5x, to corresponding data of ProFrac based on the ProFrac Forecasts for 2022E EBITDA, 2023E EBITDA and 2024E EBITDA, respectively, to determine a range of implied enterprise values for ProFrac. Jefferies then subtracted ProFrac's projected net debt and corporate adjustments as of September 30, 2022, as provided by ProFrac management, to calculate a range of implied equity values, and divided the result by the number of fully diluted shares of ProFrac Class A Common Stock to calculate a range of implied equity values per share for ProFrac. This analysis indicated the reference ranges of implied equity values per share set forth in the table below, as compared to the ProFrac Class A Common Stock closing price of \$21.26 as of June 17, 2022.

#### Selected Public Companies Analysis - ProFrac

		Implied Equity Valu		
Financial Metric	Selected Multiple Range		Per Share	
2022E EBITDA	4.25x - 6.25x	\$	18.70 - \$28.79	
2023E EBITDA	3.00x - 5.00x	\$	18.08 - \$31.96	
2024E EBITDA	2.50x - 4.50x	\$	17.38 - \$33.47	

No company utilized in the selected public companies analysis is identical to ProFrac. In evaluating the selected public companies, Jefferies made judgments and assumptions with regard to industry performance, general business, economic, market and financial conditions and other matters, many of which are beyond ProFrac's and Jefferies' control.

Discounted Cash Flow Analysis. Jefferies performed a discounted cash flow analysis of ProFrac by calculating the estimated present value of the standalone unlevered, after-tax free cash flows that ProFrac was forecasted to generate during the fourth calendar quarter of 2022 and the calendar years ending December 31, 2023 through December 31, 2025 based on the ProFrac Forecasts. Terminal values of ProFrac were calculated by applying to ProFrac's calendar year ending December 31, 2025 estimated adjusted EBITDA, based on the ProFrac Forecasts, a range of enterprise value to adjusted EBITDA multiples of 4.25x to 6.25x, selected based on Jefferies' professional judgment and taking into account, among other things, observed implied enterprise value to EBITDA multiples of the ProFrac selected companies. The present values (as of September 30, 2022) of the unlevered free cash flows and terminal values were then calculated using a selected discount rate range of 14.9% to 16.9%, based on an estimate of ProFrac's weighted average cost of capital, to determine a range of implied enterprise values for ProFrac. Jefferies then subtracted ProFrac's projected net debt and corporate adjustments as of September 30, 2022, as provided by ProFrac management, to calculate a range of implied equity values, and divided the result by the number of fully diluted shares of ProFrac Class A Common Stock to calculate a range of implied equity values per share for ProFrac. This analysis indicated a reference range of implied equity values per share of ProFrac Class A Common Stock closing price of \$21.26 as of June 17, 2022.

### **USWS Financial Analyses**

Selected Public Companies Analysis. Jefferies reviewed publicly available financial, stock market and operating information of USWS and the following selected publicly traded companies in the oilfield services industry that Jefferies considered generally relevant for purposes of analysis, based on Jefferies' professional judgment and taking into account, among other things, business and financial profiles and market capitalizations of such companies, collectively referred to as the "USWS selected companies":

- Liberty Energy Inc.
- NexTier Oilfield Solutions Inc.
- ProFrac Holding Corp.
- ProPetro Holding Corp.
- RPC, Inc.

Jefferies reviewed, among other information and to the extent publicly available, enterprise values of the USWS selected companies, calculated as fully diluted equity values based on closing stock prices on June 17, 2022, plus total debt, preferred equity and non-controlling interests, minus cash and cash equivalents (as applicable), as a multiple of such company's estimated EBITDA for the calendar years 2022, 2023 and 2024, respectively. Financial data of the USWS selected companies were based on publicly available research analysts' estimates, public filings and other publicly available information.

The results of this analysis were as follows:

### Selected Public Companies Analysis - USWS

Financial Metric	Low	High	Mean	Median
2022E EBITDA	3.6x	6.9x	5.3x	5.3x
2023E EBITDA	2.6x	5.3x	3.8x	3.7x
2024E EBITDA	2.4x	5.0x	3.5x	3.4x

Jefferies applied a selected range of enterprise value to estimated EBITDA multiples of 4.25x to 6.25x, 3.0x to 5.0x, and 2.5x to 4.5x, to corresponding data of USWS based on the USWS Forecasts for 2022E EBITDA, 2023E EBITDA and 2024E EBITDA, respectively, to determine a range of implied enterprise values for USWS. Jefferies then subtracted USWS' projected net debt and corporate adjustments as of September 30, 2022, as provided by USWS management, to calculate a range of implied equity values, and divided the result by the number of fully diluted shares of USWS Common Stock to calculate a range of implied equity values per share for USWS. This analysis indicated the reference ranges of implied equity values per share set forth in the table below, as compared to the Implied Merger Consideration of \$1.19 per share.

# Selected Public Companies Analysis - USWS

		Impl	ied Equity Value
Financial Metric	Selected Multiple Range		Per Share
2022E EBITDA	4.25x - 6.25x	\$	0.12 - \$0.62
2023E EBITDA	3.00x - 5.00x	\$	1.27 - \$2.60
2024E EBITDA	2.50x - 4.50x	\$	1.21 - \$2.75

No company utilized in the selected public companies analysis is identical to USWS. In evaluating the selected public companies, Jefferies made judgments and assumptions with regard to industry performance, general business, economic, market and financial conditions and other matters, many of which are beyond USWS' and Jefferies' control.

Discounted Cash Flow Analysis. Jefferies performed a discounted cash flow analysis of USWS by calculating the estimated present value of the standalone unlevered, after-tax free cash flows that USWS was forecasted to generate during the fourth calendar quarter of 2022 and the calendar years ending December 31, 2023 through December 31, 2025 based on the USWS Forecasts. Terminal values of USWS were calculated by applying to USWS' calendar year ending December 31, 2025 estimated adjusted EBITDA, based on the USWS Forecasts, a range of enterprise value to adjusted EBITDA multiples of 4.25x to 6.25x, selected based on Jefferies' professional judgment and taking into account, among other things, observed implied enterprise value to EBITDA multiples of the USWS selected companies. The present values (as of September 30, 2022) of the unlevered free cash flows and terminal values were then calculated using a selected discount rate range of 18.3% to 20.3%, based on an estimate of USWS' weighted average cost of capital, to determine a range of implied enterprise values for USWS. Jefferies then subtracted USWS' projected net debt and corporate adjustments as of September 30, 2022, as provided by USWS management, to calculate a range of implied equity values, and divided the result by the number of fully diluted shares of USWS Common Stock to calculate a range of implied equity values per share for USWS. This analysis indicated a reference range of implied equity values per share for USWS. Common Stock of \$2.14 to \$3.16, as compared to Implied Merger Consideration of \$1.19 per share.

## Implied Exchange Ratio Analysis

Jefferies calculated reference ranges of implied exchange ratios, based on the selected public companies analyses and discounted cash flow analyses described above, by using the range of implied per share equity values for ProFrac and the range of implied per share equity values for USWS. Jefferies calculated the range of implied exchange ratios by (i) dividing the low end of the reference ranges of per share equity values for USWS by the

high end of the reference ranges of per share equity values for ProFrac, and (ii) dividing the high end of the reference ranges of per share equity values for USWS by the low end of the reference ranges of per share equity values for ProFrac. This analysis indicated the reference ranges of implied exchange ratios set forth in the table below, as compared to the Exchange Ratio provided for in the Merger Agreement as of June 21, 2022 of 0.0561.

# Implied Exchange Ratio Analysis

Financial Metric	Im	ProFrac plied Equity Value Per Share	S Implied Equity	Implied Exchange Ratio
Selected Public Companies Analysis			 	
2022E EBITDA	\$	18.70 - \$28.79	\$ 0.12 - \$0.62	0.0042x - 0.0330x
2023E EBITDA	\$	18.08 - \$31.96	\$ 1.27 - \$2.60	0.0398x - 0.1440x
2024E EBITDA	\$	17.38 - \$33.47	\$ 1.21 - \$2.75	0.0360x - 0.1584x
Discounted Cash Flow Analysis	\$	29.06 - \$41.11	\$ 2.14 - \$3.16	0.0522x - 0.1089x

#### Miscellaneous

ProFrac has agreed to pay Jefferies for its financial advisory services to the ProFrac Special Committee in connection with the Merger an aggregate fee of \$1 million, which was payable upon delivery of Jefferies' opinion to the ProFrac Special Committee. In addition, ProFrac agreed to reimburse Jefferies for expenses, including fees and expenses of counsel, incurred in connection with Jefferies' engagement and to indemnify Jefferies and related parties against liabilities, including liabilities under federal securities laws, arising out of or in connection with the services rendered and to be rendered by Jefferies under its engagement.

As the ProFrac Special Committee was aware, although Jefferies and its affiliates did not provide financial advisory or financing services to ProFrac or USWS unrelated to the Merger during the two-year period prior to the date of Jefferies' opinion for which Jefferies and its affiliates received compensation, Jefferies and its affiliates may provide such services to ProFrac or USWS and/or their respective affiliates in the future, for which services Jefferies and its affiliates would expect to receive compensation. In the ordinary course of business, Jefferies and its affiliates may trade or hold securities or financial instruments (including loans and other obligations) of ProFrac or USWS, and/or their respective affiliates for Jefferies' own account and for the accounts of Jefferies' customers and, accordingly, may at any time hold long or short positions in those securities.

Jefferies was selected as the ProFrac Special Committee's independent financial advisor in connection with the Merger because, among other things, Jefferies is an internationally recognized investment banking firm with substantial experience in merger and acquisition transactions and based on its familiarity with ProFrac's business and industry. Jefferies is regularly engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, leveraged buyouts, negotiated underwritings, competitive bids, secondary distributions of listed and unlisted securities and private placements.

## Interests of ProFrac Executive Officers and Directors in the Merger

The ProFrac Stockholders should note that some directors and executive officers of ProFrac have interests in the Merger that are different from, or are in addition to, the interests of ProFrac's stockholders generally. These interests may present them with actual or potential conflicts of interest, and these interests, to the extent material, are described below.

## Family Relationships

Ladd Wilks, the Chief Executive Officer of ProFrac, is the son of Farris Wilks and nephew of Dan Wilks, and Matthew D. Wilks, the Executive Chairman of the ProFrac Board, is the son of Dan Wilks and nephew of Farris Wilks. Additionally, Matthew D. Wilks is the V.P. of Investments of THRC Holdings, an affiliate of Dan Wilks.

The Wilks Parties collectively hold a controlling interest in ProFrac and beneficially own certain securities of USWS, as described below. THRC Management is the general partner of THRC Holdings. Dan Wilks is the sole member of THRC Management. As a result, each of Dan Wilks and THRC Management may be deemed to have shared voting and dispositive power over, and to share beneficial ownership of, the securities owned by THRC Holdings.

### Wilks Brothers, LLC

Sergei Krylov, one of ProFrac's directors, serves as Investment Partner and Chief Financial Officer of Wilks Brothers, LLC (an entity controlled by Dan Wilks and Farris Wilks).

### Wilks Parties' Investments in USWS Securities

Unless noted otherwise, the figures in this section are not adjusted for the 1-for-6 reverse stock split implemented by USWS on August 4, 2022.

On June 24, 2021, THRC Holdings purchased an aggregate of \$25,000,000 in principal amount of USWS Equity Linked Convertible Notes from USWS.

On July 1, 2021, THRC Holdings assigned and transferred \$12,500,000 of the USWS Equity Linked Convertible Notes to Farris Wilks in exchange for \$12,500,000.

On July 12, 2021, THRC Holdings and Farris Wilks each purchased an additional \$2,750,000 in principal amount of USWS Equity Linked Convertible Notes from USWS.

On March 1, 2022, USWS issued to THRC Holdings 6,976,744 USWS February Term C Loan Warrants exercisable for no more than 6,976,744 shares of USWS Common Stock in connection with the entry by THRC Holdings into a certain Joinder to Senior Secured Term Loan Agreement. For more information about the Senior Secured Term Loan and the Term C Loans, see "Management's Discussion and Analysis of Financial Condition and Results of Operations of USWS."

Upon the consummation of the Merger, the Wilks Parties will receive an aggregate of 4,138,596 shares of ProFrac Class A Common Stock as Merger Consideration (assuming, solely for the purpose of this calculation, a hypothetical closing date of October 31, 2022), which, based on the ProFrac Class A Common Stock June 21, 2022 closing price, would be approximately \$88.9 million.

The table below summarizes THRC Holdings' and Farris Wilks's holdings of USWS securities, on an individualized basis, and the number of shares of ProFrac Class A Common Stock that will be issued by ProFrac to each of THRC Holdings and Farris Wilks as a result of the Merger and the related transactions contemplated thereby.

USWS Security	USWS Common Stock (on an as converted basis) <sup>(1)</sup>	ProFrac Class A Common Stock (0.3366 Exchange Ratio)
9,568 shares of USWS Common Stock	THRC Holdings: 9,568 shares	THRC Holdings: 3,220
	• Farris Wilks: 0	• Farris Wilks: 0
5,198 shares of USWS Series A Preferred Stock <sup>2)</sup>	<ul><li>THRC Holdings: 1,125,194</li><li>Farris Wilks: 0</li></ul>	<ul><li>THRC Holdings: 378,740</li><li>Farris Wilks: 0</li></ul>
2,666,669 USWS Series A Warrants	THRC Holdings: 126,985	THRC Holdings: 42,744
	• Farris Wilks: 0	• Farris Wilks: 0
6,976,744 USWS Term C Loan Warrants	6,976,744 shares of USWS Common Stock	0 shares of ProFrac Class A Common Stock(3)
\$80,765,592 in aggregable principal amount of USWS	• THRC Holdings: 8,578,481	• THRC Holdings: 2,887,516
Equity Linked Convertible Notes (including the PIK interest thereunder as of July 9, 2022)	• Farris Wilks: 2,455,070	• Farris Wilks: 826,376
Total		• THRC Holdings: 3,312,220
		• Farris Wilks: 826,376
		<ul> <li>Wilks Parties: 4,138,596</li> </ul>

- All shares of USWS Common Stock reflect the 1-for-3.5 reverse stock split effected by USWS on September 30, 2021 and the 1-for-6 reverse stock split effected by USWS on August 4, 2022.
- (2) Assumes a liquidation preference as of October 31, 2022.
- (3) THRC Holdings will receive \$1,227,907 in consideration for its Term C Loan Warrants in the Warrant Sale.

# USWS February Term C Loan Warrants

As described under the section titled "The Merger Agreement — Related Agreements," pursuant to the Warrant Purchase Agreement, immediately prior to the Effective Time and conditioned upon the consummation of the Merger, ProFrac will purchase from THRC Holdings 6,976,744 USWS February Term C Loan Warrants for an aggregate purchase price of \$1,227,907.

# USWS Series A Warrants

As described under the section titled "The Merger Agreement — Treatment of USWS Warrants — Rollover Warrants," at the Effective Time, the USWS Series A Warrants, including USWS Series A Warrants held by THRC Holdings, will be canceled and converted into the right to receive a warrant to purchase a number of shares of ProFrac Class A Common Stock equal to (i) the number of shares of USWS Common Stock underlying the USWS Series A Warrant multiplied by (ii) the Exchange Ratio. The exercise price of the USWS Series A Warrant will be the exercise price of the USWS Series A Warrant divided by the Exchange Ratio.

After giving effect to the Merger, THRC Holdings and Farris Wilks will own 30,805,623 shares of ProFrac Class A Common Stock and 97,447,865 shares of ProFrac Class B Common Stock, or approximately 82.7% of ProFrac's total outstanding shares. Please see "Security Ownership of Certain Owners and Management."

#### Related Agreements in Connection with Merger Agreement

Concurrently with the execution and delivery of the Merger Agreement, (i) THRC Holdings and certain other stockholders of USWS entered into a Voting Agreement with ProFrac and (ii) THRC Holdings and certain other holders of the USWS Term C Loan Warrants entered into a Warrant Purchase Agreement. For more information about the Voting Agreement and the Warrant Purchase Agreement, please see "The Merger Agreement — Related Agreements" on page 148.

On June 21, 2022, THRC Holdings and Farris Wilks entered into Convertible Note Amendments with USWS. For more information about the Convertible Note Amendments, please see "The Merger Agreement — Related Agreements" on page 148.

Concurrently with the execution of the Merger Agreement, each of the lenders (the "Term C Lenders"), including THRC Holdings, that made Term C Loans under the Consent and Sixth Amendment to USWS' Senior Secured Term Loan Credit Agreement (the "Credit Agreement"), entered into a Letter Agreement, dated June 21, 2022 (the "PIK Interest Letter Agreement"), pursuant to which, among other things, each Term C Lender, including THRC Holdings, covenanted and agreed that, with respect to interest accruing and becoming due and payable on its applicable Term C Loans under the Credit Agreement on and after June 30, 2022 ("Subject Interest"), such Term C Lender (i) would not (A) pursue any claim or (B) seek, demand or request any payment, return, exchange or other recovery, in each case, solely on account of any Subject Interest and (ii) would waive, reject and/or return to the loan parties, as applicable, (A) any offered or distributed payment solely on account of any Subject Interest or (B) the proceeds of any return, exchange or other recovery described in the foregoing clause (i). The terms of the PIK Interest Letter Agreement further provide that, (x) notwithstanding such covenants and agreements, Subject Interest shall continue to accrue in accordance with the Credit Agreement unless and until the Credit Agreement is waived, amended or otherwise modified in accordance with its terms and (y) the covenants and agreements of each Term C Lender described in clauses (i) and (ii) of the immediately preceding sentence shall terminate automatically, without notice or any other action, and shall have no further force or effect, in the event that the Merger Agreement is terminated in accordance with its terms.

## Commercial Arrangements with USWS

Stockholders should also note that ProFrac Manufacturing, LLC, an affiliate of ProFrac, previously entered into a Patent License Agreement with U.S. Well Services, LLC, to purchase three licenses to manufacture electric frac fleets (one license per fleet) using USWS Clean Fleet® technology. The purchase price was \$7.5 million per license for a total of \$22.5 million. Under the terms of the Patent License Agreement, ProFrac Manufacturing, LLC has the option, but no obligation, to purchase seven additional licenses for \$7.5 million each, and ten additional licenses for \$9.0 million each thereafter.

Additionally, on July 18, 2022, U.S. Well Services, LLC, a subsidiary of USWS, entered into a \$12.5 million promissory note (the **Equify Note**") with Equify Financial, as lender, an affiliate of the Wilks Parties. The Equify Note provides for the debt financing of certain equipment to be used by U.S. Well Services, LLC in building an electric frac fleet. The Equify Note matures August 1, 2027. The Equify Note is payable by U.S. Well Services, LLC to Equify Financial in equal monthly installments of principal in the amount of \$208,334.00, together with all accrued and unpaid interest on the outstanding principal balance of the Equify Note, commencing on September 1, 2022, and continuing thereafter until the maturity date. The Equify Note bears interest at a rate per annum equal to the lesser of (i) the sum of (A) "prime rate" as published in the Wall Street Journal from time to time plus (B) 9.25%, and (ii) the maximum amount of interest allowed by applicable law.

## USWS Board and USWS Special Committee Reasons for the Merger

In evaluating the Merger, each of the USWS Board and the USWS Special Committee consulted with USWS' management, as well as with its financial and legal advisors. In reaching its determinations and

recommendations, the USWS Board and USWS Special Committee evaluated a variety of factors that weighted in favor of the Merger, including, without limitation, the material factors listed below:

- Strategic Considerations. The USWS Board and the USWS Special Committee considered a number of factors pertaining to the strategic rationale for the merger as supporting its decision to enter into the Merger Agreement, including the following:
  - Significant scale and value of the combined company in the United States, with a strong presence in the most active U.S. unconventional oil and gas basins, including the Permian, Appalachia, the Eagle Ford, Mid-Continent and Haynesville, benefitting from improved efficiencies through increased geographical density and operating leverage.
  - An improved ability to service customers given the vertically integrated business model of ProFrac, which offersin-house
    manufacturing and refurbishment, proppant production and chemical solutions.
  - The combined company will benefit from an enhanced financial position, with a strong balance sheet capable facilitating further innovation and the full prosecution of the electric hydraulic fracturing opportunity.
  - The expectation that the combined company would generate \$35 million of expected annualized cost synergies resulting from reduced
    corporate and field overhead, decreased repair and maintenance expenses from internalizing manufacturing and refurbishment
    capabilities, utilizing pro forma company's buying power to secure best-in-class pricing from suppliers and capitalizing on internal
    power end manufacturing to minimize capital expenditures.
  - The expectation that the merger will be accretive to Adjusted EBITDA of the combined company beginning in fiscal year 2023.
  - The alignment of complementary cultures and operating philosophies, including a shared commitment to safety and integrity, employee development, partnerships with blue-chip customers, operating efficiency and service quality, technical innovation and community engagement.
  - The positive anticipated reaction from employees and other key stakeholders to the merger.
- The USWS Special Committee. In addition to the foregoing, the USWS Special Committee also considered, among other things, the procedural safeguards and processes implemented to enable the USWS Special Committee to determine the fairness of the Merger Agreement and transactions contemplated thereby for all of USWS stockholders, including the following:
  - · The USWS Special Committee consists of three independent directors, each of whom is unaffiliated with Crestview Partners;
  - The USWS Special Committee was empowered with full, plenary and exclusive power of the USWS Board to review, evaluate and investigate the proposed transaction with ProFrac and negotiate the terms of the proposed transaction with ProFrac or any potential alternative transactions;
  - The USWS Special Committee retained, and was advised by, experienced and qualified outside legal and financial advisors;
  - The USWS Special Committee requested and received a fairness opinion from Piper Sandler;
  - The terms and conditions of the Merger Agreement were determined througharm's-length negotiations conducted at the direction of the USWS Special Committee and its representatives and advisors;
  - The compensation of the members of the USWS Special Committee was in no way contingent on their approval of any transaction; and

- The frequency and extent of the USWS Special Committee's deliberation and its access to USWS management and advisors in connection with the evaluation of the potential transaction with ProFrac.
- Strategic Alternatives. The USWS Board and the USWS Special Committee considered the strategic alternatives available to USWS.
  - These alternatives include remaining a stand-alone entity, and the determination of the USWS Board and the USWS Special
    Committee was that a combination with ProFrac was a superior option when considering that there are, in general, business, financial,
    market and execution risks associated with remaining independent and successfully implementing USWS' stand-alone business plan.
  - Based on the fact that USWS had conducted a thorough strategic alternative review process in 2022 and no formal bids were submitted, it was the opinion of Piper Sandler, the USWS Board and the USWS Special Committee that it was unlikely that any other strategic buyers or financial sponsors would be willing to submit a competing offer with superior terms.
- *Financial Considerations*. The USWS Board and the USWS Special Committee considered a number of financial factors pertaining to the Merger as supporting its decision to enter into the Merger Agreement, including the following:
  - The financial terms of the transaction which specifies the fixed exchange rate of 0.3366 of ProFrac Class A Common Stock for each outstanding share of USWS Common Stock. Based upon ProFrac's closing stock price of \$21.49 per share on June 21, 2022 (the last trading day before the Merger Agreement was announced), the transaction represented a purchase price of \$1.21 per share of USWS Common Stock or a 68% premium to the USWS Common Stock closing price on June 21, 2022 and representing premiums of 54%, 59%, 39% and 21% over the average closing share price during the 15-day, 1-month, 3-month and 6-month periods ended June 21, 2022, respectively.
  - The exchange ratio is fixed and will not fluctuate as a result of changes in the price of ProFrac Class A Common Stock or USWS
    Common Stock prior to the consummation of the Merger, which limits the impact of external factors on the Merger Consideration.
  - That USWS Stockholders will own approximately 8% of the combined company's outstanding shares immediately after the Effective
    Time of the Merger (assuming conversion of all of USWS' Convertible Notes and Preferred Stock), which will allow USWS
    Stockholders to participate in all of the benefits of the combined company, including future growth and expected synergies while
    retaining the flexibility of selling all or a portion of those shares for cash at any time.
  - The opportunity for USWS Stockholders to benefit from any increase in the trading of ProFrac Class A Common Stock between announcement of the Merger and the Closing of the Merger.
  - The USWS Board and the USWS Special Committee also considered the fact that ProFrac had increased its purchase price during negotiations and the board's belief that the consideration was a full and fair price to acquire USWS.
- Financial Advisor Opinion. The USWS Board and USWS Special Committee considered Piper Sandler's opinion (the 'Piper Sandler Opinion'), dated June 21, 2022, as to the fairness, from a financial point of view and as of the date of the Piper Sandler Opinion, of the consideration to be received by USWS Stockholders.
  - The USWS Board and the USWS Special Committee considered the financial presentation of Piper Sandler, and its written opinion to the effect that, as of June 21, 2022 and based upon and subject to the various assumptions, considerations, qualifications and limitations set forth in the opinion, the Merger consideration to be received by holders of shares of USWS Common Stock

pursuant to the Merger Agreement was fair from a financial point of view to such holders. The opinion of Piper Sandler is addressed to, and for the use and benefit of, the USWS Board in connection with and for purposes of its evaluation of the Merger and does not constitute a recommendation as to how any holder of USWS Common Stock should vote with respect to the Merger. The full text of the Piper Sandler Opinion, which sets forth the assumptions made, procedures followed, matters considered and limitations on the review undertaken in connection with such opinion is attached as <u>Annex D</u> to this proxy statement/information statement/prospectus and is incorporated herein by reference.

- Terms of the Merger Agreement. The USWS Board and the USWS Special Committee considered that the terms of the Merger Agreement are reasonable, including:
  - The fact that a Special Committee comprised of the independent and disinterested directors of the USWS Board (supported by its financial and legal advisors) negotiated the Merger Agreement and the transactions contemplated thereby on behalf of USWS.
  - The fact that the size of the termination fees are reasonable in light of the size (particularly on an enterprise value basis) and benefits of
    the Merger and the USWS Board and the USWS Special Committee concluded that a termination fee of this size for the transactions
    contemplated by the Merger Agreement should not unduly deter a third party from making, or inhibit the USWS Board and the USWS
    Special Committee in evaluating, negotiating, and, if appropriate, terminating the Merger Agreement to enter into a transaction that is,
    a superior proposal.
  - The USWS Board and the USWS Special Committee considered the fact that the consummation of the proposed Merger would require
    the affirmative vote of the holders of a majority of the outstanding shares of USWS' Common Stock and that although approximately
    44% of such outstanding shares were subject to Voting Agreements, that figure would be cut back to 35% in the event of a change in
    the recommendation of the USWS Special Committee or the USWS Board.
  - The USWS Board and the USWS Special Committee considered the regulatory approvals required to complete the Merger, including expiration or termination of any waiting period or extension thereof under the HSR Act, and the belief that the Merger would be approved by the requisite authorities, without the imposition of conditions sufficiently material to preclude or materially diminish the benefits expected from the Merger, and would otherwise be completed in accordance with the terms of the Merger Agreement.
  - That the Merger Agreement provides USWS sufficient operating flexibility to conduct its business generally in the ordinary course between the signing of the Merger Agreement and the completion of the Merger.

# · Negative Factors Relating to the Merger that were Considered.

- That the completion of the Merger will preclude stockholders from having the opportunity to receive 100% of the benefits of USWS' future earnings growth and the future appreciation of the value of its capital stock that could be anticipated if its strategic plan were successfully implemented on a stand-alone basis.
- The fact that not all conditions to the Closing of the Merger, including the required approvals of governmental authorities, are within USWS' control and the risk that the Merger may not be completed despite the parties' efforts or that the Closing may be unduly delayed and the effects on USWS as a stand-alone company because of such failure or delay.
- The potential negative effect of the pendency of the Merger on the business and relationships with employees, customers, providers, suppliers, regulators and the communities in which it operates, including the risk that certain key members of senior management might choose not to remain employed with USWS prior to the completion of the Merger, regardless of the completion of the Merger.

- The challenges inherent in the combination of two businesses of the size and scope of USWS and ProFrac and the possible diversion of
  management's attention for an extended period of time.
- The risk of not capturing all of the anticipated synergies between USWS and ProFrac and the risk that other anticipated benefits of the Merger might not be realized.
- Because the Exchange Ratio is fixed, in the event of a decrease in the price of ProFrac Class A Common Stock between the date of execution of the Merger Agreement and the Closing of the Merger, USWS Stockholders may receive less value for their shares upon the Closing of the Merger than calculated pursuant to the Exchange Ratio on the date of the execution of the Merger Agreement.
- That certain provisions of the Merger Agreement could have the effect of discouraging proposals for alternative transactions involving USWS, including a termination fee of \$8 million.
- The potential for litigation relating to the Merger and the associated costs, burden and inconvenience involved in defending those proceedings.
- · The transaction costs, including severance costs, to be incurred in connection with the Merger.
- The possible diversion of management and employee attention for an extended period of time during the pendency of the Merger.
- Interests of Certain Persons. In addition to considering the factors described above, the USWS Special Committee and the USWS Board also considered that some officers and directors of USWS, including affiliates of the directors, may have interests in the business combination that are in addition to, and that may be different from, the interests of USWS' other stockholders (see section entitled "Interests of USWS Executive Officers and Directors in the Merger"). The USWS Special Committee consisting of the independent and disinterested directors reviewed and considered these interests during the negotiation of the transactions and in evaluating and unanimously approving, as members of the USWS Special Committee and the USWS Board, the Merger Agreement and related transactions.

# Recommendations of the USWS Board and the USWS Special Committee

On June 20, 2022, the USWS Special Committee unanimously (i) determined that the Merger Agreement and the transactions contemplated thereby were fair to, and in the best interest of, USWS and the USWS Stockholders and (ii) approved and declared advisable the Merger Agreement and the transactions contemplated thereby. The USWS Special Committee further unanimously resolved, subject to the terms and conditions set forth in the Merger Agreement, to (i) recommend that the USWS Board (x) determine that the Merger Agreement and the transactions contemplated thereby were fair to, and in the best interests of USWS and the USWS Stockholders and (y) approve and declare advisable the Merger Agreement and the transactions contemplated thereby and (ii) recommend that the USWS Board (x) direct that the USWS Merger Proposal be submitted to the USWS Stockholders for their approval and (y) recommend that USWS Stockholders approve the USWS Merger Proposal.

The USWS Board, acting on the unanimous recommendation and approval of the USWS Special Committee, unanimously determined that the Merger Agreement and the transactions contemplated thereby, including the Merger, are advisable and fair to and in the best interests of USWS and the USWS Stockholders and unanimously recommends that the USWS Stockholders vote "FOR" the proposal to adopt the Merger Agreement and the transactions contemplated thereby, including the Merger, "FOR" the issuance of shares of USWS Common Stock to be issued by USWS upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, "FOR" the approval of the amendment to USWS' Certificate of Designations of Series A Preferred Stock, "FOR" the approval of the amendment to the A&R LTIP, and "FOR" the proposal to adjourn the USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of such adoption.

#### **USWS Prospective Financial Information**

USWS does not as a matter of course make public projections as to future performance, earnings or other results beyond the current fiscal year due to the unpredictability of the underlying assumptions and estimates. However, USWS provided to ProFrac, in connection with its due diligence review, non-public standalone financial forecasts for the period from 2022—2026 that had been prepared by USWS' management for internal planning purposes and that are subjective in many respects (the "base case projections"). The base case projections assumed that USWS was able to complete the buildout of all four of its new Nyx Clean Fleets, an increase in margins resulting from the additional fleets and upward pricing pressure and that USWS continued as a stand-alone business with constant selling, general and administrative expenses through the period. USWS management also prepared certain downside projections (the "downside projections" and, together with the base case projections, the "projections"), which were not provided to ProFrac. The downside projections assumed no licensing revenue, an increase of litigation costs and professional fees through certain portions of the relevant period, the laying down of an electric fleet at the end of 2022, and the reduction in pricing and hours with respect to certain of USWS' fleets. These projections do not give effect to the Merger. USWS has included below a summary of these financial forecasts, which it refers to as the projections. In compiling the projections, USWS management took into account historical performance, combined with estimates regarding revenues, gross profit, adjusted EBITDA, as defined below, capital expenditures and total free cash flows. USWS also provided the projections to Piper Sandler for its use in connection with the rendering of its fairness opinion to the USWS Board and performing its related financial analysis. USWS believes that the projections reflect the current operating plan for USWS on a going-forward standalone basis. The summary of the projections is not being included in this proxy statement/information statement/prospectus to influence a stockholder's decision whether to vote for or against the proposal to adopt the Merger Agreement, but is being included because the projections were provided to ProFrac and Piper Sandler.

The projections included in this proxy statement/information statement/prospectus have been prepared by, and are the responsibility of, USWS' management. The projections were not prepared with a view toward public disclosure or toward complying with GAAP, the published guidelines of the SEC regarding projections or the guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information. KPMG LLP, USWS' independent registered public accounting firm, has not examined, compiled or performed any procedures with respect to the projections and, accordingly, KPMG LLP does not express an opinion or any other form of assurance with respect thereto.

The projections, while presented with numerical specificity, necessarily were based on numerous variables and assumptions that are inherently uncertain and many of which are beyond the control of USWS' management. Since the projections cover multiple years, by their nature, they become subject to greater uncertainty with each successive year. The assumptions upon which the projections were based necessarily involve judgments with respect to, among other things, industry performance, general business, economic, regulatory, market and financial conditions, as well as matters specific to USWS' business, all of which are difficult or impossible to predict accurately and many of which are beyond USWS' control. The projections also reflect assumptions as to certain business decisions that are subject to change. In addition, the projections may be affected by USWS' ability to achieve strategic goals, objectives and targets over the applicable periods. The projections are susceptible to interpretations and periodic revision based on actual experience and business developments.

As a result, there can be no assurance that the projections will be realized, and actual results may be significantly higher or lower than projected. The inclusion of the projections in this proxy statement/information statement/prospectus should not be regarded as an indication that USWS or any of its affiliates, advisors, officers, directors or representatives considered, or now consider, the projections to be predictive of actual future results, and the projections should not be relied upon as such. Neither USWS nor any of its affiliates, advisors, officers, directors or representatives can give any assurance that actual results will not differ from the projections, and none of them undertakes any obligation to update or otherwise revise or reconcile the projections to reflect circumstances existing after the date the projections were generated or to reflect the occurrence of future events, even in the event that any or all of the assumptions underlying the projections are shown to be in error. USWS

does not intend to make publicly available any update or other revision to the projections, except as otherwise required by applicable securities laws. Neither USWS nor any of its affiliates, advisors, officers, directors or representatives has made or makes any representation to any stockholder of USWS or to any other person regarding the ultimate performance of USWS compared to the information contained in the projections or that the projections will be achieved. USWS has made no representation to ProFrac, Merger Sub or their affiliates, in the Merger Agreement or otherwise, concerning the projections.

In light of the foregoing factors and the uncertainties inherent in the projections, stockholders are cautioned not to place undue, if any, reliance on the projections.

The following table and other information below is a summary of the base case projections (dollars in millions, except as noted).

	Management Projections				
	Fiscal Year Ending December 31				31
	2022E 2023E 2024E 2025E				2026E
Revenue	306	471	537	555	565
Gross Profit <sup>(1)</sup>	80	191	215	217	220
Adjusted EBITDA <sup>(2)</sup>	56	166	193	195	196
Total Capital Expenditures <sup>(3)</sup>	(138)	(52)	(29)	(30)	(30)
Total Free Cash Flows <sup>4)</sup>	(82)	114	164	165	167

- (1) Excludes stock-based compensation.
- (2) Adjusted EBITDA is defined as non-GAAP net income (loss) plus depreciation and amortization, income tax expense (benefit), stock-based compensation expense, net interest expense, other non-cash charges and other management adjustments.
- (3) Total Capital Expenditures is defined as maintenance capital expenditures plus growth capital expenditures.
- (4) Total Free Cash Flows defined as Adjusted EBITDA minus maintenance and growth capital expenditures.

The following table and other information below is a summary of the downside projections (dollars in millions, except as noted).

	2022	2023	2024	2025	2026
Revenue	\$299	\$380	\$429	\$448	\$459
Gross Profit <sup>(1)</sup>	74	118	134	138	142
Adj. EBITDA <sup>(2)</sup>	49	91	112	115	118
Total Capex <sup>(3)</sup>	138	49	26	27	27
Total Free Cash Flows <sup>(4)</sup>	(88)	42	86	89	92

- (1) Excludes stock-based compensation.
- (2) Adjusted EBITDA is defined as non-GAAP net income (loss) plus depreciation and amortization, income tax expense (benefit), stock-based compensation expense, net interest expense, other non-cash charges and other management adjustments.
- (3) Total Capital Expenditures is defined as maintenance capital expenditures plus growth capital expenditures.
- (4) Total Free Cash Flows defined as Adjusted EBITDA minus maintenance and growth capital expenditures.

The projections should be evaluated, if at all, in conjunction with the historical financial statements and other information regarding USWS contained elsewhere in this proxy statement/information statement/prospectus and USWS' public filings with the SEC. See "Where You Can Find More Information" beginning on page 276 of this proxy statement/information statement/prospectus.

#### Opinion of the Financial Advisors to the USWS Special Committee

USWS retained Piper Sandler to act as financial advisor to the USWS Special Committee in connection with USWS consideration of a possible business combination. USWS selected Piper Sandler to act as a financial advisor because Piper Sandler is a nationally recognized investment banking firm with substantial experience in transactions similar to the Merger and is familiar with USWS, ProFrac and their respective businesses. In the ordinary course of its investment banking business, Piper Sandler is regularly engaged in the valuation of oilfield services companies and their securities in connection with mergers and acquisitions and other corporate transactions.

Piper Sandler acted as an independent financial advisor to the USWS Special Committee in connection with the proposed Merger and participated in certain of the negotiations leading to the execution of the Merger Agreement. On June 21, 2022, prior to the execution of the Merger Agreement, Piper Sandler delivered to the USWS Special Committee and the USWS Board its written opinion, to the effect that, as of such date, the Exchange Ratio was fair, from a financial point of view, to the holders of USWS Common Stock issued and outstanding immediately prior to the Effective Time, in their capacity as holders of USWS Common Stock. The implied Merger Consideration to be received by the holders of shares of USWS Common Stock in the Merger, as of the date of the opinion, was equal to \$1.21 per share of USWS Common Stock (calculated as the product of the implied value of 0.0561 shares of ProFrac Class A Common Stock based on the closing trading price of ProFrac Class A Common Stock as of June 21, 2022. The full text of Piper Sandler's opinion is attached as Annex D to this proxy statement/information statement/prospectus. The opinion outlines the procedures followed, assumptions made, matters considered and qualifications and limitations on the review undertaken by Piper Sandler in rendering its opinion. The description of the opinion set forth below is qualified in its entirety by reference to the full text of the opinion. Holders of USWS Common Stock are urged to read the entire opinion carefully in connection with their consideration of the proposed Merger.

Piper Sandler's opinion was directed to the USWS Special Committee in connection with its consideration of the Merger and the Merger Agreement and does not constitute a recommendation to any USWS Stockholder as to how any such USWS Stockholder should vote at any meeting of stockholders called to consider and vote upon the approval of the Merger or the Merger Agreement. Piper Sandler's opinion was directed only to the fairness, from a financial point of view, of the Exchange Ratio to the holders of USWS Common Stock issued and outstanding immediately prior to the Effective Time, in their capacity as holders of USWS Common Stock, and did not address the underlying business decision of USWS to engage in the Merger, the form or structure of the Merger or any other transactions contemplated in the Merger Agreement, the relative merits of the Merger as compared to any other alternative transactions or business strategies that might exist for USWS or the effect of any other transaction in which USWS might engage. Piper Sandler also did not express any opinion as to the fairness of the amount or nature of the compensation to be received in the Merger by any officer, director or employee of USWS, or class of such persons, if any, relative to the amount of compensation to be received in the Merger by any other stockholder. Piper Sandler's opinion was approved by Piper Sandler's fairness opinion committee. In connection with its opinion, Piper Sandler has, among other things:

- reviewed and analyzed the financial terms of a draft of the Merger Agreement dated as of June 21, 2022;
- · reviewed and analyzed certain financial and other data with respect to USWS and ProFrac which was publicly available;
- reviewed and analyzed certain information, including financial forecasts, relating to the business, earnings, cash flow, assets, liabilities and
  prospects of USWS and ProFrac that were publicly available, as well as those that were furnished to Piper Sandler by USWS;
- conducted discussions with members of senior management and representatives of USWS and ProFrac concerning the matters described in
  each of the two bullets directly above, as well as their respective business prospects before and after giving effect to the Merger;

- reviewed the current and historical reported prices and trading activity of shares of USWS Common Stock and ProFrac Class A Common Stock and similar information for certain other companies deemed by Piper Sandler to be comparable to USWS;
- compared the financial performance of USWS and ProFrac with that of certain other publicly-traded companies that Piper Sandler deemed relevant:
- reviewed the financial terms, to the extent publicly available, of certain business combination transactions that Piper Sandler deemed relevant;
- performed a relative discounted cash flow analysis with respect to USWS and ProFrac;
- · performed a contribution analysis; and
- conducted such other analyses, examinations and inquiries and considered such other financial, economic and market criteria as Piper Sandler deemed necessary in arriving at its opinion.

Piper Sandler also discussed with certain members of the senior management of USWS and its representatives the business, financial condition, results of operations and prospects of USWS and held similar discussions with certain members of the senior management of ProFrac and its representatives regarding the business, financial condition, results of operations and prospects of ProFrac.

In arriving at its opinion, Piper Sandler relied upon and assumed, without assuming liability or responsibility for independent verification, the accuracy and completeness of all information that was publicly available or was furnished, or otherwise made available, to Piper Sandler or discussed with or reviewed by Piper Sandler. Piper Sandler further relied upon the assurances of the management of USWS that the financial information provided was prepared on a reasonable basis in accordance with industry practice, and that the management of USWS was not aware of any information or facts that would make any information provided to Piper Sandler incomplete or misleading. For the purpose of its opinion, Piper Sandler assumed that with respect to the financial forecasts, estimates and other forward-looking information reviewed by Piper Sandler, that such information was reasonably prepared based on assumptions reflecting the best currently available estimates and judgments of the management of USWS as to the expected future results of operations and financial condition of USWS to which such financial forecasts, estimates and other forward-looking information relate. Piper Sandler expressed no opinion as to any such financial forecasts, estimates or forward-looking information or the assumptions on which they were based. Piper Sandler did not consider the United States federal income tax treatment of the Merger and express no opinion as to the impact of any such tax treatment. Piper Sandler relied, with USWS' consent, on advice of the outside counsel and the independent accountants to USWS and on the assumptions of USWS management as to all accounting, legal, tax and financial reporting matters with respect to USWS, ProFrac and the Merger Agreement.

In arriving at its opinion, Piper Sandler assumed that the executed Merger Agreement would be in all material respects identical to the last draft reviewed by it. Piper Sandler relied upon and assumed, without independent verification, that (i) the representations and warranties of all parties to the Merger Agreement and all other related documents and instruments that are referred to therein are true and correct, (ii) each party to such agreements will fully and timely perform all of the covenants and agreements required to be performed by such party, (iii) the Merger will be consummated pursuant to the terms of the Merger Agreement without amendments thereto and (iv) all conditions to the consummation of the Merger will be satisfied without waiver by any party of any conditions or obligations thereunder, in each case in all respects material to Piper Sandler's analysis. Additionally, Piper Sandler assumed that all the necessary regulatory approvals and consents required for the Merger would be obtained in a manner that would not affect USWS, ProFrac or the contemplated benefits of the Merger.

In arriving at its opinion, Piper Sandler did not perform any appraisals or valuations of any specific assets or liabilities (fixed, contingent or other) of USWS or ProFrac, and was not furnished or provided with any such

appraisals or valuations, nor did it evaluate the solvency of USWS or ProFrac under any state or federal law relating to bankruptcy, insolvency or similar matters. The analyses performed by Piper Sandler in connection with its opinion were going concern analyses. Piper Sandler expressed no opinion regarding the liquidation value of USWS, ProFrac or any other entity. Without limiting the generality of the foregoing, Piper Sandler undertook no independent analysis of any pending or threatened litigation, regulatory action, possible unasserted claims or other contingent liabilities, to which USWS, ProFrac or any of their affiliates is a party or may be subject, and at the direction of USWS and with its consent, Piper Sandler's opinion made no assumption concerning, and therefore did not consider, the possible assertion of claims, outcomes or damages arising out of any such matters. Piper Sandler also assumed that neither USWS nor ProFrac is party to any material pending transaction, including without limitation any financing, recapitalization, acquisition or merger, divestiture or spin-off, other than the Merger.

Piper Sandler's opinion was necessarily based upon the information available to it and facts and circumstances as they existed and were subject to evaluation as of the date of the opinion; events occurring after the date of the opinion could have materially affected the assumptions used in preparing Piper Sandler's opinion. Piper Sandler expressed no opinion as to the price at which shares of USWS Common Stock or ProFrac Class A Common Stock may trade following the announcement of the Merger or at any future time. Piper Sandler did not undertake to reaffirm or revise its opinion or otherwise comment upon any events occurring after the date of the opinion and does not have any obligation to update, revise or reaffirm its opinion.

In arriving at its opinion, Piper Sandler performed a variety of financial analyses. The summary below is not a complete description of all the analyses underlying Piper Sandler's opinion or the presentation made by Piper Sandler to the USWS Special Committee, but is a summary of the material analyses performed and presented by Piper Sandler. The summary includes information presented in tabular format. In order to fully understand the financial analyses, these tables must be read together with the accompanying text. The tables alone do not constitute a complete description of the financial analyses. The preparation of a fairness opinion is a complex process involving subjective judgments as to the most appropriate and relevant methods of financial analysis and the application of those methods to the particular circumstances. The process, therefore, is not necessarily susceptible to a partial analysis or summary description. Piper Sandler believes that its analyses must be considered as a whole and that selecting portions of the factors and analyses to be considered without considering all factors and analyses, or attempting to ascribe relative weights to some or all such factors and analyses, could create an incomplete view of the evaluation process underlying its opinion. In connection with certain of its analyses relating to USWS, Piper Sandler used financial forecasts, including base case and downside projections, prepared by USWS management referred to in this proxy statement/information statement/prospectus in the section titled "USWS Prospective Financial Information" beginning on page 113. Also, no company included in Piper Sandler's comparative analyses described below is identical to USWS or ProFrac and no transaction is identical to the Merger. Accordingly, an analysis of comparable companies or transactions involves complex considerations and judgments concerning differences in financial and operating characteristics of the companies and other factors that could affect the public trading values or transaction values, as the case may be, of USWS and ProFrac and the companies to which they were compared. In arriving at its opinion, Piper Sandler did not attribute any particular weight to any analysis or factor that it considered. Rather, Piper Sandler made qualitative judgments as to the significance and relevance of each analysis and factor. Piper Sandler did not form an opinion as to whether any individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinion; rather, Piper Sandler made its determination as to the fairness of the Exchange Ratio, from a financial point of view, to the holders of USWS Common Stock, in their capacity as holders of USWS Common Stock, on the basis of its experience and professional judgment after considering the results of all its analyses taken as a whole.

In performing its analyses, Piper Sandler also made numerous assumptions with respect to industry performance, business and economic conditions and various other matters, many of which cannot be predicted and are beyond the control of USWS, ProFrac, and Piper Sandler. The analyses performed by Piper Sandler are not necessarily indicative of actual values or future results, both of which may be significantly more or less favorable than suggested by such analyses. Piper Sandler prepared its analyses solely for purposes of rendering its opinion and provided such analyses to the USWS Special Committee at its June 21, 2022 meeting. Estimates

on the values of companies do not purport to be appraisals or necessarily reflect the prices at which companies or their securities may actually be sold. Such estimates are inherently subject to uncertainty and actual values may be materially different. Accordingly, Piper Sandler's analyses do not necessarily reflect the actual value of USWS Common Stock or ProFrac Class A Common Stock or the prices at which USWS or ProFrac shares may be sold at any time. The analyses of Piper Sandler and its opinion were among a number of factors taken into consideration by the USWS board in making its determination to approve the Merger and entry into the Merger Agreement and the analyses described below should not be viewed as determinative of the decision of the USWS Special Committee with respect to the fairness of the Exchange Ratio.

### Summary of Material Financial Analyses

Piper Sandler reviewed the financial terms of the proposed Merger and the Merger Agreement, pursuant to which, at the Effective Time, each share of USWS Common Stock issued and outstanding immediately prior to the Effective Time, except for certain shares as set forth in the Merger Agreement, will be converted into the right to receive a number of shares of ProFrac Class A Common Stock equal to the Exchange Ratio, with cash paid in lieu of the issuance of any fractional shares of ProFrac Class A Common Stock. Based on ProFrac's closing price of \$21.26 on June 17, 2022, and the Exchange Ratio as defined in the Merger Agreement as 0.0561x, Piper Sandler calculated an implied share price for USWS of approximately \$1.19 per share. Based on the foregoing, Piper Sandler calculated an aggregate implied enterprise value of USWS at approximately \$401 million.

On August 4, 2022, USWS effected a 1-for-6 reverse split of its Common Stock. After giving effect to USWS' reverse stock split, the Exchange Ratio as defined in the Merger Agreement has been adjusted from 0.0561x to 0.3366x, resulting in an implied share price for USWS of approximately \$7.16 per share. For the purpose of the financial analyses described below, the share price for the USWS Class A Common Stock and the Exchange Ratio of 0.0561x, as determined prior to the 1-for-6 reverse split of USWS Class A Common Stock are used.

### Select Comparable Company Analyses

Piper Sandler reviewed and compared certain financial information, ratios and public market multiples for USWS, ProFrac or both to corresponding information, ratios and public market multiples for the following publicly traded corporations that provide comparable onshore hydraulic fracturing services in North America:

- Liberty Energy Inc. ("Liberty")
- NexTier Oilfield Solutions Inc. ("NexTier")
- ProPetro Holding Corp. ("ProPetro")

Although none of the selected companies is directly comparable to USWS or ProFrac, the companies included were chosen because they are publicly traded companies with operations that, in Piper Sandler's experience and professional judgment for purposes of this analysis, may be considered similar to certain aspects of USWS or ProFrac's operations, financial profile, size, service profile, geographic exposure and end market exposure.

Piper Sandler also calculated and compared various financial multiples and ratios based on the USWS management base case and downside case scenarios information from publicly available historical data and consensus analyst estimates. Enterprise value ("EV") multiples were calculated for EBITDA, and hydraulic horsepower ("HHP"). The foregoing calculations resulted in range of implied exchange ratios, based on peer low and peer high, of 0.0927x to 0.2601x for 2021 EBITDA, a base case range of implied exchange ratios of a negative equity value to 0.1294x, a downside case range of implied, exchange ratios of negative equity values,

and a range of implied exchange ratios, based on HHP, of a negative equity value to 0.0536x, as compared to the current exchange ratio of 0.0561x. The multiples and ratios were calculated using the applicable closing market prices as of June 17, 2022. The results of these calculations are summarized as follows:

	Enterprise Value To:(1)				
	EBITDA			Total	
	2021	2022P(2)	2023P(2)	HHP <sup>(3)</sup>	
Pressure Pumping				_	
Median	22.3x	4.8x	3.6x	1,196	
Low	11.9x	4.4x	3.1x	772	
High	22.5x	5.7x	3.9x	1,222	
ProFrac (Consensus)	21.4x <sup>(4)</sup>	5.2x	3.5x	1,936	
ProFrac (Management)	$21.4x^{(4)}$	5.2x	3.5x	1,936	
USWS at Offer (Consensus)	10.0x	25.2x	4.4x	1,238	
USWS at Offer (Management Base)	10.0x	8.7x	3.0x	1,238	
USWS at Offer (Management Downside)	10.0x	9.8x	5.4x	1,238	

- (1) Source: Capital IQ as of June 17, 2022.
- (2) USWS EV includes an extra \$83 million and \$90 million in 2022 and 2023, respectively, to account for capex requirements in order to meet EBITDA projections.
- (3) Source: Company provided information and Rystad Energy Inc.'s Q2 2022 report as of Q1 2022.
- (4) ProFrac net debt adjusted to exclude total purchase price of FTSI, \$407.5 million. EBITDA reflects standalone 2021 EBITDA per S-1.

#### Select Comparable Transaction Analyses

Using publicly available information and based on the USWS management base case and downside case scenarios, Piper Sandler evaluated the following seven transactions in the North American pressure pumping industry:

- ProFrac/FTSI
- NexTier Oilfield Solutions, Inc./Alamo Pressure Pumping
- Liberty/OneStim (Schlumberger)
- ProPetro/Pioneer Natural Resources Pressure Pumping Assets
- Matlin & Partners Acq. Corp./USWS
- STEP Energy Services/Tucker Energy Services
- Keane Group/RockPile Energy

		Enterprise Value To:					
	Transaction Value Per HHP (\$, millions)	Last Recorded Historical EBITDA	First Full- Year Projected EBITDA				
Median:	653	4.3x	3.1x				
Minimum:	313	1.6x	2.4x				
Maximum:	1.429	7.4x	4.2x				

			Er	nterprise Value To	o:
	Transaction Value	Estimated	USWS Fleet (HHP)	2022P	2023P
	(\$, millions)	Fleet (HHP)	(\$, millions)	EBITDA(1)	EBITDA <sup>(2)</sup>
USWS at Offer (Management Base):	401	323,750	1,238	8.7x	3.0x
USWS at Offer (Management Downside):	401	323,750	1,238	9.8x	5.4x
USWS at Offer (Consensus):	401	323,750	1,238	25.2x	4.4x

Source: Public filings, press releases, investor presentations and Capital IQ as of June 17, 2022.

- (1) Source: USWS provided financial model. 2022P EV includes \$83 million of capex needed to meet projections.
- (2) Source: USWS provided financial model. 2023P EV includes \$90 million of capex needed to meet projections.

No transaction utilized as a comparison in the precedent transaction analysis is identical to the merger. In evaluating the Merger, Piper Sandler made numerous judgments and assumptions with regard to industry performance, general business, economic, market and financial conditions and other matters. Mathematical analysis, such as determining the high, median and low, is not in itself a meaningful method of using comparable transaction data. The foregoing calculations resulted in a base case range of implied exchange ratios, based on precedent low and precedent high, of a negative equity value to 0.1510x, a downside case range of implied exchange ratios of negative equity values and a range of implied exchange ratios, based on HHP, of a negative equity value to 0.0870x, compared to the current exchange ratio of 0.0561x.

# Discounted Cash Flow Analysis

Piper Sandler performed a discounted cash flow analysis of USWS and ProFrac to calculate the estimated net present value as of June 30, 2022 of the unlevered, after-tax free cash flows that USWS and ProFrac were projected to generate from July 1, 2022 through December 31, 2026, in each case, based on the USWS management base case and downside case projections for USWS and the USWS management projections for

ProFrac. Piper Sandler calculated a terminal value of each company based on an EBITDA exit multiple, using its professional judgment given the nature and business of each of USWS and ProFrac and the industry in which both operate, of 4.0x, to the projected standalone EBITDA of both companies in the terminal year. The terminal value was then discounted to present value using a discount rate of 22.5% for USWS and 12.5% for ProFrac based on an estimate of each company's weighted average cost of capital calculated using the capital asset pricing model, to derive a net present value for USWS of \$653 million (base case) and \$394 million (downside case), and \$4,592 million for ProFrac, respectively. Piper Sandler then ran a series of sensitivities on the discounted cash flow analysis based on a range of terminal value multiples for both companies of 3.0x to 5.0x, a range of discount rates of 20.0% to 25.0% for USWS and a range of discount rates of 10.0% to 15.0% for ProFrac. The foregoing calculations resulted in a range of implied EVs for USWS of \$539 million to \$785 million (base case) and \$327 million to \$473 million (downside case), and a range of EVs for ProFrac of \$3,649 million to \$5,710 million. The foregoing calculations resulted in a range of implied exchange ratios of 0.0913x to 0.1187x, based on USWS management base case projections depending on potential value of NOLs, as well as a range of implied exchange ratios of a negative equity value to 0.0369x, based on USWS management downside case projections depending on potential value of NOLs, compared to the current exchange ratio of 0.0561x.

### Contribution Analysis

Piper Sandler analyzed the relative contribution of USWS and ProFrac to the discounted cash flow (**DCF**"), adjusted EBITDA for 2021, projected 2022 and projected 2023, and total HHP of the pro forma combined company. Piper Sandler calculated and compared various contribution percentages and ratios based on the 2021 EBITDA for USWS and ProFrac, the USWS management base case and downside case scenarios, USWS management projections for ProFrac, and consensus analyst estimates. Piper Sandler calculated the relative implied exchange ratios under the various contribution percentages, which resulted in 0.2426x for 2021 EBITDA, a range of implied exchange ratios of a negative equity value to 0.0974x, based on USWS management base case projections, a range of implied exchange ratios of negative equity values, based on USWS management downside case projections, and a range of implied exchange ratios of a negative equity value to 0.0005x, based on consensus analyst estimates and an implied exchange ratio of 0.1519x based on total HHP, compared to the prescribed exchange ratio of 0.0561x.

## Other Information

Piper Sandler also included premiums paid analysis for public energy services deals from 2010 to 2022, excluding mergers of equals transactions. The resulting premiums resulted in medians of between 10% to 20% over a one-day period, 20% to 30% over a one-week period, and 20% to 30% over a one-month period. Piper Sandler also included broker research, showing price targets ranged from \$1.00 to \$1.30 per share of USWS Common Stock (unadjusted for the 1-for-6 reverse split of USWS Common Stock effected by USWS on August 4, 2022).

## Piper Sandler's Relationship

Piper Sandler is acting as the USWS financial advisor in connection with the Merger and will receive a fee from USWS for providing such services, a portion of which is contingent upon the Closing of the Merger. Piper Sandler's transaction fee will be approximately \$7.5 million. Piper Sandler also received a \$1.0 million fee from USWS upon rendering its opinion, \$1.0 million of which will be credited towards any advisory fee that may become due and payable to Piper Sandler upon Closing of the Merger. USWS has also agreed to indemnify Piper Sandler against certain claims and liabilities arising out of Piper Sandler's engagement and to reimburse Piper Sandler for certain of its out-of-pocket expenses incurred in connection with Piper Sandler's engagement.

In the past, Piper Sandler has provided financial advisory and financing services to USWS, ProFrac and certain of their affiliates and may continue to do so and have received, and may receive, fees for the rendering of

such services. Piper Sandler has provided a variety of financial advisory services to ProFrac and to parties with significant investments in ProFrac, including: (i) acting as co-manager in arranging a senior secured credit facility for ProFrac in 2018; (ii) advising an entity on the sale of a large undeveloped property to ProFrac in 2021; (iii) advising an entity on the sale of such entity's preferred equity held by a third party to ProFrac in 2022; (iv) selling a minority equity stake in an entity to ProFrac in 2022; (v) advising an entity regarding a long-term supply agreement with ProFrac in return for convertible preferred equity securities of ProFrac in 2022; (vi) serving as exclusive financial advisor on ProFrac's acquisition of an entity in 2022; (vii) acting as sole debt placement agent in arranging a senior secured term loan for ProFrac in 2022 to fund a portion of the ProFrac's acquisition described in the preceding clause; (viii) acting as sole debt placement agent in 2022 in arranging an upsize of the ProFrac's existing revolving credit facility; and (ix) serving as an active bookrunner on ProFrac's initial public offering in 2022. With respect to the services described in clauses (i), (vi), (vii), (viii) and (ix), Piper Sandler realized aggregate gross fees of approximately \$12.5 million. Piper Sandler currently owns 24,254 shares of USWS Common Stock (after giving effect to the 1-for-6 revise split of Common Stock effected by USWS on August 4, 2022), which Piper Sandler received in connection with its role as financial advisor to USWS in its sale to Matlin & Partners Acquisition Corporation in 2018. Certain members of Piper Sandler's deal team have a direct interest in a portion of this position. In addition, in the ordinary course of its business, Piper Sandler and its affiliates may actively trade securities of USWS and ProFrac for its own account or the account of its customers and, accordingly, may at any time hold a long or short position in such securities. Piper Sandler may also, in the fut

### Interests of USWS Executive Officers and Directors in the Merger

In considering the recommendation of the USWS Board that USWS Stockholders vote in favor of the USWS Merger Proposal, the proposal to approve the issuance of shares of USWS Common Stock in connection with the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, the proposal to approve an amendment to the Certificate of Designations and the proposal to amend the A&R LTIP, USWS Stockholders should be aware of and take into account the fact that certain USWS directors and executive officers have interests in the Merger that may be different from, or in addition to, the interests of USWS Stockholders generally. The USWS Special Committee and the USWS Board were aware of and carefully considered these interests, among other matters, in evaluating the terms and structure, and overseeing the negotiation, of the Merger, in approving the Merger Agreement and the transactions contemplated thereby, including the Merger, and in recommending that the USWS Stockholders approve the USWS Merger proposal and the USWS compensation proposal.

At the Effective Time of the Merger, Joel Broussard, Chairman of the Board of Directors of USWS and USWS' former President and Chief Executive Officer, will receive a cash payment under the terms of a separation agreement he entered into with USWS on April 30, 2022. In addition, certain stock and performance awards held by Kyle O'Neill, a director of USWS and the current President and Chief Executive Officer of USWS, and Joshua Shapiro, Chief Financial Officer of USWS, will vest at the Effective Time. In connection with the entry into the Merger Agreement, USWS entered into amendments to the performance awards held by certain individuals, including Messrs. Broussard, O'Neill and Shapiro, which amendments provide that immediately prior to the Effective Time of the Merger, (A) the Pool A Performance Awards held by such individuals will be cancelled and converted into the right to receive the Merger Consideration in an amount equal to the accumulated award value as of July 19, 2022 divided by \$7.32, and (B) the Pool B Performance Awards held by such individuals will be cancelled and converted into the right to receive the Merger Consideration in an amount equal to the accumulated award value as of July 19, 2022 divided by \$6.468.

In addition, David Matlin, a member of the USWS Board, is a lender to USWS and holds certain USWS March Term C Loan Warrants. In connection with the entry into the Merger Agreement, the holders of the USWS Term C Loan Warrants, including Mr. Matlin, entered into an agreement with ProFrac pursuant to which ProFrac agreed to purchase the USWS Term C Loan Warrants from each holder for an amount of cash equal to \$0.176 multiplied by the number of USWS Term C Loan Warrants held by each holder.

Furthermore, Adam Klein, a member of the USWS Board, is a partner at, and a member of the investment committee of, Crestview Partners, which is a lender to USWS, holds certain securities in USWS and has the right to designate two directors to the USWS Board pursuant to the terms and subject to the conditions of that certain Subscription Agreement, dated as of July 13, 2018, by and among Crestview Partners, Matlin & Partners Acquisition Corporation and the other parties thereto. As of the date of the execution of the Merger Agreement, Adam Klein and Eddie Watson were the two Crestview Partners designees serving on the USWS Board.

As reported by Crestview Partners in its amended statement of beneficial ownership on Schedule 13D filed with the SEC on June 23, 2022, after giving effect to adjustments to reflect the 1-for-6 reverse stock split implemented by USWS on August 4, 2022, Crestview Partners beneficially owned 6,503,611 aggregate shares of USWS Common Stock at such time (the "Crestview USWS Shares"), consisting of 3,299,402 directly held shares of USWS Common Stock (the "Crestview USWS Common Shares") and the following convertible securities in USWS (collectively, the "Crestview USWS Common Stock then issuable upon exercise of 7,250,005 directly held USWS SPAC Warrants, (ii) 126,985 shares of USWS Common Stock then issuable upon exercise of 7,250,005 directly held USWS SPAC Warrants, (ii) 126,985 shares of USWS Common Stock then issuable upon exercise of 6,976,744 directly held USWS Term Loan C Warrants (the "Crestview Term Loan C Warrants"), (iv) 46,420 shares of USWS Common Stock then issuable upon conversion of 4,412 directly held shares of USWS Series A Preferred Stock and (v) 1,695,393 shares of USWS Common Stock then issuable upon conversion of \$45,186,976 in aggregate principal amount of directly held USWS Equity Linked Convertible Notes. The Crestview USWS Securities reported as beneficially owned by Crestview Partners represent approximately [\_]% of the total outstanding shares of USWS Common Stock as of the USWS record date.

Assuming the conversion and treatment of the Crestview USWS Securities (other than the Crestview Term Loan C Warrants) in the manner contemplated by the Merger Agreement, in exchange therefor, Crestview Partners would be entitled under the terms of the Merger Agreement to receive the same consideration for each class of Crestview USWS Securities as all holders of the same class of USWS securities, as more fully described in "The Merger Agreement" under the subheadings "Merger Consideration," "Treatment of USWS Series A Preferred Stock," "Treatment of USWS Equity Linked Convertible Notes" and "Treatment of USWS Warrants." In addition, upon consummation of the Closing, Crestview Partners will sell all of the Crestview Term Loan C Warrants to ProFrac pursuant to the Warrant Purchase Agreement and on the terms set forth therein. See "The Merger Agreement — Related Agreements — Warrant Purchase Agreement."

In the event that, in connection with the Closing, the USWS borrower elects under the Senior Secured Term Loan Agreement to prepay or cause to be prepaid, in whole or ratably in part, the outstanding aggregate principal amounts of the Term C Loans and accrued interest thereon, including the Term C Loan held by Crestview Partners (the "Crestview Term C Loan") then, pursuant to the Term C Loan Side Letter, Crestview Partners would also be entitled to receive payment in cash by the USWS Borrower of a premium equal to the amount of the aggregate principal of the Crestview Term C Loan so prepaid.

The following table sets forth the payments and other benefits that each of the members of the board of directors and executive officers of USWS will receive in connection with the consummation of the Merger:

Name	Title	I	Cash Payments		Stock Awards		formance Awards		Term C Lender Varrants	To	tal (\$)
Joel Broussard	Chairman of the Board	\$	633,340(1)	\$2	,762,474	\$3	,131,254	\$	_	\$6,5	27,068
Richard Burnett	Director	\$	_	\$	_	\$	_	\$	_	\$	_
Ryan Carroll	Director	\$	_	\$	_	\$	_	\$	_	\$	_
Steve Habachy	Director	\$	_	\$	_	\$	_	\$	_	\$	_
Adam Klein	Director	\$	_	\$	_	\$	_	\$1	,227,907	\$1,2	27,907
David Matlin	Director	\$	_	\$	_	\$	_	\$	122,791	\$ 1	22,791
Kyle O'Neill	Director, President and Chief Executive Officer	\$2	2,104,274(2)	\$1	,277,570	\$ 2	,351,572	\$	_	\$5,7	33,417
David Treadwell	Director	\$	_	\$	_	\$	_	\$	_	\$	_
Eddie Watson	Director	\$	_	\$	_	\$	_	\$	_	\$	_
Josh Shapiro	Chief Financial Officer	\$1	1,381,644 <sup>(2)</sup>	\$	753,322	\$ 1	,274,002	\$	_	\$3,4	108,967

- (1) Represents remaining separation payments.
- (2) Represents cash payments due per employment agreements related to reduction in responsibilities.

Additionally, Ryan Carroll, a member of the USWS Board, is currently a Managing Director and Head of Portfolio Management within the Private Credit Group at TCW Asset Management Company, LLC ("TCW"). TCW is a lender under ProFrac's term loan credit agreement, for which Piper Sandler is the administrative agent and collateral agent. THRC Holdings has also made an investment in a rescue fund at TCW for which Mr. Carroll is the portfolio manager.

## Indemnification, Exculpation and Insurance

In connection with the Merger, ProFrac has agreed that all rights to indemnification, advancement and exculpation existing in favor of the current or former directors and officers of USWS and its subsidiaries in the USWS certificate of incorporation or the USWS bylaws or the corresponding governing documents of its subsidiaries (as in effect on the date of the Merger Agreement) for acts or omissions occurring prior to the Effective Time are assumed and performed by the surviving company, with any payments thereof guaranteed by ProFrac, and continue in full force and effect for a period of six years after the Effective Time, except as otherwise required by applicable law.

The Merger Agreement requires ProFrac and the surviving company to maintain in effect USWS's current directors' and officers' liability insurance covering each person currently covered by such liability insurance policy for acts or omissions occurring prior to the Effective Time for a period of six years after the Effective Time, except that ProFrac may (i) substitute therefor policies of an insurance company the material terms of which, including coverage and amount, are no less favorable in any material respect to USWS' existing policies or (ii) request that USWS obtain such extended reporting coverage under its existing insurance programs. In no event shall the aggregate cost of such policy exceed three times the aggregate annual premium paid by USWS for fiscal year 2021 for such purpose.

#### Board of Directors and Management of USWS Following Completion of the Merger

Following completion of the Merger, USWS will be an indirect subsidiary of ProFrac. The officers and directors of Merger Sub immediately prior to the Effective Time shall be the officers and directors of USWS following the consummation of the Merger. The business and affairs of USWS will be managed by or under the direction of its board of directors. It is expected that the sole director and executive officers of USWS following completion of the Merger will be as follows:

Name	Age	Title
Ladd Wilks	36	Chief Executive Officer
Lance Turner	42	Chief Financial Officer
Coy Randle	60	Chief Operations Officer
Robert Willette	46	Chief Legal Officer
Matthew D. Wilks	39	Director

For additional information regarding the expected sole director and executive officers of USWS following completion of the Merger, please see "ProFrac's Management and Board of Directors."

## Board of Directors and Management of ProFrac Following the Completion of the Merger

Following completion of the Merger, the current directors and executive officers of ProFrac are expected to continue in their current positions, other than for changes that may be publicly announced by ProFrac in the future in the normal course. For additional information regarding the directors and executive officers of ProFrac, please see "Information about ProFrac—ProFrac's Management and Board of Directors."

## **Accounting Treatment**

ProFrac prepares its financial statements in accordance with GAAP. The Merger will be accounted for using the purchase method of accounting. Based on the facts that ProFrac board members and senior management will control and represent a majority of the board of directors and senior management of the combined company, as well as based on the terms of the exchange, pursuant to which USWS Stockholders will receive a premium of approximately 68% (as of the date preceding the Merger announcement) over the fair market value of their shares on such date, ProFrac is considered to be the acquirer of USWS for accounting purposes. ProFrac will allocate the purchase price to the fair value of USWS' assets and liabilities at the Effective Time, with the excess purchase price being recorded as goodwill.

## Federal Securities Laws Consequences; Stock Transfer Restriction Agreements

All shares of ProFrac Class A Common Stock that USWS Stockholders receive in the Merger will be freely transferable, except for shares of ProFrac Class A Common Stock received by persons who are deemed to be "affiliates" of USWS under the 1933 Act, at the time of the USWS Special Meeting. These "affiliates" may resell their shares of ProFrac Class A Common Stock only in transactions permitted by Rule 145 under the 1933 Act or as otherwise allowed under the 1933 Act. Persons who may be deemed to be affiliates of USWS for these purposes generally include individuals or entities that control, are controlled by, or are under common control with USWS and may include some officers, directors and principal stockholders of USWS.

Under Rule 145 of the 1933 Act, former USWS Stockholders who are affiliates of USWS at the time of the USWS Special Meeting and who are not affiliates of ProFrac after completion of the Merger, may sell their ProFrac Class A Common Stock at any time subject to the volume and manner of sale limitations and the current public information requirements of Rule 144 under the 1933 Act. Further, so long as such former USWS affiliates are not considered affiliates of ProFrac following completion of the Merger, and a period of at least one year has elapsed from completion of the Merger, such former affiliates may sell their ProFrac Class A Common

Stock without regard to the volume and manner of sale limitations of Rule 144 under the 1933 Act so long as there is adequate current public information available about ProFrac in accordance with Rule 144 under the 1933 Act. After a period of two years has elapsed from the completion of the Merger, and so long as such former affiliates are not, and have not been, for at least three months prior to such sale, affiliates of ProFrac, such former affiliates may freely sell their ProFrac Class A Common Stock. Former USWS Stockholders who become affiliates of ProFrac after completion of the Merger will be subject to the volume and manner of sale limitations and the current public information requirements of Rule 144 under the 1933 Act until each such stockholder is no longer an affiliate of ProFrac.

This proxy statement/information statement/prospectus does not cover any resales of the shares of ProFrac Class A Common Stock to be received by USWS Stockholders upon completion of the Merger, and no person is authorized to make any use of this proxy statement/information statement/prospectus in connection with a resale.

## **Appraisal Rights**

USWS Stockholders. USWS Stockholders are not entitled to appraisal rights in connection with the Merger Agreement and the transactions contemplated thereby.

**ProFrac Stockholders.** ProFrac Stockholders are not entitled to appraisal rights in connection with the Merger Agreement and the transactions contemplated thereby.

## **Delisting and Deregistration of USWS Common Stock**

If the Merger is consummated, USWS Common Stock will be delisted from Nasdaq and deregistered under the Exchange Act, and USWS will no longer file periodic reports with the SEC.

## MATERIAL UNITED STATES FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER

This section describes certain material U.S. federal income tax considerations of the Merger for beneficial owners of USWS Common Stock that exchange their USWS Common Stock for Merger Consideration. This discussion applies only to USWS Common Stock held as a capital asset for U.S. federal income tax purposes (generally, property held for investment) and does not address all aspects of U.S. federal income taxation that might be relevant to holders in light of their particular circumstances or status, including the Medicare contribution tax on net investment income, or holders who are subject to special rules, including:

- · brokers or dealers;
- traders in securities that elect to use amark-to-market method of accounting for their securities holdings;
- tax-exempt organizations, qualified retirement plans, individual retirement accounts or othertax-deferred accounts;
- · banks or other financial institutions, underwriters, insurance companies, real estate investment trusts or regulated investment companies;
- U.S. expatriates or former long-term residents of the U.S.;
- persons holding USWS Common Stock as part of a straddle, hedging or conversion transaction, constructive sale, or other arrangement involving more than one position;
- persons who own or have owned (directly, indirectly, or constructively) 5% or more of the shares of USWS Common Stock or ProFrac Class A Common Stock (by vote or value) or otherwise exercises control over USWS or ProFrac's corporate affairs;
- Holders of USWS Warrants;
- U.S. holders (as defined below) whose functional currency is not the U.S. dollar;
- "specified foreign corporations" (including "controlled foreign corporations"), "passive foreign investment companies" and corporations that accumulate earnings to avoid U.S. federal income tax;
- governments or agencies or instrumentalities thereof;
- · persons who received USWS Common Stock as compensation for services; and
- partnerships or other pass-through entities for U.S. federal income tax purposes;

If a partnership (including an entity or arrangement treated as a partnership for U.S. federal income tax purposes) holds USWS Common Stock, the U.S. federal income tax treatment of a partner in such partnership generally will depend on the status of the partner and the activities of the partnership. Partners should consult their tax advisors regarding the U.S. federal income tax treatment of the Merger and of holding the USWS Common Stock.

This discussion is based on the Code, its legislative history, existing and proposed Treasury Regulations, published guidance by the IRS and court decisions, all as of the date hereof. These laws are subject to change, possibly on a retroactive basis. This discussion is necessarily general and does not address all aspects of U.S. federal income taxation, including the effect of the U.S. federal alternative minimum tax, or U.S. federal estate and gift tax, or any state, local, or non-U.S. tax laws to a holder of USWS Common Stock. USWS and ProFrac have not sought and do not intend to seek any rulings from the IRS regarding the Merger. There is no assurance that the IRS will not take positions concerning the tax consequences of the Merger that are different from those discussed below, or that any such different positions would not be sustained by a court.

ALL HOLDERS OF USWS COMMON STOCK SHOULD CONSULT WITH THEIR TAX ADVISORS REGARDING THE TAX CONSEQUENCES OF THE MERGER TO THEM IN LIGHT OF THEIR PARTICULAR SITUATIONS, INCLUDING THE APPLICABLITY AND EFFECTS OF ANY U.S. FEDERAL, STATE, AND LOCAL AND NON-U.S. TAX LAWS.

#### Tax Consequences to U.S. Holders

This section applies to you if you are a "U.S. holder." For purposes of this discussion, a U.S. holder is a beneficial owner of USWS Common Stock that is, for U.S. federal income tax purposes:

- an individual who is a citizen or resident of the United States;
- a corporation (or other entity taxable as a corporation for U.S. federal income tax purposes) created or organized in or under the laws of the United States, any state thereof or the District of Columbia;
- an estate the income of which is subject to U.S. federal income taxation purposes regardless of its source; or
- an entity treated as a trust for U.S. federal income tax purposes if (i) a court within the United States is able to exercise primary supervision over the administration of such trust, and one or more such U.S. persons have the authority to control all substantial decisions of such trust or (ii) it has a valid election in effect under applicable Treasury Regulations to be treated as a U.S. person.

The Merger is expected to be treated as a taxable transaction for U.S. federal income tax purposes. A USWS Stockholder that is a U.S. holder generally will recognize gain or loss, if any, in an amount equal to the difference between (i) the sum of the fair market value as of the effective time of the Merger of any ProFrac Class A Common Stock received by such U.S. holder in the Merger and any cash received by such U.S. holder in the Merger in lieu of fractional shares of ProFrac Class A Common Stock, and (ii) such U.S. holder's adjusted tax basis in the USWS Common Stock exchanged therefor. Such gain or loss will generally be capital gain or loss, and will be long-term capital gain or loss only if the U.S. holder's holding period for the USWS Common Stock exceeds one year at the time of the Merger. Long-term capital gains of certain noncorporate U.S. holders, including individuals, may be subject to U.S. federal income taxation at preferential rates. The deductibility of capital losses is subject to limitations. If a U.S. holder acquired different blocks of USWS Common Stock at different times or different prices, the U.S. holder must determine its tax basis and holding period separately for each block of USWS Common Stock.

A U.S. holder's aggregate tax basis in any ProFrac Class A Common Stock received in the Merger generally will equal the fair market value of such stock as of the effective time of the Merger. A U.S. holder's holding period in any ProFrac Class A Common Stock received in the Merger generally will begin the day after the Merger.

Notwithstanding the above, if one or more persons control USWS and ProFrac prior to the merger, then Section 304 of the Code may apply to the transaction. For this purpose, "control" generally means ownership of more than 50% of the outstanding stock by vote or value, certain constructive ownership and aggregation rules apply, and ProFrac Class A Common Stock received in the merger is taken into account. If Section 304 applies to the Merger, then if a holder of USWS Common Stock also actually or constructively owns ProFrac stock (other than ProFrac Class A Common Stock received pursuant to the merger), instead of recognizing gain or loss as described above, such holder may recognize dividend income depending on the application of the tests set forth in Section 302 of the Code. Because the possibility of dividend treatment depends upon each holder's particular circumstances, including the application of constructive ownership rules, holders of USWS Common Stock that also actually or constructively own ProFrac stock should consult their tax advisors regarding the application of the foregoing rules to their particular circumstances.

## Tax Consequences to Non-U.S. Holders

This section applies to you if you are a "Non-U.S. holder." A Non-U.S. holder is a beneficial owner (other than a partnership or an entity or arrangement characterized as a partnership for U.S. federal income tax purposes) of USWS Common Stock that is not a U.S. holder.

A Non-U.S. holder generally will not be subject to U.S. federal income taxes on any gain recognized pursuant to the Merger unless:

- such gain is effectively connected with the Non-U.S. holder's conduct of a trade or business within the United States (or if a tax treaty applies, is attributable to a U.S. permanent establishment or fixed base maintained by the Non-U.S. holder in the United States); or
- the Non-U.S. holder is an individual who is present in the United States for 183 days or more during the taxable year that includes the date of the
  merger, and certain other conditions are satisfied.

Gain described in the first bullet point above will generally be subject to regular U.S. income tax under the rules described above as though such Non-U.S holder were a U.S. holder of USWS Common Stock, unless an applicable income tax treaty provides otherwise. ANon-U.S. holder that is a corporation may also be subject to an additional "branch profits tax" at a rate of 30% (or a lower treaty rate).

If the Non-U.S. holder is described in the second bullet, then such Non-U.S. holder will generally be subject to U.S. federal income tax at a 30% rate (or such lower rate as may be specified by an applicable income tax treaty) on any gain, which may be offset by certain U.S. source capital losses of the Non-U.S. holder.

As discussed above under "—*Tax Consequences to U.S. Holders*," in certain circumstances, Merger Consideration received by certain holders of USWS Common Stock may be treated as a dividend, depending on the application of the tests set forth in Section 302 of the Code. Any amount so treated generally would be subject to U.S. withholding tax at a rate of 30% (or such lower rate as may be specified by an applicable income tax treaty) unless such dividend is effectively connected with the Non-U.S. holder's conduct of a trade or business within the United States (and, if required by an applicable income tax treaty, is attributable to a permanent establishment or fixed base of the Non-U.S. holder in the United States). Because the application of Section 304 depends on a holder's particular circumstances, withholding agents may not be able to determine whether any particular holder is treated as receiving a dividend for U.S. federal income tax purposes. Accordingly, an applicable withholding agent may withhold at a rate of 30% (or such lower rate as may be specified by an applicable income tax treaty) on the entire amount Merger Consideration payable to a Non-U.S. holder. If a withholding agent withholds excess amounts from the Merger Consideration so payable to a Non-U.S. holder, such Non-U.S. holder may seek a refund of any such excess amounts by timely filing an appropriate claim with the IRS. In addition, for Non-U.S. holders, the receipt of any amounts treated as a dividend will generally be subject to Sections 1471 through 1474 of the Code and the U.S. Treasury regulations and administrative guidance issued thereunder ("FATCA").

Non-U.S. holders are urged to consult their own tax advisors regarding the potential applicability of these rules as well as any income tax treaty that may be applicable in their particular circumstances.

## Information Reporting and Backup Withholding

Any shares of ProFrac Class A Common Stock and/or cash received by a U.S. holder or aNon-U.S. holder in the merger may be subject to information reporting and backup withholding. To avoid backup withholding, a U.S. holder that does not otherwise establish an exemption should timely complete and return an IRS Form W-9 (or other successor form), certifying under penalties of perjury that such U.S. holder is a "United States person" (within the meaning of the Code), that the taxpayer identification number provided is correct and that such U.S. holder is not subject to backup withholding. A Non-U.S. holder generally may establish an exemption from backup withholding by certifying itsnon-U.S. person status under penalties of perjury on a properly completed applicable IRS Form W-8 (or other successor form). Backup withholding is not an additional tax. The amount of any backup withholding from a payment to a holder will be allowed as a credit against such holder's U.S. federal income tax liability and a holder may obtain a refund of any excess amounts withheld under the backup withholding rules by timely filing the appropriate claim for a refund with the IRS and furnishing any required information.

THE PRECEDING DISCUSSION IS INTENDED ONLY AS A SUMMARY OF THE MATERIAL UNITED STATES FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER AND DOES NOT PURPORT TO BE A COMPLETE ANALYSIS OR DISCUSSION OF ALL POTENTIAL INCOME TAX EFFECTS RELEVANT THERETO OR A DISCUSSION OF ANY OTHER TYPE OF TAXES. ALL USWS STOCKHOLDERS ARE URGED TO CONSULT THEIR OWN TAX ADVISORS AS TO THE SPECIFIC TAX CONSEQUENCES TO THEM OF THE MERGER, INCLUDING TAX RETURN REPORTING REQUIREMENTS, THE APPLICABILITY AND EFFECT OF ANY NON-U.S., FEDERAL, STATE, LOCAL AND OTHER APPLICABLE TAX LAWS, AND THE EFFECT OF ANY PROPOSED CHANGES IN THE TAX LAWS.

### REGULATORY AND OTHER APPROVALS REQUIRED FOR THE MERGER

### **Antitrust Approvals**

Pursuant to the terms of the Merger Agreement, the Merger cannot be completed until the waiting period applicable to the consummation of the transactions under the HSR Act has expired or been terminated and all other specified required approvals have been obtained or any applicable waiting period thereunder has expired or been terminated. Under the HSR Act and the rules promulgated thereunder, if certain HSR transaction size thresholds are triggered, the Merger cannot be completed until the parties to the Merger Agreement have given notification and furnished information to the Federal Trade Commission (the "FTC") and the United States Department of Justice (the 'DOJ"), and until the applicable waiting period has expired or has been terminated. Each of ProFrac and USWS filed a Pre-Merger Notification and Report Form pursuant to the HSR Act with the DOJ and the FTC on July 6, 2022. On August 6, 2022, the applicable waiting period under the HSR Act expired without a request for additional information.

Additionally, at any time before or after consummation of the Merger, notwithstanding any prior expiration or termination of the applicable waiting period under the HSR Act, the FTC, the DOJ or any state could take such action under antitrust laws as it deems necessary or desirable in the public interest, including seeking to enjoin the completion of the Merger or seeking the divestiture of substantial assets of ProFrac or USWS or their respective subsidiaries. Private parties may also seek to take legal action under the antitrust laws under certain circumstances.

Under the terms of the Merger Agreement, each of ProFrac and USWS has agreed to use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under the applicable law to complete the Merger and the other transactions contemplated by the Merger Agreement, including (i) preparing and filing as promptly as reasonably practicable with any governmental authority or other third party all documentation to effect all necessary, proper or advisable filings, notices, petitions, statements, registrations, submissions of information, applications and other documents, and (ii) obtaining and maintaining all approvals, consents, registrations, permits, authorizations and other confirmations required to be obtained from any governmental authority or other third party that are necessary, proper or advisable to consummate the Merger and the other transactions contemplated by the Merger Agreement.

A condition to the respective obligations of ProFrac and USWS to complete the Merger is that any waiting period applicable to the merger under the HSR Act will have been expired or terminated. On August 6, 2022, the applicable waiting period under the HSR Act expired without a request for additional information. A further condition is that all required governmental regulatory consents and approvals will have been obtained except those that individually or in the aggregate fall below the materiality standard described in the Merger Agreement. See "The Merger Agreement – Conditions to Completion of the Merger" beginning on page 144.

There can be no assurance that the Merger will not be challenged on antitrust grounds or, if such a challenge is made, that the challenge will not be successful. In addition, state antitrust authorities and private parties in certain circumstances may bring legal action under the antitrust laws seeking to enjoin the Merger or seeking conditions to the completion of the Merger.

## **Securities and Exchange Commission**

In connection with the proposed Merger, ProFrac has filed a registration statement on FormS-4, of which this proxy statement/information statement/prospectus forms a part of, with the SEC under the Securities Act, that must be declared effective by the SEC and pursuant to which the issuance at the Effective Time of shares of ProFrac Class A Common Stock comprising the Merger Consideration, the USWS SPAC Warrants (which shall remain outstanding after Closing of the Merger) and the shares of ProFrac Class A Common Stock issuable upon exercise of the USWS SPAC Warrants will be registered with the SEC.

### NASDAQ

In addition, the consummation of the Merger is subject to approval for listing of the shares of ProFrac Class A Common Stock to be issued in the Merger and reserved for issuance in connection with the Merger on the Nasdaq Global Select Market, subject to official notice of issuance.

## Listing of shares of ProFrac Class A Common Stock; Delisting and Deregistration of USWS Common Stock and Warrants

Publicly traded shares of USWS Common Stock and publicly traded USWS Warrants are currently listed on the Nasdaq Capital Market under the symbol "USWS" and "USWSW," respectively. Upon consummation of the Merger, USWS Common Stock and Warrants will be delisted from the Nasdaq Capital Market and will be subsequently deregistered under the Exchange Act. It is anticipated that upon consummation of the Merger, the publicly traded warrants of USWS shall become publicly traded warrants of ProFrac and shall be listed on the Nasdaq Global Select Market under the symbol "PFHCW."

## Other Approvals

Other than the filings described above, neither ProFrac nor USWS is aware of any regulatory approvals required to be obtained, or waiting periods required to expire, to consummate the Merger. If ProFrac and USWS discover that other approvals or waiting periods are necessary, ProFrac and USWS will seek to obtain or comply with them in accordance with the Merger Agreement.

### THE COMPANIES

### ProFrac

ProFrac is a growth-oriented, vertically integrated and innovation-driven energy services company providing hydraulic fracturing, completion services and other complementary products and services to leading upstream oil and gas companies engaged in the E&P of North American unconventional oil and natural gas resources. Founded in 2016, ProFrac was built to be the go-to service provider for E&P companies' most demanding hydraulic fracturing needs. ProFrac is focused on employing new technologies to significantly reduce GHG emissions and increase efficiency in what has historically been an emissions-intensive component of the unconventional E&P development process.

ProFrac's principal executive offices are located at 333 Shops Boulevard, Suite 301, Willow Park, Texas 76087 and its telephone number is (254) 776-3722.

## Merger Sub

Merger Sub is an indirect subsidiary of ProFrac that was incorporated in Delaware in June 2022. Merger Sub does not engage in any operations and exists solely to facilitate the Merger.

### USWS

USWS is a technology-focused oilfield service company focused on electric pressure pumping services for oil and natural gas E&P companies in the United States. USWS' patented electric pressure pumping technology provides one of the first fully electric, mobile well stimulation systems powered by locally supplied natural gas including field gas sourced directly from the wellhead. USWS' electric pressure pumping technology decreases emissions, sound pollution and truck traffic while generating operating efficiencies including significant customer fuel cost savings versus conventional diesel-powered fleets. The USWS fleet consists mostly of all-electric, mobile pressure pumping equipment and other auxiliary heavy equipment that performs stimulation services. USWS' Clean Fleet® electric fleets replace the traditional engines, transmissions, and radiators used in conventional diesel fleets with electric motors powered by electricity.

#### THE MERGER AGREEMENT

The following is a summary of the material terms of the Merger Agreement. This summary does not purport to describe all the terms of the Merger Agreement and is qualified by reference to the complete Merger Agreement, a copy of which is attached as Annex A to this proxy statement/information statement/prospectus and incorporated by reference. The rights and obligations of the parties are governed by the express terms and conditions of the Merger Agreement and not by this summary or any other information contained in this proxy statement/information statement/prospectus. All stockholders of ProFrac and USWS are urged to read the Merger Agreement carefully and in its entirety.

### Explanatory Note Regarding Summary of Merger Agreement and Representations and Warranties in the Merger Agreement

The summary of the terms of the Merger Agreement is intended to provide information about the terms of the Merger. The terms and information in the Merger Agreement should not be relied on as disclosures about ProFrac or USWS without consideration of the entirety of public disclosure by ProFrac and USWS as set forth in their respective public reports with the SEC. The terms of the Merger Agreement (such as the representations and warranties) govern the contractual rights and relationships and allocate risks between the parties in relation to the Merger. In particular, the representations and warranties made by the parties to each other in the Merger Agreement have been negotiated between the parties with the principal purpose of setting forth their respective rights with respect to their obligation to close the Merger should events or circumstances change or be different from those stated in the representations and warranties. The representations and warranties may also be subject to a contractual standard of materiality different from those generally applicable to stockholders under federal securities laws. Matters may change from the state of affairs contemplated by the representations and warranties. ProFrac and USWS will provide additional disclosure in their public reports to the extent that they are aware of the existence of any material facts that are required to be disclosed under federal securities laws and will update such disclosure as required by federal securities laws.

### Structure of the Merger

The Merger Agreement provides for the merger of Merger Sub with and into USWS, with USWS surviving the Merger as the surviving corporation and an indirect subsidiary of ProFrac. The Merger will become effective when the parties to the Merger Agreement file a certificate of merger with the Secretary of State of the State of Delaware (or at such later time as ProFrac and USWS mutually agree and specify in the certificate of merger).

### Merger Consideration

At the Effective Time, each issued and outstanding share of USWS Common Stock will be converted automatically into the right to receive 0.3366 shares of ProFrac Class A Common Stock (as adjusted for the 1-for-6 reverse split of Common Stock effected by USWS on August 4, 2022). USWS Stockholders will receive cash in lieu of any fractional shares of ProFrac Class A Common Stock that would have otherwise been received in the Merger. See "—Fractional Shares" beginning on page 135. If between the date of the Merger Agreement and the Effective Time there is a reclassification, recapitalization, stock split or combination, exchange or readjustment of shares, or any stock dividend with respect to the capital stock of ProFrac or USWS, the Merger Consideration will be appropriately adjusted.

Immediately following the completion of the Merger, based on the aggregate amount of capital stock and shares of ProFrac Class A Common Stock outstanding as of August 8, 2022, it is anticipated that persons who were stockholders of ProFrac and USWS immediately prior to the Merger will own approximately 91.76% and 8.24%, respectively, of the then outstanding ProFrac capital stock and will own approximately 76.34% and 23.66%, respectively, of the then outstanding ProFrac Class A Common Stock.

### **Exchange of Shares**

Prior to the Effective Time, ProFrac will enter into a customary exchange agreement with a nationally recognized financial institution designated by ProFrac and reasonably acceptable to USWS to act as an exchange agent to handle the exchange of ProFrac Class A Common Stock in the Merger for Merger Consideration. ProFrac will instruct the exchange agent to mail, as soon as reasonably practicable after the Effective Time and in any event not later than the fifth business day following the Closing Date, to each holder of USWS Common Stock at the Effective Time a letter of transmittal for use in the exchange and instructions explaining how to surrender stock certificates or transfer uncertificated shares of USWS Common Stock to the exchange agent. Holders of USWS Common Stock that surrender their stock certificates or transfer their uncertificated shares to the exchange agent, together with a properly completed letter of transmittal, will be entitled to receive the appropriate Merger Consideration. You should not return stock certificates with the enclosed proxy card.

After the Effective Time, holders of unexchanged shares of USWS Common Stock will not be entitled to receive any dividends or other distributions payable by ProFrac after the Closing until their shares of USWS Common Stock are surrendered. However, once those shares are surrendered, ProFrac will pay or will cause the exchange agent to pay the holder, without interest, any dividends or other distributions with a record date after the Effective Time that were previously paid to ProFrac Stockholders or are payable at the time the shares of USWS Common Stock are surrendered.

### Fractional Shares

ProFrac will not issue fractional shares of USWS Common Stock in the Merger. All fractional shares of ProFrac Class A Common Stock that a holder of shares of USWS Common Stock would otherwise be entitled to receive as a result of the Merger will be aggregated. For any fractional share that results from such aggregation, the exchange agent will pay the holder an amount of cash, without interest, equal to the fraction of a share of ProFrac Class A Common Stock to which the USWS Stockholder would otherwise be entitled in the Merger multiplied by the ProFrac Trading Price.

#### Treatment of USWS Series A Preferred Stock

Pursuant to the Merger Agreement, USWS agreed to take all requisite action so that, effective as of immediately prior to the Effective Time, (i) each holder of USWS Series A Preferred Stock issued and outstanding at such time may convert such stock into shares of USWS Common Stock at the Merger Conversion Ratio (as defined below), and (ii) any shares of USWS Series A Preferred Stock issued and outstanding immediately prior to the Effective Time not so converted at the Merger Conversion Ratio will automatically convert into shares of USWS Common Stock at the then-effective conversion rate as calculated pursuant to USWS' Certificate of Designations.

The "Merger Conversion Ratio" means, for each share of USWS Series A Preferred Stock, the quotient of (i) its liquidation preference as of the date of the conversion and (ii) Merger Conversion Price. The "Merger Conversion Price" will be \$7.32, subject to adjustment.

## **Treatment of USWS Equity Linked Convertible Notes**

Pursuant to the Merger Agreement, USWS agreed to take all requisite action so that, effective as of immediately prior to the Effective Time, each USWS Equity Linked Convertible Note issued and outstanding at such time shall automatically convert into a number of shares of USWS Common Stock equal to the quotient obtained by dividing (i) the amount of outstanding aggregate principal amount, plus accrued and unpaid interest, owing under such USWS Equity Linked Convertible Note through July 9, 2022, by (ii) \$7.32.

#### Treatment of USWS Warrants

USWS Term C Loan Warrants

Subject to the consummation of the Warrant Sale, at the Effective Time, each USWS Term C Loan Warrants issued and outstanding immediately prior to the Effective Time (which shall be held by ProFrac following the consummation of the Warrant Sale) will be automatically canceled and will cease to exist and no consideration will be delivered in exchange therefor. For more information about the Warrant Sale, please see "— Related Agreements" on page 148.

#### Rollover Warrants

Prior to the Effective Time, ProFrac will execute an amendment to each of the warrant agreements that govern the USWS Rollover Warrants as necessary to assume each warrant as of the Effective Time in accordance with the terms thereof and the Merger Agreement.

Each USWS Rollover Warrant shall remain outstanding following the Merger through its prescribed term but shall represent the right to receive upon valid exercise thereof shares of ProFrac Class A Common Stock equal to the product of (A) the number of shares of USWS Common Stock subject to such USWS Rollover Warrant immediately prior to the Effective Time and (B) the Exchange Ratio. Additionally, each USWS Rollover Warrant will be amended such that the exercise price of such USWS Rollover Warrant will equal the current exercise price of such USWS Rollover Warrant divided by the Exchange Ratio, as described below (as adjusted for the 1-for-6 reverse split of Common Stock effected by USWS on August 4, 2022):

- The USWS SPAC Warrants' per-share exercise price will become \$717.47 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$241.50 per full share exercise price of such warrants by the 0.3366 Exchange Ratio;
- The USWS Series A Warrants' per-share exercise price will become \$477.89 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$160.86 per full share exercise price of such warrants by the 0.3366 Exchange Ratio;
- The USWS Private Placement Investor Warrants' per-share exercise price will become \$31.43 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$10.58 per full share exercise price of such warrants by the 0.3366 Exchange Ratio; and
- The USWS Private Placement Agent Warrants' per-share exercise price will become \$39.28 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$13.22 per full share exercise price of such warrants by the 0.3366 Exchange Ratio.

The amendments to each of the Rollover Warrants are filed as Exhibits 10.31, 10.32, 10.33, and 10.34 to this proxy statement/information statement/prospectus and incorporated herein by reference. The foregoing descriptions of the amendments to the USWS Rollover Warrants are qualified in their entirety by reference to such Exhibits.

## Treatment of USWS Equity Awards

USWS Restricted Shares

Pursuant to the Merger Agreement, USWS will take all requisite action so that, at the Effective Time, each USWS Restricted Share that is outstanding and unvested under the USWS LTIP immediately prior to the Effective Time will, by virtue of the Merger and without any action on the part of the holder thereof, be canceled in exchange for the right to receive the Merger Consideration and, in lieu of any fractional shares, cash (treating such unvested USWS Restricted Share in the same manner as all other outstanding shares of USWS Common Stock for such purposes), less applicable withholding taxes.

### USWS DSUs

Pursuant to the Merger Agreement, USWS will take all requisite action so that immediately prior to the Effective Time, each then-outstanding USWS DSU will, by virtue of the Merger and without any action on the part of the holder thereof, be cancelled and converted into the right to receive the Merger Consideration and, in lieu of any fractional shares, cash (treating such USWS DSU in the same manner as if it were an outstanding share of USWS Common Stock for such purposes).

USWS Pool A Performance Awards and USWS Pool B Performance Awards

Pursuant to the Merger Agreement, USWS will take all requisite action so that, immediately prior to the Effective Time, by virtue of the Merger and without any action on the part of the holder thereof:

- (i) each then-outstanding USWS Pool A Performance Award will be canceled and converted into the right to receive (A) for recipients of USWS Pool A Performance Awards who consent to the terms of the Award Amendment, the Merger Consideration in an amount equal to the accumulated award value as of July 19, 2022 divided by \$7.32, and (B) with respect to each USWS Pool A Performance Award not amended by an Award Amendment, the Merger Consideration in an amount equal to the accumulated award value as of the Effective Time divided by \$7.32; and
- (ii) each then-outstanding USWS Pool B Performance Award shall be canceled and converted into the right to receive (A) with respect
  to each USWS Pool B Performance Award amended by an Award Amendment, the Merger Consideration in an amount equal to the
  accumulated award value as of July 19, 2022 divided by \$6.468, and (B) with respect to each USWS Pool B Performance Award not
  amended by an Award Amendment, the Merger Consideration in an amount equal to the accumulated award value as of the Effective
  Time divided by \$6.468.

## Representations and Warranties

The Merger Agreement contains customary representations and warranties of ProFrac and USWS relating to their respective businesses. The representations and warranties in the Merger Agreement do not survive the Effective Time.

Each of ProFrac and USWS has made representations and warranties to the other regarding, among other things:

- · due organization, good standing and qualification;
- · corporate authority to enter into the Merger Agreement and complete the Merger;
- required consents and filings with government entities;
- absence of any breach of organizational documents, laws or agreements as a result of the Merger;
- · capitalization;
- · accuracy and sufficiency of documents filed with the SEC;
- conformity of the financial statements with applicable accounting principles and that the financial statements fairly present, in all material respects, the consolidated financial positions of ProFrac and USWS, as the case may be;
- information supplied for use in this proxy statement/information statement/prospectus;
- · absence of certain changes or events;
- absence of undisclosed liabilities;
- compliance with laws and court orders;

- · absence of pending or threatened legal proceedings; and
- · opinions of financial advisors.

Additional representations and warranties were made by USWS to ProFrac relating to:

- · real property matters;
- · intellectual property matters;
- · data privacy and security matters;
- taxes;
- · employees and employee benefit plans;
- · environmental matters;
- material contracts;
- finders' fees;
- · inapplicability of antitakeover statutes to the Merger; and
- insurance matters.

Additional representations and warranties were made by ProFrac to USWS relating to the solvency of ProFrac as of the Effective Time immediately after giving effect to the transactions contemplated by the Merger Agreement.

The representations and warranties set forth in the Merger Agreement as described above should not be relied upon by any person as statements of factual information. Please see "—Explanatory Note Regarding Summary of Merger Agreement and Representations and Warranties in the Merger Agreement" on page 134.

Many of the representations and warranties are qualified by materiality, Company Material Adverse Effect or Parent Material Adverse Effect (as such terms are defined in the Merger Agreement).

### Covenants

USWS agreed to certain covenants in the Merger Agreement, including, among others:

USWS Conduct of Business Pending the Merger

USWS agreed that during the period between the signing of the Merger Agreement until the earlier of the Effective Time and the date of termination of the Merger Agreement in accordance with its terms, USWS will, and will cause each of its subsidiaries to, conduct its business in all material respects in the ordinary course. USWS also agreed that, with certain exceptions as may be required by law or the Merger Agreement or as set forth in disclosure schedules to the Merger Agreement, and except with ProFrac's prior written consent, USWS will not, and will not permit any of its subsidiaries to undertake the following actions, among other things:

- · amend its or any of its subsidiaries organizational documents;
- split, combine or reclassify its shares of capital stock or those of its subsidiaries or declare, set aside or pay any dividend or other
  distribution in respect of the capital stock of USWS or its subsidiaries, or redeem, repurchase or otherwise acquire or offer to redeem,
  repurchase or otherwise acquire any of its securities or those of its subsidiaries, except for the declaration, setting aside or payment of any
  dividends or other distributions by any of its subsidiaries in the ordinary course of business consistent with past practice;

- issue, deliver, sell or authorize the issuance, delivery or sale of any equity securities, options or other securities convertible into or
  exchangeable or exercisable for equity securities, except under limited circumstances as set forth in the Merger Agreement;
- incur or commit to any capital expenditures or any obligations or liabilities, except for such expenditures, obligations or liabilities as contemplated by USWS' fiscal 2022 budget and capital expenditure plan (as provided by USWS to ProFrac prior to the execution and delivery of the Merger Agreement) (the "USWS Budget Plan") or any other capital expenditures not to exceed \$750,000 in the aggregate;
- acquire (by merger, consolidation, acquisition of stock or assets or otherwise), directly or indirectly, any assets, securities or businesses, other than (i) in the ordinary course of business consistent with past practice, (ii) pursuant to contracts in effect on June 21, 2022, (iii) any other acquisitions for consideration that is not in excess of \$500,000 individually or in the aggregate, (iv) between USWS and any of its subsidiaries or between its subsidiaries and (v) as contemplated in the USWS Budget Plan;
- sell, assign, license, lease or otherwise transfer, or abandon or create or incur any material Lien (as defined in the Merger Agreement) on any USWS' or its subsidiaries assets, securities, properties, interests or businesses in excess of \$500,000 in the aggregate, except under limited circumstances as set forth in the Merger Agreement.
- make any loans, advances or capital contributions to, or investments in, any other person or entity in excess of \$500,000 in the aggregate, other than intercompany indebtedness;
- create, incur, suffer to exist or assume any indebtedness for borrowed money or guarantees thereof, other than intercompany indebtedness and as contemplated by the USWS Budget Plan and as set forth in the disclosure schedules to the Merger Agreement;
- other than in the ordinary course of business consistent with past practice, enter into any agreement or arrangement that limits or otherwise restricts in any material respect USWS or any of its subsidiaries from engaging or competing in any line of business, in any location or with any person or entity, or would purport to limit, after the Effective Time, ProFrac or any of its subsidiaries from engaging or competing in any line of business in any material respect;
- other than in the ordinary course of business consistent with past practice (including renewals consistent with the terms thereof), (i) amend or modify in any material respect or terminate (excluding terminations or renewals upon expiration of the term thereof in accordance with the terms thereof) any Company Material Contract (as defined in the Merger Agreement) or waive, release or assign any material rights, claims or benefits under any Company Material Contract or (ii) enter into any contract that would have been a Company Material Contract had it been entered into prior to June 21, 2022;
- recognize any new labor organization, union, employee association, trade union, works council or other similar employee representative, or negotiate, enter into, amend, modify or terminate any Collective Bargaining Agreement (as defined in the Merger Agreement);
- · grant or modify any equity or equity-based awards;
- except as required pursuant to a Company Plan (as defined in the Merger Agreement) or a contract in effect prior to June 21, 2022, or as
  required by applicable law, (i) grant or provide any severance, retention or termination payments or benefits to any current or former
  employee, officer, non-employee director, independent contractor or consultant of USWS or any of its subsidiaries, (ii) accelerate the
  time of payment or vesting of, or the lapsing of restrictions with respect to, or fund or otherwise secure the payment of, any compensation
  or benefits to any current or former employee, officer, non-employee director, independent contractor or consultant of USWS or any of its
  subsidiaries, (iii) increase the compensation payable to any current or former employee, officer,

non-employee director, independent contractor or consultant of USWS or any of its subsidiaries, other than routine increases in base salaries or hourly base wage rates, as applicable, to employees in the ordinary course of business consistent with past practice and not in excess of the USWS Budget Plan as set forth in the in the disclosure schedules to the Merger Agreement, (iv) establish, adopt, terminate or amend any Company Plan or any plan, program, arrangement, policy or agreement that would be a Company Plan if it were in existence on June 21, 2022, (v) hire any employee of USWS or any of its subsidiaries or engage any other individual to provide services to USWS or any of its subsidiaries, other than as set forth in disclosure schedules to the Merger Agreement and under limited circumstances, (vi) terminate the employment of any current employee or the engagement of any individual independent contractor of USWS or any of its subsidiaries other than for cause or for performance-related reasons or a non-Merger related reason, or (vii) promote any employee of USWS or any of its subsidiaries to a position that reports directly to the Chief Executive Officer of USWS;

- knowingly waive, release, limit or condition any restrictive covenant obligation of any current or former employee or independent
  contractor of USWS or any of its subsidiaries, in each case that would reasonably be expected to have more than a de minimis adverse
  effect on USWS or any of its subsidiaries, taken as a whole;
- change any method of financial accounting, other than any method adopted or changed pursuant to a request made by a taxing authority or as required by applicable law;
- change any material tax election or settle or compromise any material income tax liability;
- adopt or publicly propose a plan of complete or partial liquidation, restructuring, recapitalization or other reorganization, in each case, of USWS or any material subsidiary of USWS;
- fail to maintain, allow to lapse or abandon any Owned Intellectual Property Rights or any Intellectual Property Rights (as such terms are defined in the Merger Agreement) that are exclusively licensed to USWS or any of its subsidiaries (other than to the extent permitted pursuant to the Merger Agreement);
- settle, or offer to propose to settle, any legal proceedings that requires payment by USWS or any of its subsidiaries in excess of \$1 million or that would include any non-monetary relief that would have more than a de minimis adverse effect on USWS and its subsidiaries, taken as a whole, or commence any proceeding other than the commencement of any proceeding seeking damages of less than \$1 million or relating to the transactions contemplated by the Merger Agreement; or
- · agree, resolve or commit to do any of the foregoing.

### Modifications of Certain Outstanding Securities and Awards

USWS agreed (i) to amend the terms of the USWS Equity Linked Convertible Notes pursuant to the Convertible Note Amendment, (ii) immediately following receipt of the USWS Stockholder Approval and at least one day prior to the Closing, amend the Certificate of Designations by filing the COD Amendment and (iii) use commercially reasonable efforts to amend the terms of the USWS Pool A Performance Awards and USWS Pool B Performance Awards not previously amended pursuant to an executed Award Agreement.

No Solicitation of Acquisition Proposal; Other Offers

The Merger Agreement provides that, from the time of the execution of the Merger Agreement until the Effective Time or the termination of the Merger Agreement, subject to certain exceptions, neither USWS nor any of its subsidiaries will, nor will USWS or any of its subsidiaries authorize any of its or their officers, agents,

control persons, employees, consultants, professional advisors and other agents (with respect to any person, the foregoing persons are referred to herein as such person's "Representatives") to:

- solicit, initiate or knowingly take any action to facilitate or encourage any inquiries regarding, or the making or submission of any
  proposal or offer, that constitutes, or could reasonably be expected to lead to, an Acquisition Proposal (as defined below);
- enter into, engage in or participate in any discussions or negotiations with, furnish any information relating to itself or any of its subsidiaries to, or afford access to the business, properties, assets, books or records of itself or any of its subsidiaries to, otherwise cooperate in any way with, or knowingly assist, participate in, knowingly facilitate or knowingly encourage any effort by any third party, that to USWS' knowledge, is seeking to make, or who has made, an Acquisition Proposal;
- make an Adverse Recommendation Change (as defined below); or
- enter into any agreement in principle, letter of intent, indication of interest, term sheet, Merger Agreement, acquisition agreement, option agreement or other contract relating to an Acquisition Proposal (other than an Acceptable Confidentiality Agreement) (as defined in the Merger Agreement).

Notwithstanding the above, at any time prior to the receipt of the USWS Stockholder Approval:

- USWS, directly or indirectly through its Representatives, may (i) engage in discussions with third parties that have made a written Acquisition Proposal that did not result from a material breach of the non-solicitation provisions of the Merger Agreement and (ii) furnish information relating to USWS or its subsidiaries pursuant to an Acceptable Confidentiality Agreement; provided such information is provided to ProFrac within 24 hours. Prior to taking the actions described in clauses (i) and (ii) in the preceding sentence, the USWS Board or USWS Special Committee must determine in good faith, after consultation with outside legal counsel, that based on the information then available and after consultation with USWS' outside financial advisor, such Acquisition Proposal is or is reasonably expected to result in a Superior Proposal (as defined below); and
- USWS may make an Adverse Recommendation Change if (i) USWS receives a Superior Proposal (that is not withdrawn) that did not
  result from a material breach of USWS' obligations under the Merger Agreement or (ii) in response to an Intervening Event (as
  defined in the Merger Agreement).

USWS may take such course of action as described in the two bullet points immediately above if, and only if, prior to taking such action, the USWS Board or the USWS Special Committee determines in good faith, after consultation with outside legal counsel that the failure to take such action would be inconsistent with its fiduciary duties under Delaware law.

USWS and the USWS Board are not prohibited from (i) taking and disclosing a position contemplated by Rule14e-2(a) or Rule 14d-9 under the 1934 Act or (ii) making any disclosure to the stockholders of USWS if the USWS Board determines in good faith, after consultation with its outside legal counsel, that the failure to make such disclosure would be inconsistent with the USWS Board's fiduciary duties under Delaware law.

USWS will, within 48 hours of receipt, notify ProFrac of any Acquisition Proposal (or any indication that a person is considering making an Acquisition Proposal). USWS will identify the third party making the Acquisition Proposal and the material terms and conditions thereof and will keep ProFrac reasonably informed on a reasonably prompt basis upon ProFrac's request, of the status of any such Acquisition Proposal.

An "Acquisition Proposal" means any proposal, inquiry, indication of interest or offer (whether or not in writing) from any person (other than ProFrac and its subsidiaries or affiliates) relating to or involving, whether in

a single transaction or series of related transactions: (i) any direct or indirect acquisition, lease, exchange, license, transfer, disposition (including by way of liquidation or dissolution of USWS or any of its subsidiaries) or purchase of any businesses or assets (including equity interests in subsidiaries but excluding sales of assets in the ordinary course of business) of USWS or any of its subsidiaries that constitute or account for twenty percent (20%) or more of the consolidated net revenues, net income or net assets of USWS and its subsidiaries, taken as a whole; (ii) any merger, consolidation, amalgamation, share exchange, business combination, issuance of securities, sale of securities, reorganization, recapitalization, tender offer, exchange offer, liquidation, dissolution, extraordinary dividend, or similar transaction involving USWS and a person or "group" (as defined in Section 13(d) of the 1934 Act) pursuant to which the stockholders of USWS immediately preceding such transaction hold less than eighty percent (80%) of the equity interests in the surviving or resulting entity of such transaction immediately following such transaction; or (iii) any combination of the foregoing.

An "Adverse Recommendation Change" means any of the following actions by the USWS Board or any committee thereof: (i) withholding or withdrawing (or amending, modifying or qualifying in a manner adverse to ProFrac) or proposing publicly to withhold or withdraw (or amend, modify or qualify in a manner adverse to ProFrac), the Company Board Recommendation (as defined in the Merger Agreement), (ii) failing to make the Company Board Recommendation in the Proxy Statement (as defined in the Merger Agreement), subject to the terms and conditions of the Merger Agreement, (iii) approving, recommending, or otherwise declaring or endorsing publicly to be advisable or publicly proposing to approve, adopt or recommend to be advisable any (A) Acquisition Proposal or (B) contract requiring USWS to abandon, terminate or fail to consummate the Merger or any other transaction contemplated by the Merger Agreement or (iv) (x) after public announcement of any Acquisition Proposal or any material modification thereto, failing to issue a press release that expressly reaffirms, without qualification, the Company Board Recommendation, within six (6) business days following ProFrac's written request to do so or (y) following the commencement of any Acquisition Proposal that is a tender offer or exchange offer relating to the securities of USWS, (A) publicly be neutral or make any recommendation in connection with such tender offer or exchange offer other than a recommendation against such offer or a temporary "stop, look and listen" communication by the USWS Board of the type contemplated by Rule 14d-9(f) under the 1934 Act or (B) fail to send to its stockholders a statement rejecting and recommending against any such tender offer or exchange offer within ten (10) business days of such commencement of such tender offer or exchange offer.

A "Superior Proposal" means a bona fide written Acquisition Proposal from any person (other than ProFrac and its subsidiaries or affiliates) (with all references to "20% or more" in the definition of Acquisition Proposal being deemed to reference "50% or more" and all references to "80% or less" in the definition of Acquisition Proposal being deemed to reference "50% or less"), which the USWS Board has, after consultation with the USWS' outside financial advisors and outside legal counsel, determined in its good faith judgment, (i) is reasonably expected to be consummated on the terms proposed and (ii) would, if consummated, result in a transaction more favorable to its stockholders (in their capacity as such) than the transactions contemplated by the Merger Agreement after taking into account all legal, financial, regulatory and other aspects of such Acquisition Proposal (including the existence of financing conditions, the conditionality of any financing commitments and the likelihood and timing of consummation (as compared to the transactions contemplated hereby)) and such other matters that the USWS Board deems relevant.

## Efforts to Hold USWS Special Meeting

USWS has agreed to duly call, give notice of, convene and hold a meeting of its stockholders for the purpose of obtaining the approval by the USWS Stockholders of the proposals described in this proxy statement/information statement/prospectus. Except as permitted in the Merger Agreement, the USWS Board must recommend that the USWS Stockholders vote in favor of the proposals described in this proxy statement/information statement/prospectus at the USWS Special Meeting and this proxy statement/information statement/prospectus is required to include such recommendation of the USWS Board. USWS may postpone or adjourn the USWS Special Meeting (i) to solicit additional proxies for the purpose of obtaining the USWS Stockholder

Approval, (ii) for the absence of quorum and (iii) as reasonably determined by USWS (after consultation with outside legal counsel) to comply with applicable law. Except as permitted in the Merger Agreement, USWS shall use its reasonable best efforts to solicit proxies to obtain the approval of the proposals described in this proxy statement/information statement/prospectus and to take all other action necessary or advisable to secure the USWS Stockholder Approval.

ProFrac and Merger Sub agreed to certain covenants in the Merger Agreement, including, among others:

## ProFrac Conduct of Business Pending the Merger

ProFrac agreed that during the period between the signing of the Merger Agreement until the earlier of the Effective Time and the date of termination of the Merger Agreement in accordance with its term, ProFrac will, and will cause each of its subsidiaries to, conduct its business in all material respects in the ordinary course. ProFrac also agreed that, with certain exceptions as may be required by law or the Merger Agreement or as set forth in disclosure schedules to the Merger Agreement, and except with USWS' prior written consent, ProFrac will not, and will not permit any of its subsidiaries to undertake the following actions, among other things:

- amend its organizational documents in a manner that would have an adverse impact on the value of ProFrac Class A Common Stock
  or that would reasonably be expected to prevent, or impede or delay, the consummation of the Merger or the transactions contemplated
  by the Merger Agreement;
- adopt or publicly propose a plan of complete or partial liquidation, restructuring, recapitalization or other reorganization;
- take any action (or omit to take any action) with the knowledge that such action (or omission) would reasonably be expected to result in a requirement to seek the approval by holders of ProFrac Class A Common Stock of the transactions contemplated hereby; or
- · agree, resolve or commit to do any of the foregoing.

## Indemnification and Insurance

From and after the Effective Time, ProFrac and the surviving corporation agree that, subject to applicable law, all rights to indemnification of each former and present director or officer of USWS or any of its subsidiaries with respect to acts or omissions occurring at or prior to the Effective Time as provided in their respective organizational documents as in effect as of June 21, 2022 or in any agreement, will survive the Merger and continue in full force and effect in accordance with their terms, and the surviving corporation will (and ProFrac will cause the surviving corporation to) honor all the terms thereof. For a period of no less than six (6) years after the Effective Time, ProFrac will cause to be maintained in effect the provisions in the organizational documents of the surviving corporation and each subsidiary of USWS regarding indemnification, exculpation and expense advancement in effect as of immediately prior to the Effective Time.

ProFrac will cause the surviving corporation to either (i) continue to maintain in effect for a period of no less than six (6) years after the Effective Time USWS' directors' and officers' insurance policies (the "**D&O Insurance**") in place as of June 21, 2022 or (ii) purchase comparable D&O Insurance for six (6) years after the Effective Time. In no event shall ProFrac or the surviving corporation be required to expend a one-time premium for such coverage in excess of in excess of 300% of the annual premiums paid by USWS prior to the date of the Merger Agreement in respect of such coverage.

# Listing of ProFrac Stock

ProFrac agreed to use its best efforts to cause the shares of ProFrac Class A Common Stock to be issued in the Merger to be listed on the Nasdaq, subject to official notice of issuance, prior to the Effective Time.

#### Certain Employee Benefits Matters

ProFrac agreed to, among other matters, provide each individual employed by USWS or its subsidiaries at the Effective Time, who continues to be employed with USWS or any of its subsidiaries immediately after the Closing Date with compensation and benefits that are substantially comparable in the aggregate to the compensation and benefits provided to similarly situated employees of ProFrac during such period.

The parties have agreed to additional covenants under the Merger Agreement and in connection with the Merger, including, among others:

## Efforts to Close the Merger

Each of the parties have agreed to use its reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper, or advisable under applicable law to consummate and make effective the Merger and the other transactions contemplated by the Merger Agreement as promptly as reasonable practicable, including, preparing and filing requisite documents with governmental entities or other third parties and obtaining and maintaining all necessary approvals, consents and authorizations.

### Other Covenants

The Merger Agreement contains certain other mutual covenants, including, among other things, covenants relating to:

- Debt Financing. ProFrac will, and will cause its affiliates to, use commercially reasonable efforts to obtain and consummate a financing to fund the repayment in connection with the Closing of certain indebtedness of USWS and its subsidiaries, which financing shall be on terms and conditions and subject to documentation, in each case, in form and substance satisfactory to ProFrac in its sole discretion, (the "Financing") as promptly as reasonably possible. Subject to the terms of the Merger Agreement and certain customary carve-outs. USWS agreed that it will, and will cause each of its subsidiaries to, use its reasonable best efforts to assist ProFrac in arranging the Financing. Obtaining the Financing or any other financing is not a condition to the consummation of the Merger;
- the preparation of a Registration Statement on Form S-4, which this proxy statement/information statement/prospectus forms part of;
- consultation with the other party in connection with public announcements relating to the Merger;
- providing notice to the other party of certain notices or communications received from any governmental entity and certain other matters with respect to the Merger Agreement and the transactions contemplated thereby;
- · access by each party to certain information about the other party;
- requirements of Section 16(a) of the 1934 Act; and
- cooperation between ProFrac and USWS in the defense or settlement of any stockholder litigation relating to the Merger, the Merger Agreement or the transactions contemplated thereby.

### **Conditions to Completion of the Merger**

The obligations of each of ProFrac and USWS to complete the Merger are subject to the satisfaction (or waiver) of the following mutual conditions:

 approval by the USWS Stockholders of the proposals required to complete the Merger (as further described in this proxy statement/information statement/prospectus);

- the absence of any applicable law or order prohibiting the consummation of the Merger or the ProFrac Stock Issuance;
- the expiration or termination of the HSR Act waiting period;
- ProFrac's Registration Statement on Form S-4, which includes this proxy statement/information statement/prospectus, being effective and not subject to any SEC stop order; and
- approval for the listing on Nasdaq of the shares of ProFrac Class A Common Stock to be issued in the Merger, subject to official notice of issuance.

The obligations of ProFrac and Merger Sub to complete the Merger are subject to the satisfaction, or waiver by ProFrac and Merger Sub, of the following further conditions:

- the accuracy of the representations and warranties of USWS set forth in the Merger Agreement, subject to the materiality standards set forth in the Merger Agreement, as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case such representations and warranties will be true and correct as of such date), and ProFrac's receipt of an officer's certificate from USWS to that effect;
- performance of, in all material respects, all covenants and obligations required to be performed pursuant to the Merger Agreement by USWS at or prior to the Closing Date, and ProFrac's receipt of an officer's certificate from USWS to that effect; and
- no Company Material Adverse (as defined in the Merger Agreement) shall have occurred since June 21, 2022, and be continuing and ProFrac's receipt of an officer's certificate from USWS to that effect.

The obligations of USWS to complete the Merger are subject to the satisfaction, or waiver by USWS, of the following further conditions:

- the accuracy of the representations and warranties of ProFrac and Merger Sub set forth in the Merger Agreement, subject to the materiality standards set forth in the Merger Agreement, as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case such representations and warranties will be true and correct as of such date), and USWS' receipt of an officer's certificate from ProFrac to that effect on behalf of ProFrac and Merger Sub;
- performance of, in all material respects, all covenants and obligations required to be performed pursuant to the Merger Agreement by ProFrac and Merger Sub at or prior to the Closing Date, and USWS' receipt of an officer's certificate from ProFrac to that effect on behalf of ProFrac and Merger Sub; and
- no Parent Material Adverse (as defined in the Merger Agreement) shall have occurred since June 21, 2022, and be continuing and USWS' receipt of an officer's certificate from ProFrac to that effect on behalf of ProFrac and Merger Sub.

### Termination of the Merger Agreement

The Merger Agreement may be terminated at any time before the Effective Time in any of the following ways:

- mutual written agreement of ProFrac and USWS; or
- by either ProFrac or USWS if:
  - the Merger has not been completed by March 21, 2023 (the 'End Date'); provided, that if on the End Date the condition to
    Closing related to antitrust approval (Section 9.1(c) of the Merger Agreement) is not satisfied or waived but all the other
    conditions to Closing have been satisfied or waived, then End Date will be extended, if ProFrac or USWS notifies the

- other party in writing on or prior to the End Date, to June 21, 2023, provided, however that the right to terminate the Merger Agreement under this bullet point shall not be available to any party whose breach of any its obligations set forth in the Merger Agreement has proximately caused the failure of the Closing to occur on or before the End Date;
- if there has been an order issued by a government entity of competent jurisdiction having jurisdiction over a material portion of the business of USWS, ProFrac or Merger Sub permanently prohibiting the consummation of the Merger or the ProFrac Stock Issuance, and such order has become final and nonappealable; provided, however, that the right to terminate the Merger Agreement under this bullet point shall not be available to any party whose breach of its obligation under the Merger Agreement had proximately caused such order to have been issued; or
- if the USWS Stockholders' Meeting concluded and the USWS Stockholder Approval was not obtained.

The Merger Agreement may also be terminated by ProFrac prior to the Effective Time if

- · an Adverse Recommendation Change has occurred, or
- there shall have been a breach of any representation, warranty, covenant, or agreement on the part of USWS such that the conditions to the Closing of the Merger related to USWS' representations and warranties or performance of covenants, as applicable, would not be satisfied and, such breach is incapable of being cured by the End Date; or, if capable of being cured by the End Date, shall not have been cured by the End Date; provided further, that ProFrac shall not have the right to terminate the Merger Agreement if ProFrac or Merger Sub is then in material breach of any representation, warranty, covenant, or obligation under the Merger Agreement that would cause any condition related to ProFrac's or Merger Sub's representations and warranties or performance of covenants not to be satisfied.

The Merger Agreement may also be terminated by USWS prior to the Effective Time if

- there shall have been a breach of any representation, warranty, covenant, or agreement on the part of ProFrac or Merger Sub such that the conditions to the Closing of the Merger related to ProFrac and Merger Sub representations and warranties or performance of covenants, as applicable, would not be satisfied and, such breach is incapable of being cured by the End Date; or, if capable of being cured by the End Date, shall not have been cured by the End Date; provided further, that USWS shall not have the right to terminate the Merger Agreement if USWS is then in material breach of any representation, warranty, covenant, or obligation under the Merger Agreement that would cause any condition related to USWS' representations and warranties or performance of covenants not to be satisfied; or
- (A) the USWS Board or USWS Special Committee authorizes USWS to enter into an agreement with respect to a Superior Proposal that did not result from a breach of Section 6.4 of the Merger Agreement, (B) substantially concurrent with the termination of the Merger Agreement, USWS enters into an agreement providing for a Superior Proposal that did not result from a breach of Section 6.4 of the Merger Agreement and (C) prior to or concurrently with the termination of the Merger Agreement, USWS pays to ProFrac the applicable fees as described below (foregoing items (A)-(C), collectively, the "Superior Proposal Termination Right").

If the Merger Agreement is validly terminated, the Merger Agreement will become void and of no effect without any liability of any party to the other party to the Merger Agreement (except as provided in "—*Termination Fees*," below) unless said party is in willful breach of its obligations.

### **Termination Fees**

USWS will pay ProFrac a termination fee of \$8,000,000 (the "USWS Termination Fee") if:

- ProFrac terminates the Merger Agreement because a USWS Adverse Recommendation Change has occurred;
- · USWS terminates the Merger Agreement pursuant to its Superior Proposal Termination Right, as described above; or
- USWS or ProFrac terminates the Merger Agreement because the USWS Stockholders' Meeting concluded and the USWS Stockholder Approval was not obtained, and at the time of such termination, ProFrac had the right to terminate the Merger Agreement.

The Merger Agreement provides that in the event that the Merger Agreement is terminated by ProFrac or USWS because (i) the Merger has not been completed by the End Date or the USWS Stockholders' Meeting concluded and the USWS Stockholder Approval was not obtained, or as in the case for ProFrac, ProFrac terminates the Merger Agreement for USWS' breach of its representation, warranty, covenant, or agreement and such breach has not been cured as specified in the Merger Agreement and (ii) at any time after June 21, 2022, prior to such termination a bona fide Acquisition Proposal has been publicly announced or publicly made known and not withdrawn, and (iii) within nine (9) months after such termination, (x) USWS has entered into an agreement with respect to any Acquisition Proposal or (y) any Acquisition Proposal has been consummated, then, in any such event, USWS is obligated pay to ProFrac the USWS Termination Fee immediately prior to or concurrently with the occurrence of either of the events described in the foregoing clauses (x) or (y).

However, for the purpose of the definition of Acquisition Proposal as used in the paragraph immediately above, references to "20%" and "80%" are replaced by "50%".

In the event that the Merger Agreement is terminated either by USWS or ProFrac because a USWS Stockholder Meeting concluded and the USWS Stockholder Approval was not obtained, then USWS will pay ProFrac and its affiliates their reasonable and documented out-of-pocket fees and expenses incurred in connection with the Merger and the transactions contemplated by the Merger Agreement (including expenses of their representatives) in an amount up to \$3,000,000.

## Other Expenses

Except as described above, all costs and expenses incurred in connection with the Merger Agreement, will be paid by the party incurring such costs or expenses. Notwithstanding the foregoing, ProFrac will pay all fees and expenses (other than USWS attorneys' fees) incurred in connection with the filing fees due in connection with filings required under the HSR Act.

### Amendments; Waivers

The Merger Agreement may be amended, modified or supplemented by the parties in writing and signed by each of the parties at any time prior to or after the adoption and approval of the Merger Agreement by USWS Stockholders, but after such adoption and approval, no amendment will be made which by applicable law would require the further approval of the USWS Stockholders without first obtaining such approval. Except as required by applicable law, no waiver of the Merger Agreement will require the approval of the stockholders of either ProFrac or USWS. Any provision of the Merger Agreement may be waived if, but only if, such waiver is in writing and is signed by each party against whom the waiver is to be effective.

#### Related Agreements

This section describes the material provisions of certain additional agreements entered into or to be entered into pursuant to the Merger Agreement (the "Related Agreements") but does not purport to describe all of the terms thereof. The following summary is qualified in its entirety by reference to such agreements filed as an exhibit to this proxy statement/information statement/prospectus. Stockholders and other interested parties are urged to read such Related Agreements in their entirety.

## Voting Agreement

Concurrently with the execution and delivery of the Merger Agreement, USWS Supporting Stockholders entered into a Voting Agreement with ProFrac. Pursuant to the Voting Agreement, each USWS Supporting Stockholder agreed to, among other things, (i) support and vote in favor of the Merger Agreement, the Merger and the other transactions contemplated by the Merger Agreement, (ii) take (and refrain from taking) certain other actions in connection with the transactions contemplated by the Merger Agreement and (iii) to the extent applicable, deliver a Conversion Notice (as defined in the Merger Agreement) to convert each USWS Supporting Stockholder's shares of USWS Series A Preferred Stock at the Merger Conversion Ratio as described above.

The USWS Supporting Stockholders together beneficially own, in the aggregate, approximately 44% of the currently outstanding shares of USWS Common Stock, which includes 9,568 shares of USWS Common Stock, or approximately 0.07%, held by THRC Holdings.

### Warrant Purchase Agreement

Concurrently with the execution and delivery of the Merger Agreement, the holders of the Term C Loan Warrants (collectively, the 'Term C Loan Warrant Holders'), including Crestview Partners, entered into a Warrant Purchase Agreement with ProFrac (the 'Warrant Purchase Agreement'). Pursuant to the Warrant Purchase Agreement, immediately prior to the Effective Time and conditioned upon the consummation of the Merger, the Term C Loan Warrant Holders will sell all of their respective Term C Loan Warrants to ProFrac or its subsidiaries in exchange for an aggregate purchase price of \$2,639,999.82.

#### Amendment to Certificate of Designations

Under the terms of the Merger Agreement, USWS is obligated to take all requisite actions so that, effective as of immediately prior to the Effective Time, each holder of shares of USWS Series A Preferred Stock issued and outstanding at such time would have the option to convert such shares into shares of USWS Common Stock pursuant to an Optional Merger Conversion (as defined below), including any actions required to cause the Certificate of Designations to be amended prior to the Closing by adoption of an amendment to the Certificate of Designations (the "Series A Certificate of Designations Amendment").

The Series A Certificate of Designations Amendment, if and when duly adopted and filed with the Delaware Secretary of State, would amend the Certificate of Designations (as so amended, the "Amended Series A Certificate of Designations") to provide each holder of shares of USWS Series A Preferred Stock with the option, exercisable by delivery of written notice to USWS (a "Conversion Notice"), to convert all but not less than all of such holder's shares of USWS Series A Preferred Stock (any such conversion, an "Optional Merger Conversion") into shares of USWS Common Stock. Under the terms of the Amended Series A Certificate of Designations, in connection with any Optional Merger Conversion, each share of USWS Series A Preferred Stock subject to such Optional Merger Conversion would be converted into a number of shares of USWS Common Stock (the "Optional Merger Conversion Shares") equal to the quotient obtained by dividing (i) the then applicable liquidation preference of such share of USWS Series A Preferred Stock (as determined in accordance with the Amended Series A Certificate of Designations), by (ii) an amount equal to \$7.32, subject from time to time to certain adjustments (such amount, the "Merger Conversion Price").

Amendments to USWS Equity Linked Convertible Notes

Under the terms of the Merger Agreement, the Company also agreed to cause the USWS Equity Linked Convertible Notes to be amended by execution of amendments thereto (the "Convertible Note Amendments"). On June 21, 2022, each of the Convertible Note Amendments was executed by USWS and the respective payees of the USWS Equity Linked Convertible Notes and became effective, including Convertible Note Amendments with respect to the USWS Equity Linked Convertible Notes held by Crestview Partners and by THRC Holdings, LP, one of the Wilks Parties. The Convertible Note Amendments amended the terms of the Equity Linked Convertible Notes in order to provide for, among other things, the conversion by USWS, pursuant to the terms of the Merger Agreement and as of immediately prior to the Effective Time (any such conversion, a "Note Merger Conversion"), of all of the outstanding principal and interest then owing under each USWS Equity Linked Convertible Note into a number of shares of USWS Common Stock equal to the quotient obtained by dividing (i) the amount of outstanding aggregate principal amount, plus accrued and unpaid interest, owing under such USWS Equity Linked Convertible Note through July 9, 2022, by (ii) \$7.32.

	THE USWS SPECIAL MEETING
Stock on the U	WS Board is using this proxy statement/information statement/prospectus to solicit proxies from the holders of shares of USWS Common JSWS record date for use at the special meeting of USWS Stockholders. USWS is first mailing this proxy statement/information spectus and accompanying form of proxy to USWS Stockholders on or about [], 2022.
Date, Time ar	nd Place
	cial meeting of USWS Stockholders will be held on [], 2022 at [] a.m., local time, solely by means of remote n in an audio-only virtual meeting format, at www.proxydocs.com/USWS.
Purpose of th	e USWS Special Meeting
At the U	ISWS Special Meeting, USWS Stockholders will be asked to
1. 8	adopt the Merger Agreement and the transactions contemplated thereby, including the Merger;
ι	to approve, for purposes of complying with the Nasdaq listing rules, the issuance of shares of USWS Common Stock to be issued by USWS apon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, in an amount equal to 20% or more of USWS Common Stock outstanding;
3. t	to approve the Series A Certificate of Designations Amendment;
	to approve an amendment to the A&R LTIP, to increase the maximum number of shares of USWS Common Stock that may be issued under the A&R LTIP by 2,000,000 shares; and
8	If submitted to a vote of USWS Stockholders, approve an adjournment of the USWS Special Meeting, including, if necessary, to solicit additional proxies in favor of the foregoing proposals if there are not sufficient votes for those proposals, as described in "Required Vote for Adjournment of the USWS Special Meeting" beginning on page 152.
Board Recom	mendations
advisable and proxy card to issuance of sha Linked Conve	WS Board has unanimously determined that the Merger Agreement and the transactions contemplated thereby, including the Merger, are fair to and in the best interests of USWS and the USWS Stockholders and unanimously recommends that the USWS Stockholders use the vote "FOR" the proposal to adopt the Merger Agreement and the transactions contemplated thereby, including the Merger, "FOR" the ares of USWS Common Stock to be issued by USWS upon the conversion of the USWS Series A Preferred Stock and the USWS Equity rtible Notes, "FOR" the approval of the Series A Certificate of Designations Amendment, "FOR" the approval of the amendment to the ad "FOR" the proposal to approve the adjournment, including, if necessary, in order to solicit additional proxies in favor of the adoption of greement.
USWS Recor	d Date; Shares Entitled to Vote
holders of reco Special Meetin	WS Board has fixed the close of business on [], 2022, as the USWS record date for the USWS Special Meeting. Accordingly, only ord of USWS Common Stock as of the close of business on the USWS record date will be entitled to notice of, and to vote at, the USWS not or any adjournment or postponement thereof. As of the USWS record date [] shares of USWS Common Stock were issued and the holders of USWS Common Stock are entitled to one vote per share on any proposal presented at the USWS Special Meeting.

#### Stock Ownership of Directors and Executive Officers

On the USWS record date, directors and executive officers of USWS and their respective affiliates owned and were entitled to vote [\_\_\_\_] shares of USWS Common Stock, or approximately [\_\_\_\_]% of the shares of USWS Common Stock outstanding on that date. To the knowledge of USWS, the directors and executive officers of USWS and their respective affiliates intend to vote their shares of USWS Common Stock in favor of all USWS proposals at the USWS Special Meeting. Two members of the USWS Board, Joel Broussard and David Matlin, are parties to the Voting Agreement with ProFrac.

### **Quorum Requirement**

In accordance with the bylaws of USWS, a quorum exists if the holders of outstanding shares of USWS Common Stock representing a majority of the voting power of all outstanding shares of USWS Common Stock entitled to vote as of the record date are present in person or by proxy at the USWS Special Meeting, including proxies on which abstentions (withholding authority to vote) are indicated. Abstentions and broker non-votes will be counted for purposes of establishing a quorum at the USWS Special Meeting. If less than a quorum is represented at the USWS Special Meeting, the presiding officer of the meeting or a majority of the shares so represented may move to adjourn the meeting. In the event a quorum is present at the USWS Special Meeting, but sufficient votes to approve any of the items proposed by the USWS Board have not been received, the presiding officer of the meeting or the persons named as proxies may propose one or more adjournments of the USWS Special Meeting to permit further solicitation of proxies. A stockholder vote may be taken on one or more of the proposals in this proxy statement/information statement/prospectus prior to such adjournment if sufficient proxies have been received and it is otherwise appropriate.

#### Votes Required to Approve USWS Proposals

Required Vote for Adoption of Merger Agreement (Proposal 1). The affirmative vote of a majority of the outstanding shares of USWS Common Stock entitled to vote is required to adopt the Merger Agreement and the transactions contemplated thereby, including the merger. Consequently, an abstention from voting or a broker non-vote on Proposal 1 will have the effect of a vote against Proposal 1.

Adoption of the Merger Agreement and the transactions contemplated thereby, including the merger, by the requisite vote of the USWS Stockholders is required to complete the merger.

Required Vote for Approval of issuance of additional shares (Proposal 2). The affirmative vote of a majority of the total votes cast at the USWS Special Meeting is required to approve the issuance of shares of USWS Common Stock in connection with the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes.

The approval of the issuance of additional shares upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes by the requisite vote of the USWS Stockholders is a condition to the Closing of the Merger.

**Required Vote for Approval of Amendment to Certificate of Designations for Series A Preferred Stock (Proposal 3).** The affirmative vote of a majority of the outstanding shares of USWS Common Stock entitled to vote is required to approve the Series A Certificate of Designations Amendment. Consequently, an abstention from voting or a broker non-vote on Proposal 3 will have the effect of a vote against Proposal 3.

The approval of the Series A Certificate of Designations Amendment by the requisite vote of the USWS Stockholders is a condition to the Closing of the Merger.

Required Vote for Approval of Amendment to the A&R LTIP (Proposal 4). The affirmative vote of a majority of the shares of USWS Common Stock present in person or represented by proxy at the meeting and

entitled to vote on the matter is required to approve the Amendment to the A&R LTIP. Abstentions will be counted in determining the total number of shares "entitled to vote" on Proposal 4 and will have the same effect as a vote "Against" Proposal 4. Broker non-votes will have no effect on the outcome of the vote on this proposal.

The approval of the Amendment to the A&R LTIP is not a condition to the Closing of the Merger. In the event that the requisite number of shares of USWS Common Stock are not voted to approve of the Amendment to the A&R LTIP, USWS will be required to settle certain equity awards in cash in connection with the Closing of the Merger.

Required Vote for Adjournment of the USWS Special Meeting (Proposal 5). Stockholder approval of any adjournment of the USWS Special Meeting, including, if necessary, to solicit additional proxies if there are not sufficient votes for the foregoing proposal, requires the affirmative vote of a majority of the votes cast by stockholders present in person or by proxy, whether or not a quorum is present at the meeting. Abstentions and broker non-votes will have no effect on the vote on Proposal 5.

## **Voting of Proxies**

By Mail. A proxy card is enclosed for your use. To submit your proxy by mail, USWS asks that you sign and date the accompanying proxy and, if you are a stockholder of record, return it as soon as possible in the enclosed postage-paid envelope or according to the instructions provided in the proxy card. If the envelope is missing, please see the instructions on your proxy card. If you hold your shares in "street name," please refer to your proxy card or the information provided to you by your bank, broker, custodian or record holder. When the accompanying proxy is returned properly executed, the shares of USWS Common Stock represented by it will be voted at the USWS Special Meeting in accordance with the instructions contained in the proxy.

If proxies are returned properly executed without indication as to how to vote, the USWS Common Stock represented by each such proxy will be considered to be voted in favor of all matters for consideration at the USWS Special Meeting as follows: "FOR" the proposal to adopt the Merger Agreement and the transactions contemplated thereby, including the Merger, "FOR" the issuance of shares of USWS Common Stock to be issued by USWS upon the conversion of the USWS Series A Preferred Stock and the USWS Equity Linked Convertible Notes, "FOR" the approval of the Series A Certificate of Designations Amendment, "FOR" the approval of the amendment to the A&R LTIP, and "FOR" the proposal to approve the adjournment, including, if necessary, in order to solicit additional proxies in favor of the adoption of the Merger Agreement.

Concurrently with the execution and delivery of the Merger Agreement, certain USWS Supporting Stockholders entered into a Voting Agreement with ProFrac. Pursuant to the Voting Agreement, each USWS Supporting Stockholder agreed to, among other things, (i) support and vote in favor of the Merger Agreement, the Merger and the other transactions contemplated by the Merger Agreement, (ii) take (and refrain from taking) certain other actions in connection with the transactions contemplated by the Merger Agreement and (iii) to the extent applicable, deliver a Conversion Notice (as defined in the Merger Agreement) to convert each USWS Supporting Stockholder's shares of USWS Series A Preferred Stock at the Merger Conversion Ratio as described above.

The USWS Supporting Stockholders together beneficially own, in the aggregate, approximately 44% of the currently outstanding shares of USWS Common Stock, which includes 9,568 shares of USWS Common Stock, or approximately 0.07%, held by THRC Holdings.

Your vote is important. Accordingly, please sign, date and return the enclosed proxy card whether or not you plan to attend the USWS Special Meeting in person.

By Telephone. If you are a stockholder of record, you may also submit your proxy by telephone by dialing the toll-free telephone number on your proxy card and providing the unique control number indicated on the

enclosed proxy card. Telephone voting is available 24 hours a day, seven days a week, and will be accessible until 11:59 p.m., New York City time, on [\_\_\_\_\_\_], 2022. Easy-to-follow voice prompts allow you to submit your proxy and confirm that your instructions have been properly recorded. If you hold your shares in "street name," please refer to your proxy card or the information provided by your bank, broker, custodian or record holder for information on telephone voting. If you are located outside the United States, Canada and Puerto Rico, see your proxy card or other materials for additional instructions. If you submit your proxy by telephone, you do not need to return your proxy card.

By Internet. If you are a stockholder of record, you may also choose to submit your proxy on the Internet. Internet voting is available 24 hours a day, seven days a week, and will be accessible until 11:59 p.m., New York City time, on [\_\_\_\_\_\_\_], 2022. Please refer to the enclosed proxy card for information about the website for Internet voting and the unique control number you will be required to provide. If you hold your shares in "street name," please refer to your proxy card or the information provided by your bank, broker, custodian or record holder for information on Internet voting. As with telephone voting, you will be given the opportunity to confirm that your instructions have been properly recorded. If you submit your proxy on the Internet, you do not need to return your proxy card.

## **Revocation of Proxies**

A proxy may be revoked by a stockholder at any time prior to it being voted by:

- delivering a revised proxy (by one of the methods described above) bearing a later date;
- · voting in person at the USWS Special Meeting; or
- notifying the Corporate Secretary of USWS of the revocation in writing at 1360 Post Oak Blvd., Suite 1800, Houston, TX 77056 in time to be received before the USWS Special Meeting.

Attendance at the meeting alone will not effectively revoke a previously executed and delivered proxy. If a proxy is properly executed and is not revoked by the stockholder, the shares it represents will be voted at the meeting in accordance with the instructions from the stockholder. If the proxy card or voting instruction form is signed and returned without specifying choices, the shares will be voted in accordance with the recommendations of the USWS Board. If your shares are held in an account at a broker, trustee or other nominee, you should contact your broker, trustee or other nominee to change your vote.

## Solicitation of Proxies

All costs of this solicitation of proxies will be borne by USWS. In addition to solicitations by mail, certain of USWS' directors, officers and regular employees, without additional remuneration, may solicit proxies in person or by telephone. Brokers, custodians and fiduciaries will be requested to forward proxy soliciting material to the owners of USWS Common Stock held in their names, and USWS will reimburse them for their reasonable out-of-pocket costs. Solicitation by USWS officers and employees may also be made of some USWS stockholders in person or by mail, telephone or telegraph following the original solicitation. In addition, USWS has engaged D.F. King & Co., Inc. to assist it in the distribution and solicitation of proxies at a fee of \$15,000, plus expenses.

Do not send any stock certificates with your proxy cards. If USWS Stockholders adopt the Merger Agreement at the USWS Special Meeting, the other conditions to Closing are satisfied and the Merger is completed, the exchange agent will mail transmittal forms with instructions for the surrender of stock certificates for shares of USWS Common Stock as soon as practicable after completion of the Merger.

# Householding

USWS has adopted a procedure approved by the SEC called "householding." For more information, see "Householding of Proxy Materials" on page 275.

## MERGER PROPOSAL (PROPOSAL 1)

USWS is submitting this Proposal 1 to its stockholders in order to obtain the requisite stockholder authorization to approve the Merger as required under Delaware law.

## Background

As described elsewhere in this proxy statement/information statement/prospectus, USWS has entered into a Merger Agreement with ProFrac and Merger Sub. The Merger Agreement provides for the merger of Merger Sub with and into USWS, with USWS surviving the Merger as the surviving corporation and an indirect subsidiary of ProFrac. The Merger will become effective when the parties to the Merger Agreement file a certificate of merger with the Secretary of State of the State of Delaware (or at such later time as ProFrac and USWS mutually agree and specify in the certificate of merger). At the Effective Time, each issued and outstanding share of USWS Common Stock will be converted automatically into the right to receive 0.3366 shares of ProFrac Class A Common Stock. USWS Stockholders will receive cash in lieu of any fractional shares of ProFrac Class A Common Stock that would have otherwise been received in the Merger. See "The Merger Agreement" for more information about the Merger and the Merger Agreement.

## Effect of Failure to Obtain Stockholder Approval

The approval of the USWS Stockholders is a condition to the Closing of the Merger. That being the case, if the USWS Stockholders do not approve this Proposal 1, the Merger will not close as a result.

## **Board Recommendation**

THE USWS BOARD RECOMMENDS A VOTE "FOR" THE APPROVAL OF THE MERGER PROPOSAL.

## ISSUANCE OF ADDITIONAL SHARES PROPOSAL (PROPOSAL 2)

USWS is submitting this Proposal 2 to its stockholders in order to obtain the requisite stockholder authorization in accordance with the Nasdaq listing rules to issue a number of shares of USWS Common Stock to the holders of USWS Series A Preferred Stock and the holders of USWS Equity Linked Convertible Notes in excess of 19.9% of the outstanding shares of USWS Common Stock (as of the date of the Merger Agreement).

#### Background

As described elsewhere in this proxy statement/information statement/prospectus, at the Effective Time, as part of the Merger, each share of USWS Common Stock will be converted automatically into the right to receive shares of the ProFrac Class A Common Stock at the Exchange Ratio. As a condition to the Merger, USWS agreed to take all necessary action to allow for (i) each holder of USWS Series A Preferred Stock issued and outstanding immediately prior to the Effective Time to convert such stock into shares of USWS Common Stock at the Merger Conversion Ratio (as set forth in the Series A Certificate of Designations Amendment described in further detail in Proposal 3), and (ii) any shares of USWS Series A Preferred Stock issued and outstanding immediately prior to the Effective Time not so converted at the Merger Conversion Ratio to automatically convert into shares of USWS Common Stock at the then-effective conversion rate as calculated pursuant to the Certificate of Designations. In addition, USWS agreed to take all necessary action to allow for each USWS Equity Linked Convertible Note issued and outstanding immediately prior to the Effective Time to automatically convert into a number of shares of USWS Common Stock equal to the quotient obtained by dividing (i) the amount of outstanding aggregate principal amount, plus accrued and unpaid interest, owing under such USWS Equity Linked Convertible Note through July 9, 2022, by (ii) \$7.32.

As a result of these conversions, USWS will issue in excess of 19.99% of the outstanding shares of the USWS Common Stock (as of the date of the Merger Agreement).

## Requirement to Seek Stockholder Approval

As a result of the listing of the USWS Common Stock on the Nasdaq, issuances of USWS Common Stock are subject to the Nasdaq listing rules, including Rule 5635(d), which requires USWS to obtain stockholder approval prior to the issuance of securities in connection with a transaction, other than a public offering, involving the sale, issuance or potential issuance of more than 19.99% of the outstanding shares of USWS Common Stock (or securities convertible into or exercisable for shares of USWS Common Stock) at a price less than the lower of (i) the closing price (as reflected on Nasdaq.com) immediately preceding the signing of the binding agreement, or (ii) the average closing price of the USWS Common Stock (as reflected on Nasdaq.com) for the five trading days immediately preceding the signing of the binding agreement (the "Nasdaq 20% Rule").

Under the Nasdaq 20% Rule, in no event may USWS issue shares of USWS Common Stock which equal more than 19.99% of the shares of USWS Common Stock outstanding immediately prior to the execution of the Merger Agreement (which was 2,564,179 shares based on 12,827,306 shares outstanding immediately prior to the execution of the Merger Agreement), unless USWS obtains stockholder approval to issue shares of common stock in excess of such share amount. Because the above described issuances will be at a price below the price required by the Nasdaq 20% rule and will be in excess of 19.99% of the outstanding shares of USWS Common Stock, USWS is required to seek shareholder approval to effect these issuances.

# Effect of Failure to Obtain Stockholder Approval

If the stockholders do not approve this Proposal 2, USWS will be unable to issue the shares of USWS Common Stock to the holders of USWS Series A Preferred Stock and the holders of USWS Equity Linked Convertible Notes in excess of 19.9% of the outstanding shares of USWS Common Stock. As this is a condition to the Closing of the Merger, the Merger may not close as a result.

# **Board Recommendation**

THE USWS BOARD RECOMMENDS A VOTE "FOR" THE APPROVAL OF THE ISSUANCE OF SHARES OF USWS COMMON STOCK TO THE HOLDERS OF USWS SERIES A PREFERRED STOCK AND THE HOLDERS OF USWS EQUITY LINKED CONVERTIBLE NOTES IN ACCORDANCE WITH NASDAQ LISTING RULE 5635(D).

## SERIES A CERTIFICATE OF DESIGNATIONS AMENDMENT PROPOSAL (PROPOSAL 3)

USWS is submitting this Proposal 3 to its stockholders in order to obtain the requisite stockholder authorization to effect the Series A Certificate of Designations Amendment.

## Background

As described elsewhere in this proxy statement/information statement/prospectus, under the terms of the Merger Agreement, USWS is obligated to take all requisite actions so that, effective as of immediately prior to the Effective Time, each holder of shares of USWS Series A Preferred Stock issued and outstanding at such time would have the option to convert such shares into shares of USWS Common Stock pursuant to an Optional Merger Conversion (as defined below), including any actions required to cause the adoption of the Series A Certificate of Designations Amendment.

The Series A Certificate of Designations Amendment will provide each holder of shares of USWS Series A Preferred Stock with the option, exercisable by delivery a Conversion Notice, to convert all but not less than all of such holder's shares of USWS Series A Preferred Stock into shares of USWS Common Stock. Under the terms of the Amended Series A Certificate of Designations, in connection with any Optional Merger Conversion, each share of USWS Series A Preferred Stock subject to such Optional Merger Conversion would be converted into the Optional Merger Conversion Shares. The description in this proxy statement/information statement/prospectus of the proposed Series A Certificate of Designations Amendment is qualified in its entirety by reference to, and should be read in conjunction with, the actual text of the amendment as set forth in the Certificate of Amendment attached as Exhibit B to Annex A to this proxy statement/information statement/prospectus.

## Requirement to Seek Shareholder Approval

Under applicable Delaware law, an amendment to the Certificate of Designations requires the approval of a majority of the shares of USWS Common Stock outstanding on the record date.

# Effect of Failure to Obtain Stockholder Approval

If the stockholders do not approve this Proposal 3, USWS will be unable to effect the Optional Merger Conversion. As this is a condition to the Closing of the Merger, the Merger may not close as a result.

## **Board Recommendation**

THE BOARD OF DIRECTORS RECOMMENDS A VOTE "FOR" THE APPROVAL OF THE SERIES A CERTIFICATE OF DESIGNATIONS AMENDMENT.

## AMENDMENT TO A&R LTIP (PROPOSAL 4)

## Background

At the USWS Special Meeting, USWS Stockholders will be asked to approve an amendment to the Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan (the "A&R LTIP"), to increase the total number of shares of USWS Common Stock authorized to be issued under the A&R LTIP by 2,000,000 in order to provide a sufficient number of shares to cover (i) equity awards made to Joel Broussard, the Chairman of the Board of USWS and the former President and Chief Executive Officer of USWS (the "Broussard RSU Award"), and (ii) shares underlying the USWS Pool A Performance Awards and the USWS Pool B Performance Awards, which were granted to certain of USWS' executive officers and other key employees. The A&R LTIP was approved by the stockholders of USWS on May 14, 2021.

The Broussard RSU Award was approved by the USWS Board on April 30, 2022. The USWS Board deemed it in the best interest of USWS to grant this award to Mr. Broussard in order to provide incentives for him to continue to provide service to USWS following his resignation as the Company's President and Chief Executive Officer on April 30, 2022. The Broussard RSU Award consisted of a grant of 190,419 restricted stock units (the "RSUs"), each representing the right to receive one share of USWS Common Stock on the terms set forth in an award grant notice and subject to the terms of the A&R LTIP. The Broussard RSU Award will vest in full in connection with the Merger. On vesting, USWS will issue Mr. Broussard a number of shares of USWS Common Stock equal to the RSUs vesting. These shares would exceed the number of shares available under the A&R LTIP and, consequently, absent approval by the shareholders of USWS, will be required to be settled in cash instead of shares of USWS Common Stock. The cash value of the Broussard RSU Award is \$1,319,076.

USWS entered into the USWS Pool A Performance Awards and the USWS Pool B Performance Awards with certain of its key employees, including certain executive officers, on November 5, 2020, August 31, 2021, April 30, 2022, and May 31, 2022. Joel Broussard and Kyle O'Neill, who are each members of the USWS Board, were granted 307,654 and 274,656 USWS Pool A Performance Awards, respectively, and 124,021 and 49,532 USWS Pool B Performance Awards, respectively. The USWS Pool A Performance Awards and the USWS Pool B Performance Awards were granted under the A&R LTIP and will become payable in connection with the Merger and, pursuant to their terms, will require the Company to issue an aggregate of 1,588,412 shares of USWS. These shares would exceed the number of shares available under the A&R LTIP and, consequently, absent approval by the shareholders of USWS, will be required to be settled in cash instead of shares of USWS Common Stock. The aggregate cash value of the USWS Pool A Performance Awards and the USWS Pool B Performance Awards is \$11,003,292.

The table below sets forth awards that the compensation committee of the USWS Board recommended, and the USWS Board subsequently approved, that are contingent on stockholder approval of the amendment to the A&R LTIP described above. If the amendment to the A&R LTIP is not approved by USWS Stockholders at the meeting, then the awards set forth in the table below will be settled in cash.

	Number of Restricted Stock Units Subject to	Value of Restricted Stock Units Subject to	Number of USWS Pool A Performance	Value of USWS Pool A Performance	Number of USWS Pool B Performance	Value of USWS Pool B Performance
Name and Position	Awards(#)	Awards(\$)(1)	Awards	Awards	Awards	Awards
Joel Broussard, Chairman of the Board of USWS	190,419	\$ 1,318,248	307,654	\$ 2,231,639	124,021	\$ 899,615
Kyle O'Neill, Director and President and Chief						
Executive Officer	_	_	274,656	\$ 1,992,280	49,532	\$ 359,292
Joshua Shapiro, Chief Financial Officer	_	_	162,058	\$ 1,175,525	13,576	\$ 98,477
Executive Group	_	_	436,714	\$ 3,167,805	63,108	\$ 457,768
Non-Executive Director Group	190,419	\$ 1,318,248	307,654	\$ 2,231,639	124,021	\$ 899,615
Non-Executive Officer Employee Group	_	_	407,321	\$ 2,954,597	128,960	\$ 935,441

#### Amendments to A&R LTIP

To effect the increase in the aggregate number of shares of USWS Common Stock that may be issued under the A&R LTIP, it is proposed that the first sentence of Section 4.1 of the A&R LTIP be deleted in its entirety and replaced with the following:

"Subject to adjustment in accordance with Section 11, no more than 2,802,365 shares of Common Stock shall be available for the grant of Awards under the A&R LTIP (the "Total Share Reserve"), and such number shall also be the maximum number of shares of Common Stock that may be issued in the aggregate pursuant to the exercise of Incentive Stock Options (the "ISO Limit")."

Other than the revision to the first sentence of Section 4.1 of the A&R LTIP, no other amendments or changes will be made to the A&R LTIP

# Effect of Failure to Obtain Stockholder Approval

The approval of this Proposal 4 is not a condition to the Closing of the Merger. If the stockholders do not approve this Proposal 4, USWS will be unable to settle the Broussard RSU Award, the USWS Pool A Performance Awards or the USWS Pool B Performance Awards in shares of USWS Common Stock upon the vesting of these awards in connection with the Merger. If USWS cannot settle these awards in USWS Common Stock, USWS would be required to settle the awards in cash at the effective time of the Merger.

## **Board Recommendation**

THE USWS BOARD RECOMMENDS A VOTE "FOR" THE APPROVAL OF THE AMENDMENT TO THE 2018 A&R LTIP IN ORDER TO INCREASE THE NUMBER OF SHARES AVAILABLE FOR ISSUANCE UNDER THE 2018 A&R LTIP IN AN AMOUNT EQUAL TO THE SHARES UNDERLYING THE AWARDS PREVIOUSLY GRANTED TO CERTAIN EXECUTIVE OFFICERS AND EMPLOYEES OF USWS.

## USWS ADJOURNMENT PROPOSAL (PROPOSAL 5)

# Background

The USWS Special Meeting may be adjourned to another time or place from time to time, if necessary or appropriate, to permit further solicitation of proxies in the event there are not sufficient votes at the time of the USWS Special Meeting to approve the Merger proposal or any other proposals put before the USWS Stockholders at the USWS Special Meeting.

If, at the USWS Special Meeting, the number of shares of USWS Common Stock present or represented and voting in favor of the Merger proposal or any other proposals put before the USWS Stockholders at the USWS Special Meeting is insufficient to approve such proposal, USWS intends to adjourn the USWS Special Meeting from time to time in order to enable the solicitation of additional proxies.

In the Adjournment Proposal, USWS is asking the USWS Stockholders to authorize the holder of any proxy solicited by the USWS Board to vote in favor of granting authority to the proxy holders, and each of them individually, to adjourn the USWS Special Meeting to another time and place from time to time for the purpose of soliciting additional proxies. If the Adjournment Proposal is approved, USWS could adjourn the USWS Special Meeting and any adjourned session of the USWS Special Meeting and use the additional time to solicit additional proxies, including the solicitation of proxies from the USWS Stockholders who have previously voted.

## **Board Recommendation**

THE USWS BOARD UNANIMOUSLY RECOMMENDS A VOTE "FOR" THE APPROVAL OF THE ADJOURNMENT PROPOSAL.

# INFORMATION ABOUT PROFRAC

Unless the context otherwise requires, all references in this section to "ProFrac," "we," "us," or "our" refer to ProFrac Holding Corp. together with its consolidated subsidiaries.

## Overview

We are a growth-oriented, vertically integrated and innovation-driven energy services company providing hydraulic fracturing, completion services and other complementary products and services to leading upstream oil and gas companies engaged in the E&P of North American unconventional oil and natural gas resources. Founded in 2016, ProFrac was built to be the go-to service provider for E&P companies' most demanding hydraulic fracturing needs. We are focused on employing new technologies to significantly reduce GHG emissions and increase efficiency in what has historically been an emissions-intensive component of the unconventional E&P development process. We believe the technical and operational capabilities of our fleets ideally position us to capture increased demand resulting from the market recovery and our customers' shifting preferences favoring the sustainable development of natural resources.

Our operations are primarily focused in the West Texas, East Texas/Louisiana, South Texas, Oklahoma, Uinta and Appalachian regions, where we have cultivated deep and longstanding customer relationships with some of those regions' most active E&P companies. We operate in three business segments: stimulation services, manufacturing and proppant production. We believe we are among the largest providers of hydraulic fracturing services in North America by hydraulic horsepower ("HHP"), with aggregate installed capacity of over 1.7 million HHP across 34 conventional fleets, of which, as of June 30, 2022, 31 were active, reflecting a net installed capacity of approximately 1.5 million HHP across our active fleets. We believe our strategy of incorporating lower-emission Tier IV diesel engines into our conventional fleets makes them among the most emissions-friendly and capable in the industry. Further, we believe that because of those fleets' capabilities and reliability, and our relentless focus on efficient and environmentally-sound energy service solutions, our high-quality customer base views us as an integral partner in their efforts to improve their ESG profiles without sacrificing service quality.

Our lower-emission conventional hydraulic fracturing fleets have been designed to reduce our customers' relative emissions footprint while handling the most demanding well completions, which are characterized by higher pumping pressures, higher pumping volumes, longer horizontal wellbores, more frac stages per lateral and increasing amounts of proppant pumped per well. As of June 30, 2022, approximately 40% of our pumps have Tier IV engines and approximately 50% have dual fuel capabilities. In addition, we have paired these technologies with our proprietary engine standby controllers ("ESCs") to reduce idle time, which is the time during which an engine generates the highest amount of emissions, by as much as 90%, and reduce fuel consumption and GHG emissions by as much as 24%. In addition, these ESCs are capable of cold starting the engines on our pumping units without the assistance of truck tractors. This technology allows us to significantly decrease the number of truck tractors required for our operations, not only further reducing overall emissions but also eliminating the capital, safety risks and operating and maintenance costs associated with operating the additional truck tractors required for fleets that do not utilize ESCs. On the whole, these cost savings are significant, allowing us to avoid an incremental \$15,000 per year in costs associated with each truck tractor eliminated from our operations. Since early 2021, we have installed ESCs throughout our fleets, and anticipate being able to realize total cost savings of approximately \$300,000 per year per fleet as a result. When further combined with our real time GHG emissions monitoring, our fleets create additional synergies in efficiency that result in cost savings for our customers. We intend to continue to upgrade and overhaul our other fleets with the goal of having all of our conventional fleets similarly equipped, a process made cheaper by our in-house manufacturing capabilitities detailed below. This strategy aligns with our ESG initiative

In addition to our existing low-emission conventional fleets, we are constructing electric powered hydraulic fracturing fleets equipped with Clean Fleet® technology licensed from USWS. Under our agreement with USWS, we have acquired 3 licenses and may acquire up to 17 additional licenses (along with certain other rights) to construct in-house new, electric-powered hydraulic fracturing fleets utilizing Clean Fleet® technology. This technology utilizes electric motors powered by lower-cost, lower-emission power solutions, including local utility-sourced line power, or on-site generation from natural gas produced and conditioned in the field, compressed natural gas ("CNG"), liquefied natural gas ("LNG"), hydrogen and/or traditional fuels, if needed. This flexibility in fuel supply can provide our customers with additional tools to meet their emissions and sustainability goals by reducing their reliance on diesel, as well as offer potentially significant fuel cost savings. We believe that our fleets equipped with Clean Fleet® technology will supplement our environmentally advantaged conventional fleets and provide our customers an optimized suite of options to satisfy their ESG objectives while maximizing operating efficiency. We expect to begin deploying the first of these electric-powered hydraulic fracturing fleets in the third quarter of 2022, and we have two more under construction, which we expect to be ready for deployment by the end of 2022. We believe that our new electric fleets, together with our existing conventional fleets, which we continue to optimize to incorporate efficiency-enhancing features, place us on the leading edge of the domestic hydraulic fracturing business and position us to maintain a high equipment utilization rate, low emissions and attractive profitability.

Facilitating the advanced technology and operational capability of our equipment is our vertically integrated business model and supply chain management, which allows us to manufacture, assemble, repair and maintain our own fleets and ancillary frac equipment, including power ends, fluid ends, flow iron and monolines. Our vertically integrated business model also allows us to offer customers a suite of ancillary services that enhance the efficacy of the well completion process, including, sand, completion chemicals and related equipment.

We operate facilities in Cisco, Aledo and Fort Worth, Texas, including an International Organization for Standardization (**1SO**") 9001 2015 certified OEM manufacturing facility, in which we manufacture and refurbish many of the components used by our fleets, including pumps, fluid ends, power ends, flow iron and other consumables and an engine and transmission rebuild facility that is licensed to provide warranty repairs on our transmissions. These facilities, which have a proven capability to manufacture up to 22 pumps, or 55,000 HHP, per month (including electric fleets) and perform substantially all of the maintenance, repair and servicing of our hydraulic fracturing fleets, provide in-house manufacturing capacity that enables cost-advantaged growth and maintenance.

Vertical integration enables us to realize a lower capital investment and operating expense by capturing the margin of manufacturing and/or maintenance, by recycling and refurbishing older machinery in our fleet, as opposed to disposing of it, and by enabling the ongoing improvement of our equipment and processes as part of a continuous research and development cycle. This combination also facilitates our "Acquire, Retire, Replace"™ approach to growing, maintaining and modernizing our fleets, and helps us mitigate supply chain constraints that have disrupted competitors' and customers' operations in the past.

Our in-house manufacturing capabilities also allow us to rapidly implement new technologies in a cost-effective manner not possible for many of our peers. We believe that as a result of this vertical integration, we are able to achieve conventional Tier IV dual fuel fleet construction costs of \$540 per HHP contrasted with an industry cost of up to \$861 per HHP, according to Daniel Energy Partners, and an average expected price to build electric fleets, excluding power generation, of \$467 per HHP inclusive of licensing costs.

Our manufacturing capabilities and control over the manufacturing process have allowed us to design and build hydraulic fracturing fleets to uniform specifications intended for deployment in resource basins requiring high levels of pressure, flow rate and sand intensity. We believe the standardized, modular configuration of our equipment provides us with several competitive advantages, including reduced repair and maintenance costs, reduced downtime, reduced inventory costs, reduced complexity in our operations, training efficiencies and the ability to redeploy equipment among operating basins. We believe that our uniform fleet specifications along

with the ability to more directly control our supply chain and end-of-life management for our equipment differentiates us from competitors who typically purchase such equipment from third party manufacturers and rely on such manufacturers or other third parties for repair and maintenance.

We also provide ancillary products and services, further increasing our value as a business partner to our customers, including frac sand, completion chemicals, frac design and related services, logistics coordination and real time data reporting, such as operational statistics, inventory management, completions updates and emissions monitoring.

Through our recent investment in Flotek, we have gained access to alow-cost, long-term supply of a full suite of completion chemicals required by our customers during the completion process, including Flotek's proprietary biodegradable complex nano-Fluid® technology, which is more environmentally friendly than commonly used alternatives. For additional information on our investment in Flotek, please see "Note 14—Acquisitions and Investments."

In addition, to meet our customers' need for proppant, we operate anapproximate three-million-ton-per-year sand mine and processing facility in Kermit, Texas, with 40.7 million tons of proved reserves as of December 31, 2021, which allows us to sell proppant to our customers in West Texas and Southeastern New Mexico. We also recently acquired approximately 6,700 acres near Lamesa, Texas, which we refer to as West Munger, that we are currently developing into an in-basin Permian Basin frac sand resource. We are in the process of installing mining and processing facilities at West Munger which, once operational, will be one of only two sand mines in the Midland Basin. West Munger and the Kermit sand mine are each located within 100 miles of approximately 98% of all horizontal rigs in the Permian Basin, providing us with ready access to potential customers. Our integrated service platform creates operational efficiencies for our customers and allows us to capture a greater portion of their development capital spending, positioning us to maintain high equipment utilization rates, low emissions and attractive profitability.

For the six months ended June 30, 2022, ProFrac Predecessor generated net income of approximately \$94.2 million and Adjusted EBITDA of approximately \$302.0 million. Annualized Adjusted EBITDA per fleet was \$22.9 million when annualizing the six months ended June 30, 2022 and \$27.2 million when annualizing the three months ended June 30, 2022. For the definitions of Adjusted EBITDA and Adjusted EBITDA per fleet and a reconciliation to their most directly comparable financial measures calculated and presented in accordance with GAAP, please read "—Selected Historical and Pro Forma Consolidated Financial Data of ProFrac."

## Competitive strengths

We believe the following characteristics differentiate us from our peers and uniquely position us to execute on our strategy to create value for our stakeholders:

• High performing, technologically advanced fleet focused on cash flow, increased efficiencies, and lower emissions. We believe we are strongly positioned to continue to respond to the increased demand for highly-efficient and environmentally advantaged energy services, which are those that produce fewer negative impacts on the environment than those provided by standard Tier II fleets. We believe that we have one of the largest fleet of low emissions and technologically advanced conventional frac equipment in the United States, with 40% of that fleet equipped with Tier IV engines and 50% with dual fuel capabilities as of June 30, 2022. We have begun to implement our "Acquire, Retire, Replace" strategy by retiring 650,000 HHP of older, emissions-intensive fleets and recycling or refurbishing equipment from such fleets.

We believe our technologically advanced fleets are among the most reliable and best performing in the industry with the capabilities to meet the most demanding pressure and flow rate requirements in the field.

For example, we are one of the few energy services companies to install 60-inch pumps in our fleets, providing for significantly higher capacity and capability. The combination of these factors provides us with an ability to operate efficiently in the most demanding environments while helping our customers meet their ESG goals.

Our standardized equipment reduces our downtime as our mechanics can quickly and efficiently diagnose and repair our equipment and reduces the amount of inventory we need on hand. We are able to easily shift equipment among operating areas as needed to take advantage of market conditions or to replace temporarily damaged equipment. This flexibility allows us to target customers that are offering higher prices for our services, regardless of the basins in which they operate. Standardized equipment also reduces the complexity of our operations, which lowers our training costs and improves our safety profile. Finally, our standardized, high specification equipment, manufacturing capabilities and direct control over significant portions of our supply chain lead to lower total cost of ownership, which we believe allows us to both increase our margins and meet increasing demand for efficient, environmentally-advantaged energy services.

To complement our modern and highly efficient conventional fleets, we expect to begin deploying the first of our electric-powered hydraulic fracturing fleets in the third quarter of 2022, and we have two more under construction, which we expect to be ready for deployment by the end of 2022. By replacing Tier II diesel engines with electric engines, we expect our fleets equipped with Clean Fleet® technology will reduce carbon emissions by up to 33% per fleet annually. These estimates are based on manufacturer specifications for fuel consumption of each engine configuration and hold constant operational factors that influence the rate of fuel consumption and emissions, such as rate and pressure. This expected reduction is equivalent to a reduction of approximately 1,700 cars on the road per year per fleet based on U.S. Environmental Protection Agency's ("EPA") estimates.

# ProFrac Cumulative Pump Configurations & Upgrades by Year:

- Vertically integrated business model enhances our ability to meet our customers' needs. We operate a vertically integrated business model that includes complementary manufacturing and ancillary products and services, including frac sand, completion chemicals, frac design and data reporting services. Our manufacturing capabilities enhance our profitability through reduced capital and maintenance expenditures, and provides a significant advantage in cost savings and supply chain management versus our peers who do not manufacture and rebuild/refurbish their own equipment and components. Furthermore, we have strategically invested in businesses providing ancillary products and services, such as our investments in West Munger, Flotek and FHE, which provides us with greater supply chain control and mitigates disruptions that have previously impacted the operations of our competitors and customers. We manufacture and refurbish many of the components used by our fleets, including pumps, fluid-ends, power-ends, certain high-pressure iron and other consumables at our facilities located in Cisco, Aledo and Fort Worth, Texas. We have the proven capability to manufacture up to 22 pumps, or 55,000 HHP per month (including electric fleets) and perform substantially all of the maintenance, repair and servicing of our hydraulic fracturing fleets in-house. We also operate an engine and transmission rebuild facility that is licensed to provide warranty repairs on our transmissions.
  - "We do the hard jobs." Vertical integration of our business enables us to take on premium frac jobs that have more demanding pressure and flow rate requirements that put extra wear and tear on frac equipment and require more frequent equipment rebuilds. We believe many competitors avoid these jobs as they lack the capital or repair capability to sustainably maintain their equipment and generate a reasonable return. At ProFrac, we find such challenging work more economically attractive than less intensive "commodity" work that is easier on equipment because we can be more competitive with higher associated profitability.
  - Rapid and cost-effective implementation of new technologies. Much of our equipment is customized for our operations and built with
    substantially uniform specifications. With our in-house manufacturing capabilities, we are able to rapidly fabricate, develop and deploy
    new equipment and rebuild/refurbish existing equipment with minimal reliance on third-party supply chains or paying a premium for
    bespoke orders or processes. In addition to manufacturing our pumping units, we have the capability to manufacture many of the other
    components of our fleets

- such as blenders and hydration units. Our manufacturing capabilities facilitated our development of the Centipede<sup>M</sup> high pressure flow system, which reduces non-productive time by reducing rig up time by up to 50% and iron connections by up to 70%, while also preventing shutdowns. We have also developed proprietary vibration monitoring technology that enables our artificial intelligence-driven predictive pre-failure maintenance, performance reporting and design customizations on core equipment. Finally, our preferred equity investment in FHE provides us with access to innovative technology, including its proprietary wellhead pressure control systems, RigLock<sup>M</sup> and FracLock<sup>M</sup>, that enhance well completion efficiency and safety and reduce emissions.
- Advantaged in tight market. Our vertical integration reduces the risk that we will be unable to source important components, such as
  fluid-ends, power-ends and other consumable parts and ancillary products and services, such as sand and chemicals. During periods of
  high demand growth for hydraulic fracturing services, external equipment vendors often report order backlogs of up to nine months,
  which can lead to increased costs or substantial delays to deploy fleets. The FTSI Acquisition strengthens our in-house repair and
  manufacturing facilities by increasing our capacity and adding a licensed transmission repair facility. We have historically
  manufactured all major consumable components and can quickly scale to support all of our fleets at full capacity.
- Insulated from supply chain issues. Our vertical integration on key completion commodities, such as chemicals and sand, mitigate our
  exposure to price spikes and supply shortages that have negatively impacted the financial results of some of our competitors. We have
  identified sources of pricing and supply chain risk and have made strategic investments to mitigate them, turning potential weaknesses
  into strengths. For example, we believe the Flotek investment, through which we monetized our procurement demand, demonstrates
  our commitment to our vertical integration strategy and provides greater control over our supply chain.
- Organizational culture based on world class service, innovation, safety, improving environmental impact and active contributions to our communities. We believe our corporate culture plays a significant role in our ability to consistently deliver excellent service to our customers, as well as our ability to attract and retain high quality personnel. We encourage innovation throughout our organization and empower our employees to innovate. For example, we maintain an innovation award program for our employees which provides cash incentives for changes to equipment and processes that improve efficiency and safety. Motivated by this program, our employees have developed numerous tools, processes and equipment enhancements that improve our operations, such as a tool for performing maintenance on fluid ends that reduces the time required for a routine maintenance procedure from 45 minutes to 15 minutes, our PadTrac system that performs live job monitoring and a tool for rebuilding butterfly valves that allows this task to be performed by a single technician. We are committed to the safety and wellness of our employees and we actively foster training, advancement and career development. We also seek to actively contribute our time and resources to positively impact the communities in which we work and live.
- Loyal and active customers that appreciate our efficiency, suite of services and ability to complete the most difficult and demanding projects. We have a strong portfolio of active customers that value our modern, technologically advanced equipment and our commitment to a more ESG-conscious service offering. As a result of the FTSI Acquisition, our customer base has expanded and diversified to include some of the larger independent exploration and production companies, in addition to our preexisting customer base consisting of leading private midsize operators. Our customers trust us to execute on their most technically demanding operations and value our unique ability to meet their needs with our vertically integrated business model. We believe our operating history combined with our emissions savings equipment and integrated supply chain has us well positioned to serve customers' needs. While certain of our customers have historically struggled with supply chain disruptions, our business model gives us an opportunity to provide these customers with bundled services, including frac sand, completion chemicals, frac design and related services, logistics and real time data reporting, helping to limit supply chain disruptions. Our track record of consistently providing high-quality, safe and reliable service has enabled us

to develop long-term partnerships with our customers, and we expect that our customers will continue to support our growth.

- Strong data and digital capabilities. Our focus on technology and innovation also underpins our efficiency through real time data analysis of
  operational statistics, inventory management, completions updates and emissions monitoring. We offer a comprehensive and competitive suite of
  data and digital solutions such as PadTrac and SOPHIA. PadTrac is a real time data stream that provides pertinent equipment data on location to our
  operators. SOPHIA is our cloud-based platform that accompanies the ESC and provides visibility into fuel savings and carbon footprint reduction.
   SOPHIA enhances the credibility, consistency, and transparency of carbon footprint quantification by following ISO standards. We believe our
  digital infrastructure saves time, money, and makes us a more productive and cost effective enterprise.
- Large scale and leading market share across most active major U.S. basins. We believe we are among the largest hydraulic fracturing providers in North America based on HHP. We operate in some of the most active basins in the United States, including in the West Texas, East Texas/Louisiana, South Texas, Oklahoma, Uinta and Appalachian regions and our operations have diversified exposure to both natural gas and oil producing areas. This geographic and commodity diversity reduces volatility in our revenue due to regional trends, relative commodity prices, adverse weather and other events. Our large footprint and standardized equipment enables us to rapidly reposition our fleets based on demand trends among different regions and allows us to spread our fixed costs over a greater number of fleets. Our large scale also strengthens our negotiating position with our suppliers and our customers. Additionally, we expect to leverage our strengths to capture market share in these regions in response to customer demand for more efficient and cleaner fleets.
- Experienced management and shareholder team that have driven extreme value creation for stakeholders in past endeavors. Our senior management team has more than 100 years of relevant experience in hydraulic fracturing and the energy industry. The management team is focused on the operational success of the Company and their interests are aligned with those of investors and customers. Additionally, our principal shareholders, the Wilks Parties, have a proven history of founding and growing pressure pumping companies. Prior to founding ProFrac, certain of the Wilks Parties founded FracTech Holdings, LLC, the predecessor to FTSI in 2000, which they grew into one of the largest North American hydraulic fracturing companies based on HHP before selling their 70% interest in that business in 2011 in a transaction that valued the business at approximately \$5 billion. The FTSI Acquisition reunites that business with a management team familiar with FTSI's personnel, culture and equipment and is well suited to execute our "Acquire, Retire, Replace"™ strategy through strategic cannibalization of FTSI's older fleets. Combined, the Wilks have more than 75 years of experience in the energy and energy services sectors. Under their leadership, we have grown our hydraulic fracturing business to a total of 34 fleets, as of June 30, 2022, with an aggregate of over 1.7 million HHP and pro forma 2021 revenues in excess of \$1.17 billion. As of June 2, 2022, the Wilks Parties owned approximately 88.6% of our voting stock. We believe that their experience will continue to benefit our operations and business. In addition, Lance Turner, FTSI's former Chief Financial Officer of FTSI from October 2015 to March 2022 will further streamline our efforts to efficiently integrate the FTSI business and operations into our business.

## **Business Strategies**

We intend to achieve our primary business objective of creating value for our stakeholders through the following business strategies:

• Position ourselves as a key partner to our customers in response to increasing focus on environmental sustainability. As the demand for energy services in the United States recovers from the lows experienced in 2020, we expect demand for our hydraulic fracturing services to continue to grow significantly. In particular, as one of the largest hydraulic fracturing service providers in North

America based on HHP, we believe our modern, technologically advanced fleets position us to capitalize on customer mandates for "next generation" frac fleets due to their lower emissions and the economic benefits of fuel cost savings. We also offer our customers a suite of ancillary products and services that we believe is responsive to our customers' evolving needs, including frac sand, completion chemicals, frac design, manufacturing and related services, logistics and real time data reporting. Rystad Energy estimates that total HHP capacity has declined by approximately 8.8 million HHP as of Q1 2022 from approximately 25 million HHP at the end of 2018, as a result of frac equipment permanently leaving the market due to scrapping, cannibalization and deferred maintenance. In addition, approximately 25% of remaining horsepower is comprised of obsolete or non-operational fleets, according to Rystad Energy. By contrast, we have focused on upgrading and expanding our fleets' capabilities and investing in ancillary products and services, and have positioned ourselves as ready to respond to our customers' needs as upstream activity returns and the focus on ESG-sensitive operations grows. Furthermore, our consistently high fleet utilization levels and 24 hours per day, seven days per week operating schedule should result in greater revenue opportunity and enhanced margins as fixed costs are spread over a broader revenue base. We believe that any incremental future fleet additions, including fleet additions expected to result from the Merger, will benefit from these trends and associated economies of scale.

Commitment to returns-driven, environmentally-advantaged investments and technology to support further emissions reduction and greater operational efficiency. We believe demand for lower emissions operations will outpace current supply and lead to further opportunities to deploy new technical solutions to our customers relative to our competition, particularly with natural gas playing an increasingly critical role in the transition away from less clean sources of energy. We have invested in various businesses and technologies that we plan to leverage to strengthen our market position and to better serve our customers as well as share in the fuel savings provided by our investments. For example, in January 2021, we acquired a 75% ownership stake in EKU Power Drives, GMBh ("EKU"), a provider of idle reduction technologies and the manufacturer of our proprietary ESCs. Engines with ESCs will automatically turn off during non-operating time, shutting down the powertrain when it is not pumping and immediately restarting it to full load upon request. This technology reduces the wear and tear on equipment, reduces fuel consumption and eliminates emissions when the engines on our pumping units are automatically turned off and on between stages. A typical frac spread will pump between 14 to 18 hours per day and idle the remaining time. As idle time widely varies between operating stages, most frac companies leave the engines in idle due to the labor-intensive process associated with using the power take-off on a truck tractor to re-start the engine. Based on our own provision of hydraulic fracturing services, we believe our ESCs eliminate roughly 90% of idle hours and result in substantially lower emissions and fuel costs. This reduction in idle time can reduce carbon dioxide emissions by up to 24% compared to standard operations in which engines generally run continuously during a frac job.

Additionally, we are supplementing our already environmentally-advantaged conventional fleets with electric fleets equipped with Clean Fleet\* technology, which will provide customers additional low emission and cost effective solutions. We intend to continue this focus on efficiency and emissions-optimized technology in order to capitalize on the increased demand for higher efficiency and higher performing hydraulic fracturing services. We believe that by pursuing the development of advanced technology in both our conventional fleets and complementary electric-powered fleets, in addition to pursuing strategic business combinations (such as the Merger) with companies that have already developed some of these advanced technologies, we will be well positioned to capture the increasing demand for highly capable and environmentally-advantaged energy services with which operators may satisfy their ESG imperatives.

We recently acquired West Munger, and made preferred equity investments in Flotek and FHE to enhance our access to products and services necessary during the well completion process in order to mitigate supply chain disruptions and improve our operational efficiencies. Flotek is a market leader in

- environmentally friendly and biodegradable chemical technologies; FHE is a pioneer in high pressure flow control equipment that is safer and more efficient than legacy industry processes; and West Munger will provide access to a geographically advantaged source of frac sand.
- Pursue accretive mix of organic growth and strategic consolidation. We plan to continue to grow our operations and fleets in response to increased customer demand as well as selectively evaluate potential strategic acquisitions that increase our scale and capabilities and diversify our operations. In response to supply constraints for frac sand, among other factors, we acquired West Munger in December 2021 and completed the SPS Acquisition during the third quarter of 2022, each of which is expected to further reduce our exposure to supply chain risks and increase our proppant production capacity. We are continuing to evaluate vertical integration of in-basin proppant and logistics opportunities in West Texas and other regions. Similarly, we anticipate that our acquisition of Best and investment in FHE will bolster our in-house manufacturing capabilities and will provide access to innovative technology. We believe opportunities exist to acquire older generation diesel frac fleets at attractive prices and use our in-house manufacturing capabilities to upgrade and maintain them, thus extending their useful life and maximizing their cash flow, after which they can be replaced with cutting edge dual fuel or electric technology as part of our "Acquire, Retire, Replace"™ strategy. We have already begun implementing this strategy with the fleets acquired in the FTSI Acquisition by retiring 650,000 HHP of older FTSI fleets and recycling or refurbishing equipment from such fleets as a source of spare parts and components in our vertically integrated manufacturing segment in connection with selectively upgrading legacy equipment to Tier IV dual fuel engines, increasing efficiency and sustainability. We estimate that FTSI's existing fleets can be converted to dual fuel capability at a cost of approximately \$2.0 million per fleet. The resulting displacement of older fleets should yield significant improvements in emissions, operating efficiency, safety and profitability and provide a source of spare parts and components that can reduce our maintenance capital expenditures. Our vertically integrated business model and in house manufacturing enables faster integration of assets we may acquire and allows us to more economically and efficiently cannibalize, refurbish, and redeploy equipment. Additionally, we expect that our technology and focus on lower emission fleets, including electric fleets we expect to acquire in connection with the Merger, will promote growth and attract new customers focused on reducing their emissions profiles.
- Continued focus on safe, efficient and reliable operations. We are an industry leader with a proven track record in safety with a Total Reportable Incident Rate ("TRIR") of 0.42 for the year ended December 31, 2021, including our manufacturing division, compared to the industry average of 0.70, according to the International Association of Oil & Gas Producers ("IOGP"). We prioritize safety in our equipment through mechanisms like AFEX fire control, which is installed on all of our field equipment and is designed to suppress fires immediately. We believe our excellent safety record is partly attributable to the standardization of our equipment, which makes it easier for mechanics and equipment operators to identify and diagnose problems with equipment before a safety hazard arises. Our fleets are also standardized to use Centipede™ mono-line, which has fewer iron connections on site and allows for a safer and quicker rig up versus traditional flow iron assemblies. Our streamlined, innovative equipment enables safer operations and time savings, mitigation of inefficiencies from shutdowns, and improvements relative to the amount of horsepower required to put down hole. Additionally, our standardized equipment and in-house manufacturing capability allows us to rapidly assess operations as well as test new equipment while also reducing the complexity of our operations and lowering our training costs.
- Focus on generating superior returns while maintaining a conservative balance sheet and financial policies. We plan to maintain a conservative balance sheet, which will allow us to better react to potential changes in industry and market conditions and opportunistically grow our business. We had \$421.3 million of net senior debt, defined as total senior debt of \$495.0 million less \$73.7 million of cash and equivalents, as of June 30, 2022. Our 2022 capital expenditure budget, excluding acquisitions, is estimated to be in a range between \$265 million and \$290 million. We have budgeted approximately

\$65 million to \$70 million to construct three electric-powered fleets. We are fully committed to building the three electric-powered fleets and have several customers interested in contracting these fleets. We intend to align fleet construction and other growth capital expenditures with visible customer demand, by strategically deploying new equipment in response to inbound customer requests and industry trends. Also included in our 2022 capital expenditure budget is \$25 million to \$30 million to construct the West Munger sand mine. The remainder of our 2022 capital expenditure budget, excluding acquisitions, will be used to fund maintenance capital expenditures, estimated to be \$2.75 million to \$3.0 million per fleet per year, and other growth initiatives such as upgrading Tier II fleets to Tier IV dual fuel fleets. We continually evaluate our capital expenditures and the amount that we ultimately spend will depend on a number of factors, including customer demand for new fleets and expected industry activity levels. We believe we will be able to fund our 2022 capital program from cash flows from operations. We are disciplined about deploying growth capital to our business, and expect investments in new fleets to have a simple payback of 2.0 years or fewer before investing. As a result of this approach, we believe that we operate one of the most profitable frac businesses and that our strategies and competitive advantages have contributed to our strong relative financial performance, as demonstrated by our history of positive EBITDA generation despite recent market volatility. Our vertical integration of key supply chains enables consistent cost management, low capital intensity and high conversion of EBITDA to cash flow, which we believe will help us deliver shareholder returns across market cycles, while maintaining a conservative balance sheet.

# **Properties**

Our corporate headquarters are located at 333 Shops Boulevard, Suite 301, Willow Park, Texas 76087. We currently own or lease the following additional principal properties:

Location	Size	Leased or owned	Purpose
Willow Park, TX	8,244 sqft	Leased	Corporate Headquarters
Smithfield, PA	47,800 sqft	Leased	Field Operations
Odessa, TX	21,100 sqft	Leased	Sales Office
Odessa, TX	50,634 sqft	Leased	Field Operations
Odessa, TX	82,800 sqft	Owned	Field Operations
Elk City, OK	42,330 sqft	Owned	Field Operations
Washington County, PA	41,660 sqft	Owned	Field Operations
Pleasanton, TX	62,950 sqft	Owned	Field Operations
Longview, TX	36,000 sqft	Owned	Field Operations
Vernal, UT	18,827 sqft	Leased	Sales Office
Aledo, TX	94,050 sqft	Owned	Manufacturing
Hobbs, NM	12,000 sqft	Leased	Field Operations
Seminole, TX	33,700 sqft	Leased	Field Operations
Marshall, TX	21,800 sqft	Leased	Field Operations
Pleasanton, TX	16,866 sqft	Leased	Field Operations
El Reno, OK	19,027 sqft	Leased	Field Operations
Dawson County, TX	6,700 acres	Owned	Raw Land
Ector County, TX	2,723 acres	Leased	Sand Mine
Winkler County, TX	641 acres	Owned	Sand Mine
Winkler County, TX	630 acres	Leased	Sand Mine
Fort Worth, TX	109,823 sqft	Leased	Manufacturing
Fort Worth, TX	78,696 sqft	Leased	Manufacturing
Fort Worth, TX	11,889 sqft	Leased	Manufacturing
Fort Worth, TX	89,522 sqft	Leased	Manufacturing
Fort Worth, TX	22,600 sqft	Leased	Corporate Office
Cisco, TX	130,000 sqft	Owned	Manufacturing

We believe that our facilities are adequate for our current operations.

#### Recent events

On August 18, 2022, we signed a non-binding letter of intent to acquire certain real property assets of a Texas limited liability company, all of the outstanding equity membership interest of a Texas limited liability company that is in our industry (the "**Proposed Transaction**"). The letter of intent provides for a purchase price of \$175 million, of which (i) \$87.5 million would be payable in cash at the closing of the Proposed Transaction, and (ii) the remaining \$87.5 million to be paid in the form of a note payable by us. The Proposed Transaction is subject to the completion of substantial due diligence, the negotiation and resolution of significant business and legal issues, the negotiation, documentation and completion of mutually satisfactory definitive agreements among the parties, the consent of our lenders, our ability to finance the cash payment at closing, and approval of the ProFrac Board. We cannot guarantee that the Proposed Transaction will be completed on acceptable terms, if at all.

# Our mining and processing facilities

We own and operate an approximately three million ton per year sand mine and processing facility located in Winkler County, Texas that we refer to as our "Kermit sand mine." The total net book value of the Kermit sand mine and its associated plant and equipment was \$51.9 million as of December 31, 2021. Our Kermit sand mine facility is located on Farm-to-Market Road (FM) 1218 approximately 14 miles north of Kermit, Texas and approximately 58 miles west of the Midland-Odessa area. The Kermit sand mine is a surface sand mine and a production stage property. Our Kermit sand mine facility features two wash plants and a dry plant with two rotary dryers that clean and classify the sand. We built the facility in 2017 and produce 40/70-mesh and 100-mesh (70/200) sand. Additional onsite facilities include a scale house, office, shop, quality laboratory and onsite housing for up to 40 employees. We believe the Kermit sand mine and its associated plant and equipment are maintained in good working condition. The plant does not crush the material, but cleans and classifies the sand. Once the product is appropriately processed, it is stored in one of eight storage silos until it is transported by truck to its destination.

The Kermit sand mine is serviced by three phase power that is routed alongFarm-to-Market Road (FM) 1218, which runs parallel to the western property line. The pipeline providing natural gas supply for the dryers is also routed along this corridor. Plant process water is supplied by 12 wells installed around the periphery of the property. Additionally, water is recycled from the wash process water after fines are removed via settling with a flocculent in a series of constructed ponds. As the mine progresses, silt ponds are constructed in mined-out pits. Wastewater disposal from offices and other buildings are collected via holding tanks and serviced on a regular basis. Potable water is provided by off-site sources (bottled water), although a public water system permit application has been filed.

We are in the process of developing 6,700 acres of raw land in Dawson County, Texas, which we refer to as "West Munger," into a sand mine that would permit us to mine and process two million tons of sand per year. We are currently installing mining and processing facilities at West Munger and expect the facility to be operational in the third quarter of 2022. The approximately 6,700 acre site is located in the Midland Basin near Lamesa, Texas. West Munger and the Kermit sand mine, which is more fully described below, are each located within 100 miles of approximately 98% of all horizontal rings in the Permian Basin, providing us with ready access to potential customers.

On July 25, 2022, the Company completed the SPS Acquisition of 100% of the issued and outstanding membership interest of each of SP Monahans and SP Sales (collectively, "Monahans Sand Mine"), the West Texas subsidiaries of Signal Peak Silica, for a purchase price of approximately \$90.0 million in cash plus working capital adjustments of approximately \$10.0 million. The Monahans Sand Mine is an approximately three

million ton per year sand mine and processing facility located in Ector County, Texas, on Interstate 20, approximately 10 miles northeast of Monahans, Texas, and approximately 50 miles southwest of the Midland-Odessa area. The Monahans Sand Mine is a surface sand mine and a production stage property that produces 40/70-mesh and 100-mesh (70/200) sand. Our logistically advantaged Monahans Sand Mine includes a wet and dry plant and storage silos, as well as automated, multi-line, drive-in, drive-out trucking capabilities.

## Our production

We produced and sold 1,646,292 tons, 736,726 tons and 1,015,665 tons of frac sand during the years ended December 31, 2021, 2020 and 2019, respectively from the Kermit sand mine.

#### Our permits

The Kermit sand mine's current NSR permit is renewable in 2028. Other permits held for such mine include: Stormwater, Above Ground Storage Tank, Aggregate Production Operation, and a Public Water System application (pending). There are no formal state or federal reclamation plans or permits required for the operation.

#### Our reserves

We believe we have a high-quality mineral reserve base. The SEC has adopted amendments to its disclosure rules to modernize the mineral property disclosure requirements, which are codified in Regulation S-K subpart 1300. This proxy statement/information statement/prospectus has been prepared in accordance with the requirements of Subpart 1300 of Regulation S-K, which first became applicable to us for the fiscal year ended December 31, 2021. As used in this proxy statement/information statement/prospectus, the terms "mineral reserve" and "proven mineral reserve" are defined and used in accordance with Subpart 1300 of Regulation S-K.

## Summary of reserves

We follow Subpart 1300 of Regulation S-K in determining our mineral reserves. Exploration samples are evaluated in our laboratory facilities to assess product quality and mining/processing parameters. Members of our sales management team assess the salability of the product(s). Geologic, topographic and site data are used to create a geologic model and mining plan. We prepare an analysis of operating costs, capital costs and long-term anticipated sales volume and price to ensure the economic viability of the reserve. To opine as to the economic viability of our mineral reserves, John T. Boyd Company ("John T. Boyd") reviewed this analysis at the time of the proven reserve determination.

The quantity and nature of the mineral reserves are estimated by our internal mine planning team and third-party companies. According to John T. Boyd, the relatively uniform nature of the sand deposit underlying the property, combined with laboratory testing results, indicate the sand deposit will produce a high-quality proppant sand product that will meet the customer specifications for regional proppant sand within the Permian Basin. Analysis of sand by independent third-party testing companies indicates that the reserves have characteristics which generally meet American Petroleum Institute specifications with regard to crush strength, turbidity and roundness and sphericity. Our internal reserve estimates are provided to John T. Boyd for review annually so that third-party approved additions or reductions can be made to our mineral reserves and mineral resource calculations due to ore extraction, additional drilling and delineation, property acquisitions and dispositions or quality adjustments. Before acquiring new mineral reserves, we perform surveying, drill core analysis and other tests to confirm the quantity and quality of the acquired mineral reserves. John T. Boyd has reviewed our December 31, 2021 mineral reserves, and we intend to continue retaining third-party engineers to review our mineral reserves on an annual basis.

The following tables provide a summary of our Kermit sand mine, which primarily provides proppant for the oil and natural gas industry, as of December 31, 2021 and 2020 based on a price of \$19.11<sup>(1)</sup> per ton:

						As of December	er 31, 2021
	Owned/	Area	Proven reserves (in thousands	Probable reserves (in thousands	Implied average reserve life		Recovery*
361 (3 (3 ()				,			
Mine/plant location	leased	(in acres)	of tons)	of tons)	(in years)	Product Size	(%)
Winkler County, TX	Owned	641	27,379	896	26	40/200-Mesh	78
Winkler County, TX	Leased	630	13,287	6,288	18	40/200-Mesh	78
Total		1,271	40,666	7,184	44		

- \* Recovery % represents the overall product yield after mining and processing losses.
  - (1) The sales price forecast, by product, is based on second quarter of 2021 average prices, and reflects a rebound from 2020 prices. John T. Boyd opines that this is a reasonable price projection.

Based on our proven reserves as of December 31, 2021, and average annual production volume for the three years ended December 31, 2021, our proven sand reserves had an implied average reserve life of more than 44 years.

						As of December	r 31, 2020
	Owned/	Area	Proven reserves (in thousands	Probable reserves (in thousands	Implied average reserve life		Recovery*
Mine/plant location	leased	(in acres)	of tons)	of tons)	(in years)	Product Size	(%)
Winkler County, TX	Owned	641	29,025	896	27	40/200-Mesh	78
Winkler County, TX	Leased	630	13,287	6,288	18	40/200-Mesh	78
Total		1,271	42,312	7,184	45		

<sup>\*</sup> Recovery % represents the overall product yield after mining and processing losses.

Based on our proven reserves as of December 31, 2020, and average annual production volume for the three years ended December 31, 2020, our proven sand reserves had an implied average reserve life of more than 45 years.

The year over year change from December 31, 2020 to December 31, 2021 of approximately 3% in reportable reserve tons is due to the normal mining production activity occurring on the property whereby the run-of-mine material is extracted for the production of saleable frac sand product.

Estimates of in-place frac sand resources for the Kermit sand mine were prepared by performing the following tasks:

- Available drilling logs and laboratory testing results were compiled and reviewed to check for accuracy and to support development of the
  geologic model. The geologic database utilized for modeling and estimation consists of results from 27 of the 35 holes as 8 holes from the
  first campaign were twinned by the second campaign for verification purposes. The geologic data were imported into Carlson Software, a
  geologic modeling and mine planning software suite that is widely used and accepted by the mining industry.
- A geologic model of the deposit was created in Carlson Software using industry-standard grid modeling methods well-suited for simple stratigraphic deposits. The geologic model delineates the top and bottom of the mineable sand horizon and the distribution of the product size fractions across the deposit. The top and bottom of the mineable frac sand interval were established thusly:
  - As there is minimal overburden material across the property, the top of the mineable sand interval was defined as the current ground surface as provided by an aerial topographic survey conducted on July 1, 2021.

- The bottom of the mineable sand interval was established by either the bottom of the drill hole, or where present by the top of
  excessively silty intervals commonly found near the bottom of the deposit.
- · After reviewing the continuity and variability of the deposit, suitable resources classification criteria were developed and applied.
- John T. Boyd then reviewed the proposed mining regions identified by management. Estimation of the thenin-place frac sand resources for
  the Kermit sand mine assumes mining operations using standard surface excavation equipment, which is widely utilized for mining of similar
  deposit types. As such, the estimates were subject to the following setbacks and slope requirements:
  - 50 ft inside of property lines.
  - 300 ft from pipeline easements.
  - 50 ft around the wet and dry process plant area, housing camp areas, and main access road/right of way.
  - An overall pit wall slope of 3:1 (approximately 19 degrees).
- In-place volumes for each of the proposed mining blocks were calculated from the geologic model within Carlson Software. A dry,in-place, bulk density of 100 pounds per cubic foot was used to calculate the in-place tonnage of frac sand.

Further information can be found in Section 6.2.1 and Section 6.2.2 of our technical report summary prepared by John T. Boyd and in the addendum to the technical report summary prepared by John T. Boyd, which are filed as Exhibits 99.5 and 99.6, respectively, to the registration statement of which this prospectus forms a part.

# Quality summary

Samples gathered during exploration were used to create 3 composite product samples that were tested by two separate third-party laboratories (PropTester and Stim-Lab) for API RP 19C/ISO 13503-2 proppant sand characteristics. ProFrac management also conducted in-house crush testing on samples obtained from their plant feed stream. Both of the independent laboratories showed similar results overall, with management's internal laboratory testing on crush strength showing roughly 1,000 psi lower than each third-party laboratory. Testing was performed on various product sizes; however, for purposes of this estimate, John T. Boyd discussed only the 40/70 mesh and 70/200 mesh samples.

	Average API/ISO Test Results By Product Size				
	40/70 mesh			70/200 mesh	
Test	Result	Specification		Result	
Sphericity	0.7	≥	0.6	0.7	
Roundness	0.6	$\geq$	0.6	0.6	
Acid Solubility (%)	2.7	≤	3.0	3.4	
Turbidity (NTU)	n/a	≥	250	n/a	
K-Value (000 psi)	7		_	10	

<sup>\* 100-</sup>mesh proppant sand material currently does not have an API/ISO specification

The composited sample testing suggests the Kermit sand mine is capable of producing frac sands which meet minimum API/ISO recommended testing characteristics.

The relatively dense drill hole coverage, combined with the frequency of sample intervals taken during exploration, provides a sufficiently detailed understanding of the extent and quality of the frac sand deposit underlying the Kermit sand mine to support the estimates of reserves reported herein.

## Data verification

John T. Boyd did not verify historic drill hole data by conducting independent drilling in areas already explored. It is customary in preparing proppant sand resource and reserve estimates to accept basic drilling and quality testing data as provided by management, subject to the reported results being judged representative and reasonable.

John T. Boyd's efforts to judge the appropriateness and reasonability of the source exploration data included reviewing provided drilling logs, sampling procedures, frac sand quality testing results, examining archival sample intervals, and discussing aspects of developing the Kermit sand mine with us during a site visit.

## Market analysis

Permit submissions for horizontal oil and gas wells in the Permian Basin indicate a continuation of strong drilling ahead. The increase in demand for frac sand for use in the hydraulic fracturing process has resulted in a significant rise in sand prices as well as constraints on supply availability. According to Lium, total U.S. frac sand demand is expected to increase by 31% in 2022 compared to 2021 and reach 117 million tons, with the Permian expected to account for approximately 57% of the total U.S. demand. Frac sand pricing has surpassed pre-COVID levels, with Permian FOB mine pricing reaching as high as \$60/ton in the spot market in the first quarter of 2022, according to Lium.

Consequently, with increases in production and well completions, activity at frac sand mines in the region have increased. Per the Mine Safety and Health Administration ("MSHA"), operating hours for the second quarter of 2021 for Permian Basin frac sand mines were up 20% sequentially. In fact, only two mines saw a material decrease in operating hours from Q1 2021 to Q2 2021 while several had increases of more than 80% sequentially. John T. Boyd anticipates stable frac sand pricing with a slight upside bias due to potential supply chain disruptions and high commodity input costs.

For more information, please see "Management's Discussion and Analysis of Financial Condition and Results of Operations of ProFrac" and Section 1.7.1 of our technical report summary prepared by John T. Boyd which is filed as Exhibit 99.5 to the registration statement of which this proxy statement/information statement/prospectus forms a part.

## Surface and mineral rights

A portion of our Kermit sand mine reserves in Winkler County, Texas are located on approximately 630 acres that we lease under a lease that terminates in 2052. The lease imposes a royalty rate of 2% of gross sales revenue and requires that we commence production from the leased premises by January 1, 2032.

The Monahans Sand Mine is located on 2,723 acres in Ector County, Texas, under a lease that expires in 2047. The lease imposes a royalty payment that is based on a formula of tonnage sold or gross sales, whichever is greater, and is subject to an annual minimum royalty throughout the lease period.

## Our resources

There are no reportable frac sand resources excluding those converted to frac sand reserves for the Kermit sand mine. Quantities of frac sand controlled by Alpine within the defined boundaries of the Kermit Property which are not reported as frac sand reserves are not considered to have potential economic viability; as such, they are not reportable as frac sand resources.

The Kermit sand mine is in an area of west Texas where the High Plains and Trans-Pecos desert regions converge. Surficial features of the High Plains region typically consist of thin clay and soil intervals covering caliche, with mesquite, juniper, and scrub grass cover. The Trans-Pecos region has numerous landform types ranging from the rising slopes of the Guadalupe Mountain Range, to the desert dune deposits of the Sand Hills areas. Just to the east of the study area, the Caprock Escarpment, a towering landform consisting of caliche capstone, abruptly stands hundreds of feet above the desert floor.

Surficial geologic units overlying the area are predominantly Quaternary age unconsolidated deposits, ranging from windblown sheet sands and dunes to alluvial sands, silts, clays, and caliche. Origins of these deposits are believed to be a combination of eroded bedrock material from the southern Rocky Mountains, and locally eroded Ogallala Formation sandstone. As portions of the southern Rockies were eroded via weathering, particles were carried to the Pecos River. Ancient flooding events of the Pecos River resulted in the suspended particles being deposited into flood plains. Once flood waters receded, winds took over, drying and further transporting these particles into the western Texas region.

The Caprock Escarpment marks the eastern-most extent of the surficial sand deposits. Winds transporting particles into the area are thought to have collided with the escarpment, slowing and dropping particles out to where they accumulated over time. Winnowing processes caused some degree of particle sorting to occur. Due to the mechanisms and long distances of particle transport, sand grains were abraded and rounded as they reached their current locations.

The Kermit sand mine contains no discernable overburden materials with the exception of sparse areas of vegetation and roots, which are easily removed during processing. The surface sheet and dune sands here are generally mineable from the surface down to the total defined depths of the deposit. As the target formation is located at the surface, there are no geologic features which would materially impact mineability of the surface sands.

Quaternary unconsolidated deposition covers nearly all of the subject property, with the surficial deposits noted as generally ranging from60-ft to over 300-ft in thickness. A generalized stratigraphic chart of the geologic units in Winkler County, Texas is presented below.

System	Series	Geologic Unit		
		Sheet and Dune Sand		
	Pleistocene / Holocene	Loess		
Quaternary	Pleistocene	Unconsolidated Alluvium		
Neogene	Pliocene	Ogallala Formation		

#### Our customers

Our customers consist primarily of E&P companies in the continental United States. Our top five customers accounted for approximately 44.9% and 51.8% of our revenue for the years ended December 31, 2021 and 2020, respectively. During the year ended December 31, 2021, Rockcliff Energy Management, LLC accounted for 15.4%, Sabine Oil & Gas Corporation accounted for 11.1%, Surge Energy America accounted for 7.4%, Henry Resources LLC accounted for 6.1% and EAP accounted for 4.9%, respectively, of our total revenue.

## Competition

The markets in which we operate are highly competitive. To be successful, an energy services company must provide services that meet the specific needs of E&P companies at competitive prices. Competitive factors impacting sales of our services are price, environmental profile of our equipment and operations, reputation and

technical expertise, service and equipment quality, and health and safety standards. Although we believe our customers consider all of these factors, we believe price and equipment capabilities are the key factors in our customers' criteria in choosing a service provider. While we seek to price our services competitively, we believe many of our customers elect to work with us based on the capabilities and environmental profile of our fleet, as well as our extensive experience and operational expertise in U.S. unconventional oil and gas production, integrated business model and uniform fleet of standardized, high specification equipment.

We provide our services primarily in the West Texas, East Texas/Louisiana, South Texas, Oklahoma, Uinta and Appalachian regions, and we compete against different companies in each of those locations. Our major competitors include Halliburton Company, Liberty Oilfield Services Inc. and NexTier Oilfield Solutions Inc.

## Seasonality

Our results of operations have historically reflected seasonal tendencies, generally in the fourth quarter, relating to the conclusion of our customers' annual capital expenditure budgets, the holidays and inclement winter weather during which we may experience declines in our operating results.

# Operating risks and insurance

Our operations are subject to hazards inherent in the energy services industry, such as accidents, blowouts, explosions, fires and spills and releases that can cause personal injury or loss of life, damage or destruction of property, equipment, natural resources and the environment and suspension of operations.

In addition, claims for loss of oil and natural gas production and damage to formations can occur in the oilfield services industry. If a serious accident were to occur at a location where our equipment and services are being used, it could result in our being named as a defendant in lawsuits asserting large claims.

Because our business involves the transportation of heavy equipment and materials, we may also experience traffic accidents which may result in spills, property damage and personal injury.

Despite what we view as our strong safety record and our efforts to maintain safety standards, we from time to time have suffered accidents in the past and anticipate that we could experience accidents in the future. In addition to the property damage, personal injury and other losses from these accidents, the frequency and severity of these incidents affect our operating costs and insurability and our relationships with customers, employees, regulatory agencies and other parties. Any significant increase in the frequency or severity of these incidents, or the general level of compensation awards, could adversely affect the cost of, or our ability to obtain, workers' compensation and other forms of insurance, and could have other material adverse effects on our financial condition and results of operations.

We maintain commercial general liability, workers' compensation, business auto, commercial property, excess liability, and directors and officers insurance policies providing coverages of risks and amounts that we believe to be customary in our industry. Further, we have pollution legal liability coverage for our business entities, which would cover, among other things, third party liability and costs of clean-up relating to environmental contamination on our premises, while our equipment is in transit and while on our customers' job site. With respect to our hydraulic fracturing operations, coverage would be available under our pollution legal liability policy for any surface environmental clean-up and liability to third parties arising from any surface contamination. We also have certain specific coverages for some of our business segments, including for our hydraulic fracturing services.

Although we maintain insurance coverage of types and amounts that we believe to be customary in the industry, we are not fully insured against all risks, either because insurance is not available or because of the high premium costs relative to perceived risk. Further, insurance rates have in the past been subject to wide fluctuation

and changes in coverage could result in less coverage, increases in cost or higher deductibles and retentions. Liabilities for which we are not insured, or which exceed the policy limits of our applicable insurance, could have a material adverse effect on us. See "Risk Factors" for a description of certain risks associated with our insurance policies.

## Environmental and occupational health and safety regulations

#### Environmental, health and safety matters and regulation

Our operations are subject to stringent laws and regulations governing the discharge of materials into the environment or otherwise relating to environmental protection, and occupational health and safety. Numerous federal, state and local governmental agencies issue regulations that often require difficult and costly compliance measures that could carry substantial administrative, civil and criminal penalties and may result in injunctive obligations for non-compliance. These laws and regulations may, for example, restrict the types, quantities and concentrations of various substances that can be released into the environment, limit or prohibit construction or drilling activities on certain lands lying within wilderness, wetlands, ecologically or seismically sensitive areas and other protected areas, or require action to prevent or remediate pollution from current or former operations. Moreover, it is not uncommon for neighboring landowners and other third parties to file claims for personal injury and property damage allegedly caused by the release of hazardous substances, hydrocarbons or other waste products into the environment. Changes in environmental, health and safety laws and regulations occur frequently, and any changes that result in more stringent and costly requirements could materially adversely affect our operations and financial position. For example, following the election of President Biden and Democratic control in both houses of Congress, it is possible that our operations may be subject to greater environmental, health and safety restrictions, particularly with regards to hydraulic fracturing, permitting and GHG emissions. We have not experienced any material adverse effect from compliance with current requirements; however, we cannot guarantee this will always be the case.

Below is an overview of some of the more significant environmental, health and safety requirements with which we must comply. Our customers' operations are subject to similar laws and regulations. Any material adverse effect of these laws and regulations on our customers' operations and financial position may also have an indirect material adverse effect on our operations and financial position.

Waste Handling. We handle, transport, store and dispose of wastes that are subject to the Resource Conservation and Recovery Act (RCRA") and comparable state laws and regulations, which affect our activities by imposing requirements regarding the generation, transportation, treatment, storage, disposal and cleanup of hazardous and non-hazardous wastes. With federal approval, the individual states administer some or all of the provisions of RCRA, sometimes in conjunction with their own, more stringent requirements. Although certain petroleum production wastes are exempt from regulation as hazardous wastes under RCRA, such wastes may constitute "solid wastes" that are subject to the less stringent requirements of non-hazardous waste provisions.

Administrative, civil and criminal penalties can be imposed for failure to comply with waste handling requirements. Moreover, the EPA or state or local governments may adopt more stringent requirements for the handling of non-hazardous wastes or re-categorize some non-hazardous wastes as hazardous wastes in the future. Indeed, legislation has been proposed from time to time in U.S. Congress to re-categorize certain oil and natural gas exploration, development and production wastes as hazardous wastes. Several environmental organizations have also at times petitioned the EPA to modify existing regulations to re-categorize certain oil and natural gas exploration, development and production wastes as hazardous. Any such changes in these laws and regulations could have a material adverse effect on our capital expenditures and operating expenses. Although we do not believe the current costs of managing our wastes, as presently classified, to be significant, any legislative or regulatory reclassification of oil and natural gas E&P wastes could increase our costs to manage and dispose of such wastes.

Remediation of Hazardous Substances. The Comprehensive Environmental Response, Compensation and Liability Act ('CERCLA" or "Superfund") and analogous state laws generally impose liability without regard to fault or legality of the original conduct, on classes of persons who are considered to be responsible for the release of a hazardous substance into the environment. These persons include the current owner or operator of a contaminated facility, a former owner or operator of the facility at the time of contamination and those persons that disposed or arranged for the disposal of the hazardous substance at the facility. Liability for the costs of removing or remediating previously disposed wastes or contamination, damages to natural resources, the costs of conducting certain health studies, amongst other things, is strict and joint and several. In the course of our operations, we use materials that, if released, would be subject to regulation under CERCLA and comparable state laws. Therefore, governmental agencies or third parties may seek to hold us responsible under CERCLA and comparable state statutes for all or part of the costs to clean up sites at which such hazardous substances have been released. Such liability could require us to make significant expenditures to investigate and remediate such contamination or attain and maintain compliance with such laws, and may otherwise have a material adverse effect on our results of operations, competitive position or financial condition.

**NORM**. In the course of our operations, some of our equipment may be exposed to naturally occurring radioactive materials (**NORM**") associated with oil and gas deposits and, accordingly, may result in the generation of wastes and other materials containing NORM. NORM exhibiting levels of radiation in excess of established state standards are subject to special handling and disposal requirements, and any storage vessels, piping and work area affected by NORM may be subject to remediation or restoration requirements.

Water Discharges. The Clean Water Act ("CWA"), SDWA, Oil Pollution Act ("OPA") and analogous state laws and regulations impose restrictions and strict controls regarding the unauthorized discharge of pollutants, including produced waters and other oil and gas wastes, into regulated waters. The discharge of pollutants into regulated waters is prohibited, except in accordance with the terms of a permit issued by the EPA or the state. The discharge of dredge and fill material into regulated waters, including wetlands, is also prohibited, unless authorized by a permit issued by the U.S. Army Corps of Engineers (the "Corps"). The scope of these regulated waters has been subject to controversy in recent years. In September 2015, the EPA and the Corps issued new rules revising the definition of "waters of the United States" (the "Clean Water Rule"), but in April 2020, the EPA and the Corps replaced the Clean Water Rule with the Navigable Waters Protection Rule, which narrows the definition of "waters of the United States" to four categories of jurisdictional waters and includes twelve categories of exclusions, including groundwater. However, these rulemakings are currently subject to litigation. In August 2021, a federal judge in the District of Arizona struck down the Navigable Waters Protection Rule, and the Biden administration and the Corps have announced that they have stopped enforcing the Navigable Waters Protection Rule nationwide, and that they are reverting back to the 1986 definition of "waters of the United States." In December 2021, the EPA and Corps published the first of two proposed rulemakings, including a definition largely in keeping with a broader pre-2015 definition and related regulatory guidance and case law. A second proposed rulemaking expanding on this definition is expected later in 2022. In January 2022, the Supreme Court agreed to hear a case regarding the jurisdictional reach of "water of the United States." To the extent any new rules or court decisions expand the scope of the CWA's juri

Noncompliance with the CWA, SDWA, OPA, or other laws or regulations relating to water discharges may result in substantial administrative, civil and criminal penalties, as well as injunctive obligations, for us or our customers. The process for obtaining permits also has the potential to delay operations. Additionally, spill prevention, control and countermeasure plan requirements require appropriate containment berms and similar structures to help prevent the contamination of regulated waters.

Air Emissions. The CAA and comparable state laws and regulations, regulate emissions of various air pollutants through the issuance of permits and the imposition of other emissions control requirements. The EPA has developed, and continues to develop, stringent regulations governing emissions of air pollutants from

specified sources. New facilities may be required to obtain permits before work can begin, and existing facilities may be required to obtain additional permits and incur capital costs in order to remain in compliance. For example, EPA has established emission control requirements for crude oil and natural gas production and processing operations and established criteria for aggregating multiple small surface sites into a single source for air quality permitting purposes, which could cause small facilities, on an aggregate basis, to be deemed a major source subject to more stringent air permitting processes and requirements. These and other laws and regulations may increase the costs of compliance for some facilities where we operate. Obtaining or renewing permits also has the potential to delay the development of oil and natural gas projects.

Climate Change. Climate change continues to attract considerable public and scientific attention. As a result, numerous proposals have been made and are likely to continue to be made at the international, national, regional and state levels of government to monitor and limit emissions of carbon dioxide, methane and other GHGs. These efforts have included consideration of cap-and-trade programs, carbon taxes, GHG reporting and tracking programs and regulations that directly limit GHG emissions from certain sources.

In the United States, no comprehensive climate change legislation has been implemented at the federal level. However, President Biden has highlighted addressing climate change as a priority of his administration and has issued several executive orders addressing climate change. Moreover, following the U.S. Supreme Court finding that GHG emissions constitute a pollutant under the CAA, the EPA has adopted regulations that, among other things, establish construction and operating permit reviews for GHG emissions from certain large stationary sources, require the monitoring and annual reporting of GHG emissions from certain petroleum and natural gas system sources in the United States, and together with the DOT, set GHG emissions and fuel economy standards for vehicles in the United States. The regulation of methane from oil and gas facilities has been subject to uncertainty in recent years. The EPA previously had promulgated NSPS imposing limitations on methane emissions from sources in the oil and gas sector. Subsequently, in September 2020, the Trump Administration rescinded those methane standards and removed the transmission and storage segments from the oil and gas source category under the CAA's NSPS. However, in June 2021, President Biden signed a resolution passed by the U.S. Congress under the Congressional Review Act nullifying the September 2020 rule, effectively reinstating the prior standards. In November 2021, as required by President Biden's executive order, the EPA proposed new regulations to expand NSPS requirements for oil and gas sector sources and establish comprehensive standards of performance and emission guidelines for methane and volatile organic compound emissions from existing operations in the oil and gas sector, including the exploration and production, transmission, processing, and storage segments. The EPA has announced that the agency hopes to finalize these rulemakings by the end of 2022. Once finalized, the regulations are likely to be subject to legal challenge and will also need to be incorporated into the states' implementation plans, which will need to be approved by the EPA in individual rulemakings that could also be subject to legal challenge. The reinstatement of direct regulation of methane emission for new sources and the promulgation of requirements for existing oil and gas customers could result in increased costs for our customers and consequently adversely affect demand for our services.

Separately, various states and groups of states have adopted or are considering adopting legislation, regulation or other regulatory initiatives that are focused on such areas as GHG cap and trade programs, carbon taxes, reporting and tracking programs, and restriction of emissions. For example, several states, including Pennsylvania and New Mexico, have proposed or adopted regulations restricting the emission of methane from E&P activities. At the international level, the United Nations-sponsored "Paris Agreement" requires member states to submit non-binding, individually-determined reduction goals known as Nationally Determined Contributions every five years after 2020. President Biden has recommitted the United States to the Paris Agreement and, in April 2021, announced a goal of reducing the United States' emissions by 50-52% below 2005 levels by 2030. In November 2021, President Biden released "The Long-Term Strategy of the United States: Pathways to Net-Zero Greenhouse Gas Emissions by 2050," which, among other things, explains that the U.S. and EU are co-leading the "Global Methane Pledge" that aims to cut global methane pollution at least 30% by 2030 relative to 2020 levels. The impacts of these orders, pledges, agreements, and any legislation or

regulation promulgated to fulfill the United States' commitments under the Paris Agreement, cannot be predicted at this time.

Governmental, scientific, and public concern over the threat of climate change arising from GHG emissions has resulted in increasing political risks in the United States, including climate change related pledges made by certain candidates now in public office. On January 27, 2021, President Biden issued an executive order that calls for substantial action on climate change, including, among other things, the increased use of zero-emission vehicles by the federal government, the elimination of subsidies provided to the fossil fuel industry, and increased emphasis on climate-related risks across government agencies and economic sectors. The Biden Administration has previously also issued orders suspending the issuance of new leases pending a study, for oil and gas development on federal lands. For more information, see our regulatory disclosure titled "Information About ProFrac - Regulation of Hydraulic Fracturing and Related Activities." As a result, we cannot predict the full impact of these developments or whether the Biden Administration may pursue further restrictions. Other actions that could be pursued by the Biden Administration may include the imposition of more restrictive requirements for the establishment of pipeline infrastructure or the permitting of LNG export facilities, as well as more restrictive GHG emission limitations for oil and gas facilities.

Additionally, the Securities and Exchange Commission recently proposed new rules relating to the disclosure of a range of climate-related risks. We are currently assessing this rule but at this time we cannot predict the costs of implementation or any potential adverse impacts resulting from the rule. To the extent this rule is finalized as proposed, we or our customers could incur increased costs related to the assessment and disclosure of climate-related risks. In addition, enhanced climate disclosure requirements could accelerate the trend of certain stakeholders and lenders restricting or seeking more stringent conditions with respect to their investments in certain carbon intensive sectors.

The adoption and implementation of new or more stringent international, federal or state legislation, regulations or other regulatory initiatives that impose more stringent standards for GHG emissions from the oil and natural gas sector or otherwise restrict the areas in which this sector may produce oil and natural gas or generate the GHG emissions could result in increased costs of compliance or costs of consuming, and thereby reduce demand for oil and natural gas, which could reduce demand for our services. Additionally, political, litigation and financial risks may result in our customers restricting or cancelling production activities, incurring liability for infrastructure damages as a result of climatic changes, or impairing their ability to continue to operate in an economic manner, which also could reduce the demand for our services. One or more of these developments could have a material adverse effect on our business, financial condition and results of operation.

Endangered and Threatened Species. Environmental laws such as the ESA and analogous state laws may impact exploration, development and production activities in areas where we operate. The ESA provides broad protection for species of fish, wildlife and plants that are listed as threatened or endangered. Similar protections are offered to migratory birds under the MBTA and various state analogs. The U.S. Fish & Wildlife Service ("FWS") may identify previously unidentified endangered or threatened species or may designate critical habitat and suitable habitat areas that it believes are necessary for survival of a threatened or endangered species. For example, the dunes sagebrush lizard, which is found only in the active and semi-stable shinnery oak dunes of southeastern New Mexico and adjacent portions of Texas (including areas where our customers operate), was a candidate species for listing under the ESA by the FWS for many years. As a result of a recent settlement with the environmental groups, the FWS, in July 2020, acted on a petition to list the dunes sagebrush lizard finding sufficient information to warrant a formal one-year review to consider listing the species. While the listing review is ongoing, FWS has also developed a conservation agreement that would implement certain protective practices for the species and authorize incidental take of the species resulting from certain covered activities, including exploration and development of oil and gas fields. The conservation agreement is known as a CCAA. We have joined the CCAA in an effort to mitigate potential impacts on our business of a listing of the dunes sagebrush lizard by the FWS.

On June 1, 2021, FWS also proposed to list two distinct population segments of the lesser prairie-chicken under the ESA, in response to a 2016 petition from conservation groups. Separately, on July 1, 2021, a lawsuit was filed by conservation groups to overturn a 2019 FWS decision that listing the eastern hellbender salamander under the ESA was not warranted. In October 2021, the Biden administration published two rules that reversed changes made by the Trump administration, namely to the definition of "habitat" and a policy that made it easier to exclude territory from critical habitat. On March 23, 2022, the FWS proposed a rule to redesignate the northern long-eared bat from a threatened species to an endangered species under ESA, and to remove its species-specific rule that excluded most development activities from the ESA's prohibition on taking listed species. To the extent any protections are implemented for these or any other species or habitat, it could cause us or our customers to incur additional costs or become subject to operating restrictions or operating bans in the affected areas.

Regulation of Hydraulic Fracturing and Related Activities Our hydraulic fracturing operations are a significant component of our business. Hydraulic fracturing is an important and common practice that is used to stimulate production of hydrocarbons, particularly natural gas, from tight formations, including shales. The process, which involves the injection of water, sand and chemicals under pressure into formations to fracture the surrounding rock and stimulate production, is typically regulated by state oil and natural gas commissions. However, federal agencies have asserted regulatory authority over certain aspects of the process. For example, the EPA has previously issued a series of rules under the CAA that establish new emission control requirements for certain oil and natural gas production and natural gas processing operations and associated equipment. BLM also finalized rules to impose new or more stringent standards for performing hydraulic fracturing on federal and American Indian lands. While this regulation was subsequently rescinded in December 2017, which rescission was upheld by the District Court of Northern California, litigation is ongoing. Additionally, the Biden Administration may seek to revisit these regulations. Separately, the Biden Administration may also pursue further restriction of hydraulic fracturing and other oil and gas development on federal lands. For example, on January 27, 2021, President Biden issued an executive order that, among other things, called for the elimination of fossil fuel subsidies from federal budget requests beginning in 2022 and suspended the issuance of new leases for oil and gas development on federal lands to the extent permitted by law and called for a review of existing leasing and permitting practices for such activities on federal lands (the order clarifies that it does not restrict such operations on tribal lands that the federal government merely holds in trust).

In response to President Biden's executive order, the Department of Interior issued a report recommending various changes to the federal leasing program, though many such changes would require Congressional action. However, legal challenges to this suspension are ongoing, and the District Court for the Western District of Louisiana has issued a preliminary injunction against the implementation of this suspension while such challenges are pending. The Biden Administration has appealed the injunction, but is complying with the injunction during the appeals process. Separately, the state of Louisiana, among other states, have challenged the Biden Administration's use of the social cost of carbon in its decision-making, including federal leasing decisions, resulting in an injunction from the Western District of Louisiana preventing the Biden Administration's use of the social cost of carbon. In response, the Biden Administration again halted leasing activities. However, the Fifth Circuit Court of Appeals has overturned the lower court's decision, and the plaintiff states have announced that they plan to seek review by the Supreme Court. On April 15, 2022, the Department of the Interior announced that it would again resume leasing on federal lands, though with significant changes to the program, including an 80% reduction in the number of approvals of applications for permits to drill on federal lands in 2022. As a result of the foregoing, there is significant uncertainty and increased regulatory risks and costs relating to onshore oil and gas exploration and production activities. These issues could result in decreased activity on federal land, adversely impacting demand for our services.

As a result, we cannot predict the final scope of regulations or restrictions that may apply to oil and gas operations on federal lands, nor the outcome of pending litigation. Although the executive order does not apply to

existing operations under valid leases, ProFrac cannot guarantee that further action will not be taken to curtail oil and gas development on federal lands. Any restrictions for new or existing production activities on federal land could adversely impact our customer's operations and consequently reduce demand for our services. The increase in royalties associated with leasing on federal lands, and any future increases that may occur, may adversely impact exploration and production activities on federal lands and reduce demand for our services. Further, legislation to amend the SDWA to repeal the exemption for hydraulic fracturing (except when diesel fuels are used) from the definition of "underground injection" and require federal permitting and regulatory control of hydraulic fracturing, as well as legislative proposals to require disclosure of the chemical constituents of the fluids used in the fracturing process, have previously been proposed in Congress. Several states and local jurisdictions in which we or our customers operate also have adopted or are considering adopting regulations that could restrict or prohibit hydraulic fracturing in certain circumstances, impose more stringent operating standards and/or require the disclosure of the composition of hydraulic fracturing fluids.

Federal and state governments have also investigated whether the disposal of produced water into underground injection wells has caused increased seismic activity in certain areas. In response to concerns regarding induced seismicity, regulators in some states have imposed, or are considering imposing, additional requirements in the permitting of produced water disposal wells or otherwise to assess any relationship between seismicity and the use of such wells. For example, Oklahoma has issued rules for wastewater disposal wells that impose permitting and operating restrictions and reporting requirements on disposal wells in proximity to faults and also, from time to time, has implemented plans directing certain wells where seismic incidents have occurred to restrict or suspend disposal well operations. In particular, the Oklahoma Corporation Commission's well completion seismicity guidelines for operators in the SCOOP and STACK require hydraulic fracturing operations to be suspended following earthquakes of certain magnitudes in the vicinity. In addition, the Oklahoma Corporation Commission's Oil and Gas Conservation Division has previously issued an order limiting future increases in the volume of oil and natural gas wastewater injected into the ground in an effort to reduce the number of earthquakes in the state. The Texas Railroad Commission has adopted similar rules.

If new laws or regulations that significantly restrict hydraulic fracturing and related activities are adopted, such laws could make it more difficult or costly to perform fracturing to stimulate production from tight formations. In addition, if hydraulic fracturing is further regulated at the federal or state level, fracturing activities could become subject to additional permitting and financial assurance requirements, more stringent construction specifications, increased monitoring, reporting and recordkeeping obligations, plugging and abandonment requirements and also to attendant permitting delays and potential increases in costs. Such legislative changes could cause us to incur substantial compliance costs, and compliance or the consequences of any failure to comply could have a material adverse effect on our financial condition and results of operations.

OSHA Matters. The Occupational Safety and Health Act ("OSHA") and comparable state statutes regulate the protection of the health and safety of workers. In addition, the OSHA hazard communication standard requires that information be maintained about hazardous materials used or produced in operations and that this information be provided to employees, state and local government authorities and the public. In March 2016, the U.S. Occupational Safety and Health Administration issued a final rule to impose stricter standards for worker exposure to silica; our sand mining operations are subject to this rule and, moreover, the rule went into effect on June 23, 2021 for hydraulic fracturing activities. As a result, we or our customers may be required to incur additional costs associated with compliance with these standards, which costs may be material.

Mining Activities. Our sand mining operations are subject to the oversight of the U.S. Mine Safety and Health Administration ('MSHA"), which is the primary regulatory agency with jurisdiction over the commercial silica industry. MSHA regulates quarries, surface mines, underground mines, and the industrial mineral processing facilities associated with quarries and mines. MSHA administers and enforces the provisions of the Federal Mine Safety and Health Act of 1977 ("FMSHA"), as amended by the Mine Improvement and New Emergency Response Act of 2006. FMSHA imposes stringent health and safety standards on numerous aspects of our operations inclusive of mineral extraction and processing operations, transportation and transloading of

silica and delivery of silica sand to well sites. These standards include, among others, the training of personnel, operating procedures, operating and safety equipment, and other matters. As part of MSHA's oversight, its representatives must perform at least two unannounced inspections annually for each surface mining facility in its jurisdiction.

## Human capital

Our employees are a critical asset which are key to our innovative culture and overall success. We are focused on our high-performance culture through attracting, engaging, developing, retaining and rewarding top talent. We strive to enhance the economic and social well-being of our employees and the communities in which we operate. We are committed to providing a welcoming, inclusive environment for our workforce, with best-in-class training and career development opportunities to enable employees to thrive and achieve their career goals.

As of June 30, 2022, we employed 2,670 people, none of whom are represented by labor unions or subject to collective bargaining agreements.

Health and Safety. The health, safety, and well-being of our employees is of the utmost importance. We are an industry leader with a proven track record in safety with a TRIR of .42 for the year ended December 31, 2021, including our manufacturing division, compared to the industry average of .70.

We provide employees the option to participate in health and welfare plans, including medical, dental, life, accidental death and dismemberment and short-term and long-term disability insurance plans. We also offer a number of health and wellness programs, including telemedicine, health screens and fitness reimbursement as well as access to the Employee Assistance Program which provides employees and their family members access to professional providers to help navigate challenging life events 24 hours a day/365 days a year.

In response to COVID-19, we adopted enhanced safety measures and practices to protect employee health and safety and continue to follow guidelines from the Centers for Disease Control to protect our employees and minimize the risk of business disruption.

## Legal proceedings

In the ordinary course of business, we are the subject of, or party to a number of pending or threatened legal actions and administrative proceedings arising in the ordinary course of our business. While many of these matters involve inherent uncertainty, we believe that, other than as described below, the amount of the liability, if any, ultimately incurred with respect to proceedings or claims will not have a material adverse effect on our consolidated financial position as a whole or on our liquidity, capital resources or future annual results of operations.

Patterson v. FTS International Manufacturing, LLC and FTS International Services, LLC: On June 24, 2015, Joshua Patterson filed a lawsuit against the Company in the 115th Judicial District Court of Upshur County, Texas, alleging, among other things, that the Company was negligent with respect to an automobile accident in 2013. Mr. Patterson sought monetary relief of more than \$1 million. On July 19, 2018, a jury returned a verdict of approximately \$100 million, including punitive damages, against the Company. The trial court reduced the judgment on November 12, 2018, to approximately \$33 million. The Company's insurance carriers appealed and the Twelfth Court of Appeals reversed the verdict in its entirety on August 26, 2020, remanding the case for a new trial. The Company's insurance carriers are currently appealing one of the appellate findings with the Texas Supreme Court. No new trial date has been set. While the outcome of this case is uncertain, the Company has met its insurance deductible for this matter and we do not expect the ultimate resolution of this case to have a material adverse effect on our consolidated financial statements.

Lonestar Prospects, Ltd. d/b/a Vista Sand v. ProFrac Services, LLC: ProFrac Services, LLC ("ProFrac Services") entered into a Master Purchase Agreement For Products And/Or Services with Lonestar Prospects, Ltd. d/b/a Vista Sand ("Vista"), dated November 27, 2017 (the "Vista MSA"), as amended by the First Addendum to Vista MSA and the First Amendment to Vista MSA, both of which are dated June 10, 2018 (collectively, the "Vista Agreement"). Under the terms of the Vista Agreement, ProFrac Services agreed to purchase certain quantities of sand from Vista. Vista filed a complaint against ProFrac Services in the United States Bankruptcy Court for the Northern District of Texas on March 15, 2021, in which it alleges that ProFrac Services breached the terms of the Vista Agreement by failing to purchase the required amount of sand or pay for the underpurchased amounts as required by the Vista Agreement. Vista is seeking damages of approximately \$8.31 million. Vista and ProFrac Services have entered into a mutually agreed upon Scheduling Order signed by the Court on February 12, 2022. Trial docket call for this matter is currently scheduled for September 6, 2022.

When deemed necessary, we establish reserves for certain legal proceedings. The establishment of a reserve is based on an estimation process that includes the advice of legal counsel and subjective judgment of management. While management believes these reserves to be adequate, it is reasonably possible we could incur additional losses with respect to those matters in which reserves have been established.

#### ProFrac's Management and Board of Directors

The following table sets forth the names, ages and titles of ProFrac's directors and executive officers.

Name	Age	Position
Ladd Wilks	37	Chief Executive Officer
Lance Turner	42	Chief Financial Officer
Coy Randle	61	Chief Operating Officer
Robert Willette	47	Chief Legal Officer, Secretary
Matthew D. Wilks	39	Executive Chairman of the Board
Sergei Krylov	44	Director
Terry Glebocki	60	Director
Stacy Nieuwoudt	42	Director
Gerald Haddock	74	Director

## Ladd Wilks

Ladd Wilks has served as ProFrac's Chief Executive Officer since May 2016. Since February 2012, Mr. Wilks has also served as Vice President of Breckenridge Geophysical, Inc. Mr. Wilks owns a controlling interest in two private E&P companies. Mr. Wilks currently sits on the Board of Directors of Cisco Safe, the Cisco Recreation Foundation and 13 Foundation. From March 2008 to July 2011, Mr. Wilks served as VP of Logistics of FTSI. Additionally, Mr. Wilks is an executive officer at Wilks Brothers, LLC and has an extensive background with ProFrac arising from his familial connection to ProFrac's founders as the son of Farris Wilks and nephew of Dan Wilks.

## Lance Turner

Lance Turner has served as ProFrac's Chief Financial Officer since March 2022. From October 2015 to March 2022, Mr. Turner served as Chief Financial Officer and Treasurer of FTSI. Mr. Turner joined FTSI in April 2014 as Director of Finance and was promoted to Vice President of Finance of FTSI in January 2015. Prior to that, Mr. Turner spent approximately 11 years with Ernst & Young LLP, with the majority of that time in its transactions services group coordinating and advising clients on buy side and sell side transactions in various industries. He earned a Bachelor of Business Administration and a Master of Professional Accounting from the University of Texas at Austin and is a Certified Public Accountant in the state of Texas.

# Coy Randle

Coy Randle joined ProFrac in May 2018 and has served as ProFrac's Chief Operating Officer since October 2018. Mr. Randle has over 39 years' experience in the energy industry. Prior to joining ProFrac, Mr. Randle provided technical consulting services for Nolan Transportation Group. Mr. Randle served as President and Chief Operating Officer of FTSI from March 2010 to October 2015 and as Senior Vice President of Operations from January 2008 to March 2010.

# Robert Willette

Robert Willette has served as ProFrac's Chief Legal Officer and Secretary since November 2021. Since October 2020, Mr. Willette has served as Assistant General Counsel of Wilks Brothers, LLC. From August 2017 to October 2020, Mr. Willette served as Senior Vice President, General Counsel, Corporate Secretary, Chief Compliance and ESG Officer of Carbo Ceramics Inc. Prior to that, Mr. Willette served as General Counsel and Corporate Secretary for Texon L.P., which transports and markets crude oil, natural gas and natural gas liquids. Mr. Willette holds a B.S., an M.B.A., and a J.D. from the University of Kansas.

#### Matthew D. Wilks

Matthew D. Wilks has served as Executive Chairman of the ProFrac Board since August 2021 and has served as ProFrac's President since October 2018. Previously, Mr. Wilks served as ProFrac's Chief Financial Officer from May 2017 to August 2021. Mr. Wilks also has served as Vice President of Investments for Wilks Brothers, LLC since January 2012. From 2010 to 2012, Mr. Wilks served as Vice President of Logistics for FTSI. Additionally, Mr. Wilks served as a member of the board of directors of Approach Resources, Inc., an E&P company focused on the exploration, development and production of unconventional oil and gas resources in the United States. Mr. Wilks' background in numerous roles specific to ProFrac and his familial connection to ProFrac's founders as the son of Dan Wilks and nephew of Farris Wilks, allow him to engage in board deliberations with valuable insight and experience.

# Sergei Krylov

Sergei Krylov has served on the ProFrac Board since May 2022. Mr. Krylov has been in the energy industry for more than 20 years, both as an investment banker and as an executive officer. Currently, Mr. Krylov serves as Investment Partner and Chief Financial Officer of Wilks Brothers, LLC (an entity controlled by Dan Wilks and Farris Wilks). From 2014 to 2020, Mr. Krylov served as an executive at Approach Resources Inc., a NASDAQ listed exploration and production company focused on Permian basin, initially as Executive Vice President and Chief Financial Officer and subsequently as President and Chief Executive Officer. From 2000 to 2013, Mr. Krylov worked at J.P. Morgan Securities LLC in the Energy Investment Banking group in New York and Houston, where he most recently served as Managing Director. During his career Mr. Krylov has executed numerous mergers and acquisitions, capital markets offerings and financial restructurings. Mr. Krylov holds a B.B.A. in finance from Pace University.

ProFrac believes that Mr. Krylov's skills and experience, particularly his financial expertise and understanding of capital markets, knowledge of the oil and gas industry and his public company executive experience make him well qualified to serve as a director.

#### Terry Glebocki

Terry Glebocki has served on the ProFrac Board since May 2022. Ms. Glebocki served as Chief Executive Officer of Ocean Casino from December 2019 through October 2021. From February 2019 to December 2019, Ms. Glebocki served as Chief Financial Officer of Ocean Casino. Prior to that, Ms. Glebocki served as Executive Vice President, Chief Financial Officer, and Treasurer of Tropicana Entertainment Inc. from April 2015 to October 2018. Ms. Glebocki has also served as Vice President of Finance and Chief Financial Officer of Revel Entertainment Group LLC from June 2007 to April 2015. Ms. Glebocki holds a B.S. from Lehigh University.

ProFrac believes that Ms. Glebocki's leadership skills and experience, including serving as an executive officer for several companies, will bring valuable insights to the ProFrac Board and qualifies her to as a director.

## Stacy Nieuwoudt

Stacy Nieuwoudt has served on the ProFrac Board since May 2022. Ms. Nieuwoudt is currently a private investor. From November 2017 to November 2019, Ms. Nieuwoudt served as a Senior Energy and Industrials Analyst at Aptigon Capital, a Citadel Company. Prior to that, Ms. Nieuwoudt served as an Energy Equities Analyst at Surveyor Capital, a Citadel Company, from September 2010 to November 2017. Since January 2021, Ms. Nieuwoudt has served as a director of Independence Contract Drilling. Ms. Nieuwoudt holds a B.A. from Rice University.

ProFrac believe that Ms. Nieuwoudt's skills and experience, particularly her specialized knowledge, background and expertise in the energy industry, combined with her understanding of capital markets and career long experience covering the energy industry qualify her to serve on the ProFrac Board.

#### Gerald Haddock

Gerald W. Haddock has served on the ProFrac Board since May 2022. Mr. Haddock founded Haddock Enterprises, LLC in 2000 and has served as its President since such time. Mr. Haddock formerly served as President and Chief Operating Officer of Crescent Real Estate Equities Company from 1994 to 1999, including serving as CEO from 1996 to 1999. He currently serves as a director of Meritage Homes Corporation and Union Acquisition Corp. II, and previously served as a director of Union Acquisition Corp. Mr. Haddock has served on the board of trustees for Baylor College of Medicine in Houston and as a member of various other committees for Baylor University and Baylor College of Medicine. Those committees included the Executive Investment Committee at Baylor University and Baylor College of Medicine. Additionally, he has served on the board of the M.D. Anderson Proton Therapy Education and Research Foundation, the CEELI Institute and the Johnny Unitas Golden Arm Educational Foundation. Mr. Haddock holds a B.B.A and J.D. from Baylor University. He also holds an L.L.M in Taxation from New York University School of Law and an M.B.A. from Dallas Baptist University.

ProFrac believe that Mr. Haddock's significant and extensive experience advising multiple companies across a breadth of industries, combined with his understanding and background in business will bring proven leadership and business and industry acumen to the ProFrac Board and us and make him strongly qualified to serve as a director.

# Family relationships

Ladd Wilks and Matthew D. Wilks are first cousins and the sons of ProFrac's founders and principal shareholders, Farris and Dan Wilks, respectively.

### Status as a controlled company

Because the ProFrac Controlling Stockholders own 26,666,228 shares of ProFrac Class A Common Stock and 97,447,865 ProFrac LLC Units (and an equal number of shares of ProFrac Class B Common Stock), representing approximately 88.6% of the voting power of ProFrac, ProFrac is a controlled company under Sarbanes-Oxley and rules of Nasdaq. A controlled company is not required to have a majority of independent directors or to maintain an independent compensation or nominating and governance committee. As a controlled company, ProFrac remains subject to rules of Sarbanes-Oxley and Nasdaq that require ProFrac to have an audit committee composed entirely of independent directors. Under these rules, ProFrac must have at least one independent director on ProFrac's audit committee by the date that ProFrac's Class A Common Stock is listed on Nasdaq, at least two independent directors on ProFrac's audit committee within 90 days of the listing date, and at least three independent directors on ProFrac's audit committee within one year of the listing date. ProFrac had three independent directors upon the closing of the IPO.

If at any time we cease to be a controlled company, we will take all action necessary to comply with Sarbanes-Oxley and rules of Nasdaq, including by appointing a majority of independent directors to our board of directors and ensuring we have a compensation committee and nominating and governance committee composed of independent directors, subject to a permitted "phase-in" period.

# **Board of directors and committees**

The ProFrac Board consists of five directors, three of whom satisfy the independence requirements of the Exchange Act and Nasdaq listing standards. The ProFrac Board has determined that Terry Glebocki, Stacy Nieuwoudt and Gerald Haddock are independent within the meaning of the Nasdaq listing standards currently in effect and Rule 10A-3 of the Exchange Act.

ProFrac's certificate of incorporation provides that the terms of office of the directors are from the time of their election until the next annual meeting of shareholders or until their successors are duly elected and qualified.

ProFrac's certificate of incorporation provides that the authorized number of initial directors will be five and thereafter that the exact number of directors will be fixed from time to time exclusively by the ProFrac Board pursuant to a resolution adopted by a majority of the whole board. In addition, ProFrac's certificate of incorporation and bylaws provide that, in general, vacancies on the board may be filled by a majority of directors in office, although less than a quorum.

#### Audit committee

The ProFrac Board has established an audit committee whose functions include the following:

- assist the ProFrac Board in its oversight responsibilities regarding the integrity of ProFrac's financial statements, ProFrac's compliance
  with legal and regulatory requirements, ProFrac's independent accountant's qualifications and independence, and ProFrac's
  accounting and financial reporting processes and the audits of ProFrac's financial statements;
- · prepare the report required by the SEC for inclusion in ProFrac's annual proxy or information statement;
- approve audit and non-audit services to be performed by ProFrac's independent accountants;
- · perform such other functions as the ProFrac Board may from time to time assign to the audit committee.

The specific functions and responsibilities of the audit committee are set forth in the charter of ProFrac's audit committee. ProFrac's audit committee consists of Terry Glebocki, Stacy Nieuwoudt and Gerald Haddock, all of whom satisfy the independence requirements of the Exchange Act and Nasdaq listing standards, all of whom qualify as audit committee financial experts as defined under these rules and listing standards, and all of whom satisfy the financial literacy standards for audit committee members under the Exchange Act and Nasdaq listing standards. Our audit committee is composed of three members that satisfy the independence requirements of the Exchange Act and Nasdaq listing standards.

Pursuant to ProFrac's bylaws, the ProFrac Board may, from time to time, establish other committees to facilitate the management of ProFrac's business and operations. Because ProFrac is considered to be controlled by the ProFrac Controlling Stockholders under Nasdaq rules, ProFrac is eligible for exemptions from provisions of these rules requiring a majority of independent directors, nominating and corporate governance and compensation committees composed entirely of independent directors and written charters addressing specified matters. ProFrac may elect to take advantage of these exemptions. In the event that ProFrac ceases to be a controlled company within the meaning of these rules, ProFrac will be required to comply with these provisions after the specified transition periods.

## Compensation committee

The ProFrac Board has established a compensation committee which is comprised of Terry Glebocki, Stacy Nieuwoudt and Gerald Haddock. This committee establishes salaries, incentives and other forms of compensation for officers and directors. The compensation committee also administers ProFrac's 2022 Long Term Incentive Plan (the "2022 Plan" or "long term incentive plan").

# Compensation committee interlocks and insider participation

None of ProFrac's executive officers serve on the ProFrac Board or compensation committee of a company that has an executive officer that serves on the ProFrac Board or ProFrac's compensation committee. No member of the ProFrac Board is an executive officer of a company in which one of ProFrac's executive officers serves as a member of the board of directors or compensation committee of that company.

### Board role in risk oversight

ProFrac's corporate governance guidelines provide that the ProFrac Board is responsible for reviewing the process for assessing the major risks facing ProFrac and the options for their mitigation. This responsibility is largely satisfied by ProFrac's audit committee, which is responsible for reviewing and discussing with management and ProFrac's independent registered public accounting firm ProFrac's major risk exposures and the policies management has implemented to monitor such exposures, including ProFrac's financial risk exposures and risk management policies.

### **Executive compensation**

ProFrac is currently considered an "emerging growth company," within the meaning of the Securities Act, for purposes of the SEC's executive compensation disclosure rules. In accordance with such rules, ProFrac is required to provide a Summary Compensation Table and an Outstanding Equity Awards at Fiscal Year End Table, as well as limited narrative disclosures regarding executive compensation for ProFrac's last completed fiscal year. Further, ProFrac's reporting obligations extend only to ProFrac's "named executive officers," who are the individuals who served as ProFrac's principal executive officer, ProFrac's next two other most highly compensated officers at the end of the last completed fiscal year and up to two additional individuals who would have been considered one of ProFrac's next two most highly compensated officers except that such individuals did not serve as executive officers at the end of the last completed fiscal year. Accordingly, ProFrac's "named executive officers" are:

- · Ladd Wilks, Chief Executive Officer;
- Brian Uhlmer, former Chief Financial Officer<sup>(1)</sup>;
- · Coy Randle, Chief Operating Officer; and
- · Matthew D. Wilks, Executive Chairman of the Board.
- (1) Mr. Uhlmer's employment with ProFrac terminated on December 30, 2021.

### Summary compensation table

The following table summarizes the compensation awarded to, earned by or paid to ProFrac's named executive officers for the fiscal years ended December 31, 2020 and December 31, 2021.

				Option	All other	
Name and Principal Position	Year	Salary (\$)	Bonus (\$)(1)	awards (\$)	compensation (\$)(3)	Total (\$)
Ladd Wilks	2021	261,538			22,185	283,723
Chief Executive Officer	2020	230,798	_	_	26,076	256,874
Brian Uhlmer	2021	293,885	39,336	_	6,047	339,268
Former Chief Financial Officer	2020	271,712	50,000	_	9,000	330,712
Coy Randle	2021	307,692	29,836	_	19,742	357,270
Chief Operating Officer	2020	289,635	_	_ (2)	6,142	295,776
Matthew D. Wilks	2021	248,378	_	_	_	248,378
Executive Chairman of the Board						
Robert Willette	2021	11,555	50,000			61,555
Chief Legal Officer						

<sup>(1)</sup> The amounts in this column reflect discretionary bonuses earned by Mr. Uhlmer, Mr. Randle and Mr. Willette during the 2021 fiscal year and by Mr. Uhlmer during the 2020 fiscal year. No other named executive officer earned a discretionary bonus during the 2021 or 2020 fiscal years.

<sup>(2)</sup> For Mr. Randle the aggregate grant date fair value of incentive units (as defined below) in ProFrac LLC granted during fiscal year 2020 was \$0. The incentive units are intended to constitute "profits interests" and

represent actual (non-voting) equity interests that have no liquidation value for U.S. federal income tax purposes on the date of grant but are designed to gain value only after the underlying assets have realized a certain level of growth and return to those persons who hold certain other classes of equity. ProFrac believes that, despite the fact that the incentive units do not require the payment of an exercise price, these awards are most similar economically to stock options and, as such, they are properly classified as "options" for purposes of the SEC's executive compensation disclosure rules under the definition provided in Item 402(m)(5)(i) of Regulation S-K since these awards have "option-like features." For more information on the incentive units, see the included under the headings "Outstanding Equity Awards at Fiscal Year-End" and "—Narrative Disclosure to Outstanding Equity Awards at 2021 Fiscal Year-End" below.

(3) Amounts in this column reflect for the 2021 fiscal year: (a) the incremental cost associated with personal use of a company vehicle by each of Messrs. Wilks and Randle equal to \$22,185 and \$5,914, respectively, and (b) the incremental cost associated with the use of corporate housing provided to each of Mr. Randle and Mr. Uhlmer.

### Narrative to summary compensation table

#### Employment agreements

Mr. Randle entered into an employment agreement with ProFrac Services on May 22, 2018 (the **Randle Agreement**"). The Randle Agreement provided for an initial term of three years (the "**Initial Term**"), and automatically renews for successive one-year periods thereafter until terminated in accordance with its terms. ProFrac has not entered into employment agreements with either of Messrs. Ladd Wilks or Matthew Wilks. The narrative below summarizes the payments and benefits that each named executive officer, with the exception of Mr. Uhlmer, is currently eligible to receive on an annual basis, whether under the Randle Agreement or otherwise, as applicable. The description below does not purport to be a complete description of all of the terms of the Randle Agreement and is qualified in its entirety by reference to the Randle Agreement, which is attached as an exhibit to this registration statement of which this proxy statement/information statement/prospectus forms a part. In the event that ProFrac enters into employment agreements with Messrs. Ladd Wilks or Matthew Wilks, such agreements will be filed and disclosed in accordance with the SEC's disclosure rules. For Mr. Uhlmer, the narrative below summarizes the payments and benefits he was entitled to receive during fiscal year 2021 prior to his termination of employment.

In addition to the payments and benefits described below, Mr. Randle is entitled to additional payments and benefits under the Randle Agreement in the event of a termination of his employment or in connection with the consummation of certain transactions by ProFrac. For more information on these amounts, see "—Additional Narrative Disclosure—Potential Payments Upon Termination or Change in Control" below.

#### Base salary

Each named executive officer's base salary is a fixed component of annual compensation for performing specific job duties and functions. The annual rate of base salary for each named executive officer was established at levels commensurate with historical compensation, with any adjustments deemed necessary to attract and retain individuals with superior talent appropriate relative to their expertise and experience.

### Annual bonus

ProFrac has historically used cash incentive awards to motivate and reward ProFrac's named executive officers. ProFrac does not maintain a formal annual cash incentive award plan; instead, such awards are determined on a discretionary basis. Under the Randle Agreement, the discretionary bonus for Mr. Randle is based on achievement of certain levels of revenue and profit for a given fiscal year.

For 2021, Mr. Uhlmer, Mr. Randle and Mr. Willette were the only named executive officers who earned an annual cash incentive award.

Following the completion of the IPO, we granted discretionary, short term cash incentive awards under the 2022 Plan to certain of our employees, including each of the named executive officers, in order to provide annual bonus awards for the 2022 calendar year. Under the terms of these awards, the named executive officers are entitled to receive, subject to their continued employment through the applicable payment date, a target bonus equal to 80% of each named executive officer's respective base salary. This target amount was determined for each named executive officer based on, and is intended to reward, both individual and Company performance.

## Equity compensation

For all periods prior to the completion of the ProFrac IPO, ProFrac did not maintain a formal equity compensation program, but historically used long term equity incentive awards on a discretionary basis to motivate and reward certain of ProFrac's executives. During 2020, Mr. Randle received an award of incentive units in ProFrac LLC. Such incentive units were converted prior to the completion of the ProFrac IPO. For more information on the incentive units, see the "Outstanding Equity Awards at Fiscal Year-End" table and "—Narrative Disclosure to Outstanding Equity Awards at 2021 Fiscal Year-End" below. ProFrac did not grant any equity-based compensation awards during 2021.

In connection with the ProFrac IPO, May 2022, ProFrac adopted the 2022 Plan (as defined below) in which certain of ProFrac's employees, including ProFrac's named executive officers, became eligible to participate. For more information on the 2022 Plan, see "—Additional Narrative Disclosure—2022 Long Term Incentive Plan" below.

In connection with and following the completion of our IPO, our named executive officers received an annual equity award in the form of restricted stock unit ("RSU") awards. The named executive officers received RSU awards in the following amounts: 55,555 RSUs for each of Mr. Ladd Wilks and Mr. Randall, 34,496 RSUs for Mr. Matthew Wilks and 41,666 RSUs for Mr. Willette. Under the terms of these RSU awards, 100% of the RSUs vest on the first anniversary of the closing of the IPO, provided the named executive officers remain continuously employed by the Company from the date of the grant through such vesting date.

In addition to the annual RSU award described above, Mr. Willette received an additional RSU award in recognition of his contributions to the completion of the IPO. Under the terms of such award, Mr. Willette received 7,882 RSUs which will vest on the first anniversary of the completion of the IPO, subject to Mr. Willette's continuous employment with the Company through such vesting date.

## Stock-based Compensation Related to Deemed Contributions

In connection with ProFrac's IPO, ProFrac's principal shareholders, Farris Wilks ("Farris") and Dan Wilks ("Dan") (together with certain family members, family trusts, or entities they control), sold PFH Units representing approximately 1% of the equity interest in ProFrac LLC to an entity controlled by our Chief Executive Officer, Ladd Wilks ("Ladd"), and our Executive Chairman, Matt Wilks ("Matt"), respectively. These equity interests in ProFrac LLC entitled each of Ladd and Matt to 1,220,978 shares of Class B Common Stock in ProFrac. These units were sold in exchange for promissory notes. While some of the documentation relating to these transfers was subject to completion, ProFrac concluded that both transactions were consummated in connection with the company's IPO and, for accounting purposes, should be treated in accordance with ASC Topic 718, Compensation — Stock Compensation, as deemed contributions to ProFrac by Farris and Dan and grants of stock-based compensation to Ladd and Matt by ProFrac similar to stock options.

Also in connection with ProFrac's IPO, Farris engaged in estate planning that may result, subject to other terms and conditions, in additional shares being transferred by Farris to Ladd if ProFrac's total market capitalization increases to certain target levels within the next five years, which resulted in a performance award being deemed granted by ProFrac to Ladd. ProFrac concluded that this arrangement should be treated, for accounting purposes, in accordance with ASC Topic 718, Compensation — Stock Compensation, as a deemed contribution to ProFrac by a related party and the grant of stock compensation with market conditions to Ladd by ProFrac.

### Other elements of compensation

Each named executive officer is entitled to vacation and paid time off in accordance with the terms and conditions of the applicable plan or policy. Subject to the terms of any applicable plans, policies or programs, each named executive officer is entitled to receive such employee benefits, including any and all medical, dental, vision, life and accidental death and disability, short-term and long-term disability, and voluntary life and disability benefits available to employees generally, and such other benefits as ProFrac may from time to time establish for the named executive officers. Each named executive officer is reimbursed by ProFrac for all ordinary and reasonable expenses incurred in the course of the performance of employment services subject to the terms of ProFrac's reimbursement policy. ProFrac currently maintains a retirement plan intended to provide benefits under section 401(k) of the Code (the "401(k) Plan") in which employees, including the named executive officers, are allowed to contribute portions of their base compensation to a tax-qualified retirement account. See "—Additional Narrative Disclosure—Retirement Benefits" for more information.

### Outstanding equity awards at fiscal year-end

The following table reflects information regarding outstanding equity-based awards held by each named executive officer as of December 31, 2021.

		Option awards <sup>(1)</sup>					
	Number of securities	Number of securities		<u> </u>			
	underlying	underlying					
	unexercised options	unexercised options	Option exercise	Option expiration			
Name Ladd Wilks	(#) Exercisable <sup>(2)</sup>	(#) Unexercisable <sup>(2)</sup>	price (\$) <sup>(3)</sup>	date <sup>(2)</sup>			
Ladd Wilks	_	_	_	_			
Brian Uhlmer	_	_	_	_			
Coy Randle	6	4	N/A	N/A			
Matthew D. Wilks	_	_	_	_			

- (1) The award described herein was issued to Mr. Randle by ProFrac LLC.
- (2) On May 28, 2020, Mr. Randle received a grant of Class B Units in ProFrac LLC (the 'Incentive units'). The number of incentive units reflected in the table represents the number of Class B Units of ProFrac LLC owned by Mr. Randle, and not a number of shares of ProFrac Common Stock. The incentive units are structured as profits interests for tax purposes. Profits interests such as the incentive units do not require the payment of an exercise price nor do they have an expiration date; instead, they only entitle the holder thereof to receive value if and to the extent the underlying security appreciates in value following the grant of the award. Because of this appreciation feature, ProFrac believes profits interest awards are economically similar to stock options or stock appreciation rights for purposes of the SEC disclosure rules. Awards reflected as "Unexercisable" are incentive units that have not yet vested. Awards reflected as "Exercisable" are incentive units that have not yet received payment in respect thereof. The incentive units were vested with respect to 40% of the award on the date of grant, and vest with respect to an additional 20% of the award on each of the first three anniversaries of the grant date, in each case subject to Mr. Randle's continued employment with us through the applicable vesting date. For a more detailed description, please see "—Narrative Disclosure to Outstanding Equity Awards at 2021 Fiscal Year-End" below.
- (3) The incentive units do not have an "exercise price" in the same sense that a true stock option award would have an exercise price. Instead, the incentive unit award has a "participation threshold" associated with the award. Each incentive unit will entitle the holder to receive distributions only if the aggregate distributions made by ProFrac LLC in respect of each Class A Unit of ProFrac LLC issued and outstanding on or prior to date of the grant of the incentive units exceeds the participation threshold. The thresholds are set at the time of grant and typically represents the estimated fair value of a common unit or a multiple of fair value on the date of grant.

### Narrative disclosure to outstanding equity awards at 2021 fiscal year-end

On May 28, 2020, Mr. Randle received a grant of incentive units from ProFrac LLC pursuant to the ProFrac LLC Agreement and a Class B Unit Award Agreement. The incentive units are intended to constitute "profits interests" and represent actual (non-voting) equity interests that have no liquidation value for U.S. federal income tax purposes on the date of grant but are designed to gain value only after the underlying assets realize a certain level of growth and return to those persons who hold certain other classes of equity. The incentive units were vested with respect to 40% of the incentive unit award on the date of grant, and vest with respect to an additional 20% of the award on each of the first three anniversaries of the grant date, subject to Mr. Randle's continued employment with us through each applicable vesting date. As such, Mr. Randle's incentive units are currently vested with respect to 60% of the total award, with vesting of an additional 20% of the award to occur on each of May 28, 2022 and May 28, 2023.

The incentive units held by Mr. Randle fully vested and were converted into ProFrac LLC Units as described above under 'Information About ProFrac—Corporate Reorganization."

## Additional narrative disclosure

### Retirement benefits

ProFrac currently maintains a 401(k) Plan under which employees, including the named executive officers, are allowed to contribute portions of their base compensation to a tax-qualified retirement account. ProFrac provides matching contributions to all employees who have completed one year of service with ProFrac in amounts equal to 100% of the employee's contribution up to 4% of total compensation. None of ProFrac's named executive officers received matching contributions during 2021.

### Potential payments upon termination or change in control

The following discussion describes the amounts and benefits that would have been owed to Mr. Randle under the Randle Agreement in the event of a termination of employment or the consummation of certain transactions by ProFrac as of December 31, 2021, as applicable. As of December 31, 2021, neither Mr. Ladd Wilks nor Mr. Matthew D. Wilks was entitled to any payments upon a termination of employment or change in control of ProFrac and no amounts or benefits had been earned by Mr. Uhlmer in connection with his termination of employment.

### **Involuntary Termination**

### Randle Agreement

In the event that Mr. Randle's employment were to be terminated by ProFrac without Cause or due to his resignation for Good Reason (as each term is defined below) prior to the expiration of the Initial Term in May 2021, then, in addition to payment of any earned but unpaid base salary and approved but unreimbursed expenses, the Randle Agreement provided that Mr. Randle would be entitled to receive an amount equal to his then-current annual base salary, paid in substantially equal installments in accordance with ProFrac's normal payroll practices for the 12 month period following such termination of employment. Mr. Randle's receipt of the severance payments would be conditioned on his continued compliance with his non-competition, non-solicitation and confidentiality obligations during the period that he receives the severance payments. However, after the expiration of the Initial Term in May 2021, the Randle Agreement provides that Mr. Randle is not entitled to the payments described in the preceding sentence or any other payments under the Randle Agreement upon a termination of his employment for any reason, other than payments for earned but unpaid base salary or unreimbursed business expenses. Because the Initial Term expired in May 2021, Mr. Randle would not have been entitled to any payments or benefits in connection with a termination of his employment as of December 31, 2021.

For purposes of the Randle Agreement, Cause and Good Reason are generally defined as follows:

- "Cause" means (A) material failure or refusal by Mr. Randle to satisfactorily perform his lawful duties, responsibilities, or authorities under the Randle Agreement as requested by the supervisor; (B) any act of gross negligence, willful misconduct, or fraudulent or criminal behavior by Mr. Randle in the performance of his duties, responsibilities, or authorities under the Randle Agreement, including without limitation any misappropriation of any funds or property owned by ProFrac Services or its affiliates, fraud, embezzlement, or theft; (C) any conviction of, guilty plea concerning, or entry into any deferred adjudication or similar diversion arrangement with respect to any felony or crime of moral turpitude or fraud by Mr. Randle; (D) any material violation of the Randle Agreement by Mr. Randle; (E) any breach of any applicable fiduciary duty by Mr. Randle to act exclusively and solely for the benefit of ProFrac Services or its affiliates in all undertakings concerning or relating to ProFrac Services or its affiliates; (F) any misconduct in the course and scope of employment, including without limitation dishonesty, disloyalty, disorderly conduct, insubordination, harassment of other employees or third parties, abuse of alcohol or controlled substances, or other material violations of ProFrac Service's policies, rules, or practices; or (G) any act or omission that is contrary to the best interests of ProFrac Services or its affiliates or is likely to damage the business, including without limitation the reputation of ProFrac Services or its affiliates.
- "Good Reason" means a material diminution in Mr. Randle's base salary by ProFrac Services without his consent, except that the foregoing shall not constitute Good Reason unless (i) Mr. Randle provides 60 days written notice to ProFrac Services that Good Reason exists, (ii) ProFrac Services fails to cure such circumstances for 30 days and (iii) Mr. Randle terminates his employment within 100 days of the existence of the circumstances that constitute Good Reason.

## Change in control or IPO

In the event of a change in control that results in the transfer of atleast fifty-one percent of the ownership interest of ProFrac or an initial public offering (each a "qualifying transaction"), Mr. Randle would have been entitled to receive a cash bonus payment under the Randle Agreement solely during the initial term of the Randle Agreement (which expired in May 2021) (the "Change in Control Bonus"). The Change in Control Bonus would have been equal to Mr. Randle's monthly base salary multiplied by the number of months remaining in the Initial Term, which expired in May 2021. The Change in Control Bonus would have been paid over eighteen months, with 20% paid at the time of the consummation of the qualifying transaction, 20% paid six months following the qualifying transaction, 20% paid on the one-year anniversary of the qualifying transaction and 40% paid eighteen months following the qualifying transaction, provided that Mr. Randle remains continuously employed by ProFrac or its affiliate through each such date. However, because the Initial Term expired in May 2021, Mr. Randle is no longer entitled to any Change in Control Bonus as of the date hereof under the terms of the Randle Agreement.

### 2022 Long term incentive plan

In connection with the ProFrac IPO, ProFrac adopted an omnibus equity incentive plan, the 2022 Plan, for employees, consultants and directors of ProFrac and ProFrac's affiliates who perform services for ProFrac. The following description summarizes the material features of the 2022 Plan, but does not purport to be a complete description of all of the material features of the 2022 Plan and is qualified in its entirety by reference to the full text of the 2022 Plan, which is attached as an exhibit to the registration statement of which this proxy statement/information statement/prospectus forms a part.

The 2022 Plan provides for potential grants of: (i) incentive stock options qualified as such under U.S. federal income tax laws (**Incentive options**"); (ii) stock options that do not qualify as incentive stock options ("nonstatutory options" and, together with incentive options, 'options"); (iii) stock appreciation rights ("SARs"); (iv) restricted stock awards ('restricted stock awards"); (v) restricted stock units ("restricted stock units") are the contraction of the contraction of

"RSUs"); (vi) bonus stock ("bonus stock awards"); (vii) dividend equivalents; (viii) other stock-based awards; (ix) cash awards; and (x) substitute awards (referred to collectively herein with the other awards as the "awards"). The vesting, exercise or settlement of awards may be subject to the achievement of one or more performance criteria selected by the Administrator (as defined below).

## Eligibility

ProFrac's employees, consultants and non-employee directors, and employees, consultants and non-employee directors of ProFrac's affiliates, are eligible to receive awards under the 2022 Plan.

#### Administration

The ProFrac Board, or a committee thereof (as applicable, the "Administrator"), administers the 2022 Plan pursuant to its terms and all applicable state, federal or other rules or laws. The Administrator has the power to determine to whom and when awards will be granted, determine the amount of awards (measured in cash or in shares of ProFrac Class A Common Stock), proscribe and interpret the terms and provisions of each award agreement (the terms of which may vary), accelerate the vesting or exercisability of an award, delegate duties under the 2022 Plan and execute all other responsibilities permitted or required under the 2022 Plan.

#### Securities to be offered

Subject to adjustment in the event of any distribution, recapitalization, split, merger, consolidation or similar corporate event, a number of shares of ProFrac Class A Common Stock equal to 3,120,708 is available for delivery pursuant to awards under the 2022 Plan. If an award under the 2022 Plan is forfeited, settled for cash or expires without the actual delivery of shares, any shares subject to such award will again be available for new awards under the 2022 Plan.

### Types of awards

Options. ProFrac may grant options to eligible persons including: (i) incentive options (only to ProFrac's employees or those of ProFrac's subsidiaries) which comply with Section 422 of the Code; and (ii) non-statutory options. The exercise price of each option granted under the 2022 Plan will be stated in the option agreement and may vary; however, the exercise price for an option must not be less than the fair market value per share of common stock as of the date of grant (or 110% of the fair market value for certain incentive options), nor may the option be re-priced without the prior approval of the ProFrac Stockholders. Options may be exercised as the Administrator determines, but not later than ten years from the date of grant. The Administrator will determine the methods and form of payment for the exercise price of an option (including, in the discretion of the Administrator, payment in common stock, other awards or other property) and the methods and forms in which ProFrac Common Stock will be delivered to a participant.

SARs. A SAR is the right to receive a share of ProFrac Class A Common Stock, or an amount equal to the excess of the fair market value of one share of ProFrac Class A Common Stock on the date of exercise over the grant price of the SAR, as determined by the Administrator. The exercise price of a share of ProFrac Class A Common Stock subject to the SAR shall be determined by the Administrator, but in no event shall that exercise price be less than the fair market value of the ProFrac Class A Common Stock on the date of grant. The Administrator has the discretion to determine other terms and conditions of an SAR award.

Restricted stock awards. A restricted stock award is a grant of shares of ProFrac Class A Common Stock subject to a risk of forfeiture, performance conditions, restrictions on transferability and any other restrictions imposed by the Administrator in its discretion. Restrictions may lapse at such times and under such circumstances as determined by the Administrator. Except as otherwise provided under the terms of the 2022 Plan or an award agreement, the holder of a restricted stock award will have rights as a ProFrac Stockholder, including the right to vote the shares of ProFrac Class A Common Stock subject to the restricted stock award or to receive dividends on the ProFrac Class A Common Stock subject to the restricted stock award during the

restriction period. The Administrator shall provide, in the restricted stock award agreement, whether the restricted stock will be forfeited upon certain terminations of employment. Unless otherwise determined by the Administrator, ProFrac Class A Common Stock distributed in connection with a stock split or stock dividend, and other property distributed as a dividend, will be subject to restrictions and a risk of forfeiture to the same extent as the restricted stock award with respect to which such ProFrac Class A Common Stock or other property has been distributed.

Restricted stock units. RSUs are rights to receive shares of ProFrac Class A Common Stock, cash, or a combination of both at the end of a specified period. The Administrator may subject RSUs to restrictions (which may include a risk of forfeiture) to be specified in the RSU award agreement, and those restrictions may lapse at such times determined by the Administrator. Restricted stock units may be settled by delivery of shares of ProFrac Class A Common Stock, cash equal to the fair market value of the specified number of shares of ProFrac Class A Common Stock covered by the RSUs, or any combination thereof determined by the Administrator at the date of grant or thereafter. Dividend equivalents on the specified number of shares of ProFrac Class A Common Stock covered by RSUs may be paid on a current, deferred or contingent basis, as determined by the Administrator on or following the date of grant.

Bonus stock awards. The Administrator is authorized to grant ProFrac Class A Common Stock as a bonus stock award. The Administrator will determine any terms and conditions applicable to grants of ProFrac Class A Common Stock, including performance criteria, if any, associated with a bonus stock award.

Dividend Equivalents. Dividend equivalents entitle a participant to receive cash, ProFrac Class A Common Stock, other awards or other property equal in value to dividends paid with respect to a specified number of shares of ProFrac Class A Common Stock, or other periodic payments at the discretion of the Administrator. Dividend equivalents may be granted on a free-standing basis or in connection with another award (other than a restricted stock award or a bonus stock award).

Other Stock-Based Awards. Other stock-based awards are awards denominated or payable in, valued in whole or in part by reference to, or otherwise based on or related to, the value of ProFrac Class A Common Stock.

Cash Awards. Cash awards may be granted on a free-standing basis, as an element of or a supplement to, or in lieu of any other award.

Substitute Awards. Awards may be granted in substitution or exchange for any other award granted under the 2022 Plan or under another equity incentive plan or any other right of an eligible person to receive payment from ProFrac. Awards may also be granted under the 2022 Plan in substitution for similar awards held for individuals who become participants as a result of a merger, consolidation or acquisition of another entity by or with ProFrac or one of ProFrac's affiliates.

#### Certain transactions

If any change is made to ProFrac's capitalization, such as a stock split, stock combination, stock dividend, exchange of shares or other recapitalization, merger or otherwise, which results in an increase or decrease in the number of outstanding shares of ProFrac Class A Common Stock, appropriate adjustments will be made by the Administrator in the shares subject to an award under the 2022 Plan. The Administrator has the discretion to make certain adjustments to awards in the event of a change in control, such as accelerating the vesting or exercisability of awards, requiring the surrender of an award, with or without consideration, or making any other adjustment or modification to the award that the Administrator determines is appropriate in light of such transaction.

Plan amendment and termination

The ProFrac Board may amend or terminate the 2022 Plan at any time; however, ProFrac Stockholder approval will be required for any amendment to the extent necessary to comply with applicable law or exchange listing standards. The Administrator will not have the authority, without the approval of the ProFrac Stockholders, to amend any outstanding stock option or stock appreciation right to reduce its exercise price per share. The 2022 Plan will remain in effect for a period of ten years (unless earlier terminated by the ProFrac Board).

#### Clawback

All awards under the 2022 Plan are subject to any clawback or recapture policy adopted by ProFrac, as in effect from time to time.

## **Director compensation**

No member of the ProFrac Board received compensation for their services to ProFrac in respect of the 2021 fiscal year. In connection with and following the closing of the IPO, ProFrac adopted a comprehensive director compensation program in order to attract and retain qualified non-employee directors who are essential to the future success, growth, and governance of ProFrac.

Under the director compensation program, ProFrac's non-employee directors are entitled to the following compensation:

- an annual cash retainer of \$142,500 for the 2022 calendar year and an annual cash retainer of \$95,000 for each calendar year thereafter,
- an additional cash retainer of \$20,000 for the chair of the Audit Committee,
- an additional cash retainer of \$15,000 for the chair of the Compensation Committee,
- an annual equity-based award with an aggregate fair market value of approximately \$150,000 on the grant date that will vest following a
  one-year vesting period, and
- an additional cash fee of \$1,500 for attendance at each meeting of the Board or a Committee thereof.

In addition to the annual equity-based award described above, following the completion of the ProFrac IPO, ProFrac's non-employee directors received a one-time equity-based award with an aggregate fair market value of approximately \$285,000 on the applicable grant date that will vest in substantially equal installments on each of the first, second and third anniversaries of the completion of the IPO. Both awards were granted in the form of RSU awards.

Under the director compensation program, members of the ProFrac Board who are also employees will not receive additional compensation for their service as directors. Each director is entitled to be reimbursed for reasonable and necessary travel and miscellaneous expenses incurred to attend meetings and activities of the board of directors or any of its committees. Each director is also indemnified by us for actions associated with serving as a director to the fullest extent permitted under Delaware law.

# **Corporate Reorganization**

ProFrac Holding Corp. was incorporated as a Delaware corporation on August 17, 2021. Pursuant to the Master Reorganization Agreement, ProFrac Holding Corp. and ProFrac LLC completed certain restructuring transactions (the "Corporate Reorganization"). As a result of the Corporate Reorganization, among other things, (i) ProFrac Holding Corp. became a holding company whose only material asset consists of ProFrac LLC Units, (ii) ProFrac LLC owns, directly or indirectly, all of the outstanding equity interests in the subsidiaries through which ProFrac operates its assets, and (iii) ProFrac Holding Corp. is the sole managing member of ProFrac LLC and is responsible for all operational, management and administrative decisions relating to ProFrac LLC's business and consolidates the financial results of ProFrac LLC and its subsidiaries.

In connection with the ProFrac IPO:

- all of the membership interests in ProFrac LLC held by the then-existing owners of ProFrac LLC (including the THRC FTSI Related Equity) were converted into a single class of ProFrac LLC Units;
- ProFrac Holding Corp. issued to each ProFrac LLC Unit Holder a number of shares of ProFrac Class B Common Stock equal to the number of ProFrac LLC Units held by such ProFrac LLC Unit Holder following the ProFrac IPO in exchange for a cash payment equal to the par value of such shares;
- ProFrac Holding Corp. issued 16,000,000 shares of ProFrac Class A Common Stock to purchasers in the ProFrac IPO at a public
  offering price of \$18.00 per share; and
- On June 6, 2022, an over-allotment option was exercised resulting in an additional 2,228,153 shares of ProFrac Class A Common Stock being priced at \$18.00 per share.

As of August 15, 2022, ProFrac Holding Corp. directly and indirectly owns an approximate 27.8% interest in ProFrac LLC, and the ProFrac LLC Unit Holders own an approximate 72.2% interest in ProFrac LLC and all of the outstanding shares of ProFrac Class B Common Stock. Please see "Security Ownership of Certain Beneficial Owners and Management."

Each share of ProFrac Class B Common Stock has no economic rights but entitles its holder to one vote on all matters to be voted on by ProFrac Stockholders generally. Holders of ProFrac Class A Common Stock and ProFrac Class B Common Stock will vote together as a single class on all matters presented to ProFrac Stockholders for their vote or approval, except as otherwise required by applicable law or by ProFrac's certificate of incorporation. ProFrac does not intend to list the ProFrac Class B Common Stock on any exchange.

Under the ProFrac LLC Agreement, each ProFrac LLC Unit Holder, subject to certain limitations, has the right, pursuant to the Redemption Right, to cause ProFrac LLC to acquire all or a portion of its ProFrac LLC Units for, at ProFrac LLC's election, (i) shares of ProFrac Class A Common Stock at a redemption ratio of one share of ProFrac Class A Common Stock for each ProFrac LLC Unit redeemed, subject to conversion rate adjustments for stock splits, stock dividends and reclassifications and other similar transactions, or (ii) an equivalent amount of cash. The independent members of the ProFrac Board will determine whether to pay cash in lieu of the issuance of shares of ProFrac Class A Common Stock based on facts in existence at the time of the decision, which ProFrac expects would include the relative value of the ProFrac Class A Common Stock (including trading prices for the ProFrac Class A Common Stock at the time), the cash purchase price, the availability of other sources of liquidity (such as an issuance of stock) to acquire the ProFrac LLC Units and alternative uses for such cash. Alternatively, upon the exercise of the Redemption Right, ProFrac Holding Corp. (instead of ProFrac LLC) will have the right, pursuant to the Call Right, to, for administrative convenience, acquire each tendered ProFrac LLC Unit directly from the redeeming ProFrac LLC Unit Holder for, at its election, (x) one share of ProFrac Class A Common Stock, subject to conversion rate adjustments for stock splits, stock dividends and reclassifications and other similar transactions, or (y) an equivalent amount of cash. In addition, ProFrac Holding Corp. has the right to require, upon the acquisition by ProFrac Holding Corp. of substantially all of the ProFrac LLC Units or upon a change of control of ProFrac Holding Corp., each ProFrac LLC Unit Holder to exercise its Redemption Right with respect to some or all of such unitholder's ProFrac LLC Units. In connection with any redemption of ProFrac LLC Units pursuant to the Redemption Right or acquisition of ProFrac LLC Units pursuant to the Call Right, a corresponding number of shares of ProFrac Class B Common Stock held by the relevant ProFrac LLC Unit Holder will be cancelled. See "Information About ProFrac-Certain Relationships and Related Party Transactions—ProFrac LLC Agreement."

Certain of the ProFrac LLC Unit Holders have the right, under certain circumstances, to cause ProFrac to register the offer and resale of their shares of ProFrac Class A Common Stock.

ProFrac Holding Corp.'s acquisition (or deemed acquisition for U.S. federal income tax purposes) of ProFrac LLC Units pursuant to an exercise of the Redemption Right or the Call Right is expected to result in adjustments to the tax basis of the tangible and intangible assets of ProFrac LLC and such adjustments will be allocated to ProFrac Holding Corp. These adjustments would not have been available to ProFrac Holding Corp. absent its acquisition or deemed acquisition of ProFrac LLC Units and are expected to reduce the amount of cash tax that ProFrac Holding Corp. would otherwise be required to pay in the future.

On May 17, 2022, in connection with the ProFrac IPO, ProFrac Holding Corp. entered into the Tax Receivable Agreement with the TRA Holders. The Tax Receivable Agreement generally provides for the payment by ProFrac Holding Corp. to each TRA Holder of 85% of the net cash savings, if any, in U.S. federal, state and local income tax and franchise tax (computed using simplifying assumptions to address the impact of state and local taxes) that ProFrac Holding Corp. actually realizes (or is deemed to realize in certain circumstances) as a result of (i) certain increases in tax basis that occur as a result of ProFrac Holding Corp.'s acquisition (or deemed acquisition for U.S. federal income tax purposes) of all or a portion of such TRA Holder's ProFrac LLC Units in connection with this the ProFrac IPO or pursuant to the exercise of the Redemption Right or the Call Right, and (ii) imputed interest deemed to be paid by ProFrac Holding Corp. as a result of, and additional tax basis arising from, any payments ProFrac Holding Corp. makes under the Tax Receivable Agreement. ProFrac Holding Corp. is dependent on ProFrac LLC to make distributions to ProFrac Holding Corp. in an amount sufficient to cover ProFrac Holding Corp.'s obligations under the Tax Receivable Agreement.

ProFrac will retain the benefit of the remaining 15% of any actual net cash tax savings.

For additional information regarding the Tax Receivable Agreement, see "Risk Factors—Risks Related to ProFrac's Class A Common Stock" and "Information About ProFrac—Certain Relationships and Related Party Transactions—Tax Receivable Agreement."

## Security Ownership of Certain Beneficial Owners and Management

Unless otherwise indicated in the footnotes below, the following table sets forth information regarding (i) the actual beneficial ownership of ProFrac Class A Common Stock and ProFrac Class B Common Stock as of August 15, 2022, which is prior to the consummation of the Merger, and (ii) expected beneficial ownership of ProFrac Class A Common Stock and ProFrac Class B Common Stock immediately following consummation of the Merger, by:

- each person known to ProFrac to beneficially own more than 5% of any class of ProFrac's outstanding voting securities;
- each of ProFrac's directors;
- ProFrac's named executive officers; and
- All of ProFrac's directors and executive officers as a group.

The pre-Merger beneficial ownership percentages set forth in the table below are based on 41,239,957 shares of ProFrac Class A Common Stock outstanding and 101,133,201 shares of ProFrac Class B Common Stock outstanding as of August 8, 2022, as reported in ProFrac's Quarterly Report on Form 10-Q filed with the SEC on August 15, 2022. The expected post-Merger beneficial ownership percentages set forth in the table below are based on the assumption that 54,018,996 shares of ProFrac Class A Common Stock and 101,133,201 shares of ProFrac Class B Common Stock will be outstanding immediately following consummation of the Merger.

All information with respect to beneficial ownership has been furnished by the respective 5% or more stockholders, directors or executive officers, as the case may be. Unless otherwise noted, the mailing address of each listed beneficial owner is c/o ProFrac Holding Corp., 333 Shops Boulevard, Willow Park, Texas 76087.

	Before the Merger				After the Merger							
Name of Beneficial Owner <sup>6</sup>	Number of shares of ProFrac Class A Common Stock <sup>(1)</sup>	%	Number of shares of ProFrac Class B Common Stock	%	Combined voting power	%	Number of shares of ProFrac Class A Common Stock <sup>(1)</sup>	%	Number of shares of ProFrac Class B Common Stock	%	Combined voting power <sup>(2)</sup>	%
5% Stockholders												
THRC Holdings, LP(3)	14,995,735	36.3%	49,939,536	49.4%	64,895,271	45.6%	18,267,955	33.8%	49,939,536	49.4%	68,207,491	43.9%
Farris Wilks <sup>(4)</sup>	11,711,292	28.4%	47,508,329	47.0%	59,219,621	41.6%	12,537,668	23.2%	47,508,329	46.9%	60,045,997	38.7%
Directors/Named Executive Officers												
Ladd Wilks	_		1,220,978	1.2%	1,220,978	0.9%	_	_	1,220,978	1.2%	1,220,978	0.8%
Brian Uhlmer	_		_	_	_	_	_	_	_	_	_	_
Coy Randle	_		1,215,603	1.2%	1,215,603	0.9%	_	_	1,215,603	1.2%	1,215,603	0.8%
Robert Willette	_		_	_	_	_	_	_	_	_	_	_
Matthew D. Wilks	_		1,220,978	1.2%	1,220,978	0.9%	_	_	1,220,978	1.2%	1,220,978	0.8%
Sergei Krylov	_		_	_	_	_	_	_	_	_	_	_
Terry Glebocki	_		_	_	_	_	_	_	_	_	_	_
Stacy Nieuwoudt	_		_	_	_	_	_	_	_	_	_	_
Gerald Haddock	_		_	_	_	_	_	_	_	_	_	_
All Directors and Executive Officers as a Group (9 persons)	_		3,657,558	3.6%	3,657,559	2.6%	_	_	3,657,558	3.6%	3,657,559	2.4%

- Entries in the table above marked "—" indicate percentage less than 1%.
- (1) Does not include shares underlying equity awards or shares reserved for issuance under equity incentive plans. Restricted stock unit awards covering 564,721 shares of ProFrac Class A Common Stock have been granted under the 2022 Plan as of August 15, 2022. 3,120,708 total shares of ProFrac Class A Common Stock (which includes the shares subject to restricted stock unit awards) are reserved for equity awards under the 2022 Plan as of August 15, 2022.
- (2) Represents percentage of voting power of ProFrac Class A Common Stock and ProFrac Class B Common Stock voting together as a single class. The ProFrac LLC Unit Holders hold one share of ProFrac Class B Common Stock for each ProFrac LLC Unit.
- (3) THRC Holdings is the record holder of the shares of ProFrac Common Stock reported herein. THRC Management is the general partner of THRC Holdings. Dan Wilks is the sole manager of THRC Management. Accordingly, Dan Wilks may be deemed to have or share beneficial ownership of the shares of ProFrac Common Stock held directly by THRC Holdings.
- (4) Represents shares of common stock owned by the Farris and Jo Ann Wilks 2022 Family Trust and Farris Wilks. Farris Wilks serves as a trustee of such trust and, in such capacity has voting and dispositive power over the shares of ProFrac Common Stock owned by such trust. Accordingly, Farris Wilks may be deemed to have or share beneficial ownership of the shares of ProFrac Common Stock owned by the Farris and Jo Ann Wilks 2022 Family Trust.

## Certain Relationships and Related Party Transactions

## **ProFrac LLC Agreement**

The ProFrac LLC Agreement is filed as an exhibit to the registration statement of which this proxy statement/information statement/prospectus forms a part, and the following description of the ProFrac LLC Agreement is qualified in its entirety by reference thereto.

## Redemption rights

Under the ProFrac LLC Agreement, each ProFrac LLC Unit Holder, subject to certain limitations, has the right, pursuant to the Redemption Right, to cause ProFrac LLC to acquire all or a portion of its ProFrac LLC Units for, at ProFrac LLC's election, (i) shares of ProFrac Class A Common Stock at a redemption ratio of one share of ProFrac Class A Common Stock for each ProFrac LLC Unit redeemed, subject to conversion rate adjustments for stock splits, stock dividends and reclassifications and other similar transactions, or (ii) an equivalent amount of cash. The independent members of the ProFrac Board will determine whether to pay cash in lieu of the issuance of shares of ProFrac Class A Common Stock based on facts in existence at the time of the decision, which ProFrac expects would include the relative value of the ProFrac Class A Common Stock (including trading prices for the ProFrac Class A Common Stock at the time), the cash purchase price, the availability of other sources of liquidity (such as an issuance of stock) to acquire the ProFrac LLC Units and alternative uses for such cash. Alternatively, upon the exercise of the Redemption Right, ProFrac Holding Corp. (instead of ProFrac LLC) will have the right, pursuant to the Call Right, to acquire each tendered ProFrac LLC Unit directly from the redeeming ProFrac LLC Unit Holder for, at its election, (x) one share of ProFrac Class A Common Stock, subject to conversion rate adjustments for stock splits, stock dividends and reclassifications and other similar

transactions, or (y) an equivalent amount of cash. In addition, ProFrac Holding Corp. has the right to require, upon the acquisition by ProFrac Holding Corp. of substantially all of the ProFrac LLC Units or upon a change of control of ProFrac Holding Corp., each ProFrac LLC Unit Holder to exercise its Redemption Right with respect to some or all of such unitholder's ProFrac LLC Units. In connection with any redemption of ProFrac LLC Units pursuant to the Redemption Right or acquisition of ProFrac LLC Units pursuant to the Call Right, a corresponding number of shares of ProFrac Class B Common Stock held by the relevant ProFrac LLC Unit Holder will be cancelled. As the ProFrac LLC Unit Holders cause their ProFrac LLC Units to be redeemed, holding other assumptions constant, ProFrac Holding Corp.'s membership interest in ProFrac LLC will be correspondingly increased, the number of shares of ProFrac Class B Common Stock outstanding will be increased and the number of shares of ProFrac Class B Common Stock will be decreased.

#### Distributions and allocations

Under the ProFrac LLC Agreement, subject to the obligations of ProFrac LLC to make tax distributions, ProFrac has the right to determine when distributions will be made to the holders of ProFrac LLC Units and the amount of any such distributions. If ProFrac authorizes a distribution, such distribution will be made to the holders of ProFrac LLC Units generally on a pro rata basis in accordance with their respective percentage ownership of ProFrac LLC Units.

ProFrac LLC will allocate its net income or net loss for each year to the holders of ProFrac LLC Units pursuant to the terms of the ProFrac LLC Agreement, and the holders of ProFrac LLC Units, including ProFrac Holding Corp., will generally incur U.S. federal, state and local income taxes on their share of any taxable income of ProFrac LLC. Net income and losses of ProFrac LLC generally will be allocated to the holders of ProFrac LLC Units on a pro rata basis in accordance with their respective percentage ownership of ProFrac LLC Units, subject to requirements under U.S. federal income tax law that certain items of income, gain, loss or deduction be allocated disproportionately in certain circumstances. To the extent ProFrac LLC has available cash, and subject to the terms of any current or future debt instruments, ProFrac intends to cause ProFrac LLC to make (i) generally pro rata distributions to the holders of ProFrac LLC Units, including ProFrac Holding Corp., in an amount at least sufficient to allow ProFrac Holding Corp. to pay its taxes and to make payments under the Tax Receivable Agreement and any subsequent tax receivable agreements that it may enter into in connection with future acquisitions and (ii) non-pro rata payments to ProFrac Holding Corp. to reimburse it for its corporate and other overhead expenses.

## Issuance of equity

The ProFrac LLC Agreement provides that at any time ProFrac Holding Corp. issues a share of ProFrac Class A Common Stock or any other equity security, the net proceeds received by ProFrac Holding Corp. with respect to such issuance, if any, shall be concurrently invested in ProFrac LLC, and ProFrac LLC shall issue to ProFrac Holding Corp. one ProFrac LLC Unit or other economically equivalent equity interest. Conversely, if at any time, any shares of ProFrac Class A Common Stock are redeemed, repurchased or otherwise acquired, ProFrac LLC shall redeem, repurchase or otherwise acquire an equal number of ProFrac LLC Units held by ProFrac Holding Corp., upon the same terms and for the same price, as the shares of ProFrac Class A Common Stock are redeemed, repurchased or otherwise acquired.

## Competition

The parties to the ProFrac LLC Agreement have agreed that the ProFrac Controlling Stockholders are permitted to engage in business activities or invest in or acquire businesses which may compete with ProFrac's business or do business with ProFrac's customers.

#### Dissolution

ProFrac LLC will be dissolved only upon the first to occur of (i) the sale of substantially all of its assets or (ii) an election by us to dissolve the company. Upon dissolution, ProFrac LLC will be liquidated and the proceeds from any liquidation will be applied and distributed in the following manner: (a) first, to creditors (including to the extent permitted by law, creditors who are members) in satisfaction of the liabilities of ProFrac LLC, (b) second, to establish cash reserves for contingent or unforeseen liabilities and (c) third, to the holders of ProFrac LLC Units in proportion to the number of ProFrac LLC Units owned by each of them.

### Tax Receivable Agreement

As described in "Information About ProFrac—Corporate Reorganization" and "Information About ProFrac—Certain Relationships and Related Party Transactions—ProFrac LLC Agreement—Redemption Rights," the ProFrac LLC Unit Holders may cause their ProFrac LLC Units to be redeemed in the future pursuant to the Redemption Right. ProFrac LLC intends to make for itself (and for each of its direct or indirect subsidiaries that is treated as a partnership for U.S. federal income tax purposes and that it controls) an election under Section 754 of the Code that will be effective for the taxable year of the ProFrac IPO and each taxable year in which a redemption of ProFrac LLC Units occurs. Pursuant to the Section 754 election, ProFrac Holding Corp.'s acquisition (or deemed acquisition for U.S. federal income tax purposes) of ProFrac LLC Units in conjunction with the ProFrac IPO or pursuant to the Redemption Right or the Call Right is expected to result in adjustments to the tax basis of the tangible and intangible assets of ProFrac LLC. These adjustments will be allocated to ProFrac Holding Corp. Such adjustments to the tax basis of the tangible and intangible assets of ProFrac LLC would not have been available to ProFrac Holding Corp. absent its acquisition or deemed acquisition of ProFrac LLC Units pursuant to the exercise of the Redemption Right or the Call Right. The anticipated basis adjustments are expected to increase (for tax purposes) ProFrac Holding Corp.'s depreciation and amortization deductions and may also decrease ProFrac Holding Corp.'s gains (or increase its losses) on future dispositions of certain assets to the extent the increase in tax basis is allocated to those assets. Such increased deductions and losses and reduced gains may reduce the amount of tax that ProFrac Holding Corp. would otherwise be required to pay in the future.

ProFrac Holding Corp. and the TRA Holders entered into the Tax Receivable Agreement at the closing of the ProFrac IPO. The Tax Receivable Agreement generally provides for the payment by ProFrac Holding Corp. to the TRA Holders of 85% of the net cash savings, if any, in U.S. federal, state and local income and franchise tax that ProFrac Holding Corp. actually realizes (computed using simplifying assumptions to address the impact of state and local taxes) or is deemed to realize in certain circumstances in periods after the ProFrac IPO as a result of, as applicable to each such TRA Holder, (i) certain increases in tax basis that occur as a result of ProFrac Holding Corp.'s acquisition (or deemed acquisition for U.S. federal income tax purposes) of all or a portion of such TRA Holder's ProFrac LLC Units in connection with this offering or pursuant to an exercise of the Redemption Right or the Call Right and (ii) imputed interest deemed to be paid by ProFrac Holding Corp. as a result of, and additional tax basis arising from, any payments ProFrac Holding Corp. makes under the Tax Receivable Agreement. ProFrac will retain the benefit of the remaining 15% of any actual net cash tax savings. Certain of the TRA Holders' rights (including the right to receive payments) under the Tax Receivable Agreement are transferable in connection with a permitted transfer of ProFrac LLC Units or if the TRA Holder no longer holds ProFrac LLC Units.

The payment obligations under the Tax Receivable Agreement are ProFrac Holding Corp.'s obligations and not obligations of ProFrac LLC, and ProFrac expects that the payments required to be made under the Tax Receivable Agreement will be substantial. Estimating the amount and timing of payments that may become due under the Tax Receivable Agreement is by its nature imprecise. For purposes of the Tax Receivable Agreement, net cash tax savings generally will be calculated by comparing ProFrac Holding Corp.'s actual tax liability (determined by using the actual applicable U.S. federal income tax rate and an assumed combined state and local income and franchise tax rate) to the amount it would have been required to pay had it not been able to utilize any of the tax benefits subject to the Tax Receivable Agreement. The actual increase in tax basis, as well as the

amount and timing of any payments, under the Tax Receivable Agreement will vary depending on a number of factors, including the timing of any redemption of ProFrac LLC Units, the price of ProFrac Class A Common Stock at the time of each redemption, the extent to which such redemptions are taxable transactions, the amount of the redeeming ProFrac LLC Unit Holder's tax basis in its ProFrac LLC Units at the time of the relevant redemption, the depreciation and amortization periods that apply to the increase in tax basis, the amount and timing of taxable income ProFrac generates in the future, the U.S. federal income tax rates then applicable, and the portion of ProFrac Holding Corp.'s payments under the Tax Receivable Agreement that constitute imputed interest or give rise to depreciable or amortizable tax basis.

Assuming no material changes in the relevant tax law, ProFrac expects that if it were to experience and change of control or the Tax Receivable Agreement was otherwise terminated immediately after the consummation of the Merger, the estimated termination payments based on the assumptions discussed below would be approximately \$487 million (calculated using a discount rate equal to (i) the greater of (A) 0.25% and (B) SOFR, plus 150 basis points, applied against an undiscounted liability of approximately \$568 million based on a 21% U.S. federal corporate income tax rate and estimated applicable state and local income tax rates).

A delay in the timing of redemptions of ProFrac LLC Units, holding other assumptions constant, would be expected to decrease the discounted value of the amounts payable under the Tax Receivable Agreement as the benefit of the depreciation and amortization deductions would be delayed and the estimated increase in tax basis could be reduced as a result of allocations of ProFrac LLC taxable income to the redeeming unit holder prior to the redemption. Stock price increases or decreases at the time of each redemption of ProFrac LLC Units would be expected to result in a corresponding increase or decrease in the undiscounted amounts payable under the Tax Receivable Agreement in an amount equal to 85% of the tax-effected change in price. Increases in the applicable corporate income tax rate and estimated applicable state and local income tax rates would be expected to result in a corresponding increase in the undiscounted amounts payable under the Tax Receivable Agreement as a result of the corresponding increase in net cash savings. The amounts payable under the Tax Receivable Agreement are dependent upon ProFrac Holding Corp, having sufficient future taxable income to utilize the tax benefits on which it is required to make payments under the Tax Receivable Agreement. If ProFrac Holding Corp.'s projected taxable income was significantly reduced, the expected payments would be reduced to the extent such tax benefits do not result in a reduction of ProFrac Holding Corp.'s future income tax liabilities. The foregoing amounts are merely estimates and the actual payments could differ materially. It is possible that future transactions or events could increase or decrease the actual tax benefits realized and the corresponding Tax Receivable Agreement payments as compared to the foregoing estimates. Moreover, there may be a negative impact on ProFrac's liquidity if, as a result of timing discrepancies or otherwise, (i) the payments under the Tax Receivable Agreement exceed the actual benefits realized in respect of the tax attributes subject to the Tax Receivable Agreement and/or (ii) distributions to ProFrac Holding Corp. by ProFrac LLC are not sufficient to permit ProFrac Holding Corp. to make payments under the Tax Receivable Agreement after it has paid its taxes and other obligations. Please read "Risk Factors—Risks Related to ProFrac's Class A Common Stock—In certain cases, payments under the Tax Receivable Agreement may be accelerated and/or significantly exceed the actual benefits, if any, ProFrac Holding Corp. realizes in respect of the tax attributes subject to the Tax Receivable Agreement." The payments under the Tax Receivable Agreement will not be conditioned upon the TRA Holders having a continued ownership interest in either ProFrac LLC or ProFrac Holding Corp.

In addition, although ProFrac is not presently aware of any issue that would cause the IRS or other relevant taxing authorities to challenge potential tax basis increases or other tax benefits covered under the Tax Receivable Agreement, the applicable ProFrac LLC Unit Holders will not reimburse ProFrac for any payments previously made under the Tax Receivable Agreement if such basis increases or other benefits are subsequently disallowed, except that excess payments made to any such holder will be netted against future payments otherwise required to be made, if any, to such holder after ProFrac's determination of such excess (which determination may be made a number of years following the initial payment and after future payments have been made). As a result, in such circumstances, ProFrac could make payments that are greater than ProFrac Holding Corp.'s actual cash tax savings, if any, and ProFrac may not be able to recoup those payments, which could adversely affect ProFrac's liquidity.

The term of the Tax Receivable Agreement commenced upon the completion of the ProFrac IPO and will continue until all tax benefits that are subject to the Tax Receivable Agreement have been utilized or expired, unless ProFrac experiences a change of control (as defined in the Tax Receivable Agreement, which includes certain mergers, asset sales, or other forms of business combinations) or the Tax Receivable Agreement otherwise terminates early (at ProFrac Holding Corp.'s election or as a result of ProFrac Holding Corp.'s breach or the commencement of bankruptcy or similar proceedings by or against ProFrac Holding Corp.) and ProFrac Holding Corp. makes the termination payments specified in the Tax Receivable Agreement in connection with such change of control or other early termination. In the event that the Tax Receivable Agreement is not terminated, the payments under the Tax Receivable Agreement are anticipated to commence in 2023 and to continue for 15 years after the date of the last redemption of the ProFrac LLC Units. Accordingly, it is expected that payments will continue to be made under the Tax Receivable Agreement for more than 20 years. Payments will generally be made under the Tax Receivable Agreement as ProFrac Holding Corp. realizes actual cash tax savings in periods after the ProFrac IPO from the tax benefits covered by the Tax Receivable Agreement. However, if ProFrac Holding Corp. experiences a change of control or the Tax Receivable Agreement otherwise terminates early, ProFrac Holding Corp.'s obligations under the Tax Receivable Agreement would accelerate and ProFrac Holding Corp. would be required to make an immediate payment equal to the present value of the anticipated future payments to be made by it under the Tax Receivable Agreement (determined by applying a discount rate equal to (i) the greater of (A) 0.25% and (B) SOFR, plus 150 basis points) and such payment is expected to be substantial. The calculation of anticipated future payments will be based upon certain assumptions and deemed events set forth in the Tax Receivable Agreement, including (i) that ProFrac Holding Corp. has sufficient taxable income to fully utilize the tax benefits covered by the Tax Receivable Agreement, and (ii) that any ProFrac LLC Units (other than those held by ProFrac Holding Corp.) outstanding on the termination date are deemed to be redeemed on the termination date. Any early termination payment may be made significantly in advance of, and may materially exceed, the actual realization, if any, of the future tax benefits to which the termination payment relates.

The Tax Receivable Agreement provides that in the event that ProFrac Holding Corp. (i) breaches any of its material obligations under the Tax Receivable Agreement, whether (A) as a result of its failure to make any payment when due (including in cases where ProFrac Holding Corp. elects to terminate the Tax Receivable Agreement early, the Tax Receivable Agreement is terminated early due to certain mergers, asset sales, or other forms of business combinations or changes of control or ProFrac Holding Corp. has available cash but fails to make payments when due in circumstances where ProFrac Holding Corp. does not have the right to elect to defer the payment, as described below), (B) as a result of ProFrac Holding Corp.'s failure to honor any other material obligation under the Tax Receivable Agreement, or (C) by operation of law as a result of the rejection of the Tax Receivable Agreement in a case commenced under the U.S. Bankruptcy Code or otherwise, or (ii) commences any proceeding, or has any proceeding commenced against it that is not dismissed or discharged following a period of 60 days, (X) for certain events of bankruptcy, dissolution or liquidation, (Y) seeking an appointment of a receiver or trustee or (Z) making a general assignment for the benefit of creditors, then all of ProFrac Holding Corp.'s payment and other obligations under the Tax Receivable Agreement will be automatically accelerated and become due and payable applying the same assumptions described above.

As a result of either a change of control (as defined in the Tax Receivable Agreement, which includes certain mergers, asset sales, or other forms of business combinations) or another early termination, ProFrac Holding Corp. could be required to make payments under the Tax Receivable Agreement that exceed its actual cash tax savings under the Tax Receivable Agreement. In these situations, ProFrac Holding Corp.'s obligations under the Tax Receivable Agreement could have a substantial negative impact on ProFrac's liquidity and could have the effect of delaying, deferring or preventing certain mergers, asset sales, or other forms of business combinations or changes of control that could be in the best interests of holders of ProFrac Class A Common Stock or alternatively reducing the consideration paid in any such transaction to holders of ProFrac Class A Common Stock. There can be no assurance that ProFrac Holding Corp. will be able to finance its obligations under the Tax Receivable Agreement.

Decisions made by ProFrac's management in the course of running ProFrac's business, such as with respect to mergers, asset sales, other forms of business combinations or other changes in control, may influence the timing and amount of payments that are received by the TRA Holders under the Tax Receivable Agreement. For example, the earlier disposition of assets following a redemption of ProFrac LLC Units may accelerate payments under the Tax Receivable Agreement and increase the present value of such payments, and the disposition of assets before a redemption of ProFrac LLC Units may increase the TRA Holders' tax liability without giving rise to any rights of the TRA Holders to receive payments under the Tax Receivable Agreement. Such effects may result in differences or conflicts of interest between the interests of the TRA Holders and other ProFrac Stockholders.

Payments generally are due under the Tax Receivable Agreement within five business days following the finalization of the schedule with respect to which the payment obligation is calculated. However, interest on such payments will begin to accrue from the due date (without extensions) of ProFrac Holding Corp.'s U.S. federal income tax return for the period to which such payments relate until such payment due date at a rate equal to (i) the greater of (A) 0.25% and (B) SOFR, plus 100 basis points. Except in cases where ProFrac Holding Corp. elects to terminate the Tax Receivable Agreement early or it is otherwise terminated as described above, generally ProFrac Holding Corp. may elect to defer payments due under the Tax Receivable Agreement if ProFrac Holding Corp. does not have available cash to satisfy ProFrac Holding Corp.'s payment obligations under the Tax Receivable Agreement or if ProFrac Holding Corp. so contractual obligations limit its ability to make these payments. Any such deferred payments under the Tax Receivable Agreement generally will accrue interest from the due date for such payment until the payment date at a rate of (i) the greater of (A) 0.25% and (B) SOFR, plus 500 basis points. However, interest will accrue from the due date for such payment until the payment date at a rate of (i) the greater of (A) 0.25% and (B) SOFR, plus 100 basis points if ProFrac Holding Corp. is unable to make such payment as a result of limitations imposed by existing credit agreements. ProFrac has no present intention to defer payments under the Tax Receivable Agreement.

Because ProFrac Holding Corp. is a holding company with no operations of its own, its ability to make payments under the Tax Receivable Agreement is dependent on the ability of ProFrac LLC to make distributions to ProFrac Holding Corp. in an amount sufficient to cover ProFrac Holding Corp.'s obligations under the Tax Receivable Agreement. This ability, in turn, may depend on the ability of ProFrac LLC's subsidiaries to make distributions to it. The ability of ProFrac LLC, its subsidiaries and other entities in which it directly or indirectly holds an equity interest to make such distributions will be subject to, among other things, the applicable provisions of Delaware law (or other applicable jurisdiction) that may limit the amount of funds available for distribution and restrictions in relevant debt instruments issued by ProFrac LLC or its subsidiaries and/other entities in which it directly or indirectly holds an equity interest. To the extent that ProFrac Holding Corp. is unable to make payments under the Tax Receivable Agreement for any reason, such payments will be deferred and will accrue interest until paid.

The Tax Receivable Agreement incorporated by reference as an exhibit to the registration statement of which this proxy statement/information statement/prospectus forms a part, and the foregoing description of the Tax Receivable Agreement is qualified by reference thereto.

## **ProFrac Founders Registration Rights Agreement**

In connection with the closing of the ProFrac IPO, ProFrac entered into a registration rights agreement with certain of the ProFrac LLC Unit Holders (the "ProFrac Founders Registration Rights Agreement"). The ProFrac Founders Registration Rights Agreement contains provisions requiring ProFrac to register under the federal securities laws the offer and resale of shares of ProFrac Class A Common Stock by such ProFrac LLC Unit Holders or certain of their affiliates or permitted transferees as more fully described below.

### Demand Rights

Under the ProFrac Founders Registration Rights Agreement, at any time after the 180-day lock-up period following the ProFrac IPO and subject to certain limitations, any ProFrac LLC Unit Holder will have the right to require ProFrac to prepare and file a registration statement registering the offer and sale of their shares of their ProFrac Class A Common Stock. Generally, ProFrac is required to provide notice of the request to certain other holders of ProFrac Class A Common Stock who may, in certain circumstances, participate in the registration. Subject to certain exceptions, ProFrac is not obligated to effect a demand registration more than four times for each ProFrac LLC Unit Holder or within 90 days after the closing of any requested underwritten offering of shares of ProFrac Class A Common Stock.

The ProFrac Founders Registration Rights Agreement also obligates ProFrac to cooperate with each ProFrac LLC Unit Holder in effecting the disposition of its shares of ProFrac Class A Common Stock by such methods as such holder may request, including through underwritten offerings and block trades.

### Piggyback Rights

The ProFrac Founders Registration Rights Agreement provides that, subject to certain exceptions, if at any time ProFrac proposes to register an offering of ProFrac Class A Common Stock or conduct an underwritten offering, whether or not for ProFrac's own account, then ProFrac must notify the Holders and allow them to include a specified number of their shares of ProFrac Class A Common Stock in that registration statement or underwritten offering, as applicable.

## Conditions and Limitations; Expenses; Indemnity

The above described registration rights under the ProFrac Founders Registration Rights Agreement are subject to certain conditions and limitations, and ProFrac will generally be obligated to pay all registration expenses in connection with these registration obligations, regardless of whether a registration statement is filed or becomes effective. The ProFrac Founders Registration Rights Agreement also requires ProFrac to indemnify each ProFrac LLC Unit Holder against certain liabilities under the Securities Act.

### ProFrac Stockholders' Agreement

In connection with the ProFrac IPO, ProFrac entered into the ProFrac Stockholders' Agreement with the ProFrac Controlling Stockholders. Among other things, the ProFrac Stockholders' Agreement provides that, prior to the Trigger Date, THRC Holdings and the Farris Trust will each have the right to designate one director for election to the ProFrac Board.

On and after the Trigger Date, the ProFrac Board will be divided into three classes, with directors of each class serving for staggered three-year terms, and THRC Holdings and the Farris Trust will each have the right to designate one class III director to the ProFrac Board for so long as each of THRC Holdings and the Farris Trust and their affiliates collectively beneficially own at least 5% of the outstanding shares of ProFrac Common Stock.

# **Shared Services Agreement**

In connection with the closing of the ProFrac IPO, ProFrac II LLC entered into a shared services agreement with Wilks Brothers (as defined below) (the "Shared Services Agreement"), pursuant to which Wilks Brothers is required to provide certain administrative and back-office support services to ProFrac II LLC, including, among others, information technology, payroll, employee benefits and treasury services. In exchange for such services, ProFrac II LLC is obligated to pay to Wilks Brothers on a quarterly basis an aggregate annual retainer fee equal to \$7 million. The Shared Services Agreement has an initial term of three years, and shall then automatically be extended for additional one-year terms unless Wilks Brothers or ProFrac II LLC provides 90 days prior written notice of its desire not to automatically extend the term. The shared services agreement may only be assigned by a party with the other party's consent. Wilks Brothers is owned by certain of the Controlling Stockholders.

### Wilks Development Lease Agreement

In connection with the completion of the FTSI Acquisition, FTSI conveyed to Wilks Development, LLC, an affiliate of ProFrac LLC, substantially all of FTSI's owned real property consisting primarily of FTSI's hydraulic fracturing equipment manufacturing facilities in exchange for net cash consideration of approximately \$44.4 million, and relatedly agreed to lease such real property from Wilks Development, LLC in exchange for aggregate lease payments of \$51.6 million through March 2032 (the "FTSI Sale Leaseback").

## **Equify Bridge Note**

On March 4, 2022, ProFrac II LLC entered into a \$45.8 million subordinated promissory note with ProFrac's affiliate, Equify Financial (as defined below), with a stated maturity date of March 4, 2027 (the "Equify Bridge Note"). The Equify Bridge Note bears interest at a percentage per annum equal to 1.00%. Interest under the Equify Bridge Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Equify Bridge Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date

quarterly interest payments may be paid in kind or in cash. In April 2022, ProFrac repaid \$25.0 million in principal under the Equify Bridge Note, resulting in an outstanding balance of \$20.8 million. In June 2022, the Equify Bridge Note was fully paid with net proceeds from the ProFrac IPO.

### **Backstop Note**

On March 4, 2022, ProFrac LLC entered into a \$22.0 million subordinated promissory note with ProFrac's affiliate, THRC Holdings, with a stated maturity date of March 4, 2027 (the "Backstop Note"). The Backstop Note bears interest at a percentage per annum equal to 1.74%. Interest under the Backstop Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Backstop Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash. In June 2022, the Backstop Note was fully paid with net proceeds from the ProFrac IPO.

## **Closing Date Note**

On March 4, 2022, ProFrac LLC entered into a \$22.0 million subordinated promissory note with ProFrac's affiliate, THRC Holdings, with a stated maturity date of March 4, 2027 (the "Closing Date Note"). The Closing Date Note bears interest at a percentage per annum equal to 1.74%. Interest under the Closing Date Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Closing Date Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash. In June 2022, the Closing Date Note was fully paid with net proceeds from the ProFrac IPO

### Policies and procedures for review of related party transactions

A "Related Party Transaction" is a transaction, arrangement or relationship in which ProFrac or any of its subsidiaries were, are or will be a participant, the amount of which involved exceeds \$120,000, and in which any related person had, has or will have a direct or indirect material interest. A "related person" means:

- any person who is, or at any time during the applicable period was, one of ProFrac's executive officers or directors;
- any person who is known by ProFrac to be the beneficial owner of more than 5% of the outstanding shares of ProFrac Class A Common Stock;

- any immediate family member of any of the foregoing persons, which means any child, stepchild, parent, stepparent, spouse, sibling, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law of a director, executive officer or a beneficial owner of more than 5% of ProFrac Class A Common Stock, and any person (other than a tenant or employee) sharing the household of such director, executive officer or beneficial owner of more than 5% of the outstanding shares of ProFrac Class A Common Stock; and
- any firm, corporation, or other entity in which any of the foregoing persons is a partner or principal or in a similar position or in which such person has a 10% or greater beneficial ownership interest.

The ProFrac Board has adopted a Code of Business Conduct and Code of Ethics for Senior Executive Officers that provides that the ProFrac Board or its authorized committee will review all transactions with related persons that are required to be disclosed under SEC rules and, when appropriate, initially authorize or ratify all such transactions. In addition, one of ProFrac's audit committee's functions is to review and approve all relationships and transactions in which ProFrac and ProFrac's directors, director nominees and executive officers and their immediate family members, as well as holders of more than 5% of any class of ProFrac's voting securities and their immediate family members, have a direct or indirect material interest.

ProFrac's Code of Business Conduct and Code of Ethics for Senior Executive Officers provides that, in determining whether or not to recommend the initial approval or ratification of a transaction with a related person, the ProFrac Board or its authorized committee should consider all of the relevant facts and circumstances available.

ProFrac's Code of Business Conduct and Code of Ethics for Senior Executive Officers described above was adopted in connection with the completion of the ProFrac IPO and, therefore, the transactions described below were not reviewed under such policies.

#### Transactions with affiliates

In the normal course of business, ProFrac has entered into transactions with related parties where the Dan and Farris Wilks (or entities they control) hold a controlling financial interest. During the years ended December 31, 2021 and 2020, ProFrac had related party transactions with the following related party entities, including:

Entity	Business & manner of relationship with the company
Automatize, LLC ("Automatize")	Logistics company that provides for the delivery of proppants on behalf of its customers, including ProFrac.
Cisco Logistics, LLC ("Cisco Logistics")	Logistics company that delivers sand and equipment on behalf of its customers, including ProFrac.
Equify Risk Services, LLC ("Equify Risk")	Insurance broker that negotiates and secures insurance policies on behalf of its customers, including ProFrac.
Equify Financial, LLC ("Equify Financial")	Finance company that provides equipment and other financing to its customers, including ProFrac.
Wilks Brothers, LLC ("Wilks Brothers")	Management company which provides administrative support to various businesses within their portfolio. Wilks Brothers and certain entities under its control will at times incur expenses on behalf of ProFrac, billing ProFrac for these expenses at cost plus a management fee.

Entity	Business & manner of relationship with the company

Interstate Explorations, LLC ("Interstate")

Exploration and development company for which ProFrac performs pressure pumping services, and from which ProFrac has a short-term

lease for certain office space.

Flying A Pump Services, LLC ('Flying A")

Oilfield services company which provides pump down and acid services,

to which ProFrac rents equipment.

MC Estates, LLC and the Shops at Willow Park, LLC ('Related Lessors'')

Own various industrial parks and office space leased by ProFrac.

Wilks Construction Company, LLC ("Wilks Construction")

Construction company that has built and made renovations to several buildings for ProFrac.

3 Twenty-Three, LLC ("3 Twenty-Three")

Payroll administrator which performs payroll services on behalf of its

customers, including ProFrac.

Carbo Ceramics Inc. ("Carbo Ceramics")

Provider of ceramic media and industrial technologies, which purchased proppant from ProFrac.

The following table summarizes expenditures with related parties for the years ended December 31, 2021 and 2020:

	Year o	Year ended		
	Decem	ber 31,		
	2021	2020		
Automatize	\$ 80,521	\$ 26,226		
Wilks Brothers	15,480	16,622		
Related Lessors	6,308	6,052		
Equify Financial	2,871	2,323		
3 Twenty-Three	1,033	1,148		
Carbo	513	_		
Cisco Logistics	509	4,181		
Interstate	80	30		
Equify Risk	3	1,602		
Wilks Construction	_	107		
Other	114			
Total	\$ 107,432	\$ 58,291		

The following table summarizes related party accounts payable as of December 31, 2021 and 2020:

	December 31, 2021	December 31, 2020
Automatize	\$ 11,198	\$ 5,633
Wilks Brothers	9,990	11,993
Wilks Construction	57	57
Carbo	10	_
Related Lessors	1	21
Cisco Logistics	_	671
Equify Financial	_	113
Other	19	
Total	\$ 21,275	\$ 18,488

The following table summarizes revenue from related parties for the years ended December 31, 2021 and 2020:

	Yea	r ended
	Dece	mber 31,
	2021	2020
Flying A	\$2,701	\$ 294
Carbo	1,025	193
Interstate	116	11
Wilks Brothers	65	_
Automatize	3	701
Other	32	84
Total	\$3,942	\$1,283

The following table summarizes related party accounts receivable as of December 31, 2021 and 2020:

	December 31,	December 31,	
	2021	2020	
Flying A <sup>(1)</sup>	\$ 2,412	\$ 549	
Cisco Logistics	1,489	_	
Carbo	591	116	
Automatize	_	191	
Other	23	24	
Total	\$ 4,515	\$ 880	

<sup>(1)</sup> The amounts above are reported net of an allowance for doubtful accounts related to Flying A, which was \$0.2 million as of December 31, 2021.

Additionally, in January and February of 2021, ProFrac LLC executed two agreements with one of ProFrac LLC's members for the sale of certain lots of equipment, in exchange for \$8.7 million in cash, an amount that approximates the net book value of the assets. Under these agreements, for any assets subsequently resold by the member, ProFrac LLC will reimburse the member for a certain percentage of the net loss, or conversely be entitled to a certain percentage of the net gain, at rates established in the agreements. As of June 30, 2022, the majority of the assets have been sold by the member, and ProFrac LLC's remaining liability of approximately \$0.2 million for assets that have not been resold has been recorded.

On February 4, 2022, THRC Holdings entered into an assignment and assumption of contract with Encantar Properties LP, one of the sellers from whom ProFrac purchased the Munger Ranch property, under which the related party was assigned rights under the Munger Right Agreement to \$8.1 million of the \$30.0 million in consideration related to the purchase of the Munger Ranch property. As part of the ProFrac IPO, the sellers of the Munger Ranch property were issued 2,114,273 shares of Class A Common Stock in exchange for the \$30.0 million consideration related to the Munger Ranch purchase. The sellers of the Munger Ranch property were subsequently issued 2,954 shares of Class A Common Stock as a true-up.

As discussed in the sections titled "The Merger — Interests of ProFrac Executive Officers and Directors in the Merger" and "Management's Discussion and Analysis of Financial Condition and Results of Operations of USWS", U.S. Well Services, LLC, a subsidiary of USWS, entered into a \$12,500,000 promissory with Equify Financial.

### MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF PROFRAC

The following discussion and analysis provides information that the management of ProFrac believes is relevant to an assessment and understanding of ProFrac's consolidated results of operations and financial condition. The discussion should be read in conjunction with the "Selected Historical and Pro Forma Consolidated Financial Data of ProFrac" and the historical audited consolidated financial statements for the years ended December 31, 2021 and 2020, and the related notes thereto, included elsewhere in this proxy statement/information statement/prospectus.

The discussion and analysis should also be read in conjunction with ProFrac's unaudited pro forma financial information as of and for the six months ended June 30, 2022, and for the year ended December 31, 2021. See "Unaudited Pro Forma Condensed Combined Financial Information." This discussion may contain forward-looking statements based upon ProFrac management's current beliefs and expectations, that involve risks, uncertainties, assumptions and other factors that could cause actual results to differ materially from those made or implied in the forward-looking statements. ProFrac's actual results may differ materially from those anticipated in these forward-looking statements as a result of various factors, including those discussed below and those set forth under "Risk Factors" or in other parts of this proxy statement/information statement/prospectus. Accordingly, you are cautioned not to place undue reliance on these forward-looking statements, which reflect ProFrac management's analysis only as of the date hereof.

Unless the context otherwise requires, references in this "Management's Discussion and Analysis of Financial Condition and Results of Operations of ProFrac" to "we," "our," "ProFrac," and "the Company" refer to the business and operations of ProFrac Holding Corp. and its consolidated subsidiaries prior to the Merger. All dollar amounts are expressed in thousands of United States dollars ("\$"), unless otherwise indicated.

### Our Predecessor and ProFrac Holding Corp.

ProFrac Holding Corp. was formed on August 17, 2021, and had not conducted any material business operations prior to the completion of the transactions described under "Information About ProFrac" other than certain activities related to our IPO. Our predecessor consists of ProFrac LLC and its subsidiaries, Best Flow and Alpine (which we refer to as "ProFrac Predecessor") on a consolidated basis. Historical periods for ProFrac Predecessor had been presented on a consolidated and combined basis given the common control ownership by the Wilks. On December 21, 2021, all of the thenoutstanding membership interests in Best Flow and Alpine were contributed to ProFrac LLC in exchange for membership interests in ProFrac LLC. Unless otherwise indicated, the historical consolidated financial information included in this proxy statement/information statement/prospectus presents the historical financial information of ProFrac Predecessor. Historical consolidated financial information is not indicative of the results that may be expected in any future periods. For more information, please see the historical consolidated financial statements and related notes thereto included elsewhere in this proxy statement/information statement/prospectus and "—Factors Affecting the Comparability of Our Financial Results."

### Overview

We are a growth-oriented, vertically integrated and innovation-driven energy services company providing hydraulic fracturing, completion services and other complementary products and services to leading upstream oil and gas companies engaged in the exploration and production ("E&P") of North American unconventional oil and natural gas resources. Founded in 2016, ProFrac was built to be the go-to service provider for E&P companies' most demanding hydraulic fracturing needs. We are focused on employing new technologies to

significantly reduce "greenhouse gas" emissions and increase efficiency in what has historically been an emissions-intensive component of the unconventional E&P development process. We believe the technical and operational capabilities of our fleets ideally position us to capture increased demand resulting from the market recovery and our customers' shifting preferences favoring the sustainable development of natural resources.

Our operations are focused on the leading unconventional regions in the United States, where we have cultivated deep and longstanding customer relationships with some of those regions' most active E&P companies. We operate in three business segments: stimulation services, manufacturing and proppant production. We believe we are one of the largest providers of hydraulic fracturing services in North America, with installed capacity of 34 conventional fleets, of which, as of June 30, 2022, 31 were active.

### **Overall Trends and Outlook**

The global public health crisis associated with the COVID-19 pandemic had an unprecedented effect on demand for energy, crude oil prices and global economic activity. In 2020, the combined effect of COVID-19 and the disruptions to the energy industry led to a rapid and significant decline in WTI crude oil prices and Henry Hub natural gas prices. In response to the significant drop in commodity prices, E&P companies acted swiftly to reduce capital budgets and drilling and completion activity. Reduced demand for services compounded by constrained capital access forced the acceleration of the attrition cycle for pressure pumping equipment as older equipment requiring higher repair and maintenance spending were used for spare parts or scrapped. These difficult industry conditions allowed us to strengthen our industry leadership position by implementing targeted and forward-looking initiatives that enhance our equipment and provide greater supply chain control.

Drilling and completion activities for oil and gas are heavily influenced by oil and gas prices. In 2022, geopolitical tensions in Eastern Europe related to Russia's invasion of Ukraine have resulted in significant supply disruptions as a broad coalition of countries have responded with sanctions and/or import bans associated with Russian oil and natural gas. This has resulted in significant tightening in the market as reflected by higher commodity prices, with oil and gas prices reaching decade highs. As a result of the FTSI acquisition, our operations have diversified exposure to both natural gas and oil producing areas. Natural gas and oil prices have increased substantially compared to year-end 2020 prices and have also surpassed year-end 2019 (pre-COVID-19) levels.

While commodity prices have returned to and exceeded pre-pandemic levels, the pandemic has nonetheless led to supply chain disruptions worldwide. Tariffs, access to employees, increased shipping rates and raw material shortages are plaguing markets. Our supply chain is either vertically integrated or predominantly U.S. based, mitigating our exposure to global disruptions and price increases and allowing us to continue to maintain attractive margins. As our operations are predominantly U.S. based, we have no direct exposure to Russia and Ukraine. We have realized indirect impacts that may have occurred as a result of the crisis, such as modest increases in the costs of certain raw materials and components we purchase for use in our manufacturing processes. However, given the inflationary climate in the United States and globally, we are unable to determine the extent to which such increased commodity prices are the result of the crisis in Ukraine or a result of other factors. Despite these increases, we have experienced improved results of operations due to increased utilization of our fleets and increased prices for our products and services which have permitted us to maintain and increase our margins notwithstanding such cost increases.

The oil and gas industry is currently undergoing significant realignment of operating practices with a focus on reducing impacts to the environment. Many E&P companies are implementing carbon tracking and reduction initiatives and are expecting oilfield service providers to deliver products and services that utilize the most advanced and environmentally friendly technologies. We believe that companies in the pressure pumping industry with the most technologically advanced fleets and lowest carbon footprint will likely see significant growth in market share at the expense of companies with less advanced equipment. We have embraced tangible initiatives that help to protect the environment and improve our environment and communities making it part of

our organizational culture since early in the life of the company. We have and intend to continue to invest in a number of industry leading advanced technologies that reduce carbon emissions while increasing profitability. We are currently upgrading five to ten engines per month from Tier II to Tier IV DGB. In June 2021 we entered into an agreement with USWS under which we have the ability to acquire up to 20 licenses to construct electric-powered hydraulic fracturing fleets utilizing Clean Fleet® technology. We believe that these initiatives and commitment to lower emissions will help us lead the energy transition of the frac industry towards cleaner and sustainable business.

### **How we Generate Revenue**

We operate three business segments: stimulation services, manufacturing and proppant production. Business activities that are not separately reportable, which only include Flotek, are classified in the other category.

Stimulation services. We own and operate a fleet of mobile hydraulic fracturing units and other auxiliary equipment that generates revenue by providing stimulation services to our customers. We also provide personnel and services that are tailored to meet each of our customers' needs. We generally do not have long-term written contractual arrangements with our customers other than standard master service agreements, which include general contractual terms between our customers and us. We charge our customers on a per-job basis, in which we set pricing terms after receiving full specifications for the requested job, including the lateral length of the customer's wellbore, the number of frac stages per well, the amount of proppant employed and other specifications of the job. Well stimulation contains complementary services that we often provide to our customers, including sand and associated logistics, chemicals and fuel. These complementary services are provided through various contractual arrangements based on our customers' needs.

Manufacturing. We primarily generate revenue through sales of highly engineered, tight tolerance machined, assembled, and factory tested products such as high horsepower pumps, valves, piping, swivels, large-bore manifold systems, seats, and fluid ends. As of June 30, 2022, we operate facilities in Cisco, Aledo and Fort Worth, Texas, including an ISO 9001 2015 certified OEM manufacturing facility, in which we manufacture and refurbish many of the components used by our fleets, including pumps, fluid ends, power ends, flow iron and other consumables and an engine and transmission rebuild facility that is licensed to provide warranty repairs on our transmissions. Additionally, we provide iron inspection, iron recertification, pump refurbishment, fluid end refurbishment, pump function testing, paint, scrap, and lube system change services. We charge our customers for equipment based on a per-order basis, in which we set pricing terms after receiving full specifications for the requested equipment. We charge our customers for our services based on the parts and labor incurred. For the three and six months ended June 30, 2022 and 2021, intersegment revenues for the manufacturing segment were 92% and 88%, and 88% and 87%, respectively.

Proppant production. We generate revenue by providing proppant to oilfield service providers and E&P companies. We own and operate the Kermit sand mine in west Texas and recently purchased and are in the process of developing the West Munger sand mine near Lamesa, Texas, and we charge our customers on a per ton of proppant basis at current market prices. We do not have long-term written contractual arrangements with our customers with fixed pricing. For the three and six months ended June 30, 2022 and 2021, intersegment revenues for the proppant production segment were 64% and 44%, and 66% and 38%, respectively.

# Cost of Conducting our Business

The principal costs of products and services involved in operating our business are expendables, personnel, equipment repairs and maintenance and fuel. Our fixed costs are relatively low and a large portion of the costs described below are only incurred as we perform jobs for our customers.

Expendables. Expendables used in our stimulation services business are the largest expenses incurred, and include the fuel, product and freight costs associated with proppant, chemicals and other consumables. Fuel is

consumed both in the operation and movement of our hydraulic fracturing fleets and other equipment. In our proppant production business, fuel to run equipment is one of our major expenses. These costs comprise a substantial variable component of our service costs, particularly with respect to the quantity and quality of sand demanded when providing hydraulic fracturing services.

Raw Materials. Our manufacturing segment relies on various raw materials, specifically various grades of steel and other raw metals, and electricity.

Direct Labor Costs. Payroll and benefit expenses directly related to the delivery of our products and services are included in our operating costs.

Other Direct Costs. We incur other expenses related to our products and service offerings, including the costs of repairs and maintenance, general supplies, equipment rental and other miscellaneous operating expenses. Capital expenditures to upgrade or extend the useful life of equipment are not included in other direct costs.

### How we Evaluate our Operations

Our management uses a variety of financial and operating metrics to evaluate and analyze the performance of our business, including Adjusted EBITDA

## Note regarding Non-GAAP financial measures

Adjusted EBITDA is a non-GAAP financial measure and should not be considered as a substitute for net income, net loss, operating loss or any other performance measure derived in accordance with GAAP or as an alternative to net cash provided by operating activities as a measure of our profitability or liquidity. Adjusted EBITDA is a supplemental measure utilized by our management and other users of our financial statements such as investors, commercial banks, research analysts and others, to assess our financial performance because it allows us to compare our operating performance on a consistent basis across periods by removing the effects of our capital structure (such as varying levels of interest expense), asset base (such as depreciation and amortization) and items outside the control of our management team (such as income tax rates).

We view Adjusted EBITDA as an important indicator of performance. We define Adjusted EBITDA as our net income (loss), before (i) interest expense, net, (ii) income tax provision, (iii) depreciation, depletion and amortization, (iv) loss on disposal of assets, (v) stock-based compensation, and (vi) other unusual or non-recurring charges, such as costs related to our initial public offering,non-recurring supply commitment charges, certain bad debt expense and gain on extinguishment of debt.

We believe that our presentation of Adjusted EBITDA will provide useful information to investors in assessing our financial condition and results of operations. Net income (loss) is the GAAP measure most directly comparable to Adjusted EBITDA. Adjusted EBITDA should not be considered as an alternative to net income (loss). Adjusted EBITDA has important limitations as an analytical tool because it excludes some but not all items that affect the most directly comparable GAAP financial measure. You should not consider Adjusted EBITDA in isolation or as a substitute for an analysis of our results as reported under GAAP. Because Adjusted EBITDA may be defined differently by other companies in our industry, our definition of this non-GAAP financial measures may not be comparable to similarly titled measures of other companies, thereby diminishing its utility.

## Factors Affecting the Comparability of our Financial Results

Our future results of operations may not be comparable to our historical results of operations for the reasons described below:

### Recent Acquisitions

We have grown recently through strategic acquisitions and investments, including through our recently completed acquisitions of SP Companies, FTSI and West Munger and our investments in BPC and Flotek. These acquisitions are not reflected in our historical results of operations and our future results will differ as a result.

In addition, in connection with our acquisitions, we have recorded the acquired assets and liabilities at fair value on the date of acquisition, which has impacted deferred revenue and deferred costs balances and increased revenue and expenses from that which would have otherwise been recognized in subsequent periods. We also recorded identifiable intangible assets that are amortized over their useful lives, increasing expenses from that which would otherwise have been recognized.

## Public company expenses

We expect to incur additional recurring administrative expenses as a result of becoming a publicly traded corporation that we have not previously incurred, including costs associated with compliance under the Exchange Act, annual and quarterly reports to shareholders, transfer agent fees, audit fees, incremental director and officer liability insurance costs, SOX compliance readiness, and director and officer compensation. We additionally expect to incur approximately \$2.5 million in incremental, non-recurring costs related to our transition to a publicly traded corporation.

#### Income taxes

ProFrac Holding Corp. is a corporation and will be subject to U.S. federal, state and local income taxes. Although the ProFrac Predecessor entities are subject to franchise tax in the State of Texas, they have historically been treated as pass-through entities for U.S. federal and other state and local income tax purposes and as such were not subject to U.S. federal income taxes or other state or local income taxes. Rather, the tax liability with respect to the taxable income of the ProFrac Predecessor entities was passed through to their owners. Accordingly, prior to the March 2022 acquisition of FTSI, a corporation subject to U.S. federal and state income tax, the financial data attributable to ProFrac Predecessor contains no provision for U.S. federal income taxes or income taxes in any state or locality (other than franchise tax in the State of Texas). Subsequent to the acquisition of FTSI in March 2022, the financial data of the ProFrac Predecessor does contain both U.S. federal and state income taxes. We estimate that we will be subject to U.S. federal and state taxes at a blended statutory rate of approximately 23% of pre-tax earnings. Additionally, with the acquisition of EKU, the Company is subject to certain foreign taxes, which were immaterial for the three and six months ended June 30, 2022.

Effective March 4, 2022 we account for income taxes under the asset and liability method. Under this method, deferred tax assets and liabilities are recognized for the estimated future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax basis. Deferred tax assets and liabilities are measured using enacted tax rates in effect for the year in which those temporary differences are expected to be recovered or settled pursuant to the provisions of Accounting Standards Codification 740, Income Taxes. The effect on deferred tax assets and liabilities of a change in tax rate is recognized in earnings in the period that includes the enactment date. Valuation allowances are established when necessary to reduce deferred tax assets to the amounts more likely than not to be realized.

We recorded a full valuation allowance on our net deferred tax assets based on our assessment that it is more likely than not that the deferred tax asset will not be realized. A change in these assumptions could cause a decrease to the valuation allowance, which could materially impact our results of operations.

## **Results of Operations**

Three and six months ended June 30, 2022 compared to three and six months ended June 30, 2021

	Three Months Ended June 30,		Six Mont Jun	hs Ended e 30,
(Dollars in thousands)	2022	2021	2022	2021
Revenues—Stimulation services	\$576,556	\$168,506	\$912,711	\$312,209
Revenues—Manufacturing	34,854	16,223	66,860	30,880
Revenues—Proppant production	17,531	7,781	29,939	13,370
Other	15,359	_	15,359	_
Eliminations	(54,456)	(17,691)	(90,045)	(32,054)
Total revenues	589,844	174,819	934,824	324,405
Cost of revenues, exclusive of depreciation, depletion, and amortization—				
Stimulation services	344,535	126,448	589,116	245,801
Manufacturing	22,104	13,916	41,477	24,566
Proppant production	9,309	4,035	13,543	6,701
Other	19,108	_	19,108	_
Eliminations	(54,456)	(17,691)	(90,045)	(32,054)
Total cost of revenues, exclusive of depreciation, depletion, and amortization	340,600	126,708	573,199	245,014
Depreciation, depletion and amortization	64,064	34,904	108,280	70,365
Loss on disposal of assets, net	2,143	1,868	1,989	4,075
Selling, general and administrative	87,548	14,094	121,675	27,872
Interest expense, net	13,451	6,187	22,723	12,222
Loss on extinguishment of debt	8,822	_	17,095	_
Other income	(989)	(53)	(9,220)	(240)
Income tax (benefit) provision	4,112	(283)	4,864	(308)
Net income (loss)	\$ 70,093	\$ (8,606)	\$ 94,219	\$ (34,595)
Less: net (income) loss attributable to ProFrac Predecessor	(56,157)	8,478	\$ (79,867)	\$ 34,476
Less: net loss attributable to noncontrolling interests	8,704	128	\$ 8,288	\$ 119
Less: net income attributable to redeemable noncontrolling interests	(16,082)		(16,082)	
Net income attributable to ProFrac Holding Corp	\$ 6,558	<u>\$</u>	\$ 6,558	<u>\$</u>
Other data:				
Adjusted EBITDA—Stimulation services	\$196,088	\$ 30,475	\$269,657	\$ 43,428
Adjusted EBITDA—Manufacturing	\$ 9,360	\$ 349	\$ 19,382	\$ 2,679
Adjusted EBITDA—Proppant production	\$ 12,574	\$ 3,246	\$ 20,459	\$ 5,652
Adjusted EBITDA—Other	<u>\$ (7,454)</u>	<u>\$</u>	\$ (7,454)	<u>\$</u>
Adjusted EBITDA (1)	\$210,568	\$ 34,070	\$302,044	\$ 51,759
Active fleets (2)	31	14	26	14
Baker Hughes Domestic Average Rig Count—Onshore (3)	655	410	615	382
Average oil price (per barrel) (4)	\$ 108.72	\$ 66.09	\$ 101.59	\$ 61.94
Average natural gas price (per thousand cubic feet) (5)	<u>\$ 7.77</u>	\$ 3.06	\$ 6.30	\$ 3.38

<sup>(1)</sup> For definitions of the non-GAAP financial measures of Adjusted EBITDA, see "Note regarding Non-GAAP financial measures" above, and for a reconciliation of Adjusted EBITDA to our most directly comparable financial measures calculated in accordance with GAAP, see Note 18 -Segment Information, in the notes to our consolidated financial statements in F-69.

- Active fleets is the average number of fleets operating in the period.
- (3)
- (4)
- Average onshore U.S. rig count published by Baker Hughes.

  Average West TX Intermediate Spot Price published by EIA.

  Average Henry Hub Natural Gas Spot Price published by EIA.

### Three and six months ended June 30, 2022 compared to three and six months ended June 30, 2021

#### Revenue

Revenues—Stimulation services. Stimulation services revenues for the second quarter and first six months of 2022 increased \$408.0 million, or 242%, and \$600.5 million, or 192%, respectively from the same periods in 2021. The increase was primarily attributable to an increase in customer activity for our stimulation services, and an increase in active fleets, pumping hours and pricing. Additionally, FTSI contributed revenue to this segment from the acquisition date.

Revenues—Manufacturing. Manufacturing revenues for the second quarter and first six months of 2022 increased \$18.6 million, or 115%, and \$36.0 million, or 117%, respectively from the same periods in 2021. The increase was primarily attributable to an increase in demand for our products due to increased activity levels from customers for manufactured components utilized in the oilfield service industry.

Revenues—Proppant production. Proppant production revenues for the second quarter and first six months of 2022 increased \$9.8 million, or 125%, and \$16.6 million, or 124%, respectively from the same periods in 2021. The increase was primarily attributable to an increase in proppant production and pricing resulting from increased proppant demand primarily in the Permian basin.

Revenues—Other. Other revenues for the second quarter and first six months of 2022 were \$15.4 million compared to zero in the same periods in 2021. The increase was attributable to the Flotek acquisition.

*Total revenues.* Total revenues for the second quarter and first six months of 2022 increased \$415.0 million and \$610.4 million, respectively, from the same periods in 2021. The increase was primarily attributable to an increase in customer activity for our stimulation services, and an increase in active fleets, pumping hours and pricing. Additionally, FTSI contributed from the acquisition date.

# Operating costs and expenses

Cost of revenues, exclusive of depreciation, depletion, and amortization—Stimulation services. Stimulation services for the second quarter and first six months of 2022 increased \$218.1 million, or 172%, and \$343.3 million, or 140%, respectively from the same periods in 2021. The increase was primarily due to an increase in activity levels, as discussed in Revenue above, and additional input cost inflation. Additionally, FTSI contributed cost to this segment from the acquisition date.

Cost of revenues, exclusive of depreciation, depletion, and amortization—Manufacturing. Manufacturing for the second quarter and first six months of 2022 increased \$8.2 million, or 59%, and \$16.9 million, or 69%, respectively from the same periods in 2021. The increase was primarily attributable to an increase in demand for our products as well as an increase in the cost of raw materials.

Cost of revenues, exclusive of depreciation, depletion, and amortization—Proppant production. Proppant production for the second quarter and first six months of 2022 increased \$5.3 million, or 131%, and \$6.8 million, or 102%, respectively from the same periods in 2021. The increase was primarily attributable to a litigation charge of \$4.0 million and an increase in proppant production and increased cost of production.

Cost of revenues, exclusive of depreciation, depletion, and amortization—Other. Other for the second quarter and first six months of 2022 was \$19.1 million compared to zero in the same periods in 2021. The increase was attributable to the Flotek acquisition.

Depreciation, depletion, and amortization. Depreciation, depletion, and amortization for the second quarter and first six months of 2022 increased \$29.2 million and \$37.9 million, respectively from the same periods in 2021, The increase was primarily due to increased capital expenditure in the first quarter of 2022 and the depreciation related to the assets acquired from FTSI acquisition.

Gain or Loss on disposal of assets, net. For the second quarter and first six months of 2022 we recorded a net loss on disposal of assets of \$2.1 million and \$2.0 million, respectively, compared with a net loss on disposal of assets of \$1.9 million and \$4.1 million, respectively, for the same periods in 2021.

Selling, general and administrative. Selling, general and administrative expenses for the second quarter and first six months of 2022 increased \$73.5 million, or 521%, and \$93.8 million, or 337%, respectively from the same periods in 2021. The increase was primarily due to a stock-based compensation charge of \$40.3 million primarily related to certain deemed contributions in the second quarter of 2022. The increase was also due to higher headcount and personnel costs, higher incentive compensation, increased non-labor costs associated with the increased activity levels, and acquisition related expenses. See Note 11 – Stock-based compensation in the notes to our consolidated financial statements for additional discussion related to the stock-based compensation charge.

Interest expense, net. Interest expense, net for the second quarter and first six months of 2022 increased \$7.3 million and \$10.5 million, respectively, from the same periods in 2021. The increase in interest expense, net was attributable to various debt transactions, which increased our debt balances, and higher average interest rates in the first quarter of 2022. See Note 7 – Indebtedness in the notes to our consolidated financial statements for additional discussion related to our debt.

Loss on extinguishment of debt. As a result of the debt refinancing transactions in the first quarter of 2022 and debt repayments in the second quarter of 2022 transactions for the second quarter and first six months of 2022, we recognized a loss on extinguishment of \$8.8 million and \$17.1 million, respectively. See Note 7 – Indebtedness in the notes to our consolidated financial statements for additional discussion related to our debt.

Other income. For the second quarter and first six months of 2022 we recognized noncash income of \$1.0 million and \$9.2 million, respectively, primarily related to the change in fair value of the Convertible Notes related to our investment in BPC and Flotek.

*Income tax benefit (provision).* Income tax expense for the second quarter and first six months of 2022 was \$4.1 million and \$4.9 million, respectively, compared to an income tax benefit of \$0.3 and \$0.3 million for the same periods in 2021. The increase in the 2022 tax provision is due to the ProFrac Predecessor being partially owned by ProFrac Corp, a taxable entity, subsequent to the IPO.

### Segment results

The performance of our segments is evaluated primarily on Segment Adjusted EBITDA. For a reconciliation of Segment Adjusted EBITDA to net income or loss, see Note 18 – Segment Information, in the notes to our consolidated financial statements in this proxy statement/information statement/prospectus, which information is incorporated herein by reference.

Adjusted EBITDA—Stimulation services. Adjusted EBITDA—Stimulation services for the second quarter and first six months of 2022 increased \$165.6 million and \$226.2 million, respectively, from the same periods in 2021. The increase was primarily attributable to an increase in average active fleets and increased pricing for our services, partially offset by costs associated with the FTSI acquisition of \$13.0 million.

Adjusted EBITDA—Manufacturing. Adjusted EBITDA—Manufacturing for the second quarter and first six months of 2022 increased \$9.0 million and \$16.7 million, respectively, from the same periods in 2021. The increase was primarily due to increased volume for our products used by our oil field services customers.

Adjusted EBITDA—Proppant production. Adjusted EBITDA—Proppant production for the second quarter and first six months of 2022 increased \$9.3 million and \$14.8 million, respectively, from the same periods in 2021. The increase was primarily attributable to an increase in proppant production and pricing resulting from increased proppant demand in the Permian basin.

Adjusted EBITDA—Other. Adjusted EBITDA—Other for the second quarter and first six months of 2022 decreased \$7.5 million from the same periods in 2021. The decrease was due to Flotek results.

Year ended December 31, 2021 compared to year ended December 31, 2020

		ended
	Decem	,
	2021	2020 nds except
	industr	1
Revenues—Stimulation services	\$745,373	\$ 538,282
Revenues—Manufacturing	76,360	46,222
Revenues—Proppant production	27,225	10,215
Eliminations	(80,605)	(47,040)
Total revenues	768,353	547,679
Cost of revenues, exclusive of depreciation, depletion and amortization—Stimulation services	570,828	433,122
Cost of revenues, exclusive of depreciation, depletion, and amortization—Manufacturing	65,849	40,424
Cost of revenues, exclusive of depreciation, depletion, and amortization—Proppant production	14,050	6,064
Eliminations	(80,605)	(47,040)
Total cost of revenues, exclusive of depreciation, depletion and amortization	570,122	432,570
Depreciation, depletion and amortization	140,687	150,662
Loss on disposal of assets, net	9,777	8,447
Selling, general and administrative	65,592	51,014
Interest expense, net	25,788	23,276
Other expense (income)	111	(324)
Income tax (benefit) provision	(186)	582
Net loss	\$ (43,538)	\$ (118,548)
Net loss attributable to noncontrolling interest	(1,118)	(1,143)
Net loss attributable to ProFrac Predecessor	\$ (42,420)	\$ (117,405)
Other data:		
Adjusted EBITDA—Stimulation services	\$122,634	\$ 68,787
Adjusted EBITDA—Manufacturing	\$ 1,382	\$ 1,325
Adjusted EBITDA—Proppant production	\$ 10,672	\$ 2,685
Adjusted EBITDA(1)	\$134,688	\$ 72,797
Baker Hughes Domestic Average Rig Count Onshore(2)	606	524
Average oil price (per barrel)(3)	\$ 67.99	\$ 39.16
Average natural gas price (per thousand cubic feet)(4)	\$ 3.91	\$ 2.03

<sup>(1)</sup> For definitions of the non-GAAP financial measures of Adjusted EBITDA and reconciliation of Adjusted EBITDA to our most directly comparable financial measures calculated in accordance with GAAP, please read "Summary—Summary Historical and Pro Forma Financial Data—Non-GAAP Financial Measures."

- Average onshore U.S. rig count published by Baker Hughes. Average West TX Intermediate Spot Price published by EIA.
- (4) Average Henry Hub Natural Gas Spot Price published by EIA.

#### Revenues

Revenues—Stimulation services. Stimulation services revenues increased 38%, or \$207.1 million, to \$745.4 million for the year ended December 31, 2021, from \$538.3 million for the year ended December 31, 2020. The increase was primarily attributable to strong recovery from the COVID-19 pandemic resulting in an increase in customer activity for our stimulation services. We increased pumping hours 31% for the year ended December 31, 2021, versus the year ended December 31, 2020. Our average marketed active fleet count increased 27% to 14 for the year ended December 31, 2021, from 11 in the year ended December 31, 2020. We define a marketed fleet as 50,000 hydraulic horsepower, three blenders, high pressure iron, one hydration unit, one data van, suction hoses, a manifold system and other ancillary equipment as needed.

Revenues—Manufacturing. Manufacturing revenues increased 65%, or \$30.1 million, to \$76.4 million for the year ended December 31, 2021, from \$46.2 million for the year ended December 31, 2020. The increase was primarily attributable to an increase in demand for our products due to an increase in commodity prices, demand for manufactured components utilized in the oilfield service industry, and the acquisition of a majority stake in EKU. For the years ended December 31, 2021 and 2020, intersegment revenues accounted for 92% and 97% of manufacturing revenues, respectively.

Revenues—Proppant production. Proppant production revenues increased 167%, or \$17.0 million, to \$27.2 million for the year ended December 31, 2021, from \$10.2 million for the year ended December 31, 2020. The increase was primarily attributable to a 123% increase in proppant production and a 19% increase in proppant pricing resulting from increases in commodity prices and proppant demand in the Permian basin. The plant operated all 12 months of 2021 as a result of the COVID-19 pandemic recovery compared to 9 months of the same period for 2020. For the years ended December 31, 2021 and 2020, intersegment revenues accounted for 40% and 21% of proppant production revenues, respectively.

Total revenues. Total revenues increased 40%, or \$220.7 million, to \$768.4 million for the year ended December 31, 2021, from \$547.7 million for the year ended December 31, 2020. The increase was primarily attributable to recovery from the COVID-19 pandemic resulting in increased demand for oilfield services. Average oil and natural gas prices have increased 73% and 92%, respectively, from the year ended December 31, 2021, to the comparative period in 2020. The Baker Hughes U.S. onshore rig count also increased 16% when comparing the same periods.

## Operating costs and expenses

Cost of revenues, exclusive of depreciation, depletion, and amortization—Stimulation services. Cost of revenues, exclusive of depreciation, depletion, and amortization—Stimulation services increased 32%, or \$137.7 million, to \$570.8 million for the year ended December 31, 2021, from \$433.1 million for the year ended December 31, 2020. The increase was primarily due to an increase in fuel, personnel, expendable and other variable costs due to higher activity levels and an increase in our average marketed fleet count from 11 in the year ended December 31, 2020, to 14 in the year ended December 31, 2021. As a percentage of revenues, Cost of revenues, exclusive of depreciation and depletion—Stimulation services was 77% for the year ended December 31, 2021, as compared to 80% for the year ended December 31, 2020.

Cost of revenues, exclusive of depreciation, depletion, and amortization—Manufacturing. Cost of revenues, exclusive of depreciation, depletion, and amortization—Manufacturing increased 63%, or \$25.4 million, to \$65.8 million for the year ended December 31, 2021, from \$40.4 million for the year ended December 31, 2020. The increase was due to higher activity levels, coupled with higher personnel headcount and our acquisition of EKU. As a percentage of revenues, Cost of revenues—manufacturing was 86% for the year ended December 31, 2021, as compared to 87% for the year ended December 31, 2020. The decrease in Cost of revenues., exclusive of depreciation and depletion—Manufacturing as a percentage of revenues resulted from a relative increase in power end and fluid end sales, which have a higher margin, compared to flow iron.

Cost of revenues, exclusive of depreciation, depletion, and amortization—Proppant production. Cost of revenues, exclusive of depreciation, depletion, and amortization—Proppant production increased 131%, or \$8.0 million, to \$14.1 million for the year ended December 31, 2021, from \$6.1 million for the year ended December 31, 2020. The increase was primarily due to a 123% increase in proppant production combined with higher personnel and repairs and maintenance costs resulting from greater market demand and higher activity. As a percentage of revenues, Cost of revenues—Proppant production was 52% for the year ended December 31, 2021, as compared to 59% for the year ended December 31, 2020. The decrease in Cost of revenues, exclusive of depreciation and depletion—Proppant production as a percentage of revenues was primarily attributable to the 19% increase in proppant pricing.

Depreciation, depletion, and amortization. Depreciation, depletion, and amortization decreased 7%, or \$10.0 million, to \$140.7 million for the year ended December 31, 2021, from \$150.7 million for the year ended December 31, 2020. The decrease was primarily due to fully depreciated high-pressure iron associated with fleets manufactured in 2018 as well as a reduced number of tractor trucks eliminated through our ESC upgrade program.

Loss on disposal of assets, net. Loss on disposal of assets, net increased 16%, or \$1.3 million, to \$9.8 million for the year ended December 31, 2021, from \$8.4 million for the year ended December 31, 2020. The increase resulted from an increase in the early failure and disposal of components of our pressure pumping equipment as a result of higher activity levels.

Selling, general and administrative. Selling, general and administrative expenses increased 29%, or \$14.6 million, to \$65.6 million for the year ended December 31, 2021, from \$51.0 million for the year ended December 31, 2020. The increase was due to higher headcount and personnel costs associated with the increased demand of our stimulation services, higher insurance costs due to increased market rates associated with business and medical insurance.

Interest expense, net. Interest expense, net increased 11%, or \$2.5 million, to \$25.8 million for the year ended December 31, 2021, from \$23.3 million for the year ended December 31, 2020. The increase in interest expense, net was attributable to upsizing our term loan for the purchase of three effact licenses from USWS.

Other income. Other expense (income) decreased to \$0.1 million for the year ended December 31, 2021 from other income of \$0.3 million for the year ended December 31, 2020.

*Income tax benefit (provision).* Income tax benefit was \$0.2 million for the year ended December 31, 2021 compared to an income tax provision of \$0.6 million for the year ended December 31, 2020.

### Segment results

The performance of our segments is evaluated primarily on Adjusted EBITDA. For definition of thenon-GAAP financial measure of Adjusted EBITDA and reconciliation of Adjusted EBITDA to our most directly comparable financial measures calculated in accordance with GAAP, please read "Summary Unaudited Pro Forma Financial Data—Non-GAAP Financial Measures."

Adjusted EBITDA—Stimulation services. Adjusted EBITDA—Stimulation services increased 78%, or \$53.8 million, to \$122.6 million for the year ended December 31, 2021, from \$68.8 million for the year ended December 31, 2020. The increase was primarily attributable to the impacts from the COVID-19 pandemic recovery resulting in an increase in customer activity.

Adjusted EBITDA—Manufacturing. Adjusted EBITDA—Manufacturing increased 8%, or \$0.1 million, to \$1.4 million for the year ended December 31, 2021, from \$1.3 million for the year ended December 31, 2020.

Adjusted EBITDA—Proppant production. Adjusted EBITDA—Proppant production increased 297%, or \$8.0 million, to \$10.7 million for the year ended December 31, 2021, from \$2.7 million for the year ended December 31, 2020. The increase was primarily attributable to a 123% increase in proppant production and a 19% increase in proppant pricing resulting from increases in commodity prices and proppant demand in the Permian basin.

### Liquidity and Capital Resources

Historically, our primary sources of liquidity and capital resources have been borrowings under our Old ABL Credit Facility, cash flows from our operations and capital contributions from our shareholders. Our primary uses of capital have been investing in and maintaining our property and equipment and repaying indebtedness.

Since our IPO, we expect that our primary sources of liquidity and capital resources will be cash on hand, including net proceeds of our IPO, cash flows generated by operating activities and borrowings under our credit facilities. We expect that our primary uses of capital will be to continue to fund our operations, support organic and strategic growth opportunities and satisfy future debt payments.

Due our determination that Flotek is a VIE of which the Company is the primary beneficiary, Flotek's financial statements have been included in our consolidated financial statements from May 17, 2022. However, we do not have the ability to access or deploy Flotek's cash or liquidity in our operations and, accordingly, have excluded Flotek's cash and other sources of liquidity from the following discussion of our liquidity and capital resources. See Note 14 – Acquisitions and investments in the notes to our consolidated financial statements for additional discussion related to the acquisition of Flotek.

Based on our current cash and cash equivalents balance, operating cash flow, current availability under our credit facilities, and the ongoing actions and financing alternatives discussed below, we believe that we will be able to maintain sufficient liquidity to fund our planned acquisitions and capital expenditures, satisfy our obligations, and remain in compliance with our existing debt covenants through the next twelve months and beyond.

## Acquisition Financing

We may need to incur significant additional indebtedness in order to finance the Merger, which will limit our operating flexibility, and could adversely affect our operations and financial results and prevent us from fulfilling our obligations. The Merger Agreement requires us to use commercially reasonable efforts to obtain and consummate a financing to fund the repayment in connection with the closing of the Merger of certain indebtedness of USWS and its subsidiaries. Our obtaining the financing necessary to consummate the Merger is not a condition to closing the Merger.

As further described below, on July 25, 2022, we amended (i) the New Term Loan Facility to increase the size of the New Term Loan Facility by \$150.0 million with an uncommitted option to obtain commitments for a potential additional \$100.0 million of delayed draw loans and (ii) the New ABL Credit Facility to add an uncommitted \$100.0 million incremental facility and increase the potential size of the New ABL Credit Facility from \$200 million to \$300 million. We used a portion of the proceeds from the increased New Term Loan Facility to fund the acquisition of SP Companies and we anticipate using the remainder of the proceeds and operating cash to finance the Merger, pay outstanding debt under New ABL Credit Facility and/or for other general corporate purposes.

In addition to the approximately \$270 million in total stock consideration payable to USWS stockholders and holders of USWS equity awards, the repayment of approximately \$170 million of USWS debt expected to be outstanding at closing, and the payment of expenses and other costs incurred in connection with the Merger, we

expect to assume approximately \$55 million of indebtedness in connection with the Merger. Although we expect to fund these costs through the New Term Loan Credit Facility and/or the New ABL Credit Facility, available cash, additional loans and/or notes and cash flows generated from operations, we cannot assure that these sources will be sufficient, or that we will be able to refinance any of our outstanding indebtedness, and/or obtain additional financing on favorable terms, or at all, or that our cash flows generated from operations will be sufficient, in order to satisfy our obligations in connection with the Merger, or otherwise. For additional discussion of risks relating to our liquidity needs, the Merger and our anticipated additional indebtedness, see "Risks Factors" in this proxy statement/information statement/prospectus.

## Tax Receivable Agreement

On May 17, 2022, in connection with our IPO, we entered the Tax Receivable Agreement with the TRA Holders. The TRA generally provides for the payment by us to each TRA Holder of 85% of the net cash savings, if any, in U.S. federal, state and local income tax and franchise tax (computed using simplifying assumptions to address the impact of state and local taxes) that we actually realize (or are deemed to realize in certain circumstances) as a result of (i) certain increases in tax basis that occur as a result of our acquisition (or deemed acquisition for U.S. federal income tax purposes) of all or a portion of such TRA Holder's common units of ProFrac LLC in connection with our IPO or pursuant to an exercise of the Redemption Right (as defined in the Tax Receivable Agreement) or the Call Right (as defined in the Tax Receivable Agreement) and (ii) imputed interest deemed to be paid by us as a result of, and additional tax basis arising from, any payments we make under the Tax Receivable Agreement. We depend on ProFrac LLC to make distributions to us in an amount sufficient to cover our obligations under the Tax Receivable Agreement.

Payments will generally be made under the Tax Receivable Agreement as we realize actual cash tax savings from the tax benefits covered by the Tax Receivable Agreement. However, if we experience a Change of Control (as defined in the Tax Receivable Agreement), or the Tax Receivable Agreement otherwise terminates early, our obligations under the Tax Receivable Agreement would accelerate and we would be required to make an immediate payment equal to the present value of the anticipated future payments to be made by us under the Tax Receivable Agreement (determined by applying a discount rate). Any such payment is expected to be substantial.

# Our 2022 Capital Budget

Our 2022 capital expenditure budget, excluding acquisitions, is estimated to be in a range between \$265 million and \$290 million, which represents the high end of the range provided previously, due to increased activity levels and costs. We have budgeted approximately \$65 million to \$70 million to construct three electric-powered fleets. We are fully committed to building the three electric-powered fleets and have several customers interested in contracting these fleets. We intend to align fleet construction and other growth capital expenditures with visible customer demand, by strategically deploying new equipment in response to inbound customer requests and industry trends. Also included in our 2022 capital expenditure budget is \$25 million to \$30 million to construct the West Munger sand mine. We estimate maintenance capital expenditures to be \$3.0 million per fleet per year. We expect that the remainder of our 2022 capital expenditure budget, excluding any potential acquisitions, will be used to fund other growth initiatives such as upgrading Tier II fleets to Tier IV dual fuel fleets. We continually evaluate our capital expenditures and the amount that we ultimately spend will depend on a number of factors, including customer demand for new fleets and expected industry activity levels. We believe we will be able to fund our 2022 capital program from cash flows from operations.

# Working Capital

Our working capital was \$283.7 million and \$5.0 million as of June 30, 2022, and December 31, 2021, respectively. The \$278.7 million increase in working capital was primarily due to higher activity levels in the first six months of 2022.

#### Cash and Cash Flows

The following table sets forth the historical cash flows for the six months ended June 30, 2022 and 2021:

	Six Months En	ded June 30,
(In thousands)	2022	2021
Net cash provided by operating activities	\$ 84,464	\$ 26,780
Net cash used in investing activities	(388,330)	(38,451)
Net cash provided by financing activities	374,244	26,101
Net increase in cash, cash equivalents, and restricted cash	\$ 70,378	\$ 14,430

## **Operating Activities**

Net cash provided by operating activities was \$84.5 million and \$26.8 million for the six months ended June 30, 2022 and 2021, respectively. The increase was primarily due to higher activity levels and profitability, offset by working capital increases in the first and second quarters of 2022.

### Investing Activities

Net cash used in investing activities was \$388.3 million and \$38.5 million for the six months ended June 30, 2022 and 2021, respectively. The increase was primarily due to \$279.0 million of cash paid for the FTSI acquisition, our \$46.0 million investment in BPC, and higher capital expenditures of \$62.5 million related to dual fuel engine upgrades, ESC installations, and our electric frac fleet build program. These uses of cash were partially offset by cash proceeds from the FTSI Sale Leaseback of real property.

#### Financing Activities

Net cash provided by financing activities was \$374.2 million and \$26.1 million for the six months ended June 30, 2022, and 2021, respectively. The increase in cash provided by financing activities was due to proceeds from the issuance of long-term debt associated with refinancing transactions related to the FTSI acquisition, proceeds from upsizing the New ABL Credit Facility (as described below), and proceeds from our IPO. These proceeds were offset by payments against our various debt instruments.

Our cash and cash equivalents were \$5.4 million and \$3.0 million at December 31, 2021 and December 31, 2020, respectively.

The following table sets forth the historical cash flows for the years ended December 31, 2021 and 2020:

	Y ear ended		
	Decemb	December 31,	
	2021	2020	
	(\$ in tho	(\$ in thousands)	
Net cash provided by operating activities	\$ 43,942	\$ 45,054	
Net cash (used in) investing activities	\$(78,383)	\$(44,617)	
Net cash provided by (used in) financing activities	\$ 36,865	<u>\$(15,322)</u>	
Net increase (decrease) in cash and equivalents	\$ 2,424	<u>\$(14,885)</u>	

## Operating activities

Net cash provided by operating activities was \$43.9 million and \$45.1 million for the years ended December 31, 2021 and 2020, respectively. Cash flows from operations were essentially flat year over year. We had a large decrease in net loss, offset by increases in accounts receivable and inventory due to the recovery of

the business. The increases in accounts receivable and inventory were offset by smaller increases in accounts payable and accrued expenses.

#### Investing activities

Net cash used in investing activities was \$78.4 million and \$44.6 million for the years ended December 31, 2021 and 2020, respectively. The increase was primarily due to higher capital expenditures related to dual fuel engine upgrades, ESC installations and efrac build program.

#### Financing activities

Net cash provided by financing activities was \$36.9 million for the year ended December 31, 2021, compared to net cash used in financing activities of \$15.3 million for the year ended December 31, 2020. In 2021, we had net borrowings of debt of \$41.3 million, compared to net borrowings of debt of \$13.7 million in 2020. The increase in cash provided by financing activities was primarily due to the \$40.0 million expansion of our Old Term Loan Credit Facility.

## Credit Facilities and Other Financing Arrangements

New ABL Credit Facility

On March 4, 2022, ProFrac LLC, ProFrac II LLC, as borrower (in such capacity, the "ABL Borrower"), and certain of the ABL Borrower's wholly owned subsidiaries as obligors, entered into a senior secured asset-based revolving credit agreement (as amended, the "New ABL Credit Facility"), with a group of lenders with JPMorgan Chase Bank N.A., as administrative agent and collateral agent. The New ABL Credit Facility initially provided for an asset-based revolving credit facility with a borrowing base and lender commitments of \$100.0 million. On April 8, 2022, the New ABL Credit Facility was amended to increase the borrowing base and lender commitments to \$200.0 million. The New ABL Credit Facility has a borrowing base composed of certain eligible accounts receivable and eligible inventory less customary reserves, as redetermined monthly. As of June 30, 2022, the maximum availability under the New ABL Credit Facility was the aggregate lender commitments of \$200.0 million with \$143.4 million of borrowings outstanding and \$9.2 million of letters of credit outstanding, resulting in approximately \$47.4 million of remaining availability. The New ABL Credit Facility matures on the earlier of (i) March 4, 2027 and (ii) 91 days prior to the stated maturity of any material indebtedness (other than the First Financial Loan).

Borrowings under the New ABL Credit Facility accrue interest based on a three-tier pricing grid tied to average historical availability, and the ABL Borrower may elect for loans to be based on either an Adjusted Term SOFR or a base rate, plus the applicable margin. The interest rate under the New ABL Credit Facility for (a) Adjusted Term SOFR is the applicable margin plus the fluctuating per annum rate equal to Adjusted Term SOFR (with an Adjusted Term SOFR Floor of 0.00%); and (b) Base Rate Loans are the applicable margin plus the fluctuating per annum rate equal to the greatest of the Prime Rate in effect on such day, or the NYFRB Rate in effect on such day plus ½% of 1% and the Adjusted Term SOFR for aone-month Interest Period as published two (2) U.S. Government Securities Business Days (as defined in the New ABL Credit Facility) prior to such day (or if such day is not a business day, the immediately preceding business day), plus 1.0%. The applicable margin for Adjusted Term SOFR Loans ranges from 1.50% to 2.00% and for Base Rate Loans ranges from 0.50% to 1.00%, depending on the average daily availability over the last three months under the New ABL Credit Facility.

The New ABL Credit Facility bears an unused line fee ranging from 0.250% to 0.375%, depending on the average daily availability over the last three months payable quarterly in arrears. The New ABL Credit Facility also bears customary letter of credit fees. The interest rate was 5.50% as of June 30, 2022.

The New ABL Credit Facility is subject to customary mandatory prepayments, including a mandatory prepayment if the aggregate unpaid principal balance of revolving loans, agent advances, swingline borrowings,

unreimbursed drawings under letters of credit and the undrawn amount of outstanding letters of credit exceeds at any time the lesser of (x) the then applicable borrowing base and (y) the then total effective commitments under the New ABL Credit Facility, in an amount equal to such excess. After the occurrence and during the continuance of a Cash Dominion Period (defined in the New ABL Credit Facility as (a) any period commencing upon the date that availability shall have been less than the greater of (i) 12.5% of the Maximum Credit (which is the lesser of the maximum revolver amount in effect at such time and the borrowing base at such time) and (ii) \$10.0 million for a period of five consecutive business days and continuing until the date on which availability shall have been at least the greater of (y) 12.5% of the Maximum Credit and (z) \$10.0 million for 20 consecutive calendar days or (b) any period commencing on the occurrence of certain specified events of default, and continuing during the period that such specified event of default shall be continuing).

The New ABL Credit Facility contains certain customary representations and warranties and affirmative and negative covenants. The negative covenants include, subject to customary exceptions, limitations on indebtedness, dividends, distributions and certain other payments, investments, acquisitions, prepayments of specified junior indebtedness, amendments of specified junior indebtedness, transactions with affiliates, dispositions, mergers and consolidations, liens, restrictive agreements, sale and leaseback transactions, changes in fiscal periods and changes in line of business.

We are required by the New ABL Credit Facility to maintain minimum liquidity of \$5.0 million at all times. Additionally, when availability is less than the greater of (i) 12.5% of the maximum credit (which is the lesser of the maximum revolver amount in effect at such time and the borrowing base at such time) and (ii) \$10.0 million and continuing until such time as availability has been in excess of such threshold for a period of 20 consecutive calendar days, we are required by the New ABL Credit Facility to maintain a springing Fixed Charge Coverage Ratio (as defined in the New ABL Credit Facility) of at least 1.0 to 1.0, which is tested quarterly during such period. ProFrac LLC was in compliance with all covenants, and there were no existing defaults or events of default related to the New ABL Credit Facility as of June 30, 2022.

On July 25, 2022, the New ABL Credit Facility was amended to add an uncommitted \$100.0 million incremental facility (the "Incremental Facility"), under the terms of which existing lenders can make additional loans (in their sole discretion) under, or new lenders can join, the Incremental Facility and increase the potential size of the New ABL Credit Facility from \$200 million to \$300 million, subject to satisfaction of certain conditions. All other terms and conditions of the New ABL Credit Facility remained substantially unchanged.

New Term Loan Credit Facility

On March 4, 2022, ProFrac LLC, ProFrac II LLC, as borrower (in such capacity, the "Term Loan Borrower"), and certain of the Term Loan Borrower's wholly owned subsidiaries as obligors, entered into a senior secured term loan credit agreement (the "New Term Loan Credit Facility"), with Piper Sandler Finance LLC, as administrative agent and collateral agent (the "Term Loan Agent"), and the lenders party thereto. The New Term Loan Credit Facility provides for a term loan facility in an aggregate principal amount of \$450.0 million. In the second quarter of 2022, the New Term Loan Facility was paid down by \$143.8 million with net proceeds from our IPO. As of June 30, 2022, the Term Loan Borrower had approximately \$450.0 million outstanding under the New Term Loan Credit Facility. The New Term Loan Credit Facility matures on March 4, 2025.

Borrowings under the New Term Loan Credit Facility accrue interest at a percentage per annum equal to (a) until October 1, 2022, (i) for SOFR Rate Loans, 8.50%, and (ii) for Base Rate Loans, 7.50% and (b) thereafter, based on a three-tier pricing grid tied to Total Net Leverage Ratio (as defined in the New Term Loan Credit Facility), and the Term Loan Borrower may elect for loans to be based on either Adjusted Term SOFR or Base Rate, plus the applicable margin. The interest rate on the New Term Loan Credit Facility for (a) SOFR Rate Loans are the applicable margin plus the fluctuating per annum rate equal to Adjusted Term SOFR (as defined in the New Term Loan Credit Facility), with a SOFR floor of 1.00% and (b) Base Rate Loans

are the applicable margin plus the fluctuating per annum rate equal to the highest of (i) the federal funds rate plus 1/2 of 1%, (ii) the interest rate quoted in the print edition of The Wall Street Journal, Money Rates Section, as the prime rate in effect, (iii) Adjusted Term SOFR for a one-month interest period as determined on such day, plus 1.0% and (iv) 2.00%.

The applicable margin for (a) SOFR Rate Loans ranges from 6.50% to 8.00% and (b) Base Rate Loans ranges from 5.50% to 7.00%, depending on the Total Net Leverage Ratio (as defined in the New Term Loan Credit Facility) as of the first day of the then-current fiscal quarter.

The New Term Loan Credit Facility is guaranteed by ProFrac LLC and all of the Term Loan Borrower's material existing subsidiaries and certain direct and indirect future U.S. restricted subsidiaries of the Term Loan Borrower. The New Term Loan Credit Facility is secured by a lien on, and security interest in, substantially all of each such guarantor's assets, which consists of:

- a perfected security interest in all present and after-acquired equipment, fixtures, fracturing equipment (in each case of the foregoing, specifically excluding Fracturing Equipment Parts), real estate, intellectual property, equity interests in all direct and indirect subsidiaries of any grantor, intercompany loans of the grantors and/or their subsidiaries, all other assets whether real, personal or mixed, to the extent not constituting ABL Priority Collateral, and, except to the extent constituting ABL Priority Collateral and to the extent evidencing or otherwise related to such items, all documents, general intangibles, instruments, investment property, commercial tort claims, letters of credit, letter-of-credit rights and supporting obligations, all books, records and documents relating to, or arising from, any of the foregoing, in each case, except to the extent such proceeds constitute ABL Priority Collateral, any fixed asset priority;
- proceeds account and the proceeds of any of the foregoing, business interruption insurance proceeds and subject to customary
  exceptions and exclusions (collectively, the "Fixed Asset Priority Collateral"), which security interest is senior to the security interest
  in the foregoing assets securing the New ABL Credit Facility; and
- a perfected security interest in the ABL Priority Collateral, which security interest is junior to the security interest in the ABL Priority Collateral securing the New ABL Credit Facility.

The respective rights of the New Term Loan Credit Facility lenders and the New ABL Credit Facility lenders in the ABL Priority Collateral and the Fixed Asset Priority Collateral are governed by an intercreditor agreement between the Term Loan Agent and the ABL Agent.

The New Term Loan Credit Facility is subject to quarterly amortization beginning in June 2022, though any excess cash flow payments, reduce the required amortization.

Additionally, the New Term Loan Credit Facility is subject to a quarterly mandatory prepayment beginning for the calendar quarter ending on December 31, 2022 in an amount equal to the Applicable ECF Percentage (as defined in the New Term Loan Credit Facility). The Applicable ECF Percentage ranges from 50% of Excess Cash Flow (as defined in the New Term Loan Credit Facility) to 25% of Excess Cash Flow depending on the Total Net Leverage Ratio as of the last day of the applicable fiscal quarter. The New Term Loan Credit Facility is subject to the right of the Term Loan Borrower and its restricted subsidiaries to reinvest such proceeds within a specified period of time, and certain other exceptions.

Voluntary prepayments of borrowings under the New Term Loan Credit Facility are permitted at any time, in specified minimum principal amounts, subject to reimbursement of the lenders' redeployment costs actually incurred in the case of a prepayment of SOFR Rate Loans other than on the last day of the relevant interest period. Between March 4, 2022 and March 4, 2023, certain prepayments of the New Term Loan Credit Facility are subject to a prepayment premium of 3.00% (or, in the case of any IPO Prepayment (as defined in the New Term Loan Credit Facility), 2.00%). Between March 5, 2023 and March 4, 2024, certain prepayments of the New

Term Loan Credit Facility are subject to a 2.00% prepayment premium. After March 4, 2024, but prior to the Stated Termination Date (as defined in the New Term Loan Credit Facility) certain prepayments of the New Term Loan Credit Facility are subject to a 1.00% prepayment premium. No payment or prepayment premium shall be due on account of any payments or prepayments made on the Stated Termination Date.

The New Term Loan Credit Facility contains certain customary representations and warranties and affirmative and negative covenants. The negative covenants include, subject to customary exceptions, limitations on indebtedness, dividends, distributions and certain other payments, investments, capital expenditures, acquisitions, prepayments of specified junior indebtedness, amendments of specified junior indebtedness, transactions with affiliates, dispositions, mergers and consolidations, liens, restrictive agreements, changes in fiscal periods and changes in line of business.

We are required by the New Term Loan Credit Facility to maintain a Total Net Leverage Ratio (as defined in the New Term Loan Credit Facility) (i) of no more than 2.00 to 1.00 for the fiscal quarter ending on June 30, 2022, (ii) of no more than 1.55 to 1.00 for the fiscal quarters ending on September 30, 2022 and December 31, 2022, and (iii) of no more than 1.25 for each fiscal quarter ending on March 31, 2023 and thereafter.

We are required by the New Term Loan Credit Facility to maintain minimum liquidity of \$30.0 million at all times.

The New Term Loan Credit Agreement prohibits Capital Expenditures (as defined in the New Term Loan Credit Facility) in excess of (i) in the case of the Fiscal Year ended December 31, 2022, the greater of (x) \$275,000,000 in the aggregate and (y) in the case of any period of four consecutive fiscal quarters ending in such Fiscal Year, 50% of Consolidated EBITDA for the Test Period most recently ended prior to the date of the applicable Capital Expenditure, and (ii) in the case of any period of four consecutive fiscal quarters ending thereafter, commencing with the period of four consecutive fiscal quarters ending on March 31, 2023, an aggregate amount equal to 50.0% of Consolidated EBITDA for the Test Period most recently ended prior to the date of the applicable Capital Expenditure, provided that if the amount of the Capital Expenditures permitted to be made in any Fiscal Year is greater than the actual amount of the Capital Expenditures actually made in such Fiscal Year, then up to \$20,000,000 of such excess amount may be carried forward to the next succeeding Fiscal Year.

The New Term Loan Credit Facility contains customary events of default. If an event of default occurs and is continuing, the lenders may declare all loans to be immediately due and payable. Some events of default require an automatic termination of the loans and become immediately due and payable. We were in compliance with all covenants, and there were no defaults or events of default related to the New Term Loan Credit Facility as of June 30, 2022

On July 25, 2022, the New Term Loan Facility was amended to increase the size of the New Term Loan Facility by \$150.0 million with an uncommitted option to obtain commitments for a potential additional \$100.0 million of delayed draw loans before the earlier to occur of (i) the consummation of the Merger and (ii) March 31, 2023. The maturity date of the New Term Loan Facility remains the same at March 4, 2025. We used a portion of the proceeds from the increased New Term Loan Facility to fund the SPS Acquisition and anticipate using the remainder of the proceeds and operating cash to facilitate the Merger, pay outstanding debt under the New ABL Credit Facility and/or for other general corporate purposes.

## First Financial Loan

On December 22, 2021, ProFrac II LLC entered into a \$30.0 million loan agreement with First Financial Bank, N.A. with a stated maturity date of January 1, 2024 (the "First Financial Loan"). The First Financial Loan bears interest at a variable rate based on the Prime rate published by the Wall Street Journal, floating daily. As of June 30, 2022, ProFrac II LLC had \$24.0 million outstanding under the First Financial Loan.

The First Financial Loan is secured by a first lien on, and security interest in, certain truck tractors and all other trailers, trucks and vehicles owned by ProFrac Services and ProFrac Manufacturing, in each case, more particularly described in the security agreement, and is guaranteed by ProFrac Services, ProFrac Manufacturing and ProFrac LLC.

The First Financial Loan is subject to monthly principal amortization beginning in February 2022.

The First Financial Loan contains certain restrictive covenants which require ProFrac LLC to maintain a Total Net Leverage Ratio, as defined in the loan agreement, of no greater than 3.00:1.00, and a Fixed Charge Coverage Ratio, as defined in the loan agreement, of at least 1.00:1.00.

#### Equify Bridge Note

On March 4, 2022, ProFrac II LLC entered into a \$45.8 million subordinated promissory note with Equify Financial with a stated maturity date of March 4, 2027 (the "Equify Bridge Note"). The Equify Bridge Note bears interest at a percentage per annum equal to 1.0%. Interest under the Equify Bridge Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Equify Bridge Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash. In April 2022, the Company repaid \$25.0 million in principal under the Equify Bridge Note.

The Equify Bridge Note is unsecured and subordinated to the indebtedness owing under the New ABL Credit Facility and the New Term Loan Credit Facility.

Until the New ABL Credit Facility and the New Term Loan Credit Facility have been terminated, prepayments of principal under the Equify Bridge Note are permitted solely to the extent permitted under the New ABL Credit Facility and New Term Loan Credit Facility. After the New ABL Credit Facility and the New Term Loan Credit Facility have been terminated, prepayments may be made at any time without prepayment penalty or premium.

In June 2022, the Equify Bridge Note was fully paid with net proceeds from our IPO.

## Backstop Note

On March 4, 2022, ProFrac LLC entered into a \$22.0 million subordinated promissory note with THRC Holdings with a stated maturity date of March 4, 2027 (the "Backstop Note"). The Backstop Note bears interest at a percentage per annum equal to 1.74%. Interest under the Backstop Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Backstop Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash.

The Backstop Note is unsecured and subordinated to the indebtedness owing under the New ABL Credit Facility and the New Term Loan Credit Facility.

Until the New ABL Credit Facility and the New Term Loan Credit Facility have been terminated, prepayments of principal under the Backstop Note are permitted solely to the extent permitted under the New ABL Credit Facility and New Term Loan Credit Facility. After the New ABL Credit Facility and the New Term Loan Credit Facility have been terminated, prepayments may be made at any time without prepayment penalty or premium.

In June 2022, the Backstop Note was fully paid with net proceeds from our IPO.

#### Closing Date Note

On March 4, 2022, ProFrac LLC entered into a \$22.0 million subordinated promissory note with THRC Holdings with a stated maturity date of March 4, 2027 (the "Closing Date Note"). The Closing Date Note bears interest at a percentage per annum equal to 1.74%. Interest under the Closing Date Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Closing Date Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash.

The Closing Date Note is unsecured and subordinated to the indebtedness owing under the New ABL Credit Facility and the New Term Loan Credit Facility.

Until the New ABL Credit Facility and the New Term Loan Credit Facility have been terminated, prepayments of principal under the Closing Date Note are permitted solely to the extent permitted under the New ABL Credit Facility and New Term Loan Credit Facility. After the New ABL Credit Facility and the New Term Loan Credit Facility have been terminated, prepayments may be made at any time without prepayment penalty or premium.

In June 2022, the Closing Date Note was fully paid with net proceeds from the IPO.

# Contractual Obligations

The following table summarizes the principal maturity schedule for our long-term debt outstanding as of June 30, 2022:

	2022	2023	2024	2025	2026	Thereafter	Total
First Financial loan	\$ 8,659	\$15,291	\$ —	\$ —	\$	\$ —	\$ 23,950
New ABL Credit Facility		_	_	_	_	143,350	143,350
New Term Loan Credit Facility	7,560	15,120	15,120	264,580	_	_	302,380
Flotek Convertible Notes	_	12,739	_	_	_	_	12,739
Other	7,803	1,908	1,907	543	79	386	12,626
Total	\$ 24,022	\$ 45,058	\$ 17,027	\$ 265,123	\$ 79	\$ 143,736	\$ 495,045

# Capital Expenditures

During the six months ended June 30, 2022 and 2021, our capital expenditures were \$116.1 million and \$53.6 million, respectively. We currently expect our capital expenditures to increase in 2022 and 2023, which are expected to be funded with cash flows from operations. The primary drivers of the increase are building electric-powered hydraulic fracturing fleets, continued engine upgrades as part of our ESG initiatives, and deployment costs associated with reactivating hydraulic fracturing fleets.

## Customer concentration

For the year ended December 31, 2021, sales to Rockcliff Energy Management, LLC accounted for 15.4% of total revenue.

# Off-Balance Sheet Arrangements

From time-to-time, we enter into off-balance sheet arrangements and transactions that can give rise tooff-balance sheet obligations. As of June 30, 2022 and December 31, 2021, the off-balance sheet arrangements and transactions that we have entered into include undrawn letters of credit. We do not believe that these arrangements are reasonably likely to materially affect our liquidity or availability of, or requirements for, capital resources.

#### Quantitative and qualitative disclosure of market risks

Market risk is the risk of loss arising from adverse changes in market rates and prices. Historically, our risks have been predominantly related to potential changes in the fair value of our long-term debt due to fluctuations in applicable market interest rates. Going forward our market risk exposure generally will be limited to those risks that arise in the normal course of business, as we do not engage in speculative, non-operating transactions, nor do we utilize financial instruments or derivative instruments for trading purposes.

#### Commodity price risk

Our material and fuel purchases expose us to commodity price risk. Our material costs primarily include the cost of inventory consumed while performing our pressure pumping services such as proppants, chemicals, trucking and fluid supplies. For our manufacturing segment, our material costs primarily include the cost of steel. For our proppant production segment, our material costs primarily include the cost of fuel. Our fuel costs consist primarily of diesel fuel used by our trucks, frac fleets and other motorized equipment. The prices for fuel and the raw materials in our inventory are volatile and are impacted by changes in supply and demand, as well as market uncertainty and regional shortages. Additionally, the market for our products and services is indirectly exposed to fluctuations in the prices of oil and natural gas to the extent such fluctuations impact well completion activity levels. Historically, we have generally been able to pass along price increases to our customers; however, we may be unable to do so in the future. We do not engage in commodity price hedging activities.

#### Interest rate risk

We are subject to interest rate risk on our variable rate debt from the New Term Loan Credit Facility and the New ABL Credit Facility. We also have fixed rate debt but does not currently utilize derivative instruments to manage the economic effect of changes in interest rates. The impact of a 1% increase in interest rates on our outstanding debt as of June 30, 2022 would have resulted in an annual increase in interest expense of approximately \$4.5 million.

#### Credit risk

Financial instruments that potentially subject us to concentrations of credit risk are trade receivables. We extend credit to customers and other parties in the normal course of business. We have established various procedures to manage our credit exposure, including credit evaluations and maintaining an allowance for doubtful accounts.

## Recent accounting pronouncements

We have not yet implemented Financial Accounting Standards Board ("FASB") Accounting Standards Update ("ASU") No. 2016-13, Financial Instruments—Credit Losses. The ASU introduces a new accounting model, the Current Expected Credit Losses model ("CECL"), which requires earlier recognition of credit losses and additional disclosures related to credit risk. The CECL model utilizes a lifetime expected credit loss measurement objective for the recognition of credit losses for loans and other receivables at the time the financial asset is originated or acquired. The expected credit losses are adjusted each period for changes in expected lifetime credit losses.

This model replaces the multiple existing impairment models previously used under U.S. GAAP, which generally require that a loss be incurred before it is recognized. The new standard also applies to financial assets arising from revenue transactions such as contract assets and accounts receivable. Implementation of this standard is currently required for fiscal years beginning after December 15, 2022. The Company does not believe implementation will have a material impact on its financial statements.

We have not yet implemented FASB ASU No. 2019-12, Income Taxes (Topic 740)—Simplifying the Accounting for Income Taxes, which removes specific exceptions to the general principles in Topic 740 under

GAAP. The new guidance also improves the application of income tax-related guidance and simplifies GAAP for franchise taxes that are partially based on income, transactions with a government that result in a step up in the tax basis of goodwill, separate financial statements of legal entities that are not subject to tax, and enacted changes in tax laws in interim periods. The new guidance will be effective for fiscal years beginning after December 15, 2021, and interim periods within fiscal years beginning after December 15, 2022; however, early adoption is permitted. The Company is currently evaluating the impact of adopting the new guidance on its consolidated financial statements.

# **Emerging growth company**

We qualify as an "emerging growth company" pursuant to the provisions of the JOBS Act, enacted on April 5, 2012. Section 102 of the JOBS Act provides that an "emerging growth company" can take advantage of the extended transition period provided in Section 7(a)(2)(B) of the Securities Act for complying with new or revised accounting standards.

We may take advantage of these provisions until we are no longer an emerging growth company, which will occur on the earliest of (i) the last day of the fiscal year following the fifth anniversary of our IPO, (ii) the last day of the fiscal year in which we have more than \$1.07 billion in annual revenue which, as a result of the FTSI Acquisition, we expect may occur as of December 31, 2022, (iii) the date on which we issue more than \$1 billion of non-convertible debt over a three-year period and (iv) the date on which we are deemed to be a "large accelerated filer," as defined in Rulel 2b-2 promulgated under the Exchange Act.

# Critical accounting policies and estimates

The discussion and analysis of our financial condition and results of operations is based on our consolidated financial statements, which have been prepared in accordance with accounting principles generally acceptable in the United States of America. The preparation of these financial statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the dates of the financial statements and the reported revenues and expenses during the reporting periods. We evaluate these estimates and assumptions on an ongoing basis and base our estimates on historical experience, current conditions and various other assumptions that we believe to be reasonable under the circumstances. The results of these estimates form the basis for making judgments about the carrying values of assets and liabilities as well as identifying and assessing the accounting treatment with respect to commitments and contingencies. Our actual results may materially differ from these estimates.

Listed below are the accounting policies that we believe are critical to our financial statements due to the degree of uncertainty regarding the estimates or assumptions involved, and that we believe are critical to the understanding of our operations.

Property, plant, and equipment

Our property and equipment are recorded at cost, less accumulated depreciation.

Upon sale or retirement of property and equipment, the cost and related accumulated depreciation are removed from the balance sheet and the net amount, less proceeds from disposal, is recognized as a gain or loss in earnings.

The estimated useful lives and salvage values of property and equipment is subject to key assumptions such as maintenance, utilization and job variation. Unanticipated future changes in these assumptions could negatively or positively impact our net income. The determination of the appropriate useful life of our property and equipment requires significant judgment resulting from the demanding operating environments in which we conduct our business as well as the significant volatility and demand fluctuations we have seen in our industry in recent years. A significant change in our established useful lives could cause depreciation expenses to fluctuate materially.

Depreciation of property and equipment is provided on the straight-line method over the following estimated useful lives:

Machinery and equipment	2 -10 years
Office equipment, software and other	3 - 7 years
Buildings and leasehold improvements	2 -40 years

## Impairment of Long-Lived Assets

In accordance with Financial Accounting Standards Board (FASB) ASC 360, Accounting for the Impairment or Disposal of Long-Lived Assets, we review our long-lived assets to be held and used whenever events or circumstances indicate that the carrying value of those assets may not be recoverable. An impairment loss is indicated if the sum of the expected future undiscounted cash flows attributable to the assets is less than the carrying amount of such assets. In this circumstance, we recognize an impairment loss for the amount by which the carrying amount of the asset exceeds the fair value of the asset. Our cash flows forecasts require us to

make certain judgements regarding long-term forecasts of future revenue and costs and cash flows related to the assets subject to review. Our fair value estimates for certain long-lived assets require us to use significant other observable and unobservable inputs among others including assumptions related to replacement cost based on actual recent auction sales of comparable equipment. The estimates of fair value are also subject to significant variability, are sensitive to changes in market conditions, and are reasonably likely to change in the future. For these reasons, the evaluation of recoverability of our long-lived assets and the measurement of any impairment, as necessary, are considered critical accounting estimates.

#### **Business Combinations**

Business combinations are accounted for under the acquisition method of accounting. Under this method, the assets acquired and liabilities assumed are recognized at their respective fair values as of the date of acquisition. The excess, if any, of the acquisition price over the fair values of the assets acquired and liabilities assumed is recorded as goodwill. For significant acquisitions, we utilize third-party appraisal firms to assist us in determining the fair values for certain assets acquired and liabilities assumed. The measurement of these fair values requires us to make significant estimates and assumptions which are inherently uncertain.

Adjustments to the fair values of assets acquired and liabilities assumed are made until we obtain all relevant information regarding the facts and circumstances that existed as of the acquisition date (the "measurement period"), not to exceed one year from the date of the acquisition. We recognize measurement-period adjustments in the period in which we determine the amounts, including the effect on earnings of any amounts we would have recorded in previous periods if the accounting had been completed at the acquisition date.

The estimation of net assets acquired in business combinations requires significant judgment in determination of the fair value of the assets and liabilities acquired. Our fair value estimates require us to use significant observable and unobservable inputs. The estimates of fair value are also subject to significant variability, are sensitive to changes in market conditions, and are reasonably likely to change in the future. A significant change in the observable and unobservable inputs and determination of fair value of the assets and liabilities acquired could significantly impact our consolidated financial statements.

## Income Taxes

ProFrac Holding Corp. is a corporation and will be subject to U.S. federal, state and local income taxes. Although the ProFrac Predecessor entities are subject to franchise tax in the State of Texas, they have historically been treated as pass-through entities for U.S. federal and other state and local income tax purposes and as such were not subject to U.S. federal income taxes or other state or local income taxes. Rather, the tax liability with respect to the taxable income of the ProFrac Predecessor entities was passed through to their owners. Accordingly, prior to the March 2022 acquisition of FTSI, a corporation subject to U.S. federal and state income

tax, the financial data attributable to ProFrac Predecessor contains no provision for U.S. federal income taxes or income taxes in any state or locality (other than franchise tax in the State of Texas). Subsequent to the acquisition of FTSI in March 2022, the financial data of the ProFrac Predecessor does contain both U.S. federal and state income taxes. We estimate that we will be subject to U.S. federal and state taxes at a blended statutory rate of approximately 23% of pre-tax earnings. Additionally, with the acquisition of EKU, the Company is subject to certain foreign taxes, which were immaterial for the three and six months ended June 30, 2022.

Effective March 4, 2022 we account for income taxes under the asset and liability method. Under this method, deferred tax assets and liabilities are recognized for the estimated future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax basis. Deferred tax assets and liabilities are measured using enacted tax rates in effect for the year in which those temporary differences are expected to be recovered or settled pursuant to the provisions of Accounting Standards Codification 740, Income Taxes. The effect on deferred tax assets and liabilities of a change in tax rate is recognized in earnings in the period that includes the enactment date. Valuation allowances are established when necessary to reduce deferred tax assets to the amounts more likely than not to be realized.

We recorded a full valuation allowance on our net deferred tax assets based on our assessment that it is more likely than not that the deferred tax asset will not be realized. A change in these assumptions could cause a decrease to the valuation allowance, which could materially impact our results of operations.

#### Variable Interest Entities

We evaluate our ownership, contractual and other interest in entities to determine if they are VIE. We evaluate whether we have a variable interest in those entities and the nature and extent of those interests. Based on our evaluation, if we determine we are the primary beneficiary of a VIE, we consolidate the entity in our financial statements.

#### INFORMATION ABOUT USWS

Unless the context otherwise requires, all references in this subsection to "we," "us," "our" and the "Company" shall mean U.S. Well Services, Inc.

#### BUSINESS

#### Overview

We are one of the pioneer companies in developing the electric pressure pumping industry. We are based in Houston, Texas and provide our services to oil and natural gas E&P companies in the United States. We are one of the first companies to develop and commercially deploy electric-powered pressure pumping technology ("Clean Fleet®"), which we believe is an industry-changing technology. Our Clean Fleet® technology has a demonstrated track record for successful commercial operation, with over 29,900 electric pressure pumping stages completed since the first Clean Fleet® was deployed in July 2014. Our Clean Fleet® technology is supported by an intellectual property portfolio, consisting of 81 granted patents and an additional 249 pending patents. We believe that the following characteristics of the Clean Fleet® technology provide the Company with a distinct competitive advantage:

- · Industry Leading Environmental Profile
- · Reduced Fuel Costs
- Enhanced Wellsite and Community Safety Profile
- · Lower Cost of Ownership
- · Superior Operational Efficiency Profile

In May 2021, we announced the next generation of our Clean Flee® technology with the unveiling of our newly designed Nyx Clean Flee®. Nyx Clean Fleet® will use our patented PowerCube, driving two independently controlled electric motors and frac pumps to provide 6,000 hydraulic horsepower on a single trailer. Our first Nyx Clean Fleet® was delivered in the second quarter of 2022 and commenced pressure pumping operations in July 2022.

Since our announcement in May 2021 of our commitment to becoming anall-electric pressure pumping services provider, we have sold most of our legacy, diesel-powered pressure pumping equipment. We have retained some of our legacy, diesel-powered pressure pumping equipment for use during our transition to support our electric fleets and bridge the time gap between our customers' current service needs and the deployment of our newbuild Nyx Clean Fleets\*. We expect to continue phasing out our remaining conventional fleet operations as we take delivery of our newbuild Nyx Clean Fleets\*.

We currently have five electric fleets representing 233,500 electric hydraulic horsepower and legacy, diesel-powered equipment representing 90,250 hydraulic horsepower.

Currently, we provide our services in the Appalachian, Permian and Uinta Basins. We have demonstrated the capability to expeditiously deploy our fleets to new oil and gas basins when requested by customers. Our customers include EQT Corporation, Range Resources, Olympus Energy and other leading E&P companies.

# Company Formation

On February 21, 2012, U.S. Well Services, LLC ("USWS LLC") was formed as a Delaware limited liability company. As part of a corporate restructuring in February 2017, all of the outstanding equity interest of USWS LLC was acquired by a newly formed entity, USWS Holdings, LLC ("USWS Holdings"), a Delaware limited liability company that was formed for the purposes of effecting the corporate restructuring and that had no

operations of its own. USWS Holdings was acquired by U.S. Well Services, Inc. (f/k/a Matlin & Partners Acquisition Corporation, or 'MPAC'') on November 9, 2018, as discussed further under "Business Combination" herein.

#### **Business Combination**

On March 10, 2016, MPAC was incorporated in Delaware as a special purpose acquisition company for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization, or other similar business combination with one or more target businesses. On March 15, 2017, MPAC consummated its initial public offering (the "IPO"), following which its shares began trading on the Nasdaq Capital Market ('Nasdaq").

On November 9, 2018, MPAC acquired USWS Holdings (the "Transaction") pursuant to a Merger and Contribution Agreement, dated as of July 13, 2018 (as amended, the "Merger and Contribution Agreement"). The Transaction was accounted for as a reverse recapitalization. Under this method of accounting, USWS Holdings was treated as the acquirer and MPAC was treated as the acquired party.

In connection with the closing of the Transaction, MPAC changed its name to U.S. Well Services, Inc. (USWS Inc.") and its trading symbols on Nasdaq from "MPAC," and "MPACW," to "USWS" and "USWSW".

Pursuant to the Merger and Contribution Agreement, on November 9, 2018, USWS Inc. issued Common Stock to certain members of USWS Holdings in exchange for their interests in USWS Holdings and Class B common stock to certain members of USWS Holdings who retained their interests in USWS Holdings.

Following the completion of the Transaction, the Company was organized as an "Up-C" structure, meaning that substantially all the Company's assets and operations are held and conducted by USWS LLC. The Transaction did not include a tax receivable agreement.

During the year ended December 31, 2021, the remaining outstanding shares of Class B common stock were converted into shares of Common Stock. As of December 31, 2021, there were no shares of Class B common stock issued and outstanding.

During the first quarter of 2021, the remaining noncontrolling interest holders of USWS Holdings exchanged all of their respective shares for USWS Common Stock. Accordingly, USWS Holdings became the wholly owned subsidiary of USWS as of December 31, 2021.

#### Organizational Structure

The following diagram illustrates the ownership structure of the Company as of June 30, 2022:



#### **Operations**

Our operations are organized into a single business segment, which consists of pressure pumping services, and we have one reportable geographical business segment, the United States.

#### Services

We provide pressure pumping services to E&P companies. Pressure pumping services are performed to enhance the production of oil and natural gas from formations with low permeability and restricted flow of hydrocarbons. Our customers benefit from our expertise in providing pressure pumping services on horizontal and vertical oil and natural gas-producing wells in shale and other unconventional geological formations.

The process of pressure pumping involves pumping a pressurized stream of fracturing fluid — typically a mixture of water, chemicals, and proppant — into a well casing or tubing to cause the underground mineral formation to fracture or crack. Fractures release trapped hydrocarbon particles and provide a conductive channel for the oil or natural gas to flow freely to the wellbore for collection. The propping agent, or proppant, becomes lodged in the cracks created by the pressure pumping process, "propping" them open to facilitate the flow of hydrocarbons from the reservoir to the well.

Our fleets consist mostly of all-electric, mobile pressure pumping equipment and other auxiliary heavy equipment to perform pressure pumping services. Our Clean Fleet® electric fleets replace the traditional engines, transmissions, and radiators used in conventional diesel fleets with electric motors powered by electricity generated by natural gas-fueled turbine generators. We utilize high-pressure hydraulic fracturing pumps mounted on trailers and refer to the group of pump trailers and other equipment necessary to perform a typical job as a "fleet" and the personnel assigned to each fleet as a "crew".

Since our announcement in May 2021 of our commitment to becoming anall-electric pressure pumping services provider, we have sold most of our legacy, diesel-powered pressure pumping equipment. We have retained some of our legacy, diesel-powered pressure pumping equipment for use during our transition to support our electric fleets and bridge the time gap between our customers' current service needs and the deployment of our newbuild Nyx Clean Fleets\*. We expect to continue phasing out our remaining conventional fleet operations as we take delivery of our newbuild Nyx Clean Fleets\*.

#### Clean Fleet® Technology

Our Clean Fleet® uses electric motors and other existing industry equipment to provide pressure pumping services with numerous advantages over conventional diesel fleets. Our Clean Fleet® technology is a proven technology that has been in commercial operations since July 2014, making us a leading provider of electric-powered pressure pumping services. Our Clean Fleet® technology is supported by an intellectual property portfolio. We have been granted, or have received notice of allowance, for 81 patents and have an additional 249 patents pending.

We believe Clean Fleet® technology provides the Company with a distinct competitive advantage over our competitors because of the following characteristics:

- Industry Leading Environmental Profile Clean Fleet® technology offers superior emissions performance as compared to traditional diesel-powered pressure pumping equipment.
- Reduced Fuel Costs Clean Fleet® technology uses electric motors to drive its frac pumps instead of internal combustion engines and transmissions used on traditional, diesel-powered pressure pumping equipment. Clean Fleet®'s electric motors are powered by mobile turbine generators or other power sources that can use natural gas produced and conditioned in the field ("Field Gas"), compressed natural gas ("CNG"), liquefied natural gas ("LNG") or diesel as a fuel source. Our customers typically provide Field Gas or CNG to fuel Clean Flee®'s turbine generators, which offers a significant cost savings relative to purchasing diesel fuel to power traditional pressure pumping equipment in the current market environment.
- Enhanced Wellsite and Community Safety Profile Clean Fleet® offers superior safety benefits relative to conventional diesel fleets for both workers at the wellsite and inhabitants of nearby communities.
- Lower Cost of Ownership We believe Clean Fleet® offers the best cost of ownership of any commercially available pressure pumping technology. Also, Clean Fleet® uses long-lived components such as electric motors, switchgears, and turbine generators that we estimate will be capable of operating for more than 15 years, while conventional diesel fleets can be expected to operate for less than five years before requiring major component replacements.
- Superior Operational Efficiency Profile Clean Fleet® technology allows us to garner operational efficiencies that are difficult to achieve with traditional diesel-powered pressure pumping technology. By eliminating moving parts associated with internal combustion engines and transmissions, as well as eliminating the need for preventative maintenance activities such as oil filter changes, we are able to reduce maintenance-related downtime. Additionally, the ability of our Clean Fleet® technology to automatically capture and transmit vast amounts of operational, logistical, and environmental data provide us with insights that offer opportunities for continuous improvements in our operational efficiencies.

## Competitive Strengths

We believe that the following strengths will position us to provide high-quality service to our customers and create value for our stockholders:

• Proprietary Clean Fleet® technology. We are a market leader in electric pressure pumping technology, with fiveall-electric fleets. Our fleets utilizing Clean Fleet® technology provide substantial

cost savings by replacing diesel fuel with natural gas and offer considerable operational, safety and environmental advantages. Clean Fleet technology offers superior operational efficiency resulting from reduced non-productive time due to repairs, maintenance and failures associated with diesel-powered engines and transmissions. Additionally, Clean Fleet technology can substantially reduce emissions of air pollutants and noise from the wellsite. With an increasing focus on ESG and increasing returns by our customers, we believe that adoption of this technology in the near term will materially increase and allow us to continue to significantly expand our market share over the next several years.

- Strong, employee-centered culture. Our employees are critical to our success and are a key source of our competitive advantage. We continuously invest in training and development for our employees, and as a result, we can provide consistent, high-quality service and safe working conditions for both employees and customers.
- Track record for safety. Safety is a critical element of our operations. We focus on providing customers with the highest quality of service by employing a trained and motivated workforce that is rigorously focused on safety. We continuously review safety data and work to develop and implement policies and procedures that ensure the safety and wellbeing of our employees, customers, and the communities in which we operate. Our field operators are empowered to stop work and question the safety of a situation or task performed. We use specialized technology to improve safety for our truck drivers and employ measures to mitigate the risk of respirable silica dust exposure on the wellsite. We believe our record of safe operations makes us an attractive partner for both our customers and our employees. Additionally, we continue to take proactive measures to safeguard the physical health of our employees in response to the novel coronavirus 2019 ("COVID-19") pandemic. Employees capable of working from home were allowed to do so and all individuals entering a Company facility or work location are required to undergo a screening process.
- Strong customer relationships supported by contracts and dedications. We have cultivated strong relationships with a diverse group of customers because of the quality of our service, safety performance and ability to work with customers to establish mutually beneficial service agreements. Our contracts and dedications provide customers with certainty of service pricing, allowing them to efficiently budget and plan the development of their wells. Additionally, our contracts and dedications allow us to maintain higher utilization of our fleet and generate revenue and cash flow through industry cycles. We believe our relationships and the structure of our contracts and dedications position us to continue to build long-term partnerships with customers and support stable financial performance.
- Modern, high-quality equipment and rigorous maintenance program. Our fleets consist of modern, well-maintained equipment. We invest in high-quality equipment from leading original equipment manufacturers. Moreover, we take proactive measures to maintain the quality of our equipment, using specialized equipment to monitor frac pump integrity and our proprietary FRAC MD® data analytics platform to support preventative maintenance efforts. We believe the quality of our equipment is critical to our ability to provide high quality service to our customers.
- Proven, cycle-tested management team. Our management team has a proven track record for building and operating oilfield service
  companies. Our operating and commercial teams have significant industry experience and longstanding relationships with our clients. We
  believe our management team's experience and relationships position us to generate business and create value for stockholders. Additionally,
  our management team's rapid response to the COVID-19 pandemic enabled us to maintain adequate liquidity and develop operational
  procedures allowing for business continuity during a turbulent market environment.

# **Customer Concentration**

Our customers include a broad range of leading E&P companies. For the six months ended June 30, 2022, five customers each comprised greater than 10% of our consolidated revenues and collectively represented

99.9% of our consolidated revenues. For the year ended December 31, 2021, three customers each comprised greater than 10% of our consolidated revenues and collectively represented 50.4% of our consolidated revenues. For the year ended December 31, 2020, five customers each comprised greater than 10% of our consolidated revenues and collectively represented 80.7% of our consolidated revenues.

#### Suppliers

We purchase a wide variety of raw materials, parts and components that are manufactured and supplied for our operations. We currently rely on a limited number of suppliers from which we procure major equipment components used to maintain, build, or upgrade our custom Clean Fleet® pressure pumping equipment. In addition, some of these components have few suppliers and long lead times to acquire. Certain materials for which we do not currently have long-term supply agreements could experience shortages and significant price increases in the future. As a result, we may be unable to mitigate any future supply shortages and our results of operations, prospects and financial condition could be adversely affected.

It is also possible that supply chain constraints or disruptions will result in significant delivery delays with respect to the equipment that we need to build our new Nyx Clean Fleets®. If we are unable to build out our new Nyx Clean Fleets® as a result of any equipment delivery delays, we may not be in position to enter into contracts with customers, which would negatively impact our results of operations.

For the year ended December 31, 2021, purchases from our three largest suppliers represented approximately 27.3% of our overall purchases.

#### Competition

The markets in which we operate are very competitive. We provide services in various geographic regions across the United States, and our competitors include many large and small oilfield service providers. Our pressure pumping services compete with large, integrated companies such as Halliburton Company, as well as other companies including Evolution Well Services, Liberty Oilfield Services Inc., NexTier Oilfield Solutions Inc., Patterson-UTI Energy Inc., ProFrac Services, LLC and ProPetro Services Inc. In addition, our industry is highly fragmented, and we compete regionally with a significant number of smaller service providers. Although several of our larger competitors have announced their intention to develop or adopt new electric well service technologies, we believe that only the Company and one privately owned competitor are currently significant providers of all-electric pressure pumping services.

We believe that the principal competitive factors in the markets we serve are technical expertise, equipment capacity, workforce competency, efficiency, safety record, reputation, experience, and price. Additionally, projects are often awarded on a bid basis, which tends to create a highly competitive environment.

## Cyclical Nature of Industry

We operate in a cyclical industry. The key factor driving demand for our services is the level of well completions by E&P companies, which in turn depends largely on the current and anticipated economics of new well completions. Global supply and demand for oil and the domestic supply and demand for natural gas are critical in assessing industry outlook. Demand for oil and natural gas is cyclical and subject to large, rapid fluctuations. E&P companies tend to increase capital expenditures in response to increases in oil and natural gas prices, which generally results in greater revenues and profits for oilfield service companies like us. Increased E&P capital expenditures ultimately lead to greater production, which historically has resulted in increased supplies and reduced prices which in turn tend to reduce demand for oilfield services. For these reasons, the results of our operations may fluctuate from quarter to quarter and from year to year, and these fluctuations may distort comparisons of results across periods.

#### Insurance

Although we maintain insurance coverage of types and amounts that we believe to be customary in the industry, we are not fully insured against all risks, either because insurance is not available or because of the high premium costs relative to perceived risk. Further, insurance rates have in the past been subject to wide fluctuation and changes in coverage could result in less coverage, increases in cost or higher deductibles and retentions. Liabilities for which we are not insured, or which exceed the policy limits of their applicable insurance, could have a material adverse effect on our business and financial condition.

#### **Environmental and Occupational Health and Safety Regulations**

Our operations are subject to stringent laws and regulations governing the discharge of materials into the environment or otherwise relating to environmental protection, and occupational health and safety. Numerous federal, state, and local governmental agencies issue regulations that often require difficult and costly compliance measures that could carry substantial administrative, civil and criminal penalties and may result in permit revocations or modifications, operational disruptions, or injunctive obligations for noncompliance. These laws and regulations may, for example, restrict the types, quantities and concentrations of various substances that can be released into the environment, limit or prohibit construction or drilling activities on certain lands lying within wilderness, wetlands, ecologically or seismically-sensitive areas and other protected areas, or require action to prevent or remediate pollution from current or former operations. Moreover, it is not uncommon for neighboring landowners and other third parties to file claims for personal injury and property damage allegedly caused by the release of hazardous substances, hydrocarbons, or other waste products into the environment. Changes in environmental, health and safety laws and regulations occur frequently, and any changes in the laws or regulations or the interpretation thereof that result in more stringent and costly requirements could materially adversely affect our operations and financial position. We have not experienced any material adverse effect from compliance with these requirements. This trend, however, may not continue in the future.

Below is an overview of some of the more significant environmental, health and safety requirements with which we must comply. Our customers' operations are subject to similar laws and regulations. Any material adverse effect of these laws and regulations on our customers' operations and financial position may also have an indirect material adverse effect on our operations and financial position.

Waste Handling. We engage third parties to handle, transport, store and dispose of wastes that are subject to the Resource Conservation and Recovery Act ("RCRA") and comparable state laws and regulations, which impose requirements regarding the generation, transportation, treatment, storage, disposal and cleanup of hazardous and nonhazardous wastes. With federal approval, the individual states administer some or all the provisions of RCRA, sometimes in conjunction with our own, more stringent requirements. Although certain petroleum production wastes are exempt from regulation as hazardous wastes under RCRA, such wastes may be regulated under state law or constitute "solid wastes" that are subject to the less stringent requirements of nonhazardous waste provisions.

Administrative, civil, and criminal penalties can be imposed for failure to comply with waste handling requirements. Moreover, the Environmental Protection Agency ("EPA") or state or local governments may adopt more stringent requirements for the handling of nonhazardous wastes one-categorize some nonhazardous wastes as hazardous for future regulation. Indeed, legislation has been proposed from time to time in Congress to re-categorize certain oil and natural gas exploration, development, and production wastes as hazardous wastes. Several environmental organizations have also petitioned the EPA to modify existing regulations to re-categorize certain oil and natural gas exploration, development, and production wastes as hazardous. Any such changes in these laws and regulations could have a material adverse effect on our capital expenditures and operating expenses. Although we do not believe the current costs of managing our wastes, as presently classified, to be significant, any legislative or regulatory reclassification of oil and natural gas exploration and production wastes could increase our costs to manage and dispose of such wastes.

Remediation of Hazardous Substances. The Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA" or "Superfund") and analogous state laws generally impose liability without regard to fault or legality of the original conduct, on classes of persons who are considered to be responsible for the release of a hazardous substance into the environment or, under some state CERCLA-analogous laws, the release of solid waste. These persons include the current owner or operator of a contaminated facility, a former owner or operator of the facility at the time of contamination, those persons that disposed or arranged for the disposal of the substance at the facility, and transporters who selected the disposal site. Liability for the costs of removing or remediating previously disposed wastes or contamination, damages to natural resources, and the costs of conducting certain health studies, amongst other things, is strict and joint and several. In addition, it is not uncommon for neighboring landowners and other third parties to file claims for personal injury and property damage allegedly caused by the hazardous substances released into the environment. During our operations, we use materials that, if released, would be subject to CERCLA and comparable state laws. Therefore, governmental agencies or third parties may seek to hold us responsible under CERCLA and comparable state statutes for all or part of the costs to clean up sites at which such substances have been released.

**NORM.** During our operations, some of our equipment may be exposed to naturally occurring radioactive materials (**NORM**") associated with oil and gas deposits and accordingly may result in the generation of wastes and other materials containing NORM. NORM exhibiting levels of naturally occurring radiation in excess of established state standards are subject to special handling and disposal requirements, and any storage vessels, piping and work area affected by NORM may be subject to remediation or restoration requirements.

Water Discharges. The Clean Water Act, Safe Drinking Water Act, Oil Pollution Act and analogous state laws and regulations impose restrictions and strict controls regarding the unauthorized discharge of pollutants, including produced waters and other gas and oil wastes, into regulated waters. The discharge of pollutants into regulated waters is prohibited, except in accordance with the terms of a permit issued by the EPA, the U.S. Army Corps of Engineers (the "Corps"), or the applicable state. The Clean Water Act has been interpreted by these agencies to apply broadly. Attempting to clarify what constitutes a regulated "water of the United States", or WOTUS, the EPA and the Corps promulgated the Clean Water Rule in 2015. Following legal challenges and political maneuverings, the Clean Water Rule was repealed and replaced with the final Navigable Waters Protection Rule (the "NWPR") in 2020. The NWPR revised and narrowed the WOTUS definition. The NWPR was vacated by an Arizona district court in August 2021, and the agencies announced they would halt implementation of the NWPR and revert to pre-2015 WOTUS interpretation, which is broader than the NWPR. On December 21, 2021, the EPA and the Corps published a proposed rule in the Federal Register that, once finalized, will formally re-establish the pre-2015 WOTUS definition, with some updates to reflect consideration of Supreme Court decisions. In the proposed rulemaking, the agencies indicated they anticipate a second, future rulemaking that would build upon the foundation of the proposed rule. We expect political maneuverings and litigation regarding the WOTUS definition to continue, creating uncertainty as to what constitutes a protected "water of the United States." In addition, spill prevention, control and countermeasure plan requirements require appropriate containment berms and similar structures to help prevent the contamination of regulated waters.

Air Emissions. The Clean Air Act ('CAA'') and comparable state laws and regulations regulate emissions of various air pollutants through the issuance of permits and the imposition of other emissions control requirements. The EPA has developed, and continues to develop, stringent regulations governing emissions of air pollutants from specified sources. New facilities may be required to obtain permits before work can begin, and existing facilities may be required to obtain additional permits and incur capital costs to remain in compliance. These and other laws and regulations may increase the costs of compliance for some facilities where we operate, and significant administrative, civil, and criminal penalties can be imposed for failure to comply with air regulations. Obtaining or renewing permits also has the potential to delay the development of oil and natural gas projects.

Climate Change. The EPA has determined that greenhouse gases ('GHGs'') present an endangerment to public health and the environment because such gases contribute to warming of the Earth's atmosphere and other climatic changes. Based on these findings, the EPA has adopted and implemented, and continues to adopt and

implement, regulations that restrict emissions of GHGs under existing provisions of the CAA. The EPA also requires the annual reporting of GHG emissions from certain large sources of GHG emissions in the United States, including certain oil and gas production facilities. The U.S. Congress has from time to time considered adopting legislation to reduce emissions of GHGs and almost one-half of the states have already taken legal measures to reduce emissions of GHGs primarily through the development of GHG emission inventories and/or regional GHG cap and trade programs and through the establishment of emissions reduction targets. In December 2015, the United States joined the international community at the 21st Conference of the Parties of the United Nations Framework Convention on Climate Change in Paris, France. The resulting Paris Agreement calls for the parties to undertake "ambitious efforts" to limit the average global temperature, and to conserve and enhance sinks and reservoirs of GHGs. The Paris Agreement entered into force in November 2016. The United States exited the Paris Agreement in November 2020, but rejoined it effective on February 19, 2021. Additional federal and/or state regulations targeting climate change could significantly affect our operations or compliance costs.

Moreover, climate change may cause more extreme weather conditions and increased volatility in seasonal temperatures. Extreme weather conditions can interfere with our operations and increase our costs, and damage resulting from extreme weather may not be fully insured.

Endangered and Threatened Species. Environmental laws such as the Endangered Species Act ("ESA") and analogous state laws may impact exploration, development, and production activities in areas where we operate. The ESA provides broad protection for species of fish, wildlife and plants that are listed as threatened or endangered. Similar protections are offered to migratory birds under the Migratory Bird Treaty Act and various state analogs. The U.S. Fish and Wildlife Service may identify previously unidentified endangered or threatened species or may designate critical habitat and suitable habitat areas that it believes are necessary for survival of a threatened or endangered species, or a state may impose new rules limiting oil and gas operations to protect wildlife or habitat, which could cause us or our customers to incur additional costs or become subject to operating restrictions or operating bans in the affected areas.

Regulation of Hydraulic Fracturing and Related Activities. Hydraulic fracturing is an important and common practice that is used to stimulate production of hydrocarbons, particularly natural gas, from tight formations, including shales. The process, which involves the injection of water, sand, and chemicals under pressure into formations to fracture the surrounding rock and stimulate production, is typically regulated by state oil and natural gas commissions. However, federal agencies have asserted regulatory authority over certain aspects of the process. For example, on January 20, 2021, the Biden administration issued an order implementing a 60-day suspension on new oil and gas leases and drilling permits for federal lands and water. On January 27, 2021, the Biden administration issued an executive order that indefinitely pauses new oil and gas leases on federal lands while the administration undertakes a comprehensive review on climate change impacts. In June 2021, a federal judge issued a preliminary injunction to block the leasing pause. The Biden administration resumed approving oil and gas drilling permits on federal lands but is appealing the ruling. Additionally, beginning in August 2012, the EPA issued a series of rules under the CAA that establish new emission control requirements for certain oil and natural gas production and natural gas processing operations and associated equipment. Various political maneuverings resulted in September 2020 amendments to New Source Performance Standards issued in 2016. These amendments were reversed in June 2021, following a joint resolution passed by Congress and signed by President Biden pursuant to the Congressional Review Act. On November 15, 2021, the EPA published in the Federal Register a proposed rule that would for the first time regulate emissions of methane (a GHG) and volatile organic compounds ("VOCs") from existing oil and gas facilities. The proposed rule would also update, strengthen and expand existing rules applicable to new, reconstructed and modified oil and gas facilities. The agency did not provide proposed regulatory text in the proposed rule, but instead sought public comments for a supplemental proposal the agency expects to issue in 2022 that may expand on or modify the 2021 proposed rule, including potentially regulating additional sources of methane and VOC emissions, such as abandoned and unplugged wells. The U.S. Department of Interior's Bureau of Land Management ("BLM") finalized similar rules in November 2016 that would have limited methane emissions from new and existing oil and natural gas operations on federal lands through limitations on the venting and flaring of gas, as well as enhanced leak

detection and repair requirements. This rule and a September 2018 amendment to it that scaled back the waste-prevention requirements of the 2016 rule were both vacated in 2020 court decisions. In a November 2021 multi-agency Emissions Reduction Action Plan, the Biden administration indicated BLM's intent to promulgate a rule to disincentivize excessive venting or flaring of gas by requiring oil and gas operators to pay royalties to the federal government for vented or flared gas, and to strengthen financial assurance requirements for oil and gas operators. Moreover, in March 2015, the BLM issued a final rule that sought to impose requirements on hydraulic fracturing activities on federal and Indian lands, including new requirements relating to public disclosure, wellbore integrity and handling of flowback water. However, the BLM rescinded this rule in December 2017. In January 2018, California and a coalition of environmental groups filed suit in the Northern District of California to challenge the BLM's rescission of the 2015 rule. In March 2020, the California district court upheld the BLM's rescission of the hydraulic fracturing rule, and the plaintiff groups appealed. This litigation and administrative review of federal hydraulic fracturing regulations is ongoing. Many states also have hydraulic fracturing regulations in place that may be duplicative of the rescinded federal regulations, and the BLM requires operators to comply with the law of the state where the federal or Indian land is located.

Further, legislation to amend the Safe Drinking Water Act to repeal the exemption for hydraulic fracturing (except when diesel fuels are used) from the definition of "underground injection" and require federal permitting and regulatory control of hydraulic fracturing have been proposed in recent sessions of Congress. Several states and local jurisdictions in which we or our customers operate also have adopted or are considering adopting regulations that could require disclosure of the chemical constituents of the fluids used in the fracturing process, restrict, or prohibit hydraulic fracturing in certain circumstances, impose more stringent operating standards and/or require the disclosure of the composition of hydraulic fracturing fluids.

Federal and state governments have also begun investigating whether the disposal of produced water into underground injection wells has caused increased seismic activity in certain areas. For example, in December 2016, the EPA released its final report regarding the potential impacts of hydraulic fracturing on drinking water resources, concluding that "water cycle" activities associated with hydraulic fracturing may impact drinking water resources under certain circumstances such as water withdrawals for fracturing in times or areas of low water availability, surface spills during the management of fracturing fluids, chemicals or produced water, injection of fracturing fluids into wells with inadequate mechanical integrity, injection of fracturing fluids directly into groundwater resources, discharge of inadequately treated fracturing wastewater to surface waters, and disposal or storage of fracturing wastewater in unlined pits. The results of these studies could lead the federal government and has led some state governments to develop and implement additional regulations, including stricter regulatory requirements relating to the location and operation of underground injection wells or requirements regarding the permitting of produced water disposal wells or otherwise.

Increased regulation of hydraulic fracturing and related activities (whether as a result of the EPA study results or resulting from other factors) could restrict or prohibit our ability to operate in certain areas or subject us and our customers to additional permitting and financial assurance requirements, more stringent construction specifications, increased monitoring, reporting and recordkeeping obligations, and plugging and abandonment requirements. New requirements could result in increased operational costs for us and our customers and reduce the demand for our services.

**OSHA Matters.** The Occupational Safety and Health Act ('**OSHA**') and comparable state statutes regulate the protection of the health and safety of workers. In addition, the OSHA hazard communication standard requires that information be maintained about hazardous materials used or produced in operations and that this information be provided to employees, state and local government authorities, and the public.

# **Employees**

As of June 30, 2022, we had 493 full-time employees and no part-time employees. We are not a party to any collective bargaining agreements and have not experienced any strikes or work stoppages. We believe our

relationships with our employees are excellent. From time to time, we will utilize the services of independent contractors to perform various field and/or other services.

# **Intellectual Property**

We have been granted or have received notice of allowance for 81 patents, which begin to expire in late 2033, and have an additional 249 patents pending. We have filed patents in an effort to protect our Clean Fleet® technology from being duplicated by competitors. These patents should help provide unique competitive advantages in operating areas where noise and emissions are key concerns. We also use proprietary FRAC MD® technology to support our preventative maintenance program and prolong equipment useful life. This technology utilizes specialized equipment to capture and analyze vibrations to identify component stress so maintenance can be performed prior to catastrophic failures. Our fleets continuously transmit data captured in the field to our IoT platform. This data is analyzed by our team of data scientists in order improve our operations by garnering insights that inform real-time decision making in the field and drive our machine learning capabilities to increase efficiency and extend equipment useful life.

#### MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF USWS

Unless the context otherwise requires, all references in this subsection to "we," "us," "our," and the "Company" shall mean U.S. Well Services, Inc.

#### Overview

We provide pressure pumping services in oil and natural gas basins. Our Clean Flee® electric fleets are among the most reliable and highest performing fleets in the industry, with the capability to meet the most demanding pressure and pump rate requirements. In May 2021, we announced the next generation of our Clean Flee® technology with the unveiling of our newly designed Nyx Clean Flee®. The Company's first Nyx Clean Flee® to commenced pressure pumping operations in July 2022.

We operate in many of the active shale and unconventional oil and natural gas basins of the United States and our clients benefit from the performance and reliability of our equipment and personnel. Specifically, all our fleets operate on a 24-hour basis and have the ability to withstand high utilization rates, which results in more efficient operations. Our senior management team has extensive industry experience providing pressure pumping services to exploration and production companies across North America. Since announcing our commitment to becoming an all-electric pressure pumping services provider in May 2021, we have sold most of our legacy, diesel-powered pressure pumping equipment. We have retained some of our legacy, diesel-powered pressure pumping equipment for use during our transition to support our electric fleets and bridge the time gap between our customers' current service needs and the deployment of our newbuild Nyx Clean Fleets. We expect to continue phasing out our remaining conventional fleet operations as we take delivery of our newbuild Nyx Clean Fleets.

#### How the Company Generates Revenue

We generate revenue by providing pressure pumping services to our customers. We own and operate fleets of pressure pumping equipment to perform these services. We seek to enter into contractual arrangements with our customers or fleet dedications, which establish pricing terms for a fixed duration. Under the terms of these agreements, we charge our customers base monthly rates, adjusted for activity and provision of materials such as proppant and chemicals, or we charge a variable rate based on the nature of the job including pumping time, well pressure, proppant and chemical volumes and transportation.

#### **Our Costs of Conducting Business**

The principal costs involved in conducting our pressure pumping services are labor, maintenance, equipment rentals, including the rental of power generation equipment, materials, and transportation costs. A large portion of our costs are variable, based on the number and requirements of pressure pumping jobs. We manage our fixed costs, other than depreciation and amortization, based on factors including industry conditions and the expected demand for our services.

Materials include the cost of proppant delivered to the basin of operations, chemicals, and other consumables used in our operations. These costs vary based on the quantity and type of proppant and chemicals utilized when providing pressure pumping services. Transportation represents the costs to transport materials and equipment from receipt points to customer locations. Labor costs include payroll and benefits related to our field crews and other employees. Most of our employees are paid on an hourly basis. Maintenance costs include preventative and other repair costs that do not require the replacement of major components of our fleets. Maintenance and repair costs are expensed as incurred.

The following table presents our cost of services for the periods indicated (in thousands):

		nths Ended e 30,		Six Months Ended June 30,		
	2022	2021	2022	2021		
Labor	\$17,352	\$24,003	\$30,532	\$ 47,689		
Maintenance	9,027	14,916	14,146	31,511		
Power generation equipment rentals	8,944	_	17,436	_		
Materials	5,092	3,873	8,551	10,903		
Transportation	2,314	3,220	4,262	6,259		
Other <sup>(1)</sup>	12,480	13,240	21,005	25,521		
Cost of services	\$55,209	\$59,252	\$95,932	\$121,883		

(1) Other consists of fuel, lubes, other equipment rentals, travel and lodging costs for our crews, site safety costs and other costs incurred in performing our operating activities.

## Significant Trends

Our revenues and operating costs remained lower in the second quarter of 2022 versus the second quarter of 2021, as we continued to experience a lower average active fleet count resulting from our ongoing exit from the diesel pressure pumping market. However, we expect to see our average active fleet count increase throughout the remainder of the 2022 as we begin taking delivery of our newbuild Nyx Clean Fleets\* and deploying these fleets to the field. During the second quarter of 2022, we continued to be impacted by tightness in the labor market and inflation across our supply chain. As a result of these conditions, we experienced an increase in personnel costs, higher third-party service costs and price increases for a variety of goods driven by rising prices for underlying commodities such as oil, nickel and steel.

We believe the current market backdrop is favorable for our business. During the second quarter of 2022, WTI crude oil and Henry Hub natural gas prices averaged \$108.55 per barrel and \$7.49 per MMBtu, respectively, as compared to \$95.01 per barrel and \$4.59 per MMBtu in the first quarter of 2022. The resulting demand for pressure pumping services, driven by improving commodity prices, significantly reduced the available supply of pressure pumping equipment and crews, as well as the availability of ancillary goods and services. As a result, pricing for pressure pumping services has increased substantially relative to average levels experienced during the first quarter of 2022. The rising cost for diesel fuel is also creating a material cost differential between conventional hydraulic fracturing fleets and all-electric fleets that do not require any diesel fuel for power. We believe this dynamic will continue to drive demand for our electric fleets and provide an opportunity to capture premium pricing as compared to prevailing market prices for diesel-powered, conventional fleets.

#### **Results of Operations**

Three months ended June 30, 2022, compared to the three months ended June 30, 2021 (in thousands, except percentages)

	Three Months Ended June 30,					
	2022	% <sup>(1)</sup>	2021	% <sup>(1)</sup>	Variance	% Variance
Revenue	\$68,764	100.0%	\$78,799	100.0%	\$(10,035)	(12.7)%
Costs and expenses:						
Cost of services (excluding depreciation and amortization)	55,209	80.3%	59,252	75.2%	(4,043)	(6.8)%
Depreciation and amortization	5,877	8.5%	9,836	12.5%	(3,959)	(40.3)%
Selling, general and administrative expenses	9,406	13.7%	7,214	9.2%	2,192	30.4%
Litigation settlement	_	0.0%	35,000	44.4%	(35,000)	100.0%
Gain on disposal of assets	(76)	(0.1)%	(545)	(0.7)%	469	(86.1)%
Loss from operations	(1,652)	(2.4)%	(31,958)	(40.6)%	30,306	*(2)
Interest expense, net	(9,562)	(13.9)%	(7,333)	(9.3)%	(2,229)	30.4%
Change in fair value of warrant liabilities	1,577	2.3%	(136)	(0.2)%	1,713	*(2)
Patent license sales	_	0.0%	22,500	28.6%	(22,500)	100.0%
Loss on extinguishment of debt, net	_	0.0%	(839)	(1.1)%	839	96.8%
Other income	302	0.4%	23	0.0%	279	*(2)
Income tax benefit		0.0%	(27)	(0.0)%	27	*(2)
Net loss	\$ (9,335)	(13.6)%	\$ (17,716)	(22.5)%	\$ 8,381	(47.3)%

- (1) As a percentage of revenues. Percentage totals or differences in the above table may not equal the sum or difference of the components due to rounding.
- (2) Not meaningful.

Revenue. The decrease in revenue was primarily attributable to a lower active fleet count in the second quarter of 2022 compared to the prior comparable period as we began to transition to an all-electric pressure pumping service provider in May 2021. As of the end of the first quarter of 2022, we had sold most of our diesel-powered pressure pumping equipment. For the three months ended June 30, 2022, our average active fleet count decreased to 6.0 fleets compared to 9.3 fleets in the prior comparable period. Additionally, during the second quarter of 2022, a larger portion of our customers elected to self-source proppant, chemicals and trucking services as compared to the prior comparable period.

Cost of services, excluding depreciation and amortization. The decrease in cost of services, excluding depreciation and amortization, was attributable to the decrease in variable costs as activity was reduced with a lower number of active fleets in the second quarter of 2022 compared to the prior comparable period. This was offset by costs associated with procuring third-party power generation services as we transitioned away from owning power generation assets starting in October 2021, as well as rising inflation across the rest of our supply chain.

Depreciation and amortization. The decrease in depreciation and amortization was primarily due to the sale of most of our diesel-powered pressure pumping equipment during the second half of 2021.

Selling, general and administrative expenses. The increase in selling, general, and administrative expenses was primarily attributable to an increase in legal fees related to patents and higher payroll costs, which were partly offset by the reversal of \$3.1 million share-based compensation expense due to forfeitures of certain unvested restricted stock.

Litigation settlement. The Company was named as a defendant in a lawsuit filed in January 2019 by a vendor alleging that the Company breached a multi-year contract. In June 2021, following entry of the final judgement by the court in favor of the vendor, the Company entered into a settlement agreement whereby it paid \$35.0 million in cash.

Loss (gain) on disposal of assets. The amount of gain or loss on disposal of assets fluctuates period over period due to differences in the operating conditions of our pressure pumping equipment, such as wellbore pressure and rate of barrels pumped per minute, which impacts the timing of disposals of our pump components. Additionally, during the three months ended June 30, 2021, we recognized a gain associated with the sale of property and equipment.

Interest expense, net. The increase in interest expense was primarily attributable to the interest associated with the Convertible Senior Notes issued during the second quarter of 2021 and interest on our Senior Secured Term Loan during the second quarter of 2022.

Patent license sales. On June 24, 2021, the Company issued a Convertible Senior Note in the principal amount of \$22.5 million that was convertible into the License Agreement. On June 29, 2021, the holder exercised its right to convert the Convertible Senior Note in full and the Company entered into the License Agreement, which provides the licensee a five-year option to purchase up to 20 licenses to build and operate electric fleets using the Company's patented Clean Fleet® technology. Upon entry into the License Agreement, the Company sold three licenses to build and operate three electric fleets, each valued at \$7.5 million.

Loss on extinguishment of debt. During the three months ended June 30, 2021, we recognized a loss on extinguishment of debt for the unamortized debt discount and issuance costs and prepayment fees associated with the early repayment of our Senior Secured Term Loan.

# Six months ended June 30, 2022, compared to the six months ended June 30, 2021 (in thousands, except percentages)

	Six Months Ended June 30,					
	2022	% <sup>(1)</sup>	2021	% <sup>(1)</sup>	Variance	% Variance
Revenue	\$109,914	100.0%	\$155,057	100.0%	\$(45,143)	(29.1)%
Costs and expenses:						
Cost of services (excluding depreciation and amortization)	95,932	87.3%	121,883	78.6%	(25,951)	(21.3)%
Depreciation and amortization	11,577	10.5%	20,942	13.5%	(9,365)	(44.7)%
Selling, general and administrative expenses	17,778	16.2%	14,604	9.4%	3,174	21.7%
Litigation settlement	_	0.0%	35,000	22.6%	(35,000)	100.0%
Loss on disposal of assets	2,980	2.7%	1,891	1.2%	1,089	57.6%
Loss from operations	(18,353)	(16.7)%	(39,263)	(25.3)%	20,910	*(2)
Interest expense, net	(17,530)	(15.9)%	(13,516)	(8.7)%	(4,014)	29.7%
Change in fair value of warrant liabilities	831	0.8%	(7,287)	(4.7)%	8,118	*(2)
Patent license sales	_	0.0%	22,500	14.5%	(22,500)	100.0%
Loss on extinguishment of debt	(1,651)	(1.5)%	(839)	(0.5)%	(812)	100.0%
Other income	1,623	1.5%	52	0.0%	1,571	*(2)
Income tax benefit		0.0%	(27)	(0.0)%	27	*(2)
Net loss	\$ (35,080)	(31.9)%	\$ (38,326)	(24.7)%	\$ 3,246	*(2)

<sup>(1)</sup> As a percentage of revenues. Percentage totals or differences in the above table may not equal the sum or difference of the components due to rounding.

<sup>(2)</sup> Not meaningful.

Revenue. The decrease in revenue was primarily attributable to a lower active fleet count during the six months ended June 30, 2022 compared to the prior comparable period as we began to transition to an all-electric pressure pumping service provider in May 2021. As of the end of the first quarter of 2022, we had sold most of our diesel-powered pressure pumping equipment. For the six months ended June 30, 2022, our average active fleet count decreased to 5.3 fleets compared to 9.7 fleets in the prior comparable period. Additionally, during the first half of 2022, revenue from proppant, chemicals and trucking services were lower compared to the prior comparable period.

Cost of services, excluding depreciation and amortization. The decrease in cost of services, excluding depreciation and amortization, was attributable to the decrease in variable costs as activity was reduced with a lower number of active fleets in the first half of 2022 compared to the prior comparable period. This was offset by costs associated with procuring third-party power generation services as we transitioned away from owning power generation assets starting in October 2021.

Depreciation and amortization. The decrease in depreciation and amortization was primarily due to the sale of most of our diesel-powered pressure pumping equipment during the second half of 2021.

Selling, general and administrative expenses. The increase in selling, general, and administrative expenses was primarily attributable to an increase in legal fees related to patents and higher payroll costs, which were partly offset by the reversal of \$3.1 million share-based compensation expense due to forfeitures of certain unvested restricted stock.

Litigation settlement. The Company was named as a defendant in a lawsuit filed in January 2019 by a vendor alleging that the Company breached a multi-year contract. In June 2021, following entry of the final judgement by the court in favor of the vendor, the Company entered into a settlement agreement whereby it paid \$35.0 million in cash.

Loss on disposal of assets. The amount of gain or loss on disposal of assets fluctuates period over period due to differences in the operating conditions of our pressure pumping equipment, such as wellbore pressure and rate of barrels pumped per minute, which impacts the timing of disposals of our pump components. The gain or loss on disposal of assets was offset by the sale of property and equipment which commenced in the second quarter of 2021 through the first quarter of 2022.

Interest expense, net. The increase in interest expense was primarily attributable to the interest associated with the Convertible Senior Notes issued during the second quarter of 2021 and the Senior Secured Term Loan in the first half of 2022.

Patent license sales. On June 24, 2021, the Company issued a Convertible Senior Note in the principal amount of \$22.5 million that was convertible into the License Agreement. On June 29, 2021, the holder exercised its right to convert the Convertible Senior Note in full and the Company entered into the License Agreement, which provides the licensee a five-year option to purchase up to 20 licenses to build and operate electric fleets using the Company's patented Clean Fleet® technology. Upon entry into the License Agreement, the Company sold three licenses to build and operate three electric fleets, each valued at \$7.5 million.

Loss on extinguishment of debt. We recognized a loss on extinguishment of debt for the unamortized debt discount and issuance costs and prepayment fees associated with the early repayment of our Senior Secured Term Loan.

# Liquidity and Capital Resources

# Overview

Our primary sources of liquidity and capital resources have historically been cash, cash flow generated from operating activities, proceeds from the issuance of debt or equity and borrowings under our ABL Credit Facility.

As of June 30, 2022, our total liquidity was \$35.9 million, consisting of \$18.0 million of cash and restricted cash and \$17.9 million of availability under our ABL Credit Facility.

Our short-term and long-term liquidity requirements consist primarily of capital expenditures, payments of contractual obligations, including debt service obligations and working capital.

We believe that our current cash position, current and expected working capital balances, favorable cash interest payment terms under our Senior Secured Term Loan Agreement, availability under our ABL Credit Facility, proceeds received from issuance of debt or equity, and other anticipated sources of credit will be sufficient to satisfy cash requirements associated with our existing operations, capital expenditures and contractual obligations for the next twelve months. Although there is no assurance, we plan to satisfy our long-term financial obligations through operating cash flows that we could generate as we increase our number of working fleets with our planned additions of Nyx Clean Fleets® and additional financing as our long-term financial obligations mature.

# Sources of Cash

On July 18, 2022, the Company and USWS LLC entered into a \$12.5 million promissory note (the "Equify Note") with Equify Financial, LLC, as lender, which matures on August 1, 2027. The Company intends to use the proceeds to fund capital expenditures related to our newbuild Nyx Clean Fleets®.

During the first quarter of 2022, we raised approximately \$68.4 million of gross proceeds from the issuance of \$46.9 million common equity through a registered direct offering and under the 2020 ATM Agreement and borrowings of \$21.5 million under the Term C Loans. We used the proceeds for general working capital, including the funding of capital expenditures related to our newbuild Nyx Clean Fleets\*.

Registered Direct Offering. On March 11, 2022, we completed a registered direct offering of 2,363,396 shares of Class A common stock and 14,180,375 RDO Investor Warrants for gross proceeds of approximately \$25.0 million, before deducting placement agent fees and other offering expenses. We used the net proceeds for working capital purposes, including the funding of certain capital expenditures.

2020 ATM Agreement. During the six months ended June 30, 2022, we sold 1,627,991 shares of Class A common stock for total net proceeds of \$21.3 million and paid \$0.6 million in commissions under the 2020 ATM Agreement. Since inception on June 26, 2020 through April 26, 2022, we sold a total of 2,514,351 shares of Class A common stock under the 2020 ATM Agreement for total net proceeds of \$36.4 million and paid \$1.1 million in commissions.

# **Cash Flows**

(in thousands)

	Six Month June	
	2022	2021
Net cash provided by (used in):		
Operating activities	\$ 3,559	\$(28,371)
Investing activities	(30,999)	(16,288)
Financing activities	36,324	97,460

Operating Activities. Net cash provided by (used in) operating activities primarily represents the results of operations exclusive ofinon-cash expenses, including depreciation, amortization, provision for losses on accounts receivable and inventory, interest, impairment losses, (gains) losses on disposal of assets, changes in fair value of warrant liabilities, loss on extinguishment of debt and share-based compensation and the impact of changes in

operating assets and liabilities. Net cash provided by operating activities was \$3.6 million for the six months ended June 30, 2022 primarily due to cash flows generated during the second quarter of 2022 as we began to deploy additional fleets.

Net cash used in operating activities was \$28.4 million for the six months ended June 30, 2021, primarily attributable to the litigation settlement of \$35.0 million and \$3.0 million of working capital payments using proceeds from our USDA Loan.

Investing Activities. Net cash used in investing activities decreased by \$14.7 million from the prior corresponding period, primarily related to an increase in purchases of property and equipment related to growth capital expenditures. Net cash used in investing activities was \$31.0 million for the six months ended June 30, 2022, primarily due to \$48.3 million in purchases of property and equipment, consisting of growth capital expenditures. This was offset by \$5.3 million in proceeds from the sale of property and equipment and \$12.0 million of insurance proceeds related to damaged property and equipment.

Net cash used in investing activities was \$16.3 million for the six months ended June 30, 2021, primarily due to \$24.8 million in purchases of property and equipment, related to maintaining and supporting our existing pressure pumping equipment and payments made to replace damaged property and equipment. This was offset by \$6.4 million of insurance proceeds related to damaged property and equipment and \$2.1 million of proceeds from the sale of property and equipment.

Financing Activities. Net cash provided by financing activities was \$36.3 million for the six months ended June 30, 2022, primarily attributable to gross proceeds of \$68.4 million from the issuance of long-term debt, common stock and warrants, offset in part by \$19.0 million of payments on our Senior Secured Term Loan, \$7.2 million of net payments on our ABL Credit Facility and debt issuance costs of \$1.1 million.

Net cash provided by financing activities was \$97.5 million for the six months ended June 30, 2021, primarily attributable to \$86.5 million of proceeds from the issuance of Convertible Senior Notes, net borrowings of \$10.0 million on our ABL Credit Facility, \$5.4 million of net proceeds from notes payable and net proceeds of \$13.6 million from the issuance of common stock, offset in part by \$12.6 million of payments on our Senior Secured Term Loan and debt issuance costs of \$6.6 million.

#### Contractual Cash Obligations and Other Commitments

Our material cash commitments from known contractual and other obligations consist primarily of debt service obligations, including interest, operating and finance leases, and purchase commitments.

Senior Secured Term Loan. As of June 30, 2022, the outstanding principal balance of our Senior Secured Term Loan was \$102.8 million, of which \$8.8 million is due within one year. The Senior Secured Term Loan matures on December 5, 2025. As of June 30, 2022, we were in compliance with all of the covenants under our Senior Secured Term Loan.

The Senior Secured Term Loan interest rate was 0.0% for the first quarter of 2022 and an interest rate of (i) 1.0% per annum in cash and (ii) 4.125% per annum PIK interest from April 1, 2022 through December 31, 2022. The Senior Secured Term Loan requires quarterly principal payments of \$1.25 million until March 31, 2023 and \$5.0 million from June 30, 2023 through September 30, 2025, with final payment due at maturity.

Term C Loans. As of June 30, 2022, the outstanding principal balance of our Term C Loans were \$22.5 million, which mature on December 5, 2025. Our Term C Loans have a PIK interest rate of 14.0% and contains provisions with up to a 100% premium payable upon any repayment, prepayment or acceleration. As of June 30, 2022, we were in compliance with all of the covenants under our Term C Loans.

ABL Credit Facility. All borrowings under our ABL Credit Facility are subject to the satisfaction of customary conditions, including the absence of a default and the accuracy of representations and warranties and

certifications regarding sales of certain inventory, and to a borrowing base. As of June 30, 2022, we had an outstanding balance of \$7.0 million on our ABL Credit Facility. The ABL Credit Facility matures on April 1, 2025. As of June 30, 2022, we were in compliance with all of the covenants under our ABL Credit Facility.

Purchase Commitments. We have entered into purchase agreements and placed orders for equipment related to the buildout of our Nyx Clean Flee®. As of June 30, 2022, we have purchase commitments of \$16.3 million for the remainder of 2022, which is included in our 2022 growth capital expenditures below, and \$16.4 million due in 2023, under these agreements and purchase orders.

Capital Expenditures. Our business requires continual investments to upgrade or enhance existing property and equipment and to ensure compliance with safety and environmental regulations. Capital expenditures primarily relate to maintenance capital expenditures, growth capital expenditures and fleet enhancement capital expenditures. Maintenance capital expenditures include expenditures needed to maintain and to support our current operations. Growth capital expenditures include expenditures for additional fleets and generate incremental distributable cash flow. Fleet enhancement capital expenditures include expenditures on new equipment related to technology enhancements to existing fleets that increase the productivity of the fleet.

During the six months ended June 30, 2022, our capital expenditures was \$48.3 million. We currently expect that growth capital expenditures, on an accrual basis, will be approximately \$65 million to \$85 million for the remainder of 2022, primarily related to the buildout of our four new Nyx Clean Fleets® and associated equipment. Capital expenditures for growth and fleet enhancement initiatives are discretionary. We continuously evaluate our capital expenditures and the amount we ultimately spend will depend on several factors, including expected industry activity levels and company initiatives.

#### DESCRIPTION OF PROFRAC CAPITAL STOCK

#### General

The following description summarizes the material terms of ProFrac's capital stock, as expected to be in effect upon consummation of the Merger. This description summarizes the provisions included in the ProFrac amended and restated certificate of incorporation and the ProFrac amended and restated bylaws. Because this description is only a summary, it does not contain all of the information that may be important to you. For a complete description of the matters set forth in this section titled "Description of ProFrac Capital Stock", you should refer to the ProFrac amended and restated certificate of incorporation and the ProFrac amended and restated bylaws, which are included as exhibits to the registration statement of which this proxy statement/information statement/prospectus forms a part, and to the applicable provisions of Delaware law.

## **Authorized and Outstanding Shares**

The ProFrac certificate of incorporation authorizes the issuance of 1,050,000,000 shares of capital stock, consisting of:

- 600,000,000 shares of ProFrac Class A Common Stock, par value \$0.01 per share,
- 400,000,000 shares of ProFrac Class B Common Stock, par value \$0.01 per share, and
- 50,000,000 shares of ProFrac Preferred Stock, par value \$0.01 per share.

As of the record date for the Special Meeting, there are expected to be 41,239,957 shares of ProFrac Class A Common stock issued and outstanding, held of record by [\_\_\_] holder of ProFrac Class A Common Stock, and 101,133,201 shares of ProFrac Class B Common Stock issued and outstanding, held of record by [\_\_\_] holders of ProFrac Class B Common Stock. There are no shares of ProFrac Preferred Stock are issued and outstanding.

The number of authorized shares of ProFrac Class A Common Stock, ProFrac Class B Common Stock or ProFrac Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the outstanding shares of stock entitled to vote thereon, and no vote of the holders of ProFrac Class A Common Stock, ProFrac Class B Common Stock or ProFrac Preferred Stock, or of any series thereof, voting separately as a class shall be required therefor.

#### No Preemptive or Preferential Rights

No stockholder shall, by reason of holding the shares of any class or series of capital stock of ProFrac, have any preemptive or preferential rights to acquire or subscribe for any shares or securities of any class (whether now or hereafter authorized), which may at any time be issued, sold or offered for sale by ProFrac, unless specifically provided for in the terms of a series of Preferred Stock.

#### ProFrac Class A Common Stock

Voting Rights. Holders of shares of ProFrac Class A Common Stock are entitled to one vote per share held of record on all matters to be voted upon by the stockholders. The holders of ProFrac Class A Common Stock do not have cumulative voting rights in the election of directors. Holders of ProFrac Class A Common Stock and ProFrac Class B Common Stock vote together as a single class on all matters presented to ProFrac Stockholders, except with respect to the amendment of certain provisions of ProFrac's certificate of incorporation (including any designation of ProFrac Preferred Stock) that relates to the terms of one or more outstanding series of ProFrac Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to the certificate of incorporation (including any designation of ProFrac Preferred Stock) or as otherwise required under the DGCL.

Dividend Rights. Holders of shares of ProFrac Class A Common Stock are entitled to receive ratably, in proportion to the number of shares of ProFrac Class A Common Stock held, such dividends and distributions (payable in cash, stock or otherwise), if any, as may be declared by the ProFrac Board at any time and from time to time out of funds legally available for that purpose, subject to any prior rights and preferences, if any, applicable to shares of Preferred Stock or any series thereof.

Liquidation Rights. In the event of any voluntary or involuntary liquidation, dissolution or winding-up of ProFrac, following the distribution of the preferential amounts, if any, to the holders of shares of Preferred Stock or any series thereof, the holders of shares of ProFrac Class A Common Stock are entitled to receive the remaining assets of ProFrac as may be available for distribution to stockholders of ProFrac, ratably in proportion to the number of shares of Class A Common Stock held.

#### ProFrac Class B Common Stock

Voting Rights. Holders of ProFrac Class B Common Stock are entitled to one vote per share held of record on all matters to be voted upon by ProFrac Stockholders. The holders of ProFrac Class A Common Stock do not have cumulative voting rights in the election of directors. Holders of ProFrac Class A Common Stock and ProFrac Class B Common Stock vote together as a single class on all matters presented to ProFrac Stockholders, except with respect to the amendment of certain provisions of ProFrac's certificate of incorporation (including any designation of Preferred Stock) that relates to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to the certificate of incorporation (including any designation of Preferred Stock) or as otherwise required under the DGCL.

Dividend Rights. Holders of ProFrac Class B Common Stock do not have any right to receive or be paid dividends and other distributions, unless (i) the dividend consists of shares of ProFrac Class B Common Stock or of rights, options, warrants or other securities convertible into or exercisable, exchangeable or redeemable for shares of ProFrac Class B Common Stock paid proportionally with respect to each outstanding share of ProFrac Class B Common Stock and (ii) a dividend consisting of shares of ProFrac Class A Common Stock or of rights, options, warrants or other securities convertible into or exercisable, exchangeable or redeemable for, shares of ProFrac Class A Common Stock on equivalent terms is simultaneously paid to the holders of ProFrac Class A Common Stock.

Liquidation Rights. In the event of any voluntary or involuntary liquidation, dissolution or winding-up of ProFrac, the holders of ProFrac Class B Common Stock are not entitled to receive any assets of ProFrac.

#### **Preferred Stock**

ProFrac's certificate of incorporation authorizes the ProFrac Board, subject to any limitations prescribed by law and the rights of any series of Preferred Stock then outstanding, if any, to designate and to issue from time to time one or more classes or series of Preferred Stock, up to an aggregate of 50,000,000 shares of Preferred Stock. Each class or series of Preferred Stock will have the powers, preferences, rights, qualifications, limitations and restrictions determined by the ProFrac Board, which may include, among others: the number of shares to constitute the series or class, dividend rights, liquidation preferences, voting rights, conversion rights, preemptive rights and redemption rights. Except as provided by law or expressly provided for in a designation of a class or series of Preferred Stock, the holders of Preferred Stock will not be entitled to vote at or receive notice of any meeting of the ProFrac Stockholders.

# Treatment of Warrants of USWS

Prior to the Effective Time, each of the issued and then-outstanding warrant agreements that govern the USWS Rollover Warrants (as such term is described elsewhere in this proxy statement/information statement/

prospectus) shall be amended in accordance with the terms of the Merger Agreement and the terms governing each of the respective USWS Rollover Warrants. Such amended USWS Rollover Warrants shall remain outstanding following the consummation of the Merger, but shall but shall represent the right to purchase, upon exercise thereof, shares of ProFrac Class A Common Stock, rather than shares of USWS Common Stock, based on a specified exchange ratio and a corresponding adjustment to the exercise price.

Each USWS Rollover Warrant shall remain outstanding following the Merger through its prescribed term but shall become the right to receive upon valid exercise thereof shares of ProFrac Class A Common Stock equal to the product of (A) the number of shares of USWS Common Stock subject to such USWS Rollover Warrant immediately prior to the Effective Time and (B) the Exchange Ratio. Additionally, each USWS Rollover Warrant will be amended such that the exercise price of such USWS Rollover Warrant will equal the current exercise price of such USWS Rollover Warrant divided by the Exchange Ratio, as described below:

- The USWS SPAC Warrants' per-share exercise price will become \$717.47 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$241.50 per full share exercise price of such warrants by the 0.3366 Exchange Ratio;
- The USWS Series A Warrants' per-share exercise price will become \$477.89 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$160.86 per full share exercise price of such warrants by the 0.3366 Exchange Ratio;
- The USWS Private Placement Investor Warrants' per-share exercise price will become \$31.43 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$10.58 per full share exercise price of such warrants by the 0.3366 Exchange Ratio; and
- The USWS Private Placement Agent Warrants' per-share exercise price will become \$39.28 per share of ProFrac Class A Common Stock. This is calculated by dividing the current \$13.22 per full share exercise price of such warrants by the 0.3366 Exchange Ratio.

The amendments to each of the Rollover Warrants are filed as Exhibits 10.31, 10.32, 10.33, and 10.34 to this proxy statement/information statement/prospectus and incorporated herein by reference. The foregoing descriptions of the amendments to the USWS Rollover Warrants are qualified in their entirety by reference to such Exhibits.

#### Anti-Takeover Effects of Certain Provisions of Delaware Law and ProFrac's Certificate of Incorporation and Bylaws

Delaware law and ProFrac's certificate of incorporation and bylaws contain provisions that could have the effect of delaying, deferring or discouraging another party from acquiring a controlling interest of ProFrac by means of a tender offer, a proxy contest or otherwise, or to remove incumbent officers. These provisions, which are summarized below, are expected to discourage coercive takeover practices and inadequate takeover bids. These provisions are also designed to encourage persons seeking to acquire control of ProFrac to first negotiate with the ProFrac Board. ProFrac believes that the benefits of increased protection of its ability to negotiate with the proponent of an unfriendly or unsolicited proposal to acquire or restructure it outweighs the disadvantages of discouraging takeover or acquisition proposals because, among other things, negotiation of such proposals could result in an improvement of their terms and conditions.

## **Delaware Business Combination Statute**

Section 203 of the DGCL prohibits a Delaware corporation, including a Delaware corporation whose securities are listed for trading on Nasdaq, from engaging in a "business combination" with an "interested stockholder" (as such terms are defined under Section 203 of the DGCL) for a period of three years following the date that the stockholder became an interested stockholder, unless:

• prior to the date of the transaction or business combination, the board of directors of the corporation approved either the business combination or the transaction that resulted in the stockholder becoming an interested stockholder;

- upon consummation of the transaction that resulted in the shareholder becoming an interested shareholder, the interested shareholder owned at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced, excluding, for purposes of determining the number of shares outstanding (i) the shares owned by persons who are directors and also officers of the corporation, and (ii) shares owned by employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to such a plan will be tendered in a tender or exchange offer; or
- on or after such time the business combination is approved by the board of directors and authorized at a meeting of stockholders by the affirmative vote of at least two-thirds of the outstanding voting stock that is not owned by the interested stockholder.

As defined under Section 203 of the DGCL, a "business combination" includes a merger, asset or stock sale, or other transactions resulting in a financial benefit to the interested stockholder. An "interested stockholder" is a person who, together with affiliates and associates, owns or, within three years prior to the determination of interested stockholder status owned, 15% or more of the corporation's outstanding voting stock.

A Delaware corporation may "opt out" of Section 203 with an express provision in its original certificate of incorporation or an express provision in its certificate of incorporation or bylaws resulting from amendments approved by the holders of at least a majority of the corporation's then outstanding voting shares. ProFrac has elected to "opt out" of Section 203 of the DGCL.

# ProFrac's Certificate of Incorporation and Bylaws

Certain provisions of ProFrac's certificate of incorporation and bylaws may have an anti-takeover effect and may delay, deter or discourage a merger, acquisition, tender offer, takeover attempt or other change of control transactions that stockholders of ProFrac may consider to be their best interest, including such attempts that might result in a premium over the market price for the shares held by ProFrac's stockholders. Among other things, ProFrac's certificate of incorporation and bylaws including the following provisions:

Advance Notice Requirements for Stockholder Proposals and Director Nominations.

The ProFrac bylaws establish advance notice procedures with regard to shareholder proposals relating to the nomination of candidates for election as directors or any other business to be brought before meetings of ProFrac Stockholders. These procedures provide that notice of shareholder proposals must be timely given in writing to the Secretary of ProFrac prior to the meeting at which the action is to be taken. Generally, to be timely, notice must be received by the Secretary of ProFrac at the principal executive offices of ProFrac not earlier than the close of business on the 120th day and not later than the close of business on the 90th day prior to the first anniversary of the annual meeting of ProFrac Stockholders for the preceding year. ProFrac's bylaws specify the requirements as to form and content of all stockholders' notices. These requirements may preclude ProFrac Stockholders from bringing matters before the ProFrac Stockholders at an annual or special meeting.

## Authorized but Unissued Shares

The ProFrac certificate of incorporation expressly grants the ProFrac Board the power to authorize the issuance and designation of Preferred Stock from time to time in one or more classes or series, without Shareholder approval. The ProFrac Board is authorized to designate the powers, preferences, rights, qualifications, limitations and restrictions for each series of Preferred Stock, subject to any limitations imposed by applicable law. The existence of these authorized but unissued shares of ProFrac Preferred Stock could make it more difficult or discourage an attempt to obtain control of ProFrac by means of a proxy contest, tender offer, merger or otherwise.

Voting Power

The ProFrac certificate of incorporation provides that until ProFrac ceases to be a "controlled company," the business and affairs of ProFrac shall be managed by or under the direction of the Board, and the members of the ProFrac Board, who are to be designated by the parties to the ProFrac Stockholders' Agreement dated as of May 17, 2022 (as it may be amended, restated, supplemented or otherwise modified from time to time), will have a majority of the voting power of the ProFrac Board.

Size of the Board

The ProFrac bylaws provide that the number of directors is to be fixed, from time to time, exclusively by a resolution adopted by a majority of the ProFrac Board, subject to applicable laws and the terms of the ProFrac Stockholders' Agreement and the rights of the holders of any series of ProFrac Preferred Stock to elect directors under specified circumstances.

Vacancies on the Board

The ProFrac certificate of incorporation provides that, after ProFrac ceases to be a "controlled company," and subject to applicable law and the rights of the holders of any series of Preferred Stock of ProFrac then outstanding and the then-applicable terms of the ProFrac Stockholders' Agreement, all vacancies on the ProFrac Board, including newly created directorships, may only be filled by the affirmative vote of a majority of directors then in office, even if less than a quorum, or by a sole remaining director, and shall not be filled by the stockholders.

Stockholder Action by Written Consent; Special Meetings

The ProFrac bylaws provide that until ProFrac ceases to be a "controlled company," any action required or permitted to be taken at any annual or special meeting of the stockholders may be taken without a meeting, without prior notice and without a vote of the stockholders, so long as a consent setting forth the actions to be taken is signed by the holders of the outstanding stock having not less than the minimum number of votes necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted. After ProFrac ceases to be a "controlled company," subject to the rights of the holders of any series of Preferred Stock with respect to such series of Preferred Stock, any action required or permitted to be taken by the stockholders of ProFrac must be taken at a duly held annual or special meeting of the stockholders, and may not be taken by any consent of such stockholders. The ProFrac certificate of incorporation and the bylaws provide that special meetings of the stockholders may only be called by the Executive Chairman, the Chief Executive Officer, or by the Board pursuant to a resolution adopted by a majority of the Board; provided, however, that until ProFrac ceases to be a "controlled company," special meetings of the stockholders may also be called by the Secretary of ProFrac at the request of the holders of more than 50% of the then outstanding shares of common stock. After ProFrac ceases to be a "controlled company," special meetings of ProFrac Stockholders may only be called by the Executive Chairman, the Chief Executive Officer, or by the Board pursuant to a resolution adopted by a majority of the Board.

Amendment of ProFrac's certificate of incorporation and bylaws

The DGCL provides, generally, that the affirmative vote of a majority of the shares entitled to vote on any matter is required to amend a corporation's certificate of incorporation or bylaws, unless a corporation's certificate of incorporation or bylaws require a greater percentage. The ProFrac certificate of incorporation and bylaws separately provide that until ProFrac ceases to be a "controlled company," the affirmative vote of the holders of a majority of the voting power of the then outstanding shares of ProFrac stock entitled to vote thereon, voting as a single class, shall be required to amend, alter or repeal any provision of the ProFrac certificate of incorporation and bylaws. The ProFrac certificate of incorporation and bylaws separately provide that, after ProFrac ceases to be a "controlled company," each of ProFrac's certificate of incorporation and ProFrac's bylaws

provide that an affirmative vote of the holders of at least 66 2/3% in voting power of the then outstanding shares of ProFrac stock entitled to vote thereon, voting together as a single class, shall be required to amend, alter or repeal any provision of the ProFrac certificate of incorporation and bylaws. Subject to the provisions of the ProFrac certificate of incorporation, the ProFrac bylaws may be amended, altered or repealed by a resolution adopted by a majority of the directors present at any special or regular meeting of the Board at which a quorum is present if, in the case of a special meeting only, notice of such amendment, alteration or repeal is contained in the notice or waiver of notice of such meeting.

## Staggered Board; Removal of Directors

The ProFrac certificate of incorporation provides that after ProFrac ceases to be a "controlled company," the ProFrac directors, other than those that may be elected by the holders of Preferred stock, shall be divided, with respect to the time for which they severally hold office, into three classes, as nearly equal in number as is reasonably possible, with the initial term of office for the first class to expire at the first annual meeting of the stockholders following the date on which ProFrac ceased to be a "controlled company," the initial term of office for the second class to expire at the second annual meeting of stockholders following the date on which ProFrac ceased to be a "controlled company," and the third initial term of office for the third class to expire at the third annual meeting of stockholders following the date on which ProFrac ceased to be a "controlled company." The ProFrac certificate of incorporation provides that until ProFrac ceases to be a "controlled company," and subject to the rights of the holders of shares of Preferred Stock and the then-applicable terms of the Stockholders' Agreement, any director may be removed at any time, either for or without cause, upon the affirmative vote of the holders of a majority of the voting power of the outstanding shares entitled to vote generally for the election of directors, voting together as a single class and acting at a meeting of the stockholders or by written consent. After ProFrac ceases to be a "controlled company," subject to the rights of the holders of shares of Preferred Stock and the then-applicable terms of the Stockholders' Agreement, any director may be removed only for cause, upon the affirmative vote of the holders of at least 66 2/3% of the outstanding shares of stock entitled to vote generally for the election of directors, voting together as a single class and acting at a meeting of the stockholders or by written consent. The classification of ProFrac's Board and the limitations on the removal of directors could make it mo

These anti-takeover provisions could make it more difficult for a third-party to acquire ProFrac, even if the third-party's offer may be considered to be beneficial by stockholders of ProFrac. Consequently, stockholders of ProFrac may be limited in their ability to obtain a premium for their shares. These provisions could also discourage proxy contests and make it more difficult for you and other stockholders to elect directors of your choosing and to cause ProFrac to take such other corporate actions that you may desire.

#### No Cumulative Voting

Under the DGCL, the right to vote cumulatively does not exist unless the certificate of incorporation specifically authorizes cumulative voting. ProFrac's certificate of incorporation does not authorize cumulative voting. Therefore, ProFrac Stockholders of the majority of the shares of stock entitled to vote in any election of directors may elect all of the ProFrac directors standing for election.

#### **Forum Selection**

ProFrac's certificate of incorporation provides that unless ProFrac consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware, or, if such court does not have subject matter jurisdiction thereof, the federal district court of the State of Delaware, will, to the fullest extent permitted by applicable law, be the sole and exclusive forum for:

· any derivative action or proceeding brought on ProFrac's behalf;

- any action or proceeding asserting a claim of breach of a fiduciary duty owed by any of ProFrac's current or former directors, officers, employees or ProFrac Stockholders to ProFrac or ProFrac Stockholders;
- any action asserting a claim arising pursuant to any provision of the DGCL, ProFrac's certificate of incorporation or ProFrac's bylaws
  (as either may be amended or restated) or as to which the DGCL confers jurisdiction on the Court of Chancery of the State of
  Delaware: or
- · any action asserting a claim that is governed by the internal affairs doctrine.

In each such case, the action or proceeding is subject to such Court of Chancery having personal jurisdiction over the indispensable parties named as defendants therein.

To the fullest extent permitted by law, ProFrac's certificate of incorporation also provides that any person or entity purchasing or otherwise acquiring any interest in shares of ProFrac's capital stock will be deemed to have notice of, and to have consented to, this forum selection provision. Although ProFrac believes these provisions will benefit ProFrac by providing increased consistency in the application of Delaware law for the specified types of actions and proceedings, the provisions may have the effect of discouraging lawsuits against ProFrac's directors, officers, employees and stockholders. However, ProFrac Stockholders will not be deemed to have waived ProFrac's compliance with the federal securities laws and the rules and regulations thereunder, and the exclusive forum provision would not apply to suits brought to enforce a duty or liability created by the 1933 Act or the 1934 Act, as amended.

ProFrac's certificate of incorporation also provides that the federal district courts of the United States will be the exclusive forum for any complaint asserting a cause of action under the 1933 Act. Section 22 of the 1933 Act creates concurrent jurisdiction for federal and state courts over all suits brought to enforce any duty or liability created by the 1933 Act or the rules and regulations thereunder. Accordingly, there is uncertainty as to whether a court would enforce this forum provision providing for exclusive jurisdiction of federal district courts with respect to suits brought to enforce any duty or liability created by the 1933 Act. The enforceability of similar exclusive forum provisions in other companies' certificates of incorporation has been challenged in legal proceedings, and it is possible that, in connection with one or more actions or proceedings described above, a court could rule that this provision in ProFrac's certificate of incorporation is inapplicable or unenforceable.

#### Limitation of Liability and Indemnification Matters

ProFrac's certificate of incorporation provides that directors of ProFrac shall not be liable to ProFrac or stockholders of ProFrac for monetary damages for a director's breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permissible under the DGCL, as amended. ProFrac's certificate of incorporation also provides that, if the DGCL is subsequently amended to permit the exculpation of officers for monetary damages for breach of their fiduciary duty as officers, then, from and after the effective time of such amendment to the DGCL, ProFrac's officers shall not be liable to ProFrac or to stockholders of ProFrac for monetary damages for breach of their fiduciary duty as officers, except to the to the extent such exemption from liability or limitation thereof is not permitted under the DGCL.

Any amendment, repeal or modification of these provisions of the ProFrac certificate of incorporation will be prospective only, and would not adversely affect any right or protection of a director or officer of ProFrac, as such exist under the certificate of incorporation of ProFrac, for acts or omissions that occurred prior to any such amendment, repeal or modification.

The ProFrac certificate of incorporation and bylaws provide that ProFrac shall indemnify and hold harmless, to the fullest extent permissible under applicable law (as it exists or may thereafter be amended) any person who was or is made a party to or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative by reason

of the fact that this he or she, or a person for whom he or she is the legal representative, is or was a director or officer of ProFrac or, while serving as a director or officer of ProFrac, is or was serving at the request of ProFrac as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, other enterprise or non-profit entity, including service with respect to an employee benefit plan, whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all expenses, liability and loss reasonably incurred in connection with such a proceeding. The ProFrac bylaws permit ProFrac to maintain insurance, at the expense of ProFrac, to protect ProFrac and any person who is or was serving as a director, officer, employee or agent of ProFrac as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise or non-profit entity, including service with respect to an employee benefit plan, against any expense, liability or loss, whether or not ProFrac would have the power to indemnify such person against any such expense, liability or loss under the DGCL.

ProFrac has entered into indemnification agreements with each of its current directors and officers, and intends to enter into indemnification agreements with each of its future directors and officers. These agreements require (and any such future agreement are expected to require) ProFrac to indemnify these individuals to the fullest extent permitted under Delaware law against liability that may arise by reason of their service to ProFrac, and to advance expenses incurred as a result of any proceeding against them, to the extent permissible under the DGCL and subject to the terms of the indemnification agreements.

These provisions may discourage stockholders from bringing actions or proceedings against ProFrac's directors for breach of their fiduciary duty. Additionally, these provisions may have the effect of reducing the likelihood of derivative claims against directors and officers of ProFrac, even though such an action, if successful, may otherwise benefit ProFrac or stockholders of ProFrac. A stockholder's investment in ProFrac may be adversely affected to the extent ProFrac pays the costs of settlement or damage awards against directors and officers under these the indemnification provisions. ProFrac believes that the limitation of liability provision in ProFrac's certificate of incorporation and the indemnification agreements described above are necessary to continue to attract and retain qualified individuals to serve as directors and officers of ProFrac.

# Dissenters' Rights of Appraisal and Payment

Under the DGCL, with certain exceptions, ProFrac Stockholders will have appraisal rights in connection with a merger or consolidation. Pursuant to the DGCL, ProFrac Stockholders who properly request and perfect appraisal rights in connection with such merger or consolidation will have the right to receive payment of the fair value of their shares as determined by the Delaware Court of Chancery.

### Sales of Restricted Shares

All shares of ProFrac Class A Common Stock held by the ProFrac LLC Unit Holders and other recipients of ProFrac Class A Common Stock in the Corporate Reorganization in connection with the ProFrac IPO are deemed to be "restricted securities" as such term is defined under Rule 144. The restricted securities were issued and sold by ProFrac in private transactions, and are eligible for public sale only if registered under the 1933 Act or if they qualify for an exemption from registration under Rule 144 under the 1933 Act, which is summarized below.

Each ProFrac LLC Unit Holder, subject to certain limitations, has the right, pursuant to the Redemption Right, to cause ProFrac LLC to acquire all or a portion of its ProFrac LLC Units for shares of ProFrac Class A Common Stock (on a one-for-one basis, subject to conversion rate adjustments for stock splits, stock dividends and reclassifications and similar transactions). See "Information About ProFrac — Certain Relationships and Related Party Transactions—ProFrac LLC Agreement." The shares of ProFrac Class A Common Stock ProFrac issues upon such redemptions would be "restricted securities" as defined in Rule 144 described below. However, in connection with the ProFrac IPO, ProFrac entered into a registration rights agreement with certain of the ProFrac LLC Unit Holders that will require ProFrac to register under the 1933 Act these shares of ProFrac

Class A Common Stock. See "Information About ProFrac — Certain Relationships and Related Party Transactions — ProFrac Founders Registration Rights Agreement."

As a result of the lock-up agreements described below and the provisions of Rule 144 under the 1933 Act, the shares of ProFrac Class A Common Stock (excluding the shares to be sold in the ProFrac IPO) that are available for sale in the public market are as follows:

- no shares will be eligible for sale prior to 180 days after the date of the ProFrac Prospectus (other than the 5,200,000 shares of ProFrac Class A Common Stock purchased by THRC Holdings, the Farris and Jo Ann Wilks 2022 Family Trust and their affiliates, which will be eligible for sale when permitted under Rule 144 or Rule 701 under the 1933 Act); and
- 122,570,855 shares (assuming redemption of all applicable ProFrac LLC Units) will be eligible for sale upon the expiration of the lock-up agreements, beginning 180 days after the date of the ProFrac Prospectus when permitted under Rule 144 or Rule 701.

### Lock-up Agreements

All of ProFrac's directors and officers and certain of the ProFrac LLC Unit Holders agreed not to sell any ProFrac Class A Common Stock for a period of 180 days from May 12, 2022, subject to certain exceptions and extensions. These agreements do not relate to the 5,200,000 shares of ProFrac Class A Common Stock purchased by THRC Holdings, the Farris and Jo Ann Wilks 2022 Family Trust and their affiliates, though these shares will remain subject to resale restrictions under federal securities laws.

### Transfer Agent and Registrar

The transfer agent and registrar for ProFrac's Class A Common Stock is American Stock Transfer & Trust Company, LLC.

# **Listing of Securities**

The ProFrac Class A Common Stock is currently listed on the Nasdaq Global Select Market under the symbol "PFHC." It is expected that following the Closing of the Merger, the ProFrac Class A Common Stock will continue to be listed on the Nasdaq Global Select Market under the symbol "PFHC." Upon consummation of the Merger and effectiveness of this Registration Statement on Form S-4, the publicly tradable warrants of USWS, currently listed on the Nasdaq Capital Market under the symbol "USWSW," shall become publicly tradable warrants of ProFrac and shall be listed on the Nasdaq Global Select Market under the symbol "PFHCW."

### SECURITIES ACT RESTRICTIONS ON RESALE OF COMMON STOCK

## **Rule 144**

Generally, pursuant to Rule 144 under the Securities Act ("Rule 144"), a person who has beneficially owned restricted shares or warrants for at least six months would be entitled to sell their securities, provided that (i) such person is not deemed to have been one of the affiliates of ProFrac at the time of, or at any time during the three months preceding, a sale and (ii) ProFrac is subject to the Exchange Act periodic requirements for at least three months before the sale and has filed all required reports under Section 13 or 15(d) of the Exchange Act during the twelve months (or for such shorter period as ProFrac was required to file reports) preceding the sale.

Persons who have beneficially owned restricted shares or warrants for at least six months, but who are affiliates of ProFrac at the time of, or at any time during the three months preceding, a sale, would be subject to

additional restrictions, by which such person would be entitled to sell within any three-month period only a number of securities that does not exceed the greater of:

- 1% of the total number of ProFrac Class A Common Stock then outstanding; or
- the average weekly reported trading volume of ProFrac Class A Common Stock during the four calendar weeks preceding the filing of a notice on Form 144 with respect to the sale.

Sale by affiliates of ProFrac under Rule 144 are also limited by manner of sale provisions and notice requirements and to the availability of current public information about ProFrac.

### COMPARISON OF STOCKHOLDER RIGHTS AND CORPORATE GOVERNANCE MATTERS

ProFrac and USWS are both incorporated under Delaware law. Any differences, therefore, in the rights of ProFrac Stockholders and USWS Stockholders arise primarily from differences in their respective certificates of incorporation and bylaws. The following table compares the current rights of ProFrac Stockholders under the ProFrac certificate of incorporation and bylaws, and the current rights of USWS Stockholders under the USWS certificate of incorporation and bylaws. The following summary is not a complete statement of the rights of ProFrac Stockholders or USWS Stockholders or a complete description of the specific provisions referred to below. This summary does not purport to be complete and is qualified in its entirety by reference to the DGCL, ProFrac's certificate of incorporation and bylaws and USWS' certificate of incorporation and bylaws, which USWS Stockholders should read. For information on how copies of these documents may be obtained, please see "Where You Can Find More Information."

Authorized Capital Stock

### ProFrac Stockholder Rights

ProFrac's certificate of incorporation authorizes
ProFrac to issue 1,000,000,000 shares of ProFrac
Common Stock, of which 400,000,000 are
designated ProFrac Class A Common Stock and
600,000,000 are designated ProFrac Class B
Common Stock. ProFrac's certificate of
incorporation also authorizes ProFrac to issue
50,000,000 shares of preferred stock. As of
August 8, 2022, there were 41,239,957 shares of
ProFrac Class A Common Stock, 101,133,201
shares of ProFrac Class B Common Stock and 0
shares of ProFrac Preferred Stock issued and
outstanding

Voting Rights

Holders of shares of ProFrac Class A Common Stock are entitled to one vote per share held of record on all matters to be voted upon by the stockholders. Holders of ProFrac Class B Common Stock are entitled to one vote per share held of record on all matters to be voted upon by the stockholders. Holders of ProFrac Class A Common Stock and ProFrac Class B Common Stock vote together as a single class on all matters presented to the stockholders for their vote or

# **USWS Stockholder Rights**

USWS' certificate of incorporation authorizes USWS to issue 430,000,000 shares of USWS Common Stock, of which 400,000,000 shares are designated USWS Class A Common Stock, 20,000,000 are designated USWS Class B Common Stock and 10,000,000 are designated USWS Class F Common Stock, USWS' certificate of incorporation also authorizes USWS to issue 10,000,000 shares of USWS Preferred Stock. As of August 15, 2022, there were 112,851,347 shares of USWS Class A Common Stock, and no shares of USWS Class B Common Stock, no shares of USWS Class F Common Stock and 19,610 shares of USWS Series A Preferred Stock issued and

Holders of shares of USWS Class A
Common Stock are entitled to one vote for
each share held on all matters to be voted on
by the stockholders. Holders of shares of
USWS Class B Common Stock are entitled
to one vote per share held of record on all
matters to be voted upon by the stockholders.
Holders of USWS Class A Common Stock
and holders of USWS Class B Common
Stock will vote together as a single class on
all matters submitted to a vote of the

Number of Directors and Size of Board

Action of Directors

### ProFrac Stockholder Rights

approval, except with respect to the amendment of certain provisions of ProFrac's certificate of incorporation that would alter or change the powers, preferences or special rights of the ProFrac Class B Common Stock. The holders of ProFrac Common Stock do not have cumulative voting rights.

The ProFrac Board currently has five (5) members. ProFrac's certificate of incorporation provides that, subject to the ProFrac Stockholders' Agreement and the rights of the holders of any series of preferred stock to elect directors under specified circumstances, if any, the number of directors shall be fixed from time to time exclusively pursuant to a resolution adopted by a majority of the ProFrac Board. ProFrac's certificate of incorporation provides that, prior to the Trigger Date, on any matter to be voted on or consented to by the ProFrac Board (i) each Non-Wilks Director (as defined in ProFrac's certificate of incorporation) shall be entitled to cast one vote, (ii) the Wilks Directors (as defined in ProFrac's certificate of incorporation) shall collectively be entitled to cast an aggregate number of votes in such an amount that, at any time, the Wilks Directors in office at such time shall collectively be entitled to cast a majority of the votes that may be cast by the directors (the "Aggregate Wilks Director Voting Power"), and (iii) each Wilks Director shall be entitled to cast a number of votes with respect to such matter (including any fractions thereof) equal to the quotient of (A) the Aggregate Wilks Director Voting Power,

### USWS Stockholder Rights

Company's stockholders, except with respect to the amendment of certain provisions of USWS' certificate of incorporation that would alter or change the powers, preferences or special rights of the USWS Class B Common Stock. The holders of USWS' Common Stock do not have cumulative voting rights.

The USWS Board currently has nine (9) members. USWS certificate of incorporation provides that the number of directors of USWS, other than those who may be elected by one or more series of preferred stock voting separately by class or series, shall be fixed from time to time exclusively pursuant to a resolution adopted by a majority of the USWS Board. USWS' bylaws provide that the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the USWS Board.

Term of Directors

Election of Directors

Removal of Directors

#### ProFrac Stockholder Rights

divided by (B) the number of Wilks Directors voting on such matter. On and after the Trigger Date, each director, including the Wilks Directors, if any, shall be entitled to cast one vote on all matters to be voted on by the ProFrac Board

ProFrac's certificate of incorporation provides that, prior to the Trigger Date, all directors are to be elected annually, and each director shall hold office until his or her successor shall have been duly elected and qualified, or otherwise until his or her death, resignation, disqualification or removal. On and after the Trigger Date, the directors shall be divided into three classes with staggered three-year terms, with each director to hold office until his or her successor shall have been duly elected and qualified, or otherwise until his or her death, resignation, disqualification or removal.

ProFrac's bylaws provide that, subject to certain exceptions in the ProFrac Stockholders' Agreement and special rights of the holders of one or more series of preferred stock to elect directors, the directors of ProFrac shall be elected by a plurality of the votes of the shares cast by the holders of shares entitled to vote for the election of directors.

ProFrac's certificate of incorporation provides that, prior to the Trigger Date, any director may be removed at any time, either for or without cause, by the holders of a majority of the shares of ProFrac stock then entitled to vote for the election of directors. On and after the Trigger Date, any

USWS Stockholder Rights

USWS' certificate of incorporation provides that the directors shall be divided into three classes with staggered three-year terms. Each director shall hold office until the annual meeting for the year in which his or her term expires and until his or her successor has been elected and qualified, or otherwise until his or her death, resignation, retirement, disqualification or removal.

USWS' bylaws provide that, subject to special rights of the holders of one or more series of preferred stock to elect directors, the directors of USWS shall be elected by a plurality of the votes of the shares cast by the holders of shares entitled to vote for the election of directors.

USWS' certificate of incorporation provides that, subject to special rights of the holders of one or more series of preferred stock, any or all of the directors may be removed from office at any time, but only for cause and only by the affirmative vote of holders of at least

Vacancies

Quorum for Board Meetings

# **ProFrac Stockholder Rights**

or all of the directors may be removed from office at any time, but only for cause and only by the affirmative vote of the holders of at least two-thirds of the voting power of all of the then outstanding shares entitled to vote for the election of directors.

ProFrac's certificate of incorporation provides that, prior to the Trigger Date, subject to certain exceptions in the ProFrac Stockholders' Agreement and special rights of the holders of one or more series of preferred stock, any vacancy on the ProFrac Board that results from the death, resignation, disqualification or removal of any director or from any other cause, and any newly-created directorships resulting from any increase in the number of directors, shall be filled by (a) the affirmative vote of a majority of the total number of directors then in office, or (b) the affirmative vote of the holders of a majority of the voting power of the then outstanding shares entitled to vote generally for the election of directors, voting together as a single class. On and after the Trigger Date, any vacancy on the ProFrac Board that results from the death. resignation, disqualification or removal of any director or from any other cause, and any newlycreated directorships resulting from any increase in the number of directors, shall be filled solely by the affirmative vote of a majority of the total number of directors then in office. ProFrac's bylaws provide that the presence of at

ProFrac's bylaws provide that the presence of at least a majority of the total number of authorized directors constitutes a quorum for

# **USWS Stockholder Rights**

two-thirds of the voting power of all of the then outstanding shares entitled to vote for the election of directors.

USWS' certificate of incorporation provides that any vacancies on the USWS Board resulting from the death, resignation, retirement, disqualification, removal of any director or from any other cause, and any newly-created directorships resulting from any increase in the number of directors, shall be filled by the affirmative vote of a majority of the total number of directors then in office.

USWS' bylaws provide that the presence of at least a majority of the total number of authorized directors constitutes a quorum for

Annual Meetings of Stockholders

Quorum for Stockholder Meetings

Notice of Annual and Special Meetings of Stockholders

Calling Special Meetings of Stockholders

#### ProFrac Stockholder Rights

the transaction of business and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the ProFrac Board; provided that, prior to the Trigger Date, unless otherwise waived by any Wilks Director not attending a Board meeting, or in the event that a Wilks Director is incapacitated, each of the Wilks Directors must be present in order to establish a quorum for the transaction of business by the ProFrac Board.

ProFrac's bylaws provide that the annual meetings of stockholders shall be held at such date, time and place as shall be designated by the ProFrac Board, the Executive Chairman or the Chief Executive Officer and stated in the notice of the meeting, at which time the stockholders may transact any business as may be properly brought before the meeting.

ProFrac's bylaws provide that the holders of a majority of the shares of ProFrac stock issued and outstanding and entitled to vote, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business.

ProFrac's bylaws provide that notice of any meeting of stockholders must be sent not less than 10 nor more than 60 days before the date of the meeting to each stockholder entitled to vote at the meeting.

ProFrac's certificate of incorporation and bylaws provide that special meetings of

### USWS Stockholder Rights

the transaction of business and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the USWS Board.

USWS' bylaws provide that the annual meeting of stockholders shall be held at such date, time and place shall be designated the USWS Board and stated in the notice of the meeting, at which time the stockholders shall elect those directors to fill any term of a directorship that expires on the date of such annual meeting and may transact any other business as may properly be brought before the meeting.

USWS' bylaws provide that the holders of a majority of the shares of USWS stock issued and outstanding and entitled to vote, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business. USWS' bylaws provide that notice of any meeting of stockholders must be sent not less than 10 nor more than 60 days before the date of the meeting to each stockholder entitled to vote at the meeting. USWS' certificate of incorporation and bylaws provide that special meetings of

Stockholder Action by Written Consent

Amendment of Bylaws

#### ProFrac Stockholder Rights

stockholders may be called only by the Executive Chairman, the Chief Executive Officer, or pursuant to a resolution adopted by a majority of the total number of directors then in office; provided that, prior to the Trigger Date, special meetings of stockholders shall also be called by the Secretary at the request of the holders of record of a majority of the outstanding shares of ProFrac Common Stock. On and after the Trigger Date, the stockholders of ProFrac shall not have the power to call special meetings of stockholders.

ProFrac's certificate and incorporation and bylaws provide that, prior to the Trigger Date, any action required or permitted to be taken at any annual meeting or special meeting of ProFrac Stockholders may be effected by written consent of such stockholders. On and after the Trigger Date, subject to the rights of holders of any series of preferred stock, any action required or permitted to be taken by the stockholders of ProFrac must be effected by a duly called annual or special meeting of stockholders and may not be effected by written consent of the stockholders. ProFrac's bylaws may be amended, altered or repealed (a) by the affirmative vote of a majority of the ProFrac Board, and (b) (i) prior to the Trigger Date, by the affirmative vote of the holders of a majority of the shares of ProFrac stock issued and outstanding and entitled to vote, voting together as a single class, and (ii) on and after the Trigger Date, by the affirmative vote of the holders of not less than two-thirds of the shares of ProFrac

# **USWS Stockholder Rights**

stockholders may be called only by the Chairman of the USWS Board, Chief Executive Officer, or pursuant to a resolution adopted by a majority of the total number of directors then in office. The stockholders of USWS do not have the power to call special meetings of stockholders.

USWS' certificate of incorporation provides that any action required or permitted to be taken by the USWS Stockholders must be effected by a duly called annual or special meeting of such stockholders and may not be effected by written consent of the stockholders other than with respect to the USWS Class B Common Stock and the USWS Class F Common Stock, with respect to which action may be taken by written consent.

USWS' bylaws may be amended, altered or repealed (a) by the affirmative vote of a majority of the USWS Board, and (b) by the affirmative vote of the holders of a majority of the shares of USWS stock issued and outstanding and entitled to vote, voting together as a single class.

Amendment of Certificate of Incorporation

Corporate Opportunities

### ProFrac Stockholder Rights

stock issued an outstanding and entitled to vote, voting together as a single class.

ProFrac's certificate of incorporation provides that ProFrac may amend its certificate of incorporation as provided therein and by the DGCL; provided that, (a) prior to the Trigger Date, the affirmative vote of a majority of shares of ProFrac stock issued and outstanding and entitled to vote, voting together as a single class, shall be required to amend, alter or repeal any provision of ProFrac's certificate of incorporation, and (b) on and after the Trigger Date, the affirmative vote of not less than two-thirds of shares of ProFrac stock issued an outstanding and entitled to vote, voting together as a single class shall be required to amend, alter or repeal any provision of ProFrac's certificate of incorporation.

ProFrac's certificate of incorporation provides that ProFrac has renounced any interest or expectancy of ProFrac and its subsidiaries in any business opportunities that are from time to time presented to any of the ProFrac Controlling Stockholders or any of their respective affiliates, agents, shareholders, members, partners, directors, officers, employees, affiliates or subsidiaries (other than ProFrac and its subsidiaries), including any director or officer of ProFrac who is also an agent, shareholder, member, partner, director, officer, employee, affiliate or subsidiary of any ProFrac Controlling Stockholder (each, a "Business Opportunities Exempt Party"), even if the business opportunity is

# **USWS Stockholder Rights**

USWS' certificate of incorporation provides that USWS may amend its certificate of incorporation as provided therein and by the DGCL.

USWS' certificate of incorporation provides the doctrine of corporate opportunity shall apply with respect to any of the directors or officers of USWS only with respect to a corporate opportunity that was offered to such person solely in his or her capacity as a director or officer of USWS and such opportunity is one that USWS is legally and contractually permitted to undertake and would otherwise be reasonable for USWS to pursue.

Limitation on Liability of Directors

Anti-takeover Provisions

#### ProFrac Stockholder Rights

one that ProFrac or its subsidiaries might reasonably be deemed to have pursued or had the ability or desire to pursue if granted the opportunity to do so, such that no Business Opportunity Exempt Party shall be liable to ProFrac or its stockholders for breach of any fiduciary duty, unless, in the case of a person who is a director or officer of ProFrac, such business opportunity is expressly offered to such director or officer in writing solely in his or her capacity as a director or officer of ProFrac.

ProFrac's certificate of incorporation provides that, to the fullest extent permitted by the DGCL, directors of ProFrac will not be liable to ProFrac or its stockholders for monetary damages for breach of fiduciary duty as a director. ProFrac's certificate of incorporation also provides that ProFrac will indemnify its directors and officers to the fullest extent permitted by applicable law. ProFrac's bylaws permit ProFrac to purchase insurance on behalf of any officer, director, employee or other agent for any liability arising out of that person's actions as ProFrac's officer, director, employee or agent, regardless of whether the DGCL would permit indemnification.

ProFrac has no shareholder rights plan in place. ProFrac maintains a dual-class common stock structure, with the ProFrac Class B Common Stock having equal voting rights to the ProFrac Class A Common Stock as described above. The ProFrac Class B Common Stock is not publicly traded and the ultimate beneficial ownership thereof is

# **USWS Stockholder Rights**

USWS' certificate of incorporation provides that, to the fullest extent permitted by the DGCL, directors of USWS will not be liable to USWS or its stockholders for monetary damages for breach of fiduciary duty as a director. USWS certificate of incorporation also provides that USWS will indemnify its directors and officers to the fullest extent permitted by applicable law. USWS' bylaws permit USWS to purchase insurance on behalf of any officer, director, employee or other agent for any liability arising out of that person's actions as USWS' officer, director, employee or agent, regardless of whether the DGCL would permit indemnification.

USWS has no shareholder rights plan in place. Certain provisions of USWS' certificate of incorporation and bylaws contain certain protective provisions, including provisions that (a) limit the ability of stockholders to call a special meeting, remove directors, or amend USWS' certificate of incorporation and bylaws, (b) divide the USWS Board into

Exclusive Forum

#### ProFrac Stockholder Rights

concentrated among the ProFrac Controlling Stockholders. In addition, ProFrac's certificate of incorporation and bylaws contain certain protective provisions, including provisions that (a) limit the ability of stockholders to call a special meeting, (b) establish advance notice requirements for stockholder proposals and director nominations, (c) provide the ProFrac Board the ability to authorize undesignated preferred stock, (d) provide that the authorized number of directors may be changed only by resolution of the ProFrac Board, and (e) may otherwise delay or discourage transactions involving an actual or potential change in control or change in ProFrac's management, as more fully described under the heading "Description of ProFrac Capital Stock" in this proxy statement/information statement/prospectus.

Under ProFrac's certificate of incorporation, unless ProFrac consents in writing to the selection of an alternative forum, to the fullest extent permitted by the applicable law, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for any ProFrac Stockholder to bring (i) any derivative action, (ii) any action asserting a claim of breach of fiduciary duty, (iii) any action asserting a claim against ProFrac, the ProFrac Board, officers or any employees arising pursuant to the certificate of incorporation, bylaws or the DGCL, or (iv) any action asserting a claim against ProFrac, the ProFrac Board, officers or employees governed by the internal affairs doctrine.

### USWS Stockholder Rights

three classes with staggered three-year terms, and (c) provide for the existence of unauthorized and unissued shares of USWS capital stock, which could be issued by the USWS Board to persons friendly to current management, which issuance could render more difficult or discourage an attempt to obtain control of USWS.

Under USWS' certificate of incorporation, unless USWS consents in writing to the selection of an alternative forum, to the fullest extent permitted by the applicable law, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for any USWS Stockholder to bring (i) any derivative action, (ii) any action asserting a claim of breach of fiduciary duty, (iii) any action asserting a claim against USWS, the USWS Board, officers or any employees arising pursuant to the certificate of incorporation, bylaws or the DGCL, or (iv) any action asserting a claim against USWS, the USWS Board, officers or employees governed by the internal affairs doctrine.

### LEGAL MATTERS

Before this registration statement becomes effective, Brown Rudnick LLP will provide an opinion regarding the validity of the securities offered by this proxy statement/information statement/prospectus and certain other legal matters related to this proxy statement/information statement/prospectus.

#### **EXPERTS**

### ProFrac

The audited financial statements of ProFrac Holdings, LLC (referred to as "ProFrac Predecessor") included in this proxy statement/information statement/prospectus and elsewhere in the registration statement have been so included in reliance upon the report of Grant Thornton LLP, independent registered public accountants, upon the authority of said firm as experts in accounting and auditing.

The information appearing in this proxy statement/information statement/prospectus concerning estimates of ProFrac's proven mineral reserves was derived from the report of John T. Boyd Company, independent mining engineers and geologists, and has been included herein on the authority of John T. Boyd Company as experts with respect to the matters covered by such report and in giving such report.

The audited balance sheet of ProFrac Holding Corp. included in this proxy statement/information statement/prospectus and elsewhere in the registration statement have been so included in reliance upon the report of Grant Thornton LLP, independent registered public accountants, upon the authority of said firm as experts in accounting and auditing.

The audited financial statements of FTS International, Inc. included in this proxy statement/information statement/prospectus and elsewhere in the registration statement have been so included in reliance upon the report of Grant Thornton LLP, independent registered public accountants, upon the authority of said firm as experts in accounting and auditing.

## USWS

The consolidated financial statements of U.S. Well Services, Inc. as of December 31, 2021 and 2020, and for the years then ended, have been included in reliance upon the report of KPMG LLP, an independent registered public accounting firm, and upon the authority of said firm as experts in accounting and auditing.

# TRANSFER AGENT AND REGISTRAR

The transfer agent and registrar for ProFrac's Class A Common Stock is American Stock Transfer & Trust Company, LLC.

### FUTURE STOCKHOLDER PROPOSALS

#### **ProFrac**

ProFrac currently intends to hold its 2023 annual meeting of stockholders on or about June 1, 2023, regardless of whether the Merger is consummated. A proposal that a stockholder intends to present at the ProFrac 2023 annual meeting of stockholders and wishes to be considered for inclusion in ProFrac's proxy materials must be received no later than February 1, 2023. All proposals must comply with Rule 14a-8 under the Exchange Act, as well as all other requirements under the ProFrac bylaws relating to stockholder proposals.

The bylaws of ProFrac contain provisions intended to promote the efficient functioning of stockholder meetings. Some of the provisions require advance notice to ProFrac of stockholder proposals or director nominations to be considered at an annual meeting. If a stockholder wishes to submit a proposal at ProFrac's 2023 annual meeting of stockholders, other than for inclusion in ProFrac's 2023 proxy statement and form of proxy, or to nominate a person for election as director, ProFrac's bylaws require the stockholder to deliver written notice thereof, setting forth the information specified in ProFrac's bylaws, to the Secretary of ProFrac not earlier than the close of business on February 1, 2023 and not later March 3, 2023; provided, however, that in the event that the date of the annual meeting is scheduled for a date that is more than 30 days before or more than 60 days after June 1, 2023, notice by the stockholder to be timely must be received no later than the 10th day following the day on which the public announcement (as defined in ProFrac's bylaws) of the date of such meeting is first made by ProFrac.

### **USWS**

USWS held its 2022 annual meeting of stockholders on May 20, 2022. If the Merger Agreement is adopted by the requisite vote of the USWS Stockholders and the Merger is completed, USWS will survive the Merger as an indirect subsidiary of ProFrac and, consequently, will not hold an annual meeting of its stockholders in 2023. USWS stockholders will be entitled to participate, as stockholders of ProFrac, in the 2023 annual meeting of stockholders of ProFrac.

If the Merger Agreement is not adopted by the requisite vote of the USWS Stockholders or if the Merger is not completed for any reason, USWS will hold an annual meeting of its stockholders in 2023.

In the event that USWS holds a 2023 annual meeting of its stockholders, stockholders interested in presenting a proposal for inclusion in USWS' proxy statement and proxy relating to USWS' 2023 annual meeting of stockholders may do so by following the procedures prescribed in Rule 14a-8 under the Exchange Act and USWS' Bylaws. To be eligible under the SEC's stockholder proposal rule (Rule 14a-8 of the Exchange Act) for inclusion in the 2023 annual meeting proxy statement and form of proxy for USWS, if applicable, a proposal must be received by the Corporate Secretary of USWS on or before December 9, 2022. If a stockholder wishes to submit a proposal at USWS' 2023 annual meeting of Stockholders, if applicable, other than for inclusion in the 2023 proxy statement and form of proxy, or to nominate a person for election as director, USWS' Bylaws, require the stockholder to deliver written notice thereof, setting forth the information specified in its Bylaws, to the Corporate Secretary at its principal executive offices no later than the close of business on the 90th day, nor earlier than the opening of business on the 120th day, prior to the first anniversary of the preceding year's annual meeting. In other words, for a stockholder proposal to be considered at the 2023 annual meeting of stockholders, if held, it should be properly submitted to the Corporate Secretary no earlier than January 20, 2023 and no later than February 19, 2023; provided, however, that such other business must otherwise be a proper matter for stockholder action. In the event that the date of the annual meeting is more than 30 days before or after May 20, 2023, however, a stockholder must deliver notice no earlier than the opening of business on the 120th day prior to such annual meeting and not later than the close of business on the later of (a) the 90th day prior to such annual meeting or (b) the 10th day following the day on which USWS first makes public announcement of the date of such meeting.

### HOUSEHOLDING OF PROXY MATERIALS

SEC rules permit companies and intermediaries such as brokers to satisfy delivery requirements for proxy statements and notices with respect to two or more stockholders sharing the same address by delivering a single proxy statement or a single notice addressed to those stockholders. This process, commonly called "householding," provides cost savings for companies. Some brokers household proxy materials, delivering a single proxy statement or notice to multiple stockholders sharing an address unless contrary instructions have been received from the affected stockholders. Once you have received notice from your broker that they will be householding materials to your address, householding will continue until you are notified otherwise or until you revoke your consent. If, at any time, you no longer wish to participate in householding and would prefer to receive a separate proxy statement or notice, or if your household is receiving multiple copies of these documents and you wish to request that future deliveries be limited to a single copy, please notify your broker.

Requests for additional copies of this proxy statement/information statement/prospectus should be directed to, as applicable, ProFrac Holding Corp., 333 Shops Boulevard, Suite 301, Willow Park, TX 76087, Telephone: (254) 776-3722, or U.S. Well Services, Inc., 1360 Post Oak Boulevard, Suite 1800, Houston, TX 77056, Telephone: (832) 562-3730. If two or more stockholders sharing the same address are receiving multiple copies of this proxy statement/information statement/prospectus, such stockholders can request delivery of a single copy of the proxy statement/information statement/prospectus from ProFrac Holding Corp., 333 Shops Boulevard, Suite 301, Willow Park, TX 76087, Telephone: (254) 776-3722, or U.S. Well Services, Inc., 1360 Post Oak Boulevard, Suite 1800, Houston, TX 77056, Telephone: (832) 562-3730.

### WHERE YOU CAN FIND MORE INFORMATION

ProFrac and USWS file annual, quarterly and periodic reports, proxy statements and other information with the SEC. You may read and copy any of this information filed at the SEC's public reference rooms located at:

Public Reference Room 100 F Street N.E. Room 1024 Washington, DC 20549

You may obtain information on the operation of the Public Reference Room by calling the SEC atl-800-SEC-0330. These SEC filings are also available to the public from commercial document retrieval services and at the website maintained by the SEC at http://www.sec.gov. The information contained on the SEC's website is expressly not incorporated by reference into this proxy statement/information statement/prospectus.

ProFrac has filed with the SEC a registration statement on FormS-4, of which this proxy statement/information statement/prospectus forms a part. The registration statement registers the issuance of shares of ProFrac Class A Common Stock in the Merger in connection with the Merger. The registration statement, including the attached exhibits, contains additional relevant information about ProFrac and ProFrac Class A Common Stock. The rules and regulations of the SEC allow ProFrac and USWS to omit certain information included in the registration statement from this proxy statement/information statement/prospectus.

At the present time, ProFrac is not eligible to incorporate by reference the information filed with the SEC into this proxy statement/information statement/prospectus and accordingly has included the required business and financial information in this proxy statement/information statement/prospectus.

The SEC allows USWS to incorporate by reference information into this proxy statement/information statement/prospectus, which means that USWS can disclose important information to you by referring you to other documents filed separately with the SEC. The information incorporated by reference is considered part of this proxy statement/information statement/prospectus, except for any information superseded by information contained directly in this proxy statement/information statement/prospectus or in later filed documents incorporated by reference into this proxy statement/information statement/prospectus.

This proxy statement/information statement/prospectus incorporates by reference the documents listed below that USWS has previously filed with the SEC. These documents contain important business and financial information about USWS.

- USWS' Annual Report on Form 10-K for the fiscal year ended December 31, 2021, filed with the SEC on March 30, 2022, including the
  portions of USWS' <u>Definitive Proxy Statement</u> on Schedule 14A filed with the SEC on April 20, 2022, incorporated by reference therein
  (File No. 001-38025);
- USWS' Quarterly Reports on Form 10-Q for the quarters ended March 31, 2022 and June 30, 2022, filed with the SEC on May 16, 2022 and August 11, 2022, respectively (File No. 001-38025);
- USWS' Current Reports on Form 8-K, filed with the SEC on February 2, 2022, March 4, 2022, March 11, 2022, March 30, 2022, April 27, 2022, May 4, 2022, May 16, 2022, May 23, 2022, June 9, 2022, June 22, 2022, June 27, 2022, July 22, 2022, August 5, 2022 and August 10, 2022 (excluding any information furnished pursuant to Item 2.02 or Item 7.01, or any corresponding information furnished under Item 9.01, of any such Current Report on Form 8-K) (File No. 001-38025); and
- Description of USWS' Common Stock and outstanding warrants contained in USWS' Registration Statement on Form 8-A filed with the SEC on March 7, 2017 (File No. 001-38025) and any amendment or report filed for purposes of updating that document.

Documents incorporated by reference are available without charge from USWS, excluding any exhibits to those documents unless the exhibit is specifically incorporated by reference as an exhibit in this proxy statement/information statement/prospectus. You can obtain documents incorporated by reference into this proxy statement/information statement/prospectus by requesting them in writing or by telephone from USWS at the following address:

U.S. Well Services, Inc. 1360 Post Oak Boulevard Suite 1800 Houston, TX 77056 Attention: Corporate Secretary Telephone: 832-562-3730

If you wish to request documents, the applicable company must receive your request by [\_\_\_\_], 2022, in order to receive them before the USWS Special Meeting.

USWS has supplied all information contained or incorporated by reference into this proxy statement/information statement/prospectus relating to USWS, and ProFrac has supplied all information contained in this proxy statement/information statement/prospectus relating to ProFrac.

Neither ProFrac nor USWS has authorized anyone to give any information or make any representation about the Merger or the two companies that is different from, or in addition to, that contained in this proxy statement/information statement/prospectus or in any of the materials that have been incorporated by reference into this proxy statement/information statement/prospectus. Therefore, if anyone gives you information of this sort, you should not rely on it. If you are in a jurisdiction where offers to exchange or sell, or solicitations of offers to exchange or purchase, the securities offered by this proxy statement/information statement/prospectus or the solicitation of proxies is unlawful, or if you are a person to whom it is unlawful to direct these types of activities, then the offer presented in this proxy statement/information statement/prospectus speaks only as of the date of this proxy statement/information statement/prospectus unless the information specifically indicates that another date applies.

# Index to financial statements ProFrac Holdings, LLC

Audited consolidated financial statements	
Report of independent registered public accounting firm	F-1
Consolidated balance sheets as of December 31, 2021 and 2020	F-2
Consolidated statements of operations for the years ended December 31, 2021 and 2020	F-3
Consolidated statement of changes in equity for the years ended December 31, 2021 and 2020	F-4
Consolidated statements of cash flows for the years ended December 31, 2021 and 2020	F-5
Notes to consolidated financial statements	F-6
ProFrac Holding Corp.	
Audited balance sheet	
Report of independent registered public accounting firm	F-28
Balance sheet as of December 31, 2021	F-29
Notes to balance sheet	F-30
Unaudited interim financial statements	
<u>Unaudited consolidated balance sheets as of June 30, 2022 and December 31, 2021</u>	F-31
Unaudited consolidated statements of operations for the three and six months ended June 30, 2022 and 2021	F-33
Unaudited consolidated statements of comprehensive income (loss) for the three and six months ended June 30, 2022 and 2021	F-34
Unaudited consolidated statements of changes in equity for the three and six months ended June 30, 2022 and 2021	F-35
Unaudited consolidated statements of cash flows for the six months ended June 30, 2022 and 2021	F-37
Notes to unaudited interim consolidated financial statements	F-38
Unaudited pro forma financial statements	
Introduction.	F-72
Introduction. Unaudited pro forma balance sheets as of June 30, 2022	F-74
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021	F-74 F-76
Introduction. Unaudited pro forma balance sheets as of June 30, 2022	F-74
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021	F-74 F-76
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc. Audited consolidated financial statements	F-74 F-76 F-78
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm	F-74 F-76
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through	F-74 F-76 F-78
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended	F-74 F-76 F-78 F-90
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)	F-74 F-76 F-78 F-90
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor) Consolidated balance sheets as of December 31, 2021 and 2020	F-74 F-76 F-78 F-90
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm  Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated balance sheets as of December 31, 2021 and 2020 Consolidated statements of cash flows for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through	F-74 F-76 F-78 F-90
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm  Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated balance sheets as of December 31, 2021 and 2020  Consolidated statements of cash flows for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended	F-74 F-76 F-78 F-90 F-93 F-94
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm  Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated balance sheets as of December 31, 2021 and 2020  Consolidated statements of cash flows for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)	F-74 F-76 F-78 F-90
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm  Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated balance sheets as of December 31, 2021 and 2020  Consolidated statements of cash flows for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated statements of stockholders' equity (deficit) for the year ended December 31, 2021 (Successor), the period from November 20, Consolidated statements of stockholders' equity (deficit) for the year ended December 31, 2021 (Successor), the period from November 20,	F-74 F-76 F-78 F-90 F-93 F-94
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm  Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated balance sheets as of December 31, 2021 and 2020  Consolidated statements of cash flows for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated statements of stockholders' equity (deficit) for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year	F-74 F-76 F-78 F-90 F-93 F-94
Introduction. Unaudited pro forma balance sheets as of June 30, 2022 Unaudited pro forma statements of operations for the six months ended June 30, 2022 and the year ended December 31, 2021 Notes to unaudited pro forma financial statements  FTS International, Inc.  Audited consolidated financial statements  Report of independent registered public accounting firm  Consolidated statements of operations for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated balance sheets as of December 31, 2021 and 2020  Consolidated statements of cash flows for the year ended December 31, 2021 (Successor), the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and for the year ended December 31, 2019 (Predecessor)  Consolidated statements of stockholders' equity (deficit) for the year ended December 31, 2021 (Successor), the period from November 20, Consolidated statements of stockholders' equity (deficit) for the year ended December 31, 2021 (Successor), the period from November 20,	F-74 F-76 F-78 F-90 F-93 F-94

# U.S. Well Services, Inc.

Audited consolidated financial statements	
Report of independent registered public accounting firm	F-128
Consolidated balance sheets as of December 31, 2021 and 2020	F-129
Consolidated statements of operations for the years ended December 31, 2021 and 2020	F-130
Consolidated statements of cash flows for the years ended December 31, 2021 and 2020	F-131
Consolidated statement of stockholders' equity (deficit) for the years ended December 31, 2021 and 2020	F-133
Notes to consolidated financial statements	F-134
Unaudited interim financial statements	
Unaudited condensed consolidated balance sheets as of June 30, 2022 and December 31, 2021	F-174
Unaudited condensed consolidated statements of operations for the three and six months ended June 30, 2022 and 2021	F-175
Unaudited condensed consolidated statements of cash flows for the six months ended June 30, 2022 and 2021	F-176
Unaudited consolidated statement of changes in equity for the three and six months ended June 30, 2022 and 2021	F-178
Notes to unaudited condensed consolidated financial statements	F-180

# REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

Board of Directors and Owners ProFrac Holdings, LLC

### Opinion on the financial statements

We have audited the accompanying consolidated balance sheets of ProFrac Holdings, LLC (referred to as "ProFrac Predecessor" in the notes to the financial statements) and subsidiaries (the "Company") as of December 31, 2021 and 2020, the related consolidated statements of operations, equity, and cash flows for each of the two years in the period ended December 31, 2021, and the related notes (collectively referred to as the "financial statements"). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2021 and 2020, and the results of its operations and its cash flows for each of the two years in the period ended December 31, 2021, in conformity with accounting principles generally accepted in the United States of America.

# Basis for opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB and in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audits we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

### /s/ GRANT THORNTON LLP

We have served as the Company's auditor since 2018.

Dallas, Texas March 31, 2022

# ProFrac Holdings, LLC Consolidated balance sheets Amounts in thousands

	December 31, 2021	December 31, 2020	
Assets			
Current assets:			
Cash and equivalents	\$ 5,376	\$ 2,952	
Accounts receivable	161,632	72,744	
Accounts receivable—related party	4,515	880	
Prepaid expenses, and other current assets	6,213	6,516	
Inventories	73,942	56,598	
Total current assets	251,678	139,690	
Property, plant, and equipment	827,865	782,249	
Accumulated depreciation and depletion	(464,178)	(352,565)	
Property, plant, and equipment, net	363,687	429,684	
Investment in associate	<u></u>	1,260	
Investments	4,244	_	
Intangible assets	27,816	_	
Other assets	17,145	6,643	
Total assets	\$ 664,570	\$ 577,277	
Liabilities and equity			
Current liabilities:			
Accounts payable	\$ 121,070	\$ 76,374	
Accounts payable—related party	21,275	18,488	
Accrued expenses	38,149	29,890	
Other current liabilities	34,400	_	
Current portion of long-term debt	31,793	15,484	
Total current liabilities	246,687	140,236	
Long-term debt	235,128	205,369	
Long-term debt—related party	34,645	54,860	
Total liabilities	516,460	400,465	
Commitments and contingencies (Note 9)			
Equity	147,015	176,812	
Noncontrolling interests	1,039	_	
Accumulated other comprehensive loss	56		
Total Equity	148,110	176,812	
Total liabilities and equity	<u>\$ 664,570</u>	\$ 577,277	

# ProFrac Holdings, LLC Consolidated statements of operations Amounts in thousands

	Year ended l	Year ended December 31,		
	2021	2020		
Revenues	<u>\$ 768,353</u>	\$ 547,679		
Operating costs and expenses:				
Cost of revenues, exclusive of depreciation, depletion and amortization	570,122	432,570		
Depreciation, depletion and amortization	140,687	150,662		
Loss on disposal of assets, net	9,777	8,447		
Selling, general, and administrative	65,592	51,014		
Total operating costs and expenses	_ 786,178	642,693		
Operating loss	(17,825)	(95,014)		
Other (expense) income:				
Interest expense, net	(25,788)	(23,276)		
Other (expense) income	(111)	324		
Loss before income tax provision	(43,724)	(117,966)		
Income tax benefit (provision)	186	(582)		
Net loss	\$ (43,538)	\$ (118,548)		
Net loss attributable to noncontrolling interests	(1,118)	(1,143)		
Net loss attributable to ProFrac Predecessor	<u>\$ (42,420)</u>	<u>\$ (117,405</u> )		
Other comprehensive income (loss)	(56)	_		
Comprehensive loss	\$ (42,476)	\$ (117,405)		
Less: Other comprehensive income (loss) attributable to noncontrolling interests	(14)	_		
Comprehensive loss attributable to ProFrac Predecessor	<u>\$ (42,462)</u>	<u>\$ (117,405)</u>		

ProFrac Holdings, LLC Consolidated statement of changes in equity Amounts in thousands

		 mulated Other		
	Equity	rehensive Loss	ontrolling iterests	Total
Balance, January 1, 2020	\$ 281,883	\$ _	\$ 2,928	\$ 284,811
Net loss	(117,405)	_	(1,143)	(118,548)
Member contribution by debt retirement	10,549	_	_	10,549
Balance, December 31, 2020	\$ 175,027	\$ _	\$ 1,785	\$ 176,812
Net loss	(42,420)	_	(1,118)	(43,538)
Member contribution by debt retirement	18,000	_	_	18,000
Purchase of noncontrolling interests	(3,592)	_	(877)	(4,469)
Currency translation adjustments	_	56	19	75
Noncontrolling interest of acquired business	<u> </u>		1,230	1,230
Balance, December 31, 2021	\$ 147,015	\$ 56	\$ 1,039	\$ 148,110

ProFrac Holdings, LLC Consolidated statements of cash flow Amounts in thousands

	Year ended I	December 31,
	2021	2020
Cash flows from operating activities:		
Net loss	\$ (43,538)	\$ (118,548)
Adjustments to reconcile net loss to cash provided by operating activities:		
Depreciation, depletion and amortization	140,687	150,662
Loss on disposal of assets	9,777	8,447
Loss on extinguishment of debt	515	
Amortization of debt issuance costs	2,154	1,958
Bad debt expense, net of recoveries	(1,164)	2,778
Provision for inventory obsolescence	_	2,804
Changes in operating assets and liabilities:	(00.620)	10.050
Accounts receivable	(89,639)	12,378
Inventories	(16,118)	(3,489)
Prepaid expenses and other assets	3,750	(1,027)
Accounts payable	31,639	(5,129)
Accrued expenses	6,162	(5,780)
Deferred revenues and other current liabilities	(283)	
Net cash provided by operating activities	43,942	45,054
Cash flows from investing activities:		
Investment in property, plant & equipment	(87,400)	(48,037)
Cash proceeds from sale of assets	17,553	4,680
Acquisitions	(4,292)	(1,260)
Investment in preferred shares	(4,244)	
Net cash used in investing activities	(78,383)	(44,617)
Cash flows from financing activities:	_ <del></del>	
Proceeds from issuance of long-term debt	223,730	187,622
Repayments of long-term debt	(182,402)	(201,316)
Payment of debt issuance costs	(1,979)	(1,043)
Purchase of noncontrolling interests	(577)	
IPO Preparation costs	(1,982)	_
Other	75	(585)
Net cash provided by (used in) financing activities	36,865	(15,322)
Net increase (decrease) in cash and cash equivalents	2,424	(14,885)
Cash and cash equivalents, beginning of period	2,424	17,837
Cash and cash equivalents, ending of period	<u>\$ 5,376</u>	<u>\$ 2,952</u>
Supplemental cash flow information:		
Cash payments during the year for interest	\$ 23,544	\$ 20,952
Cash payments during the year for taxes	(212)	391
Non-cash transactions:		
Acquisition of land in exchange for other current liability	\$ 30,000	\$ —
Acquisition of non-controlling interests in exchange for other current liability	3,893	
Retirement of long-term debt by member	<u> 18,000</u>	10,549

ProFrac Holdings, LLC Notes to the consolidated financial statements For the year ended December 31, 2021 and 2020 (Amounts in thousands, except as noted)

### 1. Organization and description of business

The consolidated financial statements presented herein are those of ProFrac Holdings, LLC and its subsidiaries, which include Best Pump & Flow LP ("Best Flow") and Alpine Silica, LLC ("Alpine"). ProFrac Holdings, LLC on a consolidated basis, is the predecessor to ProFrac Holding Corp. and is referred to herein as "ProFrac Predecessor," "we," "us," "our," or the "Company." Prior to December 21, 2021, Dan Wilks and Farris Wilks (or entities they control) (collectively, the "Wilks") held a controlling interest in ProFrac LLC, Best Flow and Alpine. Historical periods for ProFrac Predecessor had been presented on a consolidated and combined basis given the common control ownership by the Wilks. On December 21, 2021, all of the then-outstanding membership interests in Best Flow and Alpine were contributed to ProFrac LLC in exchange for membership interests in ProFrac LLC. Accordingly, all prior periods have been retrospectively adjusted to present the operations of ProFrac LLC, Best Flow and Alpine on a consolidated basis. The acquisitions of Best Flow and Alpine have been accounted for in a manner consistent with the pooling of interest method of accounting, as the transaction was a combination of entities under common control. Under this method of accounting, the statements of operations, equity and cash flows have been adjusted to include all activities of the commonly controlled groups for all periods in which common control existed.

ProFrac Predecessor is a vertically integrated and innovation-driven energy services company providing hydraulic fracturing, completion services and other complementary products and services to leading upstream oil and gas companies engaged in the exploration and production ("E&P") of North American unconventional oil and natural gas resources. The operations of the ProFrac Predecessor are primarily focused in the Appalachian, East Texas/Louisiana and Permian Basins, where it has longstanding customer relationships. The ProFrac Predecessor operates in three business segments: stimulation services, manufacturing and proppant production.

### 2. Summary of significant accounting policies

### Basis of presentation

The consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP") and the consolidated financial statements include the accounts of the Company and all majority-owned subsidiaries. All significant intercompany accounts and transactions have been eliminated in consolidation.

### Use of estimates

The preparation of financial statements in accordance with U.S. GAAP requires management to make estimates and assumptions that affect (1) the reported amounts of assets and liabilities and (2) the reported amounts of revenue and expenses during the reporting period. We base these estimates on historical results and various other assumptions believed to be reasonable, all of which form the basis for making estimates concerning the carrying values of assets and liabilities that are not readily available from other sources. Ultimate results could differ materially from these estimates.

# Concentration of credit risk

The Company's assets that are potentially exposed to concentrations of credit risk consist primarily of cash, cash equivalents, and accounts receivable. The financial institutions in which the Company transacts business are "well capitalized" under applicable federal capital adequacy guidelines, thereby reducing, but not eliminating its exposure to credit risks for deposits in excess of federally insured amounts.

The market for the Company's services and products is the oil and gas industry in the United States. Oil and gas companies make capital expenditures on exploration, development and production operations. The level of these expenditures has historically been characterized by significant volatility.

The majority of the Company's business is conducted with oil and gas E&P companies. For the year ended December 31, 2021, sales to Rockcliff Energy Management, LLC and Sabine Oil & Gas Corporation accounted for 15% and 10%, respectively, of total revenue. For the year ended December 31, 2020, sales to Rockcliff Energy Management, LLC, Alta Resources Development LLC and Sabine Oil & Gas Corporation accounted for 15%, 11% and 10%, respectively, of total revenue. The majority of these revenues were related to our Stimulation services segment.

As of December 31, 2021, receivables from Comstock Resources, Inc., Sabine Oil & Gas Corporation and Rockcliff Energy Management, LLC accounted for 13%, 11% and 10%, respectively, of total accounts receivable. As of December 31, 2020, receivables from Rockcliff Energy Management, LLC, Comstock Resources, Inc. and Blue Dome Operating, LLC accounted for 19%, 11% and 11%, respectively, of total accounts receivable. The Company does not believe this concentration of credit risk represents material risk of loss with respect to its financial position due to the credit worthiness and the nature of the receivables.

### Cash and equivalents

The Company considers all highly liquid investments with an original maturity of three months or less to be cash equivalents.

#### Inventories

Inventories, which consist of raw materials, work in process and finished goods, are carried at the lower of cost or net realizable value. Cost represents the purchase price or production cost and includes material, labor, and manufacturing overhead. Fixed manufacturing overhead is allocated to the cost of inventory based on the normal capacity of production facilities. Unallocated overhead during periods of abnormally low production levels is recognized as cost of sales in the period in which it is incurred. Our inventory is recorded using the first-in, first-out method. We regularly review quantities on-hand and reduce the value of inventory for excess or obsolescence based primarily upon historical usage, estimated demand, and technological developments. See Note 3 for additional information related to our inventory.

# Property, plant, and equipment

Property, plant, and equipment is stated at cost less accumulated depreciation, which is generally provided by using the straight-line method over the estimated useful lives of the individual assets. We manufacture our hydraulic fracturing units and the cost of this equipment, which includes direct and indirect manufacturing costs, is capitalized, and carried as construction in progress until it is completed and placed into service. Expenditures for renewals and betterments that extend the lives of our service equipment, which includes the replacement of significant components of service equipment, are capitalized and depreciated. Other repairs and maintenance costs are expensed as incurred. Mining property costs are amortized using the units-of-production method on estimated measured tons in proved reserves. The impact of revisions to reserve estimates is recognized on a prospective basis. When assets are disposed or retired, the cost and accumulated depreciation are netted against any sale proceeds, and the resulting gains or losses are included in the results of operations.

## Intangible assets

We amortize intangible assets with a finite life on a straight-line basis over the period which the asset is expected to contribute to cash flows, ranging from seven years to 17 years. The components of our intangible assets generally consist of license agreements and technology. See Note 5—Intangible assets for additional information.

### Impairment of long-lived assets

Recoverability of assets to be held and used is assessed whenever events or circumstances suggest that the fair value of the underlying asset or asset group may have significantly declined. Recoverability is evaluated by comparing the carrying amount of an asset to the estimated undiscounted future cash flows expected to be generated by the asset. If the carrying amount of an asset exceeds its estimated undiscounted future cash flows, an impairment charge is recognized for the amount by which the carrying amount of the asset exceeds the fair value of the asset. During the years ended December 31, 2021 and 2020, no impairments of long-lived assets have been recorded.

## Accounts receivable and allowance for doubtful accounts

Accounts receivable are recorded at invoiced amounts and generally do not bear interest. We routinely assess the financial strength of our customers and establish an allowance for doubtful accounts based upon various factors including historical collection experience, current aging status of customer accounts, current financial condition, as well as other factors. Losses are charged against the allowance when the customer accounts are determined to be uncollectible. The following table summarizes the rollforward of the allowance for doubtful accounts for the years ended December 31, 2021 and 2020:

	Year ended D	ecember 31,
	2021	2020
Beginning balance	\$ (1,846)	\$ (316)
Bad debt expense, net of recoveries	1,164	(2,778)
Write-offs	16	1,248
Ending balance	\$ (666)	\$ (1,846)

# Revenue recognition

Effective January 1, 2019, the Company adopted the requirements of Accounting Standards Update ("ASU") No. 2014-09, Revenue from Contracts with Customers ("ASC 606"), which defines the criteria required to recognize revenue when a performance obligation is satisfied by transferring control over a product or service to a customer.

The Company's services are sold based upon contracts with customers. The Company recognizes revenue as it satisfies performance obligations by transferring control over a service or product to a customer. Payment terms are specified in each customer agreement and are typically a specific number of days following satisfaction of the performance obligation. The following are descriptions of the principal activities of each reportable segment from which the Company generates its revenue.

Stimulation services. We generate revenue through the provision of hydraulic fracturing services, which involves the injection of water, sand and chemicals under high pressure into formations to optimize hydrocarbon flow paths during the completion phase of wellbores. Our contracts with customers are short term in nature, typically less than four weeks, and have a single performance obligation, which is the contracted total stages, satisfied over time. Once a stage has been completed, a field ticket is created which includes charges for services performed and any inputs consumed during the service. The signing of the field ticket by a customer representative represents their acceptance of the service and agreement to the amounts to which the Company has the right to invoice and recognize as revenue. We believe that recognizing revenue based on actual stages completed, upon receipt of a signed field ticket, appropriately depicts how our hydraulic fracturing services are transferred to our customers over time.

Manufacturing. We generate revenue through sales of equipment used to perform oilfield services. The performance obligation is satisfied and revenues are recognized at the point-in-time that control of goods are

transferred to the customer, generally upon shipment from our manufacturing facility. Payment terms are specified in each customer agreement and are typically a specific number of days following satisfaction of the performance obligation.

Proppant production. We generate revenue through the sale of frac sand to oilfield service providers and E&P companies. The performance obligation is satisfied and revenue is recognized at the point-in-time that control of the product is transferred to the customer, generally upon shipment from our facility. We charge our customers on a per-ton basis at current market prices. Payment terms are specified in each customer agreement and are typically a specific number of days following satisfaction of the performance obligation.

Taxes collected from customers and remitted to governmental authorities are accounted for on a net basis and are therefore excluded from revenues in the consolidated statements of operations.

### **Business Combinations**

Business combinations are accounted for under the acquisition method of accounting. Under this method, the assets acquired and liabilities assumed are recognized at their respective fair values as of the date of acquisition. The excess, if any, of the acquisition price over the fair values of the assets acquired and liabilities assumed is recorded as goodwill. For significant acquisitions, we utilize third-party appraisal firms to assist us in determining the fair values for certain assets acquired and liabilities assumed. The measurement of these fair values requires us to make significant estimates and assumptions which are inherently uncertain.

Adjustments to the fair values of assets acquired and liabilities assumed are made until we obtain all relevant information regarding the facts and circumstances that existed as of the acquisition date (the "measurement period"), not to exceed one year from the date of the acquisition. We recognize measurement-period adjustments in the period in which we determine the amounts, including the effect on earnings of any amounts we would have recorded in previous periods if the accounting had been completed at the acquisition date.

See Note 8—Investments and Acquisitions for information on acquisitions completed during the historical period.

### Fair value measurements

Fair value is defined as the price that would be received from selling an asset or paid to transfer a liability in an orderly transaction between market participants at a measurement date. We apply the following fair value hierarchy, which prioritizes the inputs used to measure fair value into three levels and bases of categorization within the hierarchy upon the lowest level input that is available and significant to the fair value measurement:

- Level One: The use of quoted prices in active markets for identical assets or liabilities.
- Level Two: Other than quoted prices included in Level One, inputs that are observable for the asset or liability, either directly or indirectly, for substantially the full term of the asset or liability.
- Level Three: The use of significant unobservable inputs that typically require the use of management's estimates of assumptions that market
  participants would use in pricing.

Our current assets and liabilities contain financial instruments, the most significant of which are trade accounts receivable and payable. We believe the carrying value of our current assets and liabilities approximate fair value. Our fair value assessment incorporates a variety of considerations, including: (i) the short-term duration of the instruments and (ii) our historical incurrence of and expectations of future bad debt expense. The book value of our floating rate debt approximates fair value because of its floating rate structure.

## Retirement benefits

The Company has a defined contribution 401(k) plan (the "Plan") covering substantially all of its employees. Employees are eligible to participate in the Plan if they are 21 years of age and have completed 90 days of

service. Eligible employees may defer up to 100% of their compensation or the maximum allowed by law on a pre-tax or Roth after-tax basis. The Company contributes a match equal to 100% of an employee's contribution up to 4% of his or her compensation. An additional discretionary employer contribution may be made. Employee contributions (both pre-tax and Roth) and the employer match are 100% vested immediately for all years prior to 2020. Effective January 1, 2020, the employer matching contribution vests ratably upon completion of each of the employees' first three years of service, additional employer matching contributions thereafter vest immediately. Between April 5, 2020 and December 27, 2020, the Company suspended matching contributions as a cost-savings measure. The Company made matching contributions of \$1.5 million and \$0.6 million during the years ended December 31, 2021 and 2020, respectively. Expense associated with the Plan is charged to current operations. All obligations within the respective years have been fully funded.

### Foreign Currency

We remeasure foreign currency transactions into the relevant functional currency and record the foreign currency transaction gains or losses as a component of other (expense) income in our consolidated statements of operations. We translate the financial statements of our non-U.S. dollar functional currency foreign subsidiary into U.S. dollars in consolidation and record the translation gains or losses as other comprehensive income (loss).

## Profits interest grants

Class B "profits interest" units of ProFrac LLC ("Class B Units") entitle the holders to their respective percentage share of cash distributions either through ordinary course distributions or in connection with a business sale after all normal capital has been distributed.

On May 28, 2020, ProFrac LLC awarded ten Class B Units (effectively 1%) (the "Grant") to an executive. Under the terms of the award agreement, after recovery of Class A members' equity, additional returns are to be available for distribution between profits interest holders and the members in accordance with their respective residual ownership. Four of such Class B Units granted vested immediately and six Class B Units vest in equal annual installments of two Class B Units on the anniversary of the agreement date in 2021, 2022 and 2023. In connection with this offering, all such Class B Units will be forfeited.

Awards of profits interests must be evaluated as either unit-based payment arrangements or profit-sharing plans. The issuance of Class B Units was determined to be akin to a share-based payment arrangement.

Accordingly, ProFrac LLC is required to estimate the value of any stock compensation as of the date of the award, and any related value is to be amortized over the vesting period. Due to the timing of the Grant issuance amid uncertainties early in the COVID-19 pandemic along with the then-depressed oil prices, the value of the vested portion of the Grant, which only becomes eligible to receive cash payouts after all Class A members' interests have been recovered, was determined to be insignificant.

## Income taxes

The ProFrac Predecessor entities were organized as limited liability companies or a limited partnership and were treated as either a disregarded entity or a partnership for U.S. federal income tax purposes, whereby the ordinary business income or loss and certain deductions were passed-through and reported on the members' income tax returns. As such, the ProFrac Predecessor is not required to provide for U.S. federal income taxes and no provision for U.S. federal income tax expense or deferred tax asset or liability is reflected in the historical financial statements. Additionally, with the acquisition of EKU, the Company is subject to certain foreign taxes, which were immaterial for the year ended December 31, 2021. However, certain states subject the ProFrac Predecessor to entity-level taxation. For the years ended December 31, 2021 and 2020, state tax (benefit) expense of \$(0.2) million and \$0.6 million was recognized. For any deferred tax assets recorded, the Company will assess the available positive and negative evidence to estimate whether sufficient future taxable income will be generated to utilize the existing deferred tax assets, and will record a valuation allowance when it is more likely than not that some portion of the deferred tax assets will not be realized.

### New accounting standards to be adopted

We have not yet implemented Financial Accounting Standards Board (FASB) Accounting Standards Update (ASU) No. 2016-02, Leases (Topic 842). The ASU introduces a new accounting model for leases, which requires recognition of a right-of-use asset and lease liability on the balance sheet for operating leases. Implementation is not expected to have a material impact on our results of operations, however the Company expects to be required to recognize material assets or liabilities associated with the right to use certain leased assets upon adoption. We are required to adopt Topic 842 using the effective date of January 1, 2022, using the modified retrospective method. Under this adoption method, all leases that are in effect and existence as of, and after the transition date, with a cumulative impact to retained earnings in that period. We expect that this standard will have a material effect on our financial statements. While we continue to assess all the effects of adoption, we currently believe the most significant effects relate to the recognition of new Right of Use ("ROU") assets and lease liabilities on our balance sheet and providing significant new disclosures about our leasing activities. On adoption, we currently expect to recognize additional operating liabilities of approximately \$30.0 million to \$40.0 million with corresponding ROU assets of the same amount based on the present value of the remaining minimum lease payments under current leasing standards for existing operating leases. This excludes ROU assets and operating lease liabilities acquired in the FTSI Acquisition (defined below). Additionally, the ROU asset to be recognized upon adoption will increase due to the lease commitments associated with the FTSI Sale Leaseback. See Note 11—Subsequent Events.

We have not yet implemented FASB ASU No. 2016-13, Financial Instruments – Credit Losses. The ASU introduces a new accounting model, the Current Expected Credit Losses model (CECL), which requires recognition of credit losses and additional disclosures related to credit risk. The CECL model utilizes a lifetime expected credit loss measurement objective for the recognition of credit losses for loans and other receivables at the time the financial asset is originated or acquired. The expected credit losses are adjusted each period for changes in expected lifetime credit losses. This model replaces the multiple existing impairment models previously used under U.S. GAAP, which generally require that a loss be incurred before it is recognized. The new standard also applies to financial assets arising from revenue transactions such as contract assets and accounts receivable.

Implementation is currently required for fiscal years beginning after December 15, 2022. The Company does not believe implementation will have a material impact on its financial statements.

We have not yet implemented FASB ASU No. 2019-12, Income Taxes (Topic 740) – Simplifying the Accounting for Income Taxes, which removes specific exceptions to the general principles in Topic 740 in GAAP. The new guidance also improves the application of income tax-related guidance and simplifies GAAP for franchise taxes that are partially based on income, transactions with a government that result in a step up in the tax basis of goodwill, separate financial statements of legal entities that are not subject to tax, and enacted changes in tax laws in interim periods. The new guidance will be effective for fiscal years beginning after December 15, 2021, and interim periods within fiscal years beginning after December 15, 2022; however, early adoption is permitted. The Company is currently evaluating the impact of adopting the new guidance on the consolidated financial statements.

# 3. Inventories

The following table summarizes the components of our inventories as of December 31, 2021 and December 31, 2020:

	December 31, 2021	December 31, 2020
Raw materials and supplies	\$ 13,911	\$ 10,492
Work in process	3,288	5,104
Finished products and parts	56,743	41,002
Total	\$ 73,942	\$ 56,598

The amounts above are reported net of an allowance for obsolete inventory, which was \$4.3 million and \$4.4 million as of December 31, 2021 and 2020, respectively.

# 4. Property, plant, and equipment

The following table summarizes the components of our property, plant, and equipment, net as of December 31, 2021 and December 31, 2020:

	December 31, 2021	December 31, 2020
Machinery and equipment	\$ 760,829	\$ 738,312
Mining property and mine development	34,809	4,809
Office equipment, software and other	5,550	4,820
Buildings and leasehold improvements	15,947	16,695
Total	817,135	764,636
Less: accumulated depreciation and depletion	(464,178)	(352,565)
Construction in progress	10,730	17,613
Property, plant, and equipment, net	\$ 363,687	\$ 429,684

Depreciation expense for the years ended December 31, 2021 and 2020 was \$140.7 million and \$150.7 million, respectively.

Major classifications of property, plant, and equipment and their respective useful lives are as follows:

Machinery and equipment	2 years—10 years
Office equipment, software, and other	3 years—7 years
Buildings and leasehold improvements	2 years—40 years

# 5. Intangible assets

The following table summarizes the components of our intangible assets at December 31, 2021 and December 31, 2020:

	December 31, 2021				December	31, 2020						
	Gross	Less:		Gross	Le	ss:	Net					
	Book Accun		Book		Accumulated Net Bool		Accumulated		Book	Accum	ulated	Book
	Value	Amor	tization	Value	Value	Amorti	ization	Value				
Electric frac licenses	\$22,500	\$		\$22,500	\$—	\$		\$—				
Acquired technology	5,905		(589)	5,316								
Intangible assets, net	\$28,405	\$	(589)	\$27,816	<u>\$—_</u>	\$		<u>\$</u> —				

In June 2021, ProFrac LLC entered into an agreement with U.S. Well Services, LLC ("USWS") to purchase three licenses to manufacture electric frac fleets (one license per fleet) using USWS CleanFleet® technology. The purchase price was \$7.5 million per license for a total of \$22.5 million. Under the terms of the agreement, ProFrac LLC has the option, but no obligation, to purchase seven additional licenses for \$7.5 million each, and ten additional licenses for \$9.0 million each thereafter.

Intangible assets are amortized over the period the Company expects the asset to generate cash flows. As such, we will amortize each electric frac license through the remaining license period, beginning when the initial fleet built under each license is placed into service, a period we estimate to be 17 years. For technology acquired during 2021, we estimated this period to be 7 years. Amortization expense related to intangible assets was \$0.6 million for the year ended December 31, 2021.

# 6. Indebtedness

The following table summarizes the components of our debt as of December 31, 2021 and December 31, 2020:

	December 31,	December 31,
ADI Credit Facility	2021	2020
ABL Credit Facility	\$ 69,000	\$ 41,000
Term Loan	171,355	137,355
First Financial loan	30,000	_
Main Street loan	_	35,000
Tractor notes	_	3,911
Best Flow Credit Facility <sup>(1)</sup>	7,101	37,303
Best Flow notes payable <sup>(1)</sup>	_	_
Best Flow Note <sup>(1)</sup>	10,827	_
Alpine Promissory Note <sup>(1)</sup>	16,717	_
Alpine Credit Facility <sup>(1)</sup>	_	21,135
Other	1,695	5,829
Total gross debt	306,695	281,533
Less: unamortized debt issuance costs	(5,129)	(5,820)
Less: current portion of long-term debt	(31,793)	(15,484)
Total long-term debt	\$ 269,773	\$ 260,229

# (1) Related party debt agreements.

### ABL Credit Facility

On March 14, 2018, ProFrac LLC entered into a senior secured asset-based revolving credit agreement (the "ABL Credit Facility"), with Barclays Bank PLC, as administrative agent and collateral agent (the "ABL Agent"), and the lenders party thereto. The ABL Credit Facility has a maturity date of March 14, 2023.

LIBOR borrowings under the ABL bear interest at the greater of LIBOR or0.00%, plus a margin of 1.50% to 2.00%, depending on facility utilization. The interest rate was 2.75% as of December 31, 2021. Base rate loans are also available at our option.

The maximum availability of credit under the ABL is limited at any time to \$105.0 million or a borrowing base, as defined in the credit agreement. The borrowing base is calculated on percentages of eligible accounts receivable and inventory and is subject to certain reserves. If the amount available under the ABL is less than either 15.0% of our maximum availability or \$10.0 million, we will be required to maintain a minimum fixed charge coverage ratio of 1.0 to 1.0. Although we had availability in excess of this threshold, we were in compliance with this covenant as of December 31, 2021. If at any time borrowings and letters of credit issued under the ABL exceed the borrowing base, we will be required to repay an amount equal to such excess.

As of December 31, 2021, the borrowing base was \$146.2 million, therefore maximum availability under the ABL was the facility maximum of \$105.0 million. There were \$69.0 million of borrowings outstanding under the ABL, and \$3.1 million of letters of credit were outstanding, resulting in \$32.9 million of availability under the ABL.

The ABL Credit Facility contains certain customary representations and warranties and affirmative and negative covenants. The negative covenants include, subject to customary exceptions, limitations on indebtedness, dividends, distributions and other restricted payments, investments, acquisitions, prepayments of specified junior indebtedness, amendments of specified junior indebtedness, transactions with affiliates, dispositions, mergers and consolidations, liens, restrictive agreements, sale and leaseback transactions, changes in fiscal periods and changes in line of business. ProFrac LLC was in compliance with all covenants as of December 31, 2021.

On March 4, 2022, the ABL Credit Facility was extinguished. See Note 11—Subsequent Events.

### Term Loan

On September 7, 2018, ProFrac LLC entered into a \$180.0 million term loan agreement ("Term Loan"), which matures on September 15, 2023, with a group of lenders with Barclays Bank, PLC as administrative agent. Principal payments are due in quarterly installments, however due to prepayments made to date, there will be no required minimum amortization (other than excess cash flow prepayments) before termination of the Term Loan. Excess cash flow is calculated quarterly to determine any additional minimum prepayments. On June 24, 2021, ProFrac LLC and its Term Loan lenders reached an agreement to expand the facility by \$40.0 million.

The Term Loan, as amended, requires minimum excess cash flow prepayments as follows, each due approximately 55 days afterperiod-end: \$0.0 million for the fiscal quarters ended March 31, 2021 through and including December 31, 2021, and \$5.0 million for the fiscal quarters ended March 31, 2022 through and including June 30, 2023.

LIBOR borrowings under the Term Loan bear interest at the greater of LIBOR or 1.25%, plus a margin of 6.25% to 8.50%, depending on the total net leverage ratio as defined under the Term Loan. The interest rate was 9.75% as of December 31, 2021.

On May 29, 2020, one of the members of ProFrac LLC purchased and simultaneously retired a portion of the Term Loan. ProFrac LLC recognized the retirement of these notes, a non-cash transaction for ProFrac LLC, as a contribution of equity and reduction of long-term debt of \$10,549, the face value of the notes retired.

The Term Loan contains certain restrictive covenants, including a financial covenant which requires ProFrac LLC to maintain a total net leverage ratio, as defined in the credit agreement, of no greater than 2.25:1.00 for the fiscal quarters ended September 30, 2019 through and including March 31, 2020, 3.50:1.00 for the fiscal quarters ended June 30, 2020 through and including March 31, 2021,3.00:1.00 for the fiscal quarter ended June 31, 2021,2.75:1.00 for the fiscal quarter ended September 30, 2021, 2.50:1.00 for the fiscal quarter ended December 31, 2021, and 2.00:1.00 for the fiscal quarter ended March 31, 2022 and thereafter. ProFrac LLC was in compliance with all required covenants as of December 31, 2021.

The Term Loan contains certain customary representations and warranties and affirmative and negative covenants. The negative covenants include, subject to customary exceptions, limitations on indebtedness, dividends, distributions and other restricted payments, investments, acquisitions, prepayments of specified junior indebtedness, amendments of specified junior indebtedness, transactions with affiliates, dispositions, mergers and consolidations, liens, restrictive agreements, changes in fiscal periods and changes in line of business.

In February 2022, ProFrac and its Term Loan lenders entered into an agreement to amend the Term Loan. The amendment expanded the facility by \$48.0 million. See Note 11—Subsequent Events.

On March 4, 2022, the Term Loan was extinguished. See Note 11—Subsequent Events.

### First Financial Loan

On December 22, 2021, the Company entered into a \$30.0 million loan agreement with First Financial Bank, N.A. ("First Financial Loan") with a scheduled maturity date of January 1, 2024. The interest rate is LIBOR plus 3.5%, and the loan is to be repaid by equal monthly payments of principal and interest beginning in February 2022. The First Financial Loan contains certain restrictive covenants which require the Company to maintain a fixed charge coverage ratio of at least 1.00:1.00 and a maximum net leverage ratio of 3.00:1.00. The Company was in compliance with all covenants as of December 31, 2021.

#### Main Street Loan

On July 22, 2020, ProFrac LLC entered into a \$35.0 million loan agreement with First Financial Bank, N.A. which matures onJuly 22, 2025 ("Main Street Loan"). The interest rate is LIBOR plus 3.00%, floating daily. In August 2021, the loan agreement was amended to delete the covenants in place prior to the amendment. As amended, the Main Street Loan contains certain restrictive covenants which require ProFrac LLC to maintain a Fixed Charge Coverage Ratio of at least 1.00:1.00, and a Maximum Leverage Ratio of 3.50:1.00. Additionally, the Main Street Loan restricts the payment of distributions or dividends, other than for the payment of taxes.

In December 2021, the Main Street Loan had a balance of \$32.2 million. The Main Street Loan was extinguished with a cash payment of \$2.2 million and the remainder refinanced with a \$30.0 million loan with First Financial Bank, N.A. The Company incurred a loss on extinguishment of debt of \$0.5 million related to the extinguishment of the Main Street Loan, reflected in other (expense) income on the consolidated statements of operations.

### Tractor notes

On May 9, 2018, ProFrac LLC entered into a \$17.5 million loan agreement with First Financial Bank, N.A. which matured onNovember 9, 2020 ("Tractor Note A"). The interest rate on Tractor Note A was 5.35%.

On May 9, 2018, ProFrac LLC entered into a \$12.5 million loan agreement with First Financial Bank, N.A. with a stated maturity date of April 9, 2021 ("Tractor Note B"). The interest rate on Tractor Note B was 5.35%. Tractor Note B was retired in January 2021.

On May 3, 2019, ProFrac LLC entered into a \$5.5 million loan agreement with First Financial Bank, N.A. with a stated maturity date of November 3, 2021 ("Tractor Note C"). The interest rate on Tractor Note C was 6.10%. Tractor Note C was retired in January 2021.

### Best Flow Credit Facility

On February 4, 2019, Best Flow entered into a revolving loan credit agreement (the "Best Flow Credit Facility"), with Equify Financial, LLC ("Equify Financial") as lender. Equify Financial is a related party. the Best Flow Credit Facility provided for a revolving credit facility in an aggregate principal amount at any time outstanding up to \$9.0 million, subject to borrowing base availability. As of December 31, 2021, the Company had \$7.1 million in outstanding borrowings and \$1.9 million of remaining availability. The Best Flow Credit Facility has a maturity date of February 4, 2026.

The interest rate under the Best Flow Credit Facility is the lesser of (i) the Prime Rate (as defined in the Best Flow Credit Facility) plus the applicable margin (3.50%) and (ii) the Maximum Rate (as defined in the Best Flow Credit Facility). All accrued but unpaid interest on the outstanding principal balance is due and payable monthly on the first day of each calendar month. The Best Flow Credit Facility is secured by a first lien on substantially all of the assets of Best Flow.

The Best Flow Credit facility is subject to mandatory prepayments of the entire amount of the excess if the unpaid balance of Advances (as defined in the Best Flow Credit Facility) exceeds the borrowing base then in effect. The Best Flow Borrower can voluntarily prepay all or any portion of the Obligations (as defined in the Best Flow Credit Facility).

The Best Flow Credit Facility contains certain customary representations and warranties and affirmative and negative covenants. The negative covenants include, subject to exceptions, limitations on indebtedness, dividends, distributions and other restricted payments, investments, acquisitions, prepayments of indebtedness, transactions with affiliates, dispositions, mergers and consolidations, liens, restrictive agreements, changes in fiscal periods and changes in line of business.

The Best Flow Credit Facility contains certain customary events of default. If an event of default occurs and is continuing, the lenders may declare all loans to be immediately due and payable. Some events of default require an automatic termination of the loans and become immediately due and payable. As of December 31, 2021, we were in compliance with all of our covenants, and there were no existing defaults or events of default, related to the Best Flow Credit Facility.

On January 31, 2022, Best Flow and Equify Financial entered into an agreement to amend the Best Flow Credit Facility. See Note 11—Subsequent Events

On March 4, 2022, the Best Flow Credit Facility was extinguished. See Note 11—Subsequent Events.

## Best Flow notes payable

In 2021, Best Flow entered into a series of related party notes payable, the proceeds from which were used to pay down the Best Flow Credit Facility.

On January 25, 2021, Best Flow entered into a \$15.0 million loan agreement with THRC Holdings and Farris Wilks and Equify Financial as an administrative agent, who are related parties, bearing interest of 8.0%, with a maturity date of January 25, 2023. This loan agreement is unsecured and is subject to monthly excess cash flow payments (as defined in the loan agreement). This new loan agreement is unsecured and is subject to monthly excess cash flow payments (as defined in the loan agreement). This loan agreement restricts the payment of distributions or dividends, other than for the payment of taxes. All amounts outstanding under this facility together with accrued interest were contributed to ProFrac LLC in exchange for equity interests in ProFrac LLC on December 15, 2021, and such debt was canceled in connection therewith, a non-cash transaction for the Company.

On June 1, 2021, Best Flow entered into a \$3.0 million credit agreement with Wilks Brothers, who is a related party, bearing interest of 8.0%, with a maturity date of January 31, 2022. All amounts outstanding under this facility together with accrued interest were contributed to ProFrac LLC in exchange for equity interests in ProFrac LLC on December 15, 2021, and such debt was canceled in connection therewith, a non-cash transaction for the Company.

### Best Flow Note

On January 28, 2021, Best Flow issued a promissory note (the "Best Flow Note"), with Equify Financial, as holder. Equify Financial is a related party. The Best Flow Note provides for a term loan in an initial aggregate principal amount equal to \$13.0 million. Proceeds from the Best Flow Note were utilized to pay down \$7.6 million of outstanding balances on the Best Flow Credit Facility and to pay down other equipment financing agreements for \$5.4 million. The Best Flow Note matures on February 1, 2026, with a fixed interest rate of 8.0%. The principal and interest are paid in equalmonthly amortizing amounts through maturity. Prepaid amounts are subject to a 0.19% prepayment premium. Best Flow was in compliance with all covenants as of December 31, 2021.

On March 4, 2022, the Best Flow Note was extinguished. See Note 11—Subsequent Events.

### Alpine Credit Facility

On October 25, 2018, Alpine entered a secured revolving credit agreement with Equify Financial, with a maximum loan availability of \$5.3 million (the "Alpine Credit Facility"). Equify Financial is a related party. The maturity date under the Alpine Credit Facility was November 25, 2020. Interest is paid monthly at a fixed rate of \$0.0%. The revolving credit note is secured by a lien on, and security interest in, all of the assets of Alpine. Alpine had outstanding borrowings under the revolving credit note of \$21.1 million as of December 31, 2020. The Alpine Credit Facility was refinanced with the Alpine Promissory Note in January 2021.

### Alpine Promissory Note

In January 2021, Alpine entered into a \$\Delta 1.4 million promissory note with Equify Financial ("Alpine Note"). Equify Financial is a related party. The Alpine Note amortizes monthly, has an interest rate of 8.0% and has a stated maturity date in February 2027. Alpine had \$16.7 in outstanding borrowings under the Alpine Note as of December 31, 2021. Alpine was in compliance with all covenants as of December 31, 2021.

On March 4, 2022, the Alpine Note was extinguished. See Note 11—Subsequent Events.

#### Other indebtedness

As of December 31, 2021 and December 31, 2020, the ProFrac Predecessor had other debt agreements outstanding with unpaid principal balances of \$1.7 million and \$5.8 million, respectively. These were primarily comprised of various equipment financing agreements. As of December 31, 2021, these agreements have a weighted average interest rate of 5.7% and a weighted average remaining term of 6.1 years.

The following table summarizes the principal maturity schedule for our long-term debt outstanding as of December 31, 2021:

	2022	2023	2024	2025	2026	Thereafter	Total
ABL Credit Facility <sup>(1)</sup>	\$ —	\$ —	<u>\$</u>	\$ —	\$	\$ 69,000	\$ 69,000
Term Loan <sup>(1)</sup>	16,875	22,500	22,500	109,480	_	_	171,355
First Financial loan	14,110	15,890	_	_	_	_	30,000
Best Flow Credit Facility <sup>(1)</sup>	_	_	_	_	_	7,101	7,101
Best Flow Note <sup>(1)</sup>	_	_	_	_	_	10,827	10,827
Alpine Promissory Note <sup>(1)</sup>	_	_	_	_	_	16,717	16,717
Other indebtedness	808	173	152	101	75	386	1,695
Total	\$31,793	\$ 38,563	\$ 22,652	\$ 109,581	\$ 75	\$ 104,031	\$ 306,695

(1) Principal maturity for these facilities reflect the terms of the New ABL Credit Facility, the New Term Loan Credit Facility and the Equify Bridge Note which refinanced these facilities subsequent to December 31, 2021, however the presented amounts due at maturity in the schedule above are limited by the balances outstanding as of December 31, 2021. See Note 11 - Subsequent Events for additional detail and a schedule of the full principal repayment obligations as of March 30, 2022.

# 7. Related party transactions

In the normal course of business, the ProFrac Predecessor has entered into transactions with related parties where the Wilks (or entities they control) hold a controlling financial interest. During the years ended December 31, 2021 and 2020, the ProFrac Predecessor had related party transactions with the following related party entities

Automatize, LLC ("Automatize") is a logistics company that provides for the delivery of proppants on behalf of its customers, including the ProFrac Predecessor. Amounts paid to Automatize are recorded in cost of revenues, exclusive of depreciation and depletion on our consolidated statements of operations.

Cisco Logistics, LLC ("Cisco Logistics") is a logistics company that delivers sand and equipment on behalf of its customers, including the ProFrac Predecessor. Amounts paid to Cisco Logistics are recorded in cost of revenues, exclusive of depreciation and depletion on our consolidated statements of operations. Additionally, during 2021 the Company incurred costs associated with the refurbishment of certain equipment owned by Cisco Logistics and recorded a receivable of \$1.5 million for reimbursement of these expenditures.

Equify Risk Services, LLC ("Equify Risk") is an insurance broker that negotiates and secures insurance policies on behalf of its customers, including the ProFrac Predecessor. Amounts paid to Equify Risk are recorded in selling, general and administrative expenses on our consolidated statements of operations.

Equify Financial, LLC ("Equify Financial") is a finance company that provides equipment and other financing to its customers, including the ProFrac Predecessor. Amounts paid to Equify Financial are recorded in interest expenses on our consolidated statements of operations, and repayments of long-term debt on our consolidated statements of cash flows. See Note 6—Indebtedness for additional disclosures related to related party credit agreements.

Wilks Brothers, LLC ("Wilks Brothers") is a management company which provides administrative support to various businesses within their portfolio. Wilks Brothers and certain entities under its control will at times incur expenses on behalf of the ProFrac Predecessor, billing the ProFrac Predecessor for these expenses at cost as well as certain management fees. Amounts paid to Wilks Brothers are generally recorded in selling, general and administrative expenses on our consolidated statements of operations. See Note 6—Indebtedness for additional disclosures related to related party credit agreements.

Interstate Explorations, LLC ("Interstate") is an exploration and development company for which ProFrac LLC performs pressure pumping services, and from which the Company has a short-term lease for certain office space.

Flying A Pump Services, LLC ("Flying A") is an oilfield services company which provides pump down and acid services, to which ProFrac LLC rents equipment.

MC Estates, LLC and the Shops at Willow Park, LLC ("Related Lessors") own various industrial parks and office space leased by the ProFrac Predecessor. Amounts paid to Related Lessors are recorded in selling, general and administrative expenses on our consolidated statements of operations.

Wilks Construction Company, LLC ("Wilks Construction") is a construction company that has built and made renovations to several buildings for the Company. Amounts paid to Wilks Construction are recorded in capital expenditures on our consolidated statements of cash flows.

3 Twenty-Three, LLC ("3 Twenty-Three") is a payroll administrator which performs payroll services on behalf of its customers, including the ProFrac Predecessor. Amounts paid to 3 Twenty-Three are recorded in cost of revenues, exclusive of depreciation and depletion and selling, general and administrative expenses on our consolidated statements of operations.

Carbo Ceramics Inc. ("Carbo") is a provider of ceramic proppant which will at times purchase conventional proppant from the Company to act as a broker for its customers. Additionally, the Company will at times purchase manufactured proppant from Carbo for the Stimulation Services segment.

The following table summarizes expenditures with related parties for the years ended December 31, 2021 and 2020:

	Year o	Year ended	
	Decem	ber 31,	
	2021	2020	
Automatize	\$ 80,521	\$ 26,226	
Wilks Brothers	15,480	16,622	
Related Lessors	6,308	6,052	
Equify Financial	2,871	2,323	
3 Twenty-Three	1,033	1,148	
Carbo	513	_	
Cisco Logistics	509	4,181	
Interstate	80	30	
Equify Risk	3	1,602	
Wilks Construction	_	107	
Other	114		
Total	\$ 107,432	\$ 58,291	

The following table summarizes related party accounts payable as of December 31, 2021 and December 31, 2020:

	December 31, 2021	December 31, 2020
Automatize	\$ 11,198	\$ 5,633
Wilks Brothers	9,990	11,993
Wilks Construction	57	57
Carbo	10	_
Related Lessors	1	21
Cisco Logistics	_	671
Equify Financial	_	113
Other	19	
Total	\$ 21,275	\$ 18,488

The following table summarizes revenue from related parties for the years ended December 31,2021 and 2020:

	Yea	r ended
	Dece	mber 31,
	2021	2020
Flying A	\$2,701	\$ 294
Carbo	1,025	193
Interstate	116	11
Wilks Brothers	65	_
Automatize	3	701
Other	32	84
Total	\$3,942	\$1,283

The following table summarizes related party accounts receivable as of December 31, 2021 and December 31, 2020:

	December 31,	December 31,		
	2021	2020		
Flying A <sup>(1)</sup>	\$ 2,412	\$ 549		
Cisco Logistics	1,489	_		
Carbo	591	116		
Automatize	_	191		
Other	23	24		
Total	\$ 4,515	\$ 880		

(1) The amounts above are reported net of a related party allowance for doubtful accounts, which was \$0.2 million as of December 31, 2021.

Additionally, In January and February of 2021, ProFrac LLC executed two agreements with one of ProFrac LLC's members for the sale of certain lots of equipment, in exchange for \$8.7 million in cash, an amount that approximates the net book value of the assets. Under these agreements, for any assets subsequently resold by the member, ProFrac LLC will reimburse the member for a certain percentage of the net loss, or conversely be entitled to a certain percentage of the net gain, at rates established in the agreements. As of December 31, 2021, the majority of the assets have been sold by the member, and ProFrac LLC's remaining liability of approximately \$0.2 million for assets that have not been resold has been recorded.

# 8. Investments and Acquisitions

Basin Production and Completion LLC

Between September 27 and December 23, 2021, the Company purchased Series B-1 Preferred Units of Basin Production and Completion LLC ("BPC"), a manufacturer of equipment used in hydraulic fracturing for \$4.2 million. This comprises approximately 20% of the BPC Series B Preferred Units. As of December 31, 2021, we do not exercise significant influence over this entity. As we determined this to be an equity security, we initially recorded our investment at cost, presented as "Investments" in our consolidated balance sheets as of December 31, 2021. As the Series B-1 Preferred Units do not have a readily determinable fair value, and do not qualify for the practical expedient to estimate fair value, after initial recognition we will measure our investment at cost minus any impairment. We did not identify any indicators of impairment of our investment as of December 31, 2021. The Company made additional investments in BPC and entered into a note receivable with a subsidiary of BPC subsequent to December 31, 2021. See Note 11 – Subsequent Events.

# **EKU**

On December 22, 2020, the Company purchased a 25% stake in EKU, GmbH ("EKU"), an equipment manufacturer based in Germany, for \$1.2 million. For the year ended December 31, 2020, we accounted for this investment using the equity method as we had significant influence over EKU, and held a voting interest of 20% or greater, but less than 50%. The results of operations for EKU between December 22 and December 31, 2020 were not material, and are presented in the consolidated statements of operations in "Cost of revenues." Investment in this associated company is presented on a one-line basis as "Investment in associate" in our consolidated balance sheets as of December 31, 2020. In January 2021, the Company obtained a controlling interest in EKU, the results of which are consolidated thereafter.

The Company obtained a 75% controlling interest in EKU in January 2021 and performed a purchase price allocation in conjunction with the consolidation of this subsidiary. We recognized net working capital of \$2.5 million, property, plant and equipment of \$0.4 million, intangible assets of \$3.5 million and debt of

\$1.4 million at estimated fair value. In consolidation, we eliminated our investment in associate, recognized equity of \$.7 million for the value of our 75% interest, and noncontrolling interest of \$1.2 million for the value of the minority shareholder positions. See Note 5—Intangible assets for additional information related to the recognized intangible assets.

### Munger Ranch

The Company entered into an agreement to acquire property in West Texas ("Munger Right Agreement") in November 2021 for a purchase price of \$30.0 million. Under the Munger Right Agreement, the sellers were given the option to receive the consideration in cash, or in the event of an initial public offering ("IPO") prior to November 17, 2022, in equity, at the sellers' election. Under the equity option, in the event the Company completes an IPO, the sellers would be entitled to 1.5% of the outstanding shares of common stock immediately following the IPO. Each seller initially elected the equity option, as such this was a non-cash transaction for the Company for the year ended December 31, 2021. If the IPO does not occur prior to November 17, 2022, then the consideration in the sale will be payable in cash. Additionally, the Munger Right Agreement includes a 'Make Whole' provision. Under the Make Whole provision, if any seller liquidates 100% of the Class A shares they are issued prior to theone-year anniversary of the IPO, then the company will pay the difference between the IPO price of the shares disposed and the disposal price of the Class A shares. As of December 31, 2021, the liability of \$30.0 million is presented within Other Current Liabilities in our Balance Sheet. The acquired property was treated as an asset acquisition and not an acquisition of a business, and is presented within Property, plant, and equipment in our consolidated balance sheets as of December 31, 2021.

Subsequent to December 31, 2021, a related party entered into a Rights Agreement with one of the sellers, under which the related party was assigned rights to \$8.1 million of the \$30.0 million in consideration related to the Munger Right Agreement. See Note 11—Subsequent Events.

### iO-TEO, LLC

The Company acquired iO-TEQ, LLC ("IOT") in October 2021 for \$2.2 million and performed a purchase price allocation at the acquisition date. We recognized net working capital of \$0.2 million, property, plant and equipment of \$0.1 million, intangible assets of \$2.4 million and debt of \$0.4 million at estimated fair value. See Note 5—Intangible assets for additional information related to the recognized intangible assets.

#### Best Flow minority interests

In December 2021, the Company purchased a 2.302% interest in Best Flow from an investor for \$0.6 million in cash. Also in December 2021, the Company entered into an agreement with Eagleton Venture, Inc. ("Eagleton") to purchase Eagleton's 15.172% interest in Best Flow ("Eagleton Right Agreement") for a purchase price of \$3.9 million. Under the Eagleton Right Agreement, in the event that the Company completes an initial public offering ("IPO"), the Company will issue Class-A shares to Eagleton with a fixed total value of \$3.9 million divided by the IPO price. If the Company has not consummated an IPO by February 15, 2022, Eagleton may elect to receive \$3.9 million in cash. As such, this was anon-cash transaction for the Company for the year ended December 31, 2021. As of December 31, 2021, this liability of \$3.9 million is presented within Other Current Liabilities in our Balance Sheet. Subsequent to December 31, 2021, Eagleton elected to receive the \$3.9 million in cash, which was paid in March 2022. See Note 11—Subsequent Events.

# 9. Commitments and contingencies

# Litigation

In the ordinary course of business, we are subject of, or party to a number of pending or threatened legal actions and administrative proceedings arising in the ordinary course of our business. While many of these matters

involve inherent uncertainty, we believe that the amount of the liability, if any, ultimately incurred with respect to proceedings or claims will not have a material adverse effect on our consolidated financial position as a whole or on our liquidity, capital resources or future annual results of operations.

When deemed necessary, we establish reserves for certain legal proceedings. The establishment of a reserve is based on an estimation process that includes the advice of legal counsel and subjective judgment of management. While management believes these reserves to be adequate, it is reasonably possible we could incur additional losses with respect to those matters in which reserves have been established. We will continue to evaluate our litigation on a quarterly basis and will establish and adjust any litigation reserves as appropriate to reflect our assessment of the then current status of litigation. As of December 31, 2021, we have recorded an estimated liability of \$4.0 million related to a contract dispute claim with a vendor, which is included in accrued expenses in our consolidated balance sheets and in selling, general and administrative expense on our consolidated statements of operations for the year ended December 31, 2020.

#### Lease commitments

We have non-cancelable operating lease commitments related to our corporate office, several of our field offices and manufacturing facilities. The following table summarizes the future minimum rental payments required under our operating lease arrangements as of December 31, 2021.

2022	\$ 6,401
2023	6,256
2024	5,844
2025	5,329
2026	5,196
Thereafter	_16,954
Total	<u>\$ 45,980</u>

The table above does not include lease commitments associated with the FTSI Sale Leaseback. See Note 11—Subsequent Events.

Rent expense under our operating lease arrangements was \$6.1 million and \$6.1 million for the years ended December 31, 2021 and 2020, respectively.

### 10. Segment information

Our business has three reportable segments: Stimulation Services, Manufacturing and Proppant production. Each reportable segment represents a separate business unit that operated as a standalone company prior to the reorganization of the Company in December 2021. Following the reorganization, each reportable segment continues to have distinct management and prepares discrete financial information for the segment (consistent with when each operated as a standalone business). The Wilks, who are our Chief Operating Decision Makers ("CODMs"), review the discrete segment financial information, including Adjusted EBITDA as the measure of profitability, to evaluate the performance of our segments and make resource allocation decisions

We account for intersegment transactions as if the transactions were with third parties, that is, at estimated current market prices. For the years ended December 31, 2021 and 2020, intersegment revenues for the manufacturing segment were 90% and 97%, respectively. For the years ended December 31, 2021 and 2020, intersegment revenues for the proppant production segment were 40% and 20%, respectively.

The performance of our segments is evaluated primarily on Adjusted EBITDA. We define Adjusted EBITDA as our net income (loss), before (i) interest expense, net, (ii) income tax benefit (provision), (iii) depreciation, depletion and amortization, (iv) loss on disposal of assets, (v) loss on extinguishment of debt, (vi) bad debt

expense, (vii) loss on foreign currency transactions and (viii) other unusual ornon-recurring charges, such as costs related to our initial public offering, severance charges, and non-recurring supply commitment charges.

Segment information as of and for the years ended December 31, 2021 and 2020 is as follows:

	Year ended I	Year ended December 31,		
	2021	2020		
Revenues				
Stimulation services	\$ 745,373	\$ 538,282		
Manufacturing	76,360	46,222		
Proppant production	27,225	10,215		
Total segments	848,958	594,719		
Eliminations	(80,605)	(47,040)		
Total	\$ 768,353	\$ 547,679		
Cost of revenues, excluding depreciation, depletion and amortization				
Stimulation services	\$ 570,828	\$ 433,122		
Manufacturing	65,849	40,424		
Proppant production	14,050	6,064		
Total segments	650,727	479,610		
Eliminations	(80,605)	(47,040)		
Total	\$ 570,122	\$ 432,570		
Adjusted EBITDA	<u></u>			
Stimulation services	\$ 122,634	\$ 68,787		
Manufacturing	1,382	1,325		
Proppant production	10,672	2,685		
Total segments	134,688	72,797		
Eliminations	_	_		
Total	\$ 134,688	\$ 72,797		
Depreciation, depletion and amortization	<u>+,</u>	<del>+</del>		
Stimulation services	\$ 128,004	\$ 138,122		
Manufacturing	3,788	2,796		
Proppant production	8,895	9,744		
Total	\$ 140,687	\$ 150,662		
Capital expenditures	<u> </u>	<u> </u>		
Stimulation services	\$ 82,545	\$ 46,371		
Manufacturing	3,500	1,661		
Proppant production	1,355	5		
Total	\$ 87,400	\$ 48,037		
Total assets	4_07,100	4_10,001		
Stimulation services	\$ 545,047	\$ 452,729		
Manufacturing	77,968	64,769		
Proppant production	100,294	73,918		
Total segment assets	723,309	591,416		
Eliminations	(58,739)	(14,139)		
Total	\$ 664,570	\$ 577,277		
10141	\$ 004,570	\$ 311,411		

The following table sets forth the reconciliation of Segment Adjusted EBITDA to net loss in our consolidated statements of operations:

	2021	2020
Stimulation services—Adjusted EBITDA	\$ 122,634	\$ 68,787
Manufacturing—Adjusted EBITDA	1,382	1,325
Proppant production—Adjusted EBITDA	10,672	2,685
Total	\$ 134,688	\$ 72,797
Interest expense, net	(25,788)	(23,276)
Income tax benefit (provision)	186	(582)
Depreciation, depletion and amortization	(140,687)	(150,662)
Loss on disposal of assets, net	(9,777)	(8,447)
Loss on extinguishment of debt	(515)	_
Bad debt expense, net of recoveries	1,164	(2,778)
Loss on foreign currency transactions	(249)	_
Reorganization costs	(2,060)	_
Severance charges	(500)	_
Supply commitment charges		(5,600)
Net loss	<u>\$ (43,538)</u>	<u>\$(118,548)</u>

### 11. Subsequent events

The ProFrac Predecessor has evaluated events subsequent to December 31, 2021 through March 31, 2022, the date these financial statements were made available to be issued, and report the following events:

- a) On January 31, 2022, Best Flow and Equify Financial entered into an agreement to amend the Best Flow Credit Facility to increase the facility by \$6.0 million to \$15.0 million. This amendment triggered a Term Loan covenant violation which restricted additional borrowings, but a waiver was received. This facility was extinguished on March 4, 2022.
- b) On February 16, 2022, Eagleton elected to receive in cash the \$3.9 million deferred purchase price consideration related to the Company's purchase of its noncontrolling interest in Best Flow. This obligation was paid March 2, 2022.
- c) On February 9, 2022, the Company entered into an agreement to purchase all the series A-1 & B-1 preferred units of BPC for \$46.0 million ("Basin Units Acquisition"), consisting of \$40.0 million to BPC for series A-1 & B-1 preferred units and \$6.0 million to selling holders of BPC series B-1 preferred units. The Company had an investment of \$4.4 million in BPC as of December 31, 2021.
- d) In February 2022, ProFrac and its Term Loan lenders entered into an agreement to amend the Term Loan. The amendment expanded the facility by \$48.0 million, the Term Loan lenders consented to the Basin Units Acquisition funded by the additional proceeds from the Term Loan, and the Term Loan lenders waived certain covenant violations that occurred in early 2022. There were no further covenant violations between this amendment and the extinguishment of the Term Loan on March 4, 2022.
- e) On February 2, 2022, ProFrac entered into an agreement with Flotek pursuant to which Flotek will provide full downhole chemistry solutions for a minimum of ten hydraulic fleets or 33% of our Pre-Acquisition Fleets for three years starting on April 1, 2022, at a price of cost plus 7.0% ("Flotek Supply Agreement"). In exchange for entry into the Flotek Supply Agreement, we received \$10.0 million in initial principal amount of notes that are convertible into Flotek common stock and acquired an additional \$10.0 million in principal amount of such notes in a related private offering transaction. On February 16, 2022, the Flotek Supply Agreement was amended to increase the term to ten years and increase the scope to 30 fleets. In exchange, Flotek agreed to issue us \$50.0 million in initial principal amount of notes that will be convertible into

Flotek common stock The Flotek Supply Agreement Amendment includes a minimum annual volume commitment whereby we will be obligated to pay Flotek liquidated damages equal to 25% of the shortfall for such year, should we fail to meet the minimum purchase amount. The notes issued to ProFrac accrue paid-in-kind interest at a rate of 10% per annum, have a maturity of one year, and convert into common stock of Flotek (a) at the holder's option at any time prior to maturity, at a price of \$1.088125 per share, (b) at Flotek's option, if the volume-weighted average trading price of Flotek's common stock equals or exceeds \$2.50 for 20 trading days during a 30 consecutive trading day period, or (c) at maturity, at a price of \$0.8705 (the "Convertible Notes").

- f) On February 4, 2022, a related party entered into a Rights Agreement with Encantor Properties LP, one of the sellers from whom the Company purchased the Munger Ranch property, under which the related party was assigned rights to \$8.1 million of the \$30.0 million in consideration related to the Munger purchase.
- g) On February 14, 2022, the Company issued an unsecured promissory note to FHE USA, LLC, a subsidiary of BPC in an aggregate principal amount of \$1.3 million. The note accrues PIK interest quarterly at a rate of \$5.0% per annum and has a stated maturity date of February 14, 2027.
- h) On March 4, 2022, the Company entered into a senior secured term loan credit agreement ("New Term Loan Credit Facility"), with Piper Sandler Finance LLC, as administrative agent and collateral agent, and the lenders party thereto. The New Term Loan Credit Facility provides for a term loan in an aggregate principal amount of \$450.0 million. The New Term Loan Credit Facility has a maturity date of March 4, 2025. Borrowings under the New Term Loan Credit Facility bear interest at the greater of Adjusted Term SOFR or 1.0%, plus a margin of 6.50% to 8.00% (8.50% until October 1, 2022), depending on the total net leverage ratio as defined under the New Term Loan Credit Facility. The interest rate was 9.5% as of the date of funding.
- i) On March 4, 2022, the Company entered into a senior secured asset-based revolving credit agreement (the "New ABL Credit Facility"), with JPMorgan Chase Bank N.A., as administrative agent and collateral agent, and the lenders party thereto. The New ABL Credit Facility provides for an asset-based revolving credit facility, originally in the amount of up to \$100.0 million, subject to borrowing base availability, and has a maturity date of March 4, 2027. The interest rate was 4.0% as of the date of funding. As of March 30, 2022, the borrowing base is estimated to be \$158.9 million, therefore maximum availability under the New ABL Credit Facility is the facility maximum of \$100.0 million. There were \$70.7 million of borrowings outstanding and \$9.2 million of letters of credit outstanding, resulting in approximately \$20.1 million of availability under the New ABL Credit Facility.
- j) On March 4, 2022, the Company entered into a \$45.8 million subordinated promissory note with Equify Financial ("Equify Bridge Loan"). The Equify Bridge loan accrues interest at a rate of 1.00% and matures in March 2027.
- k) On March 4, 2022, the Company entered into a \$22.0 million subordinated promissory note with THRC Holdings, L.P., a related party ("Backstop Note"). The Backstop Note accrues interest at a rate of 1.74% and matures in March 2027.
- l) On March 4, 2022, the Company entered into a \$22.0 million subordinated promissory note with THRC Holdings, L.P. ("THRC Holdings"), a related party ("Closing Date Note"). The Closing Date Note accrues interest at a rate of 1.74% and matures in March 2027.

The following table summarizes the principal maturity schedule for our long-term debt outstanding as of March 30, 2022:

	2022	2023	2024	2025	2026	Thereafter
New ABL Credit Facility	<u>\$</u>	\$ —	\$ —	\$ —	\$	\$ 70,706
New Term Loan Credit Facility	16,875	22,500	22,500	388,125	_	_
First Financial loan	11,627	15,890	_	_	_	_
Equify Bridge Loan	_	_	_	_	_	45,800
Backstop Note	_	_	_	_	_	22,000
Closing Date Note	_	_	_	_	_	22,000
Other indebtedness	136	173	152	101	75	386
Total	\$28,638	\$ 38,563	\$ 22,652	\$ 388,226	\$ 75	\$ 160,892

- m) On March 4, 2022 the Company extinguished the Term Loan, ABL Credit Facility, Best Flow Credit Facility, Best Flow Note and Alpine Promissory Note. All were refinanced by the New ABL Credit Facility, New Term Loan Credit Facility and Equify Bridge Loan, and as such the extinguishments were non-cash transactions for the Company. The Company recognized a loss on extinguishment of debt of \$8.3 million associated with these extinguishments.
- n) On March 4, 2022, the Company acquired the outstanding stock of FTS International, Inc. ("FTSI") for a purchase price of \$407.5 million, consisting of cash consideration of \$334.6 million, and certain equity interests of \$72.9 million (the "FTSI Acquisition").

We will account for the acquisition of FTSI using the acquisition method of accounting. We use our best estimates and assumptions to assign fair value to the tangible and intangible assets expected to be acquired and liabilities expected to be assumed at the acquisition date. The Company is still evaluating the purchase price allocation and the initial accounting for the business combination. Neither were completed prior to March 31, 2022, the date these financial statements were made available to be issued. The determination and preliminary allocation of the purchase consideration are subject to change during the measurement period (up to one year from the date the acquisition closes) as we finalize the valuations of the net tangible and intangible assets acquired and liabilities assumed. The estimated purchase price allocation is not necessarily indicative of or intended to represent the results that would have been achieved had the FTSI Acquisition been consummated as of the dates indicated or that may be achieved in the future.

The Company has performed a preliminary valuation analysis of the fair market value of FTSI's assets and liabilities. The following table summarizes the allocation of the preliminary purchase price as of December 31, 2021 (in thousands):

Assets acquired:	
Cash and cash equivalents	\$ 85,149
Accounts receivable	67,303
Prepaid expense and other assets	4,238
Inventories	38,540
Property, plant and equipment	284,108
Goodwill and intangible assets	36,500
Other assets	1,473
Total assets acquired	517,311
Liabilities assumed:	
Accounts payable	70,591
Accrued expenses	37,932
Other current liabilities	309
Other non-current liabilities	979
Total liabilities assumed	109,811
Net assets acquired	\$ 407,500

The final purchase price allocation will be determined when the Company has completed the detailed valuations and necessary calculations. The final allocation could differ materially from the preliminary allocation used in the pro forma adjustments. The final allocation may include (i) changes in the fair value of property, plant and equipment, (ii) changes to intangible assets and goodwill and (iii) other changes to assets and liabilities.

o) In connection with the completion of the FTSI Acquisition, FTSI conveyed to Wilks Development, LLC, an affiliate of ProFrac LLC, substantially all of FTSI's owned real property consisting primarily of FTSI's manufacturing facilities in exchange for net cash consideration of \$44.4 million. The Company leased such real property from Wilks Development, LLC in exchange for aggregate lease payments of \$51.6 million through March 2032 (the "FTSI Sale Leaseback").

### Report of independent registered public accounting firm

Board of Directors and Owners ProFrac Holding Corp.

# Opinion on the financial statements

We have audited the accompanying balance sheet of ProFrac Holding Corp. (the "Company") as of December 31, 2021, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2021, in conformity with accounting principles generally accepted in the United States of America.

### **Basis for opinion**

The financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audit included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audit provides a reasonable basis for our opinion.

### /s/ GRANT THORNTON LLP

We have served as the Company's auditor since 2021.

Dallas, Texas March 31, 2022

# **ProFrac Holding Corp.**

# **Balance sheet**

	mber 31, 2021
Assets	
Receivable from stockholder	\$ 10
Total assets	\$ 10
Stockholder's equity	
Common stock, \$0.01 par value; authorized 1,000 shares; 1,000 issued and outstanding	\$ 10
Total stockholder's equity	\$ 10

 $See\ accompanying\ notes\ to\ this\ balance\ sheet.$ 

# ProFrac Holding Corp. Notes to the balance sheet

# 1. Nature of operations

ProFrac Holding Corp. (the "Company") was formed on August 17, 2021, pursuant to the laws of the State of Delaware to become a holding company for ProFrac Holdings, LLC and certain other affiliates.

# 2. Summary of significant accounting policies

### Basis of presentation

This balance sheet has been prepared in accordance with accounting principles generally accepted in the United States of America. Separate statements of operations, statements of changes in stockholder's equity and statements of cash flows have not been presented because the Company has had no business transactions or activities to date.

### 3. Subsequent events

We have evaluated subsequent events subsequent to December 31, 2021 and through March 31, 2022, the date these financial statements were available to be issued and are not aware of any events that have occurred subsequent to December 31, 2021 that would require recognition or disclosure in these financial statements.

ProFrac Holding Corp. Consolidated balance sheets (Unaudited)

(In thousands)	June 30, 2022	December 31, 2021
Assets		
Current assets:		
Cash and cash equivalents	\$ 73,653	\$ 5,376
Accounts receivable, net	444,997	161,632
Accounts receivable—related party	3,637	4,515
Prepaid expenses, and other current assets	19,331	6,213
Assets held for sale	1,805	_
Inventories	192,377	73,942
Total current assets	735,800	251,678
Property, plant, and equipment	1,231,205	827,865
Accumulated depreciation and depletion	(566,960)	(464,178)
Property, plant, and equipment, net	664,245	363,687
Operating lease right-of-use assets	80,664	_
Deferred tax assets	3,316	_
Investments	49,752	4,244
Intangible assets, net	28,241	27,816
Goodwill	82,340	_
Other assets	19,267	17,145
Total assets	\$1,663,625	\$ 664,570

ProFrac Holding Corp. Consolidated balance sheets (Unaudited)

(In thousands)	June 30, 2022	December 31, 2021
Liabilities, redeemable noncontrollable interest, and stockholders' and members' (deficit) equity		
Current liabilities:		
Accounts payable	\$ 178,905	\$ 121,070
Accounts payable—related party	37,577	21,275
Current portion of operating lease liabilities	9,160	_
Accrued expenses	159,304	38,149
Other current liabilities	15,835	34,400
Current portion of long-term debt	51,329	31,793
Total current liabilities	452,110	246,687
Long-term debt	427,961	235,128
Long-term debt—related party	_	34,645
Operating lease liabilities	75,397	
Total liabilities	955,468	516,460
Commitments and contingencies (Note 17)		
Redeemable noncontrolling interest	2,024,687	_
Stockholders' and members' equity		
Members' equity	_	147,015
Preferred stock, \$0.01 par value, 50,000,000 authorized, none issued and outstanding		
Class A common stock, \$0.01 par value, 600,000,000 authorized, 41,237,003 issued and outstanding at June 30, 2022		
and none issued and outstanding at December 31, 2021	412	_
Class B common stock, \$0.01 par value, 400,000,000 authorized, 101,133,201 issued and outstanding at June 30, 2022		
and none issued and outstanding at December 31, 2021	1,011	_
Additional paid-in capital	_	_
Accumulated deficit	(1,410,780)	_
Accumulated other comprehensive (loss) income	(36)	56
Total stockholders' and members' equity attributable to ProFrac Holding Corp.	(1,409,393)	147,071
Noncontrolling interests	92,863	1,039
Total stockholders' and members' (deficit) equity	(1,316,530)	148,110
Total liabilities, redeemable noncontrollable interest, and stockholders' and members' (deficit) equity	\$ 1,663,625	\$ 664,570

 $\label{thm:companying} \textit{In accompanying notes are an integral part of these consolidated financial statements}$ 

ProFrac Holding Corp. Consolidated statements of operations (Unaudited)

	Three Mon June		Six Months Ended June 30,		
(In thousands)	2022	2021	2022	2021	
Revenues	\$589,844	\$174,819	\$934,824	\$324,405	
Operating costs and expenses:					
Cost of revenues, exclusive of depreciation, depletion and amortization	340,600	126,708	573,199	245,014	
Depreciation, depletion and amortization	64,064	34,904	108,280	70,365	
Loss on disposal of assets, net	2,143	1,868	1,989	4,075	
Selling, general, and administrative	87,548	14,094	121,675	27,872	
Total operating costs and expenses	494,355	177,574	805,143	347,326	
Operating income (loss)	95,489	(2,755)	129,681	(22,921)	
Other (expense) income:					
Interest expense, net	(13,451)	(6,187)	(22,723)	(12,222)	
Loss on extinguishment of debt	(8,822)	_	(17,095)	_	
Other income, net	989	53	9,220	240	
Income (loss) before income tax provision	74,205	(8,889)	99,083	(34,903)	
Income tax (provision) benefit	(4,112)	283	(4,864)	308	
Net income (loss)	70,093	(8,606)	94,219	(34,595)	
Less: net (income) loss attributable to ProFrac Predecessor	(56,157)	8,478	(79,867)	34,476	
Less: net loss attributable to noncontrolling interests	8,704	128	8,288	119	
Less: net income attributable to redeemable noncontrolling interests	(16,082)		(16,082)		
Net income attributable to ProFrac Holding Corp.	\$ 6,558	<u>\$</u> —	\$ 6,558	<u>\$</u>	
Basic and diluted earnings per Class A share	\$ 0.16		\$ 0.16		
Weighted average shares used in computing basic earnings per Class A share	40,123		40,123		
Weighted average shares used in computing diluted earnings per Class A share	40,145		40,145		

ProFrac Holding Corp. Consolidated statements of comprehensive income (loss) (Unaudited)

	Three Mont	ths Ended	Six Montl	hs Ended
	June	30,	June 30,	
(In thousands)	2022	2021	2022	2021
Net income (loss)	\$ 70,093	\$(8,606)	\$ 94,219	\$(34,595)
Foreign currency translation adjustments	13	4	(123)	10
Comprehensive income	70,106	(8,602)	94,096	(34,585)
Less: comprehensive income attributable to ProFrac Predecessor	(56,157)	8,474	(79,765)	34,466
Less: comprehensive income attributable to noncontrolling interest	8,701	128	8,319	119
Less: comprehensive income attributable to redeemable noncontrolling interest	(16,082)		(16,082)	
Comprehensive income attributable to ProFrac Holding Corp.	\$ 6,568	<u>\$</u>	\$ 6,568	<u> </u>

ProFrac Holding Corp. Consolidated statements of changes in equity (Unaudited)

			ss A on Stock	Class B C		Additional		Accumulated Other		Total
(In thousands)	Members' Equity	Shares	Amount	Shares		Paid in Capital	Retained Earnings	Comprehensive Income (Loss)	Noncontrolling Interests	Equity (Deficit)
Balance, January 1, 2022	\$ 147,015		<u>\$</u> —		<del>\$ —</del>	\$ —	\$ —	\$ 56	\$ 1,039	\$ 148,110
Net income	23,710								416	24,126
Member contributions	5,000	_	_	_	_	_	_	_	_	5,000
Deemed distribution	(3,664)	_	_	_	_	_	_	_	_	(3,664)
THRC related equity	72,931	_	_	_	_	_	_	_	_	72,931
Foreign currency translation								(102)	(34)	(136)
Balance, March 31, 2022	\$ 244,992		\$ —		\$ —	<u> </u>	<u> </u>	\$ (46)	\$ 1,421	\$ 246,367
Net income prior to corporate										
reorganization	56,157	_	_	_	_	_	_	_	(46)	56,111
Issuance of Class A shares in										
IPO	_	18,228	182	_	_	227,548	_	_	_	227,730
Issuance of Class B shares	_	_	_	101,133	1,011	_	_	_	_	1,011
Effect of corporate										
reorganization and										
reclassification to redeemable										
noncontrolling interest	(301,149)	20,895	209	_	_	(81,110)	_	_	_	(382,050)
Adjustment of redeemable										
noncontrolling interest to										
redemption amount at IPO						(146,438)	(1,291,910)			(1,438,348)
Class A shares issued to settle										4.6.60.6
asset purchase	_	2,114	21	_	_	16,675	_	_	_	16,696
Net income after corporate							6.550		(0.650)	(2.100)
reorganization							6,558	_	(8,658)	(2,100)
Recognition of Flotek									00.617	00.617
noncontrolling interest	_	_	_	_	_	260	_	_	99,617	99,617
Stock-based compensation		_		_		269			526	795
Stock-based compensation related to deemed contribution						11,252				11,252
Foreign currency translation	_	_	_	_	_	11,252	_	10		11,252
Additional paid-in capital								10	3	13
related to tax receivable										
agreement						609				609
Adjustment of redeemable	_	_	_	_	_	009	_	_	_	009
noncontrolling interest to										
redemption amount	_	_	_	_	_	(28,805)	(125,428)			(154,233)
*	<u> </u>	41,237	\$ 412	101,133	¢ 1 011	\$ —	\$(1,410,780)	\$ (36)	\$ 92,863	
Balance, June 30, 2022	Φ —	41,237	<b>9</b> 412	101,133	φ 1,011	<u> </u>	\$(1,410,780)	\$ (36)	\$ 92,863	<u>\$(1,316,530)</u>

ProFrac Holding Corp. Consolidated statements of changes in equity (Unaudited)

			Common		Common	Additional		Accumulated Other		Total
(In thousands)	Members' Equity	Shares	Amount	Shares	Amount	Paid in Capital	Retained Earnings	Comprehensive Income (Loss)	Noncontrolling Interests	Equity (Deficit)
Balance, January 1, 2021	<u>\$175,027</u>		<u>\$</u>		<u>\$</u>	<u>\$</u>	<u>\$</u>	<u>\$</u>	\$ 1,785	<u>\$176,812</u>
Net (loss) income	(25,998)	_	_	_	_	_	_	_	9	(25,989)
Foreign currency translation	_	_	_	_	_	_	_	6	_	6
Noncontrolling interest of acquired										
business									1,228	1,228
Balance, March 31, 2021	\$149,029		<u>\$</u>		<u>\$</u>	<u>\$</u>	<u>\$</u>	\$ 6	\$ 3,022	\$152,057
Net loss	(8,478)	_	_	_	_	_	_	_	(128)	(8,606)
Foreign currency translation								4		4
Balance, June 30, 2021	\$140,551		<u>\$</u>		<u>\$</u>	<u> </u>	<u>\$</u>	\$ 10	\$ 2,894	\$143,455

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these consolidated financial statements}$ 

ProFrac Holding Corp Consolidated statements of cash flow (Unaudited)

	Six Months E	Six Months Ended June 30,			
(In thousands)	2022	2021			
Cash flows from operating activities:	<u></u>				
Net income (loss)	\$ 94,219	\$ (34,595)			
Adjustments to reconcile net income (loss) to cash provided by operating activities:					
Depreciation, depletion and amortization	108,280	70,365			
Stock based compensation	40,304	_			
Loss on disposal of assets, net	1,989	4,075			
Non-cash loss on extinguishment of debt	10,230	_			
Amortization of debt issuance costs	2,729	1,063			
Bad debt expense, net of recoveries	5	83			
Deferred tax expense	1,024	_			
Unrealized gain on investments, net	(8,526)	_			
Changes in operating assets and liabilities:					
Accounts receivable	(174,371)	(25,317)			
Inventories	(63,881)	(7,423)			
Prepaid expenses and other assets	(7,108)	201			
Accounts payable	(12,750)	(286)			
Accrued expenses	82,629	18,614			
Deferred revenues and other liabilities	9,691				
Net cash provided by operating activities	84,464	26,780			
Cash flows from investing activities:	<del></del>				
Investment in property, plant & equipment	(116,069)	(53,607)			
Proceeds from sale of assets	46,101	17,586			
Acquisitions, net of cash acquired	(257,267)	(2,430)			
Investment in preferred shares of BPC	(47,202)				
Initial investment in Flotek	(10,000)	_			
Other investments	(3,893)	_			
Net cash used in investing activities	(388,330)	(38,451)			
Cash flows from financing activities:					
Proceeds from issuance of long-term debt	587,560	42,084			
Repayments of long-term debt	(497,825)	(18,856)			
Borrowings from revolving credit agreements	197,233	14,000			
Repayments to revolving credit agreements	(122,883)	(10,000)			
Payment of debt issuance costs	(23,584)	(1,127)			
Member contribution	5.000				
Proceeds from issuance of common stock	329,118	_			
Payment of THRC related equity	(72,931)	_			
Payment of common stock issuance costs	(27,444)	_			
Net cash provided by financing activities	374,244	26,101			
Net increase in cash, cash equivalents, and restricted cash	70,378	14.430			
, 1		2,952			
Cash, cash equivalents, and restricted cash beginning of period	5,376				
Cash, cash equivalents, and restricted cash end of period	<u>\$ 75,754</u>	\$ 17,382			
Supplemental cash flow information:					
Cash payments for interest	\$ 13,139	\$ 12,624			
Cash payments for taxes	\$ 118	\$ —			
Non-cash investing and financing activities:					
Capital expenditures included in accounts payable	\$ 27,377	\$ 7,837			
Operating lease liabilities incurred from obtaining right-of-use-assets	\$ 44,968	\$ —			

ProFrac Holding Corp Notes to the consolidated financial statements

(Amounts in thousands, except as noted) (unaudited)

### 1. Organization and description of business

ProFrac Holding Corp. ("ProFrac Corp.") and its consolidated subsidiaries, including ProFrac Holdings, LLC ("ProFrac LLC"), is a vertically integrated and innovation-driven energy services company providing hydraulic fracturing, completion services and other complementary products and services to leading upstream oil and gas companies engaged in the exploration and production ("E&P") of North American unconventional oil and natural gas resources. The Company operates in three business segments: stimulation services, manufacturing, and proppant production.

### **Company Formation**

ProFrac Corp. was incorporated as a Delaware corporation on August 17, 2021, to become a holding corporation for ProFrac LLC and its subsidiaries upon completion of a corporate reorganization in conjunction with a planned initial public offering ("IPO"). On May 12, 2022, ProFrac Corp. completed its IPO and corporate reorganization and became the managing member of ProFrac LLC.

The unaudited consolidated financial statements presented herein are those of ProFrac Corp. subsequent to the corporate reorganization on May 12, 2022, and ProFrac LLC before that date. In these notes to the consolidated financial statements, ProFrac Corp. and ProFrac LLC together are also referred to as "we," "us," "our," or the "Company" and ProFrac LLC is also referred to as "ProFrac Predecessor." For all periods presented, the unaudited consolidated financial statements presented herein include the controlled subsidiaries of ProFrac LLC, which include Best Pump & Flow LP ("Best Flow") and Alpine Silica, LLC ("Alpine").

Prior to December 21, 2021, Dan Wilks and Farris Wilks (or entities they control) (collectively, the "Wilks") held a controlling interest in each of ProFrac LLC, Best Flow and Alpine. Historical periods for ProFrac Predecessor had been presented on a consolidated and combined basis given the common control ownership by the Wilks. On December 21, 2021, all of the then-outstanding membership interests in Best Flow and Alpine were contributed to ProFrac LLC in exchange for membership interests in ProFrac LLC. Accordingly, the results for the three and six months ended June 30, 2021 have been retrospectively adjusted to present the operations of ProFrac LLC, Best Flow and Alpine on a combined basis. The acquisitions of Best Flow and Alpine have been accounted for in a manner consistent with the pooling of interest method of accounting, as the transaction was a combination of entities under common control. Under this method of accounting, the statements of operations, comprehensive income, equity and cash flows have been adjusted to include all activities of the commonly controlled groups for all periods in which common control existed.

# Initial Public Offering

On May 12, 2022, ProFrac Corp. completed its IPO of16,000,000 shares of its Class A common stock, par value \$0.01 per share (the "Class A Common Stock") at a public offering price of \$18.00 per share. On June 6, 2022, an over-allotment option was exercised resulting in an additional, 228,153 shares of Class A Common Stock being priced at \$18.00 per share. The IPO and exercise of the over-allotment option generated combined net proceeds of \$301.7 million, after deducting underwriter discounts and commissions and estimated offering costs. The Company used \$72.9 million of the net proceeds to redeem the membership ownership interests from the then-existing owners of THRC FTSI Related Equity (as described in Note 14 – Acquisitions and investments) and contributed the remaining proceeds to ProFrac Holdings, LLC. The Company used the remaining proceeds (i) to pay down \$143.8 million of the outstanding borrowings under the New Term Loan Credit Facility (as defined herein) (ii) to fully pay the \$22.0 million of the outstanding borrowings of the Backstop Note (as defined

herein) (iii) pay down \$22.0 million of the outstanding borrowings of the Closing Date Note (as defined herein) (iv) to pay down \$20.8 million of the outstanding borrowings of the Equify Bridge Note (as defined herein) and (v) with the remaining proceeds to be used for general corporate uses and additional repayment of debt.

### Redeemable Noncontrolling Interests

ProFrac Corp's only material asset is an equity interest consisting of units representing limited liability company interests in ProFrac LLC ("PFH Units"). As the sole managing member of ProFrac LLC, ProFrac Corp. operates and controls all of the business and affairs of ProFrac LLC and conducts its business through ProFrac LLC and its subsidiaries. As a result, ProFrac Corp. consolidates the financial results of ProFrac LLC and its subsidiaries and reports a noncontrolling interest related to the portion of PFH Units not owned by ProFrac Corp., which reduces net income attributable to holders of ProFrac Corp.'s Class A common stock. The holders of PFH Units not owned by ProFrac Corp. also hold shares of ProFrac Corp.'s Class B common stock, par value \$0.01 per share (the "Class B Common Stock") such that a single share of Class B Common Stock is issued for each PFH Unit not owned by ProFrac Corp.

The holders of PFH Units not owned by ProFrac Corp. may redeem all or a portion of their PFH Units, together with a corresponding number of shares of Class B Common Stock, for either shares of Class A Common Stock or an approximately equivalent amount of cash, at the election of the Company. In connection with the exercise of such redemption, a corresponding number of shares of Class B Common Stock will be cancelled. The redemption election is not considered to be within the control of the Company because the holders of Class B Common Stock and their affiliates control the Company through direct representation on ProFrac Corp.'s Board of Directors. As a result, we present the noncontrolling interests in ProFrac LLC as redeemable noncontrolling interests outside of permanent equity.

From January 1, 2022 through June 30, 2022, we recorded adjustments to the value of our redeemable noncontrolling interests as shown below:

(in thousands)	ble Noncontrolling Interests 2022
Balance as of January 1, 2022	\$ _
Effect of corporate reorganization and reclassification to	
redeemable noncontrolling interest	382,050
Adjustment of redeemable noncontrolling interest to	
redemption amount at IPO(1)	1,438,348
Class A common stock issued to settle asset purchase	21,361
Net income after corporate reorganization	16,082
Stock-based compensation	660
Stock-based compensation related to deemed contribution	27,597
Accrued distribution related to income taxes	(15,644)
Adjustment of redeemable noncontrolling interest to	
redemption amount <sup>(2)</sup>	154,233
Balance as of June 30, 2022	\$ 2,024,687

(1) Based on 101,133,201 shares of Class B Common Stock outstanding and the \$18.00 per share IPO price.

(2) Based on 101,133,201 shares of Class B Common Stock outstanding and the 10-day VWAP of Class A Common Stock of \$20.02 at June 30, 2022.

As of June 30, 2022, ProFrac Corp. owned29.0% of ProFrac LLC with the remaining 71.0% owned directly by other PFH Unit holders. As of June 30, 2022, ProFrac Corp. had outstanding 41.2 million shares of Class A Common Stock (representing approximately 29.0% of the total voting power) and 101.1 million shares of Class B Common Stock (representing approximately 71.0% of the total voting power).

### 2. Summary of significant accounting policies

### Basis of presentation

The unaudited consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP") for interim financial reporting and pursuant to the rules and regulations of the SEC. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. These consolidated financial statements should be read in conjunction with the audited consolidated financial statements and notes thereto for the year ended December 31, 2021 included within the Company's final prospectus filed with the SEC on May 16, 2022, pursuant to Rule 424(b) under the Securities Act. The consolidated financial statements include the accounts of the Company and all majority-owned subsidiaries. All significant intercompany accounts and transactions have been eliminated in consolidation. In our opinion, all adjustments considered necessary for a fair statement have been included in the accompanying unaudited financial statements. Operating results for the three and six-months ended June 30, 2022, are not necessarily indicative of the results that may be expected for the full year ending December 31, 2022. The December 31, 2021, balance sheet information has been derived from the 2021 audited financial statements of ProFrac Predecessor.

# Use of estimates

The preparation of financial statements in accordance with GAAP requires management to make estimates and assumptions that affect (1) the reported amounts of assets and liabilities and (2) the reported amounts of revenue and expenses during the reporting period. We base these estimates on historical results and various other assumptions believed to be reasonable, all of which form the basis for making estimates concerning the carrying values of assets and liabilities that are not readily available from other sources. It is at least reasonably possible these estimates could be revised in the near term, and these revisions could be material.

#### Leases

The Company determines if an arrangement is a lease at inception of the arrangement. To the extent that we determine an arrangement represents a lease, we classify that lease as an operating lease or a finance lease. We capitalize operating and finance leases on our consolidated balance sheets through a right-of-use ("ROU") asset and a corresponding lease liability. ROU assets represent our right to use an underlying asset for the lease term, and lease liabilities represent our obligation to make lease payments arising from the lease. See Note 12—Leases for additional information.

Operating and finance lease ROU assets and liabilities are recognized at the commencement date of the lease based on the present value of lease payments over the lease term. Lease expense for operating leases is recognized on a straight-line basis over the lease term.

# Revenue recognition

The Company's products and services are sold based upon contracts with customers. The Company recognizes revenue as it satisfies performance obligations by transferring control over a service or product to a customer.

Payment terms are specified in each customer agreement and are typically a specific number of days following satisfaction of the performance obligation. The following are descriptions of the principal activities of each reportable segment from which the Company generates its revenue.

Stimulation services. We generate revenue through the provision of hydraulic fracturing services, which involves the injection of water, sand and chemicals under high pressure into formations to optimize hydrocarbon flow paths during the completion phase of wellbores. Our contracts with customers are short term in nature, typically less than four weeks, and have a single performance obligation, which is the contracted total stages, satisfied over time. Once a stage has been completed, a field ticket is created which includes charges for services performed and any inputs consumed during the service. The signing of the field ticket by a customer representative represents their acceptance of the service and agreement to the amounts to which the Company has the right to invoice and recognize as revenue. We believe that recognizing revenue based on actual stages completed, upon receipt of a signed field ticket, appropriately depicts how our hydraulic fracturing services are transferred to our customers over time.

Manufacturing. We generate revenue through sales of equipment used to perform oilfield services. The performance obligation is satisfied and revenues are recognized at the point-in-time that control of goods are transferred to the customer, generally upon shipment from our manufacturing facility. Payment terms are specified in each customer agreement and are typically a specific number of days following satisfaction of the performance obligation.

Proppant production. We generate revenue through the sale of frac sand to oilfield service providers and E&P companies. The performance obligation is satisfied and revenue is recognized at the point-in-time that control of the product is transferred to the customer, generally upon shipment from our facility. We charge our customers on a per-ton basis at current market prices. Payment terms are specified in each customer agreement and are typically a specific number of days following satisfaction of the performance obligation.

Taxes collected from customers and remitted to governmental authorities are accounted for on a net basis and are therefore excluded from revenues in the consolidated statements of operations.

#### **Business Combinations**

Business combinations are accounted for under the acquisition method of accounting. Under this method, the assets acquired and liabilities assumed are recognized at their respective fair values as of the date of acquisition. The excess, if any, of the acquisition price over the fair values of the assets acquired and liabilities assumed is recorded as goodwill. For significant acquisitions, we utilize third-party appraisal firms to assist us in determining the fair values for certain assets acquired and liabilities assumed. The measurement of these fair values requires us to make significant estimates and assumptions which are inherently uncertain.

Adjustments to the fair values of assets acquired and liabilities assumed are made until we obtain all relevant information regarding the facts and circumstances that existed as of the acquisition date (the "measurement period"), not to exceed one year from the date of the acquisition. We recognize measurement-period adjustments in the period in which we determine the amounts, including the effect on earnings of any amounts we would have recorded in previous periods if the accounting had been completed at the acquisition date.

See Note 14—Acquisitions and investments for information on acquisitions completed during the historical period.

### Goodwill

We have acquired goodwill related to business acquisitions. Goodwill represents the excess of the purchase price over the fair value of net assets acquired. We review our goodwill on an annual basis, at the beginning of the

fourth quarter, and whenever events or changes in circumstances indicate the carrying value of goodwill may exceed its fair value. If the carrying value of goodwill exceeds its fair value, we recognize an impairment loss for this difference.

### Variable Interest Entities

We evaluate our ownership, contractual and other interest in entities to determine if they are variable interest entities ("VIE"). We evaluate whether we have a variable interest in those entities and the nature and extent of those interests. Based on our evaluation, if we determine we are the primary beneficiary of a VIE, we consolidate the entity in our financial statements.

### Fair value measurements

Fair value is defined as the price that would be received from selling an asset or paid to transfer a liability in an orderly transaction between market participants at a measurement date. We apply the following fair value hierarchy, which prioritizes the inputs used to measure fair value into three levels and bases of categorization within the hierarchy upon the lowest level input that is available and significant to the fair value measurement:

- Level 1: The use of quoted prices in active markets for identical assets or liabilities.
- Level 2: Other than quoted prices included in Level 1, inputs that are observable for the asset or liability, either directly or indirectly, for substantially the full term of the asset or liability. At June 30, 2022, we had no Level 2 measurements.
- Level 3: The use of significant unobservable inputs that typically require the use of management's estimates of assumptions that market participants would use in pricing. See Note 16—Fair Value of Financial Interests for more information on our investments using Level 3 measurements.

Our current assets and liabilities contain financial instruments, the most significant of which are trade accounts receivable and payable. We believe the carrying value of our current assets and liabilities approximate fair value. Our fair value assessment incorporates a variety of considerations, including: (i) the short-term duration of the instruments and (ii) our historical incurrence of and expectations of future bad debt expense. The book value of our floating rate debt approximates fair value because of its floating rate structure.

#### Income taxes

Before May 12, 2022, the ProFrac Predecessor entities were organized as limited liability companies or a limited partnership and were treated as either a disregarded entity or a partnership for U.S. federal income tax purposes, whereby the ordinary business income or loss and certain deductions were passed-through and reported on the members' income tax returns. As such, the Company was not required to account for U.S. federal income taxes in the consolidated financial statements. Certain state income-based taxes are imposed on the Company which are reflected as income tax expense or benefit in historical periods.

In connection with the IPO in May 2022, the Company reorganized and ProFrac LLC became partially owned by ProFrac Corp., a U.S. Internal Revenue Code Subchapter C corporation ("C-Corporation"). ProFrac Corp. is a taxable entity and is required to account for income taxes under the asset and liability method for periods subsequent to May 12, 2022.

Under the asset and liability method, deferred tax assets and liabilities are recognized for the estimated future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax basis. Deferred tax assets and liabilities are measured using enacted tax rates in effect for the year in which those temporary differences are expected to be recovered or settled pursuant to the provisions of Accounting Standards Codification ("ASC") 740, Income Taxes. The effect on deferred tax assets

and liabilities of a change in tax rate is recognized in earnings in the period that includes the enactment date. Valuation allowances are established when necessary to reduce deferred tax assets to the amounts more-likely-than-not to be realized.

# Recently adopted accounting standards

On January 1, 2022, we adopted the Financial Accounting Standards Board ("FASB") Accounting Standards Update ("ASU") for "Leases," which amended existing guidance to require lessees to recognize liabilities and ROU assets on the balance sheet for the rights and obligations created by long-term leases and to disclose additional quantitative and qualitative information about leasing arrangements. We adopted this guidance using a modified retrospective approach on January 1, 2022 using the transition method that allows a cumulative-effect adjustment to the opening balance of retained earnings in the period of adoption.

We have completed our process to implement this standard, and we have designed processes and internal controls necessary for adoption of this standard. We have made policy elections to (i) not capitalize short-term leases for all asset classes, (ii) not separate non-lease components from lease components for all of our existing asset classes, (iii) apply the package of practical expedients that allows us to not reassess: whether any expired or existing contracts contain leases, lease classification for any expired or existing leases and initial direct costs for existing leases and (iv) apply the practical expedient to apply hindsight in estimating lease term and impairment.

The impact of applying this standard is not expected to significantly impact our results of operations or cash flows. As of January 1, 2022, we recognized ROU assets and liabilities of approximately \$35.8 million from operating leases on our consolidated balance sheet. See Note 12—Leases for additional disclosures related to our adoption this accounting standards update.

### New accounting standards to be adopted

We have not yet implemented FASB ASU No. 2016-13, Financial Instruments – Credit Losses. The ASU introduces a new accounting model, the Current Expected Credit Losses model ("CECL"), which requires recognition of credit losses and additional disclosures related to credit risk. The CECL model utilizes a lifetime expected credit loss measurement objective for the recognition of credit losses for loans and other receivables at the time the financial asset is originated or acquired. The expected credit losses are adjusted each period for changes in expected lifetime credit losses. This model replaces the multiple existing impairment models previously used under GAAP, which generally require that a loss be incurred before it is recognized. The new standard also applies to financial assets arising from revenue transactions such as contract assets and accounts receivable.

Implementation is currently required for fiscal years beginning after December 15, 2022. The Company does not believe implementation will have a material impact on its financial statements.

We have not yet implemented FASB ASU No. 2019-12, Income Taxes (Topic 740)—Simplifying the Accounting for Income Taxes, which removes specific exceptions to the general principles in Topic 740 under GAAP. The new guidance also improves the application of income tax-related guidance and simplifies GAAP for franchise taxes that are partially based on income, transactions with a government that result in a step up in the tax basis of goodwill, separate financial statements of legal entities that are not subject to tax, and enacted changes in tax laws in interim periods. The new guidance will be effective for fiscal years beginning after December 15, 2021, and interim periods within fiscal years beginning after December 15, 2022; however, early adoption is permitted. The Company is currently evaluating the impact of adopting the new guidance on its consolidated financial statements.

# 3. Restricted cash

The following table provides a reconciliation of cash, cash equivalents, and restricted cash reported within the consolidated statement of cash flows as of June 30, 2022, and December 31, 2021:

(In thousands)	June 30, 2022	mber 31, 2021
Cash and cash equivalents	\$73,653	\$ 5,376
Restricted cash included in prepaid expenses and other current assets	2,101	 
Total cash, cash equivalents, and restricted cash shown in the consolidated		
statements of cash flows	<u>\$75,754</u>	\$ 5,376

As of June 30, 2022, restricted cash included cash used as collateral for our credit card program.

# 4. Supplemental balance sheet information

# Inventories

The following table summarizes the components of our inventories as of June 30, 2022, and December 31, 2021:

(In thousands)	June 30, 2022	Dec	ember 31, 2021
Raw materials and supplies	\$ 82,865	\$	13,911
Work in process	8,769		3,288
Finished products and parts	_100,743		56,743
Total	\$192,377	\$	73,942

The increase in inventory from December 31, 2021, to June 30, 2022 was partially due to acquired inventory of \$4.5 million related to our acquisitions of FTS International, Inc. ("FTSI") and Flotek Industries, Inc. ("Flotek"). The remaining increase is related to our increased activity levels in 2022.

### Accrued Expenses

The following table summarizes our accrued expenses as of June 30, 2022, and December 31, 2021:

(In thousands)	June 30, 2022	December 31, 2021
Accrual for purchased materials	\$ 64,617	\$ 15,600
Employee compensation and benefits	27,134	8,107
Sales, use, and property taxes	12,007	5,974
Interest	9,058	879
Income taxes	3,600	_
Tax receivable agreement	3,448	_
Tax distribution to redeemable noncontrolling interests	15,644	_
Other	23,796	7,589
Total	<u>\$159,304</u>	\$ 38,149

# 5. Property, plant, and equipment

The following table summarizes the components of our property, plant, and equipment, net as of June 30, 2022, and December 31, 2021:

	June 30,	December 31,
(In thousands)	2022	2021
Machinery and equipment	\$1,070,672	\$ 760,829
Mining property and mine development	48,464	34,809
Office equipment, software and other	14,303	5,550
Land	950	_
Buildings and leasehold improvements	24,587	15,947
Total	1,158,976	817,135
Less: accumulated depreciation and depletion	(566,960)	(464,178)
Construction in progress	72,229	10,730
Property, plant, and equipment, net	\$ 664,245	\$ 363,687

The increase in net property, plant, and equipment from December 31, 2021, to June 30, 2022, was due to acquired assets of \$28.7 million related to our FTSI and Flotek acquisitions. This increase was partially offset by \$48.2 million of FTSI assets sold in the sale lease-back transaction immediately following the FTSI acquisition.

Depreciation expense for the three months ended June 30, 2022 and 2021, was \$63.6 million and \$34.7 million, respectively. Depreciation expense for the six months ended June 30, 2022 and 2021, was \$107.8 million and \$70.1 million, respectively.

Major classifications of property, plant, and equipment and their respective useful lives are as follows:

Machinery and equipment	2 years—10 years
Office equipment, software, and other	3 years—7 years
Buildings and leasehold improvements	2 years—40 years

# 6. Intangible assets

The following table summarizes the components of our finite-lived intangible assets as of June 30, 2022, and December 31, 2021:

		June 30, 2022			December 31, 2021			
	Gross	Gross Less:		Gross	Less:	Net		
	Book	Accumulated	Net Book	Book	Accumulated	Book		
(In thousands)	Value	Amortization	Value	Value	Amortization	Value		
Electric frac licenses	\$22,500	\$ —	\$22,500	\$22,500	\$ —	\$22,500		
Acquired technology	7,144	(1,403)	5,741	5,905	(589)	5,316		
Intangible assets, net	\$29,644	\$ (1,403)	\$28,241	\$28,405	\$ (589)	\$27,816		

Intangible assets are amortized over the period the Company expects the asset to generate cash flows. As such, we amortize each electric frac license through the remaining license period, beginning when the initial fleet built under each license is placed into service, a period we estimate to be 17 years. For technology acquired during 2021, we estimated this period to be seven years. For technology acquired in 2022 related to the acquisition of FTSI, we estimated the period to be three years. Amortization expense related to intangible assets was \$0.4 million and \$0.2 million for the three months ended June 30, 2022 and 2021, respectively. Amortization expense related to intangible assets was \$0.8 million and \$0.3 million for the six months ended June 30, 2022 and 2021, respectively.

#### 7. Indebtedness

The following table summarizes the components of our debt as of June 30, 2022 and December 31, 2021:

(In thousands)	June 30, 2022	December 31, 2021	
Old ABL Credit Facility	\$ —	\$ 69,000	
Old Term Loan	_	171,355	
First Financial Loan	23,950	30,000	
New ABL Credit Facility	143,350	_	
New Term Loan Credit Facility	302,380	_	
Best Flow Credit Facility <sup>(1)</sup>	_	7,101	
Best Flow Note <sup>(1)</sup>	_	10,827	
Alpine Promissory Note(1)	_	16,717	
Flotek Convertible Notes	12,739	_	
Other	12,626	1,695	
Total gross debt	495,045	306,695	
Less: unamortized debt issuance costs	(15,755)	(5,129)	
Less: current portion of long-term debt	(51,329)	(31,793)	
Total long-term debt	\$427,961	\$ 269,773	

# (1) Related party debt agreements

# Old ABL Credit Facility

On March 14, 2018, ProFrac LLC entered into a senior secured asset-based revolving credit agreement (the "Old ABL Credit Facility"), with Barclays Bank PLC, as administrative agent and collateral agent, and the lenders party thereto. The Old ABL Credit Facility had a maturity date of March 14, 2023.

LIBOR borrowings under the Old ABL Credit Facility bore interest at the greater of LIBOR or 0.00%, plus a margin of 1.50% to 2.00%, depending on facility utilization. The interest rate was 2.75% as of December 31, 2021.

As of December 31, 2021, the borrowing base was \$146.2 million, therefore maximum availability under the Old ABL Credit Facility was the facility maximum of \$105.0 million. There were \$69.0 million of borrowings outstanding and \$3.1 million of letters of credit outstanding, resulting in \$32.9 million of availability under the Old Credit Facility ABL.

On March 4, 2022, the Old ABL Credit Facility was refinanced and replaced with the New ABL Credit Facility (as described below).

### Old Term Loan

On September 7, 2018, ProFrac LLC entered into a \$180.0 million term loan agreement ("Old Term Loan"), which matures on September 15, 2023, with a group of lenders with Barclays Bank, PLC as administrative agent. Principal payments are due in quarterly installments, however due to prepayments made to date, there will be no required minimum amortization (other than excess cash flow prepayments) before termination of the Old Term Loan. Excess cash flow is calculated quarterly to determine any additional minimum prepayments. On June 24, 2021, ProFrac LLC and its Old Term Loan lenders reached an agreement to expand the facility by \$40.0 million.

The Old Term Loan, as amended, requires minimum excess cash flow prepayments as follows, each due approximately 55 days afterperiod-end: \$0.0 million for the fiscal quarters ended March 31, 2021 through and including December 31, 2021, and \$5.0 million for the fiscal quarters ended March 31, 2022 through and including June 30, 2023.

LIBOR borrowings under the Old Term Loan bear interest at the greater of LIBOR or 1.25%, plus a margin of 6.25% to 8.50%, depending on the total net leverage ratio as defined under the Old Term Loan. The interest rate was 9.75% as of December 31, 2021.

The Old Term Loan contained certain restrictive covenants, including a financial covenant which required ProFrac LLC to maintain a total net leverage ratio, as defined in the credit agreement, of no greater than 2.25:1.00 for the fiscal quarters ended September 30, 2019 through and including March 31, 2020, 3.50:1.00 for the fiscal quarter ended June 30, 2020 through and including March 31, 2021,3.00:1.00 for the fiscal quarter ended June 31, 2021, 2.75:1.00 for the fiscal quarter ended September 30, 2021,2.50:1.00 for the fiscal quarter ended December 31, 2021, and 2.00:1.00 for the fiscal quarter ended March 31, 2022 and thereafter. ProFrac LLC was in compliance with all required covenants as of December 31, 2021.

In February 2022, ProFrac and its Old Term Loan lenders entered into an agreement to amend the Old Term Loan. The amendment expanded the facility by \$48.0 million.

On March 4, 2022, the Old Term Loan was extinguished resulting in loss on debt extinguishment of \$.9 million.

#### First Financial Loan

On July 22, 2020, ProFrac LLC entered into a \$35.0 million loan agreement with First Financial Bank, N.A. which had a maturity of July 22, 2025 ("Main Street Loan"). In August 2021, the loan agreement was amended to remove the covenants in place prior to the amendment. As amended, the Main Street Loan contained certain restrictive covenants which required ProFrac LLC to maintain a Fixed Charge Coverage Ratio of at least 1.00:1.00, and a Maximum Leverage Ratio of 3.50:1.00. Additionally, the Main Street Loan restricted the payment of distributions or dividends, other than for the payment of taxes.

On December 22, 2021, the Main Street Loan had a balance of \$32.2 million. The Main Street Loan was extinguished with a cash payment of \$2.2 million and the remainder refinanced with a \$30.0 million loan with First Financial Bank, N.A. ("First Financial Loan").

The First Financial Loan has a maturity date of January 1, 2024 with an interest rate of LIBOR plus 3.5%, and the loan is to be repaid by equal payments of principal and interest beginning in February 2022. The First Financial Loan contains certain restricted covenants which require the Company to maintain a fixed charge ratio of at least 1.00:1.00 and a maximum net leverage ratio of 3.00:1.00. The Company was in compliance with all covenants as of June 30, 2022.

# New ABL Credit Facility

On March 4, 2022, ProFrac LLC, ProFrac Holdings II, LLC ("ProFrac II LLC"), as borrower (in such capacity, the "ABL Borrower"), and certain of the ABL Borrower's wholly owned subsidiaries as obligors, entered into a senior secured asset-based revolving credit agreement (as amended, the "New ABL Credit Facility"), with a group of lenders with JPMorgan Chase Bank N.A., as administrative agent and collateral agent. The New ABL Credit Facility initially provided for an asset-based revolving credit facility with a borrowing base and lender commitments of \$100.0 million. On April 8, 2022, the New ABL Credit Facility was amended to increase the borrowing base and lender commitments to \$200.0 million. The New ABL Credit Facility has a borrowing base composed of certain eligible accounts receivable and eligible inventory less customary reserves, as redetermined monthly. As of June 30, 2022, the maximum availability under the New ABL Credit Facility was the aggregate lender commitments of \$200.0 million with \$143.4 million of borrowings outstanding and \$9.2 million of letters of credit outstanding, resulting in approximately \$47.4 million of remaining availability. The New ABL Credit Facility matures on the earlier of (i) March 4, 2027 and (ii) 91 days prior to the stated maturity of any material indebtedness (other than the First Financial Loan).

Borrowings under the New ABL Credit Facility accrue interest based on a three-tier pricing grid tied to average historical availability, and the ABL Borrower may elect for loans to be based on either an Adjusted Term SOFR or a base rate, plus the applicable margin. The interest rate under the New ABL Credit Facility for (a) Adjusted Term SOFR is the applicable margin plus the fluctuating per annum rate equal to Adjusted Term SOFR (with an Adjusted Term SOFR Floor of 0.00%); and (b) Base Rate Loans are the applicable margin plus the fluctuating per annum rate equal to the greatest of the Prime Rate in effect on such day, or the NYFRB Rate in effect on such day plus 1/2% of 1% and the Adjusted Term SOFR for a one-month Interest Period as published two (2) U.S. Government Securities Business Days prior to such day (or if such day is not a Business Day, the immediately preceding Business Day), plus 1.0%. The applicable margin for Adjusted Term SOFR Loans ranges from 1.50% to 2.00% and for Base Rate Loans ranges from 0.50% to 1.00%, depending on the average daily availability over the last three months under the New ABL Credit Facility. The New ABL Credit Facility bears an unused line fee ranging from 0.250% to 0.375%, depending on the average daily availability over the last three months payable quarterly in arrears. The New ABL Credit Facility also bears customary letter of credit fees. The interest rate was 5.50% as of June 30, 2022.

The New ABL Credit Facility is subject to customary mandatory prepayments, including a mandatory prepayment if the aggregate unpaid principal balance of revolving loans, agent advances, swingline borrowings, unreimbursed drawings under letters of credit and the undrawn amount of outstanding letters of credit exceeds at any time the lesser of (x) the then applicable borrowing base and (y) the then total effective commitments under the New ABL Credit Facility, in an amount equal to such excess. After the occurrence and during the continuance of a Cash Dominion Period (defined in the New ABL Credit Facility as (a) any period commencing upon the date that availability shall have been less than the greater of (i) 12.5% of the Maximum Credit (which is the lesser of the maximum revolver amount in effect at such time and the borrowing base at such time) and (ii) \$10.0 million for a period of five consecutive business days and continuing until the date on which availability shall have been at least the greater of (y) 12.5% of the Maximum Credit and (z) \$10.0 million for 20 consecutive calendar days or (b) any period commencing on the occurrence of certain specified events of default, and continuing during the period that such specified event of default shall be continuing).

The New ABL Credit Facility contains certain customary representations and warranties and affirmative and negative covenants. The negative covenants include, subject to customary exceptions, limitations on indebtedness, dividends, distributions and certain other payments, investments, acquisitions, prepayments of specified junior indebtedness, amendments of specified junior indebtedness, transactions with affiliates, dispositions, mergers and consolidations, liens, restrictive agreements, sale and leaseback transactions, changes in fiscal periods and changes in line of business.

We are required by the New ABL Credit Facility to maintain minimum liquidity of \$5.0 million at all times. Additionally, when availability is less than the greater of (i) 12.5% of the maximum credit (which is the lesser of the maximum revolver amount in effect at such time and the borrowing base at such time) and (ii) \$10.0 million and continuing until such time as availability has been in excess of such threshold for a period of 20 consecutive calendar days, we are required by the New ABL Credit Facility to maintain a springing Fixed Charge Coverage Ratio (as defined in the New ABL Credit Facility) of at least 1.0 to 1.0, which is tested quarterly during such period. The Company was in compliance with all covenants, and there were no existing defaults or events of default related to the New ABL Credit Facility as of June 30, 2022.

On July 25, 2022, the New ABL Credit Facility was amended to add an uncommitted \$100.0 million incremental facility (the "Incremental Facility"), under the terms of which existing lenders can make additional loans (in their sole discretion) under, or new lenders can join, the Incremental Facility and increase the potential size of the New ABL Credit Facility from \$200 million to \$300 million, subject to satisfaction of certain conditions. All other terms and conditions of the New ABL Credit Facility remained substantially unchanged. See Note 19—Subsequent Events.

#### New Term Loan Credit Facility

On March 4, 2022, ProFrac LLC, ProFrac II, LLC, as borrower (in such capacity, the "Term Loan Borrower"), and certain of the Term Loan Borrower's wholly owned subsidiaries as obligors, entered into a senior secured term loan credit agreement (the "New Term Loan Credit Facility"), with Piper Sandler Finance LLC, as administrative agent and collateral agent, and the lenders party thereto. The New Term Loan Credit Facility provides for a term loan facility in an aggregate principal amount of \$450.0 million. In the second quarter of 2022, the New Term Loan Facility was paid down by \$143.8 million with net proceeds from the IPO and an additional \$3.8 million in scheduled payments. As of June 30, 2022, the Term Loan Borrower had approximately \$302.4 million outstanding under the New Term Loan Credit Facility. The New Term Loan Credit Facility matures on March 4, 2025.

Borrowings under the New Term Loan Credit Facility accrue interest at a percentage per annum equal to (a) until October 1, 2022, (i) for SOFR Rate Loans, 8.50%, and (ii) for Base Rate Loans, 7.50% and (b) thereafter, based on a three-tier pricing grid tied to Total Net Leverage Ratio (as defined in the New Term Loan Credit Facility), and the Term Loan Borrower may elect for loans to be based on either Adjusted Term SOFR or Base Rate, plus the applicable margin. The interest rate on the New Term Loan Credit Facility for (a) SOFR Rate Loans are the applicable margin plus the fluctuating per annum rate equal to Adjusted Term SOFR (as defined in the New Term Loan Credit Facility), with a SOFR floor of 1.00% and (b) Base Rate Loans are the applicable margin plus the fluctuating per annum rate equal to the highest of (i) the federal funds rate plus 1/2 of 1%, (ii) the interest rate quoted in the print edition of *The Wall Street Journal*, Money Rates Section, as the prime rate in effect, (iii) Adjusted Term SOFR for aone-month interest period as determined on such day, plus 1.0% and (iy) 2.00%.

The applicable margin for (a) SOFR Rate Loans ranges from 6.50% to 8.00% and (b) Base Rate Loans ranges from 5.50% to 7.00%, depending on the Total Net Leverage Ratio (as defined in the New Term Loan Credit Facility) as of the first day of the then-current fiscal quarter. The interest rate was 10.0% as of June 30, 2022.

The New Term Loan Credit Facility is guaranteed by ProFrac LLC and all of the Term Loan Borrower's material existing subsidiaries and certain direct and indirect future U.S. restricted subsidiaries of the Term Loan Borrower. The New Term Loan Credit Facility is secured by a lien on, and security interest in, substantially all of each such guarantor's assets.

The New Term Loan Credit Facility is subject to quarterly amortization beginning in June 2022, though any excess cash flow payments, reduce the required amortization. Additionally, the New Term Loan Credit Facility is subject to a quarterly mandatory prepayment beginning for the calendar quarter ending on December 31, 2022, in an amount equal to the Applicable ECF Percentage (as defined in the New Term Loan Credit Facility). The Applicable ECF Percentage ranges from 50% of Excess Cash Flow (as defined in the New Term Loan Credit Facility) to 25% of Excess Cash Flow depending on the Total Net Leverage Ratio as of the last day of the applicable fiscal quarter.

Voluntary prepayments of borrowings under the New Term Loan Credit Facility are permitted at any time, in specified minimum principal amounts, subject to reimbursement of the lenders' redeployment costs actually incurred in the case of a prepayment of SOFR Rate Loans other than on the last day of the relevant interest period. Between March 4, 2022 and March 4, 2023, certain prepayments of the New Term Loan Credit Facility are subject to a prepayment premium of 3.00% (or, in the case of any IPO Prepayment (as defined in the New Term Loan Credit Facility).2.00%). Between March 5, 2023 and March 4, 2024, certain prepayments of the New Term Loan Credit Facility are subject to a 2.00% prepayment premium. After March 4, 2024, but prior to the Stated Termination Date (as defined in the New Term Loan Credit Facility) certain prepayments of the New Term Loan Credit Facility are subject to a 1.00% prepayment premium. No payment or prepayment premium shall be due on account of any payments or prepayments made on the Stated Termination Date.

The New Term Loan Credit Facility contains certain customary representations and warranties and affirmative and negative covenants. The negative covenants include, subject to customary exceptions, limitations on

indebtedness, dividends, distributions and certain other payments, investments, capital expenditures, acquisitions, prepayments of specified junior indebtedness, amendments of specified junior indebtedness, transactions with affiliates, dispositions, mergers and consolidations, liens, restrictive agreements, changes in fiscal periods and changes in line of business.

We are required by the New Term Loan Credit Facility to maintain a Total Net Leverage Ratio (as defined in the New Term Loan Credit Facility) (i) of no more than 2.00 to 1.00 for the fiscal quarter ending on June 30, 2022, (ii) of no more than 1.55 to 1.00 for the fiscal quarters ending on September 30, 2022 and December 31, 2022, and (iii) of no more than 1.25 for each fiscal quarter ending on March 31, 2023 and thereafter.

We are required by the New Term Loan Credit Facility to maintain minimum liquidity of \$0.0 million at all times.

The New Term Loan Credit Facility contains customary events of default. If an event of default occurs and is continuing, the lenders may declare all loans to be immediately due and payable. Some events of default require an automatic termination of the loans and become immediately due and payable.

The Company was in compliance with all covenants, and there were no defaults or events of default related to the New Term Loan Credit Facility, as of June 30, 2022.

On July 25, 2022, the New Term Loan Facility was amended to increase the size of the New Term Loan Facility by \$150.0 million with an uncommitted option to obtain commitments for a potential additional \$100.0 million of delayed draw loans before the earlier to occur of (i) the consummation of the Merger (as defined herein) and (ii) March 31, 2023. The maturity date of the New Term Loan Facility remains the same at March 4, 2025. The Company used a portion of the proceeds from the increased New Term Loan Facility to fund the acquisition of Monahans (as defined herein) and anticipates using the remainder of the proceeds and operating cash to facilitate the Merger, pay outstanding debt under the New ABL Credit Facility and/or for other general corporate purposes. See Note 19—Subsequent Events.

# **Backstop Note**

On March 4, 2022, ProFrac LLC borrowed \$22.0 million pursuant to a subordinated promissory note with THRC Holdings, LP ("THRC Holdings") with a stated maturity date of March 4, 2027 (the "Backstop Note"). The Backstop Note bears interest at a percentage per annum equal tol.74%. Interest under the Backstop Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Backstop Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash.

The Backstop Note is unsecured and subordinated to the indebtedness owing under the New ABL Credit Facility and the New Term Loan Credit Facility.

In June 2022, the Backstop Note was fully paid with net proceeds from the IPO.

#### Closing Date Note

On March 4, 2022, ProFrac LLC borrowed \$22.0 million pursuant to a subordinated promissory note with THRC Holdings with a stated maturity date of March 4, 2027 (the "Closing Date Note"). The Closing Date Note bears interest at a percentage per annum equal tol.74%. Interest under the Closing Date Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Closing Date Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash.

The Closing Date Note is unsecured and subordinated to the indebtedness owing under the New ABL Credit Facility and the New Term Loan Credit Facility.

In June 2022, the Closing Date Note was fully paid with net proceeds from the IPO.

### Equify Bridge Note

On March 4, 2022, ProFrac II LLC entered into a \$45.8 million subordinated promissory note with Equify Financial with a stated maturity date ofMarch 4, 2027 (the "Equify Bridge Note"). The Equify Bridge Note bears interest at a percentage per annum equal td .0%. Interest under the Equify Bridge Note is paid on a quarterly basis and is solely payable in kind, with such interest amounts being added to the outstanding principal amount of the Equify Bridge Note, until the date that both the New ABL Credit Facility and the New Term Loan Credit Facility shall have been terminated, after which date quarterly interest payments may be paid in kind or in cash. In April 2022, the Company repaid \$25.0 million in principal under the Equify Bridge Note.

The Equify Bridge Note is unsecured and subordinated to the indebtedness owing under the New ABL Credit Facility and the New Term Loan Credit Facility.

In April 2022, the Company repaid \$25.0 million in principal under the Equify Bridge Note, resulting in an outstanding balance of \$20.8 million. In June 2022, the Equify Bridge Note was fully paid with net proceeds from the IPO.

### Best Flow Credit Facility

On February 4, 2019, Best Flow entered into a revolving loan credit agreement (the "Best Flow Credit Facility"), with Equify Financial, LLC ("Equify Financial") as lender. Equify Financial is a related party. The Best Flow Credit Facility provided for a revolving credit facility in an aggregate principal amount at any time outstanding up to \$9.0 million, subject to borrowing base availability. The Best Flow Credit Facility had a maturity date of February 4, 2026.

The interest rate under the Best Flow Credit Facility was the lesser of (i) the Prime Rate (as defined in the Best Flow Credit Facility) plus the applicable margin (3.50%) and (ii) the Maximum Rate (as defined in the Best Flow Credit Facility). All accrued but unpaid interest on the outstanding principal balance is due and payable monthly on the first day of each calendar month. The Best Flow Credit Facility was secured by a first lien on substantially all of the assets of Best Flow.

On March 4, 2022, the Best Flow Credit Facility was extinguished resulting in loss of extinguishment of debt of \$3.3 million.

### Best Flow Note

On January 28, 2021, Best Flow issued a promissory note (the "Best Flow Note"), with Equify Financial, as holder. Equify Financial is a related party. The Best Flow Note provided for a term loan in an initial aggregate principal amount equal to \$13.0 million. Proceeds from the Best Flow Note were utilized to pay down \$7.6 million of outstanding balances on the Best Flow Credit Facility and to pay down other equipment financing agreements for \$5.4 million. The Best Flow Note matures on February 1, 2026, with a fixed interest rate of 8.0%. The principal and interest are paid in equalmonthly amortizing amounts through maturity. Prepaid amounts are subject to a 0.19% prepayment premium.

On March 4, 2022, the Best Flow Note was extinguished resulting in loss of extinguishment of debt of \$.4 million.

#### Alpine Promissory Note

In January 2021, Alpine entered into a \$\infty 1.4 million promissory note with Equify Financial ("Alpine Note"). Equify Financial is a related party. The Alpine Note amortizes monthly, had an interest rate of 8.0% and had a stated maturity date in February 2027.

On March 4, 2022, the Alpine Note was extinguished resulting in loss on extinguishment of debt of \$ .8 million.

### Flotek Convertible Notes

On February 2, 2022, Flotek entered into a private investment in public equity transaction (the "PIPE Transaction") with a consortium of investors to secure growth capital. Pursuant to the PIPE Transaction, Flotek issued \$11.2 million in aggregate initial principal amount of convertible notes payable ("Flotek Convertible Notes"). The Flotek Convertible Notes accrue paid-in-kind interest at a rate of 10% per annum, have a maturity of one year, and are convertible into common stock of Flotek (a) at Flotek's option if Flotek's common stock equals or exceeds \$2.50 for 20 trading days during a 30 consecutive trading day period, (b) at the holder's option at any time prior to maturity, at a price of \$1.088125 per share or (c) at maturity at a price of \$1.088125 per share or (c) at maturity at a price of \$1.088125 per share or (c) at maturity at a price of \$1.088125 per share or (c) at maturity at a price of \$1.088125 per share or (c) at maturity at a price o

At June 30, 2022, there was \$8.2 million principal amount of Flotek Convertible Notes outstanding, which were included in the Company's consolidated financial statements at a carrying value of \$12.3 million.

#### Other indebtedness

As of June 30, 2022 and December 31, 2021, the Company had other debt agreements outstanding with unpaid principal balances of \$2.6 million and \$1.7 million, respectively. At June 30, 2022, other indebtedness included various equipment financing agreements of \$6.8 million. In July 2022, we repaid \$3.4 million under these agreements and the remaining \$3.4 million is due October 1, 2022 at an interest rate of 5.0% At June 30, 2022, other indebtedness also included a \$4.8 million loan under the Paycheck Protection Program established under the Coronavirus Aid, Relief, and Economic Securities Act held by Flotek.

The following table summarizes the principal maturity schedule for our long-term debt outstanding as of June 30, 2022:

	2022	2023	2024	2025	2026	Thereafter	Total
First Financial loan	\$ 8,659	\$ 15,291	\$ —	\$ —	\$	\$ —	\$ 23,950
New ABL Credit Facility	_	_	_	_	_	143,350	143,350
New Term Loan Credit Facility	7,560	15,120	15,120	264,580	_	_	302,380
Flotek Convertible Notes	_	12,739	_	_	_	_	12,739
Other.	7,803	1,908	1,907	543	79	386	12,626
Total	\$ 24,022	\$45,058	\$17,027	\$ 265,123	\$ 79	\$ 143,736	\$ 495,045

# 8. Income taxes

Before May 12, 2022, the ProFrac Predecessor entities were organized as limited liability companies or a limited partnership and were treated as either a disregarded entity or a partnership for U.S. federal income tax purposes, whereby the ordinary business income or loss and certain deductions were passed-through and reported on the members' income tax returns. As such, the Company was not required to account for U.S. federal income taxes in the consolidated financial statements. Certain state income-based taxes are imposed on the Company which are reflected as income tax expense or benefit in historical periods. In connection with the IPO in May 2022, the

Company reorganized and ProFrac LLC became partially owned by ProFrac Corp., aC-Corporation. ProFrac Corp. is a taxable entity and is required to account for income taxes under the asset and liability method for periods subsequent to May 12, 2022.

During the first six months of 2022, ProFrac LLC and its members completed the acquisition of FTSI, aC-Corporation. Through a series of transactions, ProFrac LLC obtained ownership of all of the assets and liabilities of FTSI, but the FTSI C-Corporation legal entity was owned by the ProFrac LLC members. In connection with the IPO, the FTSI C-Corporation legal entity merged with a subsidiary of ProFrac Corp., which inherited the tax attributes of the FTSI legal entity.

As of June 30, 2022, the Company had approximately \$158.3 million of net deferred tax assets. The deferred tax assets are driven by the outside basis difference between tax and GAAP in the Company's outside basis in ProFrac LLC (including \$22.7 million subject to the Tax Receivable Agreement), generated by both the FTSI transaction and the IPO. We have recorded a valuation allowance on the Company's net deferred tax assets based on our assessment that it is more likely than not that the deferred tax assets will not be realized, with the exception of certain deferred tax assets that are expected to be utilized in the 2022 tax year of approximately \$4.1 million. This valuation allowance assessment is based on the cumulative losses incurred by the Company in recent years. A change in our assessment could cause a decrease to the valuation allowance, which could materially impact our results of operations.

The Company's effective tax rate ("ETR") from continuing operations is expected to be4.9% for tax year 2022. After consideration of a2.4% net state tax rate (inclusive of the Texas Margin tax), the majority of the remaining difference between the federal corporate income tax rate of 21% and our effective tax rate is related to the income that is earned within the financial statement consolidated group that is not subject to tax within the financial statement consolidated group. This is due to the Company not being subject to tax prior to IPO, as well as the amount of post-IPO income that is attributable to the Class B shareholders. As such, the Company's income tax expense for the six months ended June 30, 2022 is \$4.9 million.

ProFrac LLC is obligated to make cash distributions to PFH Unit holders to fund their respective income tax liabilities relating to their share of the income of ProFrac LLC. At June 30, 2022, the Company recorded a liability for accrued distributions of \$15.6 million to be made to redeemable noncontrolling interest holders to fund their estimated tax payments.

# 9. Tax receivable agreement

In connection with our IPO, ProFrac Corp. entered into a tax receivable agreement (the "TRA") with certain PFH Unit holders (the "TRA Holders"). The TRA generally provides for payment by ProFrac Corp. to the TRA Holders of 85% of the net cash savings, if any, in U.S. federal, state and local income tax and franchise tax (computed using simplifying assumptions to address the impact of state and local taxes) that ProFrac Corp. actually realizes (or is deemed to realize in certain circumstances) as a result of (i) certain increases in tax basis that occur as a result of ProFrac Corp.'s acquisition (or deemed acquisition for U.S. federal income tax purposes) of all or a portion of such TRA Holder's PFH Units in connection with the IPO or the exercise of the Redemption Right (as defined in the TRA) or the Call Right (as defined in the TRA), and (ii) imputed interest deemed to be paid by ProFrac Corp. as a result of, and additional tax basis arising from, any payments ProFrac Corp. makes under the TRA. ProFrac Corp. will be dependent on ProFrac LLC to make distributions to ProFrac Corp. in an amount sufficient to cover ProFrac Corp.'s obligations under the TRA. ProFrac Corp. will retain the benefit of the remaining 15% of any actual net cash tax savings. The payment obligations under the TRA are ProFrac Corp.'s obligations and not obligations of ProFrac LLC, and we expect that the payments required to be made under the TRA could be substantial.

The term of the TRA commenced upon the completion of the IPO and will continue until all tax benefits that are subject to the TRA have been utilized or expired, unless we experience a Change of Control (as defined in the

TRA, which includes certain mergers, asset sales, or other forms of business combinations) or the TRA otherwise terminates early (at our election or as a result of our breach or the commencement of bankruptcy or similar proceedings by or against us) and ProFrac Corp. makes the termination payments specified in the TRA in connection with such Change of Control or other early termination. In the event that the TRA is not terminated, the payments under the TRA could commence in 2023 and will continue for 15 years after the date of the last redemption of the PFH Units.

Payments will generally be made under the TRA as we realize actual cash tax savings from the tax benefits covered by the TRA. However, if we experience a Change of Control or the TRA otherwise terminates early, ProFrac Corp.'s obligations under the TRA would accelerate and ProFrac Corp. would be required to make an immediate payment equal to the present value of the anticipated future payments to be made by it under the TRA. We estimate that such an immediate payment could range up to more the \$500 million. There can be no assurance that we will be able to satisfy our obligations under the TRA.

Estimating the amount and timing of payments that may become due under the TRA is by its nature imprecise. For purposes of the TRA, net cash tax savings generally are calculated by comparing ProFrac Corp.'s actual tax liability (determined by using the actual applicable U.S. federal income tax rate and an assumed combined state and local income and franchise tax rate) to the amount ProFrac Corp. would have been required to pay had it not been able to utilize any of the tax benefits subject to the TRA. The actual increases in tax basis covered by the TRA, as well as the amount and timing of any payments under the TRA, will vary depending on a number of factors, including the timing of any redemption of PFH Units, the price of ProFrac Corp.'s Class A Common Stock at the time of each redemption, the extent to which such redemptions are taxable transactions, the amount of the redeeming PFH Unit holder's tax basis in its PFH Units at the time of the relevant redemption, the depreciation and amortization periods that apply to the increase in tax basis, the amount and timing of taxable income we generate in the future, the U.S. federal income tax rates then applicable, and the portion of ProFrac Corp.'s payments under the TRA that constitute imputed interest or give rise to depreciable or amortizable tax basis.

We account for amounts payable under the TRA when we determine that a liability is probable and the amount is reasonably estimable. At June 30, 2022, the liability from the TRA was \$3.4 million.

### 10. Earnings per share

The numerators and denominators of the basic and diluted earnings per share ("EPS") computations for our Class A Common Stock are calculated as follows:

	Three Months Ended June 30,		Six Months Ended June 30,		
(In thousands)	2022	2021	2022	2021	
Numerator:					
Net income (loss)	\$ 70,093	\$(8,606)	\$ 94,219	\$(34,595)	
Less: net (income) loss attributable to ProFrac Predecessor	(56,157)	8,478	(79,867)	34,476	
Less: net loss attributable to noncontrolling interests	8,704	128	8,288	119	
Less: net income attributable to redeemable noncontrolling interests	(16,082)		(16,082)		
Net income attributable to ProFrac Holding Corp.	\$ 6,558	<u>\$</u>	\$ 6,558	<u>\$</u>	
Denominator:					
Weighted average Class A shares used for basic EPS computation	40,123		40,123		
Dilutive potential of employee restricted stock units	22		22		
Weighted average Class A shares used for diluted EPS computation	40,145		40,145		
Basic and diluted EPS—Class A Common Stock	\$ 0.16		\$ 0.16		

### 11. Stock-based compensation

### Stock-based Compensation Related to Deemed Contributions

In connection with the Company's IPO, our majority shareholders, Farris Wilks ("Farris") and Dan Wilks ("Dan") (together with certain family members or entities they control), sold PFH Units representing approximately 1% of the equity interest in ProFrac LLC to an entity controlled by our Chief Executive Officer, Ladd Wilks ("Ladd"), and our Executive Chairman, Matt Wilks ("Matt"), respectively. These equity interests in ProFrac LLC entitled each of Ladd and Matt to 1,220,978 shares of Class B common stock in ProFrac Corp. These units were sold in exchange for promissory notes. While some of the documentation relating to these transfers was subject to completion, we concluded that both transactions were consummated in connection with the Company's IPO and, for accounting purposes, should be treated in accordance with ASC Topic 718, Compensation — Stock Compensation, as deemed contributions to the Company by Farris and Dan and grants of stock-based compensation to Ladd and Matt by the Company similar to stock options. As no future service period was required and because the promissory notes are prepayable at any time, all related stock-based compensation expense was recognized in the second quarter of 2022. The stock-based compensation expense was \$33.7 million using the Black-Scholes-Merton option-pricing model with an average contractual term of 16.5 years, a volatility rate of 64%, and a 0% dividend yield.

Also in connection with the IPO, Farris engaged in estate planning that may result, subject to other terms and conditions, in additional shares being transferred by Farris to Ladd if the Company's total market capitalization increases to certain target levels within the next five years, which resulted in a performance award being deemed granted by the Company to Ladd. We concluded that this arrangement should be treated, for accounting purposes, in accordance with ASC Topic 718, Compensation—Stock Compensation, as a deemed contribution to the Company by a related party and the grant of stock compensation with market conditions to Ladd by the Company.

The grant date fair value of this award was estimated to be \$45.2 million and will be recognized over the estimated derived service period of approximately one year. The grant date fair value and the derived service period of this award was determined using a Monte Carlo simulation method, which incorporates the possibility that the market capitalization targets may not be satisfied. The Monte Carlo simulation is affected by a number of variables, including the fair value of our underlying common shares (\$18.00 at grant date), the expected common share price volatility over the expected term (9.2%), the expected dividend yield of our common shares over the expected term (0.0%), the risk-free interest rates over the expected term (2.86%), and the performance period of the award (five years).

The derived service period for the award was determined based on the median vesting time for the simulations that achieved the vesting hurdleStock-based compensation expense associated with this award will be recognized over the earlier of (i) the derived service period and (ii) the date on which the market condition is satisfied. Stock-based compensation expense of \$5.1 million was recognized in the second quarter of 2022. At June 30, 2022, there was \$40.1 million of total unrecognized compensation cost related to this award, which is expected to be recognized over a weighted average period of 1.0 years.

### 2022 Equity and Incentive Compensation Plan

In May 2022, the Company adopted the ProFrac Holding Corp. 2022 Long Term Incentive Plan ("2022 Plan") to attract and retain officers, employees, directors, and other key personnel and to provide those persons incentives and awards for performance. The 2022 Plan originally allocated 3,121 thousand shares of our Class A Common Stock in the form of incentive stock options, non-qualified stock options, restricted stock, restricted stock units ("RSUs"), stock appreciation rights, or other stock-based awards. As of June 30, 2022, up to approximately 2,611 thousand shares were available for future grants under the 2022 Plan.

On May 24, 2022, the Company issued 509 thousand RSUs, which had a grant date fair value of \$9.1 million. The RSUs were valued at \$17.90 per share, the market price of our Class A Common Stock on the date of grant. Awards granted vest from one to three years from the date of grant.

Stock-based compensation expense for the Company in the second quarter of 2022 for these RSUs was \$0.9 million. The weighted-average grant-date fair value per share of RSUs granted was \$17.90. The fair value of RSUs vested in the second quarter of 2022 waszero. At June 30, 2022, there was \$8.2 million of total unrecognized compensation cost related to unvested RSUs, which is expected to be recognized over a weighted average period of 1.2 years.

### 12. Leases

Effective January 1, 2022, we adopted the new lease accounting guidance under ASC Topic 842, Leases. The details of the significant changes to our accounting policies resulting from the adoption of the new lease standard are set out below. We adopted the standard using the optional modified retrospective transition method; accordingly, the comparative information as of December 31, 2021, has not been adjusted and continues to be reported under the previous lease standard. Under the new standard, assets and liabilities that arise from all leases are required to be recognized on the balance sheet for lessees. Previously, only capital leases, which are now referred to as finance leases, were recorded on the balance sheet. The adoption of this standard resulted in the recognition of approximately \$35.8 million of operating lease right-of-use assets and operating lease liabilities on our consolidated balance sheet as of January 1, 2022. The adoption of this standard did not materially impact our consolidated results of operations for the three and six months ended June 30, 2022.

Beginning January 1, 2022, for all leases with a term in excess of 12 months, we recognized a lease liability equal to the present value of the lease payments and right-of-use asset representing our right to use the underlying asset for the lease term. For operating leases, lease expense for lease payments is recognized on a straight-line basis over the lease term and accretion of the lease liability, while finance leases (of which we have none) include both an operating expense and an interest expense component. For all leases with a term of 12 months or less, we elected the practical expedient to not recognize lease assets and liabilities. We recognize lease expense for these short-term leases on a straight-line basis over the lease term.

We are a lessee for several operating leases, related primarily to real estate and light duty vehicles. The majority of our operating leases have remaining lease terms of 10 years or less. None of our leases include options to extend the leases, nor do any include options to terminate the leases. The accounting for leases may require judgment, which includes determining the incremental borrowing rates to utilize in our net present value calculation of lease payments for lease agreements which do not provide an implicit rate.

In connection with the completion of the FTSI acquisition, FTSI conveyed to Wilks Development, LLC, an affiliate of ProFrac LLC, substantially all of FTSI's owned real property, consisting primarily of FTSI's hydraulic fracturing equipment manufacturing facilities, in exchange for cash consideration of approximately \$44.4 million (the "FTSI Sale Leaseback"). We will lease such real property from Wilks Development, LLC in exchange for aggregate monthly lease payments of \$51.6 million through March 2032. The cash consideration received was \$3.7 million less than the carrying value of these assets. Because this sale was to an affiliate under common control, we accounted for the \$3.7 million as an equity transaction recorded as a deemed distribution within our consolidated statements of changes in equity.

Our leasing activities primarily consist of operating leases for administrative offices, manufacturing and maintenance facilities along with some light duty vehicles. We do not lease any equipment on a long-term basis.

The following table summarizes the components of our lease costs:

(In thousands)	Three Months Ended June 30, 2022		Months ed June 30, 2022
Operating lease costs	\$ 3,400	\$	5,601
Short-term lease costs	1,017		1,356
Total lease costs	\$ 4,417	\$	6,957

The following table includes other supplemental information for our operating leases:

	Six Months Ended June 30,	
(Dollars in thousands)		2022
Cash paid for amounts included in the measurement of our lease		_
obligations	\$	5,251
Right-of-use assets obtained in exchange for lease obligations	\$	44,968
Right-of-use assets recognized upon adoption of the leasing standard	\$	35,817
Weighted-average remaining lease term		8.1 years
Weighted-average discount rate		5.0%

The following table summarizes the maturity of our operating leases as of June 30, 2022:

(In thousands)	
Remainder of 2022	\$ 7,129
2023	13,095
2024	12,642
2025	11,496
2026	11,537
2027	11,803
2028 and thereafter	38,474
Total lease payments	106,176
Less imputed interest	(21,619)
Total lease liabilities	\$ 84,557

### 13. Related party transactions

In the normal course of business, the Company has entered into transactions with related parties where the Wilks (or entities they control) hold a controlling financial interest. During the three and six months ended June 30, 2022 and 2021, the Company had related party transactions with the following related party entities:

Automatize, LLC ("Automatize") is a logistics broker that facilitates the last-mile delivery of proppants on behalf of its customers, including the Company. Amounts paid to Automatize include costs passed through to third-party trucking companies and a commission retained by Automatize. These payments are recorded in cost of revenues, exclusive of depreciation and depletion on our consolidated statements of operations.

Cisco Logistics, LLC ("Cisco Logistics") is a logistics company that delivers sand and equipment on behalf of its customers, including the Company. Amounts paid to Cisco Logistics are recorded in cost of revenues, exclusive of depreciation and depletion on our consolidated statements of operations.

Equify Risk Services, LLC ("Equify Risk") is an insurance broker that negotiates and secures insurance policies on behalf of its customers, including the Company. Amounts paid to Equify Risk are recorded in selling, general and administrative expenses on our consolidated statements of operations.

Equify Financial, LLC ("Equify Financial") is a finance company that provides equipment and other financing to its customers, including the Company. Amounts paid to Equify Financial are recorded in interest expenses on our consolidated statements of operations, and repayments of long-term debt on our consolidated statements of cash flows. See Note 7—Indebtedness for additional disclosures related to related party credit agreements.

Wilks Brothers, LLC ("Wilks Brothers") is a management company which provides administrative support to various businesses within its portfolio. Wilks Brothers and certain entities under its control will at times incur expenses on behalf of the Company, billing the Company for these expenses at cost as well as certain management fees. Amounts paid to Wilks Brothers are generally recorded in selling, general and administrative expenses on our consolidated statements of operations.

Interstate Explorations, LLC ("Interstate") is an exploration and development company for which the Company performs pressure pumping services, and from which the Company has a short-term lease for certain office space.

Flying A Pump Services, LLC ("Flying A") is an oilfield services company which provides pump down and acid services, to which the Company rents and sells equipment and frac fleet components.

MC Estates, LLC, The Shops at Willow Park, and FTSI Industrial, LLC (collectively, the "Related Lessors") own various industrial parks and office space leased by the Company. Amounts paid to the Related Lessors are recorded in selling, general and administrative expenses on our consolidated statements of operations.

Wilks Construction Company, LLC ("Wilks Construction") is a construction company that has built and made renovations to several buildings for the Company, including construction of a new sand plant. Amounts paid to Wilks Construction are recorded in capital expenditures on our consolidated statements of cash flows.

3 Twenty-Three, LLC ("3 Twenty-Three") is a payroll administrator which performs payroll services on behalf of its customers, including the Company. Amounts paid to 3 Twenty-Three are recorded in cost of revenues, exclusive of depreciation and depletion and selling, general and administrative expenses on our consolidated statements of operations.

Carbo Ceramics Inc. ("Carbo") is a provider of ceramic proppant which will at times purchase conventional proppant from the Company to act as a broker for its customers. Additionally, the Company will at times purchase manufactured proppant from Carbo for the stimulation services segment.

FHE USA LLC ("FHE") is a provider of production and well completion equipment used at the wellsite. Its RigLock and FracLock systems remotely connect surface equipment to the wellhead that keeps crews safer and speeds up operations while also reducing the volume of high-pressure iron. Amounts paid to FHE are recorded in capital expenditures on our consolidated statements of cash flows.

The following table summarizes expenditures with related parties for the three and six months ended June 30, 2022 and 2021:

	Three Months Ended			hs Ended
	Jun	e 30,	Jun	e 30,
(In thousands)	2022	2022 2021		2021
Automatize	\$35,576	\$18,512	\$49,851	\$31,268
FHE	4,027	_	7,277	_
Wilks Brothers	8,542	1,953	8,799	4,507
Related Lessors	3,425	1,577	4,648	3,115
Wilks Construction	12,448	_	13,389	_
Equify Financial	301	_	986	_
3 Twenty-Three	_	_	247	_
Carbo	78	_	78	353
Cisco Logistics	_	160	_	424
Interstate	_	24	20	32
Equify Risk	_	_	_	3
Other	82	42	120	52
Total	<u>\$64,479</u>	\$22,268	\$85,415	\$39,754

The following table summarizes related party accounts payable as of June 30, 2022 and December 31, 2021:

	June 30,	December 31,
(In thousands)	2022	2021
Automatize	\$18,013	\$ 11,198
Wilks Brothers	13,518	9,990
Wilks Construction	5,545	57
Cisco Logistics	27	_
Carbo	455	10
Related Lessors	19	1
Other	0	19
Total	\$37,577	\$ 21,275

The following table summarizes revenue from related parties for the three and six months ended June 30, 2022 and 2021:

	Three Months Ended June 30,			Six Months Ended June 30,		
(In thousands)	2022	2021	2022	2021		
Flying A	\$ 548	\$ 1,177	\$1,920	\$2,128		
Carbo	504	159	753	179		
Wilks Brothers	2	7	3	5		
Interstate	_	111	_	111		
Other	1	25	1	34		
Total	<u>\$ 1,055</u>	\$ 1,479	\$2,677	\$2,457		

The following table summarizes related party accounts receivable as of June 30, 2022 and December 31, 2021:

	June 30,	Dece	ember 31,
(In thousands)	2022		2021
Flying A	\$ 1,438	\$	2,412
Cisco Logistics	1,493		1,489
Carbo	255		591
Interstate	310		_
Other	141		23
Total	\$ 3,637	\$	4,515

Additionally, in January and February of 2021, ProFrac LLC executed two agreements with one of ProFrac LLC's members for the sale of certain lots of equipment, in exchange for \$8.7 million in cash, an amount that approximates the net book value of the assets. Under these agreements, for any assets subsequently resold by the member, ProFrac LLC will reimburse the member for a certain percentage of the net loss, or conversely be entitled to a certain percentage of the net gain, at rates established in the agreements. As of June 30, 2022, the majority of the assets have been sold by the member, and ProFrac LLC's remaining liability of approximately \$0.2 million for assets that have not been resold has been recorded.

On February 4, 2022, THRC Holdings entered into a Rights Agreement with Encantor Properties LP, one of the sellers from whom the Company purchased the Munger Ranch property, under which the related party was assigned rights to \$8.1 million of the \$30.0 million in consideration related to the Munger purchase. As part of the IPO completed in May 2022, the sellers of Munger Ranch were issued 2,114,273 shares of Class A Common Stock in exchange for the \$30.0 million consideration related to the Munger Ranch purchase.

### 14. Acquisitions and investments

FTS International, Inc.

On March 4, 2022, the Company acquired the outstanding stock of FTSI for a purchase price of \$405.7 million, consisting of cash consideration of \$332.8 million, and THRC Holdings' equity interest of \$72.9 million ("THRC FTSI Related Equity").

Immediately following the closing of the cash acquisition pursuant to the Agreement and Plan of Merger, dated as of October 21, 2021, by and among FTSI, ProFrac LLC and ProFrac Acquisitions, Inc. (the "FTSI Merger Agreement"), ProFrac LLC distributed the 80.5% of the FTSI equity it acquired in such merger to Farris Wilks and THRC Holdings in a manner that resulted in each of them owning 50% of FTSI (the "FTSI Distribution"), with THRC Holdings receiving a smaller share of the FTSI Distribution and instead retaining certain preferred equity in ProFrac LLC in lieu of its redemption in connection with such distribution. The THRC FTSI Related Equity was the result of a transaction whereby THRC Holdings, which owned approximately 19.5% of FTSI, agreed to retain that interest in FTSI in lieu of receiving cash pursuant to the FTSI Merger Agreement.

We have accounted for the acquisition of FTSI using the acquisition method of accounting. We used our best estimates and assumptions to assign fair value to the tangible and intangible assets expected to be acquired and liabilities expected to be assumed at the acquisition date. The majority of the measurements of assets acquired and liabilities assumed, are based on inputs that are not observable in the market and thus represent Level 3 inputs.

The following table summarizes the preliminary allocation of the purchase price:

(In thousands)	
Assets acquired:	
Cash and cash equivalents	\$ 53,771
Accounts receivable	89,268
Prepaid expense and other assets	4,037
Inventories	42,344
Property, plant and equipment	307,113
Operating lease ROU asset	2,748
Intangible assets	1,239
Other assets	1,583
Total assets acquired	502,103
Liabilities assumed:	
Accounts payable	62,985
Accrued expenses	19,308
Operating lease liability current	1,235
Current portion of debt	10,136
Other current liabilities	309
Operating lease liability non-current	1,512
Other non-current liabilities	928
Total liabilities assumed	96,413
Net assets acquired	<u>\$ 405,690</u>

The allocation of the purchase price to FTSI's net tangible assets and liabilities and identifiable intangible assets as of March 4, 2022, is preliminary and subject to revisions to the fair value calculations for the identifiable assets and liabilities. The determination and allocation of the purchase consideration are subject to change during the measurement period, up to one year from the date the acquisition closed.

For the three months ended March 31, 2022, our revenues and pretax earnings included \$48.6 million and \$0.1 million loss, respectively, associated with the FTSI acquired operations after the closing on March 4, 2022. In addition, FTSI acquisition-related costs of approximately \$3.7 million were incurred during the three months ended March 31, 2022, consisting of external legal and consulting fees. These costs are classified in selling, general and administrative expense in the consolidated statements of operations. Additionally, we incurred \$9.3 million in severance costs in connection with the FTSI acquisition, which are classified in selling, general and administrative expense in the consolidated statements of operations. Throughout the second quarter of 2022, we integrated FTSI's operations. As a result, we track all stimulation services assets as one group and it would be impracticable to separately report FTSI revenues or pretax earnings subsequent to March 31, 2022.

### Flotek Industries, Inc.

On February 2, 2022, we entered into an agreement with Flotek Industries, Inc. ("Flotek"), pursuant to which Flotek will provide full downhole chemistry solutions for a minimum of ten hydraulic fleets for three years starting on April 1, 2022, at a price of cost plus 7% ("Flotek Supply Agreement"). In exchange for entry into the Flotek Supply Agreement, we received \$10 million in initial principal amount of Flotek Convertible Notes and acquired an additional \$10 million in principal amount of Flotek Convertible Notes in the PIPE Transaction. Our equity ownership in Flotek on a fully diluted basis as a result of this investment is approximately 17%. In addition, we received the right to designate up to two directors to Flotek's board of directors.

On February 16, 2022, we and Flotek agreed to amend the Flotek Supply Agreement to increase the term toten years and increase the scope to 30 fleets. In exchange for our entry into the amendment to the Flotek Supply Agreement (the "Flotek Supply Agreement Amendment"), Flotek agreed to issue us \$50 million in initial principal amount of Flotek Convertible Notes that will be convertible into Flotek common stock. The Flotek Supply Agreement Amendment and issuance to us of additional Flotek Convertible Notes were conditioned upon customary closing conditions including the approval of Flotek's shareholders. In May 2022, the Flotek shareholders approved the Convertible Notes issuance and the Flotek Supply Agreement Amendment. Our equity ownership in Flotek on a fully diluted basis after the consummation of these transactions is approximately 43%, and we are permitted to designate two additional directors, or up to four directors to Flotek's board of directors. Because of our power to appoint directors to the board of directors without a direct equity interest in Flotek, we determined that Flotek is a VIE. We further determined that the Company is the primary beneficiary of the VIE, primarily due to our ability to appoint four of seven directors to Flotek's board of directors. As a result, subsequent to May 17, 2022, we have accounted for this transaction as a business combination using the acquisition method of accounting and Flotek's financial statements have been included in our consolidated financial statements from May 17, 2022, we allocated 100% of Flotek's loss to noncontrolling interests in our consolidated financial statements.

The Flotek Supply Agreement Amendment includes a minimum annual volume commitment whereby we will be obligated to pay Flotek liquidated damages equal to 25% of the shortfall for such year, should we fail to meet the minimum purchase amount. At May 17, 2022, we had a supply agreement contract liability of \$9.9 million, which was included as purchase consideration for Flotek as a settlement of apre-existing relationship. All effects of the Supply Agreement have been eliminated from our consolidated financial statements subsequent to May 17, 2022.

The notes issued to ProFracaccrue paid-in-kind interest at a rate of 10% per annum, have a maturity of one year, and convert into common stock of Flotek (a) at the holder's option at any time prior to maturity, at a price of \$1.088125 per share, (b) at Flotek's option, if the volume-weighted average trading price of Flotek's common stock equals or exceeds \$2.50 for 20 trading days during a 30 consecutive trading day period, or (c) at maturity, at a price of \$0.8705 (the "Convertible Notes"). We initially recognized the Convertible Notes with an initial principal balance of \$0 million at \$20 million. At May 17, 2022, we estimated the fair value of these Convertible Notes to be \$30.2 million, which was included as purchase consideration for Flotek as a settlement of a pre-existing relationship. All effects of the Flotek Convertible Notes have been eliminated from our consolidated financial statements subsequent to May 17, 2022.

Before May 17, 2022, we designated our investment in the Flotek Convertible Notes as trading securities. Securities designated as trading securities are reported at fair value, with gains or losses resulting from changes in fair value recognized in net investment income on the consolidated statements of operations. For the period from February 2, 2022 through May 17, 2022 we recognized noncash income of \$10.2 million as other (expense) income on our consolidated statements of operations related to the change in fair value of the Flotek Convertible Notes.

In June 2022, Flotek issued and sold to ProFrac II LLC, a wholly-owned subsidiary of ProFrac LLC, pre-funded warrants to purchase from Flotek up to approximately 13.1 million shares of Flotek common stock at any time and at an exercise price equal to \$0.0001 per share, in exchange for \$19.5 million in cash. ProFrac II LLC and its affiliates may not receive any voting or consent rights in respect of these warrants or the underlying shares unless and until (i) Flotek has obtained approval from a majority of its shareholders excluding ProFrac II LLC and its affiliates and (ii) ProFrac II LLC has paid an additional \$4.5 million to Flotek. We entered into this transaction to provide additional working capital to Flotek to enable it to perform under the Flotek Supply Agreement Amendment. These pre-funded warrants have been eliminated from our consolidated financial statements.

The following table summarizes the preliminary allocation of the fair value of Flotek's assets, liabilities and noncontrolling interest:

(In thousands)	
Settlement of pre-existing relationships	
Accounts payable	\$ (2,713)
Supply Agreement contract liability	(9,874)
Fair value of previously held interest in 10% Convertible PIK Notes	30,220
Settlement of pre-existing relationships	\$ 17,633
Assets acquired	
Cash and cash equivalents	\$ 21,725
Restricted cash	40
Accounts receivable	18,853
Inventories	12,210
Assets held for sale	1,805
Other current assets	3,405
Property and equipment	21,551
Operating lease right-of-use assets	3,884
Deferred tax assets	282
Goodwill	82,340
Other long-term assets	17
Total assets acquired	166,112
Liabilities assumed:	
Accounts payable and accrued liabilities	24,203
Operating lease liabilities	7,394
Finance lease liabilities	79
Long-term debt	17,101
Other liabilities	85
Total liabilities assumed	48,862
Noncontrolling interests	99,617
Assets acquired less liabilities assumed and noncontrolling interests	\$ 17,633

We used our best estimates and assumptions to assign fair value to the tangible and intangible assets acquired and liabilities assumed at the acquisition date. The majority of the measurements of assets acquired and liabilities assumed, are based on inputs that are not observable in the market and thus represent Level 3 inputs. The fair value of the noncontrolling interest was based on the Flotek common stock price reported by the New York Stock Exchange at the date of the acquisition, which represented Level 1 inputs. No portion of the recorded goodwill is tax deductible.

For the three months ended June 30, 2022, our revenues and pretax earnings included \$5.1 million and \$8.5 million loss, respectively, associated with the Flotek acquired operations after May 17, 2022.

The allocation of the purchase price to Flotek's net tangible assets and liabilities and identifiable intangible assets as of May 17, 2022, is preliminary and subject to revisions to the fair value calculations for the identifiable assets and liabilities. The determination and allocation of the purchase consideration are subject to change during the measurement period, up to one year from the date the acquisition closed.

The following unaudited pro forma results of operations have been prepared as though the FTSI and Flotek acquisitions had been completed on January 1, 2021. Pro forma amounts are based on the purchase price allocation of the significant acquisition and are not necessarily indicative of the results that may be reported in the future.

	Three Mor June		Six Months Ended June 30,		
(In thousands)	2022	2021	2022	2021	
Revenue	\$ 596,774	\$ 283,811	\$ 1,030,161	\$ 541,074	
Net income (loss)	\$ 66,662	\$ (34,189)	\$ 50,203	\$ (92,743)	

### Agreement to Acquire U.S. Well Services, Inc.

As previously disclosed in the Company's Current Report on Form8-K filed with the SEC on June 24, 2022 (the "USWS Merger8-K"), on June 21, 2022, the Company entered into an Agreement and Plan of Merger (the "Merger Agreement"), by and among the Company, U.S. Well Services, Inc., a Delaware corporation ("USWS"), and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of the Company ("Merger Sub"). The Merger Agreement provides for, among other things, the merger of Merger Sub Inc. with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of the Company (the "Merger").

Subject to the terms and conditions of the Merger Agreement, at the effective time and as a result of the Merger, each share of Class A Common Stock of USWS, par value \$0.0001 per share (the "USWS Common Stock"), issued and outstanding immediately prior to the effective time of the Merger will be cancelled and converted automatically into the right to receive 0.3366 shares of the Company's Class A Common Stock.

The acquisition is expected to be completed in the fourth quarter of 2022, subject to the satisfaction of customary closing conditions, including the approval of USWS stockholders, as described in the USWS Merger 8-K.

Based on the closing price of the Company's Class A Common Stock (as reported on the Nasdaq Global Select Market) of \$1.49 per share on June 21, 2022, the last trading day prior to announcement of the Merger Agreement, the transaction represents aggregate stock consideration of approximately \$93 million and a consideration per share of USWS Common Stock of \$7.23, after giving effect to USWS' recent 1-for-6 reverse stock split. After giving effect to the conversions of certain securities of USWS, as described in the USWS Merger 8-K, the total stock consideration payable to USWS stockholders and holders of USWS equity awards, based on the Company's Class A Common Stock June 21, 2022 closing price, would be approximately \$270 million. In addition, the Company will assume approximately \$55 million of USWS' outstanding

indebtedness and repay approximately \$170 million of USWS debt expected to be outstanding at closing. We anticipate that the total consideration to be paid in connection with the closing of the Merger will be funded through the New Term Loan Credit Facility and/or the New ABL Credit Facility, available cash, additional loans and/or notes and cash flows generated from operations.

The Wilks Parties hold a controlling interest in the Company. Certain Wilks Parties also own certain securities of USWS. Upon the consummation of the Merger, the Wilks Parties will receive an aggregate of 4,138,596 shares of the Company's Class A Common Stock as merger consideration (assuming, solely for the purpose of this calculation, a hypothetical closing date of October 31, 2022), which, based on the Company's Class A Common Stock June 21, 2022 closing price, would be approximately \$88.9 million.

The consummation of the Merger is subject to customary conditions, including, among others, (a) receipt of the approval of the stockholders of USWS, (b) the absence of any applicable law or order prohibiting the consummation of the Merger or the issuance of Company stock in connection with the Merger, (c) the expiration or termination of the HSR Act waiting period (which has been satisfied), (d) the effectiveness of the registration statement in connection with the Merger, (e) approval of the issuance of Company stock in connection with the Merger for listing on Nasdaq, (f) subject to certain exceptions, the accuracy of the representations and warranties of each party, (g) the performance in all material respects of each party of its obligations under the Merger Agreement and (h) the absence of a material adverse effect on the Company and USWS.

The Merger Agreement contains certain termination rights for both the Company and USWS and further provides that, upon termination of the Merger Agreement under specified circumstances, USWS may be required to pay the Company a termination fee of \$8.0 million or reimburse the Company for certain expenses in an amount up to \$3.0 million, in each case, depending on the termination event.

### Basin Production and Completion LLC

During the year ended December 31, 2021, the Company purchased Series B-1 Preferred Units of Basin Production and Completion LLC ("BPC"), a manufacturer of equipment used in hydraulic fracturing for \$4.2 million. This comprises approximately 20% of the BPC Series B Preferred Units. As of December 31, 2021, we did not exercise significant influence over this entity. As we determined this to be an equity security, we initially recorded our investment at cost, presented as "Investments" in our consolidated balance sheets as of December 31, 2021.

On February 9, 2022, the Company entered into an agreement to purchase all the series A-1 and B-1 preferred units of BPC for \$46.0 million ("Basin Units Acquisition"), consisting of \$40.0 million to BPC for series A-1 and B-1 preferred units and \$6.0 million to selling holders of BPC series B-1 preferred units. Additionally, on February 14, 2022, the Company made a loan to FHE, a subsidiary of BPC for \$1.25 million. The loan bears interest at the rate of 5% per annum. Interest is either paid at each calendar quarter end or added to the principal balance at the election of BPC. The loan matures on February 14, 2027.

Subsequent to February 9, 2022, our investments in BPC provide the Company the ability to have significant influence, but not control over BPC's operations. BPC's business and affairs are managed under the direction of its board of directors, which the Company does not control. Based on our evaluation, we determined that BPC is a VIE, but the Company is not the primary beneficiary of the VIE. We have elected the fair value option to account for our equity method investment in BPC. See Note 16 — Fair Value of Financial Interests for more information on our investments using Level 3 measurements. At June 30, 2022, the estimated fair value of these investments in BPC was \$49.8 million.

### **EKU**

On December 22, 2020, the Company purchased a 25% stake in EKU, an equipment manufacturer based in Germany, for \$1.2 million. For the year ended December 31, 2020, we accounted for this investment using the

equity method as we had significant influence over EKU, and held a voting interest of 20% or greater, but less than 50%. In January 2021, the Company obtained a controlling interest in EKU, the results of which are consolidated thereafter.

The Company obtained a 75% controlling interest in EKU in January 2021 and performed a purchase price allocation in conjunction with the consolidation of this subsidiary. We recognized net working capital of \$2.5 million, property, plant and equipment of \$0.4 million, intangible assets of \$3.5 million and debt of \$1.4 million at estimated fair value. In consolidation, we eliminated our investment in associate, recognized equity of \$.7 million for the value of our 75% interest, and noncontrolling interest of \$1.2 million for the value of the minority shareholder positions. See Note 6 – Intangible Assets for additional information related to the recognized intangible assets.

### Munger Ranch

The Company entered into an agreement to acquire property in West Texas ("Munger Right Agreement") in November 2021 for a purchase price of \$30.0 million. Under the Munger Right Agreement, the sellers were given the option to receive the consideration in cash, or in the event of an IPO prior to November 17, 2022, in equity, at the sellers' election. Under the equity option, in the event the Company completes an IPO, the sellers would be entitled to 1.5% of the outstanding shares of common stock immediately following the IPO. Each seller ultimately elected the equity option, as such this was a non-cash transaction for the Company for the year ended December 31, 2021.

The Munger Right Agreement includes a 'Make Whole' provision. Under the Make Whole provision, if any seller liquidates 100% of the shares of our Class A Common Stock they are issued prior to the one-year anniversary of the IPO and the value of the shares sold does not equal such seller's share of the \$30.0 million purchase price, then the Company will pay the difference between the amount of cash the seller would have received had they elected the cash option and the amount they ultimately received upon the sale of the Class A shares issued under the equity option. This Make Whole provision is accounted for as a written put option with a fair value of \$4.0 million as of June 30, 2022 and is presented within Other Current Liabilities in our Balance Sheet. The acquired property was treated as an asset acquisition and not an acquisition of a business, and is presented within Property, plant, and equipment in our consolidated balance sheets.

On February 4, 2022, THRC Holdings entered into a Rights Agreement with Encantor Properties LP, one of the sellers from whom the Company purchased the Munger Ranch property, under which the related party was assigned rights to \$8.1 million of the \$30.0 million in consideration related to the Munger Ranch purchase.

As part of the IPO completed in May 2022, the Company issued2,114,273 shares of Class A Common Stock worth \$38.1 million as consideration for the Munger Ranch purchase.

## iO-TEQ, LLC

The Company acquired iO-TEQ, LLC ("IOT") in October 2021 for \$2.2 million and performed a purchase price allocation at the acquisition date. We recognized net working capital of \$0.2 million, property, plant and equipment of \$0.1 million, intangible assets of \$2.4 million and debt of \$0.4 million at estimated fair value. See Note 6 – Intangible Assets for additional information related to the recognized intangible assets.

### Best Flow minority interests

In December 2021, the Company entered into an agreement with Eagleton Venture, Inc. ("Eagleton") to purchase Eagleton's 15.172% interest in Best Flow for a purchase price of \$3.9 million, which the company paid in cash, during the three months ended March 31, 2022.

### 15. Goodwill

The changes in the carrying amount of goodwill by reportable segment were as follows for the six months ended June 30, 2022.

	Stimulation				Pro	ppant		
	services	1	Manufa	cturing	proc	duction	Other	Total
Balances at December 31, 2021	\$ —	5	\$	_	\$	_	<del>\$</del> —	<del>\$</del> —
Acquisition of Flotek	_			_		_	82,340	82,340
Balances at June 30, 2022	\$ —	9	\$	_	\$	_	\$82,340	\$82,340

### 16. Fair value of financial instruments

Assets and liabilities recorded on a recurring basis in the balance sheets are categorized based upon the level of judgment associated with the inputs used to measure their fair values.

The Company's financial instruments and investments that are carried at fair value consist mainly of Level 3 assets. Level 3 assets that have been measured on a recurring basis during 2022 relate to the Company's investments in (i) Convertible Notes of Flotek, designated as trading securities up to the acquisition date of Flotek and (ii) the equity method investment in BPC, for which we elected the fair value option, as described in Note 14—Acquisitions and investments for disclosure of the terms of these investments. The Company did not have any assets or liabilities measured at fair value using Level 2 of the fair value hierarchy at June 30, 2022. We had no Level 2 or Level 3 assets or liabilities as of December 31, 2021.

The estimated fair value of the BPC investment as of June 30, 2022 was determined using a combination of the market and income approaches.

The following table sets forth the fair value of the Company's financial instruments within Level 3 of the fair value hierarchy.

(In thousands)	June 30, 2022
<u>(</u>	Level 3
BPC Investment	_49,752
Total	\$ 49,752

The following is a reconciliation of the beginning and ending balances for assets measured at fair value on a recurring basis using significant unobservable inputs (Level 3) from inception to June 30, 2022:

(In thousands)	
	Level 3
Fair value as of January 1, 2022	\$ —
Acquisition of Flotek Convertible Notes	20,000
Acquisition of investment in BPC	47,202
Transfer of cost method investment to Level 3 fair value measurement	4,244
Change in fair value of Level 3 fair value measurements	8,100
Fair value as of March 31, 2022	\$ 79,546
Change in Flotek fair value up to acquisition date	2,120
Elimination of Flotek convertible notes at acquisition date	(30,220)
Change in BPC fair value	(1,694)
Fair value as of June 30, 2022	\$ 49,752

The estimated fair value of the Flotek Convertible Notes prior to our consolidation of Flotek on May 17, 2022 was valued using a Monte Carlo simulation with inputs such as the market trading price of the Flotek's common stock, the expected volatility of the Flotek's stock price based on historical trends, a risk-free rate of interest based on US Treasury note rates and the term of the debt, the time to liquidation based on the maturity date of the notes, and a discount rate adjusted based on the credit risk of Flotek.

The key inputs into the Monte Carlo simulation used to estimate the fair value the Convertible Notes were as follows:

	May 17,	March 31,
	2022	2022
Risk-free interest rate	1.82%	1.63%
Expected volatility	90.0%	90.0%
Term until liquidation (years)	0.72	0.84
Stock price	\$ 1.29	\$ 1.26

### 17. Commitments and contingencies

### Litigation

In the ordinary course of business, we are the subject of, or party to a number of pending or threatened legal actions and administrative proceedings arising in the ordinary course of our business. While many of these matters involve inherent uncertainty, we believe that, other than as described below, the amount of the liability, if any, ultimately incurred with respect to proceedings or claims will not have a material adverse effect on our consolidated financial position as a whole or on our liquidity, capital resources or future annual results of operations.

Patterson v. FTS International Manufacturing, LLC and FTS International Services, LLC: On June 24, 2015, Joshua Patterson filed a lawsuit against the Company in the 115th Judicial District Court of Upshur County, Texas, alleging, among other things, that the Company was negligent with respect to an automobile accident in 2013. Mr. Patterson sought monetary relief of more than \$1 million. On July 19, 2018, a jury returned a verdict of approximately \$100 million, including punitive damages, against the Company. The trial court reduced the judgment on November 12, 2018, to approximately \$33 million. The Company's insurance carriers appealed and the Twelfth Court of Appeals reversed the verdict in its entirety on August 26, 2020, remanding the case for a new trial. The Company's insurance carriers are currently appealing one of the appellate findings with the Texas Supreme Court. No new trial date has been set. While the outcome of this case is uncertain, the Company has met its insurance deductible for this matter and we do not expect the ultimate resolution of this case to have a material adverse effect on our consolidated financial statements.

Lonestar Prospects, Ltd. d/b/a Vista Sand v. ProFrac Services, LLC: ProFrac Services, LLC ("ProFrac Services") entered into a Master Purchase Agreement For Products And/Or Services with Lonestar Prospects, Ltd. d/b/a Vista Sand ("Vista"), dated November 27, 2017 (the "Vista MSA"), as amended by the First Addendum to Vista MSA and the First Amendment to Vista MSA, both of which are dated June 10, 2018 (collectively, the "Vista Agreement"). Under the terms of the Vista Agreement, ProFrac Services agreed to purchase certain quantities of sand from Vista. Vista filed a complaint against ProFrac Services in the United States Bankruptcy Court for the Northern District of Texas on March 15, 2021, in which it alleges that ProFrac Services breached the terms of the Vista Agreement by failing to purchase the required amount of sand or pay for the underpurchased amounts as required by the Vista Agreement. Vista is seeking damages of approximately \$8.31 million. Vista and ProFrac Services have entered into a mutually agreed upon Scheduling Order signed by the Court on February 12, 2022. Trial docket call for this matter is currently scheduled for September 6, 2022.

We estimate and provide for potential losses that may arise out of legal proceedings and claims to the extent that such losses are probable and can be reasonably estimated. Significant judgment is required in making these estimates and our final liabilities may ultimately be materially different from these estimates. When preparing

our estimates, we consider, among other factors, the progress of each legal proceeding and claim, our experience and the experience of others in similar legal proceedings and claims, and the opinions and views of legal coursel. Legal costs related to litigation contingencies are expensed as incurred.

### 18. Segment information

Our business has three reportable segments: Stimulation services, Manufacturing and Proppant production. Each reportable segment represents a separate business unit that operated as a standalone company prior to the reorganization of the Company in December 2021. Following the reorganization, each reportable segment continues to have distinct management and prepares discrete financial information for the segment (consistent with when each operated as a standalone business). FTSI is part of our stimulation services segment. Amounts in the other category reflect our business activities that are not separately reportable, which only included Flotek for the periods presented. Our chief operating decision makers review the discrete segment financial information, including Adjusted EBITDA as the measure of profitability, to evaluate the performance of our segments and make resource allocation decisions.

We account for intersegment transactions as if the transactions were with third parties, that is, at estimated current market prices. For the three and six months ended June 30, 2022 and 2021, intersegment revenues for the manufacturing segment were 92% and 88% and 88% and 87%, respectively. For the three and six months ended June 30, 2022 and 2021, intersegment revenues for the proppant production segment were 64% and 44% and 66% and 38%, respectively.

The performance of our segments is evaluated primarily on Adjusted EBITDA. We define Adjusted EBITDA as our net income (loss), before (i) interest expense, net, (ii) income tax provision, (iii) depreciation, depletion and amortization, (iv) loss on disposal of assets, and (v) stock-based compensation, and (vi) other unusual or non-recurring charges, such as costs related to our initial public offering,non-recurring supply commitment charges, certain bad debt expense and gain on extinguishment of debt.

Segment information, and a reconciliation of Adjusted EBITDA, for the three and six months ended June 30, 2022 and 2021 is as follows:

	Three Months Ended June 30,		Six Months Ended June 30,		
(In thousands)	2022	2021	2022	2021	
Revenues					
Stimulation services	\$576,556	\$168,506	\$ 912,711	\$312,209	
Manufacturing	34,854	16,223	66,860	30,880	
Proppant production	17,531	7,781	29,939	13,370	
Other	15,359		15,359		
Total segments	644,300	192,510	1,024,869	356,459	
Eliminations	(54,456)	(17,691)	(90,045)	(32,054)	
Total	\$589,844	<u>\$174,819</u>	\$ 934,824	\$324,405	
Adjusted EBITDA					
Stimulation services	\$196,088	\$ 30,475	\$ 269,657	\$ 43,428	
Manufacturing	9,360	349	19,382	2,679	
Proppant production	12,574	3,246	20,459	5,652	
Other	(7,454)		(7,454)		
Adjusted EBITDA for reportable segments	210,568	34,070	302,044	51,759	
Interest expense, net	(13,451)	(6,187)	(22,723)	(12,222)	
Depreciation, depletion and amortization	(64,064)	(34,904)	(108,280)	(70,365)	
Income tax benefit (provision)	(4,112)	283	(4,864)	308	
Loss on disposal of assets, net	(2,143)	(1,868)	(1,989)	(4,075)	
Loss on extinguishment of debt	(8,822)	_	(17,095)	_	
Litigation accrual	(4,000)	_	(4,000)	_	
Stock compensation expense	(1,455)	_	(1,455)	_	
Stock compensation expense related to deemed contributions	(38,849)	_	(38,849)	_	
Bad debt expense, net of recoveries	_	_	(5)	_	
Loss on foreign currency transactions	58	_	46	_	
Reorganization costs	_	_	(55)	_	
Acquisition related expenses	(4,063)	_	(17,082)	_	
Unrealized gain on investments, net	426		8,526		
Net income (loss)	\$ 70,093	\$ (8,606)	\$ 94,219	\$ (34,595)	

Segment information as of June 30, 2022 and December 31, 2021 is as follows:

(In thousands)	June 30, 2022	December 31, 2021
Total assets		
Stimulation services	\$1,401,665	\$ 510,579
Manufacturing	113,147	77,968
Proppant production	133,235	100,294
Other	193,297	
Total segment assets	1,841,344	688,841
Eliminations	(177,719)	(24,271)
Total	\$1,663,625	\$ 664,570

## 19. Subsequent events

The following events occurred after June 30, 2022:

On July 25, 2022, the Company acquired 100% of the issued and outstanding membership interests of each of SP Silica of Monahans, LLC and SP Silica Sales, LLC (collectively "Monahans"), the West Texas subsidiaries of Signal Peak Silica, for a purchase price of \$90.0 million in cash plus working capital adjustments of approximately \$10.0 million.

On July 25, 2022, the New Term Loan Facility was amended to increase the size of the New Term Loan Facility by \$150.0 million with an uncommitted option to obtain commitments for a potential additional \$100.0 million of delayed draw loans before the earlier to occur of (i) the Merger and (ii) March 31, 2023. The maturity date of the New Term Loan Facility remains the same at March 4, 2025. The Company used a portion of the proceeds from the increased New Term Loan Facility to fund the acquisition of Monahans and anticipates using the remainder of the proceeds and operating cash to facilitate the Merger, pay outstanding debt under the New ABL Credit Facility and/or for other general corporate purposes.

On July 25, 2022, the New ABL Credit Facility was amended to add an uncommitted \$100.0 million incremental facility (the "Incremental Facility", under the terms of which existing lenders can make additional loans (in their sole discretion) under, or new lenders can join, the Incremental Facility and increase the potential size of the New ABL Credit Facility from \$200 million to \$300 million, subject to satisfaction of certain conditions. All other terms and conditions of the New ABL Credit Facility remained substantially unchanged.

## ProFrac Holding Corp. Unaudited Pro Forma Condensed Combined Financial Information

#### Introduction

ProFrac Holding Corp. (together with its wholly owned subsidiaries, "ProFrac" or the "Company") is a holding company formed to own an interest in, and act as the sole managing member of, ProFrac Holdings, LLC ("ProFrac LLC"). Our predecessor consists of ProFrac Holdings, LLC, and its subsidiaries ("ProFrac LLC"), Best Pump & Flow LP ("Best Flow") and Alpine Silica, LLC ("Alpine" and, together with ProFrac LLC and Best Flow, "ProFrac Predecessor" and "we," "us," "our" and the "Company") on a combined basis. Prior to December 21, 2021, Dan Wilks, and Farris Wilks (or entities they control) (collectively, the "Wilks") held a controlling interest in ProFrac LLC, Best Flow and Alpine. Historical periods for ProFrac Predecessor had been presented on a consolidated and combined basis given the common control ownership by the Wilks. On December 21, 2021, all of the then-outstanding membership interests in Best Flow and Alpine were contributed to ProFrac LLC in exchange for membership interests in ProFrac LLC.

On March 4, 2022, ProFrac LLC completed its acquisition of FTSI for a purchase price of approximately \$407.5 million (the "FTSI Acquisition").

On May 12, 2022 ProFrac Holding Corp. completed its initial public offering ("IPO and Restructuring"). As described in Note 4, immediately prior to completion of the IPO, we engaged in certain refinancing transactions and we also used certain net proceeds from the IPO to repay certain debt post IPO (all the refinancing transactions and debt repayment from IPO are collectively referred to as "IPO and Refinancing").

On June 21, the Company announced its acquisition of U.S. Well Services. Inc or USWS ("USWS Acquisition").

The USWS Acquisition will be accounted for under the acquisition method of accounting, which requires determination of the accounting acquirer. The accounting guidance provides that in identifying the acquiring entity in a business combination effected through an exchange of equity interests, all pertinent facts and circumstances must be considered, including the relative voting rights of the stockholders of the constituent companies in the combined company, the existence of a large minority voting interest in the combined entity if no other owner or organized group of owners has a significant voting interest, the composition of the board of directors and senior management of the combined company, the relative size of each company and the terms of the exchange of equity securities in the business combination, including payment of any premium.

Immediately following the completion of the Merger, it is anticipated that persons who were stockholders of ProFrac and USWS immediately prior to the Merger will own approximately 91.76% and 8.24%, respectively, of the then outstanding ProFrac capital stock and will own approximately 76.34% and 23.66%, respectively, of the then outstanding ProFrac Class A Common Stock. ProFrac is considered to be the acquirer of USWS for accounting purposes and will allocate the purchase price to the fair value of USWS's assets and liabilities as of the acquisition date, with any excess purchase price recorded as goodwill.

The unaudited pro forma condensed combined balance sheet (the "Pro Forma Balance Sheet") as of June 30, 2022 is based upon the historical financial statements of the Company and USWS, after giving effect to USWS Acquisition contemplated by the Merger Agreement and its related adjustments as described in Note 2 to the unaudited pro forma financial statements, as if it occurred on that date.

The unaudited pro forma condensed combined statements of operations (the "Pro Forma Statements of Operations") for the six months ended June 30, 2022 and for the year ended December 31, 2021 are based upon the historical financial statements of the Company, its predecessor, USWS and FTSI, after giving effect to USWS Acquisition, FTSI Acquisition and the IPO and Restructuring (all transactions described herein are collectively referred to as "Transactions") as if they had been completed on January 1, 2021.

The unaudited pro forma financial statements have been prepared in accordance with Article 11 ofRegulation S-X as amended by the final rule, Release 33-10786, Amendments to Financial Disclosures about Acquired and Disposed Businesses. The unaudited pro forma adjustments reflecting the USWS Acquisition and FTSI Acquisition have been prepared in accordance with business combination accounting guidance as provided in FASB ASC Topic 805 and reflect the preliminary allocation of the estimated Merger Consideration to the assets acquired and liabilities assumed based upon their estimated fair values, using the assumptions set forth in the notes to the unaudited pro forma condensed combined financial information.

The pro forma data presented reflect events directly attributable to the described transactions and certain assumptions the Company believes are reasonable. The pro forma data are not necessarily indicative of financial results that would have been attained had the described transactions occurred on the date indicated or which could be achieved in the future because they necessarily exclude various operating expenses, such as incremental general and administrative expenses associated with being a public company. The adjustments are based on currently available information and certain estimates and assumptions. Therefore, the actual adjustments may differ from the pro forma adjustments. However, management believes that the assumptions provide a reasonable basis for presenting the significant effects of the transactions as contemplated and the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma financial statements.

The unaudited pro forma financial statements and related notes are presented for illustrative purposes only. If the USWS Acquisition and other transactions contemplated herein had occurred in the past, the Company's operating results might have been materially different from those presented in the unaudited pro forma financial statements. The unaudited pro forma financial statements should not be relied upon as an indication of operating results that the Company would have achieved if the USWS Acquisition and other transactions contemplated herein had taken place on the specified date. In addition, future results may vary significantly from the results reflected in the unaudited pro forma financial statement of operations and should not be relied upon as an indication of the future results the Company will have after the contemplation of the USWS Acquisition and the other transactions contemplated by these unaudited pro forma financial statements.

The pro forma financial statements should be read in conjunction with the separate historical consolidated financial statements and related notes of each of ProFrac, USWS and FTSI included elsewhere in this proxy statement/prospectus and incorporated herein. For more information, see the sections titled "Unaudited Pro Forma Condensed Combined Financial Information," "Summary historical and pro forma financial data," and the risk factors described in the section titled "Risk Factors."

# ProFrac Holding Corp. Unaudited Pro Forma Condensed Combined Balance Sheet As of June 30, 2022 (In thousands)

	ProFrac Historical	USWS Historical	USWS Pro Forma Adjustments (Note 2)	Notes	Pro Forma Combined
Assets					
Current assets:					
Cash and cash equivalent	\$ 73,653	\$ 17,268	\$ (157,329)	(2A)	\$ 69,548
			(16,519)	(2B)	
			(22,525)	(2C)	
			175,000	(2D)	
Accounts receivable, net	444,997	31,721	_		476,718
Accounts receivable-related party	3,637	<i></i>	_		3,637
Prepaid expenses, and other current assets	19,331	7,109	_		26,440
Assets held for sale	1,805	<u> </u>			1,805
Inventories	192,377	8,074			200,451
Total current assets	735,800	64,172	(21,373)		778,599
Property, plant and equipment, net	664,245	194,943	45,199	(2E)	904,387
Operating lease right-of-use assets	80,664	18,197	_	` ′	98,861
Finance lease right-of-use assets	_	3,246	_		3,246
Investments	49,752	. —	_		49,752
Intangible assets, net	28,241	12,017	214,393	(2E)	254,651
Deferred tax assets	3,316	<u> </u>	_		3,316
Goodwill	82,340	4,971	(4,971)	(2E)	82,340
Other assets	19,267	1,119			20,386
Total assets	\$ 1,663,625	\$ 298,665	\$ 233,248		\$ 2,195,538

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these unaudited pro forms financial statements.}$ 

# ProFrac Holding Corp. Unaudited Pro Forma Condensed Combined Balance Sheet As of June 30, 2022 (In thousands)

	ProFrac Historical	USWS Historical	USWS Pro Forma Adjustments (Note 2)	Notes	Pro Forma Combined
Liabilities, redeemable noncontrolling interests and stockholders' equity (deficit)					
Current liabilities:	A 150.005	0 40 027	Φ 0.500	(2E)	Ф 244 422
Accounts payable	\$ 178,905	\$ 49,027	\$ 9,500	(2F)	\$ 244,432
A	27 577		7,000	(2G)	27 577
Accounts payable-related party	37,577 159,304	14,301			37,577 173,605
Accrued expenses Current portion of operating lease liabilities	9,160	9,605			18,765
Current portion of finance lease liabilities	9,100	1,168			1,168
Other current liabilities	15,835	1,108	_		15,835
Current portion of long-term debt	51,329	15,001	(8,750)	(2A)	57,580
				(2A)	
Total current liabilities	452,110	89,102	7,750	(2.1)	548,962
Long-term debt	427,961	277,934	(158,423)	(2A)	599,289
			175,000	(2D)	
			(7,000)	(2G)	
			(116,183)	(2H)	
Operating lease liabilities	75,397	8,748			84,145
Finance lease liabilities	_	2,176			2,176
Other liabilities	_	10,660	(7,927)	(2I)	2,652
			(81)	(2J)	
Total liabilities	955,468	388,620	(106,864)		1,237,224
Redeemable noncontrolling interests	2,024,687	26,092	(26,092)	(2K)	2,024,687
Stockholders' equity (deficit)					
Class A Common Stock, \$0.01 par value; 600,000,000 shares authorized, 41,237,003 shares issued and outstanding (Actual Historical); 54,016,039 shares issued and outstanding, As					
Adjusted)	412	1	43	(2E)	540
			62	(2H)	
			8	(2I)	
			14	(2K)	
Class B Common Stock, \$0.01 par value; 400,000,000 shares authorized, 101,133,201 shares					
issued and outstanding, Actual Historical and As Adjusted)	1,011	_	_		1,011
Additional paid-in capital	_	312,571	9,844	(2A)	275,177
			(16,519)	(2B)	
			(174,040)	(2E)	
			(8,000)	(2F)	
			116,121	(2H)	
			7,919	(2I)	
			80	(2J)	
			26,078	(2K)	
			1,123	(2L)	
Accumulated deficit	(1,410,780)	(428,619)	(22,525)	(2C)	(1,435,928)
			428,619	(2E)	
			(1,500)	(2F)	
			(1,123)	(2L)	
Noncontrolling interests	92,863				92,863
Accumulated other comprehensive (loss) income	(36)	_	_		(36)
Total stockholders' deficit	(1,316,530)	(116,047)	366,204		(1,066,373)
Total liabilities, redeemable noncontrolling interests, and stockholders' deficit	\$ 1,663,625	\$ 298,665	\$ 233,248		\$ 2,195,538

 $\label{thm:companying} \textit{ notes are an integral part of these unaudited pro forma financial statements}.$ 

## **ProFrac Holding Corp.**

## Unaudited Pro Forma Condensed Combined Statement of Operations Six Months Ended June 30, 2022 (In thousands, except per share data)

	ProFrac <u>Historical</u>	FTSI <u>Historical</u>	FTSI Pro Forma Adjustments (Note 3)	Notes	IPO and Financing Adjustments (Note 4)	Notes	ProFrac Pro Forma Prior to USWS Acquisition	USWS Historical	USWS Pro Forma Adjustments (Note 2)	Notes	Pro Forma Combined
Revenues	\$934,824	\$ 76,635	\$ _		\$ _		\$1,011,459	\$109,914	\$ _		\$1,121,373
Operating costs and expenses:	_	_	_		_		_	_	_		_
Cost of revenues, exclusive of depreciation, depletion and											
amortization	573,199	59,679	_		_		632,878	95,932	_		728,810
Depreciation, depletion, and amortization	108,280	7,142	5,655 (225)	(3A) (3B)	_		120,852	11,577	11,588	(2M)	144,017
Gain (loss) on disposal of assets, net	1,989	(5)	_	()	_		1,984	2,980	_		4,964
Selling, general and administrative	121,675	14,629	9,890	(3C)	_		147,054	17,778	_		164,832
55			860	(3B)			·				
Impairments and other charges	_	9,890	(9,890)	(3C)	_		_	_	_		_
Total operating costs and expenses	805,143	91,335	6,290				902,768	128,267	11,588		1,042,623
Operating income (loss)	129,681	(14,700)	(6,290)				108,691	(18,353)	(11,588)		78,750
Other (expense) income:	ĺ	, ,	,				ĺ	, , ,	( , ,		,
Interest expense, net	(22,723)	(119)	_		(4,075)	(4A)	(21,350)	(17,530)	6,697	(2N)	(32,561)
					5,567	(4B)			10,414	(2O)	
									(9,625)	(2P)	
									(1,167)	(2Q)	
Loss on extinguishment of debt, net	(17,095)	_	_		_		(17,095)	(1,651)	_		(18,746)
Other (expense) income	9,220	120			_		9,220	2,454			11,674
Reorganization items, net		129					129				129
Income (loss) before income tax provision	99,083	(14,690)	(6,290)		1,492		79,595	(35,080)	(5,269)		39,246
Income tax provision	(4,864)						(4,864)				(4,864)
Net income (loss)	94,219	(14,690)	(6,290)		1,492		74,731	(35,080)	(5,269)		34,382
Less: net (income) loss attributable to ProFrac	(50.065)				70.067	(400					
Predecessor	(79,867)	_	_		79,867	(4C)	_	_	_		_
Less: net loss attributable to noncontrolling interests	8,288	_	_		_		8,288	_	_		8,288
Less: net income attributable to redeemable noncontrolling interests	(16,082)	_	_		(38,070)	(4C)	(54,152)	_	26,319	(2R)	(27,833)
Net income (loss) attributable to ProFrac Holding Corp.	\$ 6,558	\$(14,690)	\$ (6,290)		\$ 43,289		\$ 28,867	(35,080)	\$ 21,050		\$ 14,837
1	φ 0,558	<u>\$(17,090)</u>	<u>Ψ (0,290)</u>		<del>φ +3,269</del>		φ 20,007	(33,080)	φ 21,030		<u>φ 14,037</u>
Earnings per Class A share: Basic											\$ 0.27
Diluted											0.27
Direct											0.27

The accompanying notes are an integral part of these unaudited pro forma financial statements.

## **ProFrac Holding Corp.**

## Unaudited Pro Forma Condensed Combined Statement of Operations Year Ended December 31, 2021 (In thousands, except per share data)

	ProFrac Historical	FTSI Historical	FTSI Pro Forma Adjustments Note 3	Notes	IPO and Financing Adjustments Note 4	Notes	ProFrac Pro Forma Prior to USWS Acquisition	USWS Historical	USWS Pro Forma Adjustments Note 2	Notes	Pro forma Combined
Revenues	\$768,353	\$405,250	\$ —		\$ —		\$1,173,603	\$250,463			\$1,424,066
Operating costs and expenses:	_	_	_		_		_	_	_		_
Cost of revenues, exclusive of depreciation,											
depletion, and amortization	570,122	316,747	_		_		886,869	221,364	_		1,108,233
Depreciation, depletion, and amortization	140,687	53,853	33,928	(3D)	_		227,118	35,444	10,885	(2S)	273,447
			(1,350)	(3E)							
Gain (loss) on disposal of assets, net	9,777	2,195					11,972	(21,896)			(9,924)
Selling, general and administrative	65,592	47,920	4,521	(3F)	_		123,192	32,578	9,500	(2T)	166,393
			5,159	(3E)					1,123	(2U)	
Impairments, litigation settlement and other		4.501	(4.501)	(2E)				25.000			25.000
charges		4,521	(4,521)	(3F)				35,000			35,000
Total operating costs and expenses	786,178	425,236	37,737				1,249,151	302,490	21,508		1,573,149
Operating income (loss)	(17,825)	(19,986)	(37,737)		_		(75,548)	(52,027)	(21,508)		(149,083)
Other (expense) income:											
Interest expense, net	(25,788)	(301)	_		(25,316)	(4D)	(36,013)	(33,370)	21,830	(2V)	(58,734)
					15,392	(4E)			10,402	(2W)	
									(19,250)	(2X)	
									(2,333)	(2Y)	
Loss on extinguishment of debt, net	(515)	_	_		_		(515)	(6,142)	(22,525)	(2Z)	(29,182)
Other (expense) income	404		_				404	20,863	(22,500)	(2AA)	(1,233)
Reorganization items, net		(894)					(894)				(894)
Income (loss) before income tax provision	(43,724)	(21,181)	(37,737)		(9,924)		(112,566)	(70,676)	(55,884)		(239,126)
Income tax benefit (provision)	186	(70)					116	27			143
Net loss	(43,538)	(21,251)	(37,737)		(9,924)		(112,450)	(70,649)	(55,884)		(238,983)
Less: net loss attributable to noncontrolling											
interests	1,118	_	_		_		1,118	44	_		1,162
Less: net loss attributable to redeemable											
noncontrolling interests	_	_			72,619	(4F)	72,619	_	82,506	(2AB)	155,125
Net income (loss) attributable to ProFrac Holding											
	\$ (42,420)	\$ (21,251)	\$ (37,737)		\$ 62,695		\$ (38,713)	\$ (70,605)	\$ 26,622		\$ (82,696)
•											
											\$ (1.53)
Diluted											(,
Net income (loss) attributable to ProFrac Holding Corp. Loss per Class A share: Basic	\$ (42,420)	<u> </u>	\$ (37,737)			(4F)		\$ (70,605)		(2AB)	

The accompanying notes are an integral part of these unaudited pro forma financial statements.

## Notes to Unaudited Pro Forma Condensed Combined Financial Statements (In thousands, except per share information)

### Note 1. Basis of Presentation, anticipated USWS Acquisition, the Offering and Reorganization, and FTSI Acquisition

## Basis of Presentation

For purposes of the unaudited pro forma balance sheet, it is assumed that the USWS Acquisition had taken place on June 30, 2022. For purposes of the unaudited pro forma statements of operations, it is assumed all Transactions described herein had taken place on January 1, 2021.

### Description of transactions

Anticipated USWS Acquisition

On June 21, 2022, the Company entered into a Merger Agreement to acquire USWS. After the consummation of the USWS Acquisition, USWS will become an indirect subsidiary of the Company. The acquisition is expected to be completed in the fourth quarter of 2022, subject to the satisfaction of customary closing conditions, including the approval of USWS stockholders, as described further below.

USWS Acquisition will be accounted for as a business combination under ASC 805, with ProFrac treated as the "acquirer" and USWS treated as the "acquiree" for accounting purposes.

Below is a summary of the key clauses in the Merger Agreement that were given effect to when we prepared the pro forma financial information:

- At the Effective Time, each share of USWS Common Stock will be converted automatically into the right to receive 0.3366 shares of ProFrac Class A Common Stock, as adjusted for the 1-for-6 reverse stock split implemented by USWS on August 4, 2022.
- USWS will take all requisite action so that, effective as of immediately prior to the Effective Time, all shares of USWS Series A Preferred Stock and all USWS Equity Linked Convertible Notes outstanding at such time will be converted into a certain number of shares of USWS Common Stock.
- At the Effective Time, the USWS Term C Loan Warrants issued and outstanding immediately prior to the Effective Time (which shall be
  held by ProFrac pursuant to the Warrant Sale) will be automatically canceled and will cease to exist and no consideration will be delivered in
  exchange therefor.
- At the Effective Time, each USWS Restricted Share that is outstanding and unvested immediately prior to the Effective Time will be
  canceled in exchange for the right to receive the Merger Consideration and, in lieu of any fractional shares, cash.
- Immediately prior to the Effective Time, each USWS DSU will be canceled and converted into the right to receive the Merger Consideration and, in lieu of any fractional shares, cash.
- Immediately prior to the Effective Time, all outstanding USWS Pool A Performance Awards and USWS Pool B Performance Awards will be canceled and converted into the right to receive a certain amount of Merger Consideration and such amount will be dependent upon the holder and the type of the performance award.

Based on the number of shares of USWS Common Stock (after giving effect to the conversion of the USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes), USWS Restricted Shares, USWS DSUs, USWS Pool A Performance Awards and USWS Pool B Performance Awards outstanding as of the date of the accompanying proxy statement/information statement/prospectus, an aggregate of approximately 12.8 million shares of ProFrac Class A Common Stock would be issued in exchange therefor.

The pro forma adjustments related to the USWS Acquisition are preliminary and based on estimates of the fair value and useful lives of the assets acquired and liabilities assumed and have been prepared to illustrate the estimated effect of the transaction and certain other adjustments. The final determination of the purchase price allocation for the USWS Acquisition will be based on the fair values of assets acquired and liabilities assumed as of the date the proposed transaction closes. The Company will continue to assess its determination of fair value of the assets acquired and liabilities assumed during the USWS Acquisition during the measurement period.

Please refer to Note 2 for adjustments and assumptions related to USWS Acquisition.

### FTSI Acquisition

On March 4, 2022, ProFrac LLC acquired FTSI for a purchase price of approximately \$405.7 million, consisting of cash consideration of \$332.8 million and certain equity interests in ProFrac LLC of \$72.9 million.

FTSI is reflected in the ProFrac historical balance sheet as of June 30, 2022 based on our preliminary assessment of the fair value of assets and liabilities assumed. Accordingly, no adjustments are made to our unaudited pro forma condensed combined balance sheet to account for the FTSI Acquisition.

We reflected certain FTSI Acquisition related adjustments in the unaudited pro forma condensed combined statements of operations for the year ended December 31, 2021, and the six months ended June 30, 2022 as if the acquisition had consummated as of January 1, 2021.

The pro forma adjustments related to the FTSI Acquisition are preliminary and based on estimates of the fair value and useful lives of the assets acquired and liabilities assumed and have been prepared to illustrate the estimated effect of the transaction and certain other adjustments. The final purchase price allocation will be determined when the Company has completed the detailed valuations and necessary calculations during the measurement period (up to one year from the date the acquisition closes). The final allocation could differ materially from the preliminary allocation used in the pro forma adjustments.

Please refer to Note 3 for adjustments and assumptions related to the FTSI Acquisition.

### IPO and Refinancing

On May 17, 2022, ProFrac Holding Corp. completed the IPO of 16,000,000 shares of Class A Common Stock, par value \$0.01 per share (the "Class A Common Stock") at a price to the public of \$18.00 per share.

Immediately prior to completion of the IPO and in connection with the IPO, we engaged in the following additional series of transactions, which, together with the IPO, is referred to as "IPO and Reorganization"):

- All of the membership interests in ProFrac LLC held by the then-existing owners of ProFrac LLC (including the THRC FTSI Related Equity) were converted into a single class of ProFrac LLC Units;
- ProFrac Holding Corp. issued to each ProFrac LLC Unit Holder a number of shares of ProFrac Class B Common Stock equal to the number of ProFrac LLC Units held by such ProFrac LLC Unit Holder following the ProFrac IPO in exchange for a cash payment equal to the par value of such shares;
- On June 6, 2022, an over-allotment option was exercised resulting in an additional 2,228,153 shares of ProFrac Class A Common Stock being priced at \$18.00 per share.

Please refer to Note 4 for adjustments and assumptions related to IPO and Reorganization.

Refinancing Transactions (together with IPO and Reorganization, collectively referred to as "IPO and Refinancing")

The following financing transactions occurred in the six months ended June 30, 2022:

- In February 2022, ProFrac LLC entered an agreement to expand its Term Loan by \$48.0 million with debt issuance costs of \$0.2 million. The
  proceeds of which were used to purchase all the series A-1 and B-1 preferred units of BPC for \$46.0 million.
- On March 4, 2022, ProFrac LLC entered into the New Term Loan Credit Facility. At closing of the New Term Loan Credit Facility, ProFrac LLC borrowed \$450.0 million. The interest rate was 9.50% as of the date of funding. The proceeds from the New Term Loan Credit Facility were used to repay in full and terminate the outstanding indebtedness under the existing term loan of ProFrac LLC and to fund the FTSI acquisition.
- On March 4, 2022, the Company entered into a \$45.8 million subordinated promissory note with Equify Financial ("Equify Bridge Note"). The Equify Bridge Note accrues interest at a rate of 1.00% and matures in March 2027. The Equify Bridge Note refinanced the Best Flow Credit Facility, Best Flow Note and Alpine Promissory Note.
- On March 4, 2022, ProFrac LLC issued the Backstop Note and the Closing Date Note in an aggregate principal amount of \$44.0 million, and an
  interest rate of 1.74%.
- On May 12, 2022, ProFrac Holding Corp, completed its IPO. The IPO and the exercise of the over-allotment option exercised generated combined
  net proceeds of \$301.7 million, after deducting underwriter discounts and commissions and estimated offering costs. The Company repaid the debt
  below using the net proceeds from the IPO:
  - a. paid down \$143.8 million of outstanding borrowings under the New Term Loan Credit Facility;
  - b. paid down \$14.0 million of outstanding borrowings under the New ABL.
  - c. fully paid \$22.0 million of the Backstop Note;
  - d. paid down \$22.0 million of the outstanding borrowings of the Closing Date Note; and
  - e. paid down \$20.8 million of the outstanding borrowings of the Equify Bridge Note.

### Note 2. Pro Forma adjustments and assumptions related to the USWS Acquisition

### **Estimated Merger Consideration**

On June 21, 2022, the Company announced the acquisition of USWS. The estimated value of the Merger Consideration for the purpose of this unaudited pro forma condensed combined financial information is approximately \$448.0 million based on the \$21.54 per share closing price of ProFrac Class A Common Stock as of August 25, 2022. The actual value of Merger Consideration will change based on changes in the market share price of ProFrac Class A Common Stock and the number of shares of USWS Common Stock outstanding prior to transaction closing (after giving effect to the conversion of the USWS Series A Preferred Stock and USWS Equity Linked Convertible Notes), USWS Restricted Shares, USWS DSUs, USWS Pool A Performance Awards and USWS Pool B Performance Awards outstanding on the closing date of the Merger.

The following table summarizes the components of the estimated Merger Consideration reflected in the unaudited pro forma condensed combined financial statements (in thousands, except share and per share data):

Total ProFrac's Class A shares issued in exchange for USWS outstanding equity		
and in settlement of the Equity Linked Convertible Notes	1	1,933,989
ProFrac's share price on August 25, 2022	\$	21.54
Total Acquisition Consideration (equity)	\$	257,058
Repayment of USWS's debt		173,847
Consideration transferred related to the settlement of equity-based compensation		17,125
Total pro forma Merger Consideration	\$	448,030

The final value of the Merger Consideration could significantly differ from the amounts presented in the unaudited pro forma condensed combined financial information due to movements in the market price of ProFrac Class A Common Stock through the closing of the USWS Acquisition. A sensitivity analysis related to the fluctuation in ProFrac Class A Common Stock price was performed to assess the impact of a hypothetical change of 10% on the closing price of such ProFrac Class A Common Stock on the estimated value of the Merger Consideration and goodwill as of an assumed closing date of the USWS Acquisition.

The following table shows the estimated Merger Consideration and increase (decrease) in net assets acquired resulting from a change in share price of the ProFrac Class A Common Stock (amounts in thousands except share price):

		Estimated	Increase (decrease) in
Change in share price	Share price	purchase consideration	net assets acquired
Increase of 10%	\$ 23.69	\$ 475,448	\$ 27,418
Decrease of 10%	\$ 19.39	\$ 420,612	\$ (27,418)

### Allocation of estimated Merger Consideration

Under the acquisition method of accounting, the identifiable assets acquired and liabilities assumed of USWS will be recognized and measured at fair value as of the closing date of the transaction and added to those of ProFrac. The determination of fair value used in the transaction-related adjustments presented herein are preliminary and based on management estimates of the fair value and useful lives of the assets acquired and liabilities assumed and have been prepared to illustrate the estimated effect of the Merger. The costs of finite-lived intangible assets are amortized through expense over their estimated lives. The final allocation of the Merger Consideration, upon the completion of the Merger, will be based on USWS's assets acquired and liabilities assumed as of the acquisition date, and will depend on a number of factors that cannot be predicted with certainty at this time. Therefore, the actual allocations will differ from the transaction accounting adjustments presented herein. The allocation is dependent upon certain valuation and other studies that have not yet been completed. Accordingly, the pro forma allocation of the Merger Consideration will be subject to further adjustments as additional information becomes available and as additional analyses and final valuations are completed. There can be no assurances that these additional analyses and final valuations will not result in significant changes to the estimates of fair value set forth below.

The following table sets forth a preliminary allocation of the estimated value of the Merger Consideration to the identifiable tangible and intangible assets acquired and liabilities assumed of USWS based on USWS's unaudited consolidated balance sheet as of June 30, 2022, with the excess recorded to goodwill (in thousands):

Assets acquired:	
Cash and cash equivalents	\$ 17,268
Accounts receivable	31,721
Prepaid expense and other assets	7,109
Inventories	8,074
Property, plant and equipment	240,142
Operating lease ROU asset	18,197
Finance lease ROU asset	3,246
Intangible assets	226,410
Goodwill	_
Other assets	1,119
Total assets acquired	553,286
Liabilities assumed:	
Accounts payable	57,027
Accrued expenses	14,301
Operating lease liability current	9,605
Finance lease liability current	1,168
Current portion of debt	6,251
Long-term debt	3,328
Operating lease liability non-current	8,748
Finance lease liability non-current	2,176
Other non-current liabilities	2,652
Total liabilities assumed	105,256
Net assets acquired	\$ 448,030

The following footnotes summarize the pro forma adjustments for the USWS Acquisition (in thousands).

### **Balance Sheet adjustments**

- (2A) Reflects the repayment of long-term debt by ProFrac in connection with the USWS Acquisition and the reversal of the unamortized debt issuance cost.
- (2B) Reflects the buyout of the USWS Term C Loan / RDO Warrant, which was an equity instrument in USWS historical balance sheet. This adjustment resulted in a decrease in additional paid-in capital ("APIC").
- (2C) Reflects the penalty that the Company would incur for the early settlement of the USWS Term C Loan.
- (2D) Reflects the estimated long-term debt financing the Company intends to draw to repay USWS existing debt and complete the USWS Acquisition.
- (2E) Reflects the fair value adjustments for identified long-lived assets related to our preliminary purchase price allocation. The estimated values for identified long-lived assets, including intangibles and goodwill could materially differ in our finalized purchase price allocation. This adjustment also reflects the reversal of historical goodwill and issuance of additional Class A Common Stock.
- (2F) Reflects transaction costs expected to be incurred by ProFrac and USWS in the amount of \$1.5 million and \$8.0 million, respectively, in connection with the USWS Acquisition.

- (2G) Reflects debt issuance cost in connection with the debt financing of \$175.0 million needed for the USWS Acquisition. The debt issuance cost is estimated to be at 4% of the debt principal amount based on our current estimation.
- (2H) Reflects the conversion of the USWS Equity Linked Convertible Notes into 6,178,856 shares of ProFrac Class A Common Stock at par value of \$0.01. The remaining is recognized in APIC.
- (2I) Reflects the conversion of USWS's Pool A Performance Awards, Pool B Performance Awards, DSU and RSU into 845,047 shares of ProFrac Class A Common Stock at par value of \$0.01. The difference between the amount of these equity instruments in USWS historical financial statements and the par value of ProFrac Class A Common Stock is recognized in APIC.
- (2J) Reflects the rollover of USWS warrant liabilities with fair value adjustment recognized in APIC.
- (2K) Reflects the conversion of USWS's Series A shares into 1,447,926 shares of ProFrac Class A Common Stock at par value of \$0.01. The remaining is recognized in APIC.
- (2L) Reflects the adjustment to share-based compensation expense related to the RSU and RSA that USWS granted to certain of its board members that contained discretionary change-in-control acceleration clauses, but were not vested as of June 30, 2022. This portion is not included in the calculation of the estimated Merger Consideration because the accelerated vesting of the awards based on the Merger Agreement was determined to primarily benefit the Company.

### Statement of Operations adjustments for the six months ended June 30, 2022

- (2M) Reflects additional depreciation, depletion, and amortization expenses, from the step-up of assets based on our preliminary purchase price allocation. The estimated values for identified long-lived assets, including intangibles and goodwill could materially differ from our finalized purchase price allocation.
- (2N) Reflects the elimination of interest expense resulting from the repayment of debt as described in Note (2A).
- (20) Reflects the elimination of interest expense in connection with the cancellation of the USWS Equity Linked Convertible Notes and conversion into ProFrac Class A Common Stock as described in Note (2H).
- (2P) Reflects the increase in interest expense at an 11% rate related to debt financing described in Note (2D).

A hypothetical increase or decrease in floating interest rates of 0.125% on the long-term debt financing would increase or decrease our pro forma interest expense by \$0.1 million for the six months ended June 30, 2022.

- (2Q) Reflects the amortization of the debt issuance cost described in Note (2G) related to the debt financing over the assumed term of 3 years on a straight-line basis.
- (2R) Reflects the adjustments to the net income (loss) attributable to the redeemable noncontrolling interests and ProFrac Holding Corp. due to the issuance of additional shares of Class A Common Stock in connection with the USWS Acquisition.

## Statement of Operations adjustments for the year ended December 31, 2021

(2S) – Reflects additional depreciation, depletion, and amortization expenses, from the step-up of assets based on our preliminary purchase price allocation. The estimated values for identified long-lived assets, including intangibles and goodwill could materially differ from our finalized purchase price allocation.

- (2T) Reflects the transaction costs that would be incurred by ProFrac and USWS in the amount of \$1.5 million and \$8.0 million, respectively, in connection with this transaction that are not expected to recur.
- (2U)—Reflects the adjustment to share-based compensation expense related to the RSU and RSA that USWS granted to certain of its board members that contained discretionary change-in-control acceleration clauses, but were not vested as of June 30, 2022. This portion is not included in the calculation of the estimated Merger Consideration because the accelerated vesting of the awards based on the Merger Agreement was determined to primarily benefit the Company. This expense is not expected to recur.
- (2V) Reflects the elimination of interest expense resulting from the repayment of debt as described in Note (2A).
- (2W) Reflects the elimination of interest expense in connection with the cancellation of the USWS Equity Linked Convertible Notes and conversion into ProFrac Class A Common Stock as described in Note (2H).
- (2X) Reflects the increase in interest expense related to debt financing described in Note (2D).

A hypothetical increase or decrease in floating interest rates of 0.125% on the long-term debt financing would increase or decrease our pro forma interest expense by \$0.2 million for the year ended December 31, 2021.

- (2Y) Reflects the amortization of the debt issuance cost described in Note (2G) related to the debt financing over the assumed term of 3 years on a straight-line basis.
- (2Z) Reflects the prepayment penalty to be incurred by the Company to early repay USWS Term C Loan. This expense is not expected to recur.
- (2AA) Reflects the elimination of intercompany income recognized by USWS related to a license charge that was recognized by USWS in other income. The license charge was capitalized within assets under construction by ProFrac and has not yet been depreciated in the Company's historical financial statements. This transaction is not expected to recur.
- (2AB) Reflects the adjustments to the net income (loss) attributable to the redeemable noncontrolling interests and ProFrac Holding Corp. due to the issuance of additional shares of Class A Common Stock in connection with the USWS Acquisition.

### Note 3. Pro forma adjustments and assumptions related to FTSI Acquisition

On March 4, 2022, ProFrac LLC acquired FTSI for a purchase price of approximately \$405.7 million, consisting of cash consideration of \$332.8 million and certain equity interests in ProFrac LLC of \$72.9 million.

We have accounted for the acquisition of FTSI using the acquisition method of accounting. We used our best estimates and assumptions to assign fair value to the tangible and intangible assets expected to be acquired and liabilities expected to be assumed at the acquisition date. The following table summarizes the allocation of the preliminary purchase price (in thousands):

Assets acquired:	
Cash and cash equivalents	\$ 53,771
Accounts receivable	89,268
Prepaid expense and other assets	4,037
Inventories	42,344
Property, plant and equipment	307,113
Operating lease ROU asset	2,748
Intangible assets	1,239
Other assets	1,583
Total assets acquired	502,103
Liabilities assumed:	
Accounts payable	62,985
Accrued expenses	19,308
Operating lease liability current	1,235
Current portion of debt	10,136
Other current liabilities	309
Operating lease liability non-current	1,512
Other non-current liabilities	928
Total liabilities assumed	96,413
Net assets acquired	\$ 405,690

The allocation of the purchase price to FTSI's net tangible assets and liabilities and identifiable intangible assets as of March 4, 2022, is preliminary and subject to revisions to the fair value calculations for the identifiable assets and liabilities. The determination and allocation of the purchase consideration are subject to change during the measurement period, up to one year from the date the acquisition closed.

The following unaudited pro forma results of operations have been prepared as though the FTSI Acquisition had been completed on January 1, 2021. Pro forma amounts are based on the purchase price allocation of the significant acquisition and are not necessarily indicative of the results that may be reported in the future.

### Statement of Operations adjustments for the six months ended June 30, 2022

- (3A) Reflects additional depreciation, depletion, and amortization expenses, from the step-up of assets based on our preliminary purchase price allocation. The estimated values for identified long-lived assets, including intangibles and goodwill could materially differ from our finalized purchase price allocation.
- (3B) Reflects the reduction in depreciation expense and incremental rent expense associated with the sale leaseback of FTSI real estate assets.
- (3C) Reflects the reclassification of FTSI's various transaction expenses to selling, general and administrative expense to reflect presentation of the Company.

### Statement of Operations adjustments for the year ended December 31, 2021

(3D) – Reflects additional depreciation, depletion, and amortization expenses, from the step-up of assets based on our preliminary purchase price allocation. The estimated values for identified long-lived assets, including intangibles and goodwill could materially differ from our finalized purchase price allocation.

- (3E) Reflects the reduction in depreciation expense and incremental rent expense associated with the sale leaseback of FTSI real estate assets.
- (3F) Reflects the reclassification of FTSI's various transaction expenses to selling, general and administrative expense to reflect presentation of the Company.

### Note 4. Pro forma adjustments and assumptions related to the refinancing transactions and the IPO

The IPO and Refinancing transactions as described in Description of Transactions in Note 1 occurred in the six months ended June 30, 2022 and the Company has recognized them in the historical unaudited consolidated balance sheet as of June 30, 2022.

In preparing the pro forma statements of operations for the six months period ended June 30, 2022, and for the year ended December 31, 2021, the Company made the assumptions that IPO and Refinancing Transactions occurred on January 1, 2021 and recorded the corresponding pro forma adjustments.

## Pro forma adjustments for the six months ended June 30, 2022

(4A) – The following pro forma adjustments are reflected on our statement of operations related to the Refinancing transactions described above for the six months ended June 30, 2022 (in thousands):

Incremental interest for \$48.0 million expansion of existing term loan at 9.75% interest	\$ (513)
Incremental interest for \$450.0 million New Term Loan Credit Facility at 9.50% interest	(7,379)
Incremental interest for \$44.0 million Backstop Note and Closing Date Note at 1.74%	
interest	(132)
Reduction of interest for \$171.4 million existing term loans at 8.75% interest	2,588
Reduction of interest for \$48.0 million existing term loans at 9.75% interest	808
Incremental interest for \$45.8 million Equify Bridge Note at 1.0% interest	(79)
Reduction of interest for Alpine Promissory Note, Best Flow Credit Facility, and Best	
Flow Note at 8.00% interest	632
Total	\$(4,075)

A hypothetical increase or decrease in floating interest rates of 0.125% on the New Term Loan Credit Facility would increase or decrease our pro forma interest expense by \$0.3 million for the six months ended June 30, 2022.

(4B) – Reflects the pro forma reduction in interest expense for the debt facilities that were repaid with net proceeds from the IPO for the six months ended June 30, 2022, as summarized in table below:

Reduction of interest for \$ 143.8 million lower New Term Loan Credit Facility at 9.5%	\$4,940
Reduction of interest for \$16.9 million lower New ABL Credit Facility at 4.5%	274
Reduction of interest for \$ 22.0 million lower Backstop Note at 1.74%	139
Reduction of interest for \$ 22.0 million lower Closing Date Note at 1.74%	139
Reduction of interest for \$ 20.8 million lower Equify Bridge Note at 1.0%	75
Total	\$5,567

(4C) – Reflects the adjustments to the net income (loss) attributable to the redeemable noncontrolling interests and ProFrac Holding Corp. due to the issuance of Class A and Class B shares in connection with the IPO and Reorganization.

## Pro forma adjustments for the year ended December 31, 2021

(4D) – The following pro forma adjustments are reflected on our statement of operations related to the Refinancing transactions for the year ended December 31, 2021 (in thousands):

Incremental interest for \$48.0 million expansion of existing term loan at 9.75% interest	\$ (4,680)
Incremental interest for \$450.0 million New Term Loan Credit Facility at 9.50% interest	(42,750)
Incremental interest for \$44.0 million Backstop Note and Closing Date Note at 1.74%	
interest	(766)
Reduction of interest for \$171.4 million existing term loans at 8.75% interest	14,994
Reduction of interest for \$48.0 million existing term loans at 9.75% interest	4,680
Incremental interest for \$45.8 million Equify Bridge Note at 1.0% interest	(458)
Reduction of interest for Alpine Promissory Note, Best Flow Credit Facility, and Best	
Flow Note at 8.00% interest	3,664
Total	\$(25,316)

A hypothetical increase or decrease in floating interest rates of 0.125% on the New Term Loan Credit Facility would increase or decrease our pro forma interest expense by \$0.6 million for the year ended December 31, 2021.

(4E) – Reflects the pro forma reduction in interest expense for the debt facilities that were repaid with net proceeds from the IPO forthe year ended December 31, 2021, as summarized in table below:

Reduction of interest for \$ 143.8 million lower New Term Loan Credit Facility at 9.5%	\$ 13,660
Reduction of interest for \$16.9 million lower New ABL Credit Facility at 4.5%	758
Reduction of interest for \$ 22.0 million lower Backstop Note at 1.74%	383
Reduction of interest for \$ 22.0 million lower Closing Date Note at 1.74%	383
Reduction of interest for \$ 20.8 million lower Equify Bridge Note at 1.0%	208
Total	\$ 15,392

(4F) – Reflects the adjustments to the net income (loss) attributable to the redeemable noncontrolling interests and ProFrac Holding Corp. due to the issuance of Class A and Class B shares in connection with the IPO and Reorganization.

## Note 5. Pro forma earnings (loss) per share

The following table reflects pro forma basic and diluted earnings (loss) per common share for the issuance of shares of Class A Common Stock in the corporate reorganization and the IPO for the six months ended June 30, 2022 as shown below (in thousands, except for per share amounts):

	ended e 30, 2022
BASIC	
Numerator:	
Net income attributable to stockholders	\$ 14,837
Denominator:	
Class A shares issued in the corporate reorganization and the IPO	41,237
Class A shares issued in the USWS Acquisition	12,779
Total shares issued	 54,016
Basic earnings per share	\$ 0.27
DILUTED	
Numerator:	
Net income attributable to stockholders	\$ 14,837
Effect of dilutive securities	_
Diluted net income attributable to stockholders	\$ 14,837
Denominator:	
Basic weighted average shares outstanding	54,016
Effect of dilutive securities	22
Diluted weighted average shares outstanding	 54,038
Diluted earnings per share	\$ 0.27

The following table reflects pro forma basic and diluted earnings (loss) per common share for the issuance of shares of Class A Common Stock in the corporate reorganization and the Offering for the year ended December 31, 2021 as shown below (in thousands, except for per share amounts):

	Year ended December 31, 2021	
BASIC		
Numerator:		
Net loss attributable to stockholders	\$	(82,696)
Denominator:		
Class A shares issued in the corporate reorganization and the IPO		41,237
Class A shares issued in the USWS Acquisition		12,779
Total shares issued		54,016
Basic loss per share	\$	(1.53)
DILUTED		
Numerator:		
Net loss attributable to stockholders	\$	(82,696)
Effect of dilutive securities		
Diluted net loss attributable to stockholders	\$	(82,696)
Denominator:		
Basic weighted average shares outstanding		54,016
Effect of dilutive securities		_
Diluted weighted average shares outstanding		54,016
Diluted loss per share	\$	(1.53)

Basic earnings (loss) per Class A share measures the performance of an entity over the reporting period. Diluted earnings (loss) per Class A share measures the performance of an entity over the reporting period while giving effect to all potentially dilutive common shares that were outstanding during the period. The Company uses the "if-converted" method to determine the potential dilutive effect of its Class B Common Stock. The Company uses the "treasury stock" method to determine the potential dilutive effect of its management incentive units. On a pro forma basis, for the six months ended June 30, 2022 and the year ended December 31, 2021 the Class B common stock was not recognized in dilutive earnings (loss) per Class A share calculations as it would have been antidilutive. Management incentive units and USWS warrant rollover were only recognized in dilutive earnings (loss) per Class A share calculations for the six months ended June 30, 2022 as they would have been antidilutive for the year ended December 31, 2021.

#### REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

Board of Directors FTS International, Inc.

#### Report on 2021 financial statements

#### Opinion

We have audited the consolidated financial statements of FTS International, Inc. (a Delaware corporation) and subsidiaries (the "Company"), which comprise the consolidated balance sheet as of December 31, 2021, and the related consolidated statements of operations, cash flows, and stockholders' equity (deficit) for the year then ended (Successor), and the related notes to the financial statements.

In our opinion, the accompanying consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2021, and the results of its operations and its cash flows for the year then ended (Successor), in accordance with accounting principles generally accepted in the United States of America.

#### Basis for opinion

We conducted our audit of the consolidated financial statements in accordance with auditing standards generally accepted in the United States of America (US GAAS). Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

#### Responsibilities of management for the financial statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for one year after the date the financial statements are available to be issued.

#### Auditor's responsibilities for the audit of the financial statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with US GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with US GAAS, we:

• Exercise professional judgment and maintain professional skepticism throughout the audit.

- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not
  for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

#### Report on 2020 and 2019 financial statements

#### Opinion on the financial statements

We have audited the accompanying consolidated balance sheet of FTS International, Inc. (a Delaware corporation) and subsidiaries (the "Company") as of December 31, 2020 (Successor), the related consolidated statements of operations, cash flows, and stockholders' equity (deficit) for the period from November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and the year ended December 31, 2019 (Predecessor), and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the

Company as of December 31, 2020 (Successor), and the results of its operations and its cash flows for the period November 20, 2020 through December 31, 2020 (Successor), the period from January 1, 2020 through November 19, 2020 (Predecessor), and the year ended December 31, 2019 (Predecessor), in conformity with accounting principles generally accepted in the United States of America.

#### **Emergence from bankruptcy**

As discussed in Note 3 to the consolidated financial statements, on November 4, 2020, the United States Bankruptcy Court for the District of Delaware entered an order confirming the plan for reorganization, which became effective on November 19, 2020. Accordingly, the accompanying consolidated financial statements have been prepared in conformity with Financial Accounting Standards Board Accounting Standards Codification® (ASC) 852, *Reorganizations*, for the Successor as a new entity with assets, liabilities and a capital structure having carrying amounts not comparable with Predecessor periods, as described in Note 3.

# Basis for opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud.

Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ GRANT THORNTON LLP

We have served as the Company's auditor since 2015.

Dallas, Texas March 2, 2022

# FTS INTERNATIONAL, INC. CONSOLIDATED STATEMENTS OF OPERATIONS

	Suc	ccessor	Predeces	sor
(In millions, except per share amounts)	Year Ended December 31, 2021	Period from November 20, through December 31, 2020	Period from January 1, through <u>November 19,</u> 2020	Year Ended December 31, 2019
Revenue				
Revenue	\$ 405.2	\$ 22.6	\$ 239.6	\$ 775.7
Revenue from related parties			0.7	0.9
Total revenue	405.2	22.6	240.3	776.6
Operating expenses				
Costs of revenue (excluding depreciation of \$51.9, \$4.6, \$65.2 and \$82.5, respectively, included in depreciation and amortization				
below)	316.7	24.1	197.2	573.9
Selling, general and administrative	47.9	4.7	47.8	89.1
Depreciation and amortization	53.9	4.8	68.5	90.0
Impairments and other charges	4.5	0.3	34.1	74.6
Loss (gain) on disposal of assets, net	2.2		0.1	(1.4)
Total operating expenses	425.2	33.9	347.7	826.2
Operating loss	(20.0)	(11.3)	(107.4)	(49.6)
Interest expense, net	(0.3)	_	(22.1)	(30.7)
Gain on extinguishment of debt, net	_	_	2.0	1.2
Equity in net income of joint venture affiliate	_	_	_	0.6
Gain on sale of equity interest in joint venture affiliate	_	_	_	7.0
Reorganization items, net	(0.9)	(2.1)	103.3	
Loss before income taxes	(21.2)	(13.4)	(24.2)	(71.5)
Income tax expense			0.2	1.4
Net loss	\$ (21.2)	<u>\$ (13.4)</u>	\$ (24.4)	\$ (72.9)
Net loss attributable to common stockholders	\$ (21.2)	\$ (13.4)	\$ (24.4)	\$ (72.9)
Basic and diluted loss per share attributable to common stockholders	\$ (1.51)	\$ (0.96)	\$ (4.54)	\$ (13.40)
Shares used in computing basic and diluted loss per share (in thousands)	14,036	13,990	5,377	5,440
			•	

The accompanying notes are an integral part of these consolidated financial statements.

# FTS INTERNATIONAL, INC. CONSOLIDATED BALANCE SHEETS

(In millions, except share amounts)	Dec	ember 31, 2021	ember 31, 2020
ASSETS			
Current assets			
Cash and cash equivalents	\$	85.1	\$ 94.0
Accounts receivable, net		67.3	26.9
Inventories		38.5	29.0
Prepaid expenses and other current assets		4.3	 19.5
Total current assets		195.2	169.4
Property, plant, and equipment, net		134.0	132.3
Operating lease right-of-use assets		2.4	4.5
Intangible assets, net		6.9	7.4
Other assets		1.5	 1.4
Total assets	\$	340.0	\$ 315.0
LIABILITIES AND STOCKHOLDERS' EQUITY			
Current liabilities			
Accounts payable	\$	70.6	\$ 26.9
Accrued expenses		18.2	12.5
Current portion of operating lease liabilities		1.3	3.0
Other current liabilities		0.3	 0.3
Total current liabilities		90.4	42.7
Long-term debt		_	_
Operating lease liabilities		1.7	3.3
Other liabilities		1.0	 2.4
Total liabilities		93.1	 48.4
Commitments and contingencies (Note 15)			
Stockholders' equity			
Preferred stock, \$0.01 par value, 5,000,000 shares authorized		_	_
Common stock Class A, \$0.01 par value, 49,000,000, authorized, 13,822,205 issued and outstanding at December 31, 2021 13,677,664 issued and outstanding at December 31, 2020 Common stock Class B, \$0.01 par value, 1,000,000, authorized, 312,306 issued and outstanding at December 31, 2021 312,306 issued and			
outstanding at December 31, 2020		0.1	0.1
Additional paid-in capital		281.4	279.9
Accumulated deficit		(34.6)	(13.4)
Total stockholders' equity		246.9	 266.6
Total liabilities and stockholders' equity	\$	340.0	\$ 315.0

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these consolidated financial statements}.$ 

# FTS INTERNATIONAL, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS

		Succ	essor	Ī		Predecessor		
(In millions)	Dece	r Ended mber 31,	Nove th Dece	ember 20, crough ember 31, 2020	J	eriod from anuary 1, through vember 19, 2020	Dece	er Ended ember 31, 2019
Cash flows from operating activities								
Net loss	\$	(21.2)	\$	(13.4)	\$	(24.4)	\$	(72.9)
Adjustments to reconcile net loss to net cash provided by (used in)								
operating activities:								
Depreciation and amortization		53.9		4.8		68.5		90.0
Stock-based compensation		4.1		0.4		10.9		15.4
Amortization of debt discounts and issuance costs		_		_		2.0		1.8
Impairment of assets		_		_		_		9.7
Loss (gain) on disposal of assets, net		2.2		_		0.1		(1.4)
Gain on extinguishment of debt, net		_		_		(2.0)		(1.2)
Non-cash provision for supply commitment charges		_		_		9.1		58.5
Cash paid to settle supply commitment charges		_		_		(31.3)		(17.6)
Gain sale of equity interest in joint venture affiliate		_		_		`_ `		(7.0)
Inventory write-down		_		_		5.1		6.4
Non-cash reorganization items		_		_		(118.7)		_
Other non-cash items		0.1		_		0.9		4.7
Changes in energting assets and liabilities:								
Changes in operating assets and liabilities: Accounts receivable		(40.6)		3.2		46.0		79.0
		(9.5)		2.2		6.9		14.0
Inventories		( )						
Prepaid expenses and other assets		1.2 27.2		(0.1)		(3.8)		(1.5)
Accounts payable		4.3		5.5 0.3		(13.9)		(47.3)
Accrued expenses and other liabilities					_	(1.9)		(6.7)
Net cash provided by (used in) operating activities		21.7		2.9		(46.5)		123.9
Cash flows from investing activities								
Capital expenditures		(43.9)		(1.5)		(19.6)		(54.4)
Proceeds from disposal of assets		3.1		_		0.2		3.3
Proceeds from sale of equity interest in joint venture affiliate		_		_		_		30.7
Net cash used in investing activities		(40.8)		(1.5)		(19.4)		(20.4)
					_			
Cash flows from financing activities								
Repayments of long-term debt		_		_		(20.6)		(46.4)
Payments to secured debtholders				_		(30.7)		
Repurchases of common stock		_		_				(9.9)
Taxes paid related to net share settlement of equity awards		(2.5)		_		(0.3)		(2.0)
Payments of revolving credit facility issuance costs				_		(0.2)		
Net cash used in financing activities		(2.5)		_		(51.8)		(58.3)
Net (decrease) increase in cash, cash equivalents, and restricted cash		(21.6)		1.4	<u>—</u>	(117.7)		45.2
Cash, cash equivalents, and restricted cash at beginning of period		106.7		105.3		223.0		177.8
Cash, cash equivalents, and restricted cash at end of period	\$	85.1	\$	106.7	<u>-</u> -	105.3	\$	223.0
Supplemental cash flow information:	Ф	03.1	Ф	100.7	•	103.3	Ф	223.0
Interest paid	\$		\$		\$	14.6	\$	31.0
Income tax payments, net	\$	0.2	\$		\$	0.4	\$	2.5
	Ψ	0.2	Ψ		ф	٠.٠	Ψ	2.3
Supplemental disclosure of noncash investing activities:	ø.	17.0	0	0.5		0.2	0	0.0
Capital expenditures included in accounts payable	\$	17.0	\$	0.5	\$	0.2	\$	0.9
Operating lease liabilities incurred from obtaining right-of-use assets	\$	1.5	\$	0.1	\$	0.6	\$	11.0
Operating lease liabilities and right-of-use assets derecognized due to lease terminations	¢	0.8	¢		ø	10.2	•	2.2
icase terminations	\$	0.8	\$		\$	10.2	\$	3.2

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these consolidated financial statements}.$ 

# FTS INTERNATIONAL, INC. CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (DEFICIT)

	Commo	n Stock			Total
			Additional Paid-in	Accumulated	Stockholders' Equity
(Dollars in millions and shares in thousands)	Shares	Amount	Capital	Deficit	(Deficit)
Balance at January 1, 2019 Predecessor	5,472	\$ 36.4	\$ 4,378.4	\$ (4,307.9)	\$ 106.9
Net loss	_	_	_	(72.9)	(72.9)
Cumulative effect of accounting change	_	_	_	0.1	0.1
Activity related to stock plans	31	_	13.5	_	13.5
Repurchases of common stock	(148)		(9.9)		(9.9)
Balance at December 31, 2019 Predecessor	5,355	36.4	4,382.0	(4,380.7)	37.7
Net loss	_	_	_	(24.4)	(24.4)
Activity related to stock plans	101		25.9		25.9
Balance at November 19, 2020 Predecessor	5,456	36.4	4,407.9	(4,405.1)	39.2
Cancellation of Predecessor Equity	(5,456)	(36.4)	(4,407.9)	4,405.1	(39.2)
Balance at November 19, 2020 Predecessor		<u>\$</u>	<u>\$</u>	<u>\$</u>	\$ (0.0)
Issuance of Successor Common Stock	13,990	0.1	279.5		279.6
Balance at November 19, 2020 Successor	13,990	0.1	279.5		279.6
Net loss	_	_	_	(13.4)	(13.4)
Activity related to stock plans			0.4		0.4
Balance at December 31, 2020 Successor	13,990	<u>\$ 0.1</u>	\$ 279.9	<u>\$ (13.4)</u>	\$ 266.6
Net loss	_	_	_	(21.2)	(21.2)
Activity related to stock plans	145		1.5		1.5
Balance at December 31, 2021 Successor	14,135	\$ 0.1	\$ 281.4	\$ (34.6)	\$ 246.9

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these consolidated financial statements}.$ 

# FTS INTERNATIONAL, INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### NOTE 1—DESCRIPTION OF BUSINESS

Throughout the notes to these consolidated financial statements, the terms "the Company," "we," "us," "our" or "ours" refer to FTS International, Inc., together with its consolidated subsidiaries.

Certain prior year financial statements are not comparable to our current year financial statements due to the adoption of fresh start accounting. References to "Successor" or "Successor Company" relate to the financial position and results of operations of the reorganized Company subsequent to November 19, 2020. References to "Predecessor" or "Predecessor Company" relate to the financial position and results of operations of the Company prior to, and including, November 19, 2020.

We are one of the largest providers of hydraulic fracturing services in North America. Our services enhance hydrocarbon flow from oil and natural gas wells drilled by exploration and production ("E&P"), companies in shale and other unconventional resource formations. Our customers include leading E&P companies that specialize in unconventional oil and natural gas resources in North America. We operate in the most active oil and gas basins in the United States. Substantially all of our business activities support our well completion services. We manage our business, allocate resources, and assess our financial performance on a consolidated basis; therefore, we do not have separate operating segments.

In 2014, the Predecessor Company entered into a joint venture agreement with the Sinopec Oilfield Service Corporation ("Sinopec"). This joint venture collaboration offered hydraulic stimulation services in China. The joint venture company was owned 55% by Sinopec and 45% by the Company. In August 2019, FTSI closed on the sale of its 45% equity ownership interest in SinoFTS, to Sinopec. In exchange, FTSI received consideration of \$26.9 million for the sale of its equity interest and received a royalty fee of \$5.8 million for a license for its intellectual property use. After conducting an analysis of the relative fair values of the equity interest and royalty fee, FTSI allocated \$2.0 million to the prepaid royalty fee. FTSI recognized a gain of \$7.0 million on the sale of its equity interest in August 2019. The prepaid royalty fee will be recognized over approximately six years.

#### Concentrations of Risk

Our business activities are concentrated in the well completion services segment of the oilfield services industry in the United States. The market for these services is cyclical, and we depend on the willingness of our customers to make operating and capital expenditures to explore for, develop, and produce oil and natural gas in the United States. The willingness of our customers to undertake these activities depends largely upon prevailing industry conditions that are predominantly influenced by current and expected prices for oil and natural gas. Historically, a low commodity-price environment has caused our customers to significantly reduce their hydraulic fracturing activities and the prices they are willing to pay for those services. During these periods, these customer actions materially adversely affected our business, financial condition and results of operations.

Our customer base has historically been concentrated. Our business, financial condition and results of operations could be materially adversely affected if one or more of our significant customers ceases to engage us for our services on favorable terms, or at all, or fails to pay, or delays in paying us significant amounts of our outstanding receivables. For the year ended December 31, 2021 we had three customers make up over 10% of our consolidated revenue. These three customers represented approximately 39% of our consolidated revenue. For the Successor period of November 20, 2020 to December 31, 2020 we had 6 customers make up over 10% of our consolidated revenue. These six customers represented approximately 81% of our consolidated revenue. This level of customer concentration is partially due to the short time period included in the Successor period and may not be representative of customer concentration levels over a longer time period, such as a full fiscal year. For the Predecessor period of January 1, 2020 through November 19, 2020 we had one customer make up over 10% of

our consolidated revenue. This customer represented approximately 12% of our consolidated revenue. For the year ended December 31, 2019 we had two customers make up over 10% of our consolidated revenue. These two customers represented approximately 26% of our consolidated revenue. The loss of any of our largest existing customers could have a material adverse effect on our results of operations. While we view revenue as an important metric in assessing customer concentration, we also compare and manage our customer portfolio based on the number of fleets we place with each customer.

#### Related Parties

The Predecessor Company historically provided services and sold equipment to Chesapeake Energy Corporation and its affiliates ("Chesapeake"), which beneficially owned approximately 20% of the Predecessor's common stock, and had the right to designate two individuals to serve on our board of directors prior to November 19, 2020. Revenue earned from Chesapeake was \$0.7 million and \$0.1 million in Predecessor periods of 2020 and 2019, respectively. At December 31, 2021, we had accounts receivable balances of zero from Chesapeake.

The Predecessor Company historically sold equipment to SinoFTS for zero and \$0.9 million in 2020 and 2019, respectively. In August 2019, FTSI closed on the sale of its equity ownership interest in SinoFTS, to Sinopec. At December 31, 2021, we had accounts receivable balances of zero from this related party.

#### Subsequent Events

The Company has evaluated subsequent events through March 2, 2022, the date the financial statements were made available to be issued and determined that there have been no events that have occurred that would require adjustments to our disclosures in the financial statements.

#### NOTE 2—PROPOSED ACQUISITION

On October 21, 2021, we entered into an agreement and plan of merger (the "Merger Agreement") to be acquired by ProFrac Holdings, LLC ("ProFrac"), a leading oilfield services company, in an all-cash transaction that values FTSI at approximately \$407.5 million, including payments to holders of outstanding warrants. Under the terms of the agreement, which has been unanimously approved by FTSI's Board of Directors (the "Board"), FTSI stockholders will receive \$26.52 per share of FTSI common stock in cash (the "Merger Consideration"). This represents approximately a 14% premium over the Company's 60-day volume-weighted average closing share price through October 21, 2021.

On March 1, 2022, we entered into an amendment to the Merger Agreement (the "First Amendment"), providing for the shares of FTSI common stock held by THRC Holdings, LP, an affiliate of ProFrac ("THRC"), to remain outstanding, and not be converted into the right to receive the Merger Consideration. Pursuant to a letter agreement dated March 1, 2022 (the "Letter Agreement"), THRC has acknowledged and agreed to this change in the treatment of the shares of FTSI common stock held by it. The First Amendment may be terminated by the Company at any time prior to the Effective Time by giving notice of termination to ProFrac. In the event that the First Amendment is so terminated it would not have the effects described herein

The agreement includes a 45-day "go-shop" period which expired December 5, 2021. During the "go-shop" period, FTSI and its financial advisor, Ducera Securities LLC, reached out to nine potential strategic counterparties and 37 potential financial counterparties, and did not receive any alternative acquisition proposals.

The Company's obligation to close the transaction is conditioned upon approval by a majority of the Company's stockholders, excluding its largest stockholder THRC Holdings, which is an affiliate of ProFrac. A special meeting of stockholders to consider and vote upon certain proposals related to the Merger Agreement is scheduled for March 3, 2022. Upon closing of the transaction, the Company's common stock will no longer be listed on any public market.

#### NOTE 3—RESTRUCTURING

On September 22, 2020, FTS International, Inc., FTS International Services, LLC, and FTS International Manufacturing, LLC filed petitions for voluntary relief (the "Chapter 11 Cases") under Chapter 11 of Title 11 of the United States Code (the "Bankruptcy Code") in the United States Bankruptcy Court for the Southern District of Texas, Houston Division (the "Bankruptcy Court"). On September 22, 2020, FTSI International, Inc., FTS International Services, LLC, and FTS International Manufacturing, LLC filed the Joint Prepackaged Chapter 11 Plan of Reorganization of FTS International, Inc. and its Debtor Affiliates (as amended, modified or supplemented, the "Plan") and the related disclosure statement (the "Disclosure Statement"). On November 4, 2020, the Bankruptcy Court entered an order (the "Confirmation Order") confirming the Plan, as modified by the Confirmation Order, and approving the Disclosure Statement.

On November 19, 2020 (the "Effective Date"), the Plan became effective in accordance with its terms and FTS International, Inc., FTS International Services, LLC and FTS International Manufacturing, LLC emerged from Chapter 11. Pursuant to the Plan, on the Effective Date, all agreements, instruments, and other documents evidencing, relating to or otherwise connection with any of our common stock or other equity interests outstanding prior to the Effective Date (collectively, the "legacy equity interests") were cancelled and such legacy equity interests have no further force or effect after the Effective Date. Holders of our legacy equity interests received (i) a number of shares of Class A common stock equal to their proportionate distribution of approximately 9.4% of our common stock under the Plan (subject to dilution by the warrants issued pursuant to the Plan and the Amended and Restated Equity and Incentive Compensation Plan (the "MIP")), (ii) their proportionate distribution of 3,888,849 Tranche 2 Warrants to acquire Class A common stock.

In addition, pursuant to the Plan, on the Effective Date, all outstanding obligations under the 6.25% senior secured notes due May 1, 2022 (the "Notes") were cancelled and the indenture governing such obligations was cancelled, and the credit agreement, dated as of April 16, 2014, by and among FTS International, Inc., the lenders party thereto, and Wilmington Savings Fund Society, FSB, as successor administrative agent (the "Term Loan Agreement"), was cancelled, in each case except to the limited extent expressly set forth in the Plan. On the Effective Date, all liens and security interests granted to secure such obligations were automatically terminated and are of no further force and effect. The holders of Notes and holders of the claims under the Term Loan Agreement received their proportionate distribution of approximately 90.1% of our common stock (subject to dilution by the warrants issued pursuant to the Plan and the MIP) plus their pro rata share of \$30.7 million cash distribution. The holders of claims in connection with the termination of the supply agreement between the Predecessor and Covia Holding Corporation received, in exchange for their claims, a \$12.5 million cash distribution and 0.5% of our common stock (subject to dilution by the warrants issued pursuant to the Plan and the MIP).

Shares of Class A common stock were also issued to a holder of certain termination claims under the Plan.

#### NOTE 4—FRESH START ACCOUNTING

#### Fresh Start Accounting

Upon emergence from bankruptcy, we met the criteria and were required to adopt fresh start accounting in accordance with ASC Topic 852, Reorganizations, which on the emergence date resulted in a new entity, the Successor, for financial reporting purposes, with no beginning retained earnings or deficit as of the fresh start reporting date. The criteria requiring fresh start accounting are: (i) the holders of then-existing voting shares of the Predecessor received less than 50 percent of the new voting shares of the Successor outstanding upon emergence from bankruptcy, and (ii) the reorganization value of the Company's assets immediately prior to confirmation of the Plan was less than the total of all post-petition liabilities and allowed claims.

The Company applied fresh start accounting effective November 19, 2020, the Effective Date. As such, fresh start accounting is reflected in the accompanying consolidated balance sheet as of December 31, 2020 and related

fresh start adjustments are included in the accompanying statement of operations for the period from January 1, 2020 through November 19, 2020. As a result of the application of fresh start accounting and the effects of the implementation of the Plan, the financial statements for the period after November 19, 2020 will not be comparable with the financial statements prior to and including November 19, 2020.

#### Reorganization Value

Reorganization value represents the fair value of the Successor's total assets and is intended to approximate the amount a willing buyer would pay for the assets immediately after restructuring. Under fresh start accounting, the Company allocated the reorganization value to its individual assets based on their estimated fair values.

The Company's reorganization value is derived from an estimate of enterprise value. Enterprise value represents the estimated fair value of an entity's long-term debt and other interest-bearing liabilities and stockholders' equity less unrestricted cash and cash equivalents. The Company estimated the enterprise value of the Successor to be \$266 million at the Effective Date, which was in the Bankruptcy Court approved range of \$190 million and \$290 million. The enterprise value was derived from an independent valuation with the assistance of a third-party valuation advisor. Specific valuation approaches and key assumptions used to arrive at the reorganization value, and the value of discrete assets and liabilities resulting from the application of fresh start accounting, are described below in greater detail within the valuation process.

Although the Company believes the assumptions and estimates used to develop enterprise value and reorganization value are reasonable and appropriate, different assumptions and estimates could materially impact the analysis and resulting conclusions. The assumptions used in estimating these values are inherently uncertain and require judgment. The following table reconciles the Company's enterprise value to the fair value of the Successor's common stock as of the Effective Date.

(In millions)	No	vember 19, 2020
Enterprise value	\$	266.3
Plus: Excess cash <sup>(1)</sup>	<del></del> -	13.3
Fair value of Successor equity	\$	279.6

The following table reconciles the Company's enterprise value to the reorganization value as of the Effective Date:

(In millions)	November 19 2020	
Enterprise value	\$	266.3
Plus: Excess cash <sup>(1)</sup>		13.3
Plus: Current liabilities		36.4
Plus: Non-interest-bearing non-current liabilities		6.2
Reorganization value of Successor assets	\$	322.2

(1) Excess cash of \$13.3 million is calculated by taking the Company's Successor cash and cash equivalents balance of \$88.3 million less \$75.0 million of minimum cash required to operate the business as determined by management. The minimum cash required to operate the business of \$75.0 million was also utilized in the estimation of the range of enterprise values included in the Plan and approved by the Bankruptcy Court.

#### Valuation Process

The fair values of our principal assets, including inventories, land, buildings, and improvements, service equipment, and intangible assets were estimated with the assistance of third-party valuation advisors as of the Effective Date. In addition, we also estimated the fair value of the Company's warrants as of the Effective Date.

#### Inventories

Inventories include parts, chemicals, and other raw materials. The fair value of the parts inventory was determined using a combination of the cost and income approaches. The cost approach estimates the fair value of an asset based on the cost to reconstruct or replace it with another of similar utility, with adjustments for physical deterioration and identified obsolescence. The income approach was used to quantify the economic support available for the parts inventory based on the enterprise value estimated for the Company. For the chemicals and other materials inventory, the Company determined that the book value as the proxy for fair value.

#### Property, Plant, and Equipment

Land, Buildings, and Improvements

The fair value of land, buildings, and improvements was determined using a combination of the cost and market approaches. To determine the fair value of buildings and site improvements, the cost and market approaches were used. As part of the valuation process, information was obtained describing physical attributes such as land and building size, construction dates, and general improvement details. In applying the cost approach, the replacement cost was determined based on the current cost to construct improvements with similar utility, using modern materials and current standards, design, and layout. To determine the fair value of land and improvements, the market approach was used based on third party databases identifying listings of recent sales and comparable properties within pertinent market areas.

#### Service Equipment and Other

The fair value of the service equipment and other was estimated using a combination of the cost, market, and income approaches. The cost approach was primarily utilized, with the application of the market approach for selected assets based upon the specific characteristics being appraised. The cost approach measures the value of an asset based on the cost to reconstruct or replace it with another of similar utility, with adjustments for physical deterioration and identified obsolescence. The market approach measures the value of an asset through an analysis of recent sales or offerings of comparable assets. The income approach was used to quantify the economic support available for the service equipment and other based on the enterprise value estimated for the Company.

#### Intangible Assets

The fair value of intangible assets such as customer relationships, in-house developed software, and tradename was determined using a combination of the cost, market, and income approaches. To determine the fair value of customer relationships, the income approach was used based on the present value of the incremental after-tax cash flows attributable to the existing customer cash flow after deducting the appropriate contributory asset charges. The discount rate utilized to present-value the after-tax cash flows was selected based on consideration of the overall business risks and risks associated with the asset being valued. To determine the fair value of the in-house developed software, the cost replacement method, a form of the cost approach, was used. To determine the fair value of the tradename, the Royalty Savings Method, a variation of the income approach, was used. The estimated royalty rate was determined by observing publicly available royalty rates information for similar companies with similar assets. The forecasted cash flows expected to be generated as a result of the royalty savings were discounted to present value utilizing a discount rate considering overall business risks and risks associated with the asset being valued.

# Warrants

The fair value of the warrants issued upon the Effective Date was estimated using the Black-Scholes-Merton option pricing model. The Black-Scholes-Merton model is an option pricing model used to estimate the fair value of options and warrants based on the following input assumptions: stock price, strike price, term, risk-free rate,

volatility, and dividend yield. In using the Black-Scholes-Merton option pricing model to fair value the warrants, the following assumptions were used: market observable stock price on the Effective Date; strike prices of \$33.04 and \$37.14 for Tranche 1 and Tranche 2 warrants, respectively; expected volatility of 65.0%; risk-free rate of 0.22%; and an expected dividend yield of 0.0%. The expected volatility assumption was estimated using market data related to the Company and certain similar publicly traded entities with considerations for differences in size and leverage of the Company versus the publicly traded entities. The strike prices and term assumptions were based on the contractual term of the warrants. The risk-free rate assumption was based on United States Constant Maturity Treasury rates.

#### Consolidated Balance Sheet

The following illustrates the effects on the Company's consolidated balance sheet due to the reorganization and fresh start accounting adjustments. The explanatory notes following the table below provide further details on the adjustments, including the assumptions and methods used to determine fair value for its assets, liabilities, and warrants.

	As of November 19, 2020 (in millions)				
	Predecessor	Reorganization	Fresh Start	Successor	
	Company	Adjustments	Adjustments	Company	
ASSETS					
Current assets					
Cash and cash equivalents	\$ 146.7	\$ (58.4)(a)	\$ —	\$ 88.3	
Accounts receivable, net	30.1	_	_	30.1	
Inventories	33.5	_	(2.2)(k)	31.3	
Prepaid expenses and other current assets	19.0	<u>10.1</u> (b)	(5.3)(1)	23.8	
Total current assets	229.3	(48.3)	(7.5)	173.5	
Property, plant, and equipment, net	177.2	_	(42.0)(m)	135.2	
Operating lease right-of-use assets	4.7	_	_	4.7	
Intangible assets, net	29.5	_	(22.1)(n)	7.4	
Other assets	1.4	0.2(c)	(0.2)(o)	1.4	
Total assets	\$ 442.1	\$ (48.1)	\$ (71.8)	\$ 322.2	
LIABILITIES AND STOCKHOLDERS' (DEFICIT) EQUITY					
Current liabilities					
Accounts payable	\$ 20.3	\$ 0.8(d)	\$ —	\$ 21.1	
Accrued expenses	11.6	0.2(e)	_	11.8	
Current portion of operating lease liabilities	3.2	_	_	3.2	
Other current liabilities	12.8	(12.5)(f)		0.3	
Total current liabilities	47.9	(11.5)	_	36.4	
Operating lease liabilities	3.4	_	_	3.4	
Other liabilities	2.8	_	_	2.8	
Liabilities subject to compromise	488.8	(488.8)(g)			
Total liabilities	542.9	(500.3)	_	42.6	
Stockholders'					
Common stock—Predecessor	36.4	(36.4)(h)	_	_	
Additional paid-in capital—Predecessor	4,392.9	(4,392.9)(h)	_	_	
Common stock—Successor	_	0.1(i)	_	0.1	
Additional paid-in capital—Successor	_	279.5(i)	_	279.5	
Accumulated deficit	(4,530.1)	4,601.9(j)	(71.8)(p)		
Total stockholders' (deficit) equity	(100.8)	452.2	(71.8)	279.6	
Total liabilities and stockholders' (deficit) equity	\$ 442.1	\$ (48.1)	\$ (71.8)	\$ 322.2	

#### Reorganization Adjustments

a) Reflects the net cash activities that occurred on the Effective Date. Of the \$9.1 million transferred from cash and cash equivalents to escrow account recorded to prepaid expenses and other current assets, \$3.8 million was related to success fees recognized upon emergence. Of the \$4.9 million payment of professional and success fees, \$1.9 million was related to success fees paid and recognized upon emergence.

(In millions)	mber 19, 2020
Transfer of payment for professional fees and success fees to escrow account	
recorded in prepaid expenses and other current assets	\$ (9.1)
Payment of professional and success fees	(4.9)
Payment to secured debtholders	(30.7)
Payment of Covia settlement	(12.5)
Payment of emergence date bonus	(1.0)
Payment of debt issuance costs related to Successor Revolving Credit Facility	 (0.2)
Change in cash and cash equivalents	\$ (58.4)

b) Reflects adjustment to prepaid expenses and other current assets for the following activities.

(In millions)	mber 19, 2020
Transfer of payment for professional fees and success fees from cash and cash	
equivalent	\$ 9.1
Payment of emergence date bonus	 1.0
Change in prepaid expenses and other current assets	\$ 10.1

- c) Reflects an adjustment to debt issuance costs related to the Successor Revolving Credit Facility of \$0.2 million.
- d) Reflects an adjustment to accounts payable related to success fees of \$3.8 million recognized upon emergence offset by a \$3.0 million payment for previously accrued professional fees.
- e) Reflects an adjustment to accrued expenses related to taxes withheld from holders of Predecessor stock-based compensation upon acceleration and immediate vesting.
- f) Reflects a \$12.5 million adjustment to other current liabilities related to payment of Covia settlement amount.

g) On the Effective Date, we settled liabilities subject to compromise per the Plan. The adjustment reflects the removal of the balance from liabilities subject to compromise and the pre-tax gain on the settlement of liabilities subject to compromise as follows.

(In millions)	November 19 2020	
2022 Senior Notes	\$	379.0
Term Loan		67.6
Supply commitment charges		42.2
Total liabilities subject to compromise		488.8
Issuance of New Common Stock to holders of 2022 Senior Notes		(202.0)
Issuance of New Common Stock to Term Loan lenders		(36.1)
Issuance of New Common Stock to unsecured claimholders		(2.2)
Payment to holders of 2022 Senior Notes		(26.0)
Payment to Term Loan lenders		(4.7)
Pre-tax gain on settlement of liabilities subject to compromise	\$	217.8

- h) Reflects the cancellation of the Predecessor's common stock and the Predecessor's additional paid-in capital, which includes the acceleration of the Predecessor's stock-based compensation of \$15.1 million.
- i) The following reconciles reorganization adjustments made to the Successor common stock and Successor additionalpaid-in capital:

(In millions)	No	vember 19, 2020
Fair value of New Common Stock issued to holders of the 2022 Senior Notes	_	2020
claims	\$	202.0
Fair value of New Common Stock issued to holders of the Term Loan claims		36.1
Fair value of New Common Stock issued to holders of the unsecured claims		2.2
Fair value of New Common Stock issued to holders of legacy equity interests		24.9
Fair value of warrants issued to legacy equity interests		14.4
Total Successor common stock and additional paid-in capital		279.6
Less: Successor common stock		(0.1)
Successor additional paid-in capital	\$	279.5

j) Reflects the cumulative net impact of the following transactions on Predecessor accumulated deficit:

(In millions)	Nov	rember 19, 2020
Pre-tax gain on settlement of liabilities subject to compromise as calculated in		,
note f)	\$	217.8
Acceleration of Predecessor stock-based compensation		(15.3)
Cancellation of Predecessor common stock and additional paid-in capital		4,444.4
Success fees recognized on the Effective Date		(5.7)
Issuance of Successor common stock to legacy equity interests		(24.9)
Issuance of warrants to legacy equity interests		(14.4)
Change in accumulated deficit	\$	4,601.9

# Fresh Start Adjustments

- k) Reflects the fair value adjustment to parts inventory.
- 1) Reflects the write-off of prepaid premiums of \$5.3 million in connection with the Predecessor's directors & officers insurance.
- m) Reflects the fair value adjustments to property, plant, and equipment, as well as the elimination of the historical accumulated depreciation.
- n) Reflects the fair value adjustment to intangible assets, net.
- o) Reflects the write-off of debt issuance costs of \$0.2 million related to the Successor Revolving Credit Facility.
- p) Reflects the cumulative effect on accumulated deficit of the fresh start accounting adjustments discussed above.

#### Reorganization Items

Reorganization items represent (i) expenses incurred associated with the Chapter 11 restructuring subsequent to the Petition Date, (ii) gains or losses from liabilities settled, and (iii) fresh start accounting adjustments, recorded in reorganization items in our consolidated statements of operations. Professional service provider charges associated with our restructuring that were incurred before the Petition Date are recorded in selling, general and administrative in our consolidated statements of operations.

(In millions)	Succe Period Novemb throu Decemb 202	from ber 20, ugh ber 31,	Perio	lecessor od from uary 1, rough mber 19,
Pre-tax gain on settlement of liabilities subject to compromise	\$	_	\$	217.8
Fresh start accounting adjustments		_		(71.8)
Professional service provider fees and other expenses		(1.9)		(9.5)
Success fees for professional service providers		_		(5.7)
Derecognition of unamortized debt discounts and issuance costs		_		(2.5)
Terminated executory contracts		_		(9.7)
Acceleration of Predecessor stock-based compensation expense		_		(15.3)
Other Costs		(0.2)		
(Loss)/gain on reorganization items, net	\$	(2.1)	\$	103.3

Contractual interest expense of \$4.4 million from the Petition Date through the Effective Date, associated with our outstanding term loan and senior notes, was not accrued or recorded in the Predecessor consolidated statement of operations as interest expense.

#### NOTE 5—SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

#### Basis of Presentation

We prepared the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America ("GAAP"). The consolidated financial statements include the accounts of the Company and all majority-owned domestic and foreign subsidiaries. Investments over which we have the ability to exercise significant influence over operating and financial policies, but do not hold a controlling interest, are accounted for using the equity method of accounting. All significant intercompany accounts and transactions have been eliminated in consolidation. There were no items of other comprehensive income in the periods presented.

#### Use of Estimates

The preparation of financial statements in accordance with GAAP requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities, related revenues and expenses, and the disclosure of gain and loss contingencies at the date of the financial statements and during the periods presented. We base these estimates on historical results and various other assumptions believed to be reasonable, all of which form the basis for making estimates concerning the carrying values of assets and liabilities that are not readily available from other sources. Actual results could differ materially from those estimates.

#### Fresh Start Accounting

Upon emergence from bankruptcy in 2020 the Company adopted fresh start accounting. Refer to Note 4 - Fresh Start Accounting for further details.

#### Cash and Cash Equivalents

Cash equivalents include only investments with an original maturity of three months or less. We occasionally hold cash deposits in financial institutions that exceed federally insured limits. We monitor the credit ratings and our concentration of risk with these financial institutions on a continuing basis to safeguard our cash deposits.

#### Accounts Receivable and Allowance for Doubtful Accounts

Accounts receivable are recorded at their invoiced amounts or amounts for which we have a right to invoice based on services completed. In June 2016, the FASB issued ASU 2016-13, Financial Instruments—Credit Losses: Measurement of Credit Losses on Financial Instruments. This standard requires companies to measure credit losses utilizing a methodology that reflects expected credit losses and requires a consideration of a broader range of reasonable and supportable information to inform credit loss estimates. The new standard also applies to financial assets arising from revenue transactions such as accounts receivables. We adopted this standard on January 1, 2020, and it had no material effect on our consolidated financial statements.

#### Inventories

Inventories primarily consist of maintenance parts that are used to service our hydraulic fracturing equipment but will also include proppants and chemicals that are used to provide hydraulic fracturing services. Inventory held as of our emergence date was remeasured to fair value. Subsequent to our emergence, inventories are stated at the lower of cost or net realizable value. The cost basis of our inventories is based on the average cost method and includes in-bound freight costs.

As necessary, we record an adjustment to decrease the value of slow moving and obsolete inventory to its net realizable value. To determine the adjustment amount, we regularly review inventory quantities on hand and compare them to estimates of future product demand, market conditions, production requirements and technological developments.

#### Restricted Cash

The Company had zero and \$12.7 million restricted cash at December 31, 2021 and 2020, respectively. At December 31, 2020 the company's restricted cash included unsettled escrow fees related to our bankruptcy emergence and cash used as collateral for other banking products.

#### Property, Plant, and Equipment

Property, plant, and equipment held as of our emergence date was remeasured to fair value and depreciated over the estimated remaining useful lives of individual assets.

Property, plant, and equipment purchased subsequent to our emergence date is stated at cost less accumulated depreciation, which is generally provided by using the straight-line method over the estimated useful lives of the individual assets. We manufacture and refurbish equipment used in our hydraulic fracturing operations and the cost of this equipment, which includes direct and indirect manufacturing costs, is capitalized and carried as construction-in-progress until it is completed. Expenditures for renewals and betterments that extend the lives of our service equipment, which includes the replacement of significant components of service equipment, are capitalized and depreciated. Other repairs and maintenance costs are expensed as incurred.

We capitalize qualifying costs related to the acquisition or development ofinternal-use software. Capitalization of costs begins after the conceptual formulation stage has been completed. Capitalized costs are amortized over the estimated useful life of the software, which ranges between three and five years. The unamortized balance of capitalized software costs at December 31, 2021 and 2020, was \$0.1 million and \$0.2 million, respectively. Amortization of computer software was \$0.1 million for 2021 (Successor), \$0.3 million for January 1 through November 19, 2020 (Predecessor), zero for November 20 through December 31, 2020 (Successor), and \$3.4 million in 2019 (Predecessor).

#### Leases

We determine if a contract contains a lease at inception. We lease certain administrative offices, sales offices, and operational facilities. We also lease some service equipment and light duty vehicles. These leases have remaining lease terms of 6 years or less. Some leases contain options to extend the leases, and some include options to terminate the leases. We do not include renewal or termination options in our assessment of the lease terms unless extension or termination for certain leases is deemed to be reasonably certain. Lease agreements with lease and non-lease components are generally accounted for as a single lease component.

Operating lease assets and liabilities are recognized at the lease commencement date, which is the date we take possession of the property. Operating lease liabilities represent the present value of lease payments not yet paid. Operating lease assets represent our right to use an underlying asset and are valued based upon the operating lease liabilities adjusted for prepayments or accrued lease payments, initial direct costs, lease incentives, and impairment of operating lease assets. To determine the present value of lease payments not yet paid, we estimate incremental secured borrowing rates corresponding to the lease term including reasonably certain renewal periods. We estimate this rate based on prevailing financial market conditions, credit analysis, and management judgment.

Our leases typically contain rent escalations over the lease term. We recognize expense for these leases on a straight-line basis over the lease term. Additionally, tenant incentives used to fund leasehold improvements are recognized when earned and reduce our right-of-use asset related to the lease. These are amortized through the operating lease asset as reductions of expense over the lease term.

We provide residual value guarantees for our leases of light-duty vehicles and certain service equipment. No amounts related to these residual value guarantees have been deemed probable and included in the lease liabilities on our consolidated balance sheet; however, if the value for all of the vehicles was zero and if we cancelled these leases at December 31, 2021, we would be required to pay a total of \$5.1 million in residual value guarantees.

#### Intangible Assets

We have historically acquired indefinite-lived intangible assets related to business acquisitions. Intangible assets with indefinite lives are not amortized. The amount of indefinite-lived intangible assets recorded in our consolidated balance sheets for December 31, 2021 and 2020, was \$7.4 million, of which 6.0 million related to our tradename and \$1.4 million related to our developed technology. Accumulated amortization of this intangible as of December 31, 2021 and December 31, 2020 was \$0.5 million and \$0.1 million, respectively. Amortization expense in 2021 (Successor) and November 20 through December 31, 2020 (Successor) was \$0.4 million and \$0.1 million, respectively. For the period January 1 through November 19, 2020 (Predecessor) and 2019 (Predecessor), amortization expense was zero. On November 19, 2020, upon emergence from Chapter 11 bankruptcy our intangible assets were identified and valued at fair value and a finite life was assigned to our developed technology. We amortize this intangible asset with a finite life on a straight-line basis over a period of three years which the asset is expected to contribute to our future cash flows.

#### Impairment of Long-Lived Assets and Intangible Assets

Long-lived assets, such as property, plant, equipment are reviewed for impairment when events or changes in circumstances indicate that the carrying value of the asset may not be recoverable. Recoverability is assessed based on the undiscounted future cash flows generated by the asset or asset group. If the carrying amount of an asset or asset group is not recoverable, we recognize an impairment loss equal to the amount by which the carrying amount exceeds fair value. We estimate fair value based on the income, market, or cost valuation techniques.

Intangible assets with indefinite lives are reviewed at least annually for impairment, and in interim periods if certain events occur indicating that the carrying value of intangible assets may be impaired. We estimate fair values utilizing valuation methods such as discounted cash flows and comparable market valuations. We perform our annual impairment tests at the beginning of the fourth quarter.

#### **Equity Method Investments**

Investments in which we have the ability to exercise significant influence, but not control, are accounted for pursuant to the equity method of accounting. We recognize our proportionate share of earnings or losses of our international affiliates three months after they occur. When events and circumstances warrant, investments accounted for under the equity method of accounting are evaluated for impairment. An impairment charge is recorded whenever a decline in value of an investment below its carrying amount is determined to be other-than-temporary.

#### Income Taxes

Income taxes are accounted for using the asset and liability method. Deferred taxes are recognized for the tax consequences of temporary differences by applying enacted statutory tax rates applicable to future years to differences between the financial statement carrying amounts and the tax bases of existing assets and liabilities. The effect on deferred taxes for a change in tax rates is recognized in earnings in the period that includes the enactment date. We recognize future tax benefits to the extent that such benefits are more likely than not to be realized.

We record a valuation allowance to reduce the value of a deferred tax asset if based on the consideration of all available evidence, it is more likely than not that all or some portion of the deferred tax asset will not be realized. Significant weight is given to evidence that can be objectively verified. We evaluate our deferred income taxes quarterly to determine if a valuation allowance is required by considering all available evidence, including historical and projected taxable income and tax planning strategies. We will adjust a previously established valuation allowance if we change our assessment of the amount of deferred income tax asset that is more likely than not to be realized.

We record uncertain tax positions in accordance with ASC 740 on the basis of atwo-step process in which (1) we determine whether it is more likely than not that the tax positions will be sustained on the basis of the technical merits of the position and (2) for those tax positions that meet the more-likely-than-not recognition threshold, we recognize the largest amount of tax benefit that is more than 50 percent likely to be realized upon ultimate settlement with the related tax authority.

#### Revenue Recognition

The Company contracts with its customers to perform hydraulic fracturing services on one or more oil or natural gas wells. Under these arrangements, we satisfy our performance obligations as services are rendered, which is generally upon the completion of a fracturing stage. We typically complete one or more stages per day. A stage is considered complete when we have met the specifications set forth by the customer, at which time we have the

right to invoice the customer and the customer is obligated to pay us for the services rendered. The price for our services typically includes an equipment charge and product charges for proppant, chemicals and other products consumed during the course of providing our services. Payment terms average approximately two months from the date a stage or well is completed. All consideration owed to us for services performed during a period is fixed and our right to receive it is unconditional.

We also contract with some customers to provide them with the exclusive use of a fracturing fleet for a period of time. Our customers can generally terminate these contracts with less than 90 days' notice. We satisfy our performance obligation as services are rendered, which is based on the passage of time rather than the completion of a stage. Under these arrangements, we have the right to receive consideration from a customer even if circumstances outside of our control prevent us from performing our work. All consideration owed to us for services performed during a period is fixed and our right to receive it is unconditional.

Pricing for our services for all contracts is frequently negotiated with our customers and is based on prevailing market rates during each reporting period. The amounts we invoice our customers for services performed during a period are directly related to the value received by the customers for the period. There is no inherent uncertainty to the amount of consideration we will receive for services performed during a period and no judgment is required to allocate a portion of the transaction price to a future period. Accordingly, we are not required to identify any unsatisfied performance obligations nor attribute any revenue to them.

During the periods presented we acted as a principal, rather than as an agent, for all of the goods and services that we provided to our customers; our customer arrangements did not include obligations for refunds or warranties of our work; our revenue does not include sales taxes collected from our customers; and we did not incur incremental costs to obtain or fulfill contracts with our customers.

To comply with the FASB disclosure objective, we are required to disaggregate our revenue into categories if it will provide an enhanced understanding of how the nature, amount, timing, and uncertainty of our revenue and cash flows are affected by economic factors. To evaluate an appropriate level of disaggregation of revenue, we considered the following aspects of our business:

- · We provide a single service to our customers.
- We only generate revenue in the U.S. onshore market.
- · We have a homogeneous customer base, which is comprised of large oil and gas exploration companies.
- We provide our service over a short period of time.
- · We do not disaggregate our revenue into categories for any external communications or to make resource allocation decisions.
- · We do not have separate operating segments.

Based on the above factors, we concluded that no additional disaggregation of revenue was necessary or meaningful to help depict the nature, amount, timing and uncertainty of our revenues and cash flows.

#### **Unconditional Purchase Obligations**

Prior to our emergence from bankruptcy on November 19, 2020 we entered into supply arrangements, primarily for sand, with our vendors that contain unconditional purchase obligations. These represented obligations to transfer funds in the future for fixed or minimum quantities of goods at fixed or minimum prices. We entered into these unconditional purchase obligation arrangements in the normal course of business to ensure that adequate levels of sourced product are available to us. We currently do not have any such supply agreements.

#### Stock-Based Compensation

We measure all employee stock-based compensation awards using a fair value method and record this cost in the consolidated financial statements. Our stock-based compensation relates to restricted stock units issued to our employees. On the date that an equity-classified award is granted, we determine the fair value of the award and recognize the compensation cost over the requisite service period, which typically is the period over which the award vests. For equity-classified awards with graded vesting based solely on the satisfaction of a service condition, we recognize compensation cost as a single award on a straight-line basis. We account for forfeited awards as forfeitures occur, which results in a reversal of stock-based compensation cost previously recognized up to the date of the forfeiture. For stock-based awards with performance conditions that affect vesting, we only recognize compensation cost when it is probable that the performance conditions will be met.

#### Fair Value Measurements

Fair value is defined as the price that would be received from selling an asset or paid to transfer a liability in an orderly transaction between market participants at a measurement date. We apply the following fair value hierarchy, which prioritizes the inputs used to measure fair value into three levels and bases the categorization within the hierarchy upon the lowest level of input that is available and significant to the fair value measurement:

- Level One: The use of quoted prices in active markets for identical financial instruments.
- Level Two: The use of quoted prices for similar instruments in active markets or quoted prices for identical or similar instruments in markets that are not active or other inputs that are observable in the market or can be corroborated by observable market data.
- Level Three: The use of significant unobservable inputs that typically require the use of management's estimates of assumptions that market
  participants would use in pricing.

Money market funds, classified as cash and cash equivalents, are the only financial instruments that are measured and recorded at fair value on the Company's balance sheets. The following table presents money market funds at their level within the fair value hierarchy.

(In millions)	Total	Level 1	Level 2	Level 3
December 31, 2021			' <u></u> '	
Money market funds	\$60.0	\$ 60.0	<u>\$ —</u>	<u>\$ —</u>
December 31, 2020				
Money market funds	<u>\$59.6</u>	\$ 59.6	<u>\$</u> —	<u>\$</u> —

#### Reclassifications

All inventory write-downs have been reclassified from costs of revenue to impairments and other charges on the statements of operations to conform to current year presentation. This reclassification had no effect on operating income (loss) or net income (loss) as previously reported.

#### New Accounting Standards Updates

In February 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU")2016-02, *Leases*. The FASB subsequently issued a number of additional ASUs to update this guidance. This standard was issued to increase transparency and comparability among organizations by requiring that a right-of-use asset and corresponding lease liability be recorded on the balance sheet for leases with terms longer than 12 months. We elected to use three practical expedients allowed under the guidance. According to these practical expedients we did not reassess whether existing contracts are or contain a lease; we did not reassess whether existing leases are operating or finance leases; and we did not reassess the accounting for initial direct

costs for existing leases. Our approach to adopting this new standard included a review of existing leases and other executory contracts that could contain embedded leases and we identified the key terms that were necessary for us to calculate the right-of-use asset and lease liability. These consolidated financial statements have been prepared in accordance with the new ASU utilizing the modified retrospective transition method, which resulted in the recording of operating lease liabilities of approximately \$38 million as of January 1, 2019 on our consolidated balance sheet with an immaterial effect on our consolidated statement of stockholders' equity (deficit) and no related effect on our consolidated statement of operations.

In November 2016, the FASB issued ASU2016-18, *Restricted Cash*. This standard was issued to change the presentation of amounts generally described as restricted cash and restricted cash equivalents to be included with cash and cash equivalents when reconciling the beginning-of-period and end-of-period total amounts shown on the statement of cash flows. We adopted this standard on January 1, 2018, and the effects of this standard and related required disclosures have been reflected in our condensed consolidated statements of cash flows. The following table provides a reconciliation of cash, cash equivalents, and restricted cash reported within the consolidated statements of cash flows:

(In millions)	mber 31, 2021	mber 31, 2020
Cash and cash equivalents	\$ 85.1	\$ 94.0
Restricted cash included in prepaid expenses and other current assets	 	 12.7
Total cash, cash equivalents, and restricted cash shown in the consolidated statements of cash flows	\$ 85.1	\$ 106.7

In June 2016, the FASB issued ASU2016-13, Financial Instruments – Credit Losses: Measurement of Credit Losses on Financial Instruments. This standard requires companies to measure credit losses utilizing a methodology that reflects expected credit losses and requires a consideration of a broader range of reasonable and supportable information to inform credit loss estimates. The new standard will also apply to financial assets arising from revenue transactions such as accounts receivables. We adopted this standard on January 1, 2020, and it had no material effect on our consolidated financial statements

In December 2019, the FASB issued ASUNo. 2019-12, Simplification of Accounting for Income Taxes, which simplifies the accounting for income taxes by providing new guidance to reduce complexity and eliminate certain exceptions to the general approach to the income tax accounting model. The Company adopted this guidance effective January 1, 2021, which did not have a material impact on the accompanying unaudited condensed consolidated financial statements.

# NOTE 6—SUPPLEMENTAL BALANCE SHEET INFORMATION

#### Accounts Receivable

The following table summarizes our accounts receivable balance:

	December 31,	December 31,		
(In millions)	2021	2020		
Trade accounts receivable	\$ 70.7	\$ 30.6		
Allowance for doubtful accounts	(3.4)	(3.7)		
Accounts receivable, net	\$ 67.3	\$ 26.9		

The change in allowance for doubtful accounts is as follows:

	Successor			Predecess					
		Period from				d from			
	Vear	Ended		nber 20, ough			ıary 1, ough	Vea	r Ended
		ber 31,		nber 31,			nber 19,		mber 31,
(In millions)	20	)21	2	020		20	020		2019
Balance at beginning of period	\$	3.7	\$	3.7		\$	3.3	\$	0.9
Provision for bad debts, net included in selling, general, and administrative									
expense		0.2		_			0.8		2.4
Uncollectible receivables written off		(0.5)					(0.4)		
Balance at end of period	\$	3.4	\$	3.7		\$	3.7	\$	3.3

# Inventories

The following table summarizes our inventories:

	December 31,	December 31,		
(In millions)	2021	2020		
Maintenance parts	\$ 37.7	\$ 27.9		
Proppants and chemicals	0.7	1.0		
Other	0.1	0.1		
Total inventories	\$ 38.5	\$ 29.0		

# Prepaid Expenses and Other Current Assets

The following table summarizes our prepaid expenses and other current assets:

(In millions)	December 31, 2021	December 31, 2020		
Restricted cash	<u> </u>	\$ 12.7		
Prepaid expenses	\$ 4.2	\$ 6.7		
Other	0.1	0.1		
Total prepaid expenses and other current assets	\$ 4.3	\$ 19.5		

# Property, Plant, and Equipment, net

The following table summarizes our property, plant, and equipment:

(Dollars in millions)	December 31, 2021	December 31, 2020	Estimated Useful Life (in years)
Service equipment	\$ 150.3	\$ 92.1	2.5 - 10
Buildings and improvements	19.2	21.6	15 - 39
Office, software, and other equipment	1.2	0.5	3 - 7
Vehicles and transportation equipment	0.3	0.3	5 - 20
Land	7.8	8.4	N/A
Construction-in-process and other	9.7	14.1	N/A
Total property, plant, and equipment	188.5	137.0	
Accumulated depreciation and amortization	(54.5)	(4.7)	
Total property, plant, and equipment, net	\$ 134.0	\$ 132.3	

Depreciation expense was \$53.9 million in 2021 (Successor), \$68.5 million for January 1 through November 19, 2020 (Predecessor), \$4.7 million for November 20 through December 31, 2020 (Successor), and \$90.0 million 2019 Predecessor).

# Accrued Expenses

The following table summarizes our accrued expenses:

	December 31,		
(In millions)	2021	2020	
Sales, use, and property taxes	\$ 4.1	\$ 4.8	
Employee compensation and benefits	11.6	5.6	
Interest	_	_	
Insurance	2.3	2.0	
Other	0.2	0.1	
Total accrued expenses	\$ 18.2	\$ 12.5	

#### Other Current Liabilities

The following table summarizes our other current liabilities:

	December 31,	December 31,
(In millions)	2021	2020
Deferred revenue	0.3	0.3
Total other current liabilities	\$ 0.3	\$ 0.3

#### Other Liabilities

The following table summarizes our other liabilities:

(In millions)	December 31, 2021	December 31, 2020		
Deferred revenue	\$ 1.0	\$ 1.3		
Deferred employer payroll taxes	<u> </u>	1.1		
Total other liabilities	\$ 1.0	\$ 2.4		

#### NOTE 7—INDEBTEDNESS AND BORROWING FACILITY

# Successor Revolving Credit Facility

On November 19, 2020, The Successor Company entered into a \$40 million revolving credit facility, with an initial maturity date of November 19, 2023, with Wells Fargo, N.A.

LIBOR borrowings under the credit facility bear interest at the greater of LIBOR or 0.75% plus a margin of 2.25% to 2.50% per annum, depending on facility utilization. Base rate loans are also available at our option. The credit facility includes a \$15 million sub-limit for outstanding letters of credit which would reduce the amount available under the facility. We also pay a commitment fee on the unused amount of the facility of 0.375% per annum, depending on facility utilization.

The obligations under the credit facility are secured by substantially all of our assets, including our working capital and equipment.

The maximum availability of credit under the credit facility is limited at any time to the lesser of \$40 million or a borrowing base. The borrowing base is based on percentages of eligible accounts receivable and is subject to certain reserves. In an event of default or if the amount available under the credit facility is less than either 12.5% of our maximum availability or \$5.0 million, we will be required to maintain a minimum fixed charge coverage ratio of 1.0 to 1.0. If at any time borrowings and letters of credit issued under the credit facility exceed the borrowing base, we will be required to repay an amount equal to such excess.

The credit facility contains covenants that could, in certain circumstances, limit our ability to issue additional debt, repurchase or pay dividends on our common stock, sell substantially all of our assets, make certain investments, or enter into certain other transactions. We were in compliance with all of the covenants in the credit facility at December 31, 2021 and 2020.

As of December 31, 2021, the borrowing base was \$34.0 million and therefore our maximum availability under the credit facility was \$34.0 million. As of December 31, 2021, there were no borrowings outstanding under the credit facility, and letters of credit totaling \$4.4 million were issued, resulting in \$29.6 million of availability under the credit facility.

#### Predecessor 2021 Term Loan

On April 16, 2014, the Predecessor Company entered into a \$550 million term loan, which matures on April 16, 2021 ("Term Loan"), with a group of lenders with Wells Fargo Bank, N.A., as administrative agent. The Term Loan bears interest at a three-month London Interbank Offered Rate ("LIBOR") plus a margin of 4.75% per annum, with a 1.00% LIBOR floor. Interest is payable on interest rate reset dates, which is generally monthly.

The Term Loan was issued at a discount of \$2.7 million for aggregate consideration of \$547.3 million and resulted in net proceeds to the Company of \$540.0 million after debt issuance costs of \$7.3 million.

In 2020, the Predecessor Company repaid \$22.6 million of aggregate principal amount of Term Loan. We recognized a gain on debt extinguishment of \$2.0 million. In 2019, we repaid \$31.0 million of aggregate principal amount of Term Loan. We recognized a loss on debt extinguishment of \$0.2 million.

On November 19, 2020, upon emergence from Chapter 11 bankruptcy, the 2021 Term Loan was cancelled as part of the Plan.

We were in compliance with all of the covenants in the Term Loan for all periods that the loan was outstanding until we filed for bankruptcy on September 22, 2020.

#### Predecessor 2022 Senior Notes

On April 16, 2014, the Predecessor Company completed an offering of \$500 million of 6.25% senior secured notes due May 1, 2022, in a private offering to qualified institutional buyers ("2022 Senior Notes"). The credit facility was amended semiannually, in arrears, on May 1 and November 1. The Company received net proceeds of \$489.7 million after debt issuance costs of \$10.3 million, which resulted in an effective interest rate of 6.58% for these notes.

In 2020 and 2019, the Predecessor Company repurchased zero and \$17.0 million, respectively, of aggregate principal amount of 2022 Senior Notes in the qualified institutional market. In 2019, the Predecessor Company recognized a gain on debt extinguishment of \$1.4 million.

On November 19, 2020, upon emergence from Chapter 11 bankruptcy, the 2022 Senior Notes were cancelled as part of the Plan.

We were in compliance with all of the covenants in the indenture governing our 2022 Senior Notes for all periods that these notes were outstanding, until be filed for bankruptcy on September 22, 2020.

#### Predecessor Revolving Credit Facility

On February 22, 2018, we entered into a \$250 million revolving credit facility, with an initial maturity date of February 22, 2023, with a group of lenders with Wells Fargo, N.A., as administrative agent.

As part of the Chapter 11 filing our revolving credit facility was terminated in the third quarter of 2020. Prior to its termination, the maximum availability of credit under our revolving credit facility was limited at any time to the lesser of \$100 million or a borrowing base. The credit facility was amended in August 2020, which among other things, reduced the maximum availability under the credit facility from \$250 million to \$100 million. The borrowing base was based on percentages of eligible accounts receivable and was subject to certain reserves. If at any time borrowings and letters of credit issued under the credit facility exceeded the borrowing base, we would be required to repay an amount equal to such excess. During 2020, the Predecessor Company had no borrowings drawn under the credit facility, and certain letters of credit were issued. At September 30, 2020, we had replaced these letters of credit with cash collateralized letters of credit.

#### NOTE 8—LEASES

We had no material amount of finance leases or subleases at December 31, 2021. The following table summarizes the components of our lease costs.

		Successor				Predecessor			
		Year Ended December 31,		Period from November 20, through December 31,		Period from January 1, through ovember 19,		ar Ended	
(In millions)		021	2	020	_	2020		2019	
Operating lease cost	\$	3.8	\$	0.3	\$	12.7	\$	21.1	
Short-term lease cost		4.3		0.4		4.7		5.8	
Total lease cost	\$	8.1	\$	0.7	\$	17.4	\$	26.9	

Short-term lease costs represent costs related to leases with terms of one year or less. We had no material variable lease costs in 2021, 2020, or 2019. The following table includes other supplemental information for our operating leases.

		Succ	Predecessor				
(Dollars in millions)		Period from November 20, through December 31, 2021 2020			Period from January 1, through November 19 2020		
Cash paid for amounts included in the		021		.020	<u> </u>	2020	
measurement of our lease liabilities	\$	3.9	\$	0.3	\$	12.9	
Right-of-use assets obtained in exchange for lease							
liabilities	\$	1.5	\$	0.1	\$	0.6	
Right-of-use assets recognized upon adoption of							
the leasing standard	\$	_	\$	_	\$	0	
Weighted-average remaining lease term	2.	5 years	2	.9 years		2.9 years	
Weighted-average discount rate		4.6%		4.9%		4.9%	

The following table summarizes the maturity of our operating leases as of December 31, 2021.

	Successor			
		mber 31,		
(In millions)		2021		
2022	\$	1.4		
2023		0.9		
2024		0.8		
2025		_		
2026		_		
2027 and thereafter				
Total lease payments		3.1		
Less imputed interest		(0.1)		
Total lease liabilities	\$	3.0		

#### NOTE 9-STOCKHOLDERS' EQUITY (DEFICIT)

#### Common and Preferred Stock

On November 19, 2020, upon emergence from Chapter 11 bankruptcy, all existing shares of Predecessor common stock were cancelled, and the Successor Company issued approximately 13.7 million shares of Class A common stock and 0.3 million shares of Class B common stock. On February 16, 2022, all issued and outstanding shares of Class B common stock were converted to Class A common stock.

On November 19, 2020, upon emergence from Chapter 11 bankruptcy, the Successor Company filed an Amended and Restated Certificate of Incorporation with the Delaware Secretary of State. Pursuant to the Certificate of Incorporation, the authorized capital stock of FTSI consists of 49,000,000 shares of New Class A Common Stock and 1,000,000 shares of New Class B Common Stock and 5,000,000 shares of Preferred Stock, par value \$0.01 per share (the "Preferred Stock"). The New Class B Common Stock is identical to the New Class A Common Stock, except that such New Class B Common Stock shall not be listed for trading on any national securities exchange or NASDAQ, nor shall it be listed over-the-counter. Upon the written request of a holder of New Class B Common Stock, and in compliance with the provisions of the Certification of Incorporation, the shares of New Class B Common Stock shall be exchangeable into the same number of shares of New Class A Common Stock.

Shares of Preferred Stock may be issued in one or more classes or series from time to time, with each such class or series to consist of such number of shares and to have such designations, powers, preferences and relative, participating, optional or other rights, if any, and the qualifications, limitations or restrictions thereof, if any, with respect to each such class or series as shall be stated in the resolution or resolutions providing for the issuance of such class or series adopted by the Board.

### 2020 Reverse Stock Split

In May 2020, the Predecessor Company's board of directors (our "Board") approved a reverse stock split of the Company's issued and outstanding common stock on a one for twenty basis. The par value of the Predecessor Company's common stock and the number of shares authorized for issuance remained unchanged as a result of the reverse stock split. All common shares and stock awards presented in the consolidated financial statements have been retrospectively adjusted for the reverse stock split. In addition, the Predecessor Company transferred the listing of the Company's common stock from the New York Stock Exchange (the "NYSE") to the NYSE American.

#### Share repurchase program

In May 2019, the Predecessor Company's board of directors approved an authorization for a total share repurchase of up to \$100 million of the Predecessor Company's common stock to be executed through open market or private transactions. The authorization expired on May 14, 2020. In 2019 we repurchased approximately 3.0 million shares of common stock at an average price of \$3.34 per share for a total of \$9.9 million.

#### NOTE 10—STOCK-BASED COMPENSATION

### Predecessor 2018 Equity and Incentive Compensation Plan

Our board of directors and stockholders adopted the 2018 Equity and Incentive Compensation Plan ("2018 Plan") to attract and retain officers, employees, directors, consultants and other key personnel and to provide those persons incentives and awards for performance. The 2018 Plan originally allocated 2.8 million shares of common stock in the form of incentive stock options, non-qualified stock options, restricted stock, restricted stock units ("RSUs"), stock appreciation rights, or other stock-based awards. In 2019 our board of directors and stockholders amended and restated the 2018 Plan to increase the number of shares available for issuance by 3.6 million shares. Any shares that become available as a result of forfeiture, cancelation, expiration or cash settlement of an award are allowed to be granted again at a future date under the 2018 Plan. This plan originally was set to expire on February 1, 2028, but was terminated upon emergence from bankruptcy on November 19, 2020.

RSUs are generally valued at the market price of a share of our common stock on the date of grant. Awards granted to employees generally vest over a three or four-year period from the date of grant and are expensed on a straight-line basis over that period, which is considered to be the requisite service period.

Stock-based compensation expense for the Predecessor Company from January 1 through November 19, 2020 for these RSUs was \$10.9 million and was \$15.4 million in 2019. There were no shares granted in 2020. The weighted-average grant-date fair value per share of RSUs granted was \$51.22 in 2019. The fair value of RSUs vested was \$36.3 million and \$15.8 million in 2020 and 2019, respectively. At November 19, 2020, there was zero total unrecognized compensation cost related to unvested RSUs.

The total income tax benefit (expense) for all stock-based compensation was \$(0.1) million and \$1.2 million in 2020 and 2019 respectively; however, such benefit (expense) was substantially offset by the valuation allowance against our deferred tax assets.

#### Successor 2020 Equity and Incentive Compensation Plan

As part of the Plan, the Successor Company adopted the 2020 Equity and Incentive Compensation Plan ("2020 Plan") to attract and retain officers, employees, directors, consultants and other key personnel and to provide those persons incentives and awards for performance. The 2020 Plan originally allocated 2,160 thousand shares of common stock in the form of incentive stock options, non-qualified stock options, restricted stock units ("RSUs"), stock appreciation rights, or other stock-based awards. As of December 31, 2021, up to approximately 1,273 thousand shares were available for future grants under this plan.

Restricted Stock Units: RSUs are valued at the market price of our common stock on the date of grant. Awards granted vest over a four-year period from the date of grant. The following table summarizes the 2021 transactions related to the RSUs granted under the Successor Company's 2020 Plan.

	Number of Units (In thousands)	A Gra	eighted- verage ant-Date ir Value
Unvested balance at January 1, 2021	540	\$	14.11
Granted	29		22.18
Vested	(118)		14.11
Forfeited	(68)		14.11
Unvested balance at December 31, 2021	383	\$	14.72

Stock-based compensation expense for the Successor Company for 2021 was \$1.9 million and from November 20 through December 31, 2020, was \$0.2 million. The weighted-average grant-date fair value per share of RSUs granted was \$22.18 in 2021. The fair value of RSUs vested was \$14.11 in 2021. The fair value of RSUs forfeited was \$14.11 in 2021. At December 31, 2021, there was \$5.1 million of total unrecognized compensation cost related to unvested RSUs, which is expected to be recognized over a weighted average period of 2.9 years.

The total income tax benefit for all stock-based compensation was \$0.5 million and \$0.1 million in 2021 and 2020; respectively; however, such benefit was substantially offset by the valuation allowance against our deferred tax assets.

Performance-based Restricted Stock Unit:. The following table summarizes the 2021 transactions related to the performance-based RSUs granted under the Successor Company's 2020 Plan.

	Number	Weighted- Average
	of Units	Grant-Date
	(In thousands)	Fair Value
Unvested balance at January 1, 2021	270	\$ 10.91
Granted	6	23.45
Vested	(120)	11.86
Forfeited	(33)	10.91
Unvested balance at December 31, 2021	123	\$ 10.62

Stock-based compensation expense for the Successor Company for 2021 was \$1.7 and from November 20 through December 31, 2020, was \$0.1 million. The weighted-average grant-date fair value per share of performance-based RSUs granted was \$23.45 in 2021. The fair value of performance based RSUs vested was \$11.86 in 2021. The fair value of performance based RSUs forfeited was \$10.91 in 2021. At December 31, 2021, there was \$1.0 million of total unrecognized compensation cost related to unvested performance based RSUs, which is expected to be recognized over a weighted average period of 3.0 years.

The fair value of the performance-based RSUs granted in 2021 was determined using a Monte Carlo simulation method. Assumptions used in calculating the fair value of the performance-based RSUs granted during the year are summarized below:

Valuation assumptions:	2021 Performance- based RSU's Granted
Expected dividend yield	0%
Expected equity volatility, including peers	61.76%
Expected term (years)	7 years
Risk-free interest rate	1.37%

**Non-qualified stock options:** Stock options granted in 2020 have a four-year vesting period. As the stock options vest, the awards can be exercised up to the expiration date of the options, which is the date of the ten year anniversary from the grant date. The following table summarizes the 2021 transactions related to the stock options granted under the Successor Company's 2020 Plan.

	Number of Units (In thousands)	Weighted- Average Grant-Date Fair Value
Unvested balance at January 1, 2021	270	\$ 7.85
Granted	_	_
Vested	(59)	7.85
Forfeited	(34)	7.85
Unvested balance at December 31, 2021	177	\$ 7.85

Stock-based compensation expense for the Successor Company for 2021 was \$0.5 million and from November 20 through December 31, 2020, was \$0.1 million. There were no options granted or exercised in 2021. The fair value of options vested was \$7.85 in 2021. The fair value of options forfeited was \$7.85 in 2021. At December 31, 2021, there was \$1.3 million of total unrecognized compensation cost related to unvested options, which is expected to be recognized over a weighted average period of 2.9 years.

The fair value of the stock options is determined by applying the Black-Scholes model. There were no stock options granted during the year.

#### NOTE 11—RETIREMENT PLAN

We offer a 401(k) defined contribution retirement plan ("401(k) Plan"), which allows a participant to defer, by payroll deductions, from 0% to 100% of the participant's annual compensation, limited to certain annual maximums set by the Internal Revenue Code. The 401(k) Plan has historically provided a discretionary matching contribution to each participant's account. Company matching contributions to the 401(k) Plan are made in cash and were \$2.5 million in 2019. The Company suspended matching contributions in January 2020. As a result, there were zero company matching contributions to the 401(k) Plan in 2021 and in the Predecessor and Successor periods of 2020.

# NOTE 12—IMPAIRMENTS AND OTHER CHARGES

The following table summarizes our impairments and other charges:

	Successor				Predecessor							
	Decei	r Ended mber 31,	Period from November 20, through December 31,		November 20, through December 31,		Janu thro Novem		Period from January 1, through November 19,		Year Ended December 31,	
(In millions)	2	2021	2	2020		2	2020	_	2019			
Supply commitment charges	\$	_	\$	_		\$	9.1	9	58.5			
Impairment of assets		_		_			_		9.7			
Inventory write-down		_		_			5.1		6.4			
Transaction costs		4.3		_			18.5		_			
Employee severance costs		_		_			1.0		_			
Loss on contract termination		0.2		0.3			0.4	_				
Total impairments and other charges	\$	4.5	\$	0.3		\$	34.1	9	74.6			

#### Transaction Costs

In 2021 the Successor Company we incurred \$4.3 million of costs related to strategic initiatives including our agreement and plan of merger to be acquired by ProFrac Holdings, LLC. From January 1 through November 19, 2020, in preparation for, and prior to filing our Chapter 11 Cases, the Predecessor Company incurred and paid \$7.0 million in legal and professional fees and \$11.5 million to certain holders of our Term Loan Agreement and Secured Notes pursuant to the Restructuring Support Agreement.

#### Loss on Contract Termination

In the first quarter of 2021 (Successor), we terminated a portion of our operating lease for certain buildings and recorded a net gain of \$0.3 million. In the fourth quarter of 2021 (Successor) we incurred costs of \$0.5 million related to end of lease repair costs. In the third quarter of 2020 (Predecessor), we terminated our operating leases for containerized proppant delivery and a sand supply contract and recorded a net loss of \$0.3. In the fourth quarter of 2020 (Predecessor), we terminated a portion of our operating lease for certain buildings and recorded a net gain of \$0.4 million

#### Supply Commitment Charges

The Predecessor Company incurred supply commitment charges when our purchases of sand from certain suppliers are less than the minimum purchase commitments in our supply contracts. According to the accounting guidance for firm purchase commitments, future losses that are considered likely are also required to be recorded in the current period.

The Predecessor Company recorded aggregate charges under these supply contracts of \$9.1 million from January 1 through November 19, 2020, and \$58.5 million in 2019. These charges relate to actual purchase shortfalls incurred, as well as forecasted losses expected to be incurred and settled in future periods. These purchase shortfalls are largely due to our customers choosing to procure their own sand, often from sand mines closer to their operating areas.

In May 2019, the Predecessor Company restructured and amended our largest sand supply contract to reduce the total remaining commitment. In connection with this amendment, we recorded a supply commitment charge of \$55.0 million in the first quarter of 2019 to accelerate expected purchase shortfalls. The remaining amount of the 2019 charges represent revised estimates of our purchase shortfalls under this contract for 2019.

The Company terminated all sand supply contracts upon emergence from bankruptcy. Any amounts due as outlined in the Plan were paid upon emergence and the Company does not expect any future commitment related to these Predecessor contracts.

#### Fleet Capacity Reduction

In the fourth quarter of 2019, the Predecessor Company disposed of certain idle equipment where we believed there was no expectation of future use. The equipment we selected for disposal was comprised primarily of hydraulic fracturing pumps that were substantially depreciated. Certain hydraulic fracturing components, such as engines and transmissions that we believe to have remaining useful lives, will be removed prior to disposing of the equipment and used in our maintenance and repair activities for our remaining fleets. These disposals reduced our capacity of equipment from 34 total fleets to 28 total fleets. The amount of proceeds we received from these disposals was not significant. We recorded an asset impairment of \$4.2 million in the third quarter of 2019 in connection with these disposals.

#### Discontinued Wireline Operations

In May 2019, the Predecessor Company discontinued our wireline operations due to financial underperformance resulting from market conditions. As a result of this decision, we recorded an asset impairment of \$2.8 million and an inventory write-down of \$1.4 million in the first quarter of 2019 to adjust these assets to their estimated fair market values and net realizable values, respectively. We sold substantially all of these assets in 2019 and received net proceeds of approximately \$3.7 million.

# Inventory Write-downs

In the first nine months of 2020 (Predecessor) and in 2019, we recorded \$5.1 million and \$5.0 million of inventory write-downs to reduce excess, obsolete and slow-moving inventory to its estimated net realizable value.

#### Other Impairments

In the second quarter of 2019, the Predecessor Company recorded \$2.7 million of impairments for certain land and buildings that we no longer use. We are closely monitoring current industry conditions and future expectations. If industry conditions decline, we may be subject to impairments of long-lived assets or intangible assets in future periods.

#### Employee Severance Cost

In the first nine months of 2020 (Predecessor), we incurred employee severance costs of \$1.0 million in connection with our cost reduction measures to mitigate losses from the decline in customer activity levels due to the low commodity price environment.

#### NOTE 13—ASSET DISPOSAL

In the third quarter of 2021 (Successor), we received \$3.1 million of proceeds and recognized a \$1.6 million loss on the sale of various assets; the majority of which is from the sale of our Shreveport facility and related assets.

#### NOTE 14—INCOME TAXES

The following table summarizes the components of income tax expense (benefit):

	Succ	essor	Predecessor			
(In millions)	Year Ended December 31, 2021	Period from November 20, through December 31, 2020	Period from January 1, through November 19, 2020	Year Ended December 31, 2019		
Current:						
Federal	\$ —	\$ —	\$ —	\$ —		
State	_		0.2	0.5		
Foreign				0.9		
Total current		_	0.2	1.4		
Total deferred						
Income tax expense	\$	<u> </u>	\$ 0.2	\$ 1.4		

Actual income tax expense differed from the amount computed by applying the statutory federal income tax rate to (loss) before income taxes as follows:

	Succe	ssor	Predecess	or
(In millions)	Year Ended December 31, 2021	Period from November 20, through December 31, 2020	Period from January 1, through November 19, 2020	Year Ended December 31, 2019
Loss before income taxes	\$ (21.2)	\$ (13.4)	\$ (24.2)	\$ (71.5)
Statutory federal income tax rate	21.0%	21.0%	21.0%	21.0%
Federal income tax benefit at statutory rate	(4.5)	(2.8)	(5.1)	(15.0)
State income tax benefit, net of federal effect	(0.2)	(0.2)	(1.6)	(0.1)
Effect of changes in income apportionment				
amongst states	(1.4)	_	(3.6)	12.0
Stock-based compensation	0.4	_	7.4	2.1
Reorganization adjustments	(541.2)	0.3	577.1	_
Expired state net operating losses	0.3	_	_	_
Other items, net	0.2	0.1	1.0	1.7
Change in valuation allowance	546.4	2.6	(575.0)	0.7
Income tax expense	<u>\$</u>	<u>\$</u>	\$ 0.2	\$ 1.4
Effective tax rate	%	%	(0.8)%	(2.0)%

Due to the mobile nature of our operations, the apportionment of annual income that we earn in a state can change as the operations of our business adjust to market conditions. States have different income tax rates and therefore the weighted-average state tax rate that we apply to our taxable and deductible temporary differences and net operating loss carryforwards can also change over time. The resulting effects of state apportionment on our deferred tax assets and deferred tax liabilities are recognized to reflect the future impact of these changes and affects our overall effective tax rate; however, these changes are currently offset by corresponding changes in our valuation allowance.

In March 2020, the President of the United States signed into law the Coronavirus Aid, Relief, and Economic Security Act ("CARES Act"). The CARES Act includes several U.S. income tax provisions related to, among

other things, net operating loss carrybacks, alternative minimum tax credits, modifications to the net interest deduction limitations, and technical amendments regarding the income tax depreciation of qualified improvement property placed in service after December 31, 2017. The CARES Act did not have a material impact on the Company's financial results.

The tax effects of temporary differences that give rise to significant portions of the deferred tax assets and deferred tax liabilities are presented below:

(In millions)	December 31, 2021	December 31, 2020
Deferred tax assets:		
Goodwill and intangible assets	\$ 184.1	\$ 19.5
Federal net operating loss carryforwards	403.8	65.6
State net operating loss carryforwards, net of federal benefit	43.7	2.6
Property, plant, and equipment	7.4	_
Interest carryforward	_	9.6
Accrued liabilities	1.6	0.8
Operating lease liability	0.7	1.4
Stock-based compensation	0.2	0.1
Other	4.7	0.8
Gross deferred tax assets	646.2	100.4
Valuation allowance	(645.7)	(99.3)
Total deferred tax assets	0.5	1.1
Deferred tax liabilities:		
Property, plant, and equipment	_	0.1
Operating lease right-of-use assets	0.5	1.0
Total deferred tax liabilities	0.5	1.1
Net deferred tax asset	<u>\$</u>	<u>\$</u>

Management assesses the available positive and negative evidence to estimate whether sufficient future taxable income will be generated to permit use of the existing deferred tax assets. A significant piece of objective negative evidence evaluated was the cumulative loss incurred over the three-year period ended December 31, 2021. Such objective evidence limits the ability to consider other subjective evidence, such as our projections for future growth. The amount of the deferred tax assets considered realizable, however, could be adjusted if estimates of future taxable income during the carryforward period are increased or if objective negative evidence in the form of cumulative losses is no longer present and additional weight is given to subjective evidence such as our projections for growth.

On August 22, 2020, we entered into a restructuring support agreement with certain of our note holders and lenders (collectively the "consenting creditors"). The restructuring support agreement included a cash premium payable to the consenting creditors. The payment of the cash premium resulted in a significant modification of the notes and term loan for tax purposes. Under the Internal Revenue Code (IRC), the significant modification resulted in cancellation of debt (COD) income, part of which is excluded from taxable income under IRC Section 108. However, the excluded COD reduces the Company's tax attributes, including its NOL.

Upon emergence from bankruptcy, the Company experienced an "ownership change" as defined by Section 382 of the Internal Revenue Code of 1986, as amended (the "Code"). Under Section 382 of the Code, the Company's net operating loss carryforwards and other tax attributes (collectively, the "Tax Benefits") are potentially subject to various limitations going forward. For the period ending December 31, 2020, the Company had not determined

whether it would utilize Section 382(1)(5) of the Code, accordingly, the limitation was previously determined under Section 382(1)(6) of the Code. If the Company qualifies for Section 382(1)(5) of the Code, the use of Tax Benefits would not be limited as a result of the ownership change occurring on the bankruptcy emergence. After additional consideration and review, the Company concluded it qualifies for and would utilize Section 382(1)(5) of the Code and filed its 2020 tax return accordingly. The determination to utilize IRC Section 382(1)(5) was made in October 2021 and relied on improving market conditions, increased levels of profitability, and probability of potential scenarios that were not present or changed subsequent to our 2020 year-end financial statements. Consequently, at year ended December 31, 2021, we reinstated the federal and state NOLs and built-in-losses that were reduced at year ended December 31, 2020, offset by a corresponding change in our valuation allowance.

The benefit of utilizing IRC Section 382(l)(5) is tempered by rules that potentially reduce the utility of Section 382(l)(5). If a second ownership change occurs within two years of an ownership change that qualifies under Section 382(l)(5), the Section 382 limitation with respect to the second ownership change is zero. If a second ownership change occurs prior to November 19, 2022, the NOL balance as of this date would be subject to a zero limitation and any built-in losses recognized within 60 months of the second ownership change would also be limited to zero.

At December 31, 2021, our gross federal net operating loss carryforwards were approximately \$1,923.1 million, of which \$1,489.5 million will expire on various dates between 2032 and 2036 with the remaining losses carried forward indefinitely. Our gross state net operating loss carryforwards were approximately \$698.5 million, of which \$633.1 million will expire at various dates between 2022-2041 with the remaining carried forward indefinitely

At December 31, 2021 and 2020, we had no material liability for uncertain tax positions. We recognize interest and penalties related to UTBs on the income tax expense line in the accompanying consolidated statement of operations. FTS International, Inc. and its U.S. subsidiaries join in the filing of a U.S. federal consolidated income tax return. Our income tax returns are currently subject to examination in federal and state jurisdictions primarily for tax years from 2017 through 2020.

If the proposed acquisition by ProFrac is approved by shareholders on March 3, 2022, this will qualify as a second ownership change within two years of the ownership change that qualified under Section 382(I)(5). The NOLs presented in the Net Deferred Tax Asset table above will have no value and the Goodwill and Intangible deferred tax assets will be severely reduced. This may negatively affect our results of operations for future periods by requiring us to pay U.S. federal and state income taxes that we would not have been required to pay had the NOL carryforwards and intangible deductions existing at the second ownership change been available to us.

#### NOTE 15—COMMITMENTS AND CONTINGENCIES

#### Purchase Obligations

At December 31, 2021, the Successor Company's future minimum purchase commitments is summarized below:

(In millions)	2022	2023	2024	2025	2026	Thereafter
Other purchase obligations	0.3	0.3	0.1			
Total purchase obligations	\$0.3	<u>\$0.3</u>	\$0.1	<u>\$</u>	<u>\$</u>	<u>\$</u>

#### Litigation

In the ordinary course of business, we are subject to various legal proceedings and claims, some of which may not be covered by insurance. Some of these legal proceedings and claims are in early stages, and many of them

seek an indeterminate amount of damages. We estimate and provide for potential losses that may arise out of legal proceedings and claims to the extent that such losses are probable and can be reasonably estimated. Significant judgment is required in making these estimates and our final liabilities may ultimately be materially different from these estimates. When preparing our estimates, we consider, among other factors, the progress of each legal proceeding and claim, our experience and the experience of others in similar legal proceedings and claims, and the opinions and views of legal counsel. Legal costs related to litigation contingencies are expensed as incurred.

With respect to the litigation matter below, if there is an adverse outcome individually or collectively, there could be a material adverse effect on the Company's consolidated financial position or results of operations. Litigation matters are subject to inherent uncertainties and management's view of these matters may change in the future. Therefore, there can be no assurance as to the ultimate outcome of litigation matters. Regardless of the outcome, any such litigation and claims can have an adverse impact on the Company because of defense and settlement costs, diversion of management resources and other factors.

Patterson v. FTS International Manufacturing, LLC and FTS International Services, LLC: On June 24, 2015, Joshua Patterson filed a lawsuit against the Company in the 115th Judicial District Court of Upshur County, Texas, alleging, among other things, that the Company was negligent with respect to an automobile accident in 2013. Mr. Patterson sought monetary relief of more than \$1 million. On July 19, 2018, a jury returned a verdict of approximately \$100 million, including punitive damages, against the Company. The trial court reduced the judgment on November 12, 2018, to approximately \$33 million. The Company's insurance carriers appealed and the Twelfth Court of Appeals reversed the verdict in its entirety on August 26, 2020, remanding the case for a new trial. The Company's insurance carriers are currently appealing one of the appellate findings with the Texas Supreme Court. No new trial date has been set. While the outcome of this case is uncertain, the Company has met its insurance deductible for this matter and we do not expect the ultimate resolution of this case to have a material adverse effect on our consolidated financial statements.

We believe that costs associated with other legal matters will not have a material adverse effect on our consolidated financial statements.

### NOTE 16—EARNINGS (LOSS) PER SHARE

The numerators and denominators of the basic and diluted earnings (loss) per share ("EPS") computations for our common stock are calculated as follows:

	Successor			1		Prede	cessor		
(In millions, except per share amounts)	Dece	r Ended ember 31, 2021	Nov tl	iod from ember 20, nrough ember 31, 2020		Jan thi Nove	od from uary 1, rough mber 19,	Dece	er Ended ember 31, 2019
Numerator:									
Net loss	\$	(21.2)	\$	(13.4)	9	\$	(24.4)	\$	(72.9)
Net loss attributable to common stockholders used for basic and diluted EPS computation	\$	(21.2)	\$	(13.4)	9	\$	(24.4)	<u>\$</u>	(72.9)
Denominator:									
Weighted average shares used for basic EPS computation <sup>(1)</sup>		14,036		13,990			5,377		5,440
Effect of dilutive securities:									
Restricted stock units <sup>(2)</sup>		_		_			_		_
Dilutive potential common shares		_		_	-	,	_		
Number of shares used for diluted EPS computation (in thousands)		14,036		13,990			5,377		5,440
Basic and diluted EPS	\$	(1.51)	\$	(0.96)	9	S	(4.54)	\$	(13.40)

<sup>(1)</sup> The weighted average shares outstanding has been adjusted to give effect the 20: 1 reverse stock split in May 2020.

<sup>(2)</sup> The dilutive effect of employee restricted stock units granted under our 2018 LTIP and 2020 LTIP was either immaterial or antidilutive for 2021, 2020 and 2019.

#### REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Stockholders and Board of Directors

U.S. Well Services, Inc.:

#### Opinion on the Consolidated Financial Statements

We have audited the accompanying consolidated balance sheets of U.S. Well Services, Inc. and subsidiaries (the Company) as of December 31, 2021 and 2020, the related consolidated statements of operations, stockholders' equity (deficit), and cash flows for the years then ended, and the related notes (collectively, the consolidated financial statements). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2021 and 2020, and the results of its operations and its cash flows for the years then ended, in conformity with U.S. generally accepted accounting principles.

#### Basis for Opinion

These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) (PCAOB) and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audits, we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the consolidated financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the consolidated financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ KPMG LLP

We have served as the Company's auditor since 2012.

Houston, Texas March 30, 2022

# U.S. WELL SERVICES, INC. CONSOLIDATED BALANCE SHEETS

(in thousands, except share and per share amounts)

	Decem	
ASSETS	2021	2020
Current assets:		
Cash and cash equivalents	\$ 6,384	\$ 3,693
Restricted cash	2,736	1,569
Accounts receivable (net of allowance for doubtful accounts of \$0 and \$12,000 as of December 31, 2021 and 2020, respectively)	25,743	44,393
Inventory, net	6,351	7,965
Assets held for sale	2,043	_
Prepaids and other current assets	18,748	10,707
Total current assets	62,005	68,327
Property and equipment, net	162,664	235,332
Intangible assets, net	12,500	13,466
Goodwill	4,971	4,971
Other assets	1,417	1,127
TOTAL ASSETS	\$ 243,557	\$ 323,223
LIABILITIES, MEZZANINE EQUITY AND STOCKHOLDERS' DEFICIT	====	=====
Current liabilities:		
Accounts payable	\$ 29,180	\$ 36,362
Accrued expenses and other current liabilities	16,842	14,781
Notes payable	2,320	998
Current portion of long-term debt	5,000	10,000
Current portion of equipment financing	3,412	3,519
Current portion of capital lease obligations	1,092	54
Total current liabilities	57,846	65,714
Warrant liabilities	3,557	1,619
Convertible senior notes	105,769	_
Long-term debt	167,507	274,555
Long-term equipment financing	5,128	9,347
Long-term capital lease obligations	2,112	_
Other long-term liabilities	6,875	3,539
Total liabilities	348,794	354,774
Commitments and contingencies (NOTE 12)		
Mezzanine equity:		
Series A Redeemable Convertible Preferred Stock, par value \$0.0001 per share; 55,000 shares authorized; 19,610 shares and 50,000 shares issued and		
outstanding as of December 31, 2021 and 2020, respectively; aggregate liquidation preference of \$27,274 and \$60,418 as of December 31, 2021 and		
2020, respectively	23,866	50,975
Series B Redeemable Convertible Preferred Stock, par value \$0.0001 per share; 22,050 shares authorized; 0 shares and 22,050 shares issued and outstanding as of December 31, 2021 and 2020, respectively; aggregate liquidation preference of \$0 and \$24,100 as of December 31, 2021 and 2020,		
respectively	_	22,686
Stockholders' deficit:		
Class A Common Stock, par value of \$0.0001 per share; 400,000,000 shares authorized; 53,148,952 shares and 20,718,659 shares issued and outstanding as of December 31, 2021 and 2020, respectively <sup>(1)</sup>	5	2
Class B Common Stock, par value of \$0.0001 per share; 20,000,000 shares authorized; 0 shares and 2,302,936 shares issued and outstanding as of December 31, 2021 and 2020, respectively		_
Additional paid in capital (1)	263,928	217,217
Accumulated deficit	(393,036)	(322,431)
Total Stockholders' deficit	(129,103)	(105,212)
TOTAL LIABILITIES, MEZZANINE EQUITY AND STOCKHOLDERS' DEFICIT	\$ 243,557	\$ 323,223

<sup>(1)</sup> Prior periods have been adjusted to reflect the 1-for-3.5 reverse stock split on September 30, 2021. See Note 2, Reverse Stock Split, for details.

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these consolidated financial statements}.$ 

# U.S. WELL SERVICES, INC. CONSOLIDATED STATEMENTS OF OPERATIONS

(in thousands, except per share amounts)

		Ended iber 31.
	2021	2020
Revenue	\$250,463	\$ 244,007
Costs and expenses:		
Cost of services (excluding depreciation and amortization)	221,364	187,803
Depreciation and amortization	35,444	80,353
Selling, general and administrative expenses	32,578	43,632
Impairment of long-lived assets	_	147,543
Litigation settlement	35,000	_
(Gain) loss on disposal of assets	(21,896)	7,112
Loss from operations	(52,027)	(222,436)
Interest expense, net	(33,370)	(25,226)
Change in fair value of warrant liabilities	(2,152)	6,342
Patent license sales	22,500	_
Loss on extinguishment of debt, net	(6,142)	_
Other income	515	108
Loss before income taxes	(70,676)	(241,212)
Income tax benefit	(27)	(824)
Net loss	(70,649)	(240,388)
Net loss attributable to noncontrolling interest	(44)	(11,048)
Net loss attributable to U.S. Well Services, Inc.	(70,605)	(229,340)
Dividends accrued on Series A preferred stock	(5,857)	(7,214)
Dividends accrued on Series B preferred stock	(4,591)	(2,049)
Deemed and imputed dividends on Series A preferred stock	(750)	(13,022)
Deemed dividends on Series B preferred stock	(7,178)	(564)
Exchange of Series A preferred stock for convertible senior notes	8,936	
Net loss attributable to U.S. Well Services, Inc. common stockholders	\$ (80,045)	\$ (252,189)
Loss per common share (See Note 15):		
Basic and diluted <sup>(1)</sup>	\$ (2.44)	\$ (13.32)
Weighted average common shares outstanding:		
Basic and diluted <sup>(1)</sup>	32,394	18,512

<sup>(1)</sup> Prior periods have been adjusted to reflect the 1-for-3.5 reverse stock split on September 30, 2021. See Note 2, Reverse Stock Split, for details.

 ${\it The\ accompanying\ notes\ are\ an\ integral\ part\ of\ these\ consolidated\ financial\ statements}.$ 

# U.S. WELL SERVICES, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

	Year Ended December 31.	
	2021	2020
CASH FLOWS FROM OPERATING ACTIVITIES:		
Net loss	\$ (70,649)	\$(240,388)
Adjustments to reconcile net loss to cash provided by (used in) operating activities:		
Depreciation and amortization	35,444	80,353
Change in fair value of warrant liabilities	2,152	(6,342)
Impairment of long-lived assets	_	147,543
Provision for losses on accounts receivable	9	12,031
Provision for losses on inventory obsolescence	2,461	620
(Gain) loss on disposal of assets	(21,896)	7,112
Convertible senior notes converted into sales of patent licenses	(22,500)	_
Amortization of debt discount, premium and issuance costs	6,401	4,896
Paid-in-kind interest on convertible senior notes	9,686	_
Loss on extinguishment of debt, net	6,142	_
Share-based compensation expense	11,694	10,056
Changes in assets and liabilities:		
Accounts receivable	18,642	23,118
Inventory	(847)	468
Prepaids and other current assets	(6,063)	6,288
Accounts payable	888	(23,999)
Accrued liabilities	(4,387)	(6,208)
Accrued interest	13,546	(6,932)
Net cash provided by (used in) operating activities	(19,277)	8,616
CASH FLOWS FROM INVESTING ACTIVITIES:	<u>(==,=, +</u> )	
Purchase of property and equipment	(57,724)	(55,943)
Proceeds from sale of property and equipment and insurance proceeds from damaged property and equipment	113,880	20,944
Net cash provided by (used in) investing activities	56,156	(34,999)
* * * * * * * * * * * * * * * * * * * *		(34,999)
CASH FLOWS FROM FINANCING ACTIVITIES:	22.160	60.055
Proceeds from revolving credit facility	32,160	68,957
Repayments of revolving credit facility	(41,700)	(85,497)
Proceeds from issuance of long-term debt	3,004	31,996
Repayments of long-term debt	(125,506)	(3,750)
Payment of fees related to debt extinguishment	(2,025)	_
Proceeds from issuance of convertible senior notes	97,500	
Proceeds from issuance of notes payable	10,699	1,121
Repayments of notes payable	(9,377)	(7,507)
Repayments of amounts under equipment financing	(4,327)	(3,199)
Principal payments under capital lease obligations	(620)	(10,474)
Proceeds from issuance of preferred stock and warrants, net	_	19,596
Proceeds from issuance of common stock, net	14,667	400
Deferred financing costs	(7,496)	(21,402)
Net cash used in financing activities	(33,021)	(9,759)
Net increase (decrease) in cash and cash equivalents and restricted cash	3,858	(36,142)
Cash and cash equivalents and restricted cash, beginning of period	5,262	41,404
Cash and cash equivalents and restricted cash, end of period	\$ 9,120	\$ 5,262
	7,120	<del>- 2,202</del>

 $\label{thm:company:c$ 

# U.S. WELL SERVICES, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (continued)

(in thousands)

	Year Ended December 31,	
	2021	2020
Supplemental cash flow disclosure:		
Interest paid	\$ 2,423	\$26,287
Income tax (refunds received) paid	(810)	144
Non-cash investing and financing activities:		
Issuance of Class A common stock to senior secured term loan lenders	_	1,438
Issuance of Series B preferred stock to senior secured term loan lenders	_	1,050
Conversion of Series A preferred stock to Class A common stock	_	4,852
Exchange of Series A preferred stock for convertible senior notes	24,780	_
Conversion of Series B preferred stock to Class A common stock	27,277	_
Deemed and imputed dividends on Series A preferred stock	750	13,022
Accrued Series A preferred stock dividends	5,857	7,214
Accrued Series B preferred stock dividends	4,591	2,049
Changes in accrued and unpaid capital expenditures	8,070	9,818
Assets under capital lease obligations	3,595	229

The accompanying notes are an integral part of these consolidated financial statements.

# U.S. WELL SERVICES, INC. CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (DEFICIT)

(in thousands, except share amounts)

				Additional Paid in Accumulated		Noncontrolling	Total Equity	
	Shares	Amount	Shares	Amount	Capital <sup>(1)</sup>	Deficit	Interest	(Deficit)
Balance at December 31, 2019	17,959,321	\$ 2	5,500,692	\$ 1	\$225,385	\$ (93,091)	\$ 10,633	\$ 142,930
Class A common stock issuance	1,806,251	_	_	_	1,817	_	_	1,817
Reclassification of warrant liability upon forfeiture	_	_	_	_	203	_	_	203
Conversion of Class B common stock to Class A								
common stock	913,645	_	(3,197,756)	(1)	1	_	_	_
Conversion of Series A preferred stock to Class A								
common stock	250,414	_	_	_	4,852	_	_	4,852
Share-based compensation	_	_	_	_	7,314	_	415	7,729
Tax withholding related to vesting of share-based								
compensation	(44,073)	_	_	_	(70)	_	_	(70)
Restricted stock forfeitures	(166,899)							_
Deemed and imputed dividends on Series A								
preferred stock	_	_	_	_	(13,022)	_	_	(13,022)
Accrued Series A preferred stock dividends					(7,214)			(7,214)
Accrued Series B preferred stock dividends	_	_	_	_	(2,049)	_	_	(2,049)
Net loss						(229,340)	(11,048)	(240,388)
Balance at December 31, 2020	20,718,659	\$ 2	2,302,936	<u>\$</u>	\$217,217	\$ (322,431)	<u> </u>	<u>\$(105,212)</u>
Class A common stock issuance	5,091,800	1			14,347			14,348
Class A common stock issuance for reverse stock								
split round up	24,218	_	_	_	_	_	_	_
Reclassification of warrant liability upon forfeiture	_	_	_	_	232	_	_	232
Conversion of Class B common stock to Class A								
common stock	657,982	_	(2,302,936)	_	_	_	_	_
Conversion of Series B preferred stock to Class A								
common stock	26,615,215	2	_	_	27,275	_	_	27,277
Exchange of Series A preferred stock for convertible								
senior notes	_	_	_	_	8,936	_	_	8,936
Share-based compensation	_	_	_	_	7,269	_	44	7,313
Restricted stock grants	88,025	_	_	_	_	_	_	_
Tax withholding related to vesting of share-based								
compensation	(36,352)	_	_	_	(150)	_	_	(150)
Restricted stock forfeitures	(10,595)	_	_	_	— (150)	_	_	— (130 <i>)</i>
Deemed and imputed dividends on Series A	(,)							
preferred stock	_	_	_	_	(750)	_	_	(750)
Accrued Series A preferred stock dividends	_	_	_	_	(5,857)	_	_	(5,857)
Accrued Series B preferred stock dividends	_		_	_	(4,591)	_	_	(4,591)
Net loss	_	_	_	_	_	(70,605)	(44)	(70,649)
Balance at December 31, 2021	53,148,952	\$ 5		\$ —	\$263,928	\$ (393,036)	\$	\$(129,103)

<sup>(1)</sup> Prior periods have been adjusted to reflect the 1-for-3.5 reverse stock split on September 30, 2021. See Note 2, Reverse Stock Split, for details.

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these consolidated financial statements}.$ 

(in thousands, except share and per share amounts, or where otherwise noted)

#### NOTE 1—DESCRIPTION OF BUSINESS

U.S. Well Services, Inc. (the "Company," "we," "us" or "our"), f/k/a Matlin & Partners Acquisition Corp, is a Houston, Texas-based technology-focused oilfield service company focused on electric powered pressure pumping services for oil and natural gas exploration and production ("E&P") companies in the United States. The process of pressure pumping involves pumping a pressurized stream of fluid—typically a mixture of water, chemicals, and proppant—into a well casing or tubing to cause the underground mineral formation to fracture or crack. Fractures release trapped hydrocarbon particles and provide a conductive channel for the oil or natural gas to flow freely to the wellbore for collection. The propping agent or proppant becomes lodged in the cracks created by the stimulation process, "propping" them open to facilitate the flow of hydrocarbons from the reservoir to the well.

The Company's fleets consist mostly of all-electric, mobile pressure pumping equipment and other auxiliary heavy equipment to perform stimulation services. The Company's Clean Fleet® electric fleets replace the traditional engines, transmissions, and radiators used in conventional diesel fleets with electric motors powered by electricity generated by natural gas-fueled turbine generators. The Company utilizes high-pressure hydraulic fracturing pumps mounted on trailers and refers to the group of pump trailers and other equipment necessary to perform a typical job as a "fleet" and the personnel assigned to each fleet as a "crew". In May 2021, the Company announced its commitment to becoming an all-electric pressure pumping services provider and since then it has sold most of its legacy, diesel-powered pressure pumping equipment. We have retained some of our legacy, diesel-powered pressure pumping equipment for use during our transition to support our electric fleets and bridge the time gap between our customers' current service needs and the deployment of our newbuild Nyx Clean Fleets®. Upon delivery, our Nyx Clean Fleets® are intended to replace any conventional fleet in operation.

#### NOTE 2—SIGNIFICANT ACCOUNTING POLICIES

#### Basis of Presentation

The accompanying consolidated financial statements have been prepared in conformity with generally accepted accounting principles in the United States of America ("GAAP") and pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC").

Our operations are organized into a single business segment, which consists of pressure pumping services, and we have one reportable geographical business segment, the United States.

#### Reverse Stock Split

At the annual meeting of the Company's stockholders held on May 14, 2021, the Company's stockholders approved a proposal to amend the Company's certificate of incorporation to effect a reverse stock split at a ratio to be determined by the Company's Board of Directors within a specified range. On September 30, 2021, the Company effected a 1-for-3.5 reverse split of its Class A common stock. All owners of record as of September 30, 2021 received one issued and outstanding share of the Company's Class A common stock in exchange for three and one half outstanding shares of the Company's Class A common stock were issued as a result of the reverse stock split. Any fractional shares in connection with the reverse stock split were rounded up to the nearest whole share and no stockholders received cash in lieu of fractional shares. The reverse stock split had no impact on the number of shares of Class A common stock the Company is authorized to issue pursuant to its certificate of incorporation or on the par value per share of the Class A common stock. Proportional adjustments were made to the number of shares of Class A common stock issuable upon exercise or conversion of the Company's equity awards, convertible preferred stock and warrants,

(in thousands, except share and per share amounts, or where otherwise noted)

as well as the applicable exercise price. All share and per share information included in this Annual Report on Forml0-K has been retroactively adjusted to reflect the impact of the reverse stock split.

#### Principles of Consolidation

The consolidated financial statements comprise the financial statements of the Company and its wholly owned subsidiaries. Subsidiaries are fully consolidated from the date of acquisition, being the date on which the Company obtains control, and continue to be consolidated until the date when such control ceases. The financial statements of the subsidiaries are prepared for the same reporting period as the Company. All significant intercompany balances and transactions are eliminated upon consolidation.

#### Use of Estimates

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. We regularly evaluate estimates and judgments based on historical experience and other relevant facts and circumstances. Significant estimates included in these financial statements primarily relate to allowance for doubtful accounts, allowance for inventory obsolescence, estimated useful lives and valuation of long-lived assets, impairment assessments of goodwill and other long-lived assets, estimates of fair value of warrant liabilities, term loan, and convertible senior notes, and the valuation of share-based compensation and certain equity instruments. Actual results could differ from those estimates.

#### Cash and Cash Equivalents

Cash equivalents are highly liquid investments with an original maturity at the date of acquisition of three months or less. Cash and cash equivalents consist of cash on deposit with domestic banks and, at times, may exceed federally insured limits.

#### Restricted Cash

Cash and cash equivalents that are restricted as to withdrawal or use under the terms of certain contractual agreements, or are reserved for a specific purpose, and not readily available for immediate or general use are recorded in restricted cash in our consolidated balance sheets. As of December 31, 2021, restricted cash consisted of \$729 transferred into a trust account to support our workers' compensation obligations and \$2,007 for use in prepayment of the Senior Secured Term Loan. As of December 31, 2020, restricted cash consisted of \$513 transferred into a trust account to support our workers' compensation obligations and \$1,056 for use in approved capital expenditures.

The following table provides a reconciliation of the amount of cash and cash equivalents and restricted cash reported on the consolidated balance sheets that sum to the total of the same amounts shown on the consolidated statements of cash flows:

	Decem	ber 31,
	2021	2020
Cash and cash equivalents	<u>\$ 6,384</u>	\$ 3,693
Restricted cash	2,736	1,569
Cash and cash equivalents and restricted cash	\$ 9,120	\$ 5,262

(in thousands, except share and per share amounts, or where otherwise noted)

#### Accounts Receivable

Accounts receivable are recorded at their outstanding balances adjusted for an allowance for doubtful accounts. The allowance for doubtful accounts is determined by analyzing the payment history and credit worthiness of each customer. Receivable balances are charged off when they are considered uncollectible by management. Recoveries of receivables previously charged off are recorded as income when received.

During the year ended December 31, 2021, the Company entered into an Assignment of Claim Agreement (the "Assignment") with a third-party, whereby the Company transferred to the third-party all right, title, and interest in the Company's claim in the amount of \$14,470 in connection with a customer's bankruptcy. The Assignment was for consideration of \$2,478, which the Company received on April 26, 2021. During the first quarter of 2021, the Company wrote-off the related receivables of \$12,000, which was the unrealized amount of the claim assigned and was previously reserved for in full as of December 31, 2020. As of December 31, 2021, the Company did not record an allowance for doubtful accounts.

The following table shows the change in allowance for doubtful accounts:

	Decembe	er 31,
	2021	2020
Balance at beginning of period	\$ 12,000	\$ 22
Charges to costs and expenses	9	12,031
Recoveries and write-offs	(12,009)	(53)
Balance at end of period	<u>\$</u>	\$12,000

#### Inventory

Inventory consists of proppant, chemicals, and other consumable materials and supplies used in our pressure pumping operations. Inventories are stated at the lower of cost or net realizable value. Cost is determined principally on a first-in-first-out cost basis. All inventories are purchased for use by the Company in the delivery of its services with no inventory being sold separately to outside parties. Inventory quantities on hand are reviewed regularly and write-downs for obsolete inventory are recorded based on our forecast of the inventory item demand in the near future.

During the year ended December 31, 2021, the Company recorded \$2,461 of reserves for losses on inventory obsolescence primarily related to conventional diesel parts. As of December 31, 2021 and 2020, the Company had established inventory reserves of \$1,272 and \$315, respectively, for obsolete and slow-moving inventory. The following table shows the change in the inventory reserves:

	Decembe	er 31,
	2021	2020
Balance at beginning of period	\$ 315	\$ 579
Charges to costs and expenses	2,461	620
Recoveries and write-offs	_(1,504)	(884)
Balance at end of period	<u>\$ 1,272</u>	\$ 315

#### Assets Held for Sale

Assets that are classified as held for sale are measured at the lower of their carrying amount or fair value less expected selling costs ("estimated selling price") with a loss recognized to the extent that the carrying amount

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

exceeds the estimated selling price. The classification is applicable at the date upon which the sale of assets is probable and the assets are available for immediate sale in their present condition. Upon determining that an asset meets the criteria to be classified as held for sale, the Company ceases depreciation and reports the assets, if material, in assets held for sale in its consolidated balance sheets.

When the net carrying value of an asset designated as held for sale exceeds its estimated fair value, which we estimate based on the estimated selling price, we recognize the difference as an impairment charge. When an impairment charge is recorded, subsequent changes to the estimated selling price of assets held for sale are recorded as gains or losses to the consolidated statements of operations wherein the recognition of subsequent gains is limited to the cumulative loss previously recognized. During the year ended December 31, 2021, the Company recorded no impairment charges on its held for sale assets

#### Property and Equipment

Property and equipment are carried at cost, with depreciation provided on a straight-line basis over their estimated useful lives. Expenditures for renewals and betterments that extend the lives of the assets are capitalized. Amounts spent for maintenance and repairs, which do not improve or extend the life of the related asset, are charged to expense as incurred.

The Company separately identifies and accounts for certain critical components of its pressure pumping units including the engine, transmission, and pump, which requires us to separately estimate the useful lives of these components. For our other service equipment, we do not separately identify and track depreciation of specific original components. When we replace components of these assets, we typically estimate the net book values of the components that are retired, which are based primarily upon their replacement costs, their ages, and their original estimated useful lives.

In the first quarter of 2020, our review of impairment of long-lived assets necessitated a review of the useful lives of our property and equipment. Current trends in pressure pumping equipment operating conditions, such as increasing treating pressures and higher pumping rates, along with the increase in daily pumping time are shortening the useful life of certain critical components we use. We determined that the average useful life of fluid ends and fuel injectors was less than one year, which resulted in our determination that costs associated with the replacement of these components would no longer be capitalized, but instead expensed as they are used in operations. This change in accounting estimate was made effective in March 2020 and accounted for prospectively.

### Impairment of Long-lived Assets

Long-lived assets, such as property and equipment and amortizable identifiable intangible assets are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. When making this assessment, the following factors are considered: current operating results, trends, and prospects, as well as the effects of obsolescence, demand, competition, and other economic factors. We determine recoverability by evaluating whether the undiscounted estimated future net cash flows of the asset or asset group are less than its carrying value. When impairment is indicated, we proceed to Step 2 of the impairment test and measure the impairment as the amount by which the assets carrying value exceeds its fair value. Management considers several factors such as estimated future cash flows, appraisals, and current market value analysis in determining fair value. Assets are written down to fair value if the concluded current fair value is below the net carrying value. Impairment loss on long-lived assets of \$147,543 was recorded during the first quarter of 2020 (refer to "Note 6 – Goodwill and Intangible Assets"). No impairment of long-lived assets was recorded for the year ended December 31, 2021.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

#### Goodwill

Goodwill is not amortized, but is reviewed for impairment annually, or more frequently when events or changes in circumstances indicate that the carrying value may not be recoverable. Judgements regarding indicators of potential impairment are based on market conditions and operational performance of the business

As of December 31 of each year, or as required, the Company performs an impairment analysis of goodwill. The Company may assess its goodwill for impairment initially using a qualitative approach to determine whether conditions exist that indicate it is more likely than not that a reporting unit's carrying value is greater than its fair value, and if such conditions are identified, then a quantitative analysis will be performed to determine if there is any impairment. The Company may also elect to perform a single step quantitative analysis in which the carrying amount of the reporting unit is compared to its fair value, which the Company estimates using a guideline public company method, a form of the market approach. The guideline public company method utilizes the trading multiples of similarly traded public companies as they relate to the Company's operating metrics. An impairment charge would be recognized for the amount by which the carrying amount of the reporting unit exceeds the reporting unit's fair value, and only limited to the total amount of goodwill allocated to the reporting unit.

#### **Deferred Financing Costs**

Costs incurred to obtain financing are capitalized and amortized to interest expense using the effective interest method over the contractual term of the debt. At the balance sheet date, deferred financing costs related to term loans are presented as a direct deduction from the debt liability, while deferred financing costs related to the revolver facility are presented as deferred financing costs, net, on the consolidated balance sheets.

#### Warrants

The Company evaluates all its financial instruments, including issued stock purchase warrants, to determine if such instruments are derivatives or contain features that qualify as embedded derivatives, pursuant to Accounting Standards Codification ("ASC") 480, Distinguishing Liabilities from Equity and ASC 815-15, Derivatives and Hedging—Embedded Derivatives. The classification of derivative instruments, including whether such instruments should be recorded as liabilities or equity is evaluated pursuant to ASC 815-40, Derivatives and Hedging—Contracts in Entity's Own Equity.

The Company issued Public Warrants and Private Placement Warrants in connection with its initial public offering (the "IPO") in November 2018. Additionally, the Company issued warrants to certain institutional investors in connection with the Company's private placement of Series A preferred stock on May 24, 2019 ("Series A Warrants," and together with the Public Warrants and Private Placement Warrants, the "warrants"). As of December 31, 2021, all our outstanding warrants are recognized as liabilities. Accordingly, we recognize the warrant instruments as liabilities at fair value upon issuance and adjust the instruments to fair value at the end of each reporting period. Any change in fair value is recognized in our consolidated statements of operations. The Public Warrants are valued using their quoted market price since they are publicly traded and thus had an observable market price. The Private Placement Warrants are valued using a Monte Carlo simulation model. The Series A Warrants are valued using the Black-Scholes option pricing model.

### Convertible Notes and Convertible Preferred Stock

When the Company issues convertible notes or convertible preferred stock, it first evaluates the balance sheet classification of the convertible instrument in its entirety to determine whether the instrument should be

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

classified as a liability under ASC 480 and second whether the conversion feature should be accounted for separately from the host instrument. A conversion feature of a convertible note instrument or certain convertible preferred stock would be separated from the convertible instrument and classified as a derivative liability if the conversion feature, were it a standalone instrument, meets the definition of an "embedded derivative" in ASC 815-15. Generally, characteristics that require derivative treatment include, among others, when the conversion feature is not indexed to the Company's equity, as defined in ASC 815-40, or when it must be settled either in cash or by issuing stock that is readily convertible to cash. When a conversion feature meets the definition of an embedded derivative, it would be separated from the host instrument and classified as a derivative liability carried on the consolidated balance sheet at fair value, with any changes in its fair value recognized in the consolidated statements of operations.

If a conversion feature does not meet the conditions to be separated and accounted for as an embedded derivative liability, the Company then determines whether the conversion feature is "beneficial". A conversion feature would be considered beneficial if the conversion feature is "in the money" when the host instrument is issued or, under certain circumstances, at a later time. The beneficial conversion feature ("BCF") for convertible instruments is recognized and measured by allocating a portion of the proceeds equal to the intrinsic value of that feature to additional paid-in capital. The intrinsic value is generally calculated at the commitment date as the difference between the conversion price and the fair value of the common stock or other securities into which the security is convertible, multiplied by the number of shares into which the security is convertible. If certain other securities are issued with the convertible security, the proceeds are allocated among the different components. The portion of the proceeds allocated to the convertible security is divided by the contractual number of the conversion shares to determine the effective conversion price, which is used to measure the BCF. The effective conversion price is used to compute the intrinsic value. The value of the BCF is limited to the basis that is initially allocated to the convertible security.

If the convertible note contains a BCF, the amount of the proceeds allocated to the BCF reduces the balance of the convertible note, creating a discount which is amortized over the note's term to interest expense in the consolidated statements of operations.

When a convertible preferred stock contains a BCF, after allocating the proceeds to the BCF, the resulting discount is either amortized as deemed dividends over the period beginning when the convertible preferred stock is issued up to the earliest date the conversion feature may be exercised, or if the conversion feature is immediately exercisable, the discount is fully amortized at the date of issuance.

#### Fair Value of Preferred Stock

The fair value of Series A preferred stock at the date of issuance was estimated by calculating the present value of itsone-year redemption cost to the Company and then discounted for lack of marketability.

The fair value of Series B preferred stock is the stated value, which is equal to the proceeds received from issuance.

### Fair Value of Financial Instruments

Fair value is defined under ASC 820, Fair Value Measurement, as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. ASC 820 also establishes a three-level hierarchy, which requires an entity to maximize the use of observable

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

inputs and minimize the use of unobservable inputs when measuring fair value. The three levels are defined as follows:

- Level 1—inputs are quoted prices (unadjusted) in active markets for identical assets or liabilities.
- · Level 2—inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly.
- Level 3—inputs are unobservable for the asset or liability.

The following is a summary of the carrying amounts and estimated fair values of our financial instruments:

Senior Secured Term Loan. The fair value of the Senior Secured Term Loan is \$106,620 and \$198,000 as of December 31, 2021 and 2020, respectively, based on the market price quoted from external sources. If the Senior Secured Term Loan was measured at fair value in the financial statements, it would be classified as Level 2 in the fair value hierarchy.

Equipment financing. The carrying value of the equipment financing notes approximates fair value as its terms are consistent with and comparable to current market rates as of December 31, 2021 and 2020, respectively.

Warrants. As of December 31, 2021, the Company's outstanding warrants are accounted for as liabilities and measured at fair value. See "Note 9 – Warrant Liabilities" for fair value measurements associated with the Company's warrants.

Convertible Senior Notes. As of December 31, 2021, the fair value of the Convertible Senior Notes is \$73,538, based on an option pricing framework using a lattice model. If the Convertible Senior Notes were measured at fair value in the financial statements, they would be classified as Level 2 in the fair value hierarchy.

#### Revenue Recognition

The Company recognizes revenue based on the customer's ability to benefit from the services rendered in an amount that reflects the consideration expected to be received in exchange for those services, pursuant to ASC 606, Revenue from Contracts with Customers. The Company's revenues consist of providing pressure pumping services for either a pre-determined term or number of stages/wells to E&P companies operating in the onshore oil and natural gas basins of the United States. In the performance of these services, and at the request of our customers, we may also provide consumables such as chemicals and sand. Revenues are earned as services are rendered, which is generally on a per stage basis, per hour rate basis or daily rate basis. Customers are invoiced according to contract terms, which is generally upon the completion of a well or monthly with payment due typically 30 days from invoice date.

The Company's performance obligations are satisfied over time, typically measured by the number of stages completed or the number of pumping days a fleet is available to pump for a customer in a month. All revenue is recognized when a contract with a customer exists, collectability of amounts subject to invoice is probable, the performance obligations under the contract have been satisfied over time, and the amount to which the Company has the right to invoice has been determined. A portion of the Company's contracts contain variable consideration; however, this variable consideration is typically unknown at the time of contract inception, and is not known until the job is complete, at which time the variability is resolved.

The Company has elected to use the "as invoiced" practical expedient to recognize revenue based upon the amount it has a right to invoice upon the completion of each performance obligation per the terms of the contract.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

The practical expedient permits an entity to recognize revenue in the amount to which it has a right to invoice the customer if that amount corresponds directly with the value to the customer of the entity's performance completed to date. The Company believes that this is an accurate reflection of the value transferred to the customer as each incremental obligation is performed.

The Company has elected to expense sales commissions paid upon the successful signing of a new customer contract as incurred if the related contract will be fully satisfied within one year. For contracts that will not be fully satisfied within one year, these incremental costs of obtaining a contract with a customer will be recognized as a contract asset and amortized on a straight-line basis over the life of the contract.

Patent License Sales. On June 24, 2021, the Company issued a Convertible Senior Note (See "Note 10—Convertible Senior Notes") convertible into a patent license agreement. On June 29, 2021, the holder exercised its right to convert the Convertible Senior Note in full and the Company entered into a Patent License Agreement (the "License Agreement"), which provides the licensee a five-year option to purchase up to 20 licenses to build and operate electric fleets using the Company's patented Clean Fleet® technology (the "licenses"). Upon entry into the License Agreement, the Company sold three licenses to build and operate three electric fleets, each valued at \$7,500.

The sales of the right to use the Company's patented Clean Fleef technology is a single performance obligation. The Company recognizes the income associated with the patent license sales at the point in time when the Company satisfies its performance obligation by granting the purchaser the right to use the patented Clean Fleef technology and transfer of control has occurred. The patent license sales are recognized as other income in our consolidated statement of operations.

#### Major Customer and Concentration of Credit Risk

The concentration of our customers in the oil and natural gas industry may impact our overall exposure to credit risk, either positively or negatively, in that customers may be similarly affected by changes in economic and industry conditions. We perform ongoing credit evaluations of our customers and do not generally require collateral in support of our trade receivables.

The following table shows the percentage of revenues from our significant customers:

	Year Ended Dec	ember 31,
	2021	2020
Customer A	*	11.2%
Customer B	*	19.7%
Customer C	14.1%	18.4%
Customer E	15.5%	13.2%
Customer F	20.8%	18.2%

An asterisk indicates that revenue is less than ten percent.

(in thousands, except share and per share amounts, or where otherwise noted)

The following table shows the percentage of trade receivables from our significant customers:

	Year Ended Dec	ember 31,
	2021	2020
Customer B	*	32.2%
Customer C	20.4%	17.0%
Customer F	24.3%	12.7%
Customer G	*	12.5%
Customer H	*	13.5%
Customer J	29.7%	*
Customer K	25.0%	*

An asterisk indicates that trade receivable is less than ten percent.

#### Share-Based Compensation

Share-based compensation is measured on the grant date and fair value is recognized as expense over the requisite service period, which is generally the vesting period of the award. The Company recognizes forfeitures as they occur rather than estimating expected forfeitures.

The fair value of time-based restricted stock, deferred stock units, or other performance incentive awards is determined based on the number of shares or units granted and the closing price of Class A common stock on the date of grant. The fair value of stock options is determined by applying the Black-Sholes model on the grant-date market value of the underlying Class A common stock.

Deferred compensation expense associated with liability-based awards, such as certain performance incentive awards that could be settled either in cash or through issuance of a variable number of shares based on a fixed monetary amount at inception, is recognized at the fixed monetary amount at inception and is amortized on a straight-line basis over the requisite service period, which is generally the vesting period. However, the Company considers any delayed settlement as a post-vesting restriction which impacts the determination of the grant-date fair value of the award. The Company estimates fair value by using a risk-adjusted discount rate, which reflects the weighted average cost of capital of similarly traded public companies.

#### Income Taxes

The Company, under ASC 740, Accounting for Income Taxes, uses the asset and liability method of accounting for income taxes, under which deferred tax assets and liabilities are recognized for the future tax consequences of (i) temporary differences between the financial statement carrying amounts and the tax bases of existing assets and liabilities and (ii) operating loss and tax credit carryforwards. Deferred income tax assets and liabilities are based on enacted tax rates applicable to the future period when those temporary differences are expected to be recovered or settled. The effect of a change in tax rates on deferred tax assets and liabilities is recognized in income in the period the rate change is enacted. A valuation allowance is provided for deferred tax assets when it is more likely than not the deferred tax assets will not be realized. Our deferred tax calculation and valuation allowance requires us to make certain estimates about future operations. Changes in state or federal tax laws, as well as changes in our financial condition or the carrying value of existing assets and liabilities, could affect those estimates. The effect of a change in tax rates is recognized as income or expense in the period that the rate is enacted.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

ASC 740 prescribes a recognition threshold and a measurement attribute for the financial statement recognition and measurement of tax positions taken or expected to be taken in a tax return. For those benefits to be recognized, a tax position must be more-likely-than-not to be sustained upon examination by taxing authorities. The Company recognizes accrued interest and penalties related to unrecognized tax benefits as income tax expense. No amounts were accrued for the payment of interest and penalties at December 31, 2021. The Company is currently not aware of any issues under review that could result in significant payments, accruals, or material deviation from its position. The Company is subject to income tax examinations by major taxing authorities since inception.

#### NOTE 3—ACCOUNTING STANDARDS

#### Recently Adopted Accounting Pronouncements

The Company is an "emerging growth company," as defined in Section 2(a) of the Securities Act of 1933, as amended, (the "Securities Act"), as modified by the Jumpstart our Business Startups Act of 2012 (the "JOBS Act"), and it may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in its periodic reports and proxy statements, and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and shareholder approval of any golden parachute payments not previously approved.

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Exchange Act) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that a company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies but any such election to opt out is irrevocable. The Company has elected not to opt out of such extended transition period, which means that when a standard is issued or revised and it has different application dates for public or private companies, the Company, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make comparison of the Company's financial statements with another public company which is neither an emerging growth company nor an emerging growth company which has opted out of using the extended transition period difficult or impossible because of the potential differences in accounting standards used.

In August 2018, the Financial Accounting Standards Board ("FASB") issued ASU2018-15, Intangibles—Goodwill and Other—Internal-Use Software (Subtopic 350-40): Customer's Accounting for Implementation Costs Incurred in a Cloud Computing Arrangement That is a Service Contract, requiring a customer in a cloud computing arrangement that is a service contract to follow the guidance in ASC 350-40 in determining the requirements for capitalizing implementation costs incurred to develop or obtain internal-use-software. The Company adopted ASU 2018-15 on January 1, 2021, and the adoption of this standard did not have a material impact on the Company's consolidated financial statements.

### Recent Accounting Pronouncements Not Yet Adopted

In February 2016, the FASB issued ASU 2016-02, *Leases (Topic 842)* and subsequent amendments to the initial guidance: ASU 2017-13, ASU 2018-10, ASU 2018-11, ASU 2018-20, ASU 2019-01 and ASU 2020-02 (collectively, "ASC 842"). ASC 842 requires companies to generally recognize on the balance sheet operating and financing lease liabilities and corresponding right-of-use assets.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

The Company adopted ASC 842 on January 1, 2022, using the modified retrospective with applied transition method and will recognize a cumulative impact to retained earnings in that period. The Company elected to apply certain practical expedients, whereby it will not reassess (i) whether any expired or existing contracts are or contain leases, (ii) the lease classification for any expired or existing leases and (iii) initial direct costs for any existing leases. Prior period financial statements would be stated under the old guidance ASC 840 with no change to prior periods or disclosures associated with prior period.

Upon adoption of the new leasing standard, the Company will recognize additional right-of-use assets and related lease liabilities of approximately \$1.6 million and \$2.1 million, respectively, on its consolidated balance sheet as of January 1, 2022, primarily for its operating leases that existed upon the effective date. The impact of adoption of the new leasing standard has no impact to the consolidated statements of operations.

In June 2016, the FASB issued ASU2016-13, Financial Instruments—Credit Losses (Topic 326), which changes the impairment model for most financial assets and certain other instruments. Specifically, this new guidance requires using a forward-looking, expected loss model for trade and other receivables, held-to-maturity debt securities, loans and other instruments. This will replace the currently used model and may result in an earlier recognition of allowance for losses. In addition, in November 2019, the FASB issued ASU 2019-11, Codification Improvements to Topic 326, Financial Instruments — Credit Losses, which clarifies guidance around how to report expected recoveries. The new guidance will be effective for emerging growth companies for fiscal years beginning after December 15, 2022, including interim periods within those fiscal years. The Company is currently evaluating the impact of adopting the new guidance on the consolidated financial statements.

In December 2019, the FASB issued ASU2019-12, *Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes* which removes specific exceptions to the general principles in Topic 740 in GAAP. The new guidance also improves the issuer's application of income tax-related guidance and simplifies GAAP for franchise taxes that are partially based on income, transactions with a government that result in a step up in the tax basis of goodwill, separate financial statements of legal entities that are not subject to tax, and enacted changes in tax laws in interim periods. The Company adopted ASU 2019-12 on January 1, 2022. The adoption is not expected to have a material impact on the Company's consolidated financial statements.

In August 2020, the FASB issued ASU 2020-06, *Debt—Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging—Contracts in Entity's Own Equity (Subtopic 815-40): Accounting for Convertible Instruments and Contracts in an Entity's Own Equity which simplifies accounting for convertible instruments by removing major separation models required under current GAAP. It also amends the accounting for certain contracts in an entity's own equity that are currently accounted for as derivatives because of specific settlement provisions. In addition, the new guidance modifies how particular convertible instruments and certain contracts that may be settled in cash or shares impact the diluted EPS computation. The new guidance will be effective for small reporting companies for fiscal years beginning after December 15, 2023, including interim periods within those fiscal years. Early adoption is permitted. The Company adopted ASU 2020-06 on January 1, 2022, using the modified retrospective method of transition. The adoption is not expected to have a material impact on the Company's consolidated financial statements.* 

(in thousands, except share and per share amounts, or where otherwise noted)

#### NOTE 4—PREPAIDS AND OTHER CURRENT ASSETS

Prepaids and other current assets consisted of the following:

	Decem	iber 31,
	2021	2020
Prepaid insurance	\$ 5,207	\$ 3,162
Recoverable costs from insurance	11,109	4,635
Income tax receivable	757	1,567
Other current assets	1,675	1,343
Total prepaid expenses and other current assets	<u>\$ 18,748</u>	<u>\$ 10,707</u>

During the year ended December 31, 2021, the Company prepaid \$13,241 in insurance premiums related to renewals of various insurance policies.

The Company has insurance coverage in place covering, among other things, property damage up to certain specified amounts. Recoverable costs from insurance as of December 31, 2021, was \$11,109, which represents net book value of equipment damaged that we expect to recover from insurance, which was collected in full during the first quarter of 2022. The \$4,635 of recoverable costs from insurance, recorded as of December 31, 2020, was collected in full during the first quarter of 2021. During the years ended December 31, 2021 and 2020, the Company received \$9,053 and \$4,854, respectively, as insurance reimbursement.

#### NOTE 5-PROPERTY AND EQUIPMENT, NET

Property and equipment consisted of the following:

	Estimated	December 31,	
	Useful Life (in years)	2021	2020
Pressure pumping equipment <sup>(1)</sup>	1.5 to 25	\$ 186,826	\$ 263,869
Light duty vehicles <sup>(2)</sup>	5	5,524	2,483
Furniture and fixtures	5	67	67
IT equipment	3	1,033	1,676
Auxiliary equipment	2 to 20	12,218	11,058
Leasehold improvements	Term of lease	276	287
		205,944	279,440
Less: Accumulated depreciation and amortization		(43,280)	(44,108)
Property and equipment, net		\$ 162,664	\$ 235,332

- (1) As of December 31, 2021, the Company had capitalized \$540 related to capital leases and the accumulated depreciation was \$270.
- (2) As of December 31, 2021 and 2020, the Company had capitalized \$3,857 and \$260, respectively, related to capital leases and the accumulated depreciation was \$489 and \$31, respectively.

Depreciation and amortization expense related to property and equipment during the years ended December 31, 2021 and 2020 was \$34,479 and \$79,263, respectively.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

During the year ended December 31, 2020, as a result of the impairment test on long-lived assets described in "Note 6—Goodwill and Intangible Assets," the Company recorded an impairment charge of \$140,273 to reduce the carrying value of property and equipment to its fair value on the date of impairment. No impairment of long-lived assets was recorded for the year ended December 31, 2021.

#### Assets Sales

In May 2021, the Company announced its commitment to becoming anall-electric pressure pumping services provider and since then has sold most of its legacy, diesel-powered pressure pumping equipment. As of December 31, 2021, the Company has classified \$2,043 in net book value of diesel pressure pumping equipment, which was subsequently sold in January 2022, as assets held for sale on the consolidated balance sheet.

During the year ended December 31, 2021, the Company received \$105,963 in proceeds from the sale of equipment, of which \$44,450 was for assets classified as held for sale and \$35,000 was a result of sale-leaseback transaction (described below). In January 2022, the Company sold its remaining assets classified as held for sale for \$2,110 in proceeds. The Company used the proceeds received from the asset sales to pay down the principal of its Senior Secured Term Loan.

The Company recognized a gain of \$21,896 and a loss of \$7,112 from disposal of assets for the years ended December 31, 2021 and 2020, respectively.

In October 2021, the Company entered into an agreement to sell certain power generation equipment for proceeds of \$35,000. Concurrent with the sale of the assets, the Company entered into a twelve month lease agreement whereby the Company agreed to lease back the assets. In accordance with ASC 840, the Company accounted for the sale as a sale-leaseback transaction and accounted for the lease as a short-term operating lease. The Company deferred \$7,410 of the gain from disposal of assets to accrued expenses and other current labilities to amortize over the minimum terms of the lease.

### NOTE 6—GOODWILL AND INTANGIBLE ASSETS

#### Goodwill

Goodwill represents the difference between the purchase price and the estimated fair value of identifiable assets acquired and liabilities assumed. As of December 31, 2021 and 2020, the Company had goodwill of \$4,971.

(in thousands, except share and per share amounts, or where otherwise noted)

#### Intangible Assets

A summary of intangible assets consisted of the following:

	Estimated Useful Life (in years)	Gross Carrying Value	Accumulated Amortization	Net Book Value
As of December 31, 2021				
Trademarks	10	\$ 1,415	\$ 362	\$ 1,053
Patents	20	12,775	1,328	11,447
		\$14,190	\$ 1,690	\$12,500
As of December 31, 2020				
Trademarks	10	\$ 1,415	\$ 156	\$ 1,259
Patents	20	12,775	568	12,207
		\$14,190	\$ 724	\$13,466

The intangible assets are amortized over the period the Company expects to receive the related economic benefit. Amortization expense related to amortizable intangible assets was \$966 and \$1,090 for the years ended December 31, 2021 and 2020, respectively, which was included as part of depreciation and amortization in the consolidated statements of operations.

During the first quarter of 2020, the Company identified a triggering event and performed a quantitative impairment test on long-lived assets. The expected present value method, a form of the income approach, was utilized to determine the fair value of long-lived assets. This method is based on expected cash flows using a risk-adjusted discount rate, which reflects the weighted average cost of capital of similarly traded public companies. As a result of the impairment test performed, during the year ended December 31, 2020, the Company recorded an impairment charge of \$7,270 to reduce the carrying value of intangible assets to its fair value on the date of impairment. No impairment of intangible assets was recorded for the year ended December 31, 2021.

As of December 31, 2021, the estimated amortization expense for future periods is as follows:

Fiscal Year	Estimated Amortization Expense
2022	\$ 966
2023	966
2024	966
2025	966
2026	966
Thereafter	7,670
Total	\$ 12,500

(in thousands, except share and per share amounts, or where otherwise noted)

#### NOTE 7—ACCRUED EXPENSES AND OTHER CURRENT LIABILITIES

Accrued expenses and other current liabilities consisted of the following:

	Decem	December 31,	
	2021	2020	
Accrued payroll and benefits	\$ 5,188	\$ 7,208	
Accrued taxes	5,041	5,380	
Accrued interest	287	317	
Deferred gain on sale-leaseback	5,557	_	
Other current liabilities	769	1,876	
Accrued expenses and other current liabilities	<u>\$ 16,842</u>	\$ 14,781	

#### NOTE 8—NOTES PAYABLE

The Company enters into various premium finance agreements with a credit finance institution to pay the premiums on insurance policies for its directors' and officers' liability, general liability, workers' compensation, umbrella, auto and pollution coverage needs. During the years ended December 31, 2021 and 2020, the aggregate amount of the premiums financed was \$10,699 and \$1,121, respectively, payable in equal monthly installments at a weighted average interest rate of 5.2% and 5.3%, respectively. These premium finance agreements are due within one year and are recorded as notes payable under current liabilities in the consolidated balance sheets. As of December 31, 2021 and 2020, the Company had a remaining balance of \$2,320 and \$998, respectively, related to notes payable.

#### NOTE 9—WARRANT LIABILITIES

#### Public Warrants and Private Placement Warrants

Pursuant to the Company's IPO, 32,500,000 warrants (the "Public Warrants") were issued and 15,500,000 warrants (the "Private Placement Warrants") were sold simultaneously to Matlin & Partners Acquisition Sponsor, LLC (the "Sponsor") and Cantor Fitzgerald (the "Underwriter"). Each Public Warrant and Private Placement Warrant entitles its holder to purchase one-seventh of a share of Class A common stock at an exercise price of \$5.75 per warrant (\$40.25 per full share equivalent), to be exercised only for a whole number of shares of our Class A common stock. The warrants became exercisable 30 days after the completion of the transaction and expire on November 9, 2023 or earlier upon redemption or liquidation.

The Company may redeem the outstanding Public Warrants at a price of \$0.01 per warrant upon a minimum of 30 days' prior written notice of redemption, if the last sale price of the Company's common stock equals or exceeds \$84.00 per share for any 20 trading days within a 30-trading day period ending on the third business day before the Company sends the notice of redemption to the warrant holders. The Private Placement Warrants, however, are nonredeemable so long as they are held by the Sponsor, the Underwriter or their permitted transferees.

During the year ended December 31, 2021, no Public or Private Placement Warrants were redeemed, exercised or forfeited. During the year ended December 31, 2020, certain warrant holders elected to forfeit 6,327,218 Private Placement Warrants for no consideration. As of December 31, 2021, 9,994,635 Public Warrants and 9,172,782 Private Placement Warrants were outstanding, and exercisable for an aggregate of 2,738,203 shares of Class A common stock.

(in thousands, except share and per share amounts, or where otherwise noted)

#### Series A Warrants

On May 24, 2019 ("Closing Date"), the Company issued 2,933,333 initial warrants to certain institutional investors in connection with the Company's private placement of Series A preferred stock (the "Series A Warrants"). The Company will issue additional warrants to the purchasers in quarterly installments beginning nine months after the Closing Date through March 31, 2022.

The Series A Warrants entitle their holders to purchasetwo-sevenths of a share of Class A common stock at an exercise price of \$7.66 per warrant (\$26.81 per full share equivalent), to be exercised only for a whole number of shares of Class A common stock. The Series A Warrants expire on November 25, 2025.

During the years ended December 31, 2021 and 2020, the Company issued 1,777,776 and 1,911,108 additional Series A Warrants to the purchasers of Series A preferred stock, respectively, in accordance with the Series A preferred stock purchase agreement. During the year ended December 31, 2021, a Series A Warrant holder elected to forfeit 399,999 Series A Warrants for no consideration. See "Note 13—Mezzanine Equity" for additional discussion of the Series A preferred stock purchase agreement. As of December 31, 2021, 6,222,218 Series A Warrants were outstanding pursuant to the Series A preferred stock purchase agreement, and exercisable for 1,777,777 shares of Class A common stock.

#### Fair Value Measurement

As of December 31, 2021, the Company's outstanding warrants are accounted for as liabilities measured at fair value upon issuance, with subsequent changes in fair value reported in the Company's consolidated statements of operations each reporting period.

The following tables present the Company's fair value hierarchy for liabilities measured at fair value on a recurring basis:

	Quoted prices in active markets (Level 1)	Other observable inputs (Level 2)	Unobservable inputs (Level 3)	Total
As of December 31, 2021		·		
Public Warrants	\$ 752	\$ —	\$ —	\$ 752
Private Placement Warrants	_	871	_	871
Series A Warrants		1,934		1,934
	<u>\$ 752</u>	\$ 2,805	<u>\$</u>	\$3,557
As of December 31, 2020	<del></del>	<del>====</del>	<del></del>	<del>===</del>
Public Warrants	\$ 254	\$ —	\$ —	\$ 254
Private Placement Warrants	_	248	_	248
Series A Warrants	<u></u>	1,117		1,117
	\$ 254	\$ 1,365	<u> </u>	\$1,619

Public Warrants. The fair value of the Public Warrants are classified as Level 1 in the fair value hierarchy and valued using quoted market prices, as they are traded in active markets.

Private Placement Warrants. The fair value of the Private Placement Warrants are classified as a Level 2 in the fair value hierarchy and determined using a Monte Carlo simulation model. Inherent in a Monte Carlo simulation

(in thousands, except share and per share amounts, or where otherwise noted)

are assumptions related to expected stock-price volatility, expected life, risk-free interest rate and dividend yield. The Company estimates the volatility of its common stock warrants based on implied volatility from the Company's publicly traded warrants and from historical volatility of select peer company common stock that matches the expected remaining life of the warrants. The risk-free interest rate is based on the U.S. Treasury zero-coupon yield curve on the grant date for a maturity similar to the expected remaining life of the warrants. The expected life of the warrants is assumed to be equivalent to their remaining contractual term. The dividend rate is based on the historical rate, which the Company anticipates remaining at zero.

Series A Warrants. The fair value of the Series A Warrants are classified as a Level 2 in the fair value hierarchy and is determined using the Black-Scholes valuation method.

The following assumptions were used to calculate the fair value for the private placement and Series A Warrants:

	Private Placement Warrants	Series A Warrants
As of December 31, 2021		
Expected remaining life	1.9 years	3.9 years
Volatility rate	227.5%	227.5%
Risk-free interest rate	0.7%	1.1%
Expected dividend rate	0%	0%
As of December 31, 2020		
Expected remaining life	2.86 years	4.9 years
Volatility rate	115.8%	115.8%
Risk-free interest rate	0.2%	0.4%
Expected dividend rate	0%	0%

The following table summarizes the activities, including changes in fair values of the Company's warrant liabilities for the periods indicated:

	Decem	December 31,	
	2021	2020	
Balance at beginning of period	\$1,619	\$ 8,147	
Reclassification of warrant liability upon forfeiture	(232)	(203)	
Change in fair value of warrant liabilities	2,152	(6,342)	
Amortization of warrant issuance costs	18	17	
Balance at end of period	<u>\$3,557</u>	\$ 1,619	
Balance at end of period	\$3,337	\$ 1,619	

### NOTE 10—CONVERTIBLE SENIOR NOTES

On June 24, 2021, the Company entered into a Note Purchase Agreement (as amended, the "Note Purchase Agreement"). As of December 31, 2021, pursuant to the Note Purchase Agreement, the Company issued \$136,500 in aggregate principal amount of 16.0% Convertible Senior Secured (Third Lien) PIK Notes (the "Convertible Senior Notes"), in a private placement to institutional investors (the "Private Placement"), comprised of Cash Notes, Exchange Notes (collectively with the Cash Notes, the "Equity Linked Notes") and a License Linked Note, as described below, which mature on June 5, 2026. The Convertible Senior Notes are secured by a third priority security interest in the collateral that secures the Company's obligations under the Senior Secured Term Loan.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

Equity Linked Notes. In June 2021 and July 2021, in connection with the Private Placement, the Company issued and sold \$75,000 in principal amount of Convertible Senior Notes that are convertible at any time at the holder's option, into shares of the Company's Class A common stock for cash (the "Cash Notes"). The conversion prices of the Cash Notes range from \$3.43 to \$4.38, subject to adjustment.

In June 2021, in connection with the Private Placement, the Company issued and sold \$39,000 in principal amount of Convertible Senior Notes that are convertible at any time at the holder's option, into shares of the Company's Class A common stock in exchange for 30,390 shares of the Company's Series A preferred stock (the "Exchange Notes"). The Exchange Notes are convertible at a conversion price of \$7.00 subject to adjustment.

License Linked Note. On June 24, 2021, in connection with the Private Placement, the Company issued and sold a Convertible Senior Note in the principal amount of \$22,500 that was convertible into a patent license agreement (the "License Linked Note"). On June 29, 2021, the holder exercised its right to convert the License Linked Note in full and the Company entered into the License Agreement, which provides the licensee a five-year option to purchase up to 20 licenses to build and operate electric fleets using the Company's patented Clean Fleet® technology (the "licenses"). Upon entry into the License Agreement, the holder purchased three licenses to build and operate three electric fleets, each valued at \$7,500. The Company recognized the \$22,500 as other income from patent license sales in its consolidated statement of operations. The debt issuance costs associated with the License Linked Note were fully amortized.

The carrying value of the Convertible Senior Notes is as follows:

	December 31, 2021
Principal	\$ 114,000
PIK interest	9,686
Unamortized debt premium	1,841
Unamortized debt discount and issuance costs	(19,758)
Net Convertible Senior Notes	\$ 105,769

During the year ended December 31, 2021, the Company received \$97,500 in cash proceeds from the issuance of the Convertible Senior Notes. The Company used a portion of the proceeds from the issuance of the Convertible Senior Notes to pay the cash settlement amount in accordance with the Settlement Agreement (as described in "Note 12—Commitments and Contingencies") and the remainder for general corporate purposes, including growth capital.

The Convertible Senior Notes bear interest at a rate of 16.0% per annum. Accrued and unpaid interest is calculated on the last day of each quarter, commencing September 30, 2021, and will be paid-in-kind ("PIK") on such date by increasing the principal amount of the outstanding Convertible Senior Notes. The Company has accrued PIK interest of \$9,686 related to the Convertible Senior Notes for the year ended December 31, 2021.

Each Equity Linked Note, subject to earlier conversion, is due and payable on June 5, 2026 in shares of Class A common stock equal to the entire outstanding and unpaid principal balance, plus any PIK interest, subject to certain limitations on the number of shares of Class A common stock that may be issued and which would require the Company to settle the conversion in payment partially in cash. The number of shares of Class A common stock will be based on the 20-day volume weighted average trading price of the Class A common stock immediately preceding the maturity date. The Equity Linked Notes are convertible at any time at the option of

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

the holder into a number of shares of Class A common stock equal to the principal amount of such notes then outstanding plus PIK interest through the conversion date divided by the then applicable conversion price as described above. If the Company experiences an event of default (as defined in the Note Purchase Agreement), which is continuing on the maturity date, then payment of principal and PIK interest shall be made in cash on any outstanding Equity Linked Notes.

Additionally, following the first anniversary of the Note Purchase Agreement, and at any time in which there are no issued and outstanding shares of Series A preferred stock or Series B preferred stock, if the 20-day volume weighted average trading price of the Class A common stock is greater than \$7.00 for 10 trading days during any 20 consecutive trading day period, the Company may deliver a notice to the holder of an Equity Linked Note to convert such Equity Linked Notes at the conversion prices set forth above.

In accordance with ASC 480, the Company evaluated the Equity Linked Notes and determined they should be classified as liabilities due to the unconditional obligation to settle the notes in a variable number of shares of the Company's Class A common stock based on a fixed monetary amount known at inception. Certain of the Equity Linked Notes issued were initially measured at fair value as they were considered new instruments issued concurrently to extinguish the Series A preferred stock. See "Note 13—Mezzanine Equity" for the discussion of Series A preferred stock exchange.

The initial measurement at fair value of those certain Equity Linked Notes resulted in the Company recording a premium of \$1,900 and a total discount of \$16,120. The Company amortizes such premium and discount as an adjustment to interest expense using the effective interest method over the term of the Equity Linked Notes.

During the year ended December 31, 2021, we incurred transaction costs related to the issuance of the Convertible Senior Notes of \$4,413 which were recorded as debt issuance costs and are presented as a direct deduction from the carrying amount of the Convertible Senior Notes on our consolidated balance sheet. The debt issuance costs are being amortized under the effective interest method over the term of the Convertible Senior Notes. Amortization expense related to the Convertible Senior Notes was \$745 for the year ended December 31, 2021 and is presented in interest expense in the consolidated statements of operations.

#### NOTE 11—DEBT

Long-term debt consisted of the following:

	Decem	December 31,	
	2021	2020	
Senior Secured Term Loan	\$120,745	\$246,250	
ABL Credit Facility	14,170	23,710	
PPP Loan	_	10,000	
USDA Loan	25,000	21,996	
Equipment financing	8,540	12,866	
Capital leases	3,204	229	
Total debt principal balance	171,659	315,051	
Senior Secured Term Loan future interest payable	24,384	10,925	
Unamortized debt discount and issuance costs	(11,792)	(28,501)	
Current maturities	(9,504)	_(13,573)	
Net long-term debt	\$174,747	\$283,902	

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

#### Senior Secured Term Loan

On May 7, 2019, the Company, USWS LLC, as the borrower, and all the other subsidiaries of the Company entered into a Senior Secured Term Loan Credit Agreement (as amended, the "Senior Secured Term Loan Agreement") with CLMG Corp., as administrative and collateral agent, and the lenders party thereto. Upon entering the Senior Secured Term Loan Agreement, the Company borrowed \$250,000 in Term A Loans and Term B Loans (collectively the "Senior Secured Term Loan"), which matures on December 5, 2025.

The Senior Secured Term Loan is not subject to financial covenants but is subject to certainnon-financial covenants, including but not limited to, reporting, insurance, notice and collateral maintenance covenants as well as limitations on the incurrence of indebtedness, permitted investments, liens on assets, dispositions of assets, paying dividends, transactions with affiliates, mergers and consolidations.

The Senior Secured Term Loan requires mandatory prepayments upon certain dispositions of property or issuance of other indebtedness, as defined, and quarterly a percentage of excess cash flow, if any, equal to 100%. Certain mandatory prepayments (excluding excess cash flows sweep) and optional prepayments are subject to a yield maintenance fee for the first two years and prepayment premium of 2.0% in year three and 1.0% in year four. Upon the final payment and termination of the Senior Secured Term Loan, we are subject to an exit fee equal to 2.0% of the principal amount of loans then outstanding and the aggregate optional prepayment of principal amounts repaid during the 120 days that occurred prior to such final payment.

During the years ended December 31, 2021 and 2020, the Company, USWS LLC, as the borrower, and all the other subsidiaries of the Company entered into multiple term loan amendments to the Senior Secured Term Loan Agreement with CLMG Corp., as administrative and collateral agent, and the lenders party thereto, to make certain modifications to the Senior Secured Term Loan.

The Senior Secured Term Loan bears interest at a variable rate per annum equal to the applicable LIBOR rate, subject to a 2.0% floor, plus 8.25%. Pursuant to the second term loan amendment, the interest rate on amounts outstanding under the Senior Secured Term Loan was reduced to 0.0% and scheduled principal amortization payments were suspended for the period beginning April 1, 2020 and ending March 31, 2022.

In accordance with the fifth term loan amendment, the deferral period for interest on the Senior Secured Term Loan was shortened by three months, to January 1, 2022 pending the outstanding principal amount being reduced to certain levels by December 31, 2021. On December 31, 2021, the outstanding principal amount of the Senior Secured Term Loan was less than \$132,000 but greater than \$110,000 and therefore the interest rate is 0.0% per annum from January 1, 2022 through March 31, 2022. If on April 1, 2022, the outstanding principal amount of the Senior Secured Term Loan is equal to or less than \$103,000 then the interest rate shall be 1.0% per annum from April 1, 2022 through December 31, 2022. If, however, the outstanding principal amount of the Senior Secured Term Loan is greater than \$103,000 on April 1, 2022, than the interest rate will resume incurring at the applicable benchmark rate, subject to a 2.0% floor, plus the applicable margin of 8.25% per annum.

Pursuant to the fourth term loan amendment, the principal amortization schedule was modified such that commencing on December 31, 2020, the Company is required to make quarterly principal payments of \$1,250 until September 30, 2025.

On April 1, 2020, in exchange for entering into the second term loan amendment, the lenders under the Senior Secured Term Loan received an extension fee comprised of a \$20,000 cash payment, 1,050 shares of Series B preferred stock valued at \$1,050 based on the stated liquidation preference of \$1,000 per share, and 1,579,892

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

shares of Class A common stock valued at \$1,438 based on the closing price of the Class A common stock at the date of issuance. The Series B preferred stock issued to the lenders under the Senior Secured Term Loan had the same terms as the Series B preferred stock issued to certain institutional investors as described in "Note 13—Mezzanine Equity". The total fair value of cash and non-cash consideration transferred to the lenders under the Senior Secured Term Loan were accounted for as discount on debt issuance and amortized to interest expense using the effective interest method over the remaining term of the Senior Secured Term Loan.

During the year ended December 31, 2020, the Company accounted for the second and fourth term loan amendments as a troubled debt restructuring under ASC 470-60, *Troubled Debt Restructurings by Debtors*, due to the level of concession provided by the lenders under the Senior Secured Term Loan. Under this guidance, the future undiscounted cash flows of the Senior Secured Term Loan, as amended, exceeded the carrying value, and accordingly, no gain was recognized, and no adjustment was made to the carrying value of the debt. Interest expense on the amended Senior Secured Term Loan was computed using a new effective rate that equated the present value of the future cash payments specified by the new terms with the carrying value of the debt under the original terms.

The fourth and fifth term loan amendments agreed to make certain modifications to the Senior Secured Term Loan Agreement to, among other things, consent to entry into the USDA Loan (described within this section) and to permit the incurrence of debt and liens in connection with the Convertible Senior Notes.

The fifth term loan amendment, entered into on June 24, 2021, did not result in a significant modification or extinguishment resulting in no change in accounting for the Senior Secured Term Loan. In connection with the fifth term loan amendment, the Company paid \$3,000 to the lenders under the Senior Secured Term Loan, which was accounted for as a debt discount and is amortized to interest expense using the effective interest method over the remaining term of the Senior Secured Term Loan.

During the year ended December 31, 2021, the Company made principal payments of \$125,506, which included prepayments of \$118,006 driven primarily by asset sales. The early repayment of debt resulted in a write-off of \$14,215 of unamortized debt discount and issuance costs and prepayment fees of \$2,025, all of which were presented as loss on extinguishment of debt in the consolidated statements of operations.

As of December 31, 2021, the outstanding principal balance of the Senior Secured Term Loan was \$120,745, of which \$5,000 was due within one year from the balance sheet date.

On February 28, 2022, the Company entered into the sixth term loan amendment to the Senior Secured Term Loan Agreement. See "Note 20—Subsequent Events" for details regarding the modifications made to the Senior Secured Term Loan Agreement pursuant to the sixth term loan amendment.

### ABL Credit Facility

On May 7, 2019, the Company, USWS LLC, and all the other subsidiaries of the Company entered into a \$75,000 ABL Credit Agreement (as amended, the "ABL Credit Facility") with the lenders party thereto and Bank of America, N.A., as the administrative agent, swing line lender and letter of credit issuer. As of December 31, 2021, the aggregate revolving commitment under the ABL Credit Facility is \$50,000 and the facility matures on April 1, 2025.

During the years ended December 31, 2021 and 2020, the Company, USWS LLC, and all the other subsidiaries of the Company entered into a multiple ABL amendments to the ABL Credit Facility with the lenders party thereto and Bank of America, N.A., as the administrative agent, swing line lender and letter of credit issuer, to make certain modifications to the ABL Credit Facility.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

The ABL Credit Facility is subject to a borrowing base which is calculated based on a formula referencing the eligible accounts receivables of the borrower. As of December 31, 2021, borrowings under the ABL Credit Facility bear interest at LIBOR, plus an applicable LIBOR rate margin of 2.0% to 2.5% or base rate margin of 1.0% to 1.5% as defined in the ABL Credit Facility. In addition, as of December 31, 2021, borrowings which are advanced in respect of the FILO Amount (as defined in the first ABL amendment) accrue interest at a rate that is 1.5% higher than the rate applicable to other loans under the ABL Credit Facility, and may be repaid only after all other loans under the ABL Credit Facility have been repaid.

All borrowings under the ABL Credit Facility are subject to the satisfaction of customary conditions, including the absence of a default and the accuracy of representations and warranties and certifications regarding sales of certain inventory, and to a borrowing base (described above). In addition, the ABL Credit Facility includes a springing consolidated fixed charge coverage ratio of 1.00 to 1.00 but only when a financial covenant trigger period is in effect as defined in the ABL Credit Facility. Borrowings under the ABL Credit Facility are fully and unconditionally guaranteed jointly and severally by the Company and its subsidiaries, other than future unrestricted subsidiaries.

Pursuant to the first ABL amendment, entered into in April 2020, the aggregate revolving commitment under the ABL Credit Facility was reduced from \$75,000 to \$60,000, the maturity date was extended and the interest rate margin applicable to borrowings under the ABL Credit Facility was increased and a LIBOR floor was added. The borrowing base under the ABL Credit Facility was amended to include a FILO Amount which increased borrowing base availability.

Pursuant to the second ABL amendment, entered into in August 2020, the aggregate revolving commitment under the ABL Credit Facility was reduced from \$60,000 to \$50,000 and certain modifications were made to eligible accounts in the borrowing base and to the applicable thresholds in the cash dominion trigger period and financial covenant trigger period, among other things. The Company's option to request an increase in commitments under the accordion feature was also removed under the terms of the second ABL amendment.

During the year ended December 31, 2020, based on ASC470-50, *Modifications and Extinguishments*, the Company accounted for the first and second ABL amendments as a modification of debt.

The third and fourth ABL amendments agreed to make certain modifications to the ABL Credit Facility to, among other things, consent to entry into the USDA Loan (described within this section) and to permit the incurrence of debt and liens in connection with the Convertible Senior Notes.

On December 31, 2021, the borrowing base was \$25,266 and the outstanding revolver loan balance was \$14,170, classified as long-term debt in the consolidated balance sheets. As of December 31, 2021, we were in compliance with all of the covenants under our ABL Credit Facility.

### Paycheck Protection Program Loan

In July 2020, the Company received an unsecured loan (the "PPP Loan") in the principal amount of \$10,000 that bore interest at a rate of 1.0% per annum and matured in five years under the Paycheck Protection Program from a commercial bank. The Paycheck Protection Program was established under the Coronavirus Aid, Relief and Economic Security Act (as amended, the "CARES Act") and is administered by the U.S. Small Business Administration. Under the terms of the CARES Act, loan recipients can apply for and be granted forgiveness for all or a portion of the loan. In August 2021, the Company was notified that the principal amount of \$10,000 and accrued interest of \$99 with respect to the PPP Loan had been forgiven. The loan amount and accrued interest was recognized as a gain on extinguishment of debt in the consolidated statement of operations.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

#### USDA Loan

On November 12, 2020, the Company, USWS LLC, and USWS Holdings entered into a Business Loan Agreement (the "USDA Loan") with a commercial bank pursuant to the United States Department of Agriculture ("USDA"), Business & Industry Coronavirus Aid, Relief, and Economic Security Act Guaranteed Loan Program, in the aggregate principal amount of up to \$25,000 for the purpose of providing long-term financing for eligible working capital

The USDA loan bears interest of 5.75% per annum and is payable according to the following schedule:

- 36 monthly consecutive interest payments, beginning on December 12, 2020
- 83 monthly consecutive principal and interest payments beginning December 12, 2023
- One final principal and interest payment of the remaining due on November 12, 2030

During the year ended December 31, 2020, the Company recorded the related debt discount and debt issuance costs of \$506 and \$558, respectively, as a direct deduction to the face amount of the USDA Loan. The amortization of debt discount and debt issue costs is recorded to interest expense based on the effective interest rate method over the term of the USDA Loan.

The USDA Loan is secured by specific equipment collateral and is guaranteed by the USDA for up to 90% of the total proceeds. The USDA Loan is subject to certain financial covenants. The Company is required to maintain a Debt Service Coverage Ratio (as defined in the USDA Loan) of not less than 1.25:1, to be monitored annually, beginning in calendar year 2021. Additionally, the Company is required to maintain a Debt to Net Worth Ratio (as defined in the USDA Loan) of not more than 9:1, to be monitored annually based upon year-end financial statements beginning in calendar year 2022.

As of December 31, 2021, the outstanding principal balance of the USDA Loan was \$25,000, presented as long-term debt in the consolidated balance sheets. As of December 31, 2021, we were in compliance with all of the covenants under our USDA Loan.

#### **Equipment Financing**

In March 2020, the Company entered into an agreement to consolidate various individual equipment financing agreements, which represented substantially all our equipment financing notes, into four notes. The new notes are held by the same lender as the original equipment financing agreements. The amendments under the consolidated equipment financing notes pertain to maturity date, interest rate, and date of first installment payment. The Company evaluated the debt modification in accordance with ASC 470-50 and concluded that the debt modification did not result in a substantially different debt, and accordingly, no gain or loss was recorded. The equipment financing notes have an interest rate of 5.75% and mature in May 2024.

During the years ended December 31, 2021 and 2020, the Company made principal payments of \$4,327 and \$3,199, respectively, on the equipment financing notes. As of December 31, 2021, the outstanding balance of the equipment financing notes was \$8,540, of which \$3,412 was due within one year from the balance sheet date.

(in thousands, except share and per share amounts, or where otherwise noted)

#### Payments of Debt Obligations due by Period

As of December 31, 2021, the schedule of the repayment requirements of long-term debt is as follows:

Fiscal Year         of Long-term Debt           2022         \$ 9,504           2023         9,642           2024         10,402           2025         123,533           2026         3,367		Principal Amount
2023       9,642         2024       10,402         2025       123,533         2026       3,367	Fiscal Year	of Long-term Debt
2024     10,402       2025     123,533       2026     3,367	2022	\$ 9,504
2025     123,533       2026     3,367	2023	9,642
2026 3,367	2024	10,402
,	2025	123,533
TTI 0	2026	3,367
Thereafter15,211	Thereafter	15,211
Total <u>\$ 171,659</u>	Total	\$ 171,659

#### NOTE 12—COMMITMENTS AND CONTINGENCIES

#### Litigation

Liabilities for loss contingencies arising from claims, assessments, litigation, fines, and penalties, and other sources are recorded when it is probable that a liability has been incurred and the amount can be reasonably estimated. Legal costs incurred in connection with loss contingencies are expensed as incurred.

The Company was named a defendant in a case filed on January 14, 2019 in the Superior Court of the State of Delaware styledSmart Sand, Inc. v. U.S. Well Services LLC, C.A. 19C-01-144 PRW. On June 1, 2021, the court ruled against the Company in the case on the breach of contract claim and subsequently, on June 17, 2021, entered judgement in favor of Smart Sand, Inc. ("Smart Sand") in the amount of approximately \$51,000. On June 28, 2021, the Company entered into a Settlement Agreement and Release (the "Settlement Agreement") with Smart Sand, pursuant to which the Company and Smart Sand reached a settlement of all matters in dispute. Pursuant to the Settlement Agreement, the Company agreed to pay \$35,000 in cash and to provide Smart Sand certain rights of first refusal related to the supply of proppant for a period of two years (the "Settlement"). The parties to the Settlement Agreement also released each other from claims arising from or related to the Smart Sand litigation or the final judgment of the court. As of December 31, 2021, the Company paid \$35,000 to Smart Sand and the settlement expense was reflected as litigation settlement on the consolidated statement of operations.

In addition to the case noted above, the Company is involved in various pending or potential legal actions in the ordinary course of business. Management is unable to predict the ultimate outcome of these actions because of the inherent uncertainty of litigation. However, management believes that the most probable, ultimate resolution of the remaining matters will not have a material adverse effect on our consolidated financial position, results of operations or cash flows.

### Purchase Commitments

The Company entered into an Equipment Purchase and Sale Agreement to purchase equipment. The Company intends to fund the commitments due in the next twelve months under the Equipment Purchase and Sale Agreement through additional financing transactions and cash on hand.

(in thousands, except share and per share amounts, or where otherwise noted)

As of December 31, 2021, future minimum purchase commitments for equipment are as follows:

Fiscal Year	
2022	\$ 32,800
2023	16,400
Total	\$ 49,200

#### Sand Purchase Agreements

The Company entered into agreements for the supply of proppant for use in its pressure pumping operations. Under the terms of these agreements, the Company is subject to minimum purchase quantities on a monthly, quarterly, or annual basis at fixed prices or may pay penalties in the event of any shortfall. As of December 31, 2021, the Company has met its obligated minimum purchase commitments under the agreement for the supply of proppant.

#### Lease Agreements

The Company has various long-term operating leases for facilities with terms ranging from 36 to 76 months.

Rent expense was \$1,275 and \$2,359 for the years ended December 31, 2021 and 2020, respectively, of which \$1,001 and \$1,655, respectively, are recorded as cost of services and \$274, and \$704, respectively, are recorded as selling, general and administrative expenses in the consolidated statements of operations.

The Company has capital leases for light duty vehicles and equipment with terms ranging from 18 to 48 months. Assets under capital leases and related accumulated amortization is recorded under property and equipment in the consolidated balance sheets.

The following is a schedule of minimum future payments on non-cancellable operating leases and capital leases as of December 31, 2021:

	Operating	Capital
Fiscal Year	Leases	Leases
2022	\$ 1,107	\$1,241
2023	308	896
2024	258	891
2025	67	447
Total	<u>\$ 1,740</u>	\$3,475

The total capital leases payments include imputed interest.

The Company has also entered into short-term and long-term equipment lease agreements with certain power generation providers. The short-term equipment lease agreement has a term of twelve months and minimum future payments of \$17,550 due in 2022. As of December 31, 2021, the long-term equipment lease agreements had not commenced, and the Company was in negotiations to amend and extend the start dates based on current supply constraints. The long-term equipment lease agreements were subsequently scheduled to commence in the first half of 2022 and will have minimum future payments of approximately \$11,700 in 2022, \$15,600 in 2023 and \$3,900 in 2024. The Company expects to offset a portion of the commitments it will owe under these lease agreements through additional customer charges.

(in thousands, except share and per share amounts, or where otherwise noted)

#### Self-insurance

The Company established a self-insured plan for employees' healthcare benefits except for losses in excess of varying threshold amounts. The Company charges to expense all actual claims made during each reporting period, as well as an estimate of claims incurred, but not yet reported. The amount of estimated claims incurred, but not reported as of December 31, 2021 and 2020 was \$288 and \$189, respectively, and was reported as accrued expenses in the consolidated balance sheets. The Company believes that the liabilities recorded are appropriate based on the known facts and circumstances and does not expect further losses materially in excess of the amounts already accrued for existing claims.

#### NOTE 13—MEZZANINE EQUITY

### Series A Redeemable Convertible Preferred Stock

The following table summarizes the Company's Series A Redeemable Convertible Preferred Stock, par value \$0.0001 per share ("Series A preferred stock") activities for the year ended December 31, 2021:

	Shares	Amount
Balance at December 31, 2020	50,000	\$ 50,975
Exchange of Series A preferred stock for Convertible Senior Notes	(30,390)	(33,716)
Deemed and imputed dividends on Series A preferred stock	_	750
Accrued Series A preferred stock dividends		5,857
Balance at December 31, 2021	19,610	\$ 23,866

On May 23, 2019, the Company entered into a Purchase Agreement (the "Purchase Agreement") with certain institutional investors (collectively, the "Purchasers") to issue and sell in a private placement 55,000 shares of Series A preferred stock for an aggregate purchase price of \$1,000 per share, for total gross proceeds of \$55,000. At the initial closing on May 24, 2019 ("Closing Date"), the Purchasers purchased all the Series A preferred stock and 2,933,333 initial Series A Warrants exercisable for shares of Class A common stock. See "Note 9 – Warrant Liabilities" for additional discussion of the Series A Warrants issued pursuant to the Series A preferred stock purchase agreement.

Holders of shares of Series A preferred stock are entitled to receive cumulative dividends, compounding and accruing quarterly in arrears, from the Closing Date until the second anniversary of the Closing Date, at an annual rate of 12.0%, and thereafter, 16.0% of the stated value of \$1,000 per share, subject to increase in connection with the payment of dividends in kind. Dividends are payable, at the Company's option, in cash from legally available funds or in kind by increasing the stated value of the outstanding Series A preferred stock by the amount per share of the dividend on February 24, May 24, August 24, and November 24 of each year. During the years ended December 31, 2021 and 2020, the Company's Board of Directors did not declare a dividend on the Series A preferred stock resulting in the dividends for these periods being paid-in-kind in accordance with the Series A preferred stock's Certificate of Designations.

The Series A preferred stock is redeemable by the Company at any time for cash equal to the stated value per share on the date of redemption. If the Company notifies the holders that it has elected to redeem the Series A preferred stock, the holder may instead elect to convert such shares into Class A common stock. If the Series A preferred stock is converted in response to a redemption notice it will net settle for a combination of cash and Class A common stock.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

Following the first anniversary of the Closing Date, each holder of Series A preferred stock may convert all or any portion of its shares of Series A preferred stock into Class A common stock based on the then-applicable liquidation preference at a conversion price of \$23.35, subject to anti-dilution adjustments, at any time, but not more than once per quarter, so long as any conversion is for an underlying conversion value of Class A common stock of at least \$1,000.

The Company has the option to force a conversion of any then outstanding shares of Series A preferred stock following the third anniversary of the Closing Date, and contingent upon (i) the closing price of the Company's Class A common stock being greater than 130% of the Conversion Price for 20 trading days during any 30-day consecutive trading day period, (ii) the average daily trading volume of the Class A common stock exceeding 250,000 for 20 trading days and (iii) the Company having an effective registration statement on file with the SEC covering resales of the underlying Class A common stock to be received upon such conversion.

On the Closing Date, the Company estimated the fair value of the Series A Warrants at \$12,786 using the Black-Scholes option pricing model, which created a corresponding preferred stock discount in the same amount.

Due to the reduction of allocated proceeds to Series A preferred stock, the effective conversion price was approximately \$18.90 per share creating a BCF of \$22,104 which further reduced the carrying value of the Series A preferred stock. Since the holders' conversion option of the Series A preferred stock could only be exercisable after the first anniversary of the Closing Date, the discount resulting from the BCF was accreted over one year as deemed preferred dividends using the effective yield method, resulting in a corresponding increase in the carrying value of the Series A preferred stock over the same time period.

The Series A preferred stock had similar characteristics of an "Increasing Rate Security" as described by SEC Staff Accounting Bulletin Topic 5Q, Increasing Rate Preferred Stock. As a result, the discount on Series A preferred stock is considered an unstated dividend cost that is amortized over the period preceding commencement of the perpetual dividend using the effective interest method, by charging imputed dividend cost against retained earnings, or additional paid in capital in the absence of retained earnings and increasing the carrying amount of the Series A preferred stock by a corresponding amount. The discount is therefore being amortized over two years using the effective yield method. The amortization in each period is the amount which, together with the stated dividend in the period, results in a constant rate of effective cost with regard to the carrying amount of the Series A preferred stock. The deemed and imputed dividends on Series A preferred stock was fully amortized as of June 30, 2021.

The Series A preferred stock was recorded as Mezzanine Equity, net of issuance cost, on the consolidated balance sheets because it has redemption features upon certain triggering events that are outside the Company's control, such as change in control.

During the year ended December 31, 2020, one of the Purchasers converted 5,000 shares of Series A preferred stock and accrued dividends into 250,414 shares of Class A common stock pursuant to the certificate of designations authorizing and establishing the rights, preferences, and privileges of the Series A preferred stock. Accordingly, the Company recorded a reduction of \$4,852 in the carrying value of the Series A preferred stock.

In June 2021, the Company exchanged 30,390 shares of Series A preferred stock for the Exchange Notes (discussed in Note 10—Convertible Senior Notes). Accordingly, the Company recorded a reduction of \$33,716 in the carrying value of the Series A preferred stock during the year ended December 31, 2021. Concurrent with the issuance of the Exchange Notes, the Company also received from such holders of the Series A preferred stock total cash proceeds of \$39,000 in consideration for additional Cash Notes. In connection with the extinguishment

(in thousands, except share and per share amounts, or where otherwise noted)

of the Series A preferred stock, the Company initially recorded the Convertible Senior Notes issued to such holders at a total fair value of \$63,780. The difference of \$8,936 between the fair value of the Convertible Senior Notes issued and the carrying amount of \$72,716 of consideration received was recorded in additional paid in capital as a return from the Series A preferred holders for the year ended December 31, 2021.

As of December 31, 2021, 19,610 shares of Series A preferred stock were outstanding and convertible into 1,168,297 shares of Class A common stock, and dividends accrued and outstanding with respect to the Series A preferred stock were \$7,664 and reflected in the carrying value of Series A preferred stock

#### Series B Redeemable Convertible Preferred Stock

The following table summarizes the Company's Series B Redeemable Convertible Preferred Stock, par value \$0.0001 per share ("Series B preferred stock") activities for the year ended December 31, 2021:

	Shares	Amount
Balance at December 31, 2020	22,050	\$ 22,686
Conversion of Series B preferred stock to Class A common stock	(22,050)	(27,277)
Accrued Series B preferred stock dividends		4,591
Balance at December 31, 2021		<u>\$</u>

On March 31, 2020, the Company entered into a purchase agreement with certain institutional investors (collectively, the "Series B Purchasers"), pursuant to which the Company agreed to issue and sell in a private placement 21,000 shares of Series B preferred stock, for an aggregate purchase price of \$21,000. On April 1, 2020 (the "Series B Closing Date"), the Series B Purchasers purchased the Series B preferred stock.

Holders of the Series B preferred stock received distributions of 12.0% per annum on the then-applicable liquidation preference until May 24, 2021 and 16.0% per annum on the liquidation preference thereafter. Distributions were not required to be paid in cash and, if not paid in cash, were automatically accrued and added to the liquidation preference.

The Company had the option, but no obligation, to redeem the Series B preferred stock for cash. If the Company notified the holders that it had elected to redeem the Series B preferred stock, a holder may instead elect to convert its shares of Series B preferred stock at the specified conversion price, which was initially \$0.308 per share. The Series B preferred stock converted in response to a redemption notice would net settle for a combination of cash and Class A common stock.

Each holder of Series B preferred stock may convert all or any portion of its Series B preferred stock into Class A common stock based on the then-applicable liquidation preference, subject to anti-dilution adjustments, at any time, but not more than once per quarter, so long as any conversion is for at least \$1,000 based on the liquidation preference on the date of the conversion notice.

The Series B preferred stock was recorded as Mezzanine Equity, net of issuance cost, on the consolidated balance sheets because it had redemption features upon certain triggering events that were outside the Company's control, such as change in control.

During the first and second quarter of 2021, an aggregate total of 1,012 shares of Series B preferred stock and related accrued dividends were converted into 1,049,508 shares of Class A common stock pursuant to the

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

certificate of designations authorizing and establishing the rights, preferences, and privileges of the Series B preferred stock.

On September 14, 2021, the Company amended the certificate of designations of the Series B preferred stock to provide that the Company could, subject to certain conditions, convert all, but not less than all, of the outstanding shares of the Series B preferred stock into shares of the Company's Class A common stock. Upon conversion, each holder of the Series B preferred stock would receive the number of shares of Class A common stock equal to the aggregate amount of Series B preferred stock dividends that would have accrued if such shares were converted as of April 1, 2022, divided by the conversion price set forth in the certificate of designations.

On September 17, 2021, the Company converted the remaining 21,038 shares of the Series B preferred stock and related accrued dividends for 25,565,707 shares of Class A common stock, pursuant to the amended certificate of designations.

The Company recorded a reduction of \$27,277 in the carrying value of the Series B preferred stock during the year ended December 31, 2021. As of December 31, 2021, there were no shares of Series B preferred stock outstanding.

# NOTE 14—STOCKHOLDERS' EQUITY

# Preferred Stock

The Company is authorized to issue 10,000,000 shares of preferred stock with a par value of \$0.0001 per share with such designation, rights and preferences as may be determined from time to time by the Company's Board of Directors. The Company adopted and filed with the Secretary of State of the State of Delaware each of the Certificate of Designations for the Series A preferred stock and the Series B preferred stock as amendments to the Company's Second Amended and Restated Certificate of Incorporation (as amended, the "Charter") each on May 24, 2019 and March 31, 2020, to authorize and establish the rights, preferences and privileges of the Series A preferred stock and Series B preferred stock, respectively. See "Note 13—Mezzanine Equity" for the discussion of preferred stock issued and outstanding.

# Class A Common Stock

The Company is authorized to issue 400,000,000 shares of Class A common stock with a par value of \$0.0001 per share. As of December 31, 2021 and 2020, there were 53,148,952 and 20,718,659 shares of Class A common stock issued and outstanding, respectively. As of December 31, 2021, 285,715 outstanding shares of Class A common stock were subject to cancellation on November 9, 2024, unless the closing price per share of the Class A common stock has equaled or exceeded \$42.00 for any 20 trading days within any 30-trading day period, and 174,194 outstanding shares of Class A common stock were subject to the same cancellation provision, but at a closing price per share of \$47.25.

ATM Agreement. On June 26, 2020, the Company entered into an Equity Distribution Agreement (the "ATM Agreement") with Piper Sandler & Co. relating to the Company's shares of Class A common stock. In accordance with the terms of the ATM Agreement, the Company may offer and sell shares of our Class A common stock over a period of time. The ATM Agreement relates to an "at-the-market" offering program. Under the ATM Agreement, the Company will pay Piper Sandler an aggregate commission of up to 3% of the gross sales price per share of Class A common stock sold under the ATM Agreement. On March 19, 2021, the Company increased the number of shares of Class A common stock that it may offer in accordance with the terms of the ATM Agreement to a total amount of \$50,000, subject to the limitations set forth in General Instruction I.B.6 of Form S-3.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

During the year ended December 31, 2021, the Company sold 5,091,800 shares of Class A common stock for total net proceeds of \$14,667 and paid \$454 in commissions under the ATM Agreement. During the year ended December 31, 2020, the Company sold 226,359 shares of Class A common stock for total net proceeds of \$400 and paid \$12 in commissions under the ATM Agreement. Since inception on June 26, 2020 through December 31, 2021, the Company has sold a total of 5,318,159 shares of Class A common stock under the ATM Agreement for total net proceeds of \$15,067 and paid \$466 in commissions.

Subsequent to December 31, 2021, we sold an additional 9,767,941 shares of Class A common stock for a total net proceeds of \$21,282 and paid \$658 in commissions.

# Class B Common Stock

The Company is authorized to issue 20,000,000 shares of Class B common stock with a par value of \$0.0001 per share. The shares of Class B common stock are non-economic; however, holders are entitled to one vote per share. Each share of Class B common stock, together with one unit of USWS Holdings, is exchangeable for one share of Class A common stock or, at the Company's election, the cash equivalent to the market value of one share of Class A common stock.

During the years ended December 31, 2021 and 2020, 2,302,936 and 3,197,756 shares of Class B common stock, respectively, were converted into 657,982 and 913,645 shares of Class A common stock, respectively, which has been adjusted to reflect the reverse stock split.

As of December 31, 2020, there were 2,302,936 shares of Class B common stock issued and outstanding. As of December 31, 2021, there were no shares of Class B common stock issued and outstanding.

# Noncontrolling Interest

The Company's noncontrolling ownership interest in consolidated subsidiaries is presented in the consolidated balance sheet within stockholders' equity (deficit) as a separate component and represents approximately 3% ownership of USWS Holdings as of December 31, 2020.

During the first quarter of 2021, the remaining noncontrolling interest holders of USWS Holdings exchanged all of their respective shares for the Company's Class A common stock. Accordingly, USWS Holdings became the Company's wholly owned subsidiary as of December 31, 2021.

# NOTE 15—EARNINGS (LOSS) PER SHARE

Basic earnings (loss) per share is computed by dividing income (loss) available to common stockholders by the weighted average number of common shares outstanding during the period. Diluted earnings per share is computed in the same manner as basic earnings per share except that the denominator is increased to include the number of additional Class A common shares that could have been outstanding assuming the exercise of stock options and warrants, conversion of Series A and Series B preferred stock, conversion of Class B common stock, vesting of restricted shares of Class A common stock, conversion of Convertible Senior Notes and issuance of Class A common stock associated with the deferred stock units and certain performance awards.

Basic and diluted net income (loss) per share excludes the income (loss) attributable to and shares associated with the 459,909 shares of Class A common stock that are subject to cancellation on November 9, 2024 if certain market conditions have not been met. The Company has included in the calculation accrued dividends on Series A and Series B preferred stock and related deemed and imputed dividends.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

The following table sets forth the calculation of basic and diluted earnings (loss) per share for the periods indicated based on the weighted average number of shares of Class A common stock outstanding:

	Year Ended December 31,		
	2021	2020	
Basic net income (loss) per share			
Numerator:			
Net loss attributable to U.S. Well Services, Inc.	\$ (70,605)	\$ (229,340)	
Net loss attributable to cancellable Class A common stock	988	5,560	
Basic net loss attributable to U.S. Well Services, Inc. shareholders	(69,617)	(223,780)	
Dividends accrued on Series A preferred stock	(5,857)	(7,214)	
Dividends accrued on Series B preferred stock	(4,591)	(2,049)	
Deemed and imputed dividends on Series A preferred stock	(750)	(13,022)	
Deemed and imputed dividends on Series B preferred stock	(7,178)	(564)	
Exchange of Series A preferred stock for Convertible Senior Notes	8,936		
Basic net loss attributable to U.S. Well Services, Inc. common			
shareholders	\$ (79,057)	\$ (246,629)	
Denominator:			
Weighted average shares outstanding	32,853,855	18,971,692	
Cancellable Class A common stock	(459,909)	(459,909)	
Basic and diluted weighted average shares outstanding	32,393,946	18,511,783	
Basic and diluted net loss per share attributable to Class A common shareholders	\$ (2.44)	\$ (13.32)	

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

A summary of securities excluded from the computation of diluted earnings per share is presented below for the applicable periods:

	Year Er Decembe	
	2021	2020
Dilutive earnings per share:		
Anti-dilutive stock options	250,649	250,649
Anti-dilutive warrants	4,515,980	4,122,328
Anti-dilutive restricted stock	388,886	414,082
Anti-dilutive deferred stock units	2,052,474	2,546,245
Anti-dilutive shares from Pool B Awards	3,387,218	2,897,855
Anti-dilutive Class B common stock convertible into Class A common stock	_	657,982
Anti-dilutive Series A preferred stock convertible into Class A common stock	1,168,297	2,588,050
Anti-dilutive Series B preferred stock convertible into Class A common stock	_	22,355,922
Anti-dilutive Convertible Senior Notes convertible into Class A common		
stock	28,056,027	
Potentially dilutive securities excluded as anti-dilutive	39,819,531	35,833,113

# NOTE 16—SHARE-BASED COMPENSATION

Share-based compensation expense consisted of the following:

	Year I	Ended
	Decemb	oer 31,
	2021	2020
Restricted stock	\$ 3,459	\$ 4,719
Stock options	866	905
Deferred stock units	1,371	984
Pool A Awards	4,381	2,328
Pool B Awards	1,617	1,120
Total	\$11,694(1)	\$10,056(2)

- (1) For the year ended December 31, 2021, \$2,097 was presented as cost of services and \$9,597 was presented as selling, general and administrative expenses in the consolidated statement of operations.
- (2) For the year ended December 31, 2020, \$1,940 was presented as cost of services and \$8,116 was presented as selling, general and administrative expenses in the consolidated statement of operations.

# Long-Term Incentive Plan

On September 21, 2020, the Board of Directors approved the Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan (as amended, the "LTIP") and on May 14, 2021 it was approved by the Company's stockholders at the 2021 Annual Meeting of Stockholders. Under the LTIP, 5,414,193 shares of Class A common

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

stock are reserved for issuance pursuant to awards under the LTIP, which includes time-based restricted stock, deferred stock units, or other performance incentive awards.

# Restricted Stock

Pursuant to the LTIP, the Company grants shares of restricted Class A common stock ("restricted stock") to certain employees and directors. Restricted stock is subject to restrictions on transfer and is generally subject to a risk of forfeiture if the award recipient is no longer an employee or director of the Company prior to the lapse of the restriction. Restricted stock granted to employees generally vests over four years in equal installments each year on the anniversary of the grant date and grants to directors generally vest in full after one year. The grant date fair value of the restricted stock is determined using the closing price of the Company's Class A common stock on the grant date.

The following table summarizes the restricted stock activity for the year ended December 31, 2021:

	Shares	Averag Fair	eighted- e Grant-Date Value per Share
Outstanding at December 31, 2020	414,071	\$	31.01
Granted	88,025		2.70
Vested	(102,615)		31.19
Forfeited	(10,595)		31.19
Outstanding at December 31, 2021	388,886	\$	24.55

The weighted average grant date fair value per share of restricted stock granted was \$2.70 during the year ended December 31, 2021. There were no restricted stock grants made during the year ended December 31, 2020. The fair value of restricted stock vested was \$3,200 and \$6,123 during the years ended December 31, 2021 and 2020, respectively. As of December 31, 2021, total unrecognized compensation cost related to restricted stock was \$3,793 which is expected to be recognized over a weighted-average period of 1.2 years.

# Stock options

In 2019, the Company granted stock options under the LTIP to certain employees of the Company. The fair value of stock options on the date of grant was calculated using the Black-Scholes valuation model. These stock options were granted with seven-year terms and vest over four years in equal installments each year on the anniversary of the grant date. The expected term of the options granted was based on the safe harbor rule of the SEC Staff Accounting Bulletin No. 107 "Share-Based Payment" as the Company lacks historical exercise data to estimate the expected term of these options. The expected stock price volatility is calculated based on the Company's peer group because the Company does not have sufficient historical data and will continue to use peer group volatility information until historical volatility of the Company is available to measure expected volatility for future grants. The exercise price for stock options granted equals the closing market price of the underlying stock on the date of grant. These options are time-based and are not based upon attainment of performance goals.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

The following table summarizes the stock option activity for the year ended December 31, 2021:

Sh		Weighted- Average Exercise Price per Share		Weighted- Average Remaining Contractual Life (in years)	
Outstanding at December 31, 2020	250,649	\$	31.19	5.21	
Granted	_		_	_	
Exercised	_		_	_	
Forfeited/Expired			_	_	
Outstanding at December 31, 2021	250,649	\$	31.19	4.21	
Exercisable at December 31, 2021	125,325	\$	31.19	4.21	

During the years ended December 31, 2021 and 2020, no stock options were granted or exercised. As of December 31, 2021, the total unrecognized compensation cost related to stock options was \$1,040 which is expected to be recognized over a weighted average period of 1.21 years.

# **Deferred Stock Units**

The Company awards deferred stock units ("DSUs") to certain key employees of the Company pursuant to the LTIP. Each DSU represents the right to receive one share of the Company's Class A common stock. DSUs generally vest over three years in equal installments each year on the anniversary of the vesting effective date, subject to the grantee's continuous service through each vesting period. The grant date fair value of the DSU is determined using the closing price of the Company's Class A common stock on the grant date.

The following table summarizes the DSUs activity for the year ended December 31, 2021:

			ed-Average int Date Value per
	Units		Unit
Outstanding at December 31, 2020	2,546,249	\$	1.16
Granted	404,294		2.87
Vested	(873,408)		1.16
Forfeited	(24,661)		1.16
Outstanding at December 31, 2021	2,052,474	\$	1.49

The weighted average grant date fair value per unit of DSUs granted was \$2.87 and \$1.16 during the years ended December 31, 2021 and 2020, respectively. The fair value of DSUs vested was \$1,009 during the year ended December 31, 2021. No DSUs vested during the year ended December 31, 2020. As of December 31, 2021, the total unrecognized compensation cost related to DSUs was \$1,730 which is expected to be recognized over a weighted average period of 1.45 years.

# Pool A Performance Awards

The Company grants Pool A Performance Awards ("Pool A Awards") to certain key employees of the Company. Each Pool A Award represents the right to receive, at the Company's election, a fixed monetary amount either in

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

cash or a variable number of shares of the Company's Class A common stock based on its closing share price on the date of settlement. The Pool A Awards vest in full one year on the anniversary of the vesting effective date specified in the applicable award agreement but settlement does not occur until the fifth anniversary of the grant date.

The Company accounts for the Pool A Awards under liability accounting as a result of the fixed monetary amount that could be settled either in cash or a variable number of shares of the Company's Class A common stock. Since the settlement will not occur until the fifth anniversary of the grant date, the Company considers the delayed settlement as a post-vesting restriction which would impact the determination of grant-date fair value of the award.

During the year ended December 31, 2021, the Company granted Pool A Awards that will fully vest on January 1, 2022 (the "2021 Pool A Awards"). During the year ended December 31, 2020, the Company granted Pool A Awards which fully vested on January 1, 2021 (the "2020 Pool A Awards"). The grant date fair value of the 2020 Pool A Awards and 2021 Pool A Awards was \$2,183 and \$2,718, respectively.

As of December 31, 2021, the fair value of the 2020 Pool A Awards and 2021 Pool A Awards liabilities were remeasured to \$3,098 and \$3,611, respectively, and is presented as other long-term liabilities in the consolidated balance sheets. The fair value was estimated using a risk-adjusted discount rate reflecting the weighted-average cost of capital of similarly traded public companies.

As of December 31, 2021, the total unrecognized compensation cost related to Pool A Awards was \$3,932, which is expected to be recognized over a weighted average period of 4.33 years.

# Pool B Performance Awards

The Company grants Pool B Performance Awards ("Pool B Awards") to certain key employees of the Company. Each Pool B Award represents the right to receive, at the Company's election, either a cash payment calculated in accordance with the award agreement, or a fixed number of shares of the Company's Class A common stock. The Pool B Awards vest over three years in equal installments each year on the anniversary of the vesting effective date specified in the applicable award agreement, subject to the grantee's continuous services through each vesting period. The grant date fair value of the Pool B Awards is determined using the closing price of the Company's Class A common stock on the grant date.

The following table summarizes the Pool B Awards activity for the year ended December 31, 2021:

	Fair Value
Outstanding at December 31, 2020	\$ 3,356
Granted	1,499
Vested	(1,151)
Forfeited	(32)
Outstanding at December 31, 2021	<u>\$</u> 3,672
	<del></del>

The grant date fair value of the Pool B Awards granted was \$1,499 and \$3,362, during the years ended December 31, 2021 and 2020, respectively. As of December 31, 2021, the total unrecognized compensation cost related to Pool B Awards was \$2,086, which is expected to be recognized over a weighted average period of 1.48 years.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

# NOTE 17—EMPLOYEE BENEFIT PLAN

On March 1, 2013, the Company established the U.S. Well Services 401(k) Plan. The Company matched 100% of employee contributions up to 6% of the employee's salary, subject to cliff vesting after two years of service. At the end of the first quarter of 2020, the Company suspended its match of employee contributions. Our matching contribution was \$976 for the year ended December 31, 2020, included in cost of services and selling, general and administrative expenses in the consolidated statements of operations.

# NOTE 18—RELATED PARTY TRANSACTIONS

Crestview III USWS TE, LLC and Crestview III USWS, L.P. and its affiliates (collectively, "Crestview Partners") are part of an affiliate group which have an ownership interest in the Company of greater than 10% and is entitled to designate for nomination by the Company for election two directors to serve on the Company's Board of Directors.

#### Convertible Senior Notes

On June 24, 2021, Crestview Partners purchased \$40,000 of Convertible Senior Notes that are convertible into shares of the Company's Class A common stock for consideration of \$20,000 in cash and in exchange for 15,588 shares of the Company's Series A preferred stock. See "Note 10—Convertible Senior Notes" and "Note 13—Mezzanine Equity" for discussion of the Cash Notes and Exchange Notes.

# Series A Redeemable Convertible Preferred Stock

On May 24, 2019, Crestview Partners purchased 20,000 shares of Series A preferred stock for a total payment of \$20,000 and received 1,066,666 initial Series A Warrants with the right to receive additional Series A Warrants in connection with the Company's private placement of Series A preferred stock. During the years ended December 31, 2021 and 2020, Crestview Partners received 711,112 additional Series A Warrants, respectively. See "Note 9—Warrant Liabilities" for discussion of the Series A Warrants.

# Series B Redeemable Convertible Preferred Stock

On April 1, 2020, Crestview Partners purchased 11,500 shares of Series B preferred stock for a total payment of \$11,500, the TCW Group, Inc. ("TCW Group") purchased 6,500 shares of Series B preferred stock for a total payment of \$6,500 and David Matlin, a member of the Company's Board of Directors, purchased 1,878 shares of Series B preferred stock for a total payment of \$1,878.

On September 17, 2021, the Company converted Crestview Partners' 11,500 shares of Series B preferred stock and related accrued dividends for 13,974,980 shares of Class A common stock, pursuant to the amended certificate of designations. On September 17, 2021, the Company converted TCW Group's 6,500 shares of Series B preferred stock and related accrued dividends for 7,898,902 shares of Class A common stock. On September 17, 2021, the Company converted David Matlin's and David Treadwell's 1,678 and 200 shares of Series B preferred stock and related accrued dividends for 2,039,132 and 243,044 shares of Class A common stock, respectively. See "Note 13—Mezzanine Equity" for discussion of the Series B preferred stock.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS (in thousands, except share and per share amounts, or where otherwise noted)

# NOTE 19—INCOME TAXES

The Company's net deferred tax assets are as follows:

	Decemb	er 31,
	2021	2020
Deferred tax assets:	<u> </u>	
Net operating loss carryforward	\$ 92,339	\$ 56,815
Startup/Organization expenses	176	177
Investment in Partnership	_	70,244
Interest expense	4,066	1,652
Property and equipment	8,752	_
Leases	1,210	_
Intangible assets	26,667	_
Sec. 743(b) adjustment	3,572	_
Inventory reserve	314	_
Stock-based compensation	957	_
Accruals and other	1,224	137
Total deferred tax assets	139,277	129,025
Deferred tax liabilities:		
Prepaids	(2,078)	
Total deferred tax liabilities	(2,078)	_
Net deferred tax asset	137,199	129,025
Valuation allowance	(137,199)	(129,025)
Net deferred tax assets	<u> </u>	<u>\$</u>

The income tax provision consists of the following:

		Year Ended December 31,	
	2021	2020	
Current income taxes:			
Federal	\$ (27)	\$(757)	
State	<u>_</u>	(67)	
Total current	(27)	(824)	
Deferred income taxes:			
Total deferred	<u></u>		
Income tax benefit	<u>\$ (27)</u>	\$(824)	

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

A reconciliation of the federal income tax rate to the Company's effective tax rate is as follows:

	December 31,	2021
Pre-tax book loss	\$ 70,676	
Federal provision (benefit)	(14,842)	-21.00%
Permanent differences	(303)	-0.43%
Return to provision, other	6,665	9.42%
Valuation allowance	8,453	11.96%
Total income tax benefit	<u>\$(27)</u>	-0.04%

On March 27, 2020, the President signed the Coronavirus Aid, Relief and Economic Security Act (as amended, the "CARES Act") into law. The CARES Act contains several corporate income tax provisions, including, among other things, providing a 5-year carryback of net operating loss ("NOL") tax carryforwards generated in tax years 2018, 2019, and 2020, removing the 80% taxable income limitation on utilization of those NOLs if carried back to prior tax years or utilized in tax years beginning before 2021, temporarily liberalizing the interest deductions rules under Section 163(j) of the Tax Cuts and Jobs Act of 2017, and making corporate alternative minimum tax credits immediately refundable. During 2020, the Company filed an application to carry back its 2018 NOLs, claiming a refund of \$757.

As of December 31, 2021, the Company had total U.S. federal net operating loss ("NOL") carryforwards of \$388,541 and \$307,413 of state NOLs available to offset future taxable income. Federal NOLs of \$28,387 would begin to expire in 2036 if unused. Federal NOLs generated after December 31, 2017 do not expire and the state rules vary by state. After consideration of all of the information available, management has established a valuation allowance against the deferred tax assets of the Company's tax loss carryforwards to the extent it is not more likely than not they will be realized. As of December 31, 2021, the valuation allowance totaled \$137,199.

In assessing the realization of the deferred tax assets, management considers whether it is more likely than not that some portion of all of the deferred tax assets will not be realized. The ultimate realization of deferred tax assets is dependent upon the generation of future taxable income during the periods in which temporary differences representing net future deductible amounts become deductible. Management considers the positive and negative evidence with respect to sources of taxable income for purposes of determining the realization of deferred tax assets. In accordance with Section 382 of the Internal Revenue Code, deductibility of the Company's NOLs may be subject to an annual limitation in the event of a change in control as defined under the regulations.

The Company files income tax returns in the U.S. federal jurisdiction and various state and local jurisdictions and is subject to examination by the taxing authorities

We follow guidance issued by the FASB in accounting for uncertainty in income taxes. This guidance clarifies the accounting for income taxes by prescribing the minimum recognition threshold an income tax position is required to meet before being recognized in the consolidated financial statements and applies to all income tax positions. Each income tax position is assessed using a two-step process. A determination is first made as to whether it is more likely than not that the income tax position will be sustained, based upon technical merits, upon examination by the taxing authorities. If the income tax position is expected to meet the more likely than not criteria, the benefit recorded in the consolidated financial statements equals the largest amount that is greater than 50% likely to be realized upon its ultimate settlement.

We have considered our exposure under the standard at both the federal and state tax levels. We did not record any liabilities for uncertain tax positions as of December 31, 2021 or 2020. We record income tax-related interest

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

and penalties, if any, as a component of income tax expense. We did not incur any material interest or penalties on income taxes.

# NOTE 20—SUBSEQUENT EVENTS

# Sixth Senior Secured Term Loan Amendment

On February 28, 2022, the Company, USWS LLC and all the other subsidiaries of the Company entered into a Consent and Sixth Amendment (the "Sixth Term Loan Amendment") to the Senior Secured Term Loan Agreement with CLMG Corp., as administrative agent and term loan collateral agent, and the lenders party thereto. Pursuant to the Sixth Term Loan Amendment, the Senior Secured Term Loan Agreement was amended in order to (i) provide for an additional tranche of last-out term loans (the "Term C Loan") of up to \$35,000 principal amount and (ii) make certain modifications to the Senior Secured Term Loan Agreement on the terms and conditions set forth in the Sixth Term Loan Amendment.

The Term C Loan was funded by a syndicate of institutions, including Crestview Partners and its affiliates, David Matlin, a member of the Company's Board of Directors, and one or more other institutions (collectively, the "Term C Loan Lenders") and was extended on a last-out basis in the payment waterfall relative to the existing Term A Loan and Term B Loan (collectively the "Senior Secured Term Loan"), and was otherwise made on the general terms and conditions consistent with the Senior Secured Term Loan.

The Term C Loan shall bear interest at a Benchmark Rate (as defined in the Senior Secured Term Loan Agreement), subject to a 2.0% floor, plus 12.0% per annum, accrued on a daily basis, to be paid-in-kind by increasing the principal amount of the outstanding Term C Loan on each interest payment date. The default rate for the Term C Loan shall be 2.0% over and above the non-default rate, subject to the same terms and conditions for the Senior Secured Term Loan.

After repayment of the Senior Secured Term Loan in full, the Company will pay to the Term C Loan Lenders the following premium upon any repayment, prepayment or acceleration of the Term C Loan:

- 30% of the repaid, prepaid, or accelerated amount, if such repayment, prepayment or acceleration occurs on or prior to May 31, 2022;
- 65% of the repaid, prepaid, or accelerated amount, if such repayment, prepayment or acceleration occurs between June 1, 2022 and August 31, 2022; and
- 100% of the repaid, prepaid, or accelerated amount, if such repayment, prepayment or acceleration occurs on or after to September 1, 2022.

As of March 15, 2022, the outstanding principal balance of the Term C Loan was \$21,500. The maturity date of the Term C Loan is December 5, 2025. The Company intends to use the proceeds for general working capital, including the funding of growth capital expenditures.

Pursuant to the Sixth Term Loan Amendment, the principal amortization schedule for the existing Senior Secured Term Loan was modified such that commencing on June 30, 2023, the Company is required to make quarterly principal payments of \$5,000 until September 30, 2025. The Senior Secured Term Loan interest payments were modified that if on April 1, 2022, the outstanding principal amount of the Senior Secured Term Loan is equal to or less than \$103,000 then the interest rate from April 1, 2022 through December 31, 2022 shall be payable as follows: (i) 1.0% per annum in cash and (ii) 4.125% per annum shall be paid-in-kind by increasing the outstanding principal amount of the Senior Secured Term Loan on each interest payment date.

# U.S. WELL SERVICES, INC. NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except share and per share amounts, or where otherwise noted)

Additionally, pursuant to the Sixth Term Loan Amendment, other covenants and terms in the Senior Secured Term Loan Agreement were amended including, but not limited to, certain covenants relating to collateral, approved growth capital expenditures, and mandatory prepayments.

# Term C Loan Warrants

In connection with the entry into the Sixth Term Loan Amendment and the Term C Loan, the Company issued warrants to the Term C Loan Lenders, including Crestview Partners and David Matlin. On February 28, 2022, the Company issued 13,953,488 warrants to certain of the Term C Loan Lenders (the "February 2022 Warrants") exercisable to purchase an equivalent number of shares of Class A common stock at an exercise price of \$1.10 per share, subject to adjustment, and expiring on February 28, 2028. On March 1, 2022, the Company issued 1,046,511 warrants to certain of the Term C Loan Lenders (the "March 2022 Warrants" and, together with the February 2022 Warrants, the "Term C Loan Warrants") exercisable to purchase an equivalent number of shares of Class A common stock at an exercise price of \$1.29, subject to adjustment, and expiring on March 1, 2028.

The Term C Loan Warrants were offered in a private offering that is exempt from registration under the Securities Act, and may not be offered or sold in the United States absent such registration or an exemption from the registration requirements of the Securities Act.

# Registered Direct Offering

On March 11, 2022, the Company completed a registered direct offering of 14,180,375 shares of Class A common stock for gross proceeds of approximately \$25,000, before deducting placement agent fees and other offering expenses. The Company intends to use the net proceeds for working capital purposes, including the funding of certain capital expenditures.

The Company issued 992,626 warrants to the placement agent as partial compensation for its services in connection with this offering (the "Placement Agent Warrants"). The Placement Agent Warrants are exercisable to purchase an equivalent number of shares of Class A common stock at an initial exercise price of \$2.2038 per share. The Placement Agent Warrants are exercisable immediately, subject to certain ownership limitations and will expire three and one-half years following the date of issuance.

On March 11, 2022, in a concurrent private placement, the Company also issued 14,180,375 warrants to the purchasers of the shares of Class A common stock in the registered direct offering (the "RDO Investor Warrants"). The RDO Investor Warrants are exercisable to purchase an equivalent number of shares of Class A common stock at an initial exercise price of \$1.763 per share. The RDO Investor Warrants are exercisable immediately, subject to certain ownership limitations and will expire three and one-half years following the date of issuance.

# U.S. WELL SERVICES, INC. CONDENSED CONSOLIDATED BALANCE SHEETS

(in thousands, except share and per share amounts) (unaudited)

	June 30, 2022	Decem	ber 31, 2021
ASSETS			
Current assets:			
Cash and cash equivalents	\$ 17,268	\$	6,384
Restricted cash	736		2,736
Accounts receivable (net of allowance for doubtful accounts of \$0 as of June 30, 2022 and December 31, 2021,			
respectively)	31,721		25,743
Inventory, net	8,074		6,351
Assets held for sale			2,043
Prepaids and other current assets	6,373		18,748
Total current assets	64,172		62,005
Property and equipment, net	194,943		162,664
Operating lease right-of-use assets	18,197		_
Finance lease right-of-use assets	3,246		_
Intangible assets (net of accumulated amortization of \$2,173 and \$1,690 as of June 30, 2022 and December 31,			
2021, respectively)	12,017		12,500
Goodwill	4,971		4,971
Other assets	1,119		1,417
TOTAL ASSETS	\$ 298,665	\$	243,557
LIABILITIES, MEZZANINE EQUITY AND STOCKHOLDERS' DEFICIT		, <u></u>	
Current liabilities:			
Accounts payable	\$ 49,027	\$	29,180
Accrued expenses and other current liabilities	14,301		16,842
Notes payable	2,814		2,320
Current portion of long-term debt	8,750		5,000
Current portion of equipment financing	3,437		3,412
Current portion of capital lease obligations	_		1,092
Current portion of operating lease liabilities	9,605		_
Current portion of finance lease liabilities	1,168		
Total current liabilities	89,102		57,846
Warrant liabilities	2,733	<u> </u>	3,557
Convertible senior notes	116,183		105,769
Long-term debt	158,423		167,507
Long-term equipment financing	3,328		5,128
Long-term capital lease obligations	_		2,112
Long-term operating lease liabilities	8,748		_
Long-term finance lease liabilities	2,176		_
Other long-term liabilities	7,927		6,875
Total liabilities	388,620		348,794
Commitments and contingencies (NOTE 12)		<u> </u>	
Mezzanine equity:			
Series A Redeemable Convertible Preferred Stock, par value \$0.0001 per share; 55,000 shares authorized;			
19,610 shares issued and outstanding as of June 30, 2022 and December 31, 2021, respectively; aggregate			
liquidation preference of \$29,499 and \$27,274 as of June 30, 2022 and December 31, 2021, respectively	26,092		23,866
Stockholders' deficit:	,		
Class A Common Stock, par value of \$0.0001 per share; 400,000,000 shares authorized; 12,827,306 shares and			
8,858,161 shares issued and outstanding as of June 30, 2022 and December 31, 2021, respectively(1)	1		1
Additional paid in capital (1)	312,571		263,932
Accumulated deficit	(428,619)		(393,036)
Total Stockholders' deficit	(116,047)		(129,103)
TOTAL LIABILITIES, MEZZANINE EQUITY AND STOCKHOLDERS' DEFICIT	\$ 298,665	\$	243,557
TOTAL BRUBERIES, WEZZANINE EQUIT AND STOCKHOLDERS DEFICIT	φ 270,003	φ	473,337

<sup>(1)</sup> Periods presented have been adjusted to reflect the 1-for-6 reverse stock split on August 4, 2022. See Note 2, Reverse Stock Splits, for details.

# U.S. WELL SERVICES, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(in thousands, except per share amounts) (unaudited)

	Three Months Ended June 30,		Six Months Endo June 30,	
	2022	2021	2022	2021
Revenue	\$ 68,764	\$ 78,799	\$109,914	\$ 155,057
Costs and expenses:				
Cost of services (excluding depreciation and amortization)	55,209	59,252	95,932	121,883
Depreciation and amortization	5,877	9,836	11,577	20,942
Selling, general and administrative expenses	9,406	7,214	17,778	14,604
Litigation settlement	_	35,000	_	35,000
Loss (gain) on disposal of assets	(76)	(545)	2,980	1,891
Loss from operations	(1,652)	(31,958)	(18,353)	(39,263)
Interest expense, net	(9,562)	(7,333)	(17,530)	(13,516)
Change in fair value of warrant liabilities	1,577	(136)	831	(7,287)
Patent license sales	_	22,500	_	22,500
Loss on extinguishment of debt	_	(839)	(1,651)	(839)
Other income	302	23	1,623	52
Loss before income taxes	(9,335)	(17,743)	(35,080)	(38,353)
Income tax benefit		(27)		(27)
Net loss	(9,335)	(17,716)	(35,080)	(38,326)
Net loss attributable to noncontrolling interest				(44)
Net loss attributable to U.S. Well Services, Inc.	(9,335)	(17,716)	(35,080)	(38,282)
Dividends accrued on Series A preferred stock	(1,135)	(1,998)	(2,226)	(3,811)
Dividends accrued on Series B preferred stock	· · · ·	(811)	_	(1,522)
Deemed and imputed dividends on Series A preferred stock	_	(286)	_	(750)
Deemed dividends on Series B preferred stock	_	(1,501)	_	(5,669)
Exchange of Series A preferred stock for convertible senior notes		8,936		8,936
Net loss attributable to U.S. Well Services, Inc. common stockholders	<u>\$(10,470)</u>	<u>\$(13,376)</u>	<u>\$ (37,306)</u>	\$ (41,098)
Loss per common share (See Note 15):				
Basic and diluted <sup>(1)</sup>	\$ (0.82)	\$ (3.10)	\$ (3.27)	\$ (10.12)
Weighted average common shares outstanding:	\$ (0.02)	\$ (5.10)	\$ (5.27)	ψ (10.12)
Basic and diluted <sup>(1)</sup>	12,720	4,219	11,347	3,991

(1) Periods presented have been adjusted to reflect the 1-for-3.5 reverse stock split on September 30, 2021 and the 1-for-6 reverse stock split on August 4, 2022. See Note 2, Reverse Stock Splits, for details.

# U.S. WELL SERVICES, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands) (unaudited)

June 3           CASH FLOWS FROM OPERATING ACTIVITIES:         \$(35,080)         \$(38,32)           Net loss         Adjustments to reconcile net loss to cash provided by (used in) operating activities:         11,577         20,94           Depreciation and amortization         1831         7,28           Provision for losses on accounts receivable         −           Provision for losses on inventory obsolescence         541         1,35
Net loss\$(35,080)\$(38,32)Adjustments to reconcile net loss to cash provided by (used in) operating activities:11,57720,94Depreciation and amortization(831)7,28Change in fair value of warrant liabilities(831)7,28Provision for losses on accounts receivable—Provision for losses on inventory obsolescence5411,35
Adjustments to reconcile net loss to cash provided by (used in) operating activities:  Depreciation and amortization  Change in fair value of warrant liabilities  Provision for losses on accounts receivable  Provision for losses on inventory obsolescence  11,577 20,94  (831) 7,28  Provision for losses on inventory obsolescence
Depreciation and amortization11,57720,94Change in fair value of warrant liabilities(831)7,28Provision for losses on accounts receivable—Provision for losses on inventory obsolescence5411,39
Change in fair value of warrant liabilities (831) 7,28 Provision for losses on accounts receivable — Provision for losses on inventory obsolescence 541 1,39
Provision for losses on accounts receivable Provision for losses on inventory obsolescence  541 1,39
Provision for losses on inventory obsolescence 541 1,39
Loss on disposal of assets 2,980 1,89
Non-cash lease expense 1,997 —
Convertible senior notes converted into sales of patent licenses — (22,50
Amortization of debt discount, premium and issuance costs 3,130 3,60
Paid-in-kind interest 12,237 25
Loss on extinguishment of debt, net 1,651 83
Share-based compensation expense 1,613 3,66
Changes in assets and liabilities:
Accounts receivable (5,978) (10,50
Inventory (2,263) (96
Prepaids and other current assets 1,445 (7,70
Accounts payable 8,902 6,42
Accrued liabilities 1,229 (2,64
Accrued interest 409 7,95
Net cash provided by (used in) operating activities $\frac{3,559}{2}$ (28,37)
CASH FLOWS FROM INVESTING ACTIVITIES:
Purchase of property and equipment (48,337) (24,84
Proceeds from sale of property and equipment and insurance proceeds from damaged property and equipment 17,338 8,55
Net cash used in investing activities (30,999) (16,28
CASH FLOWS FROM FINANCING ACTIVITIES:
Proceeds from revolving credit facility 7,344 24,72
Repayments of revolving credit facility (14,495) (14,75
Proceeds from issuance of long-term debt – 3,00
Proceeds from issuance of long-term debt and warrants 21,500 —
Repayments of long-term debt (19,022) (12,56
Payment of fees related to debt extinguishment (96) (4
Proceeds from issuance of convertible senior notes — 86,50
Proceeds from issuance of notes payable 3,826 9,13
Repayments of notes payable (3,332) (3,69
Repayments of amounts under equipment financing (1,774) (1,73
Principal payments under capital lease obligations — (10
Principal payments on finance lease liabilities (560) —
Proceeds from issuance of common stock and warrants in registered direct offering, net 22,730 —
Proceeds from issuance of common stock via the 2020 ATM Agreement, net 21,298 13,56
Deferred financing costs (1,095) (6,56
Net cash provided by financing activities 36,324 97,46
Net increase in cash and cash equivalents and restricted cash  8,884  52,80
Cash and cash equivalents and restricted cash, beginning of period 9,120 5,26
Cash and cash equivalents and restricted cash, end of period \$ 18,004 \$ 58,06

# U.S. WELL SERVICES, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (continued) (in thousands)

(unaudited)

		ths Ended ie 30,
	2022	2021
Supplemental cash flow disclosure:		
Interest paid	\$1,411	\$ 1,276
Income tax refunds received	(757)	(810)
Non-cash investing and financing activities:		
Issuance of warrants to term C loan lenders	6,491	_
Issuance of warrants to purchasers of Class A common stock in registered direct offering	7,044	_
Issuance of warrants to placement agent in registered direct offering	470	_
Exchange of Series A preferred stock for convertible senior notes	<u> </u>	24,780
Conversion of Series B preferred stock to Class A common stock	_	1,067
Deemed and imputed dividends on Series A preferred stock	_	750
Accrued Series A preferred stock dividends	2,226	3,811
Accrued Series B preferred stock dividends	<del>_</del>	1,522
Changes in accrued and unpaid capital expenditures	8,927	7,543
Assets under capital lease obligations	_	1,113

# U.S. WELL SERVICES, INC. CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' DEFICIT

(in thousands, except share amounts) (unaudited)

	Three Months Ended June 30, 2022							
			Cla	iss B				
	Class A Cor	nmon	Con	nmon	Additional			
	Stock(1	)	St	ock	Paid in	Accumulated	Noncontrolling	Total
	Shares	Amount	Shares	Amount	Capital <sup>(1)</sup>	Deficit	Interest	Deficit
Balance at March 31, 2022	12,844,437	\$ 1	_	\$ —	\$314,979	\$ (419,284)	\$ —	\$(104,304)
Share-based compensation	_	_	_	_	(1,273)	_	_	(1,273)
Restricted stock forfeitures	(17,131)	_	_	_	_	_	_	_
Accrued Series A preferred stock dividends	_	_	_	_	(1,135)	_	_	(1,135)
Net loss						(9,335)		(9,335)
Balance at June 30, 2022	12,827,306	<u>\$ 1</u>		<u>\$ —</u>	\$312,571	<u>\$ (428,619)</u>	<u>\$</u>	<u>\$(116,047</u> )

	Three Months Ended June 30, 2021							
				ass B				
	Class A Co Stock			nmon ock	Additional	4 1 . 4 . 1	N	77.4.1
	Shares	Amount	Shares	Amount	Paid in Capital <sup>(1)</sup>	Accumulated Deficit	Noncontrolling Interest	Total Deficit
Balance at March 31, 2021	4,288,969	\$ —		\$ —	\$226,749	\$ (342,997)	\$ —	\$(116,248)
Class A common stock issuance	113,412	_	_	_	2,891		_	2,891
Conversion of Series B preferred stock to Class A								
common stock	44,167	_	_	_	270	_	_	270
Exchange of Series A preferred stock for								
convertible senior notes	_	_	_	_	8,936	_	_	8,936
Share-based compensation	_	_	_	_	1,617	_	_	1,617
Deemed and imputed dividends on Series A								
preferred stock	_	_	_	_	(286)	_	_	(286)
Accrued Series A preferred stock dividends	_	_	_	_	(1,998)	_	_	(1,998)
Accrued Series B preferred stock dividends	_	_	_	_	(811)	_	_	(811)
Net loss						(17,716)		(17,716)
Balance at June 30, 2021	4,446,548	<u>\$</u>		<u>\$</u>	\$237,368	\$ (360,713)	<u>\$</u>	\$(123,345)

<sup>(1)</sup> Periods presented have been adjusted to reflect the 1-for-3.5 reverse stock split on September 30, 2021 and the 1-for-6 reverse stock split on August 4, 2022. See Note 2, Reverse Stock Splits, for details.

# U.S. WELL SERVICES, INC. CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' DEFICIT (continued)

(in thousands, except share amounts) (unaudited)

	Six Months Ended June 30, 2022							
	Class A Co Stock		Class B Co Stoc		Additional Paid in	Accumulated	Noncontrolling	Total
	Shares	Amount	Shares	Amount	Capital(1)	Deficit	Interest	Deficit
Balance at December 31, 2021	8,858,161	\$ 1		\$ —	\$ 263,932	\$ (393,036)	\$ —	\$(129,103)
Adoption of ASC 842, Leases on January 1, 2022	_	_	_	_	_	(503)	_	(503)
Class A common stock issuance	3,991,387	_	_	_	36,494		_	36,494
Warrants issuance	_	_	_	_	14,005	_	_	14,005
Share-based compensation	_	_	_	_	396	_	_	396
Tax withholding related to vesting of share-based								
compensation	(4,383)	_	_	_	(30)	_	_	(30)
Restricted stock forfeitures	(17,859)	_	_	_		_	_	
Accrued Series A preferred stock dividends	_	_	_	_	(2,226)	_	_	(2,226)
Net loss	_	_	_	_		(35,080)	_	(35,080)
Balance at June 30, 2022	12,827,306	\$ 1		\$ —	\$ 312,571	\$ (428,619)	\$ —	\$(116,047)

	Six Months Ended June 30, 2021							
	Class A Co Stock		Class B Co Stock		Additional Paid in	Accumulated	Noncontrolling	Total
	Shares	Amount	Shares	Amount	Capital(1)	Deficit	Interest	Deficit
Balance at December 31, 2020	3,453,111	\$ —	2,302,936	\$ —	\$ 217,219	\$ (322,431)	\$ —	\$(105,212)
Class A common stock issuance	714,586	_	_	_	13,242	_	_	13,242
Conversion of Class B common stock to Class A								
common stock	109,664	_	(2,302,936)	_	_	_	_	_
Conversion of Series B preferred stock to Class A								
common stock	174,919	_	_	_	1,068	_	_	1,068
Exchange of Series A preferred stock for								
convertible senior notes	_	_	_	_	8,936	_	_	8,936
Share-based compensation	_	_	_	_	3,136	_	44	3,180
Tax withholding related to vesting of share-based								
compensation	(4,938)	_	_	_	(150)	_	_	(150)
Restricted stock forfeitures	(794)	_	_	_	_	_	_	
Deemed and imputed dividends on Series A								
preferred stock	_	_	_	_	(750)	_	_	(750)
Accrued Series A preferred stock dividends	_	_	_	_	(3,811)	_	_	(3,811)
Accrued Series B preferred stock dividends	_	_	_	_	(1,522)	_	_	(1,522)
Net loss						(38,282)	(44)	(38,326)
Balance at June 30, 2021	4,446,548	<u>\$</u>		<u> </u>	\$ 237,368	\$ (360,713)	<u> </u>	<u>\$(123,345)</u>

<sup>(1)</sup> Periods presented have been adjusted to reflect the 1-for-3.5 reverse stock split on September 30, 2021 and the 1-for-6 reverse stock split on August 4, 2022. See Note 2, Reverse Stock Splits, for details.

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

# NOTE 1—DESCRIPTION OF BUSINESS

U.S. Well Services, Inc. ("USWS," the "Company," "we," "us" or "our"), is a Houston, Texas-based technology-focused oilfield service company focused on electric powered pressure pumping services for oil and natural gas exploration and production companies in the United States. The process of pressure pumping involves pumping a pressurized stream of fluid—typically a mixture of water, chemicals, and proppant—into a well casing or tubing to cause the underground mineral formation to fracture or crack. Fractures release trapped hydrocarbon particles and provide a conductive channel for the oil or natural gas to flow freely to the wellbore for collection. The propping agent or proppant becomes lodged in the cracks created by the stimulation process, "propping" them open to facilitate the flow of hydrocarbons from the reservoir to the well.

The Company's fleets consist mostly of all-electric, mobile pressure pumping equipment and other auxiliary heavy equipment to perform stimulation services. The Company's Clean Fleet® electric fleets replace the traditional engines, transmissions, and radiators used in conventional diesel fleets with electric motors powered by electricity. The Company utilizes high-pressure hydraulic fracturing pumps mounted on trailers and refers to the group of pump trailers and other equipment necessary to perform a typical job as a "fleet" and the personnel assigned to each fleet as a "crew". In May 2021, the Company announced its commitment to becoming an all-electric pressure pumping services provider and since then it has sold most of its legacy, diesel-powered pressure pumping equipment. We have retained some of our legacy, diesel-powered pressure pumping equipment for use during our transition to support our electric fleets and bridge the time gap between our customers' current service needs and the deployment of our newbuild Nyx Clean Fleets®. We expect to continue phasing out our remaining conventional fleet operations as we take delivery of our newbuild Nyx Clean Fleets®.

# Proposed Merger with ProFrac

On June 21, 2022, USWS, ProFrac Holding Corp. ("ProFrac"), and Thunderclap Merger Sub I, Inc., an indirect subsidiary of ProFrac ("Merger Sub"), entered into an agreement and plan of merger (the "Merger Agreement"), providing that, among other things and subject to the terms and conditions of the Merger Agreement, at the effective time (the "Effective Time") of the Merger (defined below):

- Merger Sub will be merged with and into USWS, with USWS surviving as an indirect subsidiary of ProFrac, subject to the terms and conditions set forth in the Merger Agreement (the "Merger").
- Each share of USWS Class A common stock issued and outstanding immediately prior to the Effective Time will be cancelled and converted into the right to receive 0.3366 shares of ProFrac Class A common stock (the "Merger Consideration").
- USWS will take all requisite action so that, effective as of immediately prior to the Effective Time:
  - (i) each holder of USWS's Series A Redeemable Convertible Preferred Stock, par value \$0.0001 per share, (the "Series A preferred stock") issued and outstanding at such time may convert such stock into shares of USWS Class A common stock at the Merger Conversion Ratio (as defined in the Merger Agreement), and (ii) any shares of Series A preferred stock issued and outstanding immediately prior to the Effective Time not so converted at the Merger Conversion Ratio will automatically convert into shares of USWS Class A common stock at the then-effective conversion rate as calculated pursuant to USWS's Certificate of Designations (as defined in the Merger Agreement); and
  - each Convertible Senior Note issued and outstanding at such time will automatically convert into a number of shares of USWS
     Class A common stock equal to the quotient obtained by dividing (i) the amount of outstanding aggregate principal amount, plus
     accrued and unpaid interest, owing under such Convertible Senior Note through the date immediately prior to the Effective Time, by
     (ii) \$7.32.

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

- At the Effective Time, each Term C Loan Warrant issued and outstanding immediately prior to the Effective Time will be automatically canceled and will cease to exist and no consideration will be delivered in exchange therefor.
- At the Effective Time, each Public Warrant, Private Placement Warrant, Series A Warrant, RDO Warrant and Placement Agent Warrant issued and outstanding immediately prior to the Effective Time (collectively, the "Rollover Warrants"), in accordance with the terms of such Rollover Warrants, will be amended to provide the right to receive a warrant to purchase a number of shares of ProFrac Class A common stock equal to (i) the number of shares of USWS Class A common stock underlying such Rollover Warrant multiplied by (ii) the Exchange Ratio. The exercise price of such Rollover Warrants will be the exercise price of such Rollover Warrant divided by the Exchange Ratio.
- At the Effective Time, each share of USWS Class A common stock subject to vesting, repurchase, or other lapse of restrictions that is
  outstanding and unvested under the Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan (the "LTIP") immediately
  prior to the Effective Time will, by virtue of the Merger and without any action on the part of the holder thereof, be canceled in exchange for
  the right to receive the Merger Consideration and, in lieu of any fractional shares, cash.
- Immediately prior to the Effective Time, each then-outstanding deferred stock unit ("DSU") or restricted stock unit ("RSU"), in each case representing a right to receive one share of USWS Class A common stock granted under the LTIP, will, by virtue of the Merger and without any action on the part of the holder thereof, be canceled and converted into the right to receive the Merger Consideration and, in lieu of any fractional shares, cash (treating such DSU or RSU in the same manner as if it were an outstanding share of USWS Class A common stock for such purposes).
- Immediately prior to the Effective Time, by virtue of the Merger and without any action on the part of the holder thereof, (i) each then outstanding Pool A Performance Award ("Pool A Award") shall be canceled and converted into the right to receive (A) for recipients of Pool A Awards who consent to the terms of that certain Amendment to Performance Awards included as an exhibit to the Merger Agreement (the "Award Amendment"), the Merger Consideration in an amount equal to the accumulated award value as of July 19, 2022 divided by \$7.32, and (B) with respect to each Pool A Award not amended by an Award Amendment, the Merger Consideration in an amount equal to the accumulated award value as of the Effective Time divided by \$7.32; and (ii) each then-outstanding Pool B Performance Award ("Pool B Award") shall be canceled and converted into the right to receive (A) with respect to each Pool B Award amended by an Award Amendment, the Merger Consideration in an amount equal to the accumulated award value as of July 19, 2022 divided by \$6.468, and (B) with respect to each Pool B Award not amended by an Award Amendment, the Merger Consideration in an amount equal to the accumulated award value as of July 19, 2022 divided by \$6.468, and (B) with respect to each Pool B Award not amended by an Award Amendment, the Merger Consideration in an amount equal to the accumulated award value as of the Effective Time divided by \$6.468.

The consummation of the Merger is subject to customary conditions, including, among others, (a) receipt of USWS stockholder approval, (b) the absence of any applicable law or order prohibiting the consummation of the Merger or the Parent Stock Issuance (as defined in the Merger Agreement), (c) the expiration or termination of the waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), (d) the effectiveness of the Registration Statement (as defined in the Merger Agreement), (e) approval of the Parent Stock Issuance for listing on Nasdaq, (f) subject to certain exceptions, the accuracy of the representations and warranties of each party to the Merger Agreement, (g) the performance in all material respects of each party of its obligations under the Merger Agreement and (h) the absence of a Company Material Adverse Effect and Parent Material Adverse Effect (as such terms are defined in the Merger Agreement).

#### U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

Dan Wilks and Farris Wilks, together with certain of their affiliates, (collectively, the "Wilks Parties") collectively hold a controlling interest in ProFrac. Certain Wilks Parties also own certain securities of USWS, including outstanding Convertible Senior Notes and Term C Loan Warrants.

Concurrently with the execution and delivery of the Merger Agreement, certain stockholders of USWS (each, a "Supporting Stockholder" and, collectively, the "Supporting Stockholders") entered into a Voting Agreement with ProFrac (the "Voting Agreements"). Pursuant to the Voting Agreements, each Supporting Stockholder agreed to, among other things, (i) support and vote in favor of the Merger Agreement, the Merger and the other transactions contemplated by the Merger Agreement, (ii) take (and refrain from taking) certain other actions in connection with the transactions contemplated by the Merger Agreement and (iii) to the extent applicable, deliver a Conversion Notice (as defined in the Merger Agreement) to convert each Supporting Stockholder's shares of USWS's Series A preferred stock at the Merger Conversion Ratio as described above.

The Supporting Stockholders together beneficially own, in the aggregate, approximately 44% of the currently outstanding shares of USWS's Class A common stock, which includes 9,568 shares of USWS's Class A common stock, or approximately 0.07%, held by THRC Holdings, LP ("THRC"), one of the Wilks Parties.

USWS has agreed to operate its business in the ordinary course during the period between the execution of the Merger Agreement and the Effective Time, subject to specific exemptions set forth in the Merger Agreement, and have agreed to certain other customary restrictions on operations, as set forth in the Merger Agreement.

The Merger Agreement and the transactions contemplated thereby were unanimously approved by the Board of Directors of USWS and the Board of Directors of ProFrac on June 21, 2022, each acting on the unanimous recommendation of their respective special committees formed for the purpose of negotiating and evaluating the fairness of the transactions contemplated by the Merger Agreement.

The foregoing description of the Merger Agreement is not complete and is qualified in its entirety by reference to the full text of the Merger Agreement, which is filed as Exhibit 2.1 to this Current Report on Form 10-Q and incorporated herein by reference.

# NOTE 2—SIGNIFICANT ACCOUNTING POLICIES

# Basis of Presentation

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with generally accepted accounting principles in the United States of America ("GAAP") for interim financial information and pursuant to the instructions to Form 10-Q and Regulation S-X issued by the Securities and Exchange Commission ("SEC"). Accordingly, these financial statements do not include all information or notes required by GAAP for annual financial statements and should be read in conjunction with the annual financial statements included in the Annual Report on Form 10-K for the year ended December 31, 2021 (the "Annual Report"), filed with the SEC on March 30, 2022.

The accompanying unaudited condensed consolidated financial statements and accompanying notes present the consolidated financial position, results of operations, cash flows, and stockholders' deficit of the Company as of June 30, 2022 and December 31, 2021, and for the three and six months ended June 30, 2022 and 2021. The interim data includes all adjustments, consisting only of normal recurring adjustments, which are, in the opinion of management, necessary for a fair presentation of the results for the interim periods. The results of operations for the three and six months ended June 30, 2022 are not necessarily indicative of the results of operations expected for the entire fiscal year ended December 31, 2022.

#### U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

Our operations are organized into a single business segment, which consists of pressure pumping services, and we have one reportable geographical business segment, the United States.

# Reverse Stock Splits

On September 30, 2021, the Company effected a 1-for-3.5 reverse split of its Class A common stock. All owners of record as of September 30, 2021 received one issued and outstanding share of the Company's Class A common stock in exchange for three and one half outstanding shares of the Company's Class A common stock.

On August 4, 2022, the Company effected a 1-for-6 reverse split of its Class A common stock. All owners of record as of August 4, 2022 received one issued and outstanding share of the Company's Class A common stock in exchange for six outstanding shares of the Company's Class A common stock.

No fractional shares of Class A common stock were issued as a result of these reverse stock splits. Any fractional shares in connection with these reverse stock splits were rounded up to the nearest whole share and no stockholders received cash in lieu of fractional shares. The reverse stock splits had no impact on the number of shares of Class A common stock that the Company is authorized to issue pursuant to its certificate of incorporation or on the par value per share of the Class A common stock. Proportional adjustments were made to the number of shares of Class A common stock issuable upon exercise or conversion of the Company's equity awards, convertible preferred stock and warrants, as well as the applicable exercise price. All share and per share information included in this Quarterly Report on Form 10-Q has been retroactively adjusted to reflect the impact of these reverse stock splits.

# Principles of Consolidation

The condensed consolidated financial statements comprise the financial statements of the Company and its wholly owned subsidiaries. Subsidiaries are fully consolidated from the date of acquisition, being the date on which the Company obtains control, and continue to be consolidated until the date when such control ceases. The financial statements of the subsidiaries are prepared for the same reporting period as the Company. All significant intercompany balances and transactions are eliminated upon consolidation.

# Use of Estimates

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. We regularly evaluate estimates and judgments based on historical experience and other relevant facts and circumstances. Significant estimates included in these financial statements primarily relate to allowance for doubtful accounts, allowance for inventory obsolescence, estimated useful lives and valuation of long-lived assets, impairment assessments of goodwill and other long-lived assets, valuation of right-of-use assets and lease liabilities, estimates of fair value of warrant liabilities, term loan, and convertible senior notes, and the valuation of share-based compensation and certain equity instruments. Actual results could differ from those estimates.

# Restricted Cash

Cash and cash equivalents that are restricted as to withdrawal or use under the terms of certain contractual agreements, or are reserved for a specific purpose, and not readily available for immediate or general use are recorded in restricted cash on our condensed consolidated balance sheets. As of June 30, 2022, restricted cash primarily consisted of \$0.7 million transferred into a trust account to support our workers' compensation

#### U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

obligations. As of December 31, 2021, restricted cash consisted of \$0.7 million transferred into a trust account to support our workers' compensation obligations and \$2.0 million for the prepayment of the Senior Secured Term Loan.

The following table provides a reconciliation of cash and cash equivalents and restricted cash reported on the condensed consolidated statements of cash flows:

	Jun	e 30,
	2022	2021
Cash and cash equivalents	\$ 17,268	\$ 57,544
Restricted cash	736	519
Cash and cash equivalents and restricted cash	<u>\$ 18,004</u>	\$ 58,063

#### Accounts Receivable

Accounts receivable are recorded at their outstanding balances adjusted for an allowance for doubtful accounts. Revenue earned and recognized in advance of invoice issuance creates assets referred to as "unbilled receivables". Unbilled receivables are presented on a combined basis with accounts receivable. Allowance for doubtful accounts is determined by analyzing the payment history and credit worthiness of each customer. Receivable balances are charged off when they are considered uncollectible by management. Recoveries of receivables previously charged off are recorded as income when received.

During the first quarter of 2021, the Company wrote-off accounts receivables of \$12.0 million, which was previously reserved in full as of December 31, 2020. As of June 30, 2022, the Company did not record an allowance for doubtful accounts.

#### Inventory

Inventory consists of proppant, chemicals, and other consumable materials and supplies used in our pressure pumping operations. Inventories are stated at the lower of cost or net realizable value. Cost is determined principally on a first-in-first-out cost basis. All inventories are purchased for use by the Company in the delivery of its services with no inventory being sold separately to outside parties. Inventory quantities on hand are reviewed regularly and write-downs for obsolete inventory are recorded based on our forecast of the inventory item demand in the near future. As of June 30, 2022 and December 31, 2021, the Company had established inventory reserves of \$1.7 million and \$1.3 million, respectively, for obsolete and slow-moving inventory.

# Property and Equipment

Property and equipment are carried at cost, with depreciation provided on a straight-line basis over their estimated useful lives. Expenditures for renewals and betterments that extend the lives of the assets are capitalized. Amounts spent for maintenance and repairs, which do not improve or extend the life of the related asset, are charged to expense as incurred.

The Company separately identifies and accounts for certain critical components of its pressure pumping units including the engine, transmission, and pump, which requires us to separately estimate the useful lives of these components. For our other service equipment, we do not separately identify and track depreciation of specific original components. When we replace components of these assets, we typically estimate the net book values of the components that are retired, which are based primarily upon their replacement costs, their ages and their original estimated useful lives.

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

#### Leases

At inception, the Company determines whether an arrangement is a lease and the appropriate lease classification as operating or finance. When a lease is identified, a right-of-use asset and the corresponding lease liability are recorded on the lease commencement date based on the present value of the remaining lease payments over the lease term on the condensed consolidated balance sheet. In the event a lease does not provide an implicit rate, the Company uses its incremental borrowing rate based on information available at the commencement date in determining the present value of the remaining lease payments. Leases may include options to extend or terminate the lease. The Company generally does not include renewal or termination options in its assessment of the leases unless extension or termination for certain assets is deemed to be reasonably certain. The Company has elected the practical expedient to not recognize lease assets and liabilities for leases with a term of 12 months or less. Operating lease expenses are recognized on a straight-line basis over the lease term. The Company also has some lease agreements with lease and non-lease components, which are accounted for as a single lease component.

Right-of-use assets are assessed periodically for impairment if events or circumstances occur that indicate the carrying amount of the asset may not be recoverable. The Company monitors events and modifications of existing lease agreements that would require reassessment of the lease. When a reassessment results in the remeasurement of a lease liability, a corresponding adjustment is made to the carrying amount of the corresponding right-of-use asset.

#### Warrants

The Company evaluates all its financial instruments, including issued stock purchase warrants, to determine if such instruments are derivatives or contain features that qualify as embedded derivatives, pursuant to Accounting Standards Codification ("ASC") 480, Distinguishing Liabilities from Equity and ASC 815-15, Derivatives and Hedging—Embedded Derivatives. The classification of derivative instruments, including whether such instruments should be recorded as liabilities or equity is evaluated pursuant to ASC 815-40, Derivatives and Hedging—Contracts in Entity's Own Equity.

The Company's issued and outstanding Public Warrants, Private Placement Warrants and Series A Warrants are recognized as liabilities. Accordingly, we recognize these warrant instruments as liabilities at fair value upon issuance and adjust the instruments to fair value at the end of each reporting period. Any change in fair value is recognized in our condensed consolidated statements of operations. The Public Warrants are valued using their quoted market price since they are publicly traded and thus had an observable market price. The Private Placement Warrants are valued using a Monte Carlo simulation model. The Series A Warrants are valued using the Black-Scholes option pricing model.

The Company's issued and outstanding Term C Loan Warrants, RDO Investor Warrants and Placement Agent Warrants are recognized as equity.

See "Note 9—Warrants" for additional disclosure of the Company's outstanding warrants.

# Fair Value of Financial Instruments

Fair value is defined under ASC 820, Fair Value Measurement, as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. ASC 820 also establishes a three-level hierarchy, which requires an entity to maximize the use of observable

#### U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

inputs and minimize the use of unobservable inputs when measuring fair value. The three levels are defined as follows:

- Level 1—inputs are quoted prices (unadjusted) in active markets for identical assets or liabilities.
- · Level 2—inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly.
- Level 3—inputs are unobservable for the asset or liability.

The following is a summary of the carrying amounts and estimated fair values of our financial instruments:

Senior Secured Term Loan. The fair value of the Senior Secured Term Loan is \$91.2 million and \$106.6 million as of June 30, 2022 and December 31, 2021, respectively, based on the market price quoted from external sources. If the Senior Secured Term Loan was measured at fair value in the financial statements, it would be classified as Level 2 in the fair value hierarchy.

Term C Loan. The fair value of the Term C Loan is \$37.9 million as of June 30, 2022 based on the market price quoted from external sources. If the Term C Loan was measured at fair value in the financial statements, it would be classified as Level 2 in the fair value hierarchy.

Equipment financing. The carrying value of the equipment financing notes approximates fair value as its terms are consistent with and comparable to current market rates as of June 30, 2022 and December 31, 2021, respectively.

Warrants. Certain of the Company's outstanding warrants are accounted for as liabilities and measured at fair value. See "Note 9—Warrants" for fair value measurements associated with the Company's warrants.

Convertible Senior Notes. The fair value of the Convertible Senior Notes is \$113.1 million and \$73.5 million as of June 30, 2022 and December 31, 2021, respectively, based on an option pricing framework using a lattice model. If the Convertible Senior Notes were measured at fair value in the financial statements, they would be classified as Level 2 in the fair value hierarchy.

#### Revenue Recognition

The Company recognizes revenue based on the customer's ability to benefit from the services rendered in an amount that reflects the consideration expected to be received in exchange for those services.

The Company's performance obligations are satisfied over time, typically measured by the number of stages completed or the number of pumping days a fleet is available to pump for a customer in a month. All revenue is recognized when a contract with a customer exists, collectability of amounts subject to invoice is probable, the performance obligations under the contract have been satisfied over time, and the amount to which the Company has the right to invoice has been determined. A portion of the Company's contracts contain variable consideration; however, this variable consideration is typically unknown at the time of contract inception, and is not known until the job is complete, at which time the variability is resolved.

The Company has elected to use the "as invoiced" practical expedient to recognize revenue based upon the amount it has a right to invoice upon the completion of each performance obligation per the terms of the contract.

Patent License Sales. The sales of the right to use the Company's patented Clean Flee® technology is a single performance obligation. The Company recognizes the income associated with the patent license sales at the point

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

in time when the Company satisfies its performance obligation by granting the purchaser the right to use the patented Clean Flee® technology and transfer of control has occurred. The patent license sales are recognized as other income in our condensed consolidated statement of operations.

# Major Customer and Concentration of Credit Risk

The concentration of our customers in the oil and natural gas industry may impact our overall exposure to credit risk, either positively or negatively, in that customers may be similarly affected by changes in economic and industry conditions. We perform ongoing credit evaluations of our customers and do not generally require collateral in support of our trade receivables.

The following table shows the percentage of revenues from our significant customers:

	Three Months En	ided June 30,
	2022	2021
Customer A	19.1%	*
Customer B	*	12.9%
Customer C	14.2%	10.6%
Customer D	17.1%	*
Customer E	*	19.6%
Customer F	32.3%	17.8%
Customer H	*	12.1%
Customer I	*	11.4%
Customer K	17.1%	*

	Six Months Ende	d June 30,
	2022	2021
Customer A	17.5%	*
Customer B	*	12.9%
Customer C	17.1%	11.6%
Customer D	12.9%	*
Customer E	*	18.1%
Customer F	35.7%	19.0%
Customer H	*	15.2%
Customer K	16.7%	*

The following table shows the percentage of trade receivables from our significant customers:

	June 30, 2022	December 31, 2021
Customer A	22.1%	*
Customer C	21.6%	20.4%
Customer D	13.7%	*
Customer F	24.4%	24.3%
Customer J	*	29.7%
Customer K	16.5%	25.0%

An asterisk indicates that trade receivable is less than ten percent.

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

# Income Taxes

The provision or benefit from income taxes for interim periods is determined using an estimate of our annual effective tax rate, adjusted for discrete items, if any, that are taken into account in the relevant period. Each quarter, the Company updates the estimate of the annual effective tax rate, and if the estimated tax rate changes, the Company records a cumulative adjustment.

The Company's effective tax rate on continuing operations for the three and six months ended June 30, 2022, was 0.00%. The Company's effective tax rate on continuing operations for the three and six months ended June 30, 2021 was (0.15)% and (0.07)%, respectively. The difference between the effective tax rate and the U.S. federal statutory rate is primarily due to a valuation allowance on the Company's federal and state net deferred tax assets and excess tax benefits related to net operating losses. Due to tax losses and offsetting valuation allowance, the Company did not record a provision for U.S. income taxes in any period.

The Company is subject to taxation in the U.S. The tax years subsequent to 2017 remain open and subject to examination by federal and state taxing authorities in which the Company is subject to tax. The Company is not under examination in any other jurisdictions.

As of June 30, 2022, the Company has provided a valuation allowance against all federal and state deferred tax assets. Management continues to evaluate the realizability of deferred tax assets and the related valuation allowance. If management's assessment of the deferred tax assets or the corresponding valuation allowance were to change, the Company would record the related adjustment to income during the period in which management makes the determination. After consideration of all the information available, management determined that a valuation allowance was appropriate, as it is more likely than not that the Company will not utilize its net deferred tax assets.

# NOTE 3—ACCOUNTING STANDARDS

Except as discussed below, there have been no recent accounting pronouncements or changes in accounting pronouncements during the six months ended June 30, 2022, as compared to the recent accounting pronouncements described in the Annual Report, that are of significance, or potential significance to the Company.

# Recently Adopted Accounting Pronouncements

In February 2016, the Financial Accounting Standards Board ("FASB") issued ASU2016-02, Leases (Topic 842) and subsequent amendments to the initial guidance: ASU 2017-13, ASU 2018-10, ASU 2018-11, ASU 2018-20, ASU 2019-01 and ASU 2020-02 (collectively, "ASC 842"). ASC 842 requires companies to generally recognize on the balance sheet operating and financing lease liabilities and corresponding right-of-use assets. On January 1, 2022, the Company adopted ASC 842, using the modified retrospective with applied transition method and recognized a cumulative impact to retained earnings in that period, which allowed the Company to continue to apply the legacy guidance in Topic 840, Leases, including its disclosure requirements, in the comparative periods presented in the year of adoption. The Company elected to apply certain practical expedients, whereby it will not reassess (i) whether any expired or existing contracts are or contain leases, (ii) the lease classification for any expired or existing leases and (iii) initial direct costs for any existing leases.

On January 1, 2022, upon adoption of the new leasing standard, the Company recognized operating right-of-use assets of \$1.6 million and operating lease liabilities of \$1.6 million. On January 1, 2022, the Company

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

recognized finance lease right-of-use assets of \$3.6 million, reclassified from property and equipment, net and finance lease liabilities of \$3.8 million, including \$3.2 million reclassified from current and long-term capital lease obligations. The impact of adoption of the new leasing standard had no material impact to the condensed consolidated statements of operations.

In December 2019, the FASB issued ASU2019-12, *Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes* which removes specific exceptions to the general principles in Topic 740 in GAAP. The new guidance also improves the issuer's application of income tax-related guidance and simplifies GAAP for franchise taxes that are partially based on income, transactions with a government that result in a step up in the tax basis of goodwill, separate financial statements of legal entities that are not subject to tax, and enacted changes in tax laws in interim periods. The Company adopted ASU 2019-12 on January 1, 2022, using the prospective method of transition. The adoption did not have a material impact on the Company's condensed consolidated financial statements.

In August 2020, the FASB issued ASU 2020-06, *Debt—Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging—Contracts in Entity's Own Equity (Subtopic 815-40): Accounting for Convertible Instruments and Contracts in an Entity's Own Equity, which simplifies accounting for convertible instruments by removing major separation models required under current GAAP. It also amends the accounting for certain contracts in an entity's own equity that are currently accounted for as derivatives because of specific settlement provisions. In addition, the new guidance of settlement provisions and certain contracts that may be settled in cash or shares impact the diluted EPS computation. The new guidance will be effective for smaller reporting companies for fiscal years beginning after December 15, 2023, including interim periods within those fiscal years. Early adoption is permitted. The Company adopted ASU 2020-06 on January 1, 2022, using the modified retrospective method of transition. The adoption did not have a material impact on the Company's condensed consolidated financial statements.* 

# NOTE 4—PREPAIDS AND OTHER CURRENT ASSETS

Prepaids and other current assets consisted of the following:

	June 30, 2022	Decer	nber 31, 2021
Prepaid insurance	\$ 4,646	\$	5,207
Recoverable costs from insurance	_		11,109
Income tax receivable	_		757
Other current assets	1,727		1,675
Total prepaid expenses and other current assets	\$ 6,373	\$	18,748

During the six months ended June 30, 2022 and 2021, the Company prepaid \$5.1 million and \$11.7 million, respectively, in insurance premiums related to renewals of various insurance policies.

The Company has insurance coverage in place covering, among other things, property damage up to certain specified amounts. During the six months ended June 30, 2022, the Company received \$12.0 million from insurance reimbursements, of which \$11.1 million was accrued for as of December 31, 2021. During the six months ended June 30, 2021, the Company received \$6.4 million from insurance reimbursements.

During 2020, the Company filed an application to carry back its 2018 NOLs, claiming a refund of approximately \$0.8 million, which was received during the three months ended June 30, 2022.

#### U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

# NOTE 5-PROPERTY AND EQUIPMENT, NET

Property and equipment consisted of the following:

	Estimated Useful Life (in years)	June 30, 2022	December 31, 2021
Pressure pumping equipment <sup>(1)</sup>	1.5 to 25	\$ 227,521	\$ 186,826
Light duty vehicles <sup>(2)</sup>	5	1,667	5,524
Furniture and fixtures	5	67	67
IT equipment	3	1,033	1,033
Auxiliary equipment	2	14,160	12,218
Leasehold improvements	Term of lease	276	276
		244,724	205,944
Less: accumulated depreciation and amortization		(49,781)	(43,280)
Property and equipment, net		\$ 194,943	\$ 162,664

- (1) As of December 31, 2021, the Company had capitalized \$0.6 million of pressure pumping equipment, related to capital leases and the accumulated depreciation was \$0.3 million.
- (2) As of December 31, 2021, the Company had capitalized \$3.9 million of light duty vehicles, related to capital leases and the accumulated depreciation was \$0.5 million.

Depreciation and amortization expense related to property and equipment was \$5.3 million and \$9.6 million during the three months ended June 30, 2022 and 2021, respectively, and \$10.5 million and \$20.5 million during the six months ended June 30, 2022 and 2021, respectively.

#### Assets Sales

In May 2021, the Company announced its commitment to becoming anall-electric pressure pumping services provider and since then has sold most of its legacy, diesel-powered pressure pumping equipment. As of June 30, 2022, the Company did not have any assets classified as held for sale. As of December 31, 2021, the Company had classified \$2.0 million in net book value of diesel pressure pumping equipment as assets held for sale on the condensed consolidated balance sheet.

During the six months ended June 30, 2022, the Company received \$5.3 million in proceeds from the sale of property and equipment, of which \$2.1 million was for assets classified as held for sale. The Company used the proceeds received from the asset sales to pay down the principal balance of its Senior Secured Term Loan.

The Company recognized a gain of \$0.1 million and a loss of \$3.0 million from disposal of assets during the three and six months ended June 30, 2022, respectively, and a gain of \$0.5 million and a loss of \$1.9 million during the three and six months ended June 30, 2021, respectively.

# NOTE 6—LEASES

As described in "Note 3—Accounting Standards" the Company adopted ASC 842, *Leases*, on January 1, 2022. Prior periods presented have not been adjusted and continue to be reported in accordance with the legacy guidance in ASC 840, *Leases*. The Company's capital lease assets and related accumulated amortization were

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

recorded as property and equipment and the capital lease liabilities were recorded as current and long-term capital lease obligations on the condensed consolidated balance sheet as of December 31, 2021. See "Note 5—Property and Equipment, Net" and "Note 11—Debt" regarding disclosure of capital leases.

The Company has operating and financing leases for office facilities, light duty vehicles, tractors and pressure pumping equipment, including power generation equipment. The Company does not have any material lessor arrangements.

The weighted average remaining lease term and discount rates used in the measurement of the Company's right-of-use assets and lease liabilities are as follows:

	June 30, 2022
Weighted average remaining lease term:	
Operating leases	1.9 years
Finance leases	3.0 years
Weighted average discount rate:	
Operating leases	9.6%
Finance leases	7.4%

The components of lease expense consisted of the following:

	Three Months Ended June 30, 2022	Six Months Ended June 30, 2022
Operating lease expense	\$ 1,725	\$ 2,041
Short-term lease expense	9,981	19,249
Finance lease expense:		
Amortization of right-of-use assets	312	620
Interest on lease liabilities	63	141
Total	\$ 12,081	\$ 22,051

In accordance with prior guidance rent expense was \$0.3 million and \$0.6 million for the three and six months ended June 30, 2021, respectively, of which \$0.2 million and \$0.5 million, respectively, was recorded as cost of services and \$0.1 million and \$0.1 million, respectively, was recorded as selling, general and administrative expenses in the condensed consolidated statements of operations.

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

Supplemental cash flow information related to leases is as follows:

	onths Ended e 30, 2022
Cash paid for amounts included in the measurement of lease liabilities	
Payments on operating lease included in operating cash flows	\$ 1,896
Interest payments on finance leases included in operating cash flows	\$ 133
Principal payments on finance leases included in financing cash	
flows	\$ 560
Right-of-use assets upon adoption of ASC 842 and obtained in exchange	
for new lease liabilities	
Operating leases	\$ 19,935
Finance leases	\$ 3,893

As of June 30, 2022, the future maturities of lease liabilities are as follows:

	Operating	Finance
Fiscal Year	Leases	Leases
Remainder of 2022	\$ 5,718	\$ 782
2023	10,346	1,080
2024	3,968	1,080
2025	67	621
2026	_	40
Thereafter	_ <u></u>	16
Total lease payments	20,099	3,619
Less: imputed interest	(1,746)	(275)
Total lease liabilities	\$ 18,353	\$3,344

In addition, the Company entered into agreements for additional power generation services and equipment. Such long-term operating leases commenced upon the delivery of the equipment, which occurred during the early part of the third quarter of 2022.

As of December 31, 2021, in accordance with prior guidance the minimum future payments onnon-cancellable operating leases and capital leases are as follows:

Fiscal Year	Operating Leases	Capital Leases
2022	\$ 1,107	\$ 1,241
2023	308	896
2024	258	891
2025	67	447
Total	\$ 1,740	\$ 3,475

The total capital lease payments include imputed interest.

#### U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

In October 2021, the Company completed a sale-leaseback transaction for a short-term operating lease. The Company deferred a \$7.4 million gain from disposal of assets to accrued expenses and other current liabilities to amortize over the minimum term of the lease. As of June 30, 2022 and December 31, 2021, the remaining deferred gain on the sale-leaseback transaction was \$1.9 million and \$5.6 million, respectively.

# NOTE 7—ACCRUED EXPENSES AND OTHER CURRENT LIABILITIES

Accrued expenses and other current liabilities consisted of the following:

	June 30, 2022	December 31, 2021
Accrued payroll and benefits	\$ 6,428	\$ 5,188
Accrued taxes	4,292	5,041
Accrued interest	173	287
Deferred gain on sale-leaseback	1,852	5,557
Other current liabilities	1,556	769
Accrued expenses and other current liabilities	\$ 14,301	\$ 16,842

# NOTE 8—NOTES PAYABLE

The Company enters into various premium finance agreements with a credit finance institution to pay the premiums on insurance policies for its directors' and officers' liability, general liability, workers' compensation, umbrella, auto and pollution coverage needs. During the six months ended June 30, 2022 and 2021, the aggregate amount of the premiums financed was \$3.8 million and \$9.1 million, respectively, payable in equal monthly installments at a weighted average interest rate of 4.0% and 5.4%, respectively. These premium finance agreements are due within one year and are recorded as notes payable under current liabilities in the condensed consolidated balance sheets. As of June 30, 2022 and December 31, 2021, the Company had a remaining balance of \$2.8 million and \$2.3 million, respectively, related to notes payable.

# NOTE 9—WARRANTS

Warrants issued and outstanding consisted of the following:

	Balance Sheet		
	Classification	June 30, 2022	December 31, 2021
Public Warrants	Liability	9,994,635	9,994,635
Private Placement Warrants	Liability	9,172,782	9,172,782
Series A Warrants	Liability	6,666,662	6,222,218
Term C Loan Warrants	Equity	14,999,999	_
RDO Investor Warrants	Equity	14,180,375	_
Placement Agent Warrants	Equity	992,626	_

# Public Warrants and Private Placement Warrants

The Company issued public and private warrants in connection with its initial public offering in November 2018 (the "Public Warrants" and "Private Placement Warrants"). As of June 30, 2022, the outstanding Public Warrants and Private Placement Warrants were exercisable for an aggregate of 456,368 shares of Class A common stock.

#### U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

Each Public Warrant and Private Placement Warrant entitles its holder to purchase one forty-second of a share of our Class A common stock at an exercise price of \$5.75 per warrant (\$241.50 per full share equivalent), to be exercised only for a whole number of shares of Class A common stock. The Public Warrants and Private Placement Warrants expire on November 9, 2023 or earlier upon redemption or liquidation. The Public Warrants and Private Placement Warrants are recognized as warrant liabilities in the condensed consolidated balance sheet.

#### Series A Warrants

The Company issued warrants to certain institutional investors in connection with the Company's private placement of Series A preferred stock on May 24, 2019 (the "Series A Warrants"). The Company issued additional warrants to the purchasers in quarterly installments beginning nine months after May 24, 2019 and ending on March 31, 2022.

During the six months ended June 30, 2022 and 2021, the Company issued 444,444 and 888,888 additional Series A Warrants to the purchasers of Series A preferred stock, respectively, in accordance with the Series A preferred stock purchase agreement.

As of June 30, 2022, the outstanding Series A Warrants were exercisable for 317,461 shares of Class A common stock. The Series A Warrants entitle their holders to purchase one twenty-first of a share of Class A common stock at an exercise price of \$7.66 per warrant (\$160.86 per full share equivalent), to be exercised only for a whole number of shares of Class A common stock. The Series A Warrants expire on November 25, 2025. The Series A Warrants are recognized as warrant liabilities in the condensed consolidated balance sheet.

# Term C Loan Warrants

On February 28, 2022, in connection with the entry into the Term C Loan, the Company issued 13,953,488 warrants to certain of the Term C Loan Lenders (the "February 2022 Warrants") exercisable to purchase 2,325,582 shares of Class A common stock at an exercise price of \$6.60 per share, subject to adjustment, and expiring on February 28, 2028.

On March 1, 2022, in connection with the entry into the Term C Loan, the Company issued 1,046,511 warrants to certain of the Term C Loan Lenders (the "March 2022 Warrants" and, together with the February 2022 Warrants, the "Term C Loan Warrants") exercisable to purchase 174,419 shares of Class A common stock at an exercise price of \$7.74, subject to adjustment, and expiring on March 1, 2028.

The Term C Loan Warrants were offered in a private offering that is exempt from registration under the Securities Act, and may not be offered or sold in the United States absent such registration or an exemption from the registration requirements of the Securities Act.

The Term C Loan Warrants are recognized as equity in the condensed consolidated balance sheet. See "Note 11—Debt" for additional disclosure on the accounting for the Term C Loan and Term C Loan Warrants.

# RDO Investor Warrants and Placement Agent Warrants

On March 11, 2022, the Company completed a registered direct offering for 2,363,396 shares of Class A common stock and in a concurrent private placement, the Company also issued 14,180,375 warrants to the purchasers of the shares of Class A common stock in the registered direct offering (the "RDO Investor Warrants").

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

The RDO Investor Warrants are exercisable to purchase 2,363,396 shares of Class A common stock at an exercise price of \$10.578 per share. The RDO Investor Warrants are exercisable immediately, subject to certain ownership limitations and will expire three and one-half years following the date of issuance.

The Company also issued 992,626 warrants to the placement agent as partial compensation for its services in connection with the registered direct offering (the "Placement Agent Warrants") on March 11, 2022. The Placement Agent Warrants are exercisable to purchase 165,438 shares of Class A common stock at an exercise price of \$13.2228 per share. The Placement Agent Warrants are exercisable immediately, subject to certain ownership limitations and will expire three and one-half years following the date of issuance.

The RDO Investor Warrants and Placement Agent Warrants are recognized as equity in the condensed consolidated balance sheet. See "Note 14—Stockholders' Equity" for additional disclosure on the accounting for the RDO Investor Warrants and Placement Agent Warrants.

# Fair Value Measurement

The Company's Public Warrants, Private Placement Warrants and Series A Warrants are accounted for as liabilities measured at fair value upon issuance, with subsequent changes in fair value reported in the Company's condensed consolidated statements of operations each reporting period.

The following tables present the Company's fair value hierarchy for liabilities measured at fair value on a recurring basis:

As of June 30, 2022	in A	d Prices Active rkets vel 1)	Obs In	Other servable nputs evel 2)	Ir	servable iputs evel 3)	<u>Total</u>
Public Warrants	\$	546	\$		\$		\$ 546
	Э	340	Ъ		Þ		
Private Placement Warrants		_		578		_	578
Series A Warrants				1,609			1,609
	\$	546	\$	2,187	\$		\$2,733
As of December 31, 2021							<del>====</del>
Public Warrants	\$	752	\$	_	\$	_	\$ 752
Private Placement Warrants		_		871		_	871
Series A Warrants				1,934		_	1,934
	\$	752	\$	2,805	\$		\$3,557

Public Warrants. The fair value of the Public Warrants are classified as Level 1 in the fair value hierarchy and is valued using quoted market prices, as they are traded in active markets.

Private Placement Warrants. The fair value of the Private Placement Warrants are classified as Level 2 in the fair value hierarchy and is determined using a Monte Carlo simulation model.

Series A Warrants. The fair value of the Series A Warrants are classified as Level 2 in the fair value hierarchy and is determined using the Black-Scholes option pricing model.

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

The following assumptions were used to calculate the fair value for the Private Placement Warrants and Series A Warrants:

	Private Placement Warrants	Series A Warrants
As of June 30, 2022		
Expected remaining life	1.4 years	3.4 years
Volatility rate	255.8%	255.8%
Risk-free interest rate	2.8%	3.0%
Expected dividend rate	0%	0%
As of December 31, 2021		
Expected remaining life	1.9 years	3.9 years
Volatility rate	227.5%	227.5%
Risk-free interest rate	0.7%	1.1%
Expected dividend rate	0%	0%

# NOTE 10—CONVERTIBLE SENIOR NOTES

On June 24, 2021, the Company entered into a Note Purchase Agreement (as amended, the "Note Purchase Agreement") to issue 16.0% Convertible Senior Secured (Third Lien) PIK Notes (the "Convertible Senior Notes"), in a private placement to institutional investors (the "Private Placement"). Two of the institutional investors were Crestview III USWS TE, LLC and Crestview III USWS, L.P. (collectively, "Crestview Partners"), which are related parties. One of the institutional investors was THRC, an affiliate of the Wilks Parties.

During the year ended December 31, 2021, pursuant to the Note Purchase Agreement, the Company issued \$136.5 million in aggregate principal amount of Convertible Senior Notes, comprised of Cash Notes, Exchange Notes (collectively with the Cash Notes, the "Equity Linked Notes") and a License Linked Note, as described below, which mature on June 5, 2026. The Convertible Senior Notes are secured by a third priority security interest in the collateral that secures the Company's obligations under the Senior Secured Term Loan Agreement. The Convertible Senior Notes bear interest at a rate of 16.0% per annum. Accrued and unpaid interest is calculated on the last day of each quarter, commencing September 30, 2021, and will be paid in kind ("PIK") on such date by increasing the principal amount of the outstanding Convertible Senior Notes.

The carrying value of the Convertible Senior Notes is as follows:

	June 30, 2022	Decer	nber 31, 2021
Principal	\$ 114,000	\$	114,000
PIK interest	19,835		9,686
Unamortized debt premium	1,764		1,841
Unamortized debt discount and issuance costs	(19,416)		(19,758)
Net convertible senior notes	<u>\$ 116,183</u>	\$	105,769

As of June 30, 2022, the Convertible Senior Notes are convertible into 5,059,728 shares of the Company's Class A common stock at the option of the holders.

Equity Linked Notes. In June 2021 and July 2021, in connection with the Private Placement, the Company issued and sold \$75.0 million in principal amount of Convertible Senior Notes that are convertible at any time at the

# U.S. WELL SERVICES, INC.

# Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

holder's option, into shares of the Company's Class A common stock for cash (the "Cash Notes"). The conversion prices of the Cash Notes range from \$20.58 to \$26.25, subject to adjustment.

In June 2021, in connection with the Private Placement, the Company issued and sold \$39.0 million in principal amount of Convertible Senior Notes that are convertible at any time at the holder's option, into shares of the Company's Class A common stock in exchange for 30,390 shares of the Company's Series A preferred stock (the "Exchange Notes"). The Exchange Notes are convertible at a conversion price of \$42.00 subject to adjustment.

In accordance with ASC 480, the Company evaluated the Equity Linked Notes and determined they should be classified as liabilities due to the unconditional obligation to settle the notes for a variable number of shares of the Company's Class A common stock based on a fixed monetary amount known at inception. Certain of the Equity Linked Notes were initially measured at fair value as they were considered new instruments issued concurrently to extinguish the Series A preferred stock. See "Note 13 – Mezzanine Equity" for the discussion of Series A preferred stock exchange.

License Linked Note. On June 24, 2021, in connection with the Private Placement, the Company issued and sold a Convertible Senior Note in the principal amount of \$22.5 million that was convertible into a patent license agreement (the "License Linked Note"). On June 29, 2021, the holder exercised its right to convert the License Linked Note in full. The Company recognized the \$22.5 million as other income from patent license sales in its condensed consolidated statement of operations. The debt issuance costs associated with the License Linked Note were fully amortized.

#### NOTE 11—DEBT

Long-term debt consisted of the following:

	June 30, 2022	December 31, 2021
Senior Secured Term Loan	\$ 102,785	\$ 120,745
Term C Loan	22,525	_
ABL Credit Facility	7,019	14,170
USDA Loan	25,000	25,000
Equipment financing	6,765	8,540
Capital leases		3,204
Total debt principal balance	164,094	171,659
Senior Secured Term Loan future interest payable	24,899	24,384
Unamortized debt discount and issuance costs	(15,055)	(11,792)
Current maturities	(12,187)	(9,504)
Net long-term debt	<u>\$ 161,751</u>	\$ 174,747

# Senior Secured Term Loan

On May 7, 2019, the Company, U.S. Well Services, LLC ("USWS LLC"), as the borrower, and all the other subsidiaries of the Company entered into a Senior Secured Term Loan Credit Agreement (as amended, the "Senior Secured Term Loan Agreement") with CLMG Corp., as administrative and collateral agent, and the lenders party thereto. Upon entering the Senior Secured Term Loan Agreement, the Company borrowed

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

\$250.0 million in Term A and Term B Loans (collectively the "Senior Secured Term Loan"), which matures on December 5, 2025.

On February 28, 2022, the Company, USWS LLC and all the other subsidiaries of the Company entered into the sixth term loan amendment to the Senior Secured Term Loan Agreement with CLMG Corp., as administrative agent and term loan collateral agent, and the lenders party thereto, to make certain modifications to the Senior Secured Term Loan Agreement, including, but not limited to, the interest rates, principal payments, certain covenants relating to collateral, approved growth capital expenditures, mandatory prepayments and an additional tranche of last-out term loans (described below).

The Senior Secured Term Loan interest rate was 0.0% for the period beginning April 1, 2020 through March 31, 2022. As of April 1, 2022, the outstanding principal balance of the Senior Secured Term Loan was reduced to less than \$103.0 million, resulting in an interest rate for the period beginning April 1, 2022 through December 31, 2022 of (i) 1.0% per annum in cash and (ii) 4.125% per annum paid-in-kind by increasing the outstanding principal amount of the Senior Secured Term Loan on each interest payment date. The Senior Secured Term Loan will resume incurring interest on January 1, 2023 at the applicable benchmark rate, subject to a 2.0% floor, plus the applicable margin of 8.25% per annum.

The Senior Secured Term Loan requires quarterly principal payments of \$1.25 million until March 31, 2023 and \$5.0 million from June 30, 2023 through September 30, 2025, with final payment due at maturity.

Since April 2020, the Company has accounted for the Senior Secured Term Loan as a troubled debt restructuring under ASC470-60, *Troubled Debt Restructurings by Debtors*. The subsequent amendments, including the sixth term loan amendment, did not result in a significant modification or extinguishment resulting in no change in accounting for the Senior Secured Term Loan.

During the six months ended June 30, 2022, the Company made principal payments of \$19.0 million, which included prepayments of \$16.5 million driven primarily by asset sales. The early repayment of debt resulted in a write-off of \$1.6 million of unamortized debt discount and issuance costs and prepayment fees of \$0.1 million, all of which were presented as loss on extinguishment of debt in the condensed consolidated statements of operations.

As of June 30, 2022, the outstanding principal balance of the Senior Secured Term Loan was \$102.8 million, of which \$8.8 million was due within one year from the balance sheet date.

#### Term C Loan

The sixth term loan amendment to the Senior Secured Term Loan Agreement also provided for an additional tranche oflast-out term loans (the "Term C Loan") of up to \$35.0 million principal amount, with a maturity date of December 5, 2025. As of June 30, 2022, the Company borrowed \$21.5 million in Term C Loans.

The Term C Loan was funded by a syndicate of institutions and individuals (collectively, the "Term C Loan Lenders"), including related parties, Crestview Partners and its affiliates and David Matlin. Additionally, THRC, an affiliate of the Wilks Parties, is a Term C Loan Lender. The Term C Loan was extended on a last-out basis in the payment waterfall relative to the existing Senior Secured Term Loan, and was otherwise made on the general terms and conditions consistent with the Senior Secured Term Loan.

The Term C Loan bears interest at a benchmark rate, subject to a 2.0% floor, plus 12.0% per annum, accrued on a daily basis, to bepaid-in-kind by increasing the principal amount of the outstanding Term C Loan on each

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

interest payment date. The default rate for the Term C Loan is 2.0% over and above thenon-default rate, subject to the same terms and conditions for the Senior Secured Term Loan.

After repayment of the Senior Secured Term Loan in full, the Company will pay to the Term C Loan Lenders the following premium upon any repayment, prepayment or acceleration of the Term C Loan:

- 30% of the repaid, prepaid, or accelerated amount, if such repayment, prepayment or acceleration occurs on or prior to May 31, 2022;
- 65% of the repaid, prepaid, or accelerated amount, if such repayment, prepayment or acceleration occurs between June 1, 2022 and August 31, 2022; and
- 100% of the repaid, prepaid, or accelerated amount, if such repayment, prepayment or acceleration occurs on or after to September 1, 2022.

In connection with the entry into the Term C Loan, the Company issued an aggregate total of 14,999,999 Term C Loan Warrants to the Term C Loan Lenders, including related parties, Crestview Partners and David Matlin. Using the accounting guidance in ASC 470, *Debt* and ASC 815, *Derivatives and Hedging*, the Term C Loan Warrants were recognized as equity in the condensed consolidated balance sheet. Accordingly, the proceeds received were allocated as \$14.7 million and \$6.8 million to the Term C Loan and Term C Loan Warrants, respectively, based on the relative fair value at issuance.

The fair value of the Term C Loan at issuance was \$22.6 million, calculated using a discounted cash flow model. The difference between the Term C Loan principal balance at issuance and the value allocated to it was \$6.8 million, which was accounted for as a debt discount. Additionally, the Company incurred \$0.7 million of transaction costs related to the Term C Loan, which were recorded as debt issuance costs. The debt discount and issuance costs are presented as a direct deduction from the carrying amount of the Term C Loan and are being amortized under the effective interest method over the term of the Term C Loan.

The fair value of the Term C Loan Warrants at issuance was \$10.5 million, calculated using the Black-Scholes option pricing model. The following assumptions were used to calculate the fair value for the Term C Loan Warrants, at issuance:

	February 2022 Warrants	March 2022 Warrants	
Exercise price	\$ 6.60	\$ 7.74	
Contractual term	6.0 years	6.0 years	
Volatility rate	55.0%	55.0%	
Risk-free interest rate	1.8%	1.6%	
Expected dividend rate	0%	0%	

See "Note 9-Warrants" for additional disclosure regarding the Term C Loan Warrants.

As of June 30, 2022, the outstanding principal balance of the Term C Loan was \$22.5 million, including \$1.0 million of PIK interest, classified as long-term debt on the condensed consolidated balance sheet.

#### ABL Credit Facility

On May 7, 2019, the Company, USWS LLC, and all the other subsidiaries of the Company entered into an ABL Credit Agreement (as amended, the "ABL Credit Facility") with the lenders party thereto and Bank of America, N.A., as the administrative agent, swing line lender and letter of credit issuer. As of June 30, 2022, the aggregate revolving commitment under the ABL Credit Facility is \$50.0 million and the facility matures on April 1, 2025.

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

The ABL Credit Facility is subject to a borrowing base which is calculated based on a formula referencing the Company's eligible accounts receivables. On June 30, 2022, the borrowing base was \$24.9 million and the outstanding revolver loan balance was \$7.0 million, classified as long-term debt on the condensed consolidated balance sheet.

#### Payments of Debt Obligations due by Period

As of June 30, 2022, the schedule of the repayment requirements of long-term debt is as follows:

Fiscal Year	Principal Amount of Long-term Debt
Remainder of 2022	\$ 4,194
2023	20,031
2024	24,533
2025	96,758
2026	3,367
Thereafter	15,211
Total	\$ 164,094

#### NOTE 12—COMMITMENTS AND CONTINGENCIES

#### Litigation

Liabilities for loss contingencies arising from claims, assessments, litigation, fines, and penalties, and other sources are recorded when it is probable that a liability has been incurred and the amount can be reasonably estimated. Legal costs incurred in connection with loss contingencies are expensed as incurred.

#### **Purchase Commitments**

During 2021, the Company entered into an Equipment Purchase and Sale Agreement to purchase equipment. The purchase commitments under this agreement are reflected in the table below. The Company intends to fund the commitments due in the next twelve months under the Equipment Purchase and Sale Agreement through additional financing transactions and cash on hand.

Additionally, during the first quarter of 2022, the Company placed orders for other equipment which will be delivered throughout 2022, related to the buildout of its new Nyx Clean Fleets®. Under the terms of the purchase orders, the Company is subject to a penalty fee for any equipment cancelled prior to delivery. As of June 30, 2022, total cost of equipment not yet received by the Company under the purchase orders was \$11.8 million. While the Company intends to take receipt of the remaining deliveries its minimum contractual commitment included in the table below was \$1.8 million, which represents the penalty fee in the event the Company cancels the equipment deliveries after June 30, 2022.

As of June 30, 2022, future minimum purchase commitments for equipment are as follows:

Fiscal Year	
Remainder of 2022	\$ 16,266
2023	16,400
Total	\$ 32,666

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

#### Self-insurance

The Company established a self-insured plan for employees' healthcare benefits except for losses in excess of varying threshold amounts. The Company charges to expense all actual claims made during each reporting period, as well as an estimate of claims incurred, but not yet reported. Estimated claims incurred, but not reported as of June 30, 2022 and December 31, 2021 was \$0.2 million and \$0.3 million, respectively, and was reported as accrued expenses in the condensed consolidated balance sheets. The Company believes that the liabilities recorded are appropriate based on the known facts and circumstances and does not expect further losses materially in excess of the amounts already accrued for existing claims.

#### NOTE 13—MEZZANINE EQUITY

#### Series A Redeemable Convertible Preferred Stock

The following table summarizes the Company's Series A Redeemable Convertible Preferred Stock, par value \$0.0001 per share ("Series A preferred stock") activities for the six months ended June 30, 2022:

	Shares	Amount
Balance at December 31, 2021	19,610	\$ 23,866
Accrued Series A preferred stock dividends		2,226
Balance at June 30, 2022	19,610	\$ 26,092

In June 2021, the Company exchanged 30,390 shares of Series A preferred stock for the Exchange Notes (discussed in "Note 10—Convertible Senior Notes"). Accordingly, the Company recorded a reduction of \$33.7 million in the carrying value of the Series A preferred stock during the three months ended June 30, 2021. Concurrent with the issuance of the Exchange Notes, the Company also received from such holders of the Series A preferred stock total cash proceeds of \$39.0 million in consideration for additional Cash Notes. In connection with the extinguishment of the Series A preferred stock, the Company initially recorded the Convertible Senior Notes issued to such holders at a total fair value of \$63.8 million. The difference of \$8.9 million between the fair value of the Convertible Senior Notes issued and the carrying amount of \$72.7 million of consideration received was recorded in additional paid in capital as a return from the Series A preferred holders for the three months ended June 30, 2021.

As of June 30, 2022, the Series A preferred stock outstanding were convertible into 210,605 shares of Class A common stock, and dividends accrued and outstanding with respect to the Series A preferred stock were \$9.9 million and reflected in the carrying value of Series A preferred stock.

## Series B Redeemable Convertible Preferred Stock

On April 1, 2020, the Company issued Series B Redeemable Convertible Preferred Stock, par value of \$0.0001 per share ("Series B preferred stock") to certain institutional investors.

During the three and six months ended June 30, 2021, 250 shares and 1,012 shares of Series B preferred stock and related accrued dividends, respectively, were converted into 44,167 shares and 174,919 shares of Class A common stock, respectively, pursuant to the certificate of designations authorizing and establishing the rights, preferences, and privileges of the Series B preferred stock. Accordingly, the Company recorded a reduction of \$0.3 million and \$1.1 million in the carrying value of the Series B preferred stock during the three and six months ended June 30, 2021, respectively.

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

As of June 30, 2022 and December 31, 2021, there were no shares of Series B preferred stock outstanding.

#### NOTE 14—STOCKHOLDERS' EQUITY

#### Preferred Stock

The Company is authorized to issue 10,000,000 shares of preferred stock with a par value of \$0.0001 per share with such designation, rights and preferences as may be determined from time to time by the Company's Board of Directors. See "Note 13—Mezzanine Equity" for disclosure regarding preferred stock issued and outstanding.

#### Class A Common Stock

The Company is authorized to issue 400,000,000 shares of Class A common stock with a par value of \$0.0001 per share. As of June 30, 2022 and December 31, 2021, there were 12,827,306 and 8,858,161 shares of Class A common stock issued and outstanding, respectively. As of June 30, 2022, 47,620 outstanding shares of Class A common stock were subject to cancellation on November 9, 2024, unless the closing price per share of the Class A common stock has equaled or exceeded \$252.00 for any 20 trading days within any 30-trading day period, and 29,033 outstanding shares of Class A common stock were subject to the same cancellation provision, but at a closing price per share of \$283.50.

ATM Agreements. On June 26, 2020, the Company entered into an Equity Distribution Agreement (the "2020 ATM Agreement") with Piper Sandler & Co. relating to the Company's shares of Class A common stock. In accordance with the terms of the 2020 ATM Agreement, the Company offered and sold shares of its Class A common stock over a period of time pursuant to an "at-the-market" offering program. The 2020 ATM Agreement expired on April 26, 2022.

On April 26, 2022, the Company entered into a new Equity Distribution Agreement (the "2022 ATM Agreement") with Piper Sandler & Co. relating to the Company's shares of Class A common stock to replace the 2020 ATM Agreement. In accordance with the terms of the 2022 ATM Agreement, the Company may offer and sell shares of our Class A common stock over a period of time pursuant to an "at-the-market" offering program. Under the 2022 ATM Agreement, the Company will pay Piper Sandler an aggregate commission of up to 3% of the gross sales price per share of Class A common stock sold under the 2022 ATM Agreement in an amount of up to \$50.0 million. No shares may be sold under the 2022 ATM Agreement until the registration statement filed on April 26, 2022 has been declared effective by the SEC.

The following table presents information with respect to shares of Class A common stock sold under the 2020 ATM Agreement during the periods indicated:

	Three M	Ionths Ended		
	J	une 30,	Six Months En	ded June 30,
	2022	2021	2022	2021
Shares of Class A common stock sold		113,412	1,627,991	714,586
Total net proceeds	\$ —	\$ 2,893	\$ 21,282	\$ 13,562
Commission paid	\$ —	\$ 89	\$ 658	\$ 419

Registered Direct Offering. On March 11, 2022, the Company completed a registered direct offering of 2,363,396 shares of Class A common stock and issuance of 14,180,375 RDO Investor Warrants for gross proceeds of \$25.0 million, before deducting placement agent fees and other offering expenses. The Company

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

issued 992,626 Placement Agent Warrants as partial compensation for the placement agent's services in connection with the registered direct offering on March 11, 2022. The Company used the net proceeds for working capital purposes, including the funding of certain capital expenditures.

The proceeds received were allocated as \$17.1 million and \$7.9 million to the Class A common stock and RDO Investor Warrants, respectively, based on the relative fair value at issuance. The fair value of the Class A common stock issued in the registered direct offering was \$16.3 million, calculated based on the closing price of the Class A common stock on the date of issuance. The fair value of the RDO Investor Warrants and Placement Agent Warrants at issuance was \$7.5 million and \$0.5 million respectively, calculated using the Black-Scholes option pricing model. The following assumptions were used to calculate the fair value for the RDO Investor Warrants and Placement Agent Warrants at issuance:

	RDO Investor Warrants	Placement Agent Warrants
Exercise price	\$10.578	\$13.2228
Contractual term	3.5 years	3.5 years
Volatility rate	80.0%	80.0%
Risk-free interest rate	1.9%	1.9%
Expected dividend rate	0%	0%

#### Class B Common Stock

The Company is authorized to issue 20,000,000 shares of Class B common stock with a par value of \$0.0001 per share. The shares of Class B common stock are non-economic; however, holders are entitled to one vote per share. Each share of Class B common stock, together with one unit of USWS Holdings, is exchangeable for one share of Class A common stock or, at the Company's election, the cash equivalent to the market value of one share of Class A common stock.

During the six months ended June 30, 2021, 2,302,936 shares of Class B common stock were converted into 109,664 shares of Class A common stock, which has been adjusted to reflect the reverse stock split. As of June 30, 2022 and December 31, 2021, there were no shares of Class B common stock issued and outstanding.

#### Warrants

As of June 30, 2022, the Company has issued and outstanding Term C Loan Warrants, RDO Investor Warrants and Placement Agent Warrants, which are recognized as equity. See "Note 9—Warrants" for disclosure regarding the Term C Loan Warrants, RDO Investor Warrants and Placement Agent Warrants.

## Noncontrolling Interest

During the first quarter of 2021, the remaining noncontrolling interest holders of USWS Holdings exchanged all of their respective shares for the Company's Class A common stock. Accordingly, USWS Holdings became the Company's wholly owned subsidiary as of March 31, 2021.

## NOTE 15—EARNINGS (LOSS) PER SHARE

Basic earnings (loss) per share is computed by dividing income (loss) available to common stockholders by the weighted average number of common shares outstanding during the period. Diluted earnings per share is

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

computed in the same manner as basic earnings per share except that the denominator is increased to include the number of additional Class A common shares that could have been outstanding assuming the exercise of stock options and warrants, conversion of Series A and Series B preferred stock, conversion of Class B common stock, vesting of restricted shares of Class A common stock, conversion of Convertible Senior Notes and issuance of Class A common stock associated with the deferred stock units and certain performance awards.

Basic and diluted net income (loss) per share excludes the income (loss) attributable to and shares associated with the 76,653 shares of Class A common stock that are subject to cancellation on November 9, 2024 if certain market conditions have not been met. The Company has included in the calculation accrued dividends on Series A and Series B preferred stock and related deemed and imputed dividends.

The following table sets forth the calculation of basic and diluted earnings (loss) per share for the periods indicated based on the weighted average number of shares of Class A common stock outstanding:

	Three Months Ended June 30,			
	2022	2021	2022	2021
Basic net income (loss) per share				
Numerator:				
Net loss attributable to U.S. Well Services, Inc.	\$ (9,335)	\$ (17,716)	\$ (35,080)	\$ (38,282)
Net loss attributable to cancellable Class A common				
stock	56	316	235	721
Basic net loss attributable to U.S. Well Services, Inc.				
shareholders	(9,279)	(17,400)	(34,845)	(37,561)
Dividends accrued on Series A preferred stock	(1,135)	(1,998)	(2,226)	(3,811)
Dividends accrued on Series B preferred stock	_	(811)	_	(1,522)
Deemed and imputed dividends on Series A preferred				
stock	_	(286)	_	(750)
Deemed and imputed dividends on Series B preferred				
stock	_	(1,501)	_	(5,669)
Exchange of Series A preferred stock for Convertible				
Senior Notes		8,936		8,936
Basic net loss attributable to U.S. Well Services, Inc.				
Class A common shareholders	\$ (10,414)	\$ (13,060)	\$ (37,071)	\$ (40,377)
Denominator:	=====	<del></del>	=====	====
Weighted average shares outstanding				
· · · · · · · · · · · · · · · · · · ·	12,796,193	4,295,379	11,423,931	4,067,616
Cancellable Class A common stock	(76,653)	(76,653)	(76,653)	(76,653)
Basic and diluted weighted average shares outstanding	12,719,540	4,218,726	11,347,278	3,990,963
Basic and diluted net loss per share attributable to	<del></del> _	<del></del> _	<del> </del>	<del></del>
Class A common shareholders	\$ (0.82)	\$ (3.10)	\$ (3.27)	\$ (10.12)
- · · · · · · · · · · · · · · · · · · ·		. (0110)		

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

A summary of securities excluded from the computation of diluted earnings per share is presented below for the applicable periods:

	Three Months Ended June 30,		Six Months End June 30,	
	2022	2021	2022	2021
Dilutive earnings per share:				
Anti-dilutive restricted stock	31,113	51,393	31,113	51,393
Anti-dilutive restricted stock units	190,419	_	190,419	_
Anti-dilutive stock options	_	41,776	_	41,776
Anti-dilutive deferred stock units	389,012	424,375	389,012	424,375
Anti-dilutive shares from Pool B Awards	598,613	482,130	598,613	482,130
Anti-dilutive warrants	5,802,664	729,384	5,802,664	729,384
Anti-dilutive Series A preferred stock convertible into Class A common				
stock	210,605	180,109	210,605	180,109
Anti-dilutive Series B preferred stock convertible into Class A common				
stock	_	3,786,314	_	3,786,314
Anti-dilutive Convertible Senior Notes convertible into Class A				
common stock	5,059,728	3,848,685	5,059,728	3,848,685
Potentially dilutive securities excluded as anti-dilutive	12,282,154	9,544,166	12,282,154	9,544,166

#### NOTE 16—SHARE-BASED COMPENSATION

Share-based compensation expense consisted of the following:

		Three Months Ended June 30,		s Ended 30,
	2022	2021	2022	2021
Restricted stock <sup>(1)</sup>	\$(2,336)	\$ 876	\$(1,600)	\$1,701
Restricted stock units	211	_	211	_
Stock options	(40)	216	173	430
Deferred stock units	401	246	731	492
Pool A Awards	858	395	1,217	481
Pool B Awards	491	277	881	557
Total	\$ (415)(2)	\$2,010(3)	\$ 1,613(4)	\$3,661(5)

- (1) During the three months ended June 30, 2022, 17,131 shares of unvested restricted stock were forfeited, resulting in the reversal of \$3.1 million share-based compensation expense.
- (2) For the three months ended June 30, 2022, \$552 was presented as cost of services and \$(967) was presented as selling, general and administrative expenses in the condensed consolidated statement of operations.
- (3) For the three months ended June 30, 2021, \$276 was presented as cost of services and \$1,734 was presented as selling, general and administrative expenses in the condensed consolidated statement of operations.
- (4) For the six months ended June 30, 2022, \$849 was presented as cost of services and \$764 was presented as selling, general and administrative expenses in the condensed consolidated statement of operations.
- (5) For the six months ended June 30, 2021, \$474 was presented as cost of services and \$3,187 was presented as selling, general and administrative expenses in the condensed consolidated statement of operations.

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

#### Restricted Stock

Pursuant to the Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan (as amended, the "LTIP"), the Company grants shares of restricted Class A common stock ("restricted stock") to certain employees and directors. Restricted stock is subject to restrictions on transfer and is generally subject to a risk of forfeiture if the award recipient is no longer an employee or director of the Company prior to the lapse of the restriction. Restricted stock granted to employees generally vests over four years in equal installments each year on the anniversary of the grant date and grants to directors generally vest in full after one year. The grant date fair value of the restricted stock is determined using the closing price of the Company's Class A common stock on the grant date.

The following table summarizes the restricted stock activity for the six months ended June 30, 2022:

	Shares	Averag Fair	Veighted- ge Grant-Date · Value per Share
Outstanding at December 31, 2021	64,842	\$	147.34
Granted	_		_
Vested	(15,870)		187.11
Forfeited	(17,859)		183.30
Outstanding at June 30, 2022	31,113	\$	106.42

As of June 30, 2022, the total unrecognized compensation cost related to restricted stock was \$2.1 million which is expected to be recognized over a weighted-average period of 0.72 years.

#### Restricted Stock Units

On April 30, 2022, the Company granted 190,419 restricted stock units (the "RSUs") to Joel Broussard, Chairman of the Board. The RSUs vestone-half six months from the grant date and the remaining one-half 18 months from the grant date, subject to Mr. Broussard's continuous service through such vesting date. Each RSU represents the right to receive one share of the Company's Class A common stock pending the Company receiving stockholder approval of certain amendments to the LTIP as may be required in order to permit the transactions contemplated by the RSU award. In the event that the Company does not receive stockholder approval, the RSUs will be settled in cash. The grant date fair value of the RSUs is determined using the closing price of the Company's Class A common stock on the grant date and are classified as equity. As of June 30, 2022, the total unrecognized compensation cost related to the RSUs was \$0.7 million which is expected to be recognized over a weighted-average period of 1.33 years.

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

#### Stock Options

The following table summarizes the stock option activity for the six months ended June 30, 2022:

	Shares	Exe	eighted- Average rcise Price er Share	Weighted- Average Remaining Contractual Life (in years)
Outstanding at December 31, 2021	41,776	\$	187.11	4.21
Exercised	_		_	_
Forfeited	(41,776)		187.11	3.71
Outstanding at June 30, 2022		\$	_	_

In May 2022, all outstanding stock options were forfeited for no consideration.

#### Deferred Stock Units

The Company awards deferred stock units ("DSUs") to certain key employees of the Company pursuant to the LTIP. Each DSU represents the right to receive one share of the Company's Class A common stock. DSUs generally vest over three years in equal installments each year on the anniversary of the vesting effective date, subject to the grantee's continuous service through each vesting period but settlement does not occur until the fifth anniversary of the grant date. The grant date fair value of the DSU is determined using the closing price of the Company's Class A common stock on the grant date.

The following table summarizes the DSUs activity for the six months ended June 30, 2022:

	Units	Average Fair '	ighted- Grant Date Value per Unit
Outstanding at December 31, 2021	342,090	\$	8.96
Granted	210,154		4.93
Vested	(159,807)		8.38
Forfeited	(3,425)		6.93
Outstanding at June 30, 2022	389,012	\$	7.04

As of June 30, 2022, the total unrecognized compensation cost related to DSUs was \$2.0 million which is expected to be recognized over a weighted average period of 1.89 years.

#### Pool A Performance Awards

The Company grants Pool A Performance Awards ("Pool A Awards") to certain key employees of the Company. Each Pool A Award represents the right to receive, at the Company's election, a fixed monetary amount either in cash or a variable number of shares of the Company's Class A common stock based on its closing share price on the date of settlement. The Pool A Awards vest in full one year on the anniversary of the vesting effective date specified in the applicable award agreement but settlement does not occur until the fifth anniversary of the grant date.

The Company accounts for the Pool A Awards under liability accounting as a result of the fixed monetary amount that could be settled either in cash or a variable number of shares of the Company's Class A common stock. Since

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

the settlement will not occur until the fifth anniversary of the grant date, the Company considers the delayed settlement as a post-vesting restriction which would impact the determination of grant-date fair value of the award.

During the third quarter of 2021, the Company granted Pool A Awards that fully vested on January 1, 2022. On April 30, 2022, the Company awarded Pool A Awards to its Chief Executive Officer and its Chief Financial Officer with a grant date fair value of \$1.3 million that vest in full one year from the grant date. In May 2022, the Company granted Pool A Awards with a grant date fair value of \$1.7 million that will vest in full on January 1, 2023.

As of June 30, 2022, the fair value of the Pool A Awards liabilities were remeasured to \$7.9 million and is presented as other long-term liabilities in the condensed consolidated balance sheet. The fair value was estimated using a risk-adjusted discount rate reflecting the weighted-average cost of capital of similarly traded public companies.

As of June 30, 2022, the total unrecognized compensation cost related to the Pool A Awards was \$9.3 million, which is expected to be recognized over a weighted average period of 4.50 years.

#### Pool B Performance Awards

The Company grants Pool B Performance Awards ("Pool B Awards") to certain key employees of the Company. Each Pool B Award represents the right to receive, at the Company's election, either a cash payment calculated in accordance with the award agreement, or a fixed number of shares of the Company's Class A common stock. The Pool B Awards vest over three years in equal installments each year on the anniversary of the vesting effective date specified in the applicable award agreement, subject to the grantee's continuous services through each vesting period but settlement does not occur until the fifth anniversary of the grant date. The grant date fair value of the Pool B Awards is determined using the closing price of the Company's Class A common stock on the grant date.

The following table summarizes the Pool B Awards activity for the six months ended June 30, 2022:

Fai	r Value
Outstanding at December 31, 2021	3,672
Granted	606
Vested	(1,586)
Forfeited	(27)
Outstanding at June 30, 2022	2,665

As of June 30, 2022, the total unrecognized compensation cost related to Pool B Awards was \$1.8 million, which is expected to be recognized over a weighted average period of 1.49 years.

## NOTE 17—RELATED PARTY TRANSACTIONS

Crestview III USWS TE, LLC and Crestview III USWS, L.P. and its affiliates (collectively, "Crestview Partners") are part of an affiliate group which have an ownership interest in the Company of greater than 10% and is entitled to designate for nomination by the Company the election of two directors to serve on the Company's Board of Directors.

David Matlin is a member of the Company's Board of Directors.

#### U.S. WELL SERVICES, INC.

#### Notes to Unaudited Condensed Consolidated Financial Statements

(in thousands, except shares and per share amounts, or where otherwise noted)

#### Convertible Senior Notes

On June 24, 2021, Crestview Partners purchased \$40.0 million of Convertible Senior Notes that are convertible into shares of the Company's Class A common stock for consideration of \$20.0 million in cash and in exchange for 15,588 shares of the Company's Series A preferred stock. As of June 30, 2022 and December 31, 2021, the Convertible Senior Notes held by Crestview Partners have a carrying value of \$39.6 million and \$35.9 million, respectively, which is included in convertible senior notes on the condensed consolidated balance sheets. See "Note 10—Convertible Senior Notes" for additional disclosure regarding the Cash Notes and Exchange Notes.

#### Term C Loan and Term C Loan Warrants

On February 28, 2022, Crestview Partners contributed \$10.0 million in Term C Loans to the Company. As of June 30, 2022, the carrying value of the Term C Loan owed to Crestview Partners was \$7.5 million, which is included in long-term debt on the condensed consolidated balance sheet. In connection with the entry into the Term C Loan, Crestview received 6,976,744 February 2022 Warrants.

On March 1, 2022, David Matlin contributed \$1.0 million in Term C Loans to the Company. As of June 30, 2022, the carrying value of the Term C Loan owed to David Matlin was \$0.8 million, which is included in long-term debt on the condensed consolidated balance sheet. In connection with the entry into the Term C Loan, David Matlin received 697,674 March 2022 Warrants.

See "Note 9—Warrants" and "Note 11—Debt" for additional disclosure regarding the Term C Loan and Term C Loan Warrants.

#### NOTE 18—SUBSEQUENT EVENTS

#### **Equipment Financing**

On July 18, 2022, the Company and USWS LLC entered into a \$12.5 million promissory note (the "Equify Note") with Equify Financial, LLC, as lender, which matures on August 1, 2027. The Company intends to use the proceeds to fund capital expenditures related to its newbuild Nyx Clean Fleets.

The Equify Note is payable in equal monthly installments of principal in the amount of \$0.2 million, together with all accrued and unpaid interest on the outstanding principal balance of the Equify Note, commencing on September 1, 2022 and continuing thereafter until the maturity date, and bears interest at a rate per annum equal to the lesser of (i) the sum of (A) "prime rate" as published in the Wall Street Journal from time to time plus (B) 9.25%, and (ii) the maximum amount of interest allowed by applicable law. The Equify Note is secured by specific equipment collateral.

The Wilks Parties hold a controlling interest in Equify Financial, LLC.

#### Reverse Stock Split

On August 4, 2022, the Company effected a 1-for-6 reverse split of its Class A common stock. All owners of record as of August 4, 2022 received one issued and outstanding share of the Company's Class A common stock in exchange for six outstanding shares of the Company's Class A common stock. See "Note 2—Significant Accounting Policies" for additional disclosure regarding the reverse stock split.

#### Annex A

[Execution]

## AGREEMENT AND PLAN OF MERGER

among

U.S. WELL SERVICES, INC.,

PROFRAC HOLDING CORP.,

and

## THUNDERCLAP MERGER SUB I, INC.

Dated as of June 21, 2022

## TABLE OF CONTENTS

		Page
ARTICLE I DEFIN	ITIONS	A-3
Section 1.1	Definitions	A-3
Section 1.2	Table of Definitions	A-14
Section 1.3	Other Definitional and Interpretative Provisions	A-15
ARTICLE II THE	MERGER	A-16
Section 2.1	The Merger	A-16
Section 2.2	The Closing	A-16
Section 2.3	Effective Times	A-16
Section 2.4	Surviving Corporation Matters	A-16
ARTICLE III EFFE	ECT ON THE CAPITAL STOCK; EXCHANGE OF CERTIFICATES	A-17
Section 3.1	Series A Preferred Stock, Equity Linked Convertible Notes and Warrants	A-17
Section 3.2	Effect of the Merger on Capital Stock	A-17
Section 3.3	Certain Adjustments	A-18
Section 3.4	Fractional Shares	A-18
Section 3.5	Appraisal Rights	A-18
Section 3.6	Exchange of Company Stock	A-18
Section 3.7	Further Assurances	A-21
Section 3.8	Company Stock-Based Awards	A-21
Section 3.9	Warrants	A-22
ARTICLE IV REPI	RESENTATIONS AND WARRANTIES OF THE COMPANY	A-22
Section 4.1	Corporate Existence and Power	A-22
Section 4.2	Corporate Authorization	A-23
Section 4.3	Governmental Authorization	A-23
Section 4.4	Non-contravention	A-24
Section 4.5	Capitalization	A-24
Section 4.6	Subsidiaries	A-25
Section 4.7	SEC Filings and the Sarbanes-Oxley Act	A-26
Section 4.8	Financial Statements	A-27
Section 4.9	Information Supplied	A-28
Section 4.10	Absence of Certain Changes	A-28
Section 4.11	No Undisclosed Material Liabilities	A-28
Section 4.12	Compliance with Laws and Court Orders; Governmental Authorizations	A-28
Section 4.13	Litigation	A-29
Section 4.14	Properties	A-29
Section 4.15	Intellectual Property	A-30
Section 4.16	Data Privacy and Security	A-34
Section 4.17	Taxes	A-34
Section 4.18	Employees and Employee Benefit Plans	A-35
Section 4.19	Environmental Matters	A-38
Section 4.20	Material Contracts	A-39
Section 4.21	Finders' Fees	A-41
Section 4.22	Opinion of Financial Advisor	A-41
Section 4.23	Antitakeover Statutes	A-41
Section 4.24	Insurance	A-41
Section 4.25	No Additional Representations	A-41

ARTICLE V REPR	ESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB INC.	A-42
Section 5.1	Corporate Existence and Power	A-42
Section 5.2	Corporate Authorization	A-42
Section 5.3	Governmental Authorization	A-42
Section 5.4	Non-contravention	A-43
Section 5.5	<u>Capitalization</u>	A-43
Section 5.6	SEC Filings and the Sarbanes-Oxley Act	A-44
Section 5.7	Financial Statements	A-45
Section 5.8	<u>Information Supplied</u>	A-46
Section 5.9	Absence of Certain Changes	A-46
Section 5.10	Solvency	A-46
Section 5.11	No Undisclosed Material Liabilities	A-46
Section 5.12	Compliance with Laws and Court Orders; Governmental Authorization	A-47
Section 5.13	Litigation	A-47
Section 5.14	Opinion of Financial Advisor	A-47
Section 5.15	No Additional Representations	A-47
	ENANTS OF THE COMPANY	A-48
Section 6.1	Conduct of the Company	A-48
Section 6.2	Modifications of Certain Outstanding Securities and Awards	A-51
Section 6.3	Information and Consultation	A-51
Section 6.4	No Solicitation; Other Offers	A-51
Section 6.5	Company Stockholders' Meeting	A-54
ARTICLE VII COVENANTS OF PARENT AND MERGER SUB INC.		A-54
Section 7.1	Conduct of Parent	A-54
Section 7.2	Obligations of Merger Sub Inc	A-55
Section 7.3	Approval by Sole Stockholder of Merger Sub Inc.	A-55
Section 7.4	Director and Officer Indemnification	A-55
Section 7.5	Stock Exchange Listing	A-56
Section 7.6	Employee Matters	A-56
	VENANTS OF PARENT AND THE COMPANY	A-57
Section 8.1	<u>Efforts</u>	A-57
Section 8.2	Financing Cooperation	A-59
Section 8.3	Registration Statement; Information Statement; Proxy Statement	A-60
Section 8.4	Public Announcements	A-61
Section 8.5	Notices of Certain Events	A-61
Section 8.6	Access to Information	A-62
Section 8.7	Section 16 Matters	A-62
Section 8.8	Stockholder Litigation	A-63
	DITIONS TO THE MERGER	A-63
Section 9.1	Conditions to Obligations of Each Party	A-63
Section 9.2	Conditions to Obligations of Parent and Merger Sub Inc.	A-63
Section 9.3	Conditions to Obligations of the Company	A-64
Section 9.4	Frustration of Closing Conditions	A-65
ARTICLE X TERM		A-65
Section 10.1	Termination PEG 4 GT is a second control of the con	A-65
Section 10.2	Effect of Termination	A-66
Section 10.3	Termination Fees; Expense Reimbursement	A-66

ARTICLE XI MISCELLANEOUS		A-68
Section 11.1	No Survival of Representations and Warranties	A-68
Section 11.2	Amendment and Modification	A-68
Section 11.3	Extension; Waiver	A-68
Section 11.4	<u>Expenses</u>	A-68
Section 11.5	Disclosure Letter References	A-68
Section 11.6	<u>Notices</u>	A-69
Section 11.7	<u>Counterparts</u>	A-70
Section 11.8	Entire Agreement; No Third Party Beneficiaries	A-70
Section 11.9	Severability	A-71
Section 11.10	Assignment	A-71
Section 11.11	Governing Law	A-71
Section 11.12	Enforcement; Exclusive Jurisdiction	A-71
Section 11.13	WAIVER OF JURY TRIAL	A-72
Section 11.14	Specific Performance	A-72
Errhibita		

Exhibits Exhibit A Exhibit B

Convertible Note Amendments COD Amendment Award Amendments Exhibit C

#### AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER (this "Agreement"), dated as of June 21, 2022, among U.S. Well Services, Inc., a Delaware corporation (the "Company"), ProFrac Holding Corp., a Delaware corporation ("Parent"), and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of Parent ("Merger Sub Inc."). Parent, Merger Sub Inc. and the Company are referred to individually as a "Party" and collectively as "Parties".

#### WITNESSETH:

WHEREAS, Merger Sub Inc. is an indirect subsidiary of Parent;

WHEREAS, it is proposed that, upon the terms and subject to the conditions set forth herein, the Parties will effect the acquisition of the Company by Parent through the merger of Merger Sub Inc. with and into the Company (the "Merger"), with the Company surviving the Merger as the surviving corporation and an indirect Subsidiary of Parent;

WHEREAS, the Special Committee (the "Company Special Committee") of the Board of Directors of the Company (the "Company Board") has unanimously (i) approved and declared advisable the execution and delivery by the Company of this Agreement, the performance by the Company of its covenants and agreements contained herein and the consummation of the transactions contemplated by this Agreement, including the Merger, upon the terms, and subject to the conditions, contained herein and (ii) resolved, subject to the terms and conditions set forth in this Agreement, to recommend that the Company Board approve this Agreement and the consummation of the transactions contemplated by this Agreement, including the Merger;

WHEREAS, upon the receipt and approval of the Company Special Committee, the Company Board has unanimously (i) determined that this Agreement and the terms of the transactions contemplated hereby, including the Merger, are fair to, and in the best interests of, the Company and its stockholders, (ii) determined that it is in the best interests of the Company and its stockholders to enter into, and has declared advisable, this Agreement, (iii) approved and declared advisable the execution and delivery by the Company of this Agreement, the performance by the Company of its covenants and agreements contained herein and the consummation of the transactions contemplated by this Agreement, including the Merger, upon the terms, and subject to the conditions, contained herein, (iv) directed that this Agreement be submitted to the Company's stockholders for their approval and adoption and (v) resolved, subject to the terms and conditions set forth in this Agreement, to recommend that the Company's stockholders approve and adopt this Agreement and the consummation of the transactions contemplated by this Agreement, including the Merger:

WHEREAS, the Special Committee (the "Parent Special Committee") of the Board of Directors of Parent (the "Parent Board") has unanimously (i) determined that this Agreement and the transactions contemplated hereby are fair to, and in the best interests of, Parent and the Parent Unaffiliated Stockholders, (ii) approved and declared advisable this Agreement and the transactions contemplated hereby, (iii) resolved, subject to the terms and conditions set forth in this Agreement, to recommend that the Parent Board (x) determine that this Agreement and the transactions contemplated hereby are fair to, and in the best interests of, Parent and the Parent Unaffiliated Stockholders and (y) approve and declare advisable this Agreement and the transactions contemplated hereby, and (iv) resolved, subject to the terms and conditions set forth in this Agreement, to recommend that the Parent Board (x) direct that the Parent Stock Issuance be submitted to the Parent's stockholders for their approval and (y) recommend that the Parent stockholders approve the Parent Stock Issuance;

WHEREAS, upon the receipt and recommendation of the Parent Special Committee, the Parent Board has unanimously (i) approved and declared advisable this Agreement and the transactions contemplated hereby,

(ii) approved and declared advisable the execution and delivery by Parent of this Agreement, the performance by Parent of its covenants and agreements contained herein and the consummation of the transactions contemplated by this Agreement upon the terms, and subject to the conditions, contained herein, (iii) determined that this Agreement and the transactions contemplated hereby are fair to, and in the best interests of, Parent and the Parent Unaffiliated Stockholders and (iv) resolved, subject to the terms and conditions set forth in this Agreement, to recommend that Parent's stockholders approve the Parent Stock Issuance;

WHEREAS, the Board of Directors of Merger Sub Inc., has unanimously (i) approved and declared advisable this Agreement and the transactions contemplated hereby, including the Merger, (ii) approved and declared advisable the execution and delivery by Merger Sub Inc. of this Agreement, the performance by Merger Sub Inc. of its covenants and agreements contained herein and the consummation of the transactions contemplated by this Agreement, including the Merger, upon the terms, and subject to the conditions, contained herein, (iii) determined that this Agreement and the transactions contemplated hereby, including the Merger, are fair to, and in the best interests of, Merger Sub Inc. and its sole stockholder, (iv) directed that this Agreement be submitted to Parent, in its capacity as the sole stockholder of Merger Sub Inc., for its adoption and approval and (v) resolved, subject to the terms and conditions set forth in this Agreement, to recommend that Parent, in its capacity as the sole stockholder of Merger Sub Inc., approve and adopt this Agreement and the transactions contemplated by this Agreement, including the Merger;

WHEREAS, ProFrac Holdings II, LLC as the sole stockholder of Merger Sub Inc., will adopt and approve this Agreement and the transactions contemplated hereby, including the Merger, immediately following its execution;

WHEREAS, Parent has obtained the Parent Stockholder Approval (as defined below), in accordance with Nasdaq listing rules, by written consent;

WHEREAS, concurrently with the execution and delivery of this Agreement, and as a condition and inducement to the willingness of Parent and Merger Sub Inc. to enter into this Agreement, certain stockholders of the Company are entering into a support agreement (collectively, the "Support Agreements") pursuant to which such stockholders, among other things, will, upon the terms and subject to the conditions contained therein, (i) agree to support and vote in favor of this Agreement, the Merger and the other transactions contemplated hereby, (ii) take (and refrain from taking) certain other actions in connection with the transactions contemplated by this Agreement and (iii) to the extent applicable, deliver a Conversion Notice (as defined in the COD Amendments) to convert their shares of Series A Preferred Stock at the Merger Conversion Ratio (as defined in the COD Amendment);

WHEREAS, concurrently with the execution and delivery of this Agreement, the holders of the February Term C Loan Warrants and the March Term C Loan Warrants are entering into a Warrant Purchase Agreement pursuant to which, immediately prior to the Effective Time and conditioned upon the Closing, such warrant holders will sell all of the February Term C Loan Warrants and the March Term C Loan Warrants held by such holders to Parent or a subsidiary of Parent in exchange for cash consideration (the "Warrant Sale");

WHEREAS, Parent, the Company and Merger Sub Inc. desire to make certain representations, warranties, covenants and agreements specified herein in connection with this Agreement and also to prescribe various conditions to the consummation of the Merger; and

NOW, THEREFORE, in consideration of the foregoing and the representations, warranties, covenants and agreements contained herein, the Parties agree as follows.

#### ARTICLE I DEFINITIONS

- Section 1.1 Definitions. As used herein, the following terms have the following meanings:
  - "1933 Act" means the Securities Act of 1933, as amended.
  - "1934 Act' means the Securities Exchange Act of 1934, as amended.
- "Accumulated Cash Award Value" means, on any date with respect to a Pool A Performance Award or Pool B Performance Award, the Award Value, as defined in the applicable award agreement.
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls or is controlled by, or is under common control with such Person. The term "control" (including its correlative meanings "controlled" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies of a Person (whether through ownership of such Person's voting securities, or by Contract or otherwise).
- "Business Day" means any day that is not a Saturday, a Sunday or other day that (i) is a statutory holiday under the federal Laws of the United States or (ii) is otherwise a day on which banks in New York, New York are authorized or obligated by Law or executive order to close.
- "Certificate of Designations" means that certain Certificate of Designations of the Series A Preferred Stock, dated May 24, 2019, as it may be amended, restated or otherwise modified from time to time following the date hereof, including pursuant to the COD Amendment.
- "Claim" means any claim, demand, cause of action, suit, dispute, proceeding, arbitration, audit, hearing, investigation or inquiry (whether formal or informal).
  - "Closing Date" means the date of the Closing.
- "Collective Bargaining Agreements" mean any Contract, work rules or any side letter to which the Company or any Subsidiaries thereof is bound or that has been entered into between the Company or any Subsidiary thereof and any labor organization, union, works council, employee association, trade union or other similar employee representative body.
- "Company Acquisition Proposal" means any proposal, inquiry, indication of interest or offer (whether or not in writing) from any Person (other than Parent and its Subsidiaries or Affiliates) relating to or involving, whether in a single transaction or series of related transactions: (i) any direct or indirect acquisition, lease, exchange, license, transfer, disposition (including by way of liquidation or dissolution of the Company or any of its Subsidiaries) or purchase of any business, businesses or assets (including equity interests in Subsidiaries but excluding sales of assets in the ordinary course of business) of the Company or any of its Subsidiaries that constitute or account for twenty percent (20%) or more of the consolidated net revenues, net income or net assets of the Company and its Subsidiaries, taken as a whole; (ii) any merger, consolidation, amalgamation, share exchange, business combination, issuance of securities, sale of securities, reorganization, recapitalization, tender offer, exchange offer, liquidation, dissolution, extraordinary dividend, or similar transaction involving the Company and a Person or "group" (as defined in Section 13(d) of the 1934 Act) pursuant to which the stockholders of the Company immediately preceding such transaction hold less than eighty percent (80%) of the equity interests in the surviving or resulting entity of such transaction immediately following such transaction; or (iii) any combination of the foregoing.
- "Company Adverse Recommendation Change" means any of the following actions by the Company Board or any committee thereof: (i) withholding or withdrawing (or amending, modifying or qualifying in a

manner adverse to Parent) or proposing publicly to withhold or withdraw (or amend, modify or qualify in a manner adverse to Parent), the Company Board Recommendation, (ii) failing to make the Company Board Recommendation in the Proxy Statement, subject to the terms and conditions of this Agreement, (iii) approving, recommending, or otherwise declaring or endorsing publicly to be advisable or publicly proposing to approve, adopt or recommend to be advisable any (A) Company Acquisition Proposal or (B) Contract requiring the Company to abandon, terminate or fail to consummate the Merger or any other transaction contemplated by this Agreement or (iv) (x) after public announcement of any Company Acquisition Proposal or any material modification thereto, failing to issue a press release that expressly reaffirms, without qualification, the Company Board Recommendation, within six (6) Business Days following Parent's written request to do so or (y) following the commencement of any Company Acquisition Proposal that is a tender offer or exchange offer relating to the securities of the Company, (A) publicly be neutral or make any recommendation in connection with such tender offer or exchange offer other than a recommendation against such offer or a temporary "stop, look and listen" communication by the Company Board of the type contemplated by Rule 14d-9(f) under the 1934 Act or (B) fail to send to its stockholders a statement rejecting and recommending against any such tender offer or exchange offer within ten (10) Business Days of such commencement of such tender offer or exchange offer.

"Company Balance Sheet' means the unaudited consolidated balance sheet of the Company as of March 31, 2022 and the footnotes thereto set forth in the Company's quarterly report on Form 10-Q for the quarter ended March 31, 2022.

"Company Balance Sheet Date" means March 31, 2022.

"Company Disclosure Letter" means the disclosure letter delivered by the Company to Parent and Merger Sub Inc. in connection with, and upon the execution of, this Agreement.

"Company Equity Award" means the Company DSUs, the Company Restricted Shares, the Pool A Performance Awards and Pool B Performance Awards.

"Company Intellectual Property Rights" means all Owned Intellectual Property Rights and all Licensed Intellectual Property Rights.

"Company LTIP" means the Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan.

"Effect") that, individually or in the aggregate with all other Effects, has had or would reasonably be expected to have a material adverse effect on the business, results of operations or financial condition of the Company and its Subsidiaries, taken as a whole; provided, however, that a Company Material Adverse Effect shall not be deemed to include any Effect (alone or in combination) arising out of, relating to, or resulting from: (i) Effects generally affecting the economy, financial or securities markets, or political conditions; (ii) the execution and delivery or performance of this Agreement, or the announcement, consummation or pendency of the transactions contemplated by this Agreement or the identity of or any facts or circumstances relating to Parent or any of its Affiliates, including the impact of any of the foregoing on the relationships, contractual or otherwise, of the Company or any of its Subsidiaries with customers, suppliers, service providers, employees, Governmental Authorities or any other Persons (provided that this clause (ii) shall not apply to the representations and warranties that, by their terms, speak specifically of the consequences arising out of the execution or performance of this Agreement or the consummation of the transactions contemplated hereby); (iii) any changes in applicable Law or GAAP, in the interpretation of GAAP, in the accounting rules and regulations of the SEC, or other applicable accounting standards; (iv) acts of war or other military actions, terrorism or the escalation thereof; (v) earthquakes, hurricanes, tornadoes or other natural disasters; (vi) general conditions in the industry in which the Company and its Subsidiaries operate; (vii) any failure, in and of itself, by the Company to meet any internal or published projections, forecasts, estimates, or predictions in respect of

revenues, earnings, or other financial or operating metrics for any period (it being understood that any Effect underlying such failure may be deemed to constitute, or be taken into account in determining whether there has been or would reasonably be expected to become, a Company Material Adverse Effect, to the extent permitted by this definition and not otherwise excepted by another clause of this proviso); (viii) any change, in and of itself, in the market price or trading volume of the Company's securities (it being understood that any Effect underlying such change may be deemed to constitute, or be taken into account in determining whether there has been or would reasonably be expected to become, a Company Material Adverse Effect, to the extent permitted by this definition and not otherwise excepted by another clause of this proviso); (ix) any Claim or proceeding for appraisal of any shares of Company Stock pursuant to the DGCL in connection herewith, (x) any epidemic, pandemic or disease outbreak (including the COVID-19 pandemic) or the evolution of any COVID-19 Measures or other restrictions that relate to, or arise out of, any epidemic, pandemic or disease outbreak (including the COVID-19 pandemic) and any COVID-19 Responses, or (xi) the taking of any action (or the omission of any action) by the Company or any Subsidiary of the Company in accordance with the terms of this Agreement to the extent the taking of such action (or omission) is expressly required or contemplated by this Agreement or such action was taken at the written request of, or with the written consent of, Parent or Merger Sub Inc. (provided that this clause (ix) shall not apply to the representations and warranties that, by their terms, speak specifically of the consequences arising out of the execution or performance of this Agreement or the consummation of the transactions contemplated hereby); provided further, however, that any Effect referred to in clauses (i), (iii), (iv), (v), (vi), or (x) immediately above shall be taken into account in determining whether a Company Material Adverse Effect has occurred or would reasonably be expected to occur to the extent it has a disproportionate effect on the Company and its Subsidiaries, taken as a whole, compared to other participants in the industries in which the Company and its Subsidiaries conduct their businesses.

"Company Product" means all products or services that are being manufactured, marketed, distributed, sold or otherwise commercialized by or on behalf of the Company or any of its Subsidiaries.

"Company Stock" means the Class A common stock, par value \$0.0001 per share, of the Company.

"Company Stockholder Approval" means at the Company Stockholders' Meeting and in accordance with the Company's organizational documents: (i) the affirmative vote of the holders of a majority of the shares of Company Stock outstanding on the record date for the Company Stockholders' Meeting and entitled to vote thereon approving and adopting this Agreement and the transactions contemplated thereby, including the Merger, (ii) the affirmative vote of the holders of a majority of the total votes cast with respect to the proposal for the holders of Company Stock to approve the issuance of shares of Company Stock pursuant to the conversion of the Series A Preferred Stock and the Equity Linked Convertible Notes in accordance with Section 3.1(a) (including for the avoidance of doubt pursuant to the COD Amendment) and Section 3.1(b), respectively, under Nasdaq listing rules and without regard to any share caps or other limitations set forth in the Certificate of Designations or the Equity Linked Convertible Notes and (iii) the affirmative vote of the holders of a majority of the shares of Company Stock outstanding on the record date for the Company Stockholders' Meeting and entitled to vote thereon approving and adopting the COD Amendment.

"Company Technology" means all Owned Technology and all Licensed Technology.

"Company Unaffiliated Stockholders" means holders of Company Stock, excluding Parent and THRC Holdings, LP and any of their Affiliates.

"Competition Laws" means the Sherman Antitrust Act, as amended, the Clayton Antitrust Act, as amended, the HSR Act, the Federal Trade Commission Act, as amended, and all other Laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization, lessening of competition or restraint of trade.

"Confidentiality Agreement" means that certain letter agreement, dated as of March 4, 2021, by and between ProFrac Services, LLC and U.S. Well Services, LLC.

- "Contract" means any agreement, arrangement, contract, understanding, instrument, note, bond, mortgage, indenture, deed of trust, lease, license or other legally binding commitment, whether written or oral.
- "COVID-19 Measures" means any quarantine, "shelter in place," "stay at home," workforce reduction, social distancing, shut down, closure, sequester or any other applicable Law related to COVID-19 (or any other related variance or strain of COVID-19).
- "COVID-19 Responses" means any reasonable action taken or omitted to be taken in good faith that is reasonably determined to be necessary or prudent to be, or omitted to be, taken in response to COVID-19 (or any other related variance or strain of COVID-19) or any of the measures described in the definition of "COVID-19 Measures", including the establishment of any reasonably necessary policy, procedure or protocol.
  - "Disclosure Letter" means, as the context requires, the Company Disclosure Letter and/or the Parent Disclosure Letter.
  - "Employee" means any employee of the Company or any of its Subsidiaries.
- "Environmental Claim" means any claim, action, suit, proceeding, investigation, Order, demand or written notice alleging potential or actual liability (including liability for investigatory costs, cleanup costs, governmental response costs, natural resources damages, property damages, personal injuries, attorneys' fees, fines or penalties) arising out of, based on, resulting from or relating to (i) the presence of, Release of, or exposure to any Hazardous Substances; (ii) circumstances forming the basis of any violation of any Environmental Law; or (iii) any other matters covered or regulated by, or for which liability could be imposed under, Environmental Law.
- "Environmental Law" means any Law or Order relating to pollution, the protection, restoration or remediation of or prevention of harm to the environment or natural resources, or the protection of human health and safety, including any Law or Order relating to: (i) the exposure to, or Releases or threatened Releases of, Hazardous Substances; (ii) the generation, manufacture, processing, distribution, use, treatment, containment, disposal, storage, transport or handling of Hazardous Substances; or (iii) recordkeeping, notification, disclosure and reporting requirements respecting Hazardous Substances.
  - "Environmental Permits" means all Governmental Authorizations relating to or required by Environmental Law.
- "Equity Linked Convertible Notes" means those outstanding convertible promissory notes convertible into Company Stock and issued in accordance with that certain Note Purchase Agreement dated as of June 24, 2021, as amended, by and among the Company, Parent Predecessor, THRC Holdings, L.P., Crestview III USWS TE, LLC, Crestview III USWS TE, LLC, and Wilmington Savings Fund Society, FSB, as collateral agent for the purchaser parties thereto, as each of such promissory notes may be amended, restated or otherwise modified from time to time following the date hereof, including the Convertible Note Amendments.
  - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations issued thereunder.
- "ERISA Affiliate" of any entity means each entity that is or was at any time treated as a single employer with such entity for purposes of Section 4001(b)(1) of ERISA or Section 414(b), (c), (m) or (o) of the Code.
- "Exploit" or "Exploitation" means to research, design, develop, implement, use, reproduce, modify and otherwise create improvements or derivative works based upon, make, have made, assemble, test, support, display, perform (whether publicly or otherwise), publish, transmit, broadcast, sell, offer to sell, import, distribute and otherwise dispose of, commercialize or exploit, or authorize any third party Person to do any of the foregoing.

"February Term C Loan Warrants" means those outstanding warrants of the Company issued in accordance with that certain Warrant Agreement, dated as of February 28, 2022, by and between the Company and Continental Stock Transfer & Trust Company, as each may be amended, restated or otherwise modified from time to time following the date hereof.

"Fraud" means actual fraud by a Person, which involves a knowing and intentional or willful misrepresentation or omission of a material fact with respect to the making of any representation or warranty set forth in this Agreement and made for the purpose of inducing the other Party to act, and upon which the other Party justifiably and actually relies with resulting losses, provided that such misrepresentation or omission shall only be deemed to exist if any of the individuals listed on Schedule 1.1 with respect to the Company or Parent, as applicable, had actual knowledge that the representations and warranties made by the applicable Party were actually breached when made with the express intention that the other Party relies thereon to its detriment and does not include any fraud claim based on negligent misrepresentation, recklessness or any equitable fraud or promissory fraud.

"GAAP" means generally accepted accounting principles in the United States.

"Governmental Authority" means any (i) nation or government, any federal, state, city, town, municipality, county, local or other political subdivision thereof or thereto and any department, commission, board, bureau, instrumentality, agency, merger control authority, in each case of competent jurisdiction, (ii) any arbitrator, tribunal or federal, state, local or foreign court, in each case of competent jurisdiction, (iii) any national securities exchange on which the securities of the Company or Parent are listed, or (iv) other governmental entity or quasi-governmental entity created or empowered under a statute (or rule, regulation or ordinance promulgated thereunder) or at the direction of any governmental authority, including those set forth in clauses (i), (ii) or (iii) of this definition, and that is empowered thereunder or thereby to exercise executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government, in each case of competent jurisdiction.

"Governmental Authorization" means any licenses, approvals, clearances, permits, certificates, waivers, consents, exemptions, variances, expirations and terminations of any waiting period requirements (including pursuant to Competition Laws), and notices, filings, registrations, qualifications, declarations and designations with, and other authorizations and approvals issued by or obtained from a Governmental Authority.

"Hazardous Substance" means any material, substance, chemical, or waste (or combination thereof) (i) to which liability or standards of conduct may be imposed, or that requires or may require reporting or investigation under Environmental Law; (ii) that is listed, defined, designated, regulated or classified as "hazardous," "toxic," "radioactive," "dangerous," a "pollutant," a "contaminant," "petroleum," "oil," or words of similar meaning or effect under Environmental Law; (iii) any emerging contaminants including, but not limited to, per- and polyfluoroalkyl substances (PFAS) or (iv) that can form the basis of any liability under Environmental Law.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"Inbound Intellectual Property Contracts" means any Contracts pursuant to which the Company or any of its Subsidiaries has the right to Exploit any Licensed Intellectual Property Rights or Licensed Technology, including all modifications, amendments and supplements thereto and waivers thereunder.

"Intellectual Property Contracts" means the Inbound Intellectual Property Contracts and the Outbound Intellectual Property Contracts.

"Intellectual Property Rights" means any and all intellectual property rights throughout the world, whether registered or not, including without limitation, all (i) patents (including all reissues, divisionals,

provisionals, continuations and continuations-in-part, re-examinations, renewals, substitutions and extensions thereof), patent applications and other patent rights and any other Governmental Authority-issued indicia of invention ownership (including inventor's certificates, petty patents and patent utility models) (collectively, "Patents"), (ii) copyrights in works of authorship of any type and all rights, title and interests in all copyrights, copyright registrations and applications for copyright registration, certificates of copyright and copyrighted interests throughout the world (collectively, "Copyrights"); (iii) trade names, trademarks and service marks, logos, corporate names, domain names and other Internet addresses or identifiers, trade dress and similar rights, and applications (including intent to use applications) to register any of the foregoing and all goodwill associated therewith throughout the world (collectively, "Marks"); (iv) rights, title and interests in all trade secrets and trade secret rights arising under common law, state law, federal law or laws of foreign countries, in each case to the extent any of the foregoing derives economic value (actual or potential) from not being generally known to other Persons who can obtain economic value from its disclosure or use (collectively, "Trade Secrets"); and (v) moral rights, publicity rights and any other intellectual property rights or other rights similar, corresponding or equivalent to any of the foregoing of any kind or nature anywhere in the world.

"Intervening Event" means any material event, condition, fact, occurrence, change or development that is not known or reasonably foreseeable to the Company Board as of the date of this Agreement, which event, condition, fact, occurrence, change or development becomes known to the Company Board prior to receipt of the Company Stockholder Approval; provided, however, that in no event shall any event, condition, fact, occurrence, change or development resulting from or arising out of any of the following give rise to an Intervening Event: (i) any Company Acquisition Proposal, (ii) any actions taken by Parent or the Company in accordance with Section 8.1 or the consequences of any such action, (iii) the fact that the Company or its Subsidiaries have exceeded or met (or the failure of Parent to meet) any internal or published projections, forecasts or predictions in respect of financial or operating performance for any period ending on or after the date hereof and (iv) any change in the market price or trading volume of the Company's securities or Parent's securities; provided, however, with respect to clauses (iii) and (iv), nothing shall prevent a Party from asserting that any event, condition, fact, occurrence, change or development that may have contributed to exceeding or meeting (or, with respect to Parent, failing to meet) such projections, forecasts or predictions or such change in market price or trading volume may be taken into account in determining whether there has been an Intervening Event if not otherwise excluded.

"Investor Persons" means Crestview III USWS, L.P. and Crestview III USWS TE, LLC.

"knowledge" means (i) with respect to the Company, the actual knowledge of each of the individuals listed in Section 1.1 of the Company Disclosure Letter, after reasonable inquiry by such individual, and (ii) with respect to Parent, the actual knowledge of each of the individuals listed in Section 1.1 of the Parent Disclosure Letter, after reasonable inquiry by such individual. With respect to Intellectual Property Rights and Technology, "knowledge" does not require the Company to conduct, have conducted, obtain, or have obtained any freedom to operate opinions or similar opinions of counsel or any Patents, trademarks or other Intellectual Property Rights clearance searches, and no knowledge of any third party Patents, trademarks, or other Intellectual Property Rights that would have been revealed by such inquiries, opinions or searches will be imputed to the Company.

"Laws" means any United States, federal, state or local or any foreign law (in each case, statutory, common or otherwise), constitution, treaty, convention, ordinance, code, rule, statute, regulation or other similar requirement enacted, issued, adopted, promulgated, entered into or applied by a Governmental Authority.

"Licensed Intellectual Property Rights" means any and all Intellectual Property Rights that are licensed or sublicensed to, or subject to a covenant not to sue for the benefit of, the Company or any of its Subsidiaries.

"Licensed Technology" means any and all Technology that is licensed or sublicensed to, or subject to a covenant not to sue for the benefit of, the Company or any of its Subsidiaries.

- "Lien" means, with respect to any property or asset, any mortgage, lien, pledge, charge, security interest, encumbrance or limitation on transfer (other than such a limitation arising under federal or state securities Laws) in respect of such property or asset. For purposes of this Agreement, a Person shall be deemed to own subject to a Lien any property or asset that it has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement relating to such property or asset.
- "March Term C Loan Warrants" means those outstanding warrants of the Company issued in accordance with that certain Warrant Agreement, dated as of March 1, 2022, by and between the Company and Continental Stock Transfer & Trust Company, as each may be amended, restated or otherwise modified from time to time following the date hereof.
  - "Nasdaq" means Nasdaq Global Select Market.
- "Open Source Software" means any software that contains or is derived in any manner (in whole or in part) from any software, code or libraries that are distributed as free software or as open source software or under any licensing or distribution models similar to open source, or any other license or distribution model which seeks to require any party which uses, modifies or distributes such software, code or libraries to make such software, code or libraries (or modification or derivative work thereof) or any other software, code or libraries which may be combined with or linked thereto available in source code form or which may impose any other obligation or restriction with respect to such party's Patent or other Intellectual Property Rights.
- "Order" means any order, writ, injunction, decree, consent decree, judgment, award, injunction, settlement or stipulation issued, promulgated, made, rendered or entered into by or with any Governmental Authority (in each case, whether temporary, preliminary or permanent).
- "Outbound Intellectual Property Contracts" means any Contracts pursuant to which the Company or any of its Subsidiaries grants to any Person (other than the Company or its Subsidiaries) any right, license, or covenant not to sue with respect to any Intellectual Property Rights or Technology, including all modifications, amendments and supplements thereto and waivers thereunder.
- "Owned Intellectual Property Rights" means any and all Intellectual Property Rights owned or purported to be owned by the Company or any of its Subsidiaries.
- "Owned Technology" means any and all Technology owned or purported to be owned by the Company or any of its Subsidiaries, including, without limitation, the Technology that the Company purports to own as set forth in its securities filings, on its website, and in other public statements.
- "Parent Balance Sheet" means the consolidated balance sheet of the Parent Predecessor as of December 31, 2021 and the footnotes thereto set forth in the Prospectus.
  - "Parent Balance Sheet Date" means December 31, 2021.
- "Parent Disclosure Letter" means the disclosure letter delivered by Parent to the Company in connection with, and upon the execution of, this Agreement.
- "Parent Material Adverse Effect" means any Effect that, individually or in the aggregate with all other Effects, has had or would reasonably be expected to have a material adverse effect on the business, results of operations or financial condition of Parent and its Subsidiaries, taken as a whole; provided, however, that a Parent Material Adverse Effect shall not be deemed to include any Effect (alone or in combination) arising out of, relating to, or resulting from: (i) Effects generally affecting the economy, financial or securities markets, or political conditions; (ii) the execution and delivery or performance of this Agreement, or the announcement,

consummation or pendency of the transactions contemplated by this Agreement or the identity of or any facts or circumstances relating to the Company or any of its Affiliates, including the impact of any of the foregoing on the relationships, contractual or otherwise, of Parent or any of its Subsidiaries with customers, suppliers, service providers, employees, Governmental Authorities or any other Persons (provided that this clause (ii) shall not apply to the representations and warranties that, by their terms, speak specifically of the consequences arising out of the execution or performance of this Agreement or the consummation of the transactions contemplated hereby); (iii) any changes in applicable Law or GAAP, in the interpretation of GAAP, in the accounting rules and regulations of the SEC, or other applicable accounting standards; (iv) acts of war or other military actions, terrorism or the escalation thereof; (v) earthquakes, hurricanes, tornadoes or other natural disasters; (vi) general conditions in the industry in which Parent and its Subsidiaries operate; (vii) any failure, in and of itself, by Parent to meet any internal or published projections, forecasts, estimates, or predictions in respect of revenues, earnings, or other financial or operating metrics for any period (it being understood that any Effect underlying such failure may be deemed to constitute, or be taken into account in determining whether there has been or would reasonably be expected to become, a Parent Material Adverse Effect, to the extent permitted by this definition and not otherwise excepted by another clause of this proviso); (viii) any change, in and of itself, in the market price or trading volume of Parent's securities (it being understood that any Effect underlying such change may be deemed to constitute, or be taken into account in determining whether there has been or would reasonably be expected to become, a Parent Material Adverse Effect, to the extent permitted by this definition and not otherwise excepted by another clause of this proviso); (ix) any Claim or proceeding for appraisal of any shares of Parent Stock pursuant to the DGCL in connection herewith, (x) any epidemic, pandemic or disease outbreak (including the COVID-19 pandemic) or the evolution of any COVID-19 Measures or other restrictions that relate to, or arise out of, any epidemic, pandemic or disease outbreak (including the COVID-19 pandemic) and any COVID-19 Responses, or (xi) the taking of any action (or the omission of any action) by Parent or any Subsidiary of Parent in accordance with the terms of this Agreement to the extent the taking of such action (or omission) is expressly required or contemplated by this Agreement or such action was taken at the written request of, or with the written consent of, the Company (provided that this clause (ix) shall not apply to the representations and warranties that, by their terms, speak specifically of the consequences arising out of the execution or performance of this Agreement or the consummation of the transactions contemplated hereby); provided further, however, that any Effect referred to in clauses (i), (iii), (iv), (v), (vi) or (x) immediately above shall be taken into account in determining whether a Parent Material Adverse Effect has occurred or would reasonably be expected to occur to the extent it has a disproportionate effect on Parent and its Subsidiaries, taken as a whole, compared to other participants in the industries in which Parent and its Subsidiaries conduct their businesses.

"Parent Predecessor" means ProFrac Holdings, LLC.

"Parent Stock" means the Class A common stock, par value \$0.01 per share, of Parent.

"Parent Stock Issuance" means the issuance of shares of Parent Stock to the holders of Company Stock and Company Equity Awards in connection with the Merger.

"Parent Stockholder Approval" means the affirmative vote of the holders of a majority of the shares of Parent Stock entitled to vote with respect to the approval of the Parent Stock Issuance.

"Parent Trading Price" means an amount equal to the average of the volume weighted average price per share of Parent Stock on Nasdaq (as reported by Bloomberg L.P. or, if not reported therein, in another authoritative source mutually selected by Parent and the Company) on each of the ten (10) consecutive trading days ending with the complete trading day immediately prior to the Effective Time.

"Parent Unaffiliated Stockholders" means the holders of Parent Stock, excluding THRC Holdings, LP and its Affiliates.

"Permitted Liens" means (i) Liens for Taxes not yet due and payable for which adequate reserves (as determined in accordance with GAAP) have been established on the Company Balance Sheet or Parent Balance

Sheet, as applicable, (ii) Liens in favor of vendors, carriers, warehousemen, repairmen, mechanics, workmen, materialmen, construction or similar Liens or other encumbrances arising in the ordinary course of business with respect to amounts not yet overdue or the validity of which is being contested in good faith by appropriate proceedings or that are otherwise not material, (iii) Liens reflected in the Company Balance Sheet or Parent Balance Sheet, as applicable, (iv) with respect to any real property, Liens that do not materially impair the value or use of such real property or are being contested in the ordinary course of business in good faith, (v) Liens imposed or promulgated by operation of applicable Law with respect to real property and improvements, including zoning regulations, permits, licenses, utility easements, rights of way and similar Liens imposed or promulgated by any Governmental Authority, and (vi) with respect to any real property, title defects or irregularities that would not, individually or in the aggregate, reasonably be expected to materially impair the continued use and operation of the assets to which they relate in the business of the Company or Parent, as applicable, and its Subsidiaries as presently conducted or are being contested in the ordinary course of business in good faith and (vii) solely with respect to Section 4.14(c), Section 4.15(a) and Section 4.18(b), grants of non-exclusive licenses or other non-exclusive rights with respect to Intellectual Property Rights or Technology that do not, in each case, otherwise contain or constitute a mortgage, lien, pledge, charge, security interest, encumbrance, or limitation on transfer.

"Person" means an individual, group (within the meaning of Section 13(d)(3) of the 1934 Act), corporation, partnership, limited liability company, association, company, joint venture, estate, trust, association, or other entity or organization of any kind or nature, including a Governmental Authority.

"Personal Information" means any information that identifies or that, in combination with other information, can be used to identify a natural person, including such information of employees, contractors and customers of the Company or its Subsidiaries, that is protected under Privacy Laws

"Placement Agent Warrants" means those outstanding placement agent common stock warrants of the Company issued on March 11, 2022, as each may be amended, restated or otherwise modified from time to time following the date hereof.

"Pool A Amended Payout Shares" means, with respect to a Pool A Performance Award, the number equal to the quotient obtained by dividing (i) the Accumulated Cash Award Value with respect to such Pool A Performance Award as of the Pool A Determination Date by (ii) the Pool A Conversion Factor.

"Pool A Conversion Factor" means \$1.22.

"Pool A Determination Date" means July 19, 2022.

"Pool A Payout Shares" means, with respect to a Pool A Performance Award, the number equal to the quotient obtained by dividing (i) the Accumulated Cash Award Value with respect to such Pool A Performance Award as of the Effective Time by (ii) the Pool A Conversion Factor.

"Pool A Performance Award" means a Performance Award (Pool A) as granted under the Company LTIP as evidenced by that certain U.S. Well Services, Inc. Terms and Conditions Performance Award (Pool A).

"Pool B Amended Payout Shares" means, with respect to a Pool B Performance Award, the number equal to the quotient obtained by dividing (i) the Accumulated Cash Award Value with respect to such Pool B Performance Award as of the Pool B Determination Date by (ii) the Pool B Conversion Factor.

"Pool B Conversion Factor" means \$1.078.

"Pool B Determination Date" means July 19, 2022.

"Pool B Payout Shares" means, with respect to a Pool B Performance Award, the number equal to the quotient obtained by dividing (i) the Accumulated Cash Award Value with respect to such Pool B Performance Award as of the Effective Time by (ii) the Pool B Conversion Factor.

- "Pool B Performance Award" means a Performance Award (Pool B) as granted under the LTIP as evidenced by that certain U.S. Well Services, Inc. Terms and Conditions Performance Award (Pool B).
- "Privacy Laws" means applicable Laws in respect of, or standards imposed by self-regulatory organizations that apply to, Processing and includes, without limitation, the Gramm-Leach-Bliley Act, 15 U.S.C. § 6801 et seq., the Health Insurance Portability and Accountability Act of 1996 (42 U.S.C. §§ 1320d—1329d-8), as amended by the Health Information Technology for Economic and Clinical Health Act (42 U.S.C. §§ 3000 et seq.), and their implementing regulations, the Data Protection Directive (95/46/EC) as implemented in each relevant EU member state and the Payment Card Industry Data Security Standard.
- "Proceeding" means any suit, action, claim, proceeding, arbitration, mediation, audit or hearing (in each case, whether civil, criminal or administrative) commenced, brought, conducted or heard by or before, or otherwise involving, any Governmental Authority.
- "Processing" means the collection, use, storage, distribution, transfer, protection, disclosure or disposal of, or other action taken regarding, (i) Personal Information or (ii) 'personal data' as that term is understood under any applicable Laws.
  - "Prospectus" means that certain prospectus filed by Parent pursuant to Rule 424(b)(4) of the 1933 Act dated May 12, 2022.
  - "RDO Warrant" means those outstanding warrants of the Company issued in a registered direct offering on March 11, 2022.
- "Release" means any release, spill, emission, discharge, leaking, pouring, dumping or emptying, pumping, injection, deposit, disposal, dispersal, leaching or migration into the indoor or outdoor environment (including soil, ambient air, surface water, groundwater and surface or subsurface strata) or into or out of any property, including the movement of Hazardous Substances through or in the air, soil, surface water, groundwater or property.
  - "Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002, as amended.
  - "SEC" means the Securities and Exchange Commission.
- "Series A Warrants" means those outstanding warrants of the Company issued in accordance with that certain Warrant Agreement, dated as of May 24, 2019, by and between the Company and Continental Stock Transfer & Trust Company, as each may be amended, restated or otherwise modified from time to time following the date hereof.
- "SPAC Warrants" means those outstanding warrants of the Company issued in accordance with that certain Warrant Agreement, dated as of March 9, 2017, by and between the Company and Continental Stock Transfer & Trust Company, as each may be amended, restated or otherwise modified from time to time following the date hereof.
- "Subsidiary" means, with respect to any Person, another Person (other than a natural person), of which such first Person (i) owns directly or indirectly (a) an aggregate amount of the voting securities, other voting ownership or voting partnership interests to elect or appoint a majority of the board of directors or other governing body or (b) if there are no such voting interests, 50% or more of the equity interests therein or (ii) has the right to appoint 50% or more of the directors or managers.
- "Superior Proposal" means a bona fide written Company Acquisition Proposal from any Person (other than Parent and its Subsidiaries or Affiliates) (with all references to "20% or more" in the definition of Company Acquisition Proposal being deemed to reference "50% or more" and all references to "80% or less" in

the definition of Company Acquisition Proposal being deemed to reference "50% or less"), which the Company Board has, after consultation with the Company's outside financial advisors and outside legal counsel, determined in its good faith judgment, (i) is reasonably expected to be consummated on the terms proposed and (ii) would, if consummated, result in a transaction more favorable to its stockholders (in their capacity as such) than the transactions contemplated by this Agreement after taking into account all legal, financial, regulatory and other aspects of such Company Acquisition Proposal (including the existence of financing conditions, the conditionality of any financing commitments and the likelihood and timing of consummation (as compared to the transactions contemplated hereby)) and such other matters that the Company Board deems relevant.

"Tax" means any tax, including gross receipts, profits, sales, use, occupation, value added, ad valorem, transfer, franchise, withholding, payroll, employment, capital, goods and services, gross income, business, environmental, severance, service use, unemployment, social security, national insurance, stamp, custom, escheat, excise or real or personal property, alternative or add-on minimum or estimated taxes, or other like assessment or charge of any kind whatsoever, together with any interest, penalty, addition to tax or additional amount imposed with respect thereto, whether disputed or not.

"Tax Return" means any report, return, document, declaration or other information or statement with respect to Taxes, including information returns, any documents with respect to or accompanying payments of estimated Taxes, or with respect to or accompanying requests for the extension of time in which to file any such report, return, document, declaration or other information, and in all cases including any schedule or attachment thereto or amendment thereof.

"Tax Sharing Agreement" means all existing agreements binding a party or any of its Subsidiaries that provide for the allocation, apportionment, sharing or assignment of any Tax liability or benefit (excluding any indemnification agreement or arrangement pertaining to the sale or lease of assets or subsidiaries and any commercially reasonable indemnity, sharing or similar agreements or arrangements where the inclusion of a Tax indemnification or allocation provision is customary or incidental to an agreement the primary nature of which is not Tax sharing or indemnification).

"Taxing Authority" means any Governmental Authority responsible for the imposition, administration or collection of any Tax, assessment or charge (domestic or foreign).

"Technology" means, collectively, any technology, information, or tangible embodiments of any Intellectual Property Rights, in electronic, written or any other form, including without limitation designs, formulae, specifications, research and development information, technical information, design and manufacturing schematics, manufacturing and other processes, procedures, algorithms, data, databases, methods, techniques, ideas, know-how, concepts, inventions, invention disclosures, discoveries, developments, creations, works of authorship, improvements, derivative works, products, hardware, and software and other similar materials (including firmware or software in the form of source code, object code, byte code or other format and any bug fixes, patches, updates, upgrades or modifications thereto).

"Third Party" means any Person other than Parent, the Company or any of their respective Affiliates.

"Treasury Regulations" means the regulations promulgated under the Code.

"Warrants" means, collectively, the SPAC Warrants, the Series A Warrants, the February Term C Loan Warrants, the March Term C Loan Warrants, the Placement Agent Warrants and the RDO Warrants.

"Willful Breach" means a material breach of this Agreement that is the consequence of an act or omission by the breaching party with the actual knowledge that the taking of such act or failure to take such action would result in or constitute such a material breach, provided that such actual knowledge shall only be deemed to exist with respect to the Company or Parent, as applicable, had actual knowledge that such action or failure to take action would result in such breach.

Section 1.2 <u>Table of Definitions</u>. Each of the following terms is defined in the Section set forth opposite such term:

Term	Section
Acceptable Confidentiality Agreement	Section 6.4(b)(i)
Agreement	Preamble
Award Amendments	Section 6.2(c)
Book-Entry Shares	Section 3.2(c)
Certificate	Section 3.2(c)
Certificate of Merger	Section 2.3
Closing	Section 2.2
COBRA	Section 4.18(d)
COD Amendment	Section 6.2(b)
Code	Recitals
Company	Preamble
Company Board	Recitals
Company Board Recommendation	Section 6.5
Company Capitalization Date	Section 4.5(a)
Company DSU	Section 3.8(b)
Company Indemnified Party	Section 7.4(a)
Company Material Contract	Section 4.20(a)
Company Plan	Section 4.18(a)
Company Preferred Stock	Section 4.5(a)
Company Real Property	Section 4.14(b)
Company Restricted Share	Section 3.8(a)
Company SEC Documents	Section 4.7(a)
Company Securities	Section 4.5(e)
Company Special Committee	Recitals
Company Stockholders' Meeting	Section 6.5
Company Subsidiary Securities	Section 4.6(b)
Company Termination Fee	Section 10.3(a)
Company Transaction Litigation	Section 8.8
Continuing Employee	Section 7.6(b)
Convertible Note Amendments	Section 6.2(a)
D&O Insurance	Section 7.4(b)
DGCL	Section 2.1(a)
Effective Time	Section 2.3
End Date	Section 10.1(b)(i)
Exchange Agent	Section 3.6(a)
Exchange Fund	Section 3.6(a)
Exchange Ratio	Section 3.2(b)
Expenses	Section 10.3(c)
Financing	Section 8.2(a)
Foreign Company Plan	Section 4.18(a)
Information Statement	Section 8.3
Internal Controls  Manager Consideration	Section 4.7(g)
Merger Consideration	Section 3.2(b)
Merger Merger Sub Inc.	Recitals Preamble
S	Section 4.18(c)
Multiemployer Plan Owned Real Property	Section 4.18(c) Section 4.14(a)
Parent	Preamble
1 di Çiit	1 ICAIIIUIC

Parent Board Recitals Parent Class B Common Stock Section 5.5(a) Section 5.5(a) Parent Preferred Stock Parent RSUs Section 5.5(a) Parent SEC Documents Section 5.6(a) Parent Securities Section 5.5(b) Parent Special Committee Recitals Party or Parties Preamble Plan Termination Notice Section 7.6(c) Section 7.4(b) Premium Cap Proxy Statement Section 8.3 RDO Warrants Section 1.1 Real Property Leases Section 4.14(b) Registration Statement Section 8.3 Representatives Section 8.6(a) Rollover Warrants Section 3.9 Series A Preferred Stock Section 4.5(a) Section 5.10 Solvent Support Agreements Recitals Section 2.1(a) Surviving Corporation Surviving Corporation Stock Section 3.2(d) U.S. Company Plan Section 4.18(a) Warrant Sale Recitals

Section 1.3 Other Definitional and Interpretative Provisions. The words "hereof," "herein" and "hereunder" and words of like import used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The descriptive headings used herein are inserted for convenience of reference only and are not intended to be part of or to affect the meaning or interpretation of this Agreement. References to Articles, Sections, Exhibits and Schedules are to Articles, Sections, Exhibits and Schedules of this Agreement unless otherwise specified. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized terms used in any Exhibit or Schedule but not otherwise defined therein, shall have the meaning as defined in this Agreement. Any singular term in this Agreement shall be deemed to include the plural, and any plural term the singular. The definitions contained in this Agreement are applicable to the masculine as well as to the feminine and neuter genders of such term. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation," whether or not they are in fact followed by those words or words of like import. References to "ordinary course of business" shall mean "ordinary course of business consistent with past practice". "Writing," "written" and comparable terms refer to printing, typing and other means of reproducing words (including electronic media) in a visible form. References to any statute shall be deemed to refer to such statute and to any rules or regulations promulgated thereunder. References to any Contract are to that Contract as amended, modified or supplemented (including by waiver or consent) from time to time in accordance with the terms hereof and thereof. References to "the transactions contemplated by this Agreement" or words with a similar import shall be deemed to include the Merger. References to any Person include the successors and permitted assigns of that Person. References herein to "\$" or dollars will refer to United States dollars, unless otherwise specified. References from or through any date mean, unless otherwise specified, from and including such date or through and including such date, respectively. References to any period of days will be deemed to be to the relevant number of calendar days unless otherwise specified. The phrase "made available" with respect to documents shall mean any documents filed or furnished with the SEC prior to the date of this Agreement or, in the case of the Company, made available in the Datasite virtual data room created by the Company on June 8, 2022 on or prior to the date that is one calendar day prior to the date hereof. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be

construed as if drafted jointly by the Parties, and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement.

#### ARTICLE II THE MERGER

#### Section 2.1 The Merger. Upon the terms and subject to the conditions of this Agreement:

(a) at the Effective Time, Merger Sub Inc. will be merged with and into the Company in accordance with the provisions of the General Corporation Law of the State of Delaware (the "DGCL"). As a result of the Merger, the separate existence of Merger Sub Inc. shall cease and the Company shall continue its existence under the laws of the State of Delaware as the surviving corporation (in such capacity, the Company is sometimes referred to herein as the "Surviving Corporation") as an indirect subsidiary of Parent; and

(b) at the Effective Time, the effect of the Merger shall be as provided in this Agreement, the Certificate of Merger, and in the applicable provisions of the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of Merger Sub Inc. and the Company shall become the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of the Surviving Corporation.

Section 2.2 The Closing. Subject to the provisions of this Agreement, the closing of the Merger (the 'Closing') shall take place at the offices of Brown Rudnick LLP, 7 Times Square, New York, NY 10036 or remotely by exchange of documents and signatures (or their electronic counterparts), promptly following the Effective Time, and in any case no later than the second (2nd) Business Day following the satisfaction or, to the extent permitted hereunder, waiver of the conditions set forth in Article IX (except for any conditions that by their nature can only be satisfied on the Closing Date, but subject to the satisfaction of such conditions or waiver by the Party entitled to waive such conditions), unless another date, time or place is agreed to in writing by Parent and the Company.

Section 2.3 Effective Times. Subject to the provisions of this Agreement, as soon as practicable on the Closing Date, the Parties shall file with the Secretary of State of the State of Delaware the certificate of merger relating to the Merger (the "Certificate of Merger") in such form as is required by, and executed and acknowledged in accordance with, the relevant provisions of the DGCL. The Merger shall become effective at the time that the Certificate of Merger has been duly filed with the Secretary of State of the State of Delaware, or at such later time as Parent and the Company shall agree and specify in the Certificate of Merger (the time the Merger becomes effective, the "Effective Time").

#### Section 2.4 Surviving Corporation Matters.

- (a) At the Effective Time, the certificate of incorporation and bylaws of the Company, as in effect immediately prior to the Effective Time, shall continue to be the certificate of incorporation and bylaws of the Surviving Corporation until further amended as provided therein or in accordance with applicable Law.
- (b) The parties shall take all requisite action so that, from and after the Effective Time, until their successors have been duly elected or appointed and qualified, or until their earlier death, resignation or removal: (i) the directors of Merger Sub Inc. immediately prior to the Effective Time shall be the directors of the Surviving Corporation and (ii) the officers of Merger Sub Inc. immediately prior to the Effective Time shall be the officers of the Surviving Corporation.

# ARTICLE III EFFECT ON THE CAPITAL STOCK; EXCHANGE OF CERTIFICATES

#### Section 3.1 Series A Preferred Stock, Equity Linked Convertible Notes and Warrants

- (a) The Company shall take all requisite action so that, effective as of immediately prior to the Effective Time, (i) each holder of Series A Preferred Stock issued and outstanding at such time may convert such stock into shares of Company Stock at the Merger Conversion Ratio (as defined in the COD Amendment), and (ii) any shares of Series A Preferred Stock issued and outstanding immediately prior to the Effective Time not so converted by the holder pursuant to the foregoing clause (i) shall automatically convert into shares of Company Stock in accordance with Section 8(b) of the Certificate of Designations. Any shares of Company Stock issued upon conversion under clause (i) and (ii) shall be subject to the treatment set forth in Section 3.2(b) (treating such shares in the same manner as all other outstanding shares of Company Stock for such purposes).
- (b) The Company shall take all requisite action so that, effective as of immediately prior to the Effective Time, each Equity Linked Convertible Note issued and outstanding at such time shall automatically convert into a number of shares of Company Stock equal to the quotient obtained by dividing (i) the amount of outstanding aggregate principal amount, plus accrued and unpaid interest, owing under such Equity Linked Convertible Note through the date immediately prior to the Closing Date, by (ii) \$1.22, which shares of Company Stock shall be subject to the treatment set forth in Section 3.2(b) (treating such shares in the same manner as all other outstanding shares of Company Stock for such purposes).
- (c) At the Effective Time, each February Term C Loan Warrant and each March Term C Loan Warrant issued and outstanding immediately prior to the Effective Time (which shall be held by Parent pursuant to the Warrant Sale) shall be automatically cancelled and shall cease to exist and no consideration shall be delivered in exchange therefor.
- Section 3.2 Effect of the Merger on Capital Stock At the Effective Time, by virtue of the Merger and without any action on the part of any holder of any securities of the Company or Merger Sub Inc.:
- (a) All shares of Company Stock that are owned, directly or indirectly, by Parent, the Company (including shares held as treasury stock or otherwise) or Merger Sub Inc. immediately prior to the Effective Time shall be automatically cancelled and retired and shall cease to exist and no consideration shall be delivered in exchange therefor.
- (b) Each share of Company Stock issued and outstanding immediately prior to the Effective Time (other than shares to be cancelled in accordance with Section 3.2(a), but including shares of Company Stock (A) deemed to have been issued in respect of any Series A Preferred Stock converted pursuant to Section 3.1(a) or any Equity Linked Convertible Notes converted pursuant to Section 3.1(b) or (B) paid to the holder of a vested Company Equity Award immediately prior to the Effective Time pursuant to Section 3.8) shall be converted automatically into the right to receive 0.0561 (the 'Exchange Ratio') of a validly issued, fully paid and non-assessable share of Parent Stock (the 'Merger Consideration').
- (c) As of the Effective Time, all shares of Company Stock converted into the Merger Consideration pursuant to this Section 3.2 shall cease to be outstanding and shall automatically be cancelled and shall cease to exist, and each holder of (1) a certificate that immediately prior to the Effective Time represented any such shares of Company Stock (a "Certificate") or (2) shares of Company Stock held in book-entry form ("Book-Entry Shares") shall cease to have any rights with respect thereto, except (subject to Section 3.6) (A) the right to receive the Merger Consideration and (B) the right to receive any other amounts expressly provided herein, in each case, without interest, subject to compliance with the procedures set forth in Section 3.6.

(d) Each share of capital stock of Merger Sub Inc. issued and outstanding immediately prior to the Effective Time shall be converted into one validly issued, fully paid and nonassessable share of common stock, par value \$0.001 per share, of the Surviving Corporation (the "Surviving Corporation Stock")

Section 3.3 Certain Adjustments. Notwithstanding anything in this Agreement to the contrary, if, from the date of this Agreement until the earlier of (i) the Effective Time or (ii) any termination of this Agreement in accordance with Section 10.1, the outstanding shares of Parent Stock or Company Stock shall have been changed into a different number of shares or a different class by reason of any reclassification, stock split (including a reverse stock split), recapitalization, split-up, combination, exchange of shares, readjustment, or other similar transaction, or a stock or cash dividend thereon shall be declared with a record date within said period, then the Exchange Ratio, the Pool A Conversion Factor, the Pool B Conversion Factor and any other similarly dependent items, as the case may be, shall be appropriately adjusted to provide Parent and the holders of Company Stock the same economic effect as contemplated by this Agreement prior to such event. Nothing in this Section 3.3 shall be construed to permit any Party to take any action that is otherwise prohibited or restricted by any other provision of this Agreement.

Section 3.4 Fractional Shares. No certificates or scrip representing fractional shares of Parent Stock shall be issued upon the conversion of shares of Company Stock pursuant to Section 3.2, and such fractional share interests shall not entitle the owner thereof to any shares of Parent Stock or to vote or to any other rights of a holder of Parent Stock. All fractional shares to which a single record holder of Company Stock would be otherwise entitled to receive shall be aggregated and calculations shall be rounded to three (3) decimal places. In lieu of any such fractional shares, each holder of Company Stock who would otherwise be entitled to such fractional shares shall be entitled to an amount in cash, without interest, rounded to the nearest cent, equal to the product of (A) the amount of the fractional share interest in a share of Parent Stock to which such holder would, but for this Section 3.4, be entitled under Section 3.2(b) and (B) the Parent Trading Price. As soon as practicable after the determination of the amount of cash, if any, to be paid to holders of Company Stock in lieu of any fractional share interests in Parent Stock, the Exchange Agent shall make available such amounts, without interest, to the holders of Company Stock entitled to receive such cash.

Section 3.5 Appraisal Rights. Appraisal rights are available in connection with the Merger solely to the extent required by the DGCL.

### Section 3.6 Exchange of Company Stock.

(a) Prior to the Effective Time, Parent shall enter into a customary exchange agreement with a nationally recognized financial institution designated by Parent and reasonably acceptable to the Company (the "Exchange Agent"), and shall deposit with the Exchange Agent for the benefit of the holders of shares of Company Stock entitled to receive the Merger Consideration, for exchange in accordance with this Article III, through the Exchange Agent, subject to Section 3.6(b)(ii), book-entry shares (or certificates if requested) representing the full number of whole shares of Parent Stock issuable pursuant to Section 3.2 in exchange for outstanding shares of Company Stock. Parent shall, after the Effective Time on the appropriate payment date, if applicable, provide or cause to be provided to the Exchange Agent any dividends or other distributions payable on such shares of Parent Stock pursuant to Section 3.6(c) which had not theretofore been surrendered for exchange or been exchanged pursuant to Section 3.6(b)(iii) (such shares of Parent Stock and any dividends or other distributions with respect thereto, are hereinafter referred to as the "Exchange Fund"). For the purposes of such deposit, Parent shall assume that there will not be any fractional shares of Parent Stock. Parent shall make available to the Exchange Agent, for addition to the Exchange Fund, from time to time as needed, cash sufficient to pay cash in lieu of fractional shares in accordance with Section 3.4. The Exchange Agent shall deliver the Merger Consideration to be issued pursuant to Section 3.2 out of the Exchange Fund. Except as provided in Section 3.6(h), the Exchange Fund shall not be used for any other purpose.

#### (b) Exchange Procedures.

- (i) Certificates. Parent shall instruct the Exchange Agent to mail, as soon as reasonably practicable after the Effective Time and in any event not later than the fifth Business Day following the Closing Date, to each holder of record of a Certificate whose shares of Company Stock were converted into the Merger Consideration pursuant to Section 3.2, (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates to the Exchange Agent and shall be in customary form) and (ii) instructions for use in effecting the surrender of the Certificates (or affidavits of loss in lieu thereof in accordance with Section 3.6(g)) in exchange for the Merger Consideration. Upon surrender of a Certificate for cancellation to the Exchange Agent or to such other agent or agents as may be appointed by Parent (or affidavits of loss in lieu thereof in accordance with Section 3.6(g)), together with such letter of transmittal, duly executed, and such other documents as may reasonably be required by the Exchange Agent, the holder of such Certificate shall be entitled to receive in exchange therefor, and Parent shall cause the Exchange Agent to pay and deliver in exchange thereof as promptly as practicable, (A) the Merger Consideration such holder has the right to receive in respect of such Certificate pursuant to Section 3.2(b) (after taking into account all other Certificates surrendered by such holder pursuant to this Section 3.6(b)(i), (B) any dividends or other distributions payable pursuant to Section 3.6(c)(i) and (C) cash in lieu of fractional shares of Parent Stock payable pursuant to Section 3.4, and the Certificate so surrendered shall forthwith be cancelled. In the event of a transfer of ownership of Company Stock that is not registered in the transfer records of the Company, payment may be made and shares may be issued to a Person other than the Person in whose name the Certificate so surrendered is registered, if such Certificate shall be properly endorsed or otherwise be in proper form for transfer and the Person requesting such payment shall pay any transfer or other similar Taxes required by reason of the payment to a Person other than the registered holder of such Certificate or establish to the reasonable satisfaction of Parent that such Tax has been paid or is not applicable. No interest shall be paid or accrue on any cash payable upon surrender of any Certificate.
- (ii) <u>Book-Entry Shares</u>. Notwithstanding anything to the contrary contained in this Agreement, any holder of Book-Entry Shares shall not be required to deliver a Certificate or an executed letter of transmittal to the Exchange Agent to receive the Merger Consideration that such holder is entitled to receive pursuant to <u>Section 3.2</u>. In lieu thereof, each holder of record of one or more Book-Entry Shares whose shares of Company Stock were converted into the Merger Consideration pursuant to <u>Section 3.2</u> shall automatically upon the Effective Time be entitled to receive, and Parent shall cause the Exchange Agent to pay and deliver as promptly as practicable after the Effective Time, (A) the Merger Consideration such holder has the right to receive in respect of such Book-Entry Shares pursuant to <u>Section 3.2(b)</u> (after taking into account all other Book-Entry Shares converted by such holder pursuant to this <u>Section 3.6(b)(ii)</u>, (B) any dividends or distributions payable pursuant to <u>Section 3.6(c)(ii)</u> and (C) cash in lieu of any fractional shares payable pursuant to <u>Section 3.4</u>, and the Book-Entry Shares of such holder shall forthwith be cancelled. No interest shall be paid or accrue on any cash payable upon conversion of any Book-Entry Shares.

#### (c) Distributions with Respect to Unexchanged Shares.

(i) <u>Certificates</u>. No dividends or other distributions with respect to Parent Stock with a record date after the Effective Time shall be paid to the holder of any unsurrendered Certificate formerly representing Company Stock, until the surrender of such Certificate (or affidavits of loss in lieu thereof in accordance with <u>Section 3.6(g)</u>) in accordance with this <u>Article III</u>. Subject to applicable Law, following surrender of any such Certificate (or affidavits of loss in lieu thereof in accordance with <u>Section 3.6(g)</u>), there shall be paid to the holder of the shares of Parent Stock issued in exchange therefor, without interest, (A) at the time of delivery of such Parent Stock by the Exchange Agent pursuant to <u>Section 3.6(b)(i)</u>, the amount of dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such shares of Parent Stock and (B) at the appropriate payment date, the amount of dividends or other distributions with a record date after the Effective Time but prior to such delivery of such Parent Stock by the Exchange Agent pursuant to <u>Section 3.6(b)(i)</u>, and a payment date subsequent to such delivery of such

Parent Stock by the Exchange Agent pursuant to Section 3.6(b)(i), payable with respect to such shares of Parent Stock.

- (ii) <u>Book-Entry Shares</u>. Subject to applicable Law, there shall be paid to the holder of the shares of Parent Stock issued in exchange for Book-Entry Shares in accordance with this <u>Article III</u>, without interest, (A) at the time of delivery of such Parent Stock by the Exchange Agent pursuant to <u>Section 3.6(b)(ii)</u>, the amount of dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such shares of Parent Stock and (B) at the appropriate payment date, the amount of dividends or other distributions with a record date after the Effective Time but prior to the time of such delivery by the Exchange Agent pursuant to <u>Section 3.6(b)(ii)</u>, and a payment date subsequent to the time of such delivery by the Exchange Agent pursuant to <u>Section 3.6(b)(ii)</u>, payable with respect to such shares of Parent Stock.
- (d) The Merger Consideration issued and paid in accordance with the terms of this Article III upon the surrender of the Certificates (or affidavits of loss in lieu thereof in accordance with Section 3.6(g)) (or, immediately, in the case of the Book-Entry Shares) shall be deemed to have been issued and paid in full satisfaction of all rights pertaining to such shares of Company Stock (other than the right to receive dividends or other distributions, if any, in accordance with Section 3.6(c)) and cash in lieu of any fractional shares payable pursuant to Section 3.4. After the Effective Time, there shall be no further registration of transfers on the stock transfer books of the Surviving Corporation of shares of Company Stock that were outstanding immediately prior to the Effective Time. If, after the Effective Time, any Certificates formerly representing shares of Company Stock are presented to the Surviving Corporation or the Exchange Agent for any reason, they shall be cancelled and exchanged as provided in this Article III.
- (e) Any portion of the Exchange Fund that remains undistributed to the former holders of Company Stock for one year after the Effective Time shall be delivered to the Surviving Corporation, upon demand, and any former holder of Company Stock who has not theretofore complied with this <u>Article III</u> shall thereafter look <u>only</u> to the Surviving Corporation for payment of its claim for the Merger Consideration, cash in lieu of fractional shares payable pursuant to <u>Section 3.4</u> and any dividends or distributions with respect to Parent Stock as contemplated by <u>Section 3.6(c)</u>.
- (f) None of Parent, the Company, the Surviving Corporation or the Exchange Agent shall be liable to any Person in respect of any shares of Parent Stock (or dividends or distributions with respect thereto) from the Exchange Fund delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law. Any Merger Consideration remaining unclaimed by former holders of Company Stock immediately prior to such time as such amounts would otherwise escheat to or become property of any Governmental Authority shall, to the fullest extent permitted by applicable Law, become the property of the Surviving Corporation free and clear of any claims or interest of any Person previously entitled thereto.
- (g) In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit, in form and substance reasonably acceptable to Parent, of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by Parent or the Exchange Agent, the posting by such Person of a bond in reasonable and customary amount as Parent or the Exchange Agent may direct, as indemnity against any claim that may be made against it or the Surviving Corporation with respect to such Certificate, the Exchange Agent will issue in exchange for such lost, stolen or destroyed Certificate the Merger Consideration, cash in lieu of fractional shares payable pursuant to Section 3.4 and any unpaid dividends or other distributions that would be payable or deliverable in respect thereof pursuant to Section 3.6(c) had such lost, stolen or destroyed Certificate been surrendered as provided in this Article III.
- (h) The Exchange Agent shall invest the cash included in the Exchange Fund as directed by Parent; provided, however, that no such investment income or gain or loss thereon shall affect the amounts payable to holders of Company Stock. Any interest, gains and other income resulting from such investments shall be the sole and exclusive property of Parent payable to Parent upon its request, and no part of such interest, gains and

other income shall accrue to the benefit of holders of Company Stock; provided, further, that any investment of such cash shall in all events be limited to direct short-term obligations of, or short-term obligations fully guaranteed as to principal and interest by, the U.S. government, in commercial paper rated A-1 or P-1 or better by Moody's Investors Service, Inc. or Standard & Poor's Corporation, respectively, or in certificates of deposit, bank repurchase agreements or banker's acceptances of commercial banks with capital exceeding \$10 billion (based on the most recent financial statements of such bank that are then publicly available), and that no such investment or loss thereon shall affect the amounts payable to holders of Company Stock pursuant to this Article III. If for any reason (including losses) the cash in the Exchange Fund shall be insufficient to fully satisfy all of the payment obligations to be made in cash by the Exchange Agent hereunder, Parent shall promptly deposit cash into the Exchange Fund in an amount which is equal to the deficiency in the amount of cash required to fully satisfy such cash payment obligations.

(i) Parent, the Surviving Corporation or the Exchange Agent shall be entitled to deduct and withhold from the consideration otherwise payable to any former holder of Company Stock pursuant to this Agreement such amounts as are required to be deducted or withheld therefrom under the Code, or under any provision of state, local or foreign Tax Laws. Any amount deducted or withheld pursuant to this Section 3.6(i) and paid over to the relevant Taxing Authority shall be treated as having been paid to the holder of Company Stock in respect of which such deduction or withholding was made. Parent shall pay, or shall cause to be paid, all amounts so deducted or withheld to the appropriate Taxing Authority within the period required under applicable Law.

Section 3.7 Further Assurances. If, at any time after the Effective Time, the Surviving Corporation shall determine that any actions are necessary or desirable to vest, perfect or confirm of record or otherwise in the Surviving Corporation its right, title or interest in, to or under any of the rights, properties or assets of the Company acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger or otherwise to carry out this Agreement, then the officers and directors of the Surviving Corporation shall be authorized to take all such actions as may be necessary or desirable to vest all right, title or interest in, to and under such rights, properties or assets in the Surviving Corporation or otherwise to carry out this Agreement.

#### Section 3.8 Company Stock-Based Awards.

- (a) <u>Company Restricted Shares</u>. The Company shall take all requisite action so that, at the Effective Time, each share of Company Stock subject to vesting, repurchase, or other lapse of restrictions (a "Company Restricted Share") that is outstanding and unvested under the Company LTIP immediately prior to the Effective Time shall, by virtue of the Merger and without any action on the part of the holder thereof, be cancelled in exchange for the right to receive the Merger Consideration pursuant to <u>Section 3.2(b)</u> and cash in lieu of fractional shares payable pursuant to <u>Section 3.4</u> (treating such unvested Company Restricted Shares in the same manner as all other outstanding shares of Company Stock for such purposes) less any Taxes required to be withheld with respect to such Company Restricted Share in accordance with <u>Section 3.8(e)</u>.
- (b) <u>Deferred Stock Units</u>; <u>Restricted Stock Units</u>. The Company shall take all requisite action so that immediately prior to the Effective Time, each then-outstanding deferred stock unit or restricted stock unit, in each case representing a right to receive one share of Company Stock granted under the Company LTIP (each, a "Company DSU"), shall, by virtue of the Merger and without any action on the part of the holder thereof, be cancelled and converted into the right to receive the Merger Consideration pursuant to <u>Section 3.2(b)</u> and cash in lieu of fractional shares payable pursuant to <u>Section 3.4</u> (treating such Company DSU in the same manner as if it were an outstanding share of Company Stock for such purposes).
- (c) <u>Pool A and B Awards</u>. The Company shall take all requisite action so that, immediately prior to the Effective Time, by virtue of the Merger and without any action on the part of the holder thereof, (i) each then-outstanding Pool A Performance Award shall be cancelled and converted into the right to receive (A) with respect to each Pool A Performance Award amended by an Award Amendment, the Merger Consideration payable pursuant to <u>Section 3.2(b)</u> with respect to the number of Pool A Amended Payout Shares subject to such

Pool A Performance Award, and (B) with respect to each Pool A Performance Award not amended by an Award Amendment, the Merger Consideration pursuant to Section 3.2(b) with respect to the number of Pool A Payout Shares subject to such Pool A Performance Award and (ii) each then-outstanding Pool B Performance Award shall be cancelled and converted into the right to receive (A) with respect to each Pool B Performance Award amended by an Award Amendment, the Merger Consideration pursuant to Section 3.2(b) with respect to the number of Pool B Payout Shares subject to such Pool B Performance Award, and (B) with respect to each Pool B Performance Award not amended by an Award Amendment, the Merger Consideration pursuant to Section 3.2(b) with respect to the number of Pool B Payout Shares subject to such Pool B Performance Award, in each of clauses (i) and (ii), without any interest thereon and less applicable Tax withholding.

- (d) <u>Resolutions and Other Company Actions</u>. At or prior to the Effective Time, the Company, the Company Board, and the compensation committee of such board, as applicable, shall adopt any resolutions and take any actions (including obtaining any employee consents) that may be necessary to effectuate the provisions of <u>clauses (a)</u>, (b) and (c) of this <u>Section 3.8</u>.
- (e) <u>Withholding</u>. The Company, the Surviving Corporation and Parent (and their respective Subsidiaries, Affiliates, agents and representatives) shall be entitled to deduct and withhold from any amounts otherwise payable pursuant to this <u>Section 3.8</u> such amounts as are required or permitted to be deducted or withheld with respect to the making of such payment under applicable Law. To the extent that amounts are so deducted or withheld by the Company, Surviving Corporation or Parent (or their respective Subsidiaries, Affiliates, agents or representatives), such deducted or withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.
- (f) <u>Section 409A</u>. Notwithstanding the anything to the contrary, amounts payable pursuant to this <u>Section 3.8</u> with respect to any Company Equity Award that constitutes, either in whole or in part, a deferral of compensation subject to Section 409A of the Code, shall be paid on such later date specified in the applicable Company Equity Award, solely to the extent necessary to avoid additional taxes pursuant to Section 409A of the Code.

Section 3.9 Warrants. At the Effective Time, each SPAC Warrant, Series A Warrant, Placement Agent Warrant and RDO Warrant issued and outstanding immediately prior to the Effective Time (collectively, the "Rollover Warrants"), in accordance with the terms of such Rollover Warrants, shall be cancelled and converted into the right to receive a warrant to purchase a number of shares of Parent Stock equal to (i) the number of shares of Company Stock underlying such Rollover Warrant multiplied by (ii) the Exchange Ratio. The exercise price of such Rollover Warrants shall be the exercise price of such Rollover Warrant divided by the Exchange Ratio. From and after the Closing, Parent shall comply with all of the terms and conditions set forth in the applicable warrant agreement, including the obligation to make the payments contemplated thereby upon exercise thereof.

# ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Subject to Section 11.5, (a) except as disclosed in the Company SEC Documents filed with or furnished to the SEC prior to the date of this Agreement; provided, that in no event shall any risk factor disclosure under the heading "Risk Factors" or disclosure set forth in any "forward looking statements" disclaimer or other general statements to the extent they are cautionary, predictive or forward looking in nature that are included in any part of any Company SEC Document be deemed to be an exception to, or, as applicable, disclosure for purposes of, any representations and warranties of the Company contained in this Agreement or (b) except as set forth in the Company Disclosure Letter, the Company represents and warrants to Parent and Merger Sub Inc. that:

Section 4.1 Corporate Existence and Power. The Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware. The Company has all corporate power and

authority to carry on its business as now conducted and is duly qualified to do business as a foreign corporation and is in good standing in each jurisdiction where such qualification is necessary, except where any failure to have such power or authority or to be so qualified would not reasonably be expected, individually or in the aggregate, to have a Company Material Adverse Effect. The copies of the certificate of incorporation and bylaws of the Company filed as exhibits to the Company's reports, schedules, forms, statements, and other documents required to be filed or furnished by the Company, under the 1933 Act and the 1934 Act are true and complete copies as of the date of this Agreement. The Company is not in violation of, in conflict with, or in default under, its certificate of incorporation or bylaws.

Section 4.2 Corporate Authorization. The execution, delivery and performance by the Company of this Agreement and the consummation by the Company of the transactions contemplated hereby are within the Company's corporate power and authority and have been duly authorized by all necessary corporate action on the part of the Company, other than, with respect to the Merger, obtaining the Company Stockholder Approval and filing the Certificate of Merger with the Secretary of State of the State of Delaware as required by the DGCL. The Company Stockholder Approval is the only vote of the holders of shares of Company Stock or other capital stock of the Company necessary to adopt this Agreement and consummate the Merger under applicable Law or the certificate of incorporation or bylaws of the Company. This Agreement has been duly and validly executed and delivered by the Company, assuming due authorization, execution and delivery by Parent and Merger Sub Inc., constitutes a valid and binding agreement of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting creditors' rights generally and by general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at Law). The Company Special Committee, at a meeting duly called and held, has unanimously (i) determined that this Agreement and the transactions contemplated hereby, including the Merger, are fair to, and in the best interests of, the Company and the Company Unaffiliated Stockholders, (ii) approved and declared advisable this Agreement and the transactions contemplated hereby, including the Merger, (iii) resolved, subject to the terms and conditions set forth in this Agreement, to recommend that the Company Board (x) determine that this Agreement and the transactions contemplated hereby, including the Merger, are fair to, and in the best interests of, the Company and the Company Unaffiliated Stockholders and (y) approve and declare advisable this Agreement and the transactions contemplated by this Agreement, including the Merger, and (iv) resolved to recommend that, subject to such Company Board approval and the terms and conditions set forth in this Agreement, the Company Board direct that this Agreement be submitted to the holders of Company Stock for their adoption and approval and recommend that the holders of Company Stock approve and adopt this Agreement and the transactions contemplated by this Agreement, including the Merger. The Company Board, at a meeting duly called and held and upon receipt and recommendation of the Company Special Committee, has unanimously (i) determined that this Agreement and the transactions contemplated hereby, including the Merger, are fair to, and in the best interests of, the Company and the Company Unaffiliated Stockholders, (ii) approved and declared advisable this Agreement and the transactions contemplated hereby, including the Merger, (iii) approved and declared advisable the execution and delivery by the Company of this Agreement, the performance by the Company of its covenants and agreements contained herein and the consummation of the transactions contemplated by this Agreement, including the Merger, upon the terms, and subject to the conditions, contained herein, (iv) directed that this Agreement be submitted to the holders of Company Stock for their adoption and approval and (v) resolved, subject to the terms and conditions set forth in this Agreement, to make the Company Board Recommendation. Each of the Support Agreements is in full force and effect and has not been rescinded, modified or withdrawn in any way.

Section 4.3 Governmental Authorization. The execution, delivery and performance by the Company of this Agreement and the consummation by the Company of the transactions contemplated hereby require no action by or in respect of, or filing with, any Governmental Authority, other than (i) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware and appropriate documents with the relevant authorities of other states in which the Company is qualified to do business, (ii) compliance with any applicable requirements of the HSR Act and any non-U.S. Competition Laws, (iii) compliance with any applicable

requirements of the 1933 Act, the 1934 Act and any other applicable state or federal securities laws, (iv) compliance with any applicable requirements of Nasdaq and (v) any actions or filings the absence of which would not reasonably be expected to have, individually or in the aggregate, (x) a Company Material Adverse Effect, or (y) an effect that would prevent, materially delay or materially impair the ability of the Company to perform its obligations under this Agreement or to consummate the Merger.

Section 4.4 Non-contravention. The execution, delivery and performance by the Company of this Agreement and the consummation of the transactions contemplated hereby do not and will not (i) assuming the authorizations, consents and approvals referred to in clauses (i) through (iv) of Section 4.3 are obtained, (A) contravene, conflict with, or result in any violation or breach of any provision of the certificate of incorporation or bylaws of the Company, (B) contravene, conflict with or result in a violation or breach of any provision of any Law or Order or (C) require any consent or other action by any Person under, constitute a default or a violation, or an event that, with or without notice or lapse of time or both, would constitute a default or a violation, under or of, or cause or permit the termination, cancellation, acceleration or other change of any right or obligation or the loss of any benefit to which the Company or any of its Subsidiaries is entitled under, any provision of any Contract binding upon the Company or any of its Subsidiaries or any license, franchise, permit, certificate, approval or other similar authorization affecting, or relating in any way to, the assets or business of the Company and its Subsidiaries or (ii) result in the creation or imposition of any Lien, other than any Permitted Lien, on any asset of the Company or any of its Subsidiaries, except, in the case of clauses (i)(B), (i)(C) and (ii), which have not had, and would not reasonably be expected to have, individually or in the aggregate, (x) a Company Material Adverse Effect or (y) an effect that would prevent, materially delay or materially impair the ability of the Company to perform its obligations under this Agreement or to consummate the Merger.

## Section 4.5 Capitalization.

- (a) The authorized capital stock of the Company consists of 400,000,000 shares of Company Stock, 20,000,000 shares of Class B common stock of the Company and 10,000,000 shares of Class F common stock of the Company, each with a par value of \$0.0001 per share and 10,000,000 shares of preferred stock of the Company ("Company Preferred Stock"), including 55,000 shares of Series A Redeemable Convertible Preferred Stock ("Series A Preferred Stock") and 22,050 shares of Series B Redeemable Convertible Preferred Stock. As of June 21, 2022 ("Company Capitalization Date"), (i) there were issued and outstanding (A) 77,060,612 shares of Company Stock and (B) 19,610 shares of Series A Preferred Stock, and (C) Company DSUs with respect to an aggregate of 2,052,474 shares of Company Stock and in of which were issued under the Company LTIP, and (ii) 5,414,193 shares of Company Stock are reserved under the Company LTIP. All the outstanding shares of Company Stock are, and all shares of Company Stock reserved for issuance as described above shall be, when issued in accordance with the respective terms thereof, duly authorized, validly issued, fully paid and nonassessable and free of preemptive rights.
- (b) Section 4.5(b) of the Company Disclosure Letter, sets forth a true and complete list, of (i) each Company Equity Award, (ii) the name of the Company Equity Award holder, (iii) the number of shares of Company Stock underlying each Company Equity Award, (iv) in the case of any Pool A Performance Award or Pool B Performance Award, the Accumulated Cash Award Value, (v) the date on which the Company Equity Award was granted (vi) the vesting schedule with respect to the Company Equity Award, including any right of acceleration of such vesting schedule, (vii) the exercise price of each Company Equity Award, if applicable, and (viii) the expiration date of each Company Equity Award, if applicable. The Company LTIP permits the treatment of Company Equity Awards described in Article III.
- (c) Section 4.5(c) of the Company Disclosure Letter sets forth a true and complete list, of (i) each Warrant, (ii) the name of the Warrant holder, (iii) the number of shares of Company Stock underlying such Warrant and (iv) the exercise price of such Warrant.

- (d) Section 4.5(d) of the Company Disclosure Letter sets forth a true and complete list of (i) each Equity Linked Convertible Note, (ii) the name of the Equity Linked Convertible Note holder, (iii) the amount outstanding under such Equity Linked Convertible Note and (iv) the conversion price of such Equity Linked Convertible Note.
- (e) Except (w) as set forth in Section 4.5(e) of the Company Disclosure Letter, (x) for any awards issued pursuant to the Company LTIP after the date of this Agreement in accordance with the terms of this Agreement (all of which will be set forth on an updated Company Disclosure Letter delivered five (5) Business Days prior to the Closing and which otherwise will comply with the last sentence of Section 4.5(a)), (y) for any shares of Company Stock issued upon the exercise of any Company Equity Award, in each case, that were outstanding on the Company Capitalization Date or subsequently granted under the Company LTIP or otherwise in accordance with the terms of this Agreement and (z) the Warrants and the Equity Linked Convertible Notes, there are no issued, reserved for issuance or outstanding (i) shares of capital stock or other voting securities of or other ownership interests in the Company, (iii) warrants, calls, options or other rights to acquire from the Company, or other voting securities of or other ownership interests in the Company or (iv) restricted shares, stock appreciation rights, performance units, restricted stock units, contingent value rights, "phantom" stock or similar securities or rights issued or granted by the Company or any of its Subsidiaries that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of, any shares of capital stock or other voting securities in the Company (the items in clauses (i) through (iv) being referred to collectively as the "Company Securities").
- (f) There are no outstanding obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any of the Company Securities, and neither the Company nor any of its Subsidiaries maintains an employee stock purchase plan. Neither the Company nor any of its Subsidiaries is a party to any voting trust, proxy, voting agreement or other similar agreement with respect to the voting of any Company Securities. All outstanding shares of capital stock of the Company have been, and all shares that may be issued pursuant to any equity compensation plan or arrangement will be, when issued in accordance with the respective terms thereof, duly authorized and validly issued, fully paid and nonassessable and free of preemptive rights. No Subsidiary of the Company owns any shares of capital stock of the Company or any Company Securities. There are no outstanding bonds, debentures, notes or other indebtedness of the Company having the right to vote (whether on an as-converted basis or otherwise) (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which stockholders of the Company may vote.

## Section 4.6 Subsidiaries.

- (a) Each Subsidiary of the Company is an entity duly incorporated or otherwise duly organized, validly existing and (where applicable or recognized) in good standing under the laws of its jurisdiction of incorporation or organization, except, in the case of any such Subsidiary, where the failure to be so incorporated, organized, existing or in good standing would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Each Subsidiary of the Company has all corporate, limited liability company or comparable powers and all Governmental Authorizations required to carry on its business as now conducted, except for those powers or Governmental Authorizations the absence of which has not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Each such Subsidiary is duly qualified to do business as a foreign entity and is in good standing in each jurisdiction where such qualification is necessary, except for those jurisdictions where failure to be so qualified has not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (b) All of the outstanding capital stock or other voting securities of or other ownership interests in each Subsidiary of the Company are owned by the Company, directly or indirectly, free and clear of any Lien and free

of any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose of such capital stock or other voting securities or other ownership interests), in each case other than (x) statutory or other liens for Taxes or assessments which are not yet due or delinquent or the validity of which is being contested in good faith by appropriate proceedings and for which adequate reserves are being maintained in accordance with GAAP, (y) transfer and other restrictions under applicable federal and state securities Laws and (z) in the case of Subsidiaries that are immaterial to the Company and its Subsidiaries, taken as a whole, immaterial Liens. Section 4.6(b) of the Company Disclosure Letter contains a complete and accurate list of the Subsidiaries of the Company, including, for each of the Subsidiaries, (i) its name and (ii) its jurisdiction of organization. Each Subsidiary is directly or indirectly wholly owned by the Company. There are no issued, reserved for issuance or outstanding (i) securities of the Company or any of its Subsidiaries convertible into, or exchangeable for, shares of capital stock or other voting securities of or other ownership interests in any Subsidiary of the Company, (ii) warrants, calls, options or other rights to acquire from the Company or any of its Subsidiaries, or other obligations of the Company or any of its Subsidiaries to issue, any shares of capital stock or other voting securities of or other ownership interests in or any securities convertible into, or exchangeable for, any shares of capital stock or other voting securities of or other ownership interests in any Subsidiary of the Company or (iii) restricted shares, stock appreciation rights, performance units, contingent value rights, "phantom" stock or similar securities or rights issued or granted by the Company or any of its Subsidiaries that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of, any capital stock or other voting securities of or other ownership interests in any Subsidiary of the Company (the items in clauses (i) through (iii) being referred to collectively as the "Company Subsidiary Securities"). There are no outstanding obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any of the Company Subsidiary Securities.

#### Section 4.7 SEC Filings and the Sarbanes-Oxley Act

- (a) The Company has filed with or furnished to the SEC (including following any extensions of time for filing provided by Rulel 2b-25 promulgated under the 1934 Act) all reports, schedules, forms, statements, prospectuses, registration statements and other documents required to be filed or furnished, as the case may be, by the Company since January 1, 2019 (collectively, together with any schedules thereto and other information incorporated therein, the "Company SEC Documents").
- (b) As of its filing date (or, if amended or supplemented, as of the date of the most recent amendment or supplement filed prior to the date of this Agreement), each Company SEC Document complied as to form in all material respects with the applicable requirements of the 1933 Act and the 1934 Act and the Sarbanes-Oxley Act, and any rules and regulations promulgated thereunder, as the case may be.
- (c) As of its filing date (or, if amended or supplemented, as of the date of the most recent amendment or supplement filed prior to the date of this Agreement), each Company SEC Document filed or furnished pursuant to the 1934 Act did not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.
- (d) Each Company SEC Document that is a registration statement, as amended or supplemented, if applicable, filed pursuant to the 1933 Act, as of the date such registration statement or amendment became effective, and as of the date of such supplement, did not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading.
- (e) The Company has heretofore furnished or made available to Parent complete and correct copies of all comment letters received from the SEC or its staff from January 1, 2019 through the date of this Agreement with respect to any of the Company SEC Documents, together with all written responses of the Company thereto, to the extent that such comment letters and written responses are not publicly available on EDGAR. As of the date of this Agreement, there are no outstanding or unresolved comments in comment letters received from the SEC

or its staff with respect to any of the Company SEC Documents, and, to the knowledge of the Company, none of the Company SEC Documents are subject to ongoing SEC review.

- (f) The Company has established and maintains disclosure controls and procedures (as defined in Rule 13a-15 under the 1934 Act). Such disclosure controls and procedures are reasonably designed to ensure that material information relating to the Company, including its consolidated Subsidiaries, is made known to the Company's principal executive officer and its principal financial officer by others within those entities and such disclosure controls and procedures are reasonably designed to ensure that all such information is communicated in a timely fashion to the Company's principal executive officer and principal financial officer such that timely decisions may be made regarding the disclosure of such information in the Company's periodic and current reports required under the 1934 Act. For purposes of this Agreement, "principal executive officer" and "principal financial officer" shall have the meanings given to such terms in the Sarbanes-Oxley Act.
- (g) The Company and its Subsidiaries have established and maintained a system of Internal Controls over financial reporting (as defined in Rule 13a-15 under the 1934 Act, "Internal Controls") sufficient to provide reasonable assurance regarding the reliability of the Company's financial reporting and the preparation of Company financial statements for external purposes in accordance with GAAP. The Company has disclosed, based on its most recent evaluation of Internal Controls prior to the date of this Agreement, to the Company's auditors and audit committee (i) any significant deficiencies and material weaknesses in the design or operation of Internal Controls which are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information and (ii) any fraud, whether or not material, that involves management or other employees who have a significant role in Internal Controls.
- (h) Neither the Company nor any of its Subsidiaries has extended or maintained credit, arranged for the extension of credit, or renewed an extension of credit, in the form of a personal loan to or for any executive officer (as defined in Rule 3b-7 under the 1934 Act) or director of the Company in violation of Section 402 of the Sarbanes-Oxley Act.
- (i) The Company is in compliance, and has complied since January 1, 2019, in each case in all material respects, with (i) the applicable provisions of the Sarbanes-Oxley Act and (ii) the applicable listing and corporate governance rules and regulations of Nasdaq.
- (j) Each of the principal executive officer and principal financial officer of the Company (or each former principal executive officer and principal financial officer of the Company, as applicable) has made all certifications required by Rule 13a-14 and 15d-14 under the 1934 Act and Sections 302 and 906 of the Sarbanes-Oxley Act and any related rules and regulations promulgated by the SEC and Nasdaq, and the statements contained in any such certifications are complete and correct in all material respects.
- (k) Since the Company Balance Sheet Date through the date of this Agreement, there has been no transaction, or series of similar transactions, agreements, arrangements or understandings, nor is there any proposed transaction as of the date of this Agreement, or series of similar transactions, agreements, arrangements or understandings to which the Company or any of its Subsidiaries was or is to be a party, that would be required to be disclosed under Item 404 of Regulation S-K promulgated under the 1933 Act that has not been disclosed in the Company SEC Documents publicly filed or furnished with the SEC following the Company Balance Sheet Date.
- Section 4.8 Financial Statements. The audited consolidated financial statements and unaudited consolidated interim financial statements of the Company included or incorporated by reference in the Company SEC Documents (including all related notes and schedules thereto) (a) fairly present in all material respects, in conformity with GAAP (except, in the case of unaudited consolidated interim financial statements, as permitted by the SEC's Form 10-Q) applied on a consistent basis (except as may be indicated therein or in the notes thereto), the consolidated financial position of the Company and its consolidated Subsidiaries as of the dates

thereof and their consolidated results of operations and cash flows for the periods then ended (subject to normalyear-end audit adjustments in the case of any unaudited interim financial statements), (b) comply in all material respects with the applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto, and (c) have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto and except, in the case of the unaudited interim statements, as may be permitted under the SEC's Form 10-Q).

Section 4.9 Information Supplied. The information relating to the Company and its Subsidiaries to be contained in, or incorporated by reference in, the Registration Statement, Proxy Statement and Information Statement will not, (i) on the date the Proxy Statement is first mailed to the holders of Company Stock or at the time of the Company Stockholders' Meeting, (ii) on the date that the Information Statement is provided to holders of Parent Stock or (iii) at the time the Registration Statement is declared effective by the SEC, contain any untrue statement of any material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not false or misleading. The Registration Statement, Proxy Statement and Information Statement will comply in all material respects as to form with the requirements of the 1933 Act and the 1934 Act, as applicable, and the rules and regulations promulgated thereunder. Notwithstanding the foregoing provisions of this Section 4.9, no representation or warranty is made by the Company with respect to information or statements made or incorporated by reference in the Registration Statement, Proxy Statement or Information Statement that were not supplied by or on behalf of the Company for use therein.

## Section 4.10 Absence of Certain Changes.

- (a) From the Company Balance Sheet Date through the date of this Agreement, (i) the business of the Company and its Subsidiaries has been conducted in the ordinary course of business in all material respects, and (ii) there has not been any action taken by the Company or any of its Subsidiaries that, if taken during the period from the date of this Agreement through the Effective Time without Parent's consent, would constitute a breach of clauses (a), (b), (e), (f), (g), (h), (k), (o), (p) or (s) of Section 6.1.
- (b) Since the Company Balance Sheet Date through the date of this Agreement, there has not been any effect, change, condition, fact, event, occurrence or development that has had or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- **Section 4.11** No Undisclosed Material Liabilities. There are no liabilities or obligations of the Company or any of its Subsidiaries that would be required by GAAP, as in effect on the date hereof, to be reflected on the consolidated balance sheet of the Company (including the notes thereto), other than:
  - (a) liabilities or obligations disclosed, reflected, reserved against or otherwise provided for in the Company Balance Sheet or in the notes thereto;
  - (b) liabilities or obligations incurred in the ordinary course of business since the Company Balance Sheet Date;
  - (c) liabilities or obligations arising out of this Agreement or the transactions contemplated hereby; and
- (d) liabilities or obligations that have not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

#### Section 4.12 Compliance with Laws and Court Orders; Governmental Authorizations

(a) Except for matters that have not had, and would not reasonably be expected to have, individually or in the aggregate, (i) a Company Material Adverse Effect or (ii) an effect that would prevent, materially delay or

materially impair the ability of the Company to perform its obligations under this Agreement or to consummate the Merger, the Company and each of its Subsidiaries are, and since January 1, 2019, have been in compliance with all applicable Laws and Orders, and to the knowledge of the Company, is not under investigation by any Governmental Authority with respect to any Law or Order. There is no Order of any Governmental Authority outstanding against the Company or any of its Subsidiaries that is material to the Company and its Subsidiaries, taken as a whole.

(b) Except as would not reasonably be expected to have, individually or in the aggregate, (x) a Company Material Adverse Effect or (y) an effect that would prevent, materially delay or materially impair the ability of the Company to perform its obligations under this Agreement or to consummate the Merger, the Company and each of its Subsidiaries has all Governmental Authorizations necessary for the ownership and operation of its business as presently conducted, and each such Governmental Authorization is in full force and effect. Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, (i) the Company and each of its Subsidiaries are, and since January 1, 2019, have been in compliance with the terms of all Governmental Authorizations necessary for the ownership and operation of its businesses and (ii) since January 1, 2019, neither the Company nor any of its Subsidiaries has received written notice from any Governmental Authority alleging any conflict with or breach of any such Governmental Authorization.

Section 4.13 Litigation. Except as has not had and would not reasonably expected to have, individually or in the aggregate, (x) a Company Material Adverse Effect or (y) an effect that would prevent, materially delay or materially impair the ability of the Company to perform its obligations under this Agreement or to consummate the Merger, there is no Proceeding or, to the knowledge of the Company, investigation, pending against, or, to the knowledge of the Company, threatened by or against, the Company, any of its Subsidiaries, or to the knowledge of the Company and insofar as any such Proceeding or investigation relates to the Company or any of its Subsidiaries, any present or former officer, director, worker or employee of the Company or any of its Subsidiaries or any other Person for whom the Company or any of its Subsidiaries may be liable before (or, in the case of threatened Proceedings or investigations, that would be before) or by any Governmental Authority.

#### Section 4.14 Properties.

- (a) Section 4.14(a) of the Company Disclosure Letter sets forth a list of the addresses of the material real properties owned by the Company or any of its Subsidiaries as of the date of this Agreement (the "Owned Real Property") and the legal name of the respective owner(s) of each Owned Real Property.
- (b) Section 4.14(b) of the Company Disclosure Letter sets forth, as of the date of this Agreement, a list of the material leases, subleases, licenses or other occupancies to which the Company or any of its Subsidiaries is a party as tenant for real property (the "Real Property Leases" and, together with the Owned Real Property, the "Company Real Property").
- (c) The Company or one of its Subsidiaries owns good and marketable fee simple title or valid leasehold title (as applicable) to the Company Real Property and all property and assets reflected on the Company Balance Sheet or acquired after the Company Balance Sheet Date, in each case, free and clear of all Liens, except (i) for Permitted Liens, (ii) for the property and assets that have been disposed of or leased since the Company Balance Sheet Date in the ordinary course of business consistent with past practice and (iii) in respects that would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (d) Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, each Real Property Lease under which the Company or any of its Subsidiaries leases, subleases, licenses or otherwise occupies any real property is valid, binding and in full force and effect. Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, neither the Company nor any of its Subsidiaries has violated any provision of, or taken or failed to take any act which, with or without notice, lapse of time, or both, would constitute a default under the provisions of such Real

Property Lease, and neither the Company nor any of its Subsidiaries has received notice that it has breached, violated or defaulted under any Real Property Lease

#### Section 4.15 Intellectual Property.

- (a) The Company and/or one or more of its Subsidiaries are the sole owners of each item of material Owned Intellectual Property Rights and material Owned Technology, free and clear of any Liens, except for Permitted Liens. As of the date hereof and except as indicated on Schedule 4.15, no item of material Owned Intellectual Property Rights or material Owned Technology is subject to any outstanding injunction, judgment, order, decree or ruling of which the Company has received written notice, and no Claim of which the Company has received written notice is pending before any court or arbitrator or was threatened in writing that challenges the validity, enforceability, or the Company's ownership of any item of material Owned Intellectual Property Rights or material Owned Technology, as applicable, which remains unresolved as of the date hereof. Section 4.15(a) of the Company Disclosure Letter contains a complete and accurate list, as of the date hereof, of (i) all material Owned Intellectual Property Rights that have been registered, issued, or applied for with any Governmental Authority and (ii) all material unregistered Marks and current invention disclosures. All fees currently due, as of the signing date of this Agreement, for maintaining any Owned Intellectual Property Rights shall have been paid in full to the proper Governmental Authority to the extent necessary to avoid lapse or abandonment of such Owned Intellectual Property Rights.
- (b) Material Licensed Intellectual Property Rights and the material Licensed Technology are duly and validly licensed to the Company or its Subsidiaries under the Inbound Intellectual Property Contracts for use in the manner as used in the past and currently used by the Company and its Subsidiaries in the conduct of the business as used in the past and as currently conducted. Each material Inbound Intellectual Property Contract is valid, binding, and in full force and effect. There are no restrictions on the transfer or assignment by the Company or its Subsidiaries of any material Inbound Intellectual Property Contract required to be listed on Section 4.15(b) of the Company Disclosure Letter which would cause the Merger (or any related transactions) to impair any rights of the Company or any of its Subsidiaries under such Inbound Intellectual Property Contract. Neither the Company nor any of its Subsidiaries will, as a result of the execution and delivery of this Agreement or the performance of the Company's and each of its Subsidiaries' obligations hereunder, lose any material rights to Exploit any material Licensed Intellectual Property Rights or material Licensed Technology pursuant to any material Inbound Intellectual Property Contract. As of the date hereof, neither the Company nor any of its Subsidiaries has received notice, or has any knowledge, that any party to any material Inbound Intellectual Property Contract intends to cancel, terminate or refuse to renew (if renewable) such Inbound Intellectual Property Contract, or to exercise or decline to exercise any option or right thereunder. Section 4.15(b) of the Company Disclosure Letter contains a complete and accurate list of all (i) material Inbound Intellectual Property Contracts, other than (x) licenses for generally available software with an annual fee of under \$25,000 that is licensed to the Company or a Subsidiary in object code form only and that is not incorporated into or otherwise used in connection with the use of any Company Product, and (y) confidentiality and non-disclosure agreements entered into in the ordinary course of business. For clarity, and without limiting the foregoing sentence, Section 4.15(b) of the Company Disclosure Letter includes a complete and accurate list of all Inbound Intellectual Property Contracts pursuant to which the Company or a Subsidiary is granted a license to material software (other than Open Source Software) that is incorporated into or otherwise used in connection with the use of any Company Product.
- (c) Each material Outbound Intellectual Property Contract is valid, binding, and in full force and effect. There are no restrictions on the transfer or assignment by the Company or its Subsidiaries of any material Outbound Intellectual Property Contract required to be listed on Section 4.15(c) of the Company Disclosure Letter which would cause the Merger (or any related transactions) to impair any rights of the Company or any of its Subsidiaries under such Outbound Intellectual Property Contract. Neither the Company nor any of its Subsidiaries will, as a result of the execution and delivery of this Agreement or the performance of the Company's and each of its Subsidiaries' obligations hereunder, lose any material rights to Exploit any material

Owned Intellectual Property Rights or material Owned Technology pursuant to any material Outbound Intellectual Property Contract. As of the date hereof, neither the Company nor any of its Subsidiaries has received notice, or has any knowledge, that any party to any material Outbound Intellectual Property Contract intends to cancel, terminate or refuse to renew (if renewable) any such Outbound Intellectual Property Contract. Section 4.15(c) of the Company Disclosure Letter contains a complete and accurate list of all material Outbound Intellectual Property Contracts (i) with any industry standards bodies, patent pools or similar organizations of which the Company or any of its Subsidiaries has been a founder, member or promotor of, or a contributor to, and which require the licensing of Intellectual Property Rights or Technology by the Company or any its Subsidiaries, including any obligations to license such Intellectual Property Rights or Technology on a royalty-free, RAND or FRAND basis; (ii) pursuant to which any material source code included in the material Owned Technology or included in any other material Technology used in or incorporated in any Company Product is licensed or otherwise made available to any other Person (other than to current or former employees within the scope of their employment or independent contractors acting for the benefit and account of the Company or any of its Subsidiaries, in each case, solely during the time period that they are employed or engaged by the Company or any of its Subsidiaries) or that include any provision for source code escrow; (iii) pursuant to which any material rights, licenses, or covenants not to sue are expressly granted with respect to Patents of the Company or any of its Subsidiaries to any other Person, including any Patent cross-licenses or Patent licenses granted in settlement agreements; (iv) pursuant to which any exclusive rights, licenses, or covenants have been or may be granted by the Company or any of its Subsidiaries, whether by region, Company Product or otherwise; (v) pursuant to which any material covenant not to sue is granted by the Company or any of its Subsidiaries to any other Person; (vi) pursuant to which the Company or any of its Subsidiaries have sold, disposed of, or otherwise transferred ownership of any material Intellectual Property Rights or material Technology that was an Owned Intellectual Property Right or Owned Technology in the three (3) year period prior to the date hereof.

(d) The Company Intellectual Property Rights constitute in all material respects all of the Intellectual Property Rights used in, held for use in, and/or necessary for the conduct of the business of the Company or any of its Subsidiaries as currently conducted. To the knowledge of the Company, the Company Technology constitutes in all material respects all of the material Technology used in, held for use in, and/or necessary for the conduct of the business of the Company or any of its Subsidiaries as currently conducted. Except for material Outbound Intellectual Property Contracts listed on Section 4.15(d) of the Company Disclosure Letter, as of the date hereof, (i) neither the Company nor any of its Subsidiaries has granted any exclusive license of or right to use, or authorized the retention of any exclusive rights to use, any material Intellectual Property Right or material Technology to any Person that are still in effect, (ii) in the three (3) year period prior to the date hereof, neither the Company nor any of its Subsidiaries has transferred to any Person ownership (whether exclusive or joint) of any material Intellectual Property Right or material Technology that was an Owned Intellectual Property Right or Owned Technology and (iii) neither the Company nor any of its Subsidiaries has granted any exclusive licenses to any Intellectual Property Rights or Technology, or granted any other licenses to any Intellectual Property Rights or Technology on a non-exclusive basis in the ordinary course of the business.

(e) The Company Products (and the Exploitation thereof by the Company and its Subsidiaries) and the operations of the Company and its Subsidiaries have not and do not infringe, misappropriate or otherwise violate any Intellectual Property Rights of a third party (provided that the representation and warranty provided in this first sentence of Section 4.15(e) is to the knowledge of the Company with respect to Patents of a Third Party). Section 4.15(e) of the Company Disclosure Letter lists (i) any written Claim or other written complaint, notice or threat of any of the foregoing (including any written notification that a license under any Patent is or may be required) received by the Company or any of its Subsidiaries in the three (3) years prior to the date hereof or, subject to the knowledge of the Company, earlier to the extent it is unresolved as of the date hereof, alleging any infringement, misappropriation or violation of any material Intellectual Property Rights of a third party, and (ii) any written request or demand for indemnification or defense of a material Intellectual Property infringement Claim received by the Company or any of its Subsidiaries in the three (3) years prior to the date hereof, or, subject to the knowledge of the Company, earlier to the extent unresolved as of the date hereof, from any reseller, distributor, channel partner or end-customer of a Company Product.

- (f) There are no royalties or other similar payments that are payable by the Company or any of its Subsidiaries as of the date hereof to any third Person (excluding, for the avoidance of doubt, salaries that are payable to employees, and contractor fees that are payable to independent contractors) for the Exploitation of any material Intellectual Property Right or material Technology of such third Person by the Company or any of its Subsidiaries, excluding any royalties or other payments that are due pursuant to licenses for generally available software with an annual fee of under \$25,000 that is licensed to the Company or a Subsidiary in object code form only and that is not incorporated into or used in connection with the use of any Company Product. Without limiting the foregoing, Section 4.15(f) of the Company Disclosure Letter sets forth the cumulative amount due or payable as of the date hereof by the Company or any of its Subsidiaries to its employees and contractors, (i) under any invention reward program, Technology development incentive program, or any other similar program or arrangement, or (ii) under any applicable Laws related to the creation or development of inventions or other Technology (and any and all Intellectual Property Rights thereto).
- (g) Each employee, worker, contractor, subcontractor, and consultant of the Company or its Subsidiaries that is involved in the development or creation of material Intellectual Property Rights or Technology by or on behalf of the Company or any of its Subsidiaries or from whom the Company or any of its Subsidiaries have acquired ownership of any material Intellectual Property Rights or Technology have, in each case, executed and delivered a valid, binding and enforceable written agreement expressly assigning and agreeing to assign to the Company or the applicable Subsidiary all of such employee's, worker's, contractor's, subcontractor's, or consultant's right, title and interest in any material Intellectual Property Rights or Technology created by such employee, worker, contractor, subcontractor, or consultant within the scope of his or her employment or engagement or otherwise acquired from such employee, contractor, subcontractor or consultant. In the three (3) years prior to the date hereof or, subject to the knowledge of the Company, earlier to the extent unresolved as of the date hereof, no Person (including, without limitation, any current or former employee or consultant of the Company or its Subsidiaries) has brought any Claim before any Governmental Authority or arbitral tribunal against the Company or any of its Subsidiaries or otherwise regarding the ownership of any Intellectual Property Rights or Technology, or otherwise has asserted in writing or has alleged in writing to own any Owned Intellectual Property Rights or Owned Technology.
- (h) The Company takes commercially reasonable actions, consistent with industry standards, to protect the confidentiality and security of its information technology systems and confidential or proprietary data (including, without limitation, any Trade Secrets of the Company or its Subsidiaries). To the knowledge of the Company, there have been no violations or unauthorized disclosure or use of or unauthorized access to any information technology systems or any confidential or proprietary data (including, without limitation, any Trade Secrets of the Company or its Subsidiaries) and all items disclosed (or required to be disclosed) on Section 4.15(h) of the Company Disclosure Letter occurring within the three (3) years prior to the date hereof have been or will be satisfactorily and finally resolved or mitigated by the End Date with no material adverse effect on the Company or any of its Subsidiaries. Without limiting the foregoing, as of the date hereof, neither the Company nor any of its Subsidiaries have (A) disclosed Trade Secrets or other material confidential or proprietary information of the Company or any of its Subsidiaries to any Person, unless such disclosure was under an appropriate written nondisclosure agreement or to a Person subject to a fiduciary duty to maintain the confidentiality thereof, or (B) deposited, disclosed or delivered to any Person outside of the Company or its Subsidiaries (for clarity, excluding current or former employees within the scope of their employment or independent contractors acting for the benefit and account of the Company or any of its Subsidiaries, in each case, solely during the time period that they are employed or engaged by the Company or any of its Subsidiaries only one of the company of the company Product. To the knowledge of the Company, neither the Company nor any of its Subsidiaries has been in breach of any obligations of confidentiality that they may have with respect to any material Trade Secrets or other material confidential or proprietary information.

- (i) Neither the Company nor any of its Subsidiaries has at any time used or otherwise Exploited any Open Source Software in such a way that creates material obligations of the Company or any of its Subsidiaries with respect to any material Owned Intellectual Property Rights or material Owned Technology or grants or purports to grant to any third Person any rights or immunities under or with respect to any material Owned Intellectual Property Rights or material Owned Technology. Neither the Company nor any of its Subsidiaries is in material violation of any license for Open Source Software. Notwithstanding any other provision hereof, no material Company Technology used or otherwise Exploited by the Company or any of its Subsidiaries, including any material Technology used in or incorporated in any Company Product, is subject to any license or other agreement that (1) requires the disclosure or distribution in source code form of any material Owned Intellectual Property Rights or material Owned Technology, including any portion of any Company Product other than such Open Source Software; (2) requires the licensing of any material Owned Intellectual Property Rights or material Owned Technology, or any portion of any Company Product other than such Open Source Software, for the purpose of making derivative works; (3) imposes any restriction on the consideration to be charged for the licensing or distribution of any material Owned Intellectual Property Rights, material Owned Technology, or Company Product or otherwise limits the Company's or its Subsidiaries' freedom to seek full compensation in connection with the marketing, licensing or distribution of any of their products or services; (4) creates obligations for the Company or any of its Subsidiaries with respect to material Owned Intellectual Property Rights or material Owned Technology or grants to any Person any rights or immunities under material Owned Intellectual Property Rights or material Owned Technology; (5) imposes any other limitation, restriction or condition on the right of the Company or any of its Subsidiaries to use or distribute any material Owned Intellectual Property Rights, material Owned Technology, or Company Product; or (6) allows a third party to decompile, disassemble or otherwise reverse engineer any material Company Intellectual Property Rights or re-license any material Company Intellectual Property Rights.
- (j) Except as set forth on Section 4.15(j) of the Company Disclosure Letter, to the knowledge of the Company, the material Company Technology and Company Products are free of material defects and errors in features, functionality, performance, programming and operation, and do not contain any disabling codes or instructions, spyware, Trojan horses, worms, viruses or other harmful, malicious, or destructive code or any software routines that are designed to permit or cause unauthorized access to, or material disruption, impairment, disablement, or destruction of, software, data, systems or other materials.
- (k) Neither the execution, delivery nor performance of this Agreement nor the consummation of the Merger or any of the other transactions contemplated hereby will (i) result in the Company or any of its Subsidiaries granting to any Person any right to, or with respect to, any material Company Intellectual Property Rights, material Company Technology, or Company Products; (ii) require the consent of any other Person in respect of the Company's and its Subsidiaries' rights to own, use, and/or hold for use any material Company Intellectual Property Rights or Company Technology; (iii) impair the right of the Company or any of its Subsidiaries to Exploit any material Company Intellectual Property Rights or Company Technology or result in any loss of, or the diminishment in value of, any material Company Intellectual Property Rights or Company Technology; or (iv) result in the Company or any of its Subsidiaries being obligated to pay (or increase the amount of) any material royalties or other material amounts to any Person.
- (l) Section 4.15(l) of the Company Disclosure Letter lists any written request or demand for indemnification or defense of a material Intellectual Property infringement Claim received by the Company or any of its Subsidiaries, in the three (3) years prior to the date hereof or, subject to the knowledge of the Company, earlier to the extent unresolved as of the date hereof, from end-customers, distributors, resellers or channel partners based on Company Products, which, in each case, remains unresolved as of the date hereof.
- (m) Notwithstanding anything to the contrary set forth herein, except as expressly set forth in Section 4.10, Section 4.13 or Section 4.20, this Section 4.15 contains all of the representations and warranties provided by the Company with respect to matters related to Intellectual Property Rights (other than those rights relating to privacy and any rights similar, which are addressed in Section 4.16).

Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, all Company information technology systems are in good working condition and are sufficient in all material respects for the operation of the Company's businesses as currently conducted, including as to capacity, scalability, and ability to process current peak volumes in a timely manner, and the Company and its Subsidiaries will continue to have rights to all Company information technology systems immediately after the Closing. Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, the Company and its Subsidiaries have (a) taken commercially reasonable steps to safeguard the confidentiality, availability, security, and integrity of the Company information technology systems, including software and hardware support arrangements, and (b) implemented and maintain commercially reasonable backup, data recovery, disaster recovery, and business continuity plans, procedures and facilities, and test such plans and procedures on a regular basis consistent with industry standards, and such plans and procedures have been proven effective in all material respects upon such testing. Except as set forth on Section 4.15 of the Company Disclosure Letter, the Company and its Subsidiaries have not been subjected to an audit of any kind pursuant to any agreement under which they use any third-party Intellectual Property and, in the past two (2) years, have not received any written notice of intent to conduct any such audit.

Section 4.16 <u>Data Privacy and Security</u>. Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, the Company and its Subsidiaries are, and have at all times since January 1, 2019, been in compliance with (i) their privacy policies and notices, including any website privacy policies which are posted and accessible to individuals at all times; (ii) all written representations, warranties, covenants and agreements of the Company and each of its Subsidiaries pursuant to any Contract related to the collection, use, storage, processing, distribution, transfer, import, export, disposal or disclosure (whether electronically or in any other form or medium) of any Personal Information; and (iii) all applicable Privacy Laws. Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, the Company and its Subsidiaries have taken reasonable steps in accordance with standard industry practices and applicable Privacy Laws to secure Personal Information from unauthorized access or use thereof by any Person.

Section 4.17 <u>Taxes</u>. Except as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect:

- (a) (i) the Company and each of its Subsidiaries have duly and timely filed (taking into account valid extensions obtained in the ordinary course of business) each income or franchise Tax Return and each other Tax Return required to be filed, and all such Tax Returns are true, accurate and complete in all respects; (ii) the Company and each of its Subsidiaries have timely paid to the appropriate Taxing Authority all Taxes due and payable; (iii) the Company and each of its Subsidiaries have complied with all applicable Laws, rules, and regulations relating to the payment and withholding of Taxes and have, within the time and in the manner prescribed by Law, withheld and paid over to the proper Governmental Authority all amounts required to be so withheld and paid over, except, in each case of clauses (iii) and (iii), with respect to matters for which adequate accruals or reserves have been established, in accordance with GAAP, on the Company Balance Sheet; (iv) there is no outstanding claim, assessment or deficiency against the Company or any of its Subsidiaries in respect of any Taxes and there is no Proceeding ongoing or, to the Company's knowledge pending or threatened, against or with respect to the Company or any of its Subsidiaries in respect of any Tax; and (v) there are no Liens for Taxes on any of the assets of the Company or any of its Subsidiaries in respect of any Tax; and (v) there are no Liens for Taxes on any of the assets of the Company or any of its Subsidiaries in respect to the Subsidiaries other than Permitted Liens;
- (b) during the three (3) year period ending on the date of this Agreement, neither the Company nor any of its Subsidiaries was a distributing corporation or a controlled corporation in a transaction intended to be governed by Section 355 of the Code (or so much of Section 356 of the Code as relates to Section 355 of the Code);
- (c) neither the Company nor any of its Subsidiaries (i) is, or has been, a party to any Tax Sharing Agreement (other than an agreement exclusively between or among the Company and its Subsidiaries or among the Company's Subsidiaries) pursuant to which it will have any obligation to make any payments for Taxes after the

Effective Time, (ii) has been a member of a group filing an affiliated consolidated, combined or unitary Tax Return, including, for the avoidance of doubt, any group that is subject to equivalent or similar such Taxes in a country other than the United States (in each case, other than a group the common parent of which is or was the Company or any of its Subsidiaries), and (iii) has any liability for the payment of any Tax imposed on any Person (other than the Company) as a transferee or successor, by Contract, assumption or operation of Law, or otherwise;

- (d) neither the Company nor any of its Subsidiaries has participated in a "listed transaction" within the meaning of Treasury Regulations Section 1.6011-4(b) or a similar provision of state, local or foreign Tax Laws;
- (e) neither the Company nor any of its Subsidiaries will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any: (i) change in method of accounting for a taxable period ending on or prior to the Closing Date; (ii) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local, or non-U.S. income Tax law) executed on or prior to the Closing Date; (iii) intercompany transactions or any excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding or similar provision of state, local, or non-U.S. income Tax law); (iv) installment sale or open transaction disposition made on or prior to the Closing Date; or (v) prepaid amount or deferred revenue received or accrued on or prior to the Closing Date;
- (f) there is not in force any waiver or agreement for any extension of time for the assessment or payment of any Tax by the Company or any of its Subsidiaries (other than pursuant to extensions of time to file Tax Returns obtained in the ordinary course of business);
- (g) no written claim has been made by any Taxing Authority within the last three (3) years in a jurisdiction where the Company or any of its Subsidiaries does not currently file a Tax Return that the Company or such Subsidiary is or may be subject to any Tax or required to file any Tax Return in such jurisdiction;
- (h) no material closing agreements, private letter rulings, technical advice memoranda or similar agreements or rulings have been entered into with or issued by any Taxing Authority within the three (3) year period immediately preceding the date of this Agreement with respect to the Company or any of its Subsidiaries.
  - (i) Each Subsidiary of the Company is classified as a disregarded entity for U.S. federal and applicable state and local income tax purposes.

#### Section 4.18 Employees and Employee Benefit Plans.

(a) Section 4.18 of the Company Disclosure Letter contains a correct and complete list identifying each Company Plan. For purposes of this Agreement, (i) "Company Plan" means each "employee benefit plan" within the meaning of ERISA Section 3(3), whether or not subject to ERISA, all equity or equity-based (including the Company LTIP), change in control, bonus or other incentive compensation, Code Section 125, fringe benefit, disability, salary continuation, employment, consulting, indemnification, severance, retention, retirement, pension, profit sharing, savings or thrift, deferred compensation, health or life insurance, employee discount or free product, vacation, sick pay or paid time off agreements, plans or policies, and each other similar benefit or compensation plan, program, policy, contract, agreement or arrangement, whether written or unwritten, in each case (A) established, maintained, sponsored or contributed to (or with respect to which any obligation to contribute has been undertaken) by the Company or any ERISA Affiliate or (B) with respect to which the Company or any ERISA Affiliate has or could reasonably be expected to have, any liability or obligation, contingent or otherwise; (ii) "Foreign Company Plan" means each Company Plan that primarily covers current or former Employees, officers, directors or other service providers of the Company or any of its Affiliates based outside of the United States and/or which is governed by the laws of any jurisdiction outside of the United States (other than any plan or program maintained by a Governmental Authority to which the Company or any of its

ERISA Affiliates contributes pursuant to applicable Laws, including, but not limited to, any benefits that are required to be provided under applicable Law, including statutory severance) and any severance, retirement, pension, or leave accruals provided by the Company that primarily covers current or former Employees, officers, directors or other service providers as required by applicable Law; and (iii) "U.S. Company Plan" means each Company Plan that is not a Foreign Company Plan. The Company has made available to Parent with respect to each U.S. Company Plan: (A) all documents setting forth the terms of each such U.S. Company Plan (or with respect to any unwritten U.S. Company Plan, a written description of each material term thereof) and all material documents relating to each such U.S. Company Plan, including but not limited to the plan document, all amendments thereto and all related trust documents, insurance contracts and/or policies, and funding instruments, (B) the most recent annual reports (Form 5500 including, if applicable, all schedules and attachments thereto) and tax return (Form 990), if any, required under ERISA or the Code in connection therewith or its related trust and any state or local tax returns, (C) the most recent actuarial report (if applicable), (D) all summary plan descriptions, together with each summary of material modifications, if any, required under ERISA, and all material written employee communications relating to each such U.S. Company Plan, (E) all material written contracts, instruments or agreements relating to each such U.S. Company Plan, including but not limited to administrative agreements, and investment management or advisory agreements, and all amendments thereto, (F) the most recent Internal Revenue Service determination or opinion letter issued with respect to each such U.S. Company Plan intended to be qualified under Section 401(a) of the Code, (G) all material correspondence to or from any Governmental Authority within the past three (3) years with respect to any U.S. Company Plan, and (H) the most recent required nondiscrimination and coverage tests performed. The Company has made available to Parent with respect to each Foreign Company Plan, summaries of material benefits and form employment agreements. The Company has separately identified in Section 4.18 of the Company Disclosure Letter each Company Plan that contains a change in control provision that accelerates vesting, payment, or provides additional benefits.

- (b) Neither the Company nor any ERISA Affiliate has contributed (or had any obligation or liability of any sort, contingent or otherwise) in the last six (6) years to or with respect to a plan that is subject to Section 412 or 430 of the Code or Section 302 or Title IV of ERISA, has not.
- (c) Neither the Company nor any ERISA Affiliate maintains, contributes to, has or has any obligation or liability of any sort with respect to, or sponsors (or has in the past six (6) years maintained, contributed to, had any obligation or liability in connection with, or sponsored) a multiemployer plan as defined in Section 3(37) of ERISA (a "Multiemployer Plan") or the local equivalent thereof under applicable Laws.
- (d) With respect to each of the U.S. Company Plans, except as would not reasonably be expected to result in a material liability to the Company and the Company's Subsidiaries, taken as a whole: (i) each U.S. Company Plan intended to qualify under Section 401(a) of the Code has received a determination or opinion letter from the Internal Revenue Service upon which it may rely regarding its qualified status under the Code or can rely on a determination or opinion letter from the Internal Revenue Service the with respect to such U.S. Company Plan and to the Company's knowledge no event has occurred that has caused or could reasonably be expected to cause the loss of such qualification, (ii) all payments required to be paid by the Company or any of its Subsidiaries or ERISA Affiliates pursuant to the terms of a U.S. Company Plan, any Collective Bargaining Agreement, or by applicable Law (including, without limitation, all contributions and insurance premiums) with respect to all prior periods have been made or provided for by the Company or its Subsidiaries or ERISA Affiliates in accordance with the provisions of such U.S. Company Plan, Collective Bargaining Agreement or applicable Law, (iii) no proceeding related to the U.S. Company Plans has been instituted or, to the knowledge of the Company, threatened or is anticipated against any of the U.S. Company Plans (other than routine claims for benefits and appeals of such claims), any trustee or fiduciaries thereof, any ERISA Affiliate, or any of the assets of any trust of any of the U.S. Company Plans or the Company or any of its Subsidiaries or ERISA Affiliates, (iv) each U.S. Company Plan complies in form and has been maintained and operated in accordance with its terms and applicable Law, including, without limitation, ERISA and the Code, (v) none of the Company, any of its Subsidiaries, and, to the knowledge of the Company, its third party fiduciaries have engaged in a non-exempt

"prohibited transaction," within the meaning of Section 4975 of the Code and Section 406 of ERISA with respect to the U.S. Company Plans, (vi) no U.S. Company Plan is currently under, and neither the Company nor its Subsidiaries has received any notice of, a pending audit or investigation by the Internal Revenue Service, U.S. Department of Labor, Pension Benefit Guaranty Corporation, or any other Governmental Authority, and (vii) no U.S. Company Plan provides any post-retirement health or welfare benefits for any current or former employee of the Company or its Subsidiaries (or their dependents), other than as required under the Consolidated Omnibus Reconciliation Act of 1985, as amended ("COBRA").

- (e) The consummation of the transactions contemplated by this Agreement will not (either alone or together with any other event) (i) entitle any Employee, officer, director, independent contractor, or other service provider of the Company or its Subsidiaries (whether current, former or retired) or their beneficiaries to any pay (including but not limited to severance pay) under a U.S. Company Plan and, except as would not reasonably be expected to result in a material liability to the Company and its Subsidiaries, taken as a whole, any Foreign Company Plan, (ii) accelerate the time of payment or vesting or trigger any payment or funding (through a grantor trust or otherwise) of material compensation or benefits under any U.S. Company Plan and, except as would not reasonably be expected to result in a material liability to the Company and its Subsidiaries, taken as a whole, any Foreign Company Plan, (iii) increase the amount payable or trigger any other financial obligation pursuant to any Company Plan or (iv) results in any amounts payable to any "disqualified individual" failing to be deductible for federal income tax purposes by virtue of Section 280G of the Code or subject to an excise tax under Section 4999 of the Code. Neither the Company nor any of its Subsidiaries has any obligation to gross-up, indemnify or otherwise reimburse any current or former employee, worker, officer, director or other independent contractor of the Company or any of its Subsidiaries for any Tax incurred by such individual under Section 409A or 4999 of the Code or otherwise.
- (f) No condition or circumstance exists that would prevent or materially impede the amendment or termination of any U.S. Company Plan unilaterally at any time by the Company or its Subsidiaries, the Surviving Corporation, Parent or their respective ERISA Affiliates. Neither the Company nor any of its Subsidiaries has made any promises or commitments to create any additional U.S. Company Plan or to modify or change any existing U.S. Company Plan other than those amendments or modifications required by Law or as contemplated herein.
- (g) Except as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect with respect to each Foreign Company Plan: (i) each Foreign Company Plan required to be registered has been registered and maintained in good standing with any applicable Governmental Authority; (ii) each Foreign Company Plan intended to receive favorable tax treatment under applicable tax Laws, to the extent applicable, has been qualified or similarly determined by applicable Governmental Authorities to satisfy the requirements of such Laws; (iii) no Foreign Company Plan is a defined benefit or similar type of plan or arrangement; and (iv) no Foreign Company Plan has had any material unfunded liabilities nor are any unfunded liabilities reasonably expected to arise in connection with the transactions contemplated by this Agreement.
- (h) Each U.S. Company Plan that is a "nonqualified deferred compensation plan" (as defined for purposes of Section 409A(d)(1) of the Code) is in material compliance with Section 409A of the Code and all applicable guidance promulgated thereunder.
- (i) As of the date of this Agreement, neither the Company nor any of its Subsidiaries or ERISA Affiliates is a party to any Collective Bargaining Agreement, and there are no labor organizations, work councils, trade unions or other employee representatives representing, or, to the knowledge of the Company, purporting to represent or seeking to represent any current Employee.
- (j) Except as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect or a material liability to the Company and its Subsidiaries taken as a whole,

none of the Company or any of its Subsidiaries has been subject to any Collective Bargaining Agreement and there are no grievances or arbitrations outstanding thereunder; (ii) to the knowledge of the Company, there are no labor organizational campaigns, corporate campaigns, petitions, demands for recognition (including demands for works council recognition), applications or other unionization activities seeking recognition of a bargaining unit at the Company or any of its Subsidiaries; (iii) none of the Company or any of its Subsidiaries has suffered) any labor dispute, any activity or proceeding by a labor union or representative thereof to organize any Employee or any picketing, lockouts, strikes, slowdowns, work stoppages, job actions or threats thereof by or with respect to any Employee, against or involving the Company or any of its Subsidiaries; (iv) there are no unfair labor practice charges, grievances, arbitrations or other complaints or union matters before the National Labor Relations Board or other labor board of Governmental Authority or works council disputes that would reasonably be expected to affect the employees of the Company and its Subsidiaries; (v) there are no current or, to the knowledge of the Company, threatened strikes, slowdowns, lockouts, organized labor disputes or work stoppages, and no such strike, slowdown, lockout, organized labor dispute or work stoppage has occurred within the two (2) years preceding the date of this Agreement; (vi) the execution of this Agreement and the consummation of the transactions contemplated by this Agreement will not result in any breach or other violation of any Collective Bargaining Agreement; and (vii) the execution of this Agreement will not result in any breach or other violation of consultation obligations under applicable labor Law.

- (k) Except as contemplated by this Agreement, to the knowledge of the Company, no director, executive officer, or other key current Employee has expressed to the Company in writing any present intention to terminate his or her employment with the Company or any of its Subsidiaries. No Employees have notified the Company or its Subsidiaries, in writing, of any obligations of confidentiality of such Employees to any other Person(s) that conflict with such Employee's work for, or obligations to, Parent, the Company, the Surviving Corporation, or any of their Subsidiaries.
- (l) Except as would not reasonably be expected to result in a material liability to the Company or any of its Subsidiaries, the Company and its Subsidiaries have complied in all material respects with all applicable Laws relating to employment, employment practices, wages, hours, mandatory insurance, and other benefits, leaves of absence, employee classification, immigration control, employee safety, bonuses and terms and conditions of employment, including laws relating to termination of employment and relating to job applicants and employee background checks. No material action, suit, Claim (or counterclaim), litigation, arbitration, or mediation, (including any civil, criminal, administrative, investigative or appellate proceeding) that arises out of the current, former or potential employment or service relationship between the Company or any of its Subsidiaries and applicants or Employees, officers, or directors, or other service providers is pending or, to the knowledge of the Company, is otherwise pending or has been threatened against the Company or its Subsidiaries. Except as otherwise set forth on Section 4.18(l) of the Company Disclosure Letter, all individuals characterized and treated by the Company as independent contractors or consultants are properly treated as independent contracts under all applicable laws. Except as would not reasonably be expected to result in a material liability to the Company or any of its Subsidiaries, all employees of the Company classified as exempt under the Fair Labor Standards Act and state and local wage and hour laws are properly classified. The Company is in compliance with and has complied with all immigration laws, including Form I-9 requirements and any applicable mandatory E-Verify obligations.

## Section 4.19 Environmental Matters.

- (a) Except as not reasonably expected to have or had, individually or in the aggregate, a Company Material Adverse Effect:
  - (i) the Company and its Subsidiaries are and, since January 1, 2019, have been in compliance with all applicable Environmental Laws;
  - (ii) there is no Environmental Claim pending or, to the knowledge of the Company, threatened against the Company or any of its Subsidiaries;

- (iii) except as listed in Section 4.19(a)(iii) of the Company Disclosure Letter, the Company and its Subsidiaries have no contractual indemnity obligation to any third party for Environmental Claims or liability under Environmental Law, other than general commercial indemnification obligations entered into in the ordinary course of business, and not for the primary purpose of indemnifying matters relating to Environmental Claims or Environmental Laws; and
- (iv) to the knowledge of the Company, there are no actions, activities, circumstances, facts, conditions, events or incidents that would be reasonably likely to form the basis of any Environmental Claim against the Company or any of its Subsidiaries.
- (b) Except as listed in Section 4.19(b) of the Company Disclosure Letter, the Company and its Subsidiaries are not required to hold any Environmental Permits in order to conduct their business or operations as their business and operations are conducted.
- (c) The Company has provided Parent with true and complete copies of all material environmental assessment reports and reports of material environmental investigations with respect to the Company and any of its Subsidiaries and for any Company Real Property.
- (d) The transactions contemplated by this Agreement do not require, under any Environmental Law, any material consent of, or material filings with, any Governmental Authority with jurisdiction over the Company or any of its Subsidiaries including, but not limited to, the transfer of any environmental permits.

#### Section 4.20 Material Contracts.

- (a) Section 4.20(a) of the Company Disclosure Letter sets forth, as of the date of this Agreement, a true and complete list of each of the following types of Contracts to which the Company or any of its Subsidiaries is a party or by which any of their respective properties or assets is bound:
  - (i) each Contract that limits or restricts in any material respect the Company or any of its Subsidiaries from (1) engaging or competing in any line of business in any location or with any Person, (2) selling any products or services of or to any other Person or in any geographic region or (3) obtaining products or services from any Person, in each case of clauses (1), (2) and (3), that is material to the Company and its Subsidiaries, taken as a whole:
  - (ii) each Contract that includes any arrangement whereby the Company grants any right of first refusal or right of first offer or similar right to a Third Party or any arrangement whereby the Company or one of its Subsidiaries is obligated to lease real property that would be material to the Company and its Subsidiaries, in each that is material to the Company and its Subsidiaries, taken as a whole;
    - (iii) each Contract that is a joint venture or partnership agreement that is material to the Company and its Subsidiaries, taken as a whole;
  - (iv) each Contract that is a loan, guarantee of indebtedness or credit agreement, note, bond, mortgage, indenture or other binding commitment (other than letters of credit and those between the Company and its wholly owned Subsidiaries) relating to indebtedness for borrowed money in an amount in excess of \$500,000 individually;
  - (v) each Contract with respect to an interest, rate, currency or other swap or derivative transaction (other than those between the Company and its Subsidiaries) with a fair value in excess of \$250,000;
  - (vi) each Contract that is an (A) Inbound Intellectual Property Contract that is required to be listed on Section 4.15(b) of the Company Disclosure Letter, or (B) Outbound Intellectual Property Contract that is required to be listed on Section 4.15(c) of the Company Disclosure Letter;
  - (vii) each Contract that is an acquisition agreement or a divestiture agreement pursuant to which (A) the Company reasonably expects that it is required to pay total consideration including assumption of

debt after the date of this Agreement to be in excess of \$1 million, (B) any other Person has the right to acquire any assets of the Company or any of its Subsidiaries after the date of this Agreement with a fair market value or purchase price of more than \$1 million or (C) any other Person has the right to acquire any interests in the Company or any of its Subsidiaries, excluding, in the case of clauses (A) and (B), acquisitions or dispositions of supplies, inventory, merchandise or products in connection with the conduct of the Company's and its Subsidiaries' business or of supplies, inventory, merchandise, products, equipment, properties or other assets that are obsolete, worn out, surplus or no longer used or useful in the conduct of business of the Company or its Subsidiaries;

- (viii) each Contract that is a settlement or similar agreement with any Governmental Authority (including any corporate integrity agreement, monitoring agreement or deferred prosecution agreement) or order or consent of a Governmental Authority (including any consent decree or settlement order) to which the Company or any of its Subsidiaries is subject involving future performance by the Company or any of its Subsidiaries which is material to the Company and its Subsidiaries, taken as a whole;
- (ix) each Contract (or series of related Contracts) pursuant to which the Company or any of its Subsidiaries has continuing "earn-out" or similar obligations that could result in payments in excess of \$1 million in the aggregate;
- (x) each Contract (or series of related Contracts) that obligates the Company or any of its Subsidiaries to make any capital commitment, loan or capital expenditure in an amount in excess of \$500,000 in the aggregate in any one year period after the date of this Agreement that cannot be terminated by the Company or any of its Subsidiaries on less than sixty (60) days' notice without material payment or penalty;
  - (xi) each Contract for the employment of, or receipt of any services from, any current Employee at the level of Vice President or higher;
- (xii) each Contract to which the Company or any of its Subsidiaries is a party that constitutes a Collective Bargaining Agreement or other labor agreement or that relates to the employment of non-U.S. individual service providers to the Company or its Subsidiaries employed by professional employer organizations ("Non-U.S. Service Providers");
- (xiii) each Contract that would be required to be filed by the Company as a "material contract" pursuant to Item 601(b)(10) of RegulationS-K under the 1933 Act or disclosed by the Company under Item 1.01 on a Current Report on Form 8-K.

Each Contract of the type described in clauses (i) through (xiii) is referred to herein as a "Company Material Contract."

- (b) Except for this Agreement or as listed in Section 4.20(b) of the Company Disclosure Letter, as of the date of this Agreement, neither the Company nor any of its Subsidiaries is a party to or bound by any "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC) that is to be performed after the date of this Agreement that has not been filed as an exhibit to or incorporated by reference in a Company SEC Document.
- (c) Except for any Company Material Contract that has terminated or expired in accordance with its terms or as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, each Company Material Contract is valid and binding and in full force and effect and, to the Company's knowledge, enforceable against the other party or parties thereto in accordance with its terms. The Company and/or its Subsidiaries party thereto, as applicable, and, to the knowledge of the Company, each other party thereto, has performed its obligations required to be performed by it, as and when required, under each Company Material Contract, except for failures to perform that have not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Except for breaches, violations or defaults which have not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, neither the Company nor any of its Subsidiaries, nor to the

Company's knowledge any other party to a Company Material Contract, is in violation of or in default under any provision of such Company Material Contract. True and complete copies of the Company Material Contracts and any material amendments thereto have been made available to Parent prior to the date of this Agreement.

Section 4.21 <u>Finders' Fees</u>. Except for Piper Sandler & Co., true and correct copies of whose engagement letter has been provided to Parent, and except as set forth on Section 4.21 of the Company Disclosure Letter, there is no investment banker, broker or finder that has been retained by or is authorized to act on behalf of the Company or any of its Subsidiaries who is entitled to any fee or commission from the Company or any of its Affiliates in connection with the transactions contemplated by this Agreement.

Section 4.22 Opinion of Financial Advisor. As of the date of this Agreement, the Company Board has received the opinion of Piper Sandler & Co., financial advisor to the Special Committee and the Company, to the effect that, as of the date of such opinion, and based upon and subject to the assumptions, qualifications, matters and limitations set forth therein, the consideration to be received by the holders of Company Stock (other than shares of Company Stock owned, directly or indirectly, by Parent, the Company (including shares held as treasury stock or otherwise), or Merger Sub Inc.) pursuant to this Agreement is fair from a financial point of view to such holders. A copy of such opinion will be made available to Parent as soon as practicable following the date of this Agreement for informational purposes only. Parent and Merger Sub Inc. acknowledge and agree that the opinion of Piper Sandler & Co. may not be relied upon by Parent or any director, manager, officer or employee of Parent and that such opinion may not be distributed by Parent or Merger Sub Inc. to any third party without the prior consent of Piper Sandler & Co.

Section 4.23 <u>Antitakeover Statutes</u>. Assuming the accuracy of Parent's and Merger Sub Inc.'s representations and warranties in <u>Article V</u>, (i) the Company has taken all action necessary to exempt the Merger, this Agreement, the Support Agreements and the transactions contemplated hereby and thereby from the restrictions of Section 203 of the DGCL, and, accordingly, neither the restrictions pursuant to such provision of the DGCL nor any other antitakeover or similar statute or regulation applies or purports to apply to any such transactions and (ii) no other "control share acquisition," "fair price," "moratorium" or other antitakeover laws enacted under U.S. state or federal laws apply to this Agreement, the Support Agreement or any of the transactions contemplated hereby or thereby.

Section 4.24 <u>Insurance</u>. Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, each of the insurance policies and self-insurance programs and arrangements relating to the business, assets and operations of the Company are in full force and effect. As of the date of this Agreement, since January 1, 2019, neither the Company nor any of its Subsidiaries has received any written notice regarding any actual or possible: (a) cancellation or invalidation of any such insurance policy, other than such cancellation or invalidation that would not constitute a Company Material Adverse Effect; or (b) written notice of refusal of any coverage or rejection of any claim under any such insurance policy that if not paid would constitute a Company Material Adverse Effect. With respect to each Proceeding that has been filed or investigation initiated against the Company or any of its Subsidiaries since January 1, 2019, no insurance carrier has issued a denial of coverage or a reservation of rights with respect to any such Proceeding or investigation, or informed any of the Company or any of its Subsidiaries of its intent to do so, other than such denial or reservation that would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

Section 4.25 No Additional Representations. Except for the representations and warranties expressly made by the Company in this Article IV and in the Company Disclosure Letter, each of Parent and Merger Sub Inc. acknowledges that neither the Company nor any other Person makes any express or implied representation or warranty whatsoever and specifically (but without limiting the foregoing), that neither the Company nor any Representative of the Company makes any representation or warranty with respect to (a) the Company or its Subsidiaries or any of their respective businesses, affairs operations, assets, liabilities, conditions (financial or otherwise), prospects or any other matter relating to the Company or its Subsidiaries or (b) any documentation.

forecasts, budgets, projections, estimates or other information (including the accuracy or completeness of, or the reasonableness of the assumptions underlying, such documentation, forecasts, budgets, projections, estimates or other information) provided by the Company or any Representative of the Company, including in any "data rooms" or management presentations.

# ARTICLE V REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB INC.

Subject to Section 11.5, (a) except as disclosed in the Parent SEC Documents publicly filed or furnished to the SEC prior to the date of this Agreement; provided, that in no event shall any risk factor disclosure under the heading "Risk Factors" or disclosure set forth in any "forward looking statements" disclaimer or other general statements to the extent they are cautionary, predictive or forward looking in nature that are included in any part of any Parent SEC Document be deemed to be an exception to, or, as applicable, disclosure for purposes of, any representations and warranties of Parent or Merger Sub Inc. contained in this Agreement or (b) except as set forth in the Parent Disclosure Letter, Parent and Merger Sub Inc. represent and warrant to the Company that:

Section 5.1 Corporate Existence and Power. Each of Parent and Merger Sub Inc. is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware. Each of Parent and Merger Sub Inc. has all corporate power and authority to carry on its business as now conducted and is duly qualified to do business as a foreign corporation and is in good standing in each jurisdiction where such qualification is necessary, except where any failure to have such power or authority or to be so qualified would not reasonably be expected, individually or in the aggregate, to have a Parent Material Adverse Effect. Prior to the date of this Agreement, Parent has delivered or made available to the Company true and complete copies of the organizational documents of Parent and Merger Sub Inc. as in effect on the date of this Agreement. None of Parent or Merger Sub Inc. is in violation of, in conflict with, or in default under, its certificate of incorporation or bylaws.

# Section 5.2 Corporate Authorization.

- (a) The execution, delivery and performance by Parent and Merger Sub Inc. of this Agreement and the consummation by Parent and Merger Sub Inc. of the transactions contemplated hereby are within the corporate power and authority of Parent and Merger Sub Inc. and, except for the approval of ProFrac Holdings II, LLC as the sole stockholder of Merger Sub Inc., the Parent Stockholder Approval and the filing the Certificate of Merger with the Secretary of State of the State of Delaware as required by the DGCL, have been duly authorized by all necessary corporate action on the part of Parent and Merger Sub Inc.. The Parent Stockholder Approval is the only vote of any holders of any of Parent's capital stock necessary in connection with the consummation of the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Parent and Merger Sub Inc. and, assuming due authorization, execution and delivery by the Company, constitutes a valid and binding agreement of each of Parent and Merger Sub Inc., enforceable against Parent and Merger Sub Inc. in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting creditors' rights generally and by general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at Law).
- (b) As of the date of this Agreement, the respective board of directors of each of Parent and Merger Sub Inc. have approved and declared advisable this Agreement and the transactions contemplated hereby.

Section 5.3 Governmental Authorization. The execution, delivery and performance by Parent and Merger Sub Inc. of this Agreement and the consummation by Parent and Merger Sub Inc. of the transactions contemplated hereby require no action by or in respect of, or filing with, any Governmental Authority, other than (i) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware and appropriate documents with the relevant authorities of other states in which Parent is qualified to do business, (ii) compliance

with any applicable requirements of the HSR Act and any non-U.S. Competition Laws, (iii) compliance with any applicable requirements of the 1933 Act, the 1934 Act and any other applicable state or federal securities laws, (iv) compliance with any applicable requirements of Nasdaq and (v) any actions or filings the absence of which would not reasonably be expected to have, individually or in the aggregate, (x) a Parent Material Adverse Effect or (y) an effect that would prevent, materially delay or materially impair the ability of Parent to perform its obligations under this Agreement or to consummate the Merger.

Section 5.4 Non-contravention. The execution, delivery and performance by Parent and Merger Sub Inc. of this Agreement and the consummation of the transactions contemplated hereby do not and will not (i) assuming the authorizations, consents and approvals referred to in clauses (i) through (iv) of Section 5.3 are obtained and, with respect to consummation of the transactions contemplated hereunder, other than any consents obtain by the Closing Date, (A) contravene, conflict with, or result in any violation or breach of any provision of the organizational documents of Parent or Merger Sub Inc., (B) contravene, conflict with or result in a violation or breach of any provision of any Law or Order or (C) require any consent or other action by any Person under, constitute a default or a violation, or an event that, with or without notice or lapse of time or both, would constitute a default or a violation, under or of, or cause or permit the termination, cancellation, acceleration or other change of any right or obligation or the loss of any benefit to which Parent or any of its Subsidiaries is entitled under any provision of any Contract binding upon Parent or any of its Subsidiaries or any license, franchise, permit, certificate, approval or other similar authorization affecting, or relating in any way to, the assets or business of Parent and its Subsidiaries or (ii) result in the creation or imposition of any Lien, other than any Permitted Lien, on any asset of Parent or any of its Subsidiaries, except, in the case of each of clauses (i)(B), (i)(C) and (ii), which have not had and would not reasonably be expected to have, individually or in the aggregate, (x) a Parent Material Adverse Effect or (y) an effect that would prevent, materially delay or materially impair the ability of Parent to perform its obligations under this Agreement or to consummate the Merger.

#### Section 5.5 Capitalization.

- (a) As of the date of this Agreement, the authorized capital stock of Parent consists of (i) 600,000,000 shares of Parent Stock, (ii) 400,000,000 shares of Class B common stock, par value \$0.01 ("Parent Class B Common Stock"), and (iii) 50,000,000 shares of preferred stock, par value \$0.01 per share ("Parent Preferred Stock"). As of June 14, 2022, (A) 41,237,003 shares of Parent Stock were issued and outstanding, (B) 101,133,201 shares of Parent Class B Common Stock were issued and outstanding, (C) no shares of Parent Preferred Stock were issued and outstanding and (D) 509,467 restricted stock units with respect to 509,467 shares of Parent Stock were issued and outstanding, all of which were issued under the Parent 2022 Long Term Incentive Plan (the "Parent RSUs").
- (b) Except upon the settlement of the Parent RSUs, in each case, that were outstanding on June 14, 2022, there are no issued, reserved for issuance or outstanding (i) shares of capital stock or other voting securities of or other ownership interests in Parent, (ii) securities of Parent convertible into or exchangeable for shares of capital stock or other voting securities of or other ownership interests in Parent, (iii) warrants, calls, options or other rights to acquire from Parent or other obligation of Parent to issue, any shares of capital stock, voting securities or securities convertible into or exchangeable for capital stock or other voting securities of or other ownership interests in Parent or (iv) restricted shares, stock appreciation rights, performance units, contingent value rights, "phantom" stock or similar securities or rights issued or granted by Parent or any of its Subsidiaries that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of, any shares of capital stock of or other voting securities of or other ownership interests in Parent (the items in clauses (i) through (iv) being referred to collectively as the "Parent Securities"). There are no outstanding obligations of Parent or any of its Subsidiaries to repurchase, redeem or otherwise acquire any of the Parent Securities, and neither Parent nor any of its Subsidiaries maintains an employee stock purchase plan. Neither Parent nor any of its Subsidiaries is a party to any voting trust, proxy, voting agreement or other similar agreement with respect to the voting of any Parent Securities. All outstanding shares of capital stock of Parent have been, and all shares that may be issued pursuant to any equity compensation plan or arrangement will be, when issued in accordance with

the respective terms thereof, duly authorized and validly issued, fully paid and nonassessable and free of preemptive rights.

- (c) As of the date of this Agreement, there are no outstanding bonds, debentures, notes or other indebtedness of Parent having the right to vote (whether on an as-converted basis or otherwise) (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which stockholders of Parent may vote.
- (d) The shares of Parent Stock to be issued as Merger Consideration, when issued and delivered in accordance with the terms of this Agreement, will have been duly authorized and validly issued, fully paid and nonassessable and free of preemptive rights.
- (e) Merger Sub Inc. is an indirect subsidiary of Parent that was formed solely for the purpose of engaging in the Merger. Since its date of incorporation and prior to the Effective Time, Merger Sub Inc. has not engaged in any activities other than the execution of this Agreement, the performance of its obligations hereunder and matters ancillary thereto, and prior to the Effective Time will have no assets, liabilities or obligations of any nature other than those incident to its formation and pursuant to this Agreement and the Merger.
- (f) The authorized capital stock of Merger Sub Inc. consists of 1,000 shares of common stock, \$0.001 par value per share, all of which have been validly issued, are fully paid and nonassessable and are owned indirectly by Parent free and clear of any Lien.

# Section 5.6 SEC Filings and the Sarbanes-Oxley Act

- (a) Parent has filed with or furnished to the SEC (including following any extensions of time for filing provided by Rulel 2b-25 promulgated under the 1934 Act) all reports, schedules, forms, statements, prospectuses, registration statements and other documents required to be filed or furnished, as the case may be, by Parent since January 1, 2019 (collectively, together with any schedules thereto and other information incorporated therein, the "Parent SEC Documents").
- (b) As of its filing date (or, if amended or supplemented, as of the date of the most recent amendment or supplement filed prior to the date of this Agreement), each Parent SEC Document complied as to form in all material respects with the applicable requirements of the 1933 Act and the 1934 Act and the Sarbanes-Oxley Act, and any rules and regulations promulgated thereunder, as the case may be.
- (c) As of its filing date (or, if amended or supplemented, as of the date of the most recent amendment or supplement filed prior to the date of this Agreement), each Parent SEC Document filed or furnished pursuant to the 1934 Act did not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.
- (d) Each Parent SEC Document that is a registration statement, as amended or supplemented, if applicable, filed pursuant to the 1933 Act, as of the date such registration statement or amendment became effective, and as of the date of such supplement, did not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading.
- (e) Parent has heretofore furnished to the Company complete and correct copies of all comment letters received from the SEC or its staff from January 1, 2019 through the date of this Agreement with respect to any of the Parent SEC Documents, together with all written responses of Parent thereto, to the extent that such comment letters and written responses are not publicly available on EDGAR. As of the date of this Agreement, there are no outstanding or unresolved comments in comment letters received from the SEC or its staff with respect to any of the Parent SEC Documents, and, to the knowledge of Parent, none of the Parent SEC Documents are subject to ongoing SEC review.

- (f) Parent has established and maintains disclosure controls and procedures (as defined in Rule 13a-15 under the 1934 Act). Such disclosure controls and procedures are reasonably designed to ensure that material information relating to Parent, including its consolidated Subsidiaries, is made known to Parent's principal executive officer and its principal financial officer by others within those entities and such disclosure controls and procedures are reasonably designed to ensure all such information is communicated in a timely fashion to Parent's principal executive officer and principal financial officer such that timely decisions may be made regarding the disclosure of such information in Parent's periodic and current reports required under the 1934 Act
- (g) Parent and its Subsidiaries have established and maintained a system of Internal Controls over financial reporting sufficient to provide reasonable assurance regarding the reliability of Parent's financial reporting and the preparation of Parent financial statements for external purposes in accordance with GAAP. Parent has disclosed, based on its most recent evaluation of Internal Controls prior to the date of this Agreement, to Parent's auditors and audit committee (i) any significant deficiencies and material weaknesses in the design or operation of Internal Controls which are reasonably likely to adversely affect Parent's ability to record, process, summarize and report financial information and (ii) any fraud, whether or not material, that involves management or other employees who have a significant role in Internal Controls.
- (h) Neither Parent nor any of its Subsidiaries has extended or maintained credit, arranged for the extension of credit, or renewed an extension of credit, in the form of a personal loan to or for any executive officer (as defined in Rule 3b-7 under the 1934 Act) or director of Parent in violation of Section 402 of the Sarbanes-Oxley Act.
- (i) Parent is in compliance, and has complied since January 1, 2019, in each case in all material respects, with (i) the applicable provisions of the Sarbanes-Oxley Act and (ii) the applicable listing and corporate governance rules and regulations of Nasdaq.
- (j) Each of the principal executive officer and principal financial officer of Parent (or each former principal executive officer and principal financial officer of Parent, as applicable) has made all certifications required by Rule 13a-14 and 15d-14 under the 1934 Act and Sections 302 and 906 of the Sarbanes-Oxley Act and any related rules and regulations promulgated by the SEC and Nasdaq, and the statements contained in any such certifications are complete and correct in all material respects.
- (k) Since the Parent Balance Sheet Date through the date of this Agreement, there has been no transaction, or series of similar transactions, agreements, arrangements or understandings, nor is there any proposed transaction as of the date of this Agreement, or series of similar transactions, agreements, arrangements or understandings to which Parent or any of its Subsidiaries was or is to be a party, that would be required to be disclosed under Item 404 of Regulation S-K promulgated under the 1933 Act that has not been disclosed in the Parent SEC Documents publicly filed or furnished with the SEC following the Parent Balance Sheet Date.
- Section 5.7 Financial Statements. The audited consolidated financial statements and unaudited consolidated interim financial statements of Parent included or incorporated by reference in the Parent SEC Documents (including all related notes and schedules thereto) (a) fairly present in all material respects, in conformity with GAAP (except, in the case of unaudited consolidated interim financial statements, as permitted by the SEC's Form 10-Q) applied on a consistent basis (except as may be indicated therein or in the notes thereto), the consolidated financial position of Parent and its consolidated Subsidiaries as of the dates thereof and their consolidated results of operations and cash flows for the periods then ended (subject to normal year-end audit adjustments in the case of any unaudited interim financial statements), (b) comply in all material respects with the applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto, and (c) have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto and except, in the case of the unaudited interim statements, as may be permitted under the SEC's Form 10-Q).

Section 5.8 Information Supplied. The information relating to Parent and its Subsidiaries to be contained in, or incorporated by reference in, the Registration Statement, Proxy Statement and Information Statement will not, (i) on the date the Proxy Statement is first mailed to the holders of Company Stock or at the time of the Company Stockholders' Meeting, (ii) on the date that the Information Statement is provided to holders of Parent Stock or (iii) at the time the Registration Statement is declared effective by the SEC, contain any untrue statement of any material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not false or misleading. The Registration Statement, Proxy Statement and Information Statement will comply in all material respects as to form with the requirements of the 1933 Act and the 1934 Act, as applicable, and the rules and regulations promulgated thereunder. Notwithstanding the foregoing provisions of this Section 5.8, no representation or warranty is made by Parent or Merger Sub Inc. with respect to information or statements made or incorporated by reference in the Registration Statement, Proxy Statement or Information Statement that were not supplied by or on behalf of Parent for use therein.

#### Section 5.9 Absence of Certain Changes.

- (a) From the Parent Balance Sheet Date through the date of this Agreement, the business of Parent and its Subsidiaries has been conducted in the ordinary course of business in all material respects.
- (b) Since the Parent Balance Sheet Date through the date of this Agreement, there has not been any effect, change, condition, fact, event, occurrence or development that has had or would reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.

Section 5.10 Solvency. Each of Parent and the Surviving Corporation will be Solvent as of the Effective Time and immediately after the consummation of the Merger and the other transactions contemplated by this Agreement. For the purposes of this Agreement, the term "Solvent", when used with respect to any Person, means that, as of any date of determination, (a) the amount of the "fair saleable value" of the assets of such Person will, as of such date, exceed the sum of (1) the value of all "liabilities of such Person, including contingent and other liabilities," as of such date, as such quoted terms are generally determined in accordance with applicable Laws governing determinations of the insolvency of debtors, and (2) the amount that will be required to pay the probable liabilities of such Person, as of such date, on its existing debts (including contingent and other liabilities) as such debts become absolute and mature, (b) such Person will not have, as of such date, an unreasonably small amount of capital for the operation of the businesses in which it is engaged or proposed to be engaged following such date, and (c) such Person will be able to pay its liabilities, as of such date, including contingent and other liabilities, as they mature. For purposes of this definition, "not have an unreasonably small amount of capital for the operation of the businesses in which it is engaged or proposed to be engaged" and "able to pay its liabilities, as of such date, including contingent and other liabilities, as they mature" means that such Person will be able to generate enough cash from operations, asset dispositions or refinancing, or a combination thereof, to meet its obligations as they become due.

Section 5.11 No Undisclosed Material Liabilities. There are no liabilities or obligations of Parent or any of its Subsidiaries that would be required by GAAP, as in effect on the date hereof, to be reflected on the consolidated balance sheet of Parent (including the notes thereto), other than:

- (a) liabilities or obligations disclosed, reflected, reserved against or otherwise provided for in the Parent Balance Sheet or in the notes thereto;
- (b) liabilities or obligations incurred in the ordinary course of business since the Parent Balance Sheet Date;
- (c) liabilities or obligations arising out of this Agreement or the transactions contemplated hereby; and
- (d) liabilities or obligations that have not had, and would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.

#### Section 5.12 Compliance with Laws and Court Orders; Governmental Authorization.

(a) Except for matters that have not had and would not reasonably be expected to have, individually or in the aggregate, (i) a Parent Material Adverse Effect or (ii) an effect that would prevent, materially delay or materially impair the ability of Parent to perform its obligations under this Agreement or to consummate the Merger, Parent and each of its Subsidiaries are, and since January 1, 2019 have been, in compliance with all applicable Laws and Orders, and to the knowledge of Parent, are not under investigation by any Governmental Authority with respect to any Law or Order. There is no Order of any Governmental Authority outstanding against Parent or any of its Subsidiaries that is material to Parent and its Subsidiaries, taken as a whole.

(b) Except as would not reasonably be expected to have, individually or in the aggregate, (i) a Parent Material Adverse Effect or (ii) an effect that would prevent, materially delay or materially impair the ability of Parent or Merger Sub Inc. to perform their respective obligations under this Agreement or to consummate the Merger, Parent and each of its Subsidiaries has all Governmental Authorizations necessary for the ownership and operation of its business as presently conducted, and each such Governmental Authorization is in full force and effect. Except as would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect, (x) Parent and each of its Subsidiaries are, and since January 1, 2019, have been in compliance with the terms of all Governmental Authorizations necessary for the ownership and operation of its businesses and (y) since January 1, 2019, neither Parent nor any of its Subsidiaries has received written notice from any Governmental Authority alleging any conflict with or breach of any such Governmental Authorization.

Section 5.13 <u>Litigation</u>. Except as has not had and would not reasonably expected to have, individually or in the aggregate, (a) a Parent Material Adverse Effect or (b) an effect that would prevent, materially delay or materially impair the ability of Parent or Merger Sub Inc. to perform their respective obligations under this Agreement or to consummate the Merger, there is no Proceeding or, to the knowledge of Parent, investigation, pending against, or, to the knowledge of Parent, threatened by or against, Parent, any of its Subsidiaries, or to the knowledge of Parent and insofar as any such Proceeding or investigation relates to Parent or any of its Subsidiaries, any present or former officer, director or employee of Parent or any of its Subsidiaries or any other Person for whom Parent or any of its Subsidiaries may be liable before (or, in the case of threatened Proceedings or investigations, that would be before) or by any Governmental Authority.

Section 5.14 Opinion of Financial Advisor. As of the date of this Agreement, the Parent Special Committee has received the opinion of Jefferies LLC, financial advisor to the Parent Special Committee, to the effect that, as of the date of such opinion, and based upon and subject to the assumptions, qualifications, matters and limitations set forth therein, the Exchange Ratio is fair, from a financial point of view, to Parent. A copy of such opinion will be made available to the Company as soon as practicable following the date of this Agreement for informational purposes only. The Company acknowledges and agrees that the opinion of Jefferies LLC may not be relied upon by the Company or any director, manager, officer or employee the Company and that such opinion may not be distributed by the Company to any third party without the prior consent of Jefferies LLC.

Section 5.15 No Additional Representations. Except for the representations and warranties expressly made by Parent and Merger Sub Inc. in this Article V, the Company acknowledges that none of Parent or Merger Sub Inc. or any other Person makes any express or implied representation or warranty whatsoever and specifically (but without limiting the foregoing), that none of Parent, Merger Sub Inc. or any of their Representatives makes any representation or warranty with respect to (a) Parent or its Subsidiaries or any of their respective businesses, affairs, operations, assets, liabilities, conditions (financial or otherwise), prospects or any other matter relating to Parent or its Subsidiaries or (b) any documentation, forecasts, budgets, projections, estimates or other information (including the accuracy or completeness of, or the reasonableness of the assumptions underlying, such documentation, forecasts, budgets, projections, estimates or other information) provided by Parent or any Representative of Parent, including in any "data rooms" or management presentations.

#### ARTICLE VI COVENANTS OF THE COMPANY

- Section 6.1 Conduct of the Company. From the date of this Agreement until the earlier to occur of the Effective Time and the date of termination of this Agreement in accordance with Article X, except as expressly contemplated by this Agreement, as set forth in Section 6.1 of the Company Disclosure Letter, as consented to in writing by Parent (such consent not to be unreasonably withheld, conditioned or delayed) or as required by applicable Law, the Company shall, and shall cause each of its Subsidiaries to, conduct its business in all material respects in the ordinary course; provided, however, that no action that is specifically permitted by any of clauses (a) through (u) of this Section 6.1 shall be deemed a breach of this sentence. Without limiting the generality of the foregoing, from the date of this Agreement until the earlier to occur of the Effective Time and the date of termination of this Agreement in accordance with Article X, except as expressly contemplated by this Agreement, as set forth in Section 6.1 of the Company Disclosure Letter, as consented to in writing by Parent (such consent, other than with respect to clause (r), not to be unreasonably withheld, conditioned or delayed) or as required by applicable Law, the Company shall not, nor shall it permit any of its Subsidiaries to:
- (a) amend the certificate of incorporation, bylaws or other similar organizational documents (whether by merger, consolidation or otherwise), in each case, of the Company or any Subsidiary of the Company;
- (b) split, combine or reclassify any shares of capital stock of the Company or any of its Subsidiaries or declare, set aside or pay any dividend or other distribution (whether in cash, stock or property or any combination thereof) in respect of the capital stock of the Company or its Subsidiaries, or redeem, repurchase or otherwise acquire or offer to redeem, repurchase, or otherwise acquire any Company Securities or any Company Subsidiary Securities, except for the declaration, setting aside or payment of any dividends or other distributions by any of its Subsidiaries to the Company or to another Subsidiary in the ordinary course of business consistent with past practice;
- (c) (i) issue, deliver or sell, or authorize the issuance, delivery or sale of, any Company Securities or Company Subsidiary Securities, other than (x) the issuance of any shares of the Company Stock upon the exercise or the settlement of Company Equity Awards that are outstanding on the date of this Agreement in accordance with the applicable terms thereof on the date of this Agreement, (y) the issuance of any shares of the Company Stock upon exercise or conversion of shares of Series A Preferred Stock or Equity Linked Convertible Notes in accordance with the terms thereof, and (z) issuances of securities of the Company's Subsidiaries to the Company or to wholly owned Subsidiaries of the Company or (ii) amend any term of any Company Security (in each case, whether by merger, consolidation or otherwise);
- (d) incur or commit to any capital expenditures or any obligations or liabilities in respect thereof, except for (i) those as may be contemplated by the Company's fiscal 2022 budget and capital expenditure plan made available to Parent prior to the date of this Agreement (whether or not such capital expenditures are made during the Company's 2022 fiscal year), or (ii) any other capital expenditures not to exceed \$750,000 in the aggregate;
- (e) acquire (by merger, consolidation, acquisition of stock or assets or otherwise), directly or indirectly, any assets (including Intellectual Property Rights, Technology and other intangible assets), securities or businesses, other than (i) acquisitions of supplies, materials and similar assets in the ordinary course of business of the Company and its Subsidiaries in a manner that is consistent with past practice, (ii) pursuant to Contracts in effect on the date of this Agreement, (iii) any other acquisitions for consideration that is not in excess of \$500,000 individually or in the aggregate, (iv) acquisitions solely involving one or more of the Company and any of its wholly owned Subsidiaries and (v) as contemplated by the Company's fiscal 2022 budget and capital expenditure plan made available to Parent prior to the date of this Agreement;
- (f) sell, assign, license, lease or otherwise transfer, or abandon or create or incur any material Lien on any of the Company's or its Subsidiaries' assets (including Intellectual Property Rights, Technology and other

intangible assets), securities, properties, interests or businesses in excess of \$500,000 in the aggregate, other than (i) sales, leases or transfers in the ordinary course of business consistent with past practice, (ii) non-exclusive licenses granted to customers, suppliers, distributors, resellers, channel partners and other third parties in the ordinary course of business consistent with past practices, (iii) Permitted Liens or (iv) the abandonment of non-material Intellectual Property Rights in the ordinary course of business consistent with past practice; provided, that in no event shall the Company or any of its Subsidiaries assign or grant any exclusive licenses of any Intellectual Property Rights or Technology, or grant any other licenses of any Intellectual Property Rights or Technology other than non-exclusive licenses granted in the ordinary course of business consistent with past practices to the extent permitted above in Section 6.1(f)(ii), for the avoidance of doubt not to include any Company Intellectual Property Rights or Technology which have been assigned or granted in the ordinary course of the business, in each case, without Parent's prior written consent (which consent shall not be unreasonably be withheld):

- (g) make any loans, advances or capital contributions to, or investments in, any other Person in excess of \$500,000 in the aggregate, other than intercompany indebtedness;
- (h) create, incur, suffer to exist or assume any indebtedness for borrowed money or guarantees thereof, other than intercompany indebtedness and as contemplated by the Company's fiscal 2022 budget and capital expenditure plan set forth on Section 6.1(h) of the Company Disclosure Letter;
- (i) other than in the ordinary course of business consistent with past practice, enter into any agreement or arrangement that limits or otherwise restricts in any material respect the Company or any of its Subsidiaries from engaging or competing in any line of business, in any location or with any Person, or would purport to limit, after the Effective Time, Parent or any of its Subsidiaries from engaging or competing in any line of business in any material respect;
- (j) other than in the ordinary course of business consistent with past practice (including renewals consistent with the terms thereof), (i) amend or modify in any material respect or terminate (excluding terminations or renewals upon expiration of the term thereof in accordance with the terms thereof) any Company Material Contract or waive, release or assign any material rights, claims or benefits under any Company Material Contract or (ii) enter into any Contract that would have been a Company Material Contract had it been entered into prior to the date of this Agreement;
- (k) recognize any new labor organization, union, employee association, trade union, works council or other similar employee representative, or negotiate, enter into, amend, modify or terminate any Collective Bargaining Agreement;
  - (l) grant or modify any equity or equity-based awards;
- (m) except (i) as required pursuant to a Company Plan or a Contract in effect prior to the date of this Agreement, or (ii) as otherwise required by applicable Law, (A) grant or provide any severance, retention or termination payments or benefits to any current or former employee, officer, non-employee director, independent contractor or consultant of the Company or any of its Subsidiaries (including any obligation to gross-up, indemnify or otherwise reimburse any such individual for any Tax incurred by any such individual, including under Section 409A or 4999 of the Code), (B) accelerate the time of payment or vesting of, or the lapsing of restrictions with respect to, or fund or otherwise secure the payment of, any compensation or benefits to any current or former employee, officer, non-employee director, independent contractor or consultant of the Company or any of its Subsidiaries, (C) increase the compensation payable to any current or former employee, officer, non-employee director, independent contractor or consultant of the Company or any of its Subsidiaries, other than routine increases in base salaries or hourly base wage rates, as applicable, to employees in the ordinary course of business consistent with past practice and not in excess of the Company's 2022 established budget as set forth on Section 6.1(m) of the Company Disclosure Letter, (D) establish, adopt, terminate or amend any

Company Plan or any plan, program, arrangement, policy or agreement that would be a Company Plan if it were in existence on the date of this Agreement, (E) hire any employee of the Company or any of its Subsidiaries or engage any other individual to provide services to the Company or any of its Subsidiaries, other than as set forth in Section 6.1(m) of the Company Disclosure Letter, and provided that (1) any such hire(s) that result in an increase of 10% or more of the total Company workforce shall be approved in writing prior to such hire; (2) any such hire will be consistent with past Company practice in connection with the needs of the business of the Company as in effect immediately prior to the date of this Agreement; and (3) any compensation or benefits provided to any newly-hired employees will be on substantially the same terms and conditions as similarly-situated employees of the Company as of the date of this Agreement, (F) terminate the employment of any current employee or the engagement of any individual independent contractor of the Company or any of its Subsidiaries other than for cause or for performance-related reasons or a non-Merger related reason, or (G) promote any employee of the Company or any of its Subsidiaries to a position that reports directly to the Chief Executive Officer of the Company;

- (n) knowingly waive, release, limit or condition any restrictive covenant obligation of any current or former employee or independent contractor of the Company or any of its Subsidiaries, in each case that would reasonably be expected to have more than a de minimis adverse effect on the Company and its Subsidiaries, taken as a whole;
- (o) change the Company's methods of financial accounting, except as required by GAAP or in RegulationS-X of the 1934 Act (or any interpretation thereof), any Governmental Authority or applicable Law;
- (p) (i) make or change any material election with respect to Taxes, (ii) amend any Tax Return that would have the effect of causing a material amount of Taxes to be due in a taxable period (or portion thereof) following the Closing Date, (iii) agree or settle any claim or assessment in respect of a material amount of Taxes, (iv) agree to an extension or waiver of the limitation period for any claim or assessment in respect of a material amount of Taxes, (v) enter into any closing agreement within the meaning of Section 7121 of the Code (or any similar provision of state, local, or non-U.S. Law) with respect to a material amount of Taxes or (vi) surrender any right to a material refund of Taxes;
- (q) adopt or publicly propose a plan of complete or partial liquidation, restructuring, recapitalization or other reorganization, in each case, of the Company or any material Subsidiary of the Company;
- (r) enter into any material interest rate swaps, foreign exchange or other similar hedging arrangements other than for purposes of offsetting a bona fide exposure (including counterparty risk);
- (s) fail to maintain, allow to lapse or abandon any Owned Intellectual Property Rights or any Intellectual Property Rights that are exclusively licensed to the Company or any of its Subsidiaries (other than to the extent permitted pursuant to Section 6.1(f) hereof);
- (t) settle, offer or propose to settle any Proceeding involving or against the Company or any of its Subsidiaries that requires payment by the Company or any of its Subsidiaries in excess of \$1 million or that would include any non-monetary relief that would have more than a de minimis adverse effect on the Company and its Subsidiaries, taken as a whole, or commence any Proceeding other than the commencement of any Proceeding seeking damages of less than \$1 million or relating to the transactions contemplated hereby;
  - (u) agree, resolve or commit to do any of the foregoing.

Parent and Merger Sub Inc. acknowledge and agree that: (i) nothing contained in this Agreement shall give Parent or Merger Sub Inc., directly or indirectly, the right to control or direct the Company's operations prior to the Closing, (ii) prior to the Closing, the Company shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and its Subsidiaries' operations and (iii) notwithstanding

anything to the contrary set forth in this Agreement, no consent of Parent or Merger Sub Inc. shall be required with respect to any matter set forth in this Section 6.1 or elsewhere in this Agreement to the extent that the requirement of such consent could violate any applicable Law. In addition, for the sake of clarity, Parent does not have the right to direct and control the litigation, including any defenses, which right remains with the Company or the applicable Subsidiary of the Company.

#### Section 6.2 Modifications of Certain Outstanding Securities and Awards.

- (a) <u>Equity Linked Convertible Notes</u>. Concurrently with the execution of this Agreement (and in any event prior to the close of trading on Nasdaq on the date such execution occurs), in order to consummate the transactions contemplated by <u>Section 3.1</u> on the terms set forth therein, the Company shall amend or cause to be amended the Equity Linked Convertible Notes, substantially in the form of <u>Exhibit A</u> hereto (the "Convertible Note Amendments").
- (b) <u>Series A Preferred Stock</u>. Immediately following receipt of Company Stockholder Approval and at least one Business Day prior to Closing, the Company shall amend or cause to be amended the Certificate of Designations, substantially in the form of <u>Exhibit B</u> hereto (the "COD Amendment").
- (c) <u>Pool A and Pool B Awards</u>. Concurrently with the execution of this Agreement, the Company has executed an amendment, substantially consistent with the form set forth on <u>Exhibit C</u>, to each Pool A Performance Award and Pool B Performance Award held by each individual whose name is set forth on Section 6.2(c) of the Company Disclosure Letter. From and after the execution of this Agreement until the Effective Time, the Company shall use commercially reasonable efforts to amend each remaining Pool A Performance Award and Pool B Performance Award, substantially in the form of Exhibit C hereto (together with those amendments executed concurrently herewith, the "Award Amendments").
- (d) Each of the Investor Persons shall be a third-party beneficiary of Section 3.1, Section 6.2(a) and (b), Section 9.1(a) and Section 11.2, and no amendment to any such Section shall be effected without the prior written consent of the Investor Persons.

Section 6.3 <u>Information and Consultation</u>. The Company's Subsidiaries shall comply with their information and consultation obligations as required under applicable works council Laws and Collective Bargaining Agreements before the Closing Date.

#### Section 6.4 No Solicitation; Other Offers.

(a) General Prohibitions. From and after the date hereof until the earlier to occur of the Effective Time or the date of termination of this Agreement in accordance with Article X, neither the Company nor any of its Subsidiaries shall, nor shall the Company or any of its Subsidiaries authorize any of its or their Representatives to, directly or indirectly, (i) solicit, initiate or knowingly take any action to facilitate or encourage any inquiries regarding, or the making or submission of any proposal or offer, that constitutes, or could reasonably be expected to lead to, a Company Acquisition Proposal, (ii) enter into, engage in or participate in any discussions or negotiations with, furnish any information relating to the Company or any of its Subsidiaries or afford access to the business, properties, assets, books, records, work papers and other documents related to the Company or any of its Subsidiaries to, otherwise cooperate in any way with, or assist, participate in, knowingly facilitate or knowingly encourage any effort by, any Third Party that, to the Company's knowledge, is seeking to make, or has made, a Company Acquisition Proposal (other than to state the terms of this Agreement prohibit such discussion) (iii) make a Company Adverse Recommendation Change or (iv) enter into any agreement in principle, letter of intent, indication of interest, term sheet, merger agreement, acquisition agreement, option agreement or other Contract relating to a Company Acquisition Proposal (other than an Acceptable Confidentiality Agreement). It is agreed that any violation of the restrictions on the Company set forth in this Section 6.4(a) by any Subsidiary of the Company or any Representative of the Company or any of its

Subsidiaries, in the case of Representatives, (x) acting at the direction or (y) to the knowledge of the Company, provided that the Company did not direct such Representatives to cease violating such restrictions reasonably promptly after acquiring such knowledge, shall be a breach of this <u>Section 6.4(a)</u> by the Company.

- (b) Exceptions. Notwithstanding Section 6.4(a), at any time prior to receipt of the Company Stockholder Approval:
- (i) the Company, directly or indirectly through its Representatives, may (A) engage in discussions with any Third Party and its Representatives that have made after the date hereof a written Company Acquisition Proposal that did not result from a material breach of Section 6.4 and (B) furnish to any such Third Party and its Representatives any information relating to the Company or any of its Subsidiaries or afford access to the business, properties, assets, books, records, work papers and other documents related to the Company or any of its Subsidiaries or other non-public information relating to the Company or any of its Subsidiaries pursuant to a confidentiality agreement (a copy of which shall be provided for informational purposes only to Parent) with such Third Party and/or such Representatives with terms not materially less favorable to the Company than those contained in the Confidentiality Agreement (an "Acceptable Confidentiality Agreement"); provided that all such information (to the extent that such information has not been previously provided or made available to Parent) is provided or made available to Parent, as the case may be, promptly (and in any event within 24 hours) following the time it is provided or made available to such Third Party and/or such Representatives; provided, further, that in each case of clause (A) and (B), if, and only if, prior to taking any action described inclauses (A) or (B), the Company Board or the Company Special Committee determines in good faith, after consultation with outside legal counsel, that based on the information then available and after consultation with its outside financial advisor, such Company Acquisition Proposal is or is reasonably expected to result in a Superior Proposal;
- (ii) subject to compliance with <u>Section 6.4(e)</u>, the Company Board may make a Company Adverse Recommendation Change (A) if the Company receives a Superior Proposal (that is not withdrawn) after the date of this Agreement that did not arise from or in connection with a material breach of the Company's obligations in this <u>Section 6.4</u> or (B) in response to an Intervening Event; and
  - (iii) subject to compliance with Section 6.4(e) and Section 10.1(d)(iii), the Company may terminate this Agreement under Section 10.1(d)(iii);

provided that, in the case of <u>clause (i), (ii)</u> and <u>(iii)</u> if, and only if, prior to taking any action described in<u>clause (i), (ii)</u> and <u>(iii)</u>, the Company Board or the Company Special Committee determines in good faith, after consultation with outside legal counsel that the failure to take such action would be inconsistent with its fiduciary duties under Delaware law.

- (c) In addition, nothing contained herein shall prevent the Company Board from (i) complying with Rule 14e-2(a) or Rule 14d-9 under the 1934 Act or (ii) making any disclosure to the stockholders of the Company if the Company Board determines in good faith, after consultation with the outside legal counsel, that the failure to make such disclosure would be inconsistent with its fiduciary duties under Delaware law; provided, that any such disclosure referred to in clause (i) or (ii) that relates to a Company Acquisition Proposal shall be deemed to be a Company Adverse Recommendation Change unless (x) the Company Board expressly reaffirms the Company Board Recommendation in or in connection with such disclosure or (y) such disclosure is a "stop, look and listen" communication to the stockholders of the Company pursuant to Rule 14d-9(f) promulgated under the 1934 Act; provided, further, that this Section 6.4(c) shall not be deemed to permit the Company Board to make a Company Adverse Recommendation Change except, in each case, to the extent permitted by Section 6.4(e).
- (d) From and after the date hereof and until the earlier to occur of the Effective Time or the date of termination of this Agreement in accordance with Article X, the Company shall notify Parent promptly (but in no event later than 48 hours) after receipt by the Company (or any of its Representatives) of any Company Acquisition Proposal or any request for information relating to the Company or any of its Subsidiaries for the

purpose of facilitating the submission of a Company Acquisition Proposal or request for access to the business, properties, assets, books, records, work papers or other documents relating to the Company or any of its Subsidiaries by any Third Party that, to the Company's knowledge, has made such request for the purpose of facilitating the submission of a Company Acquisition Proposal. The Company shall provide such notice orally and in writing and shall identify the Third Party making, and the material terms and conditions of, any such Company Acquisition Proposal. The Company shall keep Parent reasonably informed, on a reasonably prompt basis upon Parent's request, of the status of any such Company Acquisition Proposal. Any material amendment to any Company Acquisition Proposal, as reasonably determined by the Company in good faith, will be deemed to be a new Company Acquisition Proposal for purposes of the Company's compliance with this Section 6.4(d).

- (e) The Company Board shall not make a Company Adverse Recommendation Change or terminate this Agreement pursuant to Section 10.1(d)(ii), unless (i) the Company notifies Parent in writing, at least three Business Days before taking that action, of its intention to do so, specifying in reasonable detail the reasons for such Company Adverse Recommendation Change (which notice shall not constitute a Company Adverse Recommendation Change), attaching (A) in the case of a Company Adverse Recommendation Change to be made in connection with a Superior Proposal or a termination of this Agreement pursuant to Section 10.1(d)(ii), the most current version (in unredacted form) of the proposed agreement under which a Superior Proposal is proposed to be consummated (and any applicable transaction and financing documents) and the identity of the Third Party making the Company Acquisition Proposal, or (B) in the case of a Company Adverse Recommendation Change to be made pursuant to an Intervening Event, a reasonably detailed description of the reasons for making such Company Adverse Recommendation Change, (ii) the Company has negotiated, and has caused its Representatives to negotiate in good faith with Parent (to the extent that Parent desires to negotiate during this period) during such notice and negotiation period any revisions to the terms of this Agreement that Parent proposes and has not withdrawn in response to such Superior Proposal and that would be binding on Parent if accepted by the Company and (iii) following the end of such notice and negotiation period, the Company Board or the Company Special Committee shall have determined, in consultation with outside legal counsel and its outside financial advisor, and giving due consideration to such binding revisions proposed by Parent, that (A) in the case of a Company Adverse Recommendation Change to be made in connection with a Superior Proposal or a termination of this Agreement pursuant to Section 10.1(d)(ii), such Superior Proposal would nevertheless continue to constitute a Superior Proposal (assuming such revisions proposed by Parent and not withdrawn were to be given effect) (it being understood and agreed that any amendment to the financial terms or other material terms of such Superior Proposal shall require a new written notification from the Company; provided that for the purposes of such new notification the reference to "three Business Days" in Section 6.4(e) shall be deemed to be "two Business Days") and (B) in the case of a Company Adverse Recommendation Change to be made pursuant to an Intervening Event, such Intervening Event would nevertheless necessitate the need for such Company Adverse Recommendation Change (it being understood and agreed that any material change to the facts and circumstances relating to such Intervening Event shall require a new written notification from the Company; provided that for the purposes of any such new notification the reference to "three Business Days" in Section 6.4(e) shall be deemed to be "two Business Days"), and in either case, the Company Board or the Company Special Committee determines in good faith, after consultation with outside legal counsel, that failure to take such action would be inconsistent with its fiduciary duties under Delaware law.
- (f) Parent will promptly notify the Parent Special Committee in the event of any such communication from the Company to Parent under this Section 6.4 and shall ensure that the Parent Special Committee and its legal advisors shall lead and direct any negotiations with the Company under this Section 6.4.
- (g) During the period commencing with the execution and delivery of this Agreement and continuing until the earlier to occur of the Effective Time or the termination of this Agreement pursuant to <u>Article X</u>, the Company shall not terminate, amend, modify or waive any provision of any confidentiality, "standstill" or similar agreement to which the Company or any of its Subsidiaries is a party and shall enforce, to the fullest extent permitted under applicable Law, the provisions of any such agreement, including by obtaining injunctions

to prevent any breaches of such agreements and to enforce specifically the terms and provisions thereof, unless the Company Board or Company Special Committee determines in good faith, after consultation with outside legal counsel, that failure to take any such action would be inconsistent with its fiduciary duties under Delaware law

Section 6.5 Company Stockholders' Meeting. The Company shall, as soon as practicable following the date of this Agreement, establish a record date for, duly call, give notice of, convene and hold a special meeting of the holders of Company Stock (such meeting, including as it may be adjourned or postponed from time to time, the "Company Stockholders' Meeting") for the purpose of obtaining the Company Stockholder Approval. Subject to Section 6.4, the Company Board shall recommend that the holders of Company Stock approve and adopt this Agreement and the transactions contemplated by this Agreement, including the Merger (the "Company Board Recommendation"). Unless the Company Board has effected a Company Adverse Recommendation Change in accordance with Section 6.4, the Company shall use its reasonable best efforts to solicit from the holders of Company Stock proxies in favor of the adoption and approval of this Agreement and to take all other action necessary or advisable to secure the Company Stockholder Approval. Subject to Section 6.4, the Proxy Statement shall include the Company Board Recommendation. Notwithstanding anything in this Agreement to the contrary, unless this Agreement is terminated in accordance with Article X, the Company shall submit this Agreement for adoption and approval by the holders of Company Stock at the Company Stockholders' Meeting. Notwithstanding anything in this Agreement to the contrary, the Company may postpone or adjourn the Company Stockholders' Meeting (i) to solicit additional proxies for the purpose of obtaining the Company Stockholder Approval, (ii) for the absence of quorum and (iii) to allow reasonable additional time for the filing and/or mailing of any supplemental or amended disclosure that the Company has determined after consultation with outside legal counsel is necessary under applicable Law and for such supplemental or amended disclosure to be disseminated and reviewed by the holders of Company Stock prior to the Company Stockholders' Meeting. The only matters to be voted upon at the Company Stockholders' Meeting are the matters required pursuant to the Company Stockholder Approval and routine proposals required in connection with such vote.

# ARTICLE VII COVENANTS OF PARENT AND MERGER SUB INC.

Section 7.1 Conduct of Parent. From the date of this Agreement until the earlier to occur of the Effective Time and the date of termination of this Agreement in accordance with Article X, except as expressly contemplated by this Agreement, as set forth in Section 7.1 of the Parent Disclosure Letter, as consented to in writing by the Company (such consent not to be unreasonably withheld, conditioned or delayed) or as required by applicable Law, Parent shall, and shall cause each of its Subsidiaries to conduct its business in all material respects in the ordinary course; provided, however, that no action that is specifically permitted by any of clauses (a) through (d) of this Section 7.1 shall be deemed a breach of this sentence. Without limiting the generality of the foregoing, from the date of this Agreement until the earlier to occur of the Effective Time and the date of termination of this Agreement in accordance with Article X, except as expressly contemplated by this Agreement, as set forth in Section 7.1 of the Parent Disclosure Letter, as consented to in writing by the Company (such consent not to be unreasonably withheld, conditioned or delayed) or as required by applicable Law, Parent shall not, nor shall it permit any of its Subsidiaries to:

(a) amend the certificate of incorporation or bylaws of Parent in a manner that would have an adverse impact on the value of Parent Stock or that would reasonably be expected to prevent, or impede or delay, the consummation of the Merger or the transactions contemplated hereby (provided, that any amendment to its certificate of incorporation solely to increase the authorized number of shares of any class or series of the capital stock of Parent shall in no way be restricted by the foregoing);

(b) adopt or publicly propose a plan of complete or partial liquidation, restructuring, recapitalization or other reorganization;

- (c) take any action (or omit to take any action) with the knowledge that such action (or omission) would reasonably be expected to result in a requirement to seek the approval by holders of Parent Stock of the transactions contemplated hereby; or
  - (d) agree, resolve or commit to do any of the foregoing.
- Section 7.2 Obligations of Merger Sub Inc. Parent shall cause Merger Sub Inc. to perform when due its obligations under this Agreement and to consummate the Merger pursuant to the terms and subject to the conditions set forth in this Agreement.
- Section 7.3 Approval by Sole Stockholder of Merger Sub Inc. Immediately following the execution and delivery of this Agreement by the Parties, ProFrac Holdings II, LLC, as the sole stockholder of Merger Sub Inc., shall adopt and approve this Agreement and the transactions contemplated hereby, including the Merger, in accordance with the DGCL, by written consent.

#### Section 7.4 Director and Officer Indemnification.

- (a) From and after the Effective Time, Parent and the Surviving Corporation agree that, subject to applicable Law, all rights to indemnification of each former and present director or officer of the Company or any of its Subsidiaries (each, together with such person's heirs, executors or administrators, a "Company Indemnified Party") with respect to acts or omissions occurring at or prior to the Effective Time as provided in their respective certificates of incorporation or by-laws (or comparable organizational documents) as in effect on the date of this Agreement or in any agreement, a true and complete copy of which agreement has been provided by the Company to Parent prior to the date of this Agreement, to which the Company or any of its Subsidiaries is a party which is in effect as of the date of this Agreement, shall survive the Merger and continue in full force and effect in accordance with their terms, and the Surviving Corporation shall (and Parent shall cause the Surviving Corporation to) honor all the terms thereof. For a period of no less than six (6) years after the Effective Time, Parent, to the fullest extent permitted under applicable Law, shall cause to be maintained in effect the provisions in the certificates of incorporation and bylaws and comparable organizational documents of the Surviving Corporation and each Subsidiary of the Company (or in such documents of any successor thereto) regarding indemnification, exculpation and expense advancement in effect as of immediately prior to the Effective Time, and, during such six (6) year period, shall not amend, repeal or otherwise modify any such provisions in any manner that would adversely affect the rights thereunder of any individual who immediately before the Effective Time was a Company Indemnified Party, except as required by applicable Law.
- (b) Parent shall cause the Surviving Corporation, and each of the Surviving Corporation hereby agrees, to either (i) continue to maintain in effect for a period of no less than six (6) years after the Effective Time the Company's directors' and officers' insurance policies (the "D&O Insurance") in place as of the date of this Agreement or (ii) purchase comparable D&O Insurance (from a carrier with the same or better credit rating as the Company's D&O Insurance carrier) for such six (6)-year period, in each case, with coverage for the persons who are covered by the Company's existing D&O Insurance, with terms, conditions, retentions and levels of coverage at least as favorable to the insured individuals as the Company's existing D&O Insurance with respect to matters existing or occurring prior to the Effective Time; provided, that in no event shall Parent or the Surviving Corporation be required to expend for such policies pursuant to this sentence an annual premium amount in excess of 300% of the amount per annum the Company paid in its last full fiscal year, which amount is set forth in Section 7.4(b) of the Company Disclosure Letter (the "Premium Cap"); provided further, that if the amount necessary to procure such insurance coverage exceeds the Premium Cap, the Company may purchase the most advantageous policy available for an amount not to exceed the Premium Cap. At the Company's option, the Company may purchase, prior to the Effective Time, a prepaid "tail policy" for a period of no more than six (6) years after the Effective Time with coverage for the persons who are covered by the Company's existing D&O Insurance, with terms, conditions, retentions and levels of coverage at least as favorable to the insured individuals as the Company's existing D&O Insurance with respect to matters existing or occurring prior to the

Effective Time, in which event Parent shall cease to have any obligations under the first sentence of this Section 7.4(b); provided, that the aggregate premium for such policies shall not exceed the Premium Cap; provided further, that if the amount of annual premiums necessary to maintain or procure such insurance coverage exceeds the Premium Cap, Parent and the Surviving Corporation shall procure and maintain for such six-year period the most advantageous policy available for an annual premium equal to the Premium Cap. In the event the Company elects to purchase such a "tail policy," the Surviving Corporation shall (and Parent shall cause the Surviving Corporation to) maintain such "tail policy" in full force and effect for a period of no less than six (6) years after the Effective Time and continue to honor its obligations thereunder.

- (c) In the event that either Parent, the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers or conveys all or substantially all of its properties, rights and other assets to any Person, then, and in each such case, Parent shall cause the successors and assigns of Parent and the Surviving Corporation, as the case may be, to succeed to or assume the applicable obligations of such Party set forth in this Section 7.4.
- (d) The provisions of this Section 7.4 shall survive consummation of the Merger, are intended to be for the benefit of, and will be enforceable by, each of the Company Indemnified Parties and are in addition to, and not in substitution for, any other rights to indemnification or contribution that any such person may have by Contract, at Law or otherwise.
- Section 7.5 <u>Stock Exchange Listing</u>. Parent shall use its best efforts to cause the shares of Parent Stock to be issued as the Merger Consideration to be approved for listing on Nasdaq, subject to official notice of issuance, prior to the Effective Time.

# Section 7.6 Employee Matters.

- (a) Parent agrees that each Continuing Employee (other than any Continuing Employee covered by a Collective Bargaining Agreement) shall, as of the Effective Time, be provided with compensation and benefits that are substantially comparable in the aggregate to the compensation and benefits provided to similarly situated employees of Parent during such period. In addition to the foregoing, Parent agrees that, at the Effective Time, Parent shall either (i) continue the agreements covering Non-U.S. Service Providers on terms that are substantially comparable in all respects to the agreements as in effect immediately prior to the Effective Time, or (ii) to the extent that Parent elects to hire any such Non-U.S. Service Provider, Parent shall provide each such Non-U.S. Service Provider with base compensation, incentive opportunities and employee benefits that are substantially comparable to the base compensation, incentive opportunities and employee benefits provided to similarly situated employees of Parent.
- (b) Parent shall give or cause to be given each Continuing Employee andNon-U.S. Service Provider service credit for such Continuing Employee's employment with (and Non-U.S. Service Provider's service to) the Company and its Subsidiaries for purposes of vesting, benefit accrual, level of benefit and eligibility to participate under each applicable Parent benefit plan, as if such service had been performed with Parent (including, for purposes of determining vacation, other paid time off and severance benefits), except for benefit accrual under defined benefit pension plans, for purposes of qualifying for subsidized early retirement benefits or to the extent it would result in a duplication of benefits. For purposes of this <u>Section 7.6</u>, "Continuing Employee" means each employee of the Company and its Subsidiaries at the Effective Time, who continues to be employed with the Company or any of its Subsidiaries immediately after the Closing Date.
- (c) Upon the written request of Parent at least three (3) Business Days prior to the Closing Date (the 'Plan Termination Notice'), effective as of immediately prior to Closing Date and contingent upon the occurrence of the Closing, the Company shall terminate or shall cause the termination of any or all U.S. tax qualified retirement plans provided to Employees of the Company and its Subsidiaries and any other Company Plan or relationship

with any professional employer organization (if applicable) as requested by the Parent, to the extent directed by Parent and in compliance with applicable Law. If the Parent provides a Plan Termination Notice to the Company, the Company shall deliver to the Parent, prior to the Closing Date, evidence that the Company Board has validly adopted resolutions to terminate the Company Plans or the relationship with any professional employer organization subject to termination as set forth in this paragraph (the form and substance of which resolutions shall be subject to review and approval of the Parent), effective no later than the date immediately preceding the Closing Date.

- (d) As soon as possible following the execution of this Agreement, the Company shall make commercially reasonable efforts to provide Parent with copies of all material documents embodying and relating to each Foreign Company Plan, including the Foreign Company Plan document, all amendments thereto and all related trust documents, the most recent summary plan description, the most recent actuarial valuation report, and the most recent tax return filing, in each case, to the extent applicable.
- (e) Parent agrees that the individuals listed on Section 7.6(e) of the Company Disclosure Letter will be eligible to receive severance benefits in accordance with the Company's Severance Benefit Plan and their individual Participation Notices as in effect on June 20, 2022 and the terms and conditions thereof during the 12 months following the Effective Time. Continuing Employees and Non-U.S. Service Providers who never had an individual Participation Notice in connection with the Company's Severance Benefit Plan shall be eligible to receive severance benefits as provided in Section 7.6(e) of the Company Disclosure Letter.
- (f) Nothing contained in this Section 7.6, expressed or implied, shall (i) be treated as the establishment, amendment or modification of any Company Plan or constitute a limitation on rights to amend, modify, merge or terminate after the Effective Time any Company Plan, (ii) give any current or former Employee, director, independent contractor, or other service provider (including Non-U.S. Service Providers) of the Company or its Subsidiaries (including any beneficiary or dependent thereof), or any labor organization, union, works council, employee association, trade union, other similar employee representative body, any third-party beneficiary or other rights or (iii) obligate Parent or any of its Affiliates to (A) maintain any particular Company Plan or any Parent plan or (B) retain the employment or services of any Employee, director, other independent contractor or other service provider (including Non-U.S. Service Providers) of the Company and its Subsidiaries.

# ARTICLE VIII COVENANTS OF PARENT AND THE COMPANY

#### Section 8.1 Efforts.

- (a) Subject to the terms and conditions of this Agreement, each of the Company and Parent shall use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, and assist and cooperate with the other in doing, all things necessary, proper or advisable under applicable Law to consummate and make effective the Merger and the other transactions contemplated by this Agreement as promptly as reasonably practicable after the date of this Agreement, including (i) preparing and filing as promptly as reasonably practicable with any Governmental Authority or other Third Party all documentation to effect all necessary, proper or advisable filings, notices, petitions, statements, registrations, submissions of information, applications and other documents and (ii) obtaining and maintaining all approvals, consents, registrations, permits, authorizations and other confirmations required to be obtained from any Governmental Authority or other Third Party that are necessary, proper or advisable to consummate and make effective the Merger and the other transactions contemplated by this Agreement; provided that no Party shall be required to pay any consideration therefor other than filing fees and other regulatory payments (and associated expenses, such as legal fees, in connection therewith).
- (b) In furtherance and not in limitation of the foregoing, each of Parent and the Company shall make, as promptly as reasonably practicable, and in any event within ten (10) Business Days of the date of this Agreement

an appropriate filing of a Notification and Report Form pursuant to the HSR Act with respect to the transactions contemplated by this Agreement. Each of the Company and Parent shall use its reasonable best efforts to supply as promptly as practicable any additional information and documentary material that may be reasonably requested pursuant to the foregoing, and use its reasonable best efforts to take all other actions necessary to cause the expiration or termination of the applicable waiting periods regarding the foregoing as soon as reasonably practicable. The Company and Parent will each request early termination of the waiting period with respect to the Merger under the HSR Act.

- (c) Except as prohibited by applicable Law or Order, each of Parent and the Company shall use its reasonable best efforts to (i) cooperate with each other in connection with any filing or submission with a Governmental Authority in connection with the transactions contemplated by this Agreement and in connection with any investigation or other inquiry by or before a Governmental Authority relating to the transactions contemplated by this Agreement, including any proceeding initiated by a private party, (ii) promptly inform the other Party of (and if in writing, supply to the other Party) any communication (other than any ministerial communications) received by such Party from, or given by such Party to, the Federal Trade Commission, the Antitrust Division of the Department of Justice, or any other similar Governmental Authority and of any material communication received or given in connection with any proceeding by a private party, in each case regarding any of the transactions contemplated by this Agreement, (iii) consult with each other prior to taking any material position with respect to the filings under the HSR Act or any other Competition Law in discussions with or filings to be submitted to any Governmental Authority, (iv) permit the other to review and discuss in advance, and consider in good faith the views of the other in connection with, any analyses, presentations, memoranda, briefs, arguments, opinions and proposals to be submitted to any Governmental Authority with respect to filings under the HSR Act or any other Competition Law, and (v) coordinate with the other in preparing and exchanging such information and promptly provide the other (and its counsel) with copies of all filings, presentations or submissions (and a summary of any oral presentations) made by such Party with any Governmental Authority relating to this Agreement or the transactions contemplated hereby under the HSR Act or any other Competition Law. Notwithstanding anything to the contrary contained herein, Parent shall, on behalf of the Parties, have control over and lead all communications and strategy relating to obtaining all approvals, consents, waivers, registrations, permits, authorizations and other confirmations under any Competition Law from any Governmental Authority or other Third Party in connection with consummating the Merger and the other transactions contemplated by this Agreement as it relates to Competition Laws or to any litigation under any Competition Law arising therefrom; provided, however, that Parent shall consult in good faith with the Company and its counsel regarding the foregoing and shall in good faith take into account the views of the Company and the Company's counsel. Notwithstanding the foregoing, matters relating to the Registration Statement, Proxy Statement and Information Statement shall be governed by Section 8.3 and not this Section 8.1(c).
- (d) Unless prohibited by applicable Law or Order or by the applicable Governmental Authority, each of the Company and Parent shall (i) to the extent reasonably practicable, not participate in or attend any meeting, or engage in any conversation (other than ministerial conversations) with any Governmental Authority in respect of the Merger (including with respect to any of the actions referred to in Section 8.1(a) but other than as contemplated by Section 8.3) without the other, (ii) to the extent reasonably practicable, give the other reasonable prior notice of any such meeting or conversation and (iii) in the event one such Party is prohibited by applicable Law or Order or by the applicable Governmental Authority from participating or attending any such meeting or engaging in any such conversation, keep the non-participating Party reasonably apprised with respect thereto.
- (e) Notwithstanding anything in this Agreement to the contrary, in no event will Parent be obligated (i) to propose or agree to accept any undertaking or condition, (ii) to enter into any consent decree, to make any divestiture, (iii) to accept any operational restriction, or (iv) to take any other action that, in the reasonable judgment of Parent, could be expected to limit the rights of Parent with respect to any product lines or assets of Parent or the Company.

#### Section 8.2 Financing Cooperation.

- (a) Parent shall, and shall cause its Affiliates to, use commercially reasonable efforts, subject to this Section 8.2, to obtain and consummate a financing to fund the repayment in connection with the Closing of certain indebtedness of the Company and its Subsidiaries, which financing shall be on terms and conditions and subject to documentation, in each case, in form and substance satisfactory to Parent in its sole discretion, (the "Financing") as promptly as reasonably possible following the date of this Agreement.
- (b) The Company shall, and shall cause each of its Subsidiaries to, and shall use its reasonable best efforts to cause its and their Representatives to, provide to Parent such customary cooperation, as may be reasonably requested by Parent to assist Parent in arranging the Financing, which reasonable best efforts shall include:
  - (i) reasonable cooperation with customary marketing efforts of Parent for all or any portion of the Financing, including causing its management team, with appropriate seniority and expertise, and use its reasonable best efforts to cause its external auditors to assist in preparation for and to participate in a reasonable number of meetings, presentations, road shows, due diligence sessions, drafting sessions and sessions with rating agencies, in each case, upon reasonable notice and at mutually agreeable dates and times;
  - (ii) providing reasonable assistance with the preparation of customary rating agency presentations, road show materials, bank information memoranda, prospectuses and bank syndication materials, offering documents, private placement memoranda and similar documents (including the delivery, within forty (40) days of the end of each applicable fiscal quarter, of unaudited consolidated balance sheets and related consolidated statements of income and cash flows of the Company for each interim fiscal quarter ended since the last audited financial statements and at least forty-five (45) days prior to the Closing Date and the delivery of customary authorization and representation letters in connection therewith) customarily required in connection with the marketing and syndication of the Financing;
  - (iii) using reasonable best efforts to furnish Parent, within a reasonable amount of time following Parent's reasonable request, with information reasonably available to the Company relating to the Company and its Subsidiaries to the extent required to consummate the Financing, and subject to clause (d) below, providing reasonable assistance to Parent's preparation of pro forma financial information; and
  - (iv) promptly furnishing Parent with all documentation and other information related to the Company and its Subsidiaries as and solely to the extent required by any Governmental Authority with respect to the Financing under applicable "know your customer" and anti-money laundering rules and regulations, including the PATRIOT Act.
- (c) Notwithstanding anything to the contrary contained in this Agreement (including this Section 8.2): (i) nothing in this Agreement (including this Section 8.2) shall require any such cooperation to the extent that (A) it would require the Company to pay any commitment or other fees, reimburse any expenses or otherwise incur any liabilities that are not reimbursed by Parent as provided in clause (d) below or give any indemnities prior to the Closing, (B) it would require the Company to take any action that in the good faith judgment of the Company unreasonably interferes in a material manner with the ongoing business or operations of the Company and/or its Subsidiaries, (C) it would require the Company or any of its Subsidiaries to enter into, execute, or approve any agreement or other documentation prior to the Closing or agree to any change or modification of any existing agreement or other documentation that would be effective prior to the Closing (other than the execution of customary authorization and representation letters and any agreements to be executed at the Closing), (D) it would require the Company to provide any financial statements, (E) it would require the Company to provide any information the disclosure of which is prohibited by or restricted under applicable Law, or (F) it would require the Company, any of its Subsidiaries or any of their respective boards of directors (or equivalent bodies) to approve or authorize the Financing or any documentation related thereto, (ii) none of the Company or any of its Subsidiaries shall be required to provide, and Parent shall be solely responsible for, (A) the preparation of pro forma financial information, including pro forma cost savings, synergies, capitalization or other pro forma

adjustments desired to be incorporated into any pro forma financial information (it being understood and agreed that the Company and its Subsidiaries shall assist Parent in Parent's preparation of the pro forma financial information), (B) any description of all or any component of the Financing, including any such description to be included in any liquidity or capital resources disclosure or any "description of notes", (C) projections, risk factors or other forward-looking statements relating to any component of such financing or (D) subsidiary financial statements or any other information of the type required by Rule 3-09, Rule 3-10 or Rule 3-16 of Regulation S-X, and (iii) no action, liability or obligation (including any obligation to pay any commitment or other fees or reimburse any expenses) of the Company, its Subsidiaries, or any of their respective Representatives under any certificate, agreement, arrangement, document or instrument relating to the Financing shall be effective until the Closing.

- (d) Parent shall (i) promptly upon request by the Company, reimburse the Company for all of its reasonable and documented out-of-pocket fees and expenses (including reasonable and documented out-of-pocket attorneys' fees) incurred by the Company, any of its Subsidiaries or any of its or their Representatives in connection with any cooperation contemplated by this Section 8.2 and (ii) indemnify and hold harmless the Company, its Subsidiaries and its and their Representatives against any claim, loss, damage, injury, liability, judgment, award, penalty, fine, Tax, cost (including cost of investigation), expense (including reasonable and documented out-of-pocket attorneys' fees) or settlement payment, in each case, incurred as a result of, or in connection with, such cooperation or the Financing, except to the extent arising out of gross negligence, bad faith or willful misconduct of the Company, its Subsidiaries or its and their Representatives. It is understood that the condition precedent set forth in Section 9.2(b), as applied to the Company's obligations under this Section 8.2, shall be deemed to be satisfied unless (i) the Company shall have Willfully Breached this Section 8.2, (ii) such Willful Breach shall not have been cured within 10 days after receipt of written notice thereof from Parent and (iii) such Willful Breach shall have been the primary cause of Parent's failure to obtain the Financing. Parent acknowledges and agrees that obtaining the Financing is not a condition to Closing.
- (e) The Company hereby consents to the use of its trademarks and logos in connection with the Financing; provided, that such trademarks and logos are used solely in a manner that is not intended to or reasonably likely to harm or disparage in a material manner the Company or any of its Subsidiaries or the reputation or goodwill of the Company or any of its Subsidiaries.
- (f) Notwithstanding anything to the contrary contained in this Agreement, nothing contained in this Section 8.2 shall require, and in no event shall the commercially reasonable efforts of Parent be deemed or construed to require Parent to pay any fees to the lenders or increase any interest rates or original issue discounts or upfront fees applicable to the Financing in excess of an amount acceptable to Parent or agree to terms not acceptable to Parent, whether to secure waiver of any conditions contained therein or otherwise.

Section 8.3 Registration Statement; Information Statement; Proxy Statement. As soon as practicable following the date of this Agreement, Parent and the Company shall jointly prepare, and Parent shall file with the SEC, a registration statement on Form S-4 to register under the 1933 Act the offer and sale of Parent Stock pursuant to the Merger (together with all amendments thereto, the "Registration Statement"). The Registration Statement will include (i) a proxy statement/prospectus containing the information required under Schedule 14A promulgated under the 1934 Act relating to the solicitation of the Company Stockholder Approval and which will be used as a prospectus of Parent with respect to the issuance of Parent Stock pursuant to the Merger (the "Proxy Statement") and (ii) an information statement containing the information required under Schedule 14C promulgated under the 1934 Act relating to the Parent Stockholder Approval (the "Information Statement"). Each of Parent and the Company shall use its reasonable best efforts to (A) have the Registration Statement declared effective under the 1933 Act as promptly as practicable after such filing and (B) if the Registration Statement is declared effective by the SEC, keep the Registration Statement effective for so long as necessary to complete the Merger. Parent and the Company shall cause the Registration Statement, Proxy Statement and Information Statement to comply as to form in all material respects with requirements of applicable Law. Each of Parent and the Company shall furnish all information concerning Parent, the Company and their respective

Affiliates, as applicable, as may be reasonably requested by the other Party to be included in the Registration Statement, Proxy Statement and Information Statement so as to enable each of Parent and the Company to comply with its obligations under this Section 8.3. Parent and the Company shall cooperate in good faith to determine the information regarding Parent and the Company that is necessary to include in the Registration Statement, Proxy Statement and Information Statement in order to satisfy applicable Laws. Each of Parent and the Company shall promptly correct any information provided by it or any of its Representatives for use in the Registration Statement, Proxy Statement and Information Statement if and to the extent that such information shall have become false or misleading in any material respect, and Parent and the Company shall take all steps necessary to cause the Registration Statement, Proxy Statement and Information Statement, as applicable, as so corrected in respect thereof, to be filed with the SEC and to be disseminated to the holders of shares of Parent Stock and Company Stock, as applicable, in each case as and to the extent required by applicable Laws, the SEC or its staff or any applicable stock exchange. Each of Parent and the Company shall provide the other Party and its counsel a reasonable opportunity to review and comment on the Registration Statement, Proxy Statement and Information Statement prior to the filing thereof with the SEC, and Parent and the Company shall give reasonable and good faith consideration to any comments made by the other Party and its counsel (it being understood that Parent, the Company and their respective counsel shall provide any comments thereon as soon as reasonably practicable). Parent and the Company shall provide to the other Party and its counsel copies of all written comments or other material communications and a description of any oral comments that Parent, the Company or their counsel receives from the SEC or its staff with respect to the Registration Statement, Proxy Statement and Information Statement promptly after such receipt, and Parent and the Company shall provide the other Party and its counsel a reasonable opportunity to participate in the formulation of any response to any such comments of the SEC or its staff (including a reasonable opportunity to review and comment on any such response or amendment to the Registration Statement, Proxy Statement or Information Statement, to which Parent and the Company shall give reasonable and good faith consideration to any comments made by the other Party and its counsel (it being understood that Parent, the Company and their respective counsel shall provide any comments thereon as soon as reasonably practicable)) and to participate in any discussions with the SEC or its staff regarding any such comments. Parent and the Company shall promptly respond to any comments of the SEC or its staff regarding the Registration Statement, Proxy Statement and Information Statement. Parent shall notify the Company promptly of the time when the Registration Statement has become effective or any supplement or amendment to the Registration Statement has become effective and of the issuance of any stop order or suspension of the qualification of the shares of Parent Stock issuable in connection with the Merger for offering or sale in any jurisdiction. Parent shall also take any other action required to be taken under the 1933 Act, the 1934 Act, any applicable foreign or state securities or blue sky Laws and the rules and regulations thereunder in connection with the issuance of the shares of Parent Stock in the Merger.

Section 8.4 <u>Public Announcements</u>. The initial press release with respect to the execution of this Agreement and the transactions contemplated hereby shall be a joint press release in a form mutually agreed to by the Parties. Thereafter, Parent and the Company will use their respective reasonable best efforts to consult with the other Party before (a) participating in any media interviews, (b) engaging in meetings or calls with analysts and (c) providing any public statements (including press releases), in any such case to the extent relating to the transactions contemplated hereby and other than engaging in discussions in which the statements made are substantially similar to previous press releases, public disclosures or public statements made by Parent and the Company. None of the limitations set forth in this <u>Section 8.4</u> shall apply to any disclosure of any information concerning this Agreement or the transactions contemplated by this Agreement (i) in connection with or following a Company Adverse Recommendation Change and (ii) in connection with any dispute between the Parties regarding this Agreement or the transactions contemplated by this Agreement.

Section 8.5 Notices of Certain Events. Each of the Company and Parent shall promptly notify and provide copies to the other of:

(a) any material written notice from any Person alleging that the approval or consent of such Person is or may be required in connection with the Merger or the other transactions contemplated by this Agreement;

- (b) any written notice or other communication from any Governmental Authority or securities exchange in connection with the Merger or the other transactions contemplated by this Agreement;
- (c) any Proceeding or investigation, commenced or, to its knowledge, threatened against, the Company or any of its Subsidiaries or Parent or any of its Subsidiaries, as the case may be, that would be reasonably likely to (i) prevent or materially delay the consummation of the Merger or the other transactions contemplated hereby or (ii) result in the failure of any condition to the Merger set forth in Article X to be satisfied; and
- (d) the occurrence of any event which would or would be reasonably likely to (i) prevent or materially delay the consummation of the Merger or the other transactions contemplated hereby or (ii) result in the failure of any condition to the Merger set forth in Article X to be satisfied;

provided, that the delivery of any notice pursuant to this Section 8.5 shall not (A) affect or be deemed to modify any representation, warranty, covenant, right, remedy, or condition to any obligation of any Party hereunder or (B) update any section of the Company Disclosure Letter or the Parent Disclosure Letter.

### Section 8.6 Access to Information.

- (a) Upon reasonable notice, and subject to applicable Law, each Party shall (and shall cause its Subsidiaries to) afford to any other Party, its Affiliates and its officers, agents, control persons, employees, consultants, professional advisers (including attorneys, accountants and financial advisors) and the Persons providing the Financing ("Representatives") reasonable access during normal business hours, to all of its and its Subsidiaries' properties. books, Contracts, commitments, records, officers and employees and, during such period as a Party may from time to time reasonably request, the Party receiving the request shall (and shall cause its Subsidiaries to) furnish promptly to the requesting Party all other information concerning it, its Subsidiaries and each of their respective businesses, properties and personnel as the requesting Party may reasonably request, including with respect to the compliance program of the Party receiving the request; provided, however, that the Party receiving such request may restrict the foregoing access and the disclosure of information to the extent that, in the good faith judgment of such Party, (i) any Law applicable to such Party or its Subsidiaries requires such Party or its Subsidiaries to restrict or prohibit access to any such properties or information, (ii) the information is subject to confidentiality obligations to a Third Party, (iii) disclosure of any such information or document could result in the loss of attorney-client privilege (provided, that such Party and/or its counsel shall use their reasonable best efforts to enter into such joint defense agreements or other arrangements, as appropriate, so as to allow for such disclosure in a manner that does not result in the loss of attorney client privilege) or (iv) such access would unreasonably disrupt the operations of such Party or any of its Subsidiaries; provided, however, that with respect to clauses (i) through (iii) of this Section 8.6(a) as it relates to access and information provided to Parent and its Representatives, the Company shall use its reasonable best efforts to (A) obtain the required consent of such Third Party to provide such access or disclosure (provided that the Company shall not be required to pay any consideration therefor) or (B) develop an alternative to providing such information so as to address such matters that is reasonably acceptable to both Parent and the Company, provided, however, this Section 8.6(a) as it relates to access and information provided to the Company and its Representatives shall be limited to that which is reasonably necessary for the Company to confirm the accuracy, at any point in time prior to the Effective Time, of the representations and warranties of Parent and Merger Sub Inc. or the compliance by Parent and Merger Sub Inc. of their respective covenants, in each case as set forth in this Agreement.
- (b) With respect to the information disclosed pursuant to <u>Section 8.6(a)</u>, each of Parent and the Company shall comply with, and shall cause such party's Representatives to comply with, all of its obligations under the Confidentiality Agreement, which agreement shall remain in full force and effect in accordance with its terms.

Section 8.7 Section 16 Matters. Prior to the Effective Time, Parent and the Company shall use reasonable best efforts to take all such steps as may be required to cause any dispositions of Company Stock (including derivative securities with respect to Company Stock) or acquisitions of Parent Stock (including derivative

securities with respect to Parent Stock) resulting from the transactions contemplated by this Agreement by each individual who is subject to the reporting requirements of Section 16(a) of the 1934 Act with respect to the Company or will become subject to such reporting requirements with respect to Parent to be exempt under Rule 16b-3 promulgated under the 1934 Act, to the extent permitted by applicable Law.

Section 8.8 Stockholder Litigation. The Company shall promptly notify Parent in writing of any litigation related to this Agreement, the Merger or the other transactions contemplated by this Agreement that is brought, or, to the knowledge of the Company, threatened in writing, in the name of or against the Company and/or its directors (any such litigation, a "Company Transaction Litigation") and shall keep Parent fully informed on a reasonably current basis with respect to the status thereof. The Company shall give Parent the opportunity to participate, subject to a customary joint defense agreement, in the defense or settlement of any Company Transaction Litigation, and the Company shall not settle, agree to any undertakings or approve or otherwise agree to any waiver that may be sought in connection with such Company Transaction Litigation, without the prior written consent of Parent (not to be unreasonably withheld, conditioned or delayed). Without limiting in any way the Parties' obligations under Section 8.1, each of the Company and Parent shall, and shall cause their respective Subsidiaries to, cooperate in the defense or settlement of any litigation contemplated by this Section 8.8.

# ARTICLE IX CONDITIONS TO THE MERGER

- Section 9.1 Conditions to Obligations of Each Party. The obligations of Parent, Merger Sub Inc., and the Company to consummate the Merger are subject to the satisfaction, at or prior to the Closing, of the following conditions (which may be waived, in whole or in part, to the extent permitted by Law, by the mutual consent of Parent and the Company; provided however that any waiver of the condition in Section 9.1(a) shall also require the written consent of the Investor Persons):
- (a) <u>Company Stockholder Approval</u>. The Company Stockholder Approval shall have been obtained in accordance with applicable Law and the organizational documents of the Company.
- (b) <u>Statutes and Injunctions</u>. No Law or Order (whether temporary, preliminary or permanent) shall have been promulgated, entered, enforced, enacted or issued or be applicable to the Merger or the Parent Stock Issuance by any Governmental Authority that prohibits or makes illegal the consummation of the Merger or the Parent Stock Issuance.
- (c) <u>Regulatory Approval</u>. The waiting period under the HSR Act relating to the transactions contemplated by the Agreement shall have expired or been terminated.
- (d) <u>Registration Statement</u>. The Registration Statement shall have become effective under the 1933 Act or shall not be the subject of any stop order that is in effect or pending Proceedings seeking a stop order.
  - (e) Nasdaq Listing. The Parent Stock Issuance shall have been approved for listing on Nasdaq, subject to official notice of issuance.
- Section 9.2 Conditions to Obligations of Parent and Merger Sub Inc. The obligations of Parent and Merger Sub Inc. to consummate the Merger are subject to the satisfaction, at or prior to the Closing, of the following conditions (which may be waived, in whole or in part, to the extent permitted by Law, by Parent):
- (a) Representations and Warranties. (i) The representations and warranties of the Company contained in Section 4.1 (Corporate Existence and Power), Section 4.2 (Corporate Authorization), Section 4.22 (Opinion of Financial Advisor) and Section 4.23 (Antitakeover Statutes) of the Agreement shall, if qualified by materiality or "Company Material Adverse Effect," be true and correct in all respects, or, if not so qualified by materiality or "Company Material Adverse Effect," be true and correct in all material respects as of Closing (other than any

such representations and warranties that by their terms address matters only at and as of another specified time, which shall be true and correct in all material respects only at and as of such time), (ii) the representations and warranties of the Company contained in Section 4.5(a) (Capitalization), clauses (iii) and (iv), in the first sentence of Section 4.5(b) (Capitalization) and Section 4.5(e) (Capitalization) of the Agreement shall be true and correct in all respects, other than any de minimis inaccuracies, in each case, at and as of Closing (other than any such representations and warranties that by their terms address matters only at and as of another specified time, which shall be true and correct in all respects, other than any de minimis inaccuracies, only at and as of such time), and (iii) all other representations and warranties of the Company contained in the Agreement shall be true and correct in all respects (disregarding all materiality and "Company Material Adverse Effect" qualifiers contained therein, other than as such qualifiers are used in Section 4.10 (Absence of Certain Changes) of the Agreement), in each case at and as of the Closing (other than any such representations and warranties that by their terms address matters only as of another specified time, which shall be true and correct in all respects (disregarding all materiality and "Company Material Adverse Effect" qualifiers contained therein, other than as such qualifiers are used in Section 4.10 (Absence of Certain Changes) of the Agreement) only at and as of such time), except, in the case of this clause (iii), where the failure of such representations and warranties to be so true and correct has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

- (b) <u>Performance of Obligations and Covenants</u>. The Company shall have performed in all material respects its covenants and obligations under the Agreement required to be performed by it at or prior to the Closing Date and any such failure to perform shall have been cured on or prior to the Closing Date.
- (c) No Company Material Adverse Effect. After the date of the Agreement, there shall not have occurred or arisen any Company Material Adverse Effect that is continuing.
- (d) Officer's Certificate. Parent shall have received a certificate signed by an executive officer of the Company certifying on behalf of the Company, and not in such officer's personal capacity, that the conditions set forth in clauses (a), (b) and (c) of this Section 9.2 shall have been satisfied.
- **Section 9.3** Conditions to Obligations of the Company. The obligations of the Company to consummate the Merger are subject to the satisfaction, at or prior to the Closing, of the following conditions (which may be waived, in whole or in part, to the extent permitted by Law, by the Company):
- (a) Representations and Warranties. (i) The representations and warranties of Parent and Merger Sub Inc. set forth in the first and second sentences of Section 5.1 (Corporate Existence and Power) and Section 5.2 (Corporate Authorization) of the Agreement shall, if qualified by materiality or "Parent Material Adverse Effect," be true and correct in all respects, or, if not so qualified by materiality or "Parent Material Adverse Effect," be true and correct in all material respects, in each case, at and as of the Closing as if made at and as of the Closing (other than any such representations and warranties that by their terms address matters only at and as of another specified time, which shall not be true and correct in all material respects only at and as of such time) and (ii) all other representations and warranties of Parent or Merger Sub Inc. contained in the Agreement shall be true and correct in all respects (disregarding all materiality and "Parent Material Adverse Effect" qualifiers contained therein, other than as such qualifiers are used in Section 5.9 (Absence of Certain Changes) of the Agreement), in each case, at and as of the Closing as if made at and as of the Closing (other than any such representations and warranties that by their terms address matters only as of another specified time, which shall be true and correct in all respects (disregarding all materiality and "Parent Material Adverse Effect" qualifiers contained therein, other than as such qualifiers are used in Section 5.9 (Absence of Certain Changes) of the Agreement) only at and as of such time), except, in the case of this clause (ii), where the failure of such representations and warranties to be so true and correct has not had and would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.

- (b) <u>Performance of Obligations and Covenants</u>. Each of Parent and Merger Sub Inc. shall have performed in all material respects its covenants and obligations under the Agreement required to be performed by it at or prior to the Closing Date and any such failure to perform shall have been cured on or prior to the Closing Date.
- (c) No Parent Material Adverse Effect. After the date of the Agreement, there shall not have occurred or arisen any Parent Material Adverse Effect that is continuing.
- (d) Officer's Certificate. The Company shall have received a certificate signed by an executive officer of Parent certifying on behalf of Parent and Merger Sub Inc., and not in such officer's personal capacity, that the conditions set forth in clauses (a), (b) and (c) of this Section 9.3 shall have been satisfied

Section 9.4 Frustration of Closing Conditions. None of Parent, Merger Sub Inc. or the Company may rely on the failure of any condition set forth in Section 9.1, Section 9.2 or Section 9.3, as the case may be, to be satisfied if such failure was caused by such Party's failure to use its reasonable best efforts to consummate the Merger and the other transactions contemplated hereby, or other breach of or noncompliance with this Agreement.

### ARTICLE X TERMINATION

Section 10.1 Termination. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time:

- (a) by mutual written consent of the Company and Parent;
- (b) by either the Company or Parent:
- (i) if the Closing shall not have occurred on or before 11:59 P.M. New York City Time on the nine-month anniversary of the date hereof (the "End Date"); provided, however, that if on the End Date the condition set forth in Section 9.1(c) shall not have been satisfied but all other conditions set forth in Section 9.1, Section 9.2 and Section 9.3 shall have been satisfied or waived in accordance with this Agreement, then the End Date shall be extended, if Parent or the Company notifies the other Party in writing on or prior to the End Date, to 11:59 P.M. New York City Time on the twelvemonth anniversary of the date hereof; provided, further, that the right to terminate this Agreement under this Section 10.1(b)(i) shall not be available to a Party if such Party's breach of any of its obligations under this Agreement shall have proximately caused the failure of the Closing to occur on or before the End Date:
- (ii) if there shall have been issued an Order by a Governmental Authority of competent jurisdiction having jurisdiction over a material portion of the business of the Company, Parent or Merger Sub Inc. (or the effect of which would have a material effect on any of the Company, Parent or Merger Sub Inc.) permanently prohibiting the Merger or the Parent Stock Issuance and such Order shall have become final and non-appealable; provided however, that the right to terminate this Agreement under this Section 10.1(b)(ii) shall not be available to a Party if such Party's breach of its obligations under this Agreement shall have proximately caused such Order to have been issued;
  - (iii) if the Company Stockholders' Meeting shall have concluded and the Company Stockholder Approval shall not have been obtained; or
- (c) by Parent:
  - (i) a Company Adverse Recommendation Change shall have occurred; or
- (ii) if the Company shall have breached or failed to perform any of its (A) representations or warranties or (B) covenants or agreements set forth in this Agreement, which breach or failure to perform (i) would

give rise to the failure of a condition set forth in Section 9.2(a) or Section 9.2(b) and is incapable of being cured by the Company by the End Date, or, if capable of being cured by the End Date, shall not have been cured by the End Date (following receipt from Parent of written notice of such breach or failure to perform); provided, that if such breach or failure to perform is capable of being cured by the Company by the End Date and the Company ceases using reasonable best efforts to cure such breach or failure to perform following written notice from Parent, Parent shall have the right to terminate this Agreement pursuant to this Section 10.1(c)(ii); provided further, that Parent shall not have the right to terminate this Agreement pursuant to this Section 10.1(c)(ii) if Parent or Merger Sub Inc. is then in breach of any of its representations, warranties, covenants or agreements such that the Company has the right to terminate this Agreement pursuant to Section 10.1(d)(i).

### (d) by the Company:

- (i) if Parent or Merger Sub Inc. shall have breached or failed to perform any of its (A) representations or warranties or (B) covenants or agreements set forth in this Agreement, which breach or failure to perform (i) would give rise to the failure of a condition set forth in Section 9.3(a) or Section 9.3(b) and (ii) is incapable of being cured by Parent or Merger Sub Inc. by the End Date, or, if capable of being cured by the End Date, shall not have been cured by the End Date (following receipt from the Company of written notice of such breach or failure to perform); provided, that if such breach or failure to perform is capable of being cured by Parent or Merger Sub Inc. by the End Date and Parent or Merger Sub Inc. cease using reasonable best efforts to cure such breach or failure to perform following written notice from the Company, the Company shall have the right to terminate this Agreement pursuant to this Section 10.1(d)(i); provided further, that the Company shall not have the right to terminate this Agreement pursuant to this Section 10.1(d)(i) if the Company is then in breach of any of its representations, warranties, covenants or agreements such that Parent has the right to terminate this Agreement pursuant to Section 10.1(c)(ii); or
- (ii) if (A) the Company Board or Company Special Committee authorizes the Company to enter into an agreement with respect to a Superior Proposal that did not result from a breach of Section 6.4 in accordance with the terms of Section 6.4, (B) substantially concurrent with the termination of this Agreement, the Company enters into an agreement providing for a Superior Proposal that did not result from a breach of Section 6.4 and (C) prior to or concurrently with the termination of this Agreement, the Company pays to Parent (or, in the event Parent has not provided wire instructions for such a payment to Company, to an account established by Company for the sole benefit of Parent) in immediately available funds any fees required to be paid pursuant to Section 10.3.

Section 10.2 Effect of Termination. In the event of the termination of this Agreement by either Parent or the Company as provided in Section 10.1, written notice thereof shall forthwith be given by the terminating Party to the other Party specifying the provision hereof pursuant to which such termination is made. In the event of the termination of this Agreement in compliance with Section 10.1, this Agreement shall be terminated and this Agreement shall forthwith become void and have no effect, without any liability or obligation on the part of any Party (or any stockholder, director, officer, employee, agent, consultant or representative of such Party), other than the Confidentiality Agreement, this Section 10.2, Section 10.3, and Article XI, which provisions shall survive such termination and remain in full force and effect; provided, however, that nothing in this Section 10.2 shall relieve any Party from liability for any Fraud or Willful Breach of this Agreement, in which case the aggrieved Party shall be entitled to all rights and remedies available at law or in equity. No termination of this Agreement shall affect the rights and remedies of the Parties contained in the Confidentiality Agreement.

### Section 10.3 Termination Fees; Expense Reimbursement.

(a) In the event that this Agreement is terminated by (i) Parent pursuant to Section 10.1(c)(i) (Company Adverse Recommendation Change), (ii) the Company pursuant to Section 10.1(d)(ii) (Superior Proposal) or (iii) Parent or the Company pursuant to Section 10.1(b)(iii) (No Company Stockholder Approval) and at the time of such termination pursuant to Section 10.1(b)(iii) (No Company Stockholder Approval), Parent had the right to

terminate this Agreement pursuant to Section 10.1(c)(i) (Company Adverse Recommendation Change), then, in each case, the Company shall pay to Parent a fee in the amount of \$8,000,000 (the "Company Termination Fee") at or prior to the termination of this Agreement in the case of a termination described in clause (ii) or as promptly as practicable (and, in any event, within two (2) Business Days following such termination) in the case of a termination described in clause (i) or (iii).

- (b) In the event that (i) this Agreement is terminated by the Company or Parent pursuant to Section 10.1(b)(i) (End Date) or Section 10.1(b)(iii) (No Company Stockholder Approval), or in the event that this Agreement is terminated by Parent pursuant to Section 10.1(c)(ii) (Company Breach), and (ii) at any time after the date of this Agreement (A) prior to such termination pursuant to Section 10.1(c)(ii) (Company Breach) or Section 10.1(b)(i) (End Date), a bona fide Company Acquisition Proposal shall have been publicly announced or publicly made known and not withdrawn or (B) prior to the Company Stockholders' Meeting (in the case of a termination pursuant to Section 10.1(b)(iii) (No Company Stockholder Approval)) a Company Acquisition Proposal shall have been publicly announced or publicly made known and not withdrawn prior to the Company Stockholders' Meeting, and (ii) within nine (9) months after such termination, (x) the Company shall have entered into an agreement with respect to any Company Acquisition Proposal or (y) any Company Acquisition Proposal shall have been consummated (in each case of clause (x) and (y), whether or not such Company Acquisition Proposal is the same as the original Company Acquisition Proposal publicly made known or publicly announced), then, in any such event, the Company shall pay to Parent the Company Termination Fee immediately prior to or concurrently with the occurrence of either of the events described in the foregoing clauses (x) or (y); provided, however, that for purposes of the definition of "Company Acquisition Proposal" in this Section 10.3(b), references to "20%" shall be replaced by "50%".
- (c) In the event that this Agreement is terminated by either the Company or Parent pursuant to Section 10.1(b)(iii) (No Company Stockholder Approval), then the Company shall pay to Parent the Expenses. Any Expenses due under this Section 10.3(c) shall be paid no later than three (3) Business Days after receipt by the Company of documentation supporting such Expenses and shall be credited against the Company Termination Fee (if any when paid). "Expenses" means the reasonable and documented out-of-pocket fees and expenses incurred or paid by or on behalf of Parent and its Affiliates in connection with the Merger or the other transactions contemplated by this Agreement, or related to the authorization, preparation, negotiation, execution and performance of this Agreement, in each case including all reasonable and documented fees and expenses of law firms, commercial banks, investment banking firms, accountants, experts and consultants to Parent and its Affiliates; provided that the aggregate amount of Expenses reimbursable shall not exceed \$3.000.000.
- (d) The Parties acknowledge and agree that the agreements contained in this Section 10.3 are an integral part of the transactions contemplated by this Agreement and that, without these agreements, the Parties would not enter into this Agreement. Accordingly, if the Company fails promptly to pay any amount due pursuant to this Section 10.3, and, in order to obtain such payment, Parent commences a suit, then thenon-prevailing party in such suit shall pay the prevailing party's costs and expenses (including reasonable attorneys' fees and expenses) in connection with such suit, together with, in the case Parent is the prevailing party, interest on the amount due pursuant to this Section 10.3 from the date such payment was required to be made until the date of payment at the prime lending rate as published in The Wall Street Journal in effect on the date such payment was required to be made. All payments under this Section 10.3 shall be made by wire transfer of immediately available funds to an account designated in writing by Parent or the Company, as applicable. In no event shall a Company Termination Fee be payable more than once.
- (e) Each Party agrees that notwithstanding anything in this Agreement to the contrary (other than with respect to claims for, or arising out of or in connection with Fraud or Willful Breach or the equitable remedies set forth in Section 11.2 and Section 11.4) (A) the payment of the Company Termination Fee shall be the sole and exclusive monetary remedy of Parent, its Subsidiaries, stockholders, Affiliates, officers, directors, employees and Representatives against the Company or any of its Representatives, Affiliates, officers, directors or employees for, and (B) in no event will Parent or any other such person seek to recover any other money damages or seek

any other remedy (including any remedy for specific performance) based on a claim in law or equity with respect to, (1) any loss suffered, directly or indirectly, as a result of the failure of the Merger to be consummated, (2) the termination of this Agreement, (3) any liabilities or obligations arising under this Agreement or (4) any claims or actions arising out of or relating to any breach, termination or failure of or under this Agreement and (C) none of the Company or any Affiliates or Representatives of the Company shall have any further liability or obligation to another Party relating to or arising out of this Agreement.

### ARTICLE XI MISCELLANEOUS

Section 11.1 No Survival of Representations and Warranties. None of the representations and warranties in this Agreement or in any schedule, certificate, instrument or other document delivered pursuant to this Agreement shall survive the Effective Time. This Section 11.1 shall not limit Section 10.2, Section 10.3 or any covenant or agreement of the Parties which by its terms contemplates performance after the Effective Time.

Section 11.2 Amendment and Modification. Subject to applicable Law, this Agreement may be amended, modified or supplemented in any and all respects by written agreement of the Parties at any time prior to or after the adoption and approval of this Agreement by the Company's stockholders, but, after any such adoption and approval, no amendment shall be made which by Law would require the further approval of the Company's stockholders without first obtaining such further approval; provided, however, that any amendment, modification or supplement to Section 3.1(a), (b), or (c), Section 6.2, Section 9.1(a) or this proviso shall also require the consent of the Investor Persons. A termination of this Agreement pursuant to Section 10.1 or an amendment or waiver of this Agreement pursuant to Section 11.2 or Section 11.3 shall, in order to be effective, require, in the case of Parent, Merger Sub Inc. and the Company, action by their respective board of directors or sole member, as applicable, and the express written approval of such termination, amendment or waiver, in the case of Parent and the Company, by the Parent Special Committee and the Company Special Committee, as applicable.

Section 11.3 Extension; Waiver. At any time prior to the Effective Time, Parent or Merger Sub Inc., on the one hand, or the Company, on the other hand, may (a) extend the time for the performance of any of the obligations or other acts of the other Parties, (b) waive any inaccuracies in the representations and warranties contained in this Agreement or in any document delivered pursuant to this Agreement of the other Parties or (c) waive compliance by the other Parties with any of the agreements or conditions contained in this Agreement. Except as required by applicable Law, no waiver of this Agreement shall require the approval of the stockholders of either Parent or the Company. Any agreement on the part of a Party to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such Party. The failure of any Party to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of such rights, nor shall any single or partial exercise by any Party of any of its rights under this Agreement preclude any other or further exercise of such rights or any other rights under this Agreement. The Parties acknowledge and agree that Parent shall act on behalf of Merger Sub Inc. and the Company may rely on any notice given by Parent on behalf of Merger Sub Inc. with respect to the matters set forth in this Section 11.3. Notwithstanding anything to the contrary herein, the condition set forth in Section 9.1(a) shall not be waived without the approval of the Investor Persons.

Section 11.4 Expenses. Except as otherwise provided herein, all costs and expenses incurred in connection with this Agreement shall be paid by the Party incurring such cost or expense. Notwithstanding the foregoing, Parent shall pay any and all fees and expenses, other than the Company's attorneys' fees, incurred in connection with the filing by the Parties of the premerger notification and report forms relating to the Merger under the HSR Act and the filing of any notice of other document under any applicable foreign Competition Law.

Section 11.5 <u>Disclosure Letter References</u>. Each Disclosure Letter shall, for all purposes in this Agreement, be arranged in numbered and lettered parts and subparts corresponding to the numbered and lettered

sections and subsections contained in this Agreement. Each item disclosed in a Party's Disclosure Letter shall constitute an exception to, or as applicable, disclosure for the purposes of, the representations and warranties (or covenants, as applicable) to which it makes express reference and shall also be deemed to be disclosed or set forth for the purposes of every other part in such Party's Disclosure Letter relating to such Party's representations and warranties (or covenants, as applicable) set forth in this Agreement to the extent a cross-reference within such Disclosure Letter is expressly made to such other part in such Disclosure Letter, as well as to the extent that the relevance of such item as an exception to, or as applicable, disclosure for purposes of, such other section of this Agreement is reasonably apparent from the face of such disclosure.

Section 11.6 Notices. All notices, requests, consents, claims, demands, waivers, and other communications hereunder shall be in writing and shall be deemed to have been given upon the earlier of actual receipt or (a) when delivered by hand or by facsimile (providing proof of delivery); (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested); or (c) on the date sent by email if sent during normal business hours of the recipient, and on the next Business Day if sent after normal business hours of the recipient. Such communications must be sent to the respective Parties at the following addresses (or to such other Persons or at such other address for a Party as shall be specified in a written notice given in accordance with this Section 11.6):

if to Parent or Merger Sub Inc., to:

ProFrac Holding Corp. 333 Shops Boulevard, Suite 301 Willow Park, TX 76087 Attention: Robert Willette Email: robert.willette@profrac.com

and

c/o Parent Special Committee 333 Shops Boulevard, Suite 301 Willow Park, TX 76087 Attention: Terry Glebocki Email: robert.willette@profrac.com

with a copy (which shall not constitute notice) to:

Brown Rudnick LLP 1 Financial Center Boston, MA 02111 Attention: Sam Williams

Facsimile:

Email: SWilliams@brownrudnick.com

and

Kirkland & Ellis LLP 609 Main Street Houston, TX 77002

Attention: Douglas E. Bacon, P.C., Debbie P. Yee, P.C., Emily

Lichtenheld

Facsimile: (713) 836-3601

Email: doug.bacon@kirkland.com, debbie.yee@kirkland.com,

emily.lichtenheld@kirkland.com

if to the Company, to:

U.S. Well Services, Inc. 1360 Post Oak Blvd., Suite 1800 Houston, TX 77056 Attention: Kyle O'Neill, Chief Executive Officer Email: koneill@uswellservices.com

And

David Treadwell 391 Gull Lake Island Richland, Michigan 49083 United States of America Email: david.treadwell@epcorp.com

with a copy (which shall not constitute notice) to:

Porter Hedges LLP 1000 Main Street, 36th Floor, Houston, Texas 77389 Attention: Corey C. Brown, Adam K. Nalley Facsimile: (713) 226-6244

Email: cbrown@porterhedges.com, analley@porterhedges.com

and

Paul Hastings LLP 101 California Street, 48th Floor San Francisco, CA 94111 Attention: Steve Camahort Facsimile: (415) 856-7361

Email: stevecamahort@paulhastings.com

if to the Investor Person, to:

Crestview Advisors, L.L.C. 590 Madison Avenue, 42nd Floor New York, New York 10022 Attention: Adam J. Klein, Ross A. Oliver

Email: aklein@crestview.com; roliver@crestview.com

with a copy (which shall not constitute notice) to:

Vinson & Elkins L.L.P. 845 Texas Avenue, Suite 4700 Houston, TX 77002

Attention: Stephen M. Gill; Crosby Scofield Email: sgill@velaw.com; cscofield@velaw.com

Section 11.7 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, it being understood that each Party need not sign the same counterpart. This Agreement shall become effective when each Party hereto shall have received a counterpart hereof signed by all of the other Parties. Signatures delivered electronically or by facsimile shall be deemed to be original signatures.

Section 11.8 Entire Agreement; No Third Party Beneficiaries This Agreement (including the Exhibits hereto and the documents and the instruments referred to herein), the Confidentiality Agreement and any

agreements entered into contemporaneously herewith (a) constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, between Parent and the Company and among the Parties with respect to the subject matter hereof and thereof (provided, that any provisions of the Confidentiality Agreement conflicting with this Agreement shall be superseded by this Agreement) and (b) are not intended to confer any rights, benefits, remedies, obligations or liabilities upon any Person other than the Parties and their respective successors and permitted assigns, except (i) for the rights of the holders of Company Stock following the Effective Time to receive the Merger Consideration in accordance with Article III, and (ii) as provided in Section 6.2(c), Section 7.4 and in Section 10.3(e) (in each case which will be to the benefit of the parties referenced in such sections); provided, that notwithstanding the foregoing clause (b), following the Effective Time, the provisions of Section 7.4 shall be enforceable by each Company Indemnified Party hereunder and his or her heirs and his or her representatives.

Section 11.9 Severability. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void, unenforceable or against its regulatory policy, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated, so long as the economic and legal substance of the transactions contemplated hereby, taken as a whole, are not affected in a manner materially adverse to any Party hereto. Upon such a determination, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

Section 11.10 Assignment. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the Parties in whole or in part (whether by operation of Law or otherwise) without the prior written consent of the other Parties (which shall include, in all cases, the prior written consent of the Company Special Committee and the Parent Special Committee), and any such assignment without such consent shall be null and void; provided, that Parent may designate, prior to the Effective Time, by written notice to the Company, another wholly owned direct or indirect Subsidiary of Parent with respect to which the representations and warranties of such Subsidiary must be true and correct in all respects hereunder to be a party to the Merger in lieu of Merger Sub Inc., in which event all references herein to Merger Sub Inc. shall be deemed references to such other Subsidiary (except with respect to representations and warranties made herein with respect to Merger Sub Inc. as of the date of this Agreement) and all representations and warranties made herein with respect to Merger Sub Inc. as of the date of this Agreement shall also be made with respect to such other Subsidiary as of the date of such designation; provided, that such assignment shall not relieve Parent of its obligations hereunder or otherwise enlarge, alter or change any obligation of any other party hereto or due to Parent or such other Subsidiary; and provided further, that such assignment does not adversely affect the Merger or the timing thereof in any way. This Agreement shall be binding upon, inure to the benefit of and be enforceable by the Parties and their respective successors and permitted assigns.

Section 11.11 Governing Law. This Agreement, and all claims or causes of action (whether in contract or tort) that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement shall be governed and construed in accordance with the Laws of the State of Delaware without giving effect to the principles of conflicts of law thereof or of any other jurisdiction.

### Section 11.12 Enforcement; Exclusive Jurisdiction.

(a) The rights and remedies of the Parties shall be cumulative (and not alternative). The Parties agree that irreparable damage would occur and that the Parties would not have any adequate remedy at law in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that, subject to the limitations in Section 10.3(d) the Parties shall be entitled to an injunction or injunctions to prevent breaches or threatened breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, including the obligations to consummate the

Merger, in the Court of Chancery of the State of Delaware or, if under applicable Law exclusive jurisdiction over such matter is vested in the federal courts, any federal court located in the State of Delaware without proof of actual damages or otherwise (and each Party hereby waives any requirement for the securing or posting of any bond in connection with such remedy), this being in addition to any other remedy to which they are entitled at law or in equity.

(b) In addition, each of the Parties irrevocably (i) consents to submit itself, and hereby submits itself, to the personal jurisdiction of the Court of Chancery of the State of Delaware and any federal court located in the State of Delaware, or, if neither of such courts has subject matter jurisdiction, any state court of the State of Delaware having subject matter jurisdiction, in the event any dispute arises out of or relates to this Agreement or any of the transactions contemplated by this Agreement, (ii) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court, and agrees not to plead or claim any objection to the laying of venue in any such court or that any judicial proceeding in any such court has been brought in an inconvenient forum, (iii) agrees that it will not bring any action, suit or proceeding arising out of or relating to this Agreement or any of the transactions contemplated by this Agreement in any court other than the Court of Chancery of the State of Delaware and any federal court located in the State of Delaware, or, if neither of such courts has subject matter jurisdiction, any state court of the State of Delaware having subject matter jurisdiction, and (iv) consents to service of process being made through the notice procedures set forth in Section 11.6.

Section 11.13 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY THAT MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY LAW ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (I) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (II) SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THE FOREGOING WAIVER, (III) SUCH PARTY MAKES THE FOREGOING WAIVER VOLUNTARILY, (IV) AND SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVER AND CERTIFICATIONS IN THIS SECTION 11.13.

Section 11.14 Specific Performance. The Parties hereto agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with the terms hereof for which money damages, even if available, would not be an adequate remedy, and that the Parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement or to enforce specifically the performance of the terms and provisions hereof in the courts referred to in Section 11.12, in addition to any other remedy to which such Party may be entitled under this Agreement. The Parties further agree to waive any requirement for the securing or posting of any bond in connection with such remedy, and that such remedy shall be in addition to any other remedy to which a Party is entitled at law or in equity. To the extent any party hereto brings a Claim to enforce specifically the performance of the terms and provisions of this Agreement (other than a Claim to enforce specifically any provision that expressly survives termination of this Agreement), the End Date shall automatically be extended to (i) the 20th Business Day following the resolution of such Claim (if the End Date would otherwise occur on or prior to such date) or (ii) such other time period established by the court presiding over such Claim.

[The remainder of this page has been intentionally left blank; the next page is the signature page.]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their respective authorized officers as of the date set forth on the cover page of this Agreement.

### U.S. WELL SERVICES, INC.

By: /s/Josh Shapiro

Name: Josh Shapiro

Title: Chief Financial Officer

### PROFRAC HOLDING CORP.

By: /s/ Matthew D. Wilks
Name: Matthew D. Wilks
Title: Executive Chairman

### THUNDERCLAP MERGER SUB I, INC.

By: /s/ Matthew D. Wilks

Name: Matthew D. Wilks
Title: Executive Chairman

[Signature Page to Agreement and Plan of Merger]

### FIRST AMENDMENT

TO

### CONVERTIBLE SENIOR SECURED (THIRD LIEN) PIK NOTE1

This FIRST AMENDMENT TO CONVERTIBLE SENIOR SECURED (THIRD LIEN) PIK NOTE dated as of June 21, 2022 (this "Amendment") is entered into by and between U.S. WELL SERVICES, INC., a Delaware corporation ("Maker"), and the undersigned payee (the "Payee").

WHEREAS, reference is made to those certain Convertible Senior Secured (Third Lien) PIK Note dated as of June 24, 2021, in each case by and between Maker and Payee (the "Note");

WHEREAS, on June 21, 2022, Maker, ProFrac Holding Corp., a Delaware corporation ('Parent'), and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of Parent ('Merger Sub Inc.'), have entered into an Agreement and Plan of Merger (as the same may be amended from time to time, the ''Merger Agreement'), providing for, among other things, the merger of Merger Sub Inc. with and into Maker, with Maker surviving the merger as the surviving corporation pursuant to the terms and conditions of the Merger Agreement; and

WHEREAS, Maker and Payee desire to amend certain provisions of the Note, as set forth in greater detail and subject to the terms and conditions outlined in this Amendment.

NOW, THEREFORE, in consideration of the foregoing and the agreements contained herein, the parties hereby agree as follows:

- 1. <u>Capitalized Terms</u>. Capitalized terms used in this Amendment without definition that are defined in the Note shall have the same meanings herein as therein.
  - 2. Interest. This note is hereby amended by amending and restating Section 2 as follows:

Interest shall begin to accrue on the unpaid principal balance of this Note, if any, commencing on June 24, 2021 and continuing through July 9, 2022 at the rate of sixteen percent (16%) per annum calculated on the basis of a 360 day year and actual days elapsed. Accrued and unpaid interest shall be calculated on this Note on the last day of each March, June, September and December, commencing September 30, 2021, and on July 9, 2022 and the Merger Agreement Termination Date (as defined below), and shall be added on each such date to the unpaid principal balance of this Note (rounded up to the nearest \$1.00) (the "PIK Interest"). PIK Interest, upon being added to the unpaid principal balance of this Note, shall no longer be deemed to be accrued and unpaid interest on the outstanding principal amount. References herein and in the Agreement to the "principal amount" of the Notes includes any increases in the principal amount of the outstanding Notes as a result of the PIK Interest. Accrued and unpaid interest on this Note shall also be due and payable on the Maturity Date under the terms set forth in Section 3. Notwithstanding anything to the contrary herein, in the event that the Merger Agreement is terminated (the date of such termination, the "Merger Agreement Termination Date"), interest shall be deemed to have accrued on the unpaid principal balance of this Note from July 9, 2022 through the Merger Agreement Termination Date and shall accrue from the Merger Agreement Termination Date until repayment of this Note in full, and in each such case otherwise in accordance with the terms set forth in the preceding sentences of this Section 2.

- 3. Conversion. The Note is hereby amended by adding a new Section 6(f) as follows:
- (f) Notwithstanding anything to the contrary herein, subject to Maker obtaining the Company Stockholder Approval (as defined in that certain Agreement and Plan of Merger, dated as of June 21, 2022, by and among ProFrac Holding Corp., a Delaware corporation, Maker and the other parties thereto (as the
- Note to Draft: To be updated for each note.

same may be amended from time to time, the 'Merger Agreement')) and in accordance with Section 3.1(b) of the Merger Agreement, immediately prior to the Effective Time (as defined in the Merger Agreement), Maker shall convert all of the outstanding principal and interest then owing under this Note into a number of shares of Class A Common Stock equal to the quotient obtained by dividing (A) the amount of such outstanding principal and interest owing through the date immediately prior to the date of conversion, by (B) the Merger Conversion Price. The "Merger Conversion Price" shall initially be \$1.22, which may be adjusted from time to time in the same manner as the Conversion Price as set forth in Section 6(d). The issuance of Class A Common Stock pursuant to this Section 6(f) shall not require Maker to comply with the NASDAQ listing requirements in Section 7(i) unless failure to do so would have an adverse effect on Payee.

4. Share Limitations. The Note is hereby amended by amending and restating the first sentence of Section 7 as follows:

Notwithstanding the provisions set forth in Section 3, Section 6 or anywhere else in this Note (but subject to the last sentence of Section 6(f)), (i) no shares of Class A Common Stock will be issued under this Note unless and until the Company shall have submitted a Listing of Additional Shares to the NASDAQ covering all the shares of Class A Common Stock issuable pursuant to this Note (the "Listing Application") and NASDAQ shall have completed its review of, and approved, such listing application, (ii) no shares of Class A Common Stock will be issued under this Note to the extent such issuance would constitute a "change of control" under the Nasdaq's listing rules (the "Change of Control Limitation") or would be in excess of the number of shares of Class A Common Stock authorized and available for issuance under the Company's charter (the "Charter Limitation"), and (iii) the total number of shares of Class A Common Stock that may be issued under the Equity Linked Notes at a price per share which is less than the Conversion Price, when combined with any other shares of Class A Common Stock which may be aggregated with such issuances under applicable NASDAQ rules for this purpose, will not exceed the number permitted under such applicable NASDAQ rules (the "Exchange Cap"), unless stockholder approval is obtained in order to comply with, satisfy or remove, as applicable, the Change of Control Limitation, the Charter Limitation, or the Exchange Cap, as applicable.

- 5. <u>Change of Control</u>. The Note is hereby amended to add the following proviso at the end of the definition of "Change of Control" in <u>Section 6(c)</u>: "; <u>provided, further</u>, that the transactions contemplated by the Merger Agreement shall be deemed to not be a Change of Control".
- 6. <u>Tax Matters</u>. Maker and Payee agree to reasonably consult with each other regarding the reporting on any U.S. federal (or applicable state or local) income tax return with respect to the Note, including the accrual of any original issue discount on the Note.
  - 7. Miscellaneous.
- (a) This Amendment and any covenants and agreements contained herein shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.
- (b) Except as otherwise expressly set forth herein, nothing herein shall be deemed to constitute an amendment, modification or waiver of any of the provisions of the Note which shall remain in full force and effect as of the date hereof.
- (c) This Amendment may be executed in any number of counterparts, each of which, when executed and delivered, shall be an original, but all counterparts shall together constitute one instrument. The words "execution," "signed," "signature," and words of similar import in this Amendment shall be deemed to include electronic or digital signatures or electronic records, each of which shall be of the same effect, validity, and enforceability as manually executed signatures or a paper-based record-keeping system, as the case may be, to the extent and as provided for under applicable law, including the Electronic Signatures in Global and National Commerce Act of 2000 (15 U.S.C. §§ 7001 to 7031), the Uniform Electronic Transactions Act (UETA), or any state law based on the UETA.

(d) This Amendment shall be governed by the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or of any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware, and shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns in accordance with the Note

[Remainder of page intentionally left blank]

A-A-3

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed by their respective authorized officers as of the day and year first above written.

	MAKER:
	U.S. WELL SERVICES, INC.
	By
	Name:
	Title:
Sionature Page to Note Ameno	lment

A-A-4

PAYEE:		
Ву:		
Name:		
Title:		

Signature Page to Note Amendment

A-A-5

Exhibit B

### U.S. WELL SERVICES, INC.

# FIRST AMENDMENT TO THE CERTIFICATE OF DESIGNATIONS

Pursuant to Section 151 of the General Corporation Law of the State of Delaware

### SERIES A REDEEMABLE CONVERTIBLE PREFERRED STOCK

(Par Value \$0.0001 Per Share)

U.S. Well Services, Inc. (the "Corporation"), a corporation organized and existing under and by virtue of the provisions of the General Corporation Law of the State of Delaware (the "General Corporation Law"), hereby certifies that, pursuant to the authority expressly granted to and vested in the Board of Directors of the Corporation (the "Board of Directors") by the Second Amended and Restated Certificate of Incorporation of the Corporation (as amended from time to time in accordance with its terms and the General Corporation Law, the "Certificate of Incorporation"), which authorizes the Board of Directors, by resolution, to provide out of the unissued shares of the preferred stock (the "Preferred Stock") for one or more series of Preferred Stock and to establish from time to time the number of shares to be included in each such series and to fix the voting rights (if any), designations, powers, preferences and relative, participating, optional, special and other rights (if any) of each such series and any qualifications, limitations and restrictions thereof, and in accordance with the provisions of Section 151 of the General Corporation Law, the Board of Directors duly adopted on June [\*], 2022 the following recitals and resolutions:

WHEREAS, the Board of Directors previously fixed the rights, preferences, restrictions and other matters relating to a Series A Redeemable Convertible Preferred Stock (hereinafter referred to as the "Series A Preferred Stock"), consisting of up to 55,000 shares of the Series A Preferred Stock which the Corporation has the authority to issue, as set forth in that certain Certificate of Designations of the Series A Preferred Stock dated May 24, 2019 (the "Certificate of Designations");

WHEREAS, pursuant to Section 7.14(b) of the Purchase Agreement among the Corporation and the purchasers party thereto dated May 23, 2019, such parties agreed not to treat the Series A Preferred Stock as "preferred stock" within the meaning of Section 305 of the Internal Revenue Code of 1986, as amended, and U.S. Treasury Regulation § 1.305-5 for U.S. federal income tax and withholding tax purposes;

WHEREAS, on June [•], 2022, the Corporation, ProFrac Holding Corp., a Delaware corporation ('Parent'), and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of Parent ("Merger Sub Inc"), have entered into an Agreement and Plan of Merger (as the same may be amended from time to time, the "Merger Agreement"), providing for, among other things, the merger of Merger Sub Inc. with and into the Corporation, with the Corporation surviving the merger as the surviving corporation pursuant to the terms and conditions of the Merger Agreement.

WHEREAS, the Board of Directors wishes to amend the Certificate of Designations in accordance with the General Corporation Law; and

WHEREAS, the Board of Directors has determined that the amendments to the terms of the Series A Preferred Stock set forth in this First Amendment to the Certificate of Designations do not adversely affect the rights, preferences, privileges or powers of the Series A Preferred Stock.

NOW, THEREFORE, BE IT RESOLVED, that the Board of Directors does hereby amend the Certificate of Designations and the rights, preferences, restrictions and other matters relating to the Series A Preferred Stock as follows:

- 1. Conversion. The Certificate of Designations is hereby amended by adding a new Section 7(q) as follows:
- (q) Each holder of Series A Preferred Stock shall have the option, exercisable by delivery of a Conversion Notice to the Corporation, to convert all of such holder's shares of Series A Preferred Stock into Class A Common Stock at the Merger Conversion Ratio (such conversion, a "Merger Conversion"). The Corporation shall cause the effective date for the delivery of shares of Class A Common Stock issued in a Merger Conversion to be the Business Day on which the Conversion Notice is received. The "Merger Conversion Ratio" means, for each share of Series A Preferred Stock, the quotient of (i) its Liquidation Preference as of the date of the conversion and (ii) Merger Conversion Price. The "Merger Conversion Price" shall initially be \$1.22, which may be adjusted from time to time in the same manner as the Conversion Price as set forth in Section 7(g).
- 2. CoC Forced Conversion. The following shall be added at the beginning of Section 7(d) of the Certificate of Designations: "Subject to a holder's right to elect a Merger Conversion,".
- 3. Effective Date.

This First Amendment to the Certificate of Designations shall become effective on [•], 2022.

[The Remainder of this Page Intentionally Left Blank]

A-B-2

IN WITNESS WHEREOF, U.S. Well Services, Inc. has caused this First Amendment to the Certificate of Designations to be duly executed this  $[\bullet]$  day of, 2022.

	U.S. WELL SERVICES, INC.	
	By:	
	Name: Title:	
[Signature Page to Certificate of Designation of Incident to the content of the c	enations1	
Total and the continue of Best	5.ranonoj	
A-B-3		

### AMENDMENT

TO

#### PERFORMANCE AWARDS

This AMENDMENT TO PERFORMANCE AWARDS dated as of June 21, 2022 (this "Amendment") is entered into by and between U.S. WELL SERVICES, INC., a Delaware corporation ("Company"), and the undersigned recipient (the "Participant") of one or more Performance Awards (Pool A) (each, a "Pool A Performance Award") and/or one or more Performance Awards (Pool B) (each, a "Pool B Performance Award"), each granted under the U.S. Well Services, Inc. Amended and Restated 2018 Stock Incentive Plan (as amended, the "Plan").

WHEREAS, on June 21, 2022, the Company, ProFrac Holding Corp., a Delaware corporation ("*Parent*"), and Thunderclap Merger Sub I, Inc., a Delaware corporation ("*Merger Sub*"), have entered into an Agreement and Plan of Merger (as the same may be amended from time to time, the *Merger Agreement*"), providing for, among other things, the merger of Merger Sub Inc. with and into the Company, with the Company surviving the merger as the surviving corporation pursuant to the terms and conditions of the Merger Agreement; and

WHEREAS, the Company and Participant desire to amend each Pool A Performance Award and/or Pool B Performance Award granted to and held by Participant as set forth in greater detail and subject to the terms and conditions outlined in this Amendment.

NOW, THEREFORE, in consideration of the foregoing and the agreements contained herein, the parties hereby agree as follows:

- 1. <u>Capitalized Terms</u>. Capitalized terms used in this Amendment without definition that are defined in the Merger Agreement shall have the same meanings herein as therein.
- 2. <u>Payment Upon Merger</u>. Each Pool A Performance Award and Pool B Performance Award (collectively, the "Awards") granted to and held by Participant is hereby amended by adding the following:

Notwithstanding anything to the contrary herein, conditioned upon the consummation of the Merger, immediately prior to the Effective Time, by virtue of the Merger and without any action on the part of Participant, (i) each then-outstanding Pool A Award granted to and held by Participant shall be cancelled and converted into the right to receive the Merger Consideration payable pursuant to Section 3.2 of the Merger Agreement with respect to the number of Pool A Amended Payout Shares subject to such Pool A Performance Award, and (ii) each then-outstanding Pool B Award granted to and held by Participant shall be cancelled and converted into the right to receive the Merger Consideration payable pursuant to Section 3.2 of the Merger Agreement with respect to the number of Pool B Amended Payout Shares subject to such Pool B Performance Award, in each of clauses (i) and (ii), payable at the time specified therein without any interest thereon and less applicable Tax withholding.

- 3. <u>Termination</u>. Notwithstanding anything to the contrary herein, in the event the Merger Agreement is terminated prior to the consummation of the Merger, this Amendment shall terminate.
  - 4. Miscellaneous.
- (a) This Amendment and any covenants and agreements contained herein shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.
- (b) Except as otherwise expressly set forth herein, nothing herein shall be deemed to constitute an amendment, modification or waiver of any of the provisions of the Awards which shall remain in full force and effect as of the date hereof. Without limiting the generality of the foregoing, any vesting or forfeiture conditions applicable to such Awards shall continue in accordance with their terms from and after the date of this Agreement.

- (c) This Amendment may be executed in any number of counterparts, each of which, when executed and delivered, shall be an original, but all counterparts shall together constitute one instrument. The words "execution," "signed," "signature," and words of similar import in this Amendment shall be deemed to include electronic or digital signatures or electronic records, each of which shall be of the same effect, validity, and enforceability as manually executed signatures or a paper-based record-keeping system, as the case may be, to the extent and as provided for under applicable law, including the Electronic Signatures in Global and National Commerce Act of 2000 (15 U.S.C. §§ 7001 to 7031), the Uniform Electronic Transactions Act (UETA), or any state law based on the UETA.
- (d) This Amendment shall be governed by the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or of any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware, and shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns in accordance with the Awards.
- (e) This Amendment is intended to comply with Section 409A of the Code, including compliance with the payment times as provided in Awards prior to this Amendment to the extent required by Section 409A of the Code. Nothing in this Amendment shall be construed to change the time or form of payment for purposes of Section 409A of the Code.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed by their respective authorized officers as of the day and year first above written.

<b>COMPANY</b> :
U.S. WELL SERVICES, INC.
ByName: Title:
PARTICIPANT:
By Name:

 $Signature\ Page\ to\ Performance\ Award\ Amendment$ 

Annex B

### PROFRAC HOLDING CORP. UNANIMOUS WRITTEN CONSENT OF THE STOCKHOLDERS

The undersigned stockholders (the "<u>Stockholders</u>") of ProFrac Holding Corp., a Delaware corporation (the "<u>Company</u>"), holding shares of capital stock of the Company representing at least a majority of the Company's outstanding shares of Class A common stock, par value \$0.01 per share, and Class B common stock, par value \$0.01 per share, do hereby consent with respect to all shares held by them, to adopt the following resolutions by written consent in lieu of a meeting in accordance with the provisions of Section 228 of the General Corporation Law of the State of Delaware (the "<u>DGCL</u>") and Section 2.14 of the Amended and Restated Bylaws of the Company, which consent may be delivered in any number of counterparts (including by electronic signature or other electronic transmission):

WHEREAS: The Board of Directors of the Company (the "Board") has approved the Agreement and Plan of Merger by and among the

Company, Thunderclap Merger Sub I, Inc. ("Merger Sub Inc.") and U.S. Well Services, Inc. (the "Target"), in the form attached hereto as <a href="Exhibit A">Exhibit A</a> (the "Merger Agreement"), pursuant to which Merger Sub Inc. will merge with and into the

Target, with the Target surviving as a wholly owned indirect subsidiary of the Company (the "Merger").

WHEREAS: Pursuant to the terms of the Merger Agreement and in connection with the Merger, the Company will issue shares of

Class A common stock, par value \$0.01, to the holders of Class A common stock of the (the "Company Share Issuance").

WHEREAS: Pursuant to the listing rules of the Nasdaq Global Select Market, the Company Share Issuance requires the approval of a

majority of the Company's stockholders entitled to vote thereon.

WHEREAS: The Board has resolved that the Merger is fair, advisable and in the best interests of the Company and its stockholders and

has recommended that the stockholders vote in favor of the approval of the Company Share Issuance.

#### NOW THEREFORE, BE IT:

### **Approval of the Company Share Issuance**

**RESOLVED:** That the Company Share Issuance be, and hereby is, authorized and approved in all respects.

**RESOLVED:** That the officers of the Company (the "Authorized Officers") be, and each acting singly hereby is, authorized and directed

to do or cause to be done any and all such acts and things as they may deem necessary or desirable for the performance in full of all obligations of the Company under the Merger Agreement in connection with the Company Share Issuance (such determination to be exclusively evidenced by the taking of such action or the execution and delivery thereof by any such

Authorized Officer).

### **Omnibus Resolutions**

RESOLVED: That the Authorized Officers, be, and they hereby are authorized, in the name and on behalf of the Company, and with

the corporate seal as necessary or desirable, to

execute, make oath to, acknowledge, and deliver any and all instruments and documents and to perform all such other acts as may be necessary or appropriate for the purpose of carrying out all of the foregoing resolutions, the execution of any such instrument or document or the performance of any such other act to be conclusive evidence of the necessity or appropriateness thereof, and the Authorized Officers are hereby authorized and directed to employ such agents, and to pay such expenses as such Authorized Officer may deem necessary or appropriate in order to fully carry out the intent and accomplish the purposes of the foregoing resolutions.

### RESOLVED:

That all acts and deeds previously performed by any officer, employee of, or counsel for, the Company prior to the date of these resolutions that are within the authority conferred hereby are hereby ratified, approved and confirmed in all respects as the authorized acts and deeds of the Company.

IN WITNESS WHEREOF, the undersigned have executed this Written Consent by unanimous written consent of the Stockholders of ProFrac Holding Corp. this 21st day of June, 2022. This Written Consent may be executed in any number of counterparts, each of which shall be deemed an original and all of which together shall constitute one instrument.

### STOCKHOLDERS:

### THRC HOLDINGS, LP

By: THRC Management, LLC, its General Partner

By: /s/ Dan H. Wilks
Name: Dan H. Wilks
Title: Manager

### FARRIS WILKS

By: /s/ Farris Wilks

### Exhibit A

Merger Agreement

(reproduced in Annex A)

Annex C

## **Jefferies**

Jefferies LLC

520 Madison Avenue New York, NY 10022 tel 212.284.2300 Jefferies.com

June 21, 2022

Special Committee of the Audit Committee of the Board of Directors of ProFrac Holding Corp. 333 Shops Boulevard, Suite 301 Willow Park, Texas 76087

Members of the Special Committee:

We understand that U.S. Well Services, Inc. (the "Company"), ProFrac Holding Corp. ("Parent"), and Thunderclap Merger Sub I, Inc., an indirect subsidiary of Parent ("Merger Sub"), propose to enter into an Agreement and Plan of Merger (the "Merger Agreement"), pursuant to which, among other things, Merger Sub will merge with and into the Company (the "Merger"), with the Company as the surviving corporation, in a transaction in which each outstanding share of Class A common stock, par value \$0.0001 per share, of the Company (the "Company Common Stock"), other than shares of Company Common Stock held in the treasury of the Company or owned, directly or indirectly, by the Company, Parent or Merger Sub immediately prior to the time of the Merger, all of which shares will be cancelled, will be converted into the right to receive 0.0561 (the "Exchange Ratio") of a share of Class A common stock, par value \$0.01 per share of Parent (the "Parent Common Stock"). The terms and conditions of the Merger are more fully set forth in the Merger Agreement.

You, the Special Committee of the Audit Committee of the Board of Directors of ProFrac Holding Corp. (the "Special Committee"), have asked for our opinion as to whether the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement is fair, from a financial point of view, to Parent.

In arriving at our opinion, we have, among other things:

- (i) reviewed a draft, dated June 21, 2022, of the Merger Agreement;
- (ii) reviewed certain publicly available financial and other information about the Company and Parent;
- (iii) reviewed certain information furnished to us and approved for our use by Parent, including financial forecasts and analyses, relating to the business, operations and prospects of the Company prepared by management of the Company (the "Company Forecasts");
- (iv) reviewed certain information furnished to us and approved for our use by Parent, including financial forecasts and analyses, relating to the business, operations and prospects of Parent prepared by management of Parent (the "Parent Forecasts");
- (v) reviewed certain estimates furnished to us and approved for our use by Parent as to potential cost savings and revenue synergies (including the amount and timing thereof) expected by the management of Parent to result from the Merger (the "Synergies");
- (vi) held discussions with members of senior management of Parent concerning the matters described in clauses (ii) through (v) above;

- (vii) reviewed the share trading price history and valuation multiples for the Company Common Stock and Parent Common Stock and compared them with those of certain publicly traded companies that we deemed relevant;
- (viii) considered certain potential pro forma financial effects of the Merger on Parent utilizing the financial forecasts and estimates relating to Parent and the Company and the potential Synergies referred to above; and
- (ix) conducted such other financial studies, analyses and investigations as we deemed appropriate.

In our review and analysis and in rendering this opinion, we have assumed and relied upon, but have not assumed any responsibility to independently investigate or verify, the accuracy and completeness of all financial and other information that was supplied or otherwise made available to us by Parent or that was publicly available to us (including, without limitation, the information described above), or that was otherwise reviewed by us. We have relied on assurances of the management of Parent that it is not aware of any facts or circumstances that would make any of the foregoing information incomplete, inaccurate or misleading. In our review, we did not obtain any independent evaluation or appraisal of any of the assets or liabilities (contingent, accrued, derivative, off-balance sheet or otherwise), nor did we conduct a physical inspection of any of the properties or facilities of, the Company or Parent, and we have not been furnished with and assume no responsibility to obtain, any such evaluations, appraisals or physical inspections. We have not evaluated the solvency or fair value of the Company, Parent or any other entity under any laws relating to bankruptcy, insolvency or similar matters.

With respect to the financial forecasts provided to and reviewed by us, we note that projecting future results of any company is inherently subject to uncertainty. You have informed us, however, and we have assumed, at your direction, that the Company Forecasts were reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of the Company as to the future financial performance of the Company and the other matters covered thereby. We have also assumed, at your direction, that the Parent Forecasts were reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of Parent as to the future financial performance of Parent and the other matters covered thereby. With respect to the future financial performance of Parent giving pro forma effect to the Merger, at your direction and with your consent, we have assumed that the Synergies will be realized in the amounts and at the times projected. We express no opinion as to the Company Forecasts or the Parent Forecasts or the assumptions on which they are based, including with respect to the Synergies.

We have relied upon the assessments of the management of Parent as to, among other things, (i) the potential impact on the Company and Parent of market, competitive, seasonal, macroeconomic, geopolitical and other conditions, trends and developments in and prospects for, and governmental, regulatory and legislative matters relating to or affecting, the energy industry and the geographic regions and local communities in which the Company and Parent operate, and (ii) the ability of Parent to integrate the businesses of the Company and Parent and to realize the potential Synergies. We have assumed that there will not be any developments with respect to any such matters that would have an adverse effect on the Company, Parent or the Merger (including the contemplated benefits thereof) or that otherwise would be meaningful in any respect to our analyses or opinion.

Our opinion is based on economic, monetary, regulatory, market and other conditions existing and which can be evaluated as of the date hereof. We expressly disclaim any undertaking or obligation to advise any person of any change in any fact or matter affecting our opinion of which we become aware after the date hereof. We further note that the current volatility and disruption in the energy, credit and financial markets relating to, among other things, the COVID-19 pandemic and armed conflict in or relating to Ukraine, may or may not have an effect on the Company or Parent, and we are not expressing an opinion as to the effects of such volatility or such disruption on the Merger or any party to the Merger.

We have made no independent investigation of, and we express no view or opinion as to, any legal, regulatory, accounting or tax matters affecting or relating to the Company or Parent, and we have assumed the correctness in all respects material to our analyses and opinion of all legal, regulatory, accounting and tax advice given to the Company, Parent and the Special Committee, including, without limitation, with respect to changes in, or the impact of, accounting standards or tax and other laws, regulations and governmental and legislative policies affecting the Company or Parent or the Merger and legal, regulatory, accounting and tax consequences of the terms of, and transactions contemplated by, the Merger Agreement and related documents to the Company and Parent and their respective stockholders. In addition, in preparing this opinion, we have not taken into account any tax consequences of the transaction to any holder of the Company Common Stock or Parent Common Stock. We have assumed that the Merger will be consummated in accordance with the terms of the Merger Agreement without waiver, modification or amendment of any term, condition or agreement and in compliance with all applicable laws, documents and other requirements and that the final form of the Merger Agreement will be substantially similar to the last draft reviewed by us. We have also assumed that in the course of obtaining the necessary governmental, regulatory or third-party approvals, consents, waivers and releases for the Merger or otherwise, including with respect to any divestitures or other requirements, no delay, limitation, restriction or condition will be imposed or occur that would have an adverse effect on the Company, Parent or the contemplated benefits of the Merger or that otherwise would be material in any respect to our analyses or opinion.

In addition, we were not requested to and did not provide advice concerning the structure, the specific amount of the Exchange Ratio, or any other aspects of the Merger, or to provide services other than the delivery of this opinion. We were not authorized to and did not solicit any expressions of interest from any other parties with respect to the purchase or sale of all or any part of the Company or Parent or any other alternative transaction. We did not participate in negotiations with respect to the terms of the Merger and related transactions. Consequently, we have assumed that such terms are the most beneficial terms from Parent's perspective that could under the circumstances be negotiated among the parties to such transactions, and no opinion is expressed whether any alternative transaction might result in consideration more favorable to Parent than that contemplated by the Merger Agreement.

Our opinion does not address the relative merits of the transactions contemplated by the Merger Agreement as compared to any alternative transaction or opportunity that might be available to the Company or Parent, nor does it address the underlying business decision by the Company or Parent to engage in the Merger or the terms of the Merger Agreement or the documents referred to therein, including the form or structure of the Merger or any term, aspect or implication of any other agreements, arrangements or understandings entered into in connection with, or contemplated by or resulting from the Merger or otherwise. We have not been asked to address, and our opinion does not address, the fairness to, or any other consideration of, the holders of any class of securities, creditors or other constituencies of the Company or Parent or any other party, other than the fairness of the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement. We express no view or opinion as to the actual value of Parent Common Stock when issued in connection with the Merger or the price at which shares of the Company Common Stock or Parent Common Stock or any other securities may trade or otherwise be transferrable at any time. Furthermore, we do not express any view or opinion as to the fairness, financial or otherwise, of the amount or nature of any compensation or other consideration payable to or to be received by any of the Company's or Parent's officers, directors or employees, or any class of such persons, in connection with the Merger relative to the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement or otherwise. Our opinion has been authorized by the Fairness Committee of Jefferies LLC.

It is understood that our opinion is for the use and benefit of the Special Committee (in its capacity as such) in its consideration of the Merger. Our opinion does not constitute a recommendation as to how any holder of shares of the Company Common Stock or Parent Common Stock should vote on the Merger or any matter related thereto.

We have been engaged by Parent to deliver an opinion to the Special Committee in connection with the Merger and will receive a fee for our services, which is payable upon delivery of this opinion. We also will be reimbursed for expenses incurred. Parent has also agreed to indemnify us against liabilities arising out of or in connection with the services rendered and to be rendered by us under such engagement. As you are aware, in the past two years we have not provided financial advisory or financing services to the Company or Parent or their respective affiliates for which we and our affiliates have received compensation. In the ordinary course of our business, we and our affiliates may trade or hold securities or financial instruments (including loans and other obligations) of the Company or Parent and/or their respective affiliates for our own account and for the accounts of our customers and, accordingly, may at any time hold long or short positions or otherwise effect transactions in those securities.

In addition, we may seek to, in the future, provide financial advisory and financing services to the Company, Parent or entities that are affiliated with the Company or Parent, for which we would expect to receive compensation. Except as otherwise expressly provided in our engagement letter with Parent, our opinion may not be used or referred to by Parent, or quoted or disclosed to any person in any manner, without our prior written consent.

Based upon and subject to the foregoing, we are of the opinion that, as of the date hereof, the Exchange Ratio provided for in the Merger pursuant to the Merger Agreement is fair, from a financial point of view, to Parent.

Very truly yours,

JEFFERIES LLC

gessis le

Annex D

## PIPER SANDLER & CO. Fairness Opinion

June 21, 2022

Board of Directors U.S. Well Services, Inc. 1360 Post Oak Boulevard Suite 1800 Houston, Texas

#### Members of the Board:

You have requested our opinion as to the fairness, from a financial point of view, to the holders (other than the Excluded Holders (as defined below)) of Class A common stock, par value \$0.0001 per share (the "Company Common Stock"), of U.S. Well Services, Inc. (the "Company"), of the Exchange Ratio (as defined below), pursuant to the Agreement and Plan of Merger, dated June 21, 2022 (the "Agreement"), to be entered into among the Company, ProFrac Holding Corp. (the "Acquiror") and Thunderclap Merger Sub I, Inc., an indirect subsidiary of the Acquiror ("Merger Sub", and collectively with the Company and Acquiror, the "Parties"). The Agreement provides for, among other things, the merger (the "Merger") of Merger Sub with and into the Company, pursuant to which each outstanding share of Company Common Stock, other than shares of Company Common Stock held by Acquiror (directly or indirectly), the Company (including shares held in treasury) or Merger Sub, will be converted into the right to receive 0.0561 (the "Exchange Ratio") of a validly issued, fully paid and non-assessable share of Class A common stock, par value \$0.01 per share, of the Acquiror (the "Acquiror Common Stock"). The terms and conditions of the Merger are more fully set forth in the Agreement. "Excluded Holders" means the Parties and each of their controlled affiliates.

In arriving at our opinion, we have: (i) reviewed and analyzed the financial terms of a draft of the Agreement dated June 21, 2022; (ii) reviewed and analyzed certain financial and other data with respect to the Company and the Acquiror which was publicly available, (iii) reviewed and analyzed certain information, including financial forecasts, relating to the business, earnings, cash flow, assets, liabilities and prospects of the Company and the Acquiror that were publicly available, as well as those that were furnished to us by the Company; (iv) conducted discussions with members of senior management and representatives of the Company and the Acquiror concerning the matters described in clauses (ii) and (iii) above, as well as their respective businesses and prospects before and after giving effect to the Merger; (v) reviewed the current and historical reported prices and trading activity of Company Common Stock and the Acquiror Common Stock and similar information for certain other companies deemed by us to be comparable to the Company; (vi) compared the financial performance of the Company and the Acquiror with that of certain other publicly traded companies that we deemed relevant; (vii) reviewed the financial terms, to the extent publicly available, of certain business combination transactions that we deemed relevant; (viii) performed a relative discounted cash flow analysis with respect to the Company and the Acquiror; and (ix) performed a contribution analysis. In addition, we have conducted such other analyses, examinations and inquiries and considered such other financial, economic and market criteria as we have deemed necessary in arriving at our opinion.

We have relied upon and assumed, without assuming liability or responsibility for independent verification, the accuracy and completeness of all information that was publicly available or was furnished, or otherwise made available, to us or discussed with or reviewed by us. We have further relied upon the assurances of the management of the Company and the Acquiror that the financial information provided has been prepared on a reasonable basis in accordance with industry practice, and that they are not aware of any information or facts that would make any information provided to us incomplete or misleading. Without limiting the generality of the foregoing, for the purpose of this opinion, we have assumed that with respect to financial forecasts, estimates and

Piper Sandler & Co. June 21, 2022 Page 2

other forward-looking information reviewed by us, that such information has been reasonably prepared based on assumptions reflecting the best currently available estimates and judgments of the management of the Company as to the expected future results of operations and financial condition of the Company and the Acquiror to which such financial forecasts, estimates and other forward-looking information relate. We express no opinion as to any such financial forecasts, estimates or forward-looking information or the assumptions on which they were based. We did not consider the United States federal income tax treatment of the Merger and express no opinion as to the impact of any such tax treatment. We have relied, with your consent, on advice of the outside counsel and the independent accountants to the Company, and on the assumptions of the management of the Company, as to all accounting, legal, tax and financial reporting matters with respect to the Company, the Acquiror and the Agreement.

In arriving at our opinion, we have assumed that the executed Agreement will be in all material respects identical to the last draft reviewed by us. We have relied upon and assumed, without independent verification, that (i) the representations and warranties of all parties to the Agreement and all other related documents and instruments that are referred to therein are true and correct, (ii) each party to such agreements will fully and timely perform all of the covenants and agreements required to be performed by such party, (iii) the Merger will be consummated pursuant to the terms of the Agreement without amendments thereto and (iv) all conditions to the consummation of the Merger will be satisfied without waiver by any party of any conditions or obligations thereunder, in each case in all respects material to our analysis. Additionally, we have assumed that all the necessary regulatory approvals and consents required for the Merger will be obtained in a manner that will not adversely affect the Company, the Acquiror or the contemplated benefits of the Merger.

In arriving at our opinion, we have not performed any appraisals or valuations of any specific assets or liabilities (fixed, contingent or other) of the Company or the Acquiror, and have not been furnished or provided with any such appraisals or valuations, nor have we evaluated the solvency of the Company or the Acquiror under any state or federal law relating to bankruptcy, insolvency or similar matters. The analyses performed by us in connection with this opinion were going concern analyses. We express no opinion regarding the liquidation value of the Company, the Acquiror or any other entity. Without limiting the generality of the foregoing, we have undertaken no independent analysis of any pending or threatened litigation, regulatory action, possible unasserted claims or other contingent liabilities, to which the Company, the Acquiror or any of their affiliates is a party or may be subject, and at the direction of the Company and with its consent, our opinion makes no assumption concerning, and therefore does not consider, the possible assertion of claims, outcomes or damages arising out of any such matters. We have also assumed that neither the Company nor the Acquiror is party to any material pending transaction, including without limitation any financing, recapitalization, acquisition or merger, divestiture or spin-off, other than the Merger.\(^1\)

This opinion is necessarily based upon the information available to us and facts and circumstances as they exist and are subject to evaluation on the date hereof; events occurring after the date hereof could materially affect the assumptions used in preparing this opinion. We are not expressing any opinion herein as to the price at which shares of Company Common Stock or Acquiror Common Stock may trade following announcement of the Merger or at any future time. We have not undertaken to reaffirm or revise this opinion or otherwise comment upon any events occurring after the date hereof and do not have any obligation to update, revise or reaffirm this opinion.

We note that Acquiror has signed a definitive purchase agreement to purchase Signal Peak Silica's Monahans sand mine for approximately \$90 million. We have not included the potential effects of such acquisition in our analysis.

Piper Sandler & Co. June 21, 2022 Page 3

We have been engaged by the Company to act as its financial advisor and we will receive a fee from the Company for providing our services, a significant portion of which is contingent upon the consummation of the Merger. We will also receive a fee for rendering this opinion. Our opinion fee is not contingent upon the consummation of the Merger or the conclusions reached in our opinion. The Company has also agreed to indemnify us against certain liabilities and reimburse us for certain expenses in connection with our services.

We currently own 145,524 shares of Class A common stock in USWS, which we received in connection with our role as financial advisor to the Company in its sale to Matlin & Partners Acquisition Corporation in 2018. Certain members of our deal team have a direct interest in a portion of this position.

We are currently engaged by the Acquiror (i) as its financial advisor in connection with its acquisition of Signal Peak Silica's Monahans sand mine, which acquisition was announced by Acquiror on June 21, 2022 (see footnote 1) and (ii) as its placement agent to upsize its existing term loan and evaluate further debt refinancing options. We are also currently separately engaged by two entities, in each of which the Acquiror has an equity interest, to evaluate strategic options and capital-raising options, and with respect to the engagement that includes capital-raising options, Acquiror is a potential investor.

In the past, we have provided financial advisory and financing services to the Company, the Acquiror and certain of their affiliates and may continue to do so and have received, and may receive, fees for the rendering of such services. We have provided a variety of financial advisory services to the Acquiror and to parties with significant investments in the Acquiror, including: (i) acting as co-manager in arranging a senior secured credit facility for Acquiror in 2018; (ii) advising an entity on the sale of a large undeveloped property to the Acquiror in 2021; (iii) advising an entity on the sale of such entity's preferred equity held by a third party to the Acquiror in 2022; (iv) selling a minority equity stake in an entity to the Acquiror in 2022; (v) advising an entity regarding a long-term supply agreement with the Acquiror in return for convertible preferred equity securities of Acquiror in 2022; (vi) serving as exclusive financial advisor on the Acquiror's acquisition of an entity in 2022; (vii) acting as sole debt placement agent in arranging a senior secured term loan for Acquiror in 2022 to fund a portion of the Acquiror's acquisition described in the preceding clause; (viii) acting as sole debt placement agent in 2022 in arranging an upsize of the Acquiror's existing revolving credit facility; and (ix) serving as an active bookrunner on the Acquiror's initial public offering in 2022. With respect to the services described in clauses (i), (vi), (vii), (viii) and (ix), we realized aggregate gross fees of approximately \$12.5 million.

In addition, in the ordinary course of our business, we and our affiliates may actively trade securities of the Company and the Acquiror for our own account or the account of our customers and, accordingly, may at any time hold a long or short position in such securities. We may also, in the future, provide investment banking and financial advisory services to the Company, the Acquiror or entities that are affiliated with the Company or the Acquiror, for which we would expect to receive compensation.

Consistent with applicable legal and regulatory requirements, Piper Sandler has adopted policies and procedures to establish and maintain the independence of its Research Department and personnel. As a result, Piper Sandler's research analysts may hold opinions, make statements or recommendations, and/or publish research reports with respect to the Company and the Merger and other participants in the Merger that differ from the views of Piper Sandler's investment banking personnel.

This opinion is provided to the Board of Directors of the Company in connection with its consideration of the Merger and is not intended to be and does not constitute a recommendation to any stockholder of the Company as

Piper Sandler & Co. June 21, 2022 Page 4

to how such stockholder should act or vote or tender their shares or otherwise participate in the Merger, or make any election with respect to the Merger or any other matter. Except with respect to the use of this opinion in connection with the proxy statement relating to the Merger in accordance with our engagement letter with the Company, this opinion shall not be disclosed, referred to, published or otherwise used (in whole or in part), nor shall any public references to us be made (including any description of or reference to Piper

Sandler, this opinion or the advice rendered by us in any proxy statement), without our prior written approval. This opinion has been approved for issuance by the Piper Sandler Opinion Committee.

This opinion addresses solely the fairness, from a financial point of view, to holders of Company Common Stock (other than the Excluded Holders) of the proposed Exchange Ratio set forth in the Agreement and does not address any other terms or agreement relating to the Merger or any other terms of the Agreement. We were not requested to opine as to, and this opinion does not address, the basic business decision to proceed with or effect the Merger, the merits of the Merger relative to any alternative transaction or business strategy that may be available to the Company, Acquiror's ability to fund the merger consideration, any other terms contemplated by the Agreement or the fairness of the Merger to any other class of securities, creditor or other constituency of the Company. Furthermore, we express no opinion with respect to the amount or nature of compensation to any officer, director or employee of any party to the Merger, or any class of such persons, relative to the Exchange Ratio to be received by holders of Company Common Stock in the Merger or with respect to the fairness of any such compensation.

Based upon and subject to the foregoing and based upon such other factors as we consider relevant, it is our opinion that the Exchange Ratio is fair, from a financial point of view, to the holders of Company Common Stock (other than the Excluded Holders) in their capacity as holders of Company Common Stock as of the date hereof.

Sincerely,

PIPER SANDLER & CO.

#### PART II INFORMATION NOT REQUIRED IN PROSPECTUS

#### Item 20. Indemnification of Directors and Officers.

The following summary is qualified in its entirety by reference to the General Corporation Law of the State of Delaware (the **DGCL**") and to ProFrac's certificate of incorporation, as amended to date, and ProFrac's bylaws, as amended to date. Section 102(b)(7) of the DGCL permits a corporation to eliminate the personal liability of its directors for monetary damages for a breach of fiduciary duty as a director, except where the director breached his or her duty of loyalty, failed to act in good faith, engaged in intentional misconduct or knowingly violated a law, authorized the payment of a dividend or approved a stock repurchase or redemption in violation of Delaware corporate law or obtained an improper personal benefit. ProFrac's certificate of incorporation provides the liability of ProFrac's directors to ProFrac or its stockholders shall be eliminated to the fullest extent permitted by the DGCL, as the same exists or may be amended from time to time.

Section 145 of the DGCL provides that a corporation has the power to indemnify a director, officer, employee or agent of the corporation and certain other persons serving at the request of the corporation in related capacities against expenses (including attorneys' fees), judgments, fines and amounts paid in settlements actually and reasonably incurred by the person in connection with an action, suit or proceeding to which he or she is or is threatened to be made a party by reason of such position, if such person acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the corporation, and, in any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful, except that, in the case of actions brought by or in the right of the corporation, indemnification is limited to expenses and no indemnification shall be made with respect to any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the Court of Chancery or other adjudicating court determines that, despite the adjudication of liability but in view of all of the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

ProFrac's bylaws include provisions that indemnify, to the fullest extent authorized or permitted by law, any person made or threated to be made a party to any action, suit, or proceeding, whether criminal, civil, administrative, or investigative, by reason of the fact that such person or such person's testator or intestate is or was a director, officer, or employee of ProFrac or serves or served at the request of ProFrac at any other enterprise (such as another corporation, partnership, joint venture, trust, other enterprise and non-profit entity, including service with respect to an employee benefit plan) as a director, officer, or employee. ProFrac's bylaws also provide that expenses incurred by any such person in defending any such action, suit, or proceeding shall be paid or reimbursed by ProFrac promptly upon receipt by it of an undertaking by such person to repay such expenses if it shall ultimately be determined that such person is not entitled to be indemnified by ProFrac. Additionally, ProFrac's bylaws expressly grant ProFrac the power to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the corporation. ProFrac has entered into indemnification agreements with each of its current directors and officers pursuant to the foregoing provisions. ProFrac will also maintain an insurance policy covering its officers and directors with respect to certain liabilities.

## Item 21. Exhibits and Financial Statement Schedules.

The following exhibits are filed herewith or incorporated herein by reference:

Exhibit No.	Document
2.1	Agreement and Plan of Merger, dated as of October 21, 2021, by and between FTS International, Inc., ProFrac Holdings, LLC and ProFrac Acquisitions, Inc. (incorporated by reference to Exhibit 2.1 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on November 22, 2021).
2.2	Master Reorganization Agreement, dated as of May 12, 2022, by and among ProFrac Holdings, LLC, ProFrac Holding Corp. and the other parties thereto (incorporated by reference to Exhibit 2.1 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
3.1	Amended and Restated Certificate of Incorporation of ProFrac Holding Corp., as filed with the Secretary of State of the State of Delaware on May 17, 2022 (incorporated by reference to Exhibit 3.1 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
3.2	Second Amended and Restated Certificate of Incorporation of U.S. Well Services, Inc. (incorporated by reference to Exhibit 3.1 to U.S. Well Services, Inc.'s Current Report on Form 8-K filed with the SEC on November 16, 2018).
3.3	Certificate of Amendment to the Second Amended and Restated Certificate of Incorporation of U.S. Well Services, Inc., dated as of August 4, 2022 (incorporated by reference to Exhibit 3.1 to U.S. Well Services, Inc.'s Current Report on Form 8-K filed with the SEC on August 5, 2022).
3.4	Certificate of Amendment to Second Amended and Restated Certificate of Incorporation of U.S. Well Services, Inc. (incorporated by reference to Exhibit 3.1 to U.S. Well Services, Inc.'s Current Report on Form 8-K filed with the SEC on October 1, 2021).
3.5	Certificate of Designations dated May 24, 2019, of U.S. Well Services, Inc. (incorporated by reference to Exhibit 3.1 to U.S. Well Services, Inc.'s Current Report on Form 8-K filed with the SEC on May 24, 2019).
3.6	Certificate of Designations dated March 31, 2020, U.S. Well Services, Inc. (incorporated by reference to Exhibit 3.1 to U.S. Well Services, Inc.'s Current Report on Form 8-K filed with the SEC on April 2, 2020).
3.7	First Amendment to Certificate of Designations of the Series B Redeemable Convertible Preferred Stock, dated September 14, 2021 (incorporated by reference to Exhibit 3.1 to U.S. Well Services, Inc.'s Current Report on Form 8-K filed with the SEC on September 17, 2021).
3.8	Amended and Restated Bylaws of ProFrac Holding Corp., effective as of May 17, 2022 (incorporated by reference to Exhibit 3.2 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
3.9	Amended and Restated Bylaws of U.S. Well Services, Inc. (incorporated by reference to Exhibit 3.4 to U.S. Well Services, Inc.'s Registration Statement on Form S-1 (File No. 333-216076), filed with the SEC on February 15, 2017).
4.1	Form of Class A Common Stock Certificate (incorporated by reference to Exhibit 4.1 to ProFrac Holding Corp.'s Registration Statement on Form S-1/A (File No. 333-261255) filed with the SEC on November 30, 2021).
4.2	Registration Rights Agreement, dated as of May 17, 2022, by and among ProFrac Holding Corp., THRC Holdings, LP, Farris C. Wilks and the other parties thereto (incorporated by reference to Exhibit 4.1 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).

Exhibit No.	Document
4.3	Third Amended and Restated Limited Liability Company Agreement of ProFrac Holdings, LLC, dated as of May 17, 2022 (incorporated by reference to Exhibit 4.2 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
4.4	Stockholders' Agreement, dated as of May 17, 2022, by and among ProFrac Holding Corp., THRC Holdings, LP, Farris C. Wilks, FARJO Holdings, LP and the Farris and Jo Ann Wilks 2022 Family Trust (incorporated by reference to Exhibit 4.3 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
4.5	Right Agreement, dated as of December 20, 2021, by and among ProFrac Holdings, LLC and Eagleton Ventures, Inc. (incorporated by reference to Exhibit 4.5 to Amendment No. 2 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on March 31, 2022).
4.6	Form of West Munger Registration Rights Agreement (incorporated by reference to Exhibit 4.6 to Amendment No. 3 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) originally filed with the SEC on November 22, 2021).
5.1*	Opinion of Brown Rudnick LLP.
10.1	Form of Voting Agreement, by and among ProFrac Holding Corp. and the stockholders of U.S. Well Services, Inc. party thereto (incorporated by reference to Exhibit 10.1 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on June 24, 2022).
10.2	Form of Warrant Purchase Agreement, by and among ProFrac Holding Corp. and the holders of warrants of U.S. Well Services, Inc. party thereto (incorporated by reference to Exhibit 10.2 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on June 24, 2022).
10.3	Tax Receivable Agreement, dated as of May 17, 2022, by and among ProFrac Holding Corp., the TRA Holders and the Agents named therein (incorporated by reference to Exhibit 10.1 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.4	Shared Services Agreement, dated as of May 3, 2022, by and between Wilks Brothers, LLC and ProFrac Holdings II, LLC (incorporated by reference to Exhibit 10.3 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.5	Indemnification Agreement (Johnathan Ladd Wilks) (incorporated by reference to Exhibit 10.4 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.6	Indemnification Agreement (Matthew D. Wilks) (incorporated by reference to Exhibit 10.5 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.7	Indemnification Agreement (James Coy Randle, Jr.) (incorporated by reference to Exhibit 10.6 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.8	Indemnification Agreement (Lance Turner) (incorporated by reference to Exhibit 10.7 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.9	Indemnification Agreement (Robert Willette) (incorporated by reference to Exhibit 10.8 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.10	Indemnification Agreement (Sergei Krylov) (incorporated by reference to Exhibit 10.9 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.11	Indemnification Agreement (Theresa Glebocki) (incorporated by reference to Exhibit 10.10 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.12	Indemnification Agreement (Stacy Nieuwoudt) (incorporated by reference to Exhibit 10.11 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).

Exhibit No.	Document
10.13	Indemnification Agreement (Gerald Haddock) (incorporated by reference to Exhibit 10.12 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.14	Patent License Agreement, effective as of June 29, 2021, between U.S. Well Services, LLC and ProFrac Manufacturing, LLC (incorporated by reference to Exhibit 10.6 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on November 22, 2021).
10.15	First Amendment to Term Loan Credit Agreement, dated as of July 25, 2022, by and among ProFrac Holdings II, LLC, ProFrac Holdings, LLC, the guarantors party thereto, the lenders party thereto, and Piper Sandler Finance LLC, as the agent and collateral agent for the lenders (incorporated by reference to Exhibit 10.1 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on July 29, 2022).
10.16	First Amendment to Credit Agreement, dated as of July 25, 2022, by and among ProFrac Holdings II, LLC, ProFrac Holdings, LLC, the guarantors party thereto, the lenders party thereto, and JPMorgan Chase Bank, N.A., as the agent and collateral agent for the lenders (incorporated by reference to Exhibit 10.2 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on July 29, 2022).
10.17	Subordinated Promissory Note among ProFrac Holdings II, LLC, as payor, and Equify Financial LLC, as payee, dated as of March 4, 2022 (incorporated by reference to Exhibit 10.11 to Amendment No. 2 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on March 31, 2022).
10.18	Subordinated Promissory Note among ProFrac Holdings, LLC, as payor, and THRC Holdings, LP, as payee, dated as of March 4, 2022 (incorporated by reference to Exhibit 10.12 to Amendment No. 2 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on March 31, 2022).
10.19	Subordinated Promissory Note among ProFrac Holdings, LLC, as payor, and THRC Holdings, LP, as payee, dated as of March 4, 2022 (incorporated by reference to Exhibit 10.13 to Amendment No. 2 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on March 31, 2022).
10.20	Securities Purchase Agreement dated February 16, 2022 by and between Flotek Industries, Inc. and ProFrac Holdings, LLC (incorporated by reference to Exhibit 10.14 to Amendment No. 2 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on March 31, 2022).
10.21	Securities Purchase Agreement between Flotek Industries, Inc. and ProFrac Holdings II, LLC dated June 17, 2022 (incorporated by reference to Exhibit 10.1 to Flotek Industries, Inc.'s Current Report on Form 8-K filed with the SEC on June 23, 2022).
10.22	Purchase and Sale Agreement, dated as of February 18, 2022, by and between ProFrac Holdings, LLC and Wilks Development, LLC (incorporated by reference to Exhibit 10.6 to Amendment No. 3 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on April 26, 2022).
10.23	Chemical Products Supply Agreement between Flotek Chemistry, LLC and ProFrac Services, LLC dated February 2, 2022 (incorporated by reference to Exhibit 10.2 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 23, 2022).
10.24	Amendment No. 1 to Chemical Products Supply Agreement between Flotek Chemistry, LLC and ProFrac Services, LLC, dated May 17, 2022 (incorporated by reference to Exhibit 10.1 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 23, 2022).

Exhibit No.	Document
10.25†	Employment Agreement, effective as of May 22, 2018, between ProFrac Services, LLC and James Coy Randle (incorporated by reference to Exhibit 10.15 to Amendment No. 3 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on April 26, 2022).
10.26†	ProFrac Holding Corp. 2022 Long Term Incentive Plan (incorporated by reference to Exhibit 10.2 to ProFrac Holding Corp.'s Current Report on Form 8-K filed with the SEC on May 18, 2022).
10.27†	Form of Restricted Stock Unit Agreement (incorporated by reference to Exhibit 10.16 to Amendment No. 3 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on April 26, 2022.
10.28†	Form of Restricted Stock Unit Agreement (Directors) (incorporated by reference to Exhibit 4.5 to ProFrac Holding Corp.'s Registration Statement on Form S-8 (File No. 333-265176) filed with the SEC on May 24, 2022).
10.29†	Assignment Agreement, dated as of May 10, 2022, by and between Farris Wilks and Jo Ann Wilks, asCo-Trustees of the Farris and Jo Ann Wilks 2022 Family Trust, created by Trust Agreement dated as of May 10, 2022, as assignor, and KWELL Holdings, LP, as assignee and Declaration of Intent (incorporated by reference to Exhibit 10.18 to ProFrac Holding Corp.'s Quarterly Report on Form 10-Q filed with the SEC on August 15, 2022).
10.30†	Assignment and Assumption Agreement, dated as of August [], 2022, by and between THRC Holdings, LP, a Texas limited liability company, as assignor, and Matthew D. Wilks, as assignee (incorporated by reference to Exhibit 10.19 to ProFrac Holding Corp.'s Quarterly Report on Form 10-Q filed with the SEC on August 15, 2022).
10.31*	Form of Amended and Restated Series A Warrant of U.S. Well Services, Inc.
10.32*	Form of Amended and Restated SPAC Warrants of U.S. Well Services, Inc.
10.33*	Form of Amended and Restated Placement Agent Warrants of U.S. Well Services, Inc.
10.34*	Form of Amended and Restated RDO Warrants of U.S. Well Services, Inc.
10.35	Form of First Amendment to Convertible Senior Secured (Third Lien) PIK Note (included as Exhibit A to Annex A to this proxy statement/information statement/prospectus).
10.36	Form of First Amendment to the Series A Certificate of Designations (included as Exhibit B to Annex A to this proxy statement/information statement/prospectus).
10.37	Form of Amendment to Performance Award Agreement (included as Exhibit C to Annex A to this proxy statement/information statement/prospectus).
10.38	Letter Agreement, dated June 21, 2022 by and among U.S. Well Services, Inc., the other Loan Parties party thereto, and the Term Loan C Lenders party thereto (incorporated by reference to Exhibit 10.3 to U.S. Well Services, Inc.'s Current Report on Form 8-K filed with the SEC on June 27, 2022).
21.1	List of Subsidiaries of ProFrac Holding Corp. (incorporated by reference to Exhibit 21.1 to the amendment to the Registrant's Form S-1 Registration Statement, File No. 333-261255, originally filed with the Commission on May 4, 2022).
23.1*	Consent of Grant Thornton LLP, independent registered public accounting firm of ProFrac Holdings, LLC.
23.2*	Consent of Grant Thornton LLP, independent registered public accounting firm of ProFrac Holding Corp.
23.3*	Consent of Grant Thornton LLP, independent registered public accounting firm of FTS International, Inc.

Exhibit No.	Document
23.4*	Consent of KPMG LLP, independent registered public accounting firm of U.S. Well Services, Inc.
23.5*	Consent of Brown Rudnick LLP (included as part of Exhibit 5.1).
23.6*	Consent of John T. Boyd Company, independent mining engineers and geologists for ProFrac Holding Corp.
24.1*	Power of Attorney (contained on the signature page of this registration statement).
99.1*	Form of U.S. Well Services, Inc. Proxy Card.
99.2*	Consent of Jefferies LLC.
99.3*	Consent of Piper Sandler & Co.
99.4	Action by Written Consent of the Stockholders of ProFrac Holding Corp. dated June 21, 2022 (included as Annex B to the proxy statement/information statement/prospectus).
99.5	Summary Reserve Report of John T. Boyd Company (incorporated by reference to Exhibit 99.1 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on November 22, 2021).
99.6	Addendum to Summary Reserve Report of John T. Boyd Company (incorporated by reference to Exhibit 99.2 to Amendment No. 3 to ProFrac Holding Corp.'s Registration Statement on Form S-1 (File No. 333-261255) filed with the SEC on April 26, 2022.
101.INS*	Inline XBRL Instance Document.
101.SCH*	Inline XBRL Taxonomy Extension Schema Document.
101.CAL*	Inline XBRL Taxonomy Extension Calculation Linkbase Document.
101.DEF*	Inline XBRL Taxonomy Extension Definition Linkbase Document.
101.LAB*	Inline XBRL Taxonomy Extension Label Linkbase Document.
101.PRE*	Inline XBRL Taxonomy Extension Presentation Linkbase Document.
107*	Filing Fee Table.

Filed herewith.

To be filed by Amendment.

Compensatory plan or arrangement.

Schedules have been omitted pursuant to Item 601(a)(5) of Regulation S-K. ProFrac Holding Corp. agrees to furnish supplementally a copy of such schedules, or any section thereof, to the SEC upon request.

#### Item 22. Undertakings.

- (a) The undersigned registrant hereby undertakes:
  - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
    - (i) to include any prospectus required by Section 10(a)(3) of the Securities Act of 1933;
  - (ii) to reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Securities and Exchange Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement; and
  - (iii) to include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement.
- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (4) The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference into the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
  - (5) The undersigned registrant undertakes as follows:
  - (i) that prior to any public reoffering of the securities registered hereunder through use of a prospectus which is a part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.
  - (ii) The undersigned registrant undertakes that every prospectus (a) that is filed pursuant to paragraph (5)(i) immediately preceding, or (b) that purports to meet the requirements of Section 10(a)(3) of the Securities Act of 1933 and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.

- (6) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act of 1933 and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act of 1933 and will be governed by the final adjudication of such issue.
- (b) The undersigned registrant hereby undertakes to respond to requests for information that is incorporated by reference into the prospectus pursuant to Items 4, 10(b), 11 or 13 of this form, within one business day of receipt of such request, and to send the incorporated documents by first-class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the registration statement through the date of responding to the request.
- (c) The undersigned registrant hereby undertakes to supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in the registration statement when it became effective.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Act of 1933, as amended, ProFrac Holding Corp. has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Willow Park, State of Texas, on August 30, 2022.

#### PROFRAC HOLDING CORP.

By: /s/ Matthew D. Wilks

Name: Matthew D. Wilks

Title: Executive Chairman and Director

#### POWER OF ATTORNEY

KNOWN ALL PERSONS BY THESE PRESENTS, that each of the undersigned constitutes and appoints Matthew D. Wilks and Lance Turner, and each of them, as his or her true and lawful attorney-in-fact and agent, with full power of substitution and resubstitution, for such person and in his or her name, place and stead, in any and all capacities, to sign this Registration Statement on Form S-4 (including all pre-effective and post-effective amendments), and to file the same, with all exhibits thereto, and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorney-in-fact and agent, acting alone, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes as he or she might or could do in person, hereby ratifying and confirming that such attorney-in-fact and agent, or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following persons in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Matthew D. Wilks Matthew D. Wilks	Executive Chairman and Director (Principal Executive Officer)	August 30, 2022
/s/ Lance Turner Lance Turner	Chief Financial Officer (Principal Financial Officer and Principal Accounting Officer)	August 30, 2022
/s/ Ladd Wilks Ladd Wilks	Chief Executive Officer	August 30, 2022
/s/ Sergei Krylov Sergei Krylov	Director	August 30, 2022
/s/ Terry Glebocki Terry Glebocki	Director	August 30, 2022
/s/ Stacy Nieuwoudt Stacy Nieuwoudt	Director	August 30, 2022
/s/ Gerald Haddock Gerald Haddock	Director	August 30, 2022

ProFrac Holding Corp. 333 Shops Boulevard, Suite 301 Willow Park, Texas 76087

Re: Registration Statement for ProFrac Holding Corp.

#### Ladies and Gentlemen:

We have acted as counsel to ProFrac Holding Corp., a Delaware corporation (the "Company"), in connection with the Registration Statement on Form S-4 (the "Registration Statement") filed by the Company with the Securities and Exchange Commission (the "Commission") under the Securities Act of 1933, as amended (the "Securities Act") relating to the proposed registration by the Company of 14,056,943 shares (the "Shares") of the Company's Class A common stock (the "Common Stock"), 19,167,417 warrants to purchase Common Stock (the "Warrants") to be issued in connection with the merger contemplated by the Agreement and Plan of Merger, dated June 21, 2022 (the "Merger Agreement"), by and among the Company, U.S. Well Services, Inc., a Delaware corporation ("USWS"), and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of the Company, and 153,613 shares of Common Stock issuable upon exercise of the Warrants (the "Warrant Shares"). The Shares, the Warrants and the Warrant Shares, when issued in accordance with the Merger Agreement, are referred to herein as the "Securities."

In connection with the registration of the Securities, we have examined, are familiar with, and have relied as to factual matters solely upon, copies of the following documents for the purpose of rendering this opinion (collectively, the "<u>Documents</u>"):

- 1. the Registration Statement;
- 2. the Merger Agreement;
- 3. the amended and restated certificate of incorporation of the Company filed as Exhibit 3.1 to the Registration Statement;
- 4. Warrant Agreement by and between USWS (f/k/a Matlin & Partners Acquisition Corporation) and Continental Stock Transfer & Trust Company dated March 9, 2017;
- 5. The form of Amended and Restated Warrant Agreement, filed as Exhibit 10.32 to the Registration Statement (the "Amended and Restated Warrant Agreement");
- 6. Specimen Stock Certificate for the Company's Common Stock;
- 7. the bylaws of the Company as presently in effect as of the date hereof as certified by the Secretary of the Company; and
- 8. the resolutions adopted by the Company's board of directors, relating to the execution and delivery of, and the performance by the Company of its obligations under, the Merger Agreement, including the issuance of the Securities.

For purposes of this opinion, we have assumed, without any investigation, (i) the legal capacity of each natural person, (ii) the full power and authority of each entity and person other than the Company to execute, deliver and perform each document heretofore executed and delivered or hereafter to be executed and delivered and to do each other act heretofore done or hereafter to be done by such entity or person, (iii) the due authorization by each entity or person other than the Company of each document heretofore executed and delivered or hereafter to be executed and delivered and to do each other act heretofore done or to be done by such entity or person, (iv) the due execution and delivery by each entity or person other than the Company of each document heretofore executed and delivered and delivered by such entity or person, (v) the legality, validity, binding effect and enforceability of each document heretofore executed and delivered or hereafter to be executed and delivered by each entity or person other than the Company and of each other act heretofore done or hereafter to be done by such entity or person, (vi) the genuineness of each signature on, and the completeness of each document submitted to us as a copy, (viii) the conformity to the original of each document submitted to us as a copy, (vii) the completeness, accuracy and proper indexing of all governmental and judicial records searched and (x) no modification of any provision of any document, no waiver of any right or remedy and no exercise of any right or remedy other than in a commercially reasonable and conscionable manner and in good faith.

The opinions expressed herein are based solely upon (i) our review of the Documents, (ii) discussions with those of our attorneys who have devoted substantive attention to the matters contained herein, and (iii) such review of published sources of law as we have deemed necessary.

We have not, except as specifically noted herein, made any independent review or investigation of orders, judgments, rules or other regulations or decrees by which the Company or any of its property may be bound. We also have not made any independent investigation as to the existence of actions, suits, investigations, or proceedings, if any, pending or threatened against the Company. Our opinions contained herein are limited to the laws of the General Corporation Law of the State of Delaware, including the statutory provisions, all applicable provisions of the Delaware Constitution, and reported judicial decisions interpreting these laws, and the federal law of the United States of America and, with respect to the opinions set forth in numbered paragraph 2 below, the internal laws of the State of New York.

Our opinion is based on these laws as in effect on the date hereof and as of the effective date of the Registration Statement, and we assume no obligation to revise or supplement this opinion after the effective date of the Registration Statement should the law be changed by legislative action, judicial decision or otherwise. Where our opinions expressed herein refer to events to occur at a future date, we have assumed that there will have been no changes in the relevant law or facts between the date hereof and such future date. The opinions hereafter expressed with respect to the enforceability of the Securities are subject to the general qualifications that such rights and remedies may be subject to and affected by: (i) applicable bankruptcy, insolvency, reorganization, receivership, moratorium, or assignment for the benefit of creditors laws and other laws affecting the rights and remedies of creditors generally, including, without limitation, laws regarding fraudulent transfers, fraudulent conveyances, preferences, avoidance, automatic stay and turn-over; (ii) general principles of equity, including, without limitation, those governing the availability of equitable remedies, affording equitable defenses, requiring good faith, fair dealing and reasonableness in the performance and enforcement of a contract, and affording defenses based upon unconscionability, lack of notice, impracticability or impossibility of performance; and (iii) general rules of contract law with respect to matters such as the election of remedies, the limits of severability, mutuality of obligations, and opportunity to cure, limitations on the enforceability of indemnification, contribution or exculpation provisions under applicable securities laws or otherwise and limitations on the enforceability of provisions which are in violation of public policy.

Based on the foregoing, and in reliance thereon, and subject to the limitations, qualifications and exceptions set forth herein, we are of the opinion that:

- 1. the Securities, when issued as described in the Registration Statement and in accordance with the terms of the Merger Agreement and the Form of Amended and Restated Warrant Agreement, as applicable, will be validly issued, fully-paid and nonassessable.
- the Warrants, when issued in accordance with their terms as described in the Registration Statement, will constitute valid and legally binding obligations of the Company.

In connection with our opinions expressed above, we have assumed, at or prior to the time of the delivery of any Securities, (i) that the Registration Statement, as then amended, will be declared effective by the Commission under the Securities Act and will remain effective on the date of issuance of the Securities and that the registration will apply to all of the Shares, the Warrants and the Warrant Shares, and will not have been modified or rescinded and that there will not have occurred any change in law affecting the validity of the issuance of such Shares, Warrants or Warrant Shares, as applicable, (ii) the execution and delivery of the Amended and Restated Warrant Agreement, (iii) that the authorization of the Securities will not have been modified or rescinded by the Board of Directors of the Company or a committee thereof and there will not have occurred any change in law affecting the validity or enforceability of such Security, (iv) that the stockholders of USWS will have approved and adopted the Merger Agreement and the transactions contemplated thereby and (v) that the other conditions to consummating the transactions contemplated by the Merger Agreement will have been satisfied and such transactions are consummated.

We express no legal opinion upon any matter other than those explicitly addressed above, and our express opinions therein contained shall not be interpreted to be implied opinions upon any other matter. Not in limitation of the foregoing, we are not rendering any opinion as to the compliance with any other federal or state law, rule or regulation relating to securities, or to the sale or issuance thereof. This opinion may be used in connection with the offering of the Shares, the Warrants and the Warrant Shares only while the Registration Statement, as it may be amended from time to time, remains in effect.

We hereby consent to the filing of this opinion with the Commission in connection with the filing of the Registration Statement. We also consent to the use of our name in the related prospectus and any related prospectus supplement under the heading "Legal Matters." In giving this consent, we do not thereby admit that we are included in the category of persons whose consent is required under Section 7 of the Securities Act or the rules and regulations promulgated thereunder.

Very truly yours,

/s/ Brown Rudnick LLP BROWN RUDNICK LLP

#### FORM OF AMENDED AND RESTATED WARRANT AGREEMENT

between

#### PROFRAC HOLDING CORP.

and

#### CONTINENTAL STOCK TRANSFER & TRUST COMPANY

THIS AMENDED AND RESTATED WARRANT AGREEMENT (this "Agreement"), dated as of [\_\_], 2022, is by and between ProFrac Holding Corp., a Delaware corporation (the "Company"), and Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "Warrant Agent"), also referred to herein as the "Transfer Agent").

WHEREAS, U.S. Well Services, Inc. ("USWS") and Transfer Agent entered into that certain Purchase Agreement (the 'Purchase Agreement'), dated as of May 23, 2019, with the purchasers named therein (collectively, the "Purchasers"), pursuant to which USWS agreed to issue and sell, and the Purchasers agreed to purchase, shares of preferred stock and warrants (the "USWS Warrants");

WHEREAS, USWS and Transfer Agent are parties to that certain Warrant Agreement, dated as of May 24, 2019 (the **Existing Warrant Agreement**");

WHEREAS, each USWS Warrant entitles the holder thereof to purchase one share of Class A common stock of USWS, par value \$0.0001 per share ("USWS Class A Common Stock"), for the Warrant Price as described in the Existing Warrant Agreement;

WHEREAS, on June 21, 2022 that certain Agreement and Plan of Merger ('Merger Agreement') was entered into by and among the Company, USWS, and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of Company ("Merger Sub");

WHEREAS, the Merger Agreement provides for, among other things, the merger of Merger Sub with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of the Company (the "Merger");

WHEREAS, the Company and Transfer Agent wish to amend the Existing Warrant Agreement and each USWS Warrant outstanding as of the effective time of the Merger (the "Existing Warrants") such that as of the effective time of the Merger, the holders of the Existing Warrants (Existing Warrant Holders") shall have the right to purchase the number of shares of Class A Common Stock of the Company, par value \$.01 per share, equal to the product of (i) the number of shares of Series A Common Stock of USWS, par value \$0.0001 per share (the "USWS Common Stock") underlying the Existing Warrants and (ii) 0.3366 (as adjusted for the 1-for-6 reverse split of USWS Common Stock effected by USWS on August 4, 2022);

WHEREAS, the Company and Transfer Agent desire to amend the Existing Warrant Agreement such that the Company assumes the Existing Warrants and they are otherwise amended as of the Effective Time in accordance with the terms hereof and of the Merger Agreement (each such amended Existing Warrant, a "Warrant");

WHEREAS, the Company desires the Warrant Agent to act on behalf of the Company, and the Warrant Agent is willing to so act, in connection with the issuance, registration, transfer, exchange, redemption and exercise of the Warrants; and

WHEREAS, the Company desires to provide for the form and provisions of the Warrants, the terms upon which they shall be issued and exercised, and the respective rights, limitation of rights, and immunities of the Company, the Warrant Agent, and the holders of the Warrants; and WHEREAS, all acts and things have been done and performed which are necessary to make the Warrants, when executed on behalf of the Company and countersigned by or on behalf of the Warrant Agent, as provided herein, the valid, binding and legal obligations of the Company, and to authorize the execution and delivery of this Agreement.

NOW, THEREFORE, in consideration of the mutual agreements herein contained, the parties hereto agree as follows:

1. **APPOINTMENT OF WARRANT AGENT**. The Company hereby appoints the Warrant Agent to act as agent for the Company for the Warrants, and the Warrant Agent hereby accepts such appointment and agrees to perform the same in accordance with the terms and conditions set forth in this Agreement.

#### 2. WARRANTS.

- 2.1 Form of Warrant. Each Warrant shall be physically certificated in the form annexed hereto as Exhibit A.
- 2.2 Effect of Countersignature. Unless and until countersigned by the Warrant Agent pursuant to this Agreement, a Warrant shall be invalid and of no effect and may not be exercised by the holder thereof.

#### 2.3 Registration.

2.3.1 Warrant Register. The Warrant Agent shall maintain books (the "Warrant Register") for the registration of original issuance and the registration of transfer of the Warrants. Upon the initial issuance of the Warrants, the Warrant Agent shall issue and register the Warrants in the names of the respective holders thereof in such denominations and otherwise in accordance with instructions delivered to the Warrant Agent by the Company.

Physical certificates shall be signed by, or bear the facsimile signature of, the Executive Chairman of the Board, Chief Executive Officer, Chief Financial Officer, Secretary or other principal officer of the Company. In the event the person whose facsimile signature has been placed upon any Warrant shall have ceased to serve in the capacity in which such person signed the Warrant before such Warrant is issued, it may be issued with the same effect as if he or she had not ceased to be such at the date of issuance.

2.3.2 Registered Holder. Prior to due presentment for registration of transfer of any Warrant, the Company and the Warrant Agent may deem and treat the person in whose name such Warrant is registered in the Warrant Register (the "Registered Holder") as the absolute owner of such Warrant and of each Warrant represented thereby (notwithstanding any notation of ownership or other writing on any physical certificate made by anyone other than the Company or the Warrant Agent), for the purpose of any exercise thereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.

#### 3. TERMS AND EXERCISE OF WARRANTS.

- 3.1 <u>Warrant Price</u>. Each Warrant shall, when countersigned by the Warrant Agent, entitle the Registered Holder thereof, subject to the provisions of such Warrant and of this Agreement, to purchase from the Company the number of shares of Class A Common Stock stated therein (the "Warrant Entitlement"), at the price of \$477.89 per share (the "Warrant Price"), subject to the adjustments provided in <u>Section 4</u> hereof.
- 3.2 <u>Duration of Warrants</u>. A Warrant may be exercised only during the period (the **Exercise Period**") commencing on the date that is six months and one day after the date of this Agreement and terminating at 5:00 p.m., New York City time, on the date that is six (6) years after such date (the "**Expiration Date**"). Each Warrant not exercised on or before the Expiration Date shall become void, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at 5:00 p.m., New York City time, on the Expiration Date.

#### 3.3 Exercise of Warrants.

3.3.1 Cashless Exercise. Subject to the provisions of the Warrant and this Agreement, a Warrant, when countersigned by the Warrant Agent, may be exercised in whole or in part from time to time by the Registered Holder thereof by surrendering it, at the office of the Warrant Agent, or at the office of its successor as Warrant Agent, in the Borough of Manhattan, City and State of New York, with the subscription form, as set forth in the Warrant, duly executed, for that number of shares of Class A Common Stock equal to the quotient obtained by dividing (a) the product of (i) the aggregate Warrant Entitlement of the Warrants subject to such surrender and exercise, multiplied by (ii) the difference between the Warrant Price and the "Fair Market Value", as defined in this Section 3.3.1, by (b) the Fair Market Value. Solely for purposes of this Section 3.3.1, "Fair Market Value" means the volume weighted average price of the Class A Common Stock as reported during the ten (10) trading day period ending on the second trading day prior to the date on which the notice of exercise is delivered by such holder to the Warrant Agent. The holder shall be responsible for any and all applicable taxes due in connection with the exercise of the Warrant, the exchange of the Warrant for shares of Class A Common Stock and the issuance of such Class A Common Stock. For the avoidance of doubt, the Warrant shall only be exercisable on a "cashless" basis in accordance with this Section 3.1.1.

#### 3.3.2 Issuance of Shares of Class A Common Stock on Exercise.

(a) As soon as practicable after the exercise of any Warrant pursuant to Section 3.3.1, the Company shall issue to the Registered Holder of such Warrant a book-entry position or certificate, as applicable, for the number of full shares of Class A Common Stock to which it is entitled, registered in such name or names as may be directed by it, and if such Warrant shall not have been exercised in full, a countersigned Warrant for the number of shares of Class A Common Stock as to which such Warrant shall not have been exercised. No Warrant shall be exercisable and the Company shall not be obligated to issue shares of Class A Common Stock upon exercise of a Warrant unless the Class A Common Stock issuable upon such Warrant exercise has been registered, qualified or deemed to be exempt from registration or qualification under the securities laws of the state of residence of the Registered Holder of the Warrants. Subject to Section 4.6 of this Agreement, a Registered Holder of Warrants may exercise its Warrants only for a whole number of shares of Class A Common Stock. If the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share, the Company shall round down to the nearest whole number of shares of Class A Common Stock to be issued to such holder.

3.3.3 <u>Valid Issuance</u>. All shares of Class A Common Stock issued upon the proper exercise of a Warrant in conformity with this Agreement shall be validly issued, fully paid and non-assessable.

3.3.4 <u>Date of Issuance</u>. Each person in whose name any book-entry position or certificate, as applicable, for shares of Class A Common Stock is issued shall for all purposes be deemed to have become the holder of record of such shares of Class A Common Stock on the date on which the Warrant was surrendered, irrespective of the date of delivery of such certificate, except that, if the date of such surrender is a date when the share transfer books of the Company of the Warrant Agent are closed, such person shall be deemed to have become the holder of such shares of Class A Common Stock at the close of business on the next succeeding date on which the share transfer books are open.

#### 4. ADJUSTMENTS.

#### 4.1 Stock Dividends.

4.1.1 Split-Ups. If after the date hereof, and subject to the provisions of Section 4.6 below, the number of outstanding shares of Class A Common Stock is increased by a stock dividend payable in shares of Class A Common Stock, or by a split-up of shares of Class A Common Stock or other similar event, then, on the effective date of such stock dividend, split-up or similar event, the Warrant Entitlement shall be increased in proportion to such increase in the outstanding shares of Class A Common Stock. A rights offering to holders of the Class A Common Stock entitling holders to purchase shares of Class A Common Stock at a price less than the "Fair Market Value" (as defined below) shall be deemed a stock dividend of a number of shares of Class A Common Stock equal to the product of (i) the number of shares of Class A Common Stock actually sold in such rights offering (or issuable under any other equity securities sold in such rights offering that are convertible into or exercisable for the Class A Common Stock) multiplied by (ii) one (1) minus the quotient of (x) the price per share of Class A Common Stock paid in such rights offering divided by (y) the Fair Market Value. For purposes of this subsection 4.1.1, (i) if the rights offering is for securities convertible into or exercisable for Class A Common Stock, in determining the price payable for Class A Common Stock, there shall be taken into account any consideration received for such rights, as well as any additional amount payable upon exercise or conversion and (ii) "Fair Market Value" means the volume weighted average price of the Class A Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the first date on which the shares of Class A Common Stock trade on the applicable exchange or in the applicable market, regular way, without the right to receive such rights.

- 4.1.2 <u>Dividends</u>. If the Company, at any time while the Warrants are outstanding and unexpired, shall pay a dividend or make a distribution in cash, securities or other assets to the holders of the Class A Common Stock on account of such shares of Class A Common Stock (or other shares of the Company's capital stock into which the Warrants are convertible), other than as described in <u>subsection 4.1.1</u> above, then the Warrant Price shall be decreased, effective immediately after the effective date of such dividend, by the amount of cash and/or the fair market value (as determined by the Company's board of directors (the "Board"), in good faith) of any securities or other assets paid on each share of Class A Common Stock in respect of such dividend.
- 4.2 <u>Aggregation of Shares</u>. If after the date hereof, and subject to the provisions of <u>Section 4.6</u> hereof, the number of outstanding shares of Class A Common Stock is decreased by a consolidation, combination, reverse stock split or reclassification of shares of Class A Common Stock or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the Warrant Entitlement shall be decreased in proportion to such decrease in outstanding shares of Class A Common Stock.
- 4.3 <u>Adjustments in Warrant Price and Warrant Entitlement</u>. Whenever the Warrant Entitlement is adjusted, as provided in <u>subsection 4.1.1</u> or <u>Section 4.2</u> above, the Warrant Price shall be adjusted (to the nearest cent) by multiplying such Warrant Price immediately prior to such adjustment by a fraction (x) the numerator of which shall be the Warrant Entitlement immediately prior to such adjustment, and (y) the denominator of which shall be the Warrant Entitlement immediately thereafter. Whenever the Warrant Price is adjusted as provided in <u>subsection 4.1.2</u> above, the Warrant Entitlement shall be adjusted by multiplying such Warrant Entitlement immediately prior to such adjustment by a fraction (x) the numerator of which shall be the Warrant Price immediately prior to such adjustment, and (y) the denominator of which shall be the Warrant Price immediately thereafter.
- 4.4 Replacement of Securities upon Reorganization, etc. In case of any reclassification or reorganization of the outstanding shares of Class A Common Stock (other than a change under <u>subsections 4.1.1</u> or <u>4.1.2</u> or <u>Section 4.2</u> hereof or that solely affects the par value of such shares of Class A Common Stock), or in the case of any merger or consolidation of the Company with or into another entity or conversion of the Company as another entity (other than a consolidation or merger in which the Company is the continuing corporation and that does not result in any reclassification or reorganization of the outstanding shares of Class A Common Stock), or in the case of any sale or conveyance to another entity of the assets or other property of the Company as an entirety or substantially as an entirety in connection with which the Company is dissolved, the holders of the Warrants shall thereafter have the right to purchase and receive, upon the basis and upon the terms and conditions specified in the Warrants and in lieu of the shares of Class A Common Stock of the Company immediately theretofore purchasable and receivable upon the exercise of the rights represented thereby, the kind and amount of shares of stock or other securities or property (including cash) receivable upon such reclassification, reorganization,

merger or consolidation, or upon a dissolution following any such sale or transfer, that the holder of the Warrants would have received if such holder had held a number of shares of Class A Common Stock equal to the aggregate then-applicable Warrant Entitlement of such Warrants immediately prior to such event (the "Alternative Issuance") and the successor or purchasing entity, as the case may be, shall execute an amendment hereto with the Warrant Agent providing for delivery of such Alternative Issuance; provided, however, that (i) if the holders of the Class A Common Stock were entitled to exercise a right of election as to the kind or amount of securities, cash or other assets receivable upon such consolidation or merger, then the kind and amount of securities, cash or other assets constituting the Alternative Issuance for which each Warrant shall become exercisable shall be deemed to be the weighted average of the kind and amount received per share by the holders of the Class A Common Stock in such consolidation or merger, and (ii) if a tender, exchange or redemption offer shall have been made to and accepted by the holders of the Class A Common Stock under circumstances in which, upon completion of such tender or exchange offer, the maker thereof, together with members of any group (within the meaning of Rule 13d-5(b)(1) under the Exchange Act (or any successor rule)) of which such maker is a part, and together with any affiliate or associate of such maker (within the meaning of Rule 12b-2 under the Exchange Act (or any successor rule)) and any members of any such group of which any such affiliate or associate is a part, own beneficially (within the meaning of Rule 13d-3 under the Exchange Act (or any successor rule)) more than 50% of the outstanding shares of Class A Common Stock, (x) the holder of a Warrant shall be entitled to receive as the Alternative Issuance, the highest amount of cash, securities or other property to which such holder would actually have been entitled as a stockholder if such Warrant holder had held a number of shares of Class A Common Stock equal to the then-applicable Warrant Entitlement prior to the expiration of such tender or exchange offer, accepted such offer and all of the Class A Common Stock held by such holder had been purchased pursuant to such tender or exchange offer, subject to adjustments (from and after the consummation of such tender or exchange offer) as nearly equivalent as possible to the adjustments provided for in this Section 4 and (y) if the issuer(s) of the securities constituting the Alternative Issuance is not the Company, then the Company and such issuer(s) shall take such action so as to ensure the availability of Section 3(a)(9) under the Securities Act of 1933, as amended (the "Securities Act") for any issuance of such securities upon exercise of the Warrants and the tacking of the "holding period" under Rule 144 under the Securities Act for the Warrants to such securities provided, further, that if less than 70% of the consideration receivable by the holders of the Class A Common Stock in the applicable event is payable in the form of common stock in the successor entity that is listed for trading on a national securities exchange or is quoted in an established over-the-counter market, or is to be so listed for trading or quoted immediately following such event, and if the Registered Holder properly exercises the Warrant within thirty (30) days following the public disclosure of the consummation of such applicable event by the Company pursuant to a Current Report on Form 8-K filed with the Commission, the Warrant Price shall be reduced by an amount (in dollars) equal to the difference of (i) the Warrant Price in effect prior to such reduction minus (ii) (A) the Per Share Consideration (as defined below) (but in no event less than zero) minus (B) the Black-Scholes Warrant Value (as defined below). The "Black-Scholes Warrant Value" means the value of a Warrant immediately prior to the consummation of the applicable event based on the Black-Scholes Warrant Model for a Capped American Call on Bloomberg Financial Markets ("Bloomberg"). For purposes of calculating such amount, (1) the price of each share of Class A Common Stock shall be the volume weighted average price of the Class A Common Stock as reported during the ten (10) trading day period ending on the trading

day prior to the effective date of the applicable event, (2) the assumed volatility shall be the 90 day volatility obtained from the HVT function on Bloomberg determined as of the trading day immediately prior to the day of the announcement of the applicable event, and (3) the assumed risk-free interest rate shall correspond to the U.S. Treasury rate for a period equal to the remaining term of the Warrant. "Per Share Consideration" means (i) if the consideration paid to holders of the Class A Common Stock consists exclusively of cash, the amount of such cash per share of Class A Common Stock, and (ii) in all other cases, the volume weighted average price of the Class A Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event. If any reclassification or reorganization also results in a change in shares of Class A Common Stock covered by subsection 4.1.1, then such adjustment shall be made pursuant tosubsection 4.1.1 or Sections 4.2, 4.3 and this Section 4.4. The provisions of this Section 4.4 shall similarly apply to successive reclassifications, reorganizations, mergers or consolidations, sales or other transfers. In no event will the Warrant Price be reduced to less than the par value per share issuable upon exercise of the Warrant.

- 4.5 Notices of Changes in Warrant. Upon every adjustment of the Warrant Price or the Warrant Entitlement under Section 4, the Company shall give written notice thereof to the Warrant Agent, which notice shall state the Warrant Price resulting from such adjustment and the increase or decrease, if any, in the Warrant Entitlement, setting forth in reasonable detail the method of calculation and the facts upon which such calculation is based. Subject to the procedures in Section 4.8 (as applicable), the Company shall be responsible for any adjustments made to the Warrant Price or Warrant Entitlement. Upon the occurrence of any event specified in Sections 4.1, 4.2, 4.3, 4.4 or 4.8, the Company shall give written notice of the occurrence of such event to each holder of a Warrant, at the last address set forth for such holder in the Warrant Register, of the record date or the effective date of the event. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such event.
- 4.6 No Fractional Shares. Notwithstanding any provision contained in this Agreement to the contrary, the Company shall not issue fractional shares of Class A Common Stock upon the exercise of Warrants. If, by reason of any adjustment made pursuant to this Section 4, the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share, the Company shall, upon such exercise, round down to the nearest whole number the number of shares of Class A Common Stock to be issued to such holder.
- 4.7 Form of Warrant. The form of Warrant need not be changed because of any adjustment pursuant to this Section 4, and Warrants issued after such adjustment may state the same Warrant Price and the same Warrant Entitlement as is stated in the Warrants initially issued pursuant to this Agreement; provided, however, that the Company may at any time in its sole discretion make any change in the form of Warrant that the Company may deem appropriate and that does not affect the substance thereof, and any Warrant thereafter issued or countersigned, whether in exchange or substitution for an outstanding Warrant or otherwise, may be in the form as so changed.

4.8 Other Events. In case any event shall occur affecting the Company as to which none of the provisions of preceding subsections of this Section 4 are strictly applicable, but which would require an adjustment to the terms of the Warrants in order to (i) avoid an adverse impact on the Warrants and (ii) effectuate the intent and purpose of this Section 4, then, in each such case, the Company shall appoint a firm of independent public accountants, investment banking or other appraisal firm of recognized national standing, which shall give its opinion as to whether or not any adjustment to the rights represented by the Warrants is necessary to effectuate the intent and purpose of this Section 4 and, if they determine that an adjustment is necessary, the terms of such adjustment. The Company shall adjust the terms of the Warrants in a manner that is consistent with any adjustment recommended in such opinion.

#### 5. TRANSFER AND EXCHANGE OF WARRANTS.

- 5.1 <u>Registration of Transfer</u>. The Warrant Agent shall register the transfer, from time to time, of any outstanding Warrant upon the Warrant Register, upon surrender of such Warrant for transfer, properly endorsed with signatures properly guaranteed and accompanied by appropriate instructions for transfer. Upon any such transfer, a new Warrant representing an equal aggregate number of Warrants shall be issued and the old Warrant shall be cancelled by the Warrant Agent. The Warrants so cancelled shall be delivered by the Warrant Agent to the Company from time to time upon request.
- 5.2 <u>Procedure for Surrender of Warrants</u>. Warrants may be surrendered to the Warrant Agent, together with a written request for exchange or transfer, and thereupon the Warrant Agent shall issue in exchange therefor one or more new Warrants as requested by the Registered Holder of the Warrants so surrendered, representing an equal aggregate number of Warrants; provided, however, that in the event that a Warrant surrendered for transfer bears a restrictive legend, the Warrant Agent shall not cancel such Warrant and issue new Warrants in exchange thereof until the Warrant Agent has received an opinion of counsel for the Company stating that such transfer may be made and indicating whether the new Warrants must also bear a restrictive legend.
- 5.3 <u>Fractional Warrants</u>. The Warrant Agent shall not be required to effect any registration of transfer or exchange which shall result in the issuance of a warrant certificate for a fraction of a warrant.
  - 5.4 Service Charges. No service charge shall be made for any exchange or registration of transfer of Warrants.
- 5.5 Warrant Execution and Countersignature. The Warrant Agent is hereby authorized to countersign and to deliver, in accordance with the terms of this Agreement, the Warrants required to be issued pursuant to the provisions of this Section 5, and the Company, whenever required by the Warrant Agent, shall supply the Warrant Agent with Warrants duly executed on behalf of the Company for such purpose.

#### 6. OTHER PROVISIONS RELATING TO RIGHTS OF HOLDERS OF WARRANTS.

6.1 No Rights as Stockholder. A Warrant does not entitle the Registered Holder thereof to any of the rights of a stockholder of the Company, including, without limitation, the right to receive dividends, or other distributions, exercise any preemptive rights to vote or to consent or to receive notice as stockholders in respect of the meetings of stockholders or the election of directors of the Company or any other matter.

6.2 <u>Lost, Stolen, Mutilated, or Destroyed Warrants</u> If any Warrant is lost, stolen, mutilated, or destroyed, the Company and the Warrant Agent may on such terms as to indemnity or otherwise as they may in their discretion impose (which shall, in the case of a mutilated Warrant, include the surrender thereof), issue a new Warrant of like denomination, tenor, and date as the Warrant so lost, stolen, mutilated, or destroyed. Any such new Warrant shall constitute a substitute contractual obligation of the Company, whether or not the allegedly lost, stolen, mutilated, or destroyed Warrant shall be at any time enforceable by anyone.

6.3 <u>Reservation of Common Stock</u>. The Company shall at all times reserve and keep available a number of its authorized but unissued shares of Class A Common Stock that shall be sufficient to permit the exercise in full of all outstanding Warrants issued pursuant to this Agreement.

6.4 <u>DTC Undertaking</u>. The Company shall use commercially reasonable efforts to exchange each holder's Warrants for book-entry interests in a global Warrant at The Depositary Trust Company with an unrestricted CUSIP no later than the date that is 370 days from such Warrants' issuance date (or such shorter period of time if permitted under the securities laws or if such Warrants are sold under Rule 144 or under a registration statement).

#### 7. CONCERNING THE WARRANT AGENT AND OTHER MATTERS.

7.1 <u>Payment of Taxes</u>. The Company shall from time to time promptly pay all taxes and charges that may be imposed upon the Company or the Warrant Agent in respect of the issuance or delivery of shares of Class A Common Stock upon the exercise of the Warrants, but the Company shall not be obligated to pay any transfer taxes in respect of the Warrants or such shares of Class A Common Stock.

#### 7.2 Resignation, Consolidation, or Merger of Warrant Agent.

7.2.1 Appointment of Successor Warrant Agent. The Warrant Agent, or any successor to it hereafter appointed, may resign its duties and be discharged from all further duties and liabilities hereunder after giving sixty (60) days' notice in writing to the Company. If the office of the Warrant Agent becomes vacant by resignation or incapacity to act or otherwise, the Company shall appoint in writing a successor Warrant Agent in place of the Warrant Agent. If the Company shall fail to make such appointment within a period of thirty (30) days after it has been notified in writing of such resignation or incapacity by the Warrant Agent or by the holder of a Warrant (who shall, with such notice, submit his Warrant for inspection by the Company), then the holder of any Warrant may apply to the Supreme Court of the State of New York for the County of New York for the appointment of a successor Warrant Agent at the Company's cost. Any successor Warrant Agent, whether appointed by the Company or by such court, shall be a corporation organized and existing under the laws of the State of New York, in good standing and having its principal office in the Borough of Manhattan, City and State of New York, and authorized under such laws to exercise corporate trust powers and subject to supervision or examination by federal or state authority. After appointment, any successor Warrant Agent shall be vested with all the authority, powers, rights, immunities, duties, and obligations of its predecessor Warrant Agent with like effect as if originally named as Warrant Agent hereunder, without any further act or deed; but if for any reason it becomes necessary or appropriate, the

predecessor Warrant Agent shall execute and deliver, at the expense of the Company, an instrument transferring to such successor Warrant Agent all the authority, powers, and rights of such predecessor Warrant Agent hereunder; and upon request of any successor Warrant Agent the Company shall make, execute, acknowledge, and deliver any and all instruments in writing for more fully and effectually vesting in and confirming to such successor Warrant Agent all such authority, powers, rights, immunities, duties, and obligations.

- 7.2.2 Notice of Successor Warrant Agent. In the event a successor Warrant Agent shall be appointed, the Company shall give notice thereof to the predecessor Warrant Agent and the Transfer Agent for the Class A Common Stock not later than the effective date of any such appointment.
- 7.2.3 Merger or Consolidation of Warrant Agent. Any corporation into which the Warrant Agent may be merged or with which it may be consolidated or any corporation resulting from any merger or consolidation to which the Warrant Agent shall be a party shall be the successor Warrant Agent under this Agreement without any further act.

#### 7.3 Fees and Expenses of Warrant Agent.

- 7.3.1 <u>Remuneration</u>. The Company agrees to pay the Warrant Agent reasonable remuneration for its services as such Warrant Agent hereunder and shall, pursuant to its obligations under this Agreement, reimburse the Warrant Agent upon demand for all expenditures that the Warrant Agent may reasonably incur in the execution of its duties hereunder.
- 7.3.2 <u>Further Assurances</u>. The Company agrees to perform, execute, acknowledge, and deliver or cause to be performed, executed, acknowledged, and delivered all such further and other acts, instruments, and assurances as may reasonably be required by the Warrant Agent for the carrying out or performing of the provisions of this Agreement.

#### 7.4 Liability of Warrant Agent.

- 7.4.1 Reliance on Company Statement. Whenever in the performance of its duties under this Agreement, the Warrant Agent shall deem it necessary or desirable that any fact or matter be proved or established by the Company prior to taking or suffering any action hereunder, such fact or matter (unless other evidence in respect thereof be herein specifically prescribed) may be deemed to be conclusively proved and established by a statement signed by the Chief Executive Officer, Chief Financial Officer, Secretary or Chairman of the Board and delivered to the Warrant Agent. The Warrant Agent may rely upon such statement for any action taken or suffered in good faith by it pursuant to the provisions of this Agreement.
- 7.4.2 <u>Indemnity</u>. The Warrant Agent shall be liable hereunder only for its own gross negligence, willful misconduct or bad faith. The Company agrees to indemnify the Warrant Agent and save it harmless against any and all liabilities, including judgments, costs and reasonable counsel fees, for anything done or omitted by the Warrant Agent in the execution of this Agreement, except as a result of the Warrant Agent's gross negligence, willful misconduct or bad faith.

7.4.3 Exclusions. The Warrant Agent shall have no responsibility with respect to the validity of this Agreement or with respect to the validity or execution of any Warrant (except its countersignature thereof). The Warrant Agent shall not be responsible for any breach by the Company of any covenant or condition contained in this Agreement or in any Warrant. The Warrant Agent shall not be responsible to make any adjustments required under the provisions of Section 4 hereof or responsible for the manner, method, or amount of any such adjustment or the ascertaining of the existence of facts that would require any such adjustment; nor shall it by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of any shares of Class A Common Stock to be issued pursuant to this Agreement or any Warrant or as to whether any shares of Class A Common Stock shall, when issued, be valid and fully paid and non-assessable.

7.5 Acceptance of Agency. The Warrant Agent hereby accepts the agency established by this Agreement and agrees to perform the same upon the terms and conditions herein set forth and among other things, shall account promptly to the Company with respect to Warrants exercised and concurrently account for, and pay to the Company, all monies received by the Warrant Agent for the purchase of shares of Class A Common Stock through the exercise of the Warrants.

7.6 Waiver. The Warrant Agent has no right of set-off or any other right, title, interest or claim of any kind ('Claim') in, or to any distribution of, the Trust Account (as defined in that certain Investment Management Trust Agreement, dated as of the date hereof, by and between the Company and the Warrant Agent as trustee thereunder) and hereby agrees not to seek recourse, reimbursement, payment or satisfaction for any Claim against the Trust Account for any reason whatsoever. The Warrant Agent hereby waives any and all Claims against the Trust Account and any and all rights to seek access to the Trust Account.

#### 8. MISCELLANEOUS PROVISIONS.

8.1 <u>Successors</u>. All the covenants and provisions of this Agreement by or for the benefit of the Company or the Warrant Agent shall bind and inure to the benefit of their respective successors and assigns.

8.2 <u>Notices</u>. Any notice, statement or demand authorized by this Agreement to be given or made by the Warrant Agent or by the holder of any Warrant to or on the Company shall be sufficiently given when so delivered if by hand or overnight delivery or if sent by certified mail or private courier service within five (5) days after deposit of such notice, postage prepaid, or by e-mail with delivery receipt obtained, addressed (until another address is filed in writing by the Company with the Warrant Agent), as follows:

ProFrac Holding Corp.
333 Shops Boulevard, Suite 301
Willow Park, Texas 76087
Email: []
Attention: []

Any notice, statement or demand authorized by this Agreement to be given or made by the holder of any Warrant or by the Company to or on the Warrant Agent shall be sufficiently given when so delivered if by hand or overnight delivery or if sent by certified mail or private courier service within five (5) days after deposit of such notice, postage prepaid, or by e-mail with delivery receipt obtained, addressed (until another address is filed in writing by the Warrant Agent with the Company), as follows:

Continental Stock Transfer & Trust Company 1 State Street, 30<sup>th</sup> Floor New York, NY 10004 Email: compliance@continentalstock.com Attention: Compliance Department

- 8.3 <u>Applicable Law.</u> The validity, interpretation, and performance of this Agreement and of the Warrants shall be governed in all respects by the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. The Company hereby agrees that any action, proceeding or claim against it arising out of or relating in any way to this Agreement shall be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. The Company hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum.
- 8.4 <u>Persons Having Rights under this Agreement.</u> Nothing in this Agreement shall be construed to confer upon, or give to, any person or corporation other than the parties hereto and the Registered Holders of the Warrants any right, remedy, or claim under or by reason of this Agreement or of any covenant, condition, stipulation, promise, or agreement hereof. All covenants, conditions, stipulations, promises, and agreements contained in this Agreement shall be for the sole and exclusive benefit of the parties hereto and their successors and assigns and of the Registered Holders of the Warrants.
- 8.5 Examination of the Warrant Agreement. A copy of this Agreement shall be available at all reasonable times at the office of the Warrant Agent in the Borough of Manhattan, City and State of New York, for inspection by the Registered Holder of any Warrant. The Warrant Agent may require any such holder to submit his Warrant for inspection by it.
- 8.6 <u>Counterparts</u>. This Agreement may be executed in any number of original or facsimile counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.
- 8.7 Effect of Headings. The section headings herein are for convenience only and are not part of this Agreement and shall not affect the interpretation thereof.

8.8 Amendments. This Agreement may be amended by the parties hereto without the consent of any Registered Holder (a) for the purpose of curing any ambiguity, or curing, correcting or supplementing any defective provision contained herein or adding or changing any other provisions with respect to matters or questions arising under this Agreement as the parties may deem necessary or desirable and that the parties deem shall not adversely affect the interest of the Registered Holders, and (b) to provide for the delivery of Alternative Issuance pursuant to Section 4.4. All other modifications or amendments, including any amendment to increase the Warrant Price or shorten the Exercise Period, shall require the vote or written consent of the Registered Holders of more than seventy-five percent (75%) of the then outstanding Warrants.

8.9 <u>Severability</u>. This Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible and be valid and enforceable.

Exhibit A Form of Warrant Certificate

Exhibit B Legend - Warrants

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

PROFRAC HOLDING CORP.	
By:	
Name:	
Title:	
CONTINENTAL STOCK TRANSFER &	
TRUST COMPANY, as Warrant Agent	
By:	
Name:	
Title:	

#### **EXHIBIT A**

[Form of Warrant Certificate]

[FACE]

Number Warrant Entitlement: [•]

# Warrants THIS WARRANT SHALL BE VOID IF NOT EXERCISED PRIOR TO THE EXPIRATION OF THE EXERCISE PERIOD PROVIDED FOR IN THE WARRANT AGREEMENT DESCRIBED BELOW

#### PROFRAC HOLDING CORP.

Incorporated Under the Laws of the State of Delaware

#### **Warrant Certificate**

This Warrant Certificate certifies that, or registered assigns, is the registered holder of warrant(s) evidenced hereby (the Warrants" and each, a "Warrant") to purchase shares of Class A common stock, par value \$0.01 per share ('Class A Common Stock'), of ProFrac Holding Corp., a Delaware corporation (the "Company"). Each Warrant entitles the holder, upon exercise during the period set forth in the Warrant Agreement referred to below, to receive from the Company that number of fully paid and non-assessable shares of Class A Common Stock equal to the Warrant Entitlement at a price (the "Warrant Price") as determined pursuant to the Warrant Agreement and through "cashless" exercise provisions set forth in Section 3.3.1 thereof, upon surrender of this Warrant Certificate at the office or agency of the Warrant Agent referred to below, subject to the conditions set forth herein and in the Warrant Agreement. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

The initial Warrant Price per share of Class A Common Stock for any Warrant is equal to \$477.89 per share; provided, however, that a Warrant may not be exercised for a fractional share. The Warrant Price and the Warrant Entitlement are subject to adjustment upon the occurrence of certain events set forth in the Warrant Agreement.

Subject to the conditions set forth in the Warrant Agreement, the Warrants may be exercised only during the Exercise Period and to the extent not exercised by the end of such Exercise Period, such Warrants shall become void.

Reference is hereby made to the further provisions of this Warrant Certificate set forth on the reverse hereof and such further provisions shall for all purposes have the same effect as though fully set forth at this place.

This Warrant Certificate shall not be valid unless countersigned by the Warrant Agent, as such term is used in the Warrant Agreement.

This Warrant Certificate shall be governed by and construed in accordance with the internal laws of the State of New York, without regard to conflicts of laws principles thereof.

PROFRAC HOLDING CORP.		
By: Name: Title:		
CONTINENTAL STOCK TRANSFER & TRUST COMPANY, as Warrant Agent		
By:		
Name:		
Title:		

#### [Form of Warrant Certificate] [Reverse]

The Warrants evidenced by this Warrant Certificate are part of a duly authorized issue of Warrants entitling the holder on exercise to receive shares of Class A Common Stock and are issued or to be issued pursuant to a Warrant Agreement dated as of [\_\_\_], 2022 (the "Warrant Agreement"), duly executed and delivered by the Company to Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "Warrant Agent"), which Warrant Agreement is hereby incorporated by reference in and made a part of this instrument and is hereby referred to for a description of the rights, limitation of rights, obligations, duties and immunities thereunder of the Warrant Agent, the Company and the holders (the words "holders" or "holder" meaning the Registered Holders or Registered Holder) of the Warrants. A copy of the Warrant Agreement may be obtained by the holder hereof upon written request to the Company. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Warrants may be exercised at any time during the Exercise Period set forth in the Warrant Agreement. The holder of Warrants evidenced by this Warrant Certificate may exercise them by surrendering this Warrant Certificate, with the form of election to purchase set forth hereon properly completed and executed, at the principal corporate trust office of the Warrant Agent through the "cashless" exercise provisions as provided for in the Warrant Agreement. In the event that upon any exercise of Warrants evidenced hereby the number of Warrants exercised shall be less than the total number of Warrants evidenced hereby, there shall be issued to the holder hereof or his, her or its assignee, a new Warrant Certificate evidencing the number of Warrants not exercised

The Warrant Agreement provides that upon the occurrence of certain events the number of shares of Class A Common Stock issuable upon exercise of the Warrants set forth on the face hereof may, subject to certain conditions, be adjusted. If, upon exercise of a Warrant, the holder thereof would be entitled to receive a fractional interest in a share of Class A Common Stock, the Company shall, upon exercise, round down to the nearest whole number of shares of Class A Common Stock to be issued to the holder of the Warrant.

Warrant Certificates, when surrendered at the principal corporate trust office of the Warrant Agent by the Registered Holder thereof in person or by legal representative or attorney duly authorized in writing, may be exchanged, in the manner and subject to the limitations provided in the Warrant Agreement, but without payment of any service charge, for another Warrant Certificate or Warrant Certificates of like tenor evidencing in the aggregate a like number of Warrants.

Upon due presentation for registration of transfer of this Warrant Certificate at the office of the Warrant Agent a new Warrant Certificate or Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee(s) in exchange for this Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any tax or other governmental charge imposed in connection therewith.

The Company and the Warrant Agent may deem and treat the Registered Holder(s) hereof as the absolute owner(s) of this Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone), for the purpose of any exercise hereof, of any distribution to the holder(s) hereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary. Neither the Warrants nor this Warrant Certificate entitles any holder hereof to any rights of a stockholder of the Company.

## Election to Purchase (To Be Executed Upon Exercise of Warrant)

of ProFrac Holding Corp. (the "Company") in accorda Common Stock be registered in the name of delivered to whose address is Common Stock purchasable hereunder, the undersigned	nce with the terms hereof. The unde , whose address is . If said number of shares of Class A d requests that a new Warrant Certifi	Varrant Certificate, to receive shares of Class A Common Stock resigned requests that a certificate for such shares of Class A and that such shares of Class A Common Stock be a Common Stock is less than all of the shares of Class A cate representing the remaining balance of such shares of		
Class A Common Stock be registered in the name of	, whose address is	and that such Warrant Certificate be delivered to		
, whose address is	<u>.</u>			
The Warrant may only be exercised on a "cashless" basis pursuant to Section 3.3.1 of the Warrant Agreement. Accordingly, (i) the number of shares of Class A Common Stock that this Warrant is exercisable for shall be determined in accordance with Section 3.3.1 of the Warrant Agreement and (ii) the holder hereof shall complete the following: The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, through the cashless exercise provisions of the Warrant Agreement, to receive shares of Class A Common Stock. If said number of shares is less than all of the shares of Class A Common Stock purchasable hereunder (after giving effect to the cashless exercise), the undersigned requests that a new Warrant Certificate representing the remaining balance of such shares of Class A Common Stock be registered in the name of, whose address is and that such Warrant Certificate be delivered to, whose address is				
[Signature Page Follows]				

Date:, 20	
	(Signature)
	(4.11)
	(Address)
	(Tax Identification Number)
Signature Guaranteed:	

THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO SEC RULE 17Ad-15 (OR ANY SUCCESSOR RULE)).

## EXHIBIT B

## LEGEND

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS, AND MAY NOT BE OFFERED, SOLD, TRANSFERRED OR OTHERWISE DISPOSED OF UNLESS REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND ANY APPLICABLE STATE SECURITIES LAWS OR AN EXEMPTION FROM REGISTRATION IS AVAILABLE."

#### FORM OF AMENDED AND RESTATED WARRANT AGREEMENT

between

#### PROFRAC HOLDING CORP.

and

# CONTINENTAL STOCK TRANSFER & TRUST COMPANY

THIS AMENDED AND RESTATED WARRANT AGREEMENT (this "Agreement"), dated as of [\_\_], 2022, is by and between ProFrac Holding Corp., a Delaware corporation (the "Company"), and Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "Warrant Agent"), also referred to herein as the "Transfer Agent").

WHEREAS, U.S. Well Services, Inc., a Delaware corporation ("USWS"), and Transfer Agent are parties to that certain Warrant Agreement, dated as of March 9, 2017 (the "Existing Warrant Agreement");

WHEREAS, on June 21, 2022 that certain Agreement and Plan of Merger ("Merger Agreement") was entered into by and among the Company, USWS, and Thunderclap Merger Sub I, Inc., a Delaware corporation and an indirect subsidiary of Company ("Merger Sub");

WHEREAS, the Merger Agreement provides for, among other things, the merger of Merger Sub with and into USWS, with USWS surviving the merger as the surviving corporation and an indirect subsidiary of the Company (the "Merger");

WHEREAS, the Company and Transfer Agent wish to amend the Existing Warrant Agreement and each warrant issued pursuant to the Existing Warrant Agreement and outstanding as of the effective time of the Merger (the "Existing Warrants") such that as of the effective time of the Merger, the holders of the Existing Warrants ("Existing Warrant Holders") shall have the right to purchase the number of shares of Class A Common Stock of the Company, par value \$.01 per share equal to the product of (i) the number of shares of Series A Common Stock of USWS, par value \$0.0001 per share (the "USWS Common Stock"), underlying the Existing Warrants and (ii) 0.3366 (as adjusted for the 1-for-6 reverse split of USWS Common Stock effected by USWS on August 4, 2022);

WHEREAS, the Company and Transfer Agent desire to amend the Existing Warrant Agreement such that the Company assumes the Existing Warrants and they are otherwise amended as of the Effective Time in accordance with the terms hereof and of the Merger Agreement (each such amended Existing Warrant, a "Warrant");

WHEREAS, the Company desires the Warrant Agent to act on behalf of the Company, and the Warrant Agent is willing to so act, in connection with the issuance, registration, transfer, exchange, redemption and exercise of the Warrants;

WHEREAS, the Company desires to provide for the form and provisions of the Warrants, the terms upon which they shall be issued and exercised, and the respective rights, limitation of rights, and immunities of the Company, the Warrant Agent, and the holders of the Warrants; and

WHEREAS, all acts and things have been done and performed which are necessary to make the Warrants, when executed on behalf of the Company and countersigned by or on behalf of the Warrant Agent, as provided herein, the valid, binding and legal obligations of the Company, and to authorize the execution and delivery of this Agreement.

NOW, THEREFORE, in consideration of the mutual agreements herein contained, the parties hereto agree as follows:

1. Appointment of Warrant Agent. The Company hereby appoints the Warrant Agent to act as agent for the Company for the Warrants, and the Warrant Agent hereby accepts such appointment and agrees to perform the same in accordance with the terms and conditions set forth in this Agreement.

#### 2. Warrants.

- 2.1 Form of Warrant. Each Warrant shall be issued in registered form only.
- 2.2 Effect of Countersignature. If a physical certificate is issued, unless and until countersigned by the Warrant Agent pursuant to this Agreement, a Warrant shall be invalid and of no effect and may not be exercised by the holder thereof.

#### 2.3 Registration.

2.3.1 Warrant Register. The Warrant Agent shall maintain books (the "Warrant Register"), for the registration of original issuance and the registration of transfer of the Warrants. Upon the initial issuance of the Warrants, the Warrant Agent shall issue and register the Warrants in the names of the respective holders thereof in such denominations and otherwise in accordance with instructions delivered to the Warrant Agent by the Company. Ownership of beneficial interests in the Public Warrants shall be shown on, and the transfer of such ownership shall be effected through, records maintained by institutions that have accounts with the Depository Trust Company (the "Depositary") (such institution, with respect to a Warrant in its account, a "Participant").

If the Depositary subsequently ceases to make its book-entry settlement system available for the Public Warrants, the Company may instruct the Warrant Agent regarding making other arrangements for book-entry settlement. In the event that the Public Warrants are not eligible for, or it is no longer necessary to have the Public Warrants available in, book-entry form, the Warrant Agent shall provide written instructions to the Depositary to deliver to the Warrant Agent for cancellation each book-entry Public Warrant, and the Company shall instruct the Warrant Agent to deliver to the Depositary definitive certificates in physical form evidencing such Warrants which shall be in the form annexed hereto as Exhibit A.

Physical certificates, if issued, shall be signed by, or bear the facsimile signature of, the Chairman of the Board, Chief Executive Officer, Chief Financial Officer, Secretary or other principal officer of the Company. In the event the person whose facsimile signature has been placed upon any Warrant shall have ceased to serve in the capacity in which such person signed the Warrant before such Warrant is issued, it may be issued with the same effect as if he or she had not ceased to be such at the date of issuance.

- 2.3.2 Registered Holder. Prior to due presentment for registration of transfer of any Warrant, the Company and the Warrant Agent may deem and treat the person in whose name such Warrant is registered in the Warrant Register (the "Registered Holder") as the absolute owner of such Warrant and of each Warrant represented thereby (notwithstanding any notation of ownership or other writing on any physical certificate made by anyone other than the Company or the Warrant Agent), for the purpose of any exercise thereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.
- 2.4 <u>Detachability of Warrants</u>. The Common Stock and Public Warrants comprising the Units shall begin separate trading on the 52nd day following the date of the Prospectus or, if such 52nd day is not on a day, other than a Saturday, Sunday or federal holiday, on which banks in New York City are generally open for normal business (a "*Business Day*"), then on the immediately succeeding Business Day following such date, or earlier (the "*Detachment Date*") with the consent of CF & Co., as representative of the several underwriters, but in no event shall the Common Stock and the Public Warrants comprising the Units be separately traded until (A) the Company has filed a current report on Form 8-K with the Commission containing an audited balance sheet reflecting the receipt by the Company of the gross proceeds of the Offering, including the proceeds received by the Company from the exercise by the underwriters of their right to purchase additional Units in the Offering (the "*Over-allotment Option*"), if the Over-allotment Option is exercised prior to the filing of the Form 8-K, and (B) the Company issues a press release and files with the Commission a current report on Form8-K announcing when such separate trading shall begin.
- 2.5 Private Placement Warrants. The Private Placement Warrants shall be identical to the Public Warrants, except that so long as they are held by the Sponsor, CF & Co. or any of their Permitted Transferees (as defined below) the Private Placement Warrants: (i) may be exercised for cash or on a cashless basis, pursuant to subsection 3.3.1(c) hereof, (ii) may not be transferred, assigned or sold until thirty (30) days after the completion by the Company of an initial Business Combination (as defined below), and (iii) shall not be redeemable by the Company; provided, however, that in the case of (ii), the Private Placement Warrants and any shares of Common Stock held by the Sponsor, CF & Co. or any of their Permitted Transferees and issued upon exercise of the Private Placement Warrants may be transferred by the holders thereof:
- (a) as gift to such person's immediate family or to a trust, the beneficiary of which is a member of such person's immediate family, an affiliate of such person or to a charitable organization;
- (b) to the Company's officers or directors, any affiliate or family member of any of the Company's officers or directors, any affiliate of the Sponsor or to any member(s) of the Sponsor or any of their affiliates, or CF & Co.'s officers, directors and direct and indirect equityholders;

- (c) by virtue of the laws of descent and distribution upon death of such person;
- (d) pursuant to a qualified domestic relations order;
- (e) by virtue of the laws of the state of Delaware or the Sponsor's limited liability company agreement upon dissolution of the Sponsor;
- (f) through private sales or transfers made in connection with the consummation of the Company's initial Business Combination at prices no greater than the price at which the Warrants were originally purchased;
  - (g) in the event of the Company's liquidation prior to consummation of the Company's initial Business Combination; or
- (h) in the event that, subsequent to the consummation of the Company's initial Business Combination, the Company consummates a merger, capital stock exchange, reorganization or other similar transaction that results in all of the holders of the Company's equity securities issued in the Offering having the right to exchange their shares of Common Stock for cash, securities or other property;

<u>provided, however</u>, that, in the case of clauses (a) through (f), these transferees (the "*Permitted Transferees*") enter into a written agreement with the Company agreeing to be bound by the transfer restrictions in this Agreement.

#### 3. Terms and Exercise of Warrants.

- 3.1 Warrant Price. Each Warrant shall, when countersigned by the Warrant Agent, entitle the Registered Holder thereof, subject to the provisions of such Warrant and of this Agreement, to purchase from the Company the number of shares of Common Stock stated therein, at the price of \$717.47 per share, subject to the adjustments provided in Section 4 hereof and in the last sentence of this Section 3.1. The term "Warrant Price" as used in this Agreement shall mean the price per share at which shares of Common Stock may be purchased at the time a Warrant is exercised. The Company in its sole discretion may lower the Warrant Price at any time prior to the Expiration Date (as defined below) for a period of not less than twenty (20) Business Days, provided, that the Company shall provide at least twenty (20) days prior written notice of such reduction to Registered Holders of the Warrants and, provided further that any such reduction shall be identical among all of the Warrants.
- 3.2 <u>Duration of Warrants</u>. A Warrant may be exercised only during the period (the *Exercise Period*") commencing on the later of: (i) the date that is thirty (30) days after the first date on which the Company completes a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination, involving the Company and one or more businesses (a "*Business Combination*"), or (ii) the date that is twelve (12) months from the date of the closing of the Offering, and terminating at 5:00 p.m., New York City time on the earlier to occur of: (x) the date that is five (5) years after the date on which the Company completes its Business Combination, (y) the liquidation of the Company in accordance with the Company's amended and restated certificate of incorporation as amended from time to time, if the Company

fails to complete a Business Combination, or (z) other than with respect to the Private Placement Warrants, the Redemption Date (as defined below) as provided in Section 6.2 hereof (the "Expiration Date"); provided, however, that the exercise of any Warrant shall be subject to the satisfaction of any applicable conditions, as set forth in subsection 3.3.2 below with respect to an effective registration statement. Except with respect to the right to receive the Redemption Price (as defined below) (other than with respect to a Private Placement Warrant) in the event of a redemption (as set forth in Section 6 hereof), each Warrant (other than a Private Placement Warrant in the event of a redemption) not exercised on or before the Expiration Date shall become void, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at 5:00 p.m. New York City time on the Expiration Date. The Company in its sole discretion may extend the duration of the Warrants by delaying the Expiration Date; provided, that the Company shall provide at least twenty (20) days prior written notice of any such extension to Registered Holders of the Warrants and, provided further that any such extension shall be identical in duration among all the Warrants.

# 3.3 Exercise of Warrants.

- 3.3.1 Payment. Subject to the provisions of the Warrant and this Agreement, a Warrant, when countersigned by the Warrant Agent, may be exercised by the Registered Holder thereof by surrendering it, at the office of the Warrant Agent, or at the office of its successor as Warrant Agent, in the Borough of Manhattan, City and State of New York, with the subscription form, as set forth in the Warrant, duly executed, and by paying in full the Warrant Price for each full share of Common Stock as to which the Warrant is exercised and any and all applicable taxes due in connection with the exercise of the Warrant, the exchange of the Warrant for the shares of Common Stock and the issuance of such Common Stock, as follows:
  - (a) in lawful money of the United States, in good certified check or good bank draft payable to the Warrant Agent;
- (b) in the event of a redemption pursuant to Section 6 hereof in which the Company's board of directors (the 'Board') has elected to require all holders of the Warrants to exercise such Warrants on a "cashless basis," by surrendering the Warrants for that number of shares of Common Stock equal to the quotient obtained by dividing (x) the product of the number of shares of Common Stock underlying the Warrants, multiplied by the difference between the Warrant Price and the "Fair Market Value", as defined in this subsection 3.3.1(b) by (y) the Fair Market Value. Solely for purposes of this subsection 3.3.1(b) and Section 6.3, the "Fair Market Value" shall mean the average last sale price of the Common Stock for the ten (10) trading days ending on the third trading day prior to the date on which the notice of redemption is sent to the holders of the Warrants, pursuant to Section 6 hereof;
- (c) with respect to any Private Placement Warrant, so long as such Private Placement Warrant is held by the Sponsor, CF & Co. or a Permitted Transferee, by surrendering the Warrants for that number of shares of Common Stock equal to the quotient obtained by dividing (x) the product of the number of shares of Common Stock underlying the Warrants, multiplied by the difference between the Warrant Price and the "Fair Market Value", as defined in this subsection 3.3.1(c), by (y) the Fair Market Value. Solely for purposes of this subsection 3.3.1(c), the "Fair Market Value" shall mean the average last sale price of the Common Stock for the ten (10) trading days ending on the third trading day prior to the date on which notice of exercise of the Warrant Agent; or

#### (d) as provided in Section 7.4 hereof.

- 3.3.2 Issuance of Shares of Common Stock on Exercise. As soon as practicable after the exercise of any Warrant and the clearance of the funds in payment of the Warrant Price (if payment is pursuant to subsection 3.3.1(a)), the Company shall issue to the Registered Holder of such Warrant a book-entry position or certificate, as applicable, for the number of full shares of Common Stock to which he, she or it is entitled, registered in such name or names as may be directed by him, her or it, and if such Warrant shall not have been exercised in full, a new book-entry position or countersigned Warrant, as applicable, for the number of shares of Common Stock as to which such Warrant shall not have been exercised. Notwithstanding the foregoing, the Company shall not be obligated to deliver any shares of Common Stock pursuant to the exercise of a Warrant and shall have no obligation to settle such Warrant exercise unless a registration statement under the Securities Act with respect to the shares of Common Stock underlying the Public Warrants is then effective and a prospectus relating thereto is current, subject to the Company's satisfying its obligations under Section 7.4. No Warrant shall be exercisable and the Company shall not be obligated to issue shares of Common Stock upon exercise of a Warrant unless the Common Stock issuable upon such Warrant exercise has been registered, qualified or deemed to be exempt under the securities laws of the state of residence of the Registered Holder of the Warrants. In the event that the conditions in the two immediately preceding sentences are not satisfied with respect to a Warrant, the holder of such Warrant shall not be entitled to exercise such Warrant and such Warrant may have no value and expire worthless, in which case the purchaser of a Unit containing such Public Warrants shall have paid the full purchase price for the Unit solely for the shares of Common Stock underlying such Unit. Subject to Section 4.6 of this Agreement, a Registered Holder of Warrants may exercise its Warrants only for a whole number of shares of Common Stock (i.e., only an even number of Warrants may be exercised at any given time by a Registered Holder). In no event will the Company be required to net cash the Warrant exercise. The Company may require holders of Public Warrants to settle the Warrant on a "cashless basis" pursuant to Section 7.4. If, by reason of any exercise of warrants on a "cashless basis", the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share, the Company shall round down to the nearest whole number, the number of shares of Common Stock to be issued to such holder.
- 3.3.3 <u>Valid Issuance</u>. All shares of Common Stock issued upon the proper exercise of a Warrant in conformity with this Agreement shall be validly issued, fully paid and non-assessable.
- 3.3.4 <u>Date of Issuance</u>. Each person in whose name any book-entry position or certificate, as applicable, for shares of Common Stock is issued shall for all purposes be deemed to have become the holder of record of such shares of Common Stock on the date on which the Warrant, or book-entry position representing such Warrant, was surrendered and payment of the Warrant Price was made, irrespective of the date of delivery of such certificate in the case of a certificated Warrant, except that, if the date of such surrender and payment is a date when the share transfer books of the Company or book-entry system of the Warrant Agent are closed, such person shall be deemed to have become the holder of such shares of Common Stock at the close of business on the next succeeding date on which the share transfer books or book-entry system are open.

3.3.5 Maximum Percentage. A holder of a Warrant may notify the Company in writing in the event it elects to be subject to the provisions contained in this subsection 3.3.5; however, no holder of a Warrant shall be subject to this subsection 3.3.5 unless he, she or it makes such election. If the election is made by a holder, the Warrant Agent shall not effect the exercise of the holder's Warrant, and such holder shall not have the right to exercise such Warrant, to the extent that after giving effect to such exercise, such person (together with such person's affiliates), to the Warrant Agent's actual knowledge, would beneficially own in excess of 9.8% (the "Maximum Percentage") of the shares of Common Stock outstanding immediately after giving effect to such exercise. For purposes of the foregoing sentence, the aggregate number of shares of Common Stock beneficially owned by such person and its affiliates shall include the number of shares of Common Stock issuable upon exercise of the Warrant with respect to which the determination of such sentence is being made, but shall exclude shares of Common Stock that would be issuable upon (x) exercise of the remaining, unexercised portion of the Warrant beneficially owned by such person and its affiliates and (y) exercise or conversion of the unexercised or unconverted portion of any other securities of the Company beneficially owned by such person and its affiliates (including, without limitation, any convertible notes or convertible preferred stock or warrants) subject to a limitation on conversion or exercise analogous to the limitation contained herein. Except as set forth in the preceding sentence, for purposes of this paragraph, beneficial ownership shall be calculated in accordance with Section 13(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). For purposes of the Warrant, in determining the number of outstanding shares of Common Stock, the holder may rely on the number of outstanding shares of Common Stock as reflected in (1) the Company's most recent annual report on Form 10-K, quarterly report on Form 10-Q, current report on Form 8-K or other public filing with the Commission as the case may be, (2) a more recent public announcement by the Company or (3) any other notice by the Company or the Transfer Agent setting forth the number of shares of Common Stock outstanding. For any reason at any time, upon the written request of the holder of the Warrant, the Company shall, within two (2) Business Days, confirm orally and in writing to such holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of equity securities of the Company by the holder and its affiliates since the date as of which such number of outstanding shares of Common Stock was reported. By written notice to the Company, the holder of a Warrant may from time to time increase or decrease the Maximum Percentage applicable to such holder to any other percentage specified in such notice; provided, however, that any such increase shall not be effective until the sixty-first (61st) day after such notice is delivered to the Company.

# 4. Adjustments.

#### 4.1 Stock Dividends.

4.1.1 <u>Split-Ups</u>. If after the date hereof, and subject to the provisions of Section 4.6 below, the number of outstanding shares of Common Stock is increased by a stock dividend payable in shares of Common Stock, or by a split-up of shares of Common Stock or other similar event, then, on the effective date of such stock dividend, split-up or similar event, the

number of shares of Common Stock issuable on exercise of each Warrant shall be increased in proportion to such increase in the outstanding shares of Common Stock. A rights offering to holders of the Common Stock entitling holders to purchase shares of Common Stock at a price less than the "Fair Market Value" (as defined below) shall be deemed a stock dividend of a number of shares of Common Stock equal to the product of (i) the number of shares of Common Stock actually sold in such rights offering (or issuable under any other equity securities sold in such rights offering that are convertible into or exercisable for the Common Stock) multiplied by (ii) one (1) minus the quotient of (x) the price per share of Common Stock paid in such rights offering divided by (y) the Fair Market Value. For purposes of this subsection 4.1.1, (i) if the rights offering is for securities convertible into or exercisable for Common Stock, in determining the price payable for Common Stock, there shall be taken into account any consideration received for such rights, as well as any additional amount payable upon exercise or conversion and (ii) "Fair Market Value" means the volume weighted average price of the Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the first date on which the shares of Common Stock trade on the applicable exchange or in the applicable market, regular way, without the right to receive such rights.

4.1.2 Extraordinary Dividends. If the Company, at any time while the Warrants are outstanding and unexpired, shall pay a dividend or make a distribution in cash, securities or other assets to the holders of the Common Stock on account of such shares of Common Stock (or other shares of the Company's capital stock into which the Warrants are convertible), other than (a) as described in subsection 4.1.1 above, (b) Ordinary Cash Dividends (as defined below), (c) to satisfy the redemption rights of the holders of the Common Stock in connection with a proposed initial Business Combination, (d) to satisfy the redemption rights of the holders of Common Stock in connection with a stockholder vote to amend the Company's amended and restated certificate of incorporation to modify the substance or timing of the Company's obligation to redeem 100% of Common Stock if the Company does not complete the Business Combination within 24 months from the closing of the Offering, or (e) in connection with the redemption of public shares upon the failure of the Company to complete its initial Business Combination and any subsequent distribution of its assets upon its liquidation (any such non-excluded event being referred to herein as an 'Extraordinary Dividend'), then the Warrant Price shall be decreased, effective immediately after the effective date of such Extraordinary Dividend, by the amount of cash and/or the fair market value (as determined by the Board, in good faith) of any securities or other assets paid on each share of Common Stock in respect of such Extraordinary Dividend. For purposes of this subsection 4.1.2, "Ordinary Cash Dividends" means any cash dividend or cash distribution which, when combined on a per share basis, with the per share amounts of all other cash dividends and cash distributions paid on the Common Stock during the 365-day period ending on the date of declaration of such dividend or distribution (as adjusted to appropriately reflect any of the events referred to in other subsections of this Section 4 and excluding cash dividends or cash distributions that resulted in an adjustment to the Warrant Price or to the number of shares of Common Stock issuable on exercise of each Warrant) does not exceed \$0.50 (being 5% of the offering price of the Units in the Offering).

4.2 <u>Aggregation of Shares</u>. If after the date hereof, and subject to the provisions of <u>Section 4.6</u> hereof, the number of outstanding shares of Common Stock is decreased by a consolidation, combination, reverse stock split or reclassification of shares of Common Stock or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the number of shares of Common Stock issuable on exercise of each Warrant shall be decreased in proportion to such decrease in outstanding shares of Common Stock.

4.3 Adjustments in Exercise Price. Whenever the number of shares of Common Stock purchasable upon the exercise of the Warrants is adjusted, as provided in <u>subsection 4.1.1</u> or <u>Section 4.2</u> above, the Warrant Price shall be adjusted (to the nearest cent) by multiplying such Warrant Price immediately prior to such adjustment by a fraction (x) the numerator of which shall be the number of shares of Common Stock purchasable upon the exercise of the Warrants immediately prior to such adjustment, and (y) the denominator of which shall be the number of shares of Common Stock so purchasable immediately thereafter.

4.4 Replacement of Securities upon Reorganization, etc. In case of any reclassification or reorganization of the outstanding shares of Common Stock (other than a change under subsections 4.1.1 or 4.1.2 or Section 4.2 hereof or that solely affects the par value of such shares of Common Stock), or in the case of any merger or consolidation of the Company with or into another entity or conversion of the Company as another entity (other than a consolidation or merger in which the Company is the continuing corporation and that does not result in any reclassification or reorganization of the outstanding shares of Common Stock), or in the case of any sale or conveyance to another entity of the assets or other property of the Company as an entirety or substantially as an entirety in connection with which the Company is dissolved, the holders of the Warrants shall thereafter have the right to purchase and receive, upon the basis and upon the terms and conditions specified in the Warrants and in lieu of the shares of Common Stock of the Company immediately theretofore purchasable and receivable upon the exercise of the rights represented thereby, the kind and amount of shares of stock or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, that the holder of the Warrants would have received if such holder had exercised his, her or its Warrant(s) immediately prior to such event (the "Alternative Issuance") and the successor or purchasing entity, as the case may be, shall execute an amendment hereto with the Warrant Agent providing for delivery of such Alternative Issuance; provided, however, that (i) if the holders of the Common Stock were entitled to exercise a right of election as to the kind or amount of securities, cash or other assets receivable upon such consolidation or merger, then the kind and amount of securities, cash or other assets constituting the Alternative Issuance for which each Warrant shall become exercisable shall be deemed to be the weighted average of the kind and amount received per share by the holders of the Common Stock in such consolidation or merger that affirmatively make such election, and (ii) if a tender, exchange or redemption offer shall have been made to and accepted by the holders of the Common Stock (other than a tender, exchange or redemption offer made by the Company in connection with redemption rights held by stockholders of the Company as provided for in the Company's amended and restated certificate of incorporation or as a result of the repurchase of shares of Common Stock by the Company if a proposed initial Business Combination is presented to the stockholders of the Company for approval) under circumstances in which, upon completion of such tender or exchange offer, the maker thereof, together with members of any group (within the meaning of Rule 13d-5(b)(1) under the Exchange Act (or any successor rule)) of which such maker is a part, and together with any affiliate or associate of such maker (within the meaning of Rule 12b-2 under the Exchange Act (or any successor rule)) and any members of any such group of which any such affiliate or associate is a part, own beneficially (within the meaning of Rule 13d-3 under the

Exchange Act (or any successor rule)) more than 50% of the outstanding shares of Common Stock, the holder of a Warrant shall be entitled to receive as the Alternative Issuance, the highest amount of cash, securities or other property to which such holder would actually have been entitled as a stockholder if such Warrant holder had exercised the Warrant prior to the expiration of such tender or exchange offer, accepted such offer and all of the Common Stock held by such holder had been purchased pursuant to such tender or exchange offer, subject to adjustments (from and after the consummation of such tender or exchange offer) as nearly equivalent as possible to the adjustments provided for in this Section 4; provided, further, that if less than 70% of the consideration receivable by the holders of the Common Stock in the applicable event is payable in the form of common stock in the successor entity that is listed for trading on a national securities exchange or is quoted in an established over-the-counter market, or is to be so listed for trading or quoted immediately following such event, and if the Registered Holder properly exercises the Warrant within thirty (30) days following the public disclosure of the consummation of such applicable event by the Company pursuant to a Current Report on Form 8-K filed with the Commission, the Warrant Price shall be reduced by an amount (in dollars) equal to the difference of (i) the Warrant Price in effect prior to such reduction minus (ii) (A) the Per Share Consideration (as defined below) (but in no event less than zero) minus (B) the Black-Scholes Warrant Value (as defined below). The "Black-Scholes Warrant Value" means the value of a Warrant immediately prior to the consummation of the applicable event based on the Black-Scholes Warrant Model for a Capped American Call on Bloomberg Financial Markets ("Bloomberg"). For purposes of calculating such amount, (1) Section 6 of this Agreement shall be taken into account, (2) the price of each share of Common Stock shall be the volume weighted average price of the Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event, (3) the assumed volatility shall be the 90 day volatility obtained from the HVT function on Bloomberg determined as of the trading day immediately prior to the day of the announcement of the applicable event, and (4) the assumed risk-free interest rate shall correspond to the U.S. Treasury rate for a period equal to the remaining term of the Warrant. "Per Share Consideration" means (i) if the consideration paid to holders of the Common Stock consists exclusively of cash, the amount of such cash per share of Common Stock, and (ii) in all other cases, the volume weighted average price of the Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event. If any reclassification or reorganization also results in a change in shares of Common Stock covered by subsection 4.1.1, then such adjustment shall be made pursuant to subsection 4.1.1 or Sections 4.2, 4.3 and this Section 4.4. The provisions of this Section 4.4 shall similarly apply to successive reclassifications, reorganizations, mergers or consolidations, sales or other transfers. In no event will the Warrant Price be reduced to less than the par value per share issuable upon exercise of the Warrant.

4.5 Notices of Changes in Warrant. Upon every adjustment of the Warrant Price or the number of shares of Common Stock issuable upon exercise of a Warrant, the Company shall give written notice thereof to the Warrant Agent, which notice shall state the Warrant Price resulting from such adjustment and the increase or decrease, if any, in the number of shares of Common Stock purchasable at such price upon the exercise of a Warrant, setting forth in reasonable detail the method of calculation and the facts upon which such calculation is based. Upon the occurrence of any event specified in Sections 4.1, 4.2, 4.3 or 4.4, the Company shall give written notice of the occurrence of such event to each holder of a Warrant, at the last address set forth for such holder in the Warrant Register, of the record date or the effective date of the event. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such event.

- 4.6 No Fractional Shares. Notwithstanding any provision contained in this Agreement to the contrary, the Company shall not issue fractional shares of Common Stock upon the exercise of Warrants. If, by reason of any adjustment made pursuant to this Section 4, the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share, the Company shall, upon such exercise, round down to the nearest whole number of shares of Common Stock to be issued to such holder.
- 4.7 Form of Warrant. The form of Warrant need not be changed because of any adjustment pursuant to this Section 4, and Warrants issued after such adjustment may state the same Warrant Price and the same number of shares of Common Stock as is stated in the Warrants initially issued pursuant to this Agreement; provided, however, that the Company may at any time in its sole discretion make any change in the form of Warrant that the Company may deem appropriate and that does not affect the substance thereof, and any Warrant thereafter issued or countersigned, whether in exchange or substitution for an outstanding Warrant or otherwise, may be in the form as so changed.
- 4.8 Other Events. In case any event shall occur affecting the Company as to which none of the provisions of preceding subsections of this Section 4 are strictly applicable, but which would require an adjustment to the terms of the Warrants in order to (i) avoid an adverse impact on the Warrants and (ii) effectuate the intent and purpose of this Section 4, then, in each such case, the Company shall appoint a firm of independent public accountants, investment banking or other appraisal firm of recognized national standing, which shall give its opinion as to whether or not any adjustment to the rights represented by the Warrants is necessary to effectuate the intent and purpose of this Section 4 and, if they determine that an adjustment is necessary, the terms of such adjustment. The Company shall adjust the terms of the Warrants in a manner that is consistent with any adjustment recommended in such opinion.

# 5. Transfer and Exchange of Warrants.

- 5.1 <u>Registration of Transfer</u>. The Warrant Agent shall register the transfer, from time to time, of any outstanding Warrant upon the Warrant Register, upon surrender of such Warrant for transfer, properly endorsed with signatures properly guaranteed and accompanied by appropriate instructions for transfer. Upon any such transfer, a new Warrant representing an equal aggregate number of Warrants shall be issued and the old Warrant shall be cancelled by the Warrant Agent. The Warrants so cancelled shall be delivered by the Warrant Agent to the Company from time to time upon request.
- 5.2 <u>Procedure for Surrender of Warrants</u>. Warrants may be surrendered to the Warrant Agent, together with a written request for exchange or transfer, and thereupon the Warrant Agent shall issue in exchange therefor one or more new Warrants as requested by the Registered Holder of the Warrants so surrendered, representing an equal aggregate number of Warrants; <u>provided, however</u>, that in the event that a Warrant surrendered for transfer bears a restrictive legend (as in the case of the Private Placement Warrants), the Warrant Agent shall not cancel such Warrant and issue new Warrants in exchange thereof until the Warrant Agent has received an opinion of counsel for the Company stating that such transfer may be made and indicating whether the new Warrants must also bear a restrictive legend.

- 5.3 <u>Fractional Warrants</u>. The Warrant Agent shall not be required to effect any registration of transfer or exchange which shall result in the issuance of a warrant certificate or book-entry position for a fraction of a warrant.
  - 5.4 Service Charges. No service charge shall be made for any exchange or registration of transfer of Warrants.
- 5.5 Warrant Execution and Countersignature. The Warrant Agent is hereby authorized to countersign and to deliver, in accordance with the terms of this Agreement, the Warrants required to be issued pursuant to the provisions of this Section 5, and the Company, whenever required by the Warrant Agent, shall supply the Warrant Agent with Warrants duly executed on behalf of the Company for such purpose.
- 5.6 <u>Transfer of Warrants</u>. Prior to the Detachment Date, the Public Warrants may be transferred or exchanged only together with the Unit in which such Warrant is included, and only for the purpose of effecting, or in conjunction with, a transfer or exchange of such Unit. Furthermore, each transfer of a Unit on the register relating to such Units shall operate also to transfer the Warrants included in such Unit. Notwithstanding the foregoing, the provisions of this <u>Section 5.6</u> shall have no effect on any transfer of Warrants on and after the Detachment Date.

#### 6. Redemption.

- 6.1 Redemption. Subject to Section 6.4 hereof, not less than all of the outstanding Warrants may be redeemed, at the option of the Company, at any time while they are exercisable and prior to their expiration, at the office of the Warrant Agent, upon notice to the Registered Holders of the Warrants, as described in Section 6.2 below, at the price of \$0.01 per Warrant (the "Redemption Price"), provided that the last sales price of the Common Stock reported has been at least \$24.00 per share (subject to adjustment in compliance with Section 4 hereof), on each of twenty (20) trading days within the thirty (30) trading-day period ending on the third Business Day prior to the date on which notice of the redemption is given and provided that there is an effective registration statement covering the shares of Common Stock issuable upon exercise of the Warrants, and a current prospectus relating thereto, available throughout the 30-day Redemption Period (as defined in Section 6.2 below) or the Company has elected to require the exercise of the Warrants on a "cashless basis" pursuant to subsection 3.3.1.
- 6.2 <u>Date Fixed for, and Notice of, Redemption</u>. In the event that the Company elects to redeem all of the Warrants, the Company shall fix a date for the redemption (the "*Redemption Date*"). Notice of redemption shall be mailed by first class mail, postage prepaid, by the Company not less than thirty (30) days prior to the Redemption Date (the "*30-day Redemption Period*") to the Registered Holders of the Warrants to be redeemed at their last addresses as they shall appear on the registration books. Any notice mailed in the manner herein provided shall be conclusively presumed to have been duly given whether or not the Registered Holder received such notice.

- 6.3 Exercise After Notice of Redemption. The Warrants may be exercised, for cash (or on a "cashless basis" in accordance with subsection 3.3.1(b) of this Agreement) at any time after notice of redemption shall have been given by the Company pursuant to Section 6.2 hereof and prior to the Redemption Date. In the event that the Company determines to require all holders of Warrants to exercise their Warrants on a "cashless basis" pursuant to subsection 3.3.1, the notice of redemption shall contain the information necessary to calculate the number of shares of Common Stock to be received upon exercise of the Warrants, including the "Fair Market Value" (as such term is defined in subsection 3.3.1(b) hereof) in such case. On and after the Redemption Date, the record holder of the Warrants shall have no further rights except to receive, upon surrender of the Warrants, the Redemption Price.
- 6.4 Exclusion of Private Placement Warrants. The Company agrees that the redemption rights provided in this Section 6 shall not apply to the Private Placement Warrants if at the time of the redemption such Private Placement Warrants continue to be held by the Sponsor, CF & Co. or their Permitted Transferees. However, once such Private Placement Warrants are transferred (other than to Permitted Transferees under Section 2.5), the Company may redeem the Private Placement Warrants, provided that the criteria for redemption are met, including the opportunity of the holder of such Private Placement Warrants to exercise the Private Placement Warrants prior to redemption pursuant to Section 6.3. Private Placement Warrants that are transferred to persons other than Permitted Transferees shall upon such transfer cease to be Private Placement Warrants and shall become Public Warrants under this Agreement.

#### 7. Other Provisions Relating to Rights of Holders of Warrants.

- 7.1 No Rights as Stockholder. A Warrant does not entitle the Registered Holder thereof to any of the rights of a stockholder of the Company, including, without limitation, the right to receive dividends, or other distributions, exercise any preemptive rights to vote or to consent or to receive notice as stockholders in respect of the meetings of stockholders or the election of directors of the Company or any other matter.
- 7.2 <u>Lost, Stolen, Mutilated, or Destroyed Warrants</u> If any Warrant is lost, stolen, mutilated, or destroyed, the Company and the Warrant Agent may on such terms as to indemnity or otherwise as they may in their discretion impose (which shall, in the case of a mutilated Warrant, include the surrender thereof), issue a new Warrant of like denomination, tenor, and date as the Warrant so lost, stolen, mutilated, or destroyed. Any such new Warrant shall constitute a substitute contractual obligation of the Company, whether or not the allegedly lost, stolen, mutilated, or destroyed Warrant shall be at any time enforceable by anyone.
- 7.3 <u>Reservation of Common Stock</u>. The Company shall at all times reserve and keep available a number of its authorized but unissued shares of Common Stock that shall be sufficient to permit the exercise in full of all outstanding Warrants issued pursuant to this Agreement.
  - 7.4 Registration of Common Stock; Cashless Exercise at Company's Option

7.4.1 Registration of the Common Stock. The Company agrees that as soon as practicable, but in no event later than fifteen (15) Business Days after the closing of its initial Business Combination, it shall use its best efforts to file with the Commission a registration statement for the registration, under the Securities Act, of the shares of Common Stock issuable upon exercise of the Warrants. The Company shall use its best efforts to cause the same to become effective and to maintain the effectiveness of such registration statement, and a current prospectus relating thereto, until the expiration of the Warrants in accordance with the provisions of this Agreement. If any such registration statement has not been declared effective by the 60th Business Day following the closing of the Business Combination, holders of the Warrants shall have the right, during the period beginning on the 61st Business Day after the closing of the Business Combination and ending upon such registration statement being declared effective by the Commission, and during any other period when the Company shall fail to have maintained an effective registration statement covering the shares of Common Stock issuable upon exercise of the Warrants, to exercise such Warrants on a "cashless basis," by exchanging the Warrants (in accordance with Section 3(a)(9) of the Securities Act (or any successor rule) or another exemption) for that number of shares of Common Stock equal to the quotient obtained by dividing (x) the product of the number of shares of Common Stock underlying the Warrants, multiplied by the difference between the Warrant Price and the "Fair Market Value" (as defined below) by (y) the Fair Market Value. Solely for purposes of this subsection 7.4.1, "Fair Market Value" shall mean the volume weighted average price of the Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the date that notice of exercise is received by the Warrant Agent from the holder of such Warrants or its securities broker or intermediary. The date that notice of cashless exercise is received by the Warrant Agent shall be conclusively determined by the Warrant Agent. In connection with the "cashless exercise" of a Public Warrant, the Company shall, upon request, provide the Warrant Agent with an opinion of counsel for the Company (which shall be an outside law firm with securities law experience) stating that (i) the exercise of the Warrants on a cashless basis in accordance with this subsection 7.4.1 is not required to be registered under the Securities Act and (ii) the shares of Common Stock issued upon such exercise shall be freely tradable under United States federal securities laws by anyone who is not an affiliate (as such term is defined in Rule 144 under the Securities Act (or any successor rule)) of the Company and, accordingly, shall not be required to bear a restrictive legend. Except as provided in subsection 7.4.2, for the avoidance of any doubt, unless and until all of the Warrants have been exercised, the Company shall continue to be obligated to comply with its registration obligations under the first three sentences of this subsection 7.4.1.

7.4.2 <u>Cashless Exercise at Company's Option</u> If the Common Stock is at the time of any exercise of a Warrant not listed on a national securities exchange such that it satisfies the definition of a "covered security" under Section 18(b)(1) of the Securities Act (or any successor rule), the Company may, at its option, (i) require holders of Public Warrants who exercise Public Warrants to exercise such Public Warrants on a "cashless basis" in accordance with Section 3(a)(9) of the Securities Act (or any successor rule) as described in <u>subsection 7.4.1</u> and (ii) in the event the Company so elects, the Company shall (x) not be required to file or maintain in effect a registration statement for the registration, under the Securities Act, of the Common Stock issuable upon exercise of the Warrants, notwithstanding anything in this Agreement to the contrary, and (y) use its best efforts to register or qualify for sale the Common Stock issuable upon exercise of the Public Warrant under the blue sky laws of the state of residence of the exercising Public Warrant holder to the extent an exemption is not available.

#### 8. Concerning the Warrant Agent and Other Matters.

8.1 <u>Payment of Taxes</u>. The Company shall from time to time promptly pay all taxes and charges that may be imposed upon the Company or the Warrant Agent in respect of the issuance or delivery of shares of Common Stock upon the exercise of the Warrants, but the Company shall not be obligated to pay any transfer taxes in respect of the Warrants or such shares of Common Stock.

#### 8.2 Resignation, Consolidation, or Merger of Warrant Agent.

- 8.2.1 Appointment of Successor Warrant Agent. The Warrant Agent, or any successor to it hereafter appointed, may resign its duties and be discharged from all further duties and liabilities hereunder after giving sixty (60) days' notice in writing to the Company. If the office of the Warrant Agent becomes vacant by resignation or incapacity to act or otherwise, the Company shall appoint in writing a successor Warrant Agent in place of the Warrant Agent. If the Company shall fail to make such appointment within a period of thirty (30) days after it has been notified in writing of such resignation or incapacity by the Warrant Agent or by the holder of a Warrant (who shall, with such notice, submit his Warrant for inspection by the Company), then the holder of any Warrant may apply to the Supreme Court of the State of New York for the County of New York for the appointment of a successor Warrant Agent at the Company's cost. Any successor Warrant Agent, whether appointed by the Company or by such court, shall be a corporation organized and existing under the laws of the State of New York, in good standing and having its principal office in the Borough of Manhattan, City and State of New York, and authorized under such laws to exercise corporate trust powers and subject to supervision or examination by federal or state authority. After appointment, any successor Warrant Agent shall be vested with all the authority, powers, rights, immunities, duties, and obligations of its predecessor Warrant Agent with like effect as if originally named as Warrant Agent hereunder, without any further act or deed; but if for any reason it becomes necessary or appropriate, the predecessor Warrant Agent all the authority, powers, and rights of such predecessor Warrant Agent hereunder; and upon request of any successor Warrant Agent all the authority, powers, and deliver any and all instruments in writing for more fully and effectually vesting in and confirming to such successor Warrant Agent all such authority, powers, rights, immuniti
- 8.2.2 Notice of Successor Warrant Agent. In the event a successor Warrant Agent shall be appointed, the Company shall give notice thereof to the predecessor Warrant Agent and the Transfer Agent for the Common Stock not later than the effective date of any such appointment.
- 8.2.3 Merger or Consolidation of Warrant Agent. Any corporation into which the Warrant Agent may be merged or with which it may be consolidated or any corporation resulting from any merger or consolidation to which the Warrant Agent shall be a party shall be the successor Warrant Agent under this Agreement without any further act.

#### 8.3 Fees and Expenses of Warrant Agent.

- 8.3.1 Remuneration. The Company agrees to pay the Warrant Agent reasonable remuneration for its services as such Warrant Agent hereunder and shall, pursuant to its obligations under this Agreement, reimburse the Warrant Agent upon demand for all expenditures that the Warrant Agent may reasonably incur in the execution of its duties hereunder.
- 8.3.2 <u>Further Assurances</u>. The Company agrees to perform, execute, acknowledge, and deliver or cause to be performed, executed, acknowledged, and delivered all such further and other acts, instruments, and assurances as may reasonably be required by the Warrant Agent for the carrying out or performing of the provisions of this Agreement.

# 8.4 Liability of Warrant Agent.

- 8.4.1 Reliance on Company Statement. Whenever in the performance of its duties under this Agreement, the Warrant Agent shall deem it necessary or desirable that any fact or matter be proved or established by the Company prior to taking or suffering any action hereunder, such fact or matter (unless other evidence in respect thereof be herein specifically prescribed) may be deemed to be conclusively proved and established by a statement signed by the Chief Executive Officer, Chief Financial Officer, Secretary or Chairman of the Board of the Company and delivered to the Warrant Agent. The Warrant Agent may rely upon such statement for any action taken or suffered in good faith by it pursuant to the provisions of this Agreement.
- 8.4.2 <u>Indemnity</u>. The Warrant Agent shall be liable hereunder only for its own gross negligence, willful misconduct or bad faith. The Company agrees to indemnify the Warrant Agent and save it harmless against any and all liabilities, including judgments, costs and reasonable counsel fees, for anything done or omitted by the Warrant Agent in the execution of this Agreement, except as a result of the Warrant Agent's gross negligence, willful misconduct or bad faith.
- 8.4.3 Exclusions. The Warrant Agent shall have no responsibility with respect to the validity of this Agreement or with respect to the validity or execution of any Warrant (except its countersignature thereof). The Warrant Agent shall not be responsible for any breach by the Company of any covenant or condition contained in this Agreement or in any Warrant. The Warrant Agent shall not be responsible to make any adjustments required under the provisions of Section 4 hereof or responsible for the manner, method, or amount of any such adjustment or the ascertaining of the existence of facts that would require any such adjustment; nor shall it by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of any shares of Common Stock to be issued pursuant to this Agreement or any Warrant or as to whether any shares of Common Stock shall, when issued, be valid and fully paid and non-assessable.
- 8.5 <u>Acceptance of Agency</u>. The Warrant Agent hereby accepts the agency established by this Agreement and agrees to perform the same upon the terms and conditions herein set forth and among other things, shall account promptly to the Company with respect to Warrants exercised and concurrently account for, and pay to the Company, all monies received by the Warrant Agent for the purchase of shares of Common Stock through the exercise of the Warrants.

8.6 Waiver. The Warrant Agent has no right of set-off or any other right, title, interest or claim of any kind ("Claim") in, or to any distribution of, the Trust Account (as defined in that certain Investment Management Trust Agreement, dated as of the date hereof, by and between the Company and the Warrant Agent as trustee thereunder) and hereby agrees not to seek recourse, reimbursement, payment or satisfaction for any Claim against the Trust Account for any reason whatsoever. The Warrant Agent hereby waives any and all Claims against the Trust Account and any and all rights to seek access to the Trust Account.

#### 9. Miscellaneous Provisions.

- 9.1 <u>Successors</u>. All the covenants and provisions of this Agreement by or for the benefit of the Company or the Warrant Agent shall bind and inure to the benefit of their respective successors and assigns.
- 9.2 Notices. Any notice, statement or demand authorized by this Agreement to be given or made by the Warrant Agent or by the holder of any Warrant to or on the Company shall be sufficiently given when so delivered if by hand or overnight delivery or if sent by certified mail or private courier service within five (5) days after deposit of such notice, postage prepaid, addressed (until another address is filed in writing by the Company with the Warrant Agent), as follows:

ProFrac Holding Corp.
333 Shops Boulevard, Suite 301
Willow Park, Texas 76087
Email: [\_\_]
Attention: [\_\_]

Any notice, statement or demand authorized by this Agreement to be given or made by the holder of any Warrant or by the Company to or on the Warrant Agent shall be sufficiently given when so delivered if by hand or overnight delivery or if sent by certified mail or private courier service within five (5) days after deposit of such notice, postage prepaid, addressed (until another address is filed in writing by the Warrant Agent with the Company), as follows:

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, NY 10004 Email: compliance@continentalstock.com Attention: Compliance Department

9.3 Applicable Law. The validity, interpretation, and performance of this Agreement and of the Warrants shall be governed in all respects by the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. The Company hereby agrees that any action, proceeding or claim against it arising out of or relating in any way to this Agreement shall be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. The Company hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum.

- 9.4 <u>Persons Having Rights under this Agreement.</u> Nothing in this Agreement shall be construed to confer upon, or give to, any person or corporation other than the parties hereto and the Registered Holders of the Warrants any right, remedy, or claim under or by reason of this Agreement or of any covenant, condition, stipulation, promise, or agreement hereof. All covenants, conditions, stipulations, promises, and agreements contained in this Agreement shall be for the sole and exclusive benefit of the parties hereto and their successors and assigns and of the Registered Holders of the Warrants.
- 9.5 Examination of the Warrant Agreement. A copy of this Agreement shall be available at all reasonable times at the office of the Warrant Agent in the Borough of Manhattan, City and State of New York, for inspection by the Registered Holder of any Warrant. The Warrant Agent may require any such holder to submit his Warrant for inspection by it.
- 9.6 <u>Counterparts</u>. This Agreement may be executed in any number of original or facsimile counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.
- 9.7 Effect of Headings. The section headings herein are for convenience only and are not part of this Agreement and shall not affect the interpretation thereof.
- 9.8 Amendments. This Agreement may be amended by the parties hereto without the consent of any Registered Holder (i) for the purpose of curing any ambiguity, or curing, correcting or supplementing any defective provision contained herein or adding or changing any other provisions with respect to matters or questions arising under this Agreement as the parties may deem necessary or desirable and that the parties deem shall not adversely affect the interest of the Registered Holders, and (ii) to provide for the delivery of Alternative Issuance pursuant to Section 4.4. All other modifications or amendments, including any amendment to increase the Warrant Price or shorten the Exercise Period and any amendment to the terms of only the Private Placement Warrants, shall require the vote or written consent of the Registered Holders of 65% of the then outstanding Public Warrants. Notwithstanding the foregoing, the Company may lower the Warrant Price or extend the duration of the Exercise Period pursuant to Sections 3.1 and 3.2, respectively, without the consent of the Registered Holders.
- 9.9 Severability. This Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible and be valid and enforceable.

Exhibit A Form of Warrant Certificate

Exhibit B Legend—Private Placement Warrants

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

PROFRAC HOLDING CORP.
By: Name: Title:
CONTINENTAL STOCK TRANSFER & TRUST COMPANY, as Warrant Agent
By: Name: Title:

[Signature Page to Warrant Agreement]

#### EXHIBIT A

[Form of Warrant Certificate] [FACE]

Number

# Warrants

# THIS WARRANT SHALL BE VOID IF NOT EXERCISED PRIOR TO THE EXPIRATION OF THE EXERCISE PERIOD PROVIDED FOR IN THE WARRANT AGREEMENT DESCRIBED BELOW PROFRAC HOLDING CORP.

Incorporated Under the Laws of the State of Delaware

CUSIP	[ ]

# Warrant Certificate

This Warrant Certificate certifies that, or registered assigns, is the registered holder of warrant(s) evidenced hereby (the 'Warrants'' and each, a "Warrant'') to purchase shares of Class A common stock, par value \$0.01 per share ('Common Stock''), of ProFrac Holding Corp., a Delaware corporation (the "Company"). Each Warrant entitles the holder, upon exercise during the period set forth in the Warrant Agreement referred to below, to receive from the Company that number of fully paid and non-assessable shares of Common Stock as set forth below, at the exercise price (the 'Exercise Price'') as determined pursuant to the Warrant Agreement, payable in lawful money (or through 'cashless exercise'' as provided for in the Warrant Agreement) of the United States of America upon surrender of this Warrant Certificate and payment of the Exercise Price at the office or agency of the Warrant Agent referred to below, subject to the conditions set forth herein and in the Warrant Agreement. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Each Warrant is initially exercisable for one-half of one fully paid and non-assessable share of Common Stock. The number of shares of Common Stock issuable upon exercise of the Warrants is subject to adjustment upon the occurrence of certain events set forth in the Warrant Agreement.

The initial Exercise Price per share of Common Stock for any Warrant is equal to \$717.47 per share; provided, however, that a Warrant may not be exercised for a fractional share, so that only an even number of Warrants may be exercised at a given time. The Exercise Price is subject to adjustment upon the occurrence of certain events set forth in the Warrant Agreement.

Subject to the conditions set forth in the Warrant Agreement, the Warrants may be exercised only during the Exercise Period and to the extent not exercised by the end of such Exercise Period, such Warrants shall become void.

Reference is hereby made to the further provisions of this Warrant Certificate set forth on the reverse hereof and such further provisions shall for all purposes have the same effect as though fully set forth at this place.

This Warrant Certificate shall not be valid unless countersigned by the Warrant Agent, as such term is used in the Warrant Agreement.

This Warrant Certificate shall be governed by and construed in accordance with the internal laws of the State of New York, without regard to conflicts of laws principles thereof.

PROFRAC HOLDING CORP.	
By: Name: Title:	
CONTINENTAL STOCK TRANSFER & TRUST COMPANY, as Warrant Agent	
By:	
Name:	
Title:	

#### [Form of Warrant Certificate]

#### [Reverse]

The Warrants evidenced by this Warrant Certificate are part of a duly authorized issue of Warrants entitling the holder on exercise to receive shares of Common Stock and are issued or to be issued pursuant to a Warrant Agreement dated as of [\_\_], 2022 (the "Warrant Agreement"), duly executed and delivered by the Company to Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "Warrant Agent"), which Warrant Agreement is hereby incorporated by reference in and made a part of this instrument and is hereby referred to for a description of the rights, obligations, duties and immunities thereunder of the Warrant Agent, the Company and the holders (the words "holders" or "holder" meaning the Registered Holders or Registered Holder) of the Warrants. A copy of the Warrant Agreement may be obtained by the holder hereof upon written request to the Company. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Warrants may be exercised at any time during the Exercise Period set forth in the Warrant Agreement. The holder of Warrants evidenced by this Warrant Certificate may exercise them by surrendering this Warrant Certificate, with the form of election to purchase set forth hereon properly completed and executed, together with payment of the Exercise Price as specified in the Warrant Agreement (or through "cashless exercise" as provided for in the Warrant Agreement) at the principal corporate trust office of the Warrant Agent. In the event that upon any exercise of Warrants evidenced hereby the number of Warrants exercised shall be less than the total number of Warrants evidenced hereby, there shall be issued to the holder hereof or his, her or its assignee, a new Warrant Certificate evidencing the number of Warrants not exercised.

Notwithstanding anything else in this Warrant Certificate or the Warrant Agreement, no Warrant may be exercised unless at the time of exercise (i) a registration statement covering the shares of Common Stock to be issued upon exercise is effective under the Securities Act and (ii) a prospectus thereunder relating to the shares of Common Stock is current, except through "cashless exercise" as provided for in the Warrant Agreement.

The Warrant Agreement provides that upon the occurrence of certain events the number of shares of Common Stock issuable upon exercise of the Warrants set forth on the face hereof may, subject to certain conditions, be adjusted. If, upon exercise of a Warrant, the holder thereof would be entitled to receive a fractional interest in a share of Common Stock, the Company shall, upon exercise, round down to the nearest whole number of shares of Common Stock to be issued to the holder of the Warrant.

Warrant Certificates, when surrendered at the principal corporate trust office of the Warrant Agent by the Registered Holder thereof in person or by legal representative or attorney duly authorized in writing, may be exchanged, in the manner and subject to the limitations provided in the Warrant Agreement, but without payment of any service charge, for another Warrant Certificate or Warrant Certificates of like tenor evidencing in the aggregate a like number of Warrants.

Upon due presentation for registration of transfer of this Warrant Certificate at the office of the Warrant Agent a new Warrant Certificate or Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee(s) in exchange for this Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any tax or other governmental charge imposed in connection therewith.

The Company and the Warrant Agent may deem and treat the Registered Holder(s) hereof as the absolute owner(s) of this Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone), for the purpose of any exercise hereof, of any distribution to the holder(s) hereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary. Neither the Warrants nor this Warrant Certificate entitles any holder hereof to any rights of a stockholder of the Company.

# Election to Purchase

# (To Be Executed Upon Exercise of Warrant)

herewith tenders payment for such shares of Common Stock to the order of ProFrac Holding Corp. (the "Company") in the amount of \$
in accordance with the terms hereof. The undersigned requests that a certificate for such shares of Common Stock be registered in the name of
, whose address is and that such shares of Common Stock be delivered to whose address is
. If said number of shares of Common Stock is less than all of the shares of Common Stock purchasable hereunder, the undersigned
requests that a new Warrant Certificate representing the remaining balance of such shares of Common Stock be registered in the name of
, whose address is and that such Warrant Certificate be delivered to, whose address is
·
In the event that the Warrant has been called for redemption by the Company pursuant to Section 6 of the Warrant Agreement and the Company has required cashless exercise pursuant to Section 6.3 of the Warrant Agreement, the number of shares of Common Stock that this Warrant is exercisable for shall be determined in accordance with subsection 3.3.1(b) and Section 6.3 of the Warrant Agreement.
In the event that the Warrant is a Private Placement Warrant that is to be exercised on a "cashless" basis pursuant to subsection 3.3.1(c) of the Warrant Agreement, the number of shares of Common Stock that this Warrant is exercisable for shall be determined in accordance with subsection 3.3.1(c) of the Warrant Agreement.
In the event that the Warrant is to be exercised on a "cashless" basis pursuant to Section 7.4 of the Warrant Agreement, the number of shares of Common Stock that this Warrant is exercisable for shall be determined in accordance with Section 7.4 of the Warrant Agreement.
In the event that the Warrant may be exercised, to the extent allowed by the Warrant Agreement, through cashless exercise (i) the number of shares of Common Stock that this Warrant is exercisable for would be determined in accordance with the relevant section of the Warrant Agreement which allows for such cashless exercise and (ii) the holder hereof shall complete the following: The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, through the cashless exercise provisions of the Warrant Agreement, to receive shares of Common Stock. If said number of shares is less than all of the shares of Common Stock purchasable hereunder (after giving effect to the cashless exercise), the undersigned requests that a new Warrant Certificate representing the remaining balance of such shares of Common Stock be registered in the name of, whose address is and that such Warrant Certificate be delivered to, whose address is
[Signature Page Follows]
D 4 20
Date:, 20
(Signature)

	(Address)	
	(Tax Identification Number)	
Signature Guaranteed:		

THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15 (OR ANY SUCCESSOR RULE)).

#### **EXHIBIT B**

# **LEGEND**

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS, AND MAY NOT BE OFFERED, SOLD, TRANSFERRED OR OTHERWISE DISPOSED OF UNLESS REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND ANY APPLICABLE STATE SECURITIES LAWS OR AN EXEMPTION FROM REGISTRATION IS AVAILABLE. IN ADDITION, SUBJECT TO ANY ADDITIONAL LIMITATIONS ON TRANSFER DESCRIBED IN THE LETTER AGREEMENT BY AND AMONG PROFRAC HOLDING CORP. (THE "COMPANY"), MATLIN & PARTNERS ACQUISITION SPONSOR LLC AND THE OTHER PARTIES THERETO, THE SECURITIES REPRESENTED BY THIS CERTIFICATE MAY NOT BE SOLD OR TRANSFERRED PRIOR TO THE DATE THAT IS THIRTY (30) DAYS AFTER THE DATE UPON WHICH THE COMPANY COMPLETES ITS INITIAL BUSINESS COMBINATION (AS DEFINED IN SECTION 3 OF THE WARRANT AGREEMENT REFERRED TO HEREIN) EXCEPT TO A PERMITTED TRANSFERE (AS DEFINED IN SECTION 2 OF THE WARRANT AGREEMENT) WHO AGREES IN WRITING WITH THE COMPANY TO BE SUBJECT TO SUCH TRANSFER PROVISIONS.

SECURITIES EVIDENCED BY THIS CERTIFICATE AND SHARES OF CLASS A COMMON STOCK OF THE COMPANY ISSUED UPON EXERCISE OF SUCH SECURITIES SHALL BE ENTITLED TO REGISTRATION RIGHTS UNDER A REGISTRATION RIGHTS AGREEMENT TO BE EXECUTED BY THE COMPANY."

NEITHER THIS SECURITY NOR THE SECURITIES FOR WHICH THIS SECURITY IS EXERCISABLE HAVE BEEN REGISTERED WITH THE SECURITIES AND EXCHANGE COMMISSION OR THE SECURITIES COMMISSION OF ANY STATE IN RELIANCE UPON AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), AND, ACCORDINGLY, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR PURSUANT TO AN AVAILABLE EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS. THIS SECURITY AND THE SECURITIES ISSUABLE UPON EXERCISE OF THIS SECURITY MAY BE PLEDGED IN CONNECTION WITH A BONA FIDE MARGIN ACCOUNT OR OTHER LOAN SECURED BY SUCH SECURITIES.

# FORM OF AMENDED AND RESTATED PLACEMENT AGENT COMMON STOCK PURCHASE WARRANT PROFRAC HOLDING CORP.

Warrant Shares:	Issue Date: [], 2022
Initial Exercise Date: [], 2022	
THIS AMENDED AND RESTATED PLACEMENT AGENT COMMON STOCK PURCHASE of that, for value received, or its assigns (the "Holder") is entitled, upon the terms and subject to the conditions hereinafter set forth, at any time on or after the date set forth above (the "Initial Exercise Date") and time) on September 11, 2025 (the "Termination Date") but not thereafter, to subscribe for and purchase from Placeroporation (the "Company"), up to shares (as subject to adjustment hereunder, the 'Warrant Shares') purchase price of one share of Common Stock under this Warrant shall be equal to the Exercise Price, as defined the terms of that certain Agreement and Plan of Merger ("Merger Agreement") entered by and among the Comporation ("USWS"), and Thunderclap Merger Sub I, Inc., a Delaware corporation.	he limitations on exercise and the on or prior to 5:00 p.m. (New York City ROFRAC HOLDING CORP., a Delaware of the Company's Common Stock. The I in Section 2(b) as amended pursuant to
This Warrant amends and restates the terms of the Warrant issued on March 11, 2022 (the 'Origina' engagement letter, dated as of March 8, 2022, by and between USWS and H.C. Wainwright & Co., LLC (the "Let Holder hereby agrees to surrender to the Company for cancellation the Original Warrant or at the request of the Coancellation in form and substance acceptable to the Company. Holder and Company hereby acknowledge and a Warrant, the Original Warrant shall be amended and restated and all of the Company's obligations under such we full without any further action on the part of the Company or Holder.	etter"). By accepting this Warrant, the Company to execute an instrument of the issuance of this

1

The following is a statement of the rights of the Holder and the conditions to which this Warrant is subject, and to which Holder, by acceptance of this Warrant, agrees:

Section 1. <u>Definitions</u>. Capitalized terms used and not otherwise defined herein shall have the meanings set forth in that certain Amended and Restated Securities Purchase Agreement (the "**Purchase Agreement**"), dated [\_\_], 2022, among the Company and the purchasers signatory thereto.

Section 2. Exercise.

a) Exercise of Warrant. Exercise of the purchase rights represented by this Warrant may be made, in whole or in part, at any time or times on or after the Initial Exercise Date and on or before the Termination Date by delivery to the Company of a duly executed facsimile copy or PDF copy submitted by e-mail (or e-mail attachment) of the Notice of Exercise in the form annexed hereto (the 'Notice of Exercise'). Within the earlier of (i) two (2) Trading Days and (ii) the number of Trading Days comprising the Standard Settlement Period (as defined in Section 2(d)(i) herein) following the date of exercise as aforesaid, the Holder shall deliver the aggregate Exercise Price for the Warrant Shares specified in the applicable Notice of Exercise by wire transfer or cashier's check drawn on a United States bank unless the cashless exercise procedure specified in Section 2(c) below is specified in the applicable Notice of Exercise; provided, however, in the event that the Holder has not delivered such aggregate Exercise Price within three (3) Trading Days following the date of such exercise as aforesaid, the Company shall not be obligated to deliver such Warrant Shares hereunder until such payment is made. No ink-original Notice of Exercise shall be required, nor shall any medallion guarantee (or other type of guarantee or notarization) of any Notice of Exercise be required. Notwithstanding anything herein to the contrary, the Holder shall not be required to physically surrender this Warrant to the Company until the Holder has purchased all of the Warrant Shares available hereunder and the Warrant has been exercised in full, in which case, the Holder shall surrender this Warrant to the Company for cancellation within three (3) Trading Days of the date on which the final Notice of Exercise is delivered to the Company. Partial exercises of this Warrant resulting in purchases of a portion of the total number of Warrant Shares available hereunder shall have the effect of lowering the outstanding number of Warrant Shares purchasable hereunder in an amount equal to the applicable number of Warrant Shares purchased. The Holder and the Company shall maintain records showing the number of Warrant Shares purchased and the date of such purchases. The Company shall deliver any objection to any Notice of Exercise within one (1) Trading Day of receipt of such notice. The Holder and any assignee, by acceptance of this Warrant, acknowledge and agree that, by reason of the provisions of this paragraph, following the purchase of a portion of the Warrant Shares hereunder, the number of Warrant Shares available for purchase hereunder at any given time may be less than the amount stated on the face hereof.

b) Exercise Price. The exercise price per share of Common Stock under this Warrant shall be \$39.28 subject to adjustment hereunder (the "Exercise Price").

c) <u>Cashless Exercise</u>. If at the time of exercise hereof there is no effective registration statement registering, or the prospectus contained therein is not available for the issuance of the Warrant Shares to the Holder, then this Warrant may also be exercised, in whole or in part, at such time by means of a "cashless exercise" in which the Holder shall be entitled to receive a number of Warrant Shares equal to the quotient obtained by dividing [(A-B) (X)] by (A), where:

- (A) = as applicable: (i) the VWAP on the Trading Day immediately preceding the date of the applicable Notice of Exercise if such Notice of Exercise is (1) both executed and delivered pursuant to Section 2(a) hereof on a day that is not a Trading Day or (2) both executed and delivered pursuant to Section 2(a) hereof on a Trading Day prior to the opening of "regular trading hours" (as defined in Rule 600(b) of Regulation NMS promulgated under the federal securities laws) on such Trading Day, (ii) at the option of the Holder, either (y) the VWAP on the Trading Day immediately preceding the date of the applicable Notice of Exercise or (z) the Bid Price of the Common Stock on the principal Trading Market as reported by Bloomberg L.P. ("Bloomberg") as of the time of the Holder's execution of the applicable Notice of Exercise is executed during "regular trading hours" on a Trading Day and is delivered within two (2) hours thereafter (including until two (2) hours after the close of "regular trading hours" on a Trading Day) pursuant to Section 2(a) hereof or (iii) the VWAP on the date of the applicable Notice of Exercise if the date of such Notice of Exercise is a Trading Day and such Notice of Exercise is both executed and delivered pursuant to Section 2(a) hereof after the close of "regular trading hours" on such Trading Day;
- (B) = the Exercise Price of this Warrant, as adjusted hereunder; and;
- (X) = the number of Warrant Shares that would be issuable upon exercise of this Warrant in accordance with the terms of this Warrant if such exercise were by means of a cash exercise rather than a cashless exercise.

"Bid Price" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed or quoted on a Trading Market, the bid price of the Common Stock for the time in question (or the nearest preceding date) on the Trading Market on which the Common Stock is then listed or quoted as reported by Bloomberg (based on a Trading Day from 9:30 a.m. (New York City time) to 4:02 p.m. (New York City time)), (b) if OTCQB or OTCQX is not a Trading Market, the volume weighted average price of the Common Stock for such date (or the nearest preceding date) on OTCQB or OTCQX as applicable, (c) if the Common Stock is not then listed or quoted for trading on OTCQB or OTCQX and if prices for the Common Stock are then reported on The Pink Open Market (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Common Stock so reported, or (d) in all other cases, the fair market value of a share of Common Stock as determined by an independent appraiser selected in good faith by the Holders of a majority in interest of the Securities then outstanding and reasonably acceptable to the Company, the fees and expenses of which shall be paid by the Company.

"YWAP" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed or quoted on a Trading Market, the daily volume weighted average price of the Common Stock for such date (or the nearest preceding date) on the Trading Market on which the Common Stock is then listed or quoted as reported by Bloomberg (based on a Trading Day from 9:30 a.m. (New York City time) to 4:02 p.m. (New York City time)), (b) if OTCQB or OTCQX is not a Trading Market, the volume weighted average price of the Common Stock for such date (or the nearest preceding date) on OTCQX as applicable, (c) if the Common Stock is not then listed or quoted for trading on OTCQB or OTCQX and if prices for the Common Stock are then reported on The Pink Open Market (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Common Stock so reported, or (d) in all other cases, the fair market value of a share of Common Stock as determined by an independent appraiser selected in good faith by the Holders of a majority in interest of the Securities then outstanding and reasonably acceptable to the Company, the fees and expenses of which shall be paid by the Company.

If Warrant Shares are issued in such a cashless exercise, the parties acknowledge and agree that in accordance with Section 3(a)(9) of the Securities Act, the Warrant Shares shall take on the characteristics of the Warrants being exercised, and the holding period of the Warrant Shares being issued may be tacked on to the holding period of this Warrant. The Company agrees not to take any position contrary to this Section 2(c).

#### d) Mechanics of Exercise.

i. Delivery of Warrant Shares Upon Exercise. The Company shall cause the Warrant Shares purchased hereunder to be transmitted by the Transfer Agent to the Holder by crediting the account of the Holder's or his designee's balance account with The Depository Trust Company through its Deposit or Withdrawal at Custodian system ("**DWAC**") if the Company is then a participant in such system and either (A) there is an effective registration statement permitting the issuance of the Warrant Shares to or resale of the Warrant Shares by the Holder or (B) the Warrant Shares are eligible for resale by the Holder without volume or manner-of-sale limitations pursuant to Rule 144 (assuming cashless exercise of the Warrants), and otherwise by physical delivery of a certificate, registered in the Company's share register in the name of the Holder or his designee, for the number of Warrant Shares to which the Holder is entitled pursuant to such exercise to the address specified by the Holder in the Notice of Exercise by the date that is the earliest of (i) two (2) Trading Days after the delivery to the Company of the Notice of Exercise, (ii) two (2) Trading Day after delivery of the aggregate Exercise Price to the Company and (iii) the number of Trading Days comprising the Standard Settlement Period after the delivery to the Company of the Notice of Exercise (such date, the "Warrant Share Delivery Date"). Upon delivery of the Notice of Exercise, the Holder shall be deemed for all corporate purposes to have become the holder of record of the Warrant Shares with respect to which this Warrant has been exercised, irrespective of the date of delivery of the Warrant Shares, provided that payment of the aggregate Exercise Price (other than in the case of a cashless exercise) is received by the Warrant Share Delivery Date. If the Company fails for any reason to deliver to the Holder the Warrant Shares subject to a Notice of Exercise by the Warrant Share Delivery Date, the Company shall pay to the Holder, in cash, as liquidated damages and not as a penalty, for each \$1,000 of Warrant Shares subject to such exercise (based on the VWAP of the Common Stock on the date of the applicable Notice of Exercise), \$10 per Trading Day (increasing to \$20 per Trading Day on the third Trading Day after the Warrant Share Delivery Date) for each Trading Day after such Warrant Share Delivery Date until such Warrant Shares are delivered or Holder rescinds such exercise. The Company agrees to maintain a transfer agent that is a participant in the FAST program so long as this Warrant remains outstanding and exercisable. As used herein, "Standard Settlement Period" means the standard settlement period, expressed in a number of Trading Days, on the Company's primary Trading Market with respect to the Common Stock as in effect on the date of delivery of the Notice of ii. <u>Delivery of New Warrants Upon Exercise</u>. If this Warrant shall have been exercised in part, the Company shall, at the request of a Holder and upon surrender of this Warrant certificate, at the time of delivery of the Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Warrant Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant.

iii. Rescission Rights. If the Company fails to cause the Transfer Agent to transmit to the Holder the Warrant Shares pursuant to Section 2(d)(i) by the Warrant Share Delivery Date, then the Holder will have the right to rescind such exercise.

iv. Compensation for Buy-In on Failure to Timely Deliver Warrant Shares Upon Exercise. In addition to any other rights available to the Holder, if the Company fails to cause the Transfer Agent to transmit to the Holder the Warrant Shares in accordance with the provisions of Section 2(d) (i) above pursuant to an exercise on or before the Warrant Share Delivery Date, and if after such date the Holder is required by his broker to purchase (in an open market transaction or otherwise) or the Holder's brokerage firm otherwise purchases, shares of Common Stock to deliver in satisfaction of a sale by the Holder of the Warrant Shares which the Holder anticipated receiving upon such exercise (a "Buy-In"), then the Company shall (A) pay in cash to the Holder the amount, if any, by which (x) the Holder's total purchase price (including brokerage commissions, if any) for the shares of Common Stock so purchased exceeds (y) the amount obtained by multiplying (1) the number of Warrant Shares that the Company was required to deliver to the Holder in connection with the exercise at issue times (2) the price at which the sell order giving rise to such purchase obligation was executed, and (B) at the option of the Holder, either reinstate the portion of the Warrant and equivalent number of Warrant Shares for which such exercise was not honored (in which case such exercise shall be deemed rescinded) or deliver to the Holder the number of shares of Common Stock that would have been issued had the Company timely complied with its exercise and delivery obligations hereunder. For example, if the Holder purchases Common Stock having a total purchase price of \$11,000 to cover a Buy-In with respect to an attempted exercise of shares of Common Stock with an aggregate sale price giving rise to such purchase obligation of \$10,000, under clause (A) of the immediately preceding sentence the Company shall be required to pay the Holder \$1,000. The Holder shall provide the Company written notice indicating the amounts payable to the Holder in respect of the Buy-In and, upon request of the Company, evidence of the amount of such loss. Nothing herein shall limit a Holder's right to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Company's failure to timely deliver shares of Common Stock upon exercise of the Warrant as required pursuant to the terms hereof. The obligation of the Company to pay compensation for Buy In under this Section 2(d)(iv) is subject to delivery by the Holder of the aggregate Exercise Price in accordance with the terms of Section 2(a) herein.

v. No Fractional Shares or Scrip. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant. As to any fraction of a share which the Holder would otherwise be entitled to purchase upon such exercise, the Company shall, at its election, either pay a cash adjustment in respect of such final fraction in an amount equal to such fraction multiplied by the Exercise Price or round up to the next whole share.

vi. Charges, Taxes and Expenses. Issuance of Warrant Shares shall be made without charge to the Holder for any issue or transfer tax or other incidental expense in respect of the issuance of such Warrant Shares, all of which taxes and expenses shall be paid by the Company, and such Warrant Shares shall be issued in the name of the Holder or in such name or names as may be directed by the Holder; provided, however, that in the event that Warrant Shares are to be issued in a name other than the name of the Holder, this Warrant when surrendered for exercise shall be accompanied by the Assignment Form attached hereto duly executed by the Holder and the Company may require, as a condition thereto, the payment of a sum sufficient to reimburse it for any transfer tax incidental thereto. The Company shall pay all Transfer Agent fees required for same-day processing of any Notice of Exercise and all fees to the Depository Trust Company (or another established clearing corporation performing similar functions) required for same-day electronic delivery of the Warrant Shares.

vii. <u>Closing of Books</u>. The Company will not close its stockholder books or records in any manner which prevents the timely exercise of this Warrant, pursuant to the terms hereof.

e) Holder's Exercise Limitations. The Company shall not effect any exercise of this Warrant, and a Holder shall not have the right to exercise any portion of this Warrant, pursuant to Section 2 or otherwise, to the extent that after giving effect to such issuance after exercise as set forth on the applicable Notice of Exercise, the Holder (together with the Holder's Affiliates, and any other Persons acting as a group together with the Holder or any of the Holder's Affiliates (such Persons, "Attribution Parties")), would beneficially own in excess of the Beneficial Ownership Limitation (as defined below). For purposes of the foregoing sentence, the number of shares of Common Stock beneficially owned by the Holder and his Affiliates and Attribution Parties shall include the number of shares of Common Stock issuable upon exercise of this Warrant with respect to which such determination is being made, but shall exclude the number of shares of Common Stock which would be issuable upon (i) exercise of the remaining, nonexercised portion of this Warrant beneficially owned by the Holder or any of his Affiliates or Attribution Parties and (ii) exercise or conversion of the unexercised or nonconverted portion of any other securities of the Company (including, without limitation, any other Common Stock Equivalents) subject to a limitation on conversion or exercise analogous to the limitation contained herein beneficially owned by the Holder or any of his Affiliates or Attribution Parties. Except as set forth in the preceding sentence, for purposes of this Section 2(e), beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder, it being acknowledged by the Holder that the Company is not representing to the Holder that such calculation is in compliance with Section 13(d) of the Exchange Act and the Holder is solely responsible for any schedules required to be filed in accordance therewith. To the extent that the limitation contained in

is exercisable shall be in the sole discretion of the Holder, and the submission of a Notice of Exercise shall be deemed to be the Holder's determination of whether this Warrant is exercisable (in relation to other securities owned by the Holder together with any Affiliates and Attribution Parties) and of which portion of this Warrant is exercisable, in each case subject to the Beneficial Ownership Limitation, and the Company shall have no obligation to verify or confirm the accuracy of such determination and shall have no liability for exercises of the Warrant that are not in compliance with the Beneficial Ownership Limitation. In addition, a determination as to any group status as contemplated above shall be determined in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder and the Company shall have no obligation to verify or confirm the accuracy of such determination and shall have no liability for exercises of the Warrant that are not in compliance with the Beneficial Ownership Limitation. For purposes of this Section 2(e), in determining the number of outstanding shares of Common Stock, a Holder may rely on the number of outstanding shares of Common Stock as reflected in (A) the Company's most recent periodic or annual report filed with the Commission, as the case may be, (B) a more recent public announcement by the Company or (C) a more recent written notice by the Company or the Transfer Agent setting forth the number of shares of Common Stock outstanding. Upon the written or oral request of a Holder, the Company shall within two (2) Trading Day confirm orally and in writing to the Holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of securities of the Company, including this Warrant, by the Holder or his Affiliates or Attribution Parties since the date as of which such number of outstanding shares of Common Stock was reported. The "Beneficial Ownership Limitation" shall be 4.99% of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock issuable upon exercise of this Warrant. The Holder, upon notice to the Company, may increase or decrease the Beneficial Ownership Limitation provisions of this Section 2(e), provided that the Beneficial Ownership Limitation in no event exceeds 9.99% of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock upon exercise of this Warrant held by the Holder and the provisions of this Section 2(e) shall continue to apply. Any increase in the Beneficial Ownership Limitation will not be effective until the 61st day after such notice is delivered to the Company. The provisions of this paragraph shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this Section 2(e) to correct this paragraph (or any portion hereof) which may be defective or inconsistent with the intended Beneficial Ownership Limitation herein contained or to make changes or supplements necessary or desirable to properly give effect to such limitation. The limitations contained in this paragraph shall apply to a successor holder of this Warrant.

#### Section 3. Certain Adjustments.

a) Stock Dividends and Splits. If the Company, at any time while this Warrant is outstanding: (i) pays a stock dividend or otherwise makes a distribution or distributions on shares of its Common Stock or any other equity or equity equivalent securities payable in shares of Common Stock (which, for avoidance of doubt, shall not include any shares of Common Stock issued by the Company upon exercise of this Warrant), (ii) subdivides outstanding shares of Common Stock into a larger number of shares, (iii) combines (including by way of reverse stock split) outstanding shares of Common Stock into a smaller number of shares, or (iv) issues by reclassification of shares of the Common Stock any shares of capital stock of the Company, then

in each case the Exercise Price shall be multiplied by a fraction of which the numerator shall be the number of shares of Common Stock (excluding treasury shares, if any) outstanding immediately before such event and of which the denominator shall be the number of shares of Common Stock outstanding immediately after such event, and the number of shares issuable upon exercise of this Warrant shall be proportionately adjusted such that the aggregate Exercise Price of this Warrant shall remain unchanged. Any adjustment made pursuant to this Section 3(a) shall become effective immediately after the record date for the determination of stockholders entitled to receive such dividend or distribution and shall become effective immediately after the effective date in the case of a subdivision, combination or re-classification.

b) Subsequent Rights Offerings. In addition to any adjustments pursuant to Section 3(a) above, if at any time the Company grants, issues or sells any Common Stock Equivalents or rights to purchase stock, warrants, securities or other property pro rata to the record holders of any class of shares of Common Stock (the "Purchase Rights"), then the Holder will be entitled to acquire, upon the terms applicable to such Purchase Rights, the aggregate Purchase Rights which the Holder could have acquired if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date on which a record is taken for the grant, issuance or sale of such Purchase Rights, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the grant, issue or sale of such Purchase Rights (provided, however, that to the extent that the Holder's right to participate in any such Purchase Right would result in the Holder exceeding the Beneficial Ownership Limitation, then the Holder shall not be entitled to participate in such Purchase Right to such extent (or beneficial ownership of such shares of Common Stock as a result of such Purchase Right to such extent) and such Purchase Right to such extent shall be held in abeyance for the Holder until such time, if ever, as his right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).

c) Pro Rata Distributions. During such time as this Warrant is outstanding, if the Company shall declare or make any dividend (other than cash) or other distribution of its assets (or rights to acquire its assets) to holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of stock or other securities, property or options by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "Distribution"), at any time after the issuance of this Warrant, then, in each such case, the Holder shall be entitled to participate in such Distribution to the same extent that the Holder would have participated therein if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date of which a record is taken for such Distribution, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the participation in such Distribution (provided, however, that to the extent that the Holder's right to participate in any such Distribution would result in the Holder exceeding the Beneficial Ownership Limitation, then the Holder shall not be entitled to participate in such Distribution to such extent (or in the beneficial ownership of any shares of Common Stock as a result of such Distribution to such extent) and the portion of such Distribution shall be held in abeyance for the benefit of the Holder until such time, if ever, as his right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).

d) Fundamental Transaction. If, at any time while this Warrant is outstanding, (i) the Company, directly or indirectly, in one or more related transactions effects any merger or consolidation of the Company with or into another Person, (ii) the Company (and all of its Subsidiaries, taken as a whole), directly or indirectly, effects any sale, lease, license, assignment, transfer, conveyance or other disposition of all or substantially all of its assets in one or a series of related transactions, (iii) any, direct or indirect, purchase offer, tender offer or exchange offer (whether by the Company or another Person) is completed pursuant to which holders of Common Stock are permitted to sell, tender or exchange their shares for other securities, cash or property and has been accepted by the holders of 50% or more of the outstanding Common Stock, (iv) the Company, directly or indirectly, in one or more related transactions effects any reclassification, reorganization or recapitalization of the Common Stock or any compulsory share exchange pursuant to which the Common Stock is effectively converted into or exchanged for other securities, cash or property, or (v) the Company, directly or indirectly, in one or more related transactions consummates a stock or share purchase agreement or other business combination (including, without limitation, a reorganization, recapitalization, spin-off, merger or scheme of arrangement) with another Person or group of Persons whereby such other Person or group acquires more than 50% of the outstanding shares of Common Stock (not including any shares of Common Stock held by the other Person or other Persons making or party to, or associated or affiliated with the other Persons making or party to, such stock or share purchase agreement or other business combination) (each a "Fundamental Transaction"), then, upon any subsequent exercise of this Warrant, the Holder shall have the right to receive, for each Warrant Share that would have been issuable upon such exercise immediately prior to the occurrence of such Fundamental Transaction, at the option of the Holder (without regard to any limitation in Section 2(e) on the exercise of this Warrant), the number of shares of Common Stock of the successor or acquiring corporation or of the Company, if it is the surviving corporation, and any additional consideration (the "Alternate Consideration") receivable as a result of such Fundamental Transaction by a holder of the number of shares of Common Stock for which this Warrant is exercisable immediately prior to such Fundamental Transaction (without regard to any limitation in Section 2(e) on the exercise of this Warrant). For purposes of any such exercise, the determination of the Exercise Price shall be appropriately adjusted to apply to such Alternate Consideration based on the amount of Alternate Consideration issuable in respect of one share of Common Stock in such Fundamental Transaction, and the Company shall apportion the Exercise Price among the Alternate Consideration in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Holder shall be given the same choice as to the Alternate Consideration he receives upon any exercise of this Warrant following such Fundamental Transaction. Notwithstanding anything to the contrary, in the event of a Fundamental Transaction, the Company or any Successor Entity (as defined below) shall, at the Holder's option, exercisable at any time concurrently with, or within 30 days after, the consummation of the Fundamental Transaction (or, if later, the date of the public announcement of the applicable Fundamental Transaction), purchase this Warrant from the Holder by paying to the Holder an amount of cash equal to the Black Scholes Value (as defined below) of the remaining unexercised portion of this Warrant on the date of the consummation of such Fundamental Transaction; provided, however, that if the Fundamental Transaction is not within the Company's control, including not approved by the Company's Board of Directors, Holder shall only be entitled to receive from the Company or any Successor Entity the same type or form of

consideration (and in the same proportion), at the Black Scholes Value of the unexercised portion of this Warrant, that is being offered and paid to the holders of Common Stock of the Company in connection with the Fundamental Transaction, whether that consideration be in the form of cash, stock or any combination thereof, or whether the holders of Common Stock are given the choice to receive from among alternative forms of consideration in connection with the Fundamental Transaction; provided, further, that if holders of Common Stock of the Company are not offered or paid any consideration in such Fundamental Transaction, such holders of Common Stock will be deemed to have received common stock of the Successor Entity (which Entity may be the Company following such Fundamental Transaction) in such Fundamental Transaction. "Black Scholes Value" means the value of this Warrant based on the Black Scholes Option Pricing Model obtained from the "OV" function on Bloomberg determined as of the day of consummation of the applicable Fundamental Transaction for pricing purposes and reflecting (A) a risk-free interest rate corresponding to the U.S. Treasury rate for a period equal to the time between the date of the public announcement of the applicable Fundamental Transaction and the Termination Date, (B) an expected volatility equal to the greater of 100% and the 100 day volatility obtained from the HVT function on Bloomberg (determined utilizing a 365-day annualization factor) as of the Trading Day immediately following the public announcement of the applicable Fundamental Transaction, (C) the underlying price per share used in such calculation shall be the greater of (i) the sum of the price per share being offered in cash, if any, plus the value of any non-cash consideration, if any, being offered in such Fundamental Transaction and (ii) the highest VWAP during the period beginning on the Trading Day immediately preceding the announcement of the applicable Fundamental Transaction (or the consummation of the applicable Fundamental Transaction, if earlier) and ending on the Trading Day of the Holder's request pursuant to this Section 3(d) and (D) a remaining option time equal to the time between the date of the public announcement of the applicable Fundamental Transaction and the Termination Date and (E) a zero cost of borrow. The payment of the Black Scholes Value will be made by wire transfer of immediately available funds (or such other consideration) within the later of (i) five (5) Trading Days of the Holder's election and (ii) the date of consummation of the Fundamental Transaction. The Company shall cause any successor entity in a Fundamental Transaction in which the Company is not the survivor (the "Successor Entity") to assume in writing all of the obligations of the Company under this Warrant and the other Transaction Documents in accordance with the provisions of this Section 3(d) pursuant to written agreements in form and substance reasonably satisfactory to the Holder and approved by the Holder (without unreasonable delay) prior to such Fundamental Transaction and shall, at the option of the Holder, deliver to the Holder in exchange for this Warrant a security of the Successor Entity evidenced by a written instrument substantially similar in form and substance to this Warrant which is exercisable for a corresponding number of shares of capital stock of such Successor Entity (or its parent entity) equivalent to the shares of Common Stock acquirable and receivable upon exercise of this Warrant (without regard to any limitations on the exercise of this Warrant) prior to such Fundamental Transaction, and with an exercise price which applies the exercise price hereunder to such shares of capital stock (but taking into account the relative value of the shares of Common Stock pursuant to such Fundamental Transaction and the value of such shares of capital stock, such number of shares of capital stock and such exercise price being for the purpose of protecting the economic value of this Warrant immediately prior to the consummation of such Fundamental Transaction), and which is reasonably satisfactory in form and substance to the Holder. Upon the occurrence of any such Fundamental Transaction, the Successor Entity shall succeed to, and be substituted for (so that from and after the date of such

Fundamental Transaction, the provisions of this Warrant and the other Transaction Documents referring to the "Company" shall refer instead to the Successor Entity), and may exercise every right and power of the Company and shall assume all of the obligations of the Company under this Warrant and the other Transaction Documents with the same effect as if such Successor Entity had been named as the Company herein.

e) <u>Calculations</u>. All calculations under this Section 3 shall be made to the nearest cent or the nearest 1/100th of a share, as the case may be. For purposes of this Section 3, the number of shares of Common Stock deemed to be issued and outstanding as of a given date shall be the sum of the number of shares of Common Stock (excluding treasury shares, if any) issued and outstanding.

## f) Notice to Holder.

i. <u>Adjustment to Exercise Price</u>. Whenever the Exercise Price is adjusted pursuant to any provision of this Section 3, the Company shall promptly deliver to the Holder by facsimile or email a notice setting forth the Exercise Price after such adjustment and any resulting adjustment to the number of Warrant Shares and setting forth a brief statement of the facts requiring such adjustment.

ii. Notice to Allow Exercise by Holder. If (A) the Company shall declare a dividend (or any other distribution in whatever form) on the Common Stock, (B) the Company shall declare a special nonrecurring cash dividend on or a redemption of the Common Stock, (C) the Company shall authorize the granting to all holders of the Common Stock rights or warrants to subscribe for or purchase any shares of capital stock of any class or of any rights, (D) the approval of any stockholders of the Company shall be required in connection with any reclassification of the Common Stock, any consolidation or merger to which the Company is a party, any sale or transfer of all or substantially all of the assets of the Company, or any compulsory share exchange whereby the Common Stock is converted into other securities, cash or property, or (E) the Company shall authorize the voluntary or involuntary dissolution, liquidation or winding up of the affairs of the Company, then, in each case, the Company shall cause to be delivered by facsimile or email to the Holder at his last facsimile number or email address as it shall appear upon the Warrant Register of the Company, at least 20 calendar days prior to the applicable record or effective date hereinafter specified, a notice stating (x) the date on which a record is to be taken for the purpose of such dividend, distribution, redemption, rights or warrants, or if a record is not to be taken, the date as of which the holders of the Common Stock of record to be entitled to such dividend, distributions, redemption, rights or warrants are to be determined or (y) the date on which such reclassification, consolidation, merger, sale, transfer or share exchange is expected to become effective or close, and the date as of which it is expected that holders of the Common Stock of record shall be entitled to exchange their shares of the Common Stock for securities, cash or other property deliverable upon such reclassification, consolidation, merger, sale, transfer or share exchange; provided that the failure to deliver such notice or any defect therein or in the delivery thereof shall not affect the validity of the corporate action required to be specified in such notice. To the extent that any notice provided in this Warrant constitutes, or contains, material, non-public information regarding the Company or any of the Subsidiaries, the Company shall simultaneously file such notice with the Commission pursuant to a Current Report on Form 8-K. The Holder shall remain entitled to exercise this Warrant during the period commencing on the date of such notice to the effective date of the event triggering such notice except as may otherwise be expressly set forth herein.

#### Section 4. Transfer of Warrant.

- a) Transferability. Subject to compliance with any applicable securities laws and the conditions set forth in Section 4(d) hereof and to the provisions of Section 4.1 of the Purchase Agreement, this Warrant and all rights hereunder (including, without limitation, any registration rights) are transferable, in whole or in part, upon surrender of this Warrant at the principal office of the Company or its designated agent, together with a written assignment of this Warrant substantially in the form attached hereto duly executed by the Holder or his agent or attorney and funds sufficient to pay any transfer taxes payable upon the making of such transfer. Upon such surrender and, if required, such payment, the Company shall execute and deliver a new Warrant or Warrants in the name of the assignee or assignees, as applicable, and in the denomination or denominations specified in such instrument of assignment, and shall issue to the assignor a new Warrant evidencing the portion of this Warrant not so assigned, and this Warrant shall promptly be cancelled. Notwithstanding anything herein to the contrary, the Holder shall not be required to physically surrender this Warrant to the Company unless the Holder has assigned this Warrant in full, in which case, the Holder shall surrender this Warrant to the Company within three (3) Trading Days of the date on which the Holder delivers an assignment form to the Company assigning this Warrant in full. The Warrant, if properly assigned in accordance herewith, may be exercised by a new holder for the purchase of Warrant Shares without having a new Warrant issued.
- b) New Warrants. This Warrant may be divided or combined with other Warrants upon presentation hereof at the aforesaid office of the Company, together with a written notice specifying the names and denominations in which new Warrants are to be issued, signed by the Holder or his agent or attorney. Subject to compliance with Section 4(a), as to any transfer which may be involved in such division or combination, the Company shall execute and deliver a new Warrant or Warrants in exchange for the Warrant or Warrants to be divided or combined in accordance with such notice. All Warrants issued on transfers or exchanges shall be dated the Issue Date of this Warrant and shall be identical with this Warrant except as to the number of Warrant Shares issuable pursuant thereto.
- c) <u>Warrant Register</u>. The Company shall register this Warrant, upon records to be maintained by the Company for that purpose (the "<u>Warrant Register</u>"), in the name of the record Holder hereof from time to time. The Company may deem and treat the registered Holder of this Warrant as the absolute owner hereof for the purpose of any exercise hereof or any distribution to the Holder, and for all other purposes, absent actual notice to the contrary.
- d) <u>Transfer Restrictions</u>. If, at the time of the surrender of this Warrant in connection with any transfer of this Warrant, the transfer of this Warrant shall not be either (i) registered pursuant to an effective registration statement under the Securities Act and under applicable state securities or blue sky laws or (ii) eligible for resale without volume or manner-of-sale restrictions or current public information requirements pursuant to Rule 144, the Company may require, as a condition of allowing such transfer, that the Holder or transferee of this Warrant, as the case may be, comply with the provisions of Section 5.7 of the Purchase Agreement.

e) <u>Representation by the Holder</u>. The Holder, by the acceptance hereof, represents and warrants that he is acquiring this Warrant and, upon any exercise hereof, will acquire the Warrant Shares issuable upon such exercise, for his own account and not with a view to or for distributing or reselling such Warrant Shares or any part thereof in violation of the Securities Act or any applicable state securities law, except pursuant to sales registered or exempted under the Securities Act.

## Section 5. Miscellaneous.

- a) No Rights as Stockholder Until Exercise; No Settlement in Cash. This Warrant does not entitle the Holder to any voting rights, dividends or other rights as a stockholder of the Company prior to the exercise hereof as set forth in Section 2(d)(i), except as expressly set forth in Section 3. Without limiting the rights of a Holder to receive Warrant Shares on a "cashless exercise," and to receive the cash payments contemplated pursuant to Sections 2(d)(i) and 2(d)(iv), in no event will the Company be required to net cash settle an exercise of this Warrant.
- b) Loss, Theft, Destruction or Mutilation of Warrant. The Company covenants that upon receipt by the Company of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Warrant or any stock certificate relating to the Warrant Shares, and in case of loss, theft or destruction, of indemnity or security reasonably satisfactory to it (which, in the case of the Warrant, shall not include the posting of any bond), and upon surrender and cancellation of such Warrant or stock certificate, if mutilated, the Company will make and deliver a new Warrant or stock certificate of like tenor and dated as of such cancellation, in lieu of such Warrant or stock certificate.
- c) <u>Saturdays, Sundays, Holidays, etc.</u> If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall not be a Trading Day, then, such action may be taken or such right may be exercised on the next succeeding Trading Day.

## d) Authorized Shares.

The Company covenants that, during the period the Warrant is outstanding, it will reserve from its authorized and unissued Common Stock a sufficient number of shares to provide for the issuance of the Warrant Shares upon the exercise of any purchase rights under this Warrant. The Company further covenants that its issuance of this Warrant shall constitute full authority to its officers who are charged with the duty of issuing the necessary Warrant Shares upon the exercise of the purchase rights under this Warrant. The Company will take all such reasonable action as may be necessary to assure that such Warrant Shares may be issued as provided herein without violation of any applicable law or regulation, or of any requirements of the Trading Market upon which the Common Stock may be listed. The Company covenants that all Warrant Shares which may be issued upon the exercise of the purchase rights represented by this Warrant will, upon exercise of the purchase rights represented by this Warrant and payment for such Warrant Shares in accordance herewith, be duly authorized, validly issued, fully paid and nonassessable and free from all taxes, liens and charges created by the Company in respect of the issue thereof (other than taxes in respect of any transfer occurring contemporaneously with such issue).

Except and to the extent as waived or consented to by the Holder, the Company shall not by any action, including, without limitation, amending its certificate of incorporation or through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of Holder as set forth in this Warrant against impairment. Without limiting the generality of the foregoing, the Company will (i) not increase the par value of any Warrant Shares above the amount payable therefor upon such exercise immediately prior to such increase in par value, (ii) take all such action as may be necessary or appropriate in order that the Company may validly and legally issue fully paid and nonassessable Warrant Shares upon the exercise of this Warrant and (iii) use commercially reasonable efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof, as may be, necessary to enable the Company to perform its obligations under this Warrant.

Before taking any action which would result in an adjustment in the number of Warrant Shares for which this Warrant is exercisable or in the Exercise Price, the Company shall obtain all such authorizations or exemptions thereof, or consents thereto, as may be necessary from any public regulatory body or bodies having jurisdiction thereof.

- e) <u>Jurisdiction</u>. All questions concerning the construction, validity, enforcement and interpretation of this Warrant shall be determined in accordance with the provisions of the Purchase Agreement.
- f) <u>Restrictions</u>. The Holder acknowledges that the Warrant Shares acquired upon the exercise of this Warrant, if not registered, and the Holder does not utilize cashless exercise, will have restrictions upon resale imposed by state and federal securities laws.
- g) Nonwaiver and Expenses. No course of dealing or any delay or failure to exercise any right hereunder on the part of Holder shall operate as a waiver of such right or otherwise prejudice the Holder's rights, powers or remedies, notwithstanding the fact that the right to exercise this Warrant terminates on the Termination Date. Without limiting any other provision of this Warrant, if the Company willfully and knowingly fails to comply with any provision of this Warrant, which results in any material damages to the Holder, the Company shall pay to the Holder such amounts as shall be sufficient to cover any costs and expenses including, but not limited to, reasonable attorneys' fees, including those of appellate proceedings, incurred by the Holder in collecting any amounts due pursuant hereto or in otherwise enforcing any of his rights, powers or remedies hereunder.
- h) Notices. Any notice, request or other document required or permitted to be given or delivered to the Holder by the Company shall be delivered to the address for the Holder that appears in the Company's Warrant Register.
- i) <u>Limitation of Liability</u>. No provision hereof, in the absence of any affirmative action by the Holder to exercise this Warrant to purchase Warrant Shares, and no enumeration herein of the rights or privileges of the Holder, shall give rise to any liability of the Holder for the purchase price of any Common Stock or as a stockholder of the Company, whether such liability is asserted by the Company or by creditors of the Company.

- j) <u>Remedies</u>. The Holder, in addition to being entitled to exercise all rights granted by law, including recovery of damages, will be entitled to specific performance of his rights under this Warrant. The Company agrees that monetary damages would not be adequate compensation for any loss incurred by reason of a breach by it of the provisions of this Warrant and hereby agrees to waive and not to assert the defense in any action for specific performance that a remedy at law would be adequate.
- k) <u>Successors and Assigns</u>. Subject to applicable securities laws, this Warrant and the rights and obligations evidenced hereby shall inure to the benefit of and be binding upon the successors and permitted assigns of the Company and the successors and permitted assigns of Holder. The provisions of this Warrant are intended to be for the benefit of any Holder from time to time of this Warrant and shall be enforceable by the Holder or holder of Warrant Shares.
- 1) <u>Amendment</u>. This Warrant may be modified or amended or the provisions hereof waived with the written consent of the Company, on the one hand, and the Holder of this Warrant, on the other hand.
- m) Severability. Wherever possible, each provision of this Warrant shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Warrant shall be prohibited by or invalid under applicable law, such provision shall be ineffective to the extent of such prohibition or invalidity, without invalidating the remainder of such provisions or the remaining provisions of this Warrant.
- n) <u>Headings</u>. The headings used in this Warrant are for the convenience of reference only and shall not, for any purpose, be deemed a part of this Warrant.

\*\*\*\*\*\*

(Signature Page Follows)

IN WITNESS WHEREOF, the Company has caused the indicated.	is Warrant to be executed by its officer thereunto duly authorized as of the date first above
PROFRAC HOLDING CORP.	
Ву:	_
Name:	_
Title:	_

## NOTICE OF EXERCISE

## TO: PROFRAC HOLDING CORP.

(1) The undersigned hereby elects to purchase Warrant Shares of the Company pursuant to the terms of the attached Warrant (only if exercised in full), and tenders herewith payment of the exercise price in full, together with all applicable transfer taxes, if any.
(2) Payment shall take the form of (check applicable box):
[] in lawful money of the United States; or
[] if permitted the cancellation of such number of Warrant Shares as is necessary, in accordance with the formula set forth in subsection 2(c), to exercise this Warrant with respect to the maximum number of Warrant Shares purchasable pursuant to the cashless exercise procedure set forth in subsection 2(c).
(3) Please issue said Warrant Shares in the name of the undersigned or in such other name as is specified below:
The Warrant Shares shall be delivered to the following DWAC Account Number:
(4) <u>Accredited Investor</u> . The undersigned is an "accredited investor" as defined in Regulation D promulgated under the Securities Act of 1933, as amended.
[SIGNATURE OF HOLDER]
Name of Investing Entity:
Signature of Authorized Signatory of Investing Entity:
Name of Authorized Signatory:
Title of Authorized Signatory:
Date:

## EXHIBIT B

## ASSIGNMENT FORM

(To assign the foregoing Warrant, execute this form and supply required information. Do not use this form to exercise the Warrant to purchase shares.)

FOR VALUE RECEIVED, the foregoing Warrant and all rights evidenced thereby are hereby assigned to

Name:	
	(Please Print)
Address:	
	(Please Print)
Phone Number:	
Email Address:	
Dated:	
Holder's Signature:	
Holder's Address:	

NEITHER THIS SECURITY NOR THE SECURITIES FOR WHICH THIS SECURITY IS EXERCISABLE HAVE BEEN REGISTERED WITH THE SECURITIES AND EXCHANGE COMMISSION OR THE SECURITIES COMMISSION OF ANY STATE IN RELIANCE UPON AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), AND, ACCORDINGLY, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR PURSUANT TO AN AVAILABLE EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS. THIS SECURITY AND THE SECURITIES ISSUABLE UPON EXERCISE OF THIS SECURITY MAY BE PLEDGED IN CONNECTION WITH A BONA FIDE MARGIN ACCOUNT OR OTHER LOAN SECURED BY SUCH SECURITIES.

#### FORM OF AMENDED AND RESTATED

#### COMMON STOCK PURCHASE WARRANT

## PROFRAC HOLDING CORP.

Issue Date: \_\_\_\_\_

	Initial Exercise Date:
	THIS AMENDED AND RESTATED COMMON STOCK PURCHASE WARRANT (the "Warrant") certifies that, for value received,
	or its assigns (the "Holder") is entitled, upon the terms and subject to the limitations on exercise and the conditions hereinafter set forth, a
any time on o	or after the date set forth above (the "Initial Exercise Date") and on or prior to 5:00 p.m. (New York City time) on September 11, 2025 (the
"Termination	n Date") but not thereafter, to subscribe for and purchase from PROFRAC HOLDING CORP., a Delaware corporation (the Company"), up to
share	es (as subject to adjustment hereunder, the "Warrant Shares") of the Company's Common Stock. The purchase price of one share of Common

Stock under this Warrant shall be equal to the Exercise Price, as defined in Section 2(b), as amended pursuant to the terms of that certain Agreement and Plan of Merger ("Merger Agreement") entered by and among the Company, U.S. Well Services, Inc., a Delaware corporation ("USWS"), and Thunderclap Merger Sub I, Inc., a Delaware corporation.

2 , , ,

Warrant Shares: \_\_\_\_

This Warrant amends and restates the terms of the Warrant to the Holder issued on March 11, 2022 (the "Original Warrant"). By accepting this Warrant, the Holder hereby agrees to surrender to the Company for cancellation the Original Warrant or at the request of the Company to execute an instrument of cancellation in form and substance acceptable to the Company. Holder and Company hereby acknowledge and agree that upon the issuance of this Warrant, the Original Warrant shall be amended and restated and all of the Company's obligations under such warrant shall be discharged and released in full without any further action on the part of the Company or Holder.

The following is a statement of the rights of the Holder and the conditions to which this Warrant is subject, and to which Holder, by acceptance of this Warrant, agrees:

Section 1. <u>Definitions</u>. Capitalized terms used and not otherwise defined herein shall have the meanings set forth in that certain Amended and Restated Securities Purchase Agreement (the "<u>Purchase Agreement</u>"), dated \_\_\_\_\_\_\_, among the Company and the purchasers signatory thereto.

#### Section 2. Exercise

a) Exercise of Warrant. Exercise of the purchase rights represented by this Warrant may be made, in whole or in part, at any time or times on or after the Initial Exercise Date and on or before the Termination Date by delivery to the Company of a duly executed facsimile copy or PDF copy submitted by e-mail (or e-mail attachment) of the Notice of Exercise in the form annexed hereto (the 'Notice of Exercise''). Within the earlier of (i) two (2) Trading Days and (ii) the number of Trading Days comprising the Standard Settlement Period (as defined in Section 2(d)(i) herein) following the date of exercise as aforesaid, the Holder shall deliver the aggregate Exercise Price for the Warrant Shares specified in the applicable Notice of Exercise by wire transfer or cashier's check drawn on a United States bank unless the cashless exercise procedure specified in Section 2(c) below is specified in the applicable Notice of Exercise; provided, however, in the event that the Holder has not delivered such aggregate Exercise Price within three (3) Trading Days following the date of such exercise as aforesaid, the Company shall not be obligated to deliver such Warrant Shares hereunder until such payment is made. No ink-original Notice of Exercise shall be required, nor shall any medallion guarantee (or other type of guarantee or notarization) of any Notice of Exercise be required. Notwithstanding anything herein to the contrary, the Holder shall not be required to physically surrender this Warrant to the Company until the Holder has purchased all of the Warrant Shares available hereunder and the Warrant has been exercised in full, in which case, the Holder shall surrender this Warrant to the Company for cancellation within three (3) Trading Days of the date on which the final Notice of Exercise is delivered to the Company. Partial exercises of this Warrant resulting in purchases of a portion of the total number of Warrant Shares available hereunder shall have the effect of lowering the outstanding number of Warrant Shares purchasable hereunder in an amount equal to the applicable number of Warrant Shares purchased. The Holder and the Company shall maintain records showing the number of Warrant Shares purchased and the date of such purchases. The Company shall deliver any objection to any Notice of Exercise within one (1) Trading Day of receipt of such notice. The Holder and any assignee, by acceptance of this Warrant, acknowledge and agree that, by reason of the provisions of this paragraph, following the purchase of a portion of the Warrant Shares hereunder, the number of Warrant Shares available for purchase hereunder at any given time may be less than the amount stated on the face hereof.

- b) Exercise Price. The exercise price per share of Common Stock under this Warrant shall be \$31.43, subject to adjustment hereunder (the "Exercise Price").
- c) <u>Cashless Exercise</u>. If at the time of exercise hereof there is no effective registration statement registering, or the prospectus contained therein is not available for the issuance of the Warrant Shares to the Holder, then this Warrant may also be exercised, in whole or in part, at such time by means of a "cashless exercise" in which the Holder shall be entitled to receive a number of Warrant Shares equal to the quotient obtained by dividing [(A-B) (X)] by (A), where:
  - (A) = as applicable: (i) the VWAP on the Trading Day immediately preceding the date of the applicable Notice of Exercise if such Notice of Exercise is (1) both executed and delivered pursuant to Section 2(a) hereof on a day that is not a Trading Day or (2) both executed and delivered pursuant to Section 2(a) hereof on a Trading Day prior to the opening of "regular trading hours" (as defined in Rule 600(b) of Regulation NMS promulgated under the federal securities laws) on such Trading Day, (ii) at the option of the Holder, either (y) the VWAP on the Trading Day immediately preceding the date of the applicable Notice of Exercise or (z) the Bid Price of the Common Stock on the principal Trading Market as reported by Bloomberg L.P. ("Bloomberg") as of the time of the Holder's execution of the applicable Notice of Exercise is executed during "regular trading hours" on a Trading Day and is delivered within two (2) hours thereafter (including until two (2) hours after the close of "regular trading hours" on a Trading Day) pursuant to Section 2(a) hereof or (iii) the VWAP on the date of the applicable Notice of Exercise if the date of such Notice of Exercise is a Trading Day and such Notice of Exercise is both executed and delivered pursuant to Section 2(a) hereof after the close of "regular trading hours" on such Trading Day;
  - (B) = the Exercise Price of this Warrant, as adjusted hereunder; and
  - (X) = the number of Warrant Shares that would be issuable upon exercise of this Warrant in accordance with the terms of this Warrant if such exercise were by means of a cash exercise rather than a cashless exercise.

"Bid Price" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed or quoted on a Trading Market, the bid price of the Common Stock for the time in question (or the nearest preceding date) on the Trading Market on which the Common Stock is then listed or quoted as reported by Bloomberg (based on a Trading Day from 9:30 a.m. (New York City time) to 4:02 p.m. (New York City time)), (b) if OTCQB or OTCQX is not a Trading Market, the volume weighted average price of the Common Stock for such date (or the nearest preceding date) on OTCQB or OTCQX as applicable, (c) if the Common Stock is not then listed or quoted for trading on OTCQB or OTCQX and if prices for the Common Stock are then reported on The Pink Open Market (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Common Stock so reported, or (d) in all other cases, the fair market value of a share of Common Stock as determined by an independent appraiser selected in good faith by the Holders of a majority in interest of the Securities then outstanding and reasonably acceptable to the Company, the fees and expenses of which shall be paid by the Company.

"VWAP" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed or quoted on a Trading Market, the daily volume weighted average price of the Common Stock for such date (or the nearest preceding date) on the Trading Market on which the Common Stock is then listed or quoted as reported by Bloomberg (based on a Trading Day from 9:30 a.m. (New York City time)), (b) if OTCQB or OTCQX is not a Trading Market, the volume weighted average price of the Common Stock for such date (or the nearest preceding date) on OTCQB or OTCQX as applicable, (c) if the Common Stock is not then listed or quoted for trading on OTCQB or OTCQX and if prices for the Common Stock are then reported on The Pink Open Market (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Common Stock so reported, or (d) in all other cases, the fair market value of a share of Common Stock as determined by an independent appraiser selected in good faith by the Holders of a majority in interest of the Securities then outstanding and reasonably acceptable to the Company, the fees and expenses of which shall be paid by the Company.

If Warrant Shares are issued in such a cashless exercise, the parties acknowledge and agree that in accordance with Section 3(a)(9) of the Securities Act, the Warrant Shares shall take on the characteristics of the Warrants being exercised, and the holding period of the Warrant Shares being issued may be tacked on to the holding period of this Warrant. The Company agrees not to take any position contrary to this Section 2(c).

#### d) Mechanics of Exercise.

i. <u>Delivery of Warrant Shares Upon Exercise</u>. The Company shall cause the Warrant Shares purchased hereunder to be transmitted by the Transfer Agent to the Holder by crediting the account of the Holder's or its designee's balance account with The Depository Trust Company through its Deposit or Withdrawal at Custodian system ("<u>DWAC</u>") if the Company is then a participant in such system and either (A) there is an effective registration statement permitting the issuance of the Warrant Shares to or resale of the Warrant Shares by the Holder or (B) the Warrant Shares are eligible for resale by the Holder without volume or manner-of-sale limitations pursuant to Rule 144 (assuming cashless exercise of the Warrants), and otherwise by physical delivery of a certificate, registered in the Company's share register in the name of the Holder or its designee, for the number of Warrant Shares to which the Holder is entitled pursuant to such exercise to the address specified by the Holder in the Notice of Exercise by the date that is the earliest of (i) two (2) Trading Days after the delivery to the Company of the Notice of Exercise, (ii) two (2) Trading

Day after delivery of the aggregate Exercise Price to the Company and (iii) the number of Trading Days comprising the Standard Settlement Period after the delivery to the Company of the Notice of Exercise (such date, the "Warrant Share Delivery Date"). Upon delivery of the Notice of Exercise, the Holder shall be deemed for all corporate purposes to have become the holder of record of the Warrant Shares with respect to which this Warrant has been exercised, irrespective of the date of delivery of the Warrant Shares, provided that payment of the aggregate Exercise Price (other than in the case of a cashless exercise) is received by the Warrant Share Delivery Date. If the Company fails for any reason to deliver to the Holder the Warrant Shares subject to a Notice of Exercise by the Warrant Share Delivery Date, the Company shall pay to the Holder, in cash, as liquidated damages and not as a penalty, for each \$1,000 of Warrant Shares subject to such exercise (based on the VWAP of the Common Stock on the date of the applicable Notice of Exercise), \$10 per Trading Day (increasing to \$20 per Trading Day on the third Trading Day after the Warrant Share Delivery Date) for each Trading Day after such Warrant Share Delivery Date until such Warrant Shares are delivered or Holder rescinds such exercise. The Company agrees to maintain a transfer agent that is a participant in the FAST program so long as this Warrant remains outstanding and exercisable. As used herein, "Standard Settlement Period" means the standard settlement period, expressed in a number of Trading Days, on the Company's primary Trading Market with respect to the Common Stock as in effect on the date of delivery of the Notice of Exercise.

- ii. <u>Delivery of New Warrants Upon Exercise</u>. If this Warrant shall have been exercised in part, the Company shall, at the request of a Holder and upon surrender of this Warrant certificate, at the time of delivery of the Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Warrant Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant.
- iii. <u>Rescission Rights</u>. If the Company fails to cause the Transfer Agent to transmit to the Holder the Warrant Shares pursuant to Section 2(d)(i) by the Warrant Share Delivery Date, then the Holder will have the right to rescind such exercise.
- iv. <u>Compensation for Buy-In on Failure to Timely Deliver Warrant Shares Upon Exercise</u>. In addition to any other rights available to the Holder, if the Company fails to cause the Transfer Agent to transmit to the Holder the Warrant Shares in accordance with the provisions of Section 2(d)(i) above pursuant to an exercise on or before the Warrant Share Delivery Date, and if after such date the Holder is required by its broker to purchase (in an open market transaction or otherwise) or the Holder's brokerage firm otherwise purchases, shares of Common Stock to

deliver in satisfaction of a sale by the Holder of the Warrant Shares which the Holder anticipated receiving upon such exercise (a "Buy-In"), then the Company shall (A) pay in cash to the Holder the amount, if any, by which (x) the Holder's total purchase price (including brokerage commissions, if any) for the shares of Common Stock so purchased exceeds (y) the amount obtained by multiplying (1) the number of Warrant Shares that the Company was required to deliver to the Holder in connection with the exercise at issue times (2) the price at which the sell order giving rise to such purchase obligation was executed, and (B) at the option of the Holder, either reinstate the portion of the Warrant and equivalent number of Warrant Shares for which such exercise was not honored (in which case such exercise shall be deemed rescinded) or deliver to the Holder the number of shares of Common Stock that would have been issued had the Company timely complied with its exercise and delivery obligations hereunder. For example, if the Holder purchases Common Stock having a total purchase price of \$11,000 to cover a Buy-In with respect to an attempted exercise of shares of Common Stock with an aggregate sale price giving rise to such purchase obligation of \$10,000, under clause (A) of the immediately preceding sentence the Company shall be required to pay the Holder \$1,000. The Holder shall provide the Company written notice indicating the amounts payable to the Holder in respect of the Buy-In and, upon request of the Company, evidence of the amount of such loss. Nothing herein shall limit a Holder's right to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Company's failure to timely deliver shares of Common Stock upon exercise of the Warrant as required pursuant to the terms hereof. The obligation of the Company to pay compensation for Buy In under this Section 2(d)(iv) is subject to delivery by the Holder of the aggregate Exercise Price in accordance with the terms of Section 2(a) herein.

- v. No Fractional Shares or Scrip. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant. As to any fraction of a share which the Holder would otherwise be entitled to purchase upon such exercise, the Company shall, at its election, either pay a cash adjustment in respect of such final fraction in an amount equal to such fraction multiplied by the Exercise Price or round up to the next whole share.
- vi. <u>Charges</u>, <u>Taxes and Expenses</u>. Issuance of Warrant Shares shall be made without charge to the Holder for any issue or transfer tax or other incidental expense in respect of the issuance of such Warrant Shares, all of which taxes and expenses shall be paid by the Company, and such Warrant Shares shall be issued in the name of the Holder or in such name or names as may be directed by the Holder; <u>provided</u>, <u>however</u>, that in the event that Warrant Shares are to be issued in a name other than the name of the Holder, this Warrant when surrendered for exercise shall be

accompanied by the Assignment Form attached hereto duly executed by the Holder and the Company may require, as a condition thereto, the payment of a sum sufficient to reimburse it for any transfer tax incidental thereto. The Company shall pay all Transfer Agent fees required for same-day processing of any Notice of Exercise and all fees to the Depository Trust Company (or another established clearing corporation performing similar functions) required for same-day electronic delivery of the Warrant Shares.

vii. <u>Closing of Books</u>. The Company will not close its stockholder books or records in any manner which prevents the timely exercise of this Warrant, pursuant to the terms hereof.

e) Holder's Exercise Limitations. The Company shall not effect any exercise of this Warrant, and a Holder shall not have the right to exercise any portion of this Warrant, pursuant to Section 2 or otherwise, to the extent that after giving effect to such issuance after exercise as set forth on the applicable Notice of Exercise, the Holder (together with the Holder's Affiliates, and any other Persons acting as a group together with the Holder or any of the Holder's Affiliates (such Persons, "Attribution Parties")), would beneficially own in excess of the Beneficial Ownership Limitation (as defined below). For purposes of the foregoing sentence, the number of shares of Common Stock beneficially owned by the Holder and its Affiliates and Attribution Parties shall include the number of shares of Common Stock issuable upon exercise of this Warrant with respect to which such determination is being made, but shall exclude the number of shares of Common Stock which would be issuable upon (i) exercise of the remaining, nonexercised portion of this Warrant beneficially owned by the Holder or any of its Affiliates or Attribution Parties and (ii) exercise or conversion of the unexercised or nonconverted portion of any other securities of the Company (including, without limitation, any other Common Stock Equivalents) subject to a limitation on conversion or exercise analogous to the limitation contained herein beneficially owned by the Holder or any of its Affiliates or Attribution Parties. Except as set forth in the preceding sentence, for purposes of this Section 2(e), beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder, it being acknowledged by the Holder that the Company is not representing to the Holder that such calculation is in compliance with Section 13(d) of the Exchange Act and the Holder is solely responsible for any schedules required to be filed in accordance therewith. To the extent that the limitation contained in this Section 2(e) applies, the determination of whether this Warrant is exercisable (in relation to other securities owned by the Holder together with any Affiliates and Attribution Parties) and of which portion of this Warrant is exercisable shall be in the sole discretion of the Holder, and the submission of a Notice of Exercise shall be deemed to be the Holder's determination of whether this Warrant is exercisable (in relation to other securities owned by the Holder together with any Affiliates and Attribution Parties) and of which portion of this Warrant is exercisable, in each case subject to the Beneficial Ownership Limitation, and the Company shall have no obligation to verify or confirm the accuracy of such determination and shall have no liability for exercises of the Warrant that are not in compliance with the Beneficial Ownership Limitation. In addition, a determination as

to any group status as contemplated above shall be determined in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder and the Company shall have no obligation to verify or confirm the accuracy of such determination and shall have no liability for exercises of the Warrant that are not in compliance with the Beneficial Ownership Limitation. For purposes of this Section 2(e), in determining the number of outstanding shares of Common Stock, a Holder may rely on the number of outstanding shares of Common Stock as reflected in (A) the Company's most recent periodic or annual report filed with the Commission, as the case may be, (B) a more recent public announcement by the Company or (C) a more recent written notice by the Company or the Transfer Agent setting forth the number of shares of Common Stock outstanding. Upon the written or oral request of a Holder, the Company shall within two (2) Trading Day confirm orally and in writing to the Holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of securities of the Company, including this Warrant, by the Holder or its Affiliates or Attribution Parties since the date as of which such number of outstanding shares of Common Stock was reported. The "Beneficial Ownership Limitation" shall be [4.99%/9.99%] of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock issuable upon exercise of this Warrant. The Holder, upon notice to the Company, may increase or decrease the Beneficial Ownership Limitation provisions of this Section 2(e), provided that the Beneficial Ownership Limitation in no event exceeds 9.99% of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock upon exercise of this Warrant held by the Holder and the provisions of this Section 2(e) shall continue to apply. Any increase in the Beneficial Ownership Limitation will not be effective until the 61st day after such notice is delivered to the Company. The provisions of this paragraph shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this Section 2(e) to correct this paragraph (or any portion hereof) which may be defective or inconsistent with the intended Beneficial Ownership Limitation herein contained or to make changes or supplements necessary or desirable to properly give effect to such limitation. The limitations contained in this paragraph shall apply to a successor holder of this Warrant.

## Section 3. Certain Adjustments.

a) Stock Dividends and Splits. If the Company, at any time while this Warrant is outstanding: (i) pays a stock dividend or otherwise makes a distribution or distributions on shares of its Common Stock or any other equity or equity equivalent securities payable in shares of Common Stock (which, for avoidance of doubt, shall not include any shares of Common Stock issued by the Company upon exercise of this Warrant), (ii) subdivides outstanding shares of Common Stock into a larger number of shares, (iii) combines (including by way of reverse stock split) outstanding shares of Common Stock into a smaller number of shares, or (iv) issues by reclassification of shares of the Common Stock any shares of capital stock of the Company, then in each case the Exercise Price shall be multiplied by a fraction of which the numerator shall be the number of shares of Common Stock (excluding treasury shares, if any) outstanding immediately before such event and of which the denominator shall be the number of

shares of Common Stock outstanding immediately after such event, and the number of shares issuable upon exercise of this Warrant shall be proportionately adjusted such that the aggregate Exercise Price of this Warrant shall remain unchanged. Any adjustment made pursuant to this Section 3(a) shall become effective immediately after the record date for the determination of stockholders entitled to receive such dividend or distribution and shall become effective immediately after the effective date in the case of a subdivision, combination or re-classification.

- b) Subsequent Rights Offerings. In addition to any adjustments pursuant to Section 3(a) above, if at any time the Company grants, issues or sells any Common Stock Equivalents or rights to purchase stock, warrants, securities or other property pro rata to the record holders of any class of shares of Common Stock (the "Purchase Rights"), then the Holder will be entitled to acquire, upon the terms applicable to such Purchase Rights, the aggregate Purchase Rights which the Holder could have acquired if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date on which a record is taken for the grant, issuance or sale of such Purchase Rights, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the grant, issue or sale of such Purchase Rights (provided, however, that to the extent that the Holder's right to participate in any such Purchase Right would result in the Holder exceeding the Beneficial Ownership Limitation, then the Holder shall not be entitled to participate in such Purchase Right to such extent (or beneficial ownership of such shares of Common Stock as a result of such Purchase Right to such extent and be held in abeyance for the Holder until such time, if ever, as its right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).
- c) Pro Rata Distributions. During such time as this Warrant is outstanding, if the Company shall declare or make any dividend or other distribution of its assets (or rights to acquire its assets) to holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of cash, stock or other securities, property or options by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "Distribution"), at any time after the issuance of this Warrant, then, in each such case, the Holder shall be entitled to participate in such Distribution to the same extent that the Holder would have participated therein if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date of which a record is taken for such Distribution, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the participation in such Distribution (provided, however, that to the extent that the Holder's right to participate in any such Distribution would result in the Holder exceeding the Beneficial Ownership Limitation, then the Holder shall not be entitled to participate in such Distribution to such extent (or in the beneficial ownership of any shares of Common Stock as a result of such Distribution to such extent) and the portion of such Distribution shall be held in abeyance for the benefit of the Holder until such time, if ever, as its right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).

d) Fundamental Transaction. If, at any time while this Warrant is outstanding, (i) the Company, directly or indirectly, in one or more related transactions effects any merger or consolidation of the Company with or into another Person, (ii) the Company (and all of its Subsidiaries, taken as a whole), directly or indirectly, effects any sale, lease, license, assignment, transfer, conveyance or other disposition of all or substantially all of its assets in one or a series of related transactions, (iii) any, direct or indirect, purchase offer, tender offer or exchange offer (whether by the Company or another Person) is completed pursuant to which holders of Common Stock are permitted to sell, tender or exchange their shares for other securities, cash or property and has been accepted by the holders of 50% or more of the outstanding Common Stock, (iv) the Company, directly or indirectly, in one or more related transactions effects any reclassification, reorganization or recapitalization of the Common Stock or any compulsory share exchange pursuant to which the Common Stock is effectively converted into or exchanged for other securities, cash or property, or (v) the Company, directly or indirectly, in one or more related transactions consummates a stock or share purchase agreement or other business combination (including, without limitation, a reorganization, recapitalization, spin-off, merger or scheme of arrangement) with another Person or group of Persons whereby such other Person or group acquires more than 50% of the outstanding shares of Common Stock (not including any shares of Common Stock held by the other Person or other Persons making or party to, or associated or affiliated with the other Persons making or party to, such stock or share purchase agreement or other business combination) (each a "Fundamental Transaction"), then, upon any subsequent exercise of this Warrant, the Holder shall have the right to receive, for each Warrant Share that would have been issuable upon such exercise immediately prior to the occurrence of such Fundamental Transaction, at the option of the Holder (without regard to any limitation in Section 2(e) on the exercise of this Warrant), the number of shares of Common Stock of the successor or acquiring corporation or of the Company, if it is the surviving corporation, and any additional consideration (the "Alternate Consideration") receivable as a result of such Fundamental Transaction by a holder of the number of shares of Common Stock for which this Warrant is exercisable immediately prior to such Fundamental Transaction (without regard to any limitation in Section 2(e) on the exercise of this Warrant). For purposes of any such exercise, the determination of the Exercise Price shall be appropriately adjusted to apply to such Alternate Consideration based on the amount of Alternate Consideration issuable in respect of one share of Common Stock in such Fundamental Transaction, and the Company shall apportion the Exercise Price among the Alternate Consideration in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Holder shall be given the same choice as to the Alternate Consideration it receives upon any exercise of this Warrant following such Fundamental Transaction. Notwithstanding anything to the contrary, in the event of a Fundamental Transaction, the Company or any Successor Entity (as defined below) shall, at the Holder's option, exercisable at any time concurrently with, or within 30 days after, the consummation of the Fundamental Transaction (or, if later, the date of the public

announcement of the applicable Fundamental Transaction), purchase this Warrant from the Holder by paying to the Holder an amount of cash equal to the Black Scholes Value (as defined below) of the remaining unexercised portion of this Warrant on the date of the consummation of such Fundamental Transaction; provided, however, that if the Fundamental Transaction is not within the Company's control, including not approved by the Company's Board of Directors, Holder shall only be entitled to receive from the Company or any Successor Entity the same type or form of consideration (and in the same proportion), at the Black Scholes Value of the unexercised portion of this Warrant, that is being offered and paid to the holders of Common Stock of the Company in connection with the Fundamental Transaction, whether that consideration be in the form of cash, stock or any combination thereof, or whether the holders of Common Stock are given the choice to receive from among alternative forms of consideration in connection with the Fundamental Transaction; provided, further, that if holders of Common Stock of the Company are not offered or paid any consideration in such Fundamental Transaction, such holders of Common Stock will be deemed to have received common stock of the Successor Entity (which Entity may be the Company following such Fundamental Transaction) in such Fundamental Transaction. "Black Scholes Value" means the value of this Warrant based on the Black Scholes Option Pricing Model obtained from the "OV" function on Bloomberg determined as of the day of consummation of the applicable Fundamental Transaction for pricing purposes and reflecting (A) a risk-free interest rate corresponding to the U.S. Treasury rate for a period equal to the time between the date of the public announcement of the applicable Fundamental Transaction and the Termination Date, (B) an expected volatility equal to the greater of 100% and the 100 day volatility obtained from the HVT function on Bloomberg (determined utilizing a 365-day annualization factor) as of the Trading Day immediately following the public announcement of the applicable Fundamental Transaction, (C) the underlying price per share used in such calculation shall be the greater of (i) the sum of the price per share being offered in cash, if any, plus the value of any non-cash consideration, if any, being offered in such Fundamental Transaction and (ii) the highest VWAP during the period beginning on the Trading Day immediately preceding the announcement of the applicable Fundamental Transaction (or the consummation of the applicable Fundamental Transaction, if earlier) and ending on the Trading Day of the Holder's request pursuant to this Section 3(d) and (D) a remaining option time equal to the time between the date of the public announcement of the applicable Fundamental Transaction and the Termination Date and (E) a zero cost of borrow. The payment of the Black Scholes Value will be made by wire transfer of immediately available funds (or such other consideration) within the later of (i) five (5) Trading Days of the Holder's election and (ii) the date of consummation of the Fundamental Transaction. The Company shall cause any successor entity in a Fundamental Transaction in which the Company is not the survivor (the "Successor Entity") to assume in writing all of the obligations of the Company under this Warrant and the other Transaction Documents in accordance with the provisions of this Section 3(d) pursuant to written agreements in form and substance reasonably satisfactory to the Holder and approved by the Holder (without unreasonable delay) prior to such Fundamental Transaction and shall, at the option of the Holder, deliver to the Holder in exchange for this Warrant a security of the Successor Entity evidenced by a written instrument substantially similar in form and substance to this Warrant which is

exercisable for a corresponding number of shares of capital stock of such Successor Entity (or its parent entity) equivalent to the shares of Common Stock acquirable and receivable upon exercise of this Warrant (without regard to any limitations on the exercise of this Warrant) prior to such Fundamental Transaction, and with an exercise price which applies the exercise price hereunder to such shares of capital stock (but taking into account the relative value of the shares of Common Stock pursuant to such Fundamental Transaction and the value of such shares of capital stock, such number of shares of capital stock and such exercise price being for the purpose of protecting the economic value of this Warrant immediately prior to the consummation of such Fundamental Transaction), and which is reasonably satisfactory in form and substance to the Holder. Upon the occurrence of any such Fundamental Transaction, the Successor Entity shall succeed to, and be substituted for (so that from and after the date of such Fundamental Transaction, the provisions of this Warrant and the other Transaction Documents referring to the "Company" shall refer instead to the Successor Entity), and may exercise every right and power of the Company and shall assume all of the obligations of the Company under this Warrant and the other Transaction Documents with the same effect as if such Successor Entity had been named as the Company herein.

e) <u>Calculations</u>. All calculations under this Section 3 shall be made to the nearest cent or the nearest 1/100th of a share, as the case may be. For purposes of this Section 3, the number of shares of Common Stock deemed to be issued and outstanding as of a given date shall be the sum of the number of shares of Common Stock (excluding treasury shares, if any) issued and outstanding.

## f) Notice to Holder.

- i. <u>Adjustment to Exercise Price</u>. Whenever the Exercise Price is adjusted pursuant to any provision of this Section 3, the Company shall promptly deliver to the Holder by facsimile or email a notice setting forth the Exercise Price after such adjustment and any resulting adjustment to the number of Warrant Shares and setting forth a brief statement of the facts requiring such adjustment.
- ii. Notice to Allow Exercise by Holder. If (A) the Company shall declare a dividend (or any other distribution in whatever form) on the Common Stock, (B) the Company shall declare a special nonrecurring cash dividend on or a redemption of the Common Stock, (C) the Company shall authorize the granting to all holders of the Common Stock rights or warrants to subscribe for or purchase any shares of capital stock of any class or of any rights, (D) the approval of any stockholders of the Company shall be required in connection with any reclassification of the Common Stock, any consolidation or merger to which the Company is a party, any sale or transfer of all or substantially all of the assets of the Company, or any compulsory share exchange whereby the Common Stock is converted into other securities, cash or property, or (E) the Company shall authorize the voluntary or involuntary dissolution, liquidation or winding up of the affairs of the Company, then, in each case, the Company

shall cause to be delivered by facsimile or email to the Holder at its last facsimile number or email address as it shall appear upon the Warrant Register of the Company, at least 20 calendar days prior to the applicable record or effective date hereinafter specified, a notice stating (x) the date on which a record is to be taken for the purpose of such dividend, distribution, redemption, rights or warrants, or if a record is not to be taken, the date as of which the holders of the Common Stock of record to be entitled to such dividend, distributions, redemption, rights or warrants are to be determined or (y) the date on which such reclassification, consolidation, merger, sale, transfer or share exchange is expected to become effective or close, and the date as of which it is expected that holders of the Common Stock of record shall be entitled to exchange their shares of the Common Stock for securities, cash or other property deliverable upon such reclassification, consolidation, merger, sale, transfer or share exchange; provided that the failure to deliver such notice or any defect therein or in the delivery thereof shall not affect the validity of the corporate action required to be specified in such notice. To the extent that any notice provided in this Warrant constitutes, or contains, material, non-public information regarding the Company or any of the Subsidiaries, the Company shall simultaneously file such notice with the Commission pursuant to a Current Report on Form 8-K. The Holder shall remain entitled to exercise this Warrant during the period commencing on the date of such notice to the effective date of the event triggering such notice except as may otherwise be expressly set forth herein.

#### Section 4. Transfer of Warrant.

a) Transferability. Subject to compliance with any applicable securities laws and the conditions set forth in Section 4(d) hereof and to the provisions of Section 4.1 of the Purchase Agreement, this Warrant and all rights hereunder (including, without limitation, any registration rights) are transferable, in whole or in part, upon surrender of this Warrant at the principal office of the Company or its designated agent, together with a written assignment of this Warrant substantially in the form attached hereto duly executed by the Holder or its agent or attorney and funds sufficient to pay any transfer taxes payable upon the making of such transfer. Upon such surrender and, if required, such payment, the Company shall execute and deliver a new Warrant or Warrants in the name of the assignee or assignees, as applicable, and in the denomination or denominations specified in such instrument of assignment, and shall issue to the assignor a new Warrant evidencing the portion of this Warrant not so assigned, and this Warrant shall promptly be cancelled. Notwithstanding anything herein to the contrary, the Holder shall not be required to physically surrender this Warrant to the Company unless the Holder has assigned this Warrant in full, in which case, the Holder shall surrender this Warrant to the Company within three

(3) Trading Days of the date on which the Holder delivers an assignment form to the Company assigning this Warrant in full. The Warrant, if properly assigned in accordance herewith, may be exercised by a new holder for the purchase of Warrant Shares without having a new Warrant issued.

- b) New Warrants. This Warrant may be divided or combined with other Warrants upon presentation hereof at the aforesaid office of the Company, together with a written notice specifying the names and denominations in which new Warrants are to be issued, signed by the Holder or its agent or attorney. Subject to compliance with Section 4(a), as to any transfer which may be involved in such division or combination, the Company shall execute and deliver a new Warrant or Warrants in exchange for the Warrant or Warrants to be divided or combined in accordance with such notice. All Warrants issued on transfers or exchanges shall be dated the Issue Date of this Warrant and shall be identical with this Warrant except as to the number of Warrant Shares issuable pursuant thereto.
- c) <u>Warrant Register</u>. The Company shall register this Warrant, upon records to be maintained by the Company for that purpose (the <u>Warrant Register</u>"), in the name of the record Holder hereof from time to time. The Company may deem and treat the registered Holder of this Warrant as the absolute owner hereof for the purpose of any exercise hereof or any distribution to the Holder, and for all other purposes, absent actual notice to the contrary.
- d) <u>Transfer Restrictions</u>. If, at the time of the surrender of this Warrant in connection with any transfer of this Warrant, the transfer of this Warrant shall not be either (i) registered pursuant to an effective registration statement under the Securities Act and under applicable state securities or blue sky laws or (ii) eligible for resale without volume or manner-of-sale restrictions or current public information requirements pursuant to Rule 144, the Company may require, as a condition of allowing such transfer, that the Holder or transferee of this Warrant, as the case may be, comply with the provisions of Section 5.7 of the Purchase Agreement.
- e) <u>Representation by the Holder</u>. The Holder, by the acceptance hereof, represents and warrants that it is acquiring this Warrant and, upon any exercise hereof, will acquire the Warrant Shares issuable upon such exercise, for its own account and not with a view to or for distributing or reselling such Warrant Shares or any part thereof in violation of the Securities Act or any applicable state securities law, except pursuant to sales registered or exempted under the Securities Act.

## Section 5. Miscellaneous.

a) No Rights as Stockholder Until Exercise; No Settlement in Cash. This Warrant does not entitle the Holder to any voting rights, dividends or other rights as a stockholder of the Company prior to the exercise hereof as set forth in Section 2(d)(i), except as expressly set forth in Section 3. Without limiting the rights of a Holder to receive Warrant Shares on a "cashless exercise," and to receive the cash payments contemplated pursuant to Sections 2(d)(i) and 2(d)(iv), in no event will the Company be required to net cash settle an exercise of this Warrant.

- b) Loss, Theft, Destruction or Mutilation of Warrant. The Company covenants that upon receipt by the Company of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Warrant or any stock certificate relating to the Warrant Shares, and in case of loss, theft or destruction, of indemnity or security reasonably satisfactory to it (which, in the case of the Warrant, shall not include the posting of any bond), and upon surrender and cancellation of such Warrant or stock certificate, if mutilated, the Company will make and deliver a new Warrant or stock certificate of like tenor and dated as of such cancellation, in lieu of such Warrant or stock certificate.
- c) <u>Saturdays, Sundays, Holidays, etc.</u> If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall not be a Trading Day, then, such action may be taken or such right may be exercised on the next succeeding Trading Day.

## d) Authorized Shares.

The Company covenants that, during the period the Warrant is outstanding, it will reserve from its authorized and unissued Common Stock a sufficient number of shares to provide for the issuance of the Warrant Shares upon the exercise of any purchase rights under this Warrant. The Company further covenants that its issuance of this Warrant shall constitute full authority to its officers who are charged with the duty of issuing the necessary Warrant Shares upon the exercise of the purchase rights under this Warrant. The Company will take all such reasonable action as may be necessary to assure that such Warrant Shares may be issued as provided herein without violation of any applicable law or regulation, or of any requirements of the Trading Market upon which the Common Stock may be listed. The Company covenants that all Warrant Shares which may be issued upon the exercise of the purchase rights represented by this Warrant and payment for such Warrant Shares in accordance herewith, be duly authorized, validly issued, fully paid and nonassessable and free from all taxes, liens and charges created by the Company in respect of the issue thereof (other than taxes in respect of any transfer occurring contemporaneously with such issue).

Except and to the extent as waived or consented to by the Holder, the Company shall not by any action, including, without limitation, amending its certificate of incorporation or through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of Holder as set forth in this Warrant against impairment. Without limiting the generality of the foregoing, the Company will (i) not increase the par value of any Warrant Shares above the amount payable therefor upon such exercise immediately prior to such increase in par value, (ii) take all such action as may be necessary or appropriate in order that the Company may validly and legally issue fully paid and nonassessable Warrant Shares upon the exercise of this Warrant and (iii) use commercially reasonable efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof, as may be, necessary to enable the Company to perform its obligations under this Warrant.

Before taking any action which would result in an adjustment in the number of Warrant Shares for which this Warrant is exercisable or in the Exercise Price, the Company shall obtain all such authorizations or exemptions thereof, or consents thereto, as may be necessary from any public regulatory body or bodies having jurisdiction thereof.

- e) <u>Jurisdiction</u>. All questions concerning the construction, validity, enforcement and interpretation of this Warrant shall be determined in accordance with the provisions of the Purchase Agreement.
- f) <u>Restrictions</u>. The Holder acknowledges that the Warrant Shares acquired upon the exercise of this Warrant, if not registered, and the Holder does not utilize cashless exercise, will have restrictions upon resale imposed by state and federal securities laws.
- g) Nonwaiver and Expenses. No course of dealing or any delay or failure to exercise any right hereunder on the part of Holder shall operate as a waiver of such right or otherwise prejudice the Holder's rights, powers or remedies, notwithstanding the fact that the right to exercise this Warrant terminates on the Termination Date. Without limiting any other provision of this Warrant, if the Company willfully and knowingly fails to comply with any provision of this Warrant, which results in any material damages to the Holder, the Company shall pay to the Holder such amounts as shall be sufficient to cover any costs and expenses including, but not limited to, reasonable attorneys' fees, including those of appellate proceedings, incurred by the Holder in collecting any amounts due pursuant hereto or in otherwise enforcing any of its rights, powers or remedies hereunder.
- h) Notices. Any notice, request or other document required or permitted to be given or delivered to the Holder by the Company shall be delivered in accordance with the notice provisions of the Purchase Agreement.
- i) <u>Limitation of Liability</u>. No provision hereof, in the absence of any affirmative action by the Holder to exercise this Warrant to purchase Warrant Shares, and no enumeration herein of the rights or privileges of the Holder, shall give rise to any liability of the Holder for the purchase price of any Common Stock or as a stockholder of the Company, whether such liability is asserted by the Company or by creditors of the Company.
- j) Remedies. The Holder, in addition to being entitled to exercise all rights granted by law, including recovery of damages, will be entitled to specific performance of its rights under this Warrant. The Company agrees that monetary damages would not be adequate compensation for any loss incurred by reason of a breach by it of the provisions of this Warrant and hereby agrees to waive and not to assert the defense in any action for specific performance that a remedy at law would be adequate.

- k) <u>Successors and Assigns</u>. Subject to applicable securities laws, this Warrant and the rights and obligations evidenced hereby shall inure to the benefit of and be binding upon the successors and permitted assigns of the Company and the successors and permitted assigns of Holder. The provisions of this Warrant are intended to be for the benefit of any Holder from time to time of this Warrant and shall be enforceable by the Holder or holder of Warrant Shares.
- l) Amendment. This Warrant may be modified or amended or the provisions hereof waived with the written consent of the Company, on the one hand, and the Holder of this Warrant, on the other hand.
- m) <u>Severability</u>. Wherever possible, each provision of this Warrant shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Warrant shall be prohibited by or invalid under applicable law, such provision shall be ineffective to the extent of such prohibition or invalidity, without invalidating the remainder of such provisions or the remaining provisions of this Warrant.
- n) Headings. The headings used in this Warrant are for the convenience of reference only and shall not, for any purpose, be deemed a part of this Warrant.

\*\*\*\*\*\*

(Signature Page Follows)

	IN WITNESS WHE	REOF, the Company	has caused this	Warrant to be	executed by its	officer thereun	to duly author	rized as of the	date first
above indicat	ted								

## PROFRAC HOLDING CORP.

By:					
	Name:				
	Title:				

#### NOTICE OF EXERCISE

# 

Name of Investing Entity:

Signature of Authorized Signatory of Investing Entity:

## ASSIGNMENT FORM

(To assign the foregoing Warrant, execute this form and supply required information. Do not use this form to exercise the Warrant to purchase shares.)

FOR VALUE RECEIVED, the foregoing Warrant and all rights evidenced thereby are hereby assigned to

Name:	
	(Please Print)
Address:	
	(Please Print)
Phone Number:	
Email Address:	
Dated:,,	
Holder's Signature:	
Holder's Address:	

## CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We have issued our report dated March 31, 2022, with respect to the consolidated financial statements of ProFrac Holdings, LLC contained in the Registration Statement and Prospectus. We consent to the use of the aforementioned report in the Registration Statement and Prospectus, and to the use of our name as it appears under the caption "Experts."

/s/ GRANT THORNTON LLP

Dallas, Texas August 30, 2022

## CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We have issued our report dated March 31, 2022, with respect to the financial statements of ProFrac Holding Corp. contained in the Registration Statement and Prospectus. We consent to the use of the aforementioned report in the Registration Statement and Prospectus, and to the use of our name as it appears under the caption "Experts."

/s/ GRANT THORNTON LLP

Dallas, Texas August 30, 2022

## CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We have issued our report dated March 2, 2022, with respect to the consolidated financial statements of FTS International, Inc. contained in the Registration Statement and Prospectus of ProFrac Holding Corp. We consent to the use of the aforementioned report in the Registration Statement and Prospectus, and to the use of our name as it appears under the caption "Experts."

/s/ GRANT THORNTON LLP

Dallas, Texas August 30, 2022



KPMG LLP 811 Main Street Houston, TX 77002

## Consent of Independent Registered Public Accounting Firm

We consent to the use of our report dated March 30, 2022, with respect to the consolidated financial statements of U.S. Well Services, Inc., included herein and to the reference to our firm under the heading "Experts" in the prospectus.

/s/ KPMG LLP

Houston, Texas August 30, 2022

> KPMG LLP, a Delaware limited liability partnership and a member firm of the KPMG global organization of independent member firms affiliated with KPMG International Limited, a private English company limited by guarantee.

Chairman James W. Boyd

President and CEO John T. Boyd II

Managing Director and COO

Ronald L. Lewis

Vice Presidents Robert J. Farmer Matthew E. Robb John L. Weiss Michael F. Wick

William P. Wolf

Managing Director - Australia

Hugh Morrison

Managing Director - China

Jisheng (Jason) Han

Managing Director - South America

Carlos F. Barrera

Managing Director - Metals

Gregory B. Sparks

Pittsburgh

4000 Town Center Boulevard, Suite 300 Canonsburg, PA 15317 (724) 873-4400 (724) 873-4401 Fax jtboydp@jtboyd.com

Denver

(303) 293-8988 jtboydd@jtboyd.com

Brisbane

61 7 3232-5000 jtboydau@jtboyd.com

Reiiina

86 10 6500-5854 jtboydcn@jtboyd.com

Bogota

+57-3115382113 jtboydcol@jtboyd.com

www.jtboyd.com

J.P. Morgan Securities LLC

August 29, 2022

File: 3976.001

Piper Sandler & Co. Morgan Stanley & Co. LLC as representatives of the several

Underwriters

c/o J.P. Morgan Securities LLC 383 Madison Avenue New York, NY 10179

c/o Piper Sandler & Co. U.S. Bancorp Center 800 Nicollet Mall Minneapolis, Minnesota 55402

c/o Morgan Stanley & Co. LLC 1585 Broadway New York, New York 10036

Subject: Consent to be Named in Registration Statement

Ladies and Gentlemen:

The undersigned hereby consents to the references to our firm in the form and context in which they appear in this Registration Statement on Form S-4 of ProFrac Holding Corp. and the related prospectus that is a part thereof (the "Registration Statement"). We hereby further consent to (i) the use in such Registration Statement of information contained in our reports setting forth the estimates of reserves of ProFrac Holding Corp. as of December 31, 2021 and (ii) the reference to us under the heading "Experts" in such Registration Statement.

Respectfully submitted,

JOHN T. BOYD COMPANY By:

/s/ Ronald L. Lewis Ronald L. Lewis Managing Director & COO

JOHN T. BOYD COMPANY



#### YOUR VOTE IS IMPORTANT! PLEASE VOTE BY:

## INTERNET



Go To: www.proxypush.com/USWS Cast your vote online

- Have your Proxy Card ready
- . Follow the simple instructions to record your vote



#### PHONE Call 1-866-870-7492

- Use any touch-tone telephone Have your Proxy Card ready
- · Follow the simple recorded instructions



- Mark, sign and date your Proxy Card
- Fold and return your Proxy Card in the postage-paid envelope provided



You must register to attend the meeting online and/or participate at www.proxydocs.com/USWS prior to the special meeting

## U.S. Well Services, Inc.

## Special Meeting of Stockholders

For Stockholders of record as of XXXXXXX XX, 202X

XXXXX, XXXXXXX XX, 202X 10:00 AM, Central Time PLACE: Special Meeting to be held live via the Internet.

Please visit www.proxydocs.com/USWS for more details.

## This proxy is being solicited on behalf of the Board of Directors

The undersigned hereby appoints Kyle O'Neill and Erin Simonson (the "Named Proxies"), and each or either of them, as the true and lawful attorneys of the undersigned, with full power of substitution and revocation, and authorizes them, and each of them, to vote all the shares of common stock of U.S. Well Services, Inc. which the undersigned is entitled to vote at said meeting and any adjournment thereof upon the matters specified and upon such other matters as may be properly brought before the meeting or any adjournment thereof, conferring authority upon such true and lawful attorneys to vote in their discretion on such other matters as may properly come before the meeting and revoking any proxy heretofore given.

THE SHARES REPRESENTED BY THIS PROXY WILL BE VOTED AS DIRECTED OR, IF NO DIRECTION IS GIVEN, SHARES WILL BE VOTED IDENTICAL TO THE BOARD OF DIRECTORS RECOMMENDATION. This proxy, when properly executed, will be voted in the manner directed herein. In their discretion, the Named Proxies are authorized to vote upon such other matters that may properly come before the meeting or any adjournment or

You are encouraged to specify your choice by marking the appropriate box (SEE REVERSE SIDE) but you need not mark any box if you wish to vote in accordance with the Board of Directors' recommendation. The Named Proxies cannot vote your shares unless you sign (on the reverse side) and return

# U.S. Well Services, Inc. Special Meeting of Stockholders

Please make your marks like this:

Signature (and Title if applicable)

THE BOARD OF DIRECTORS RECOMMENDS A VOTE:

FOR ON PROPOSALS 1, 2, 3, 4 AND 5

	PROPOSAL		YOUR VOTE		BOARD OF DIRECTORS RECOMMENDS
1.	To approve the Agreement and Plan of Merger, dated as of June 21, 2022, among U.S. Well Services, Inc.("USWS"), ProFrac Holding Corp. ("ProFrac") and Thunderclap Merger Sub I, Inc., an indirect subsidiary of ProFrac ("Merger Sub"), and the transactions contemplated thereby, including the merger, pursuant to which Merger Sub will merge with and into USWS, with USWS surviving as an indirect subsidiary of ProFrac.	FOR	AGAINST	ABSTAIN	FOR
2.	To approve, for purposes of complying with the Nasdaq listing rules, the issuance of shares of USWS Class A common stock, par value \$0.0001 per share (the "USWS Common Stock"), to be issued by USWS upon the conversion of the USWS Series A Redeemable Convertible Preferred Stock and the Convertible Senior Secured (Third Lien) PIK Notes, in an amount equal to 20% or more of USWS Common Stock outstanding.				FOR
3.	To approve an amendment to the USWS Certificate of Designations of the Series A Redeemable Convertible Preferred Stock to modify certain terms relating to the conversion rights of USWS Series A Redeemable Convertible Preferred Stock.				FOR
4.	To approve an amendment to our Amended and Restated U.S. Well Services, Inc. 2018 Stock Incentive Plan (the "A&R LTIP") to increase the maximum number of shares of USWS Common Stock that may be issued under the A&R LTIP by 2,000,000 shares.				FOR
5.	If submitted to a vote of USWS stockholders, to approve an adjournment of the special meeting of USWS stockholders including, if necessary, to solicit additional proxies in favor of the foregoing proposals if there are not sufficient votes for those proposals.				FOR
	You must register to attend the meeting online and/or participate at www.prox special meeting Authorized Signatures - Must be completed for your instructions to be executed. Please sign exactly as your name(s) appears on your account. If held in joint tenancy, all persons should should include title and authority. Corporations should provide full name of corporation and title of authority.	ld sign. Tr	ustees, admin	istrators, et	ε,

Signature (if held jointly)

Date

Date

August 30, 2022

Special Committee of the Audit Committee of the Board of Directors of ProFrac Holding Corp.
333 Shops Boulevard, Suite 301
Willow Park, Texas 76087

Members of the Special Committee:

We hereby consent to the inclusion of our opinion letter, dated June 21, 2022, to the Special Committee of the Audit Committee of the Board of Directors (in its capacity as such) of ProFrac Holding Corp. ("ProFrac") included as Annex C to, and to the reference to such opinion letter under the captions "SUMMARY—Opinions of Financial Advisors—Opinion of the Financial Advisor to the ProFrac Special Committee", "THE MERGER—ProFrac Board and ProFrac Special Committee Reasons for the Merger—Factors Considered by the ProFrac Special Committee", and "THE MERGER—Opinion of the Financial Advisor to the ProFrac Special Committee" in, the proxy statement/information statement/prospectus relating to the proposed merger involving U.S. Well Services, Inc. ("USWS"), which proxy statement/information statement/prospectus forms a part of the Registration Statement on Form S-4 of ProFrac filed with the Securities and Exchange Commission on the date hereof (the "Registration Statement"). By giving the foregoing consent, we do not thereby admit that we are experts with respect to any part of such Registration Statement within the meaning of the term "expert" as used in, or that we come within the category of persons whose consent is required under, the Securities Act of 1933, as amended, or the rules and regulations of the Securities and Exchange Commission promulgated thereunder.

Very truly yours,

/s/ JEFFERIES LLC JEFFERIES LLC

## CONSENT OF PIPER SANDLER & CO.

We hereby consent to the inclusion of our opinion letter, dated June 21, 2022, to the Board of Directors of U.S. Well Services, Inc., a Delaware corporation ("<u>USWS</u>"), as Annex D to the joint proxy statement/prospectus contained in that certain Registration Statement of ProFrac Holding Corp. ("<u>ProFrac</u>") on Form S-4 (the "<u>Registration Statement</u>") relating to the agreement and plan of merger involving USWS and ProFrac, and references made to our firm and such opinion in the Registration Statement. In giving such consent, we do not admit that we come within the category of persons whose consent is required under, nor do we admit that we are "experts" for purposes of, the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

/s/ Piper Sandler & Co.

August 30, 2022 Houston, Texas

#### Calculation of Filing Fee Tables

## Form S-4 (Form Type)

#### ProFrac Holding Corp.

(Exact Name of Registrant as Specified in its Charter)

#### Table 1: Newly and Carry Forward Registered Securities

	Security Type	Security Class Title	Fee Calculation or Carry Forward Rule	Amount Registered(1)	Proposed Maximum Offering Price Per Unit	Maximum Aggregate Offering Price	Fee Rate	Amount of Registration Fee		
	Newly Registered Securities									
Fees to be paid	Equity	ProFrac Class A Common Stock	457(c) 457(f)(1)	14,056,943 (2)	N/A	\$278,549,638.55 (3)	0.0000927	\$25,821.55		
	Equity	Warrants	457(c) 457(f)(1)	19,167,417 (4)	N/A	\$1,035,040.52 (5)	0.0000927	\$95.95		
	Equity	ProFrac Class A Common Stock issuable upon exercise of Warrants	457(g)	153,613 (6)	N/A	110,212,719.11 (7)	0.0000927	\$10,216.72		
Fees Previously Paid	_	_				_		_		
			C	arry Forward Securi	ties					
Carry Forward Securities	_	_		_		_				
		Total Offering	Amounts			\$389,797,398.18		\$36,134.22		
	Total Fees Previously		iously Paid					_		
	Total Fee Offsets						_			
	Net Fee Due					\$36,134.22				

- (1) Pursuant to Rule 416(a) of the Securities Act, there are also being registered an indeterminable number of additional securities as may be issued to prevent dilution resulting from stock splits, stock dividends or similar transactions.
- (2) Represents the estimated maximum number of shares of Class A Common Stock, par value \$0.01 per share, of ProFrac Holding Corp. ("ProFrac") expected to be issuable by ProFrac upon completion of the Merger. The number of shares of ProFrac Class A Common Stock being registered is based upon the product of (a) 41,761,565, the estimated maximum number of shares of Class A Common Stock, par value \$0.0001 per share, of U.S. Well Services, Inc. ("USWS") expected to be outstanding at the Effective Time multiplied by (b) 0.3366 (the "Exchange Ratio").
- (3) Estimated solely for the purpose of calculating the registration fee required by Section 6(b) of the Securities Act and calculated in accordance with Rule 457(c) and 457(f) of the Securities Act. The proposed maximum aggregate offering price of the ProFrac Class A Common Stock to be registered is based on the product of (a) \$6.67, the average of the high and low sale prices of USWS Common Stock as reported on the Nasdaq Capital Market on August 23, 2022, and (b) 41,761,565, the estimated maximum number of shares of USWS Common Stock expected to be exchanged in connection with the Merger.
- (4) The number of warrants to acquire shares of ProFrac Class A Common Stock being registered represents the number of USWS SPAC Warrants outstanding as of June 30, 2022, consisting of (a) 9,994,635 warrants to acquire shares of USWS Common Stock issued in connection with USWS' initial public offering and (b) 9,172,782 warrants to acquire shares of USWS Common Stock issued in a private placement that closed simultaneously with the closing of USWS' initial public offering. Prior to the Effective Time, ProFrac intends to execute an amendment to each of the warrant agreements that govern the USWS SPAC Warrants to, among other things, (a) assume each USWS SPAC Warrant, (b) provide that the exercise price of each USWS SPAC Warrant will equal the exercise price of such warrant divided by the Exchange Ratio and (c) provide that each USWS SPAC Warrant, as amended, will become the right to receive, upon valid exercise thereof, shares of ProFrac Class A Common Stock in an amount equal to the product of (i) the number of shares of USWS Common Stock subject to such USWS SPAC Warrant immediately prior to the Effective Time and (ii) the Exchange Ratio.
- (5) Estimated solely for the purpose of calculating the registration fee required by Section 6(b) of the Securities Act and calculated in accordance with Rule 457(c) and 457(f) of the Securities Act. The proposed maximum aggregate offering price of the warrants to be registered is based on the product of (a) \$0.054, the average of the high and low sales prices of the USWS SPAC Warrants as reported on the Nasdaq Capital Market on August 23, 2022 and (b) 19,167,417, the estimated maximum number of USWS SPAC Warrants expected to be assumed by ProFrac prior to the Effective Time (as described in footnote 4 above).
- (6) Represents the estimated maximum number of shares of ProFrac Class A Common Stock expected to be issuable by ProFrac upon exercise of the USWS SPAC Warrants, as amended (as described in footnote 4 above).
- (7) Estimated solely for the purpose of calculating the registration fee required by Section 6(b) of the Securities Act and calculated in accordance with Rule 457(g) of the Securities Act. The proposed maximum aggregate offering price of the ProFrac Class A Common Stock to be registered is based on the product of (a) \$717.47, the aggregate of the exercise prices of the number of USWS SPAC Warrants, as amended (as described in footnote 4 above), that a holder will be required to surrender in order to acquire one share of ProFrac Class A Common Stock and (b) 153,613, the estimated maximum number of shares of ProFrac Class A Common Stock expected to be issuable upon exercise of the USWS SPAC Warrants, as amended (as described in footnote 4 above).